#### Securities Act 1997

#### FORM 4

Act, Section 116 (2)

## **NOTICE OF CHANGE IN SUBSTANTIAL SHAREHOLDING**

Note: To be given to the company, the stock exchange on which it is listed and the Securities Commission by a person who is a substantial shareholder where there is a change of 1% or more in the relevant interests of the person. This form must be given within two days after the person becomes aware of the change. Use a separate form for each substantial shareholder that has a change of 1% or more in their relevant interests. Use a Form 5 where the person ceases to be a substantial shareholder as a result of the change.

1.	Name of listed
	company

Oil Search Limited

Note: Insert the name of the listed company in which the substantial shareholding is held.

2. Name of stock exchange

Port Moresby Stock Exchange, Australian Securities Exchange

Note: Insert the name of the stock exchange on which the company is listed.

## 3. Details of person that is a substantial shareholder

Given names (natural persons only)	Surname or corporate name and registration number, if applicable	Residential address or address of registered office	Postal address
N/A	UBS Group AG	Bahnhofstrasse 45, 8001 Zurich, Switzerland	Singapore Branch, One Raffles Quay, #50-01 North Tower, Singapore 048583

4. Date of change in relevant interests.

23 December 2014

Note: Insert the date of the change of the relevant interest of the person named at Item 3 in the listed company at Item 1, not the date on which the person became aware of the change.

# 5. Date of last notice given by substantial shareholder

#### 29 October 2014

Note:Insert the date on which the person named it Item 3 last gave a notice in accordance with Section 115, 116 or 117 of the Act in relation to the voting shares i.e. a Form 3, 4 or 5.

# 6. Voting shares in which a relevant interest is held.

	Number of shares	% of shares in company
Voting shares before the change	206,303,686	13.55%
Voting shares after the change	193,521,205	12.71%

Note: Insert the number and % of voting shares in which the person has a relevant interest, before and after the date specified at Item 5. The terms "voting shares" and "relevant interest" are defined in Sections 112–114 of the Act.

# 7. Total number of voting shares issued by the listed company

	Before the change	After the change
Total number of voting shares issued	1,522,692,587	1,522,692,587

Note: Insert the total number of voting shares issued by the listed company used for the purpose of calculating the percentages at Item 6.

# 8. Particulars of relevant interests after the change.

Class of voting shares in which relevant interest now held	Name(s) of registered holder(s)	Brief description of the nature of the relevant interests in the voting shares	Brief description of the transaction under which the relevant interests changed e.g. purchase / sale on stock exchange	Number of voting shares affected by the transaction	The consideration for the transaction (Kina)	Name of the other party to the transaction (if known) *
Ordinary	UBS AG	Fund Manager with power to exercise control over voting shares	Ordinary course transaction (e.g. stock exchange transactions) entered into for portfolio	53,560 shares	Refer to Annexure F (Australian Dollars only)	N/A

			management purposes			
Ordinary	UBS Fund Management (Switzerland) AG	Fund Manager with power to exercise control over voting shares	Ordinary course transaction (e.g. stock exchange transactions) entered into for portfolio management purposes	176,190 shares	Refer to Annexure F (Australian Dollars only)	N/A
Ordinary	UBS Fund Services (Luxembourg ) SA	Fund Manager with power to exercise control over voting shares	Ordinary course transaction (e.g. stock exchange transactions) entered into for portfolio management purposes	1,941,468 shares	Refer to Annexure F (Australian Dollars only)	N/A
Ordinary	UBS Global Asset Management (Australia) Ltd	Fund Manager with power to exercise control over voting shares	Ordinary course transaction (e.g. stock exchange transactions) entered into for portfolio management purposes	13,919,359 shares	Refer to Annexure F (Australian Dollars only)	N/A
Ordinary	UBS Global Asset Management Life Limited	Fund Manager with power to exercise control over voting shares	Ordinary course transaction (e.g. stock exchange transactions) entered into for portfolio management purposes	200,642 shares	Refer to Annexure F (Australian Dollars only)	N/A
Ordinary	UBS Global Asset Management (Singapore) Ltd	Fund Manager with power to exercise control over voting shares	Ordinary course transaction (e.g. stock exchange transactions) entered into for portfolio management purposes	7,649 shares	Refer to Annexure F (Australian Dollars only)	N/A
Ordinary	UBS Global Asset Management (UK) Limited	Fund Manager with power to exercise control over voting shares	Ordinary course transaction (e.g. stock exchange transactions) entered into for portfolio management purposes	1,633,475 shares	Refer to Annexure F (Australian Dollars only)	N/A

ſ <del></del>	T		<u></u>	1	Τ	I
Ordinary	UBS AG Australia Branch	Prime Broker with power to control the exercise of the power to dispose of shares (see Annexure E)	These holdings are held by UBS AG Australia Branch on behalf of Prime Brokerage clients	18,033,357 shares	Refer to Annexure F (Australian Dollars only)	Refer to Annexure E- 1
Ordinary	UBS AG London Branch	Prime Broker with power to control the exercise of the power to dispose of shares (see Annexure E)	These holdings are held by UBS AG London Branch on behalf of Prime Brokerage clients	321,513 shares	Refer to Annexure F (Australian Dollars only)	Refer to Annexure E- 1
Options	UBS Securities Australia Ltd	Beneficial Owner	Ordinary course transactions (e.g. on market purchases)	5,411,043 shares	Refer to Annexure F (Australian Dollars only)	N/A
Ordinary	UBS Wealth Management Australia Ltd	Broker with power to exercise discretion over account	Discretionar y client accounts	2,432,705 shares	Refer to Annexure F (Australian Dollars only)	N/A
Ordinary	UBS AG, Australia Branch	See Section 11 below	See Section 11 below	12,377,994 shares	Entry into an equity derivative financing transaction between NPCP Investments and UBS relating to 12,377,994 shares in Oil Search Limited	NPCP Investments Limited
Ordinary	UBS Nominees Pty Ltd	See Section 11 below	See Section 11 below	137,012,250 shares	The assumption of liability by NPCP Investments Limited for the obligations owed to UBS under an equity derivative financing transaction previously entered into with the	NPCP Investments Limited

	Independent
	State of Papua
	New Guinea

<sup>\*</sup> This information is not required where the transaction took place on a stock exchange. Otherwise, if not known, insert "unknown".

Note: Where there is insufficient space on the form to supply the information required, use additional forms or annexe a separate sheet in the same format containing the information.

# 9. Details of other person(s) giving a notice for the same transaction

Given names (natural persons only)	Surname or corporate name and registration number, if applicable	Residential address or address of registered office	Postal address
N/A	NPCP Investments Limited (NPCP Investments)	Level 7, Deloitte Tower Douglas Street Port Moresby, NCD Papua New Guinea	Level 7, Deloitte Tower Douglas Street Port Moresby, NCD Papua New Guinea
N/A	The Independent State of Papua New Guinea (State)	N/A	Department of Treasury PO Box 542 Waigani 131 NCD Papua New Guinea

Note: If known, insert the name and address of any other person(s) believed to have given, or to be intending to give, notice, whether a Form 3, 4 or 5, in relation to the transaction to which this notice relates. If not known, insert, "unknown".

# 10. Details of any associate with a relevant interest in the voting shares after the change

Given names (natural persons only)	Surname or corporate name and registration number, if applicable	Residential address or address of registered office	Postal address
N/A	See item 8	Details of all UBS offices can to following link: http://apps2.ubs	

Nature of Association: Related bodies corporate				
Details of Relevant Interest of Associate:				

Note: Provide the name and address of each associate of the person submitting this notice who has a relevant interest in voting shares in the company, and the nature of the association with the associate and the details of the relevant interest of the associate.

# 11. Prescribed particulars of any agreement under which substantial shareholding changed

Date of any agreement	Material terms of any agreement (whether written or oral)
17 December 2014	New Collar Confirmation Letter for an ISDA equity derivative financing transaction relating to 12,377,994 shares in Oil Search Limited, including the 1995 ISDA Credit Support Annex (CSA), between NPCP Investments and UBS. [Annexure A]
	Under the CSA, the title to the shares in Oil Search Limited which are the subject of the equity derivative financing was transferred to UBS as eligible credit support for the financial obligations of NPCP Investments under the equity derivative financing, and UBS has an obligation to transfer equivalent credit support (being shares in Oil Search Limited) back to NPCP Investments upon settlement of the equity derivative financing.
17 December 2014	Collar Confirmation Letter for an ISDA equity derivative financing transaction relating to 137,012,250 shares in Oil Search Limited, including the 1995 ISDA Credit Support Annex (CSA), as novated and amended between NPCP Investments and UBS. [Annexure B]
	Under the CSA, the title to the shares in Oil Search Limited which are the subject of the equity derivative financing was transferred to UBS as eligible credit support for the financial obligations of the counterpart (originally the State now NPCP Investments) under the equity derivative financing, and UBS has an obligation to transfer equivalent credit support (being shares in Oil Search Limited) back to NPCP Investments upon settlement of the equity derivative financing.
17 December 2014	Specific Security Deed (CHESS Securities – Collar) as novated and amended between NPCP Investments and UBS. [Annexure C]
	Under the Specific Security Deed, NPCP Investments has created security interests in the nature of fixed charges over the shares acquired in Oil Search Limited in favour of UBS to secure all amounts of money payable or owing under the equity derivative financing.
17 December 2014	Nominee Deed as novated and amended between NPCP Investments and UBS Nominees Pty Ltd. [Annexure D]
	Under the Nominee Deed, UBS Nominees Pty Ltd acquired 137,012,250 shares in Oil Search Limited to be transferred, dealt with or otherwise disposed of as directed under the Collar Confirmation Letter. UBS Securities Australia Limited is a party to this Deed as a market participant under the ASX Settlement Operating Rules.

Note: Provide the details of the material terms of any agreement, whether written or oral, and the details of the material terms of any relevant agreement through which a relevant interest, the subject of the notice, was changed. True copies of the documents must be submitted with this form.

# 12. The following documents must be submitted with this form -

- A. a true copy of any document setting out the material terms of any agreement that -
  - (a) contributed to the person having to submit this form;
  - (b) is in writing or in a document in which the material terms of an oral agreement have been reduced to writing in accordance with Section 40 (5) (c) of the Regulation; and
  - (c) is readily available to the person submitting the notice; or
- B. where a document in Paragraph A is not readily available to the person submitting this form, a memorandum in writing detailing the material terms of the document.

The number of pages that accompany, or are annexed to, this notice is N/A.

#### 13. **Declaration**

I declare that to the best of my knowledge and belief the information in this form is true and correct.

Signature: ————————————————————————————————————		
Full Name of Person Signing: Chandrachud Giri	iappa	
Role: Authorized Signatory	_ Date:	29 December 2014
Signature:		
Full Name of Person Signing: Wuen-Yi Toh		
Role: Authorized Signatory	Date:	29 December 2014

Note: Where this notice is given by a body corporate it must be signed by a director or secretary who is authorized by the board of the body corporate to give the notice. Initials are not sufficient for the full name of the person signing this form.

# Annexure A



New Collar Confirmation

UBS AG, Australia Branch ABN 47 088 129 613 AFSL No. 231087 Level 16 Chifley Tower 2 Chifley Square Sydney NSW 2000 Australia

> OTC Operations Tel. +612-9324 3455 Fax. +612-9324 2880

Date:

17 December 2014

To:

NPCP Investments Limited of Level 7 Deloitte Tower, Douglas Street, Port Moresby

NCD, Papua New Guinea (Counterparty or Party B)

E-mail:

robert.acevski@npcp.com

Attention: Tel:

Robert Acevski +675 320 2253

Fax:

+675 3200238

From:

UBS AG, Australia Branch (ABN 47 088 129 613) of Level 16, Chifley Tower, 2 Chifley

Square, Sydney NSW 2000 (UBS or Party A)

Subject:

Transaction in respect of OSH

Dear Sirs/Madams,

The purpose of this letter agreement (this "Confirmation") is to confirm the terms and conditions of the above referenced transaction entered into between Counterparty and UBS on the Trade Date specified below (the "Transaction").

This Confirmation evidences a complete binding agreement between the parties as to the terms of the Transaction to which this Confirmation relates. Until such time as the parties execute an ISDA Master Agreement and 1995 ISDA Credit Support Annex, the parties agree that this Confirmation, and any future Confirmation(s) entered into between the parties thereafter shall supplement, form part of, and be subject to, an agreement in the form of the 2002 ISDA Master Agreement as if on the Trade Date the parties had executed such agreement but without any Schedule, except for the election of Australian Dollars as the Termination Currency and New South Wales as the governing law and such other elections and modifications detailed herein referring to the ISDA Form (the "Deemed Agreement") and that each Confirmation entered into between the parties shall each be supplemented by and be subject to the terms of each 1995 ISDA Credit Support Annex with the elections and modifications provided in that Confirmation. The parties agree that the terms of each Annex to the 2002 Master Agreement Protocol published by the International Swaps and Derivatives Association, Inc ("Protocol") apply to the agreement as if the parties had adhered to the Protocol without amendment.

The definitions and provisions contained in the 2002 ISDA Equity Derivatives Definitions (the "Equity Definitions") and the 2006 ISDA Definitions (the "2006 Definitions", and together with the Equity Definitions, the "Definitions"), in each case, as published by the International Swaps and Derivatives Association, Inc., are incorporated into this Confirmation. In the event of any inconsistency between the 2006 Definitions and the Equity Definitions, the Equity Definitions will govern. In the event of any inconsistency between the Definitions and this Confirmation, this Confirmation will govern. References herein to a "Transaction" shall be deemed references to a "Share Option Transaction" for the purposes of the Equity Definitions.

In the event of any inconsistency between the provisions of the Deemed Agreement and this Confirmation, this Confirmation will prevail for the purpose of this Transaction.

The terms of the particular Transaction to which this Confirmation relates are as follows:

#### **GENERAL TERMS**

Trade Date :	2014

Effective Date: The Effective Date (as defined in the Global Novation and Amendment

Deed).

Tax: Has the meaning given in the ISDA Form and includes any stamp,

registration, documentation or similar tax, duty, levy, fee or other

imposition.

Shares: Fully paid ordinary shares of Oil Search Limited (ARBN 055 079 868).

ASX code "OSH".

Business Days: Sydney

Currency Business Days: Sydney

Convention: If a payment or delivery is due, or a notice must be given or received,

under this Confirmation on a day that is not a Business Day, a Local Business Day or a Currency Business Day, as the case may be, then the payment or delivery must be made, or the notice must be given or received, on the next Business Day, Local Business Day or Currency

Business Day, as the case may be, immediately after that day.

Calculation Agent: Party A, provided that if Party A is the sole Defaulting Party, then the

Calculation Agent shall be a leading independent dealer in the relevant market mutually selected by the parties (the "Substitute Calculation Agent"). The Substitute Calculation Agent shall be mutually selected by the parties within one Business Day of the event that requires Calculation Agent determination. If the parties are unable to agree on a Substitute Calculation Agent, each party shall select an independent leading dealer and such independent leading dealers shall agree on an independent third party within two Business Days of the event that requires Calculation Agent determination. Unless there is a clear error, the calculations and determinations of the Substitute Calculation Agent are binding and conclusive. The fees and expenses of using the Substitute Calculation Agent, if any, shall be paid equally by Party A

and Party B.

**TERMS** 

Initial Exchange Payer (acting as lender):

Party A

Initial Exchange: Party A shall pay to Party B the Net Aggregate Initial Exchange

Amount on the Initial Exchange Date, where such amount is notionally split into 3 equal tranches ("Tranches") corresponding to 3 separate

Final Exchange Amounts.

Net Aggregate Initial Exchange Amount:

The amount specified as such in the Initial Price Notification Letter, being the sum of the Initial Exchange Amounts across all the Tranches

minus the Aggregate Premium Amount.

Initial Exchange Amount per Tranche:

In respect of each Tranche, the Initial Exchange Amount is calculated

Tranche Notional x (1 - Rate x t/365)

where:

"Tranche Notional" is the sum of the Put Strikes of all of the notional Put Options in each Tranche, as set out in the Initial Price Notification Letter:

"Rate" is the rate described as such in the Side Letter; and

"t" is the number of days from but excluding the Initial Exchange Date to and including the Collar Group Final Exchange Date for the Collar Group to which the Tranche belongs.

The Effective Date (as defined in the Global Novation and Amendment Initial Exchange Date:

Deed).

Final Exchange Payer (acting as borrower):

Party B

Final Exchange Amount per Tranche:

Collar Group Final Exchange Amount: The Final Exchange Amount for each Tranche is the amount described as such in the Initial Price Notification Letter.

Means, in respect of each Collar Group, the amount described as such in the Initial Price Notification Letter, which is the sum of the Final Exchange Amounts of the Tranches that make up the Collar Group.

Final Exchange Payment Date:

If Cash Settlement is applicable in respect of a Collar Group, then Party B must pay to Party A the Collar Group Final Exchange Amount on the Collar Group Final Exchange Date for the relevant Collar Group.

If Physical Settlement is applicable to a Tranche, then Party B must pay to Party A the Final Exchange Amount for that Tranche on the Settlement Date for that Tranche.

The Collar Group Final Exchange Date and Expiration Dates of each

Tranche are set out in the Side Letter.

Collar Group Final Exchange Date:

Means, in respect of a Collar Group, the collar group final exchange date for that Collar Group as set out in the Side Letter.

If any terms of the notional Put Options and Call Options are adjusted Adjustments:

by the Calculation Agent in accordance with this Confirmation, Party A shall be entitled to adjust the financing terms above including but not

limited to the Final Exchange Amount.

## **OPTION TERMS**

#### **General Terms**

The amount specified as such in the Initial Price Notification Aggregate Premium Amount :

Single Transaction: Party A and Party B acknowledge and agree that:

> (a) the Share Option Transaction the subject of this Confirmation is a "Step Collar" option comprising 1 "Collar Group";

(b) the Collar Group comprises three Tranches each with a different Expiration Date;

(c) each Tranche is made up of five "Component Collars" with

- the same Expiration Date and the applicable strike prices set out in (the Side Letter and Initial Price Notification Letter;
- (d) each Component Collar is made up of notional "**Put Options**" and "**Call Options**" over the same number of Shares;
- (e) the Step Collar, the Tranches, the Component Collars and the notional Put Options and notional Call Options comprised within them are not separate Transactions for the purposes of the Deemed Agreement, cannot be traded or dealt with separately and all payment and delivery obligations described herein form one Transaction for the purposes of the Deemed Agreement; and
- (f) without limiting paragraph (e) above, the Component Collars that make up a Tranche and the Initial Exchange Amount and Final Exchange Amount that correspond to that Tranche cannot be terminated independently of each other (each being a "Related Transaction" of the others), and "Related Options" mean the Put Options and Call Options that make up the Component Collars in each Tranche.

Additional details of the Component Collars, Tranches and the Collar Group are set out in the Side Letter and Initial Price Notification Letter.

This Confirmation must be construed and interpreted accordingly.

Put Option Seller:
Party A
Put Option Buyer:
Party B
Call Option Seller:
Party B
Call Option Buyer:
Party A
Option Style:
European

Option Entitlement: 1 Share per Option.

Strike Price: Means the Put Strike for each notional Put Option or the Call Strike

for each notional Call Option that make up each Component Collar

set out in the Initial Price Notification Letter.

Aggregate Number of Shares: The sum of the Tranche Quantities for all outstanding Tranches,

which as at the Effective Date is the quantity specified as such in

the Side Letter.

Collar Group Quantity: The number specified as such in the Side Letter

Tranche Quantity: The number specified as such in the Side Letter

Number of Put Options per Component

Collar:

The number specified as such in Schedule 1 to the Side Letter

Number of Call Options per Component The

Collar:

The number specified as such in Schedule 1 to the Side Letter

Expiration Date: The Expiration Date for each Tranche as set out in the Side Letter

is the Expiration Date of the Component Collars in that Tranche and all the notional Put Options and notional Call Options that

make up the Component Collars in that Tranche.

Exchange: Australian Securities Exchange ("ASX") or any successor to such

exchange or quotation system.

Related Exchange: Not applicable

# **Procedure for Exercise of Put Options and Call Options**

Expiration Time: The Scheduled Closing Time on the Exchange on the Expiration

Date of the Put Option or Call Option.

Exercise Date: The Expiration Date of the Put Option or Call Option.

Automatic Exercise: Applicable, provided that Section 3.4(c) of the Equity Definitions

shall be deleted in its entirety and replaced with the following: "In-the-Money" means, in respect of a Call Option, that the Reference Price is greater than the applicable Strike Price and, in respect of a Put Option, that the Reference Price is equal to or less

than the applicable Strike Price."

Valuation (for the purposes of Settlement and Exercise)

Valuation Time: The Expiration Time
Valuation Date: The Expiration Date

**Settlement Terms** 

Settlement Method Election: Applicable in respect of each Collar Group, provided that any

election is irrevocable, given in respect of only one Collar Group at any one time, and the election (or default position in the absence of any election) shall apply to each Collar Group separately. Once a Settlement Method is elected for a Collar Group it shall apply to all Tranches (and therefore to all Component Collars and Related

Options) in the Collar Group.

Electing Party: Party B.

Settlement Method Election Date: The date falling 10 Scheduled Trading Days prior to the earliest

Expiration Date of each Collar Group (or such later date as Party A

may agree following a request by Party B).

Default Settlement Method: Cash Settlement

Reference Price: For the purposes of determining whether a Put Option or a Call

Option is "In-the-Money", the Reference Price is the Volume Weighted Average Price of the Share on the Exchange at the Expiration Time on the Expiration Date for that Put Option or Call

Option.

Volume Weighted Average Price (for

purposes of Reference Price):

Means the price referenced for OSH AU Equity VWAP on

Bloomberg (code RQ017).

Settlement Price: For each Put Option and Call Option in respect of which Cash

Settlement is applicable, the price per Share in Australian dollars

that Party A has achieved or considers in a commercially

reasonable manner that it would have achieved, in terminating or liquidating applicable Hedge Positions on the Expiration Date for that Put Option or Call Option, plus applicable commissions and Taxes and any stamp, registration, documentation or similar tax,

duty, levy, fee or other imposition.

Cash Settlement: If Cash Settlement is applicable in respect of a Tranche, then the

Seller must pay to the Buyer the Cash Settlement Amount for that Tranche, if any, on the Cash Settlement Payment Date for that

Cash Settlement Amount:

Tranche, and Section 8.1 of the Equity Definitions shall not apply.

In respect of each Tranche, the Cash Settlement Amount payable by Party B to Party A is an amount equal to the Aggregate Call Option Cash Settlement Amount minus the Aggregate Put Option Cash Settlement Amount, subject to a minimum of zero; and the Cash Settlement Amount payable by Party A to Party B is an amount equal to the Aggregate Put Option Cash Settlement Amount minus the Aggregate Call Option Cash Settlement Amount, subject to a minimum of zero.

For the purposes of this calculation:

- the Aggregate Call Option Cash Settlement Amount is the sum of the Option Cash Settlement Amounts in respect of each Call Option comprised in that Tranche which is exercised or deemed to be exercised on the Exercise Date of such Tranche, calculated in accordance with Section 8.2(b) of the Equity Definitions; and
- the Aggregate Put Option Cash Settlement Amount is the sum of the Option Cash Settlement Amounts in respect of each Put Option comprised in that Tranche which is exercised or deemed to be exercised on the Exercise Date of such Tranche, calculated in accordance with Section 8.2(b) of the Equity Definitions.

The intention of the parties is that following such calculations, only one net amount is payable by one party to the other in respect of each Tranche.

An amount equal to the greater of (a) the excess of (i) in the case of a Call Option, the relevant Settlement Price over the Call Strike, or (ii) in the case of a Put Option, the Put Strike over the relevant Settlement Price, and (b) zero.

Where Cash Settlement applies in respect of a Tranche, the date occurring one Settlement Cycle after the Expiration Date of that Tranche, provided that if such date is not a Business Day then the Cash Settlement Payment Date shall be the immediately following Currency Business Day.

Where Physical Settlement applies in respect of a Tranche, the date that falls one Settlement Cycle following the Expiration Date of that Tranche. If such date is not a Currency Business Day then the Settlement Date shall be the following Currency Business Day.

Applicable if Physical Settlement is applicable, provided that Party A shall be the Determining Party.

#### Dividends

Dividend Amounts:

Without prejudice to Schedule 1 to this Confirmation, and except to the extent included as an adjustment pursuant to "Adjustments" below (pursuant to Section 11.2 of the Equity Definitions), no Dividend Amounts are payable by Party A or Party B in respect of the Transaction.

# Adjustments

Method of Adjustment:

Calculation Agent Adjustment; provided, however, that:

 a) Section 11.2(c) of the Equity Definitions is deemed amended by the deletion of the words "(provided that no adjustments will be made to account solely for changes in volatility, expected dividends, stock loan rate or liquidity relative to the relevant

231777706.13

Strike Price Differential:

Cash Settlement Payment Date:

Settlement Date:

Failure to Deliver:

Share)" and replacing them with the words "(including adjustments to account for changes in volatility, expected dividends, stock loan rate or liquidity relative to the relevant Share)"; and

b) Section 11.2(e)(iii) of the Equity Definitions shall be deleted.

## **Extraordinary Events**

Consequences of Merger Events

Share-for-Share: Modified Calculation Agent Adjustment
Share-for-Other: Modified Calculation Agent Adjustment
Share-for-Combined: Modified Calculation Agent Adjustment

Tender Offer: Applicable

Consequences of Tender Offers

Share-for-Share: Modified Calculation Agent Adjustment
Share-for-Other: Modified Calculation Agent Adjustment
Share-for-Combined: Modified Calculation Agent Adjustment

Composition of Combined Consideration: Not Applicable

Nationalization, Insolvency or Delisting: Cancellation and Payment - Calculation Agent Determination

Determining Party: Party A, in all cases

Additional Disruption Events

Change in Law: Applicable, <u>provided</u> that Section 12.9(a)(ii)(B) of the Equity

Definitions is replaced in its entirety as follows: "(B) due to the promulgation of or any change in the interpretation by any court, tribunal or regulatory authority with competent jurisdiction of any applicable law or regulation (including any action taken by a taxing authority), the Calculation Agent determines in good faith that (X) it has become illegal to hold, acquire or dispose of Hedge Positions relating to such Transaction or (Y) a party to the

Transaction will incur materially increased cost in performing its obligations under such Transaction (including, without limitation, due to any increase in tax liability, decrease in tax benefit or other

adverse effect on its tax position)".

Hedge Positions: The definition of "Hedge Positions" in Section 13.2(b) of the

Equity Definitions shall be amended by inserting the words "or an

Affiliate thereof" after the words "a party" in the third line.

Insolvency Filing: Applicable

Hedging Disruption: Applicable, provided that Section 12.9(a)(v) of the Equity

Definitions is replaced in its entirety as follows: "(v) Hedging Disruption" means that the Hedging Party is unable, after using commercially reasonable efforts, to either (i) acquire, establish, re-establish, substitute, maintain, unwind or dispose of any transaction(s) or asset(s) it deems necessary to hedge the equity price risk (or any other relevant price risk, including, but not limited to, any currency risk) of entering into and performing its obligations with respect to this Transaction, or (ii) freely realise, recover, receive, repatriate, remit or transfer the proceeds of Hedge Positions or this Transaction between accounts within the jurisdiction of the Hedge Positions (the "Affected Jurisdiction")

or from accounts within the Affected Jurisdiction to accounts outside of the Affected Jurisdiction."

Consequences of Hedging Disruption:

Section 12.9(b)(iii) of the Equity Definitions is replaced with the following:

"(iii) If 'Hedging Disruption' is specified in the relevant Confirmation to be applicable to a Transaction, then upon the occurrence of such an event, the Hedging Party may elect to terminate the Transaction upon notice to the Non-Hedging Party specifying the date of such termination, which may be the day on which the notice of termination is issued, in which event the Determining Party will determine the Cancellation Amount

payable by one party to the other."

Increased Cost of Hedging: Applicable, <u>provided</u> that Section 12.9(a)(vi) of the Equity

Definitions is replaced as follows: "(vi) "Increased Cost of Hedging" means that the Hedging Party would incur a materially increased (as compared with the circumstances that existed on the Trade Date) amount of tax, duty, expense or fee (other than brokerage commissions) to (A) acquire, establish, re-establish, substitute, maintain, unwind or dispose of any transaction(s) or asset(s) it deems necessary to hedge the equity price risk (or any other relevant price risk including, but not limited to, currency risk) of entering into and performing its obligations with respect to this Transaction or (B) freely realise, recover, receive, repatriate, remit or transfer the proceeds of the Hedge Positions or

this Transaction between accounts within the Affected

Jurisdiction or from accounts within the Affected Jurisdiction to

accounts outside of the Affected Jurisdiction."

Loss of Stock Borrow: Applicable, <u>provided</u> however that Section 12.9(b)(iv)(B) of the

Equity Definitions is deemed amended to include after the words "(B) refer the Hedging Party to a Lending Party" and before the words "that will lend", the words "acceptable to the Hedging

Party in its sole discretion".

Maximum Stock Loan Rate: 0% per annum Increased Cost of Stock Borrow: Applicable

Initial Stock Loan Rate: 0% per annum

Hedging Party: Party A

Determining Party: Party A

Non-Reliance: Applicable
Agreements and Acknowledgements Applicable

Additional Acknowledgements: Applicable

**ACCOUNT DETAILS** 

Regarding Hedging Activities:

**Account for payments to Party B:** To be advised

Account for payments to Party A: Beneficiary Bank: UBS AG Australia Bank

Beneficiary Bank SWIFT: UBSWAU2S Account holder name: UBS AG Australia Correspondent: UBS AG, Australia Branch

BSB: 946 612 A/c: 242624 Swift: UBSWAU2S

#### ADDITIONAL PROVISIONS

For the purposes of this Transaction:

## 1) Disclosure requirement and additional representations and warranties

Party B represents and warrants to Party A:

- (a) that, as at the Trade Date for each Component Collar and on the Initial Exchange Date, there are no security interests (as defined in the *Personal Property Securities Act 2009* (Cth) or any other encumbrance of any nature whatsoever) in respect of the Shares that would be delivered to Party A under the terms of this Confirmation (including in accordance with Schedule 1 to this Confirmation);
- (b) that it will seek its own advice on the appropriateness and suitability of the Transaction and any disclosure obligations it may have relating to its dealings in Shares and Transaction in respect of Shares:
- (c) at the time of entry into the Transaction and on a continuous basis during the term of the Transaction that, by entering into and performing its obligations under the Transaction, it is not, and will not be, acting in breach of any:
  - i. laws, regulations or Authorisations binding on it or any of its property or to which any of its property may be subject;
  - ii. agreements, instruments, undertakings or other restrictions binding upon it or any of its property (other than any negative pledges) in a manner that could have a Material Adverse Effect; or
  - iii. negative pledges binding on or applicable to it;
- (d) that neither Counterparty nor any of its Affiliates nor anyone acting on Counterparty's or any Affiliate's behalf has engaged in any behaviour which is designed to cause, has caused, or might reasonably be expected to cause manipulation of the price of any security of the Issuer;
- (e) it will record in its books and account for the value of this Transaction in accordance with the policies and rules applicable to it;
- (f) it has complied and will comply with all laws, regulations and administrative provisions applicable to it in connection with the Transaction, including making all relevant disclosures to, or obtaining any required approvals from all regulatory authorities in respect of its holding in the Shares (if any) and in respect of the Transaction as required by, and within the time limits required by, such regulatory authorities or pursuant to applicable laws or regulations in this regard;
- (g) neither this Transaction nor any delivery of Shares or payment of cash contemplated hereby will constitute a violation by Counterparty or any Affiliate of any applicable law or regulations of any applicable jurisdiction prohibiting "insider dealing" in, or market abuse in respect of, securities;
- (h) it is not in possession of any material information about the Shares or the Issuer which has not been publicly disclosed, which might have influenced its decision to enter into the Transaction or any particular element thereof and it is not entering into the Transaction on the basis of any such information;
- (i) it has full authority and capacity to enter into the Transaction and the execution and delivery of this Confirmation and the consummation by it of the Transaction contemplated hereby have been duly authorised by all necessary corporate action. All consents, orders, approvals, and other authorisations, whether governmental, corporate or other, necessary for the Transaction have been obtained or made and are in full force and effect;
- (j) the purpose and effect of the Transaction and the manner in which it intends to account for the Transaction are permissible and appropriate as a matter of local law, custom and practice in jurisdictions applicable to it in this regard;
- (k) it has entered into this Transaction as principal for its own account in the normal and ordinary course of its business; and
- (l) this Transaction, including the accounting and tax treatment to be accorded to the Transaction, is consistent with all regulatory requirements arising from or applicable to this Transaction and it has taken all steps necessary to ensure that this Transaction complies with such requirements, and it will ensure that such accounting and tax treatment is appropriately reflected, if required, with the proper regulatory authorities in the applicable jurisdiction.

Party B represents and warrants to Party A and undertakes to Party A at all times:

- (m) (status) it is a company validly existing under the laws of the place of its incorporation;
- (n) (power) it has the power to enter into and perform its obligations under the Equity Derivative Financing Documents, to carry out the Equity Derivative Transactions and to carry on its business as now conducted or contemplated;
- (o) (corporate authorisation) it has taken all necessary corporate action to authorise the entry into and performance of the Equity Derivative Financing Documents, and to carry out the Transactions;
- (p) (documents binding) each Equity Derivative Financing Document is its valid and binding obligation enforceable against it in accordance with its terms, subject to the application of equitable principles and creditors rights generally and any necessary stamping and registration;
- (q) (transactions permitted) the execution and performance by it of the Equity Derivative Financing Documents and the Transaction did not and will not violate in any respect any existing provision of a law or treaty or a judgment, ruling, order or decree of a government agency binding on it, its constitution or other constituent documents, or any other document or agreement which is binding on it or its assets;
- (r) (filings and stamp taxes) all filings and registrations which are required to be effected, and all stamp, registration or similar Taxes which are required to be paid, to ensure that the Equity Derivative Financing Documents are legal, valid, binding, enforceable and admissible in evidence and have the priority that they contemplate have been effected and paid or will, within the time prescribed by law, be effected and paid;
- (s) (pari passu ranking) its payment obligations under the Equity Derivative Financing Documents rank at least pari passu with the claims of all its other unsecured and unsubordinated creditors;
- (t) (other information)
  - the information and reports (if any) that it has given to Party A or any in connection with the Equity Derivative Financing Documents are true and accurate in all material respects and not misleading in any material respect (including by omission); and
  - (ii) any forecasts, projections and opinions in them are fair and reasonable (and were made or formed on the basis of recent historical information and reasonable assumptions after due inquiry and consideration by appropriate officers of Party B),
  - as at the date of this document or, if given later, when given;
- (u) (disclosure of relevant information) Party B has disclosed to Party A all the information that is reasonably likely to be material to an assessment by it of the risks that it assumes by entering into the Equity Derivative Financing Documents;
- (v) (no default) Party B has notified Party A of any Event of Default or Potential Event of Default that has occurred and is continuing, and it is not in breach of any other agreement in a manner that could have a Material Adverse Effect;
- (v) (commercial activity) Party B is subject to civil and commercial law with respect to its obligations under the Equity Derivative Financing Documents. The execution and delivery of the Equity Derivative Financing Documents constitute, and Party B's performance of and compliance with its obligations under those documents will constitute, private and commercial acts rather than public or governmental acts;
- (w) (immunity) neither Party B nor any of its assets has any right of immunity from suit, execution, attachment or other legal process in any legal proceedings in relation to the Equity Derivative Equity Derivative Financing Documents taken in any jurisdiction, including, without limitation, in Party B's country; and
- (x) (processing of transactions) the processing of any transaction by Party A in accordance with Party B's instructions will not contravene any laws or regulations in Australia, Papua New Guinea or any other applicable jurisdiction.

These representations and warranties are repeated each time Party B enters into a Transaction with Party A and on each date on which a payment or delivery is due from Party B under the Transaction.

Each party will be deemed to represent to the other party on the date on which it enters into each Transaction that (in the absence of any written agreement between the parties which expressly imposes affirmative obligations to the contrary for this Transaction):-

(A) Non-Reliance. It is acting for its own account, and it has made its own independent decisions to enter into this Transaction and as to whether this Transaction is appropriate or proper for it based upon its

own judgment and upon advice from such advisers as it has deemed necessary. It is not relying on any communication (written or oral) of the other party as investment advice or as a recommendation to enter into this Transaction, it being understood that information and explanations related to the terms and conditions of this Transaction will not be considered investment advice or a recommendation to enter into this Transaction. No communication (written or oral) received from the other party will be deemed to be an assurance or guarantee as to the expected results of this Transaction.

- (B) Assessment and Understanding. It is capable of assessing the merits of and understanding (on its own behalf or through independent professional advice), and understands and accepts, the terms, conditions and risks of this Transaction. It is also capable of assuming, and assumes, the risks of this Transaction.
- (C) Status Of The Parties. The other party is not acting as a fiduciary for or an adviser to it in respect of this Transaction.

The parties further agree that details of this Transaction (including the identity of the counterparty) may, (1) upon request or order by any competent authority, regulatory or enforcement organisation, governmental or otherwise, including the securities exchange on which the underlying shares are listed, and (2) as required by applicable law, rules, regulations, codes or guidelines (whether having the force of law or otherwise), be disclosed in accordance with such request, order, law, rules, regulations, codes or guidelines (whether such disclosure is to be made to third parties or otherwise). By entering into the Transaction, each party agrees to such disclosure and releases the other party and any of its subsidiaries and affiliates from any duty of confidentiality owed to it in relation to such information.

#### 2) Additional Termination Event

If any of the events, circumstances or conditions referred to in Part I (*Additional Termination Events*) of Schedule 2 (*Additional Provisions*) occurs it shall constitute an Additional Termination Event with all Transactions between Party A and Party B as Affected Transactions and Party B as the sole Affected Party.

# 3) Credit Support

- (a) Credit Support Document means: in relation to Party A, not applicable and in relation to Party B, the Specific Security Deed and the Existing Nominee Deed.
- (b) Credit Support Provider means: in relation to Party A, not applicable and in relation to Party B, not applicable.
- (c) Paragraph 11 of the 1995 ISDA Credit Support Annex to the ISDA Form ("CSA") is included as Schedule 1 hereto. However, it is acknowledged and agreed by the parties that the CSA shall operate one-way in the sense that only Party B is required to transfer Eligible Credit Support (and Party A shall not) and the CSA is modified in accordance with Schedule 1.

## 4) Voting Arrangements

Party B may, by providing at least 10 Business Days' prior written notice, request a transfer of Shares included in the Credit Support Balance for the sole purpose of enabling Party B to vote at a properly convened meeting of members of the Issuer. Subject to availability of stock borrow in the Shares on terms reasonably acceptable to Party A, Party A will use its best efforts to borrower Shares from third parties to comply with any such request, provided also that:

- (a) no Potential Event of Default, Event of Default or Termination Event has occurred in respect of which Party B is the Defaulting Party or in respect of which Party B is the Affected Party;
- (b) the number of Shares requested does not exceed the number of Shares included in the Credit Support Balance at that time;
- (c) the Shares are delivered to UBS Nominees Pty Ltd (the "Nominee") who will hold the Shares in accordance with the Existing Nominee Deed;
- (d) Party B is eligible to vote at the meeting of members of the Issuer during the time when the Shares are held by the Nominee;
- (e) the Shares are to be held by the Nominee for the shortest period of time necessary to give Party B the entitlement to direct the Nominee to vote at the meeting of members referred to in paragraph (d) above, and the Nominee is subject to an enforceable, irrevocable instruction to redeliver the Shares to Party A

- as soon as possible after the record time for the member vote and Party B shall pay to Party A any costs incurred by Party A related to a delayed delivery;
- (f) an Ex-Distribution Date for the Share is not scheduled to occur during the period when Shares are held by the Nominee, and if an Ex-Distribution Date does occur during that period then the Nominee is subject to an enforceable, irrevocable instruction to promptly redeliver the Shares to Party A prior to the Ex-Distribution Date and Party B shall bear any costs incurred by UBS related to any delayed delivery relating to an Ex-Distribution Date;
- (g) if Shares are not redelivered to Party A prior to an Ex-Distribution Date and Party B (or the Nominee as nominee for Party B) becomes entitled to receive, and does receive, any Distribution, then Party B agrees to deliver to Party A, or pay to Party A an amount equal to, and the Nominee is subject to an enforceable, irrevocable instruction from Party B to promptly deliver to Party A, or pay to Party A an amount equal to, such Distribution following the making of such Distribution by the Issuer; and
- (h) Party B pays Party A the stock borrow cost incurred by Party A to procure the delivery of Shares to be held by the Nominee, by no later than the third Business Day after the Shares are returned to Party A. Such borrow cost shall be calculated on no more than the Delta Quantity of Shares.

Subject to the foregoing procedures, Party B may in accordance with the Existing Nominee Deed direct, in writing, the Nominee to exercise any voting rights it has in respect of the Shares held by the Nominee.

#### 5) Events of Default

Section 5(a) is amended as follows:

- (i) by deleting the following words from subsection (i) thereof: "if such failure is not remedied on or before the first Local Business Day in the case of any such payment or the first Local Delivery Day in the case of any such delivery after, in each case, notice of such failure is given to the party"
- (ii) by deleting the number "30" appearing on line five of subsection (ii)(1) thereof and substituting the number "10" therefor; and
- (iii) by inserting the following at the end of subsection (vii)(3) thereof:

"or a notice is sent convening a meeting to propose a voluntary arrangement of its creditors".

## 6) Cross-Default

The "Cross-Default" provisions of Section 5(a)(vi) of the ISDA Form will apply to Party B and will not apply to Party A.

## "Specified Entity" means:

- (a) In relation to Party A for the purpose of Section 5(a)(vi), none; or
- (b) In relation to Party B for the purpose of Section 5(a)(vi), none.

"Specified Indebtedness" means any obligation (whether present or future, contingent or otherwise, as principal or surety or otherwise) in respect of money borrowed or raised or under any finance lease, redeemable preference share, letter of credit, futures contract, bills facility, guarantee, indemnity or a transaction of a type described in sub-paragraphs (a)(i) and (ii), (b) and (c) of the definition of Specified Transaction.

#### "Threshold Amount":

- (a) does not apply to Party A; and
- (b) with respect to:
  - (i) Party B, means AUD25,000,000 or its equivalent in any other currency or currencies; or

(ii) a Specified Entity of Party B, means AUD25,000,000 or its equivalent in any other currency or currencies.

#### 7) Market Disruption Events

If the Expiration Date for a Tranche is a Disrupted Day then to the extent the Expiration Date is deferred to a later date in accordance with Section 3.1(f) of the Equity Definitions, the Expiration Date(s) for the unexpired Tranche(s) in the same Collar Group shall be deferred by the same number of Scheduled Trading Days.

#### 8) Miscellaneous

#### Offices:

The Office of Party A for the Transaction is Australia. The Office of Party B for the Transaction is Port Moresby.

#### **Addresses for Notices:**

The addresses for notices for the purpose of section 12(a) of the Agreement are set out on page 1 of this Confirmation.

#### **Process Agent:**

- (a) Party B irrevocably appoints Gadens Lawyers Sydney Pty Ltd ACN 100 963 308 of Level 16, 77 Castlereagh Street, Sydney NSW 2000 as its agent under this document for service of process in any proceedings in the State of New South Wales.
- (b) If any person appointed as process agent is unable for any reason to act as agent for service of process, Party B must immediately appoint another agent on terms acceptable to Party A. Failing this, Party A may appoint another person for this purpose.
- (c) Party B undertakes to maintain the appointment of the process agent until all money owed by Party A to Party B under the Equity Derivative Financing Documents is finally and irrevocably paid in full, and agrees that any process served on that person is taken to be served on it.
- (d) This clause does not affect any other method of service allowed by law.

### **Governing Law and Consent to Jurisdiction**

This Confirmation will be governed by, and construed in accordance with, the laws in force in the State of New South Wales and each party submits to the non-exclusive jurisdiction of the courts of that State without reference to choice of law doctrine.

Notwithstanding Section 13(b)(i) of the ISDA Form, each party submits to the non-exclusive jurisdiction of the courts of the State of New South Wales, and any court that may hear appeals from any of those courts, for any proceedings in connection with this Confirmation.

Party B irrevocably waives any objection to the venue of any proceedings on the ground that they have been brought in an inconvenient forum.

#### 9) FATCA - HIRE Act.

(a) FATCA PROTOCOL PROVISION. "Withholding Tax imposed on payments to non-US counterparties under the United States Foreign Account Tax Compliance Act. "Tax" as used in Part 2(a) of the Schedule (Payer Tax Representation), if applicable, and "Indemnifiable Tax" as defined in Section 14 of this Agreement shall not include any U.S. federal withholding tax imposed or collected pursuant to Sections 1471 through 1474 of the U.S. Internal Revenue Code of 1986, as amended (the "Code"), any current or future regulations or official interpretations thereof, any agreement entered into pursuant to Section 1471(b) of the Code, or any fiscal or regulatory legislation, rules or practices adopted pursuant to any intergovernmental agreement entered into in connection with the implementation of such Sections of the Code (a "FATCA Withholding Tax"). For the avoidance of doubt, a FATCA Withholding Tax is a Tax the deduction or withholding of which is required by applicable law for the purposes of Section 2(d) of this

agreement. If the parties each independently decide to adhere to any ISDA Protocol on FATCA Withholding Tax, upon effective adherence by both parties, the provisions of such Protocol shall supersede the foregoing provision

(b) SHORT FORM HIRE ACT PROTOCOL PROVISION. The parties agree that the definitions and provisions contained in the Attachment to the 2010 Short Form HIRE Act Protocol published by the International Swaps and Derivatives Association, Inc. on November 30, 2010 are incorporated into and apply to this Agreement as if set forth in full herein. The definition of "Indemnifiable Tax" shall not include any Dividend Equivalent Tax.

## 10) Termination and close-out under Related Agreements

#### (a) Application of this paragraph 10

This paragraph 10 only applies for such time that UBS AG, Australia Branch ABN 47 088 129 613 ("UBS AG") is a party to this Confirmation.

Each Party agrees that this Confirmation and all Transactions under this Confirmation and the Related Agreements are entered into in reliance on the fact that this Confirmation and the Related Agreements constitute a single agreement between the Parties and the Parties would not otherwise enter into any Transaction.

#### (b) Close-Out Date

- (i) (Designation of Close-Out Date) If an Event of Default (Related Agreement) or a Termination Event (Related Agreement) occurs, the Designating Party (Related Agreement) may, by notice to the Non-Designating Party (Related Agreement), designate a day (being a day not earlier than the date the notice is given) as the Close-Out Date.
- (ii) (Significance of the Close-Out Date) The parties agree that the terms of each Related Agreement are amended with effect from the date of this Confirmation so that a Close-Out Date is taken to be the date on which all outstanding transactions under each such Related Agreement are accelerated, terminated, liquidated or cancelled (as applicable) in accordance with the terms of the Related Agreement, despite any notification requirements under the Related Agreement that would otherwise apply.

# (c) Close-out netting

- (i) (*Termination*) If a Close-Out Date occurs, then the Terminated Obligations shall terminate immediately on such date and be replaced with the obligations in this paragraph 10(c).
- (ii) (*Termination Amounts*) The termination values of the Terminated Obligations are to be determined by the Designating Party (Related Agreement) as the net amount which would, but for this paragraph 10(c), be payable by one party to the other under the terms of the Related Agreements as determined in accordance with the terms of the Related Agreements in the relevant circumstances.

## (d) Net Termination Amount

As soon as reasonably practicable following the determination of the Termination Amount in respect of each Terminated Obligation by the party making the determinations as contemplated by the Related Agreements (the "Relevant Determining Party"), the aggregate of the Termination Amounts due from UBS AG to Party B shall be set-off against the aggregate of the Termination Amounts due from Party B to UBS AG and only the difference between such amounts (the "Net Termination Amount") shall be payable by the party from whom the larger sum is due.

#### (e) Payment of Net Termination Amount

(i) The Relevant Determining Party shall notify the other party in writing of its calculation of each Termination Amount and the Net Termination Amount promptly following its

- determination. The Net Termination Amount shall be payable by Party B to UBS AG or by UBS AG to Party B (as applicable) on the date on which such notice is effective.
- (ii) The Net Termination Amount shall accrue interest at the Default Rate from (and including) the Close-Out Date to (but excluding) the date on which the Net Termination Amount is paid.
- (iii) Upon the payment of the Net Termination Amount (together with any interest accrued on that amount), Party B's or UBS AG's obligation to pay any Net Termination Amount, as applicable, under this document shall be taken to have been satisfied and discharged, but without prejudice to any other sums that may be payable or owed under this document.
- (f) Set-off rights under Related Agreements

If a Related Agreement includes a set-off right which allows or requires its Termination Amount to be set-off in whole or in part ("**Set-Off Right**"), the Set-Off Right is deemed not to apply in respect of the Termination Amount and to apply instead in respect of the Net Termination Amount.

#### (g) Definitions

For the purposes of this paragraph 10:

"Affected Party (Related Agreement)" has the meaning given to "Affected Party" in a Related Agreement.

"Close-Out Date" means the date which is the "Close-Out Date" as a result of paragraph 10(b) (Close-Out Date).

"Defaulting Party (Related Agreement)" means the party in respect of whom an Event of Default (Related Agreement) has occurred.

"Designating Party (Related Agreement)" means (a) in respect of an Event of Default (Related Agreement), the Non-Defaulting Party (Related Agreement) or, (b) in respect of a Termination Event (Related Agreement), (i) the Non-Affected Party (Related Agreement) if there is one Affected Party (Related Agreement), or (ii) any Affected Party (Related Agreement) if there are two Affected Parties (Related Agreement).

"Event of Default (Related Agreement)" means an "Event of Default" under a Related Agreement.

"Net Termination Amount" has the meaning given to it in paragraph 10(d) (Net Termination Amount).

"Non-Affected Party (Related Agreement)" means the party to a Related Agreement which is not the Affected Party (Related Agreement).

"Non-Defaulting Party (Related Agreement)" means the party to a Related Agreement which is not the Defaulting Party (Related Agreement).

## "Related Agreements" means:

- (i) the Deemed Agreement (including, without limitation, this Confirmation);
- (ii) the Existing Equity Derivative Financing Agreement; and
- (iii) any replacement document or agreement that Party A and Party B agree in writing is to be the Related Agreement for the purposes of this document.

"Set-Off Right" has the meaning given to it in paragraph 10(f) (Set-off rights under Related Agreements).

"Termination Amount" means, in respect of a Terminated Obligation, the termination value determined under paragraph 10(c)(ii) (*Termination Amounts*) in respect of such Terminated Obligation.

"Termination Event (Related Agreement)" means a "Termination Event" under a Related Agreement.

"Terminated Obligations" means all the parties' outstanding payment and delivery obligations under each Related Agreement, in each case, other than their obligations under paragraph 10(c) (Close-out netting) immediately prior to termination under paragraph 10(c)(i) (Close-out netting).

#### 11) Definitions

Unless the otherwise defined in this Confirmation and unless the context otherwise requires, capitalized terms and expressions used in this Confirmation have the respective meanings given to them in the Global Novation and Amendment Deed. In this Confirmation the following terms also have the following meanings:

### "Equity Derivative Financing Document" means:

- (a) each Existing Equity Derivative Financing Document;
- (b) each New Equity Derivative Financing Document;
- (d) any document or agreement that is entered into under any of the above;
- (e) any document or agreement that amends, supplements, replaces or novates any of the above; and
- (f) any undertaking (whether or not in writing) by Party B that is given under or relates to any of the above.

"Ex-Distribution Date" means, in respect of any Distribution in respect of Shares, the date on which the Shares commence trading on the Exchange without entitlement to the Distribution.

# "Existing Equity Derivative Financing Documents" means:

- (a) the Existing Equity Derivative Financing Agreement;
- (b) the Existing Nominee Deed; and
- (c) the Existing Specific Security Deed.

"Existing Equity Derivative Financing Agreement" means the confirmation letter dated 12 March 2014 between Party A and the Independent State of Papua New Guinea, including any ISDA master agreement or credit support annex entered into or deemed to be entered into under or in relation to the letter or the transactions contemplated therein, as novated from the Independent State of Papua New Guinea to Party B pursuant to the Global Novation and Amendment Deed.

"Existing Nominee Deed" means the document entitled "Nominee Deed" dated 12 March 2014 between Party A, the Independent State of Papua New Guinea and UBS Nominees Pty Ltd (ABN 32 001 450 522), as novated from the Independent State of Papua New Guinea to Party B pursuant to the Global Novation and Amendment Deed (as amended, restated, novated, supplemented, modified or varied from time to time).

"Existing Side Letter" means the letter entitled "Financing transaction in respect of OSH – Confirmation Side Letter" dated 12 March 2014 between Party A and the Independent State of Papua New Guinea, as novated from the Independent State of Papua New Guinea to Party B pursuant to the Global Novation and Amendment Deed (as amended, restated, novated, supplemented, modified or varied from time to time).

"Existing Specific Security Deed" means the "Specific Security Deed (CHESS Securities – Collar)" dated 12 March 2014 between Party A and the Independent State of Papua New Guinea, as novated from the Independent State of Papua New Guinea to Party B pursuant to the Global Novation and Amendment Deed (as amended, restated, novated, supplemented, modified or varied from time to time).

"Global Novation and Amendment Deed" means the document of that title dated on or about 12 December 2014 between, among others, the Independent State of Papua New Guinea, Party A and Party B.

#### "Material Adverse Effect" means a material adverse effect on:

- (a) Party B's business, operation, property or condition (financial or otherwise);
- (b) Party B's ability to perform its obligations under the Equity Derivative Financing Documents;
- (c) the attachment, perfection, effectiveness or priority of any Security Interest given under the Equity Derivative Financing Documents; or
- (d) the validity or enforceability of any Equity Derivative Financing Document or the rights and remedies of Party A under any Equity Derivative Financing Document.

#### "New Equity Derivative Financing Documents" means:

- (a) the Deemed Agreement;
- (b) the Side Letter; and
- (c) the "Initial Price Notification Letter" given by Party A to Party B in relation to the Deemed Agreement and the Side Letter;
- (d) any document or agreement that is entered into under any of the above;
- (e) any document or agreement that amends, supplements, replaces or novates any of the above; and
- (f) any undertaking (whether or not in writing) by Party B that is given under or relates to any of the above.

"Side Letter" means the letter entitled "Transaction in respect of OSH – Confirmation Side Letter" dated on or about 12 December 2014 between Party A and Party B.

"State" means the Independent State of Papua New Guinea.

## 12) Stamp duty

Party B is solely responsible for, and must indemnify Party A against, any stamp duty including any interest or penalty that is payable on or in relation to:

- (a) this Confirmation;
- (b) the sale, purchase, assignment, transfer or declaration of trust of or over any property under this Confirmation; and
- (c) any instrument or transaction that this Confirmation contemplates (including any other Equity Derivative Financing Document).

#### SCHEDULE 1 - CREDIT SUPPORT ANNEX PARAGRAPH 11

#### Paragraph 11. Elections and Variables

- (a) Base Currency and Eligible Currency.
  - (i) "Base Currency" means Australian Dollars.
  - (ii) "Eligible Currency" means the Base Currency.

#### (b) Credit Support Obligations.

(i) Delivery Amount, Return Amount and Credit Support Amount.

#### (A) "Delivery Amount"

Party A is not required to deliver any Eligible Credit Support (whether by way of Delivery Amounts or Return Amounts or otherwise) to Party B in respect of the Transaction covered by this Confirmation.

Despite any other provision of this Confirmation, Party B must on the Initial Exchange Date deliver to Party A an initial Delivery Amount comprised of Shares in an amount equal to the Aggregate Number of Shares. No other Delivery Amounts are required to be delivered by Party B to Party A except:

(aa) in circumstances where Eligible Credit Support in the form of Shares has been transferred by Party A to the Nominee in accordance with "Additional Provisions paragraph 4 – Voting Arrangements" of this Confirmation, in which case the required Delivery Amount is equivalent to the number of Shares so transferred to the Nominee. The Delivery Amount must be delivered to Party A in accordance with "Additional Provisions paragraph 4 – Voting Arrangements" of this Confirmation and the Existing Nominee Deed; and

(bb) in circumstances where Party A has agreed to the delivery or substitution of collateral other than the Shares, in which case any required Delivery Amounts are determined in accordance with the agreement between Party A and Party B in respect of that alternative collateral.

# (B) "Return Amount"

Despite any other provision of this Confirmation, Party A must on the Cash Settlement Payment Date (if Cash Settlement applies) or Settlement Date (if Physical Settlement applies) for a Tranche, deliver a Return Amount comprised of Shares in an amount equal to the Tranche Quantity for that Tranche to Party B, provided that no Event of Default (as defined in the ISDA Form) has occurred in respect of which Party B is the Defaulting Party or Termination Event has occurred in respect of which Party B is the Affected Party.

To the extent that, on a Settlement Date, Party A is required to deliver a Return Amount to Party B (such quantity of Shares being "Quantity X") and Party B is on that date required to deliver Shares to Party A in connection with the physical settlement of the Related Options (such quantity of Shares being "Quantity Y"), Party A's obligation to deliver Shares under this Confirmation in respect of the relevant Tranche shall be satisfied by the delivery of a number of Shares to Party B on the Settlement Date for that Tranche, equal to the excess, if any, of Quantity X over Quantity Y.

- (C) "Credit Support Amount" in Paragraph 10 does not apply and, instead, for the purposes of this Confirmation, it means at any time an amount equal to the Aggregate Number of Shares comprised in the Delivery Amount on the Initial Exchange Date less any Shares delivered to Party B under paragraph (B) ("Return Amount") above.
- (ii) Eligible Credit Support. The following items will qualify as "Eligible Credit Support":

	Party B	Valuation Percentage
Fully paid ordinary shares of the Issuer, or any other collateral agreed to in advance in writing by Party A	✓	100%
Australian Dollars cash	✓	100%

"Issuer" means Oil Search Limited (ARBN 055 079 868). ASX code "OSH".

- (iii) Thresholds.
  - (A) "Independent Amount" means with respect to Party A: zero "Independent Amount" means with respect to Party B: zero
  - (B) "Threshold" means zero for both Party A and party B.
  - (C) "Minimum Transfer Amount" means AUD 1,000,000 for both Party A and Party B.
  - (D) **Rounding**. AUD 10,000
- (c) Valuation and Timing.
  - (i) "Valuation Agent" means Party A.
  - (ii) "Valuation Date" means any Local Business Day.
  - (iii) "Valuation Time" means the close of business in the Relevant Market on the Local Business Day first preceding the Valuation Date or date of calculation, as applicable.

For the purposes of this provision, "Relevant Market" means:

- (a) with respect to the calculation of Value, the principal market in which the Eligible Credit Support is traded; and
- (b) with respect to the calculation of Exposure, the location most closely associated with the relevant Transaction;

each as determined by the Valuation Agent, or as otherwise agreed between the parties.

- (iv) "Notification Time" means 3:00 p.m., Sydney time, on the Local Business Day.
- (d) Exchange Date. "Exchange Date" has the meaning specified in Paragraph 3(c)(ii).

#### (e) Dispute Resolution.

- (i) "**Resolution Time**" means 3:00 p.m., Sydney time, on the Local Business Day following the date on which the notice is given that gives rise to a dispute under Paragraph 4.
- (ii) Value. For the purpose of Paragraphs 4(a)(4)(i)(C) and 4(a)(4)(ii), , disputes over value will be resolved by the Valuation Agent seeking three bid quotes as of the relevant Valuation Date or date of Transfer from parties that regularly act as dealers in the securities or other property in question. The Value will be the Base Currency Equivalent of the arithmetic mean of the bid prices obtained by the Valuation Agent multiplied by the applicable Valuation Percentage.
- (iii) *Alternative*. The provisions of Paragraph 4 will apply.

### (f) Distributions and Interest Amount.

Despite anything to the contrary in the 1995 ISDA Credit Support Annex forming part of the ISDA Form, Party A is only required to pay "Equivalent Distributions" to Party B as contemplated under paragraph 5(c) of the 1995 ISDA Credit Support Annex as follows:

- a) If the ex-date of an Extraordinary Dividend declared in respect of the Shares occurs during the Dividend Period then Party B agrees that:
  - (i) Party A shall not be required to pay an Equivalent Distribution to Party B in respect of a quantity of Shares equal to the Delta Quantity;
  - (ii) Party A shall pay to Party B an Equivalent Distribution equal to the Actual Dividend Amount per Share in respect of the Non-Delta Quantity; and
  - (iii) payment under paragraph (ii) shall be made by no later than the second Business Day after the payment date of such Extraordinary Dividend or cash return by the Issuer.
- b) If the ex-date of an ordinary dividend declared in respect of the Shares occurs during the Dividend Period then Party B agrees that:
  - (i) Party A shall pay to Party B an Equivalent Distribution equal to \$x per Share multiplied by the Delta Quantity, where "\$x" is A\$0.08 if the ex-date is for the 2014 Final Dividend, or A\$0.10 if the ex-date is for the 2015 Interim Dividend or 2015 Final Dividend;
  - (ii) Party A shall pay to Party B an Equivalent Distribution equal to the Actual Dividend Amount per Share in respect of the Non-Delta Quantity; and
  - (iii) payment under paragraphs (i) and (ii) shall be made by no later than the second Business Day after the payment date of such ordinary dividend by the Issuer.

No Interest Amounts are payable by Party A to Party B.

For the purposes of the above:

"Delta Quantity" is a number determined by UBS that is between zero and the Aggregate Number of Shares.

"Non-Delta Quantity" is calculated as the Aggregate Number of Shares minus the Delta Quantity.

"Actual Dividend Amount" is the Australian dollar dividend amount per Share declared by the Issuer, or if the Issuer does not declare a dividend in Australian dollars then it is the dividend amount per share declared by the Issuer multiplied by an exchange rate determined by the Calculation Agent on the ex-dividend date to convert that amount into Australian dollars, less any withholding that would be applicable to Party A when it makes a payment to

Party B under this Confirmation, and less any withholding that would be applicable to a shareholder in the same position and with the same characteristics as Party A.

"**Dividend Period**" means the period from but excluding the Trade Date, to and including the Expiration Date for the relevant Component Collar.

# (g) Demands and Notices / Address for Transfers.

(i) Any demand, specification or notice under this Annex (each, a "**Notice**") must be delivered in writing which, for the avoidance of doubt, shall include electronic messaging or email. All Notices shall be delivered to the following email addresses and physical addresses:

#### With respect of Party A:

Level 16 Chifley Tower 2 Chifley Square Sydney NSW 2000 Australia

Email address: <a href="mailto:elaine.chow@ubs.com">elaine.chow@ubs.com</a> / <a href="mailto:kevin.bayett@ubs.com">kevin.bayett@ubs.com</a> / <a href="mailto:sh-aus-eq-support@ubs.com">sh-aus-eq-support@ubs.com</a> / <a href="mailto:jalpa.shillig@ubs.com">jalpa.shillig@ubs.com</a> / <a href="mailto:sh-syd-rmp-op@ubs.com">sh-syd-rmp-op@ubs.com</a> / <a href="mailto:green.gen">green.gen</a> / <a href="mailto:g

Telephone: +61 2 9324 2216 / +61 2 9324 3140 / +61 2 9324 3193

Fax number: +61-2-9324-2880

Attention: Elaine Chow / Kevin Bayett / / Jalpa Shillig / Greg Jansz / Grant

Schwulst

# With respect to Party B:

Address: Level 7 Deloitte Tower,

Douglas Street, Port Moresby NCD Papua New Guinea

Email address: robert.acevski@npcp.com

Fax number: +675 3200238 Attention: Robert Acevski

## (ii) Addresses and account details for Transfers.

## With respect of Party A:

UBS AG, Australia Branch

BSB: 946-612 Account#: 242888

Attn: Collateral Mgr Sydney Exigo Code: UBSB20 For delivery of Shares:

PID: 1505

HIN: X0000103462

#### With respect to Party B:

To be advised by Party B to Party A

#### (h) Other Provisions.

This Annex, including the act of transfer (disposition) itself, will be governed and construed in accordance with New South Wales law.

#### **SCHEDULE 2 – ADDITIONAL PROVISIONS**

#### Part I – Additional Termination Events

Each of the following events or circumstances is an Additional Termination Event:

- (a) (non-payment) Party B fails to pay any amount that is due and payable by it under an Equity Derivative Financing Document when it is due unless payment is made within 1 Local Business Day of its due date;
- (b) (other obligations) Party B fails to comply with any of its obligations under an Equity Derivative Financing Document (including Part II of this Schedule 2) (other than a failure referred to elsewhere in this Part I) and:
  - (i) Party A considers that the failure cannot be remedied; or
  - (ii) Party A considers that the failure can be remedied, and the failure is not remedied within 10 days after it occurs;
- (c) (misrepresentation) any representation, warranty or statement made or repeated by Party B in or in connection with an Equity Derivative Financing Document is untrue or misleading (whether by omission or otherwise), when so made or repeated, in any material respect;

#### (d) (Insolvency Event)

- (i) a moratorium is declared in respect of the financial indebtedness (including granting a guarantee of financial indebtedness and including any obligation (whether present or future, actual or contingent) to pay or deliver any money or commodity under or in respect of any financial accommodation or derivative transaction) of Party B;
- (ii) Party B commences negotiations with any one or more of its foreign creditors with a view to the general readjustment or rescheduling of its financial indebtedness (including granting a guarantee of financial indebtedness and including any obligation (whether present or future, actual or contingent) to pay or deliver any money or commodity under or in respect of any financial accommodation or derivative transaction):
- (iii) an Insolvency Event occurs in respect of Party B;
- (e) (Other default) an event of default, a termination event or any other event giving rise to termination or cancellation of a transaction under another Equity Derivative Financing Document (however that event may be defined or described) occurs;
- (f) (Material Adverse Effect) an event or a change occurs (whether or not foreseeable) which has a Material Adverse Effect;

# (g) (compulsory acquisition)

- (i) all or a material part of the property of Party B is compulsorily acquired by any government agency; or
- (ii) Party B sells or divests itself of all or a material part of its property because it is required to do so by a binding order from a government agency,

and Party B does not receive compensation for the acquisition, sale or disposal which is acceptable to Party A;

(h) (inability to perform) if Party B ceases for any reason to be able lawfully to carry out all the transactions which an Equity Derivative Financing Document contemplates may be carried out by it;

- (i) (provisions void) if all or any material provision of an Equity Derivative Financing Document is or becomes void, voidable, illegal or unenforceable or of limited force (other than because of equitable principles or laws affecting creditors' rights generally), or Party B claims this to be the case;
- (j) (exchange controls) any restriction is imposed on the ability of Party B to hold, deal with and remit foreign currencies which would make it impossible for it to perform its obligations under an Equity Derivative Financing Document; and
- (k) (Share Sale and Purchase Agreement):
  - (i) the Share Sale and Purchase Agreement is vitiated, terminated, rescinded or repudiated by the State for any reason; or
  - (ii) a material provision of the Share Sale and Purchase Agreement is or becomes wholly or partly invalid, void, voidable or unenforceable in any respect,

in each case without Party A's prior written consent.

#### Part II - Undertakings

Party B must:

- (a) (comply with law) comply with all applicable laws;
- (b) (hold Authorisations) obtain and maintain each Authorisation that is necessary or desirable to:
  - (i) execute the Equity Derivative Financing Documents to which it is a party and to carry out the transactions that those documents contemplate; or
  - (ii) ensure that the Equity Derivative Financing Documents to which it is a party are legal, valid, binding and admissible in evidence,

and must comply with any conditions to which any of these Authorisations is subject;

- (c) (compliance with and enforcement of the Share Sale and Purchase Agreement)
  - (i) comply fully with its obligations under the Share Sale and Purchase Agreement; and
  - (ii) enforce the Share Sale and Purchase Agreement and exercise its rights, authorities and discretions under that document prudently and in accordance with good industry practice;

#### **EXECUTION PAGES**



Please confirm that the foregoing correctly sets forth the terms of our agreement by executing the copy of this Confirmation enclosed for that purpose and returning it to us at your earliest convenience.

and accepted for and on benait of
Investments Limited
Sleaves.
FRANK KRAMER
DIRECTOR

A award and accounted for and an habalf of

Agreed and accepted for and on behalf of UBS AG, AUSTRALIA BRANCH

By:	
Name: Title:	
Ву:	
Name: Title:	

\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*

### Disclaimer

This information is communicated by UBS AG and/or its affiliates ("UBS"). UBS may from time to time, as principal or agent, have positions in, or may buy or sell, or make a market in any securities, currencies, financial instruments or other assets underlying the transaction or product to which this document relates. UBS may provide investment banking and other services to and/or have officers who serve as directors of the companies referred to in this document. UBS may pay or receive brokerage or retrocession fees in connection with this transaction. UBS' trading and/or hedging activities related to this transaction or product may have an impact on the price of the underlying asset and may affect the likelihood that any relevant barrier is crossed. UBS has policies and procedures designed to minimise the risk that officers and employees are influenced by any conflicting interest or duty and that confidential information is improperly disclosed or made available.

Structured products and OTC transactions are complex and may involve a high risk of loss. Prior to entering into a transaction or a product you should consult with your own legal, regulatory, tax, financial and accounting advisors to the extent you consider it necessary, and make your own investment, hedging and trading decisions (including decisions regarding the suitability of this transaction or product) based upon your own judgement and advice from those advisers you consider necessary.

#### **EXECUTION PAGES**

# December 2014

Please confirm that the foregoing correctly sets forth the terms of our agreement by executing the copy of this Confirmation enclosed for that purpose and returning it to us at your earliest convenience.

Agreed and accepted for and on behalf of

**NPCP** Investments Limited

By:

Name:

Title:

Agreed and accepted for and on behalf of

UBS AG, AUSTRALIA BRANCH

By: Elawelhor

Name: ELATINE OTTECTOS

\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*

Ву:

Name: CRAIG ROBERTS
Title: MANAGING OILECTOR

#### Disclaimer

This information is communicated by UBS AG and/or its affiliates ("UBS"). UBS may from time to time, as principal or agent, have positions in, or may buy or sell, or make a market in any securities, currencies, financial instruments or other assets underlying the transaction or product to which this document relates. UBS may provide investment banking and other services to and/or have officers who serve as directors of the companies referred to in this document. UBS may pay or receive brokerage or retrocession fees in connection with this transaction. UBS' trading and/or hedging activities related to this transaction or product may have an impact on the price of the underlying asset and may affect the likelihood that any relevant barrier is crossed. UBS has policies and procedures designed to minimise the risk that officers and employees are influenced by any conflicting interest or duty and that confidential information is improperly disclosed or made available.

Structured products and OTC transactions are complex and may involve a high risk of loss. Prior to entering into a transaction or a product you should consult with your own legal, regulatory, tax, financial and accounting advisors to the extent you consider it necessary, and make your own investment, hedging and trading decisions (including decisions regarding the suitability of this transaction or product) based upon your own judgement and advice from those advisers you consider necessary.

By entering into any transaction with or purchasing any product from UBS you acknowledge and agree that UBS is acting, and will at all times act, as an independent contractor on an arm's-length basis and is not acting, and will not act, in any other capacity, including as your financial adviser or fiduciary, unless expressly agreed in writing. UBS in its capacity as principal or agent is involved in a wide range of commercial banking and investment banking activities globally from which conflicting interests or duties may arise. UBS may provide services to your affiliates or any other entity or person (a "Third Party"), and may engage in any transaction (on its own account or otherwise) with respect to you or a Third Party, or act in relation to any matter for itself or any Third Party, notwithstanding that such services, transactions or actions may be adverse to you or your affiliates, and UBS may retain for its own benefit any related remuneration or profit. You should not rely on any communication (written or oral) of UBS or its employees as investment advice or as a recommendation to enter into the proposed transaction or to purchase the proposed product.

No communication (written or oral and including this document) received from UBS will be deemed to be an assurance or guarantee as to the expected results of the proposed transaction or product.

UBS makes no representation or warranty relating to any information herein which is derived from independent sources. This document shall not be copied or reproduced without UBS' prior written permission.

No action has been or will be taken in any jurisdiction that would permit a public offering of the products described herein, save where explicitly stated in the product offer document. The products or transactions described herein must be sold or offered in accordance with all applicable selling restrictions in the jurisdictions in which they are made available.

Copyright 2014 UBS AG, Australia Branch. All rights reserved. No part of this document may be reproduced or distributed in any manner without the prior written permission of UBS. UBS accepts no liability whatsoever for the actions of third parties in this respect.

# **UBS**

## Annexure B

UBS AG, Australia Branch ABN 47 088 129 613 AFSL No. 231087 Level 16 Chifley Tower 2 Chifley Square Sydney NSW 2000 Australia

> OTC Operations Tel. +612-9324 3455 Fax. +612-9324 2880

**Date:** 12 March 2014, as novated / amended and restated on or about 12 December 2014

To: NPCP Investments Limited (*Counterparty* or *Party B*)

E-mail: Robert.acevski@npcp.com

Attention: Robert Acevski Tel: +675 320 2253

Fax: +675 320 0238

From: UBS AG, Australia Branch (ABN 47 088 129 613) of Level 16, Chifley Tower, 2 Chifley

Square, Sydney NSW 2000 (UBS or Party A)

Subject: Financing transaction in respect of OSH

\_\_\_\_\_

Dear Sirs/Madams.

The purpose of this letter agreement (this "Confirmation") is to confirm the terms and conditions of the above referenced transaction entered into between Counterparty and UBS on the Trade Date specified below (the "Transaction").

This Confirmation evidences a complete binding agreement between the parties as to the terms of the Transaction to which this Confirmation relates. Until such time as the parties execute an ISDA Master Agreement and 1995 ISDA Credit Support Annex, the parties agree that this Confirmation, and any future Confirmation(s) entered into between the parties thereafter shall supplement, form part of, and be subject to, an agreement in the form of the 2002 ISDA Master Agreement as if on the Trade Date the parties had executed such agreement but without any Schedule, except for the election of Australian Dollars as the Termination Currency and New South Wales as the governing law and such other elections and modifications detailed herein referring to the ISDA Form (the "Deemed Agreement") and that each Confirmation entered into between the parties shall each be supplemented by and be subject to the terms of each 1995 ISDA Credit Support Annex with the elections and modifications provided in that Confirmation. The parties agree that the terms of each Annex to the 2002 Master Agreement Protocol published by the International Swaps and Derivatives Association, Inc ("Protocol") apply to the agreement as if the parties had adhered to the Protocol without amendment.

The definitions and provisions contained in the 2002 ISDA Equity Derivatives Definitions (the "Equity Definitions") and the 2006 ISDA Definitions (the "2006 Definitions", and together with the Equity Definitions, the "Definitions"), in each case, as published by the International Swaps and Derivatives Association, Inc., are incorporated into this Confirmation. In the event of any inconsistency between the 2006 Definitions and the Equity Definitions, the Equity Definitions will govern. In the event of any inconsistency between the Definitions and this Confirmation, this Confirmation will govern. References herein to a "Transaction" shall be deemed references to a "Share Option Transaction" for the purposes of the Equity Definitions.

In the event of any inconsistency between the provisions of the Deemed Agreement and this Confirmation, this Confirmation will prevail for the purpose of this Transaction.

The terms of the particular Transaction to which this Confirmation relates are as follows:

#### **GENERAL TERMS**

Trade Date: 6 March 2014

Effective Date: The date specified as such by Party A in a notice given to Party B

confirming that all the conditions precedent to Party A entering into the Transaction specified in Schedule 2 to the Existing Side Letter have

been satisfied, or waived by Party A.

Tax: Has the meaning given in the ISDA Form and includes any stamp,

registration, documentation or similar tax, duty, levy, fee or other

imposition.

Shares: Fully paid ordinary shares of Oil Search Limited (ARBN 055 079 868).

ASX code "OSH".

Business Days: Sydney

Currency Business Days: Sydney

Convention: If a payment or delivery is due, or a notice must be given or received,

under this Confirmation on a day that is not a Business Day, a Local Business Day or a Currency Business Day, as the case may be, then the payment or delivery must be made, or the notice must be given or received, on the next Business Day, Local Business Day or Currency

Business Day, as the case may be, immediately after that day.

Calculation Agent: Party A, provided that if Party A is the sole Defaulting Party, then the

Calculation Agent shall be a leading independent dealer in the relevant market mutually selected by the parties (the "Substitute Calculation Agent"). The Substitute Calculation Agent shall be mutually selected by the parties within one Business Day of the event that requires Calculation Agent determination. If the parties are unable to agree on a Substitute Calculation Agent, each party shall select an independent leading dealer and such independent leading dealers shall agree on an independent third party within two Business Days of the event that requires Calculation Agent determination. Unless there is a clear error, the calculations and determinations of the Substitute Calculation Agent are binding and conclusive. The fees and expenses of using the Substitute Calculation Agent, if any, shall be paid equally by Party A

and Party B.

TERMS

Initial Exchange Payer (acting as lender):

Party A

Initial Exchange: Party A shall pay to Party B the Aggregate Initial Exchange Amount

on the Initial Exchange Date, where such amount is notionally split into 30 equal tranches ("Tranches") corresponding to 30 separate Final

Exchange Amounts.

Aggregate Initial Exchange The amount specified as such in the Existing Side Letter.

Amount:

Initial Exchange Amount per Tranche:

In respect of each Tranche, the Initial Exchange Amount is calculated

Tranche Notional x (1 - Rate x t/365)

where:

"Tranche Notional" is the amount described as such in the Existing Side Letter;

"Rate" is the rate described as such in the Existing Side Letter; and "t" is the number of days from but excluding the Initial Exchange Date to and including the Collar Group Final Exchange Date for the Collar Group to which the Tranche belongs.

Initial Exchange Date: The Completion Date (as defined in the Subscription Agreement).

Final Exchange Payer (acting as borrower):

Party B

Final Exchange Amount per

Tranche:

The Final Exchange Amount for each Tranche is the amount described as such in the Existing Side Letter.

Collar Group Final Exchange Amount:

Means, in respect of each Collar Group, the amount described as such in the Existing Side Letter, which is the sum of the Final Exchange Amounts of the Tranches that make up the Collar Group.

Final Exchange Payment Date:

If Cash Settlement is applicable in respect of a Collar Group, then Party B must pay to Party A the Collar Group Final Exchange Amount on the Collar Group Final Exchange Date for the relevant Collar Group.

If Physical Settlement is applicable to a Tranche, then Party B must pay to Party A the Final Exchange Amount for that Tranche on the Settlement Date for that Tranche.

Collar Group Final Exchange Dates and Expiration Dates of each Tranche are set out in Schedule 1 to the Existing Side Letter.

Collar Group Final Exchange Date:

Means, in respect of a Collar Group, the collar group final exchange date for that Collar Group as set out in section (B) of Schedule 1 to the Existing Side Letter.

Adjustments:

If any terms of the notional Put Options and Call Options are adjusted by the Calculation Agent in accordance with this Confirmation, Party A shall be entitled to adjust the financing terms above including but not limited to the Final Exchange Amount.

### **OPTION TERMS**

### **General Terms**

Single Transaction:

Party A and Party B acknowledge and agree that

- (a) the Share Option Transaction the subject of this Confirmation is a "**Step Collar**" option comprising 6 "**Collar Groups**";
- (b) each Collar Group comprises five Tranches each with a different Expiration Date;
- (c) each Tranche is made up of five "Component Collars" with the same Expiration Date and the applicable strike prices set out in Schedule 1 to the Existing Side Letter;

- (d) each Component Collar is made up of notional "**Put Options**" and "**Call Options**" over the same number of Shares;
- (e) the Step Collar, the Collar Groups, the Tranches, the Component Collars and the notional Put Options and notional Call Options comprised within them are not separate Transactions for the purposes of the Deemed Agreement, cannot be traded or dealt with separately and all payment and delivery obligations described herein form one Transaction for the purposes of the Deemed Agreement; and
- (f) without limiting paragraph (e) above, the Component Collars that make up a Tranche and the Initial Exchange Amount and Final Exchange Amount that correspond to that Tranche cannot be terminated independently of each other (each being a "Related Transaction" of the others), and "Related Options" mean the Put Options and Call Options that make up the Component Collars in each Tranche.

Additional details of the Component Collars, Tranches and Collar Groups are set out in Schedule 1 to the Existing Side Letter.

This Confirmation must be construed and interpreted accordingly.

Put Option Seller:

Put Option Buyer:

Party A

Put Option Buyer:

Party B

Call Option Seller:

Party B

Call Option Buyer:

Party A

Option Style:

European

Option Entitlement: 1 Share per Option.

Strike Price: Means the Put Strike for each notional Put Option or the Call Strike

for each notional Call Option that make up each Component Collar set out in column 3 and column 5 (respectively) of the table in

section (A) of Schedule 1 to the Existing Side Letter.

Aggregate Number of Shares: The sum of each Tranche Quantity for all outstanding Tranches,

which as at the Effective Date is 137,012,250.

Collar Group Quantity: The number specified as such in the Existing Side Letter

Tranche Quantity The number specified as such in the Existing Side Letter

Number of Put Options per Component Th

Collar:

The number specified as such in the Existing Side Letter

Number of Call Options per Component

Collar:

The number specified as such in the Existing Side Letter

Premium: Not applicable. There is no Premium payable by Party A or Party

B in respect of the Transaction.

Expiration Date: The Expiration Date for each Tranche as set out in section (B) of

Schedule 1 to the Existing Side Letter is the Expiration Date of the Component Collars in that Tranche and all the notional Put Options and notional Call Options that make up the Component Collars in

that Tranche.

Exchange: Australian Securities Exchange ("ASX") or any successor to such

exchange or quotation system.

Related Exchange: Not applicable

## **Procedure for Exercise of Put Options and Call Options**

Expiration Time: The Scheduled Closing Time on the Exchange on the Expiration

Date of the Put Option or Call Option.

Exercise Date: The Expiration Date of the Put Option or Call Option.

Automatic Exercise: Applicable, provided that Section 3.4(c) of the Equity Definitions

shall be deleted in its entirety and replaced with the following: "In-the-Money" means, in respect of a Call Option, that the Reference Price is greater than the applicable Strike Price and, in respect of a Put Option, that the Reference Price is equal to or less

than the applicable Strike Price."

# Valuation (for the purposes of Settlement and Exercise)

Valuation Time: The Expiration Time
Valuation Date: The Expiration Date

### **Settlement Terms**

Settlement Method Election: Applicable in respect of each Collar Group, provided that any

election is irrevocable, given in respect of only one Collar Group at any one time, and the election (or default position in the absence of any election) shall apply to each Collar Group separately. Once a Settlement Method is elected for a Collar Group it shall apply to all Tranches (and therefore to all Component Collars and Related

Options) in the Collar Group.

Electing Party: Party B.

Settlement Method Election Date: The date falling 10 Scheduled Trading Days prior to the earliest

Expiration Date of each Collar Group (or such later date as Party A

may agree following a request by Party B).

Default Settlement Method: Cash Settlement

Reference Price: For the purposes of determining whether a Put Option or a Call

Option is "In-the-Money", the Reference Price is the Volume Weighted Average Price of the Share on the Exchange at the Expiration Time on the Expiration Date for that Put Option or Call

Option.

Volume Weighted Average Price (for

purposes of Reference Price):

Means the price referenced for OSH AU Equity VWAP on

Bloomberg (code RQ017).

Settlement Price: For each Put Option and Call Option in respect of which Cash

Settlement is applicable, the price per Share in Australian dollars

that Party A has achieved or considers in a commercially

reasonable manner that it would have achieved, in terminating or liquidating applicable Hedge Positions on the Expiration Date for that Put Option or Call Option plus applicable commissions and Taxes and any stamp, registration, documentation or similar tax,

duty, levy, fee or other imposition.

Cash Settlement: If Cash Settlement is applicable in respect of a Tranche, then the

Seller must pay to the Buyer the Cash Settlement Amount for that Tranche, if any, on the Cash Settlement Payment Date for that Tranche, and Section 8.1 of the Equity Definitions shall not apply.

Cash Settlement Amount:

In respect of each Tranche, the Cash Settlement Amount payable by Party B to Party A is an amount equal to the Aggregate Call Option Cash Settlement Amount minus the Aggregate Put Option Cash Settlement Amount, subject to a minimum of zero; and the Cash Settlement Amount payable by Party A to Party B is an amount equal to the Aggregate Put Option Cash Settlement Amount minus the Aggregate Call Option Cash Settlement Amount, subject to a minimum of zero.

For the purposes of this calculation:

- the Aggregate Call Option Cash Settlement Amount is the sum of the Option Cash Settlement Amounts in respect of each Call Option comprised in that Tranche which is exercised or deemed to be exercised on the Exercise Date of such Tranche, calculated in accordance with Section 8.2(b) of the Equity Definitions; and
- the Aggregate Put Option Cash Settlement Amount is the sum of the Option Cash Settlement Amounts in respect of each Put Option comprised in that Tranche which is exercised or deemed to be exercised on the Exercise Date of such Tranche, calculated in accordance with Section 8.2(b) of the Equity Definitions.

The intention of the parties is that following such calculations, only one net amount is payable by one party to the other in respect of each Tranche.

An amount equal to the greater of (a) the excess of (i) in the case of a Call Option, the relevant Settlement Price over the Call Strike, or (ii) in the case of a Put Option, the Put Strike over the relevant Settlement Price, and (b) zero.

Where Cash Settlement applies in respect of a Tranche, the date occurring one Settlement Cycle after the Expiration Date of that Tranche, provided that if such date is not a Business Day then the Cash Settlement Payment Date shall be the immediately following Currency Business Day.

Where Physical Settlement applies in respect of a Tranche, the date that falls one Settlement Cycle following the Expiration Date of that Tranche. If such date is not a Currency Business Day then the Settlement Date shall be the following Currency Business Day.

Applicable if Physical Settlement is applicable, provided that Party A shall be the Determining Party.

Dividend Amounts:

Settlement Date:

Failure to Deliver:

**Dividends** 

Strike Price Differential:

Cash Settlement Payment Date:

Without prejudice to Schedule 1 to this Confirmation, and except to the extent included as an adjustment pursuant to "Adjustments" below (pursuant to Section 11.2 of the Equity Definitions), no Dividend Amounts are payable by Party A or Party B in respect of the Transaction.

### Adjustments

Method of Adjustment:

Calculation Agent Adjustment; provided, however, that:

a) Section 11.2(c) of the Equity Definitions is deemed amended by the deletion of the words "(provided that no adjustments will be made to account solely for changes in volatility, expected dividends, stock loan rate or liquidity relative to the relevant Share)" and replacing them with the words "(including adjustments to account for changes in volatility, expected dividends, stock loan rate or liquidity relative to the relevant Share)"; and

b) Section 11.2(e)(iii) of the Equity Definitions shall be deleted.

### **Extraordinary Events**

Consequences of Merger Events

Share-for-Share: Modified Calculation Agent Adjustment
Share-for-Other: Modified Calculation Agent Adjustment
Share-for-Combined: Modified Calculation Agent Adjustment

Tender Offer: Applicable

Consequences of Tender Offers

Share-for-Share: Modified Calculation Agent Adjustment
Share-for-Other: Modified Calculation Agent Adjustment
Share-for-Combined: Modified Calculation Agent Adjustment

Composition of Combined Consideration: Not Applicable

Nationalization, Insolvency or Delisting: Cancellation and Payment - Calculation Agent Determination

Determining Party: Party A, in all cases

**Additional Disruption Events** 

Change in Law: Applicable, <u>provided</u> that Section 12.9(a)(ii)(B) of the Equity

Definitions is replaced in its entirety as follows: "(B) due to the promulgation of or any change in the interpretation by any court, tribunal or regulatory authority with competent jurisdiction of any applicable law or regulation (including any action taken by a taxing authority), the Calculation Agent determines in good faith that (X) it has become illegal to hold, acquire or dispose of Hedge Positions relating to such Transaction or (Y) a party to the

Transaction will incur materially increased cost in performing its obligations under such Transaction (including, without limitation, due to any increase in tax liability, decrease in tax benefit or other

adverse effect on its tax position)".

Hedge Positions: The definition of "Hedge Positions" in Section 13.2(b) of the

Equity Definitions shall be amended by inserting the words "or an Affiliate thereof" after the words "a party" in the third line.

Insolvency Filing: Applicable

Hedging Disruption: Applicable, provided that Section 12.9(a)(v) of the Equity

Definitions is replaced in its entirety as follows: "(v) Hedging Disruption" means that the Hedging Party is unable, after using commercially reasonable efforts, to either (i) acquire, establish, re-establish, substitute, maintain, unwind or dispose of any transaction(s) or asset(s) it deems necessary to hedge the equity price risk (or any other relevant price risk, including, but not limited to, any currency risk) of entering into and performing its obligations with respect to this Transaction, or (ii) freely realise, recover, receive, repatriate, remit or transfer the proceeds of Hedge Positions or this Transaction between accounts within the jurisdiction of the Hedge Positions (the "Affected Jurisdiction") or from accounts within the Affected Jurisdiction to accounts

outside of the Affected Jurisdiction."

Consequences of Hedging Disruption:

Section 12.9(b)(iii) of the Equity Definitions is replaced with the

following:

"(iii) If 'Hedging Disruption' is specified in the relevant Confirmation to be applicable to a Transaction, then upon the occurrence of such an event, the Hedging Party may elect to terminate the Transaction upon notice to the Non-Hedging Party specifying the date of such termination, which may be the day on which the notice of termination is issued, in which event the Determining Party will determine the Cancellation Amount

payable by one party to the other."

Increased Cost of Hedging: Applicable, provided that Section 12.9(a)(vi) of the Equity

Definitions is replaced as follows: "(vi) "Increased Cost of Hedging" means that the Hedging Party would incur a materially increased (as compared with the circumstances that existed on the Trade Date) amount of tax, duty, expense or fee (other than brokerage commissions) to (A) acquire, establish, re-establish, substitute, maintain, unwind or dispose of any transaction(s) or asset(s) it deems necessary to hedge the equity price risk (or any other relevant price risk including, but not limited to, currency risk) of entering into and performing its obligations with respect to this Transaction or (B) freely realise, recover, receive,

repatriate, remit or transfer the proceeds of the Hedge Positions or

this Transaction between accounts within the Affected

Jurisdiction or from accounts within the Affected Jurisdiction to

accounts outside of the Affected Jurisdiction."

Loss of Stock Borrow: Applicable, provided however that Section 12.9(b)(iv)(B) of the

Applicable

Equity Definitions is deemed amended to include after the words "(B) refer the Hedging Party to a Lending Party" and before the words "that will lend", the words "acceptable to the Hedging

Party in its sole discretion".

Maximum Stock Loan Rate: 0% per annum Increased Cost of Stock Borrow:

Initial Stock Loan Rate: 0% per annum

Hedging Party: Party A **Determining Party:** Party A

Non-Reliance: **Applicable** Applicable

Agreements and Acknowledgements Regarding Hedging Activities:

**Applicable** Additional Acknowledgements:

ACCOUNT DETAILS

Account for payments to Party B: To be advised

Beneficiary Bank: UBS AG Australia Bank Account for payments to Party A:

Beneficiary Bank SWIFT: UBSWAU2S Account holder name: UBS AG Australia UBS AG, Australia Branch Correspondent:

BSB: 946 612 A/c: 242624 Swift: UBSWAU2S

### ADDITIONAL PROVISIONS

For the purposes of this Transaction:

### 1) Disclosure requirement and additional representations and warranties

Party B represents and warrants to Party A:

- (a) that, as at the Trade Date for each Component Collar and on the Initial Exchange Date, there are no security interests (as defined in the *Personal Property Securities Act 2009* (Cth) or any other encumbrance of any nature whatsoever) in respect of the Shares that would be delivered to Party A under the terms of this Confirmation (including in accordance with Schedule 1 to this Confirmation);
- (b) that it will seek its own advice on the appropriateness and suitability of the Transaction and any disclosure obligations it may have relating to its dealings in Shares and Transaction in respect of Shares;
- (c) at the time of entry into the Transaction and on a continuous basis during the term of the Transaction that, by entering into and performing its obligations under the Transaction, it is not, and will not be, acting in breach of any:
  - i. laws, regulations or Authorisations binding on it or any of its property or to which any of its property may be subject;
  - ii. agreements, instruments, undertakings or other restrictions binding upon it or any of its property (other than any negative pledges) in a manner that could have a Material Adverse Effect; or
  - iii. negative pledges binding on or applicable to it;
- (d) that neither Counterparty nor any of its Affiliates nor anyone acting on Counterparty's or any Affiliate's behalf has engaged in any behaviour which is designed to cause, has caused, or might reasonably be expected to cause manipulation of the price of any security of the Issuer;
- (e) it will record in its books and account for the value of this Transaction in accordance with the policies and rules applicable to it;
- (f) it has complied and will comply with all laws, regulations and administrative provisions applicable to it in connection with the Transaction, including making all relevant disclosures to, or obtaining any required approvals from all regulatory authorities in respect of its holding in the Shares (if any) and in respect of the Transaction as required by, and within the time limits required by, such regulatory authorities or pursuant to applicable laws or regulations in this regard;
- (g) neither this Transaction nor any delivery of Shares or payment of cash contemplated hereby will constitute a violation by Counterparty or any Affiliate of any applicable law or regulations of any applicable jurisdiction prohibiting "insider dealing" in, or market abuse in respect of, securities;
- (h) it is not in possession of any material information about the Shares or the Issuer which has not been publicly disclosed, which might have influenced its decision to enter into the Transaction or any particular element thereof and it is not entering into the Transaction on the basis of any such information;
- (i) it has full authority and capacity to enter into the Transaction and the execution and delivery of this Confirmation and the consummation by it of the Transaction contemplated hereby have been duly authorised by all necessary corporate action. All consents, orders, approvals, and other authorisations, whether governmental, corporate or other, necessary for the Transaction have been obtained or made and are in full force and effect;
- (j) the purpose and effect of the Transaction and the manner in which it intends to account for the Transaction are permissible and appropriate as a matter of local law, custom and practice in jurisdictions applicable to it in this regard;
- (k) it has entered into this Transaction as principal for its own account in the normal and ordinary course of its business; and
- (l) this Transaction, including the accounting and tax treatment to be accorded to the Transaction, is consistent with all regulatory requirements arising from or applicable to this Transaction and it has taken all steps necessary to ensure that this Transaction complies with such requirements, and it will ensure that such accounting and tax treatment is appropriately reflected, if required, with the proper regulatory authorities in the applicable jurisdiction.

Party B represents and warrants to Party A and undertakes to Party A at all times:

- (m) (status) it is a company validly existing under the laws of the place of its incorporation;
- (n) (power) it has the power to enter into and perform its obligations under the Equity Derivative Financing Documents, to carry out the Equity Derivative Transactions and to carry on its business as now conducted or contemplated;
- (o) (corporate authorisation) it has taken all necessary corporate action to authorise the entry into and performance of the Equity Derivative Financing Documents, and to carry out the Equity Derivative Transactions;
- (p) (documents binding) each Equity Derivative Financing Document is its valid and binding obligation enforceable against it in accordance with its terms, subject to the application of equitable principles and creditors rights generally and any necessary stamping and registration;
- (q) (transactions permitted) the execution and performance by it of the Equity Derivative Financing Documents and the Transaction did not and will not violate in any respect any existing provision of a law or treaty or a judgment, ruling, order or decree of a government agency binding on it, its constitution or other constituent documents, or any other document or agreement which is binding on it or its assets;
- (r) (filings and stamp taxes) all filings and registrations which are required to be effected, and all stamp, registration or similar Taxes which are required to be paid, to ensure that the Equity Derivative Financing Documents are legal, valid, binding, enforceable and admissible in evidence and have the priority that they contemplate have been effected and paid or will, within the time prescribed by law, be effected and paid;
- (s) (pari passu ranking) its payment obligations under the Equity Derivative Financing Documents rank at least pari passu with the claims of all its other unsecured and unsubordinated creditors;
- (t) (other information)
  - (i) the information and reports (if any) that it has given to Party A or any in connection with the Equity Derivative Financing Documents are true and accurate in all material respects and not misleading in any material respect (including by omission); and
  - (ii) any forecasts, projections and opinions in them are fair and reasonable (and were made or formed on the basis of recent historical information and reasonable assumptions after due inquiry and consideration by appropriate officers of Party B),
  - as at the date of this document or, if given later, when given;
- (u) (disclosure of relevant information) Party B has disclosed to Party A all the information that is reasonably likely to be material to an assessment by it of the risks that it assumes by entering into the Equity Derivative Financing Documents;
- (v) (no default) Party B has notified Party A of any Event of Default or Potential Event of Default that has occurred and is continuing, and it is not in breach of any other agreement in a manner that could have a Material Adverse Effect;
- (v) (commercial activity) Party B is subject to civil and commercial law with respect to its obligations under the Equity Derivative Financing Documents. The execution and delivery of the Equity Derivative Financing Documents constitute, and Party B's performance of and compliance with its obligations under those documents will constitute, private and commercial acts rather than public or governmental acts;
- (w) (immunity) neither Party B nor any of its assets has any right of immunity from suit, execution, attachment or other legal process in any legal proceedings in relation to the Equity Derivative Equity Derivative Financing Documents taken in any jurisdiction, including, without limitation, in Party B's country; and
- (x) (processing of transactions) the processing of any transaction by Party A in accordance with Party B's instructions will not contravene any laws or regulations in Australia, Papua New Guinea or any other applicable jurisdiction.

These representations and warranties are repeated each time Party B enters into a Transaction with Party A and on each date on which a payment or delivery is due from Party B under the Transaction.

Each party will be deemed to represent to the other party on the date on which it enters into each Transaction that (in the absence of any written agreement between the parties which expressly imposes affirmative obligations to the contrary for this Transaction):-

- (A) Non-Reliance. It is acting for its own account, and it has made its own independent decisions to enter into this Transaction and as to whether this Transaction is appropriate or proper for it based upon its own judgment and upon advice from such advisers as it has deemed necessary. It is not relying on any communication (written or oral) of the other party as investment advice or as a recommendation to enter into this Transaction, it being understood that information and explanations related to the terms and conditions of this Transaction will not be considered investment advice or a recommendation to enter into this Transaction. No communication (written or oral) received from the other party will be deemed to be an assurance or guarantee as to the expected results of this Transaction.
- (B) Assessment and Understanding. It is capable of assessing the merits of and understanding (on its own behalf or through independent professional advice), and understands and accepts, the terms, conditions and risks of this Transaction. It is also capable of assuming, and assumes, the risks of this Transaction.
- (C) Status Of The Parties. The other party is not acting as a fiduciary for or an adviser to it in respect of this Transaction.

The parties further agree that details of this Transaction (including the identity of the counterparty) may, (1) upon request or order by any competent authority, regulatory or enforcement organisation, governmental or otherwise, including the securities exchange on which the underlying shares are listed, and (2) as required by applicable law, rules, regulations, codes or guidelines (whether having the force of law or otherwise), be disclosed in accordance with such request, order, law, rules, regulations, codes or guidelines (whether such disclosure is to be made to third parties or otherwise). By entering into the Transaction, each party agrees to such disclosure and releases the other party and any of its subsidiaries and affiliates from any duty of confidentiality owed to it in relation to such information.

### 2) Additional Termination Event

If any of the events, circumstances or conditions referred to in Part I (*Additional Termination Events*) of Schedule 2 (*Additional Provisions*) occurs it shall constitute an Additional Termination Event with all Transactions between Party A and Party B as Affected Transactions and Party B as the sole Affected Party.

### 3) Credit Support

- (a) Credit Support Document means: in relation to Party A, not applicable and in relation to Party B, the Existing Specific Security Deed and the Existing Nominee Deed.
- (b) Credit Support Provider means: in relation to Party A, not applicable and in relation to Party B, not applicable.
- (c) Paragraph 11 of the 1995 ISDA Credit Support Annex to the ISDA Form ("CSA") is included as Schedule 1 hereto. However, it is acknowledged and agreed by the parties that the CSA shall operate one-way in the sense that only Party B is required to transfer Eligible Credit Support (and Party A shall not) and the CSA is modified in accordance with Schedule 1.

### 4) Voting Arrangements

Party B may, by providing at least 10 Business Days' prior written notice, request a transfer of Shares included in the Credit Support Balance for the sole purpose of enabling Party B to vote at a properly convened meeting of members of the Issuer. Subject to availability of stock borrow in the Shares on terms reasonably acceptable to Party A, Party A will use its best efforts to borrower Shares from third parties to comply with any such request, provided also that:

- (a) no Potential Event of Default, Event of Default or Termination Event has occurred in respect of which Party B is the Defaulting Party or in respect of which Party B is the Affected Party;
- (b) the number of Shares requested does not exceed the number of Shares included in the Credit Support Balance at that time;
- (c) the Shares are delivered to UBS Nominees Pty Ltd (the "Nominee") who will hold the Shares in accordance with the Existing Nominee Deed;
- (d) Party B is eligible to vote at the meeting of members of the Issuer during the time when the Shares are held by the Nominee;
- (e) the Shares are to be held by the Nominee for the shortest period of time necessary to give Party B the entitlement to direct the Nominee to vote at the meeting of members referred to in paragraph (d) above,

- and the Nominee is subject to an enforceable, irrevocable instruction to redeliver the Shares to Party A as soon as possible after the record time for the member vote and Party B shall pay to Party A any costs incurred by Party A related to a delayed delivery;
- (f) an Ex-Distribution Date for the Share is not scheduled to occur during the period when Shares are held by the Nominee, and if an Ex-Distribution Date does occur during that period then the Nominee is subject to an enforceable, irrevocable instruction to promptly redeliver the Shares to Party A prior to the Ex-Distribution Date and Party B shall bear any costs incurred by UBS related to any delayed delivery relating to an Ex-Distribution Date;
- (g) if Shares are not redelivered to Party A prior to an Ex-Distribution Date and Party B (or the Nominee as nominee for Party B) becomes entitled to receive, and does receive, any Distribution, then Party B agrees to deliver to Party A, or pay to Party A an amount equal to, and the Nominee is subject to an enforceable, irrevocable instruction from Party B to promptly deliver to Party A, or pay to Party A an amount equal to, such Distribution following the making of such Distribution by the Issuer; and
- (h) Party B pays Party A the stock borrow cost incurred by Party A to procure the delivery of Shares to be held by the Nominee, by no later than the third Business Day after the Shares are returned to Party A. Such borrow cost shall be calculated on no more than the Delta Quantity of Shares.

Subject to the foregoing procedures, Party B may in accordance with the Existing Nominee Deed direct, in writing, the Nominee to exercise any voting rights it has in respect of the Shares held by the Nominee.

### 5) Events of Default

Section 5(a) is amended as follows:

- (i) by deleting the following words from subsection (i) thereof: "if such failure is not remedied on or before the first Local Business Day in the case of any such payment or the first Local Delivery Day in the case of any such delivery after, in each case, notice of such failure is given to the party"
- (ii) by deleting the number "30" appearing on line five of subsection (ii)(1) thereof and substituting the number "10" therefor; and
- (iii) by inserting the following at the end of subsection (vii)(3) thereof:

"or a notice is sent convening a meeting to propose a voluntary arrangement of its creditors".

### 6) Cross-Default

The "Cross-Default" provisions of Section 5(a)(vi) of the ISDA Form will apply to Party B and will not apply to Party A.

### "Specified Entity" means:

- (a) In relation to Party A for the purpose of Section 5(a)(vi), none; or
- (b) In relation to Party B for the purpose of Section 5(a)(vi), none.

"Specified Indebtedness" means any obligation (whether present or future, contingent or otherwise, as principal or surety or otherwise) in respect of money borrowed or raised or under any finance lease, redeemable preference share, letter of credit, futures contract, bills facility, guarantee, indemnity or a transaction of a type described in sub-paragraphs (a)(i) and (ii), (b) and (c) of the definition of Specified Transaction.

### "Threshold Amount":

- (a) does not apply to Party A; and
- (b) with respect to:
  - (i) Party B, means AUD25,000,000 or its equivalent in any other currency or currencies; or

(ii) a Specified Entity of Party B, means AUD25,000,000 or its equivalent in any other currency or currencies.

### 7) Market Disruption Events

If the Expiration Date for a Tranche is a Disrupted Day then to the extent the Expiration Date is deferred to a later date in accordance with Section 3.1(f) of the Equity Definitions, the Expiration Date(s) for the unexpired Tranche(s) in the same Collar Group shall be deferred by the same number of Scheduled Trading Days.

### 8) Miscellaneous

### Offices:

The Office of Party A for the Transaction is Australia. The Office of Party B for the Transaction is Port Moresby.

### **Addresses for Notices:**

The addresses for notices for the purpose of section 12(a) of the Agreement are set out on page 1 of this Confirmation.

### **Process Agent:**

- (a) Party B irrevocably appoints Gadens Lawyers Sydney Pty Ltd ACN 100 963 308 of Level 16, 77 Castlereagh Street, Sydney NSW 200 as its agent under this document for service of process in any proceedings in the State of New South Wales.
- (b) If any person appointed as process agent is unable for any reason to act as agent for service of process, Party B must immediately appoint another agent on terms acceptable to Party A. Failing this, Party A may appoint another person for this purpose.
- (c) Party B undertakes to maintain the appointment of the process agent until all money owed by Party A to Party B under the Equity Derivative Financing Documents is finally and irrevocably paid in full, and agrees that any process served on that person is taken to be served on it.
- (d) This clause does not affect any other method of service allowed by law.

### **Governing Law and Consent to Jurisdiction**

This Confirmation will be governed by, and construed in accordance with, the laws in force in the State of New South Wales and each party submits to the non-exclusive jurisdiction of the courts of that State without reference to choice of law doctrine.

Notwithstanding Section 13(b)(i) of the ISDA Form, each party submits to the non-exclusive jurisdiction of the courts of the State of New South Wales, and any court that may hear appeals from any of those courts, for any proceedings in connection with this Confirmation.

Party B irrevocably waives any objection to the venue of any proceedings on the ground that they have been brought in an inconvenient forum.

### 9) FATCA - HIRE Act.

(a) FATCA PROTOCOL PROVISION. "Withholding Tax imposed on payments to non-US counterparties under the United States Foreign Account Tax Compliance Act. "Tax" as used in Part 2(a) of the Schedule (Payer Tax Representation), if applicable, and "Indemnifiable Tax" as defined in Section 14 of this Agreement shall not include any U.S. federal withholding tax imposed or collected pursuant to Sections 1471 through 1474 of the U.S. Internal Revenue Code of 1986, as amended (the "Code"), any current or future regulations or official interpretations thereof, any agreement entered into pursuant to Section 1471(b) of the Code, or any fiscal or regulatory legislation, rules or practices adopted pursuant to any intergovernmental agreement entered into in connection with the implementation of such Sections of the Code (a "FATCA Withholding Tax"). For the avoidance of doubt, a FATCA Withholding Tax is a Tax the deduction or withholding of which is required by applicable law for the purposes of Section 2(d) of this

agreement. If the parties each independently decide to adhere to any ISDA Protocol on FATCA Withholding Tax, upon effective adherence by both parties, the provisions of such Protocol shall supersede the foregoing provision

(b) SHORT FORM HIRE ACT PROTOCOL PROVISION. The parties agree that the definitions and provisions contained in the Attachment to the 2010 Short Form HIRE Act Protocol published by the International Swaps and Derivatives Association, Inc. on November 30, 2010 are incorporated into and apply to this Agreement as if set forth in full herein. The definition of "Indemnifiable Tax" shall not include any Dividend Equivalent Tax.

### 10) Definitions

Unless the otherwise defined in this Confirmation and unless the context otherwise requires, capitalized terms and expressions used in this Confirmation have the respective meanings given to them in the Global Novation and Amendment Deed. In this Confirmation the following terms also have the following meanings:

### "Equity Derivative Financing Document" means:

- (a) each Existing Equity Derivative Financing Document;
- (b) each New Equity Derivative Financing Document;
- (c) any document or agreement that is entered into under any of the above;
- (d) the Global Novation and Amendment Deed;
- (e) any document or agreement that amends, supplements, replaces or novates any of the above; and
- (f) any undertaking (whether or not in writing) by Party B that is given under or relates to any of the above.

"Ex-Distribution Date" means, in respect of any Distribution in respect of Shares, the date on which the Shares commence trading on the Exchange without entitlement to the Distribution.

### "Existing Equity Derivative Financing Documents" means:

- (a) the Deemed Agreement;
- (b) the Existing Nominee Deed; and
- (c) the Existing Specific Security Deed.

"Existing Nominee Deed" means the document entitled "Nominee Deed" dated 12 March 2014 between Party A, the Independent State of Papua New Guinea and UBS Nominees Pty Ltd (ABN 32 001 450 522), as novated from the Independent State of Papua New Guinea to Party B pursuant to the Global Novation and Amendment Deed (as amended, restated, novated, supplemented, modified or varied from time to time).

**"Existing Side Letter"** means the letter entitled **"Financing transaction in respect of OSH – Confirmation Side Letter"** dated 12 March 2014 between Party A and the Independent State of Papua New Guinea, as novated from the Independent State of Papua New Guinea to Party B pursuant to the Global Novation and Amendment Deed (as amended, restated, novated, supplemented, modified or varied from time to time).

"Existing Specific Security Deed" means the "Specific Security Deed (CHESS Securities – Collar)" dated 12 March 2014 between Party A and the Independent State of Papua New Guinea, as novated from the Independent State of Papua New Guinea to Party B pursuant to the Global Novation and Amendment Deed (as amended, restated, novated, supplemented, modified or varied from time to time).

"Global Novation and Amendment Deed" means the document of that title dated on or about 12 December 2014 between, among others, the Independent State of Papua New Guinea, Party A and Party B.

### "Material Adverse Effect" means a material adverse effect on:

- (a) Party B's business, operation, property or condition (financial or otherwise);
- (b) Party B's ability to perform its obligations under the Equity Derivative Financing Documents;
- (c) the attachment, perfection, effectiveness or priority of any Security Interest given under the Equity Derivative Financing Documents; or
- (d) the validity or enforceability of any Equity Derivative Financing Document or the rights and remedies of Party A under any Equity Derivative Financing Document.

### "New Equity Derivative Financing Documents" means:

- (a) the confirmation letter dated on or about 12 December 2014 between Party A and Party B, including any ISDA master agreement or credit support annex entered into or deemed to be entered into under or in relation to the letter or the transactions contemplated therein;
- (b) the letter entitled "Transaction in respect of OSH Confirmation Side Letter" dated on or about 12 December 2014 between Party A and Party B;
- (c) the "Initial Price Notification Letter" given by Party A to Party B in relation to the Deemed Agreement and the Side Letter;
- (d) any document or agreement that is entered into under any of the above;
- (e) any document or agreement that amends, supplements, replaces or novates any of the above; and
- (f) any undertaking (whether or not in writing) by Party B that is given under or relates to any of the above.
- "Side Letter" means the letter entitled "Transaction in respect of OSH Confirmation Side Letter" dated on or about 12 December 2014 between Party A and Party B relating to the confirmation letter dated on or about 12 December 2014 between Party A and Party B.

"State" means the Independent State of Papua New Guinea.

### 11) Stamp duty

Party B is solely responsible for, and must indemnify Party A against, any stamp duty including any interest or penalty that is payable on or in relation to:

- (a) this Confirmation;
- (b) the sale, purchase, assignment, transfer or declaration of trust of or over any property under this Confirmation; and
- (c) any instrument or transaction that this Confirmation contemplates (including any other Equity Derivative Financing Document).

### SCHEDULE 1 - CREDIT SUPPORT ANNEX PARAGRAPH 11

### Paragraph 11. Elections and Variables

- (a) Base Currency and Eligible Currency.
  - (i) "Base Currency" means Australian Dollars.
  - (ii) "Eligible Currency" means the Base Currency.

### (b) Credit Support Obligations.

(i) Delivery Amount, Return Amount and Credit Support Amount.

### (A) "Delivery Amount"

Party A is not required to deliver any Eligible Credit Support (whether by way of Delivery Amounts or Return Amounts or otherwise) to Party B in respect of the Transaction covered by this Confirmation.

Despite any other provision of this Confirmation, Party B must on the Initial Exchange Date deliver to Party A an initial Delivery Amount comprised of Shares in an amount equal to the Aggregate Number of Shares. No other Delivery Amounts are required to be delivered by Party B to Party A except:

(aa) in circumstances where Eligible Credit Support in the form of Shares has been transferred by Party A to the Nominee in accordance with "Additional Provisions paragraph 4 – Voting Arrangements" of this Confirmation, in which case the required Delivery Amount is equivalent to the number of Shares so transferred to the Nominee. The Delivery Amount must be delivered to Party A in accordance with "Additional Provisions paragraph 4 – Voting Arrangements" of this Confirmation and the Existing Nominee Deed; and

(bb) in circumstances where Party A has agreed to the delivery or substitution of collateral other than the Shares, in which case any required Delivery Amounts are determined in accordance with the agreement between Party A and Party B in respect of that alternative collateral.

### (B) "Return Amount"

Despite any other provision of this Confirmation, Party A must on the Cash Settlement Payment Date (if Cash Settlement applies) or Settlement Date (if Physical Settlement applies) for a Tranche, deliver a Return Amount comprised of Shares in an amount equal to the Tranche Quantity for that Tranche to Party B, provided that no Event of Default (as defined in the ISDA Form) has occurred in respect of which Party B is the Defaulting Party or Termination Event has occurred in respect of which Party B is the Affected Party.

To the extent that, on a Settlement Date, Party A is required to deliver a Return Amount to Party B (such quantity of Shares being "Quantity X") and Party B is on that date required to deliver Shares to Party A in connection with the physical settlement of the Related Options (such quantity of Shares being "Quantity Y"), Party A's obligation to deliver Shares under this Confirmation in respect of the relevant Tranche shall be satisfied by the delivery of a number of Shares to Party B on the Settlement Date for that Tranche, equal to the excess, if any, of Quantity X over Quantity Y.

- (C) "Credit Support Amount" in Paragraph 10 does not apply and, instead, for the purposes of this Confirmation, it means at any time an amount equal to the Aggregate Number of Shares comprised in the Delivery Amount on the Initial Exchange Date less any Shares delivered to Party B under paragraph (B) ("Return Amount") above.
- (ii) Eligible Credit Support. The following items will qualify as "Eligible Credit Support":

	Party B	Valuation Percentage
Fully paid ordinary shares of the Issuer, or any other collateral agreed to in advance in writing by Party A	✓	100%
Australian Dollars cash	✓	100%

<sup>&</sup>quot;Issuer" means Oil Search Limited (ARBN 055 079 868). ASX code "OSH".

- (iii) Thresholds.
  - (A) "Independent Amount" means with respect to Party A: zero "Independent Amount" means with respect to Party B: zero
  - (B) "Threshold" means zero for both Party A and party B.
  - (C) "Minimum Transfer Amount" means AUD 1,000,000 for both Party A and Party B.
  - (D) **Rounding**. AUD 10,000
- (c) Valuation and Timing.
  - (i) "Valuation Agent" means Party A.
  - (ii) "Valuation Date" means any Local Business Day.
  - (iii) "Valuation Time" means the close of business in the Relevant Market on the Local Business Day first preceding the Valuation Date or date of calculation, as applicable.

For the purposes of this provision, "Relevant Market" means:

- (a) with respect to the calculation of Value, the principal market in which the Eligible Credit Support is traded; and
- (b) with respect to the calculation of Exposure, the location most closely associated with the relevant Transaction;

each as determined by the Valuation Agent, or as otherwise agreed between the parties.

- (iv) "Notification Time" means 3:00 p.m., Sydney time, on the Local Business Day.
- (d) **Exchange Date**. "**Exchange Date**" has the meaning specified in Paragraph 3(c)(ii).
- (e) Dispute Resolution.

- (i) "**Resolution Time**" means 3:00 p.m., Sydney time, on the Local Business Day following the date on which the notice is given that gives rise to a dispute under Paragraph 4.
- (ii) Value. For the purpose of Paragraphs 4(a)(4)(i)(C) and 4(a)(4)(ii), disputes over value will be resolved by the Valuation Agent seeking three bid quotes as of the relevant Valuation Date or date of Transfer from parties that regularly act as dealers in the securities or other property in question. The Value will be the Base Currency Equivalent of the arithmetic mean of the bid prices obtained by the Valuation Agent, multiplied by the applicable Valuation Percentage.
- (iii) *Alternative*. The provisions of Paragraph 4 will apply.

### (f) Distributions and Interest Amount.

Despite anything to the contrary in the 1995 ISDA Credit Support Annex forming part of the ISDA Form, Party A is only required to pay "Equivalent Distributions" to Party B as contemplated under paragraph 5(c) of the 1995 ISDA Credit Support Annex as follows:

- a) If the ex-date of an Extraordinary Dividend declared in respect of the Shares occurs during the Dividend Period then Party B agrees that:
  - (i) Party A shall not be required to pay an Equivalent Distribution to Party B in respect of a quantity of Shares equal to the Delta Quantity;
  - (ii) Party A shall pay to Party B an Equivalent Distribution equal to the Actual Dividend Amount per Share in respect of the Non-Delta Quantity; and
  - (iii) payment under paragraph (ii) shall be made by no later than the second Business Day after the payment date of such Extraordinary Dividend or cash return by the Issuer.
- b) If the ex-date of an ordinary dividend declared in respect of the Shares occurs during the Dividend Period then Party B agrees that:
  - (i) Party A shall pay to Party B an Equivalent Distribution equal to A\$0.022 per Share in respect of the Delta Quantity;
  - (ii) Party A shall pay to Party B an Equivalent Distribution equal to the Actual Dividend Amount per Share in respect of the Non-Delta Quantity; and
  - (iii) payment under paragraphs (i) and (ii) shall be made by no later than the second Business Day after the payment date of such ordinary dividend by the Issuer.

No Interest Amounts are payable by Party A to Party B.

For the purposes of the above:

"Delta Quantity" is a number determined by UBS that is between zero and the Aggregate Number of Shares.

"Non-Delta Quantity" is calculated as the Aggregate Number of Shares minus the Delta Quantity.

"Actual Dividend Amount" is the Australian dollar dividend amount per Share declared by the Issuer, or if the Issuer does not declare a dividend in Australian dollars then it is the dividend amount per share declared by the Issuer multiplied by an exchange rate determined by the Calculation Agent on the ex-dividend date to convert that amount into Australian dollars, less any withholding that would be applicable to Party A when it makes a payment to Party B under this Confirmation, and less any withholding that would be applicable to a shareholder in the same position and with the same characteristics as Party A.

"**Dividend Period**" means the period from but excluding the Trade Date, to and including the Expiration Date for the relevant Component Collar.

### (g) Demands and Notices / Address for Transfers.

(i) Any demand, specification or notice under this Annex (each, a "**Notice**") must be delivered in writing which, for the avoidance of doubt, shall include electronic messaging or email. All Notices shall be delivered to the following email addresses and physical addresses:

### With respect of Party A:

Level 16 Chifley Tower 2 Chifley Square Sydney NSW 2000 Australia

Email address: <a href="mailto:elaine.chow@ubs.com">elaine.chow@ubs.com</a> / <a href="mailto:kevin.bayett@ubs.com">kevin.bayett@ubs.com</a> / <a href="mailto:sh-aus-eq-support@ubs.com">sh-aus-eq-support@ubs.com</a> / <a href="mailto:grant.schwulst@ubs.com">grant.schwulst@ubs.com</a> / <a href="mailto:sh-syd-rmp-op@ubs.com">sh-syd-rmp-op@ubs.com</a> / <a href="mailto:s

Telephone: +61 2 9324 2216 / +61 2 9324 3140 / +61 2 9324 3193

Fax number: +61-2-9324-2880

Attention: Elaine Chow / Kevin Bayett / Greg Jansz / Grant Schwulst / Jalpa Shillig

### With respect to Party B:

NPCP Investments Limited

Level 7 Deloitte Tower Douglas Street Port Moresby N.C.D. Papua New Guinea

Email address: robert.acevski@npcp.com

Telephone: +675 320 2253 Fax Number: +675 3200238 Attention: Robert Acevski

### (ii) Addresses and account details for Transfers.

### With respect of Party A:

UBS AG, Australia Branch

BSB: 946-612 Account#: 242888

Attn: Collateral Mgr Sydney Exigo Code: UBSB20 For delivery of Shares:

PID: 1505

HIN: X0000103462

### With respect to Party B:

To be advised by Party B to Party A

### (h) Other Provisions.

This Annex, including the act of transfer (disposition) itself, will be governed and construed in accordance with New South Wales law.

### **SCHEDULE 2 – ADDITIONAL PROVISIONS**

### Part I – Additional Termination Events

Each of the following events or circumstances is an Additional Termination Event:

- (a) (non-payment) Party B fails to pay any amount that is due and payable by it under an Equity Derivative Financing Document when it is due unless payment is made within 1 Local Business Day of its due date;
- (b) (other obligations) Party B fails to comply with any of its obligations under an Equity Derivative Financing Document (other than a failure referred to elsewhere in this Part I) and:
  - (i) Party A considers that the failure cannot be remedied; or
  - (ii) Party A considers that the failure can be remedied, and the failure is not remedied within 10 days after it occurs;
- (c) (misrepresentation) any representation, warranty or statement made or repeated by Party B in or in connection with an Equity Derivative Financing Document is untrue or misleading (whether by omission or otherwise), when so made or repeated, in any material respect;

### (d) (Insolvency Event)

- (i) a moratorium is declared in respect of the financial indebtedness (including granting a guarantee of financial indebtedness and including any obligation (whether present or future, actual or contingent) to pay or deliver any money or commodity under or in respect of any financial accommodation or derivative transaction) of Party B;
- (ii) Party B commences negotiations with any one or more of its foreign creditors with a view to the general readjustment or rescheduling of its financial indebtedness (including granting a guarantee of financial indebtedness and including any obligation (whether present or future, actual or contingent) to pay or deliver any money or commodity under or in respect of any financial accommodation or derivative transaction);
- (iii) an Insolvency Event occurs in respect of Party B;
- (e) (Other default) an event of default, a termination event or any other event giving rise to termination or cancellation of a transaction under another Equity Derivative Financing Document (however that event may be defined or described) occurs;
- (f) (Material Adverse Effect) an event or a change occurs (whether or not foreseeable) which has a Material Adverse Effect;

### (g) (compulsory acquisition)

- (i) all or a material part of the property of Party B is compulsorily acquired by any government agency; or
- (ii) Party B sells or divests itself of all or a material part of its property because it is required to do so by a binding order from a government agency,

and Party B does not receive compensation for the acquisition, sale or disposal which is acceptable to Party A;

(h) (inability to perform) if Party B ceases for any reason to be able lawfully to carry out all the transactions which an Equity Derivative Financing Document contemplates may be carried out by it;

- (i) (provisions void) if all or any material provision of an Equity Derivative Financing Document is or becomes void, voidable, illegal or unenforceable or of limited force (other than because of equitable principles or laws affecting creditors' rights generally), or Party B claims this to be the case; and
- (j) (exchange controls) any restriction is imposed on the ability of Party B to hold, deal with and remit foreign currencies which would make it impossible for it to perform its obligations under an Equity Derivative Financing Document.

### Part II - Undertakings

Party B must:

- (a) (comply with law) comply with all applicable laws; and
- (b) (hold Authorisations) obtain and maintain each Authorisation that is necessary or desirable to:
  - (i) execute the Equity Derivative Financing Documents to which it is a party and to carry out the transactions that those documents contemplate; or
  - (ii) ensure that the Equity Derivative Financing Documents to which it is a party are legal, valid, binding and admissible in evidence,

and must comply with any conditions to which any of these Authorisations is subject.

### **EXECUTION PAGES**

### [EXECUTION BLOCKS INTENTIONALLY OMITTED]

\*\*\*\*\*\*\*\*\*\*\*\*\*\*

### Disclaimer

This information is communicated by UBS AG and/or its affiliates ("UBS"). UBS may from time to time, as principal or agent, have positions in, or may buy or sell, or make a market in any securities, currencies, financial instruments or other assets underlying the transaction or product to which this document relates. UBS may provide investment banking and other services to and/or have officers who serve as directors of the companies referred to in this document. UBS may pay or receive brokerage or retrocession fees in connection with this transaction. UBS' trading and/or hedging activities related to this transaction or product may have an impact on the price of the underlying asset and may affect the likelihood that any relevant barrier is crossed. UBS has policies and procedures designed to minimise the risk that officers and employees are influenced by any conflicting interest or duty and that confidential information is improperly disclosed or made available.

Structured products and OTC transactions are complex and may involve a high risk of loss. Prior to entering into a transaction or a product you should consult with your own legal, regulatory, tax, financial and accounting advisors to the extent you consider it necessary, and make your own investment, hedging and trading decisions (including decisions regarding the suitability of this transaction or product) based upon your own judgement and advice from those advisers you consider necessary.

By entering into any transaction with or purchasing any product from UBS you acknowledge and agree that UBS is acting, and will at all times act, as an independent contractor on an arm's-length basis and is not acting, and will not act, in any other capacity, including as your financial adviser or fiduciary, unless expressly agreed in writing. UBS in its capacity as principal or agent is involved in a wide range of commercial banking and investment banking activities globally from which conflicting interests or duties may arise. UBS may provide services to your affiliates or any other entity or person (a "Third Party"), and may engage in any transaction (on its own account or otherwise) with respect to you or a Third Party, or act in relation to any matter for itself or any Third Party, notwithstanding that such services, transactions or actions may be adverse to you or your affiliates, and UBS may retain for its own benefit any related remuneration or profit. You should not rely on any communication (written or oral) of UBS or its employees as investment advice or as a recommendation to enter into the proposed transaction or to purchase the proposed product.

No communication (written or oral and including this document) received from UBS will be deemed to be an assurance or guarantee as to the expected results of the proposed transaction or product.

UBS makes no representation or warranty relating to any information herein which is derived from independent sources. This document shall not be copied or reproduced without UBS' prior written permission.

No action has been or will be taken in any jurisdiction that would permit a public offering of the products described herein, save where explicitly stated in the product offer document. The products or transactions described herein must be sold or offered in accordance with all applicable selling restrictions in the jurisdictions in which they are made available.

Copyright 2014 UBS AG, Australia Branch. All rights reserved. No part of this document may be reproduced or distributed in any manner without the prior written permission of UBS. UBS accepts no liability whatsoever for the actions of third parties in this respect.

Specific Security Deed (CHESS Securities - Collar)

## **NPCP Investments Limited**

and

**UBS AG, Australia Branch** 

ABN 47 088 129 613

Project Kutubu

12 March 2014, as novated / amended and restated on or about 12 December 2014

### **CONTENTS**

CLAUS	SE		PAGE	
1.	INTERPRETATION			
	1.1 1.2	Definitions  Definitions in ASX Settlement Rules and Equity Derivative Financing		
	1.3	Agreement	7	
	1.4 1.5	PPSA terms  Non Business Days		
2.	SECUI	RITY	8	
	2.1	Security clause	8	
	2.2	Priority		
	2.3	Transferred Securities		
	2.4	Documents relating to New Rights		
	2.5 2.6	Dividends and votes  Secured Party entitled to exercise rights on default		
	2.0	Acknowledgment of no subordination		
	2.7	Collateral Security		
	2.9	Nominees and custodians		
3.		NGS WITH SECURED PROPERTY		
	3.1	Restricted dealings		
4.	REPRE	ESENTATIONS AND WARRANTIES	10	
	4.1	General representations and warranties of the Grantor		
	4.2	Representations and warranties regarding Secured Property		
	4.3	Repetition of representations and warranties		
	4.4	No representations by the Secured Party	12	
5.	GRAN	TOR'S UNDERTAKINGS	12	
	5.1	Undertakings of the Borrower	12	
	5.2	General undertakings		
	5.3	Other undertakings regarding Secured Property	12	
6.	SECUI	RED PARTY'S POWERS	14	
	6.1	Exercise of rights by Secured Party	14	
	6.2	No notice required unless mandatory		
	6.3	Secured Party's right to make good a default	14	
	6.4	Powers on enforcement	15	
7.	POWE	R OF ATTORNEY	15	
	7.1	Appointment of Attorneys	15	
	7.2	General	16	
8.	ENFO	RCEMENT	16	
	8.1	Circumstances when this document may be enforced		
	8.2	Enforcement despite earlier payment		
	8.3	Exclusion of PPSA provisions	16	
9.	DEFAL	JLT INTEREST	17	
	9.1	Grantor must pay interest	17	
	9.2	Interest after judgment		
	9.3	Accrual and calculation of interest	17	
10.	APPOI	INTMENT OF RECEIVER	17	

10.1 10.2 10.3	Appointment Receiver other than as Grantor's agent Powers of Receiver	. 18
11. PROT	ECTION OF SECURED PARTY AND APPOINTEES	. 20
11.1 11.2	Protection of Secured Party and Receiver	. 20
11.3	Liability for loss	
	ECTION OF THIRD PARTIES	
12.1 12.2	Dealings under this document  Receipts	
13. APPL	ICATION OF MONEY	. 21
13.1 13.2 13.3 13.4 13.5 13.6	Order Only actual receipts credited Compensation Certificates and disputes No interest Payment into bank account Amounts contingently due	. 22 . 22 . 22 . 23 . 23
14. CONT	INUING SECURITY	. 23
14.1 14.2 14.3 14.4	Continuity Limitations on Grantor's rights No marshalling Effect of Insolvency Event	. 23 . 24
15. RECC	VERY OF GST	. 24
16. INDE	MNITIES	. 25
16.1 16.2 16.3 16.4	General indemnities Indemnity for exercise of rights or proceedings Currency indemnity Recovery from Secured Property.	. 25 . 26
17. DISC	HARGE	. 26
18. CONF	IDENTIALITY	. 27
18.1 18.2	General confidentiality	
19. NOTI	CES	. 28
19.1 19.2 19.3 19.4	How to give a notice  When a notice is given  Notices to the Secured Party  Address for notices	. 28 . 28
20. AMEN	NDMENT AND ASSIGNMENT	. 29
20.1 20.2	Amendment	
21. GENE		. 29
21.1 21.2 21.3 21.4 21.5 21.6	Governing law Liability for own expenses Giving effect to this document Authority to register and waiver of right to receive verification statements Variation of rights Operation of this document	30 30 30 30

21.7	Operation of indemnities	31
21.8	Consents	31
21.9	Statements by Secured Party	32
21.10	Set-off	32
21.11	No merger	32
21.12	Exclusion of contrary legislation	32
21.13	Counterparts	32
21.14	Execution by fewer than all parties	32

**THIS DEED** is made on 12 March 2014, as novated / amended and restated on or about 12 December 2014

### **BETWEEN:**

- (1) **NPCP Investments Limited** (the **Grantor**); and
- (2) **UBS AG, Australia Branch** ABN 47 088 129 613 of Level 16, Chifley Tower, 2 Chifley Place, Sydney NSW 2000, Australia (the **Secured Party**).

### **RECITALS**

The parties enter into this document in connection with the Equity Derivative Financing Documents.

### THE PARTIES AGREE AS FOLLOWS:

### 1. INTERPRETATION

### 1.1 **Definitions**

The following definitions apply in this document.

ASX Settlement means ASX Settlement Pty Limited (ABN 49 008 504 532).

**ASX Settlement Rules** means the ASX Settlement Operating Rules and any other operating rules, procedures, directions, decisions, requirements, customs, usages and practices of ASX Settlement, as amended from time to time.

**Attorney** means an attorney appointed under an Equity Derivative Financing Document and any attorney's substitute or delegate.

**CHESS Security** means a Marketable Security that is registered on a CHESS Subregister.

**CHESS Subregister** means, for the Issuer, that part of its share register that is maintained by ASX Settlement under CHESS.

**Collateral Security** means a Security Interest or Guarantee (other than the Security Interest granted in clause 2.1) from any person that secures or otherwise provides for payment of any Secured Money.

**Controlling Participant** means UBS Securities Australia Limited ABN 62 008 586 481 or any other person acceptable to the Secured Party who is able to sponsor Marketable Securities on a CHESS Subregister.

**Event of Default** means an "Event of Default" (as defined in the Equity Derivative Financing Agreement) in respect of which the Grantor is the Defaulting Party or a "Termination Event" (as defined in the Equity Derivative Financing Agreement) in respect of which the Grantor is the Affected Party.

### Equity Derivative Financing Agreement means each of:

(a) the confirmation letter dated 12 March 2014 between the Independent State of Papua New Guinea and the Secured Party, including any ISDA master agreement or credit support annex incorporated in or entered into under or in relation to the letter or the transactions contemplated therein, as novated from the Independent State of Papua New Guinea to the Grantor pursuant to the Global Novation and Amendment Deed (as amended, restated, novated, supplemented, modified or varied from time to time); and (b) the confirmation letter dated on or about 12 December 2014 between the Grantor and the Secured Party, including any ISDA master agreement or credit support annex incorporated in or entered into under or in relation to the letter or the transactions contemplated therein (as amended, restated, novated, supplemented, modified or varied from time to time).

**Global Novation and Amendment Deed** means the document of that title dated on or about 12 December 2014 between, among others, the Independent State of Papua New Guinea, the Secured Party and the Grantor.

**Issuer** means Oil Search Limited, a company incorporated under the laws of the Independent State of Papua New Guinea with ARBN 055 079 868.

### Marketable Security means:

- (a) an intermediated security;
- (b) a debenture, stock or bond, unit in a unit trust, or other marketable security; and
- (c) any other investment instrument.

**New Right** means a present or future right of the Grantor:

- (a) to or in any money, dividend (including any return of capital), interest, offer, bonus, note or other Marketable Security, or any entitlement to subscribe for any of them;
- (b) resulting from any substitution, conversion, redemption, forfeiture, cancellation, reclassification, consolidation or subdivision; or
- (c) resulting from a reduction of capital, liquidation or scheme of arrangement,

in connection with the Secured Property.

Nominee means UBS Nominees Pty Ltd ABN 32 001 450 522.

**Nominee Deed** means the Nominee Deed dated 12 March 2014 between the Independent State of Papua New Guinea, the Secured Party, the Nominee and the Controlling Participant, as novated from the Independent State of Papua New Guinea to the Grantor pursuant to the Global Novation and Amendment Deed (as amended, restated, novated, supplemented, modified or varied from time to time).

### Permitted Security Interest means:

- (a) a Security Interest (if any) created under an Equity Derivative Financing Document;
- (b) a lien that arises by operation of law in the ordinary course of ordinary business, where the amount secured is not overdue or is being diligently contested in good faith and appropriately provisioned;
- (c) a Security Interest:
  - (i) existing on the date of this document that has been approved by the Secured Party; or
  - (ii) that arises after the date of this document and that the Secured Party approves before it arises,

where the amount secured does not increase, and the time for payment of that amount is not extended beyond the amount and time approved by the Secured Party.

**Receiver** means a receiver or a receiver and manager.

**Secured Money** means all amounts (including damages) that are payable, owing but not payable, or that otherwise remain unpaid by the Grantor to the Secured Party and the Finance Parties on any account at any time under or in connection with the Equity Derivative Financing Agreement or any other Equity Derivative Financing Document:

- (a) whether present or future, actual or contingent;
- (b) whether incurred alone, jointly, severally or jointly and severally;
- (c) whether the Grantor is liable on its own account or for the account of, or as surety for, another person and without regard to the capacity in which the Grantor is liable;
- (d) whether due to the Secured Party alone or with another person;
- (e) whether the Secured Party is entitled for its own account or the account of another person;
- (f) whether arising from a banker and customer relationship or any other relationship;
- (g) whether originally contemplated by the Grantor or the Secured Party or not;
- (h) whether the Secured Party is the original person in whose favour the undertakings in this document, the Equity Derivative Financing Agreement or any other Equity Derivative Financing Document were given or an assignee and, if the Secured Party is an assignee:
  - (i) whether or not the Grantor consented to or knew of the assignment;
  - (ii) no matter when the assignment occurred; and
  - (iii) whether or not the entitlements of that original person were assigned with the Security Interest granted in clause 2.1; and
- (i) if determined pursuant to any award, order or judgment against the Grantor, whether or not the Grantor was a party to the court proceedings, arbitration or other dispute resolution process in which that award, order or judgment was made.

**Secured Property** means all the Grantor's present and future interest in, to, under or derived from all:

- (a) Transferred Securities; and
- (b) New Rights.

**Transfer** means, in respect of a Marketable Security or a New Right, an executed document of transfer sufficient to transfer all the legal and beneficial ownership of that Marketable Security or New Right to the Secured Party or its nominee.

**Transferred Security** means, at any time, each Marketable Security of the Issuer held by, the Nominee as trustee for the Grantor pursuant to the Nominee Deed at that time.

### 1.2 **Definitions in ASX Settlement Rules and Equity Derivative Financing Agreement**

Unless otherwise defined in this document, any capitalised term used in this document which is defined in the:

- (a) ASX Settlement Rules has the meaning given in the ASX Settlement Rules; and
- (b) an Equity Derivative Financing Agreement, including by incorporation, has the same meaning given in the Equity Derivative Financing Agreement.

### 1.3 Rules for interpreting this document

Headings are for convenience only, and do not affect interpretation. The following rules also apply in interpreting this document, except where the context makes it clear that a rule is not intended to apply.

- (a) A reference to:
  - (i) a legislative provision or legislation (including subordinate legislation) is to that provision or legislation as amended, re-enacted or replaced, and includes any subordinate legislation issued under it;
  - (ii) a document (including this document) or agreement, or a provision of a document (including this document) or agreement, is to that document, agreement or provision as amended, supplemented, replaced or novated;
  - (iii) a party to this document or to any other document or agreement includes a successor in title, permitted substitute or a permitted assign of that party;
  - (iv) a person includes any type of entity or body of persons, whether or not it is incorporated or has a separate legal identity, and any executor, administrator or successor in law of the person; and
  - (v) anything (including a right, obligation or concept) includes each part of it and any part of it.
- (b) A singular word includes the plural, and vice versa.
- (c) A word which suggests one gender includes the other genders.
- (d) If a word or phrase is defined, any other grammatical form of that word or phrase has a corresponding meaning.
- (e) If an example is given of anything (including a right, obligation or concept), such as by saying it includes something else, the example does not limit the scope of that thing.
- (f) The word **agreement** includes an undertaking or other binding arrangement or understanding, whether or not in writing.
- (g) The expression **this document** includes the agreement, arrangement, understanding or transaction recorded in this document.
- (h) An Event of Default "continues" until the Secured Party notifies the Grantor that it has been:
  - (i) remedied to the satisfaction of the Secured Party; or
  - (ii) waived by the Secured Party.

- (i) Unless otherwise defined in this document, words that are defined in an applicable Accounting Standard have the same meaning in this document.
- (j) Unless otherwise defined in this document, words defined in the GST Law have the same meaning in clauses concerning GST.
- (k) If a person is a member of a GST group, references to GST for which the person is liable and to input tax credits to which the person is entitled include GST which the representative member of the GST group is liable and input tax credits to which the representative member is entitled.

### 1.4 **PPSA terms**

In this document, unless the context requires otherwise, the following terms have the meanings given to them in the PPSA:

- (a) amendment demand;
- (b) attach;
- (c) control;
- (d) financing change statement;
- (e) financing statement;
- (f) intermediated security;
- (g) investment instrument;
- (h) registration; and
- (i) verification statement.

### 1.5 **Non Business Days**

If the day on or by which a person must do something under this document is not a Business Day:

- (a) if the act involves a payment that is due on demand, the person must do it on or by the next Business Day; and
- (b) in any other case, the person must do it on or by the previous Business Day.

### 2. **SECURITY**

### 2.1 **Security clause**

The Grantor grants a security interest in the Secured Property to the Secured Party to secure the payment of the Secured Money. This security interest is a fixed charge.

### 2.2 **Priority**

The Security Interest granted in clause 2.1 is intended to take effect as a first ranking security subject only to those Permitted Security Interests which the Secured Party agrees in writing rank in priority to it.

### 2.3 Transferred Securities

The Grantor must:

- (a) at the time it executes this document, enter into the Nominee Deed with the Secured Party, the Nominee and the Controlling Participant in respect of the Transferred Securities; and
- (b) at each time that any transfer of Marketable Securities of the Issuer from the Secured Party to the Nominee occurs under the Nominee Deed:
  - (i) give the Secured Party, the Nominee and the Controlling Participant the applicable holder or other identification numbers and information sufficient to identify the Transferred Securities (including a copy of the relevant holding statement);
  - (ii) do everything within its power that is necessary to ensure that the relevant Transferred Securities are recorded on the Issuer's CHESS Subregister; and
  - (iii) do everything within its power to transfer sponsorship of the relevant Transferred Securities to the Controlling Participant.

This clause does not apply to the extent that the Secured Party agrees otherwise under clause 2.9.

### 2.4 **Documents relating to New Rights**

The Grantor must deposit with the Secured Party or the Secured Party's nominee:

- (a) all certificates, transfers and other documents or agreements evidencing title to New Rights; and
- (b) Transfers (in form and number satisfactory to the Secured Party) with the name of the transferee, the consideration and the date left blank, in respect of those new Rights,

immediately it acquires those New Rights.

### 2.5 **Dividends and votes**

The Grantor may exercise rights (including voting rights) in connection with the Secured Property to the extent contemplated by the Nominee Deed.

### 2.6 Secured Party entitled to exercise rights on default

If an Event of Default occurs and is continuing:

- (a) the Grantor must, if the Secured Party gives notice to the Grantor requiring it to do so, do everything necessary to ensure that the Secured Property is registered in the name of the Secured Party or its nominee in accordance with any directions contained in that notice;
- (b) the Grantor's rights under clause 3.2 (*Voting*) of the Nominee Deed immediately cease;
- (c) the Secured Party may exercise or refrain from exercising any rights in connection with the Secured Property; and

(d) the Secured Party may receive all New Rights and apply them (or their sale proceeds) in accordance with clause 13.1.

### 2.7 Acknowledgment of no subordination

The Grantor acknowledges that the Secured Party has not agreed to subordinate its Security Interest in the Secured Property to any other interest in the Secured Property, except to the extent (if any) expressly provided by an Equity Derivative Financing Document.

### 2.8 Collateral Security

The Security Interest granted in clause 2.1 is collateral to and secures the same money as is secured by Collateral Securities.

### 2.9 Nominees and custodians

If the Secured Party agrees that any of the Secured Property can be held by a nominee or custodian on behalf of the Grantor other than the Nominee, the Grantor undertakes to procure that the nominee or custodian:

- (a) acknowledges and consents to the Security Interest created by clause 2.1;
- (b) agrees to hold the Secured Property on such terms as the Secured Party requires, including by agreeing not to:
  - (i) create or allow another interest in any Secured Property other than a Permitted Security Interest; or
  - (ii) dispose, or part with possession, of any Secured Property,

without the consent of the Secured Party; and

(c) agrees to do anything, including executing documents, that the Secured Party may require to enable it to perfect its Security Interest by control.

### 3. **DEALINGS WITH SECURED PROPERTY**

### 3.1 Restricted dealings

The Grantor must not do, or agree to do, any of the following unless it is permitted to do so by another provision in an Equity Derivative Financing Document:

- (a) create or allow another interest in any Secured Property other than a Permitted Security Interest; or
- (b) dispose, or part with possession, of any Secured Property.

### 4. REPRESENTATIONS AND WARRANTIES

### 4.1 General representations and warranties of the Grantor

The provisions of paragraph 1 (*Disclosure requirement and additional representations and warranties*) of the section entitled "Additional Provisions" of the Equity Derivative Financing Agreement apply to this document as if set out in full in this document, except that references to "Party B" are to the "Grantor", references to "Party A" are to the "Secured Party" and with any other necessary changes.

### 4.2 Representations and warranties regarding Secured Property

The Grantor represents and warrants to the Secured Party that:

### (a) (Secured Property)

- (i) it is the beneficial owner of the Secured Property (and the Secured Property is held by the Nominee in accordance with the Nominee Deed), and
- (ii) it has full power to grant a Security Interest in the Secured Property in the manner provided in this document; and
- (iii) the Security Interest granted in clause 2.1 is an effective security in the Secured Property, except to the extent (if any) set out in any Equity Derivative Financing Document or that the Secured Party may otherwise agree;
- (b) **(no other interest)** no other person has any interest in or other right over the Secured Property except:
  - (i) the Nominee, pursuant to the terms of the Nominee Deed;
  - (ii) to the extent (if any) set out in any other Equity Derivative Financing Document:
  - (iii) Permitted Security Interests; or
  - (iv) as otherwise agreed by the Secured Party;
- (c) (**location of Secured Property**) at the date of execution of this document, none of the Secured Property is located, or taken for the purposes of New South Wales stamp duty law to be located, in New South Wales;
- (d) (no escrow or other conditions) the Secured Property is not subject to any escrow or other conditions imposed by the Corporations Act or under the rules of any stock exchange;
- (e) (all calls satisfied) all calls made in respect of the Secured Property have been satisfied;
- (f) (fully paid Marketable Securities) the Marketable Securities in the Secured Property are all fully paid; and
- (g) **(no money owing to Issuer)** no money is owing by the Grantor to the Issuer in connection with the Secured Property.

### 4.3 Repetition of representations and warranties

The representations and warranties in:

- (a) clause 4.1 (*General representation and warranties of the Grantor*) are taken to be repeated on each date that the corresponding representation or warranty in the Equity Derivative Financing Agreement is repeated under the terms of the Equity Derivative Financing Agreement, on the basis of the facts and circumstances as at that date: and
- (b) clause 4.2 (*Representations and warranties regarding Secured Property*) are taken to be repeated on each date from the date of this document until the date on which

all amounts outstanding under the Equity Derivative Financing Documents have been paid in full, on the basis of the facts and circumstances as at that date.

### 4.4 No representations by the Secured Party

The Grantor acknowledges that it has not relied and will not rely on any financial or other advice, representation, statement or promise provided or made by or on behalf of the Secured Party in deciding to enter into this document or to exercise any right or perform any obligation under it.

### 5. GRANTOR'S UNDERTAKINGS

### 5.1 **Undertakings of the Borrower**

Part II (*Undertakings*) of Schedule 2 (*Additional Provisions*) of the Equity Derivative Financing Agreement is incorporated into this document as if set out in full, except that references to the "Party B" are to the "Grantor", references to "this document" are to the "Equity Derivative Financing Documents", and with any other necessary changes.

### 5.2 **General undertakings**

The Grantor must:

- (a) (**obligation to pay**) punctually pay the Secured Money when it becomes payable in accordance with the Equity Derivative Financing Documents or, in the absence of any agreement or after default under any agreement, on demand by the Secured Party:
- (b) (registration and stamping) at its own cost ensure that:
  - (i) this document is immediately registered with any Government Agency specified by the Secured Party if the Secured Party determines that registration is necessary to perfect the Security Interest granted in clause 2.1 or to protect the rights or priority of the Secured Party; and
  - (ii) this document is stamped for the proper amount within the period provided by law in each state and territory of Australia in which this document is required to be stamped; and
- (c) (ensure no Event of Default) ensure that no Event of Default in respect of which the Grantor is the Defaulting Party occurs.

### 5.3 Other undertakings regarding Secured Property

The Grantor must:

### (a) (outgoings)

- (i) punctually pay all outgoings (including Taxes) in respect of the Secured Property payable by it, except to the extent that:
  - (A) these are being diligently contested in good faith and by appropriate proceedings;
  - (B) it has made adequate reserves for them; and
  - (C) failure to pay them will not have a Material Adverse Effect on it or prejudice the Secured Property;

- (ii) pay the contested amount after the final determination or settlement of the relevant contest; and
- (iii) on request by the Secured Party, immediately hand to the Secured Party evidence of every payment covered by this undertaking or required under this document:
- (b) **(preserve and protect security)** promptly do everything necessary or reasonably required by the Secured Party to:
  - (i) preserve and protect the value of the Secured Property; or
  - (ii) protect and enforce its title and rights and the Secured Party's title as secured party to the Secured Property;
- (c) (issuer information) give to the Secured Party as soon as it receives them copies of:
  - (i) each document that the Issuer gives to the Grantor in connection with the Secured Property; and
  - (ii) such other information in its possession relating to the financial condition, business, property and affairs of the Issuer as the Secured Party reasonably requests;
- (d) (comply with constitution) do everything the Grantor is required to do (including the payment of calls or instalments) by the constitution of the Issuer;
- (e) (perfection by control) to the extent that any Secured Property is of a type in which a security interest can be perfected by control under the PPSA, do anything that the Secured Party may reasonably require to enable it to perfect its Security Interest by control;
- (f) (**Nominee Deed**) without limiting clause 5.3(e), ensure that:
  - (i) the Nominee Deed is in force at all times in relation to all CHESS Securities in the Secured Property;
  - (ii) it complies with all its obligations under, and does not do or omit to do anything which would cause any other party to the Nominee Deed to breach or to be unable to comply with its obligations under, the Nominee Deed;
  - (iii) it does nothing, and it does not instruct or otherwise cause the Nominee to do anything, without the consent of the Secured Party to remove any CHESS Securities from the CHESS Subregister or the control of the Controlling Participant (even if the Controlling Participant is suspended from CHESS participation); and
  - (iv) it does everything required by the Secured Party, if sponsorship of any CHESS Securities in the Secured Property is under the terms of the ASX Settlement Rules to be transferred to another Controlling Participant, to transfer that sponsorship to the new Controlling Participant;
- (g) (New Rights) give the Secured Party particulars of all New Rights and documentary or other evidence of New Rights immediately after becoming aware of the New Rights;
- (h) **(take up)** at the Secured Party's request take up New Rights if, in the Secured Party's opinion, failure to do so could mean the Secured Property or the Security

Interest granted in clause 2.1 may become materially lessened in value or prejudicially affected; and

### (i) (**location of registers**) ensure that:

- (i) as at the date on which a Security Interest is granted over a Marketable Security in accordance with clause 2.1, it is recorded on the CHESS Subregister for the issuer of the Marketable Security or another register approved by the Secured Party in writing; and
- (ii) ensure that a CHESS Security is not removed from that register without the Secured Party's consent.

### 6. **SECURED PARTY'S POWERS**

### 6.1 **Exercise of rights by Secured Party**

If the Secured Party exercises a power, right, discretion or remedy in connection with this document, that exercise is taken not to be an exercise of a power, right, discretion or remedy under the PPSA unless the Secured Party states otherwise at the time of exercise. However, this clause does not apply to a power, right, discretion or remedy which can only be exercised under the PPSA.

### 6.2 No notice required unless mandatory

To the extent the law permits, the Grantor waives:

- (a) its rights to receive any notice that is required by:
  - (i) any provision of the PPSA (including a notice of a verification statement); or
  - (ii) any other law before a secured party or Receiver exercises a power, right, discretion or remedy; and
- (b) any time period that must otherwise lapse under any law before a secured party or Receiver exercises a power, right, discretion or remedy.

If the law which requires a period of notice or a lapse of time cannot be excluded, but the law provides that the period of notice or lapse of time may be agreed, that period or lapse is one day or the minimum period the law allows to be agreed (whichever is the longer).

However, nothing in this clause prohibits the Secured Party or any Receiver from giving a notice under the PPSA or any other law.

### 6.3 Secured Party's right to make good a default

- (a) If the Grantor breaches this document, the Secured Party may do everything it considers to be necessary or desirable to attempt to remedy the breach to the Secured Party's satisfaction. The Secured Party is not obliged to do so. Any liabilities or expenses incurred by the Secured Party in attempting to remedy any such breach must be reimbursed by the Grantor on demand and, pending reimbursement, will be part of the Secured Money.
- (b) Clause 6.3(a) does not limit any other right the Secured Party has under this document or at law.

### 6.4 Powers on enforcement

If this document has become enforceable, the Secured Party or any of its Authorised Representatives, without notice to the Grantor, may:

- (a) exercise any of the powers that might be exercised by a Receiver even if a Receiver has not been appointed; and
- (b) complete any Transfer, other transfer or instrument of any nature executed by or on behalf of the Grantor in blank and deposited with the Secured Party as Collateral Security in favour of, or take any other action required to transfer any Secured Property to, the Secured Party or any appointee of the Secured Party or any other person.

# 7. **POWER OF ATTORNEY**

### 7.1 **Appointment of Attorneys**

The Grantor irrevocably appoints the Secured Party and each Authorised Representative of the Secured Party and as an independent appointment appoints any Receiver, severally its attorney, at the Grantor's cost, to:

- (a) (all acts necessary) do anything necessary or desirable in the opinion of the Secured Party or the Attorney to:
  - (i) complete this document;
  - (ii) give full effect to this document;
  - (iii) better secure the Secured Property to the Secured Party in a manner consistent with this document; or
  - (iv) assist in the execution or exercise of any power under this document,

including give directions to the Nominee or any Controlling Participant in respect of CHESS Securities in the Secured Property, and execute any Transfer (including a Transfer in blank) or other document;

- (b) do all or any of the following while an Event of Default in respect of which the Grantor is the Defaulting Party is continuing:
  - (i) (recover Secured Property) demand, sue for, recover and give discharge for the Secured Property;
  - (ii) (commence actions) commence, carry on , enforce, settle, arrange and compromise any proceedings to obtain or enforce the payment or delivery of the Secured Property;
  - (iii) (compound debts) compound, settle or compromise any debt of the Grantor in connection with the Secured Property;
  - (iv) (execute deeds) execute any agreement including any deed of assignment, composition or release in connection with the Secured Property;
  - (v) (exercise Marketable Security owner's rights) exercise all or any powers, rights, discretions and remedies available to any owner or holder (whether beneficial or otherwise) of any Marketable Securities included in the Secured Property (including any right to give directions to the Nominee or any Controlling Participant or to redeem any Marketable Security and

rights available to an owner of Marketable Securities under the Corporations Act or any other statute, or under the rules of any stock exchange); and

(vi) (**general**) do anything else that the Grantor must or may do, or that the Secured Party may do, under this document or by law.

# 7.2 **General**

- (a) Each Attorney may appoint and remove substitutes, and may delegate its powers (including this power of delegation) and revoke any delegation.
- (b) An Attorney may do anything contemplated by this clause even if the Attorney is affected by an actual or potential conflict of interest or duty, or might benefit from doing it.
- (c) An Attorney may do anything contemplated by this clause in its name, in the name of the Grantor or in the name of both of them.
- (d) The Grantor must ratify anything done by an Attorney under this clause.
- (e) The Grantor gives the power of attorney in this clause:
  - (i) to secure performance by the Grantor of its obligations to the Secured Party under this document and any property interest of the Secured Party under this document; and
  - (ii) for valuable consideration, receipt of which is acknowledged by the Grantor.

### 8. **ENFORCEMENT**

# 8.1 Circumstances when this document may be enforced

The Secured Money will immediately become payable at the Secured Party's option (despite any delay or previous waiver of the right to exercise that option) without the need for any demand or notice under this document or under another Equity Derivative Financing Document, and this document will immediately become enforceable (whether or not the Secured Money has become payable in this manner) if an Event of Default occurs.

# 8.2 Enforcement despite earlier payment

This document may be enforced:

- (a) even if the Secured Party accepts a payment of interest or other amount after the occurrence of any Event of Default; and
- (b) without the need for any notice to, or of any consent or agreement of, the Grantor or any other person.

# 8.3 Exclusion of PPSA provisions

To the extent the law permits:

- (a) for the purposes of sections 115(1) and 115(7) of the PPSA:
  - (i) the Secured Party need not comply with sections 95, 118, 121(4), 125, 130, 132(3)(d) or 132(4); and
  - (ii) sections 142 and 143 are excluded:

- (b) for the purposes of section 115(7) of the PPSA, the Secured Party need not comply with sections 132 and 137(3); and
- (c) if the PPSA is amended after the date of this document to permit the Grantor and the Secured Party to agree to not comply with or to exclude other provisions of the PPSA, the Secured Party may notify the Grantor that any of these provisions is excluded, or that the Secured Party need not comply with any of these provisions, as notified to the Grantor by the Secured Party.

# 9. **DEFAULT INTEREST**

# 9.1 **Grantor must pay interest**

- (a) The Grantor must pay interest on each amount that is not paid when due (unless the Grantor is already required to pay interest on the unpaid amount by the terms of any other Equity Derivative Financing Document), from (and including) the day on which it falls due to (but excluding) the day on which it is paid in full, at the rate calculated in accordance with clause 9.1(b). This interest must be paid on demand.
- (b) Interest on an unpaid amount accrues each day at a rate equal to the sum of the Secured Party's cost of funding that amount over that day (as the Secured Party conclusively certifies to the Grantor) plus 2.00% per annum, and is capitalised (if not paid) every 30 days.
- (c) This subclause does not affect the Grantor's obligation to pay each amount under this document when it is due.

# 9.2 Interest after judgment

If a liability of the Grantor becomes merged in a judgment or order, the Grantor, as an independent obligation, must pay interest on the amount of that liability, from (and including) the date of the judgment or order until it is paid in full, at the higher of the rate that applies under the judgment or order and the rate calculated in accordance with clause 9.1.

# 9.3 **Accrual and calculation of interest**

Interest under this clause:

- (a) accrues daily; and
- (b) is calculated on the basis of the actual number of days on which interest has accrued and of a 365 day year.

# 10. **APPOINTMENT OF RECEIVER**

# 10.1 **Appointment**

If this document has become enforceable (whether or not the Secured Party has entered into possession of all or any of the Secured Property) the Secured Party or any Authorised Representative of the Secured Party may at any time:

- (a) appoint any person or any two or more persons jointly and severally to be a receiver or receiver and manager (or an additional receiver or receiver and manager) of Secured Property;
- (b) remove the Receiver and in case of the removal, retirement or death of any Receiver appoint another as a replacement; and

(c) fix the remuneration of the Receiver.

Subject to clause 10.2, every Receiver appointed under this subclause will be the Grantor's agent and the Grantor alone will be responsible for the Receiver's acts and defaults and remuneration.

# 10.2 Receiver other than as Grantor's agent

- (a) The Secured Party by notice to the Grantor and the Receiver may require the Receiver to act as the Secured Party's agent.
- (b) The power to appoint a Receiver under this clause may be exercised even though:
  - (i) at the time when this document becomes enforceable or when an appointment is made, an order may have been made or a resolution may have been passed to wind up the Grantor; or
  - (ii) a Receiver appointed in the circumstances specified in the preceding paragraph may not, or may not in some respects, act as the Grantor's agent.

### 10.3 **Powers of Receiver**

The Receiver will have full power to do all or any of the following:

- (a) (possession and control) take possession of, or take control of, collect and get any Transfers or other documents held or entitled to be held by the Secured Party under the Security Interest granted in clause 2.1 and the Secured Property and for that purpose to take proceedings (in the name of the Grantor or otherwise);
- (b) (**give up possession**) give up possession of any Transfers or other documents held or entitled to be held by the Secured Party under the Security Interest granted in clause 2.1 and the Secured Property;
- (c) (exercise Secured Party's rights)
  - (i) exercise all or any of the Secured Party's powers, rights, discretions and remedies under this document; and
  - (ii) comply with the directions given by the Secured Party;
- (d) (exercise Marketable Security owner's rights) exercise all or any powers, rights, discretions and remedies available to any owner or holder (whether beneficial or otherwise) of any Marketable Securities included in the Secured Property (including any right to give directions to the Nominee or any Controlling Participant or to redeem any Marketable Security and rights available to an owner of Marketable Securities under the Corporations Act or any other statute, or under the rules of any stock exchange);
- (e) **(registration**) do everything necessary to obtain registration of the Secured Property in the name of the Secured Party or its nominee and to enable the Secured Party or its nominee to receive New Rights;
- (f) (New Rights) receive all New Rights;

# (g) (settle disputes)

- (i) settle, arrange and compromise any accounts, claims, questions or disputes that may arise in connection with the Secured Property or in any way relating to this document; and
- (ii) execute releases or other discharges in relation to the settlement, arrangement, or compromise;
- (h) (sell) sell (whether or not the Receiver has taken possession), exchange or otherwise dispose of (absolutely or conditionally) the Secured Property (or agree to do so):
  - (i) with or without any other property;
  - (ii) by public auction, private sale or tender for cash or on credit;
  - (iii) whether or not the reserve price for a sale by auction or tender is disclosed;
  - (iv) in one lot or in parcels;
  - (v) with or without special conditions, (such as conditions as to title or time or method of payment of purchase money) including by allowing the purchase money to remain:
    - (A) outstanding on any security over the property sold or over any other property; or
    - (B) owing without any security; and
  - (vi) on other terms the Receiver considers desirable,

without being responsible for any loss;

- (i) **(transfer on sale)** execute Transfers, other transfers and assignments of Secured Property (including in the name of the Grantor), and do everything to complete any sale under clause 10.3(h) that the Receiver thinks necessary;
- (j) **(employees and agents)** engage employees, agents, consultants, lawyers, advisers and contractors for any of the purposes of this clause on terms that the Receiver thinks appropriate;
- (k) (**give receipts**) give receipts for all money and other property that may come into the hands of the Receiver in exercise of any power given by this document;
- (I) (enforce contracts) carry out and enforce or otherwise obtain the benefit of all contracts:
  - (i) entered into or held by the Grantor in connection with the Secured Property; or
  - (ii) entered into in exercise of the powers given by this document;
- (m) (make debtors bankrupt) wind up the issuer and do everything in connection with any winding up that the Receiver thinks desirable to recover or protect Secured Property;
- (n) (**perform undertakings**) do everything necessary to perform any undertaking of the Grantor in this document;

(o) (**receive money**) receive all money or other property payable or deliverable to the Grantor from the Secured Property;

### (p) (desirable or incidental matters)

- (i) do or cause to be done everything that the Receiver thinks desirable in the interests of the Secured Party; and
- (ii) do anything incidental to the exercise of any other power;
- (q) (take legal proceedings) take proceedings (including in the name of the Grantor) in connection with any of the above; and
- (r) (**delegate**) with the Secured Party's consent delegate any of the powers given to the Receiver by this clause to any person.

### 11. PROTECTION OF SECURED PARTY AND APPOINTEES

# 11.1 Protection of Secured Party and Receiver

- (a) The Secured Party is not obliged to, but may, do the following:
  - (i) notify any debtor or member of the Grantor or any other person of this document; or
  - (ii) enforce payment of any money payable to the Grantor, or take any step or proceeding for any similar purpose.
- (b) None of the Secured Party, any of its Authorised Representatives or agents, any Attorney or any Receiver is liable for any omission or delay in exercising any power, right, discretion or remedy under this document or for any involuntary loss or irregularity that may occur in relation to the exercise or non-exercise of any of them except to the extent that it is a direct and foreseeable result of its own fraud, gross negligence or wilful misconduct.

# 11.2 **Conflict of interests**

The Secured Party, an Authorised Representative or agent of the Secured Party, an Attorney, Receiver or other person appointed by the Secured Party under this document may exercise or agree to exercise a power given by this document or by law even though that person may have a conflict of interests in exercising the power.

# 11.3 Liability for loss

- (a) None of the Secured Party, an Authorised Representative or agent of the Secured Party, an Attorney, a Receiver or any other person appointed by the Secured Party under this document is liable for any loss that the Grantor suffers as a direct or indirect result of:
  - (i) the exercise or attempted exercise of, or failure to exercise, any of its rights contained in this document; and
  - (ii) any release or dealing with any other Guarantee or Security Interest (including any prejudice to or loss of the Grantor's rights of subrogation),

except to the extent that such loss is a direct and foreseeable result of its own fraud, gross negligence or wilful misconduct.

- (b) If the Secured Party, any agent of the Secured Party or a Receiver enters into possession of Secured Property, none of the Secured Party, any of its Authorised Representatives or agents, any Attorney or any Receiver is liable:
  - (i) to account as secured party in possession or for anything except actual receipts; or
  - (ii) for any loss on realisation or for any default or omission for which a secured party in possession might be liable, except to the extent that it is a direct and foreseeable result of its own fraud, gross negligence or wilful misconduct.

# 12. **PROTECTION OF THIRD PARTIES**

### 12.1 **Dealings under this document**

A purchaser or other party to a disposal or dealing in attempted exercise of a power contained in this document is not:

- (a) bound to enquire whether an Event of Default has occurred, whether this document has become enforceable, whether a Receiver has been properly appointed or about the propriety or regularity of a sale, disposal or dealing; or
- (b) affected by notice that a sale, disposal or dealing is unnecessary or improper.

Despite any irregularity or impropriety in a sale, disposal or dealing, it is to be treated, for the protection of the purchaser or other party to the disposal or dealing, as being authorised by this document and valid.

# 12.2 **Receipts**

A receipt that the Secured Party, one of its Authorised Representatives or agents or a Receiver gives for any money payable to or receivable by the Secured Party or the Receiver because of this document will:

- (a) relieve the person paying or handing over money or other property from all liability:
  - (i) for the application (or any loss or misapplication) of the money or other property;
  - (ii) to enquire whether the Secured Money has become payable; and
  - (iii) (where appropriate) as to the propriety or regularity of the appointment of the Receiver; and
- (b) discharge the person paying that money from its liability to pay that money.

### 13. **APPLICATION OF MONEY**

## 13.1 **Order**

(a) Money that the Secured Party or a Receiver receives under or because of this document is to be applied, after satisfaction of any claims that the Secured Party or the Receiver is aware is a claim that ranks in priority to the Security Interest granted in clause 2.1, in the following order or in any other order that the Secured Party or the Receiver may determine in its absolute discretion, subject to any Equity Derivative Financing Documents or applicable law to the contrary (for example, section 140 of the PPSA where relevant):

- (i) (expenses) first in payment of all expenses that the Secured Party or a Receiver incurs in or incidental to the exercise or attempted exercise of a power or otherwise in relation to any Equity Derivative Financing Document;
- (ii) (outgoings) then in payment of any other outgoings that the Receiver or the Secured Party thinks it appropriate to pay;
- (iii) (**Receiver**) then in payment to the Receiver of any remuneration (whether by way of commission or otherwise);
- (iv) (indemnities) then in payment to the Secured Party or a Receiver of any amount necessary to give effect to any indemnity contained in this document; and
- (v) (**Secured Money**) then in payment to the Secured Party of the Secured Money.
- (b) Any surplus will belong to the Grantor or other persons entitled to it. The Secured Party or the Receiver may pay the surplus to the credit of a bank account in the name of the Grantor or other person entitled to it or into court and will then be under no further liability in relation to it. The surplus will not accrue interest.
- (c) When the Secured Party or a Receiver receives money under or because of this document, and applies it in payment to the Secured Party of the Secured Money, the Secured Party or the Receiver (as the case may be) may apply different parts of the money received to different parts of the Secured Money, regardless of any appropriation by the Grantor, as the Secured Party or Receiver chooses.

# 13.2 Only actual receipts credited

In applying any money towards the Secured Money, the Grantor's account will be credited only with the amount of the money that the Secured Party actually receives for that purpose. The credit will date from the time of receipt.

### 13.3 **Compensation**

If any compensation becomes payable for the Secured Property, the Secured Party may:

- (a) apply the sum received on account of any compensation, at the Secured Party's option, in or towards repayment of the Secured Money;
- (b) make, enforce, settle or compromise any claims relating to compensation; and
- (c) execute any necessary assurances and releases in the names of the Grantor and the Secured Party.

If any compensation comes into the hands of the Grantor before a final irrevocable discharge of this document, the Grantor must immediately pay it to the Secured Party.

# 13.4 **Certificates and disputes**

- (a) The Secured Party may rely on a certificate issued by any person who claims to be entitled to any money received from the exercise of any right in relation to the Secured Property which states that the Grantor owes it a certain amount of money, without making any further enquiry.
- (b) If there is any dispute between any persons (other than the Secured Party) regarding an entitlement to receive any money received from the exercise of any right in relation to the Secured Property, the Secured Party may pay that money

into court, and after doing so does not have any further obligation in respect of that money.

### 13.5 **No interest**

The Secured Party is not obliged to pay interest to any person on any money received from the exercise of any right in relation to the Secured Property.

# 13.6 Payment into bank account

The Secured Party or the Receiver may pay any money to the credit of a bank account in the name of a person to whom it is obliged to pay any money received from the exercise of any right in relation to the Secured Property, and having done so is under no further liability in respect of that money.

# 13.7 Amounts contingently due

- (a) If any part of the Secured Money is contingently owing to the Secured Party when money is being applied under clause 13.1 the Secured Party or Receiver may:
  - retain an amount equal to the amount contingently owing, or any part of it; and
  - (ii) put that amount in an interest-bearing account, payable at call.
- (b) If the amount which is contingently owing:
  - (i) becomes payable; or
  - (ii) ceases to be contingently owing,

the Secured Party or Receiver must apply the amount retained (and any interest earned on it) in accordance with clause 13.1.

# 14. **CONTINUING SECURITY**

### 14.1 **Continuity**

The Security Interest granted in clause 2.1:

- (a) is a continuing security, and remains in full force until a final irrevocable discharge of that Security Interest is given to the Grantor under clause 17 despite any transaction or other thing (including a settlement of account or intervening payment); and
- (b) will apply to the present and future balance of the Secured Money.

# 14.2 Limitations on Grantor's rights

Until the Secured Money has been irrevocably paid and discharged in full, the Grantor may not:

- (a) share in any Guarantee, Security Interest or money received or receivable by the Secured Party in relation to the Secured Money or stand in the place of the Secured Party in relation to any Guarantee, Security Interest or right to receive money;
- (b) in reduction of its liability under this document, raise a defence, set off or counterclaim against the Secured Party or claim a set off or make a counterclaim against the Secured Party; or

(c) claim to be entitled by way of contribution, indemnity, subrogation, marshalling or otherwise to the benefit of any document or agreement to which the Secured Party is a party.

# 14.3 No marshalling

The Secured Party is not under any obligation to marshal or appropriate in favour of the Grantor or to exercise, apply, perfect or recover any Security Interest that the Secured Party holds at any time or any funds or property that the Secured Party may be entitled to receive or have a claim on.

# 14.4 **Effect of Insolvency Event**

- (a) If an Insolvency Event has occurred in relation to an Obligor, any amount paid by that Obligor (**relevant payment**) will only be applied against any Secured Money if:
  - (i) the Secured Party forms the opinion in good faith (which will be conclusively binding on the Grantor) that it will not be required to pay the relevant payment to any person under any law relating to bankruptcy, winding up or the protection of creditors; or
  - (ii) a final judgment is given by a court of competent jurisdiction in favour of the Secured Party that it is not required to pay the relevant payment to any person under any law relating to bankruptcy, winding up or the protection of creditors.
- (b) If an amount is applied against any Secured Money and the Secured Party pays or determines that it is obliged to pay the relevant amount to any person under any law relating to bankruptcy, winding up or the protection of creditors:
  - (i) the Secured Party's rights are to be reinstated and will be the same in relation to that amount as if the application, or the payment or transaction giving rise to it, had not been made; and
  - (ii) the Grantor must immediately do anything (including the signing of documents) required by the Secured Party to restore to the Secured Party any Guarantee or Security Interest to which it was entitled immediately before that application or the payment or transaction giving rise to it.
- (c) Any discharge or release between the Secured Party and the Grantor is subject to reinstatement of the Secured Party's rights under this subclause.

### 15. **RECOVERY OF GST**

- (a) Unless otherwise indicated all amounts referred to in this document are exclusive of GST.
- (b) If the Secured Party makes a taxable supply under or in connection with this document for consideration that is exclusive of GST, the Grantor must:
  - (i) pay to the Secured Party an amount equal to any GST for which the Secured Party is liable in relation to that supply; and
  - (ii) make that payment as and when the consideration or part of it must be paid or provided.
- (c) If requested by the Grantor, the Secured Party must issue a tax invoice for a taxable supply to the person to whom it made the supply.

(d) The Grantor's obligation to reimburse the Secured Party for an amount paid or payable to a third party (including an obligation to pay the Secured Party's or another party's legal costs) includes GST on the amount paid or payable to the third party except to the extent that the Secured Party is entitled to an input tax credit for that GST. Unless notified otherwise by the Secured Party, the Grantor must assume that the Secured Party is not entitled to any input tax credit for that GST.

### 16. **INDEMNITIES**

# 16.1 **General indemnities**

The Grantor must indemnify the Secured Party against, and must pay the Secured Party on demand the amount of, all losses (including loss of profit), liabilities, costs, expenses and Taxes (other than Excluded Taxes) incurred in connection with each of the following:

- (a) any Secured Money being paid or becoming due for payment other than on its due date, or any amount required to be paid under any Equity Derivative Financing Document not being paid on its due date, including losses (including loss of profit), liabilities, costs, expenses and Taxes (other than Excluded Taxes) incurred because of:
  - (i) the cancellation, termination, unwinding or alteration of any swap or other arrangement made by the Secured Party to fund the Secured Money or other amount; or
  - (ii) any liquidation or re-employment of deposits or other funds acquired by the Secured Party to fund the Secured Money or other amount;
- (b) the preparation, negotiation, execution, stamping and registration of the Equity Derivative Financing Documents and the satisfaction of any conditions precedent;
- (c) the transactions that the Equity Derivative Financing Documents contemplate;
- (d) any amendment to, or any consent, approval, waiver, release or discharge of or under, an Equity Derivative Financing Document;
- (e) any Event of Default or Potential Event of Default;
- (f) the administration, and any actual or attempted preservation or enforcement, of any rights under the Equity Derivative Financing Documents;
- (g) an enquiry by a Government Agency involving the Grantor; and
- (h) any action taken by the Secured Party under or in relation to the PPSA, including any registration, or any response to an amendment demand or a request under section 275 of the PPSA,

including legal expenses on a full indemnity basis (provided that the Borrower shall only be liable for reasonable legal expenses incurred in connection with paragraph (b)), the Secured Party's internal administration and legal costs at the rate and on the basis determined by the Secured Party and expenses incurred by the Secured Party in engaging investigative accountants and financial consultants (but only where such engagement is made after the occurrence of an Event of Default or a Potential Event of Default).

### 16.2 Indemnity for exercise of rights or proceedings

To the extent permitted by law, the Grantor must indemnify each of the Secured Party, each Authorised Representative and agent of the Secured Party, each Receiver and

Attorney of the Grantor and any other person appointed under this document or the Corporations Act by or on behalf of the Secured Party as secured party under this document against, and must pay each of them on demand the amount of all losses, liabilities, costs, expenses and Taxes (other than Excluded Taxes) that they each incur:

- (a) (directly or indirectly) in the exercise or attempted exercise of any of the powers, rights, discretions or remedies (express or implied) vested in them under this document or the Corporations Act; and
- (b) in connection with all proceedings, expenses, claims and demands in relation to anything done or omitted in any way relating to Secured Property,

including legal expenses on a full indemnity basis and expenses incurred in engaging investigative accountants and financial consultants (but only where such engagement is made after the occurrence of an Event of Default or a Potential Event of Default) except to the extent that any such losses, liabilities, costs, expenses and Taxes are a direct and foreseeable result of the fraud, gross negligence or wilful misconduct of that person.

# 16.3 **Currency indemnity**

If, for any reason (including as a result of a judgment or order), an amount payable by the Grantor under or in respect of this document (**Relevant Amount**) is received by the Secured Party in a currency (**Payment Currency**) that is not the currency in which the amount is expressed to be payable under this document (**Required Currency**) then the Grantor, as an independent obligation, must indemnify the Secured Party against, and must pay the Secured Party on demand the amount of, any shortfall between:

- (a) the amount of Required Currency which the Secured Party receives on converting the amount it received in the Payment Currency into an amount in the Required Currency in accordance with its usual practice; and
- (b) the Relevant Amount in the Required Currency.

# 16.4 Recovery from Secured Property

A person who is entitled to be indemnified for a loss, liability, expense or Tax under clauses 16.1, 16.2 or 16.3 may recover the amount to be indemnified direct from the Secured Property.

### 17. **DISCHARGE**

The Secured Party must at the request and cost of the Grantor reconvey, surrender or release any remaining Secured Property (as appropriate) to the Grantor and the Secured Property will then be discharged from this document:

- (a) when the Secured Party is satisfied that:
  - (i) all the Secured Money has been irrevocably paid and discharged in full; and
  - (ii) no amount remains contingently payable or may become payable on the security of this document (including under an indemnity); and
- (b) on payment or retention of all expenses incurred by or payable to the Secured Party, its Authorised Representatives or any Receiver or Attorney.

Any discharge is subject to clause 14.4.

#### 18 **CONFIDENTIALITY**

### 18.1 **General confidentiality**

The parties to this document agree that the Secured Party must not disclose any information provided by the Grantor under the Equity Derivative Financing Documents to any person who is not a party, and the Grantor must not disclose any information concerning the contents of or the transactions contemplated by any Equity Derivative Financing Document, including in each case information of the kind mentioned in section 275(1) of the PPSA, to any person who is not a party, except to the extent that:

- (a) (**permitted by documents**) the disclosure is expressly permitted by an Equity Derivative Financing Document or required by law to ensure the enforceability or priority of any Equity Derivative Financing Document;
- (b) (consent of other party) the other party consents to the disclosure;
- (c) (**public domain**) the information is already in the public domain, unless it entered the public domain because of a breach of confidentiality by the disclosing party;
- (d) (employees and advisers) the disclosure is made on a confidential basis to the officers, employees, agents, lawyers, auditors or other professional advisers of the party or its holding company. For the avoidance of doubt, it is not mandatory for a receiving party to enter into any written confidentiality undertaking in respect of any disclosure made under this paragraph;
- (e) (**comply with laws**) the disclosure is necessary to comply with any applicable law, or an order of a court or tribunal;
- (f) (**comply with directives**) the disclosure is necessary to comply with a directive or request of any Government Agency (whether or not having the force of law);
- (g) **(obtain Authorisations)** the disclosure is necessary or desirable to obtain an Authorisation from any Government Agency;
- (h) (discovery and litigation) the disclosure is necessary or desirable in relation to any discovery of documents, or any proceedings before a court, tribunal or other Government Agency; or
- (i) **(Secured Party's purposes)** it is in the Secured Party's interest to disclose the information, including:
  - (i) to persons considering purchasing the Secured Party or who are involved in valuing the Secured Party for ratings purposes; or
  - (ii) to persons considering taking an assignment of, interest in or exposure to the Equity Derivative Financing Documents, including under securitisation or sub-participation arrangements.

Any disclosure made under this clause 18.1(i) must be made on the basis that the receiving party must keep the information confidential as required by this clause.

### 18.2 **No request**

The Grantor agrees not to exercise its rights to make any request of the Secured Party under section 275 of the PPSA, to authorise the disclosure of any information under that section or to waive any duty of confidence that would otherwise permit non-disclosure under that section (but this does not limit the Grantor's rights to request information other than under section 275).

### 19. **NOTICES**

### 19.1 How to give a notice

A notice, consent or other communication under this document is only effective if it is:

- (a) in writing, signed by or on behalf of the person giving it;
- (b) addressed to the person to whom it is to be given; and
- (c) given in one of the following ways:
  - (i) sent by prepaid mail (by airmail, if the addressee is overseas) or delivered to that person's address;
  - (ii) sent by fax to that person's fax number and the machine from which it is sent produces a report that states that it was sent in full without error;
  - (iii) given personally; or
  - (iv) given in any other manner permitted by law.

## 19.2 When a notice is given

Subject to clause 19.3, a notice, consent or other communication that complies with this clause is conclusively regarded as given and received:

- (a) if it is sent by fax or delivered, if received:
  - (i) by 5.00 pm (local time in the place of receipt) on a Business Day on that day; or
  - (ii) after 5.00 pm (local time in the place of receipt) on a Business Day, or on a day that is not a Business Day on the next Business Day;
- (b) if it is sent by mail, when it would be delivered in the ordinary course of post, but in any event:
  - (i) not later than three Business Days after posting within Australia; or
  - (ii) not later than seven Business Days after posting to or from a place outside Australia;
- (c) if given personally, when actually received by that person; and
- (d) if it is given in any other manner permitted by law, when actually received by that person, unless a later time of receipt is specified in it.

## 19.3 Notices to the Secured Party

Any notice, consent or other communication to be made or delivered to the Secured Party will be effective:

- (a) only when actually received by it; and
- (b) only if it is expressly marked for the attention of the department or officer specified in the following clause (or any substitute department or officer as it specifies for this purpose).

### 19.4 Address for notices

A person's address and fax number are those set out below, or as the person notifies the sender:

## **Secured Party**

Address: Level 16

Chifley Tower 2 Chifley Square Sydney NSW 2000

Australia

Fax number: +61-2-9324-2880

Email: <u>elaine.chow@ubs.com</u> / <u>kevin.bayett@ubs.com</u> / <u>sh-aus-eq-</u>

support@ubs.com / greq.jansz@ubs.com / grant.schwulst@ubs.com/ sh-

syd-rmp-op@ubs.com

Attention: Elaine Chow/Kevin Bayett/Greg Jansz/Grant Schwulst

# Grantor

Address: Level 7

Deloitte Tower Port Moresby

 $\mathsf{N}.\mathsf{C}.\mathsf{D}.$ 

Papua New Guinea

Fax number: +675 3200238

Email: <a href="mailto:robert.acevski@npcp.com">robert.acevski@npcp.com</a>

Attention: Robert Acevski

# 20. **AMENDMENT AND ASSIGNMENT**

# 20.1 Amendment

This document can only be amended or replaced by another document executed by the parties.

# 20.2 Assignment

- (a) The Grantor may only assign, encumber, declare a trust over or otherwise deal with its rights under any Equity Derivative Financing Document with the written consent of the Secured Party.
- (b) The Secured Party may assign, encumber, declare a trust over or otherwise deal with its rights under this document without the consent of the Grantor, and may disclose to any potential holder of the right, or an interest in the right, any information relating to this document or any party to it.

# 21. **GENERAL**

# 21.1 Governing law

(a) This document is governed by the laws of the State of New South Wales.

- (b) Each party submits to the non-exclusive jurisdiction of the courts of the State of New South Wales, and any court that may hear appeals from any of those courts, for any proceedings in connection with any Equity Derivative Financing Document.
- (c) The Grantor irrevocably waives:
  - (i) any objection to the venue of any proceedings on the ground that they have been brought in an inconvenient forum; and
  - (ii) any immunity from set off, suits, proceedings and execution to which it or any of its property may now or in the future be entitled under any applicable law.
- (d) The Grantor confirms the appointment under the "Process Agent" provisions of the Equity Derivative Financing Agreement of Gadens Lawyers Sydney Pty Ltd ACN 100 963 308 of Level 16, 77 Castlereagh Street, Sydney NSW 2000 as its agent to receive service of process for any proceedings in connection with any Equity Derivative Financing Document. The Grantor undertakes to maintain this appointment until all Secured Money is repaid, and agrees that any process served on that person is taken to be served on it.

# 21.2 Liability for own expenses

The Grantor is liable for its own costs and expenses in complying with this document, including where it does so at the Secured Party's request or for the Secured Party's benefit.

### 21.3 Giving effect to this document

- (a) The Grantor must do anything (including executing any Transfer and other transfer in blank, or any other document, and perfecting and protecting any Security Interest intended to be created by or pursuant to this document), and must ensure that its employees and agents do anything, that the Secured Party may reasonably require to:
  - (i) give full effect to this document; or
  - (ii) more fully secure the rights, remedies and powers of the Secured Party under this document or to enable the Secured Party to exercise those rights, remedies and powers,

including giving directions to the Nominee or any Controlling Participant in respect of any CHESS Securities in the Secured Property.

(b) The Secured Party may, at the Grantor's cost, do anything which the Grantor should have done under this document if the Grantor does not do so promptly or, if in the Secured Party's opinion, the Grantor does not do so properly.

### 21.4 Authority to register and waiver of right to receive verification statements

The Grantor acknowledges that the Secured Party may, at the Grantor's cost, register one or more financing statements in relation to its Security Interest. If permitted by the PPSA, the Grantor waives its right under section 157 of the PPSA to receive notice of any verification statement relating to the registration of any such financing statement or any related financing change statement.

# 21.5 **Variation of rights**

The exercise of a right partially or on one occasion does not prevent any further exercise of that right in accordance with the terms of this document. Neither a forbearance to exercise a right nor a delay in the exercise of a right operates as an election between rights or a variation of the terms of this document.

### 21.6 **Operation of this document**

- (a) Subject to clause 21.6(b), the Equity Derivative Financing Documents contain the entire agreement between the parties about their subject matter. Any previous understanding, agreement, representation or warranty relating to that subject matter is replaced by the Equity Derivative Financing Documents and has no further effect.
- (b) Any right that the Secured Party may have under the Equity Derivative Financing Documents is in addition to, and does not replace or limit, any other right that the Secured Party may have.
- (c) Any provision of this document which is unenforceable or partly unenforceable is, where possible, to be severed to the extent necessary to make this document enforceable, unless this would materially change the intended effect of this document.

# 21.7 **Operation of indemnities**

- (a) Each indemnity in this document survives the expiry or termination of this document.
- (b) The Secured Party may recover a payment under an indemnity in this document before it makes the payment in respect of which the indemnity is given.
- (c) If a provision of this document is expressed to:
  - (i) indemnify;
  - (ii) exclude or limit any liability of; or
  - (iii) otherwise benefit,

a person who is not a party to this document, the Grantor agrees that the Secured Party holds the benefit of that indemnity, exclusion, limitation or other benefit for that person and may enforce this document on their behalf and for their benefit.

### 21.8 Consents

Where this document contemplates that the Secured Party may agree or consent to something (however it is described), the Secured Party may:

- (a) agree or consent, or not agree or consent, in its absolute discretion; and
- (b) agree or consent subject to conditions,

unless this document expressly contemplates otherwise.

# 21.9 Statements by Secured Party

A statement by an Authorised Representative of the Secured Party on any matter relating to any Equity Derivative Financing Document (including any amount owing by the Grantor) is conclusive unless clearly wrong on its face.

## 21.10 **Set-off**

If an Event of Default occurs and is continuing, the Secured Party, without notice to the Grantor, may combine any account that the Grantor holds at any branch or office (in Australia or elsewhere) of the Secured Party with, or set off any amount in any currency that is or may become owing in any currency by the Secured Party to the Grantor against, any amount owing by the Grantor to the Secured Party under the Equity Derivative Financing Documents. For this purpose the Secured Party may:

- (a) change the terms (including the repayment date) of any account or other payment obligation between the parties;
- (b) convert amounts into different currencies in accordance with the Secured Party's usual practice; and
- (c) do anything (including execute any document) in the name of the Grantor that the Secured Party considers necessary or desirable.

This subclause overrides any other document or agreement to the contrary.

### 21.11 **No merger**

Nothing in this document merges with any other Security Interest, or any Guarantee, judgment or other right or remedy, that the Secured Party may hold at any time.

### 21.12 Exclusion of contrary legislation

Any legislation that affects an obligation of the Grantor in a manner that is adverse to the interests of the Secured Party, or adversely affects the exercise by the Secured Party of a right or remedy, under or relating to this document is excluded to the full extent permitted by law.

# 21.13 Counterparts

This document may be executed in counterparts. Delivery of a counterpart of this document by email attachment or fax constitutes an effective mode of delivery.

# 21.14 Execution by fewer than all parties

This document binds the Grantor even if the Secured Party does not execute or only subsequently executes this document.

**EXECUTED** as a deed.

[EXECUTION BLOCKS INTENTIONALLY OMITTED]

# Annexure D



# **Nominee Deed**

# **NPCP Investments Limited**

# **UBS AG, Australia Branch**

ABN 47 088 129 613

# **UBS Nominees Pty Ltd**

ABN 32 001 450 522

# **UBS Securities Australia Limited**

ABN 62 008 586 481

# Project Kutubu

12 March 2014, as novated / amended and restated on or about 12 December 2014

# **CONTENTS**

CLA	USE		PAGE	
1.	INTERPRETATION			
	1.1 1.2	Definitions		
2.	DECLARATION			
	2.1 2.2 2.3 2.4 2.5 2.6 2.7	Declaration of Trust Present entitlement Termination No disposal Beneficial interest Liability of Indemnified Parties Encumbrances		
3.	VOTING AT MEETINGS OF THE ISSUER			
	3.1 3.2 3.3	RequestVotingTransfer of Shares back to UBS	8	
4.	UNDERTAKING			
	4.1 4.2 4.3	Payment undertaking NPCP undertakings. Nominee undertakings.	9	
5.	POWERS AND DUTIES OF THE NOMINEE			
	5.1 5.2 5.3 5.4	Limitations  Transfer of Property  Directions of the Court  Knowledge of the Nominee.	10 10	
6.	OBLIGATIONS OF THE NOMINEE			
	6.1 6.2 6.3 6.4 6.5	Extent of discretion In funds No responsibility for deed Exclusion of further obligations Set Off	10 11	
7.	LIMITATION OF LIABILITY			
	7.1 7.2 7.3 7.4 7.5 7.6 7.7	Indemnity Limitation of liability and obligations Nominee's lien as trustee Reliance on documents and experts Independent Investigation No monitoring Replacement of Nominee		
8.	TAXE	ES	13	
	8.1 8.2 8.3 8.4 8.5	Obligation to pay Tax  Notice to NPCP  Withholding.  Time for payment  GST	14 14	
9	ΡΔΥΝ/	PAYMENTS		

10.	NOTICES			
	10.1 10.2 10.3 10.4 10.5	Notice by UBS and the Nominee Time of receipt Notice by NPCP Notices Between UBS and Nominee Addresses for notices	15 16 16	
11.	PROC	PROCESS AGENT		
12.	PARTI	CIPANT		
13.	GENE	RAL	19	
	13.1 13.2 13.3 13.4 13.1 13.2	Stamp duty Telephone recording Amendments to deed Waiver Waiver of immunity. Governing Law	19 19 19 19	
	10.2	Governing Law		

**THIS DEED** is made on 12 March 2014, as novated / amended and restated on or about 12 December 2014

### **BETWEEN**

- (1) NPCP Investments Limited (NPCP);
- (2) **UBS AG, Australia Branch** ABN 47 088 129 613 (**UBS**);
- (3) **UBS Nominees Pty Ltd** ABN 32 001 450 522 (**Nominee**); and
- (4) **UBS Securities Australia Limited** ABN 62 008 586 481 (**Participant**).

### **RECITALS**

The Nominee agrees to act on the terms set out in this deed as trustee for NPCP.

### THE PARTIES AGREE AS FOLLOWS:

### 1. **INTERPRETATION**

### 1.1 **Definitions**

Unless otherwise defined in this deed, or unless the context otherwise requires, capitalised terms and expressions used in this deed have the respective meanings given to them in a Confirmation (including by reference to one or more other documents). The following definitions also apply in this deed.

**Amounts Payable** means all amounts (including damages) that are payable, owing but not payable, or that otherwise remain unpaid by the NPCP to UBS on any account at any time (whether present or future, actual or contingent) under or in connection with the Equity Derivative Financing Documents or the transactions which they contemplate.

**ASIC** means the Australian Securities and Investments Commission.

ASX Settlement means ASX Settlement Pty Ltd (ABN 49 008 504 532).

**ASX Settlement Operating Rules** means the operating rules of the clearing and settlement facility operated by ASX Settlement.

**Beneficial Interest** means the beneficial interest of NPCP in the Trust Property.

# **Confirmation** means:

- (a) the long-form confirmation in respect of a step collar derivative transaction dated 12 March 2014 between UBS and the Independent State of Papua New Guinea, as novated from the Independent State of Papua New Guinea to NPCP pursuant to the Global Novation and Amendment Deed (as amended, restated, novated, supplemented, modified or varied from time to time); or
- (b) the long-form confirmation in respect of a step collar derivative transaction dated on or about 12 December 2014 between UBS and NPCP,

or both of the above.

Corporations Act means the Corporations Act 2001 (Cth).

### **Encumbrance** means:

(a) a security interest that is subject to the PPS Act;

- (b) any other mortgage, pledge, lien or charge;
- (c) an easement, restrictive covenant, caveat or similar restriction over property; or
- (d) any other interest or arrangement of any kind that in substance secures the payment of money or the performance of an obligation, or that gives a creditor priority over unsecured creditors in relation to any property,

in each case, whether existing or agreed to be granted or created.

**Equity Derivative Financing Agreement** means the confirmation letter dated 12 March 2014 between Party A and the Independent State of Papua New Guinea, including any ISDA master agreement or credit support annex entered into or deemed to be entered into under or in relation to the letter or the transactions contemplated therein, as novated from the Independent State of Papua New Guinea to Party B pursuant to the Global Novation and Amendment Deed.

**Equity Derivative Financing Document** has the meaning given to it in the Equity Derivative Financing Agreement.

**Global Novation and Amendment Deed** means the document of that title dated on or about 12 December 2014 between, among others, Independent State of Papua New Guinea, UBS and NPCP.

**Government Agency** means a government or government department, or governmental or semi-governmental person and a person (whether autonomous or not) charged with administration of any applicable law and includes without limitation the Australian Tax Office and any revenue authority of Papua New Guinea.

**GST** has the same meaning as given to that term in the GST Law.

**GST Law** has the meaning given to the term "GST law" in *A New Tax System (Goods & Services Tax) Act 1999* (Cth).

# **Indemnified Party** means:

- (a) the Nominee;
- (b) UBS; and
- (c) any successor, substitute or assign of any of the above.

**Nominee** means the trustee of the Trust from time to time being UBS Nominees Pty Ltd (ABN 32 001 450 522) at the date of this deed.

**Payment Undertaking** has the meaning given to that term in clause 4.1 (*Payment undertaking*).

PPS Act means the Personal Property Securities Act 2009 (Cth).

**Tax** includes any tax, levy, impost, deduction, charge, rate, duty, compulsory loan or withholding which is levied or imposed by a Government Agency including GST, and any related interest, penalty, charge, fee or other amount, excluding stamp duty payable on this deed, or on the transfer of property to or by UBS or the Nominee.

**Tax Invoice** means a tax invoice complying with the requirements of any law about GST.

**Trust** means the trust constituted under this deed.

**Trust Property** means the sum of A\$10 and all other property transferred to and held by the Nominee pursuant to this deed from time to time (including any Shares) and all related rights, benefits and interests.

**UBS Entity** means UBS and any of its related bodies corporate.

# 1.2 Interpretation - references

Unless the contrary intention appears, a reference in this deed to:

- (a) (variations or replacements) a document (including this deed) includes any variation or replacement of it;
- (b) (clauses, annexures and schedules) a clause, annexure or schedule is a reference to a clause in or annexure or schedule to this deed:
- (c) (**reference to statutes**) a statute, ordinance, code or other law or the ASX Settlement Operating Rules includes regulations and other instruments under it and consolidations, amendments, re-enactments or replacements of any of them;
- (d) (law) law means common law, principles of equity, and laws made by parliament (and laws made by parliament include State, Territory and Commonwealth laws and regulations and other instruments under them, and consolidations, amendments, re-enactments or replacements of any of them);
- (e) (singular includes plural) the singular includes the plural and vice versa;
- (f) **(person)** the word "person" includes an individual, a firm, a body corporate, a partnership, a joint venture, an unincorporated body or association, or any government agency;
- (g) (executors, administrators, successors) a particular person includes a reference to the person's executors, administrators, successors, substitutes (including persons taking by novation) and assigns;
- (h) **(two or more persons)** an agreement, representation or warranty in favour of two or more persons is for the benefit of them jointly and each of them individually;
- (i) (**jointly and severally**) an agreement, representation or warranty by two or more persons binds them jointly and each of them individually except that the obligations of UBS and the Nominee under this deed are several and not joint;
- (j) (dollars) Australian dollars, dollars, A\$ or \$ is a reference to the lawful currency of Australia;
- (k) (meaning not limited) the words "including", "for example" or "such as" when introducing an example, do not limit the meaning of the words to which the example relates to that example or examples of a similar kind;
- (I) (time of day) a date or time is a reference to a date or time in Sydney;
- (m) (next Business Day) under this deed if an event must occur on a stipulated day which is not a Business Day then the stipulated day will be taken to be the next Business Day;
- (n) (headings) headings (including those in brackets at the beginning of paragraphs) are for convenience only and do not affect the interpretation of this deed; and

(o) (**definitions**) a word or phrase defined in the Corporations Act or ASX Settlement Operating Rules, has the same meaning in this deed.

### 2. **DECLARATION**

### 2.1 **Declaration of Trust**

The Nominee declares that it holds each item of Trust Property as a trustee on the terms of this deed on trust for NPCP solely.

Subject to this deed, the Nominee undertakes to NPCP that it will:

- (a) transfer, deal with or otherwise dispose of the Trust Property as NPCP may direct and not otherwise; and
- (b) recognise the right of NPCP to dividends or distributions on Trust Property and will account to NPCP or as NPCP may direct for all Income or other distributions paid or made from time to time in respect of the Trust Property; and
- (c) deliver promptly or procure the transfer on demand to NPCP, or as it may direct, legal title to the Trust Property.

### 2.2 **Present entitlement**

NPCP is presently entitled to and has an absolute vested and indefeasible interest in the income of the Trust.

### 2.3 **Termination**

The Trust under this deed will terminate on the earlier of:

- (a) the date 80 years from the date of this deed; and
- (a) any earlier date determined by the Nominee in its absolute discretion with the consent of UBS.

### 2.4 No disposal

- (a) Except as expressly permitted by this deed, neither the Nominee nor NPCP shall dispose of any Trust Property during the continuation of the Trust.
- (b) In consideration for the transactions contemplated under the Equity Derivative Financing Documents, NPCP gives to UBS, and UBS has, irrevocable authority to give to the Nominee, on behalf of NPCP, such direction and instructions as to the disposal of Trust Property as NPCP is entitled to give under this deed. Without limiting the generality of this paragraph (b), this includes any such direction and instructions as to the disposal of Trust Property following UBS or any agent on its behalf appointing any controller or taking possession of any of NPCP's property (or any analogous event occurring under the law of any applicable jurisdiction).

# 2.5 **Beneficial interest**

- (a) NPCP is absolutely entitled pursuant to this deed as against the Nominee to the Trust Property.
- (b) NPCP must forebear from requiring the transfer of Trust Property to it or its nominee (other than the Nominee) prior to the termination date of the Trust.

# 2.6 Liability of Indemnified Parties

To the extent permitted by law, no Indemnified Party will be liable in respect of any conduct, delay, negligence or breach of duty in the exercise or non-exercise of any power, nor for any loss (including direct and indirect, economic and consequential loss) which results, except where it arises from fraud, gross negligence, or wilful misconduct by or on behalf of the Indemnified Party.

### 2.7 Encumbrances

- (a) The Nominee and NPCP must not create, or take or omit to take any action to permit to exist, any Encumbrance or other third party interest (other than an Encumbrance or third party interest created under an Equity Derivative Financing Document) over or in respect of any Trust Property or any interest in the Trust Property. Any such Encumbrance or other third party interest shall be invalid and unenforceable as between and/or against UBS, the Nominee and NPCP (or any person claiming under or through any of them).
- (b) Each of the Nominee and NPCP represents and warrants to UBS that no Trust Property is subject to any Encumbrance or other third party interest created by it (other than an Encumbrance or third party interest created under an Equity Derivative Financing Document).

### 3. **VOTING AT MEETINGS OF THE ISSUER**

### 3.1 Request

- (a) If NPCP is, or would be, eligible to vote at the meeting of members of the Issuer:
  - (i) at the record time for the members' vote; and
  - (ii) during the time when the Shares are proposed to be held by the Nominee in accordance with this clause 3 (*Voting at meetings of the Issuer*),

and provided that:

- (iii) no Event of Default or Termination Event has occurred in respect of NPCP;
- (iv) an Ex-Distributions Date in respect of the Shares is not scheduled or anticipated to occur during the time when the Shares are proposed to be held by the Nominee in accordance with this clause 3 (*Voting at meetings of the Issuer*); and
- (v) UBS determines (acting reasonably) that no Extraordinary Event would occur or is likely to occur during the time when the Shares are proposed to be held by the Nominee in accordance with this clause 3 (*Voting at meetings* of the Issuer),

then NPCP may request in writing, at least 10 Business Days prior to the proposed transfer date, that UBS transfers to the Nominee a specified number of Shares which Shares must not exceed the number of Shares in the Collateral Balance of NPCP at the time of the request.

- (b) UBS will use its best commercial efforts (as determined by UBS) to comply with a written request from NPCP under paragraph (a), subject to the terms of the Equity Derivative Financing Documents.
- (c) NPCP must pay to UBS the stock borrow cost incurred by UBS to procure the delivery of Shares to be held by the Nominee, by no later than the third Business

Day after the Shares are transferred to UBS as contemplated under clause 3.3 (*Transfer of Shares back to UBS*). The borrow cost shall be calculated on no more than the Delta Quantity of Shares.

(d) To the fullest extent permitted by law, UBS has and is under no liability (whether in negligence or otherwise) to NPCP or the Nominee for any costs, liabilities, losses or expenses incurred or suffered by NPCP or the Nominee in connection with or relating to any ability or inability of the Nominee or NPCP to cast any vote at any meeting of members (or a class of members) of the Issuer if Shares are not delivered to the Nominee as contemplated by this clause 3 (Voting at meetings of the Issuer).

### 3.2 Voting

In respect of any Shares comprised in the Trust Property:

- (a) subject to paragraph (b), the Nominee shall not cast any vote at any meetings of members (or a class of members) of the Issuer; and
- (b) NPCP may direct the Nominee to cast a vote at any meeting of members (or a class of members) of the Issuer in respect of any such Shares, in which case the Nominee must vote in accordance with the direction of NPCP.

### 3.3 Transfer of Shares back to UBS

- (a) NPCP acknowledges and agrees that for so long as any Shares are held by the Nominee, NPCP has an outstanding obligation to re-deliver Equivalent Credit Support under the terms of a Confirmation.
- (b) Accordingly, the parties acknowledge and agree that the Shares are to be held by the Nominee for the shortest period of time necessary to give NPCP the entitlement to direct the Nominee in accordance with clause 3.2 (*Voting*) to vote at the meeting of members referred to in clause 3.1 (*Request*). NPCP hereby irrevocably and unconditionally instructs and directs the Nominee to transfer to UBS (which transfer is undertaken in the ordinary exercise of the Nominee's lien as trustee) all Shares held by the Nominee as soon as practicable and in any event within one Business Day after the record time for the member's vote.
- (c) Without limiting clause 3.1(a) or the other provisions of this clause 3.3 (*Transfer of Shares back to UBS*), if an Ex-Distributions Date for the Shares is scheduled or anticipated to occur at any time while the Shares are held by the Nominee, then NPCP hereby irrevocably and unconditionally instructs and directs the Nominee to transfer to UBS (which transfer is undertaken in the ordinary exercise of the Nominee's lien as trustee) all Shares held by the Nominee as soon as practicable and in any event prior to the relevant Ex-Distributions Date in respect of the Shares.
- (d) All Shares transferred to UBS in accordance with this clause 3.3 (*Transfer of Shares back to UBS*) will constitute the delivery of Eligible Credit Support under and for the purposes of a Confirmation.
- (e) NPCP must indemnify the Indemnified Parties against, and must pay to the Indemnified Parties on demand the amount of, all costs, liabilities, losses or expenses incurred by an Indemnified Party in respect of any delay in transfer of the Shares caused directly or indirectly by NPCP.

- (f) If, notwithstanding and without limiting paragraph (c):
  - (i) the Shares are held by the Nominee on an Ex-Distributions Date of a Distribution declared or resolved to be paid or made in respect of the Shares; and
  - (ii) the Nominee receives any Distribution,

then NPCP agrees that:

- (iii) the Nominee shall deliver to UBS, or pay to UBS an amount equal to, the entirety of such Distribution received from the Issuer by no later than the second Business Day after the making of such Distribution by the Issuer; and
- (iv) the provisions of a Confirmation in respect of Distributions in respect of Shares shall apply, as if the Shares remained held by UBS pursuant to the terms of a Confirmation, and with any other necessary changes.

NPCP hereby irrevocably and unconditionally instructs and directs the Nominee to make each such payment contemplated by this paragraph (f).

### 4. UNDERTAKING

# 4.1 **Payment undertaking**

At any time while the Trust Property consists of Shares, the Nominee indemnifies UBS against, and must pay to UBS on demand, the amount of all Amounts Payable (**Payment Undertaking**). The Payment Undertaking is a continuing indemnity and will (subject to the Nominee's limitation of liability in clause 7.2 (*Limitation of liability and obligations*)) extend to the ultimate balance of sums payable by NPCP under the Equity Derivative Financing Documents, regardless of any intermediate payment or discharge in whole or in part.

# 4.2 **NPCP undertakings**

NPCP agrees:

- (a) to comply with all obligations affecting the Trust Property and pay on time all amounts for which the holder of the Trust Property is liable, including calls, instalments and Taxes;
- (b) at UBS's request in its discretion, to take up any rights attaching to Trust Property;
- (c) to promptly provide to UBS on request any information which UBS reasonably requests about the Trust Property or anything in relation to it; and
- (d) not to provide any instructions to the Nominee relating to the Trust Property that are inconsistent with this deed or any other Equity Derivative Financing Document.

### 4.3 **Nominee undertakings**

The Nominee and NPCP each acknowledges and agrees for the benefit of UBS in respect of each Share comprised in the Trust Property that:

(a) the Nominee will not comply with instructions given by NPCP without seeking the consent of UBS except as expressly contemplated under clause 3 (*Voting at meetings of members*); and

(b) the Nominee must comply with instructions (including instructions to transfer the Shares) given by UBS in accordance with the Equity Derivative Financing Documents without seeking the consent of **NPCP**.

# 5. **POWERS AND DUTIES OF THE NOMINEE**

### 5.1 **Limitations**

The Nominee shall have no powers, duties, rights or discretions in respect of the Trust except those expressly set out in this deed. In particular the Nominee shall have no power to dispose of Trust Property otherwise than in accordance with this deed.

# 5.2 **Transfer of Property**

Any power or duty of the Nominee to transfer any property under this deed (including the transfer of Trust Property) may be exercised or satisfied by instructing the Participant to transfer or dispose of the relevant property and/or posting or delivering to the registered address of the recipient an instrument of transfer of the relevant property in addition to any other mechanism or process permitted by law. Compliance by the Nominee with this clause will be a complete discharge of the Nominee's obligations with respect to the transfer of the relevant property.

### 5.3 **Directions of the Court**

UBS and/or the Nominee and/or NPCP may apply to a court for directions as to any matter arising in connection with the exercise of the powers and functions of UBS and/or the Nominee and/or NPCP under this deed, and the applicant shall not be responsible for any delay arising as a result.

# 5.4 Knowledge of the Nominee

The Nominee will only be considered to have knowledge, awareness or notice of a thing, or grounds to believe any thing, by virtue of the officers of the Nominee having day to day responsibility for the administration of the Trust having actual knowledge, actual awareness or actual notice of that thing, or grounds to believe that thing (and similar references will be interpreted in this way).

### 6. **OBLIGATIONS OF THE NOMINEE**

### 6.1 Extent of discretion

The Nominee need not:

- (a) act except when required to do so by this deed and then at the discretion of the Nominee as to the manner and time of acting; or
- (b) consult with **NPCP** before acting or giving any consent, approval or agreement or making any determination under this deed except where this deed expressly provides otherwise (including under clause 3.2 (*Voting*)).

# 6.2 In funds

The Nominee shall not act or fail to act if by doing so it will knowingly incur any material liability for which it is not satisfied that it is sufficiently indemnified from the Trust Property or otherwise.

# 6.3 No responsibility for deed

The Nominee has no responsibility for the form or contents of this deed or any other Equity Derivative Financing Document and will have no liability arising as a result of or in connection with any inadequacy, invalidity or unenforceability of any provision of this deed or any other Equity Derivative Financing Document.

### 6.4 Exclusion of further obligations

Despite any provision of this deed, the Nominee:

- (a) is not required to do or omit to do anything or incur any liability unless the Nominee's liability is limited in a manner satisfactory to the Nominee;
- (b) will not be under any obligations to advance or use its own funds for the payment of any costs, expenses or facilities;
- (c) will have no liability for the acts or omissions of any brokers or other agents whose acts or omissions are not reasonably capable of supervision by the Nominee;
- (d) has no responsibility for the performance by UBS and/or NPCP of their obligations under or in connection with this deed, the Trust Property, and the Nominee will have no liability arising as a result of or in connection with any act or omission of UBS and/or NPCP; and
- (e) will have no obligations to prepare accounts or tax returns in respect of the Trust.

### 6.5 **Set Off**

The Nominee may set off or withhold any amount payable to it by UBS and/or NPCP against or from any amount payable by the Nominee to UBS and/or NPCP (as the case may be).

### 7. **LIMITATION OF LIABILITY**

### 7.1 **Indemnity**

- (a) The Nominee:
  - (i) is fully indemnified out of the Trust Property in respect of any liability or obligation that the Nominee incurs while acting or purporting to act as Nominee; and
  - (ii) may reimburse itself from the Trust Property for all expenses incurred or payments made by it in respect of the Trust, except in relation to any liability, obligation or expense incurred as a result of the Nominee's fraud, gross negligence or wilful default.

The Trustee may apply either or both capital and income of the Trust Fund to satisfy its rights under this clause.

(b) Without limiting paragraph (a) above, NPCP acknowledges and agrees that the Nominee's liabilities and undertakings under this deed including in respect of the Payment Undertaking are expenses of the Trust for which the Nominee is entitled to be indemnified out of Trust Property. Accordingly, NPCP acknowledges and agrees that that if UBS makes a demand on the Nominee under the Payment Undertaking given under clause 4.1 (Payment undertaking), the Nominee (under

its trustee's lien) may (without first notifying **NPCP**) dispose of the Trust Property and use the proceeds of that disposal to satisfy that demand.

# 7.2 Limitation of liability and obligations

- (a) Despite any other provision of this deed or any other Equity Derivative Financing Document, a liability or obligation of the Nominee arising under or in connection with this deed or any other Equity Derivative Financing Document (including in connection with the Payment Undertaking) is strictly limited to the extent to which (and can be enforced against the Nominee only to the extent to which) the Nominee is actually indemnified for the liability and the liability or obligation can lawfully be satisfied of the Trust Property. No person may seek to recover any shortfall in the amounts owing by the Nominee by bringing proceedings against the Nominee or applying to have the Nominee wound up.
- (b) When the Nominee, whether on instructions or under the Nominee's lien as trustee, exercise a power to dispose of Trust Property, the Nominee has no liability to **NPCP** for any delay in disposal or any failure to obtain a particular price or for obtaining difference prices for the Trust Property.

### 7.3 Nominee's lien as trustee

**NPCP** acknowledges and agrees that the Nominee's common law lien as trustee is a proprietary interest in the Trust Property that ranks in priority to the Beneficial Interest. However, nothing in this deed is intended to create a charge or security interest separate from the Trustee's rights at common law.

## 7.4 Reliance on documents and experts

The Nominee may rely on:

- (a) any document (including any facsimile transmission or telegram) unless it knows or has reasonable grounds to believe that the document is not genuine and correct; and
- (b) advice, information and statements of UBS or other persons employed or retained by UBS,

and shall not be liable for any act or omission undertaken in reliance on that document, advice, information or statement. Without limitation, the Nominee may rely on such documents, advice, information and statements for the purpose of exercising and performing its powers and duties to make payments to **NPCP** and to transfer or dispose of Trust Property in accordance with this deed. Where the Nominee considers that it requires any advice, information or statement from any of the persons referred to in paragraph (b) of this clause, for the purpose of exercising or performing its powers and duties under this deed, the Nominee will have no liability in connection with any delay in the exercise or performance of the relevant powers or duties pending receipt of the relevant advice, information or statements.

# 7.5 **Independent Investigation**

**NPCP** confirms that it has made and will continue to make, independently and without reliance on the Nominee or UBS:

- (a) its own investigations into the affairs of UBS and the Nominee; and
- (b) its own analyses and decisions whether to take or not take action under this deed.

# 7.6 **No monitoring**

The Nominee is not required to keep itself informed as to any Issuer or the compliance by UBS and/or NPCP with this deed or any other document or agreement or to inspect any property or book of UBS and/or NPCP.

# 7.7 **Replacement of Nominee**

- (a) Subject to the appointment of a successor Nominee as provided in this clause:
  - (i) the Nominee may resign at any time by giving not less than 30 days written notice to UBS and/or NPCP; and
  - (ii) UBS may remove the Nominee from office if the Nominee breaches any of its obligations under this deed and the breach remains unremedied after UBS has given not less than 30 days written notice to the Nominee.
- (b) On notice of resignation or removal UBS has the right to appoint a successor Nominee who accepts the appointment.
- (c) If no successor Nominee is appointed within 30 days after notice, the retiring Nominee may appoint a successor Nominee who accepts the appointment.
- (d) On its appointment the successor Nominee will have all the rights, powers and obligations of the retiring Nominee (except liabilities arising from defaults).
- (e) The retiring Nominee shall execute and deliver all documents or agreements which UBS considers necessary or desirable in UBS's opinion to transfer to the successor Nominee this deed or to effect the appointment of the successor Nominee (subject to satisfaction of all liabilities owed to the Nominee on its own account under this deed or for which the Nominee may be personally liable).
- (f) After any retiring Nominee's resignation or removal, this clause will continue in effect in respect of anything done or omitted to be done by it while it was acting as Nominee, and the retiring Nominee retains the rights and remedies available to it under this deed or at law in relation to the performance and exercise of its powers, duties and functions while Nominee.
- (g) UBS shall not unreasonably withhold its approval of any proposed successor Nominee. It shall respond as soon as practicable to any request for approval.
- (h) **NPCP** agrees not to remove the Nominee as its nominee under this deed during the term of the Equity Derivatives Financing Documents.

## 8. TAXES

### 8.1 **Obligation to pay Tax**

(a) Where the Nominee has received notice requiring the payment of any Tax on behalf or in respect of **NPCP** or Trust Property, the Nominee shall determine, on advice, whether it has any obligation to pay or provide for the amount of the Tax. The Nominee may also at any time obtain advice as to whether it has any obligation to pay any Tax on behalf of or in respect of **NPCP** or any Trust Property, whether or not it has received notice requiring payment. If the Nominee determines that it has an obligation to pay the Tax, or there are reasonable grounds to argue that it has an obligation to pay the Tax, the Nominee shall give notice to **NPCP**, with a copy of that notice to be provided to UBS.

(b) **NPCP** agrees to indemnify the Nominee in respect of any tax referred to in clause 8.1(a).

### 8.2 Notice to NPCP

A notice given by the Nominee to **NPCP** pursuant to clause 8.1 (*Obligation to pay Tax*) shall set forth reasonable particulars pertaining to the obligation to make the relevant Tax payment.

### 8.3 Withholding

- (a) If the Nominee receives advice that it is prudent to withhold any amount of Tax in respect of any amount otherwise payable to **NPCP** then, notwithstanding any other provision in this deed, the Nominee will withhold that amount for so long as it remains prudent to do so.
- (b) Despite any other provision in this deed, if at any time an applicable law obliges the Nominee to make a deduction or withholding in respect of Taxes from a payment under this document, the Nominee:
  - (i) must notify NPCP of the obligation (with a copy of that notification to be provided to UBS);
  - (ii) must ensure that the deduction or withholding does not exceed the minimum amount required by law; and
  - (iii) must pay to the relevant Government Agency on time the full amount of the deduction or withholding and promptly deliver to NPCP a copy of any receipt, certificate or other proof of payment.
- (c) If any amount is deducted or withheld in accordance with paragraph (b) above, the Nominee is not required to gross up any payment or future payment either under this deed or any other Equity Derivative Document to **NPCP** in respect of that amount deducted or withheld.

## 8.4 Time for payment

**NPCP** must make any payment due pursuant to the indemnity in clause 8.1(b) on or before the day that is five Business Days before the last date on which payment of the relevant Tax may lawfully be made without incurring penalties or interest for late payment.

## 8.5 **GST**

If GST is or will be payable on a supply by UBS or **NPCP** made under or in connection with this deed, to the extent that the consideration otherwise provided for that supply under this deed is not stated to include an amount in respect of GST on the supply, **NPCP** must:

- (a) pay to UBS or the Nominee an additional amount equal to any GST which UBS or the Nominee must pay for any supply by UBS or the Nominee to NPCP under this deed;
- (b) pay that additional amount as and when **NPCP** must pay the other consideration for the supply; and

(c) indemnify UBS and the Nominee from any GST and against any damage or cost directly or indirectly arising from or caused by the failure of **NPCP** to pay any amount as and when required by this clause.

UBS or the Nominee will issue NPCP with a Tax Invoice in respect of any additional amount on account of GST that UBS or Nominee may recover from NPCP under this clause.

### 9. **PAYMENTS**

All monies payable under this deed shall be paid by electronic funds transfer to an account designated by the payee unless otherwise agreed.

### 10. **NOTICES**

# 10.1 Notice by UBS and the Nominee

Except where otherwise provided in this deed, all notices required or permitted to be given by UBS or the Nominee to **NPCP** pursuant to this deed must be in writing and will be duly given if:

- (a) left at NPCP's address as set out in clause 10.5 (Addresses for notices); or
- (b) sent by prepaid mail to **NPCP** 's address (which must be airmail if that address is not within Australia);
- (c) sent by facsimile transmission to NPCP 's facsimile number; or
- (d) sent in electronic form (such as email), with the subject matter line of an email specifying **NPCP**'s name, this deed and a brief description of the subject matter of the communication.

# 10.2 Time of receipt

A notice given by UBS or the Nominee in accordance with clause 10.1 is treated as having been duly given and received:

- (a) when delivered, in the case of a notice left at NPCP 's address;
- (b) on the second Business Day after posting, in the case of being sent by pre-paid mail to an address in Australia;
- (c) on the fifth Business Day after posting, in the case of being sent by pre-paid mail to an address outside Australia; and
- (d) at the time of confirmation of transmission, in the case of being sent by facsimile transmission; and
- (e) when sent in readable form:
  - (i) by 5.00 pm (local time in the place of receipt) on a Business Day on that Business Day; or
  - (ii) after 5.00 pm (local time in the place of receipt) on a Business Day, or on a day that is not a Business Day on the next Business Day,

in the case of being sent in electronic form.

# 10.3 **Notice by NPCP**

All notices required or permitted to be given by **NPCP** to UBS or the Nominee pursuant to this deed must be in writing and will be treated as being duly given if and only if they are actually received by UBS or the Nominee at the address of that party as set out in clause 10.5 (*Addresses for notices*) or at such other address as may have been notified to the **NPCP**.

### 10.4 Notices Between UBS and Nominee

Notices required or permitted to be given by UBS to the Nominee or by the Nominee to UBS may be given orally provided UBS and the Nominee are related bodies corporate.

### 10.5 Addresses for notices

### **NPCP**

Address: Level 7 Deloitte Tower,

Douglas Street Port Moresby NCD Papua New Guinea

Telephone: +675 320 2253

Fax number: +675 320 0238

Email: robert.acevski@npcp.com

Attention: Robert Acevski

**UBS** 

Address: Level 16

Chifley Tower 2 Chifley Square Sydney NSW 2000

Australia

Fax number: +61-2-9324-2880

Email: elaine.chow@ubs.com / kevin.bayett@ubs.com / sh-aus-eq-

support@ubs.com / justin.churchill@ubs.com/luke.goldsworthy@ubs.com/

jalpa.schillig@ubs.com / sh-syd-rmp-op@ubs.com

Attention: Elaine Chow / Kevin Bayett / Justin Churchill / Luke Goldsworthy / Jalpa

Schillig

## **Nominee**

Address: Level 16

Chifley Tower 2 Chifley Square Sydney NSW 2000

Australia

Fax number: +61-2-9324-2880

Email: <u>elaine.chow@ubs.com</u> / <u>kevin.bayett@ubs.com</u> / <u>sh-aus-eq-</u>

support@ubs.com / justin.churchill@ubs.com/luke.goldsworthy@ubs.com/

jalpa.schillig@ubs.com / sh-syd-rmp-op@ubs.com

Attention: Elaine Chow / Kevin Bayett / Justin Churchill / Luke Goldsworthy / Jalpa

Schillig

## **Participant**

Address: Level 16

Chifley Tower 2 Chifley Square Sydney NSW 2000

Australia

Fax number: +61-2-9324-2880

Email: <u>elaine.chow@ubs.com</u> / <u>kevin.bayett@ubs.com</u> / <u>sh-aus-eq-</u>

support@ubs.com / justin.churchill@ubs.com/luke.goldsworthy@ubs.com/

jalpa.schillig@ubs.com / sh-syd-rmp-op@ubs.com

Attention: Elaine Chow / Kevin Bayett / Justin Churchill / Luke Goldsworthy / Jalpa

Schillig

#### 11. PROCESS AGENT

(a) NPCP irrevocably appoints Gadens Lawyers Sydney Pty Ltd ACN 100 963 308 of Level 16, 77 Castlereagh Street, Sydney NSW 2000 as its agent under this deed for service of process in any proceedings in the State of Victoria.

- (b) If any person appointed as process agent is unable for any reason to act as agent for service of process, NPCP must immediately appoint another agent on terms acceptable to UBS and the Nominee. Failing this, UBS may appoint another person for this purpose.
- (c) **NPCP** undertakes to maintain the appointment of the process agent until all money owed by **NPCP** to UBS is paid to UBS, and agrees that any process served on that person is taken to be served on it.
- (d) This clause does not affect any other method of service allowed by law.

#### 12. **PARTICIPANT**

- (a) Each of NPCP and the Nominee notify the Participant that NPCP has granted or will grant a security interest in its present and future right, title and interest in, to, under and derived from all Shares transferred to, and held by, the Nominee as trustee for NPCP pursuant to this deed (**Security**).
- (b) Each of **NPCP** and the Nominee irrevocably and unconditionally:
  - (i) authorise any representative of UBS (the **Authorised Signatories**) severally to be the sole signatories authorised on behalf of NPCP and the Nominee to instruct the Participant from time to time in relation to the Shares (except as contemplated by clause 3.2);
  - (ii) (except as contemplated by clause 3.2) direct the Participant to only act on the instructions and directions of an Authorised Signatory in respect of the

Shares, including instructions and directions by which the Shares can be transferred or otherwise dealt with, and to only act in accordance with any such instructions and/or directions from the Authorised Signatories including the provision of any information that they request on the Participant's holding of the Shares (for the avoidance of doubt, any such instructions or directions may include the giving of communications by means of electronic messages or other electronic communications); and

- (iii) direct the Participant to pay or deliver (as appropriate):
  - (A) the proceeds of enforcement of the Security; or
  - (B) (subject to any contrary directions of UBS) any dividends, rights, payments, distributions, other amounts and other entitlements to which UBS is entitled under the terms of any Equity Derivative Financing Document in respect of any Shares,

into the following account (Nominated Bank Account):

Bank: To be advised by UBS to the other parties
Account name: To be advised by UBS to the other parties
SWIFT ID: To be advised by UBS to the other parties
Account no: To be advised by UBS to the other parties
Payment directions: To be advised by UBS to the other parties,

or as an Authorised Signatory may direct from time to time.

- (c) The Participant confirms that it:
  - accepts the terms, authorisations and directions contained in paragraph (b)
    and undertakes to act only in accordance with the instructions and directions
    of an Authorised Signatory (including where such instructions relate to the
    sending of electronic messages or other electronic communications) in
    relation to the Shares and will provide information on the Shares to an
    Authorised Signatory upon request;
  - (ii) will pay any proceeds of:
    - (A) enforcement of the Security; or
    - (B) (subject to any contrary directions of UBS) any dividends, rights, payments, distributions and other amounts to which UBS is entitled under the terms of any Equity Derivative Financing Document in respect of any Shares,

into the Nominated Account or as an Authorised Signatory may direct from time to time.

(d) Each of NPCP, the Nominee and UBS acknowledge that the Participant is entitled to assume the genuineness and authenticity of any instruction purported to be given by an Authorised Signatory and each of NPCP and the Nominee is deemed to have ratified and confirmed any such instruction. The Participant is not liable for any act or loss that results from the Participant having acted on an instruction given by a person it reasonably believes to be an Authorised Signatory for the purposes of this clause.

#### 13. **GENERAL**

## 13.1 **Stamp duty**

NPCP shall be responsible for the payment of all stamp duties and documentary taxes (including any interest or penalty that is payable) assessed by any revenue authority in Australia and/or Papua New Guinea upon the transfer of Trust Property to or by the Nominee, the constitution of the Trust, the execution of this deed or any other transactions contemplated by this deed.

## 13.2 **Telephone recording**

NPCP agrees to:

- (a) the tape recording by UBS or its agents of any telephone conversation with NPCP;
- (b) the retention of any tape recording so made; and
- (c) the use of any tape recording so made as evidence of the contents of the telephone conversation.

## 13.3 Amendments to deed

UBS, **NPCP** and the Nominee may amend this deed from time to time by supplemental deed signed by the parties.

#### 13.4 Waiver

No failure to exercise and no delay in exercising any power operates as a waiver. Nor does any single or partial exercise of any power preclude any other or further exercise of that power or any other power. The powers in this deed are in addition to and do not exclude or limit any right, power or remedy provided by law.

## 13.1 Waiver of immunity

Each party irrevocably waives, to the extent permitted by applicable law, with respect to itself and its revenues and assets (irrespective of their use or intended use), all immunity on the grounds of sovereignty or other similar grounds from:

- (a) suit;
- (b) jurisdiction of any court;
- (c) relief by way of injunction or order for specific performance or recovery of property;
- (d) attachment of its assets (whether before or after judgment); and
- (e) execution or enforcement of any judgment,

to which it or its revenues or assets might otherwise be entitled in any Proceedings in the courts of any jurisdiction and irrevocably agrees, to the extent permitted by applicable law, that it will not claim any such immunity in any Proceedings.

# 13.2 **Governing Law**

- (a) This deed shall be governed by the laws applying in Victoria.
- (b) The parties submit to the exclusive jurisdiction of the courts of Victoria.

(c)	Nothing in this clause 13.2 limits the right of UBS or the Nominee to institute proceedings arising out of or in connection with this deed or any Beneficial Interest against <b>NPCP</b> in any manner permitted by law.

**EXECUTED** as a deed.

[EXECUTION BLOCKS INTENTIONALLY OMITTED]

Holder of relevant interest	UBS AG, Australia Branch
Type of agreement	Prime Brokerage Agreement
Parties to agreement	(i) UBS AG, Australia Branch ("UBS AG")  (ii) Client (Please refer to Annexure D-1.)
Transfer date	Please refer to Annexure D-1.
Holder of voting rights	UBS AG
Are there any restrictions on voting rights?	Please refer to the details below.
If yes, detail	Since all right and title in the securities passes to the transferee (i.e. UBS AG), the transferee has the right to vote. However, in certain circumstances, the transferee may seek to arrange for instructions to be exercised in accordance with the instructions of the transferor (i.e. Client).
Scheduled return date (if any)	None, subject to the terms of the relevant loan.
Does the borrower have the right to return early?	Yes.
If yes, detail	The borrower (i.e. UBS AG) has the right to return at its discretion.
Does the lender have the right to recall early?	Yes.
If yes, detail	The lender (i.e. Client) can recall at its request subject to compliance with margin requirements and the terms of the relevant loan.
Will the securities be returned on settlement?	Yes.
If yes, detail any exceptions	None.

# Prime Brokerage Agreement - UBS AG, Australia Branch

Parties to agreement	Transfer date
Australian Leaders Fund Limited	(i) 12 September 2014
	(ii) 23 September 2014
	(iii) 3 November 2014
	(iv) 11 November 2014
	(v) 3 December 2014
	(vi) 4 December 2014
Perpetual Investment Management Limited as responsible entity of Perpetual's	(i) 12 September 2014
Wholesale Geared Australian Fund	(ii) 15 September 2014
	(iii) 16 September 2014
	(iv) 19 September 2014
	(v) 2 October 2014
	(vi) 8 October 2014
	(vii) 15 October 2014
	(viii) 7 November 2014
	(ix) 17 November 2014
	(x) 20 November 2014
	(xi) 25 November 2014
	(xii) 4 December 2014
	(xiii) 5 December 2014
	(xiv) 8 December 2014
Regal Funds Management Pty Limited as trustee for Regal Australian Long Short	(i) 12 September 2014
Geared Equity Fund	(ii) 16 September 2014
	(iii) 19 September 2014
	(iv) 9 October 2014
	(v) 15 October 2014
	(vi) 28 November 2014
	(vii) 16 December 2014
Regal Funds Management Pty Ltd as trustee and manager of Atlantic Absolute Return	(i) 12 September 2014
Fund	(ii) 3 December 2014
	(iii) 15 December 2014
	(iv) 16 December 2014

# Annexure E-1

Regal Funds Management Pty Limited as trustee and manager of Tasman Market	(i)	16 September 2014
Neutral Fund	(ii)	9 October 2014
	(iii)	15 October 2014
	(iv)	4 December 2014
	(v)	5 December 2014
	` ´	10 December 2014
		15 December 2014
Bennelong Funds Management Pty Ltd as trustee of Bennelong Long Short Equity Fund	(i)	15 September 2014
the state of the s	(ii)	16 September 2014
		15 October 2014
	(iii)	
	(iv)	19 November 2014
Watermark Market Neutral Fund Limited	(i)	12 September 2014
	(ii)	3 November 2014
	(iii)	3 December 2014
	(iv)	4 December 2014
Atrium Investment Management Pty Ltd in its capacity as trustee of the scheme Atrium	(i)	15 September 2014
Benn L-S Fund	(ii)	16 September 2014
	(iii)	28 October 2014
	(iv)	30 October 2014
Watermark Funds Management Pty Limited as responsible entity for Watermark Market	(i)	7 October 2014
Neutral Fund	(ii)	3 November 2014
	(iii)	3 December 2014
	(iv)	4 December 2014
Bennelong Funds Management Ltd as Trustee for Bennelong Alpha 200 Fund	(i)	14 October 2014
	(ii)	15 October 2014
	(iii)	30 October 2014
	\/	

	UBS Securities Australia Ltd	las	0	(1)	
H	UBS Securities Australia Ltd	Sell	102,125	(11,752)	Ordinary
09-Oct-14	UBS Securities Australia Ltd	Sell	11,088	(1,276)	Ordinary
-	UBS Securities Australia Ltd	Sell	11,627	(1,338)	Ordinary
- 1	UBS Securities Australia Ltd	Setl	37,315	(4,294)	Ordinary
	UBS Securities Australia Ltd	Seli	175,547	(20.201)	Ordinary
	UBS Securities Australia Ltd	les	650'8	(352)	Ordinary
	UBS Securities Australia Ltd	Sell	146,157	(618'91)	Ordinary
-	UBS Securities Australia Ltd	Sell	7,187	(827)	Ordinary
1	UBS Securities Australia Ltd	Sell	25,323	(2,914)	Ordinary
1	UBS Securities Australia Ltd	Sell	8,003	(921)	Ordinary
ŀ	UBS Securities Australia Ltd	Sell	2,607	(300)	Ordinary
-	UBS Securities Australia Ltd	Sell	6,587	(758)	Ordinary
-	UBS Securities Australia Ltd	Self	166,248	(601,61)	Ordinary
	UBS Securities Australia Ltd	Seth	29,484	(3,389)	Ordinary
	UBS Securities Australia Ltd	Setil	6,595	(758)	Ordinary
	UBS Securities Australia Ltd	Sell	922	(106)	Ordinary
	UBS Securities Australia Ltd	Sell	4,350	(200)	Ordinary
	UBS Securities Australia Ltd	Sell	6.421	(738)	Ordinary
39-Oct-14	UBS Securities Australia Ltd	Sell	3.480	(400)	Ordinary
	UBS Securities Australia Ltd	Buv	143 385	16 500	Ordinan
0-Oct-14	UBS Global Asset Management	Biv	1 246	000	Ordinary
0-Oct-14	18S Global Asset Management	Coll	2000 03	607	Ordinary
	TIPE Consister Australia Ind	liac liac	/60'80	(6,9,3)	Ordinary
T	Upp Securities Australia Ltd	Sell	84,100	(10,000)	Ordinary
T	UBS Securities Australia Ltd	Sell	188,112	(22,309)	Ordinary
	UBS Securities Australia Ltd	Buy	450,147	53,633	Ordinary
	UBS Securities Australia Ltd	Buy	23,571	2,796	Ordinary
	UBS Securities Australia Ltd	Sell	136.448	(16.228)	Ordinary
0-Oct-14	UBS Securities Australia Ltd	Buy	27.718	3 294	Ordinary
	UBS Securities Australia Ltd	8uv	23 183	2 750	Ordinan
	UBS Securities Australia Ltd	Seli	797 48	(10.068)	Ordinan
	UBS Securities Australia Ltd	Buy	292 025	24.670	Columny
	UBS Securities Australia Ltd	Sell	81 418	10,0,0	Ordinary
	UBS Securities Australia Ltd	Sell	420 740	(50,000)	Ordinan
	UBS Securities Australia Itel	193	405 114	(500,55)	Ciciliary
Γ	UBS AG Australia Branch	2.0	200,000	(166,14)	Ordinary
	IRS AG Australia Branch	in a	7 103 465	000 010	Ordinary
l	UBS AC Australia Branch	(ng	2,103,465	000,052	Ordinary
	USC Months Management Acceptance	Jac.	262,654	(67,021)	Ordinary
Ī	CDS Weekling and and an	eu,	838	001	Ordinary
10-041-14	Obsignation in an agement Australia Ltd	Buy	1,090	130	Ordinary
1	UBS wealth Management Australia Ltd	Buy	1,408	168	Ordinary
	UBS Wealth Management Australia Ltd	Buy	964	115	Ordinary
1	UBS Wealth Management Australia Ltd	Buy	838	100	Ordinary
Ī	UBS Wealth Management Australia Ltd	Buy	1,584	189	Ordinary
7	UBS Wealth Management Australia Ltd	Buy	863	103	Ordinary
٦	UBS Wealth Management Australia Ltd	Buy	3,177	379	Ordinary
	UBS Wealth Management Australia Ltd	Buy	302	108	Ordinary
	UBS Wealth Management Australia Ltd	Buy	1,098	131	Ordinary
	UBS Wealth Management Australia Ltd	Buy	1,525	182	Ordinary
	UBS Wealth Management Australia Ltd	Buy	545	92	Ordinary
	UBS Wealth Management Australia Ltd	Buy	1,609	192	Ordinary
	UBS Wealth Management Australia Ltd	Buy	22.370	2.669	Ordinary
	UBS Wealth Management Australia Ltd	Buy	1,375	164	Ordinary
	UBS Wealth Management Australia Ltd	Buy	964	115	Ordinary
	UBS Wealth Management Australia Ltd	Buy	13,201	1,575	Ordinary
-	UBS Wealth Management Australia Ltd	Buy	738	88	Ordinary
10-0ct-14	UBS Wealth Management Australia Ltd	Buy	1,785	213	Ordinary
	UBS Wealth Management Australia Ltd	Buy	3,336	398	Ordinary
ı	UBS Wealth Management Australia Ltd	Buy	4,249	205	Ordinary
Ī	UBS Wealth Management Australia Ltd	Buy	2,188	261	Ordinary
_	UBS Wealth Management Australia Ltd	Buy	2,255	569	Ordinary
	UBS Wealth Management Australia Ltd	Buy	1,182	141	Ordinary
10-Oct-14	UBS Wealth Management Australia Ltd	ANA.	302	36	Ordinary
Ī	UBS Wealth Management Australia Ltd	Buy	3.235	386	Ordinary
Ī	UBS Wealth Management Australia Ltd	Buv	1676	200	Ordinary
Ť	IRS Wealth Management Australia Ltd	200	020	2 .	Comment
Ť	JBS Wealth Management Australia Ltd	Briv	1 383	150	Ordinary
T	IRS Wealth Management Australia Ind	ind a	1,505	00.	Ordinary
Ť	202 Western Management Australia Ltd	Ang	1980	681	Ordinary
T	USS Wealth Management Australia Ltd	Buy	1,475	176	Ordinary
Ť	UBS vvearin Management Australia Ltd	Buy	2,112	252	Ordinary
	JBS Wealth Management Australia Ltd	Buy	981	117	Ordinan
ľ					5000
_	JBS Wealth Management Australia Ltd	Buv	1,106	132	Ordinary

	1985 Global Asset Management UBS Securities Australia Ltd Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	Sell	4,505 12,588 87,000	(518)	Ordinary
	urrities Australia Ltd Lurrities Australia Ltd Australia Ltd Lurrities Australia Ltd	Bin	12,588	1,452	Ordinary
	turities Australia Ltd unities Australia Ltd unities Australia Ltd		87 000		
	urities Australia Ltd	Sell	1 000'10	(10,000)	Ordinary
	urities Australia Ltd	Sell	485,338	(55,883)	Ordinary
		Buy	445,271	51,230	Ordinary
	urities Australia Ltd	Sell	16,191	(1,866)	Ordinary
	urrites Australia Ltd	Sell	886,535	(102,186)	Ordinary
	urnes Australia Ltd	Buy	009'269	80,389	Ordinary
	unities Australia Ltd	and	74,604	8,596	Ordinary
	unities Australia (+d	Dia.	580,0	(00)	Cidiliary
	urities Australia Ltd	ing in	320 BCC	(75, 257)	Ordinary
	urities Australia I td	Sell	750 152	(20,20)	Crainary
	urities Australia Itd	Sell	903,022	(501, VOL)	Ordinary
	urities Australia I td	Rin	2 190,232	752 182	Ordinary
	urities Australia I1d	la S	2 201 335	(253 572)	Ordinary
	urities Australia I th	9	782 837	(30,503)	Color
	urities Australia (10	وا	964 769	(111.251)	Ordinary
	urities Australia Ltd	Riv	281 102	22 363	ordinary.
	JBS AG. London Branch	Seil	750.156	(86 438)	Ordinary
	UBS AG, London Branch	Buv	931 107	86.438	Ordinary
	JBS AG, London Branch	Buv	1 100 864	102 186	Ordinary
	JBS AG, London Branch	Sell	886 535	(102 186)	Ordinary
	JBS AG Australia Branch	Buy	1,301,294	150.000	Ordinary
	Australia Branch	Sell	42,243	(4.865)	Ordinary
Ī	Australia Branch	Seil	48,115	(5.573)	Ordinary
	JBS AG Australia Branch	Sell	769,513	(88,749)	Ordinary
39-Oct-14 UBS AG	AG Australia Branch	Sell	830,343	(208,26)	Ordinary
09-Oct-14 UBS AG	JBS AG Australia Branch	Buy	1,002,928	115,455	Ordinary
39-Oct-14 UBS AG	UBS AG Australia Branch	Sell	975,873	(112,422)	Ordinary
09-Oct-14 UBS AG	UBS AG Australia Branch	Sell	12,378	(1,426)	Ordinary
	UBS Wealth Management Australia Ltd	, Ang	451	52	Ordinary
	UBS Wealth Management Australia Ltd	Sell	16,161	(1,864)	Ordinary
	UBS Wealth Management Australia Ltd	Sell	22,126	(2,552)	Ordinary
9-Oct-14 UBS Wea		Buy	1,987	229	Ordinary
	stralia Ltd	Buy	494	25	Ordinary
-Oct-14 UBS Secu	UBS Securities Australia Ltd	Sell	156,945	(18,050)	Ordinary
1		Sell	78,472	(9,025)	Ordinary
		Sell	34,397	(3,956)	Ordinary
1		Sell	26	(3)	Ordinary
١		Sell	661	(92)	Ordinary
1		Sell	17,632	(5,029)	Ordinary
		Sell	7,013	(807)	Ordinary
1		Sell	287	(33)	Ordinary
I		Sell	16,111	(1,854)	Ordinary
		Sell	6,370	(733)	Ordinary
		Sell	25,557	(2,941)	Ordinary
-Oct-14 UBS Secu		Sell	17,198	(1,979)	Ordinary
		Sell	3,956	(455)	Ordinary
	UBS Securities Australia Ltd	Sell	79,062	(860'6)	Ordinary
		Sell	76,568	(8,811)	Ordinary
١		Sell	86,422	(9,945)	Ordinary
09-Oct-14 UBS Secu		Sell	40,191	(4,625)	Ordinary
1		Sell	7,143	(822)	Ordinary
09-Oct-14 UBS Secu		Sell	23,254	(2,676)	Ordinary
-		Sell	7,812	(668)	Ordinary
1		Sell	9,220	(1,061)	Ordinary
ı		Sell	12,548	(1,444)	Ordinary
		Sell	1,469	(169)	Ordinary
		Sell	6	(ι)	Ordinary
	UBS Securities Australia Ltd	Sell	15,433	(1,776)	Ordinary
	UBS Securities Australia Ltd	Sell	292	(88)	Ordinary
	UBS Securities Australia Ltd	Sell	28	(10)	Ordinary
		Sell	2,555	(294)	Ordinary
Ī		Sell	930	(107)	Ordinary
Ī		Sell	2,607	(300)	Ordinary
09-Oct-14 UBS Secu		Sell	365	(42)	Ordinary
		Self	504	(58)	Ordinary
	UBS Securities Australia I td	les	5,214	(600)	Ordinary
l		-	1 564	(190)	O'T'
1			#00°'	(001)	Ordinary

13-0ct-14	UBS Securities Australia Ltd	35	287 983	(747 45)	Ordinan
13-0ct-14	UBS Securities Australia Ltd	Buy	364,080	43,683	Ordinary
13-Oct-14	UBS Securities Australia Ltd	Buy	1,874		Ordinary
13-00-14	UBS Securities Australia Ltd	ll Sell	378,416		
13-00-14	UBS Securities Australia Ltd	Buy	175,470	21,111	
13-00-14	UBS Securities Australia Ltd	Sell	176,239		
13-00-14	UBS Securities Australia Ltd	Sell	82,817	(10,000)	
13-001-14	Obs securities Australia Ltd	le R	85,101		
13-001-14	OBS Securities Australia (10	Buy	8,140		Ordinary
13-00-14	OBS Securities Australia Ltd	Seil	14,7/3	(1,781)	
13-001-14	OBS Securities Australia Ltd	Se.	48,0/1	(66,73)	Ordinary
13.00	Obsideration Australia Liu	pro)	122,410	/88'51	١
13-001-14	UBS Securities Australia Lto	Seil	16/,882	(20,190)	
13-00-14	UBS Securities Australia Ltd	Sei	355,411		
13-00-14	UBS AG, London Branch	Buy	466,264		
13-00-14	UBS AG, London Branch	Sell	378,416		
13-0ct-14	UBS AG Australia Branch	Buy	223,054	26,717	
13-Oct-14	UBS AG Australia Branch	Buy	65,671		
13-0ct-14	UBS AG Australia Branch	Buy	472,699		
13-Oct-14	UBS AG Australia Branch	Buy	1,605,532	192,308	Ordinary
13-Oct-14	UBS Wealth Management Australia Ltd	Buy	626	75	Ordinary
13-0ct-14	UBS Wealth Management Australia Ltd	Buy	8,776	1,051	Ordinary
13-00-14	UBS Wealth Management Australia Ltd	Buy	5,077	809	Ordinary
13-Oct-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	A/N	825.200	Options
:		Enter Into Derivative			
13-00-14	UBS Securities Australia Ltd	Contract	NA	825,200	Options
13-Oct-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	ΑN	825,200	Options
		Enter Into Derivative			
13-Oct-14	UBS Securities Australia Ltd	Contract	WA	825,199	Options
13-0ct-14	UBS Securities Australia Ltd	Contract	N/A	825.199	Options
		Enter Into Derivative			
13-Oct-14	UBS Securities Australia Ltd	Contract	WA	825,200	Options
13-Oct-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	ΑN	825.200	Options
		Enter Into Derivative			
13-0ct-14	UBS Securities Australia Ltd	Contract	NA	825,200	Options
13-Oct-14	UBS Securities Australia Ltd	Contract	N.	825,199	Options
		Enter Into Derivative			
13-Oct-14	UBS Securities Australia Ltd	Contract	NA	825,199	Options
13-Oct-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	NA NA	825,200	Options
		Enter Into Derivative			
13-04-14	UBS Securities Australia Ltd	Contract	N/A	825,200	Options
13-0a-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	A.N.	825.200	Options
13-04-14	I BS Securities Australia 12d	Enter Into Derivative	Š	975 100	1
	מיים מיים מיים מיים מיים מיים מיים מיים	Enter Into Derivative		661,030	Should
13-0ct-14	UBS Securities Australia Ltd	Contract	N/A	825,199	Options
13-0ct-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	Ž	825.200	Options
		Enter Into Derivative			
13-001-14	ODS Securities Australia Ltd	Conde	W.	825,200	Options
13-Oct-14	UBS Securities Australia Ltd	Contract	WA	825,200	Options
13-04-14	LBS Securities Australia I to	Enter into Derivative	ØN.	875 199	Constitution
		Enter Into Derivative		60,000	c condo
13-Oct-14	UBS Securities Australia Ltd	Contract	N/A	825,199	Options
13-Oct-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	A/A	825,200	Options
13.04.14	I IRS Securities Australia 14d	Enter Into Derivative	S.Z	OOL 368	C
		Enter Into Derivative		050,500	Spiral
13-Oct-14	UBS Securities Australia Ltd	Contract	N/A	825,200	Options
13-0ct-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	AN	825.199	Options
		Enter Into Derivative			
13-Oct-14	UBS Securities Australia Ltd	Contract	N/A	825,199	Ontions

1,442   1,442   1,442   1,442   1,475   1,47	agement Australia Ltd		1,542	184	Ordinary
Buy         4,005         479           Buy         4,005         479           Buy         1,475         1,475           Buy         1,475         1,475           Buy         1,475         1,475           Buy         1,475         1,475           Buy         1,475         1,628           Buy         1,475         1,638           Buy         1,445         1,475           Buy         1,475         1,475           Buy         1,475         1,475	Lating America	Buy	2000	100	0
BLY         4903         585           BLY         4903         585           BLY         1,475         174           BLY         1,488         174           BLY         1,488         174           BLY         1,488         174           BLY         1,488         173           BLY         1,381         1,58           BLY         1,381         1,68           BLY         1,381         1,68           BLY         1,381         1,68           BLY         1,381         1,64           BLY         1,381         1,64           BLY         1,381         1,68           BLY         1,381         1,68           BLY         1,382         1,68           BLY         1,415         1,68           BLY         1,415         1,68           BLY         1,415         1,68           BLY         1,427         1,68           BLY         1,48         1,72           BLY         1,48         1,72           BLY         1,48         1,72           BLY         1,48         1,72	agement Australia Ltd	Buy	4,015	479	Ordinary
BLY         175           BLY         1475         176           BLY         1689         173           BLY         1689         174           BLY         1689         174           BLY         1689         173           BLY         1689         173           BLY         1689         177           BLY         1689         177           BLY         1781         172           BLY         1781         172           BLY         1781         178           BLY         1781         178           BLY         1781         178           BLY         1782         300           BLY         1741         1783           BLY         1742         1784           BLY         1782         1	agement Australia Ltd	Buy	4,903	285	Ordinary
BAY         855         102           BAY         1,688         103           BAY         1,688         174           BAY         2,037         243           BAY         1,281         172           BAY         1,316         1,22           BAY         1,316         1,23           BAY         2,526         300           BAY         2,526         300           BAY         1,316         1,53           BAY         2,526         300           BAY         2,526         300           BAY         2,526         300           BAY         2,526         300           BAY         2,572         303           BAY         2,572         303           BAY         1,472         1,432	agement Australia Ltd	Buy	1,475	176	Ordinary
Buy         1,688         199           Buy         1,688         199           Buy         1,688         194           Buy         1,688         174           Buy         1,688         174           Buy         1,316         175           Buy         1,318         1,72           Buy         1,381         1,68           Buy         1,381         1,68           Buy         2,526         300           Buy         2,105         3,74           Buy         1,381         1,68           Buy         1,381         1,68           Buy         1,382         1,68           Buy         1,382         1,68           Buy         1,372         1,68           Buy         1,372         1,68           Buy         1,372         1,68           Buy         1,372         3,74           Buy         1,325         1,68           Buy         1,322         1,68           Buy         1,322         1,68           Buy         2,526         3,06           Buy         2,344         3,74	agement Australia Ltd	Buy	855	102	Ordinary
Buy         1,034         743           Buy         1,048         1,148           Buy         1,081         1,17           Buy         1,081         1,17           Buy         1,081         1,17           Buy         2,256         300           Buy         2,148         1,17           Buy         2,148         1,000           Buy         3,148         3,148           Buy         3,149         3,149           Buy         1,1372         1,638           Buy         1,1372         1,638           Buy         1,1372         1,638           Buy         1,256         300           Buy         1,256         300           Buy         1,247         1,638           Buy         1,372         1,638           Buy         1,372         1,638           Buy         1,372         300           Buy         1,372         300           Buy         2,526         300           Buy         2,528         600           Buy         2,528         600           Buy         2,528         600	agement Australia Ltd	Buy	1,668	199	Ordinary
Buy         1,73         1,74           Buy         1,73         1,74           Buy         1,316         1,73           Buy         1,316         1,53           Buy         1,316         1,53           Buy         2,526         300           Buy         2,525         300           Buy         2,922         3,53           Buy         1,415         1,63           Buy         1,2421         1,537           Buy         1,2421         1,537           Buy         1,2421         1,415           Buy         1,326         3,06           Buy         1,324         3,24           Buy         1,326         3,06           Buy         2,586         3,06           Buy         2,327,60         3,00           Buy         2,327,60         3,00           Buy         2,327,60         3,00	agement Australia Ltd	Buy	2,03/	243	Ordinary
Buy         478         17           Buy         478         17           Buy         1316         159           Buy         1316         159           Buy         1316         159           Buy         2,526         300           Buy         2,105         374           Buy         2,105         378           Buy         2,105         378           Buy         1,1381         1,189           Buy         2,105         378           Buy         1,242         378           Buy         1,342         374           Buy         1,342         1,452           Buy         1,342         374           Buy         1,342         374           Buy         1,327         374           Buy         2,342         3,04           Buy         1,342         1,452           Buy         1,452         1,452           Buy <td>agement Australia Ltd</td> <td>euy Pr.::</td> <td>1,438</td> <td>1,1</td> <td>Ordinary</td>	agement Australia Ltd	euy Pr.::	1,438	1,1	Ordinary
Buy         1,081         1,28           Buy         1,081         1,29           Buy         3,621         400           Buy         3,621         400           Buy         2,256         300           Buy         2,143         154           Buy         2,143         154           Buy         3,143         3,14           Buy         3,143         3,14           Buy         1,145         1,688           Buy         1,145         1,688           Buy         1,372         42         2,24           Buy         1,372         42         2,24           Buy         1,374         1,472         400           Buy         1,374         1,472         400           Buy         2,586         300         800           Buy         2,523         600           Buy         2,523         600<	agement Australia Ltd	ng a	196	2	Ordinary
early         1,316         1,57           Buy         1,316         1,57           Buy         2,526         300           Buy         2,572         300           Buy         2,972         353           Buy         2,972         353           Buy         2,972         353           Buy         2,972         353           Buy         1,475         1,68           Buy         1,472         1,587           Buy         1,472         1,472           Buy         2,525         3,00           Buy         2,526         3,00     <	agenterit Australia Ltd	ina and	1,001	00,	Ordinary
Buy         3.621         430           Buy         3.72         3.04           Buy         1.381         1.48           Buy         2.025         3.00           Buy         2.02         2.50           Buy         2.02         2.03           Buy         2.02         2.03           Buy         1.93         1.189           Buy         1.0442         1.59           Buy         1.372         1.59           Buy         1.3447         1.69           Buy         1.377         1.69           Buy         1.377         1.69           Buy         1.377         1.69           Buy         1.377         2.244           Buy         1.366         3.06           Buy         1.387         3.04           Buy         2.586         3.06           Buy         2.387         6.00           Buy         2.387         6.00 <t< td=""><td>acement Australia I td</td><td>Buy</td><td>1316</td><td>157</td><td>Ordinary</td></t<>	acement Australia I td	Buy	1316	157	Ordinary
947         2.526         300           8by         2.526         300           8by         3.143         164           8by         3.145         3.53           8by         2.105         2.50           8by         2.105         2.50           8by         1.004         1.638           8by         1.004         1.638           8by         1.372         1.638           8by         1.372         1.638           8by         1.372         1.638           8by         1.376         1.672           8by         2.526         300           8by         2.523         66           8by         2.523         600           8by         2.523         300           8by         2.523         300           8by         2.523         300	stralia Itd	Bis	169.5	057	Ordinary
Buy         1,381         164           Buy         2,972         35.4           Buy         2,972         35.4           Buy         2,972         35.4           Buy         2,972         35.4           Buy         2,972         35.5           Buy         1,415         1,68           Buy         1,372         1,63           Buy         1,372         1,52           Buy         1,374         1,52           Buy         1,372         1,63           Buy         1,344         1,52           Buy         1,344         1,52           Buy         1,342         1,45           Buy         1,344         1,45           Buy         1,342         1,45           Buy         1,342         1,45           Buy         2,52         300           Buy         2,52         300	stralia 11d	B <sub>10</sub>	2 526	000	Ordinary
Buy         3,143         374           Buy         2,105         353           Buy         2,105         353           Buy         2,105         2,52           Buy         2,105         2,53           Buy         1,415         1,63           Buy         1,045         1,193           Buy         1,045         1,63           Buy         1,347         1,475           Buy         1,347         1,475           Buy         1,347         1,475           Buy         1,347         1,475           Buy         1,326         30           Buy         1,326         30           Buy         2,526         30           Buy         2,526         30           Buy         2,526         30           Buy         2,528         6,00           Buy         2,136         3,14	certain Ltd	Busy	1301	164	Coding
649         2,072         35.3           849         2,972         35.3           849         2,972         35.3           849         1,415         1,688           849         1,3782         1,638           849         1,3782         1,587           849         1,3782         1,587           849         1,347         1,587           849         1,347         2,274           849         1,347         2,274           849         1,347         2,274           849         1,347         2,274           840         1,347         2,274           840         1,347         2,274           840         1,347         3,44         3,44           840         1,347         3,44         3,44           840         1,347         3,44         3,24           840         1,347         3,44         3,44           840         1,347         3,44         3,44           840         1,347         3,44         3,44           840         2,357,600         26,000           840         2,357,600         26,000	and and and	000	000	100	Ordinary
649         2,105         250           849         8,40         1,000           849         1,103         1,103           849         1,1005         1,103           849         1,1005         1,103           849         1,1005         1,103           849         1,372         1,587           849         1,3447         1,587           849         1,347         1,587           849         1,347         1,587           840         1,347         1,587           840         1,347         1,587           840         1,347         1,587           840         1,347         1,587           840         1,347         1,587           840         1,347         1,025           840         1,347         1,025           840         1,347         1,025           840         1,348         3,148           840         2,587         6,000           840         2,587         6,000           840         2,587         6,000           840         2,533         3,04           840         2,533         3	Strain Ltd	ana b	2,142	250	Codinary
649         6,102         1,000           649         6,102         1,000           849         13,722         1,638           849         13,722         1,638           849         13,722         1,638           849         13,722         1,638           849         13,722         1,638           849         78,685         9,447         1,527           849         78,685         9,345         80           849         78,685         9,345         80           849         78,685         9,345         80           840         13,747         2,74         80           844         78,685         9,345         80           849         7,528         80         80           840         2,526         3,06         80           844         2,526         3,00         80           844         2,527         3,00         80           844         2,527         3,00         80           844         2,357,600         2,50         3,00           845         3,45         3,44         3,44         3,44           846	on all all	no d	276,2	222	Cidilialy
64y         1,415         1,68           84y         1,415         1,68           84y         1,342         1,69           84y         13,792         1,587           84y         13,792         1,587           84y         13,742         1,587           84y         13,742         1,587           84y         13,747         1,587           84y         1,347         1,587           84y         1,347         1,587           84y         1,347         1,482           84y         1,347         1,482           84y         1,346         3,066           84y         1,356         3,066           84y         1,356         3,066           84y         2,556         300           84y         2,357,600         2,000           84y         2,357,600         2,000           84y         2,357,600         3,04           84y         2,357,600         3,04           84y         2,103         3,04           84y         2,103         3,04           84y         2,103         3,04           84y         2,133	יייייייייייייייייייייייייייייייייייייי	ana Ora:	2,102	2000	Codinary
Buy         1,0,441         1,0,442         1,193           Buy         1,0,442         1,193         1,638           Buy         1,3,422         1,538         1,638           Buy         1,3,422         1,538         1,4752           Buy         1,3,472         1,638         1,432           Buy         1,3,472         1,638         1,432           Buy         1,3,472         2,274         8,25           Buy         1,3,472         8,623         1,432           Buy         1,2,52         3,00         8,21         4,00           Buy         2,524         3,00         8,00         8,00         8,00           Buy         2,526         3,00         8,00         8,00         8,00         8,00           Buy         2,526         3,00         8,00         <	Istralia Ltd	áng	024,8	000'-	o dilliary
Buy         1,1782         1,158           Buy         1,3782         1,158           Buy         1,3782         1,587           Buy         78,665         9,342           Buy         78,665         9,342           Buy         78,665         9,342           Buy         1,347         1,572           Buy         1,347         2,214           Buy         2,877         6,02           Buy         2,587         6,02           Buy         2,587         306           Buy         2,587         306           Buy         2,587         306           Buy         2,587         300           Buy         2,587         300           Buy         2,587         300           Buy         2,103         374           Buy         2,357,600         2,500,000           Buy         2,357,600         2,500,000           Buy         2,357,600         2,500           Buy         2,357,600         2,500           Buy         2,357,600         2,500           Buy         2,357,600         2,500           Buy         2,	ואון פווים דונים	, and	1,400	00 5	Croinaly
Buy         1,1,2,2         1,5,9           Buy         13,447         1,59           Buy         13,447         1,59           Buy         13,447         1,52           Buy         12,272         1,40           Buy         12,272         1,40           Buy         13,47         2,74           Buy         1,36         3,62           Buy         2,52         3,06           Buy         3,143         3,14           Buy         3,143         3,14           Buy         2,52         3,00           Buy         2,52         3,00           Buy         2,52         3,00           Buy         2,52         3,00           Buy         2,53         6,80           Buy         2,53         6,80           Buy         2,53         6,80           Buy         2,53         3,00           Buy         2,53         3,00	Jstralia Ltd	Ruy	10,045	E81,1	Crainary
Buy         78,685         9,485           Buy         17,447         1,542           Buy         17,447         1,542           Buy         173,776         8,762           Buy         13,776         8,762           Buy         13,776         8,762           Buy         1,356         161           Buy         2,526         300           Buy         2,527         300           Buy         3,445         374           Buy         2,526         300           Buy         2,526         300           Buy         3,445         374           Buy         3,455         374           Buy         2,523         6,00           Buy         2,103         300           Buy         2,37,60         2,50           Buy         2,37,60         2,50           Buy         3,445         3,74           Buy         2,37         2,28	Jstralia Ltd	Ruy	13,/92	859,1	Ordinary
Buy         17,422         19,42           Buy         12,421         14,422           Buy         13,776         8762           Buy         19,177         22,44           Buy         19,177         22,44           Buy         19,177         2,24           Buy         1,356         30.66           Buy         1,356         30.66           Buy         2,526         30.0           Buy         2,526         30.0           Buy         2,149         374           Buy         2,149         374           Buy         2,355,600         280,00           Buy         2,357,600         280,00           Buy         2,375,600         280,00           Buy         2,375,600         280,00           Buy         2,375,600         280,00           Buy         2,375,600         2,000           Buy         2,375         6,00           Buy         2,375         6,00           Buy         2,375         6,00           Buy         2,373         6,00           Buy         2,233         300           Buy         2,	stralia Ltd	Buy	13,447	1,597	Ordinary
Buy         14,722         14,722           Buy         19,147         2,724           Buy         19,147         2,724           Buy         19,147         2,724           Buy         19,147         2,724           Buy         1,1356         161           Buy         2,526         300           Buy         2,528         5,000           Buy         2,103         3,148         3,14           Buy         2,237         5,00           Buy         2,133         1,138           Buy         2,237         5,00           Buy         2,238         6,00           Buy         2,223         5,00           Buy         2,238         5,00           Buy         2,238         <	ıstralıa Ltd	Buy	589'8/	9,345	Ordinary
Buy   7.376   8.762     Buy   19.147   2.244     Buy   3.621   4.20     Buy   2.526   3.066     Buy   2.526   3.066     Buy   2.527,600   2.60.000     Buy   2.357,600   2.50.000     Buy   2.357,600   2.234     Buy   2.357,600   2.20.000     Buy   2.358   3.74     Buy   2.358   3.75     Buy   2.358   3.75     Buy   2.358   3.75     Buy   2.358   3.75     Buy   2.358     Buy   2.358   3.75     Buy   2.358     Buy	ustralia Ltd	Buy	124,212	14,752	Ordinary
Buy   19,147   2,274   Buy   Buy   19,147   2,274   Buy   Buy   5,827   620   Buy   2,526   300   Buy   2,523   300   Buy   2,535   300   Buy	ustralia Ltd	Buy	73,776	8,762	Ordinary
8y         362         430           8y         5,872         642           8uy         5,872         662           8uy         2,587         662           8uy         2,587         3,066           8uy         2,587         3,066           8uy         2,587         3,046           8uy         3,148         3,74           8uy         3,148         3,74           8uy         3,148         3,74           8uy         3,149         3,74           8uy         3,148         3,74           8uy         3,148         3,74           8uy         3,148         3,74           8uy         3,145         3,74           8uy         2,103         2,50,000           8uy         2,357,600         2,50,000           8uy         2,103         3,14           8uy         2,357,600         2,50           8uy         2,103         3,14           8uy         2,357,600         2,50           8uy         2,357,600         2,50           8uy         2,35         6,00           8uy         2,35         6,00 <td>ıstralia Ltd</td> <td>Buy</td> <td>19,147</td> <td>2,274</td> <td>Ordinary</td>	ıstralia Ltd	Buy	19,147	2,274	Ordinary
Buy         587         692           Buy         1,356         666           Buy         25,816         3,066           Buy         25,816         3,066           Buy         25,26         300           Buy         2,25         300           Buy         733         874           Buy         3,149         374           Buy         3,145         374           Buy         3,145         374           Buy         2,535         300           Buy         2,532         300           Buy         2,533         300           Buy         2,533         300           Buy         4,155         494           Buy         4,155         494           Buy         6,333         1,26           Buy </td <td>ustralia Ltd</td> <td>Buy</td> <td>3,621</td> <td>430</td> <td>Ordinary</td>	ustralia Ltd	Buy	3,621	430	Ordinary
Buy   25,816   9.06     Buy   2,528   3.00     Buy   8,631   1,025     Buy   8,631   1,025     Buy   8,631   1,025     Buy   2,357,600   2,50,000     Buy   3,364   4,135   2,26     Buy   3,364   4,135   2,26     Buy   6,335   7,28     Buy   6,335   7,29     Buy   6,335   7,20     Buy   6,335   7,20     Buy   7,722   7,20     Buy	ustralia Ltd	Buy	5,827	269	Ordinary
Buy         25.26         3.066           Buy         25.26         3.0           Buy         8.41         1.025           Buy         3.149         37.4           Buy         3.149         37.4           Buy         3.149         37.4           Buy         2.357,600         28.0           Buy         2.357,600         28.0           Buy         2.357,600         28.0           Buy         2.357,600         28.0           Buy         2.103         2.50           Buy         2.103         2.50           Buy         2.103         2.0           Buy         2.103         2.0           Buy         2.103         2.0           Buy         2.133         2.0           Buy         2.133         2.0           Buy         2.23         3.0           Buy         4.155         2.2           Buy         4.155         2.2           Buy         4.155         2.2           Buy         4.155         2.2           Buy         2.28         2.0           Buy         2.28         2.0	stralia Ltd	Buy	1,356	161	Ordinary
Buy         2,556         300           Buy         8,631         1,025           Buy         3,149         374           Buy         3,149         374           Buy         3,149         374           Buy         3,149         374           Buy         2,105         250           Buy         2,105         200           Buy         3,149         374           Buy         2,103         200           Buy         3,146         3,74           Buy         2,210         200           Buy         4,205         5,000           Buy         4,205         5,000           Buy         4,235         5,00           Buy         4,235         5,00           Buy         4,235         5,00           Buy         4,155         7,28           Buy         1,131         1,256           Buy         1,131         1,256           Buy         1,131         1,256           Buy         1,231         1,256           Buy         1,244         3,74           Buy         1,264         3,74 <tr< td=""><td>stralia Ltd</td><td>Buy</td><td>25,816</td><td>3,066</td><td>Ordinary</td></tr<>	stralia Ltd	Buy	25,816	3,066	Ordinary
Buy         8.631         1.05           Buy         8.631         1.025           Buy         3.149         3.74           Buy         3.149         3.74           Buy         3.149         3.74           Buy         3.149         3.74           Buy         2.05         600           Buy         2.355,600         280,000           Buy         10,069         1.138           Buy         4,2,025         5,000           Buy         2,103         2,50           Buy         2,103         2,03           Buy         2,103         3.00           Buy         3,344         40           Buy         3,344         40           Buy         4,135         2,23           Buy         4,155         2,23           Buy         2,235         300           Buy         1,272         2,23 <td>ustralia Ltd</td> <td>Buy</td> <td>2,526</td> <td>300</td> <td>Ordinary</td>	ustralia Ltd	Buy	2,526	300	Ordinary
Buy         8,631         1,025           Buy         7,149         374           Buy         7,33         874           Buy         3,149         374           Buy         3,149         374           Buy         3,149         374           Buy         2,105         250           Buy         3,145         3,145           Buy         3,145         3,24           Buy         2,535,600         280,000           Buy         3,145         3,145           Buy         4,265         5,000           Buy         2,533         300           Buy         2,533         300           Buy         3,364         4,00           Buy         4,155         2,23           Buy         3,364         4,00           Buy         4,155         2,23           Buy         4,155         2,23           Buy         6,337         7,28           Buy         4,155         2,23           Buy         6,335         1,29           Buy         6,533         1,26           Buy         1,75         2,23	stralia Ltd	Buy	42	2	Ordinary
Buy   3,149   374   87	stralia Ltd	Buy	8,631	1.025	Ordinary
Buy         535         67           Buy         5148         574           Buy         3,148         374           Buy         2,357,600         280,000           Buy         2,357,600         280,000           Buy         2,533         5,000           Buy         2,523         5,000           Buy         2,523         300           Buy         2,533         6,806           Buy         2,533         300           Buy         2,533         300           Buy         2,533         300           Buy         2,533         300           Buy         3,364         1,138           Buy         4,153         4,00           Buy         4,153         4,00           Buy         4,155         4,00           Buy         4,155         2,00           Buy         4,155         2,00           Buy         6,335         7,73           Buy         1,732         2,23           Buy         1,732         2,23           Buy         2,236         3,00           Buy         2,286         3,17 <td>stralia Ltd</td> <td>58</td> <td>3.149</td> <td>374</td> <td>Ordinary</td>	stralia Ltd	58	3.149	374	Ordinary
Buy   2,555   566     Buy   2,105   250     Buy   2,3145   374     Buy   2,3145   374     Buy   10,069   1,198     Buy   42,025   5,000     Buy   42,025   5,000     Buy   2,323   300     Buy   2,323   300     Buy   2,323   300     Buy   3,364   346     Buy   3,364   404     Buy   3,364   404     Buy   6,335   7,28     Buy   6,335   7,29     Buy   7,722   7,28     Buy   7,722   7,727     Buy	stralia Ltd	Buy	733	87	Ordinary
Buy         3.149         3.74           Buy         2,105         2.50           Buy         2,355,600         280,000           Buy         1,345         1,344           Buy         1,006-6         1,134           Buy         1,007-6         1,17           Buy         2,103         2,50           Buy         2,133         1,26           Buy         1,341         1,26           Buy         4,155         2,23           Buy         4,155         2,23           Buy         4,155         2,23           Buy         4,155         2,23           Buy         1,732         2,23           Buy         1,732         2,23           Buy         1,732         2,23           Buy         2,235         3,74           Buy         2,235         3,74           Buy         2,235         2,72	stralia Ltd	Buy	525	99	Ordinary
Buy         2,105         250           Buy         2,357,600         2,80,000           Buy         3,145         374           Buy         1,099         1,198           Buy         2,035         5,000           Buy         2,103         250           Buy         2,728         6,806           Buy         2,728         6,806           Buy         3,364         1,113           Buy         3,364         1,113           Buy         1,134         400           Buy         1,256         400           Buy         1,231         1,256           Buy         1,331         1,256           Buy         1,331         1,256           Buy         1,331         1,256           Buy         1,532         200           Buy         6,335         758           Buy         6,335         778           Buy         6,335         778           Buy         6,335         778           Buy         1,732         200           Buy         6,335         778           Buy         1,732         200	stralia Ltd	Buy	3,149	374	Ordinary
Buy         2,357,600         280,000           Buy         31,45         324           Buy         13,46         1,138           Buy         42,025         5,000           Buy         2,103         250           Buy         2,133         200           Buy         6,375         5,000           Buy         6,375         7,88           Buy         6,375         7,88           Buy         1,131         1,226           Buy         1,236         404           Buy         1,237         7,88           Buy         1,237         7,88           Buy         1,237         2,23           Buy         1,237         2,23           Buy         1,236         404           Buy         1,237         7,28           Buy         1,732         2,23           Buy         1,732         2,23           Buy         1,732         2,23           Buy         1,732         2,28           Buy         1,732         2,28           Buy         1,732         2,23           Buy         2,238         2,700 </td <td>stralia Ltd</td> <td>Buy</td> <td>2,105</td> <td>250</td> <td>Ordinary</td>	stralia Ltd	Buy	2,105	250	Ordinary
Buy   1,000   1,198	stralia Ltd	Buy	2.357,600	280,000	Ordinary
Buy   10,099   1,198	stralia Ltd	Buy	3,145	374	Ordinary
8uy         42,025         5,000           8uy         2,103         2,500           8uy         2,103         2,500           8uy         2,103         2,500           8uy         2,103         2,50           8uy         2,233         6,806           8uy         6,375         7,88           8uy         3,364         40           8uy         3,364         40           8uy         4,155         2,23           8uy         1,272         2,23           8uy         1,272         2,23           8uy         1,732         2,00           8uy         1,732         2,00           8uy         2,335         7,28           8uy         2,33         7,23           8uy         2,34         3,14           8uy         2,34         3,14           8uy         2,33         7,23           8uy         2,33         7,27 <tr< td=""><td>stralia Ltd</td><td>Buy</td><td>10,069</td><td>1,198</td><td>Ordinary</td></tr<>	stralia Ltd	Buy	10,069	1,198	Ordinary
Buy         2,005         5,000           Buy         2,523         300           Buy         2,523         300           Buy         2,523         300           Buy         6,325         5,866           Buy         45,4         5,41           Buy         10,311         1,133           Buy         10,311         1,256           Buy         1,131         1,256           Buy         2,523         300           Buy         2,523         300           Buy         2,523         300           Buy         6,337         7,28           Buy         6,337         7,88           Buy         6,335         7,78           Buy         1,732         206           Buy         1,732         206           Buy         1,732         206           Buy         1,732         206           Buy         1,732         258           Buy         1,732         206           Buy         1,732         206           Buy         1,732         206           Buy         1,732         200	stralia Ltd	Buy	85	7	Ordinary
Buy   2,103   250     Buy   2,523   300     Buy   6,375   5,84     Buy   6,375   5,84     Buy   1,0311   1,256     Buy   2,523   300     Buy   6,335   7,28     Buy   6,335   7,28     Buy   6,335   7,28     Buy   6,335   7,29     Buy   6,335   7,20     Buy   6,333   7,20     Buy   1,72   206     Buy   2,095   250     Buy   1,529   21,070     Buy   1,56   21,070     Buy   1,1,001	stralia Ltd	Buy	42,025	9,000	Ordinary
8uy         2,523         300           8uy         6,375         6,806           8uy         6,375         6,806           8uy         6,375         7,88           8uy         9,380         1,113           8uy         3,364         400           8uy         3,364         400           8uy         2,523         300           8uy         2,523         300           8uy         1,875         2,53           8uy         1,875         7,88           8uy         6,375         7,78           8uy         1,732         2,64           8uy         1,732         2,64           8uy         1,732         2,78           8uy         1,732         2,64           8uy         1,732         2,88           8uy         1,732         2,73           8uy         1,75,935         2,100           8uy         1,750         1,1,76	stralia I td	Blav	2 103	250	Ordinary
Buy   57,238   6,806   Buy   6,375   7,88   Buy   6,375   7,88   Buy   6,375   7,88   Buy   6,375   7,88   Buy   6,375   7,134   1,134   Buy   7,134   1,256   Buy   7,136   Buy   8,136   B	stralja Ltd	8nv	2 523	300	Ordinary
Buy   6,375   786   84   84   84   84   84   84   84	stralia Itd	Buy	57 738	908.9	Ordinary
Buy   45.4   7.4     Buy   45.4   1.13     Buy   10.311   1.256     Buy   10.311   1.256     Buy   2.523   300     Buy   6.335   7.58     Buy   6.335   7.58     Buy   7.732   2.06     Buy   7.733   7.753     Buy   7.733   7.753     Buy   7.733   7.753     Buy   7.7535   7.1070	pri milato	in a	375.3	0000	ordina)
Buy   9,360   1,133   1,126   1,134   1,256   1,134   1,256   1,134   1,256   1,134   1,256   1,134   1,256   1,134   1,256   1,134   1,256   1,134   1,256   1,134   1,1256   1,134	Strain Ltd	b.u.	676,0	00.7	Oldinary
Buy   10,311   1,226     Buy   10,311   1,226     Buy   10,311   1,226     Buy   2,523   300     Buy   2,523   300     Buy   1,875   2,948     Buy   6,375   7,38     Buy   6,375   7,78     Buy   7,78     Buy   7,78   7,78     Buy   7,78	SSILAINA LIG	eny	454	7 .	Ordinary
Buy   1,256     Buy   1,031   1,256     Buy   1,031   1,256     Buy   2,523   300     Buy   4,155   494     Buy   6,375   7,28     Buy   6,375   7,28     Buy   6,375   7,28     Buy   7,78   7,78     Buy   7,78     Buy   7,78   7,78	Istralia Ltd	eny.	095'6	1,113	Ordinary
Buy   10,314   4,000     Buy   10,314   1,226     Buy   4,155   2.23   300     Buy   4,155   2.24     Buy   1,875   2.24     Buy   1,875   2.24     Buy   1,875   2.24     Buy   1,722   2.26     Buy   2,286   2.77     Buy   6,335   7,78     Buy   6,335   7,78     Buy   1,722   2.06     Buy   2,286   2,77     Buy   1,722   2.06     Buy   1,723   2.05     Buy   1,723   2.00     Buy   1,75,935   2,1,070     Buy   1,75,935   2,1,070     Buy   8,5,000     Buy   8,5,000     Buy   8,5,000     Buy   8,5,000     Buy   8,5,000     Buy   8,7,000     Buy	Istralia Ltd	an)	10,311	977'	Ordinary
Buy   1,213   1,229     Buy   1,431   1,229     Buy   1,435   264     Buy   1,732   7.58     Buy   1,732   7.58     Buy   1,732   7.78     Buy   1,732   7.78     Buy   1,732   7.78     Buy   1,732   7.78     Buy   2095500   25,000     Buy   1,75,595   21,070     Buy   1,75,595   21,070     Buy   1,75,595   21,070     Buy   1,75,995   21,070     Buy   1,75,995   1,176     Buy   1,76   1,176     Buy   1,176   1,176	Istralia Ltd	Buy	3,364	400	Ordinary
Buy   4,523 300	istralia Ltd	Buy	10,311	977'	Ordinary
Buy   1,155   294     Buy   1,875   228     Buy   6,375   7.28     Buy   6,375   7.28     Buy   6,375   7.28     Buy   7,238   7.78     Buy   7,238   7.73     Buy   7,248   7.77     Buy   7,595   7,100     Buy   7,595   7,100     Buy   8,35,000   10,000     Buy   8,35,000   10,00	Istralia Ltd	Buy	2,523	300	Ordinary
Buy   6,375   728     Buy   6,375   728     Buy   6,375   778     Buy   1,732   206     Buy   3,145   377   378     Buy   3,145   377   378     Buy   6,335   775     Buy   209,500   25,000     Buy   175,935   21,070     Buy   175,935	stralia Lto	Buy	4,155	\$4	Ordinary
Buy   15,4375   758     Buy   16,643   1,979     Buy   16,643   1,979     Buy   2,288   2,272     Buy   2,188   2,72     Buy   3,145   3,74     Buy   1,488   1,73     Buy   1,488   1,73     Buy   1,595   21,070     Buy   175,935   21,070     Buy   175,935   21,070     Buy   175,935   21,070     Buy   1,75,935   21,070     Buy   1,75,935   21,070     Buy   1,75,935   1,070     Buy   1,75,935   1	istralia Ltd	Buy	5/8/1	223	Ordinary
Buy   1,574   1,979     Buy   1,732   2.06     Buy   1,732   2.06     Buy   1,732   2.06     Buy   3,145   3.74     Buy   1,75,935   7,73     Buy   1,75,935   21,070     Buy   1,75,935   21,070     Buy   1,75,935   21,070     Buy   1,75,935   1,75,000     Buy   1,75,100     Bu	istralia Ltd	Buy	6,375	758	Ordinary
Buy   1,732   206     Buy   1,732   206     Buy   3,145   374     Buy   6,333   753     Buy   175,935   21,070     Buy   175,935   21,070     Buy   175,935   21,070     Buy   8,34   17,06	stralia Ltd	Buy	16,643	1,979	Ordinary
Buy   1/722   2/06     Buy   2/288   2/27     Buy   3/145   3/24     Buy   3/145   3/24     Buy   1/438   1/53     Buy   1/75,935   21/070     Buy   895,000   100,000	istralia Ltd	Buy	6,375	758	Ordinary
Buy   2.288   2.789   2.484	straka Ltd	Buy	1,732	506	Ordinary
Buy   5,145   3,145	istralia Ltd	Buy	2,288	272	Ordinary
Buy   1,438   753   754   754   755   75	stralia Ltd	Buy	3,145	3/4	Ordinary
Buy   1/438   1771   1781   1781   1781   1781   1781   1781   1881	stralia Ltd	Buy	6,333	753	Ordinary
Buy   2095/00   25,000     Buy   175,935   21,070     Buy   835,000   100,000     Buy   97,766   11,761     Call   14,000   100,000     Call   15,000   100,000     Call	istralia Ltd	Buy	1,438	171	Ordinary
Buy   175,935 21,070     Buy   813,000   100,000     Buy   97,86   11,761     Buy   97,86   11,761     Con   14,000	stralia Ltd	Buy	209,500	25,000	Ordinary
Buy   835,000   100,000   1	istralia Ltd	8uy	175,935	21,070	Ordinary
8uy 8 1,716 8uy 97,766 11,761 8uy 74,003 60,0304	stralia Ltd	Buy	835,000	100,000	Ordinary
8uy 97,766 11,761	istralia Ltd	Buy	8	-	Ordinary
(100)	Management	Birv	992 / 6	11 761	Ordinary
The state of the s	de alle de	000	24,000	(150.0)	) deline
(8,931) (8,931)	stralia Ltd	Seli	/4,003	(186,831)	Ordinary
UBS Securities Au		tratala i tot stratala i tot		Buy   Buy	649         1,415           849         1,415           849         1,3045           849         13,792           849         13,792           849         13,747           849         13,477           849         13,477           849         19,477           849         19,477           849         1,356           849         2,526           849         2,526           849         2,526           849         2,526           849         2,526           840         3,149           840         3,149           840         3,149           840         2,526           840         2,526           840         2,526           840         3,149           840         2,526           840         2,527           840         2,523           840         2,523           840         2,523           840         2,523           840         2,523           840         2,523           840         2,523 <th< td=""></th<>

		Forter Into Derivative			
13-Oct-14	UBS Securities Australia Ltd	Contract	NA	825,200	Options
13-Oct-14	UBS Securities Australia Ltd	Enter into Derivative Contract	N/A	825,200	Options
13-Oct-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	N/A	825,200	Options
13-Oct-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	ΑN	825.199	Options
13-Oct-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	4 X	825.199	Options
13-Oct-14	UBS Securities Australia Ltd	Sell	6,951	(839)	Ordinary
13-Oct-14	UBS Securities Australia Ltd	Sell	5,510	(599)	Ordinary
13-Oct-14	UBS Securities Australia Ltd	Sell	1,320	(129)	Ordinary
13-Oct-14	UBS Securities Australia Ltd	lles :	10,300	(1,241)	Ordinary
3-Oct-14	UBS Securities Australia Ltd	¥ 3	15,637	(1,884)	Ordinary
3-0d-14	UBS Securities Australia Ind	Sell	830	(100)	Ordinary
3-Oct-14	UBS Securities Australia Ltd	Sell	1,353	(163)	Ordinary
3-Oct-14	UBS Securities Australia Ltd	Sell	199	(24)	Ordinary
3-Oct-14	UBS Securities Australia Ltd	Sell	6,777	(818)	Ordinary
3-Oct-14	UBS Securities Australia Ltd	Sell	157	(1,448)	Ordinary
3-Oct-14	UBS Securities Australia Ltd	Sell	9,635	(1,163)	Ordinary
3-Oct-14	UBS Securities Australia Ltd	Sell	24,134		Ordinary
	UBS Securities Australia Ltd	Sell	8,425		Ordinary
1	UBS Securities Australia Ltd	J. 9	2,532	(305)	Ordinary
ı	UBS Securities Australia Ltd	Sell	3.967		Ordinary
	UBS Securities Australia Ltd	Sell	3,320		Ordinary
13-Oct-14	UBS Securities Australia Ltd	Sell	1,726		Ordinary
	UBS Securities Australia Ltd	les	6,623		Ordinary
3-00-14	UBS Securities Australia Ltd	lie S	2/2/2	(310)	Ordinary
ı	UBS Securities Australia Ltd	Sell	10,196		Ordinary
	UBS Securities Australia Ltd	Sell	4,911		Ordinary
Т	UBS Securities Australia Ltd	Zell Zell	183	(22)	Ordinary
	UBS Securities Australia Ltd	Seli	4,936	:	Ordinary
	UBS Securities Australia Ltd	Sell	1,238		Ordinary
1	UBS Securities Australia Ltd	Sell	6,935		Ordinary
1	UBS Securities Australia Itd	Sell	10.155		Ordinary
	UBS Securities Australia Ltd	Sell	5,944		Ordinary
П	UBS Securities Australia Ltd	Sell	2,114.		Ordinary
13-Oct-14	UBS Securities Australia Ltd	Sell	4,145	(200)	Ordinary
	1085 Securities Australia Ltd	lies Seit	067'8	(000(1)	Ordinary
3-04-14	UBS Securities Australia Ltd	Sell	102.795	(12,370)	Ordinary
3-Oct-14	UBS Securities Australia Ltd	Sell	12,631	(1,520)	Ordinary
3-Oct-14	UBS Securities Australia Ltd	Sell	219,450	(26,408)	Ordinary
3-Oct-14	UBS Securities Australia Ltd	Sell	968'99	(8,050)	Ordinary
3-Oct-14	UBS Securities Australia Ltd	Suy.	44,049	5,385	Ordinary
13-0ct-14	UBS Securities Australia Ltd	Buy	113.784	13.910	Ordinary
3-Oct-14	UBS Securities Australia Ltd	Buy	121,612	14,867	Ordinary
13-Oct-14	UBS Securities Australia Ltd	Buy	26,331	3,219	Ordinary
3-Oct-14	UBS Securities Australia Ltd	Buy	6,528	798	Ordinary
3-0ct-14	UBS Securities Australia Ltd	Buy	12,875	1,574	Ordinary
3.04.14	UBS Securities Australia 11d	Enter Into Derivative	650,62	Mond Ond AOS)	Ordinary
	ברכת וורכז אתפונו מום בנת	Enter Into Derivative		(00000000000000000000000000000000000000	a podo
13-0ct-14	UBS Securities Australia Ltd	Enter Into Derivative	N/N	000,000,000	Options
13-Oct-14	UBS Securities Australia Ltd	Contract	NA	(200'000'000)	Options
3-Oct-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	AVA	500,000,000	Options
1	USS Global Asset Management	) Seil	4,424	(575)	Ordinary
14-Oct-14	UBS Securities Australia Ltd	Buy	15,037	1,797	Ordinary
	UBS Securities Australia Ltd	Buy	181,181	11,559	Ordinary
	UBS Securities Australia Ltd	Buy	54,155	005'9	Ordinary
П	UBS Securities Australia Ltd	Sell	086'59	(8,000)	Ordinary
	UBS Securities Australia Ltd	Buy	14,928	1 781	Ordinary

Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinany	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary
(60)	(451)	(279)	(149)	(53)	(332)	(407)	(62)	(523)	(12)	(155)	(401)	(548)	(9)	(252)	(382)	(100,000)	(67)	(414)	(051)	(63)	(406)	(275)	(56)	(519)	(202)	(/3)	(100)	(647)	(645)	(4)	(Z) (Z)	(317)	(420)	(131)	(649)	(645)	(643)	(364)	(648)	(151)	(52)	(149)	(42)	(711)	(31)	(577)	(37)	(275)	(87)	(147)	(347)	(62)	(300)	(44b)	(448)	(371)	(77)	(301)	(34)	(110)	(310)	(48)	(20)	(435)
504	2,780	2,342	1,251	445	2.812	3,413	243	4,380	101	219,6	3,358	2 210	05	4,682	3,238	850,000	263	3,476	657.1	1 283	3,404	2,306	470	4,354	4,217	6.470	1,047	5,438	5,421	32	17	842	3.536	1,103	5,461	5,428	1,004	3.063	5,453	1,271	437	1,253	723	3 878	260	4,844	311	2,309	730	1,235	2,915	521	2,523	3.759	3.768	3,118	647	2,533	286	926	2,609	404	421	3,656
Sell	Sell	Sell	Seil	Sell	Sell	Sell	Seil	Sell	Sell	lies lies	liac liac	Sall	Sell	Sell	Sell	Seli	Sell	Sell	lec les	Cell	Sell	Sell	Sell	Sell	Sell	Jan Jan	Spl	Sell	Sell	Sell	Sell	Sell	Sell	Sell	Sell	Sell	la de	Sell	Seil	Sell	Sell	Sell	Sel.	Sell	Sell	Sell	Sell	Seli	Sell	Sell	Sell	Sell	Sell	100	Sell	Sell	Sell	Sell						
UBS Securities Australia Ltd	UBS Securities Australia Ltd	ies Australia Ltd	ies Australia Ltd	UBS Securities Australia Ltd	ies Australia Ltd	ies Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	IBS Securities Australia Ind	DS Securities Australia Ind	es Australia Itd	es Australia Ltd	UBS Securities Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	es Australia Ltd	ss Australia Ltd	UBS Securities Australia Ltd	ss Australia Ltd	UBS Securities Australia Ltd	USS Securities Australia Itd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	s Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	USS Securities Australia Ltd	JBS Securities Australia Ltd	Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd											
T	Sign						14-Oct-14 UBS Securitie	Ť	1		1	14-Oct-14 UBS Securities	l				-	14-Oct-14 UBS Securitie	ı		SBO	Ī	NBS	nes	4-Oct-14 UBS Securitie	J. J.		NBS	П	T	14-Oct-14 UBS Securitie		l	14-Oct-14 UBS Securitie	1	1		l	Ш	14-Oct-14 UBS Securities		١	Ī	Ī	14-Oct-14 UBS Securities	Ī	Ī		Ī		14-Oct-14 UBS Securities	T	T		ľ			14-Oct-14 UBS Securities		Ī		14-Oct-14 UBS Securities	Ī	

14-Oct-14	UBS Securities Australia Ltd	Sell	1 562	(186)	Ordinan
1-14	UBS Securities Australia Ltd	Sell	546	(65)	Ordinary
t-14	UBS Securities Australia Ltd	Sell	2,520	(300)	Ordinary
ct-14	UBS Securities Australia Ltd	Sell	2,100	(250)	Ordinary
ct-14	UBS Securities Australia Ltd	Sell	311	(37)	Ordinary
α-14	UBS Securities Australia Ltd	Sell	. 19	(8)	Ordinary
ct-14	UBS Securities Australia Ltd	Sell	4,701	(523)	Ordinary
d-14	UBS Securities Australia Ltd	Sell	3,749	(446)	Ordinary
4 - 14	UBS Securities Australia Ltd	Sell	437	(52)	Ordinary
177	LIBS Securities Australia Ltd	Jell Sell	65	(2)	Ordinary
4-14	IIIRS Securities Australia 14d	100	567	(45)	Ordinary
d-14	UBS Securities Australia Ltd	la s	8 6	(41)	Ordinary
d-14	UBS Securities Australia Ltd	Sell	3 791	(451)	Ordinary
ct-14	UBS Securities Australia Ltd	Seti	3,287	(392)	Ordinary
ct-14	UBS Securities Australia Ltd	Self	3,274	(06E)	Ordinary
d-14	UBS Securities Australia Ltd	Seil	3,215	(383)	Ordinary
ct-14	UBS Securities Australia Ltd	Sell	3,165	(377)	Ordinary
G-14	UBS Securities Australia Ltd	Sell	2,015	(240)	Ordinary
14-Oct-14	UBS Securities Australia Ltd	Sell	512	(61)	Ordinary
41-14	UBS Securities Australia Ltd	Sell	1,302	(155)	Ordinary
14	UBS Securities Australia Itd	Sell	420	(04)	Ordinary
DCI-14	UBS Securities Australia Ltd	Sell	17 171	(EV)	Ordinary
14-Oct-14	UBS Securities Australia Ltd	Sett	1,379	(164)	Ordinary
t-14	UBS Securities Australia Ltd	Sell	2,944	(320)	Ordinary
14-Oct-14	UBS Securities Australia Ltd	Sell	3,202	(381)	Ordinary
kt-14	UBS Securities Australia Ltd	Sell	1,447	(172)	Ordinary
4-0ct-14	UBS Securities Australia Ltd	Sell	3,202	(381)	Ordinary
4-Oct-14	UBS Securities Australia Ltd	Nell Sell	2,262	(569)	Ordinary
177	LIBS Securities Australia Ltd	Sell	040	(100)	Ordinary
14-14	LIRS Securities Australia Ltd	Sell	4,707	(095)	Ordinary
G-14	UBS Securities Australia Itd		203 4	(000)	Ordinary
d-14	UBS Securities Australia Ltd	les	3.774	(449)	Ordinary
ct-14	UBS Securities Australia Ltd	Sell	3,685	(439)	Ordinary
ct-14	UBS Securities Australia Ltd	Sell	3,618	(431)	Ordinary
ct-14	UBS Securities Australia Ltd	Sell	4,642	(553)	Ordinary
4-14	UBS Securities Australia Ltd	Sell	160,1	(130)	Ordinary
d-14	UBS Securities Australia Ltd	Sell	1,738	(207)	Ordinary
G-14	UBS Securities Australia Ltd	Sell	5,129	(611)	Ordinary
14	UBS Securities Australia Ltd	Sell	588	(02)	Ordinary
4 1 4	UBS Securities Australia Ltd	Sell	569	(32)	Ordinary
	UBS Securities Australia Ltd	Sell	773	(92)	Ordinary
ı	UBS Securities Australia Ltd	Hac I	3,512	(430)	Ordinary
1	UBS Securities Australia Ltd	lies	2,0/4	(247)	Ordinary
14-00-14	UBS Securities Australia Itd	3	/66'7	(357)	Ordinary
	UBS Securities Australia Itd	3	0 2 3	(1)	Ordinary
ı	UBS Securities Australia Itd	3	1,009	(679)	Ordinary
ļ	UBS Securities Australia Ltd	Sell	4 070	(484)	Ordinary
	UBS Securities Australia Ltd	Seil	4,336	(515)	Ordinary
П	UBS Securities Australia Ltd	Sell	808	(96)	Ordinary
П	UBS Securities Australia Ltd	lleS.	353	(42)	Ordinary
	UBS Securities Australia Ltd	Sell	1,372	(163)	Ordinary
-	UBS Securities Australia Ltd	Sell	1,869	(222)	Ordinary
	UBS Securities Australia Ltd	Sell	2,491	(296)	Ordinary
4-14	UBS Securities Australia Ltd	Sell	6,269	(745)	Ordinary
G-14	UBS Securities Australia Ltd	Sell	1,851	(220)	Ordinary
ct-14	UBS Securities Australia Ltd	Sell	2,812	(334)	Ordinary
71-17	UBS Securities Australia Ltd	Seil	6,236	(741)	Ordinary
41-14	UBS Securities Australia Ltd	Sell	999	(64)	Ordinary
41.14	UBS Securities Australia Ltd	, sell	6,273	(745)	Ordinary
117	VIDS Securities Australia Ind	Sell	758,1	(077)	Ordinary
1-14	UBS Securities Australia Ltd	Sell	3,528	(419)	Ordinary
4-0ct-14	LIRS Securities Australia 11d	Sell	9,6/5	(6/4)	Ordinary
4-Oct-14	LIRS Securities Australia 11d	Sell	576	(45)	Ordinary
1-14	UBS Securities Australia 11d	le de la company	400°C	(756)	Ordinary
14-Oct-14	UBS Securities Australia Ltd	Sell	6/5/9	(771)	Ordinary
t-14	UBS Securities Australia Ltd	les.	0000		
				(929)	Ordinary
41-5	UBS Securities Australia Ltd	105	1/7'c	(629)	Ordinary

UBS Securities Australia Ltd         Sell         53           UBS Securities Australia Ltd         Sell         138           UBS Securities Australia Ltd         Sell         139           UBS Securities Australia Ltd         Sell         139           UBS Securities Australia Ltd         Sell         139           UBS Securities Australia Ltd         Sell         130           UBS Securities Australia Ltd         Sell         2.5           UBS Securities Australia Ltd         Sell         2.6           UBS Securities Australia Ltd         Sell         2.6	ct-14	USS SECURITIES AUSTRAINA LTD	Jes	\$	:	
URS Securities Automia 10   Self   Control Securities Automia 10   Self   Securities Automia 10   Self   Securities Automia 10   Self   Securities Automia 10   Self   Self   Securities Automia 10   Self   Self   Securities Automia 10   Self   Self   Securities Automia 10   Self   Self   Securities Automia 10   Self   Securiti	ct-14	UBS Securities Australia Ltd	Sell	3,776		
Mis Securine Audition to   Mis Securine Auditi	41.4	UBS Securities Australia Ltd	Sell	16,555	`	
1855 Secretive Automits Ltd   544   1,528   1,625	4 1 4	UBS Securities Australia Ltd	Sell	3,893		
655 Securine Autorials Ltd   541   542   543	14	UBS Securities Australia Ltd	Sell	3,859		
Mis Securine Automia Ltd   Self   Self   Self   Securine Automia Ltd   Self	4 :	UBS Securities Australia Ltd	Seil	3,834		
10.00 Securine Automited 10.00   2.00   1.	14	UBS Securities Australia Ltd	Self	3,656		
10.50   10.5	# # # # #	UBS Securities Australia Ltd	)eil	120,2		
165 Secretic Automita 127   161   162   162   162   162   163   164   163   164   163   164	4-14	LIRC Securities Australia Ltd	Sell	1,/94 cco.c		
1655 Securine Autorities Later	4-14	LIRS Securities Australia I to	Coll	700 1		
165 Secretic Automita 127   164	114	IIRS Securities Australia Itd	Coll	1,00,1		
USS Scorring Automita to 1   Sept   2,999   (4.59)   (4	1.14	LIBS Securities Australia Ltd	9	3.475	7.60	
USS Scorticies Australia Ltd   Seil   2731 (445)   (	1-14	UBS Securities Australia Ltd	100	2,800		
1585 Scortifice Australia Ltd   1581   1,055   1,057	1-14	UBS Securities Australia Ltd	0.00	3 731		
USS Securine Australia Ltd   Self   Color	ct-14	UBS Securities Australia Ltd	Sel	755		
USS Scorrines Autoralia Ltd         Seal         2054         (450)           LBS Scorrines Autoralia Ltd         Seal         3910         (450)           LBS Scorrines Autoralia Ltd         Seal         372, 47         (471)           LBS Scorrines Autoralia Ltd         Seal         423         (51)           LBS Scorrines Autoralia Ltd         Seal         423         (58)           LBS Scorrines Autoralia Ltd         Seal         1,756         (1730)           LBS Scorrines Autoralia Ltd         Seal         1,620         (1780)           LBS Scorrines Autoralia Ltd         Seal         1,756         (1740)           LBS Scorrines Autoralia Ltd	r-14	UBS Securities Australia Ltd	Sell	1.065		
USS Scormines Australia Ltd         Seal         4.56         (6.50)           USS Scormines Australia Ltd         Seal         3.901         (6.50)           USS Scormines Australia Ltd         Seal         1.622         (7.1)           USS Scormines Australia Ltd         Seal         1.625         (7.60)           USS Scormines Australia Ltd         Seal         2.54         (6.60)           USS Scormines Australia Ltd         Seal         3.06         (6.60)           USS Scormines Australia Ltd         Seal         2.54         (7.2)           USS Scormines Australia Ltd         Se	ct-14	UBS Securities Australia Ltd	Sell	2,054		
UBS Securities Australia Led         Sell         3 95.0         (46.6)           UBS Securities Australia Led         Sell         3 95.0         (47.0)           UBS Securities Australia Led         Sell         2.25         (5.0)           UBS Securities Australia Led         Sell         5.21         (6.0)           LOSS Securities Australia Led         Sell         5.24         (5.0)           LOSS Securities Australia Led         Sell         3.09         (7.190)           LOSS Securities Australia Led         Sell         3.09         (7.278)           LOSS Securities Australia Led         Sell         3.09         (7.29)           LOSS Securities Australia Led         Sell         3.09         (7.29)           LOSS Sec	ct-14	UBS Securities Australia Ltd	Sell	436		ļ
URS Securities Australia Ind         Sell         3.95.2         (67)           URS Securities Australia Ind         Sell         2.28         (59)           URS Securities Australia Ind         Sell         1.62         (19)           LBS Securities Australia Ind         Sell         1.62         (19)           LBS Securities Australia Ind         Sell         1.62         (19)           LBS Securities Australia Ind         Sell         1.62         (1,50)           LBS Securities Australia Ind         Sell         1.62         (1,190)           LBS Securities Australia Ind         Sell         1.62         (1,50)           LBS Securities Australia Ind         Sell         1.62         (1,50)           LBS Securities Australia Ind         Sell         2.34         (1,52)           LBS Securities Australia Ind         Sell         2.34         (1,52)           LBS Securities Australia Ind         Sell         2.34         (1,52)           LBS Securities Australia Ind         Sell         2.34         (2,52)           LBS Securities Australia Ind         Sell         2.34         (2,52)           LBS Securities Australia Ind         Sell         2.34         (2,52)           LBS Securities Australia Ind	ct-14	UBS Securities Australia Ltd	Sell	3,910		
URS Securities Australia In UM Securities Australia	١	UBS Securities Australia Ltd	Sell	3,952		
URS Securine Australia Ind 10         Sell         1,528         (1,99)           URS Securine Australia Ind 10         Sell         1,500         (1,99)           URS Securine Australia Ind 10         Sell         5,800         (1,99)           URS Securine Australia Ind 10         Sell         5,800         (1,99)           URS Securine Australia Ind 10         Sell         3,300         (1,99)           URS Securine Australia Ind 10         Sell         3,300         (1,90)           URS Securine Australia Ind 2         Sell         2,300         (1,90)           URS Securine Australia Ind 2         Sell         2,300         (1,90)           URS Securine Australia Ind 2         Sell         2,300         (1,90)           URS Securine Australia Ind 2         Sell         3,400         (1,90)           UR	-	UBS Securities Australia Ltd	Sell	445		
UBS Securities Autorials Ltd         Sell         1,622         (198)           UBS Securities Autorials Ltd         Sell         5,77         (198)           UBS Securities Autorials Ltd         Sell         1,275         (1,590)           UBS Securities Autorials Ltd         Sell         1,796         (1,523)           UBS Securities Autorials Ltd         Sell         1,796         (1,523)           UBS Securities Autorials Ltd         Sell         1,796         (1,523)           UBS Securities Autorials Ltd         Sell         2,34         (2,50)           UBS Securities Autorials Ltd         Sell         2,544         (2,50)           UBS Securities Autorials Ltd         Sell         2,66         (2,53)           UBS Securities Autorials Ltd         Sell         2,66         (2,53)           UBS Securities Autor	1	UBS Securities Australia Ltd	Self	428		١
USS Securities Autorating Ltd         Sell         \$2.50         (4.52)           UBS Securities Autorating Ltd         Sell         1.57.266         (1.42)           UBS Securities Autorating Ltd         Sell         1.57.266         (1.42)           UBS Securities Autorating Ltd         Sell         3.927         (4690)           UBS Securities Autorating Ltd         Sell         3.927         (4690)           UBS Securities Autorating Ltd         Sell         3.927         (4690)           UBS Securities Autorating Ltd         Sell         3.924         (520)           UBS Securities Autorating Ltd         Sell         3.924         (520)           UBS Securities Autorating Ltd         Sell         3.924         (520)           UBS Securities Autorating Ltd         Sell         2.944         (520)           UBS Securities Autorating Ltd         Sell         2.945         (520)           UBS Securities Autorating Ltd         Sell         2.945         (520)           U	ı	UBS Securities Australia Ltd	Sell	1,662		l
UBS Securines Australia Ltd         Seil         1,250         (1,190)           UBS Securines Australia Ltd         Seil         1,266         (1,190)           UBS Securines Australia Ltd         Seil         1,266         (1,650)           UBS Securines Australia Ltd         Seil         3,937         (465)           UBS Securines Australia Ltd         Seil         3,934         (2,70)           UBS Securines Australia Ltd         Seil         2,344         (2,70)           UBS Securines Australia Ltd         Seil         2,544         (2,70)           UBS Securines Australia Ltd         Seil         2,661         (4,10)           UBS Securines Australia Ltd         Seil         2,661         (4,10)           UBS Securines Australia Lt	1	LIBS Securities Australia Ltd	lles o	2003		1
UBS Securities Australia Ltd         Seil         1,277         (1.42)           UBS Securities Australia Ltd         Seil         1,276         (1,423)           UBS Securities Australia Ltd         Seil         3,065         (1,523)           UBS Securities Australia Ltd         Seil         4,198         (500)           UBS Securities Australia Ltd         Seil         2,34         (2,27)           UBS Securities Australia Ltd         Seil         3,40         (3,27)           UBS Securities Australia Ltd         Seil         3,40         (3,27)           UBS Securities Australi		LIBS Securities Australia 11d	las	076'6	001 1/	
UBS Securities Australia Ltd         Sell         1,2786         (1,523)           UBS Securities Australia Ltd         Sell         3,937         (4659)           UBS Securities Australia Ltd         Sell         3,937         (4659)           UBS Securities Australia Ltd         Sell         2,344         (500)           UBS Securities Australia Ltd         Sell         2,334         (278)           UBS Securities Australia Ltd         Sell         2,334         (371)           UBS Securities Australia Ltd         Sell         2,544         (372)           UBS Securities Australia Ltd         Sell         2,544         (373)           UBS Securities Australia Ltd         Sell         2,544         (373)           UBS Securities Australia Ltd         Sell         3,408         (457)           UBS Securities Australia Ltd         Sell         3,641         (457)           UBS Securities Australi		UBS Securities Australia Ltd	Seil	1217	(145	l
UBS Securities Australia Ltd         Self         3566         (565)           UBS Securities Australia Ltd         Self         4,198         (669)           UBS Securities Australia Ltd         Self         4,198         (500)           UBS Securities Australia Ltd         Self         2,334         (277)           UBS Securities Australia Ltd         Self         2,334         (278)           UBS Securities Australia Ltd         Self         2,348         (523)           UBS Securities Australia Ltd         Self         3,041         (405)           UBS Securities Australia Ltd         Self         5,243         (523)           UBS Securities Australia Ltd         Self         4,375         (406)           UBS Securities Australia Ltd<		UBS Securities Australia Ltd	Sell	12,786	(1.523	
UBS Securities Australia Ltd         Sell         3937         (4690)           UBS Securities Australia Ltd         Sell         4,198         (500)           UBS Securities Australia Ltd         Sell         1,500         (527)           UBS Securities Australia Ltd         Sell         2,244         (278)           UBS Securities Australia Ltd         Sell         2,334         (278)           UBS Securities Australia Ltd         Sell         2,344         (278)           UBS Securities Australia Ltd         Sell         2,344         (278)           UBS Securities Australia Ltd         Sell         3,407         (353)           UBS Securities Australia Ltd         Sell         3,407         (353)           UBS Securities Australia Ltd         Sell         2,586         (373)           UBS Securities Australia Ltd         Sell         2,586         (373)           UBS Securities Australia Ltd         Sell         3,407         (353)           UBS Securities Australia Ltd         Sell         3,609         (369)           UBS Securities Australia Ltd         Sell         3,609         (369)           UBS Securities Australia Ltd         Sell         4,378         (369)           UBS Securities Australia Ltd		UBS Securities Australia Etd	Sell	3,066	(365	
UBS Securities Australia Ltd         Sell         4,198         (500)           UBS Securities Australia Ltd         Sell         2,20         (500)           UBS Securities Australia Ltd         Sell         2,34         (278)           UBS Securities Australia Ltd         Sell         2,34         (278)           UBS Securities Australia Ltd         Sell         2,344         (201)           UBS Securities Australia Ltd         Sell         2,344         (203)           UBS Securities Australia Ltd         Sell         2,394         (253)           UBS Securities Australia Ltd         Sell         3,089         (352)           UBS Securities Australia Ltd         Sell         3,089         (352)           UBS Securities Australia Ltd         Sell         4,729         (351)           UBS Securities Australia Ltd         Sell         4,729         (351)           UBS Securities Australia Ltd         Sell         4,729         (351)           UBS Securities Australia Ltd <td></td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>3,937</td> <td>697)</td> <td></td>		UBS Securities Australia Ltd	Sell	3,937	697)	
UBS Securities Australia Ltd         Sell         4.200         (500)           UBS Securities Australia Ltd         Sell         2.94         (327)           UBS Securities Australia Ltd         Sell         2.344         (278)           UBS Securities Australia Ltd         Sell         2.340         (273)           UBS Securities Australia Ltd         Sell         2.560         (331)           UBS Securities Australia Ltd         Sell         2.560         (331)           UBS Securities Australia Ltd         Sell         2.560         (327)           UBS Securities Australia Ltd         Sell         2.540         (35)           UBS Securities Australia Ltd         Sell         2.540         (35)           UBS Securities Australia Ltd         Sell         3.241         (35)           UBS Securities Australia Ltd         Sell         3.241         (35)           UBS Securities Australia Ltd		UBS Securities Australia Ltd	Sell	4,198	005)	
UBS Securities Australia Ltd         Sell         29.4         (25)           UBS Securities Australia Ltd         Sell         2.334         (278)           UBS Securities Australia Ltd         Sell         2.344         (279)           UBS Securities Australia Ltd         Sell         2.344         (273)           UBS Securities Australia Ltd         Sell         2.447         (273)           UBS Securities Australia Ltd         Sell         2.467         (273)           UBS Securities Australia Ltd         Sell         3.408         (273)           UBS Securities Australia Ltd         Sell         3.408         (373)           UBS Securities Australia Ltd         Sell         3.068         (373)           UBS Securities Australia Ltd         Sell         3.041         (352)           UBS Securities Australia Ltd         Sell         4.596         (431)           UBS Securities Australia Ltd         Sell         4.596         (431)           UBS Securities Australia Ltd </td <td>-</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>4,200</td> <td>(500</td> <td></td>	-	UBS Securities Australia Ltd	Sell	4,200	(500	
UBS Securities Australia Ltd         Sell         1,909         (227)           UBS Securities Australia Ltd         Sell         2,544         (203)           UBS Securities Australia Ltd         Sell         2,544         (203)           UBS Securities Australia Ltd         Sell         2,544         (203)           UBS Securities Australia Ltd         Sell         2,643         (255)           UBS Securities Australia Ltd         Sell         2,643         (257)           UBS Securities Australia Ltd         Sell         3,086         (373)           UBS Securities Australia Ltd         Sell         3,086         (373)           UBS Securities Australia Ltd         Sell         3,089         (373)           UBS Securities Australia Ltd         Sell         2,695         (373)           UBS Securities Australia Ltd         Sell         3,089         (352)           UBS Securities Australia Ltd         Sell         3,081         (55)           UBS Securities Australia Ltd         Sell         4,784         (154)           UBS Securities Australia Ltd         Sell         4,784         (154)           UBS Securities Australia Ltd         Sell         4,578         (54)           UBS Securities Australia Ltd </td <td>1</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>294</td> <td>(35</td> <td></td>	1	UBS Securities Australia Ltd	Sell	294	(35	
UBS Securities Australia Ltd         Sell         2.34         (27)           UBS Securities Australia Ltd         Sell         2.34         (29)           UBS Securities Australia Ltd         Sell         2.348         (27)           UBS Securities Australia Ltd         Sell         4,407         (523)           UBS Securities Australia Ltd         Sell         4,393         (203)           UBS Securities Australia Ltd         Sell         4,393         (523)           UBS Securities Australia Ltd         Sell         2,806         (334)           UBS Securities Australia Ltd         Sell         2,806         (334)           UBS Securities Australia Ltd         Sell         2,806         (334)           UBS Securities Australia Ltd         Sell         2,806         (321)           UBS Securities Australia Ltd         Sell         1,784         (154)           UBS Securities Australia Ltd         Sell         5,806         (321)           UBS Securities Australia Ltd         Sell         5,806         (321)           UBS Securities Australia Ltd         Sell         3,041         (352)           UBS Securities Australia Ltd         Sell         3,041         (352)           UBS Securities Australia Ltd		UBS Securities Australia Ltd	Seil	1,909		
UBS Securities Australia Ltd         Sell         2,434         (243)           UBS Securities Australia Ltd         Sell         2,434         (291)           UBS Securities Australia Ltd         Sell         2,467         (252)           UBS Securities Australia Ltd         Sell         2,667         (273)           UBS Securities Australia Ltd         Sell         4,427         (257)           UBS Securities Australia Ltd         Sell         2,666         (33)           UBS Securities Australia Ltd         Sell         2,669         (33)           UBS Securities Australia Ltd         Sell         3,089         (58)           UBS Securities Australia Ltd         Sell         3,089         (58)           UBS Securities Australia Ltd         Sell         1,284         (14)           UBS Securities Australia Ltd         Sell         3,089         (58)           UBS Securities Australia Ltd         Sell         3,041         (35)           UBS Securities Australia Ltd         Sell         3,041         (35)           UBS Securities Australia Ltd         Sell         4,598         (41)           UBS Securities Australia Ltd         Sell         4,578         (56)           UBS Securities Australia Ltd	ı	UBS Securities Australia Ltd	) seil	2,334		
UBS Securities Australia Ltd         Sell         2,744.3         (259.9)           UBS Securities Australia Ltd         Sell         2,734         (259.1)           UBS Securities Australia Ltd         Sell         2,61         (371)           UBS Securities Australia Ltd         Sell         3,408         (4,593)           UBS Securities Australia Ltd         Sell         3,408         (4,606)           UBS Securities Australia Ltd         Sell         2,609         (334)           UBS Securities Australia Ltd         Sell         2,699         (337)           UBS Securities Australia Ltd         Sell         2,695         (327)           UBS Securities Australia Ltd         Sell         2,695         (327)           UBS Securities Australia Ltd         Sell         2,695         (327)           UBS Securities Australia Ltd         Sell         4,798         (557)           UBS Securities Australia Ltd         Sell         4,798         (571)           UBS Securities Australia Ltd         Sell         4,798         (571)           UBS Securities Australia Ltd         Sell         4,798         (571)           UBS Securities Australia Ltd         Sell         4,578         (541)           UBS Securities Aust		UBS Securities Australia Ltd	Sell	2,544		
UBS Securities Australia Ltd         Sell         4,754         (252)           UBS Securities Australia Ltd         Sell         4,407         (5.55)           UBS Securities Australia Ltd         Sell         4,393         (5.53)           UBS Securities Australia Ltd         Sell         4,393         (5.53)           UBS Securities Australia Ltd         Sell         2,698         (589)           UBS Securities Australia Ltd         Sell         2,698         (581)           UBS Securities Australia Ltd         Sell         2,698         (581)           UBS Securities Australia Ltd         Sell         2,698         (581)           UBS Securities Australia Ltd         Sell         1,734         (154)           UBS Securities Australia Ltd         Sell         1,734         (154)           UBS Securities Australia Ltd         Sell         1,734         (154)           UBS Securities Australia Ltd         Sell         8,79         (557)           UBS Securities Australia Ltd         Sell         8,79         (567)           UBS Securities Australia Ltd         Sell         4,79         (567)           UBS Securities Australia Ltd         Sell         4,78         (567)           UBS Securities Australia Ltd<	ı	UBS Securities Australia Ltd	Sell	2,443		
UBS Securities Australia Ltd         Sell         2,561         (373)           UBS Securities Australia Ltd         Sell         4,393         (523)           UBS Securities Australia Ltd         Sell         4,427         (527)           UBS Securities Australia Ltd         Sell         2,666         (527)           UBS Securities Australia Ltd         Sell         2,669         (527)           UBS Securities Australia Ltd         Sell         2,695         (53)           UBS Securities Australia Ltd         Sell         1,7294         (14)           UBS Securities Australia Ltd         Sell         1,7294         (15)           UBS Securities Australia Ltd         Buy         4,986         (10)           UBS Securities Australia Ltd         Buy         4,986         (10)           UBS Securities Australia Ltd         Buy         4,986         (10)           UBS Securities Australia Ltd         Sell         3,041         (56)           UBS Securities Australia Ltd         Sell         4,576         (58)           UBS Securities Australia Ltd         Sell         4,576         (58)           UBS Securities Australia Ltd         Sell         4,577         (58)           UBS Securities Australia Ltd		UBS Securities Australia (1d	Sell	7077		
UBS Securities Australia Ltd         Sell         4,393         (5.23)           UBS Securities Australia Ltd         Sell         3,408         (406)           UBS Securities Australia Ltd         Sell         2,806         (334)           UBS Securities Australia Ltd         Sell         2,809         (334)           UBS Securities Australia Ltd         Sell         2,690         (319)           UBS Securities Australia Ltd         Sell         2,690         (319)           UBS Securities Australia Ltd         Sell         2,690         (319)           UBS Securities Australia Ltd         Buy         4,72         (45)           UBS Securities Australia Ltd         Buy         4,78         (55)           UBS Securities Australia Ltd         Sell         4,78         (57)           UBS Securities Australia Ltd         Sell         4,78         (51)           UBS Securities Australia Ltd         Sell         4,78         (51)           UBS Securities Australia Ltd         Sell         4,78         (51)           UBS Securities Australia Ltd         Sell         4,78         (54)           UBS Securities Australia Ltd         Sell         4,578         (54)           UBS Securities Australia Ltd <t< td=""><td></td><td>UBS Securities Australia Ltd</td><td>100</td><td>7 661</td><td></td><td></td></t<>		UBS Securities Australia Ltd	100	7 661		
UBS Securities Australia Ltd         Sell         3,233         1,232           UBS Securities Australia Ltd         Sell         4,27         (577)           UBS Securities Australia Ltd         Sell         2,806         (344)           UBS Securities Australia Ltd         Sell         2,699         (371)           UBS Securities Australia Ltd         Sell         2,699         (371)           UBS Securities Australia Ltd         Sell         7,798         (371)           UBS Securities Australia Ltd         Sell         7,798         (371)           UBS Securities Australia Ltd         Sell         7,794         (154)           UBS Securities Australia Ltd         Buy         5,21         4,9           UBS Securities Australia Ltd         Sell         3,041         (362)           UBS Securities Australia Ltd         Sell         3,041         (362)           UBS Securities Australia Ltd         Sell         3,451         (412)           UBS Securities Australia Ltd         Sell         4,578         (441)           UBS Securities Australia Ltd         Sell         4,578         (542)           UBS Securities Australia Ltd         Sell         4,578         (542)           UBS Securities Australia Ltd		LIBS Securities Australia 14d	100	7 303		
UBS Securities Australia Ltd         Self         4,427         (527)           UBS Securities Australia Ltd         Self         2,806         (334)           UBS Securities Australia Ltd         Self         2,699         (339)           UBS Securities Australia Ltd         Self         2,699         (331)           UBS Securities Australia Ltd         Self         680         (319)           UBS Securities Australia Ltd         Self         680         (35)           UBS Securities Australia Ltd         Bby         4,798         (49)           UBS Securities Australia Ltd         Bby         4,798         (49)           UBS Securities Australia Ltd         Self         4,798         (101)           UBS Securities Australia Ltd         Self         4,798         (101)           UBS Securities Australia Ltd         Self         4,578         (54)           UBS Securities Australia Ltd         Self         4,578         (54)           UBS Securities Australia Ltd	ı	UBS Securities Australia 11d	i a	3 400		
List Securities Australia Lid         Sell         2,806         (3.94)           List Securities Australia Ltd         Sell         2,808         (3.98)           18B Securities Australia Ltd         Sell         2,699         (3.91)           18B Securities Australia Ltd         Sell         2,695         (3.91)           18B Securities Australia Ltd         Sell         1,734         (1.54)           18B Securities Australia Ltd         Buy         4,72         62           18B Securities Australia Ltd         Buy         4,72         62           18B Securities Australia Ltd         Buy         4,72         62           18B Securities Australia Ltd         Sell         4,576         (557)           18B Securities Australia Ltd         Sell         4,576         (557)           18B Securities Australia Ltd         Sell         3,641         (3.5)           18B Securities Australia Ltd         Sell         4,542         (5.61)           18B Securities Australia Ltd         Sell         4,542         (5.64)           18B Securities Australia Ltd         Sell         4,578         (5.64)           18B Securities Australia Ltd         Sell         4,578         (5.65)           18B Securities Australia Ltd <td>l</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>4 427</td> <td></td> <td></td>	l	UBS Securities Australia Ltd	Sell	4 427		
UBS Securities Australia Ltd         Sell         3,089         (968)           UBS Securities Australia Ltd         Sell         2,680         (31)           UBS Securities Australia Ltd         Sell         2,695         (321)           UBS Securities Australia Ltd         Sell         738         (83)           UBS Securities Australia Ltd         Bby         4,294         (85)           UBS Securities Australia Ltd         Bby         5,21         45           UBS Securities Australia Ltd         Bby         5,21         65           UBS Securities Australia Ltd         Sell         3,41         (56)           UBS Securities Australia Ltd         Sell         3,451         (56)           UBS Securities Australia Ltd         Sell         3,451         (42)           UBS Securities Australia Ltd         Sell         3,461         (41)           UBS Securities Australia Ltd         Sell         4,573         (56)           UBS Securities Australia Ltd         Sell <td></td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>2 806</td> <td>(33)</td> <td></td>		UBS Securities Australia Ltd	Sell	2 806	(33)	
UBS Securities Australia Ltd         Sell         2.695         (319)           UBS Securities Australia Ltd         Sell         6.695         (321)           UBS Securities Australia Ltd         Sell         1.294         (184)           UBS Securities Australia Ltd         Buy         4.72         4.6           UBS Securities Australia Ltd         Buy         4.72         4.6           UBS Securities Australia Ltd         Sell         4.72         6.2           UBS Securities Australia Ltd         Sell         4.78         (58)           UBS Securities Australia Ltd         Sell         3.041         (58)           UBS Securities Australia Ltd         Sell         3.041         (58)           UBS Securities Australia Ltd         Sell         4.572         (541)           UBS Securities Australia Ltd         Sell         4.572         (541)           UBS Securities Australia Ltd         Sell         4.572         (541)           UBS Securities Australia Ltd         Sell         4.572         (545)           UBS Securities Australia Ltd         Sell         4.572         (546)           UBS Securities Australia Ltd         Sell         4.572         (546)           UBS Securities Australia Ltd         <		UBS Securities Australia Ltd	Sell	3.089	(368)	
UBS Securities Australia Ltd         Sel         (321)           UBS Securities Australia Ltd         Sel         (89)         (81)           UBS Securities Australia Ltd         Sel         (84)         (85)           UBS Securities Australia Ltd         Bay         4124         (184)           UBS Securities Australia Ltd         BBy         4,298         (85)           UBS Securities Australia Ltd         Sel         3,041         (85)           UBS Securities Australia Ltd         Sel         8,10         (10)           UBS Securities Australia Ltd         Sel         3,041         (85)           UBS Securities Australia Ltd         Sel         3,041         (87)           UBS Securities Australia Ltd         Sel         3,451         (4,12)           UBS Securities Australia Ltd         Sel         4,542         (54)           UBS Securities Australia Ltd         Sel         4,542         (54)           UBS Securities Australia Ltd         Sel         4,578         (54)           UBS Securities Australia Ltd         Sel         4,578         (54)           UBS Securities Australia Ltd         Sel         4,578         (55)           UBS Securities Australia Ltd         Sel         4,578		UBS Securities Australia Ltd	Sell	2.580	(319)	
UBS Securities Australia Ltd         Sell         F89         (87)           UBS Securities Australia Ltd         Sell         794         (85)           UBS Securities Australia Ltd         Bay         4,294         (144)           UBS Securities Australia Ltd         Bay         4,394         (85)           UBS Securities Australia Ltd         Sell         3,41         (85)           UBS Securities Australia Ltd         Sell         3,67         (86)           UBS Securities Australia Ltd         Sell         3,041         (87)           UBS Securities Australia Ltd         Sell         3,041         (87)           UBS Securities Australia Ltd         Sell         3,461         (41)           UBS Securities Australia Ltd         Sell         4,572         (54)           UBS Securities Australia Ltd         Sell         4,572         (54)           UBS Securities Australia Ltd         Sell         4,573         (56)           UBS Securities Australia Ltd         Sell	ĺ	UBS Securities Australia Ltd	Sell	2.695	(321	
UBS Securities Australia Ltd         Sell         798         (95)           UBS Securities Australia Ltd         Bay         43         41           UBS Securities Australia Ltd         Bay         4 98         15           UBS Securities Australia Ltd         Sell         4 98         56           UBS Securities Australia Ltd         Sell         4 98         56           UBS Securities Australia Ltd         Sell         4 576         (527)           UBS Securities Australia Ltd         Sell         3 461         (412)           UBS Securities Australia Ltd         Sell         4 542         (412)           UBS Securities Australia Ltd         Sell         4 542         (412)           UBS Securities Australia Ltd         Sell         4 542         (58)           UBS Securities Australia Ltd         Sell         4 578         (58)           UBS Securities Australia Ltd         Sell         4 578         (58)           UBS Securities Australia Ltd         Sell         4 578         (487)           UBS Securities Australia Ltd         Sell         4 578         (487)           UBS Securities Australia Ltd         Sell         4 578         (487)           UBS Securities Australia Ltd         Sell <td></td> <td>UBS Securities Australia Ltd</td> <td>Seil</td> <td>089</td> <td>(81)</td> <td></td>		UBS Securities Australia Ltd	Seil	089	(81)	
UBS Securities Australia Ltd         Sell         1,294         (1,54)           UBS Securities Australia Ltd         Buy         521         49           UBS Securities Australia Ltd         Buy         521         65           UBS Securities Australia Ltd         Sell         3,041         (562)           UBS Securities Australia Ltd         Sell         4,676         (557)           UBS Securities Australia Ltd         Sell         3,041         (567)           UBS Securities Australia Ltd         Sell         4,676         (577)           UBS Securities Australia Ltd         Sell         3,461         (412)           UBS Securities Australia Ltd         Sell         4,572         (543)           UBS Securities Australia Ltd         Sell         4,572         (543)           UBS Securities Australia Ltd         Sell         4,573         (569)           UBS Securities Australia Ltd         Sell         4,578         (546)           UBS Securities Australia Ltd		UBS Securities Australia Ltd	Sell	862	(95)	
UBS Securities Australia Ltd         Bay         412         49           UBS Securities Australia Ltd         Buy         521         62           UBS Securities Australia Ltd         Sell         4,988         555           UBS Securities Australia Ltd         Sell         4,578         (101)           UBS Securities Australia Ltd         Sell         848         (101)           UBS Securities Australia Ltd         Sell         3,461         (45)           UBS Securities Australia Ltd         Sell         4,542         (54)           UBS Securities Australia Ltd         Sell         4,542         (54)           UBS Securities Australia Ltd         Sell         4,573         (56)           UBS Securities Australia Ltd         Sell         4,573         (56)           UBS Securities Australia Ltd         Sell         4,578         (56)           UBS Securities Australia Ltd         Sell <td>ļ</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>1,294</td> <td>(154)</td> <td></td>	ļ	UBS Securities Australia Ltd	Sell	1,294	(154)	
UBS Securities Australia Ltd         BBy         62           UBS Securities Australia Ltd         6By         4,986         585           UBS Securities Australia Ltd         5B         1,041         (557)           UBS Securities Australia Ltd         5B         (157)         (157)           UBS Securities Australia Ltd         5B         (157)         (157)           UBS Securities Australia Ltd         5B         (157)         (157)           UBS Securities Australia Ltd         5B         (157)         (158)           UBS Securities Australia Ltd         5B         (158)	1-14	UBS Securities Australia Ltd	Buy	412	49	
UBS Securities Australia Ltd         Sell         4,998         5.95           UBS Securities Australia Ltd         Sell         4,978         (577)           UBS Securities Australia Ltd         Sell         4,676         (577)           UBS Securities Australia Ltd         Sell         3,471         (101)           UBS Securities Australia Ltd         Sell         4,773         (412)           UBS Securities Australia Ltd         Sell         4,773         (58)           UBS Securities Australia Ltd         Sell         4,773         (58)           UBS Securities Australia Ltd         Sell         4,273         (58)           UBS Securities Australia Ltd         Sell         4,273         (58)           UBS Securities Australia Ltd         Sell         4,273         (58)           UBS Securities Australia Ltd         Sell         4,578         (546)           UBS Securities Australia Ltd         Sell         4,578         (546)           UBS Securities Australia Ltd         Sell         4,506         (556)           UBS Securities Australia Ltd         Sell         2,457         (290)           UBS Securities Australia Ltd         Sell         2,457         (293)           UBS Securities Australia Ltd	1-14	UBS Securities Australia Ltd	Buy	521	62	
UBS Securities Australia Ltd         Seal         3 Q41         (552)           UBS Securities Ltd         Seal         4,676         (557)           UBS Securities Australia Ltd         Seal         (101)           UBS Securities Australia Ltd         Seal         3,451         (41)           UBS Securities Australia Ltd         Seal         4,542         (41)           UBS Securities Australia Ltd         Seal         4,542         (54)           UBS Securities Australia Ltd         Seal         4,573         (56)           UBS Securities Australia Ltd         Seal         4,578         (56)           UBS Securities Australia Ltd         Seal         2,570         (58)           UBS Securities Australia Ltd         Seal         2,570         (58)           UBS Securities Australia Ltd         Seal         2,570	t-14	UBS Securities Australia Ltd	Buy	4,998	595	
UBS Securities Australia Ltd         Sell         4,676         (557)           UBS Securities Australia Ltd         Sell         848         (10)           UBS Securities Australia Ltd         Sell         4,676         (412)           UBS Securities Australia Ltd         Sell         4,522         (412)           UBS Securities Australia Ltd         Sell         4,273         (56)           UBS Securities Australia Ltd         Sell         4,273         (56)           UBS Securities Australia Ltd         Sell         4,273         (56)           UBS Securities Australia Ltd         Sell         4,578         (56)           UBS Securities Australia Ltd         Sell         3,270         (58)           UBS Securities Australia Ltd         Sell         3,270         (59)           UBS Securities Australia Ltd         Sell         2,457         (23)           UBS Securities Australia Ltd         Sell         2,457         (23)           UBS Securities Australia Ltd	t-14	UBS Securities Australia Ltd	Seli	3,041	(362)	
UBS Securities Australia Ltd         Sell         R48         (101)           UBS Securities Australia Ltd         Sell         3.02         (56)           UBS Securities Australia Ltd         Sell         4.273         (412)           UBS Securities Australia Ltd         Sell         4.273         (541)           UBS Securities Australia Ltd         Sell         4.273         (56)           UBS Securities Australia Ltd         Sell         4.273         (56)           UBS Securities Australia Ltd         Sell         4.273         (56)           UBS Securities Australia Ltd         Sell         4.578         (546)           UBS Securities Australia Ltd         Sell         4.562         (556)           UBS Securities Australia Ltd         Sell         4.096         (487)           UBS Securities Australia Ltd         Sell         3.70         (390)           UBS Securities Australia Ltd         Sell         2.457         (233)           UBS Securities Australia Ltd         Sell         2.457         (239)           UBS Securities Australia Ltd         Sell         2.457         (239)           UBS Securities Australia Ltd         Sell         2.457         (239)           UBS Securities Australia Ltd	t-14	UBS Securities Australia Ltd	lleS	4,676	(557)	
UBS Securities Australia Ltd         Sell         3.02         (36)           UBS Securities Australia Ltd         Sell         3.461         (4.12)           UBS Securities Australia Ltd         Sell         4.542         (541)           UBS Securities Australia Ltd         Sell         4.737         (59)           UBS Securities Australia Ltd         Sell         4.737         (56)           UBS Securities Australia Ltd         Sell         4.157         (56)           UBS Securities Australia Ltd         Sell         4.157         (56)           UBS Securities Australia Ltd         Sell         4.578         (55)           UBS Securities Australia Ltd         Sell         2.487         (529)           UBS Securities Australia Ltd         Sell         2.487         (229)           UBS Securities Australia Ltd         Sell         2.487         (229)           UBS Securities Australia Ltd         Sell         2.487         (229)           UBS Securities Australia Ltd	t-14	UBS Securities Australia Ltd	Sell	848	(101)	
UBS Securities Australia Inf         Sell         3.451         (47.2)           UBS Securities Australia Ltd         Sell         4.542         (54)           UBS Securities Australia Ltd         Sell         4.737         (59)           UBS Securities Australia Ltd         Sell         4.757         (59)           UBS Securities Australia Ltd         Sell         4.757         (455)           UBS Securities Australia Ltd         Sell         4.757         (455)           UBS Securities Australia Ltd         Sell         4.578         (546)           UBS Securities Australia Ltd         Sell         4.662         (556)           UBS Securities Australia Ltd         Sell         4.662         (556)           UBS Securities Australia Ltd         Sell         3.270         (590)           UBS Securities Australia Ltd         Sell         2.4657         (529)           UBS Securities Australia Ltd         Sell         2.467         (229)           UBS Securities Australia L	1-14	UBS Securities Australia Ltd	Sell	302	(36)	L
UBS Securities Australia Ltd         Sell         4,542         (541)           UBS Securities Australia Ltd         Sell         277         (33)           UBS Securities Australia Ltd         Sell         4,273         (569)           UBS Securities Australia Ltd         Sell         4,578         (569)           UBS Securities Australia Ltd         Sell         4,578         (569)           UBS Securities Australia Ltd         Sell         4,578         (556)           UBS Securities Australia Ltd         Sell         4,578         (556)           UBS Securities Australia Ltd         Sell         4,578         (467)           UBS Securities Australia Ltd         Sell         4,578         (487)           UBS Securities Australia Ltd         Sell         3,70         (359)           UBS Securities Australia Ltd         Sell         2,457         (239)           UBS Securities Australia Ltd         Sell         2,457         (229)	t-14	UBS Securities Australia Ltd	Sell	3,461	(412)	Ordinary
UBS Securities Australia Ltd         Sell         277         (33)           UBS Securities Australia Ltd         Sell         4,273         (56)           UBS Securities Australia Ltd         Sell         4,151         (465)           UBS Securities Australia Ltd         Sell         4,157         (465)           UBS Securities Australia Ltd         Sell         4,578         (546)           UBS Securities Australia Ltd         Sell         4,662         (556)           UBS Securities Australia Ltd         Sell         4,662         (556)           UBS Securities Australia Ltd         Sell         3,706         (390)           UBS Securities Australia Ltd         Sell         2,457         (293)	t-14	UBS Securities Australia Ltd	Sell	4,542	(541)	L
UBS Securities Australia Ltd         Sell         4,273         (569)           UBS Securities Australia Ltd         Sell         4,273         (569)           UBS Securities Australia Ltd         Sell         4,151         (459)           UBS Securities Australia Ltd         Sell         4,578         (546)           UBS Securities Australia Ltd         Sell         4,662         (556)           UBS Securities Australia Ltd         Sell         4,662         (559)           UBS Securities Australia Ltd         Sell         4,086         (487)           UBS Securities Australia Ltd         Sell         2,467         (259)	t-14	UBS Securities Australia Ltd	Sell	772	(33)	L
UBS Securities Australie Ltd         Sell         467         (58)           UBS Securities Australie Ltd         Sell         4,151         (465)           UBS Securities Australie Ltd         Sell         4,578         (546)           UBS Securities Australie Ltd         Sell         4,622         (556)           UBS Securities Australie Ltd         Sell         4,622         (556)           UBS Securities Australie Ltd         Sell         4,622         (556)           UBS Securities Australie Ltd         Sell         3,70         (487)           UBS Securities Australie Ltd         Sell         2,457         (293)           UBS Securities Australie Ltd         Sell         2,457         (593)	t-14	UBS Securities Australia Ltd	les.	4273	(505)	Ordinary
UBS Securities Australia Ltd         Sell         4,151         (455)           UBS Securities Australia Ltd         Sell         4,578         (546)           UBS Securities Australia Ltd         Sell         4,622         (556)           UBS Securities Australia Ltd         Sell         4,662         (556)           UBS Securities Australia Ltd         Sell         4,662         (556)           UBS Securities Australia Ltd         Sell         3,706         (390)           UBS Securities Australia Ltd         Sell         2,457         (291)           UBS Securities Australia Ltd         Sell         2,457         (292)	t-14	UBS Securities Australia Ltd	Seil	487	(58)	
UBS Securities Australia Ltd         Sell         4,578         (5,46)           UBS Securities Australia Ltd         Sell         4,662         (5,56)           UBS Securities Australia Ltd         Sell         4,662         (5,56)           UBS Securities Australia Ltd         Sell         4,682         (487)           UBS Securities Australia Ltd         Sell         3,270         (5,90)           UBS Securities Australia Ltd         Sell         2,457         (2,93)	t-14	UBS Securities Australia Ltd	Se	4151	(495)	
UBS Securities Australia Ltd         Sell         4,662         (5.56)           UBS Securities Australia Ltd         Sell         4,662         (5.56)           UBS Securities Australia Ltd         Sell         4,085         (487)           UBS Securities Australia Ltd         Sell         3,270         (390)           UBS Securities Australia Ltd         Sell         2,487         (290)           UBS Securities Australia Ltd         Sell         2,487         (290)           UBS Securities Australia Ltd         Sell         2,487         (290)           UBS Securities Australia Ltd         Sell         2,457         (290)           UBS Securities Australia Ltd         Sell         2,457         (290)	1-14	UBS Securities Australia Ltd	Sell	4,578	(546)	
UBS Securities Australie Ltd         Sell         4,622         (556)           UBS Securities Australie Ltd         Sell         4,095         (487)           UBS Securities Australia Ltd         Sell         2,70         (390)           UBS Securities Australia Ltd         Sell         2,457         (293)           UBS Securities Australia Ltd         Sell         2,487         (293)           UBS Securities Australia Ltd         Sell         2,457         (293)           UBS Securities Australia Ltd         Sell         2,457         (293)           UBS Securities Australia Ltd         Sell         2,457         (293)	1-14	UBS Securities Australia Ltd	Sell	4,662	(556)	
UBS Securities Australia Ltd         Sell         4,086         (467)           UBS Securities Australia Ltd         Sell         3,770         (390)           UBS Securities Australia Ltd         Sell         2,457         (293)           UBS Securities Australia Ltd         Sell         2,482         (296)           UBS Securities Australia Ltd         Sell         2,487         (296)           UBS Securities Australia Ltd         Sell         2,451         (297)           UBS Securities Australia Ltd         Sell         2,451         (202)	41-1	UBS Securities Australia Ltd	lles	4 662	(556)	
UBS Securities Australia Ltd         Sell         3,270         (390)           UBS Securities Australia Ltd         Sell         2,457         (293)           UBS Securities Australia Ltd         Sell         2,482         (296)           UBS Securities Australia Ltd         Sell         2,482         (296)           UBS Securities Australia Ltd         Sell         2,457         (297)           UBS Securities Australia Ltd         Sell         420         (50)	1-14	UBS Securities Australia Ltd	lleS	4.086	(487)	
UBS Securities Australia Ltd         Sell         2,457         (293)           UBS Securities Australia Ltd         Sell         2,482         (296)           UBS Securities Australia Ltd         Sell         2,482         (296)           UBS Securities Australia Ltd         Sell         2,451         (292)           UBS Securities Australia Ltd         Sell         420         (50)	14	UBS Securities Australia Ltd	lleS.	3 2 7 0	(390)	
UBS Securities Australia Ltd         Sell         2,482         (296)           UBS Securities Australia Ltd         Sell         2,451         (292)           UBS Securities Australia Ltd         Sell         2,451         (50)	-14	UBS Securities Australia Ltd	Sell	7.457	(263)	
UBS Securities Australia Ltd         Sell         2,451         (292)           UBS Securities Australia Ltd         Sell         420         (50)	-14	UBS Securities Australia Ltd	Sell	2.482	(2962)	
UBS Securities Australia Ltd Sell 420 (50)	-14	UBS Securities Australia I td	la y	2 451	(505)	
(OC) OZA		IIBS Securities Augustia ind	nac Coll	104,2	(267)	
Million A. Lateral A. Canada and A. Canada a	<u>.</u>	Obs securines washand tru	lac!	107 <del>4</del>	170	

(54) Ordinary		(707) Ordinary	(25,000) Ordinary		(Society)			(650) Ordinary			(54) Ordinary	(565) Ordinary		(798) Ordinary		١		(184) Ordinary		(82) Ordinary		(213) Ordinary								(30) Ordinary			(50) Ordinary	(55) Ordinary				(35) Ordinally		(247) Ordinary		(149) Ordinary			(54) Ordinary				(679) Ordinary					(56) Ordinary		(591) Ordinary			(33) Ordinary							(SA)									
455	169	2,960	212,500		750	667	20404	5,615	497	1,595	455	4,754	5.680	6715	000 0	2,800	840	1,549	5,2/9	069	278	1,793	6,500	1,044	27.2	3,520	180,7	191	150//	557	3,722	3,048	421	403	E24 E30 V	(98'+	006	150	2040	2,080	6 509	1,255	6,003	5,995	455	5,286	3,707	2,224	5,721	295	5,156	4,836	514	472	3.842	4.979	59	930	5/7	194	4,550	5,114	5,257	5,004	278	755	400	4,389							
Sell	Sell	Sell	Sell	las	760	-	liac liac	Hac .	lisc.	Seil	Sell	Sell								Sell			Sell	Sell	Sell		Jes 5		136	liec	Sell		ii o	lloc o	llec o	lle3	iec o	1705	Sell	lles	Sell	les	Sell	Sell	Sell	[es]	Sell	J	35	30		Sell	Sell	Sell	Sell	Sell	Sell	las		Sell	Sell	Sell Sell	Sell Sell Sell Sell Sell	Sell Sell Sell Sell Sell Sell Sell Sell	las	las Sellas									
UBS Securities Australia Ltd	IDS Countries Australia 110	The second secon	Obs securities Australia tita	OBS Securities Australia Lid	OBS Securities Australia Ltd	UBS Securities Australia Ltd	190 Securities Australia 14d	Obs. Describings Appendix Ltd	Obs Securities Australia Ltd	UBS Securities Australia Ltd	USC Cocurition Australia 144	OSS SECULINES AUSTRAINS LICE	UDS Securities Australia Liu	Obs Securities According to	UBS Securities Australia Ltd	DE Counties Australia Ltd	DDS Securities Australia Ltd	DC Countries Australia Ltd	DDS Securities Australia Ltd	DDS Securities Australia Ltd	DC Cocurition Australia 14d	185 Securities Australia Ltd	URS Securities Australia Itcl	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	IDS Committee Australia led	OBS Securities Australia Ltd	JBS securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	IDC Commission A retrain 144	JBS Securities Australia Lita	JBS Securities Australia Ltd	UBS Securities Australia Ltd UBS Securities Australia Ltd	UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia trd UBS Securities Australia Ltd	UBS Securities Australia trd	USDS Secturines Australia Ltd. USS Securities Australia Ltd.	UBS Securities Australia Ltd																														
14-Oct-14	П	14-Oct-14	Γ	Γ	Ī	T	T	14-001-14	Ī					14-04-14		Т		-	1	1	-	14-Oct-14	-	-	ł	ı	14-Oct-14	ı	T		T	Ī	T	Ť	Ť	140414		14 04 14		l	ŀ		14-Oct-14				14-Oct-14	14-Oct-14	14-Oct-14	Γ	Г	Ī	Ť	Ī	Ī	Ī			4-001-14					Γ											

(89) Ordinary							,232) Ordinary		L	L	L							(2,210) Ordinary			(320) Ordinary			(25) Ordinary	748) Ordinary				(375) Ordinary				١	ı	(83) Ordinary								(62) Ordinary					İ	(55) Ordinary		(362) Ordinary		(2) Ordinary						1												(12) Ordinary (128) Ordinary (128) Ordinary (129) Ordinary (129) Ordinary (129) Ordinary (139) Ordinary (139) Ordinary (129) O
750							10,386																	070		6.790			3,165									1,943			1 167				L)		4,323		466					3,895																	
	5,7		4.8		5,1		10,	7	2,5	2,0	2					77	77	181	13,	15.2	2,699	,			-	9		6,7	3,1	4,0		1,7		3,1		3,5	.,,		139			2.5		4,5	16,156	2	4,3	4,5	7		3,6	1,6		3,8	2,5	2,7		5'5		2,3	2,3	2,3	2,3	2,3	2,3 1 5 5,8 1	2,3	2,3 1 1 5,8 5,8 5,8 6,3 2,7 2,7 2,7	5.3 5.8 5.8 5.8 5.8 6.3 7.7 7.7 7.7 7.7	2.3 5.8 5.8 6.3 7.7 4.1	2.3	23.00
Sell	Sell	Sell	Sell	Sell	Seli	Sell	Sell	Sell	Sell	Sell	Sett	) Joseph	Cod	1961	Jell Cell	E-X	Jell Sell	le S		le C	Jan 3	lle0	100	<u> </u>	les.	Sell	Sell	Seil	Sell	Seli	Sell	Sell Sell	Sell	E S	E S	- N	- T	les o	وا	وا	J.	les	Sell	Sell	Sell	Sell	Sell	Sell	Sell	Sell	Sell	Sell	Sell	Sell	Sell	Sell	Sell	Sell	Cell	3	les.	les Jes	Sell Sell	Sell Sell	Sell Sell Sell	26   26   36   36   36   36   36   36		Sel			
a Ltd	a Ltd	a Ltd	a Ltd	a Ltd	altd	a Ltd	a Ltd	a Ltd	a Ltd	a Ltd	a Ltd	altd	114	110	114	9 110	9 110	0110	a rid	a Ltd	a Ltd	0114	a Itd	11d	altd	a Ltd	a Ltd	a Ltd	a Ltd	a Ltd	b ttd	a ttd	a Ltd	e utd	e Ltd	S Ltd	a Lita	Tion a ring	114	ltd	150	) Itd	ttd	b Ltd	e ttd	a Ltd	a Ltd	ltd	Ltd	ttd	Ltd	ltd	Ltd	ltd	ttd	Ltd	Ltd	Ltd	Ltd		, Ltd	ı Ltd	ı Lid Lid Lid	1 trd 1 trd 1 trd 1 trd	11d 11d 11d 11d 11d	1	114 114 117 117 117 117 117	11q 11q 11q 11q 11q 11q	114 114 117 117 117 117 117 117 117 117	1149 1119 1119 1119 1119 1119 1119 1119	1114 1114 1114 1114 1114 1114 1114 111
UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australi	UBS Securities Australi	UBS Securities Australia Ltd	UBS Securities Australi	UBS Securities Australia Ltd	UBS Securities Australi	UBS Securities Australia Ltd	I BS Securities Australi	130 Securities Australia Ltd	I IBC Securities Augustial	I IDC Counties Australia	UBS Securities Australia Life	100 Securities Australia Ltd	UBS Securities Australi	UBS Securities Australia Ltd	UBS Securities Australia Ltd	I IBS Securities Australia	LIBS Securities Australi	UBS Securities Australia	UBS Securities Australia	UBS Securities Australi	UBS Securities Australia Ltd	UBS Securities Australia	UBS Securities Australi	UBS Securities Australia Ltd	UBS Securities Australi	UBS Securities Australia Ltd	UBS Securities Australia	UBS Securities Australia Ltd	I IBS Securities Australia Ltd		UBS Securities Australia Ltd	UBS Securities Australia Ltd	<b>UBS Securities Australia Ltd</b>	UBS Securities Australia	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia		UBS Securities Australia	UBS Securities Australia Ltd UBS Securities Australia Ltd	UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd UBS Securities Ltd UBS Securities Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia	188 Securites Australia Ltd 188 Securities Australia Ltd																			
15-Oct-14	15-0ct-14	15-Oct-14	15-0ct-14	15-0ct-14	15-0ct-14	15-0ct-14	15-0ct-14	15-Oct-14	15-Oct-14	15-0ct-14	15-0ct-14	15-Oct-14	15-Oct.14	15.04.14	15.04.14	1000	\$1.00 cc	15 04 14	15 Oct 14	15-Oct-14	15 Oct 14		l		l				15-Oct-14	-	15-0ct-14	15-Oct-14	15-0ct-14	15-00-14	15-04-14	13-Oct-14	12-00t-14	15.04.14	15-04-14	15-04-14	15-Oct-14	15-Oct-14	15-Oct-14	15-Oct-14	15-0ct-14		- 1	-	١		15-0ct-14	-	-	-																	

		######################################
	-	
1		
	1 1	
	. 1	
	- 1	
	1	
_		
		Sell
		Sell Sell
		Sell
		Je J
	- 1	Sell
		Sell
	1	<u></u>
	ı	Sell
	ı	
	1	
	1	lles les
		Sell Sell
	1	Sell Sell Sell
	1 1	Sell Sell Sell Sell Sell Sell Sell Sell

				l
UBS Securities Australia Ltd	Sell	195	(23)	
UBS Securities Australia Ltd	Sell	4,752	(562)	Ordinary
UBS Securities Australia Ltd	Sell	93		Ordinary
UBS Securities Australia Ltd	Sell	3,452		Ordinary
UBS Securities Australia Ltd	Sell	17,241	(2,038)	Ordinary
UBS Securities Australia Ltd	Sell	1,362		Ordinary
UBS Securities Australia Ltd	Sell	390		
Upo Securities Australia Liu	Det.	ope, y	(51,1)	
nes secunities Australia Lita	nac .	4,423		١
UBS Securities Australia Ltd	) Seit	8EC,2		1
Upo percuines Australia Ltd	liac .	/27,1		
Ups securines Australia Ltd	nec .	190		
UBS Securities Australia Ltd	Sell	34		
UBS Securities Australia Ltd	Sell	1,311		Ordinary
UBS Securities Australia Ltd	Sell	11,463		Ordinary
UBS Securities Australia Ltd	Sell	2,115	(250)	Ordinary
UBS Securities Australia Ltd	Sell	4,721		Ordinary
UBS Securities Australia Ltd	Sell	2,919		Ordinary
UBS Securities Australia Ltd	Sell	1,083		Ordinary
UBS Securities Australia Ltd	Sell	4,670	(883)	Ordinary
UBS Securities Australia Ltd	Sell	262	(31)	Ordinary
UBS Securities Australia Ltd	Sell	1,714	)	Ordinary
UBS Securities Australia Ltd	les.	4.738	(561)	Ordinary
UBS Securities Australia Ltd	les	866 2		Ordinary
UBS Securities Australia Ltd	les	4 687	(555)	Ordinary
LIRS Securities Australia Ltd	3	4,667		Ordinary
LIRS Securities Australia 11d	9	3 203	(379)	Ordinary
I IRS Securities Australia I to	Coll	735		Ordinary
II BS Securities Australia I to	190	752		Ordinary
JIRS Securities Australia 11d	Sell S	902	(09)	Ordinary
I IRS Securities Australia 11d		A 202	(00)	Crainary
I IRC Cocurities Australia 14d	100	7,233		Ordinary
LIBS Securities Australia 14d	lac.	7 521		Ordinary
Tipe Countries Australia Ltd	liac	1,331	(000)	Ordinary
Opp Securities Australia Ltd	liac liac	2,634		Ordinary
Con Securities Australia Liu	200	334		Ordinary
Cop securities Australia Ltd	Sell	280,0		Ordinary
Obs securities Australia Ltd	) Sell	1///9		Ordinary
UBS Securities Australia Ltd	Sell	12,204	(1,446)	Ordinary
Obs securities Australia Ltd	Sell	431		Ordinary
UBS securities Australia Ltd	) Sell	7,043		Ordinary
UBS Securities Australia Ltd	Nell Nell	2,531		Ordinary
UBS Securities Australia Ltd	Sell	658		Ordinary
UBS Securities Australia Ltd	Self	4,971		Ordinary
UBS Securities Australia Ltd	Sell	4,819		Ordinary
UBS Securities Australia Ltd	Seli	3,758		Ordinary
UBS Securities Australia Ltd	Sell	11,197	(1,329)	Ordinary
UBS Securities Australia Ltd	Sell	7,587	(006)	Ordinary
UBS Securities Australia Ltd	Sell	3,490	(414)	Ordinary
UBS Securities Australia Ltd	Sell	7,587	(006)	Ordinary
UBS Securities Australia Ltd	Sell	3,072	(364)	Ordinary
UBS Securities Australia Ltd	Sell	219	(52)	Ordinary
UBS Securities Australia Ltd	Sell	464	(55)	Ordinary
UBS Securities Australia Ltd	Sell	2.532	(300)	Ordinary
UBS Securities Australia Ltd	Sell	4,566	(541)	Ordinary
UBS Securities Australia Ltd	Sell	110	(13)	Ordinary
UBS Securities Australia Ltd	Sell	135	(16)	Ordinary
UBS Securities Australia I td	P	7 198	(841)	Ordinan
UBS Securities Australia Ltd	98	3517	(417)	Ordinan
UBS Securities Australia (td	Seil	261	(31)	Ordinary
HBS Securities Australia Ltd	Jes	781	(5)	Ordinan
I IRS Societies Australia Ltd	Call	2007	(600)	Ordinary
LIBS Securities Australia Ltd	Cell	2,03	(040)	Ordinary
IIRS Securities Australia 11d	ag S	200	(01)	Ordinary
UBS Securities Australia Ltd	i de	2 524	(300)	Codinary
UBS Securities Australia I to	195	387	(97)	Ordinary
LIBS Securities Australia Hol	100	2 540	(606)	Ordinary
IRC Securities Australia Ltd	Coll	Shr'7	(coc)	Ordinary
Upo Committee Australia Ltd	IIAC	477	(C)	Ordinary
UBS Securities Australia Ltd	Sell	2,703	(057)	Ordinary
UBS Securities Australia Ltd	Sell	740	(88)	Ordinary
UBS Securities Australia Ltd	Sell	13,305	(1,582)	Ordinary
UBS Securities Australia Ltd	Sell	800	(62)	Ordinary
UBS Securities Australia Ltd	Sell	749	(68)	Ordinary
PID Counties America 144	100			
FIDS Contraction Assets and				-

Sell	UBS Securities Australia Ltd	Sell	2,616	(118)	Ordinary
Self         76           Self         3.226           Self         2.02           Self         2.03           Self         2.04           Self         2.04           Self         2.04           Self         2.04           Self         2.04           Self         2.04           Self <td>Securities Australia Ltd</td> <td>Sell</td> <td>656</td> <td>(82)</td> <td>Ordinary</td>	Securities Australia Ltd	Sell	656	(82)	Ordinary
Self   3029     Self   3029     Self   3029     Self   2,002     Self   2,023     Self	Securities Australia Ltd	Self	76	(6)	Ordinary
Self   240     Self   2,103     Self   2,105     Self	Securities Australia Ltd	Self	3,229	(384)	Ordinary
Sell         740           Sell         2,133           Sell         2,523           Sell         2,523           Sell         2,449           Sell         2,449           Sell         2,449           Sell         3,263           Sell         3,452           Sell         3,462           Sell         3,472           Sell         3,472           Sell         3,473           Sell         3,377           Sell         3,377           Sell         3,377           Sell         3,377           Sell         3,377           Sell         3,377           Sell         3,420           Sell         3,420           Sell         3,420           Sell         3,420           Sell         3,420           Sell         3,420           Sell         3,420 <td>Securities Australia Ltd</td> <td>Sell</td> <td>202</td> <td>(09)</td> <td>Ordinary</td>	Securities Australia Ltd	Sell	202	(09)	Ordinary
Sell         2.103           Sell         2.523           Sell         2.523           Sell         2.523           Sell         2.627           Sell         2.627           Sell         3.265           Sell         3.265           Sell         3.265           Sell         2.022           Sell         2.03           Sell         2.03           Sell         2.03           Sell         2.03           Sell         3.317           Sell         3.32           Sell         3.317           Sell         3.32           Sell         3.317           Sell         3.32           Sell         1.92           Sell         1.23           Sell         2.34           Sell         2.34           Sell         2.34           Sell         2.34           Sell         2.105           Sell         2.105           Sell         2.115           Sell         2.115           Sell         2.115           Sell         2.115	Securities Australia Ltd	Sell	740	(88)	Ordinary
Sell         2,523           Sell         2,523           Sell         2,627           Sell         2,627           Sell         2,627           Sell         2,627           Sell         3,262           Sell         3,262           Sell         3,022           Sell         2,023           Sell         3,142           Sell         3,37           Sell         3,37           Sell         3,37           Sell         3,37           Sell         1,73           Sell         3,37           Sell         1,427           Sell         1,427           Sell         1,427           Sell         1,427           Sell         1,427           Sell         1,427           Sell         2,329           Sell         2,132	Securities Australia Ltd	Sell	2,103	(250)	Ordinary
Sell         2523           Sell         2523           Sell         2,627           Sell         2,627           Sell         2,627           Sell         2,627           Sell         2,627           Sell         3,452           Sell         3,452           Sell         7,33           Sell         2,623           Sell         2,524           Sell         3,37           Sell         3,37 <t< td=""><td>Securities Australia Ltd</td><td>Sell</td><td>121</td><td>(2)</td><td>Ordinary</td></t<>	Securities Australia Ltd	Sell	121	(2)	Ordinary
Sell         657           Sell         2627           Sell         2449           Sell         3.265           Sell         160           Sell         3.265           Sell         160           Sell         3.062           Sell         20.533         CG           Sell         20.533         CG           Sell         20.02         3.37           Sell         20.02         3.37           Sell         3.37         3.37           Sell         3.37         3.44           Sell         3.37         3.42           Sell         3.37         3.42           Sell         3.37         3.42           Sell         1.912         3.42           Sell         1.912         3.42           Sell         2.34         4.42           Sell         2.34         3.42         4.42           Sell         2.105         2.42           Sell         2.105         2.42           Sell         2.105         2.42           Sell         2.105         2.42           Sell         2.105 <th< td=""><td>Securities Australia Ltd</td><td>Sell</td><td>2,523</td><td>(300)</td><td>Ordinary</td></th<>	Securities Australia Ltd	Sell	2,523	(300)	Ordinary
Sell         2.627           Sell         2.627           Sell         2.449           Sell         3.265           Sell         3.265           Sell         3.265           Sell         3.265           Sell         2.0623         CR           Sell         7.42         2.05           Sell         7.23         2.06           Sell         7.23         2.52           Sell         7.23         3.062           Sell         3.317         3.317           Sell         3.317         3.31           Sell         3.317         3.31           Sell         3.31         4.42           Sell         3.31         4.43           Sell         3.31         4.43           Sell         3.31         4.43           Sell         3.31         4.43           Sell         3.32         3.32           Sell         3.32	Securities Australia Ltd	Sell	657	(28)	Ordinary
Self         547           Self         5448           Self         2,449           Self         120           Self         3,462           Self         160           Self         3,462           Self         3,472           Self         3,422           Self         3,420           Self         3,420           Self         1,231           Self         1,231           Self         1,232           Self         1,242           Self         1,382           Self         1,382           Self         2,340           Self         2,342           Self         1,372           Self         2,342           Self         2,342           Self         2,342           Self         2,105           Self         2,105      Self         2,132	Securities Australia Ltd	Sel	2 627	(312)	Ordinary
Self   3,249   C     Self   3,265   C     Self   3,265   C     Self   3,265   C     Self   3,765   C     Self	Securities Australia 11d	Sell	547	(65)	Ordinac
Self   3,265   6,65     Self   3,27     Self   3,27	Securities Australia Ltd	a S	2 449	(102)	Ordinary
Self   3,252     Self   3,452     Self   2,062     Self   2,062     Self   2,053     Self   2,054     Self   2,054     Self   3,317     Self	Securities Australia Ltd	Coll	100	(00)	(in in
Self	Commission Australia I and	1170	100	(600)	Oldinary C
Self	Secondary Australia Lid	nec .	5,203	(000)	Ordinary
Self   2.0529   C2     Self   2.052   C2     Self   2.052   C2     Self   2.052   C3     Self   2.052   C4     Self   2.052	securines Australia Ltd	Sell	091	(61)	Ordinary
Self   2,052   Co     Self   2,053   Co     Self   2,053   Co     Self   2,053   Co     Self   3,17   Co     Sel	Securities Australia Ltd	Sell	3,452	(410)	Ordinary
Self   20,503   C2,     Self   2,10   C3,     Self   2,10   C4,     Self   3,10   C4,	Securities Australia Ltd	Sel	2,062	(245)	Ordinary
Self   20,503   C2,     Self   2,134   C3,     Self   2,134   C4,     Self   2,134   C4,     Self   2,134   C4,     Self   2,134   C4,     Self   2,132   C4,	Securities Australia Ltd	Sell	379	(45)	Ordinary
Self   249   249   240   254	Securities Australia Ltd	Seil	20.503	(2 435)	Ordinary
Self         210           Self         210           Self         2.534           Self         3.062           Self         3.062           Self         3.17           Self         3.31           Self         3.31           Self         3.31           Self         3.37           Self         3.37           Self         3.31           Self         2.340           Self         2.350           Self         2.350           Self         2.350           Self         2.350           Self         2.350 <tr< td=""><td>Securities Australia Ltd</td><td>lleS</td><td>5749</td><td>(08)</td><td>Ordinary</td></tr<>	Securities Australia Ltd	lleS	5749	(08)	Ordinary
Self   2.534     Self   1.731     Self   1.731     Self   1.731     Self   1.731     Self   1.427     Self   1.427     Self   1.432     Self   1.433     Self   1.433     Self   1.433     Self   1.433     Self   1.433     Self   1.435     Self	Securities Australia I tol	ll eo	010	(30)	(10)
Self   2,534     Self   1,731     Self   1,731     Self   1,731     Self   1,731     Self   1,912     Self   1,913     Self   1,913     Self   1,230     Self   1,213     Self	Security Automotive and	liac liac	OLZ	(67)	Ordinary
Sell	Securities Australia Ltd	Sell	56	(11)	Ordinary
Sell         772           Sell         782           Sell         3.062           Sell         3.317           Sell         3.317           Sell         3.317           Sell         3.317           Sell         3.317           Sell         4.22           Sell         1.912           Sell         1.912           Sell         2.340           Sell         2.350	Securities Australia Ltd	Sell	2,534	(301)	Ordinary
Sel         752           Sel         3.062           Sel         886           Sel         3.37           Sel         3.37           Sel         3.37           Sel         3.37           Sel         4.47           Sel         1.888           Sel         4.02           Sel         1.888           Sel         2.349           Sel         2.349           Sel         2.349           Sel         2.120           Sel         2.132           Sel         1.512           Sel	Securities Australia Ltd	Sell	1,731	(502)	Ordinary
Sell         3,002           Sell         866           Sell         3,317           Sell         3,317           Sell         3,317           Sell         3,317           Sell         1,912           Sell         1,912           Sell         1,912           Sell         1,912           Sell         1,912           Sell         1,821           Sell         2,349           Sell         3,12           Sell         1,821           Sell         2,105           Sell         2,132	Securities Australia Ltd	Sell	752	(68)	Ordinary
Sell         886           Sell         3.37           Sell         3.37           Sell         3.37           Sell         3.37           Sell         1.888           Sell         1.888           Sell         1.883           Sell         2.349           Sell         2.349           Sell         2.349           Sell         2.349           Sell         2.340           Sell         2.330           Sell         2.330           Sell         2.330           Sell         2.132           Sell         2.132           Sell         1.912           Sell         1.512           Sell         1.512           Sell         1.573           Sell         1.575           Sell         1.575	Securities Australia Ltd	Sell	3.062	(E9E)	Ordinan
Self   3,317     Self   3,317     Self   3,317     Self   3,317     Self   4,20     Self   4,20     Self   4,20     Self   4,20     Self   3,012     Self   2,132     Self   3,132     Self   3	Securities Australia I td	lles	886	(105)	Ordinan
Self   3,377     Self   9,022     Self   9,022     Self   1,427     Self   2,132     Self   2,132     Self   2,132     Self   2,132     Self   2,132     Self   1,273     Self	Cocurities Australia 14d	100	35	(COL)	Cidilaly
Self   3.317     Self   3.317     Self   3.317     Self   3.317     Self   3.317     Self   3.318     Self	Securities Australia Liu	liac .	57	(5)	Ordinary
Sell         9,024           Sell         9,024           Sell         1,427           Sell         1,427           Sell         1,427           Sell         1,427           Sell         1,912           Sell         1,912           Sell         2,103           Sell         2,105           Sell         2,105           Sell         2,105           Sell         2,102           Sell         2,103           Sell         2,103           Sell         2,103           Sell         2,102           Sell         2,132           Sell         2,132           Sell         2,132           Sell         2,132           Sell         2,132           Sell         1,512           Sell         1,512           Sell         1,513           Sell         1,513 </td <td>securines Australia Ltd</td> <td>Sell</td> <td>3,317</td> <td>(565)</td> <td>Ordinary</td>	securines Australia Ltd	Sell	3,317	(565)	Ordinary
Self   8,002     Self   1,427     Self   2,349     Self   2,136     Self   2,132     Self   3,132     Self	Securities Australia Ltd	Sell	34	(5)	Ordinary
Seri   1,823	Securities Australia Ltd	Sell	9,022	(1,069)	Ordinary
Self   1,427     Self   1,598     Self   1,598     Self   2,349     Self   2,140     Self   3,042     Self   3,042     Self   3,042     Self   2,132     Self   1,230     Self	Securities Australia Ltd	Sell	8,043	(623)	Ordinary
Sell   1,996     Sell   1,972     Sell   1,972     Sell   2,249     Sell   3,014     Sell   3,014     Sell   3,014     Sell   3,014     Sell   2,132     Sell   1,230     Sell	Securities Australia Ltd	Self	1,427	(691)	Ordinary
Self   422     Self   2.449     Self   2.449     Self   2.449     Self   2.449     Self   2.449     Self   2.105     Self   2.105     Self   2.105     Self   2.105     Self   2.105     Self   2.105     Self   2.132     Self   1.215     Self	Securities Australia Ltd	Sell	1.898	(222)	Ordinary
Self   1,912     Self   2,340     Self   2,340     Self   3,042     Self   3,042     Self   3,042     Self   3,105     Self   1,821     Self   1,821     Self   2,132     Self   1,230     Self	Securities Australia Ltd	Sell	422	(50)	Ordinary
Self   1,249     Self   2,349     Self   2,349     Self   3,014     Self   2,102     Self   1,215     Self	Securities Australia 14d	3	1 913	Ecc.	in in
Self   2,349   0   1   1   1   1   1   1   1   1   1	Dia management and a second and	100	216,1	(777)	Ordinary
Self   8.430	Securities Australia Ltd	Sell	404	(48)	Ordinary
Self   8,420 (1)     Self   1,821 (1)     Self   1,105 (1)     Self   1,105 (1)     Self   2,105 (1)     Self   2,105 (1)     Self   2,132 (1)     Self   3,132 (1)     Self	securities Australia Ltd	seit	2,349	(279)	Ordinary
Sell   1,3014   1,5	Securities Australia Ltd	Seil	8,420	(1,000)	Ordinary
Sell   1921	Securities Australia Ltd	Sell	3,014	(358)	Ordinary
Self   2,105     Self   7,17     Self   7,18     Self   7,18     Self   7,19     Self   7,19	Securities Australia Ltd	Sell	1,821	(216)	Ordinary
Self   177   178   188	Securities Australia Ltd	Sell	2.105	(250)	Ordinary
Self   244     Self   247     Self   248     Self   273     Self   2,082     Self   2,132     Self   2,132     Self   2,132     Self   2,132     Self   2,132     Self   2,132     Self   1,230     Self   1,230	Securities Australia I to	(e)	127	(14)	Coding
Self	Securities Australia I to		770	(00)	Ordinaly
Self	Securities Australia 14d	100	110	(67)	Ordinary
Sell	Secondes Assemble and	ilac i	01.	(41)	Ordinary
Sell   6.312     Sell   2.082     Sell   2.082     Sell   2.132     Sell   2.115     Sell   2.115     Sell   2.115     Sell   1.230     Sell   1.213     Sell	Securities Australia Ltd	Sell	5/3	(89)	Ordinary
Sell   2,002     Sell   2,300     Sell   2,132     Sell   2,132     Sell   2,132     Sell   2,132     Sell   2,132     Sell   2,132     Sell   1,23     Sell	Securities Australia Ltd	Sell	8,312	(986)	Ordinary
Sell         2,082           Sell         2,082           Sell         2,132           Sell         2,135           Sell         2,136           Sell         2,137           Sell         1,572           Sell         1,572           Sell         1,573           Sell         2,61           Sell         2,61           Sell         1,230           Sell         1,230           Sell         1,233           Sell         1,235           Sell         1,235           Sell         1,235           Sell         1,235           Sell         1,235           Sell         1,535	Securities Australia Ltd	Sell	29	(8)	Ordinary
Sell         2,300           Sell         2,132           Sell         1,873           Sell         2,132           Sell         2,132           Sell         4,37           Sell         1,52           Sell         1,23           Sell         2,13           Sell         2,61           Sell         2,61           Sell         1,23           Sell         1,23           Sell         1,23           Sell         1,213           Sell         1,213           Sell         1,213           Sell         1,019	Securities Australia Ltd	Seit	2,082	(247)	Ordinary
Sell   2,132     Sell   2,132     Sell   2,132     Sell   2,145     Sell   2,145     Sell   2,145     Sell   1,230     Sell	Securities Australia Ltd	Sel	2 300	(822)	Ordinary
Self   1,662     Self   2,132     Self   2,115     Self   1,115     Self   1,120     Self   1,20     Self	Securities Australia Ltd	les	2 132	(523)	Ordinary
Self   2,132     Self   2,132     Self   2,115     Self   1,912     Self   1,230     Self   1,235     Self	Cacurities Australia 14d	liv3	1000	(100)	Commany
Sell	Committee A second data	1100	200,1	(177)	Contain
Sell   2.115     Sell   1.912     Sell   1.912     Sell   2.87     Sell   1.290     Sell   1.290     Sell   1.290     Sell   1.291     Sell   1.575     Sell   1.575     Sell   1.578     Sell	Securities Australia Liu	)ell	7,132	(553)	Orginary
Sell   1,912     Sell   447     Sell   447     Sell   1,230     Sell   1	Securities Australia Ltd	Sell	270	(32)	Ordinary
Sell   1912     Sell   287     Sell   287     Sell   1230     Sell   251     Sell   1578     Sell   1218     Sell   1578     Sell   1218     Sell   1218     Sell   1218     Sell   1218     Sell   1218     Sell   1218     Sell   1019	Securities Australia Ltd	Sell	2,115	(221)	Ordinary
Self   447   287	Securities Australia Ltd	Seli	1,912	(722)	Ordinary
Self   287   152	Securities Australia Ltd	Sei	447	(53)	Ordinary
Self   152     Self   1230     Self   1230     Self   1213     Self   1213     Self   1213     Self   1213     Self   1213     Self   1019     Self   1019	Securities Australia Ltd	les	787	We	Ordinary
Sell   1,230     Sell   1,230     Sell   251     Sell   51     Sell   1,575     Sell   1,575     Sell   817     Sell   1,019     Sell   1,01	Securities Australia 11d		153	(01)	6
Sell   1,230	Securities Australia Ltd	Dell Communication	70.	(01)	Ordinary
Sell   261	securities Australia Ltd	Sell	1,230	(146)	Ordinary
Sell   Sell   Sell   15.75   Sell	Securities Australia Ltd	Sell	261	(31)	Ordinary
Self   1,575   Self   1,273   Self   1,273   Self   1,213   Self   1,015   Self	Securities Australia Ltd	Sell	51	(9)	Ordinary
Sel   1,213	Securities Australia Ltd	Self	1,575	(187)	Ordinary
Self   93   817     817       817	Securities Australia Ltd	Sell	1,213	(144)	Ordinary
Sell   1,03   Sell   1,03   Sell   1,03   Sell   1,039	Securities Australia 14d	log	60	/44/	
Sell 2.105 Sell 1,019 Sell 4,352	Securities Australia Ltd	liac liac	SS II	(11)	Ordinary
Sell 2,105   Sell 2,105   Sell 4,362   Sel	ecurines Australia Ltd	Sell Sell	81/	(76)	Ordinary
Sel 1,019 Sel 4,362	securities Australia Ltd	Sell	2,105	(250)	Ordinary
Sell 4,362	securities Australia Ltd	Sell	1,019	(121)	Ordinary
	Securities Australia Ltd	Sell	4,362	(518)	Ordinary
Sel	UBS Securities Australia Ltd	lles	7 300	(867)	Ordinary
0000	To a mixture A make the latest		52.	(100)	S CELED S
Sell Sell	ecurities Australia Ltd	Se			
1955	The American Park		50	(£)	Ordinary

Sell Sell Sell Sell Sell Sell Sell Sell	UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd
Sell Sell	1 1 1 1
E E	
Jas Jas	
ķ	
-	
3	
les.	
les	
Nell Sell	
Sell	
Sell Sell	
Sell Sell	
Sell	
Sell	
Sell	
Set	
Sell	
Jes.	
Sel	
ll d'S	
Je Je	
105	l
il d'y	
E Contraction	l
170	l
120	İ
, Se	1
E .	
Je j	1
Sell	
- N	
, .	
Zell	
Zell Zell	1
la l	1
J.	l
ē, i	
la Coll	
100	
II O	1
3	ı
Sell	1
<b>3</b>	
وا	1
Į	1
Call	1
100	П
# P	1
nac S	П
H .	
Jell Nell	1
,	-
Sell	-
Zell	
Sell	
Sell	
Sell	
leg.	
100	١
100	
III.	ı
Sell	
Sell	١
Sell	
<u>i</u>	
195	l
38	ı
<u>s</u>	1
وا	
- - - - - - - - - - - - - - - - - -	- 1
Sell	
	8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8

150.014   100 Security Accordance Administry A   150.014   150.0	71-17			75	3	
565 Securiter Australia Ltd   561   252   202	5-0ct-14	UBS Securities Australia Ltd	Sel	15 577	(1.850)	
Dist Scurringer Australia Ltd   Self   2.506   CR07     Dist Scurringer Australia Ltd   Self   Sel	5-Oct-14	UBS Securities Australia Ltd	Sell	2,105		
Dis Securiter Australia Ltd   Self   5,949   1,549	5-Oct-14	UBS Securities Australia Ltd	Sell	2,526		
Dis Schmidter Autrainis Ltd   Self    5-Oct-14	UBS Securities Australia Ltd	Sell	2,080			
Bits Securine Auxilia Let   Self	5-Oct-14	UBS Securities Australia Ltd	Sell	696'6	٤	
Dis Scennine Autralia   100   541   55.204   650   6	5-Oct-14	UBS Securities Australia Ltd	Sell	29		
USS Securine Automala Led         Seal         S, 044         6509           USS Securine Automala Led         Seal         S, 159         (662)           USS Securine Automala Led         Seal         S, 159         (663)           USS Securine Automala Led         Seal         S, 150         (673)           USS Securine Automala Led         Seal         S, 150         (673)           USS Securine Automala Led         Seal         S, 150         (673)           USS Securine Automala Led         Seal         S, 150         (73)           USS Securine Automala Led         Seal <t< td=""><td>5-Oct-14</td><td>UBS Securities Australia Ltd</td><td>Sell</td><td>5,204</td><td></td><td></td></t<>	5-Oct-14	UBS Securities Australia Ltd	Sell	5,204		
USS Securines Automate Lot         Sed         5,100         (652)           USS Securines Automate Lot         Sed         5,100         (652)           USS Securines Automate Lot         Sed         5,547         (653)           USS Securines Automate Lot         Sed         5,547         (653)           USS Securines Automate Lot         Sed         5,547         (673)           USS Securines Automate Lot         Sed         5,742 <td>5-Oct-14</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>5,044</td> <td></td> <td></td>	5-Oct-14	UBS Securities Australia Ltd	Sell	5,044		
USS Securine Automala LO         SM         \$159         (61)           USS Securine Automala LO         SM         \$150         (61)           USS Securine Automala LO         SM         \$150         (61)           USS Securine Automala LO         SM         \$150         (61)           USS Securine Automala LO         SM         \$10         \$10           USS SECURE Automala LO         SM         \$10         \$10           USS SECURDA CONTRICA Automala LO	-Oct-14	UBS Securities Australia Ltd	Set	5,100		
USS Securine Automise to 10 bits         Death of 15 bits         5 bits         5 bits         15 bits	000-14	UBS Securities Australia Ltd	Sell	5,159		
ONS Securine Automise to 10 Sept         SEP SEP SEP SEP SEP SEP SEP SEP SEP SEP	0-0ct-14	Obsider Australia Etc	liac li	5,547		
(10)         (10)         (10)         (10)           (10)         (10)         (10)         (10)         (10)           (10)         (10)         (10)         (10)         (10)         (10)           (10)         (10)         (10)         (10)         (10)         (10)         (10)           (10)         (10)         (10)         (10)         (10)         (10)         (10)         (10)           (10)	-Oct-14	ODS Securities Australia Ltd	Sell S	016,6		
(15)         (15)         (15)         (15)           (15)         (15)         (15)         (15)         (15)           (15)         (15)         (15)         (15)         (15)           (15)         (15)         (15)         (15)         (15)           (15)         (15)         (15)         (15)         (15)           (15)         (15)         (15)         (15)         (15)           (15)         (15)         (15)         (15)         (15)           (15)         (15)         (15)         (15)         (15)           (15)         (15)         (15)         (15)         (15)           (15)         (15)         (15)         (15)         (15)           (15)         (15)         (15)         (15)         (15)           (15)         (15)         (15)         (15)         (15)           (15)         (15)         (15)         (15)         (15)           (15)         (15)         (15)         (15)         (15)           (15)         (15)         (15)         (15)         (15)           (15)         (15)         (15)         (15)         (15)	04.14	USS Securities Australia Ltd	Sell	/9 P		
USS Securities Automita Ltd         Seal         2.7.2         COS           USS Securities Automita Ltd         Seal         4.34         (5.9)           USS Securities Automita Ltd         Seal         4.33         (4.0)           USS Securities Automita Ltd         Seal         5.645         (6.7)           USS Securities Automita Ltd         Seal         5.702         (6.7)           USS Securities Automita Ltd         Se	20ct-14	UBS Securities Australia Ltd	llac	900,1		
USS Securities Autoritation	04 14	DDS Securities Australia Ltd	liac o	9/9/5		
USS Securities Australia Ltd         Seil         A.73         (49)           USS Securities Australia Ltd         Seil         5.45         (40)           USS Securities Australia Ltd         Seil         5.645         (612)           USS Securities Australia Ltd         Seil         5.645         (617)           USS Securities Australia Ltd         Seil         5.642         (617)           USS Securities Australia Ltd         Seil         5.643         (617)           USS Securities Australia Ltd         Seil         5.641         (617)           USS Securities Australia Ltd         Seil         5.641         (617)           USS Securities Australia Ltd         Seil         5.641         (618)           USS Securities Australia Ltd <td>170</td> <td>LIDE Securities Australia Lid</td> <td>liec S</td> <td>214</td> <td></td> <td></td>	170	LIDE Securities Australia Lid	liec S	214		
URS Securities Anatolist Ltd         Seil         6.13         (6.12)	04-14	LIBS Securities Australia 14d	les of	32		
USS Securities Australia List         Sell         5,545         (6/20)           USS Securities Australia List         Sell         5,645         (6/20)           USS Securities Australia List         Sell         5,545         (6/20)           USS Securities Australia List         Sell         5,543         (6/20)           USS Securities Australia List         Sell         2,522         (6/20)           USS Securities Australia List         Sell         2,523         (6/20)           USS Securities Australia List         Sell         2,523         (6/20)           USS Securities Australia List         Sell         5,230         (6/20)           USS Securities Australia List         Sell         5,230         (6/20)           USS Securities Australia List         Sell         5,240         (7/20)           USS Securities Australia List         Sell         5,240         (7/20) <t< td=""><td>04.14</td><td>UDC Countities Australia 14d</td><td>100</td><td>0/</td><td></td><td></td></t<>	04.14	UDC Countities Australia 14d	100	0/		
UBS Securities Australia Ltd         Sell         5,577         (662)           UBS Securities Australia Ltd         Sell         5,577         (662)           UBS Securities Australia Ltd         Sell         2,572         (673)           UBS Securities Australia Ltd         Sell         2,722         (673)           UBS Securities Australia Ltd         Sell         2,732         (73)           UBS Securities Australia Ltd <td>1,1</td> <td>URC Counties Australia 14d</td> <td>Call</td> <td>413</td> <td></td> <td></td>	1,1	URC Counties Australia 14d	Call	413		
URS Securities Australia Ltd         Sell         5.527         (662)           URS Securities Australia Ltd         Sell         5.729         (677)           URS Securities Australia Ltd         Sell         2.529         (670)           URS Securities Australia Ltd         Sell         2.524         (682)           URS Securities Australia Ltd         Sell         2.524         (672)           URS Securities Australia Ltd         Sell         2.524         (702)           URS Securities Australia Ltd	Dt.14	LIRS Securities Australia (1d	) and	5,645		
UBS Securities Australia Ltd         Sell         6,002         (475)           UBS Securities Australia Ltd         Sell         2,512         (2,524)           UBS Securities Australia Ltd         Sell         2,512         (2,524)           UBS Securities Australia Ltd         Sell         2,572         (2,524)           UBS Securities Australia Ltd         Sell         1,1330         (1,180)           UBS Securities Australia Ltd         Sell         2,702         (6,502)           UBS Securities Australia Ltd         Sell         2,504         (7,03)           UBS	-Oct-14	IIRS Securities Australia Itd		5,577		
UBS Securities Australia Ltd         Sell         CA 234         CA 240           UBS Securities Australia Ltd         Sell         25.73         CA 240           UBS Securities Australia Ltd         Sell         2.57         CA 240           UBS Securities Australia Ltd         Sell         5.73         CA 250           UBS Securities Australia	Ort-14	IIRS Securities Australia Ltd	3	4 002	ŀ	l
USS Securities Australia LEG         Sell         2,529         (2,528)           USS Securities Australia LEG         Sell         2,529         (300)           USS Securities Australia LEG         Sell         2,529         (300)           USS Securities Australia LEG         Sell         2,529         (300)           USS Securities Australia LEG         Sell         1,130         (118)           USS Securities Australia LEG         Sell         2,157         (650)           USS Securities Australia LEG         Sell         2,590         (7,10)           USS Securities Australia LEG         Sell         3,590         (7,10)           USS Securities Australia LEG         Sell         4,14         (4,3)           USS Securities Austral	Oct-14	UBS Securities Australia Ltd	lleS	5 198		
USS Securities Australia Ltd         Sell         2.572         C/29B           USS Securities Australia Ltd         Sell         2.523         (1,00)           USS Securities Australia Ltd         Sell         1.523         (1,00)           USS Securities Australia Ltd         Sell         7.02         (1,540)           USS Securities Australia Ltd         Sell         5.02         (1,50)           USS Securities Australia Ltd         Sell         5.02         (1,50)           USS Securities Australia Ltd         Sell         5.02         (1,50)           USS Securities Australia Ltd         Sell         5.04         (1,50)           USS Securities Australia Ltd         Sell         5.04         (1,50)           USS Securities Australia Ltd         Sell         5.04         (1,00)           USS Securities Australia	-00-14	UBS Securities Australia 11d	Sell	20.434		
UBS Securities Australia Ltd         Sell         2,529         (568)           UBS Securities Australia Ltd         Sell         2,529         (658)           UBS Securities Australia Ltd         Sell         1,1330         (1,1340)           UBS Securities Australia Ltd         Sell         2,108         (550)           UBS Securities Australia Ltd         Sell         2,159         (650)           UBS Securities Australia Ltd         Sell         2,159         (650)           UBS Securities Australia Ltd         Sell         2,250         (650)           UBS Securities Australia Ltd         Sell         2,540         (670)           UBS Securities Australia Ltd         Sell         5,581         (670)           UBS Securities Australia Ltd         Sell         5,581         (670)           UBS Securities Australia Ltd         Sell         3,397         (497)           UBS Securities Australia Ltd         Sell         5,581         (670)           UBS Securities Australia Ltd         Sell         5,582         (706)           UBS Securities Australia Ltd         Sell         5,582         (707)           UBS Securities Australia Ltd         Sell         5,582         (708)           UBS Securities Australia	-Oct-14	UBS Securities Australia 11d	Sell	2 512		
UBS Securities Augralia Ltd         Self         5,228         (1,540)           UBS Securities Augralia Ltd         Self         4,743         (1,540)           UBS Securities Augralia Ltd         Self         5,702         (6,65)           UBS Securities Augralia Ltd         Self         5,702         (6,69)           UBS Securities Augralia Ltd         Self         5,801         (7,03)           UBS Securities Augralia Ltd         Self         5,801         (7,03)           UBS Securities Augralia Ltd         Self         5,802         (7,02)           UBS Securities Augralia Lt	-Oct-14	UBS Securities Australia Ltd	Sell	2.529		l
UBS Securities Australia Ltd         Sell         1,1330         (1,344)           UBS Securities Australia Ltd         Sell         2,702         (6,50)           UBS Securities Australia Ltd         Sell         5,800         (7,03)           UBS Securities Australia Ltd         Sell         6,900         (7,03)           UBS Securities Australia Ltd         Sell         6,900         (7,03)           UBS Securities Australia Ltd         Sell         6,900         (7,03)           UBS Securities Australia Ltd         Sell         6,200         (7,03)           UBS Securit	Oct-14	UBS Securities Australia Ltd	Sell	5,628		
UBS Securities Australia Ltd         Sell         1985         (188)           UBS Securities Australia Ltd         Sell         5,702         (6,50)           UBS Securities Australia Ltd         Sell         5,702         (6,50)           UBS Securities Australia Ltd         Sell         5,231         (6,50)           UBS Securities Australia Ltd         Sell         2,395         (7,50)           UBS Securities Australia Ltd         Sell         5,890         (7,03)           UBS Securities Australia Ltd         Sell         1,039         (7,122)           UBS Securities Australia Ltd         Sell         1,133         (7,60)           UBS Securities Australia Ltd         Sell         1,133         (7,60)           UBS Securities Australia Ltd         Sell         1,133         (7,60)           UBS Securities	-Oct-14	UBS Securities Australia Ltd	Sell	11,330		
UBS Securities Australia Ltd         Sell         5,702         (5,62)           UBS Securities Australia Ltd         Sell         5,702         (6,69)           UBS Securities Australia Ltd         Sell         2,108B         (6,19)           UBS Securities Australia Ltd         Sell         5,290         (271)           UBS Securities Australia Ltd         Sell         5,391         (6,69)           UBS Securities Australia Ltd         Sell         5,391         (6,69)           UBS Securities Australia Ltd         Sell         3,397         (402)           UBS Securities Australia Ltd         Sell         3,397         (402)           UBS Securities Australia Ltd         Sell         3,397         (402)           UBS Securities Australia Ltd         Sell         5,922         (706)           UBS Securities Australia Ltd         Sell         3,397         (402)           UBS Securities Australia Ltd         Sell         1,741         (706)           UBS Securities Australia Ltd         Sell         1,741         (707)           UBS Securities Australia Ltd         Sell         1,741         (709)           UBS Securities Australia Ltd         Sell         1,741         (709)           UBS Securities Austral	-Oct-14	UBS Securities Australia Ltd	Sell	966		
UBS Securities Australia Ltd         Sell         2,702         (676)           UBS Securities Australia Ltd         Sell         2,108         (2,500)           UBS Securities Australia Ltd         Sell         2,239         (7,502)           UBS Securities Australia Ltd         Sell         2,290         (7,03)           UBS Securities Australia Ltd         Sell         2,394         (7,03)           UBS Securities Australia Ltd         Sell         2,397         (7,03)           UBS Securities Australia Ltd         Sell         3,397         (402)           UBS Securities Australia Ltd         Sell         4,14         (4,22)           UBS Securities Australia Ltd         Sell         1,103         (1,123)           UBS Securitie	-Oct-14	UBS Securities Australia Ltd	Sell	4,743		
UBS Securities Australia Ltd         Sell         2.197         (2500)           UBS Securities Australia Ltd         Sell         2.197         (260)           UBS Securities Australia Ltd         Sell         2.197         (260)           UBS Securities Australia Ltd         Sell         1.335         (703)           UBS Securities Australia Ltd         Sell         1.335         (703)           UBS Securities Australia Ltd         Sell         3.397         (403)           UBS Securities Australia Ltd         Sell         3.397         (405)           UBS Securities Australia Ltd         Sell         1.039         (1.182)           UBS Securities Au	-Oct-14	UBS Securities Australia Ltd	Sell	5,702		
UBS Securities Australia Ltd         Sell         5.22         (619)           UBS Securities Australia Ltd         Sell         2.790         (277)           UBS Securities Australia Ltd         Sell         7.290         (773)           UBS Securities Australia Ltd         Sell         5.819         (689)           UBS Securities Australia Ltd         Sell         6.694         (703)           UBS Securities Australia Ltd         Sell         3.37         (402)           UBS Securities Australia Ltd         Sell         3.37         (402)           UBS Securities Australia Ltd         Sell         3.37         (402)           UBS Securities Australia Ltd         Sell         4.200         (709)           UBS Securities Australia Ltd         Sell         4.200         (709)           UBS Securities Australia Ltd         Sell         4.200         (709)           UBS Securities Australia Ltd         Sell         1.331         (1.60)           UBS Securities Australia Ltd         Sell         1.741         (1.60)           UBS Securities Australia Ltd         Sell         1.741         (1.60)           UBS Securities Australia Ltd         Sell         7.522         (7.00)           UBS Securities Australia Ltd	-Oct-14	UBS Securities Australia Ltd	Sell	21,088		
UBS Securities Australia Ltd         Sell         2.197         (260)           UBS Securities Australia Ltd         Sell         2.290         (718)           UBS Securities Australia Ltd         Sell         5.819         (703)           UBS Securities Australia Ltd         Sell         (689)           UBS Securities Australia Ltd         Sell         (706)           UBS Securities Australia Ltd         Sell         (72)           UBS Securities Australia Ltd         Sell         (720)           UBS Securities Australia Ltd         Sell         (730)           UBS Securities Australia Ltd         Sell         (730) </td <td>-Oct-14</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>5,221</td> <td></td> <td></td>	-Oct-14	UBS Securities Australia Ltd	Sell	5,221		
UBS Securities Australia Ltd         Sell         1,250         (277)           UBS Securities Australia Ltd         Sell         1,335         (1028)           UBS Securities Australia Ltd         Sell         5,940         (7029)           UBS Securities Australia Ltd         Sell         5,861         (649)           UBS Securities Australia Ltd         Sell         3,397         (40)           UBS Securities Australia Ltd         Sell         6,940         (702)           UBS Securities Australia Ltd         Sell         6,940         (497)           UBS Securities Australia Ltd         Sell         4,200         (497)           UBS Securities Australia Ltd         Sell         1,039         (1,182)           UBS Securities Australia Ltd         Sell         1,039         (1,182)           UBS Securities Australia Ltd         Sell         1,039         (1,182)           UBS Securities Australia Ltd         Sell         1,303         (1,182)           UBS Securities Australia Ltd         Sell         1,303         (1,182)           UBS Securities Australia Ltd         Sell         1,303         (1,182)           UBS Securities Australia Ltd         Sell         6,300         (1,69)           UBS Securities	-0ct-14	UBS Securities Australia Ltd	Sell	2,197		1
UBS Securities Australia Ltd         Sell         5.819         (158)           UBS Securities Australia Ltd         Sell         5.819         (709)           UBS Securities Australia Ltd         Sell         5.819         (709)           UBS Securities Australia Ltd         Sell         3.534         (800)           UBS Securities Australia Ltd         Sell         3.397         (402)           UBS Securities Australia Ltd         Sell         4.109         (706)           UBS Securities Australia Ltd         Sell         4.200         (497)           UBS Securities Australia Ltd         Sell         4.200         (497)           UBS Securities Australia Ltd         Sell         1.039         (7.12)           UBS Securities Australia Ltd         Sell         1.039         (7.12)           UBS Securities Australia Ltd         Sell         1.039         (7.12)           UBS Securities Australia Ltd         Sell         1.03         (1.12)           UBS Securities Australia Ltd         Sell         7.71         (8.97)           UBS Securities Australia Ltd         Sell         7.01         (8.97)           UBS Securities Australia Ltd         Sell         7.01         (8.97)           UBS Securities Australia	-0ct-14	UBS Securities Australia Ltd	Sell	2,290		
UBS Securities Australia Ltd         Sell         5,940         (703)           UBS Securities Australia Ltd         Sell         5,819         (699)           UBS Securities Australia Ltd         Sell         5,834         (402)           UBS Securities Australia Ltd         Sell         3,397         (402)           UBS Securities Australia Ltd         Sell         5,962         (706)           UBS Securities Australia Ltd         Sell         6,93         (402)           UBS Securities Australia Ltd         Sell         6,200         (497)           UBS Securities Australia Ltd         Sell         1,039         (1,182)           UBS Securities Australia Ltd         Sell         1,039         (1,182)           UBS Securities Australia Ltd         Sell         1,337         (1,540)           UBS Securities Australia Ltd         Sell         1,339         (1,697)           UBS Securities Australia Ltd         Sell         1,741         (7,69)           UBS Securities Australia Ltd         Sell         7,592         (600)           UBS Securities Australia Ltd         Sell         7,592         (600)           UBS Securities Australia Ltd         Sell         7,592         (600)           UBS Securities Austr	-Oct-14	UBS Securities Australia Ltd	Sell	1,335		
UBS Securities Australia Ltd         Sell         5.819         (699)           UBS Securities Australia Ltd         Sell         5.81         (690)           UBS Securities Australia Ltd         Sell         3.37         (402)           UBS Securities Australia Ltd         Sell         3.37         (402)           UBS Securities Australia Ltd         Sell         6.93         (706)           UBS Securities Australia Ltd         Sell         4.14         4.09           UBS Securities Australia Ltd         Sell         4.200         (497)           UBS Securities Australia Ltd         Sell         4.200         (497)           UBS Securities Australia Ltd         Sell         4.200         (497)           UBS Securities Australia Ltd         Sell         1.131         (1.74)           UBS Securities Australia Ltd         Sell         1.131         (1.74)           UBS Securities Australia Ltd         Sell         1.7412         (2.05)           UBS Securities Australia Ltd         Sell         1.7412         (2.05)           UBS Securities Australia Ltd         Sell         7.522         (2.04)           UBS Securities Australia Ltd         Sell         6.318         (4.01)           UBS Securities Australia L	-Oct-14	UBS Securities Australia Ltd	Sell	5,940		
UBS Securities Australia Ltd         Sell         5.861         (6.94)           UBS Securities Australia Ltd         Sell         5.32         (900)           UBS Securities Australia Ltd         Sell         3.397         (402)           UBS Securities Australia Ltd         Sell         5.962         (706)           UBS Securities Australia Ltd         Sell         4.20         (723)           UBS Securities Australia Ltd         Sell         4.20         (727)           UBS Securities Australia Ltd         Sell         4.20         (727)           UBS Securities Australia Ltd         Sell         13.03T         (7.544)           UBS Securities Australia Ltd         Sell         13.03T         (7.544)           UBS Securities Australia Ltd         Sell         13.04         (7.544)           UBS Securities Australia Ltd         Sell         17.01         (8.2)           UBS Securities Australia Ltd         Sell         7.52         (7.69)           UBS Securities Australi	-Oct-14	UBS Securities Australia Ltd	Sell	5,819		
UBS Securities Australia Ltd         Sell         3.397         (402)           UBS Securities Australia Ltd         Sell         3.397         (402)           UBS Securities Australia Ltd         Sell         5.962         (706)           UBS Securities Australia Ltd         Sell         6.93         (42)           UBS Securities Australia Ltd         Sell         4,200         (497)           UBS Securities Australia Ltd         Sell         4,200         (497)           UBS Securities Australia Ltd         Sell         1,639         (1,182)           UBS Securities Australia Ltd         Sell         1,639         (1,182)           UBS Securities Australia Ltd         Sell         1,039         (1,182)           UBS Securities Australia Ltd         Sell         1,039         (1,182)           UBS Securities Australia Ltd         Sell         1,7412         (2,602)           UBS Securities Australia Ltd         Sell         7,71         (48)           UBS Securities Australia Ltd         Sell         6,326         (7,69)           UBS Securities Australia Ltd         Sell         6,326         (7,69)           UBS Securities Australia Ltd         Sell         6,326         (2,000)           UBS Securities Au	-Oct-14	UBS Securities Australia Ltd	Setl	5,861		
UBS Securities Australia Ltd         Sell         3.39         (402)           UBS Securities Australia Ltd         Sell         3.34         (40)           UBS Securities Australia Ltd         Sell         6.93         (8.20)           UBS Securities Australia Ltd         Sell         4.109         (7.23)           UBS Securities Australia Ltd         Sell         4.20         (497)           UBS Securities Australia Ltd         Sell         1.039         (7.23)           UBS Securities Australia Ltd         Sell         1.37         (7.27)           UBS Securities Australia Ltd         Sell         1.30         (1.122)           UBS Securities Australia Ltd         Sell         1.74         (2.063)           UBS Securities Australia Ltd         Sell         7.71         (2.063)           UBS Securities Australia Ltd         Sell         7.71         (2.063)           UBS Securities Australia Ltd         Sell         7.71         (2.063)           UBS Securities Australia Ltd         Sell         7.52         (2.00)           UBS Securities Australia Ltd         Sell         7.52         (2.00)           UBS Securities Australia Ltd         Sell         7.52         (2.00)           UBS Securities Australia	-Oct-14	UBS Securities Australia Ltd	Sell	2,534		
UBS Securities Australia Ltd         Sell         59.2         (40)           UBS Securities Australia Ltd         Sell         69.3         (70)           UBS Securities Australia Ltd         Sell         69.3         (82)           UBS Securities Australia Ltd         Sell         4.200         (1723)           UBS Securities Australia Ltd         Sell         4.200         (1723)           UBS Securities Australia Ltd         Sell         1.83.1         (1717)           UBS Securities Australia Ltd         Sell         1.95         (1.822)           UBS Securities Australia Ltd         Sell         1.01         (1.122)           UBS Securities Australia Ltd         Sell         1.03         (1.122)           UBS Securities Australia Ltd         Sell         7.01         (1.65)           UBS Securities Australia Ltd         Sell         7.01         (1.65)           UBS Securities Australia Ltd         Sell         7.01         (1.65)           UBS Securities Australia Ltd         Sell         6.236         (2.000)           UBS Securities Australia Ltd         Sell         6.236         (2.00)           UBS Securities Australia Ltd         Sell         6.236         (2.00)           UBS Securities Australi	-Oct-14	UBS Securities Australia Ltd	Sell	3,397		
UBS Securities Australia (14)         Sell         5922         (705)           UBS Securities Australia (14)         Sell         693         (822)           UBS Securities Australia (14)         Sell         4.14         (497)           UBS Securities Australia (14)         Sell         4.200         (497)           UBS Securities Australia (14)         Sell         1.831         (7.17)           UBS Securities Australia (14)         Sell         1.831         (7.17)           UBS Securities Australia (14)         Sell         1.97         (1.182)           UBS Securities Australia (14)         Sell         1.74,412         (7.05)           UBS Securities Australia (14)         Sell         7.71         (1.05)           UBS Securities Australia (14)         Sell         7.71         (7.05)           UBS Securities Australia (14)         Sell         7.52         (7.04)           UBS Securities Australia (14)         Sell         7.52         (7.09)           UBS Securities Australia (14)         Sell         7.52         (7.09)           UBS Securities Australia (14)         Sell         6.318         (4.01)           UBS Securities Australia (14)         Sell         6.32         (7.02)           UBS Secur	Oct-14	UBS Securities Australia Ltd	Sell	34		
UBS Securities Autoratile Ltd         Sell         693         (82)           UBS Securities Autoratile Ltd         Sell         4.14         (49)           UBS Securities Autoratile Ltd         Sell         4,200         (497)           UBS Securities Autoratile Ltd         Sell         4,200         (497)           UBS Securities Autoratile Ltd         Sell         1,831         (2.77)           UBS Securities Autoratile Ltd         Sell         1,039         (1,122)           UBS Securities Autoratile Ltd         Sell         1,031         (1,542)           UBS Securities Autoratile Ltd         Sell         1,17,412         (2,683)           UBS Securities Autoratile Ltd         Sell         1,7412         (2,683)           UBS Securities Autoratile Ltd         Sell         1,7412         (2,683)           UBS Securities Autoratile Ltd         Sell         1,7412         (2,693)           UBS Securities Autoralie Ltd         Sell         4,14         (4,61)	Oct-14	UBS Securities Australia Ltd	Sell	5,962	(907)	
UBS Securities Australia Ltd         Sell         414         (49)           UBS Securities Australia Ltd         Sell         4,200         (497)           UBS Securities Australia Ltd         Sell         4,200         (497)           UBS Securities Australia Ltd         Sell         1,837         (273)           UBS Securities Australia Ltd         Sell         1,637         (1,182)           UBS Securities Australia Ltd         Sell         1,7412         (1,66)           UBS Securities Australia Ltd         Sell         1,7412         (2,66)           UBS Securities Australia Ltd         Sell         7,701         (83)           UBS Securities Australia Ltd         Sell         7,592         (1,66)           UBS Securities Australia Ltd         Sell         7,592         (600)           UBS Securities Australia Ltd         Sell         6,380         (7,49)           UBS Securities Australia Ltd         Sell         6,380         (7,592)           UBS Securities Australia Ltd         Sell         6,380         (7,69)           UBS Securities Australia Ltd         Sell         6,380         (7,69)           UBS Securities Australia Ltd         Sell         4,077         (4,69)           UBS Securities Aust	-Oct-14	UBS Securities Australia Ltd	Sell	669	(83)	
UBS Securities Australia Ltd         Sell         4,200         (#272)           UBS Securities Australia Ltd         Sell         4,200         (#372)           UBS Securities Australia Ltd         Sell         1,331         (*271)           UBS Securities Australia Ltd         Sell         1,303         (*1,182)           UBS Securities Australia Ltd         Sell         1,303         (*1,544)           UBS Securities Australia Ltd         Sell         7,71         (*105)           UBS Securities Australia Ltd         Sell         7,71         (*105)           UBS Securities Australia Ltd         Sell         7,71         (*105)           UBS Securities Australia Ltd         Sell         7,52         (*106)           UBS Securities Australia Ltd         Sell         7,52         (*106)           UBS Securities Australia Ltd         Sell         7,52         (*106)           UBS Securities Australia Ltd         Sell         7,52         (*107)           UBS Securities Australia Ltd         Sell         7,52         (*107)           UBS Securities Australia Ltd         Sell         4,14         (*107)           UBS Securities Australia Ltd         Sell         4,14         (*107)           UBS Securities Aust	-Oct-14	UBS Securities Australia Ltd	Sell	414	(48)	Ordinary
UBS Securities Australia (td.)         Sell         4,200         (497)           UBS Securities Australia (td.)         Sell         1,831         (217)           UBS Securities Australia (td.)         Sell         1,831         (217)           UBS Securities Australia (td.)         Sell         1,975         (1,182)           UBS Securities Australia (td.)         Sell         1,071         (1,182)           UBS Securities Australia (td.)         Sell         1,01         (1,122)           UBS Securities Australia (td.)         Sell         1,741         (2,683)           UBS Securities Australia (td.)         Sell         1,580         (7,693)           UBS Securities Australia (td.)         Sell         6,380         (7,49)           UBS Securities Australia (td.)         Sell         6,381         (4,69)           UBS Securities Australia (td.)         Sell         6,381         (4,69)           UBS Securities Australia (td.)         Sell         6,411         (7,69)	-Oct-14	UBS Securities Australia Ltd	Sell	1,039	(123)	Ordinary
UBS Securities Australia Ltd         Sell         4.27         (50)           UBS Securities Australia Ltd         Sell         1,831         (27)           UBS Securities Australia Ltd         Sell         13,033         (1,182)           UBS Securities Australia Ltd         Sell         13,033         (1,544)           UBS Securities Australia Ltd         Sell         17,741         (1,054)           UBS Securities Australia Ltd         Sell         701         (105)           UBS Securities Australia Ltd         Sell         701         (105)           UBS Securities Australia Ltd         Sell         6,380         (7,49)           UBS Securities Australia Ltd         Sell         6,380         (7,49)           UBS Securities Australia Ltd         Sell         7,592         (90)           UBS Securities Australia Ltd         Sell         414         (49)           UBS Securities Australia Ltd         Sell         6,318         (40)           UBS Securities Australia Ltd <td>-Oct-14</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>4,200</td> <td>(497)</td> <td>Ordinary</td>	-Oct-14	UBS Securities Australia Ltd	Sell	4,200	(497)	Ordinary
UBS Securities Australia Ltd         Sell         1931         (277)           UBS Securities Australia Ltd         Sell         13,031         (1,544)           UBS Securities Australia Ltd         Sell         101         (1,544)           UBS Securities Australia Ltd         Sell         17,412         (2,683)           UBS Securities Australia Ltd         Sell         17,412         (2,683)           UBS Securities Australia Ltd         Sell         17,412         (2,683)           UBS Securities Australia Ltd         Sell         17,512         (169)           UBS Securities Australia Ltd         Sell         1,522         (7,44)           UBS Securities Australia Ltd         Sell         1,522         (7,44)           UBS Securities Australia Ltd         Sell         7,522         (7,69)           UBS Securities Australia Ltd         Sell         7,522         (4,60)           UBS Securities Australia Ltd         Sell         4,14         (4,61)           UBS Securities Australia Ltd         Sell         4,14         (4,61)           UBS Securities Australia Ltd         Sell         8,25         (4,60)           UBS Securities Australia Ltd         Sell         8,25         (4,61)           UBS Securiti	Oct-14	UBS Securities Australia Ltd	Sell	422	(20)	Ordinary
UBS Securities Australia Ltd         Sell         19,90         (1,182)           UBS Securities Australia Ltd         Sell         1,031         (1,544)           UBS Securities Australia Ltd         Sell         1,071         (1,644)           UBS Securities Australia Ltd         Sell         701         (1,653)           UBS Securities Australia Ltd         Sell         701         (1,653)           UBS Securities Australia Ltd         Sell         701         (1,653)           UBS Securities Australia Ltd         Sell         6,2360         (7,24)           UBS Securities Australia Ltd         Sell         6,2360         (7,24)           UBS Securities Australia Ltd         Sell         6,236         (7,24)           UBS Securities Australia Ltd         Sell         6,236         (7,24)           UBS Securities Australia Ltd         Sell         4,14         (4,91)           UBS Securities Australia Ltd         Sell         4,14         (4,91)           UBS Securities Australia Ltd         Sell         8,24         (100)           UBS Securities Australia Ltd         Sell         6,11         (7,69)           UBS Securities Australia Ltd         Sell         8,24         (100)           UBS Securities Aus	-Oct-14	UBS Securities Australia Ltd	Sell	1,831	(217)	Ordinary
UBS Securities Australie Ltd         Sell         13,031         (1,544)           UBS Securities Australia Ltd         Sell         101         (12)           UBS Securities Australia Ltd         Sell         771         (2,683)           UBS Securities Australia Ltd         Sell         701         (83)           UBS Securities Australia Ltd         Sell         (70)         (83)           UBS Securities Australia Ltd         Sell         (73)         (74)           UBS Securities Australia Ltd         Sell         (74)         (74)           UBS Securities Australia Ltd         Sell         7,522         (74)           UBS Securities Australia Ltd         Sell         6,380         (749)           UBS Securities Australia Ltd         Sell         6,318         (749)           UBS Securities Australia Ltd         Sell         6,318         (749)           UBS Securities Australia Ltd         Sell         8,41         (10)           UBS Securities Australia Ltd         Sell         8,41         (10)           UBS Securities Australia Ltd         Sell         8,51         (40)           UBS Securities Australia Ltd         Sell         8,51         (40)           UBS Securities Australia Ltd	Oct-14	UBS Securities Australia Ltd	Sell	9,976	(1,182)	Ordinary
UBS Securines Australia trd         Self         101         (12)           UBS Securines Australia trd         Self         17412         2.0633           UBS Securines Australia trd         Self         701         (83)           UBS Securines Australia trd         Self         135         (76)           UBS Securines Australia trd         Self         155         (74)           UBS Securines Australia trd         Self         63,860         (749)           UBS Securines Australia trd         Self         63,860         (749)           UBS Securines Australia trd         Self         63,860         (749)           UBS Securines Australia trd         Self         4,057         (49)           UBS Securines Australia trd         Self         4,057         (49)           UBS Securines Australia trd         Self         4,057         (49)           UBS Securines Australia trd         Self         8,41         (40)           UBS Securines Australia trd         Self         8,41         (40)           UBS Securines Australia trd         Self         8,42         (40)           UBS Securines Australia trd         Self         8,54         (40)           UBS Securines Australia trd         Self	Oct-14	UBS Securities Australia Ltd	Sell	13,031	(1,544)	Ordinary
UBS Securities Australia Ltd         Sell         (106)           UBS Securities Australia Ltd         Sell         (17412         (2,653)           UBS Securities Australia Ltd         Sell         701         (63)           UBS Securities Australia Ltd         Sell         6,350         (76,90)           UBS Securities Australia Ltd         Sell         6,350         (7,00)           UBS Securities Australia Ltd         Sell         6,350         (7,00)           UBS Securities Australia Ltd         Sell         6,318         (7,00)           UBS Securities Australia Ltd         Sell         4,057         (4,00)           UBS Securities Australia Ltd         Sell         4,057         (4,00)           UBS Securities Australia Ltd         Sell         4,057         (4,00)           UBS Securities Australia Ltd         Sell         8,41         (100)           UBS Securities Australia Ltd         Sell         8,41         (7,00)           UBS Securities Australia Ltd         Sell         8,41         (7,00)           UBS Securities Australia Ltd         Sell         8,52         (4,00)           UBS Securities Australia Ltd         Sell         8,53         (4,00)           UBS Securities Australia Ltd <t< td=""><td>Oct-14</td><td>UBS Securities Australia Ltd</td><td>Sell</td><td>101</td><td>(12)</td><td>Ordinary</td></t<>	Oct-14	UBS Securities Australia Ltd	Sell	101	(12)	Ordinary
UBS Securities Australia Ltd         Sell         17,412         (2,063)           UBS Securities Australia Ltd         Sell         701         (83)           UBS Securities Australia Ltd         Sell         135         (16)           UBS Securities Australia Ltd         Sell         16,360         (754)           UBS Securities Australia Ltd         Sell         62,80         (74)           UBS Securities Australia Ltd         Sell         62,80         (74)           UBS Securities Australia Ltd         Sell         4,057         (49)           UBS Securities Australia Ltd         Sell         4,12         (40)           UBS Securities Australia Ltd         Sell         4,14         (40)           UBS Securities Australia Ltd         Sell         3,18         (40)           UBS Securities Australia Ltd         Sell         3,18         (40)           UBS Securities Australia Ltd         Sell         8,41         (10)           UBS Securities Australia Ltd         Sell         8,41         (10)           UBS Securities Australia Ltd         Sell         8,51         (40)           UBS Securities Australia Ltd         Sell         6,41         (70)           UBS Securities Australia Ltd         Sel	Oct-14	UBS Securities Australia Ltd	Sell	568	(106)	Ordinary
UBS Securities Australia Ltd         Self         701         (82)           UBS Securities Australia Ltd         Self         135         (16)           UBS Securities Australia Ltd         Self         6,380         (74)           UBS Securities Australia Ltd         Self         6,280         (74)           UBS Securities Australia Ltd         Self         6,29         (74)           UBS Securities Australia Ltd         Self         6,38         (74)           UBS Securities Australia Ltd         Self         6,38         (49)           UBS Securities Australia Ltd         Self         4,14         (49)           UBS Securities Australia Ltd         Self         8,41         (40)           UBS Securities Australia Ltd         Self         8,41         (70)           UBS Securities Australia Ltd         Self         8,41         (70)           UBS Securities Australia Ltd         Self         8,41         (71)           UBS Securities Australia Ltd         Self	Oct-14	UBS Securities Australia Ltd	Sell	17,412	(2.063)	Ordinary
UBS Securities Australia Ltd         Sell         (15)           UBS Securities Australia Ltd         Sell         (5,360)         (7,54)           UBS Securities Australia Ltd         Sell         (7,52)         (7,49)           UBS Securities Australia Ltd         Sell         (7,52)         (7,49)           UBS Securities Australia Ltd         Sell         (7,49)         (7,49)           UBS Securities Australia Ltd         Sell         (7,49)         (7,49)           UBS Securities Australia Ltd         Sell         41,4         (4,9)           UBS Securities Australia Ltd         Sell         8,44         (1,00)           UBS Securities Australia Ltd         Sell         8,41         (1,00)           UBS Securities Australia Ltd         Sell         8,41         (1,00)           UBS Securities Australia Ltd         Sell         8,51         (4,00)           UBS Securities Australia Ltd         Sell         8,51         (7,60)           UBS Securities Australia Ltd         Sell         8,51         (7,60)           UBS Securities Australia Ltd         Sell         8,51         (7,60)           UBS Securities Australia Ltd         Sell         8,51         (7,61)           UBS Securities Australia Ltd	Oct-14	UBS Securities Australia Ltd	Sell	102	(883)	Ordinary
UBS Securities Australia Ltd         Sell         (6,360)         (754)           UBS Securities Australia Ltd         Sell         16,370         (2,000)           UBS Securities Australia Ltd         Sell         6,360         (7,49)           UBS Securities Australia Ltd         Sell         6,38         (900)           UBS Securities Australia Ltd         Sell         4,057         (48)           UBS Securities Australia Ltd         Sell         4,057         (48)           UBS Securities Australia Ltd         Sell         8,41         (40)           UBS Securities Australia Ltd         Sell         8,41         (40)           UBS Securities Australia Ltd         Sell         8,41         (70)           UBS Securities Australia Ltd         Sell         8,41         (70)           UBS Securities Australia Ltd         Sell         8,41         (70)           UBS Securities Australia Ltd         Sell         8,51         (42)           UBS Securities Australia Ltd         Sell         8,51         (42)           UBS Securities Australia Ltd         Sell         17,887         (2,11)           UBS Securities Australia Ltd         Sell         17,897         (2,11)           UBS Securities Australia Ltd	-Oct-14	LiBS Securities Australia Ltd	lles	135	(91)	Ordinan
UBS Securities Australia Ltd         Sell         (7.67)           UBS Securities Australia Ltd         Sell         (5.500)           UBS Securities Australia Ltd         Sell         (5.200)           UBS Securities Australia Ltd         Sell         (7.49)           UBS Securities Australia Ltd         Sell         40.57         (7.49)           UBS Securities Australia Ltd         Sell         41.4         (4.9)           UBS Securities Australia Ltd         Sell         84.4         (10.0)           UBS Securities Australia Ltd         Sell         87.9         (8.0)           UBS Securities Australia Ltd         Sell         87.9         (8.0)           UBS Securities Australia Ltd         Sell         87.9         (8.0)           UBS Securities Australia Ltd         Sell         87.9         (4.0)           UBS Securities Australia Ltd         Sell         87.9         (4.2)           UBS Securities Australia Ltd         Sell         87.5         (4.2)           UBS Securities Australia Ltd         Sell         87.1         (7.50)           UBS Securities Australia Ltd         Sell         87.1         (7.2)           UBS Securities Australia Ltd         Sell         (7.1)         (7.1)	Oct.14	LIRS Securities Australia Ltd	lo de la companya de	095.9	(OL)	Ordinary
UBS Securities Australia Ltd         Sell         CVS 2000           UBS Securities Australia Ltd         Sell         CS 20           UBS Securities Australia Ltd         Sell         CS 20           UBS Securities Australia Ltd         Sell         CS 20           UBS Securities Australia Ltd         Sell         4,057         (483)           UBS Securities Australia Ltd         Sell         4,057         (481)           UBS Securities Australia Ltd         Sell         4,057         (40)           UBS Securities Australia Ltd         Sell         3,844         (10)           UBS Securities Australia Ltd         Sell         6,401         (40)           UBS Securities Australia Ltd         Sell         6,411         (70)           UBS Securities Australia Ltd         Sell         6,411         (70)           UBS Securities Australia Ltd         Sell         8,53         (4,2)           UBS Securities Australia Ltd         Sell         17,887         (2,1)           UBS Securities Australia Ltd         Sell         17,887         (2,1)           UBS Securities Australia Ltd         Sell         17,887         (2,1)           UBS Securities Australia Ltd         Sell         17,887         (2,1) <tr< td=""><td>04-14</td><td>IRC Securities Australia and</td><td>Soll</td><td>16 870</td><td>(450)</td><td>Ordinary</td></tr<>	04-14	IRC Securities Australia and	Soll	16 870	(450)	Ordinary
URB Securities Australia Ltd         Sell         0,23         (1,49)           URB Securities Australia Ltd         Sell         6,318         (7,49)           UBS Securities Australia Ltd         Sell         6,318         (7,49)           UBS Securities Australia Ltd         Sell         41,4         (4,61)           UBS Securities Australia Ltd         Sell         84,4         (100)           UBS Securities Australia Ltd         Sell         336         (40)           UBS Securities Australia Ltd         Sell         8,4         (100)           UBS Securities Australia Ltd         Sell         8,4         (100)           UBS Securities Australia Ltd         Sell         8,4         (40)           UBS Securities Australia Ltd         Sell         8,4         (42)           UBS Securities Australia Ltd         Sell         8,5         (42)           UBS Securities Australia Ltd         Sell         8,5         (42)           UBS Securities Australia Ltd         Sell         17,887         (4,1)           UBS Securities Australia Ltd         Sell         17,887         (4,1)           UBS Securities Australia Ltd         Sell         17,897         (4,1)           UBS Securities Australia Ltd         <	11.00	Upp Commission America and	liac	0/9/01	(000,2)	Orginary
URS Securities Australia Ltd         Self         (472)         (474)           URS Securities Australia Ltd         Self         (457)         (481)           URS Securities Australia Ltd         Self         414         (481)           URS Securities Australia Ltd         Self         844         (410)           URS Securities Australia Ltd         Self         824         (40)           URS Securities Australia Ltd         Self         709         (84)           URS Securities Australia Ltd         Self         641         (760)           URS Securities Australia Ltd         Self         835         (42)           URS Securities Australia Ltd         Self         85         (42)           URS Securities Australia Ltd         Self         85         (42)           URS Securities Australia Ltd         Self         17,887         (7,18)           URS Securities Australia Ltd         Self         17,887         (7,18)           URS Securities Australia Ltd         Self         (7,27)         (42)           URS Securities Australia Ltd         Self         (7,27)         (42)           URS Securities Australia Ltd         Self         (7,27)         (4,27)           URS Securities Australia Ltd <td< td=""><td>24.14</td><td>LIBS Securities Australia Ltd</td><td>liac</td><td>679</td><td>(74)</td><td>Ordinary</td></td<>	24.14	LIBS Securities Australia Ltd	liac	679	(74)	Ordinary
USS Securines Australia Ltd         Sell         4,057         (481)           UBS Securines Australia Ltd         Sell         414         (49)           UBS Securines Australia Ltd         Sell         844         (100)           UBS Securines Australia Ltd         Sell         844         (100)           UBS Securines Australia Ltd         Sell         843         (40)           UBS Securines Australia Ltd         Sell         6411         (760)           UBS Securines Australia Ltd         Sell         6411         (760)           UBS Securines Australia Ltd         Sell         835         (42)           UBS Securines Australia Ltd         Sell         17,887         (2,1)	04-14	LIBS Securities Australia 14d	list of	250,1	(000)	Grainary
UBS Securities Australia Ltd         Sell         4.05 / (489)           UBS Securities Ltd         Sell         844         (100)           UBS Securities Australia Ltd         Sell         844         (100)           UBS Securities Australia Ltd         Sell         338         (40)           UBS Securities Australia Ltd         Sell         709         (84)           UBS Securities Australia Ltd         Sell         641         (765)           UBS Securities Australia Ltd         Sell         835         (42)           UBS Securities Australia Ltd         Sell         855         (42)           UBS Securities Australia Ltd         Sell         17,887         (4.1)	04.14	UDS Securities Australia teu	)AEI	010,0	(142)	Cramary
URS Securities Australia Ltd         Self         338         (40)           UBS Securities Australia Ltd         Self         338         (40)           UBS Securities Australia Ltd         Self         709         (84)           UBS Securities Australia Ltd         Self         709         (84)           UBS Securities Australia Ltd         Self         6.411         (760)           UBS Securities Australia Ltd         Self         835         (42)           UBS Securities Australia Ltd         Self         835         (42)           UBS Securities Australia Ltd         Self         17,867         (2,18)           UBS Securities Australia Ltd         Self         17,867         (2,18)           UBS Securities Australia Ltd         Self         12,034         (4,3)	04.14	UDS Sectioners Australia Ltd	) cell	410	(104)	Ordinary
UBS Securines Australia Ltd         Sell         338         (40)           UBS Securines Australia Ltd         Sell         6411         (64)           UBS Securines Australia Ltd         Sell         6411         (760)           UBS Securines Australia Ltd         Sell         6411         (760)           UBS Securines Australia Ltd         Sell         835         (42)           UBS Securines Australia Ltd         Sell         17,887         (2,1)8           UBS Securines Australia Ltd         Sell         12,034         (41)8	Oct-14	LIBS Securities Australia I to	3	VVX	(001)	Ordinan
UBS Securities Australia Ltd         Sell         703         (843)           UBS Securities Australia Ltd         Sell         6,411         (765)           UBS Securities Australia Ltd         Sell         835         (42)           UBS Securities Australia Ltd         Sell         355         (42)           UBS Securities Australia Ltd         Sell         17,887         (2,118)           UBS Securities Australia Ltd         Sell         121,034         (4,13)	Oct-14	LIBS Securities Australia 1td	9	855	(40)	Ordinary
UBS Securities Australia Ltd         Sell         6.411         (760)           UBS Securities Australia Ltd         Sell         895         (99)           UBS Securities Australia Ltd         Sell         355         (42)           UBS Securities Australia Ltd         Sell         17,887         (2,18)           UBS Securities Australia Ltd         Sell         17,867         (3,18)	04.14	LIBS Societies Australia 14d	la S	900	(Pat)	Ordinary
URS Securities Australia Ltd         Sell         (42)           UBS Securities Australia Ltd         Sell         355         (42)           UBS Securities Australia Ltd         Sell         17,887         (2,118)           UBS Securities Australia Ltd         Sell         17,887         (2,118)           UBS Securities Australia Ltd         Sell         12,034         (4,13)	14.14	TIPS Cocurities Australia Lea	100	100	(40)	Ordinary
UBS Securities Australia Ltd         Sell         355         (42)           UBS Securities Australia Ltd         Sell         17,887         (2,118)           UBS Securities Australia Ltd         Sell         17,034         (4,432)	1,10	HIS Countries Australia 14d	3511	200	(00/)	Ordinary
UBS Securities Australia Ltd         Sell         17,887         (2,1)8           UBS Securities Australia Ltd         Sell         121,034         (4,432)	ı	UBS Securities Australia Ltd	Sell	835	(66)	Ordinary
UBS Securities Australia (td         Sell         121,034         (4,132)	١	Obs securities Australia Ltd	Sell	565	(42)	Ordinary
UBS Securities Australia Ltd   Sell   (14,332)	Ì	Obs securities Australia Ltd	Sell	/88′/1	(2,118)	Ordinary
		UBS Securities Australia Ltd	Sell	121,034	114 3321	

			103	69)	Ordinary
5.04-14	UBS Securities Australia Ltd	Sell	8 50,	0.00	
α-14	UBS Securities Australia Ltd	3	15,619	(1 844)	Ordinary
Ct-14	UBS Securities Australia Ltd	Sett	2.355	(2.78)	Ordinary
d-14	UBS Securities Australia Ltd	Seil	1,728	(202)	Ordinary
G-14	UBS Securities Australia Ltd	Sell	5 522	(652)	Ordinary
d-14	UBS Securities Australia Ltd	Sell	449	(23)	Ordinary
ct-14	UBS Securities Australia Ltd	Self	9,402	(1,110)	Ordinary
ct-14	UBS Securities Australia Ltd	Sell	2,456	(230)	Ordinary
d-14	UBS Securities Australia Ltd	Sell	4,537	(536)	Ordinary
14	UBS Securities Australia Ltd	Sell	6,163	(728)	Ordinary
4 1 4	UBS Securities Australia Ltd	Sell	2,540	(300)	Ordinary
1 1 1	UBS Securities Australia Ltd	Sell	24,012	(2,835)	Ordinary
17 17	UBS Securities Australia Ltd	3	12,865	(1,519)	Ordinary
4-14	URS Securities Australia 11d	Sell	3,465	(000,1)	Ordinary
4-14	UBS Securities Australia Ltd	100	7750	(920)	Ordinary
ct-14	UBS Securities Australia Ltd	Sell	4 323	(517)	Ordinary
ct-14	UBS Securities Australia Ltd	Sell	3.587	(424)	Ordinary
ct-14	UBS Securities Australia Ltd	Sell	11,387	(1,346)	Ordinary
ct-14	UBS Securities Australia Ltd	Sell	34,947	(4,126)	Ordinary
15-Oct-14	UBS Securities Australia Ltd	Sell	8,465	(1,000)	Ordinary
4-14	UBS Securities Australia Ltd	Self	3,515	(415)	Ordinary
5.04.14	UBS Securities Australia Ltd	Set	839	(66)	Ordinary
114	URS Securities Australia Ltd	i S	59,654	(7,043)	Ordinary
5-0ct-14	UBS Securities Australia Ltd	3	197'6	(1501)	Ordinary
ct-14	UBS Securities Australia Ltd	Sell	13.680	(1,617)	Ordinary
5-Oct-14	UBS Securities Australia Ltd	Sell	6,615	(781)	Ordinary
ct-14	UBS Securities Australia Ltd	Sell	771,1	(139)	Ordinary
5-Oct-14	UBS Securities Australia Ltd	Sell	5,726	(929)	Ordinary
5-Oct-14	UBS Securities Australia Ltd	Sell	50,447	(956'5)	Ordinary
5-Oct-14	UBS Securities Australia Ltd	Sell	84,700	(10,000)	Ordinary
1 1 1	UBS Securities Australia Ltd	Nell Nell	22,818	(2,694)	Ordinary
5-001-14 5-04-14	TIDE Counties Australia Ltd	le N	2,541	(300)	Ordinary
1 2	UDC Securities Australia Ltd	nec o	18,211	(2,150)	Ordinary
14	UBS Securities Australia 11d		6,093	(2514)	Ordinary
15-Oct-14	UBS Securities Australia Ltd	Sell	2.948	(348)	Ordinary
ct-14	UBS Securities Australia Ltd	Sell	29,560	(3,490)	Ordinary
ct-14	UBS Securities Australia Ltd	Sell	6,132	(724)	Ordinary
ct-14	UBS Securities Australia Ltd	les	2,541	(300)	Ordinary
ct-14	UBS Securities Australia Ltd	Sell	2,118	(250)	Ordinary
ct-14	UBS Securities Australia Ltd	Sell	1,143	(135)	Ordinary
14	UBS Securities Australia Ltd	Sell	847	(100)	Ordinary
ct-14	UBS Securities Australia Ltd	Sell	5,497	(649)	Ordinary
ct-14	UBS Securities Australia Ltd	Sell	200	(65)	Ordinary
41.4	UBS Securities Australia Ltd	Sell	3,392	(401)	Ordinary
4 14	UBS Securities Australia Ltd	Sell	2,812	(332)	Ordinary
41.17	UBS Securities Australia Ltd	les Sell	127	(15)	Ordinary
1 2	LIBS Securities Australia Ltd	Sell	940	(100)	Ordinary
4-14	UBS Securities Australia Ltd	Seil	00/1	(201)	Ordinary
4-14	UBS Securities Australia Itd	Sell	27 50	6 6	Ordinary
41-14	UBS Securities Australia Ltd	Sel	212	(52)	Ordinary
t-14	UBS Securities Australia Ltd	Sell	618	(23)	Ordinary
ct-14	UBS Securities Australia Ltd	Sell	87,984	(10,400)	Ordinary
ct-14	UBS Securities Australia Ltd	Sell	211,500	(25,000)	Ordinary
5-Oct-14	UBS Securities Australia Ltd	Sell	2,985	(357)	Ordinary
5-Oct-14	UBS Securities Australia Ltd	Seli	1,162	(139)	Ordinary
5-Oct-14	UBS Securities Australia Ltd	Sell	16,825	(2,015)	Ordinary
5-Oct-14	UBS Securities Australia Ltd	Sell	1,328	(159)	Ordinary
5-0d-14	UBS Securities Australia Ltd	Sell	4,142	(496)	Ordinary
41-14	UBS Securities Australia Ltd	Sell	4,623	(223)	Ordinary
4 :	UBS Securities Australia Ltd	- Xell	1,455	(174)	Ordinary
4 14	UBS Securities Australia Ltd	Sell Sell	8,365	(1,000)	Ordinary
5.04.14	LIBS Securities Australia Ltd	iles iles	505.9	(826)	Ordinary
t-14	UBS Securities Australia Itd	100	765	(1/2)	Ordinary
15-Oct-14	UBS Securities Australia Ltd	Sell	4.034	(482)	Ordinary
1.14	UBS Securities Australia Ltd	Seli	4 400	/ / /	(in in)
				(173)	Ordinan
S.T.	I IBS Securities Australia 14d	3	1,450	(173)	Ordinary

1 3				7001	
15-Oct-14	UBS Securities Australia Ltd	Sell	85		Ordinary
Oct-14	UBS Securities Australia Ltd	Sell	1,381		Ordinary
41-14	USS Securities Australia Ltd	Sel.	2,940	(707)	Ordinary
04:14	LIBS Securities Australia Ind	io jo	11 721		Ordinary
Oct.14	IIRS Societies Australia Itd	100	5 QAK		Ordinary
04-14	II BS Securities Australia I td	100	17.667	(780 C)	Ordinary
ı	UBS Securities Australia Ltd	Sell	6.721	(762)	Ordinary
ı	UBS Securities Australia Ltd	Sell	804	(56)	Ordinary
ļ	1085 Securities Australia Ltd	Sell	8.465	(000 t)	Ordinary
ĺ	UBS Securities Australia Ltd	Sell	2,666	(315)	Ordinary
	UBS Securities Australia Ltd	Sell	21.916	(2.589)	Ordinary
ı	UBS Securities Australia Ltd	35	6 780	(801)	Ordinary
l	UBS Securities Australia Ltd	Jes	13.036	(1540)	Ordinary
ı	UBS Securities Australia Ltd	Seil	1.084	(128)	Ordinary
15-Oct-14	UBS Securities Australia Ltd	Sell	203	(24)	Ordinary
ı	UBS Securities Australia Ltd	Sell	10.088	(1.191)	Ordinary
l	1085 Securities Australia Ltd	Sell	8.470	(000 t)	Ordinary
1	UBS Securities Australia Ltd	Sell	3.396	(401)	Ordinary
15-Oct-14	UBS Securities Australia Ltd	Sell	2 541	(JOE)	Ordinary
l	UBS Securities Australia Ltd	Sell	7.589	(968)	Ordinary
	UBS Securities Australia Ltd	Sell	881	(104)	Ordinary
۱	UBS Securities Australia Ltd	Se	881	(104)	Ordinary
	1185 Securities Australia 11d	las	25.5	(656)	Ordinary
	LIRS Securities Australia 11d	100	19 744	(0.00)	Ordinary
5 Oct 14	LIBS Securities Australia 11d	la Control	12.485	(1 474)	Ordinary
5.Oct-14	IIBS Securities Australia Itd	Coll	5 531	(653)	Ordinary
27.77	II IBS Societies Australia (+d)	Sell	7 673	(GGB)	Ordina
5-04-14	IIIRS Securities Australia I to	la)	2,029	(050)	Ordinary
24.14	UDC Countries Australia 12d	Cell	1013	(000)	Ordinani
5-Oct-14	I IBS Securities Australia I to	100	5,073		Ordinary
.Oct.14	I IRC Securities Australia I tol	Coll	NEC 2		Credinary
00.13	LIBS Societies Australia 14d	100	900.01	(1010)	Ordinary
1 1	Life Countries Australia Ltd	Coll	25,000		Guillay
1 1 2	Cop Securities Australia Lid	136	15.72	(300)	Ordinary
ı	USS Securities Australia Ltd	) dell	811,2		Ordinary
	UBS Securities Australia Ltd	Set	11,130		Ordinary
5-Oct-14	UBS Securities Australia Ltd	Sell	11,943		Ordinary
1	UBS Securities Australia Ltd	Sell	16,821		Ordinary
	UBS Securities Australia Ltd	Sell	2,633	(311)	Ordinary
	UBS Securities Australia Ltd	Sell	7,788		Ordinary
	UBS Securities Australia Ltd	Sell	4,910		Ordinary
5-0ct-14	UBS Securities Australia Ltd	Sell	6,831		Ordinary
	UBS Securities Australia Ltd	Sell	2 540		Ordinary
5.Oct.14	I BS Socurdios Australia I tol	Coll	000 99		Ordinan
l	Libo Copusition Australia 144	70	17 494		
ı	Coo Securities Australia Ltd	300	404'/	(2,003)	Countries
1	UBS Securities Australia Ltd	iac ii	16,730		Ordinary
1	UBS Securities Australia Ltd	les	38,909		Ordinary
1	UBS Securities Australia Ltd	Sell	52,231	(6,163)	Ordinary
	UBS Securities Australia Ltd	Sell	22,162	(2,615)	Ordinary
	UBS Securities Australia Ltd	Sell	7,085	(836)	Ordinary
l	UBS Securities Australia Ltd	Sell	5,866	(663)	Ordinary
	UBS Securities Australia Ltd	Sell	5.876	(569)	Ordinary
Oct-14	UBS Securities Australia Ltd	Seil	75	9	Ordinary
15-Oct-14	LIRS Securities Australia Ltd	Sell	16 902	(1 000)	Ordinary
17	PS Conviction August 144	100	12 596	(605/1)	Ordinan
111	und described Academic Ltd.	11-3	Occ.,c.	(900,1)	Grany
70-14	USS Securities Australia Ltd	iiac	175	(38)	Ordinary
2ct-14	UBS Securities Australia Ltd	Sell	22,719	(2,687)	Ordinary
Jet-14	UBS Securities Australia Ltd	Sell	2,504	(962)	Ordinary
Oct-14	UBS Securities Australia Ltd	Sell	14,762	(1,746)	Ordinary
24-14	UBS Securities Australia Itd	Sell	16716	(1 977)	Ordinary
110 011	I IDC Constitute Australia 144	100	3.440	(100)	O'CLINE Y
11.14	Obstantials Australia Liu	i i i	614	(100)	Grandary
15-Oct-14	UBS Securities Australia Ltd	Sell .	8118	(14)	Oronary
Jct-14	UBS Securities Australia Ltd	Sell	4,230	(200)	Ordinary
2ct-14	UBS Securities Australia Ltd	Sell	ccc	(76)	Ordinary
)ct-14	UBS Securities Australia Ltd	Sell	3,384	(400)	Ordinary
)ct-14	UBS Securities Australia Ltd	Sell	3,866	(457)	Ordinary
Oct-14	UBS Securities Australia Ltd	Jes .	1.134	(134)	Ordinary
Det-14	I IRS Securities Australia I to	9	11 176	(1 221)	Ordinary
YEL:		368	9/1/1	(176'1)	Clumary
15-Oct-14		Sell	31,564	(15/3)	Ordinary
2ct-14		Sell	833	(66)	Ordinary
)ct-14	UBS Securities Australia Ltd	Seli	280	(33)	Ordinary
71-17	UBS Securities Australia Ltd	Sell	2,914	(344)	Ordinary
Dr.+-14	LIDS Counties Australia 14st	3	203	100/	
71.5	I III Societies Australia	3		******	

OS SCORING ARM Management         Sent         10,550           USS Scoringe Abart Nationagement         BAI         10,500           USS Scoringe Abart Nationagement         BAI         10,000           USS Scoringe Abart Nationagement         BAI         10,000           USS Scoringe Abart National Lead         BAI         10,000 <th>anagement</th> <th>ne.</th> <th>C8C'Q)</th> <th></th> <th>Ordinary</th>	anagement	ne.	C8C'Q)		Ordinary
1675 Securiors Australia 11	21126111	Bin	160 967		Ordinan
1915 Securities Autoriale Ltd   804   810   100 (2021   101 (202	alia Ltd	Sell	58.488	(106.9)	Ordinary
1955 Securities Autorities 1441   Seal     1955 Securities Autorities 1441   Seal     1955 Securities Autorities 1441   Seal     1955 Securities Autorities 1441   Seal     1955 Securities Autorities 1441   Seal     1955 Securities Autorities 1441   Seal     1955 Securities Autorities 1441   Seal   Seal   Seal   Seal     1955 Securities Autorities 1441   Seal   Seal   Seal   Seal   Seal     1955 Securities Autorities 1441   Seal   Seal   Seal   Seal   Seal     1955 Securities Autorities 1441   Seal   Seal   Seal   Seal   Seal     1955 Securities Autorities 1441   Seal   Seal   Seal   Seal   Seal   Seal   Seal     1955 Securities Autorities 1441   Seal   alia i tri	Ruv	8 500		Ordinany	
ONE Securines Autorities (LORG)         Securine Autorities (LORG)	official and	, no.	000,0	-	Ordinary
USS Securine Autoritia (10)   Serior    alia titu	Jen J	104,023	(12,140)	Ordinary	
UNE Securines Australia Ltd         Buy         B.02           USE Securines Australia Ltd         Serial         8.40           USE Securines Australia Ltd         Serial         8.40           USE Securines Australia Ltd         Serial         8.60           USE Securines Australia Ltd         Buy         8.62           USE Securines Australia Ltd         Buy         8.63           USE Securines Australia Ltd         Buy         8.63           USE Securines Australia Ltd         Buy         8.63           USE Securines Australia Ltd         Serial         8.61           USE Securines Australia Ltd         Serial         8.61 </td <td>alla Liu</td> <td>pro)</td> <td>254,243</td> <td></td> <td>Ordinary</td>	alla Liu	pro)	254,243		Ordinary
USS Securities Autorials Ltd	atia Liu	proj.	870'6		Ordinary
URS Securities Australia to 10         Seal         24,210           URS Securities Australia to 10         Seal         24,210           URS Securities Australia to 10         Seal         70,227           URS Securities Australia to 10         Seal         70,227           URS Securities Australia to 10         Seal         70,227           URS Securities Australia to 10         Seal         80,9         68,64           URS Securities Australia to 10         Buy         68,43         81,246           URS Securities Australia to 10         Buy         68,548         81,246           URS Wealth Management Australia to 10         Buy         68,548         81,246           URS Securities Australia to 10         Seal         1,149         81,248           URS Weath Management Australia to 10         Seal         2,150         91,248           URS Weath Management Australia to 10         Seal         1,140         91,248           URS Weath Management Australia to 10         Seal         1,140 <td>alia Liu</td> <td>) aer</td> <td>8,420</td> <td>(1)</td> <td>Ordinary</td>	alia Liu	) aer	8,420	(1)	Ordinary
URS Securities Australia tot         Sell         6.6564           URS Securities Australia tot         Sell         7.227           URS Securities Australia tot         Sell         6.64381           URS Securities Australia tot         Buy         6.64381           URS Act London Branch         Buy         6.64381 <t< td=""><td>alia Ltd</td><td>Buy</td><td>4,210</td><td></td><td>Ordinary</td></t<>	alia Ltd	Buy	4,210		Ordinary
URS Securities Australia to 10         Seril         70,237           URS Securities Australia to 10         Sini Total Seril         70,237           URS Securities Australia to 10         Buy         684,858           URS Securities Australia to 10         Buy         684,858           URS Securities Australia to 10         Buy         664,881           URS ACL London Branch         Buy         664,881           URS Securities Australia to 10         Buy         664,881           URS Securities Australia to 10         Buy         66,886           URS Securities Australia to 10         Buy         6,886           URS Securities Australia to 10         Seril         86,986           URS Securities Australia to 10         Seril         2,150	alia Ltd	Sei	249,694		Ordinary
URS Scormine Australia Ind D         Serial         70.227           URS Scormine Australia Ind D         Buy         258.258           URS Scormine Australia Ind D         Buy         66.438           URS Mealth Management Australia Ind D         Buy         66.438           URS Wealth Management Australia Ind D         Buy         16.731           URS Wealth Management Australia Ind D         Buy         1.149           URS Wealth Management Australia Ind D         Sell         1.020           URS Scormine Australia Ind D         Sell         1.020           URS Scormine Australia Ind D         Sell         2.580           URS Scormine Australia Ind D         Sell         2.150           URS	alia Ltd	Seil	6,856		Ordinary
URS Securities Australia It UD         Buy         558.255           URS Securities Australia It UD         Buy         664.380           URS Securities Australia It UD         Buy         664.381           URS Act London Branch         Buy         10.09           URS Act London Branch         Buy         664.868           URS Act London Branch         Buy         664.868           URS Water In Management Australia It UD         Buy         673.289           URS Securities Australia It UD         Sell         7783.00           URS Securities Australia It UD         Sell         77.380           URS Securities Australia It UD         Sell         7.340 <td>alia Ltd</td> <td>Sell</td> <td>70,227</td> <td></td> <td>Ordinary</td>	alia Ltd	Sell	70,227		Ordinary
URS Securities Australia Ind 1         Sel1         Sel2 58 591           URS Securities Australia Ind 1         Buy 1         SER 58 501           URS Securities Australia Ind 1         Buy 1         SER 58 300           URS AG, London Branch 1         Buy 1         SER 58 501           URS AG, London Branch 1         Buy 1         SER 58 501           URS AG, London Branch 1         Buy 1         SER 58 501           URS AG, London Branch 1         Buy 1         SER 58 501           URS Machin Management Australia Ind 1         Buy 1         SER 58 501           URS Weath Management Australia Ind 1         Buy 1         SER 501           URS Weath Management Australia Ind 1         Buy 1         SER 500           URS Securities Australia Ind 2         Sel1         1 149           URS Securities Australia Ind 2         Sel1         1 150           URS Securities Australia Ind 2         Sel1         2 150           URS Securities Australia Ind 2         Sel1         2 150           URS Securities Australia Ind 2         Sel1         2 150           URS Securities Australia Ind 2         Sel1         2 150           URS Securities Australia Ind 2         Sel1         2 150           URS Securities Australia Ind 2         Sel1         2 15	alia Ltd	Buy	258,285	30,641	Ordinary
UBS Securities Australia Ltd         Buy         654,280           UBS Securities Australia Ltd         Self         654,280           UBS Securities Australia Ltd         Buy         654,280           UBS ACG, London Branch         Buy         658,680           UBS Wealth Management Australia Ltd         Buy         102,000           UBS Wealth Management Australia Ltd         Buy         10,000           UBS Wealth Management Australia Ltd         Buy         1,1,400           UBS Wealth Management Australia Ltd         Buy         1,1,600           UBS Wealth Management Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         2,1,580           UBS Securit	alia Ltd	Sell	86,491	(10,176)	Ordinary
USS Securities Australia Ltd         Buy         654,380           USS AG, Lordon Branch         Bold         654,380           USS AG, Lordon Branch         Buy         654,381           USS AG, Lordon Branch         Buy         654,381           USS AG, Lordon Branch         Buy         658,687           USS AG, Lordon Branch         Buy         658,886           USS AG, Lordon Branch         Buy         658,886           USS AG, Lordon Branch         Buy         658,886           USS Wealth Management Australia Ltd         Buy         1021           USS Wealth Management Australia Ltd         Buy         15,731           USS Wealth Management Australia Ltd         Buy         15,731           USS Wealth Management Australia Ltd         Buy         1,532           USS Securities Australia Ltd         See         1,532           USS Securities Australia Ltd         See         2,580           USS Securities Australia Ltd         See         2,156           USS Securities Australia Ltd         See         <	alia Ltd	Buy	658,663	161,77	Ordinary
UBS ACC, Lordon Rearch         Seri         SE1.265           UBS ACC, Lordon Rearch         Buy         654.28           UBS ACC, Lordon Rearch         Buy         658.66           UBS ACC, Lordon Rearch         Buy         658.64           UBS Wealth Management Australia Ltd         Buy         658.64           UBS Wealth Management Australia Ltd         Buy         689.65           UBS Wealth Management Australia Ltd         Buy         1,449           UBS Wealth Management Australia Ltd         Buy         1,539           UBS Securities Australia Ltd         Buy         1,530           UBS Securities Australia Ltd         Sell         2,386           UBS Securities Australia Ltd         Sell	alia Ltd	Buy	654,380	77,192	Ordinary
USS ACC (Lordon Banch)         Bay         654,359           USS ACC, Lordon Banch         Bay         654,359           USS ACC, Lordon Banch         Bay         654,359           USS ACC, Lordon Banch         Bay         10.73           USS ACC, Lordon Banch         Bay         10.73           USS Wealth Management Australia Ltd         Bay         15,731           USS Wealth Management Australia Ltd         Bay         1,149           USS Wealth Management Australia Ltd         Bay         1,149           USS Wealth Management Australia Ltd         Bay         1,149           USS Securities Australia Ltd         Bay         1,149           USS Securities Australia Ltd         Sell         2,550           USS Securities Australia Ltd         Sell         2,550           USS Securities Australia Ltd         Sell         3,550           USS Securities Australia Ltd         Sell </td <td>anch</td> <td>Sell</td> <td>812,461</td> <td>(77, 192)</td> <td>Ordinary</td>	anch	Sell	812,461	(77, 192)	Ordinary
USS ACC, London Branch         Buy         CSS BEST           USS ACC, London Branch         Buy         B17,783           USS ACC, London Branch         Buy         B17,783           USS ACC, London Branch         Buy         B17,783           USS Wealth Management Australia Ltd         Buy         B17,783           USS Wealth Management Australia Ltd         Buy         B2,583           USS Wealth Management Australia Ltd         Buy         L1,489           USS Wealth Management Australia Ltd         Buy         L1,489           USS Wealth Management Australia Ltd         Buy         L1,489           USS Securities Australia Ltd         Seel         B2,500           USS Securities Au	anch	Buv	654 381	77 192	Ordinary
USS NG, London Banch         Sell         817,788           USS NG, London Banch         Bay         817,788           USS Wealth Management Australia Ltd         Bay         48,648           USS Wealth Management Australia Ltd         Bay         1,449           USS Wealth Management Australia Ltd         Bay         1,449           USS Wealth Management Australia Ltd         Bay         1,449           USS Wealth Management Australia Ltd         Sell         1,449           USS Wealth Management Australia Ltd         Sell         1,430           USS Wealth Management Australia Ltd         Sell         2,528           USS Securities Australia Ltd         Sell         2,528           USS Securities Australia Ltd         Sell         2,750           USS Securi	anch.	Riv	658 667	77 191	Ordinan
USS Yeachth Management Australia Ltd         Buy         66.448           USS Weathh Management Australia Ltd         Buy         66.448           USS Weathh Management Australia Ltd         Buy         10.93           USS Securities Australia Ltd         Sell         20.50           USS Securities Australia Ltd         Sell         20.50           USS Securities Australia Ltd         Sell         21.50	400	Coll	700,000	(101.77)	Cidinaly
USS Yealth Management Australia Ltd         Buy         65,743           USS Wealth Management Australia Ltd         Buy         65,743           USS Wealth Management Australia Ltd         Buy         1,149           USS Wealth Management Australia Ltd         Sel         1,149           USS Wealth Management Australia Ltd         Sel         2,188           USS Securities Australia Ltd         Sel         2,180           USS Securities Australia Ltd         Sel         2,150           USS Securities Australia Ltd         Sel         2,132           USS	mont Authority 144		100,	(161,77)	Ordinary
USS Vestiti Namesperiest Australia Ltd         Buy         15, 234           USS Vestiti Namesperiest Australia Ltd         Buy         15, 234           USS Vestiti Namesperiest Australia Ltd         Buy         1, 14, 49           USS Vestiti Namesperiest Australia Ltd         Buy         1, 14, 49           USS Vestiti Namesperiest Australia Ltd         Buy         1, 14, 49           USS Vestiti Namesperiest Australia Ltd         Sel         2, 58           USS Vestitives Australia Ltd         Sel         2, 58           USS Securities Australia Ltd         Sel         2, 150           USS Securities Australia Ltd         Sel         2, 250           USS Securities Australia Ltd         Sel         1, 422     <	ment Australia Ltd	ond.	1,021	071	Ordinary
USS Weath Management Australia Ltd         Buy         11.49           USS Weath Management Australia Ltd         Buy         11.49           USS Weath Management Australia Ltd         Buy         1.149           USS Weath Management Australia Ltd         Sel         1.149           USS Weath Management Australia Ltd         Sel         2.989           USS Weath Management Australia Ltd         Sel         2.189           USS Weath Management Australia Ltd         Sel         2.189           USS Securities Australia Ltd         Sel         2.189           USS Securities Australia Ltd         Sel         2.150           USS Securities Australia Ltd         Sel         2.100           USS Securities Australia Ltd         Sel         2.100           USS Securities Australia Ltd         Sel         2.100           USS Securities Australia Ltd         Sel         2.200           USS Securities Aust	STIEST AUSTRALIA LIG	Buy	48,548	51/15	Ordinary
USS Weath Management Australia Ltd   Buy   1.499   1	ment Australia Ltd	any	15,/31	1,848	Ordinary
UBS Securities Australia Ltd         Buy         1,149           UBS Securities Australia Ltd         Buy         1,980           UBS Securities Australia Ltd         Seli         4,980           UBS Securities Australia Ltd         Seli         2,980           UBS Securities Australia Ltd         Seli         7,923           UBS Securities Australia Ltd         Seli         7,520           UBS Securities Australia Ltd         Seli         2,580           UBS Securities Australia Ltd         Seli         2,703           UBS Securities Australia Ltd         Seli         2,704           UBS Securities Australia Ltd <th< td=""><td>ement Australia Ltd</td><td>Buly</td><td>808</td><td>95</td><td>Ordinary</td></th<>	ement Australia Ltd	Buly	808	95	Ordinary
UBS Yeachtife Australia Ltd         Buy         1,090           UBS Wealth Management Australia Ltd         Sein         2,988           UBS Securities Australia Ltd         Seil         2,988           UBS Securities Australia Ltd         Seil         782,000           UBS Securities Australia Ltd         Seil         7,32,89           UBS Securities Australia Ltd         Seil         7,32,89           UBS Securities Australia Ltd         Seil         2,150           UBS Securities Australia Ltd         Seil         3,050           UBS Securities Australia Ltd         Seil         3,050           UBS Securities Australia Ltd         Seil         1,342           UBS Securities Australia Ltd         Seil         2,047           UBS Securities Australia Ltd         Seil         1,320           UBS Securities Australia Ltd <td>ement Australia Ltd</td> <td>Buy</td> <td>1,149</td> <td>135</td> <td>Ordinary</td>	ement Australia Ltd	Buy	1,149	135	Ordinary
UBS Securities Australia Ltd         Bluy         4,988           UBS Securities Australia Ltd         Sell         430,000           UBS Securities Australia Ltd         Sell         78,300           UBS Securities Australia Ltd         Sell         73,289           UBS Securities Australia Ltd         Sell         73,230           UBS Securities Australia Ltd         Sell         2,500           UBS Securities Australia Ltd         Sell         2,709           UBS Securities Australia Ltd         Sell         2,709           UBS Securities Australia Ltd         Sell         1,542           UBS Securities Australia Ltd         Sell         2,709           UBS Securities Australia Ltd         Sell         1,542           UBS Securities Australia Ltd         Sell         2,709           UBS Securities Australia Ltd         Sell         2,709           UBS Securities Australia Ltd         Sell         3,700           UBS Securities Australia Ltd         Sell         2,700           UBS Securities Australia Ltd	ement Australia Ltd	Buy	060'1	128	Ordinary
UBS Securities Australia Ltd         Seli         48,000           UBS Securities Australia Ltd         Seli         78,000           UBS Securities Australia Ltd         Seli         73,269           UBS Securities Australia Ltd         Seli         7,269           UBS Securities Australia Ltd         Seli         2,269           UBS Securities Australia Ltd         Seli         2,150           UBS Securities Australia Ltd         Seli         2,709           UBS Securities Australia Ltd         Seli         2,500           UBS Securities Australia Ltd         Seli         3,000           UBS Securities Australia Ltd         Seli         3,000           UBS Securities Australia Ltd         Seli         1,362           UBS Securities Australia Ltd	ement Australia Ltd	Buy	2,988		Ordinary
UBS Securities Australia tod         Sel         866.500           UBS Securities Australia tod         Sell         73,269           UBS Securities Australia tod         Sell         23,269           UBS Securities Australia tod         Sell         21,50           UBS Securities Australia tod         Sell         2,50           UBS Securities Australia tod         S	alia Ltd	Sell	430,000		Ordinary
UBS Securities Australia Ltd         Seil         789,000           UBS Securities Australia Ltd         Seil         2,589           UBS Securities Australia Ltd         Seil         2,580           UBS Securities Australia Ltd         Seil         2,150           UBS Securities Australia Ltd         Seil         2,709           UBS Securities Australia Ltd         Seil         3,00           UBS Securities Australia Ltd         Seil         2,00           UBS Securities Australia Ltd         Seil         2,00           UBS Securities Australia Ltd         Seil         2,00           UBS Securities Australia Ltd         Sei	alia Ltd	Sett	856,500		Ordinary
UBS Securities Australia Ltd         Seil         22.89           UBS Securities Australia Ltd         Seil         15.00           UBS Securities Australia Ltd         Seil         15.00           UBS Securities Australia Ltd         Seil         27.00           UBS Securities Australia Ltd         Seil         27.00           UBS Securities Australia Ltd         Seil         27.00           UBS Securities Australia Ltd         Seil         27.50           UBS Securities Australia Ltd         Seil         27.30           UBS Securities Australia Ltd         Seil         2.50           UBS Securities Australia Ltd         Se	alia Ltd	Sell	783,000		Ordinary
UBS Securities Australia Ltd         Seil         15,500           UBS Securities Australia Ltd         Seil         13,023           UBS Securities Australia Ltd         Seil         2,500           UBS Securities Australia Ltd         Seil         2,709           UBS Securities Australia Ltd         Seil         2,709           UBS Securities Australia Ltd         Seil         2,709           UBS Securities Australia Ltd         Seil         2,500           UBS Securities Australia Ltd         Seil         2,500           UBS Securities Australia Ltd         Seil         2,500           UBS Securities Australia Ltd         Seil         3,000           UBS Securities Australia Ltd         Seil         3,000           UBS Securities Australia Ltd         Seil         3,000           UBS Securities Australia Ltd         Seil         1,462           UBS Securities Australia Ltd         Seil         2,500           UBS Securities Australia Ltd         Seil         2,500           UBS Securities Australia Ltd         Seil         1,204           UBS Securities Australia Ltd         Seil         2,500           UBS Securities Australia Ltd         Seil         1,200           UBS Securities Australia Ltd	alia Itd	Şef	73 789		Ordinan
UBS Securities Australia Ltd         Sell         13,923           UBS Securities Australia Ltd         Sell         3,566           UBS Securities Australia Ltd         Sell         27,709           UBS Securities Australia Ltd         Sell         27,709           UBS Securities Australia Ltd         Sell         1,548           UBS Securities Australia Ltd         Sell         1,548           UBS Securities Australia Ltd         Sell         2,150           UBS Securities Australia Ltd         Sell         2,150           UBS Securities Australia Ltd         Sell         3,000           UBS Securities Australia Ltd         Sell         1,482           UBS Securities Australia Ltd         Sell         2,300           UBS Securities Australia Ltd         Sell         1,300           UBS Securities Australia Ltd         Sell         1,300           UBS Securities Australia Ltd	D+1 0 0	4 0	007/07		O C
UBS Securities Australia Ltd         Sell         2.150           UBS Securities Australia Ltd         Sell         3.566           UBS Securities Australia Ltd         Sell         3.566           UBS Securities Australia Ltd         Sell         1.548           UBS Securities Australia Ltd         Sell         1.548           UBS Securities Australia Ltd         Sell         1.548           UBS Securities Australia Ltd         Sell         2.786           UBS Securities Australia Ltd         Sell         2.787           UBS Securities Australia Ltd         Sell         2.788           UBS Securities Australia Ltd         Sell         2.780           UBS Securities Australia Ltd         <	alia Ital	Self.	2,300		Ordinary
UBS Securities Australia Ltd         Sell         27.709           UBS Securities Australia Ltd         Sell         27.709           UBS Securities Australia Ltd         Sell         27.709           UBS Securities Australia Ltd         Sell         15.48           UBS Securities Australia Ltd         Sell         2.750           UBS Securities Australia Ltd         Sell         2.750           UBS Securities Australia Ltd         Sell         2.786           UBS Securities Australia Ltd         Sell         3.900           UBS Securities Australia Ltd         Sell         3.000           UBS Securities Australia Ltd         Sell         2.780           UBS Securities Australia Ltd         Sell         2.780           UBS Securities Australia Ltd         Sell         2.780           UBS Securities Australia Ltd         Sell         1.7340           UBS Securities Australia Ltd         Sell         1.200           UBS Securities Australia Ltd	200	350	13,523		Cicioary
UBS Securities Australia Ltd         Sell         3.68           UBS Securities Australia Ltd         Sell         3.58           UBS Securities Australia Ltd         Sell         41,667           UBS Securities Australia Ltd         Sell         15,48           UBS Securities Australia Ltd         Sell         2,79           UBS Securities Australia Ltd         Sell         2,786           UBS Securities Australia Ltd         Sell         2,786           UBS Securities Australia Ltd         Sell         2,786           UBS Securities Australia Ltd         Sell         3,70           UBS Securities Australia Ltd         Sell         3,70           UBS Securities Australia Ltd         Sell         1,462           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         1,597           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         1,204           UBS Securities Australia Ltd         Sell         1,204           UBS Securities Australia Ltd         Sell         1,204           UBS Securities Australia Ltd         S	alia Ltd	Sell	051,2		Ordinary
UBS Securities Australia Ltd         Sell         23,586           UBS Securities Australia Ltd         Sell         27,709           UBS Securities Australia Ltd         Sell         1,548           UBS Securities Australia Ltd         Sell         1,548           UBS Securities Australia Ltd         Sell         2,786           UBS Securities Australia Ltd         Sell         2,786           UBS Securities Australia Ltd         Sell         3,000           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         1,587           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         1,2,000           UBS Securities Australia Ltd	alia Ltd	Sell	9,813		Ordinary
UBS Securities Australia Ltd         Sell         2.709           UBS Securities Australia Ltd         Sell         1,546           UBS Securities Australia Ltd         Sell         1,546           UBS Securities Australia Ltd         Sell         2,786           UBS Securities Australia Ltd         Sell         2,786           UBS Securities Australia Ltd         Sell         2,786           UBS Securities Australia Ltd         Sell         30,900           UBS Securities Australia Ltd         Sell         2,647           UBS Securities Australia Ltd         Sell         3,595           UBS Securities Australia Ltd         Sell         1,442           UBS Securities Australia Ltd         Sell         1,242           UBS Securities Australia Ltd         Sell         1,387           UBS Securities Australia Ltd         Sell         1,387           UBS Securities Australia Ltd         Sell         1,204           UBS Securities Australia Ltd	alia Ltd	Sell	3,586		Ordinary
UBS Securities Australia Ltd         Sell         41,667           UBS Securities Australia Ltd         Sell         15,88           UBS Securities Australia Ltd         Sell         2,786           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         30,900           UBS Securities Australia Ltd         Sell         3,047           UBS Securities Australia Ltd         Sell         1,342           UBS Securities Australia Ltd         Sell         1,342           UBS Securities Australia Ltd         Sell         1,362           UBS Securities Australia Ltd         Sell         2,500           UBS Securities Australia Ltd         Sell         1,204           UBS Securities Australia Ltd         Sell         1,204           UBS Securities Australia Ltd         Sell         2,500           UBS Securities Australia Ltd         Sell         1,204           UBS Securities Australia Ltd	alia Ltd	Sell	27,709		Ordinary
UBS Securities Australia Ltd         Sell         1,548           UBS Securities Australia Ltd         Sell         1,548           UBS Securities Australia Ltd         Sell         2,786           UBS Securities Australia Ltd         Sell         2,786           UBS Securities Australia Ltd         Sell         30,900           UBS Securities Australia Ltd         Sell         3,000           UBS Securities Australia Ltd         Sell         2,047           UBS Securities Australia Ltd         Sell         1,482           UBS Securities Australia Ltd         Sell         1,482           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         1,332           UBS Securities Australia Ltd         Sell         2,587           UBS Securities Australia Ltd         Sell         1,287           UBS Securities Australia Ltd         Sell         1,290           UBS Securities Australia Ltd         Sell         2,232           UBS Securities Australia Ltd	əlia Ltd	Sell	41,667		Ordinary
UBS Securities Australia Ltd         Sell         19.135           UBS Securities Australia Ltd         Sell         2.786           UBS Securities Australia Ltd         Sell         30.900           UBS Securities Australia Ltd         Sell         30.900           UBS Securities Australia Ltd         Sell         3.00           UBS Securities Australia Ltd         Sell         2.647           UBS Securities Australia Ltd         Sell         1.342           UBS Securities Australia Ltd         Sell         1.462           UBS Securities Australia Ltd         Sell         2.580           UBS Securities Australia Ltd         Sell         2.731           UBS Securities Australia Ltd         Sell         2.730           UBS Securities Australia Ltd         Sell         1.204           UBS Securities Australia Ltd         Sell         2.730           UBS Securities Australia Ltd         Sell         2.730           UBS Securities Australia Ltd	alia Ltd	Sell	1.548		Ordinary
UBS Securities Australia Ltd         Sell         2,150           UBS Securities Australia Ltd         Sell         2,786           UBS Securities Australia Ltd         Sell         30,900           UBS Securities Australia Ltd         Sell         30,900           UBS Securities Australia Ltd         Sell         3,040           UBS Securities Australia Ltd         Sell         1,482           UBS Securities Australia Ltd         Sell         1,482           UBS Securities Australia Ltd         Sell         1,482           UBS Securities Australia Ltd         Sell         1,582           UBS Securities Australia Ltd         Sell         1,282           UBS Securities Australia Ltd         Sell         1,290           UBS Securities Australia Ltd         Sell         1,2,00           UBS Securities Australia Ltd	alia Ltd	Sell	19.135		Ordinary
UBS Securities Australia Ltd         Sell         2,786           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         370           UBS Securities Australia Ltd         Sell         2,047           UBS Securities Australia Ltd         Sell         33,953           UBS Securities Australia Ltd         Sell         1,462           UBS Securities Australia Ltd         Sell         1,650           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         2,322           UBS Securities Australia Ltd         Sell         1,204           UBS Securities Australia Ltd         Sell         1,200           UBS Securities Australia Ltd <t< td=""><td>alia Ltď</td><td>Sell</td><td>2.150</td><td></td><td>Ordinary</td></t<>	alia Ltď	Sell	2.150		Ordinary
UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         30,900           UBS Securities Australia Ltd         Sell         30,900           UBS Securities Australia Ltd         Sell         1,342           UBS Securities Australia Ltd         Sell         1,342           UBS Securities Australia Ltd         Sell         1,462           UBS Securities Australia Ltd         Sell         1,462           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         1,204           UBS Securities Australia Ltd         Sell         1,204           UBS Securities Australia Ltd         Sell         1,2,000           UBS Securities Australia Ltd         Sell         2,530           UBS Securities Australia Ltd         Sell         1,2,000           UBS Securities Australia Ltd         Sell         2,500           UBS Securities Australia Ltd	aliaito	97	787.0		Ordinan
UBS Securities Australia Ltd         Sell         20,300           UBS Securities Australia Ltd         Sell         370           UBS Securities Australia Ltd         Sell         2,047           UBS Securities Australia Ltd         Sell         1,342           UBS Securities Australia Ltd         Sell         33,953           UBS Securities Australia Ltd         Sell         1,442           UBS Securities Australia Ltd         Sell         1,987           UBS Securities Australia Ltd         Sell         1,987           UBS Securities Australia Ltd         Sell         1,387           UBS Securities Australia Ltd         Sell         1,387           UBS Securities Australia Ltd         Sell         8,310           UBS Securities Australia Ltd         Sell         1,204           UBS Securities Australia Ltd         Sell         1,200           UBS Securities Australia Ltd         Sell         2,200           UBS Securities Australia Ltd         Sell         2,200           UBS Securities Australia Ltd         <	100	1140	20,77		Cidinaly
UBS Securities Australia Ltd         Sell         30,900           UBS Securities Australia Ltd         Sell         30,900           UBS Securities Australia Ltd         Sell         1,342           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         1,204           UBS Securities Australia Ltd         Sell         1,204           UBS Securities Australia Ltd         Sell         1,2,00           UBS Securities Australia Ltd         Sell         2,530           UBS Securities Australia Ltd         Sell         2,500           UBS Securities Australia Ltd         Sell         2,500           UBS Securities Australia Ltd	alia tito	Jell Sell	7,580		Ordinary
UBS Securities Australia Ltd         Sell         370           UBS Securities Australia Ltd         Sell         2,047           UBS Securities Australia Ltd         Sell         1,342           UBS Securities Australia Ltd         Sell         1,462           UBS Securities Australia Ltd         Sell         1,462           UBS Securities Australia Ltd         Sell         1,462           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         1,33           UBS Securities Australia Ltd         Sell         9,73           UBS Securities Australia Ltd         Sell         8,910           UBS Securities Australia Ltd         Sell         1,204           UBS Securities Australia Ltd         Sell         2,205           UBS Securities Australia Ltd         Sell         2,205           UBS Securities Australia Ltd         Sell         2,200           UBS Securities Australia Ltd         S	alia Ltd	Sell	30,900		Ordinary
UBS Securities Australia Ltd         Sell         1,342           UBS Securities Australia Ltd         Sell         1,342           UBS Securities Australia Ltd         Sell         33,953           UBS Securities Australia Ltd         Sell         1,462           UBS Securities Australia Ltd         Sell         1,462           UBS Securities Australia Ltd         Sell         2,590           UBS Securities Australia Ltd         Sell         2,590           UBS Securities Australia Ltd         Sell         9,133           UBS Securities Australia Ltd         Sell         8,913           UBS Securities Australia Ltd         Sell         8,910           UBS Securities Australia Ltd         Sell         1,204           UBS Securities Australia Ltd         Sell         1,204           UBS Securities Australia Ltd         Sell         1,204           UBS Securities Australia Ltd         Sell         2,300           UBS Securities Australia Ltd	alia Ltd	Sell	370		Ordinary
UBS Securities Australia Ltd         Sell         1,342           UBS Securities Australia Ltd         Sell         5,108           UBS Securities Australia Ltd         Sell         1,462           UBS Securities Australia Ltd         Sell         1,462           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         731           UBS Securities Australia Ltd         Sell         731           UBS Securities Australia Ltd         Sell         8,910           UBS Securities Australia Ltd         Sell         8,910           UBS Securities Australia Ltd         Sell         1,7540           UBS Securities Australia Ltd         Sell         1,724           UBS Securities Australia Ltd         Sell         1,724           UBS Securities Australia Ltd         Sell         1,204           UBS Securities Australia Ltd         Sell         1,204           UBS Securities Australia Ltd         Sell         2,305           UBS Securities Australia Ltd         Sell         2,505           UBS Securities Australia Ltd         Sell         2,505           UBS Securities Australia Ltd         Sell         2,505           UBS Securities Australia Ltd	alia Ltd	Seli	2,047		Ordinary
UBS Securities Australie Ltd         Sell         33,953           UBS Securities Australie Ltd         Sell         33,953           UBS Securities Australia Ltd         Sell         1,462           UBS Securities Australia Ltd         Sell         2,500           UBS Securities Australia Ltd         Sell         2,500           UBS Securities Australia Ltd         Sell         1,387           UBS Securities Australia Ltd         Sell         23           UBS Securities Australia Ltd         Sell         89,133           UBS Securities Australia Ltd         Sell         2,43           UBS Securities Australia Ltd         Sell         1,204           UBS Securities Australia Ltd         Sell         1,204           UBS Securities Australia Ltd         Sell         1,206           UBS Securities Australia Ltd         Sell         2,500           UBS Securities Australia Ltd <t< td=""><td>alia Ltd</td><td>Sell</td><td>1 342</td><td></td><td>Ordinary</td></t<>	alia Ltd	Sell	1 342		Ordinary
UBS Securities Australia Ltd         Sell         33,953           UBS Securities Australia Ltd         Sell         1,462           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         731           UBS Securities Australia Ltd         Sell         9,133           UBS Securities Australia Ltd         Sell         9,133           UBS Securities Australia Ltd         Sell         8,910           UBS Securities Australia Ltd         Sell         1,794           UBS Securities Australia Ltd         Sell         1,794           UBS Securities Australia Ltd         Sell         1,204           UBS Securities Australia Ltd         Sell         1,204           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         2,500           UBS Securities Australia Ltd         Sell         2,500           UBS Securities Australia Ltd         Sell         2,500           UBS Securities Australia Ltd         Sell         2,524           UBS Securities Australia Ltd         Sell         4,555           UBS Securities Australia Ltd <t< td=""><td>alia Itd</td><td>(a)</td><td>5 108</td><td>(Pos)</td><td>Ordinan</td></t<>	alia Itd	(a)	5 108	(Pos)	Ordinan
UBS Securities Australia Ltd         Sell         1,462           UBS Securities Australia Ltd         Sell         9,000           UBS Securities Australia Ltd         Sell         9,000           UBS Securities Australia Ltd         Sell         1,387           UBS Securities Australia Ltd         Sell         22,200           UBS Securities Australia Ltd         Sell         22,20           UBS Securities Australia Ltd         Sell         8,910           UBS Securities Australia Ltd         Sell         2,32           UBS Securities Australia Ltd         Sell         1,204           UBS Securities Australia Ltd         Sell         1,2,900           UBS Securities Australia Ltd         Sell         1,2,900           UBS Securities Australia Ltd         Sell         2,300           UBS Securities Australia Ltd         Sell         2,500           UBS Securities Australia Ltd	7	100	200-10	(45C)	Cidillaly
UBS Securities Australia Ltd         Sell         1,482           UBS Securities Australia Ltd         Sell         1,587           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         731           UBS Securities Australia Ltd         Sell         9,133           UBS Securities Australia Ltd         Sell         9,133           UBS Securities Australia Ltd         Sell         8,910           UBS Securities Australia Ltd         Sell         17,940           UBS Securities Australia Ltd         Sell         17,940           UBS Securities Australia Ltd         Sell         17,940           UBS Securities Australia Ltd         Sell         12,500           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         4,555           UBS Securities Australia Ltd         Sell         4,555           UBS Securities Australia Ltd         Sell         8,524           UBS Securities Australia Ltd         Sell         6,552           UBS Securities Australia Ltd	alla Lio	261	606,66	(3,948)	Ordinary
UBS Securities Australia Ltd         Self         9080           UBS Securities Australia Ltd         Self         9,080           UBS Securities Australia Ltd         Self         1,387           UBS Securities Australia Ltd         Self         222           UBS Securities Australia Ltd         Self         28,390           UBS Securities Australia Ltd         Self         8,910           UBS Securities Australia Ltd         Self         2,23,380           UBS Securities Australia Ltd         Self         1,204           UBS Securities Australia Ltd         Self         1,2,900           UBS Securities Australia Ltd         Self         1,2,900           UBS Securities Australia Ltd         Self         2,320           UBS Securities Australia Ltd         Self         2,530           UBS Securities Australia Ltd         Self         2,530           UBS Securities Australia Ltd         Self         2,530           UBS Securities Australia Ltd	alia Ltd	Sell	1,462	(170)	Ordinary
UBS Securities Australia Ltd         Sell         2.550           UBS Securities Australia Ltd         Sell         731           UBS Securities Australia Ltd         Sell         732           UBS Securities Australia Ltd         Sell         9133           UBS Securities Australia Ltd         Sell         9133           UBS Securities Australia Ltd         Sell         8,910           UBS Securities Australia Ltd         Sell         17,940           UBS Securities Australia Ltd         Sell         17,940           UBS Securities Australia Ltd         Sell         17,940           UBS Securities Australia Ltd         Sell         12,900           UBS Securities Australia Ltd         Sell         2,500           UBS Securities Australia Ltd         Sell         3,524           UBS Securities Australia Ltd         Sell         6,5524           UBS Securities Australia Ltd         Sell         6,552           UBS Securities Australia Ltd         Sell         6,552           UBS Securities Australia Ltd <t< td=""><td>alia Ltd</td><td>Sell</td><td>060'6</td><td>(1,057)</td><td>Ordinary</td></t<>	alia Ltd	Sell	060'6	(1,057)	Ordinary
UBS Securities Australia Ltd         Sell         1.987           UBS Securities Australia Ltd         Sell         222           UBS Securities Australia Ltd         Sell         283           UBS Securities Australia Ltd         Sell         8 910           UBS Securities Australia Ltd         Sell         28,389           UBS Securities Australia Ltd         Sell         28,380           UBS Securities Australia Ltd         Sell         1,204           UBS Securities Australia Ltd         Sell         1,290           UBS Securities Australia Ltd         Sell         1,2,900           UBS Securities Australia Ltd         Sell         2,500           UBS Securities Australia Ltd         Sell         6,5524           UBS Securities Australia Ltd         Sell         6,5524           UBS Securities Australia Ltd         Sell         6,5524           UBS Securities Australia Ltd	alia Etd	Sell	2.580	(300)	Ordinary
UBS Securities Australia Ltd         Sell         731           UBS Securities Australia Ltd         Sell         913           UBS Securities Australia Ltd         Sell         913           UBS Securities Australia Ltd         Sell         913           UBS Securities Australia Ltd         Sell         12,04           UBS Securities Australia Ltd         Sell         28,380           UBS Securities Australia Ltd         Sell         17,940           UBS Securities Australia Ltd         Sell         12,900           UBS Securities Australia Ltd         Sell         12,900           UBS Securities Australia Ltd         Sell         2,500           UBS Securities Australia Ltd         Sell         2,500           UBS Securities Australia Ltd         Sell         2,500           UBS Securities Australia Ltd         Sell         3,524           UBS Securities Australia Ltd         Sell         8,524           UBS Securities Australia Ltd         Sell         8,5524           UBS Securities Australia Ltd         Sell         8,5524           UBS Securities Australia Ltd         Sell         6,510           UBS Securities Australia Ltd         Sell         6,510           UBS Securities Australia Ltd <td< td=""><td>aja i td</td><td>رواا</td><td>1 987</td><td>(150)</td><td>Ordinan</td></td<>	aja i td	رواا	1 987	(150)	Ordinan
UBS Securities Australia Ltd         541         292           UBS Securities Australia Ltd         541         9.133           UBS Securities Australia Ltd         541         8.910           UBS Securities Australia Ltd         541         28,380           UBS Securities Australia Ltd         541         7,340           UBS Securities Australia Ltd         541         7,340           UBS Securities Australia Ltd         541         1,290           UBS Securities Australia Ltd         541         1,290           UBS Securities Australia Ltd         541         2,300           UBS Securities Australia Ltd         541         2,300           UBS Securities Australia Ltd         541         2,300           UBS Securities Australia Ltd         541         3,500           UBS Securities Australia Ltd         541         3,500           UBS Securities Australia Ltd         541         3,500           UBS Securities Australia Ltd         541         3,600           UBS Securities Australia Ltd         541         3,400           UBS Securities Australia Ltd         541         3,400           UBS Securities Australia Ltd         541         3,510           UBS Securities Australia Ltd         541	afia I tol	Cell	721	(30)	ordinary.
UBS Securities Australia Ltd         Sell         9.24           UBS Securities Australia Ltd         Sell         9.133           UBS Securities Australia Ltd         Sell         1.204           UBS Securities Australia Ltd         Sell         7.8,380           UBS Securities Australia Ltd         Sell         7.340           UBS Securities Australia Ltd         Sell         17,340           UBS Securities Australia Ltd         Sell         12,500           UBS Securities Australia Ltd         Sell         47,205           UBS Securities Australia Ltd         Sell         4,535           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         4,535           UBS Securities Australia Ltd         Sell         3,524           UBS Securities Australia Ltd         Sell         3,524           UBS Securities Australia Ltd         Sell         3,5524           UBS Securities Australia Ltd         Sell         3,406           UBS Securities Australia Ltd	-1:-1:-1:-	100	167	(00)	Countries
UBS Securities Australia Ltd         Sell         9,133           UBS Securities Australia Ltd         Sell         8,910           UBS Securities Australia Ltd         Sell         1,204           UBS Securities Australia Ltd         Sell         17,940           UBS Securities Australia Ltd         Sell         17,940           UBS Securities Australia Ltd         Sell         12,900           UBS Securities Australia Ltd         Sell         12,900           UBS Securities Australia Ltd         Sell         2,325           UBS Securities Australia Ltd         Sell         2,320           UBS Securities Australia Ltd         Sell         2,322           UBS Securities Australia Ltd         Sell         3,552           UBS Securities Australia Ltd         Sell         3,552           UBS Securities Australia Ltd         Sell         3,552           UBS Securities Australia Ltd         Sell         3,406           UBS Securities Australia Ltd         Sell         5,523           UBS Securities Australia Ltd         Sell         6,552	alia Lta	yell	767	(34)	Ordinary
UBS Securities Australia Ltd         Sell         8-910           UBS Securities Australia Ltd         Sell         1,204           UBS Securities Australia Ltd         Sell         728,380           UBS Securities Australia Ltd         Sell         7,240           UBS Securities Australia Ltd         Sell         12,500           UBS Securities Australia Ltd         Sell         47,205           UBS Securities Australia Ltd         Sell         47,205           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         4,535           UBS Securities Australia Ltd         Sell         8,552           UBS Securities Australia Ltd         Sell         4,535           UBS Securities Australia Ltd         Sell         8,552	alia Ltd	Sell	9,133	(1,062)	Ordinary
UBS Securities Australia Ltd         Sell         1,204           UBS Securities Australia Ltd         Sell         1,204           UBS Securities Australia Ltd         Sell         17,940           UBS Securities Australia Ltd         Sell         12,900           UBS Securities Australia Ltd         Sell         12,900           UBS Securities Australia Ltd         Sell         27,205           UBS Securities Australia Ltd         Sell         27,205           UBS Securities Australia Ltd         Sell         2,322           UBS Securities Australia Ltd         Sell         4,655           UBS Securities Australia Ltd         Sell         3,406           UBS Securities Australia Ltd         Sell         3,406           UBS Securities Australia Ltd         Sell         6,524           UBS Securities Australia Ltd         Sell         3,406           UBS Securities Australia Ltd         Sell         6,510           UBS Securities Australia Ltd         Sell         6,510	alia Ltd	Sell	8,910	(1,036)	Ordinary
UBS Securities Australia Ltd         Sell         28,380           UBS Securities Australia Ltd         Sell         17,940           UBS Securities Australia Ltd         Sell         12,900           UBS Securities Australia Ltd         Sell         12,900           UBS Securities Australia Ltd         Sell         47,205           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         4,535           UBS Securities Australia Ltd         Sell         36,524           UBS Securities Australia Ltd         Sell         3,506           UBS Securities Australia Ltd         Sell         3,406           UBS Securities Australia Ltd         Sell         3,510	alia Ltd	Sell	1 204	(140)	Ordinan
UBS Securities Australia Ltd         Sell         17,940           UBS Securities Australia Ltd         Sell         17,940           UBS Securities Australia Ltd         Sell         12,900           UBS Securities Australia Ltd         Sell         12,900           UBS Securities Australia Ltd         Sell         2,320           UBS Securities Australia Ltd         Sell         2,320           UBS Securities Australia Ltd         Sell         4,635           UBS Securities Australia Ltd         Sell         8,635           UBS Securities Australia Ltd         Sell         3,606           UBS Securities Australia Ltd         Sell         8,635           UBS Securities Australia Ltd         Sell         8,652	List 144	CAL	OBE BC	(00 c)	Civilian
UBS Securities Australia Ltd         Sell         17,940           UBS Securities Australia Ltd         Sell         12,900           UBS Securities Australia Ltd         Sell         12,900           UBS Securities Australia Ltd         Sell         27,250           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         86,524           UBS Securities Australia Ltd         Sell         36,524           UBS Securities Australia Ltd         Sell         36,524           UBS Securities Australia Ltd         Sell         3,406	alid Ltu	, sen	200,02	(0.00,0)	Ordinary
UBS Securities Autoralia Ltd         Sell         421           UBS Securities Autoralia Ltd         Sell         12,900           UBS Securities Autoralia Ltd         Sell         12,900           UBS Securities Autoralia Ltd         Sell         47,205           UBS Securities Autoralia Ltd         Sell         2,320           UBS Securities Autoralia Ltd         Sell         4,635           UBS Securities Autoralia Ltd         Sell         3,652.4           UBS Securities Autoralia Ltd         Sell         3,652.4           UBS Securities Autoralia Ltd         Sell         3,652.4           UBS Securities Autoralia Ltd         Sell         6,510           UBS Securities Autoralia Ltd         Sell         6,510           UBS Securities Autoralia Ltd         Sell         6,510	alia Ltd	Sell	17,940	(2,086)	Ordinary
UBS Securities Australia Ltd         Sell         12,900           UBS Securities Australia Ltd         Sell         12,900           UBS Securities Australia Ltd         Sell         27,205           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         4,635           UBS Securities Australia Ltd         Sell         36,524           UBS Securities Australia Ltd         Sell         36,524           UBS Securities Australia Ltd         Sell         3,406           UBS Securities Australia Ltd         Sell         5,406           UBS Securities Australia Ltd         Sell         5,406           UBS Securities Australia Ltd         Sell         5,510	alia Ltd	Sell	421	(49)	Ordinary
UBS Securities Australia trd         Sell         12,900           UBS Securities Australia trd         5481         47,205           UBS Securities Australia trd         5481         2,530           UBS Securities Australia trd         5481         4,655           UBS Securities Australia trd         5481         4,655           UBS Securities Australia trd         5481         3,652,44           UBS Securities Australia trd         5481         3,406           UBS Securities Australia trd         5481         5,510           UBS Securities Australia trd         5481         5,510           UBS Securities Australia trd         5481         5,510	alia Ltd	Sell	12,900	(1,500)	Ordinary
UBS Securities Australia Ltd         Sell         47,205           UBS Securities Australia Ltd         Sell         2,580           UBS Securities Australia Ltd         Sell         2,322           UBS Securities Australia Ltd         Sell         4,635           UBS Securities Australia Ltd         Sell         36,524           UBS Securities Australia Ltd         Sell         3,405           UBS Securities Australia Ltd         Sell         5,405           UBS Securities Australia Ltd         Sell         5,510           UBS Securities Australia Ltd         Sell         5,510	bit 1td	Çe!	12 900	/1 500)	Ordinan
UBS Securities Australia Ltd         Sell         4.7.00           UBS Securities Australia Ltd         Sell         2.530           UBS Securities Australia Ltd         Sell         4.655           UBS Securities Australia Ltd         Sell         4.655           UBS Securities Australia Ltd         Sell         3.6524           UBS Securities Australia Ltd         Sell         3.406           UBS Securities Australia Ltd         Sell         6.510           UBS Securities Australia Ltd         Sell         6.510	P+1 v2v	n-a	302 27	(E 490)	Civilian C
UBS Securities Australia trd         2481         2,580           UBS Securities Australia trd         5481         2,332           UBS Securities Australia trd         5481         3,6524           UBS Securities Australia trd         5481         3,406           UBS Securities Australia trd         5481         5,406           UBS Securities Australia trd         5481         6,510           UBS Securities Australia trd         5481         6,510	-II- 1+A	Jet. 1	003 C	(000)	Cidinary
UBS Securities Australia Ltd         Sell         2.322           UBS Securities Australia Ltd         Sell         4.635           UBS Securities Australia Ltd         Sell         36,524           UBS Securities Australia Ltd         Sell         3.406           UBS Securities Australia Ltd         Sell         6,510           UBS Securities Australia Ltd         Sell         7.31	alla Ltd	Jec	7,080	(aux)	Ordinary
UBS Securities Australia trd         Sell         4.635           UBS Securities Australia trd         Sell         36.524           UBS Securities Australia trd         Sell         3.406           UBS Securities Australia trd         Sell         6.510           UBS Securities Australia trd         Sell         731	alia Ltd	Sell	2,322	(0,2)	Ordinary
UBS Securities Australia Ltd         Sell         36.524           UBS Securities Australia Ltd         Sell         3.406           UBS Securities Australia Ltd         Sell         5.10           UBS Securities Australia Ltd         Sell         73.1	elia Ltd	Sell	4,635	(623)	Ordinary
UBS Securities Australia Ltd         Sell         3,406           UBS Securities Australia Ltd         Sell         6,510           UBS Securities Australia Ltd         Sell         731	alia Ltd	Sell	36,524	(4,247)	Ordinary
UBS Securities Australia Ltd         Sell         5-100           UBS Securities Australia Ltd         Sell         5-10           UBS Securities Australia Ltd         Sell         731	lin 14d	1000	3 406	(302)	, arrier C
UBS Securities Australia Ltd 5,510 UBS Securities Australia Ltd 5ell 731	alid Ltu	Sell 2	204,0	(vect)	Ordinary
UBS Securities Australia Ltd Sell 731	alia Ltd	Sell	V16,0	i/c/)	Orginary
	alia Ltd	Sell	731	(82)	Ordinary
UBS Securities Australia Ltd Sell Sell 5,212	alia Ltd	Sell	5,212	(909)	Ordinary
I IRS Contrition Australia 174 Call 22 950	10 14 1	117	030 66	1025 C/	, man 1
UBS Securities Australia Ltd Sell 22,859	alia Ltd	Sell	22,859	(2,658)	Ordinary
16-Oct-14 UBS Securities Austra		methormen Australia Ltd methormen Australia Ltd methormen Australia Ltd methormen Australia Ltd methormen Australia Ltd methormen Australia Ltd methormen Australia Ltd methormen Australia Ltd methormen Australia Ltd milla Ltd	Straile Ltd Straile Ltd Straile Ltd Straile Ltd Straile Ltd Straile Ltd Straile Ltd	1979   1979	100   100

22,380 25,804 35,804 31,335 4,656	(2,669) (4,270) (4,270) (553) (553) (63) (490) (159) (159)
35,804 31,335 4,656 528	(4,270) (3,737) (555) (63) (490) (159) (159)
31,335	(3,737) (555) (63) (490) (159) (159)
4,656	(555) (63) (490) (159) (347)
528	(63) (490) (159) (347)
	(490) (159) (347)
4,109	(159)
1,335	(347)
2,915	
1,537	(183)
1,344	(160)
4,435	(528)
3,108	(370)
3,204	(381)
7,092	(2,036)
160	(6)
1,151	(137)
3,259	(388)
1,815	(216)
1,051	(125)
[65]	8
2,616	(311)
73,766	(2,831)
20	(9)
4 401	(5.35)
7077	(5.27)
7447	(523)
	(500)
101,1	(197)
4,469	(237)
4,460	(531)
8,400	(1,000)
1,873	(223)
4,469	(532)
2,621	(312)
1 949	(030)
7.769	(533)
2 5 2 0	(300)
3 533	(000)
1,613	(100)
2530	(000)
020,2	(000)
(60)0	(707)
3,3/	(402)
3,299	(393)
21,928	(2,612)
4,494	(535)
3,889	(463)
3,780	(450)
101	(12)
1266	(1.187)
3010	(350)
2000	(600)
20,354	(2,493)
6/6/07	(2,496)
833	(66)
20	(9)
1,447	(172)
3,900	(464)
5.040	(009)
404 292	(48 130)
8 837	(CSO 1)
1 308	(404)
607'1	(101)
8,400	(nnn' i )
8,400	(1,000)
4,323	(514)
7,560	(006)
26	(11)
1 310	(155)
	200
, CO2	(2)
200	(00)
1 043	(150)
2 5 2 2	(2000)
526,2	(000)
169,7	(320)
	1.151 1.151 1.1815 1.1815 1.1815 1.1815 1.1817 1.1811 1.1817 1.1817 1.1817 1.1817 1.1818 1.1817 1.1818 1.1817 1.1818 1.1817 1.1818 1.1817 1.1818 1.1817 1.1818 1.1817 1.1818 1.1817 1.1818 1.18

-OCI-14		2		107	,
'-Oct-14	UBS Wealth Management Australia Ltd	Buy	748	98	Ordina
7-0d-14	UBS Wealth Management Australia Ltd	Buy	1,983	228	Ordina
1-14	UBS Wealth Management Australia Ltd	Buy	1,331	153	Ordina
7-OCI-14	UBS wealth Management Australia Ltd	Buy	861	66	Ordina
1 100	UBS Wealth Management Australia Ltd	Buy	618	12	Ordinary
704.14	LiBC Wealth Management Australia 11d	buy D	950	79 000	Signa
7-Drt-14	URS Wealth Management Australia 11d	Busy	100,000	120	Orgina
-Oct-14	UBS Securities Australia Ltd	Jag.	605 500	(000 02)	
7-Oct-14	UBS Securities Australia Ltd	Sell	3,404	(391)	Ordinary
7-Oct-14	UBS Securities Australia Ltd	Sell	9,306	(1,069)	Ordinary
ct-14	UBS Securities Australia Ltd	Sell	908'6	(1,069)	Ordinary
t-14	UBS Securities Australia Ltd	Buy	50,118	5,774	Ordinary
4-14	UBS Securities Australia Ltd	Buy	123,482	14,226	Ordinary
171	UBS Securities Australia Ltd	Buy	1,708	197	Ordina
114	URS Securities Australia 11d	Bring	2 566	064	Ordinary
d-14	UBS Securities Australia Ltd	Boy	3 789	437	Ordinary
7-Oct-14	UBS Securities Australia Ltd	Buy	1,379	159	Ordina
ct-14	UBS Securities Australia Ltd	Buy	2,575	297	Ordina
7-Oct-14	UBS Securities Australia Ltd	Buy	53,355	6,154	Ordinary
7.04.14	USS Securities Australia Ltd	Buy	9,138	1,054	Ordina
7-04-14	UBS Securities Australia Ltd	Buy	1,665	192	Ordinary
2.04-14	UBS Securities Australia Ltd	buy Buy	3,147	363	Ordina
7-Oct-14	UBS Securities Australia Ltd	Buv	2,5,5	020	S C
-Oct-14	UBS Securities Australia Ltd	Buv	5,020	579	O C
7-Oct-14	UBS Securities Australia Ltd	Buy	1.899	219	Ordina
7-Oct-14	UBS Securities Australia Ltd	Buy	2,601	300	Ordinar
7-Oct-14	UBS Securities Australia Ltd	Buy	4,335	200	Ordinar
-Oct-14	UBS Securities Australia Ltd	Buy	139	16	Ordina
'-Oct-14	UBS Securities Australia Ltd	Buy	147	17	Ordinary
Ct-14	UBS Securities Australia Ltd	Buy	460	23	Ordinar
00-14	UBS Securities Australia Ltd	Buy	3,147	363	Ordinar
7.04.14	UBS Securities Australia Ltd	Buy	3,728	430	Ordinary
177	USS Securities Australia Ltd	Buy	338	39	Ordinary
41-14	URS Securities Australia Ltd	Buy	9	- 600	Ordinary
q-14	UBS Securities Australia Ltd	Buy	3 147	575	Ordinary
d-14	UBS Securities Australia Ltd	Buy	3 147	363	Ordinary
-Oct-14	UBS Securities Australia Ltd	8uv	61	2	Ordinan
ct-14	UBS Securities Australia Ltd	Buy	9.025	1.041	Ordinary
7-Oct-14	UBS Securities Australia Ltd	Buy	2,168	250	Ordinar
d-14	UBS Securities Australia Ltd	8uy	3,147	363	Ordinar
7-Oct-14	UBS Securities Australia Ltd	Buy	1,179	136	Ordinary
Q-14	UBS Securities Australia Ltd	Buy	4,153	479	Ordinary
4 1	UBS Securities Australia Ltd	Buy	3,147	363	Ordinar
7-00-14 7-00-14	UBS Securities Australia Ltd	Buy	4,084	471	Ordinary
7-00-14	UBS Securities Australia Ltd	Buy	27,163	3,133	Ordinary
7-00-14	UBS Securities Australia Ltd	Sell	100,946	(11,670)	Ordinary
7-04-14	UBS Securities Australia Itd	Sell Sell	3,750	(430)	Ordinary
41-14	UBS Securities Australia Ltd	100	015.7	(005)	o color
7-0ct-14	UBS Securities Australia Ltd	, es	2 241	(500)	Ordina
ct-14	UBS Securities Australia Ltd	les	2 180	(050)	Ordinar
d-14	UBS Securities Australia Ltd	Sell	39 777	(4 561)	Ordinar
ct-14	UBS Securities Australia Ltd	Sell	44,812	(5.139)	Ordinar
-Oct-14	UBS Securities Australia Ltd	Sell	1,029	(118)	Ordinary
-Oct-14	UBS Securities Australia Ltd	Sell	9,182	(1,053)	Ordinar
-Oct-14	UBS Securities Australia Ltd	Sell	28,750	(3,297)	Ordinary
7-Oct-14	UBS Securities Australia Ltd	Sell	26,160	(3,000)	Ordinary
7-0ct-14	UBS Securities Australia Ltd	Sell	26,160	(3,000)	Ordinary
G-14	UBS Securities Australia Ltd	Sell	26,160	(3,000)	Ordinary
14-14	UBS Securities Australia Ltd	Sell	3,750	(430)	Ordinary
414	UBS Securities Australia Ltd	Zell Sell	2,180	(250)	Ordinary
41-14	UBS Securities Australia Ltd	ll Sell	46,871	(5,372)	Ordinary
04.14	UBS Securities Australia Ltd	Sell	23,221	(699,2)	Ordinary
1-14	UBS Securities Australia Ltd	las.	24,006	(5,500)	Ordinary
17-0ct-14	UBS Securities Australia IId	Sell	220	(CE)	Colina
ا	The state of the s				
V . 1	11:00 Commission America   40	3	2536	(000)	Ordinary

24 14	1 IDE Cognition Australia I to	ll co	000	1117777)	The second secon
6-0d-14	UBS Securities Australia Ltd	[e] Sell	18,4300	(500)	Ordinary
16-Oct-14	UBS Securities Australia Ltd	¥ 34	1729	(201)	Ordinary
5-Oct-14	UBS Securities Australia Ltd	lleS.	2.150	0220	Ordinary
16-0ct-14	UBS Securities Australia Ltd	Sell	43,000	(5.000)	Ordinary
6-0ct-14	UBS Securities Australia Ltd	Sel	464	(54)	Ordinary
6-0-14	(IRS Securities Australia Ltd	Sall	4 352	(505)	Ordinary
6-Oct-14	(IBS Securities Australia Itd	Sell	7576	(1007)	Ordinary
6-Oct-14	UBS Securities Australia Ltd	Sell	3.440	(400)	Ordinary
6-Oct-14	UBS Securities Australia Ltd	Sell	808	(96)	Ordinary
6-Oct-14	UBS Securities Australia Ltd	Sell	28,380	(3,300)	Ordinary
6-Oct-14	UBS Securities Australia Ltd	Sell	110,725	(12,875)	Ordinary
16-Oct-14	UBS Securities Australia Ltd	Sell	1.187	(138)	Ordinary
16-Oct-14	UBS Securities Australia Ltd	Sell	46.543	(5.412)	Ordinary
16-Oct-14	UBS Securities Australia Ltd	Sell	2.150	(250)	Ordinary
-Oct-14	UBS Securities Australia Ltd	Sell	4.300	(200)	Ordinary
16-0-14	I IRC Cocurities Australia 114	Coll	10,217	(1 188)	Ordinan
16-Oct-14	UBS Securities Australia 1td	llas	2 580	(00E)	Ordinary
6-Ort-14	1185 Securities Australia 11d	las	13 545	(1 575)	Ordinary
F.04-14	1100 Cocurition Australia 144	Coll	0.440	(000)	Ordinary
0.00114	Tipe Countries August in 164	Coll	371	(100)	Ordinary
6.04.17	HBS Securities Australia 14d	Soli	7 046	(7/20)	Ordinary
	Let Consider Assessing the	501	OFF OR	(176)	Company
1 1 1 1	Data Securities Australia Ltd	Coll	0/#/08	(1003)	Colonialy
6-0ct-14	Upo Securities Australia Liu	Sell	4,500	(00c)	Ordinary
1000	Day Securities Adad and Lind	i cell	151,00	(76.70)	Countary
6-Oct-14	UBS Securities Australia Ltd	Sell	1,324	(300)	Ordinary
1	opo secultura Australia Liu	361	00000	(066)	Ordinary
-Oct-14	obs securities Australia Ltd	3411	196'907	(31,027)	Ordinary
-001-14	UBS Securities Australia Ltd	Self.	N	(76)	Ordinary
1 130	Opp Securities Australia Liu	361	676	(10)	Cidinary
6-00-14 6-04-14	Obsideration Australia Ltd	Sell	439	(3.1.0)	Ordinary
	Obsideration Australia Ltd	261	218,00	(10,10)	Orginary
1	UDS Securities Australia Lito	inal 3	12,413	0,440	Ordinary
	OBS Securities Australia Ltd	) ell	950,01	(7/1/7)	Orginary
	UBS Securities Australia Ltd	Sell	52,823	(b,542)	Ordinary
1	UBS Securities Australia Ltd	Ang.	1/8,138	855,02	Ordinary
-Oct-14	UBS Securities Australia Ltd	Sell	48,465	(2,590)	Ordinary
-Oct-14	UBS Securities Australia Ltd	Suy	8,680	1,000	Ordinary
-Oct-14	UBS Securities Australia Ltd	Sell	6,644	(769)	Ordinary
Oct-14	UBS Securities Australia Ltd	Sell	82,000	(10,000)	Ordinary
Oct-14	UBS Securities Australia Ltd	Buy	347,200	40,000	Ordinary
7-Oct-14	UBS Securities Australia Ltd	Sell	318,941	(36,752)	Ordinary
7-Oct-14	UBS Securities Australia Ltd	Sell	313,200	(36,078)	Ordinary
7-Oct-14	UBS Securities Australia Ltd	Buy	1,035	119	Ordinary
7-Oct-14	UBS Securities Australia Ltd	Sell	33,843	(3,899)	Ordinary
7-Oct-14	UBS Securities Australia Ltd	Sell	1,491,581	(172,079)	Ordinary
7-Oct-14	UBS AG Australía Branch	Buy	26,429	3,044	Ordinary
7-Oct-14	UBS AG Australia Branch	Buy	38,107	4,389	Ordinary
7-0ct-14	UBS AG Australia Branch	Seli	1,108,711	(127,921)	Ordinary
7-0ct-14	UBS Wealth Management Australia Ltd	Buy	1,365	157	Ordinary
7-Oct-14	UBS Wealth Management Australia   td	Bery	1 244	143	Ordinary
7-04-14	IRC Woolth Management Australia 14	B: W	171.01	305 4	Cedimon
1000	DOS WESTER MANISON OF THE LIGHT	ouy D	12,141	2500	Countries
7.00-14	UBS Wearth Management Australia Ltd	any.	2,400	9/7	Ordinary
- CG-14	UBS Wealth Management Australia Ltd	Buy	/04	18	Ordinary
-Oct-14	UBS Wealth Management Australia Ltd	Buy	531	19	Ordinary
7-0ct-14	UBS Wealth Management Australia Ltd	Buy	11,124	1,279	Ordinary
7-Oct-14	UBS Wealth Management Australia Ltd	Buy	1,113	128	Ordinary
-Oct-14	UBS Wealth Management Australia Ltd	Buy	12,133	1,395	Ordinary
'-Oct-14	UBS Wealth Management Australia Ltd	Buy	1,044	120	Ordinary
'-Oct-14	UBS Wealth Management Australia Ltd	Buy	1,592	183	Ordinary
7-Oct-14	UBS Wealth Management Australia Ltd	Buv	808	693	Ordinary
7-Ort-14	URS Wealth Management Australia Ltd	Rin	5 453	109	Ordinary
7-0ct-14	IIRs Wealth Management Australia Itd	Br.v	708.9	785	Ordinary
1 1 2	COC Months Management Australia Ltd	600	0,000	200	The state of the s
1 250	INS Wealth Management Australia 14d	Serv.	10,202	1 403	Ordinary
24.13	1955 Weddill Washingsenreit Australia Ltd	la	1 006	361	Ordinary
- CC-14	DDS Wearth Management Australia Ltd	and a	950'1	971	Oroinary
Oct-14	UBS Wealth Management Australia LTG	gnà	519	Si	Ordinary
Oct-14	UBS Weaith Management Australia Ltd	Buy	1,009	116	Ordinary
Oct-14	UBS Wealth Management Australia Ltd	Buy	10,993	1,264	Ordinary
Oct-14	UBS Wealth Management Australia Ltd	Buy	1,722	198	Ordinary
7-Oct-14	UBS Wealth Management Australia Ltd	Buy	5,184	965	Ordinary
-Ort-14	1185 Wealth Management Australia 11d	RIN	2 261	090	Ordinary
	OCC SALES A SALES AND ADDRESS OF THE PARTY O	6.5	4 200	5,	(in the control of
CG14	לחפס על אישנינו ועושנושלושנושניו שחסנושוש דנח	gný	8/7'	/4/	
					Olumary

UBS Securities Australia Ltd	Sell	4,959	(572)	Ordinary
UBS Securities Australia Ltd	Sell	165	(61)	Ordinary
alia Ltd	Ne Ne	3,047	(1351)	Ordinary
alia Ltd	Self	321	(37)	Ordinary
ralia Ltd	Sell	4,982	(574)	Ordinary
alia Lid	Sell	4,9/4	(573)	Ordinary
alia Ltd	195	7.821	(ODE)	Ordinary
ralia Ltd	Sell	165	(61)	Ordinary
UBS Securities Australia Ltd	Sell	21,591	(2,486)	Ordinary
ralia Ltd	Sell	1,868	(215)	Ordinary
ralia Ltd	Jes S	2,485	(286)	Ordinary
UBS Securities Australia Ind	- X	1,529	(1/6)	Ordinary
ralia Itd	3	1 738	(000)	Colinary
ralia Ltd	Sell	16.850	(1 939)	Ordinary
ralia Ltd	Sell	4,284	(493)	Ordinary
alia Ltd	Sell	17	Ø	Ordinary
alia Ltd	Sell	4,455	(513)	Ordinary
alia Ltd	Sell	4,475	(515)	Ordinary
alia Ltd	Sell	4,467	(514)	Ordinary
alia Ltd	Sell	4,988	(574)	Ordinary
UBS Securities Australia Ltd	Sell	2,607	(300)	Ordinary
one tito	Hac I	1,043	(021)	Ordinary
IRS Securities Australia I+d	Sell.	300	(991)	Ordinary
UBS Securities Australia Itcl	Hay	292	(/7)	Ordinary
UBS Securities Australia 11d	200	1 443	(100)	Ordinary
UBS Securities Australia Ltd	les	766	(88)	Ordinary
lia Ltd	Sell	3.715	(427)	Ordinary
lia Ltd	Sell	4,976	(572)	Ordinary
ilia Ltd	Sell	435	(20)	Ordinary
lia Ltd	Sell	2,175	(250)	Ordinary
ia Ltd	Sell	2,175	(220)	Ordinary
lia Ltd	Sell	284	(67)	Ordinary
illa Ltd	Nell Sell	4,407	(205)	Ordinary
Mile Ltd	Jan Jan Jan Jan Jan Jan Jan Jan Jan Jan	3,649	(419)	Ordinary
alia ital	iiac	516,1	(151)	Ordinary
alia LtG	Hac	4,155 84E	(7/4)	Ordinary
UBS Securities Australia Ltd	Sell	3 484	(400)	Ordinary
alia Ltd	Sell	2,178	(250)	Ordinary
UBS Securities Australia Ltd	Sell	10,408	(1,195)	Ordinary
alia Ltd	Sell	2,613	(300)	Ordinary
UBS Securities Australia Ltd	Sell	3,484	(400)	Ordinary
UBS Securities Australia Ltd	Sell	801	(26)	Ordinary
UBS Securities Australia Ltd	Sell	2,944	(338)	Ordinary
UBS Securities Australia Ltd	Sell	202	(58)	Ordinary
JBS Securities Australia Ltd	Sell	4,976	(571)	Ordinary
JBS Securities Australia Ltd	Sell	218	(22)	Ordinary
JBS Securities Australia Ltd	Sell	1,927	(221)	Ordinary
UBS Securities Australia Ltd	Sell	1,727	(198)	Ordinary
DBS Securities Australia Ltd	Set.	1,107	(127)	Ordinary
UBS Securities Australia Ltd	Sell	4,959	(695)	Ordinary
JBS Securities Australia Ltd	Sell	6,767	(776)	Ordinary
alia Liū	Nell Sell	180,1	(124)	Ordinary
DS Securities Australia Ltd	Dell Coll	2,616	(300)	Ordinary
IBS Societies Australia 14d	100	1/0	000	Ordinary
DS Securities Australia Ind	llac	2005	(000)	Ordinary
alia ttu	100	2,006	(306)	Ordinary
UBS Securities Australia Itd	Sell Sell	3 750	(061)	Ordinary
UBS Securities Australia Ltd	Sell	2.616	(300)	Ordinary
alia Ltd	Sell	872	(100)	Ordinary
JBS Securities Australia Ltd	Sell	610	(02)	Ordinary
JBS Securities Australia Ltd	Sell	4,746	(544)	Ordinary
UBS Securities Australia Ltd	Sell	4,136	(474)	Ordinary
JBS Securities Australia Ltd	Sell	4,767	(547)	Ordinary
UBS Securities Australia Ltd	96	75. 4	(502)	Ordinary
			(100)	Cicinal

7-00-14	UBS Securities Australia Ltd	Sell	3,750		Ordinary
17-04-14	UBS Securities Australia Ltd	Set	1 884	(3,900)	Ordinary
G-14	UBS Securities Australia Ltd	les	314		Ordinary
ct-14	UBS Securities Australia Ltd	Sell	2.183		Ordinary
7-0ct-14	UBS Securities Australia Ltd	lay.	2.619		Ordinary
14	UBS Securities Australia Itd	lay	44		Ordinary
17-Oct-14	UBS Securities Australia Ltd	38	1,694	(194)	Ordinary
rt-14	UBS Securities Australia Ltd	Sell	236		Ordinary
1-14	UBS Securities Australia Ltd	Sell	5 648	(647)	Ordinary
144	IIBS Securities Australia   14	lia)	27.79		Ordinan
1.14	IIIR Securities Australia I to	3	500 00	(10,525)	Ordinary
1.14	ILIBS Securities Australia   14	7	2,619		Ordinan
	IIRS Contridice Australia I to	Coll	67 483	(000)	Ordings
1.14	DIDS Global Areat Management	100	075		Cochian
	LIDE Countries Australia 144	les l	240.0		Closial
117	UBS Securities Australia Ltd	liac	302 200	(50,00)	Ordinary
	TIDE Countries Australia Ltd	nec o	627,122		Ordinary
	Obs Securities Australia Ltd	liac d	500,711		Ordinary
1	Opposed and all Liu	Buy	16,52	050,5	Ordinary
41-14	UBS Securities Australia Ltd	Sell	115,201	(13,320)	Ordinary
* 1	U.D.C. Contribut Australia ciu	) and	102,261		Ordinary
	Obs Securities Abstraira Ltd	liac	58/1		Orginary
4 1	Obs securities Adstraina Ltd	suy 	7//77		Ordinary
1	ODS Securities Australia Liu	Sell	05,550	(10,302)	Ordinary
	Opp Securities Australia Liu	100	506,662		Ordinary
20-04-14	UBS Securities Australia Ltd	seil	4,598		Ordinary
	CD3 Securities Australia Liu	ing.	3,470		Ordinary
t-14		Sell	338,074	(38,972)	Ordinary
4-14	UBS Securities Australia LTG	Sell	360,140		Ordinary
0-0ct-14	UBS Securities Australia Ltd	BUN	287,707		Ordinary
Oct-14	UBS Securities Australia Ltd	Sell	3,532		Ordinary
t-)4	UBS Securities Australia Ltd	Sell	658'09		Ordinary
20-Oct-14	UBS Securities Australia Ltd	Stock borrowed	N/A		Ordinary
1-14	UBS AG Australia Branch	Buy	346,170	40,000	Ordinary
0-Oct-14	UBS AG Australia Branch	Buy	86,543	000'01	Ordinary
0-Oct-14	UBS AG Australia Branch	Buy	1,126,728	130,000	Ordinary
0-Oct-14	UBS AG Australia Branch	Seli	78,209		Ordinary
)-Oct-14	UBS AG Australia Branch	Sell	19,984		Ordinary
0-Oct-14	UBS Wealth Management Australia Ltd	Sell	32,963		Ordinary
0-Oct-14	UBS Securities Australia Ltd	Sell	346.000		Ordinary
14	UBS Securities Australia Ltd	Sell	3 438		Ordinary
14	UBS Securities Australia Ltd	lles	966		Ordinary
-14	I IRS Securities Australia 1td	100	20 654	(585 6)	Ordinary
	TIDS Countries Assessed and	11-0	FC0,03	(500/3)	or all all
:	ODS Securities Australia Liu	261	708,1	(6) 7)	Ordinary
<u>+</u>	Obs securities Australia Ltd	Sell	1,022	(811)	Ordinary
4	UBS Securities Australia Ltd	Sell	4,339	(501)	Ordinary
14	UBS Securities Australia Ltd	Sell	4,339	(501)	Ordinary
20-Oct-14	UBS Securities Australia Ltd	Sell	346	(40)	Ordinary
-14	UBS Securities Australia Ltd	Self	1,604	(581)	Ordinary
-14	UBS Securities Australia Ltd	Sell	2,714	(813)	Ordinary
-14	UBS Securities Australia I td	idy	3 778	(UEV)	Ordinan
	I IRC Securities Australia 14d	Coll	27,75	(000)	Ordinary
	Obs securines Australia Ltd	Sell Sell	1,601	(300)	Crainary
-	OBS Securities Australia Ltd	Sell	3,199	(392)	Ordinary
14	UBS Securities Australia Ltd	Sell	6,945	(801)	Ordinary
-14	UBS Securities Australia Ltd	Sell	4,318	(498)	Ordinary
714	UBS Securities Australia Ltd	Seli	1,153	(133)	Ordinary
14	UBS Securities Australia Ltd	las	4 318	(867)	Ordinary
-14	URS Securities Australia Itd	Sel.	9057	(607)	Ordinan
	COO Decolines Australia Ltd	nac .	075'4	(664)	Ordinary
4	UBS Securities Australia Ltd	Sell	4,324	(499)	Ordinary
Oct-14	UBS Securities Australia Ltd	Sell	4,352	(202)	Ordinary
-Oct-14	UBS Securities Australia Ltd	Sell	8,618	(966)	Ordinary
-Oct-14	UBS Securities Australia Ltd	Sell	2,895	(PEE)	Ordinary
-0-t-14	I ISS Securities Australia 11d	9	1 440	(291)	Ordinary
1	Con Securities Australia Louis	i ei	0++,	(101)	Cidinary
<u> </u>	UBS Securities Australia Ltd	Seri	555,7	(300)	Ordinary
-14	IUBS Securities Australia Ltd	Sell	1,751	(202)	Ordinary
20-Oct-14	UBS Securities Australia Ltd	Sell	4,500	(519)	Ordinary
-14	UBS Securities Australia Ltd	Sell	6	(E)	Ordinary
-14	UBS Securities Australia Itd	les.	1947	(ACC)	Ordinary
2	IIRS Securities Australia 114	Cell	3000 8	(272)	Craining
	Opp Securities Australia Liu	341	oon's	(340)	Ordinary
4	UBS Securities Australia Lto	yell	1,601	(300)	Ordinary
4	UBS Securities Australia Ltd	Seil	2,601	(300)	Ordinary
-14	UBS Securities Australia Ltd	Sell	2,168	(250)	Ordinary
20-Oct-14	UBS Securities Australia Ltd	Sell	5.679	(655)	Ordinary
1.4	I IBS Securities Australia I to	100	4 077	(WZ3)	Codings
4	UBS Securities Australia Ltd	Sell	1/5'4	(h/¢)	Ordinary

OCCURT         1885 SCHORTERA ALMERINE LINE         BAY         1,100		Catholic Commission Assessed Commission	(na			
Bit Securine Auctivide Ltd   Buy   1,555   1	14	LIBS Securities Australia Ltd	Buy	26	3	Ordinary
Bits Securine Australia Ltd   Bay   1,000   2,000   2,000   1,000   2,000	14-14	RS Securities Australia ind	bug.	13 655	17	Ordinary
Dist Scientine Auctivide LED   Bigs   Comparison   Comp	kt-14	UBS Securities Australia Itd	Bire	3 150	1,349	Ordinary
Bits Securine Australia Ltd   Bay   145	14	UBS Securities Australia 11d	But	7,180	057	Ordinary
Bits Securine Authrisk LTC   Bits	11-14	I IRS Securities Australia 14d	Pour Branch	24.5	75	Ordinary
1615 Securities Australia Int	20-14	UBS Securities Australia Ltd	Bury	715,1	8 8	Ordinary
bits         Good Societies Australia Incl.         Bay         2552         25           Lobs Societies Australia Incl.         Bay         2,304         25,30           Lobs Societies Australia Incl.         Bay         2,304         25,00           Lobs Societies Australia Incl.         Bay         2,304         25,00           Lobs Societies Australia Incl.         Bay         2,304         25,00           Lobs Societies Australia Incl.         Bay         2,106         25,00           Lobs Societies Australia Incl.         Bay         2,106         25,00           Lobs Societies Australia Incl.         Bay         2,106         25,00           Lobs Societies Australia Incl.         Bay         1,920         25,00           Lobs Societies Australia Incl.         Bay         2,107         1,116           Lobs Societies Australia Incl.         Bay         2,127         1,16	7t-14	UBS Securities Australia Ltd	S S	900	16	Ordinary
BOS SCURING AMERINE INTERNAL INTER	2d-14	UBS Securities Australia Ltd	Biv	892	20 02	Ordinary
Loss Securines Australia Led         Boy         8.506         8.506           Loss Securines Australia Led         Boy         3.02         3.02           Loss Securines Australia Led         Boy         2.02         3.02           Loss Securines Australia Led         Boy         2.05         3.03           Loss Securines Australia Led         Boy         2.05         2.05           Loss Securines Australia Led         Boy         2.05         2.05           Loss Securines Australia Led         Boy         2.02         2.05           Los Securines Australia Led         Boy         2.02         2.05           Los Securines Australia Led         Boy         2.02         2.05           Los Securines Australia Led         Boy         2.02 <td>ht-14</td> <td>UBS Securities Australia Ltd</td> <td>Sur.</td> <td>2 912</td> <td>7337</td> <td>Ordinary</td>	ht-14	UBS Securities Australia Ltd	Sur.	2 912	7337	Ordinary
USS Securities Automita Intel   Bay   1,394   1,595   1,595   1,995	1d-14	UBS Securities Australia Ltd	800	8 250	950	Ordinary
USS Scorning Australia Ltd   Bay   2,126   250     USS Scorning Australia Ltd   Bay   2,125   250     USS Scorning Australia Ltd   Bay   2,125   250     USS Scorning Australia Ltd   Bay   2,125   2,125     USS Scorning Australia Ltd   Bay   2,125   2,126     USS Scorning Australia Ltd   Bay   2,126   2,126     USS Scorning Australia Ltd   Self   Self   3,126     USS Scorning Australia Ltd   Self   Self   3,126     USS Scorning Australia Ltd   Self   Self   3,126     USS Scorning Aust	t-14	UBS Securities Australia Ltd	Buv	3.048	35.7	Ordinary
UBS Scornines Autorials Ltd         Boy         2.598         300           UBS Scornines Autorials Ltd         Boy         2.509         3.05           LBS Scornines Autorials Ltd         Boy         2.099         2.09           LBS Scornines Autorials Ltd         Boy         2.099         2.05           LBS Scornines Autorials Ltd         Boy         2.095         2.05           LBS Scornines Autorials Ltd         Boy         2.097         2.05           LBS Scornines Autorials Ltd         Boy         2.097         2.05           LBS Scornines Autorials Ltd         Boy         2.097         2.05           LBS Scornines Autorials Ltd         Boy         2.035         2.05           LBS Scornines Autorials Ltd         Boy         2.037 <td>)d-14</td> <td>UBS Securities Australia Ltd</td> <td>Buv</td> <td>3.126</td> <td>361</td> <td>Ordinary</td>	)d-14	UBS Securities Australia Ltd	Buv	3.126	361	Ordinary
UBS Securines Autoritida LOG         Biny         3.35         500           UBS Securines Autoritida LOG         Boy         3.13         500           UBS Securines Autoritida LOG         Boy         1.08         1.38           UBS Securines Autoritida LOG         Boy         2.083         3.13           UBS Securines Autoritida LOG         Boy         2.083         3.13           UBS Securines Autoritida LOG         Boy         2.033         3.00           UBS Securities Autoritida LOG         Boy	ct-14	UBS Securities Australia Ltd	Buy	2,598	300	Ordinary
USE Scormines Autoralia Ltd         Bibs         13.15         5.86           USE Scormines Autoralia Ltd         Buy         1.36         1.36           USE Scormines Autoralia Ltd         Buy         2.16         1.27           USE Scormines Autoralia Ltd         Buy         2.60         2.70           USE Scormines Autoralia Ltd         Buy         2.60         2.70           USE Scormines Autoralia Ltd         Buy         2.60         2.70           USE Scormines Autoralia Ltd         Buy         2.70         2.70           USE Scormines Autoralia Ltd         Buy         2.70 <t< td=""><td>tt-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>4,330</td><td>200</td><td>Ordinary</td></t<>	tt-14	UBS Securities Australia Ltd	Buy	4,330	200	Ordinary
UNES Scorrintes Autoriale Ltd         Buy         12,089         13,986           UNES Scorrintes Autoriale Ltd         Buy         2,105         25,03           UNES Scorrintes Autoriale Ltd         Buy         2,683         2,21           UNES Scorrintes Autoriale Ltd         Buy         2,683         2,50           UNES Scorrintes Autoriale Ltd         Buy         2,683         2,50           UNES Scorrintes Autoriale Ltd         Buy         2,683         2,50           UNES Scorrintes Autoriale Ltd         Buy         2,29         2,50           UNES Scorrintes Autoriale Ltd         Buy         2,23         2,50           UNES Scorrintes Autoriale Ltd         Buy         2,33         2,68           UNES Scorrintes Autoriale Ltd         Buy         2,59         2,578           UNES Scorrintes Autoriale Ltd         Buy         2,59         2,578           UNES Scorrintes Autoriale Ltd         Buy         2,58         2,58           UNES Scorrintes Autoriale Ltd	tt-14	UBS Securities Australia Ltd	Buy	3,135	362	Ordinary
UBS Securities Autorials Ltd         Bay         2.152         3.9           UBS Securities Autorials Ltd         Bay         2.162         2.3           UBS Securities Autorials Ltd         Bay         2.6         2.2           UBS Securities Autorials Ltd         Bay         2.6         2.0           UBS Securities Autorials Ltd         Bay         2.1373         2.768           UBS Securities Autorials Ltd         Bay         2.1373         2.738           UBS Securities Autorials Ltd         Bay         1.0         2.1373         2.0           UBS Securities Autorials Ltd         Bay         1.0         2.1373         2.0           UBS Securities Autorials Ltd         Bay         1.0         2.1         2.0           UBS Securities Autorials Ltd         Bay         1.0         2.0         2.0           UBS Securities Autorials Ltd         <	ct-14	UBS Securities Australia Ltd	Buy	12,089	1,396	Ordinary
UBS Securities Australia Ltd         Buy         2165         250           UBS Securities Australia Ltd         Buy         2909         311           UBS Securities Australia Ltd         Buy         23971         2289           UBS Securities Australia Ltd         Buy         23971         2289           UBS Securities Australia Ltd         Buy         23791         2379           UBS Securities Australia Ltd         Buy         23791         1718           UBS Securities Australia Ltd         Buy         23791         1718           UBS Securities Australia Ltd         Buy         2379         1718           UBS Securities Australia Ltd         Buy         2379         1730           UBS Securities Australia Ltd         Sell <td< td=""><td>t-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>329</td><td>38</td><td>Ordinary</td></td<>	t-14	UBS Securities Australia Ltd	Buy	329	38	Ordinary
URS Securities Australia (14)         Bay         2,699         311           URS Securities Australia (14)         Bay         2,693         311           URS Securities Australia (14)         Bay         2,665         2,901           URS Securities Australia (14)         Bay         2,665         2,901           URS Securities Australia (14)         Bay         2,133         2,148           URS Securities Australia (14)         Bay         2,133         2,148           URS Securities Australia (14)         Bay         2,133         2,148           URS Securities Australia (14)         Bay         4,131         1,116           URS Securities Australia (14)         Bay         4,931         5,70           URS Securities Australia (14)         Bay         4,931         5,70           URS Securities Australia (14)         Bay         4,931         1,131           URS Securities Australia (14)         Bay         2,538         5,70           URS Securities Australia (14)         Bay         4,936         1,135           URS Securities Australia (14)         Bay         2,538         1,136           URS Securities Australia (14)         Bay         4,936         1,137           URS Securities Australia (14)	ct-14	UBS Securities Australia Ltd	Buy	2,165	250	Ordinary
URS Securities Australia Ltd         Bay         2,953         2,788           URS Securities Australia Ltd         Bay         2,957         2,788           URS Securities Australia Ltd         Bay         2,168         2,00           URS Securities Australia Ltd         Bay         2,168         2,00           URS Securities Australia Ltd         Bay         2,252         2,68           URS Securities Australia Ltd         Bay         2,252         2,68           URS Securities Australia Ltd         Bay         2,252         2,68           URS Securities Australia Ltd         Bay         2,582         2,57           URS Securities Australia Ltd         Bay         2,582         1,135           URS Securities Australia Ltd         Bay         2,582         1,135           URS Securities Australia Ltd         Bay         2,582         1,522           URS Securities Australia Ltd         Bay         2,583         1,135           URS Securities Australia Ltd         Bay         2,583         1,135           URS Securities Australia Ltd         Sel         2,544         (350)           URS Securities Australia Ltd         Sel         2,544         (350)           URS Securities Australia Ltd         Sel	kt-14	UBS Securities Australia Ltd	Buy	199	23	Ordinary
URS Securities Australia Ltd         Buy         1,959         2,951         2,558           URS Securities Australia Ltd         Buy         1,992         2,950         2,900           URS Securities Australia Ltd         Buy         2,232         2,500         2,500           URS Securities Australia Ltd         Buy         2,232         2,500         1,179         2,500         1,179         2,500         1,170         1,271         1,170         1,272         2,500         1,170         1,271         1,170         1,271         1,170         1,272         2,500         1,170         1,170         1,170         1,170         1,170         1,271         1,170	0t-14	UBS Securities Australia Ltd	Buy	2,693	311	Ordinary
URS Securines Australia Ltd         Bay         1,922         250           URS Securines Australia Ltd         Bay         2,168         2,578           URS Securines Australia Ltd         Bay         2,108         2,578           URS Securines Australia Ltd         Bay         2,332         2,578           URS Securines Australia Ltd         Bay         2,333         2,578           URS Securines Australia Ltd         Bay         2,582         1,135           URS Securines Australia Ltd         Bay         2,582         1,273           URS Securines Australia Ltd         Bay         2,673         1,252           URS Securines Australia Ltd         Bay         2,673         1,252           URS Securines Australia Ltd         Bay	1 2	UBS Securities Australia Ltd	Buy	23,971	2,768	Ordinary
USS Securities Australia Ltd         Buy         6,105         970         970           USS Securities Australia Ltd         Buy         2,105         970         970           USS Securities Australia Ltd         Buy         2,233         2,578         1,119           USS Securities Australia Ltd         Buy         2,1373         2,582         5,00           USS Securities Australia Ltd         Buy         2,1373         1,137         1,137           USS Securities Australia Ltd         Buy         2,582         2,592         1,137           USS Securities Australia Ltd         Buy         2,137         1,137         1,137           USS Securities Australia Ltd         Buy         2,137         1,137         1,137         1,137           USS Securities Australia Ltd         Buy         2,137         1,137         1,137         1,137         1,137         1,137         1,137         1,137         1,137         1,137         1,	1 1	UBS Countries Australia Ltd	Buy	799.	230	Ordinary
USES Scorrintes Autorials Ltd         Boy         2,500           USES Scorrintes Autorials Ltd         Bluy         2,335         5,70           USES Scorrintes Autorials Ltd         Bluy         2,335         5,70           USES Scorrintes Autorials Ltd         Bluy         1,211         1,105           USES Scorrintes Autorials Ltd         Bluy         5,622         1,523           USES Scorrintes Autorials Ltd         Bluy         5,623         1,524           USES Scorrintes Autorials Ltd         Bluy         5,633         1,525           USES Scorrintes Autorials Ltd         Bluy         5,633         1,520           USES Scorrintes Autorials Ltd         Bluy         5,633         1,520           USES Scorrintes Autorials Ltd         Bluy         5,634         1,155           USES Scorrintes Autorials Ltd         Sell         2,644         1,155           USES Scorrintes Autorials Ltd         Sell         2,105         1,155           USES Scorrintes Autorials Ltd	177	LIBS Securities Australia Ltd	Buy	2,165	250	Ordinary
UBS Securities Australia Ltd         Buy         2,325         2,30           UBS Securities Australia Ltd         Buy         2,322         2,30           UBS Securities Australia Ltd         Buy         2,313         2,486           UBS Securities Australia Ltd         Buy         2,1373         2,486           UBS Securities Australia Ltd         Buy         2,532         2,53           UBS Securities Australia Ltd         Buy         2,532         2,53           UBS Securities Australia Ltd         Buy         2,532         2,53           UBS Securities Australia Ltd         Buy         2,532         1,135           UBS Securities Australia Ltd         Buy         2,532         1,523           UBS Securities Australia Ltd         Buy         2,532         1,523           UBS Securities Australia Ltd         Buy         2,532         1,523           UBS Securities Australia Ltd         Buy         2,732         1,523           UBS Securities Australia Ltd         Buy         2,732         1,523           UBS Securities Australia Ltd         Sell         2,734         1,523           UBS Securities Australia Ltd         Sell         2,735         1,735           UBS Securities Australia Ltd <th< td=""><td>14-14</td><td>IIRS Securities Australia Ltd</td><td>Bury</td><td>8,400</td><td>0/6</td><td>Ordinary</td></th<>	14-14	IIRS Securities Australia Ltd	Bury	8,400	0/6	Ordinary
UBS Securities Australia Ltd         Buy         1,108         1,23         1,248           UBS Securities Australia Ltd         Buy         1,108         1,109         1,109           UBS Securities Australia Ltd         Buy         9,679         1,119         1,109           UBS Securities Australia Ltd         Buy         2,552         2,552         2,593           UBS Securities Australia Ltd         Buy         2,552         2,552         2,593           UBS Securities Australia Ltd         Buy         2,552         2,544         1,572           UBS Securities Australia Ltd         Buy         4,520         1,523         1,523           UBS Securities Australia Ltd         Buy         4,520         1,523         1,523           UBS Securities Australia Ltd         Buy         4,520         1,523         1,523           UBS Securities Australia Ltd         Buy         1,526         1,523         1,523           UBS Securities Australia Ltd         Sell         1,422.4         1,131         1,131         1,131         1,131         1,131         1,131         1,131         1,131         1,131         1,131         1,132         1,132         1,132         1,132         1,132         1,132         1,132	14	UBS Securities Australia 11d	Bita	4,530	0000	Ordinary
UBS Securities Australia Ltd         Buy         2,1,37         2,468           UBS Securities Australia Ltd         Buy         2,1,37         2,468           UBS Securities Australia Ltd         Buy         4,931         5,70           UBS Securities Australia Ltd         Buy         2,552         2,552           UBS Securities Australia Ltd         Buy         2,552         2,552           UBS Securities Australia Ltd         Buy         2,552         2,552           UBS Securities Australia Ltd         Buy         6,23         2,552           UBS Securities Australia Ltd         Buy         6,23         2,552           UBS Securities Australia Ltd         Buy         6,261         1,135           UBS Securities Australia Ltd         Buy         10,516         1,135           UBS Securities Australia Ltd         Sell         2,613         (7,702)           UBS Securities Australia Ltd         Sell         2,613         (7,002)           UBS Securities Australia Ltd         Sell         2,613         (7,002)           UBS Securities Australia Ltd         Sell         2,614         (7,002)           UBS Securities Australia Ltd         Sell         2,615         (7,002)           UBS Securities Australia Ltd	14	UBS Securities Australia Ltd	Bliv	1 108	0/2/2	Ordinary
LIBS Securities Australia Ltd         Buy         9,679         1,19           UBS Securities Australia Ltd         Buy         1,211         1,140           UBS Securities Australia Ltd         Buy         2,552         2,55           UBS Securities Australia Ltd         Buy         2,552         2,55           UBS Securities Australia Ltd         Buy         2,552         2,644           UBS Securities Australia Ltd         Buy         99,752         1,132           UBS Securities Australia Ltd         Buy         99,752         1,135           UBS Securities Australia Ltd         Sell         2,613         (5,50)           UBS Securities Australia Ltd         Sell         2,613         (5,50)           UBS Securities Australia Ltd         Sell         2,613         (6,50)           UBS Securities Australia Ltd         Sell         2,613         (6,50)           UBS Securities Australia Ltd         Sell         2,613         (6,50)           UBS Securities Australia Ltd </td <td>tt-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>21 373</td> <td>837 C</td> <td>Ordinary</td>	tt-14	UBS Securities Australia Ltd	Buy	21 373	837 C	Ordinary
UBS Securities Australia Ltd         Bay         1,211         1,40           UBS Securities Australia Ltd         Buy         535         570           UBS Securities Australia Ltd         Buy         538         575           UBS Securities Australia Ltd         Buy         1,498         1,232           UBS Securities Australia Ltd         Buy         47,091         1,222           UBS Securities Australia Ltd         Buy         47,091         1,425           UBS Securities Australia Ltd         Buy         47,091         1,425           UBS Securities Australia Ltd         Buy         47,091         1,425           UBS Securities Australia Ltd         Buy         5,73         1,135           UBS Securities Australia Ltd         Buy         5,73         1,135           UBS Securities Australia Ltd         Sell         3,798         (1,702)           UBS Securities Australia Ltd         Sell         3,798         (1,602)           UBS Securities Australia Ltd         Sell         3,798         (1,602)           UBS Securities Australia Ltd         Sell         3,798         (1,602)           UBS Securities Australia Ltd         Sell         3,775         (2,500)           UBS Securities Australia Ltd	kt-14	UBS Securities Australia Ltd	Buv	6296	1119	Ordinary
UBS Securities Australia Ltd         Buy         4 931         570           UBS Securities Australia Ltd         Buy         2,552         255           UBS Securities Australia Ltd         Buy         2,552         255           UBS Securities Australia Ltd         Buy         673         173           UBS Securities Australia Ltd         Buy         4,091         1,552           UBS Securities Australia Ltd         Buy         7,14         (1,707)           UBS Securities Australia Ltd         Sell         7,14         (300)           UBS Securities Australia Ltd         Sell         2,178         (2,500)           UBS Securities Australia Ltd         Sell         2,178         (2,500)           UBS Securities Australia Ltd         Sell         3,200         (3,00)           UBS Securities Australia Ltd         Sell         2,175         (2,500)           UBS Securities Australia Ltd         Sell         3,200         (1,001)           UBS Securities Australia Ltd	Mt-14	UBS Securities Australia Ltd	Buv	1211	140	Ordinary
UBS Securities Australia Ltd         Buy         558         62           UBS Securities Australia Ltd         Buy         2,552         295           UBS Securities Australia Ltd         Buy         623         173           UBS Securities Australia Ltd         Buy         99,752         1,135           UBS Securities Australia Ltd         Buy         99,752         1,135           UBS Securities Australia Ltd         Buy         1,636         1,659           UBS Securities Australia Ltd         Buy         1,632         1,523           UBS Securities Australia Ltd         Sell         1,432         1,639           UBS Securities Australia Ltd         Sell         2,178         (2,20)           UBS Securities Australia Ltd         Sell         2,178         (2,20)           UBS Securities Australia Ltd         Sell         2,178         (2,20)           UBS Securities Australia Ltd         Sell         2,178         (2,50)           UBS Securities Australia Ltd	kt-14	UBS Securities Australia Ltd	Buy	4.931	570	Ordinary
UBS Securities Australia Ltd         Buy         1,595         173           UBS Securities Australia Ltd         Buy         673         1,522           UBS Securities Australia Ltd         Buy         673         1,532           UBS Securities Australia Ltd         Buy         67,753         1,532           UBS Securities Australia Ltd         Buy         67,753         1,532           UBS Securities Australia Ltd         Buy         97,753         1,532           UBS Securities Australia Ltd         Sell         1,532         1,532           UBS Securities Australia Ltd         Sell         3,782         (1,702)           UBS Securities Australia Ltd         Sell         2,613         (300)           UBS Securities Australia Ltd         Sell         2,613         (300)           UBS Securities Australia Ltd         Sell         2,613         (300)           UBS Securities Australia Ltd         Sell         2,614         (50)           UBS Securities Australia Ltd         Sell         2,615         (50)           UBS Securities Australia Ltd         Sell         2,175         (2,50)           UBS Securities Australia Ltd         Sell         2,175         (2,50)           UBS Securities Australia Ltd	kt-14	UBS Securities Australia Ltd	Buy	536	29	Ordinary
UBS Securities Australie Ltd         Buy         1,495         173           UBS Securities Australia Ltd         Buy         62,75         11,532           UBS Securities Australia Ltd         Buy         99,752         11,532           UBS Securities Australia Ltd         Buy         10,916         1,532           UBS Securities Australia Ltd         Buy         10,916         1,532           UBS Securities Australia Ltd         Buy         1,425         11,315           UBS Securities Australia Ltd         Sell         7,79         1,425           UBS Securities Australia Ltd         Sell         7,79         1,625           UBS Securities Australia Ltd         Sell         7,17         1,620           UBS Securities Australia Ltd         Sell         2,674         (5,20)           UBS Securities Australia Ltd         Sell         2,675         (5,20)           UBS Securities Australia Ltd         Sell         2,175         (5,20)           UBS Securities Australia Lt	kt-14	UBS Securities Australia Ltd	Buy	2,552	295	Ordinary
UBS Securities Australia Ltd         Buy         99,752         11,532           UBS Securities Australia Ltd         Buy         99,752         11,532           UBS Securities Australia Ltd         Buy         47,091         1,262           UBS Securities Australia Ltd         Buy         97,875         1,1352           UBS Securities Australia Ltd         Sell         1,422-4         (1,702)           UBS Securities Australia Ltd         Sell         3,798         (4,300)           UBS Securities Australia Ltd         Sell         2,798         (4,300)           UBS Securities Australia Ltd         Sell         2,798         (4,300)           UBS Securities Australia Ltd         Sell         2,774         (3,500)           UBS Securities Australia Ltd         Sell         2,774         (3,500)           UBS Securities Australia Ltd         Sell         2,774         (3,500)           UBS Securities Australia Ltd         Sell         2,775         (3,500)           UBS Securities Australia Ltd         Sell         2,775         (3,500)           UBS Securities Australia Ltd         Sell         2,775         (3,500)           UBS Securities Australia Ltd         Sell         4,350         (3,000)           U	ct-14	UBS Securities Australia Ltd	Buy	1,496	173	Ordinary
UBS Securities Australia Ltd         Buy         40,595         11,522           UBS Securities Australia Ltd         Buy         40,595         5,444           UBS Securities Australia Ltd         Buy         70,576         1,315           UBS Securities Australia Ltd         Sell         1,4820         (1,702)           UBS Securities Australia Ltd         Sell         7,14         (82)           UBS Securities Australia Ltd         Sell         7,17         (1,702)           UBS Securities Australia Ltd         Sell         2,178         (3,50)           UBS Securities Australia Ltd         Sell         2,178         (3,50)           UBS Securities Australia Ltd         Sell         3,50         (4,50)           UBS Securities Australia Ltd         Sell         3,50         (4,50)           UBS Securities Australia Ltd         Sell         3,50         (4,50)           UBS Securities Australia Ltd         Sell         2,175         (2,50)           UBS Securities Australia Ltd         Sell         3,50         (4,00)           UBS Securities Australia Ltd         Sell         2,175         (2,50)           UBS Securities Australia Ltd         Sell         3,50         (1,00)           UBS Securities Austra	ct-14	UBS Securities Australia Ltd	Buy	623	72	Ordinary
UBS Securities Australia Ltd         Bby         47/951         5,444           UBS Securities Australia Ltd         Buy         1,252         1,252           UBS Securities Australia Ltd         Buy         1,4350         1,659           UBS Securities Australia Ltd         Sell         3,798         (4,350)           UBS Securities Australia Ltd         Sell         2,178         (2,20)           UBS Securities Australia Ltd         Sell         2,178         (2,20)           UBS Securities Australia Ltd         Sell         2,674         (4,50)           UBS Securities Australia Ltd         Sell         2,674         (4,50)           UBS Securities Australia Ltd         Sell         4,446         (5,10)           UBS Securities Australia Ltd         Sell         2,178         (5,20)           UBS Securities Australia Ltd         Sell         2,178         (5,20)           UBS Securities Australia Ltd         Sell         2,175         (2,50)           UBS Securities Australia Ltd         Sell         2,175         (2,50)           UBS Securities Australia Ltd         Sell         2,175         (2,50)           UBS Securities Australia Ltd         Sell         2,175         (2,00)           UBS Securities A	ct-14	UBS Securities Australia Ltd	Buv	99 752	11532	Ordinary
UBS Securities Australia Ltd         Bby         910516         11,252           UBS Securities Australia Ltd         Buy         97,875         11,315           UBS Securities Australia Ltd         Sell         14,350         11,315           UBS Securities Australia Ltd         Sell         3,782         (1,702)           UBS Securities Australia Ltd         Sell         2,613         (300)           UBS Securities Australia Ltd         Sell         2,613         (300)           UBS Securities Australia Ltd         Sell         2,778         (350)           UBS Securities Australia Ltd         Sell         3,546         (5,60)           UBS Securities Australia Ltd         Sell         3,50         (6,60)           UBS Securities Australia Ltd         Sell         3,50         (6,00)           UBS Securities Australia Ltd         Sell         2,175         (2,50)           UBS Securities Australia Ltd         Sell         2,175         (2,50)           UBS Securities Australia Ltd         Sell         2,175         (3,00)           UBS Securities Australia Ltd         Sell         2,175         (3,00)           UBS Securities Australia Ltd         Sell         3,50         (4,00)           UBS Securities Au	ct-14	UBS Securities Australia Ltd	Buy	160.74	5 444	Ordinary
UBS Securities Australia Ltd         Buy         97,875         11,315           UBS Securities Australia Ltd         Sell         14,350         1,635           UBS Securities Australia Ltd         Sell         37.98         (435)           UBS Securities Australia Ltd         Sell         2,613         (300)           UBS Securities Australia Ltd         Sell         2,613         (300)           UBS Securities Australia Ltd         Sell         2,445         (520)           UBS Securities Australia Ltd         Sell         4,455         (511)           UBS Securities Australia Ltd         Sell         4,455         (520)           UBS Securities Australia Ltd         Sell         4,455         (500)           UBS Securities Australia Ltd         Sell         1,168         (194)           UBS Securities Australia Ltd         Sell         2,175         (250)           UBS Securities Australia Ltd         Sell         2,175         (250)           UBS Securities Australia Ltd         Sell         2,175         (250)           UBS Securities Australia Ltd         Sell         1,168         (141)           UBS Securities Australia Ltd         Sell         1,775         (260)           UBS Securities Australia L	ct-14	UBS Securities Australia Ltd	Buy	10.916	1262	Ordinary
UBS Securities Australia Ltd         Bay         14,350         1,559           UBS Securities Australia Ltd         Sell         3,798         (4,300)           UBS Securities Australia Ltd         Sell         2,613         (300)           UBS Securities Australia Ltd         Sell         2,774         (350)           UBS Securities Australia Ltd         Sell         2,774         (350)           UBS Securities Australia Ltd         Sell         3,744         (350)           UBS Securities Australia Ltd         Sell         4,466         (51)           UBS Securities Australia Ltd         Sell         4,466         (51)           UBS Securities Australia Ltd         Sell         2,175         (300)           UBS Securities Australia Ltd         Sell         2,175         (250)           UBS Securities Australia Ltd         Sell         2,175         (250)           UBS Securities Australia Ltd         Sell         2,175         (250)           UBS Securities Australia Ltd         Sell         3,202         (368)           UBS Securities Australia Ltd         Sell         3,202         (360)           UBS Securities Australia Ltd         Sell         3,202         (402)           UBS Securities Australia Ltd	d-14	UBS Securities Australia Ltd	Buy	57.875	11.315	Ordinary
UBS Securities Australia Ltd         Sell         1,702           UBS Securities Australia Ltd         Sell         7,14         (8.2)           UBS Securities Australia Ltd         Sell         2,178         (3.50)           UBS Securities Australia Ltd         Sell         2,178         (3.50)           UBS Securities Australia Ltd         Sell         2,178         (3.50)           UBS Securities Australia Ltd         Sell         4,426         (5.01)           UBS Securities Australia Ltd         Sell         4,200         (3.60)           UBS Securities Australia Ltd         Sell         4,200         (3.60)           UBS Securities Australia Ltd         Sell         1,66         (1.54)           UBS Securities Australia Ltd         Sell         2,175         (2.50)           UBS Securities Australia Ltd         Sell         3,202         (3.60)           UBS Securities Australia Ltd         Sell         3,558         (1.03)           UBS Securities Australia Ltd	ct-14	UBS Securities Australia Ltd	Buy	14.350	1659	Ordinary
UBS Securities Australia Ltd         Sell         7798         (4350)           UBS Securities Australia Ltd         Sell         7.174         (82)           UBS Securities Australia Ltd         Sell         2.178         (250)           UBS Securities Australia Ltd         Sell         2.874         (320)           UBS Securities Australia Ltd         Sell         4.446         (6.15)           UBS Securities Australia Ltd         Sell         4.350         (6.00)           UBS Securities Australia Ltd         Sell         2.178         (6.20)           UBS Securities Australia Ltd         Sell         2.175         (7.50)           UBS Securities Australia Ltd         Sell         2.00         (400)           UBS Securities Australia Ltd         Sell         1.775         (7.02)           UBS Securities Australia Ltd         Sell         1.775         (7.02)           UBS Securities Austr	ct-14	UBS Securities Australia Ltd	3	14.824	(1 702)	Ordinary
UBS Securities Australia Ltd         Sell         2,513         (820)           UBS Securities Australia Ltd         Sell         2,7178         (350)           UBS Securities Australia Ltd         Sell         3,746         (5,50)           UBS Securities Australia Ltd         Sell         4,446         (6,11)           UBS Securities Australia Ltd         Sell         4,446         (6,11)           UBS Securities Australia Ltd         Sell         4,250         (5,00)           UBS Securities Australia Ltd         Sell         1,688         (15,40)           UBS Securities Australia Ltd         Sell         2,175         (250)           UBS Securities Australia Ltd         Sell         2,610         (300)           UBS Securities Australia Ltd         Sell         2,610         (300)           UBS Securities Australia Ltd         Sell         3,520         (403)           UBS Securities Australia Ltd         Sell         3,555         (403)           UBS Securities Austra	ct-14	UBS Securities Australia Ltd	3	3 798	(989)	Ordinary
UBS Securities Australia Ltd         Sell         2,613         (300)           UBS Securities Australia Ltd         Sell         2,778         (350)           UBS Securities Australia Ltd         Sell         2,474         (350)           UBS Securities Australia Ltd         Sell         4,446         (6,11)           UBS Securities Australia Ltd         Sell         4,446         (6,11)           UBS Securities Australia Ltd         Sell         2,610         (300)           UBS Securities Australia Ltd         Sell         1,165         (134)           UBS Securities Australia Ltd         Sell         2,175         (250)           UBS Securities Australia Ltd         Sell         2,175         (250)           UBS Securities Australia Ltd         Sell         4,350         (500)           UBS Securities Australia Ltd         Sell         4,350         (500)           UBS Securities Australia Ltd         Sell         1,277         (200)           UBS Securities Australia Ltd         Sell         1,775         (200)           UBS Securities Australia Ltd         Sell         1,775         (200)           UBS Securities Australia Ltd         Sell         1,775         (200)           UBS Securities Australia L	d-14	UBS Securities Australia Ltd	9	714	(60)	Ordinary
UBS Securities Australia Ltd         Sel         2,178         (250           UBS Securities Australia Ltd         Sel         2,178         (250           UBS Securities Australia Ltd         Sel         5,446         (51)           UBS Securities Australia Ltd         Sel         4,466         (51)           UBS Securities Australia Ltd         Sel         4,00         (46)           UBS Securities Australia Ltd         Sel         1,168         (154)           UBS Securities Australia Ltd         Sel         2,175         (250)           UBS Securities Australia Ltd         Sel         2,175         (250)           UBS Securities Australia Ltd         Sel         4,350         (500)           UBS Securities Australia Ltd         Sel         2,175         (250)           UBS Securities Australia Ltd         Sel         3,002         (48)           UBS Securities Australia Ltd         Sel         2,000         (400)           UBS Securities Australia Ltd         Sel         3,002         (402)           UBS Securities Australia Ltd         Sel         1,775         (200)           UBS Securities Australia Ltd         Sel         1,775         (200)           UBS Securities Australia Ltd         Sel<	ct-14	UBS Securities Australia 11d	3	2,842	(700)	Ordinary
UBS Securities Australia Ltd         Sell         2,874         (356)           UBS Securities Australia Ltd         Sell         4,446         (510)           UBS Securities Australia Ltd         Sell         4,446         (510)           UBS Securities Australia Ltd         Sell         4,350         (66)           UBS Securities Australia Ltd         Sell         1,65         (134)           UBS Securities Australia Ltd         Sell         2,715         (250)           UBS Securities Australia Ltd         Sell         3,775         (250)           UBS Securities Australia Ltd         Sell         3,775         (50)           UBS Securities Australia Ltd         Sell         3,202         (50)           UBS Securities Australia Ltd         Sell         3,202         (50)           UBS Securities Australia Ltd         Sell         1,277         (20)           UBS Securities Australia Ltd         Sell         1,775         (20)           UBS Securities Australia Ltd	41-14	IUBS Securities Australia I1d	3	2,23	(350)	Ordinary
USS Securities Australia Ltd         Sell         6,259           USS Securities Australia Ltd         Sell         4,446         (6,10)           USS Securities Australia Ltd         Sell         4,350         (6,00)           USS Securities Australia Ltd         Sell         2,610         (6,00)           USS Securities Australia Ltd         Sell         2,775         (7,00)           USS Securities Australia Ltd         Sell         2,775         (7,00)           USS Securities Australia Ltd         Sell         4,350         (5,00)           USS Securities Australia Ltd         Sell         4,350         (5,00)           USS Securities Australia Ltd         Sell         4,350         (5,00)           USS Securities Australia Ltd         Sell         1,277         (1,41)           USS Securities Australia Ltd         Sell         1,277         (1,03)           USS Securities Australia Ltd         Sell         1,775         (2,02)           USS Securities Australia Ltd	41-14	IUBS Securities Australia Itd	9	2 074	(000)	Cidilary
UBS Securities Australia Ltd         Sell         4,446         (517)           UBS Securities Australia Ltd         Sell         4,446         (517)           UBS Securities Australia Ltd         Sell         4,000         (46)           UBS Securities Australia Ltd         Sell         1,668         (154)           UBS Securities Australia Ltd         Sell         2,175         (250)           UBS Securities Australia Ltd         Sell         2,175         (250)           UBS Securities Australia Ltd         Sell         4,350         (50)           UBS Securities Australia Ltd         Sell         4,370         (56)           UBS Securities Australia Ltd         Sell         3,202         (40)           UBS Securities Australia Ltd         Sell         1,775         (200)           UBS Securities Australia Ltd         Sell         1,775         (200)           UBS Securities Australia Ltd         Sell         1,775         (200)           UBS Securities Australia Ltd         Sell         1,803         (103)           UBS Securities Australia Ltd         Sell         829         (103)           UBS Securities Australia Ltd         Sell         4,356         (403)           UBS Securities Australia Ltd	41-14	UBS Securities Australia Itd	lles	2,012	(959)	Ordinary
UBS Securities Australia Ltd         Sell         4,55G         (50D)           UBS Securities Australia Ltd         Sell         2,610         (50D)           UBS Securities Australia Ltd         Sell         1,166         (1,34)           UBS Securities Australia Ltd         Sell         1,168         (1,34)           UBS Securities Australia Ltd         Sell         2,175         (2,50)           UBS Securities Australia Ltd         Sell         4,350         (50D)           UBS Securities Australia Ltd         Sell         4,350         (50D)           UBS Securities Australia Ltd         Sell         3,222         (56D)           UBS Securities Australia Ltd         Sell         2,175         (2,00)           UBS Securities Australia Ltd         Sell         2,510         (40)           UBS Securities Australia Ltd         Sell         2,610         (30D)           UBS Securities Australia Ltd         Sell         1,775         (202)           UBS Securities Australia Ltd         Sell         1,577         (202)           UBS Securities Australia Ltd         Sell         1,577         (202)           UBS Securities Australia Ltd         Sell         1,577         (202)           UBS Securities Australia	4-14	UBS Securities Australia (1d)		7,446	(070)	Ordinary
UBS Securities Australia Ltd         Sell         4.00         (4.00)           UBS Securities Australia Ltd         Sell         1,168         (1.54)           UBS Securities Australia Ltd         Sell         1,168         (1.54)           UBS Securities Australia Ltd         Sell         2,175         (2.50)           UBS Securities Australia Ltd         Sell         2,175         (2.50)           UBS Securities Australia Ltd         Sell         4,350         (5.00)           UBS Securities Australia Ltd         Sell         4,350         (5.00)           UBS Securities Australia Ltd         Sell         2,000         (3.00)           UBS Securities Australia Ltd         Sell         1,277         (4.01)           UBS Securities Australia Ltd         Sell         1,775         (2.00)           UBS Securities Australia Ltd         Sell         1,775         (2.00)           UBS Securities Australia Ltd         Sell         1,775         (2.00)           UBS Securities Australia Ltd         Sell         3,556         (4.02)           UBS Securities Australia Ltd         Sell         3,556         (4.02)           UBS Securities Australia Ltd         Sell         4,386         (4.02)           UBS Securities	4-14	I IRS Securities Australia 14d		1,4	(100)	Crairiary
UBS Securities Australia Ltd         Sell         2,610         (409)           UBS Securities Australia Ltd         Sell         1,166         (134)           UBS Securities Australia Ltd         Sell         1,166         (134)           UBS Securities Australia Ltd         Sell         2,175         (250)           UBS Securities Australia Ltd         Sell         2,175         (250)           UBS Securities Australia Ltd         Sell         4,825         (500)           UBS Securities Australia Ltd         Sell         2,175         (250)           UBS Securities Australia Ltd         Sell         3,202         (360)           UBS Securities Australia Ltd         Sell         2,610         (300)           UBS Securities Australia Ltd         Sell         2,610         (300)           UBS Securities Australia Ltd         Sell         1,775         (200)           UBS Securities Australia Ltd         Sell         1,603         (103)           UBS Securities Australia Ltd         Sell         1,57         (200)           UBS Securities Australia Ltd         Sell         1,57         (200)           UBS Securities Australia Ltd         Sell         1,57         (200)           UBS Securities Australia Ltd <td>177</td> <td>I IDS Securities Australia Ind</td> <td>liac</td> <td>4,550</td> <td>(nne)</td> <td>Ordinary</td>	177	I IDS Securities Australia Ind	liac	4,550	(nne)	Ordinary
USS Securities Australia Ltd         Sell         (400)           USS Securities Australia Ltd         Sell         (188         (194)           USS Securities Australia Ltd         Sell         2,175         (250)           USS Securities Australia Ltd         Sell         2,175         (250)           USS Securities Australia Ltd         Sell         2,175         (250)           USS Securities Australia Ltd         Sell         4350         (500)           USS Securities Australia Ltd         Sell         1,227         (141)           USS Securities Australia Ltd         Sell         2,610         (300)           USS Securities Australia Ltd         Sell         1,775         (204)           USS Securities Australia Ltd         Sell         1,775         (204)           USS Securities Australia Ltd         Sell         1,775         (204)           USS Securities Australia Ltd         Sell         1,803         (207)           USS Securities Australia Ltd         Sell         1,803         (402)           USS Securities Australia Ltd         Sell         1,803         (402)           USS Securities Australia Ltd         Sell         1,803         (402)           USS Securities Australia Ltd         Sell <td>1 1</td> <td>Upp Securities Australia Ind</td> <td>Jen C</td> <td>400</td> <td>(46)</td> <td>Ordinary</td>	1 1	Upp Securities Australia Ind	Jen C	400	(46)	Ordinary
UBS Securities Australia Ltd         Sell         1,10-9         (1,49-)           UBS Securities Australia Ltd         Sell         2,17-5         (250)           UBS Securities Australia Ltd         Sell         2,17-5         (250)           UBS Securities Australia Ltd         Sell         4,350         (500)           UBS Securities Australia Ltd         Sell         3,202         (500)           UBS Securities Australia Ltd         Sell         1,227         (641)           UBS Securities Australia Ltd         Sell         2,610         (300)           UBS Securities Australia Ltd         Sell         2,610         (300)           UBS Securities Australia Ltd         Sell         1,775         (200)           UBS Securities Australia Ltd         Sell         1,809         (103)           UBS Securities Australia Ltd         Sell         1,809         (103)           UBS Securities Australia Ltd         Sell         3,558         (409)           UBS Securities Australia Ltd         Sell         3,558         (409)           UBS Securities Australia Ltd         Sell         3,558         (409)           UBS Securities Australia Ltd         Sell         4,386         (50)           UBS Securities Australia		Tipe Committee Australia Lid	Dell Sell	01972	(006)	Ordinary
USS Securities Australia tod         Sell         1,088         (1940)           USS Securities Australia tod         Sell         2,175         (250)           USS Securities Australia tod         Sell         4,350         (500)           USS Securities Australia tod         Sell         4,350         (500)           USS Securities Australia tod         Sell         4,37         (500)           USS Securities Australia tod         Sell         2,175         (500)           USS Securities Australia tod         Sell         2,610         (300)           USS Securities Australia tod         Sell         2,610         (300)           USS Securities Australia tod         Sell         1,775         (200)           USS Securities Australia tod         Sell         1,603         (103)           USS Securities Australia tod         Sell         1,503         (103)           USS Securities Australia tod         Sell         3,560         (103)           USS Securities Australia tod         Sell         1,577         (200)           USS Securities Australia tod         Sell         1,577         (200)           USS Securities Australia tod         Sell         4,167         (409)           USS Securities Australia tod		ODS SECURICIES AUSTRAINA LICE	Sell	1,165	(134)	Ordinary
UBS Securities Australia Ltd         Sell         2.775         CS-50           UBS Securities Australia Ltd         Sell         4.350         (502)           UBS Securities Australia Ltd         Sell         4.350         (502)           UBS Securities Australia Ltd         Sell         3.202         (588)           UBS Securities Australia Ltd         Sell         2.610         (300)           UBS Securities Australia Ltd         Sell         1.275         (300)           UBS Securities Australia Ltd         Sell         1.775         (207)           UBS Securities Australia Ltd         Sell         1.803         (103)           UBS Securities Australia Ltd         Sell         1.805         (103)           UBS Securities Australia Ltd         Sell         4.385         (103)           UBS Securities Australia Ltd         Sell         4.386         (103)           UBS Securities Australia Ltd	1 1	OBS Securities Australia Ltd	Sell	1,688	(194)	Ordinary
UBS Securities Australia Ltd         Sell         4,350         (500)           UBS Securities Australia Ltd         Sell         4,350         (500)           UBS Securities Australia Ltd         Sell         3,202         (560)           UBS Securities Australia Ltd         Sell         1,227         (141)           UBS Securities Australia Ltd         Sell         2,510         (300)           UBS Securities Australia Ltd         Sell         1,775         (200)           UBS Securities Australia Ltd         Sell         1,775         (200)           UBS Securities Australia Ltd         Sell         895         (103)           UBS Securities Australia Ltd         Sell         895         (103)           UBS Securities Australia Ltd         Sell         8,35         (609)           UBS Securities Australia Ltd         Sell         4,385         (609)           UBS Securities Australia Ltd         Sell         4,385         (609)           UBS Securities Australia Ltd         Sell         4,385         (609)           UBS Securities Australia Ltd         Sell         4,167         (479)           UBS Securities Australia Ltd         Sell         4,167         (479)	41.14	UBS Securities Australia Ltd	>e#	2,175	(250)	Ordinary
UBS Securities Australia Ltd         Self         43 SO         (500)           UBS Securities Australia Ltd         Self         43 SO         (560)           UBS Securities Australia Ltd         Self         1,277         (368)           UBS Securities Australia Ltd         Self         2,610         (300)           UBS Securities Australia Ltd         Self         1,775         (200)           UBS Securities Australia Ltd         Self         1,775         (200)           UBS Securities Australia Ltd         Self         1,775         (202)           UBS Securities Australia Ltd         Self         895         (103)           UBS Securities Australia Ltd         Self         3,556         (402)           UBS Securities Australia Ltd         Self         1,757         (202)           UBS Securities Australia Ltd         Self         1,757         (202)           UBS Securities Australia Ltd         Self         4,385         (402)           UBS Securities Australia Ltd         Self         4,385         (504)           UBS Securities Australia Ltd         Self         4,385         (504)           UBS Securities Australia Ltd         Self         4,385         (504)           UBS Securities Australia Ltd </td <td>4-14</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>2,175</td> <td>(250)</td> <td>Ordinary</td>	4-14	UBS Securities Australia Ltd	Sell	2,175	(250)	Ordinary
UBS Securities Australia Ltd         Seli         492         (56)           UBS Securities Australia Ltd         Sell         3,202         (368)           UBS Securities Australia Ltd         Sell         1,227         (141)           UBS Securities Australia Ltd         Sell         2,610         (300)           UBS Securities Australia Ltd         Sell         1,775         (204)           UBS Securities Australia Ltd         Sell         895         (103)           UBS Securities Australia Ltd         Sell         895         (103)           UBS Securities Australia Ltd         Sell         895         (103)           UBS Securities Australia Ltd         Sell         1,757         (204)           UBS Securities Australia Ltd         Sell         1,757         (204)           UBS Securities Australia Ltd         Sell         1,757         (204)           UBS Securities Australia Ltd         Sell         4,385         (504)           UBS Securities Australia Ltd         Sell         4,385         (504)           UBS Securities Australia Ltd         Sell         4,167         (479)           UBS Securities Australia Ltd         Sell         4,167         (479)	r-14	UBS Securities Australia Ltd	Sell	4,350	(200)	Ordinary
UBS Securities Australia Ltd         Sell         3.222         G889           UBS Securities Australia Ltd         Sell         1,227         (141)           UBS Securities Australia Ltd         Sell         2,610         (300)           UBS Securities Australia Ltd         Sell         1,775         (200)           UBS Securities Australia Ltd         Sell         1,775         (200)           UBS Securities Australia Ltd         Sell         897         (103)           UBS Securities Australia Ltd         Sell         897         (103)           UBS Securities Australia Ltd         Sell         3,558         (409)           UBS Securities Australia Ltd         Sell         1,757         (200)           UBS Securities Australia Ltd         Sell         1,757         (200)           UBS Securities Australia Ltd         Sell         1,757         (200)           UBS Securities Australia Ltd         Sell         4,385         (504)           UBS Securities Australia Ltd         Sell         4,167         (479)           UBS Securities Australia Ltd         Sell         4,167         (479)	d-14	UBS Securities Australia Ltd	Seli	487	(99)	Ordinary
UBS Securities Australia (Ed.)         Sell         1.227         (44)           UBS Securities Australia (Ed.)         Sell         2.610         (300)           USS Securities Australia (Ed.)         Sell         7.75         (200)           UBS Securities Australia (Ed.)         Sell         1.75         (200)           UBS Securities Australia (Ed.)         Sell         897         (103)           UBS Securities Australia (Ed.)         Sell         896         (103)           UBS Securities Australia (Ed.)         Sell         3,558         (409)           UBS Securities Australia (Ed.)         Sell         4,385         (500)           UBS Securities Australia (Ed.)         Sell         4,585         (52)           UBS Securities Australia (Ed.)         Sell         4,5167         (479)	ct-14	UBS Securities Australia Ltd	Sell	3,202	(368)	Ordinary
UBS Securities Australia 144         Sell         2,610         (300)           USS Securities Australia 144         Sell         2,610         (300)           USS Securities Australia 144         Sell         1,775         (204)           USS Securities Australia 144         Sell         893         (103)           USS Securities Australia 144         Sell         896         (103)           USS Securities Australia 144         Sell         896         (103)           USS Securities Australia 144         Sell         1,757         (202)           USS Securities Australia 144         Sell         4,385         (504)           USS Securities Australia 144         Sell         4,385         (77)           USS Securities Australia 144         Sell         4,167         (479)           USS Securities Australia 144         Sell         4,167         (479)	ct-14	UBS Securities Australia Ltd	Sell	1,227	(141)	Ordinary
UBS Securities Australia Ltd         Sell         2,610         (300)           UBS Securities Australia Ltd         Sell         1,775         (204)           UBS Securities Australia Ltd         Sell         897         (103)           UBS Securities Australia Ltd         Sell         895         (103)           UBS Securities Australia Ltd         Sell         856         (402)           UBS Securities Australia Ltd         Sell         1,757         (202)           UBS Securities Australia Ltd         Sell         1,757         (202)           UBS Securities Australia Ltd         Sell         4,385         (504)           UBS Securities Australia Ltd         Sell         4,385         (504)           UBS Securities Australia Ltd         Sell         4,385         (504)           UBS Securities Australia Ltd         Sell         4,5167         (479)           UBS Securities Australia Ltd         Sell         4,167         (479)	ct-14	UBS Securities Australia Ltd	Sell	2.610	(300)	Ordinary
USS Securites Australia Ltd         Sell         1,775         (204)           UBS Securites Australia Ltd         Sell         1,775         (207)           UBS Securites Australia Ltd         Sell         1,803         (207)           UBS Securites Australia Ltd         Sell         897         (103)           UBS Securites Australia Ltd         Sell         1,757         (202)           UBS Securites Australia Ltd         Sell         1,757         (202)           UBS Securites Australia Ltd         Sell         4,385         (504)           UBS Securites Australia Ltd         Sell         4,385         (77)           UBS Securites Australia Ltd         Sell         4,386         (77)           UBS Securites Australia Ltd         Sell         4,377         (479)           UBS Securites Australia Ltd         Sell         4,387         (479)	1-14	UBS Securities Australia Ltd	Sell	2,610	(300)	Ordinary
UBS Securities Australia Ltd         Sell         (207)           UBS Securities Australia Ltd         Sell         897         (103)           UBS Securities Australia Ltd         Sell         896         (103)           UBS Securities Australia Ltd         Sell         3,558         (409)           UBS Securities Australia Ltd         Sell         1,557         (202)           UBS Securities Australia Ltd         Sell         4,385         (504)           UBS Securities Australia Ltd         Sell         4,385         (7)           UBS Securities Australia Ltd         Sell         4,167         (479)	1-14	UBS Securities Australia Ltd	Sell	1775	(200)	Ordinary
UBS Securities Australia Ltd         Seli         897         (103)           UBS Securities Australia Ltd         Sell         896         (103)           UBS Securities Australia Ltd         Sell         3,558         (403)           UBS Securities Australia Ltd         Sell         1,757         (202)           UBS Securities Australia Ltd         Sell         4,385         (504)           UBS Securities Australia Ltd         Sell         67         (7)           UBS Securities Australia Ltd         Sell         4,385         (504)           UBS Securities Australia Ltd         Sell         4,387         (52)           UBS Securities Australia Ltd         Sell         4,167         (479)	1-14	UBS Securities Australia Ltd	les	1,803	(202)	Ordinary
UBS Securities Australia Ltd         Self         (103)           UBS Securities Australia Ltd         Self         3.558         (409)           UBS Securities Australia Ltd         Self         1,757         (202)           UBS Securities Australia Ltd         Self         4,385         (504)           UBS Securities Australia Ltd         Self         4,385         (504)           UBS Securities Australia Ltd         Self         4,385         (77)           UBS Securities Australia Ltd         Self         4,387         (479)           UBS Securities Australia Ltd         Self         4,167         (479)	1-14	UBS Securities Australia (1d	la y	200	(103)	Ordinary
UBS Securities Australia tot   UBS Securities Australia tot   UBS Securities Australia tot   Seli   1,757 (202)   UBS Securities Australia tot   Seli   1,757 (479)   UBS Securities Australia tot   1,757 (479)   U		TIDS Countries Australia (44)	100	/60	(601)	Ordinary
UBS Securines Australia Ltd         Sell         4,395         (4,02)           UBS Securines Australia Ltd         Sell         4,385         (5,04)           UBS Securines Australia Ltd         Sell         61         (7)           UBS Securines Australia Ltd         Sell         62         (5,04)           UBS Securines Australia Ltd         Sell         4,385         (5,04)           UBS Securines Australia Ltd         Sell         4,167         (479)	1 7 7	Tibe Convision Australia Ltd	nac.	968	(103)	Ordinary
UBS Securities Australia Ltd         Sell         1,557         (202)           UBS Securities Australia Ltd         Sell         6,000         (7)           UBS Securities Australia Ltd         Sell         4,52         (7)           UBS Securities Australia Ltd         Sell         4,167         (479)           UBS Securities Australia Ltd         Sell         4,167         (479)		Obsideration Australia Lid		800,5	(409)	Ordinary
UBS Securities Australia Ltd         Sell         4,365         (5,04)           UBS Securities Australia Ltd         Sell         6.1         (7)           UBS Securities Australia Ltd         Sell         452         (22)           UBS Securities Australia Ltd         Sell         4,167         (479)		obs seculates Australia Lid	= 1	/5/1	(707)	Ordinary
UBS Securities Australia Ltd   Sell   Sell   4167   (479)	14	18C Cocuring Australia Ind	1136	4,363	(504)	Ordinary
UBS Securities Australia Ltd   Sell   4,167 (479)		Libe Cocuries Australia Ital	llac	10		Ordinary
UBS Securities Australia Ltd   Sell (479)	ſ	UBS Securities Austraria Ltd	Sell	452	(52)	
					(200)	Crairialy

Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	1000	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	
(218)	(787)	(42)	(156)	(368)	(671)	(197)	(604)	(604)	(1)	(1)	(000)	(430)	(300)	(315)	(200)	(250),	(287)	(665)	(009)	(2,687)	(548)	(1,760)	(2,450)	(2,939)	(655)	(260)	(045)	(5.584)	(19,416)	(543)	(2,948)	(548)	(17)	(366)	(546)	(2)	(C) (C)	(562)	(300)	(429)	(201)	(236)	(8)	(34)	(822.887)	255	787	645	225	674	430	250	90	8//	65	1,000	/71	162	20.	280	538	382	498	16	114	3	88	516	
1,903	2,506	/96	1,363	3,213	1,563	1,722	5.270	5.270	σ	7 543	245.0	3,745	2,613	2,744	4,355	2,178	2,500	5,214	5,223	23,390	4,770	15,321	21,327	25,569	4,069	4,869	4,747	48 581	168,919	4,724	25,648	4,768	148	3,188	4,733	σ	2 178	2,604	2,613	3,737	1,751	2,056	70	796	7.171.460	2,203	6,800	5,573	1,942	5,817	3,711	2,158	111	b,/14	100 0	1,007	0 467	1 400	43	2,419	4,648	3,300	4,303	786	382	36	260	4,458	
														les.		Sel	Set				Sell	Sell	Sell	Sel	lec S	los los	lley							Sell				Sell		Sell	Sell	Se	Sell	las	Sell				Buy	Buy	Buy	Buy	Buy	Buy.	Ang and	Bus	ing a	9 8	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	
UBS Securities Australia Ltd	UBS Securities Australia Ltd	bs securities Australia Ltd	BS Securities Australia Ltd	BS Securities Australia Ltd	UBS Securities Australia Ltd	BS Securities Australia Ltd	BS Securities Australia Ltd	UBS Securities Australia Ltd	BS Securities Australia Ltd	BS Securities Australia 14d	DS Securities Australia Ltd	DO DECUMES AUGUST BUT	BS Securities Australia Ltd	BS Securities Australia Ltd	85 Securities Australia Ltd	BS Securities Australia Ltd	BS Securities Australia Ltd	BS Securities Australia Ltd	BS Securities Australia Ltd	BS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UDS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	BS Securities Australia Ltd	85 Securities Australia Ltd	BS Securities Australia Ltd	BS Securities Australia Ltd	BS Securities Australia Ltd	BS Securities Australia Ltd	BS Securities Australia Ltd	BS Securities Australia Ltd	BS Securities Australia Ltd	UBS Securities Australia Ltd	BS Securities Australia Ltd	UBS Securities Australia Ltd	35 Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	35 Securities Australia Ltd	35 Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	DE Securities Australia Ltd	UBS Securities Australia Ltd	IBS Securities Australia Ltd	IBS Securities Australia 11d	385 Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	S Securities Australia Ltd	S Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	S Securities Australia Ltd	UBS Securities Australia Ltd	
	20-0ct-14	1	ľ					ı		l		ı		1		-							1	20.00-14	T	Ī	Ť	l					١		ı		ı	20-Oct-14 UE				T	20-04-14		Γ					Ī	٦	20-0ct-14 UB		20 OCT-14		T	T	Ť	20-Oct-14 UB	Ť		20-Oct-14 UB				20-Oct-14 UB			l

165 Securine Autorials Ltd   541   645		UBS Wealth Management Australia Ltd	Sell	4,350	(200)	Ordina
USS Securine Authorise Ltd.   Self   1,000		JBS Securities Australia Ltd	Sell	653	(75)	Ordinary
19   Security Activities   19   19   19   19   19   19   19   1	T	JBS Securities Australia Ltd	Sell	1,479	(170)	Ordinary
18   Securitaria Australia LEG   Securitaria Australia L	T	IBS Securities Australia Ind	Sell	559'77	(2,604)	Ordina
1855 Securities Australia Ltd   541   542   54		IDS Contribes Australia Ltd	nac 1 och	744	(28)	Ordina
1855 Securine Australia Live   1841   1,2424   1,845	T	IRC Securities Australia 144	Coll	995	(59)	Crdina
Bits Sectioner Authorise is 10		IRS Securities Australia IId	3	2100	(100)	Ordinary
Bits Securine Automisk Led   Sel   2,219   C,200	l	JBS Securities Australia Ind	3	3510	(700)	Cignaly
Bit Securior Automita Line   Securior Automi		IRS Securities Australia 14d	3	010,2	(005)	Ordinary
Big Securities Automists to the part   Big 256   (4,434)		IRC Countries Australia Ind	Cell	617'6	(0/5)	Crana
Bis Securities Automitia bit of Securities Automitia bit	1	IRC Committee Americalia (4)	Cell	/97	(32)	Crana
1855 Securities Automitation   2440   4450	١	IDC Committee Americality and	Call	0/5'05	(4,434)	Cdina
(1) SES scenarios Americanis (14)         (24)         (24)         (24)           (15) Securios Augustia (14)         (24)         (24)         (24)           (15) Securios Augustia (14)         (24)         (24)         (25)           (16) Securios Augustia (14)         (24)	ļ	IDC Cocuration Australia 144	ll ac	3,741	(430)	Ordinary
USS Securities Automated Ltd         Seat         3,149         (550)           USS Securities Automated Ltd         Seat         1,183         (199)           USS Securities Automated Ltd         Seat         1,183         (199)           USS Securities Automated Ltd         Seat         1,183         (199)           USS Securities Automated Ltd         Seat         2,178         (199)           USS Securities Automated Ltd         Seat         4,785         (199)           USS Securities Automated Ltd         Seat         4,785         (199)           USS Securities Automated Ltd         Seat         5,786         (103)           USS Securities Automated Ltd	l	DD Securities Australia Ltd	11-0	348	(40)	Ordina
1955 Securities Australia Ltd         Self         3,045         (1950)           1955 Securities Australia Ltd         Self         1,731         (199)           195 Securities Australia Ltd         Self         1,731         (199)           195 Securities Australia Ltd         Self         1,737         (190)           195 Securities Australia Ltd         Self         2,737         (190)           195 Securities Australia Ltd         Self         2,737         (190)           195 Securities Australia Ltd         Self         2,738         (1,650)           195 Securities Australia Ltd         Self         2,738         (1,650)           195 Securities Australia Ltd         Self         2,738         (1,650)           195 Securities Aust	١	us securities Australia Ltd	Jeil Jeil	4,350	(200)	Ordinary
USS Securities Autorials Ltd   Sell   173149   15420	1	JBS Securities Australia Ltd	Seif	3,045	(320)	Ordinary
URS Securities Australia (14)   Sell   1731   (15)		JBS Securities Australia Ltd	Sell	3,149	(362)	Ordinary
UBS Securities Australia Ltd         Sell         1.193         (1.50           UBS Securities Australia Ltd         Sell         2.173         (1.50           UBS Securities Australia Ltd         Sell         49.053         (1.020)           UBS Securities Australia Ltd         Sell         6.050         (1.021)           UBS Securities Australia Ltd         Sell         4.055         (1.021)           UBS Securities Australia Ltd         Sell         4.056         (1.021)           UBS Securities Australia Ltd         Sell         4.056         (1.021)           UBS Securities Australia Ltd         Sell         4.056         (1.022)           UBS Securities Australia Ltd         Sell         4.056         (1.020)           UBS Securities Australia Ltd         Sell         4.056         (1.020)           UBS Securities Australia Ltd         Sell         2.271         (1.020)           UBS Securities Australia Ltd         Sell         2.050         (1.020)           U		JBS Securities Australia Ltd	Sell	1,731	(199)	Ordina
URS Securities Australia Ltd         Sel         2.175         CGSD           URS Securities Australia Ltd         Sel         4.176         (103 b)           URS Securities Australia Ltd         Sel         4.176         (103 b)           URS Securities Australia Ltd         Sel         6.050         (103 b)           URS Securities Australia Ltd         Sel         6.050         (103 b)           URS Securities Australia Ltd         Sel         6.050         (103 b)           URS Securities Australia Ltd         Sel         7.650         (685)           URS Securities Australia Ltd         Sel         7.650         (685)           URS Securities Australia Ltd         Sel         7.650         (680)           URS Securities Australia Ltd         Sel         7.650         (680)           URS Securities Australia Ltd         Sel         7.650         (7.520)           URS Securities Australia Ltd         Sel         7.650         (7.520)           URS Securities Australia Ltd         Sel         7.760         (7.520)           URS Securities Australia Ltd         Sel         7.760         (7.520)           URS Securities Australia Ltd         Sel         7.760         (7.520)           URS Securities Australia		JBS Securities Australia Ltd	Sell	139	(16)	Ordinary
URS Securines Australia Ltd         Sell         4176         (138)           URS Securines Australia Ltd         Sell         4176         (140)           URS Securines Australia Ltd         Sell         600         (100)           URS Securines Australia Ltd         Sell         600         (100)           URS Securines Australia Ltd         Sell         40,581         (572)           URS Securines Australia Ltd         Sell         40,581         (573)           URS Securines Australia Ltd         Sell         2,100         (650)           URS Securines Australia Ltd         Sell         2,118         (250)           URS Securines Australia Ltd         Se		JBS Securities Australia Ltd	Sell	2.175	(050)	Ordina
UBS Securities Australia Ltid         Sell         4176         (480)           UBS Securities Australia Ltid         Sell         6,030         (700)           UBS Securities Australia Ltid         Sell         40,031         (5,538)           UBS Securities Australia Ltid         Sell         5,050         (700)           UBS Securities Australia Ltid         Sell         5,050         (850)           UBS Securities Australia Ltid         Sell         2,175         (5,520)           UBS Securities Australia Ltid         Sell         1,778         (7,520)           UBS Securities Australia Ltid         Sell         1,778         (7,520)		JBS Securities Australia Ltd	Se	1 183	(136)	Continue
UBS Securities Australia Ltd         Sell         6,590         (102)           UBS Securities Australia Ltd         Sell         6,590         (102)           UBS Securities Australia Ltd         Sell         49,051         (5,538)           UBS Securities Australia Ltd         Sell         40,551         (5,538)           UBS Securities Australia Ltd         Sell         4,585         (6,590)           UBS Securities Australia Ltd         Sell         4,585         (6,590)           UBS Securities Australia Ltd         Sell         2,717         (7,52)           UBS Securities Australia Ltd         Sell         2,718         (7,52)           UBS S	l	JBS Securities Australia I tol	las	3617	(150)	500
USS Securities Australia Ltd         Sell         6,590         (700)           UBS Securities Australia Ltd         Sell         6,590         (700)           UBS Securities Australia Ltd         Sell         6,590         (700)           UBS Securities Australia Ltd         Sell         5,0051         (6,578)           UBS Securities Australia Ltd         Sell         6,500         (650)           UBS Securities Australia Ltd         Sell         7,75         (650)           UBS Securities Australia Ltd         Sell         6,500         (650)           UBS Securities Australia Ltd         Sell         7,70         (1,000)           UBS Securities Australia Ltd         Sell         8,70         (1,000)           UBS Securities Australia Ltd         Sell         8,70         (1,000)           UBS Securities Australia Ltd         Sell         8,70         (1,000)           UBS Securities Australia Ltd         Sell         2,715         (2,520)           UBS Securities Australia Ltd         Sell         2,715         (2,520)           UBS Securities Australia Ltd         Sell         2,715         (2,520)           UBS Securities Australia Ltd         Sell         1,705         (1,520)           UBS Securities	ĺ	IR Cocurities Australia 14d		9/1/4	(480)	Craina
URS Securities Australia Ltd         Seil         6,590         (7,00)           UBS Securities Australia Ltd         Seil         6,590         (7,00)           UBS Securities Australia Ltd         Seil         40,551         (5,53)           UBS Securities Australia Ltd         Seil         40,533         (6,50)           UBS Securities Australia Ltd         Seil         2,700         (600)           UBS Securities Australia Ltd         Seil         2,600         (600)           UBS Securities Austral	1	As securities Australia cita	lac l	896	(103)	Ordina
UBS Securities Australia Ltd         Sell         4,955.1         6,55.8           UBS Securities Australia Ltd         Sell         4,05.1         6,57.9           UBS Securities Australia Ltd         Sell         6,00.1         6,57.9           UBS Securities Australia Ltd         Sell         7,75         6,60.0           UBS Securities Australia Ltd         Sell         4,25.0         (4,65.0)           UBS Securities Australia Ltd         Sell         4,25.0         (6,00.0)           UBS Securities Australia Ltd         Sell         2,715         (2,50.0)           UBS Securities Australia Ltd         Sell         2,610         (3,00.0)           UBS Securities Australia Ltd         Sell         2,610         (3,00.0)           UBS Securities Australia Ltd         Sell         2,610         (3,00.0)           UBS Securities Australia Ltd         Sell         2,1,186         (7,23.0)           UBS Securities Australia Ltd         Sell         2,1,186         (1,25.0)           UBS Securities Australia Ltd         Sell         2,1,186         (1,25.0)           UBS Securities Australia Ltd         Sell         2,1,186         (1,25.0)           UBS Securities Australia Ltd         Sell         2,09         (1,25.0)		JBS Securities Australia Ltd	sell	060'9	(200)	Ordina
UBS Securities Australia Ltd         Sell         5,0051         (5/5)           UBS Securities Australia Ltd         Sell         4,033         (4,656)           UBS Securities Australia Ltd         Sell         4,033         (4,656)           UBS Securities Australia Ltd         Sell         7,659         (556)           UBS Securities Australia Ltd         Sell         7,659         (620)           UBS Securities Australia Ltd         Sell         2,750         (620)           UBS Securities Australia Ltd         Sell         2,750         (620)           UBS Securities Australia Ltd         Sell         2,610         (300)           UBS Securities Australia Ltd         Sell         2,630         (300)           UBS Securities Australia Ltd         Sell         3,930         (3,649)           UBS Securities Austra	١	JBS Securities Australia Ltd	Sell	49,051	(8:638)	Ordina
UBS Securities Autorials Ltd         Self         478         (500)           UBS Securities Autorials Ltd         Self         5220         (600)           UBS Securities Autorials Ltd         Self         6,388         (4,530)           UBS Securities Autorials Ltd         Self         6,388         (732)           UBS Securities Autorials Ltd         Self         4,753         (4,630)           UBS Securities Autorials Ltd         Self         4,753         (4,630)           UBS Securities Autorials Ltd         Self         7,145         (7,22)           UBS Securities Autorials Ltd         Self         2,510         (600)           UBS Securities Autorials Ltd         Self         2,510         (600)           UBS Securities Autorials Ltd         Self         2,510         (7,22)           UBS Securities Autorials Ltd         Self         3,480         (4,581)           UBS Securities Autorials Ltd         Self         1,682         (1,582)           UBS Securitie	_	JBS Securities Australia Ltd	Seli	50,051	(5,753)	Ordina
UBS Securities Autstrale Ltd         Self         5,220         (650)           UBS Securities Autstrale Ltd         Self         7,699         (688)           UBS Securities Autstrale Ltd         Self         7,699         (690)           UBS Securities Autstrale Ltd         Self         2,175         (690)           UBS Securities Autstrale Ltd         Self         2,610         (690)           UBS Securities Autstrale Ltd         Self         3,659         (7,232)           UBS Securities Autstrale Ltd         Self         3,659         (7,232)           UBS Securities Autstrale Ltd         Self         3,659         (7,232)           UBS Securities Autstrale Ltd         Self         2,610         (300)           UBS Securities Autstrale Ltd         Self         5,639         (4,480)           UBS Securities Autstrale Ltd         Self         5,639         (4,480)           UBS Securities Aut		JBS Securities Australia Ltd	Self	478	(52)	Ordina
UBS Securities Autstrale Ltd         Sell         6,0333         (4,636)           UBS Securities Autstrale Ltd         Sell         6,755         (4,636)           UBS Securities Autstrale Ltd         Sell         2,510         (5,00)           UBS Securities Autstrale Ltd         Sell         3,510         (1,000)           UBS Securities Autstrale Ltd         Sell         3,610         (1,000)           UBS Securities Autstrale Ltd         Sell         2,510         (2,00)           UBS Securities Autstrale Ltd         Sell         2,510         (2,00)           UBS Securities Autstrale Ltd         Sell         2,510         (2,00)           UBS Securities Autstrale Ltd         Sell         1,679         (1,000)           UBS Securities Autstrale Ltd         Sell         1,679         (1,000)           UBS Securities Autstrale Ltd         Sell         1,000         (1,000)           UBS Securities Autstrale Ltd         Sell         2,00         (1,000)           UB		JBS Securities Australia Ltd	Sell	5 220	(600)	Croins
UBS Securities Australia Ltd         Sell         7,659         (869)           UBS Securities Australia Ltd         Sell         2,732         (820)           UBS Securities Australia Ltd         Sell         2,610         (300)           UBS Securities Australia Ltd         Sell         2,610         (300)           UBS Securities Australia Ltd         Sell         2,610         (300)           UBS Securities Australia Ltd         Sell         3,149         (1,000)           UBS Securities Australia Ltd         Sell         3,149         (1,000)           UBS Securities Australia Ltd         Sell         3,149         (3,23)           UBS Securities Australia Ltd         Sell         3,149         (3,00)           UBS Securities Australia Ltd         Sell         8,00         (1,059)           UBS Securities Australia Ltd         Sell         8,00         (1,050)           UBS Securities Australia Ltd         Sell         8,00         (1,050)           UBS Securities A	Ī	JBS Securities Australia Ltd	Sell	255 OF	(959 (/)	, de
UBS Securities Australia Ltd         Sell         6,368         7,75         7,32           UBS Securities Australia Ltd         Sell         2,610         (1,000)           UBS Securities Australia Ltd         Sell         2,610         (1,000)           UBS Securities Australia Ltd         Sell         3,78         (2,520)           UBS Securities Australia Ltd         Sell         2,610         (1,000)           UBS Securities Australia Ltd         Sell         2,610         (3,000)           UBS Securities Australia Ltd         Sell         2,610         (3,000)           UBS Securities Australia Ltd         Sell         3,480         (4,000)           UBS Securities Australia Ltd         Sell         3,480         (4,000)           UBS Securities Australia Ltd         Sell         3,98         (4,000)           UBS Securities Australia Ltd         Sell         3,98         (4,000)           UBS Securities Australia Ltd         Sell         3,98         (4,000)           UBS Securities Australia Ltd         Sell         5,93         (1,059)           UBS Securities Australia Ltd         Sell         5,93         (1,059)           UBS Securities Australia Ltd         Sell         Sell         2,51         (1,050)<	Ī	JBS Securities Australia 11d	Sell	7,690	(200)	5 6
USE Securities Australia Ltd         Sell         2,175         (7.25)           USE Securities Australia Ltd         Sell         2,175         (7.25)           USE Securities Australia Ltd         Sell         2,175         (7.25)           USE Securities Australia Ltd         Sell         2,610         (3.00)           USE Securities Australia Ltd         Sell         2,610         (3.00)           USE Securities Australia Ltd         Sell         2,610         (3.00)           USE Securities Australia Ltd         Sell         3,193         (3.00)           USE Securities Australia Ltd         Sell         3,193         (3.00)           USE Securities Australia Ltd         Sell         1,679         (2.29)           USE Securities Australia Ltd         Sell         1,679         (3.00)           USE Securities Australia Ltd         Sell         1,679         (3.00)           USE Securities Australia Ltd         Sell         1,679         (1,29)           USE Securities Australia Ltd         Sell         5,933         (4,62)           USE Securities Australia Ltd         Sell         5,939         (1,42)           USE Securities Australia Ltd         Sell         5,939         (1,42)           USE Securitie	-	BS Securities Australia Ital	Sell	0303	(000)	
USS Securities Autarials Ltd         Sell         4,283         (4990)           UBS Securities Autarials Ltd         Sell         8,700         (1,000)           UBS Securities Autarials Ltd         Sell         8,700         (1,000)           UBS Securities Autarials Ltd         Sell         2,718         (2,530)           UBS Securities Autarials Ltd         Sell         2,718         (2,530)           UBS Securities Autarials Ltd         Sell         2,718         (2,530)           UBS Securities Autarials Ltd         Sell         2,610         3,000           UBS Securities Autarials Ltd         Sell         2,610         3,000           UBS Securities Autarials Ltd         Sell         2,610         3,000           UBS Securities Autarials Ltd         Sell         1,679         (1,090)           UBS Securities Autarials Ltd         Sell         1,679         (1,090)           UBS Securities Autarials Ltd         Sell         1,679         (1,090)           UBS Securities Autarials Ltd         Sell         5,533         (6,690)           UBS Securities Autarials Ltd         Sell         5,533         (6,690)           UBS Securities Autarials Ltd         Sell         5,533         (6,690)           UBS S	T	BS Sociution Australia Ital	Call	0,000	(725)	5 6
USS Securities Australia Ltd         Sell         4,729         (4,00)           UBS Securities Australia Ltd         Sell         3,055         (1,72)           UBS Securities Australia Ltd         Sell         3,055         (1,72)           UBS Securities Australia Ltd         Sell         2,510         (300)           UBS Securities Australia Ltd         Sell         2,511         (7,23)           UBS Securities Australia Ltd         Sell         2,610         (300)           UBS Securities Australia Ltd         Sell         2,610         (300)           UBS Securities Australia Ltd         Sell         1,679         (1,059)           UBS Securities Australia Ltd         Sell         5,59         (1,059)           UBS Securities Australia Ltd         Sell         5,939         (1,42)           UBS Securities Australia Ltd         Sell         5,939         (1,42)           UBS Securit	Ť	ADC Contraction Australia 14d	Coll	571,2	(nc7)	Orgina
UBS Securities Australia Ltd         Sell         8,700         (300)           UBS Securities Australia Ltd         Sell         8,700         (1,000)           UBS Securities Australia Ltd         Sell         2,510         (3,234)           UBS Securities Australia Ltd         Sell         2,510         (300)           UBS Securities Australia Ltd         Sell         2,510         (300)           UBS Securities Australia Ltd         Sell         2,610         (300)           UBS Securities Australia Ltd         Sell         2,610         (300)           UBS Securities Australia Ltd         Sell         2,69         (2,584)           UBS Securities Australia Ltd         Sell         1,705         (1,99)           UBS Securities Australia Ltd         Sell         1,705         (1,99)           UBS Securities Australia Ltd         Sell         5,533         (1,500)           UBS Securities	Ť	on permittee Australia cito		4,263	(490)	Ordina
UBS Securities Australia Ltd         Sell         8,700         (1,000)           UBS Securities Australia Ltd         Sell         1,496         (1,12)           UBS Securities Australia Ltd         Sell         2,511         (5,553)           UBS Securities Australia Ltd         Sell         2,610         (300)           UBS Securities Australia Ltd         Sell         2,610         (400)           UBS Securities Australia Ltd         Sell         2,610         (400)           UBS Securities Australia Ltd         Sell         1,673         (4,02)           UBS Securities Australia Ltd         Sell         1,673         (1,03)           UBS Securities Australia Ltd         Sell         2,939         (1,05)           UBS Securities Australia Ltd         Sell         5,933         (1,62)           UBS Securities Australia Ltd         Sell         5,939         (1,42)           UBS Securities Australia Ltd         Sell         5,939         (1,42)           UBS Securities	1	is securities Australia Ltd	Sell	2,610	(300)	Ordina
UBS Securities Autorabile Ltd         Sell         3,659         (3,524)           UBS Securities Autorabile Ltd         Sell         1,456         (1,732)           UBS Securities Autorabile Ltd         Sell         2,211         (2,533)           UBS Securities Autorabile Ltd         Sell         3,440         (400)           UBS Securities Autorabile Ltd         Sell         3,440         (2,580)           UBS Securities Autorabile Ltd         Sell         3,93         (367)           UBS Securities Autorabile Ltd         Sell         2,60         (37)           UBS Securities Autorabile Ltd         Sell         1,705         (1,99)           UBS Securities Autorabile Ltd         Sell         1,705         (1,99)           UBS Securities Autorabile Ltd         Sell         1,705         (1,99)           UBS Securities Autorabile Ltd         Sell         2,205         (1,058)           UBS Securities Autorabile Ltd         Sell         5,209         (3,400)           UBS Securities Autorabile Ltd         Sell         5,209         (1,402)           UBS Securities Autorabile Ltd         Sell         5,213         (1,502)           UBS Securities Autorabile Ltd         Sell         5,213         (1,502)		IBS Securities Australia Ltd	Sell	8,700	(1,000)	Ordina
UBS Securities Autarials Ltd   Self   1,456 (172)   1,45	-	IBS Securities Australia Ltd	Sell	30,659	(3,524)	Ordina
UBS Securities Autarials Ltd         Self         27,185         (2,439)           UBS Securities Autarials Ltd         Self         2,511         (2,539)           UBS Securities Autarials Ltd         Self         3,449         (400)           UBS Securities Autarials Ltd         Self         3,449         (400)           UBS Securities Autarials Ltd         Self         1,705         (2,284)           UBS Securities Autarials Ltd         Self         1,705         (1,99)           UBS Securities Autarials Ltd         Self         5,533         (6,82)           UBS Securities Autarials Ltd         Self         2,610         (1,00)           UBS Securit		IBS Securities Australia Ltd	Sell	1,496	(172)	Ordinary
UBS Securities Australia Ltd         Sell         22,211         C5539           UBS Securities Australia Ltd         Sell         3,493         (400)           UBS Securities Australia Ltd         Sell         3,493         (400)           UBS Securities Australia Ltd         Sell         2,61         (200)           UBS Securities Australia Ltd         Sell         1,679         (23)           UBS Securities Australia Ltd         Sell         1,679         (1,99)           UBS Securities Australia Ltd         Sell         1,679         (1,99)           UBS Securities Australia Ltd         Sell         2,905         (1,08)           UBS Securities Australia Ltd         Sell         2,905         (1,08)           UBS Securities Australia Ltd         Sell         2,905         (1,08)           UBS Securities Australia Ltd         Sell         5,933         (1,08)           UBS Securities Australia Ltd         Sell         5,933         (1,08)           UBS Securities Australia Ltd         Sell         5,933         (1,08)           UBS Securities Australia Ltd         Sell         2,610         (1,09)           UBS Securities Australia Ltd         Sell         2,610         (1,09)           UBS Securities Aust		BS Securities Australia 11d	Seil	21 185	(36) (7)	Croling
UBS Securities Australia Ltd         Sell         2,610         (400)           UBS Securities Australia Ltd         Sell         3,480         (400)           UBS Securities Australia Ltd         Sell         3,480         (400)           UBS Securities Australia Ltd         Sell         1,705         (23)           UBS Securities Australia Ltd         Sell         1,705         (199)           UBS Securities Australia Ltd         Sell         1,705         (199)           UBS Securities Australia Ltd         Sell         1,705         (1,59)           UBS Securities Australia Ltd         Sell         2,905         (1,59)           UBS Securities Australia Ltd         Sell         2,905         (1,59)           UBS Securities Australia Ltd         Sell         2,905         (1,59)           UBS Securities Australia Ltd         Sell         5,533         (6,62)           UBS Securities Australia Ltd         Sell         5,533         (6,62)           UBS Securities Australia Ltd         Sell         5,533         (6,62)           UBS Securities Australia Ltd         Sell         2,533         (6,62)           UBS Securities Australia Ltd         Sell         2,610         (1,000)           UBS Securities Aust		BS Securities Australia 11d	9	25,100	(004,2)	Clarina
USS Securities Australia Ltd         Sell         3,480         (400)           USS Securities Australia Ltd         Sell         3,480         (400)           USS Securities Australia Ltd         Sell         26         (23)           USS Securities Australia Ltd         Sell         1,705         (1,98)           USS Securities Australia Ltd         Sell         1,705         (1,98)           USS Securities Australia Ltd         Sell         6,780         (1,705)           USS Securities Australia Ltd         Sell         2,205         (1,058)           USS Securities Australia Ltd         Sell         2,205         (1,058)           USS Securities Australia Ltd         Sell         2,205         (1,058)           USS Securities Australia Ltd         Sell         5,233         (1,058)           USS Securities Australia Ltd         Sell         5,233         (1,058)           USS Securities Australia Ltd         Sell         5,233         (1,058)           USS Securities Australia Ltd         Sell         2,210         (1,020)           USS Securities Australia Ltd         Sell         2,210         (1,020)           USS Securities Australia Ltd         Sell         2,510         (1,020)           USS Securi		BS Societies Australia 14d	Cell	112,22	(000,	500
USS Securities Autorials Ltd   Sell   31439 (600)   (201)	ı	DD Counties Australia 144	1135	2,810	(300)	Ordinary
UBS Securities Australia Ltd         Sell         19,871         (3.6)           UBS Securities Australia Ltd         Sell         1,679         (2.34)           UBS Securities Australia Ltd         Sell         1,679         (3.24)           UBS Securities Australia Ltd         Sell         1,705         (1,931)           UBS Securities Australia Ltd         Sell         1,705         (1,658)           UBS Securities Australia Ltd         Sell         2,205         (1,531)           UBS Securities Australia Ltd         Sell         1,705         (1,658)           UBS Securities Australia Ltd         Sell         1,205         (1,558)           UBS Securities Australia Ltd         Sell         5,533         (6,820)           UBS Securities Australia Ltd         Sell         5,533         (6,820)           UBS Securities Australia Ltd         Sell         5,533         (6,820)           UBS Securities Australia Ltd         Sell         2,610         (7,02)           UBS Securities Australia Ltd         Sell         2,610         (3,00)           UBS Securities Australia Ltd         Sell         2,610         (3,00)           UBS Securities Australia Ltd         Sell         2,610         (3,00)           UBS S	ı	ios securities Australia Ltd	- N	3,480	(400)	Ordinary
UBS Securities Australia Ltd         Sell         19871         (2.84)           UBS Securities Australia Ltd         Sell         1,679         (1.93)           UBS Securities Australia Ltd         Sell         1,679         (1.93)           UBS Securities Australia Ltd         Sell         1,679         (1.93)           UBS Securities Australia Ltd         Sell         3,205         (1.93)           UBS Securities Australia Ltd         Sell         3,205         (1.93)           UBS Securities Australia Ltd         Sell         2,209         (1.93)           UBS Securities Australia Ltd         Sell         2,209         (1.050)           UBS Securities Australia Ltd         Sell         5,533         (1.620)           UBS Securities Australia Ltd         Sell         5,533         (1.620)           UBS Securities Australia Ltd         Sell         5,533         (1.620)           UBS Securities Australia Ltd         Sell         2,543         (1.620)           UBS Securities Australia Ltd         Sell         2,610         (1.200)           UBS Securities Australia Ltd         Sell         2,610         (1.200)           UBS Securities Australia Ltd         Sell         2,610         (1.620)           UBS S	ı	Bs securities Australia Ltd	Nell Nell	3,193	(367)	Ordinary
UBS Securities Autarials Ltd         Self         26         (3)           UBS Securities Autarials Ltd         Self         1,679         (199)           UBS Securities Autarials Ltd         Self         1,775         (199)           UBS Securities Autarials Ltd         Self         4,770         (1,99)           UBS Securities Autarials Ltd         Self         4,780         (1,595)           UBS Securities Autarials Ltd         Self         2,939         (1,698)           UBS Securities Autarials Ltd         Self         5,533         (6,682)           UBS Securities Autarials Ltd         Self         5,533         (6,820)           UBS Securities Autarials Ltd         Self         5,533         (6,820)           UBS Securities Autarials Ltd         Self         5,533         (6,820)           UBS Securities Autarials Ltd         Self         2,610         (7,20)           UBS Securities		los securities Australia Ltd	Ne l	19,871	(2,284)	Ordinary
UBS Securities Autstralia Ltd         Sell         1,209         (2.9)           UBS Securities Autstralia Ltd         Sell         1,705         (193)           UBS Securities Autstralia Ltd         Sell         6,700         (1,050)           UBS Securities Autstralia Ltd         Sell         2,205         (1,050)           UBS Securities Autstralia Ltd         Sell         2,593         (6,448)           UBS Securities Autstralia Ltd         Sell         5,933         (6,50)           UBS Securities Autstralia Ltd         Sell         5,933         (6,50)           UBS Securities Autstralia Ltd         Sell         5,933         (6,30)           UBS Securities Autstralia Ltd         Sell         5,239         (1,420)           UBS Securities Autstralia Ltd         Sell         2,510         (1,050)           UBS Securities Autstralia Ltd         Sell         2,510         (1,050)           UBS Securities Autstralia Ltd         Sell         2,610         (1,050)	-	BS Securities Australia Ltd	Sell	26	(3)	Ordinary
UBS Securities Autarials Ltd         Stell         1,679         (199)           UBS Securities Autarials Ltd         Stell         1,679         (199)           UBS Securities Autarials Ltd         Stell         46,780         (1,537)           UBS Securities Autarials Ltd         Stell         9,205         (1,058)           UBS Securities Autarials Ltd         Stell         2,998         (1,489)           UBS Securities Autarials Ltd         Stell         5,533         (6,627)           UBS Securities Autarials Ltd         Stell         5,533         (6,627)           UBS Securities Autarials Ltd         Stell         5,533         (6,627)           UBS Securities Autarials Ltd         Stell         2,533         (6,627)           UBS Securities Autarials Ltd         Stell         2,453         (7,052)           UBS Securities Autarials Ltd         Stell         2,453         (7,052)           UBS Securities Autarials Ltd         Stell         2,453         (7,052)           UBS Securities Autarials Ltd         Stell         2,610         (1,020)           UBS Securities Autarials Ltd         Stell         2,610         (1,020)           UBS Securities Autarials Ltd         Stell         2,610         (1,020) <tr< td=""><td></td><td>BS Securities Australia Ltd</td><td>Sell</td><td>500</td><td>(24)</td><td>Ordina</td></tr<>		BS Securities Australia Ltd	Sell	500	(24)	Ordina
UBS Securities Autarials Ltd   Sell   1,705   1,596   1,596   1,505   1,596   1,596   1,505   1,596	-	BS Securities Australia Ltd	Self	1,679	(193)	Ordinary
UBS Securities Australia Ltd         Sell         170S         (199           UBS Securities Australia Ltd         Sell         9,20S         (1,058)           UBS Securities Australia Ltd         Sell         29,99B         (1,058)           UBS Securities Australia Ltd         Sell         29,99B         (1,49)           UBS Securities Australia Ltd         Sell         5,533         (1,69C)           UBS Securities Australia Ltd         Sell         5,933         (1,68Z)           UBS Securities Australia Ltd         Sell         5,239         (1,42)           UBS Securities Australia Ltd         Sell         2,239         (1,42)           UBS Securities Australia Ltd         Sell         2,453         (1,42)           UBS Securities Australia Ltd         Sell         2,453         (1,202)           UBS Securities Australia Ltd         Sell         2,453         (1,202)           UBS Securities Australia Ltd         Sell         2,610         (1,202)           UBS Securities Australia Ltd         Sell         2,610         (1,202)           UBS Securities Australia Ltd         Sell         2,610         (1,202)           UBS Securities Australia Ltd         Sell         3,741         (1,300)           UBS		BS Securities Australia Ltd	Sell	818	(54)	Ordinary
UBS Securities Autarials Ltd         Sell         46/780         (6/397)           USS Securities Autarials Ltd         Sell         9.05         (1.058)           USS Securities Luttralia Ltd         Sell         2.998         (3.448)           USS Securities Autarials Ltd         Sell         5.933         (6.830)           USS Securities Autarials Ltd         Sell         5.933         (6.820)           USS Securities Autarials Ltd         Sell         5.239         (1.062)           USS Securities Autarials Ltd         Sell         5.253         (6.830)           USS Securities Autarials Ltd         Sell         2.453         (7.062)           USS Securities Autarials Ltd         Sell         2.450         (7.062)           USS Securities Autarials Ltd         Sell         2.450         (7.062)           USS Securities Autarials Ltd         Sell         2.450         (7.072)           USS Securities Autarials Ltd         Sell         2.449         (7.072)           USS Securities Autarials Ltd         Sell         2.610         (7.072)           USS Securities Autarials Ltd         Sell         2.610         (7.072)           USS Securities Autarials Ltd         Sell         3.741         (7.52) <td< td=""><td>-</td><td>BS Securities Australia Ltd</td><td>Seil</td><td>1,705</td><td>(961)</td><td>Ordinary</td></td<>	-	BS Securities Australia Ltd	Seil	1,705	(961)	Ordinary
UBS Securities Autstrale Ltd         Sell         9.05         (1,058)           UBS Securities Autstrale Ltd         Sell         2.29,938         (3,448)           UBS Securities Autstrale Ltd         Sell         1,035         (1,19)           UBS Securities Autstrale Ltd         Sell         5,933         (6,82)           UBS Securities Autstrale Ltd         Sell         5,933         (6,82)           UBS Securities Autstrale Ltd         Sell         1,235         (1,42)           UBS Securities Autstrale Ltd         Sell         2,510         (1,02)           UBS Securities Autstrale Ltd         Sell         2,610         (1,03)           UBS Securities Autstrale Ltd         Sell         3,741         (4,03)           UBS Securities Autstrale Ltd         Sell         3,741         (4,03)           UBS Secur	i	BS Securities Australia Ltd	Sell	46,780	(5,377)	Ordinary
USS Securities Autarials Ltd   Sell	BS Securities Australia Ltd	Sell	9,205	(1.058)	Ordinary	
URS Securities Australia Ltd         Sell         29,998         (2,448)           UBS Securities Australia Ltd         Sell         5,533         (6,30)           UBS Securities Australia Ltd         Sell         5,533         (6,82)           UBS Securities Australia Ltd         Sell         9,239         (1,062)           UBS Securities Australia Ltd         Sell         1,235         (1,022)           UBS Securities Australia Ltd         Sell         2,439         (1,022)           UBS Securities Australia Ltd         Sell         2,439         (1,020)           UBS Securities Australia Ltd         Sell         2,439         (1,020)           UBS Securities Australia Ltd         Sell         2,439         (1,020)           UBS Securities Australia Ltd         Sell         3,459         (1,570)           UBS Securities Australia Ltd         Sell         3,561         (1,573)           UBS Securities Australia Ltd         Sell         1,679         (1,830)           UBS Securities Australia Ltd         Sell         1,679         (1,830)           UBS Securities Australia Ltd         Sell         3,741         (4,300)           UBS Securities Australia Ltd         Sell         3,741         (4,300) <td< td=""><td></td><td>BS Securities Australia Ltd</td><td>Sell</td><td>313</td><td>(36)</td><td>Ordina</td></td<>		BS Securities Australia Ltd	Sell	313	(36)	Ordina
USS Securities Australia Ltd         Sell         (1.035         (1.19)           UBS Securities Australia Ltd         Sell         5.933         (6.82)           UBS Securities Australia Ltd         Sell         5.933         (6.82)           UBS Securities Australia Ltd         Sell         2.25         (1.062)           UBS Securities Australia Ltd         Sell         1.235         (1.421)           UBS Securities Australia Ltd         Sell         2.453         (1.202)           UBS Securities Australia Ltd         Sell         2.453         (1.202)           UBS Securities Australia Ltd         Sell         2.460         (1.002)           UBS Securities Australia Ltd         Sell         2.469         (1.202)           UBS Securities Australia Ltd         Sell         2.469         (1.502)           UBS Securities Australia Ltd         Sell         3.361         (1.502)           UBS Securities Australia Ltd         Sell         3.3741         (4.300)	1	BS Securities Australia Ltd	Sell	99996	(3.448)	Ordinary
UBS Securities Autorials Ltd         Sell         5,533         (630)           UBS Securities Autorials Ltd         Sell         5,933         (682)           UBS Securities Autorials Ltd         Sell         5,233         (1,020)           UBS Securities Autorials Ltd         Sell         1,235         (1,020)           UBS Securities Autorials Ltd         Sell         2,610         (300)           UBS Securities Autorials Ltd         Sell         1,679         (1,573)           UBS Securities Autorials Ltd         Sell         1,679         (1,573)           UBS Securities Autorials Ltd         Sell         1,679         (1,530)           UBS Securities Autorials Ltd         Sell         1,679         (1,530)           UBS Securities Autorials Ltd         Sell         3,741         (4,300)           UBS Securities Autorials Ltd         Sell         3,741         (4,300)           UBS Securities Autorials Ltd         Sell         2,715         (2,500)           UBS Securi		BS Securities Australia Ltd	Sell	1 035	(011)	i cuipo
Self         5.933         (682)           Self         5.935         (1.625)           Self         6.25         (1.025)           Self         1.235         (1.027)           Self         2.451         (200)           Self         2.453         (282)           Self         2.453         (1.202)           Self         4.489         (515)           Self         2.560         (300)           Self         1.679         (152)           Self         1.679         (153)           Self         3.741         (430)           Self         3.741         (430)           Self         2.560         (300)           Self         3.741         (430)           Self         3.751         (430)           Self         2.560         (300)           Self         2.775         (200)           Self         2.777         (300)           Self         2.670         (300)	Γ	BS Securities Australia Ltd	Sell	5,033	(636)	Ordinary
UBS Securities Australia Ltd         Sell         9,335         (1,062)           UBS Securities Australia Ltd         Sell         6,25         (1,22)           UBS Securities Australia Ltd         Sell         1,235         (1,42)           UBS Securities Australia Ltd         Sell         2,510         (300)           UBS Securities Australia Ltd         Sell         2,635         (1,202)           UBS Securities Australia Ltd         Sell         1,045         (1,202)           UBS Securities Australia Ltd         Sell         8,396         (5,65)           UBS Securities Australia Ltd         Sell         1,579         (1,573)           UBS Securities Australia Ltd         Sell         3,741         (430)           UBS Securities Australia Ltd         Sell         3,741         (430)           UBS Securities Australia Ltd         Sell         2,717         (250)           UBS Securities Australia Ltd         Sell         2,777         (250)           UBS Securiti		BS Securities Australia Ltd	lles	F 023	(503)	O C
Sell	1	RS Securities Australia Itol	ii d	0000	(200)	Couriery
UBS Securities Australia Ltd         Sell         0.22         (1.23)           UBS Securities Australia Ltd         Sell         2.453         (1.20)           UBS Securities Australia Ltd         Sell         2.453         (1.20)           UBS Securities Australia Ltd         Sell         2.453         (1.202)           UBS Securities Australia Ltd         Sell         2.610         (1.202)           UBS Securities Australia Ltd         Sell         2.610         (300)           UBS Securities Australia Ltd         Sell         2.610         (300)           UBS Securities Australia Ltd         Sell         1.673         (1.522)           UBS Securities Australia Ltd         Sell         3.741         (4.30)           UBS Securities Australia Ltd         Sell         3.741         (4.30)           UBS Securities Australia Ltd         Sell         2.751         (6.50)           UBS Securities Australia Ltd         Sell         2.751         (6.50)           UBS Securi	, =	DC Contribute Australia Hall	100	65,6	(200,1)	Ordinary
URS Securities Australia Ltd         Sell         2,453         (1,42)           URS Securities Australia Ltd         Sell         2,453         (282)           URS Securities Australia Ltd         Sell         1,0,457         (1,002)           URS Securities Australia Ltd         Sell         4,889         (5,16)           URS Securities Australia Ltd         Sell         2,610         (300)           URS Securities Australia Ltd         Sell         8,396         (6,573)           URS Securities Australia Ltd         Sell         1,578         (1,373)           URS Securities Australia Ltd         Sell         1,679         (1,330)           URS Securities Australia Ltd         Sell         3,741         (4,300)           URS Securities Australia Ltd         Sell         3,741         (4,300)           URS Securities Australia Ltd         Sell         3,741         (4,300)           URS Securities Australia Ltd         Sell         2,775         (250)           URS Securities Australia Ltd         Sell         2,775         (250)           URS Securities Australia Ltd         Sell         2,775         (250)		Di Secondo Adolonio Lid	===	070	(77)	Ordinar
UBS Securities Australia Ltd         Seal         2,610         (300)           UBS Securities Australia Ltd         Seal         1,659         (1,202)           UBS Securities Australia Ltd         Seal         1,0457         (1,202)           UBS Securities Australia Ltd         Seal         3,610         (300)           UBS Securities Australia Ltd         Seal         3,561         (300)           UBS Securities Australia Ltd         Seal         3,56         (1,522)           UBS Securities Australia Ltd         Seal         1,679         (1,522)           UBS Securities Australia Ltd         Seal         1,679         (1,522)           UBS Securities Australia Ltd         Seal         3,741         (430)           UBS Securities Australia Ltd         Seal         2,510         (300)	Ī	Do seculties Australia Ltd	) act	1,235	(142)	Ordinar
UBS Securities Australia Ltd         Sell         4.953         (282)           UBS Securities Australia Ltd         Sell         1.0457         (1.202)           UBS Securities Australia Ltd         Sell         4.489         (516)           UBS Securities Australia Ltd         Sell         8.360         (550)           UBS Securities Australia Ltd         Sell         13.665         (1,573)           UBS Securities Australia Ltd         Sell         1,579         (1,573)           UBS Securities Australia Ltd         Sell         1,579         (1,630)           UBS Securities Australia Ltd         Sell         3,741         (430)           UBS Securities Australia Ltd         Sell         3,747         (430)           UBS Securities Australia Ltd         Sell         2,717         (250)           UBS Securities Australia Ltd         Sell         2,717         (250)           UBS Securities Australia Ltd         Sell         2,717         (250)	1	BS Securities Australia Ltd	)eii	2,610	(300)	Ordinar
Sell	٥	BS Securities Australia Ltd	Sell	2,453	(282)	Ordina
UBS Securities Australia Ltd         Sell         4.489         (516)           UBS Securities Australia Ltd         Sell         2,010         (300)           UBS Securities Australia Ltd         Sell         (1,573)         (1,573)           UBS Securities Australia Ltd         Sell         (1,573)         (1,573)           UBS Securities Australia Ltd         Sell         1,573         (1,573)           UBS Securities Australia Ltd         Sell         3,741         (430)           UBS Securities Australia Ltd         Sell         3,741         (430)           UBS Securities Australia Ltd         Sell         2,717         (250)           UBS Securities Australia Ltd         Sell         2,717         (250)           UBS Securities Australia Ltd         Sell         2,717         (250)	2	BS Securities Australia Ltd	Sell	10,457	(1,202)	Ordinar
Sell         2,610         (500)           Sell         8,396         (465)           Sell         13,685         (153)           Sell         1,679         (193)           Sell         1,679         (193)           Sell         3,741         (430)           Sell         2,175         (550)           Sell         2,510         (300)	2	BS Securities Australia Ltd	Sell	4 489	(516)	Ordina
Sell   8,396 (1573)     Sell   13,685 (1573)     Sell   17 (23)     Sell   3,741 (430)     Sell   2,175 (230)     Sell   2,0510 (300)	ľ	BS Securities Australia Ltd	Sell	2,510	(300)	
Self	1	RS Securities Australia Itel	===	302.8	(000)	0
Sell 1,679 (1,573)     Sell 1,679 (1,573)     Sell 2,741 (430)     Sell 8,335 (958)     Sell 2,775 (250)     Sell 2,775 (250)	1	Social Control of the	1100	akt'o	(cqs)	Color
Sell   1,679 (193)     Sell   1,679 (193)     Sell   3,741 (430)     Sell   2,175 (250)     Sell   2,610 (300)	2	65 Securities Australia Ltd	Seil	13,685	(1,573)	Ordina
5e    3,741 (430)   5e    6,335 (958)   5e    2,175 (250)   5e    2,500 (300)   3e	2	BS Securities Australia Ltd	Sell	1,679	(193)	Ordina
Sell         3,741         (430)           Sell         8,335         (958)           Sell         2,775         (250)           Sell         2,610         (50)	₽	BS Securities Australia Ltd	Sell	17	(2)	Ordina
UBS Securities Australia Ltd         Sell         8,335         (958)           UBS Securities Australia Ltd         Sell         2,175         (250)           UBS Securities Australia Ltd         Sell         2,175         (350)	Ī	BS Securities Australia Ltd	Sell	3 741	(08//)	Color
Sell 2,775 (250)   Sell 2,610 (300)	f	RS Securities Australia 14d	i j	362.0	(054)	
UBS Securities Australia td Sell 2,610 (300)	T	po securities Australia Lid	100	8,530	(858)	Crdina
Sell 2,610 (300)	١	bs securities Australia Ltd	Sell	2,175	(250)	Ordina
	2	85 Securities Australia Ltd	Sell	2,610	(300)	Ordinar
Sell (46)	-	BS Securities Australia Ltd	Sell	400	(46)	Ordinary
440 (45)	Ť	בא אחזים שחזים שונים	201	U01	(44p)	CIGINA
(1.784)				10001	(4 DOA)	3

2006.11   2018 Securities Australia It of 2006.13   2018	974 974 3 454 121 171 171 171 172 2,595 4,437 8,762 2,695 1,546 1,546 1,720 1,720 1,546 1,720 1,720 2,167 2,168 2,	(112) Ordensy (192) Ordensy (170) Ordensy (170) Ordensy (170) Ordensy (170) Ordensy (180) Ordensy (1	Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities A	974  3 454  1171  1171  1171  1174  2 556  4 477  2 556  1 1,557  7 2 72  7 2 72  7 2 72  7 2 72  7 2 72  7 3 72  8 766  1 1,730  1 1,730  1 1,730  2 16  2 16  2 16  2 16  2 16  2 16  2 16  2 16  2 16  2 16  2 16  3 100  3 100  5 100		Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
185 Securites Australia Ltd 186 Securites Australia Ltd 18	1,17.7 1,471 1,471 1,471 1,471 1,527 1,527 1,527 1,727 1,727 1,727 1,727 1,727 1,720 1,720 1,720 1,720 1,720 1,720 1,720 1,720 1,720 1,720 2,119 2,100 2,119 2,100 2,119 2,100 2,119 2,100 2		Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
185 Securities Australia Ltd 185 Securities A	1,471 1,471 1,471 1,471 1,472 2,596 4,473 1,527		Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	1,447 1,447 2,556 2,556 1,557 2,556 1,436 8,762 8,762 4,437 2,26 4,437 2,26 1,346 1,346 1,346 2,166 2,1		Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	8,140 8,170 2,595 2,595 2,163 2,163 2,163 8,762 8,762 8,762 4,437 4,437 4,437 4,437 1,456 8,762 4,437 1,456 8,762 3,366 1,566 1,566 1,566 2,16 2		Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	2.595 4.477 2.596 4.30 1.557 7.27 7.27 7.27 7.27 7.27 7.27 7.27		Odinary Odinary Ordinary
USS Securities Australia Ltd USS Securities A	2.595 4.295 4.305 2.163 2.163 8.762 8.762 8.762 8.763 8.303 3.30 1.703 2.103 2.103 2.105 2.306 8.303 2.306 8.303 2.306 8.303 2.306 8.303 8		Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
185 Securites Australia Ltd 185 Securites Australia Ltd 185 Securites Australia Ltd 185 Securites Australia Ltd 185 Securites Australia Ltd 185 Securites Australia Ltd 185 Securites Australia Ltd 185 Securites Australia Ltd 185 Securities Australia	1,557 1,557 1,727 2,163 1,436 1,436 1,236 1,236 1,236 1,236 1,730 1,730 1,730 1,730 1,730 1,730 1,730 1,730 1,730 1,730 1,730 1,730 1,730 1,730 1,730 1,730 2,16		Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securines Australia Ltd UBS Securines Australia Ltd	1,557 212 2162 2162 2162 8,762 4386 4386 4386 346 346 346 346 2,566 1,346 2,033 2,19 2,19 2,19 2,19 2,19 2,19 2,19 2,19		Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Itid UBS Securities Itid UBS Securities Australia Itid UBS Securities Australia Itid UBS Securities Australia Itid UBS Securities Australia Itid	2.162 8.762 8.762 4.437 4.437 4.438 3.46 3.43 3.46 3.43 1.566 2.666 1.730 2.666 2.66		Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	2.163 1.456 8.762 8.762 9.364 9.366 9.366 9.367 9.368 9.3720 9.312 9.313 9.329 9.314 9.326 9.338		Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd  UBS Securities Australia Ltd	8.762 8.762 4.437 4.386 3.36 3.36 1.587 1.587 1.739 2.16		Ordinary Ordinary Ordinary Ordinary
UBS Securite Australia Ltd  UBS Securite Australia Ltd	4.872 4.872 4.386 3.46 3.48 3.48 3.48 3.48 5.686 5.686 5.686 5.687 2.033 2.16 2.16 2.16 2.16 2.16 2.16 2.18 2.18 2.18 2.18 2.18 2.18 2.18 2.18 2.18 2.18 2.19 2.19 2.19 2.19 2.19 2.19 2.19 2.19 2.19 2.19 2.19 2.19 2.19 2.10		Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia tod UBS Securities Australia tod	4437 28 28 44,386 4,386 346 346 347 1,346 1,346 2,033 2,16 2,16 2,16 2,19 2,19 2,19 2,19 2,19 2,19 2,19 2,19		Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	4,386 4,386 340 1,687 1,580 1,730 1,730 1,730 2,165 2,165 2,166 2,166 2,166 2,166 2,166 2,166 2,166 2,166 2,166 2,166 2,166 2,166 2,166 2,166 2,166 2,166 2,166 2,166 2,173 2,170 2,170 2,18		Ordinary Ordinary
UBS Securities Australia tod UBS Securities Australia tod	346 303 303 1687 5666 1,246 2,033 2,033 2,19 2,19 2,19 2,19 2,19 2,19 2,19 2,19		Ordinary
UBS Securities Australia trd UBS Securities Australia Ltd	346 303 1,087 1,087 1,087 1,046 2,068 2,168 2,16 2,		
UBS Securities Australia Itd UBS Securities Australia Ltd	303 1,5687 5,668 1,730 1,730 2,033 2,16 2,16 2,1		Ordinary
188 Securities Australia Ltd 188 Securities Australia Ltd 185 Securities Australia Ltd 186 Securities Australia Ltd 187 Securities Australia Ltd 187 Securities Australia Ltd 187 Securities Australia Ltd 187 Securities Australia Ltd 187 Securities Australia Ltd 187 Securities Australia Ltd 187 Securities Australia Ltd	5.666 1,786 1,786 2,033 2,16 2,16 2,19 3,720 2,19 2,19 6,335 7,630 5,700 5,700 5,700 10,64 9,514 6,9514		Ordinary
UBS Securities Australia trd UBS Securities Australia trd	5,666 1,730 1,730 2,033 2,16 2,16 2,173 3,720 2,365 2,366 51,880 61,335 76,305 77,307 310,646 39,514 39,514		Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	1,730 1,546 2,033 2,16 216 1,773 3,720 2,19 2,19 2,19 2,19 2,19 2,19 2,19 2,19		Ordinary
1.085 Securities Australia Ltd 1.085 Securities Australia Ltd	1,946 2,033 216 216 217 217 217 2,119 2,11		Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	2.033 216 216 2173 1,720 2,720 2,360 51,880 47,335 76,305 77,007 310,646 9,514 89,514		Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	216 216 1,773 1,720 2,119 2,366 2,366 2,366 15,880 6,335 7,6305 7,6305 310,646 9,514 9,514		Ardinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	216 1773 3720 3720 2,119 2,119 2,103 2,365 51,880 47,335 76,305 77,007 310,546 9,514		Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	1,733 1,720 2,119 3,633 1,863 1,864 1,864 1,315 7,6305 1,310,646 1,310,646 1,310,646 1,310,646		Adinary Adinary
UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS AGL, London Banch UBS AGL, London Banch UBS AGL, London Banch UBS AGL, London Banch UBS AGL, London Banch UBS AGL, London Banch UBS AGL, London Banch UBS AGL, London Banch UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd	3,700 2,119 2,119 3,603 5,363 5,189 47,335 76,305 76,305 76,305 77,007 310,646 9,551 8,951		- Judipari
UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Gobal Asser Management UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd	2,11.2 2,11.2 2,163 2,365 51,880 47,335 76,305 77,007 310,646 9,514 9,514		Ordinary
10.95 Securine Australia too 10.95 Securine Australia too 10.95 Securine Australia too 10.95 Securine Australia too 10.95 Securine Australia too 10.95 Securine Australia too 10.95 Securine Australia too 10.95 Securine Australia too 10.95 Securine Australia too 10.95 Securine Australia too 10.95 Securine Australia too 10.95 Securine Australia too 10.95 Securine Australia too 10.95 Securine Banch 10.95 AG, London Banch 10.95 AG, London Banch 10.95 AG, London Banch 10.95 AG, London Banch 10.95 AG, London Banch 10.95 AG, London Banch 10.95 AG, London Banch 10.95 AG, London Banch 10.95 AG, London Banch 10.95 AG, London Banch 10.95 AG, London Banch 10.95 AG, London Banch 10.95 AG, London Banch 10.95 AG, London Banch 10.95 AG, London Banch 10.95 AG, London Banch 10.95 Securine Australia too	2,113 3,603 2,366 2,366 47,335 77,007 310,646 9,514 9,514	l	Juli lai y
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	2,366 2,366 2,366 47,335 76,305 77,007 310,646		Juliary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS AG, London Blanch UBS AG, London Blanch UBS AG, London Blanch UBS AG, London Blanch UBS AG, London Blanch UBS AG, London Blanch UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	51,880 51,880 57,007 57,007 59,514 59,514		Jrdinary
UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS AG, London Banch  UBS Securities Australia Ltd	47,335 47,335 76,305 77,007 310,646 55,514		Ordinary
UBS Security Australia Ltd  UBS Security Australia Ltd  UBS Security Australia Ltd  UBS Security Australia Ltd  UBS Security Australia Ltd  UBS Security Australia Ltd  UBS Security Australia Ltd  UBS Security Australia Ltd  UBS AG, London Banch  UBS AG, Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd	47,335 76,305 57,007 310,646 59,514		Ordinary
UBS Securites Australia Ltd  UBS Securites Australia Ltd  UBS Securites Australia Ltd  UBS Securites Australia Ltd  UBS Securites Australia Ltd  UBS Securites Australia Ltd  UBS Securites Australia Ltd  UBS Securites Australia Ltd  UBS AG, London Banch  UBS Securites Australia Ltd	76,305 57,007 310,646 59,514		Ordinary
UBS Security Australia tod UBS Security Australia tod UBS Security Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod	57,007 310,646 59,514		Ordinary
UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Darich UBS AG, Lordon Barch UBS Securities Barch UBS Securities Lord UBS Securities Australia tod	310,646		Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS AGI, London Branch UBS AGI, London Branch UBS AGI, London Branch UBS AGI, London Branch UBS AGI, London Branch UBS AGI, London Branch UBS AGI, London Branch UBS AGI, London Branch UBS AGI, London Branch UBS AGI, London Branch UBS AGI, London Branch UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	59,514		Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd			Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS AG, London Blanch UBS AG, London Blanch UBS AG, London Blanch UBS AG, London Blanch UBS AG, London Blanch UBS AG, London Blanch UBS AG, London Blanch UBS AG, London Blanch UBS AG, London Blanch UBS Securities Blanch UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	127,584		Ordinary
UBS Securities Australia Ltd  UBS AG, London Branch  UBS AG, Australia Branch  UBS AG Australia Branch  UBS Securities Australia Ltd	186,071		Ordinary
UBS AG, London Blanch UBS AG, London Blanch UBS AG, London Blanch UBS AG, London Blanch UBS AG, London Blanch UBS AG, Australia Blanch UBS AG, Australia Blanch UBS Sweathn Management Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	163,433		Ordinary
UBS AG, Lordon Branch UBS AG, Lordon Branch UBS AG, Lordon Branch UBS AG, Lordon Branch UBS AG Australie Branch UBS AG Australie Branch UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	70,628		Ordinary
UBS AG, Lordon Branch UBS AG, Lordon Branch UBS AG, Lordon Branch UBS AG, Australia Branch UBS AG, Australia Branch UBS AG, Australia Branch UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	57,007		Ordinary
UBS AG, London Branch UBS AG, Austrialia Branch UBS AG Austrialia Branch UBS AG Austrialia Branch UBS Vesaltri Management Austrialia Ltd UBS Securities Austrialia Ltd UBS Securities Austrialia Ltd UBS Securities Austrialia Ltd UBS Securities Austrialia Ltd UBS Securities Austrialia Ltd UBS Securities Austrialia Ltd UBS Securities Austrialia Ltd UBS Securities Austrialia Ltd UBS Securities Austrialia Ltd UBS Securities Austrialia Ltd UBS Securities Austrialia Ltd UBS Securities Austrialia Ltd	94,431		Ordinary
UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	76 305		Indinan
UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	170,667	(19 704)	Ordinary
UBS Vecative Management Australia Ltd UBS Vecative Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	218 318	l	Ordinary
USS Sweath Management Australia Ltd USS Sweath Management Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd	0.00.2		Countries
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	0/1/916	1	Juliary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	76,222	8,863 O	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd			
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	N/A		Options
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	170,000		Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd		IO 665,11	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd			Ordinary
UBS Securites Australia 11d UBS Securites Australia 11d UBS Securites Australia 11d	100.920	(11,600)	Ordinary
UBS Securities Australia 1td UBS Securities Australia 1td UBS Securities Australia 1td			
UBS Securities Australia Ltd UBS Securities Australia Ltd	VIV	000 000 00	Justicus
UBS Securities Australia Ltd	10000		pirons
UBS Securities Australia Ltd	10//67		ramary
			Irdinary
UBS Securities Australia Ltd			rdinary
UBS Securities Australia Ltd			hdinary
I IRS Securities Australia I to			t in the second
(2-OCT-14 DBS Securities Australia Ltd Sell		1	urdinary
UBS Securities Australia Ltd			roinary
UBS Securities Australia Ltd			Ardinary
UBS Securities Australia Ltd			Irdinary
UBS Securities Australia Ltd			irdinary
UBS Securities Australia Ltd			rdinary
UBS Securities Australia Ltd			rdinary
I IRS Securities Australia (10			Indinary
A DO Commission According to the		١	dina y
	989,595	(64,812) Or	Ordinary
UBS Securities Australia Ltd		1	rdinary
UBS AG Australia Branch			rdinary
UBS AG Australia Branch			rdinary
List Wealth Management Australia 14d			
UBS Wealth Management Australia Ltd			rdinary

	30,807  400  400  400  400  8,700  8,700  9,800  8,918  8,918  8,918  8,918  8,918  8,918  8,918  1,332  1,332  1,332	(3,541) (840) (98) (300) (46) (1,000)	
	8,208 2,510 2,610 2,610 8,700	l)	
	2 863 400 8,700 8,	l)	
	8700 8700 8700 8700 9700	1)	
	8,700 8,700 2,884 1,714 1,375 1,	1)	
	2,870 2,870 2,072 2,072 3,417 3,341 3,341 3,341 3,341 3,341 3,341 3,341 3,341 3,341		
	2,2824 2,029 1,029 1,039 1		
	5,029 174 174 8,331 34,525 34,525 1,392 2,625 2,625 2,527 2,341 2,341 2,341 2,341 2,341 2,341 2,341 2,341 2,341 3,341		
	1,022 1,032 1,		Ciciliary
	8 978 8 978 331 337 8 940 8 96 8 96 8 96 8 972 2 22 2 23 2 3341 7 1.17 7 1.17 7 1.17 7 1.17 7 1.17	(5/8)	Ordinary
	8978 8978 940 940 940 1,992 1,992 2,62 2,532 1,711 7,111 7,111 7,111 7,111 7,111 7,111		Ordinary
	34752 896 34752 2.532 2.25 2.25 2.27 7.11 7.11 7.11 3.341 3.341	(1)	Ordinary
	896 84752 1,532 2,532 2,627 1,717 7,117 3,341 9		Ordinary
	34,525 34,525 1,920 2,62 2,62 2,62 2,71 7,17 7,11 7,11 8,341		
	34,724 1,1952 2,932 2,932 2,932 2,932 3,117 3,117 3,241 3,241 3,241	(103)	Ordinary
	2,932 2,932 26 26 27 17 2,117 2,347 9		
	2,932 202 17 17 1,17 1,117 3,341 3,341 3,341 3,341 3,341		
	26 26 27,117 7,117 3,341 9	(3)	Ordinary
	7,17 7,117 3,341 3,341 2,9 659		Ordinary
	3,341		Ordinary
	3,341	(8)	Ordinary
	9 29 667		Ordinary
	79 657		Ordinary
		(3,410)	Ordinary
	3,950	(454)	Ordinary
	929		Ordinary
UBS Securities Australia Ltd Sell	52	(9)	Ordinary
UBS Securities Australia Ltd	6	Đ	Ordinary
UBS Securities Australia Ltd	26	6	Ordinary
1185 Securities Australia I td	3 828	(070)	Ordinary
	900		Ordinary
	5.781	(503)	Ordinary
I IRS Securities Australia Ind	731		Ordinary
IRS Societies Aistralia 14d	24 800	27	Ordinary
	34,800	(4)	Ordinary
	226		Ordinary
	4,976		Ordinary
	5,124	(583)	Ordinary
UBS Securities Australia Ltd	505	(89)	Ordinary
	35	(4)	Ordinary
UBS Securities Australia Ltd Sell	444	(51)	Ordinary
	6		Ordinary
	1.775	0	Ordinary
	18.479		Ordinary
	6118	(202)	Ordinary
	4611		Ordinary
	200		O'Gillary O'Gillary
	000		Ordinary
	CEO		Ordinary
UBS Securities Australia Ltd	17		Ordinary
	400	(46)	Ordinary
	131		Ordinary
	148		Ordinary
	6,542	(752)	Ordinary
	452		Ordinary
UBS Securities Australia Ltd Sell	1,575		Ordinary
	2 845	(225)	Ordinary
	6.438		Ordinan
LIBC Countries Australia 144	24.2		Column y
			Ordinary
	6,682	(768)	Ordinary
	55,967	(6,433)	Ordinary
	5,351	(615)	Ordinary
UBS Securities Australia Ltd	8	(11)	Ordinary
UBS Securities Australia Ltd	444	(51)	Ordinary
	26 665	(3.065)	Ordinary
	7 543	(CDD'C)	Coding
LIBS Contribute Australia 144	C36 231	(199)	Ordinary
Securities Australia Ltd	756,457	(505,11)	1
Securities Australia Ltd	3,480	(400)	Ordinary
Securities Australia Ltd	426	(49)	
UBS Securities Australia Ltd Sell	2,236	(526)	
UBS Securities Australia Ltd	1.572	(180)	Ordinary
	7777	(540)	Ordinate
TIBC Countries Australia 144	0000	(304)	Countary
Securities Australia Ltd	875'5	(381)	Ordinary
	4,365	(200)	Ordinary
	2 160	(696)	Ordinari
ODS Securities Australia LTO	3,169	(363)	Ordinary
	22 314	(355.0)	Ordinary

Buy   1,0,0,0,0   1,0,0,0,0   1,0,0,0,0   1,0,0,0,0   1,0,0,0,0   1,0,0,0,0   1,0,0,0,0   1,0,0,0,0   1,0,0,0,0   1,0,0,0,0   1,0,0,0,0   1,0,0,	37 Ordinary	77 Ordinary	00 Ordinary	9 Ordinary	9 Ordinary	8 Ordinary				L		L		L				7 Ordinary		Ordinary		11 Ordinary							l		1			S Ordinary					0 Ordinary	ı								1) Ordinary				İ					Options			O) Ordinary				Ordinary
Buy   Buy	9	3 2,13	4 40	1,26	1,26	8								3					7																			ļ																					200					263
Titalia Lid	2,02	18,29	3,42	10,85	10,85	9	8 52	23.58	45	2,24	1,82	3,70	1,20	31,60	7,67	8,18	70,7	12,T	20,83	1,15	17,65	1,63	5,90	61,2	2,30	6.93	29,600	231,39%	39,81	37,03.	2,06	331,100	59,93	93.51	59,786	10,28	2,739	60,19	14,73	75/dC1	579.23	666,208	85,406	823,70	78,016	110,54	7.82(	5,614	296	2,76	888	17,28	18,542	704 DPC	46.830		20	N/N	1,735	17,706	340,437	131,834	126,241	1777
1955 Securities Australia Ltd 1955 S	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Bury	Suv	Buy	Buy	Buy	Buy	Riv	Buy	Buy	Buy	Sell	Buy	Sell	Buy	Sell	Soil	Buy	Sell	Buy	Sell	lles .	Buy	lle?	Buy	Sell	Sell	Buy	Suy.	lley Vel	Buy	Sell	Buy	Buy	Buy	Buy	Ruy	Buy	Buy	Enter Into Derivative	Contract	Contract	Buy	Sell	Buy	Sell	Buy	Buy
	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	LIRS Securities Australia 14d	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ind	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ind	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ind	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS AG Australia Branch	UBS AG Australia Branch	UBS Wealth Management Australia Ind	UBS Wealth Management Australia Ltd	UBS Wealth Management Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia It d	UBS Securities Australia Ltd	UBS Securities Australia Ltd	Lat Joseph A. and jobs and O. O. O.	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Global Asset Management	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	ILIRS Securities Australia Ltd

22.77	UBS Securities Australia Ltd	Sell	6	(1)	Ordinary
22-0ct-14	UBS Securities Australia Ltd	Sell	3,715	(4	Ordinary
22-Oct-14	UBS Securities Australia Ltd	Sell	113	(13)	Ordinary
22-Oct-14	UBS Securities Australia Ltd	Sell	122	(14)	Ordinary
22-Oct-14	UBS Securities Australia Ltd	Sell	113	(13)	Ordinary
22-Oct-14	UBS Securities Australia Ltd	Sell	20,884	(2,395)	Ordinary
22-Oct-14	UBS Securities Australia Ltd	Sell	54,099	(6,204)	Ordinary
22-Oct-14	UBS Securities Australia Ltd	Sell	113	(13)	Ordinary
22-0ct-14	UBS Securities Australia Ltd	Sell	21,599	(7,477)	Ordinary
22-Oct-14	UBS Securities Australia Ltd	Sell	113	(13)	Ordinary
22-Oct-14	UBS Securities Australia Ltd	Buy	36,973	4,240	Ordinary
22-Oct-14	UBS Securities Australia Ltd	Sell	37,618	(4,314)	Ordinary
22-Oct-14	UBS Securities Australia Ltd	Sell	2,433		Ordinary
22-Oct-14	UBS Securities Australia Ltd	Buy	845,970	97,800	Ordinary
23-Oct-14	UBS Global Asset Management	Buy	3,811	443	Ordinary
23-Oct-14	UBS Securities Australia Ltd	Sell	37.339	(4	Ordinary
23-Oct-14	UBS Securities Australia Ltd	Buy	60,046	ĺ	Ordinary
23-Oct-14	UBS Securities Australia Ltd	Sell	43.300	(5,000)	Ordinary
23-Oct-14	UBS Securities Australia Ltd	Sell	154 410		Ordinary
23-Oct-14	UBS Securities Australia Ltd	Buv	508.A		Ordinary
23-04-14	IIIRS Securities Australia Itd	Run	209 00		Ordinary
73,04-14	LIBS Securities Australia 144	B:n/	86,003		Ordinary
23-Oct.14	I IRC Cocurding Australia 144	Rin	200,51		Ordinary
23 Oct 14	Upo percentes Australia cia	and a	987'11;		Ordinary
20 00 14	Other Property Annual Communication of the Communic	1120	240,01		Ordinary
23-04-14	Upos pecurines Australia Lid	lac U	6/,0/2		Odinary
23-001-14	UBS Securities Australia Ltd	ouy	2/1,458		Ordinary
11 10 00	UBS Securities Australia Ltd	ing.	997'/8		Ordinary
3-001-14	Obs securities Australia LIO	Jac.	4/2,/02		Ordinary
23-Oct-14	CDS Securities Australia Ltd	Jell C-11	97,/36		Ordinary
23-046-14	Upo securities Australia cro	lac lac	/61,634		Ordinary
3-Oct-14	UBS AG AUSTRAIIA BRANCH	Jacil	1,713,169		Ordinary
3-UCI-14	UBS AG Australia Branch	Suy	246,538		Ordinary
3-00-14	UBS AG Australia Branch	Sell	385,390	(44,913)	Ordinary
3-Oct-14	UBS Securities Australia Ltd	Buy	2,410		Ordinary
3-Oct-14	UBS Securities Australia LTG	Buly	5,148	ŀ	Ordinary
3-Oct-14	UBS Securities Australia Ltd	Buy	7,855	916	Ordinary
3-Oct-14	UBS Securities Australia Ltd	Buy	14,835	1,729	Ordinary
3-00-14	Ubs securities Australia Ltd	ânâ	2,574	300	Ordinary
3-Oct-14	UBS Securities Australia Ltd	Buy	3,432	400	Ordinary
204.14	PIDE Consider America 144	Box	2,145	067	Ordinary
0.000	opp securities Australia Lto	ina	CCB'/	976	Ordinary
3-00-14 0-0-14	Obs Securities Australia Ltd	áng	/57	30	Ordinary
2-00-14	Obs securities Australia Ltd	ána	26,323	3,068	Ordinary
1	UBS Securities Australia Ltd	Buy	8,580	1,000	Ordinary
1	UBS Securities Australia Ltd	euy.	16,414	1,913	Ordinary
	UBS Securities Australia Ltd	Buy	257	30	Ordinary
3-0ct-14	UBS Securities Australia Ltd	Buy	4,290	200	Ordinary
	UBS Securities Australia Ltd	Buy	85,800	10,000	Ordinary
1	UBS Securities Australia Ltd	Buy	2,711	316	Ordinary
1	UBS Securities Australia Ltd	Buy	67,782	7,900	Ordinary
	UBS Securities Australia Ltd	Buy	3,689	430	Ordinary
3-Oct-14	UBS Securities Australia Ltd	Buy	13,471	1,570	Ordinary
Ì	UBS Securities Australia Ltd	Buy	515	09	Ordinary
	UBS Securities Australia Ltd	Buy	3,149	367	Ordinary
	UBS Securities Australia Ltd	Buy	98	101	Ordinary
3-Oct-14	UBS Securities Australia Ltd	Buy	16 105	1 877	Ordinary
3-Oct-14	UBS Securities Australia Ltd	Buv	21 973	2 561	Ordinary
	UBS Securities Australia Ltd	Buv	73 445	095.8	Ordinary
	LIBS Securities Australia Itd	Buy	703 3	9,200 6,800	Ordinary
	100 Counties Autralia Ind	(50)	100,0	200	Clumacy
ı	Upo Securities Australia pro	ína d	/1977	COS.	Ordinary
	Obsiderations Australia Ltd	ina a	6/5/1	184	Ordinary
1000	COD Securities Adultalia LIG	ána a	6967	34b	Ordinary
ł	Obsiderational Ltd	anà	191,25	2,936	Ordinary
23-001-14	UBS Securities Australia Ltd	ony o	1,275,000	150,000	Ordinary
2 Oct 14	DES SCULINES AUSTRAINE LIG	and.	180,61	40/	Ordinary
3-00-14	Obsiderations Australia LIG	aug.	/88,19	1,231	Ordinary
3-04-14	Opp Securines Australia Ltd	on)	15,836	/82/	Ordinary
3-Oct-14	UBS Securities Australia Ltd	Buy	18,995	2,219	Ordinary
3-0ct-14	UBS Securities Australia Ltd	Buly	21,383	2,498	Ordinary
3-Oct-14	UBS Securities Australia Ltd	Buy	29,686	3,468	Ordinary
3-Oct-14	UBS Securities Australia Ltd	Buy	2,029	237	Ordinary
3-Oct-14	UBS Securities Australia Ltd	903	6,274	733	Ordinary
3-Oct-14	UBS Securities Australia Ltd	Buy	14,715	1,719	Ordinary
23-Oct-14	UBS Securities Australia Ltd	Buy	4,280	200	Ordinary
				1022	, min

UBS Securities Australia Ltd	Sell	1 557	(500)	Cierrary
Australia Ltd	Sell	3,741	(430)	Ordinary
UBS Securities Australia Ltd	Sell	78	(6)	Ordinary
s Australia Ltd	lis Sell	8,700	(1,000)	Ordinary
UBS Securities Australia Ltd	Sell	78.848	(6,007)	Ordinary
UBS Securities Australia Ltd	Sell	85,095	(9,781)	Ordinary
UBS Securities Australia Ltd	Sell	105,270	(12,100)	Ordinary
es Australia Ltd	Sell	869,500	(100,000)	Ordinary
es Australia Ltd	Sell	126,395	(14,559)	Ordinary
ies Australia Ltd	Sell	64,860	(7,481)	Ordinary
ties Australia Ltd	Sell	643,141	(74,180)	Ordinary
UBS Securities Australia Ltd	lies S	1,740,000	(200,000)	Ordinary
JBS Securities Australia Ltd	Sell	870,000	(100,000)	Ordinary
UBS Securities Australia Ltd	Sell	367,721	(42,413)	Ordinary
ties Australia Ltd	Seil	870,000	(100,000)	Ordinary
ties Australia Ltd	Sell	868,000	(100,000)	Ordinary
ties Australia Ltd	Sell	11,358	(1,310)	Ordinary
ties Australia Ltd	Sell	4,439	(512)	Ordinary
ties Australia Itd	3	2,601	(005)	Ordinary
ties Australia Itd	3	7 225	(4,587)	Ordinary
ties Australia Ltd	Sell	13.837	(1.596)	Ordinan
JBS Securities Australia Ltd	Sell	10,413	(1,201)	Ordinary
JBS Securities Australia Ltd	Sell	2,168	(250)	Ordinary
UBS Securities Australia Ltd	Sell	2,601	(300)	Ordinary
ties Australia Ltd	Seil	2,601	(300)	Ordinary
ties Australia Ltd	Sell	2,766	(319)	Ordinary
JBS Securities Australia Ltd	Sell	4,335	(200)	Ordinary
IBS Securities Australia Ltd	lac	3,893	(449)	Ordinary
JBS Securities Australia Itd	la S	11 012	(300)	Ordinary
JBS Securities Australia Ltd	Sell	71811	(1 363)	Ordinary
JBS Securities Australia Ltd	Sell	2,168	(250)	Ordinary
rities Australia Ltd	Sell	11,817	(1,363)	Ordinary
rities Australia Ltd	Sell	11,323	(1,306)	Ordinary
ities Australia Ltd	Seli	6	(i)	Ordinary
thes Australia Ltd	>e+	12,199	(1,407)	Ordinary
tion Australia (10	Seil	11,6%	(1,349)	Ordinary
ities Australia Itd	Sell	755,11	(1,314)	Ordinary
ities Australia Ltd	Sell	5 202	(009)	Ordinary
UBS Securities Australia Ltd	Sell	10.404	(1 200)	Ordinary
ities Australia Ltd	Sell	6.069	(2007)	Ordinary
UBS Securities Australia Ltd	Sell	19,178	(2,212)	Ordinary
rities Australia Ltd	Sell	104	(12)	Ordinary
rities Australia Ltd	Sell	35	(11)	Ordinary
rities Australia Ltd	Sell	9,234	(1,065)	Ordinary
UBS Securities Australia Ltd	Sell	19,525	(2,252)	Ordinary
UBS Securities Australia Ltd	Sell	13,005	(1,500)	Ordinary
JBS Securities Australia Ltd	Sell	20,000	(2,767)	Ordinary
rities Australia Ltd	Sell	52	(9)	Ordinary
UBS Securities Australia Ltd	Sell	104		Ordinary
rities Australia Ltd	Sell	19,742	(7,777)	Ordinary
UBS Securities Australia Ltd	Sell	95	(11)	Ordinary
Securities Australia Ltd	Sell	7,760	(882)	Ordinary
UBS Securities Australia Ltd	Sell	192,110	(22,158)	Ordinary
UBS Securities Australia Ltd	Sell	8,670	(1,000)	Ordinary
JBS Securities Australia Ltd	Sell	95	(11)	Ordinary
Securities Australia Ltd	Sell	10,491	(1,210)	Ordinary
JBS Securities Australia Ltd	Sell	104	(12)	Ordinary
JBS Securities Australia Ltd	Sell	94,858	(10,941)	Ordinary
UBS Securities Australia Ltd	Sell	191,226	(22,056)	Ordinary
ties Australia Ltd	Sell	104	(12)	Ordinary
ties Australia Ltd	Sell	909'59	(7.567)	Ordinary
UBS Securities Australia Ltd	Sell	104	(12)	Ordinary
ties Australia Ltd	Sell	10,404	(1,200)	Ordinary
Obs securities Australia Ltd	-	071	(665)	
		667'61	(175'1)	Urdinary

				2000	(in incide
7-Oct-14	UBS Securities Australia Ltd	Buy	7.083	816	Ordinary
27-Oct-14	UBS Securities Australia Ltd	Seł	1,712,441	(197,486)	Ordinary
114	UBS Securities Australia Ltd	Sell	78,737	(25,341)	Ordinary
t-14	UBS AG Australia Branch	Buy	712,27	8,318	Ordinary
1-14	UBS AG Australia Branch	Sell	669,146	(77,173)	Ordinary
114	UBS Wealth Management Australia Ltd	Sell	1,481	(171)	Ordinary
1-14	UBS Wealth Management Australia Ltd	Buy	27,690	3,194	Ordinary
27-0ct-14	UBS Securities Australia Itd	Enter Into Derivative	δŅ	900 000	Ontions
		Enter Into Derivative		200,000	chical
27-Oct-14	UBS Securities Australia Ltd	Contract	N/A	(60,000,000)	Options
27-0ct-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	AN N	(3.000.000)	Options
		Enter Into Derivative			
27-0ct-14	UBS Securities Australia Ltd	Contract	N/A	300,000,000	Options
/-Oct-14	UBS Securities Australia Ltd	Sel.	16,748	(1,925)	Ordinary
7 04 14	UBS Securities Australia Ltd	Sel.	3,/41	(430)	Ordinary
177	Tilos Cocuritios Australia Ind	Coll	100.01	(100,7)	Orginary
1-1-1	UBS Securities Australia Ltd	Sell	14 285	(1 642)	Ordinary
7-Oct-14	UBS Securities Australia Ltd	Sell	3,149	(362)	Ordinary
7-Oct-14	UBS Securities Australia Ltd	Sell	11,310	(1,300)	Ordinary
7-Oct-14	UBS Securities Australia Ltd	Sell	270	(31)	Ordinary
7-Oct-14	UBS Securities Australia Ltd	Seil	1,488	(171)	Ordinary
-0ct-14	UBS Securities Australia Ltd	Sell	7,082	(814)	Ordinary
11.14	I IRS Securities Australia +td	الله الله	73.741	(315)	Ordinary
1-14	UBS Securities Australia 11d	35	20540	(05)	Ordinary
1-14	UBS Securities Australia Ltd	Self	2,175	(250)	Ordinary
1-14	UBS Securities Australia Ltd	Seli	3,149	(362)	Ordinary
:t-14	UBS Securities Australia Ltd	Sell	2,201	(253)	Ordinary
1-14	UBS Securities Australia Ltd	Sell	2,140	(246)	Ordinary
1-14	UBS Securities Australia Ltd	Sell	12,841	(1,476)	Ordinary
1-14	UBS Securities Australia Ltd	lia C	998'6	(1,134)	Ordinary
7-Oct-14	USS Securities Australia Ltd	Sell	060,6	(00/)	Ordinary
1-14	HRS Securities Australia 14d	la S	707	(67)	Ordinary
1.14	I IRS Securities Australia 11d	Tê'S	0 083	(C) (O)	Codinary
7-Oct-14	UBS Securities Australia Ltd	Self	4.507	(518)	Ordinary
t-14	UBS Securities Australia Ltd	Sell	42,578	(4,894)	Ordinary
t-14	UBS Securities Australia Ltd	Sell	2,610	(300)	Ordinary
t-14	UBS Securities Australia Ltd	Sell	20,784	(2,389)	Ordinary
1-14	UBS Securities Australia Ltd	Sell	10,440	(1,200)	Ordinary
1-14	UBS Securities Australia Ltd	260	792	(91)	Ordinary
114	UBS Securities Australia Ltd	Sel	17 400	(10,467)	Ordinary
7-0ct-14	UBS Securities Australia Ltd	Sell	25t,	(A)	Ordinary
7-0ct-14	UBS Securities Australia Ltd	Seli	46,145	(5,304)	Ordinary
7-Oct-14	UBS Securities Australia Ltd	Sell	2,819	(324)	Ordinary
t-14	UBS Securities Australia Ltd	Sell	43,500	(000)(2)	Ordinary
t-14	UBS Securities Australia Ltd	Sell	24,047	(2,764)	Ordinary
7-Oct-14	UBS Securities Australia Ltd	Sell	4,437	(510)	Ordinary
t-14	UBS Securities Australia Ltd	Sell	25	(9)	Ordinary
t-14	UBS Securities Australia Ltd	Sell	1,035	(119)	Ordinary
27-Oct-14	UBS Securities Australia Ltd	Sell	444	(51)	Ordinary
t-14	UBS Securities Australia Ltd	Sell	15,042	(67,1)	Ordinary
1-14	UBS Securities Australia Ltd	.Sell	29,449	(3,385)	Ordinary
-Oct-14	UBS Securities Australia Ltd	Sell	10,275	(1,181)	Ordinary
-Oct-14	UBS Securities Australia Ltd	Sell	2,253	(259)	Ordinary
27-Oct-14	UBS Securities Australia Etd	Self	2,140	(246)	Ordinary
7-Oct-14	UBS Securities Australia Ltd	Sell	2,227	(256)	Ordinary
	UBS Securities Australia Ltd	Dell Coll	278	(32)	Ordinary
7-Oct-14	UBS Securities Australia Ltd	Sell	7,421	(853)	Ordinary
	Upo Securities Australia Ital	ligo de la companya d	17 400	(990'c)	Ordinary
1	Upo Securities Australia trd	100	17,400	(2,000)	Orginary
ľ	LIBS Securities Australia Ltd	i i	4,550	(nnc)	Ordinary
27-Oct-14	LIBS Securities Australia Ltd	197	2,010	(000)	Ordinary
l	118S Securities Australia Itd	Jac.	771.0	(050)	Ordinary
27-0ct-14	IUBS Securities Australia Ltd	3,	61 796	7 103	Ordinary
l	UBS Securities Australia Itd	3	771.7	(875)	Occinially
l	מסק מברתו ווידיו לימינו שוני היים		(//:/		
7/-04-14	Libe Constitute Ametralia Ltd	9	271.6	(350)	Ordinary

8-Oct-14		λ	אאס בכ	908 6	Ordinan
	UBS Securities Australia Ltd	Buy	51,051	5,999	Ordinary
28-Oct-14	UBS Securities Australia Ltd	Buy	43	2	Ordinary
8-Oct-14	UBS Securities Australia Ltd	Sell	3,864	(454)	Ordinary
8-Oct-14	UBS Securities Australia Ltd	Buy	83,534	9,816	Ordinary
8-Oct-14	UBS Securities Australia Ltd	Buy	129	73	Ordinary
8-00-14 0-0-14	UBS Securities Australia Ltd	Sell	7,931	(932)	Ordinary
04.14	CIDS Securities Australia Ltd	Sell	3,413	(401)	Ordinary
8.04-14	LIBS Securities Australia Ind	روا ورا	1,030	(171)	Ordinary
28-Oct-14	UBS Securities Australia Ltd	Buv	110,12	5,000	Ordinary
8-Oct-14	UBS Securities Australia Ltd	Sell	σ	(1)	o de la constantina della cons
9-Oct-14	UBS Securities Australia Ltd	Sell	5 569	(651)	Ordinary
9-Oct-14	UBS Securities Australia Ltd	Sell	50.713	(5 940)	Ordinary
9-Oct-14	UBS Securities Australia Ltd	Buy	129,593	15,208	Ordinary
9-Oct-14	UBS Securities Australia Ltd	Buy	53,615	6,292	Ordinary
9-Oct-14	UBS Securities Australia Ltd	Buy	1,684	196	Ordinary
9-Oct-14	UBS Securities Australia Ltd	Sell	115,615	(13,540)	Ordinary
9-Oct-14	UBS Securities Australia Ltd	Buy	85,481	10,000	Ordinary
29-Oct-14	UBS Securities Australia Ltd	Buy	25,650	3,000	Ordinary
9-00-14	UBS Securities Australia Ltd	Nell C	38,615	(4,500)	Ordinary
9-OCI-14	use securities Australia Ltd	buy coll	152,719	17,855	Ordinary
9-04-14	198 Securities Australia 144	Din.	105,430	(12,331)	Ordinary
9-04-14	LIBS Securities Australia Ind	Sall	68,037	666,7	Ordinary
9-0d-14	USS AG Australia Branch	Birv	677,900	(94,103)	Organary
9-Oct-14	UBS AG Australia Branch	Sell	640,884	(74 994)	Ordinary
9-Oct-14	UBS Wealth Management Australia Ltd	Sell	4,087	(478)	Ordinary
1000		Enter Into Derivative			
11.000	ODS Securities Australia Ltd	Enter Into Derivative	NA	20/,443	Options
29-Oct-14	UBS Securities Australia Ltd	Contract	N/A	(20,744,300)	Options
29-Oct-14	UBS Securities Australia 11d	Enter Into Derivative Contract	dy'z	000 000 1	orditor
		Enter Into Derivative		(000'000'1)	Options
29-Oct-14	UBS Securities Australia Ltd	Contract	NA	200,000	Options
1000	1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 -	Enter Into Derivative			
29-04-14	UBS Securities Australia Ltd	Contract	AN S	(20,000,000)	Options
9-0d-14	LIBS Securities Australia Itd	Jel.	137	(q1)	Ordinary
9-04-14	LIRS Securities Australia Itd	Sett	001.0	(000)	Ordinary
29-Oct-14	UBS Securities Australia Itd	3	2027	(A5C)	Ordinary
29-Oct-14	UBS Securities Australia Ltd	Sell	727	(30)	Ordinary
9-Oct-14	UBS Securities Australia Ltd	Sell	6.130	(717)	Ordinary
:9-Oct-14	UBS Securities Australia Ltd	Sell	21,375	(2,500)	Ordinary
29-Oct-14	UBS Securities Australia Ltd	Sell	222	(56)	Ordinary
9-Oct-14	UBS Securities Australia Ltd	Sell	10,106	(1,182)	Ordinary
9-Oct-14	UBS Securities Australia Ltd	<b>.</b>	2,565	(300)	Ordinary
9-Oct-14	UBS Securities Australia Ltd	lax 3	9,559	(1,118)	Ordinary
9-0d-14	UBS Securities Australia Ltd	Jac S	26,667	(3,119)	Ordinary
9-00-14	UBS Securities Australia L(d	Jel 9	0,840	(800)	Ordinary
9-Oct-14	UBS Securities Australia Ltd	Sel S	4 775	(5,00)	Ordinary
9-0ct-14	UBS Securities Australia Ltd	Sell	17,134	(2.004)	Ordinary
9-Oct-14	UBS Securities Australia Ltd	Sell	3,155	(369)	Ordinary
9-Oct-14	UBS Securities Australia Ltd	Sell	5,113	(208)	Ordinary
9-Oct-14	UBS Securities Australia Ltd	Sell	15,775	(1,845)	Ordinary
9-0ct-14	UBS Securities Australia Ltd	Sell	39,954	(4,673)	Ordinary
9-0ct-14	UBS Securities Australia Ltd	Sell	1,163	(136)	Ordinary
9-Oct-14	UBS Securities Australia Ltd	Nell Sell	44,554	(5,211)	Ordinary
9-00-14	UBS Securities Australia Ltd	yell	3,155	(698)	Ordinary
9-00-14	UBS Securities Australia Ltd	Zell	1,163	(136)	Ordinary
9-Ort-14	LiBS Securities Australia Itch	, a	3 078	(005)	Ordinary
9-Oct-14	UBS Securities Australia Ltd	les	718	(88)	Ordinary
9-0d-14	UBS Securities Australia Ltd	les.	209	(72)	Ordinary
29-Oct-14	UBS Securities Australia Ltd	Sell	3.677	(430)	Ordinary
9-Oct-14	UBS Securities Australia Ltd	Sell	3,155	(69E)	Ordinary
9-Oct-14	UBS Securities Australia Ltd	Seil	2,138	(250)	Ordinary
9-Oct-14	UBS Securities Australia Ltd	Sell	8,969	(1,049)	Ordinary
9-Oct-14	UBS Securities Australia Ltd	Sell	26,667	(3,119)	Ordinary
9-Oct-14	UBS Securities Australia Ltd	Sell	5,113	(288)	Ordinary
9.04.14					

Section	Ordinary	Ordinary	Ordinary	Ordinary	Company	Clossary	Clumary	Cidinally	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	300	Options	Options		Options	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary		Ordinary	Ordinary	Ordinary Ordinary Ordinary	Ordinary Ordinary Ordinary Ordinary	Ordinary Ordinary Ordinary Ordinary Ordinary	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary	Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary	ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
1000	(73.486)	(7.596)	(25,099)	(1540)	(0454)	(404)	(060)	(000,02)	(085,5)	/16,1	84,510	175,000	(ι)	000 000 002	000'000'000	(300,000,000)		(300,000,000)	430	20.355	670	700	5,000	21,213	20,000	(DEC 96)	30,906	2,008	(87,547)	222,619	25,263	52,118	100,000	3 4	(CI)	001	360	389	292	292	1,107	309	150	699	08	1,794	5,000	80	108	87	48	442		100	5,000	5,000	5,000 5,000 5,331 4,840 1,875	5,000 5,331 4,840 1,875 4,365	100 5,000 5,331 4,840 1,875 4,365	100 5,000 5,331 4,840 1,875 4,365 118	100 5,000 5,331 4,840 1,875 4,365 118 116	100 5,000 5,331 4,840 1,875 4,365 118 116 901 302	5,000 5,000 4,840 1,875 1,865 1,16 901 901 901 302 7,200	5,331 4,340 1,875 1,875 1,187 1,16 1,16 1,16 1,16 1,16 1,16 1,16 1,1	5,000 5,000 6,331 1,4840 1,485 4,365 118 118 116 901 2,200 2,200 2,200 837 2,200 837 2,200 839	5,331 4,875 1,875 1,875 1,16 1,16 1,16 2,20 2,20 2,20 2,79 89	5,000 5,000 4,340 1,875 1,875 1,875 1,18
0 527	203 624	65.857	217 608	13 352	300.5	0000	970,00	00,012	40,040	261,61	/32,702	1,517,250	6	S Z		N/A		WA	3,661	173 274	5,727	5,964	42,650	181,172	2C,201	846 238	263,931	17,106	746,540	1,906,524	216,349	446,332	850,000	/	971	853	3.074	3,318	2,237	4,842	9,454	151,5	1 281	5,713	683	15,312	42,675	89	922	743	410	3,775	700	110 07	42,675	42,675	42,675 45,500 41,309 16,003	42.675 45.500 41,309 16,003 37,255	42,675 45,500 41,309 16,003 37,255 1,007	42,675 45,500 41,309 16,003 37,255 1,007 990	42,675 45,500 41,309 16,003 37,255 1,007 1,007 990 7,690	42,675 45,500 41,509 16,003 37,255 1,007 950 7,690 2,257 8,440 8,40 8,	42,675 45,500 41,500 16,003 37,255 10,007 1007 2579 2,579 2,579 18,864	42,675 45,500 41,509 16,003 37,255 1,007 1,007 2,500 5,440 18,554 2,387 2,387 2,387 2,387	42,675 45,500 41,509 16,003 37,255 37,255 990 7,680 2,5440 18,854 18,854 2,854 18,854 18,854 18,854 18,854 18,854 6,905 6,905 6,905	42,675 45,500 41,500 16,003 1,007 1,00 1,00	42,675 45,500 41,509 16,003 37,255 7,680 7,680 7,680 7,680 1,007 1,007 1,007 1,007 1,007 2,509 1,800 1
	3.	Sell	Sell	lei.	Sell	5 7	100			on)	, cry	Şriv	Sell	Enter Into Derivative	Enter Into Derivative	Contract	Enter into Derivative	ontract	Buy	an Buy	8uy	Buy	luy	Buy	65	Sel	Buy	'n	ell	Luy.	Buy	ry.	án =	5	M. M.	6 5	, in	ý	Ly,	·uy	Buy	in a	5 3	Buy	'n	ń	λ'n	Buy	uy	ń	Buy	Buy	Buy	- -	2	Buy	Buy Buy Buy	Buy Buy Buy Buy	Buy Buy Buy Buy	Buy Buy Buy Buy Buy Buy	Buy Buy Buy Buy Buy Buy Buy Buy	Buy Buy Buy Buy Buy Buy Buy Buy Buy Buy	Buy Buy Buy Buy Buy Buy Buy Buy Buy Ruy Ruy	8 8 9 9 8 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	89.9 60.9	8 8 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	899 809 809 809 809 809 809 809
bt Legisting Australia 140	URS Securities Australia	Securities	UBS Securities Australia	URS Securities Australia	I IBS Securities Australia	UBS Securities Australia L	UBS Securities Australia L	Obs securities Australia L	USS Securities Australia	OBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia L	ht Leilenburg seithungs 2811	n pilense vaninas cao	UBS Securities Australia Ltd		UBS Securities Australia L	UBS Global Asset Management	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia I	UBS AG Australia Branch	UBS AG Australia Branch	UBS AG Australia Branch	UBS AG Australia Branch	UBS AG Australia Branch	UBS AG Australia Branch	UBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia	UBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia L	UBS Securities Australia Ltd	UBS Securities Australia Li	UBS Securities Australia Li	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia L	UBS Securities Australia Li	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	one secondes most one a	185 Securities Australia 11	UBS Securities Australia Ltd UBS Securities Australia Ltd	UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	UBS Securities Australia Ital UBS Securities Australia Ital	UBS Securities Australia I. 1985 Securities Australia I. 1985 Securities Australia I. 1985 Securities Australia I. 1985 Securities Australia I. 1985 Securities Australia I. 1985 Securities Australia I. 1985 Securities Australia II.	UBS Securities Australia Itol UBS Securities Australia Itol UBS Securities Australia Itol UBS Securities Australia Itol UBS Securities Australia Itol UBS Securities Australia Itol UBS Securities Australia Itol UBS Securities Australia Itol UBS Securities Australia Itol UBS Securities Australia Itol UBS Securities Australia Itol UBS Securities Australia Itol UBS Securities Australia Itol UBS Securities Australia Itol UBS Securities Australia Itol UBS Securities Australia Itol UBS Securities Australia Itol UBS Securities Australia Itol UBS Securities Australia Itol	108 Securities Australia Ltd. 108 Securities Australia Ltd.
27.04.14	27-Oct-14	ı	27-Oct-14	l	ı		ı	1	ı	1	١	27-Oct-14	-		1,000	27-Oct-14	:	27-Oct-14	28-Oct-14	28-Oct-14				28-Oct-14	ı	1	l				H		-	1		28-Oct-14	ı		Ш	1	28-Oct-14		ı	28-Oct-14	l			28-Oct-14		1	28-Oct-14	28-Oct-14	28-Oct-14	28-Oct-14		28-04-14	28-Oct-14 28-Oct-14	28-Oct-14 28-Oct-14 28-Oct-14	28-0ct-14 28-0ct-14 28-0ct-14 28-0ct-14	28-0ct-14 28-0ct-14 28-0ct-14 28-0ct-14 28-0ct-14	28-0ct-14 28-0ct-14 28-0ct-14 28-0ct-14 28-0ct-14 28-0ct-14	28-0ct-14 28-0ct-14 28-0ct-14 28-0ct-14 28-0ct-14 28-0ct-14 28-0ct-14	28-0ct-14 28-0ct-14 28-0ct-14 28-0ct-14 28-0ct-14 28-0ct-14 28-0ct-14 28-0ct-14	28-0ct-14 28-0ct-14 28-0ct-14 28-0ct-14 28-0ct-14 28-0ct-14 28-0ct-14 28-0ct-14 28-0ct-14			

29-0c-14 UBS securities Australia ted 29-0c-14 UBS securities Australia ted 29-0c-14 UBS securities Australia ted 29-0c-14 UBS securities Australia ted 29-0c-14 UBS Securities Australia ted 29-0c-14 UBS Securities Australia ted 29-0c-14 UBS Securities Australia ted 29-0c-14 UBS Securities Australia ted 29-0c-14 UBS Securities Australia ted 29-0c-14 UBS Securities Australia ted 29-0c-14 UBS Securities Australia ted 29-0c-14 UBS Securities Australia ted 29-0c-14 UBS Securities Australia ted 29-0c-14 UBS Securities Australia ted 29-0c-14 UBS Securities Australia ted 20-0c-14 UBS Wealth Management Australia ted 20-0c-14 UBS Wealth Management Australia ted 20-0c-14 UBS Wealth Management Australia ted 20-0c-14 UBS Wealth Management Australia ted 20-0c-14 UBS Wealth Management Australia ted 20-0c-14 UBS Wealth Management Australia ted 20-0c-14 UBS Wealth Management Australia ted 20-0c-14 UBS Wealth Management Australia ted 20-0c-14 UBS Wealth Management Australia ted 20-0c-14 UBS Wealth Management Australia ted 20-0c-14 UBS Wealth Management Australia ted 20-0c-14 UBS Wealth Management Australia ted 20-0c-14 UBS Wealth Management Australia ted 20-0c-14 UBS Wealth Management Australia ted 20-0c-14 UBS Wealth Management Australia ted 20-0c-14 UBS Wealth Management Australia	Sell Sell Sell Sell Buy		2,565	(300)	Ordinary
	Sell Sell Buy		50	2 22	- 1
	8uy Buy		071	(14)	Ordinary
	louy Buy		14,535	(00/1)	Ordinary
	Buy		854,000	000,001	Ordinary
			35,800	4,192	Ordinary
	Buy		51	9	Ordinary
	Buy		102	12	Ordinary
	Buy		1,290	152	Ordinary
	Eut	Enter Into Derivative			
	COL	Contract	N/A	20,744,300	Options
	Ent	Enter Into Derivative			
	כמ	Contract	ΑV	(100,000,000)	Options
	Ent	Enter Into Derivative			
	Co	itract	NA	20'000'000	Options
	Buy		113,626	13,208	Ordinary
	Buy		191,942	22,303	Ordinary
	Sell		99,554	(11,543)	Ordinary
	Buy		87,450	10,162	Ordinary
	Sell		57,346	(6.675)	Ordinary
	Sel		85.611	(10,000)	Ordinary
	Buy		158.124	18.394	Ordinary
	Sel		122.576	(14.238)	Ordinary
	le S		2 619 406	(304 887)	Ordinary
	Stor	rk borrowed	Α'N	130 000	Ordinary
	Coll		15 984	(1.960)	Ordinan
	las		2576 186	(390 000)	Contrary.
	100		020,020,	(000,462)	Collinary
	nac o		6/97,00	(00,00)	Columary
			476,67	5,004	Ordinary
			2,493	(289)	Ordinary
			3,467	(402)	Ordinary
	ralia Ltd Sell		3,433	(368)	Ordinary
			1,190	(138)	Ordinary
			3 7 7 8	(438)	Ordinary
	ľ		4 477	(5,10)	Ordinan
	Ī		1000	(4,0)	( in the contract of the contr
			7,341	(341)	Ordinary
			18,881	(2,189)	Ordinary
			2,683	(311)	Ordinary
			2,665	(309)	Ordinary
	alia Ltd Sell		165,075	(19,138)	Ordinary
			3.968	(460)	Ordinary
	Ī		123 914	(14 366)	Ordinary
			1000	(000)	O'CLI-TIC
	all city		420.40	(677)	Countries
			107'17	(510/7)	Commany
	alia Ltd		7,846	(330)	Ordinary
			2,510	(291)	Ordinary
			60,844	(7,054)	Ordinary
	alia Ltd Sell		3,045	(323)	Ordinary
			2.511	791	Ordinary
	Call Call		020 6	(1947)	Ordinary
	T		6/0/2	(741)	Ordinary
			8,600	(997)	Ordinary
			25,924	3,004	Ordinary
	alia Ltd		3.657	(474)	Ordinary
$1 \mid \cdot \mid \cdot \mid$			107.001	(17 469)	o dina
			100,421	(904/41)	Contrary.
			1,725	(200)	Ordinary
	alia Ltd Sell		3,355	(386)	Ordinary
	alia Ltd Sell		1,699	(197)	Ordinary
	Ī		1 204	10000	
l			D6/1	(807)	Ordinary
			2,200	(255)	Ordinary
	alia Ltd Sell		2,312	(268)	Ordinary
			3 735	(433)	Ordinan
ı	T		00000	(000 00)	
1			431,000	(000,05)	Ordinary
3	Sell		11,624	(1,350)	Ordinary
	Sell		4.322	(202)	Ordinary
1	00		121	(10)	Ordinary
3 5			-	(4.1)	
FOCT-14 UBS Securities Australia Ltd	Sell		98	(OL)	Ordinary
NBS	Sett		14,155	(1,644)	Ordinary
I	lo2		1171	(361)	Ordinary
3	200		171,1	(001)	Ciciliany
SS	Sel		1/7'51	11,542,1)	Ordinary
30-Oct-14 UBS Securities Australia Ltd	Sett		904	(105)	Ordinary
	Sell		089	(62)	Ordinary
IBS	la's		783	(0.8)	Ordinary
901	173		201.01	2000	
30-Oct-14 UBS Securities Australia Ltd	Sell		16,746	(1,945)	Ordinary
	Btry		24 476	2 846	Ordinary

** 10 00	ISS SECTION ACCUSED TO	Ver	1 445	(150)	Cedimon
20-00-14	UBS Securities Australia Ltd	J.	075 74	(109)	Ordinary
30-Oct-14	UBS Securities Australia Ltd	les	379.050	(36,90)	Ordinary
30-Oct-14	UBS Securities Australia Ltd	Sell	616,050	(66,600)	Ordinary
		Enter Into Derivative			
30-Oct-14	UBS Securities Australia Ltd	Contract	NA	(2,000)	Options
30-0ct-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	A/N	(000)8(2)	Ontions
71 000	Fel Signature A society and 2 Oct	Enter Into Derivative			
41.0000	Opp Security Australia Ltd	Enter into Derivative	Y.	(37,000)	Options
30-0ct-14	UBS Securities Australia Ltd	Contract	N/A	39,900	Options
30-0ct-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	2	(2.000)	Ontions
30-0ct-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	2	(00005)	Ontions
		Enter Into Derivative		(months)	SICINO.
30-0ct-14	UBS Securities Australia Ltd	Contract	₹N	(15,000)	Options
30-Oct-14	UBS Securities Australia Ltd	Contract	NVA	(8,000)	Options
20.04.14	Sociation 2003	Enter Into Derivative	1		
31-Oct-14	UBS Global Asset Management	Sell	N/A 282 679	92,900	Options
31-Oct-14	UBS Securities Australia Ltd	Buy	105,817	12.316	Ordinary
31-Oct-14	UBS Securities Australia Ltd	Sell	77,938	(1,071)	Ordinary
31-Oct-14	UBS Securities Australia Ltd	Sell	28,522	(3,313)	Ordinary
31-00-14	UBS Securities Australia Ltd	Sell	646,860	(75,225)	Ordinary
31-0ct-14	UBS Securities Australia I td	Buy	8,620	1,000	Ordinary
31-Oct-14	UBS Securities Australia Ltd	Sell	264,165	(30.662)	Ordinary
31-Oct-14	UBS AG, London Branch	Buy	797,313	75,225	Ordinary
31-Oct-14	UBS AG, London Branch	Sell	645,860	(75,225)	Ordinary
31-0d-14	UBS AG Australia Branch	Buy	412,330	47,700	Ordinary
31-Oct-14	UBS AG Australia Branch	Sell	304,314	(35.342)	Ordinary
31-Oct-14	UBS Wealth Management Australia Ltd	Sell	5,253	(613)	Ordinary
31-0ct-14	UBS Securities Australia Ltd	Sell	652,500	(75,000)	Ordinary
31-04-14	UBS Securities Australia Ltd	N	298,719	(34,614)	Ordinary
31-0ct-14	UBS Securities Australia Ltd	Sell	19,495	(2,252)	Ordinary
31-Oct-14	UBS Securities Australia Ltd	Sell	1,346	(156)	Ordinary
31-0ct-14	UBS Securities Australia Ltd	Sell	5,601	(649)	Ordinary
03-Nov-14	UBS Global Asset Management	Riv	0707	(24)	Ordinary
03-Nov-14	UBS Securities Australia Ltd	Sell	40,928	(4,741)	Ordinary
03-Nov-14	UBS Securities Australia Ltd	Buy	18,279	2,123	Ordinary
03-Nov-14	UBS Securities Australia Ltd	Sell	140,997	(16,357)	Ordinary
03-NOV-14	UBS Securities Australia Ltd	Sell	21,495	(2,485)	Ordinary
03-Nov-14	UBS Securities Australia Ltd	Buy	30,036	3.480	Ordinary
03-Nov-14	UBS Securities Australia Ltd	Sell	138,429	(16,038)	Ordinary
1	UBS Securities Australia Ltd	Sell	3,456	(400)	Ordinary
03-Nov-14	UBS Securities Australia Ltd	Sell	6,048	(700)	Ordinary
l	UBS Securities Australia Ltd	Stock borrowed	N/A	215.000	Ordinary
	UBS AG Australia Branch	ΙI	499,058	57,842	Ordinary
03-Nov-14	UBS Wealth Management Australia Ltd	Buy	69,414	8,062	Ordinary
03-Nov-14	UBS Securities Australia Ltd	Sell	286	(33)	Ordinary
33-Nov-14	UBS Securities Australia Ltd	Buv	3.151	366	Ordinary
33-Nov-14	UBS Securities Australia Ltd	Buy	3,151	366	Ordinary
03-Nov-14	UBS Securities Australia Ltd	Buy	17,125	1,989	Ordinary
03-Nov-14	UBS Securities Australia Ltd	Buy	3,151	366	Ordinary
03-Nov-14	UBS Securities Australia Ltd	Buy	603	02	Ordinary
03-Nov-14	UBS Securities Australia Ltd	Buy	2,153	250	Ordinary
03-Nov-14	UBS Securities Australia Ltd	Buy	3,151	366	Ordinary
03-Nov-14	UBS Securities Australia Ltd	Buy	3,151	365	Ordinary
03-Nov-14	UBS Securities Australia Ltd	Buy	26	m	Ordinary
03-Nov-14	UBS Securities Australia Ltd	Buy	52	48	Ordinary
03-Nov-14	UBS Securities Australia Ltd	Buý	921	101	Ordinary
03-Nov-14	UBS Securities Australia Ltd	Buy	026'98	10,101	Ordinary
3-Nov-14	UBS Securities Australia Ltd	Buy	2/6	32	Ordinary

	1443 135 93 232 2532 65,199 4,039 2,232 2,100 4,220 2,100 2,100 2,100 2,100 2,100 2,100 2,100 4,200 3,788 186 186 196 196 196 197 198 198 198 198 198 198 198 198 198 198	16 16 300	Ordinary Ordinary Ordinary Ordinary
	935 937 938 938 938 938 938 938 938 938	300	Ordinary Ordinary Ordinary
	2.532 65.199 65.199 7.034 7.034 7.20 6.317 2.337 2.337 2.337 2.337 4.369 1.86 4.369 1.35	300	Ordinary
	2532 65,199 4,094 3,269 4,270 2,317 2,317 2,317 2,318 1,378 1,38 1,38 1,38 1,38 1,38 1,38 1,38 1,3	300	Ordinary
	65,199 4,034 3,629 4,220 5,317 5,317 2,110 2,110 2,110 4,220 1,369 4,260 1,369 1	1771	
	4,034 3,659 4,220 5,317 2,532 2,110 4,220 3,798 1,86 4,963 4,963 4,963 9,98	(7/,7	Ordinary
	4,250 4,270 5,317 2,532 2,110 4,270 3,788 186 4,363 135 99	478	Ordinary
	4,20 5,377 2,537 2,110 4,200 1,360 1,360 1,361 1,351 1	430	Ordinary
	2,537 2,532 2,110 4,220 3,798 1,86 4,963 4,963 9,98 9,98	200	Ordinary
	2.532 2.110 4.220 3.798 1.86 4.363 1.35 9.9	630	Ordinary
	2.110 2.110 3.798 1.86 4.363 1.35 1.35	300	Ordinary
	4,220 3,798 1,186 4,163 135 93	250	Ordinary
	3,798 186 186 4,363 135 93	200	Ordinary
	4,363 135 93	450	Ordinary
	4,363 135 93	22	Ordinary
	135	517	Ordinary
	86	16	Ordinary
		=	Ordinary
	5 824	Con	Ordinan
	8	1	Calpio
	325 350	25025	Cidinary
	220,740	200,12	Ordinary
	40,004	4,818	Ordinary
	19,/50	2,340	Ordinary
	105,905	12,548	Ordinary
	40,588	4,809	Ordinary
	25,826	3,060	Ordinary
	2,532	300	Ordinary
	27,143	3.216	Ordinary
	16 956	2000	Ordinan
	70.803	0.466	Croling
	21.042	023.0	
	100 31	0.00	100
	2000	600,	Clemary
	300	3 5	Ordinary
	59,430	8,227	Ordinary
	90,916	10,772	Ordinary
	422	20	Ordinary
	15,512	1,839	Ordinary
	44,774	5,305	Ordinary
	19,682	2,332	Ordinary
	85,219	10,097	Ordinary
	1,747	202	Ordinary
	2,203	261	Ordinary
	29.633	3 511	Ordinary
	9639	057	Ordinary
	19 707	2 335	Ordinary
	AE 027	2007	Commission
	2,522	300	Oldinary
	250,2	300	Ordinary
	108'/	265	Ordinary
	25,970	3,077	Ordinary
	4,220	200	Ordinary
	2,110	250	Ordinary
	39,820	4,718	Ordinary
	228	27	Ordinary
	354	42	Ordinary
	844	501	Ordinary
	253	CE.	Ordinan
	2532	000	Ordinary
	10 211	3 5	Countary
	11/0	7177	Ordinary
	3,629	430	Ordinary
	2,110	250	Ordinary
	27,008	3,200	Ordinary
	12.660	1.500	Ordinary
	3.308	392	Ordinary
	6,650	200	Cading
	050.	60,	Countries
	oce'i	157	Clerary
	177	21	Ordinary
	844	100	Ordinary
1085 Securities Australia Ltd	3 187	377	Ordinary
	2,102	115	Clulial y
	0000	428	Ordinary
	26,046	3,086	Ordinary
UBS Securities Australia Ltd	93	11	Ordinary
	00014		1
UBS Securities Australia Ltd	17,808	2,110	Ordinary

_				_	_	_	1	т-	т-	т-	_	т-		_	_	_	_	_	_	r	r	_		-1	_	_	_	_		_				_	_	_			_	_	_	_	_	_		_		_		_		_	_	_	_	1	_			<del>,</del>		ı	_		_	_	_	_	_
Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordings	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary
1,786	2	366	367	32	3998	300	531	100	367	366	366	366	3.859	367	843	7,983	25	(11)	348	6,664	(211)	163,246	3,840	(7,846)	2,997	(38,904)	(14,406)	(20,000)	233,953	20,000	(22,500)	139,039	166,047	(10,000)	(664)	(1,234)	8	212	(966)	(100,842)	000,000	21.623	(109,303)	10,000	(504,205)	(7,724)	16,353	(3,100)	13,814	(2,895)	153,080	(504,205)	204,203	(15,880)	(18.136)	327 009	15,000	20,000	(95,864)	(51,437)	291	321	σ .	4,591	720	430	242	579	250
15,377	603	3,151	3,160	576	3.151	2.583	4.572	198	3,160	3,151	3,151	3,151	33,226	3.160	7,258	68'893	43	95	3,003	57,510	1,831	1,411,849	33,255	68,057	1,991	337,239	124/19	2 151	2,027,367	431,145	N/A	1,202,618	1,439,388	86.800	5.764	10,711	69	1,803	8,452	853,325	100 001	181 891	927,939	85,488	4,279,490	65,574	138,569	26,219	116,487	24,620	1,296,833	4,2/9,490	143 570	243,570	153.989	2.770.386	129,193	171,657	808,926	434,039	2,456	2,709	76	38,748	2,710	9,629	4,574	4,887	2,110
Buy	Buy	Buy	Buv	Buy	Buy	Buy	Buv	Buy	Buy	Buy	Buy	BUV	Buy	Buy	Buy	Buy	Buy	Seil	Buy	Buy	Sell	Buy	Buy	Sell	Buy	Seil	Sell	Self	Buy	Buy	Stock returned	Buy	Buy	Zell Xell	les.	Sell	Sell	Buy	Seli	Sell	goà.	Buy	Sell	Buy	Sell	Sell	Buy	Sell	Buy	Sell	Buy	Seil Control	ani,	Sell	100	Buv	Buy	Buy	Sell	Sell	Buy	Buy	Buy	Buy	eny	Suy	Buy	Buy	Buy
ties Australia Ltd	UBS Securities Australia Ltd	ties Australia Ltd	ties Australia Ltd	ties Australia (td	ties Australia Ltd	ties Australia Ltd	ties Australia (1d	ties Australia Ltd	ties Australia Ltd	ties Australia Ltd	ties Australia Ltd	ties Australia Ltd	ties Australia Ltd	ties Australia Ltd	ties Australia Ltd	ties Australia Ltd	ties Australia Ltd	ties Australia Ltd	ties Australia Ltd	ties Australia Ltd	Asset Management	ties Australia Ltd	UBS Securities Australia Ltd	ties Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS AG Australia Branch	ustralia Branch	USS Securities Australia Ltd	UBS Securities Australia Ltd	ties Australia Ltd	ties Australia Ltd	Asset Management	ties Australia Ltd	ties Australia Ltd	ties Australia Ltd	ies Australia Itd	ties Australia Ltd	ies Australia Ltd	UBS Securities Australia Ltd	ties Australia Ltd	ties Australia Ltd	ties Australia Ltd	ies Australia Ltd	JBS Securities Australia Ltd	nes Australia Ltd	UBS AG, London Branch	UBS AG, London Branch	JBS AG Australia Branch	stralia Branch	JBS AG Australia Branch	JBS AG Australia Branch	UBS AG Australia Branch	UBS AG Australia Branch	JBS AG Australia Branch	JBS Securities Australia Ltd	UBS Securities Australia Ltd	ies Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	ies Australia Ltd	UBS Securities Australia Ltd	ies Australia Ltd					
	3-Nov-14 UBS Securiti		l			1						l		ı	ı	l		l		1							T	Ī	Ī	04-Nov-14 UBS Securiti		-	04-Nov-14 UBS AG Aus			1	4-Nov-14 UBS Securitie		-	1	1	ı	ı	l						-Nov-14 UBS Securitie		T	PINOV-14 UBS AG, LON		Ť	Ī	Ī	05-Nov-14 UBS AG Aust		05-Nov-14 UBS AG Aust	٦	1				1	1		05-Nov-14 UBS Securitie

UBS AG, London Banch UBS AG, L	431 431 44069 4.069		Colorac
UBS AG, London Branch UBS AG, London Branch	4,069		Configura
UBS AG, London Branch UBS AG, London Branch	20,538	(577)	Ordinary
UBS AG, London Branch UBS AG, London Branch	000,00	Ę	Grainary
UBS AG, London Branch UBS AG, London Branch			Orginary
UBS AG, London Barnch UBS AG, London Barnch	860,7		Ordinary
1 UBS AG, London Banch	/01./	(824)	Ordinary
1985 AG, London Barnch 1985 AG, London Barnch	380		Ordinary
1985 AG, London Banch 1985 AG, London Banch	816,6		Ordinary
UBS AG, London Banch UBS AG, L	\$ 55.	(4)	Ordinary
UBS AG, London Banch UBS AG, London Banch	621,6		Ordinary
1985 AG, London Banch 1985 AG, London Banch	4,255		Ordinary
UBS AG, London Barnch UBS AG, London Barnch	680'9	(90/)	Ordinary
UBS AG, London Barnch UBS AG, London Barnch			Ordinary
UBS AG, London Banch UBS AG, London Banch		(711)	Ordinary
UBS AG, London Barnch UBS AG, London Barnch			Ordinary
USS AG, London Banch USS AG, London Banch			Ordinary
USS AG, London Barnch USS AG, London Branch		(/0/)	Ordinary
UBS AG, London Banch UBS AG, London Banch	04/		Ordinary
UBS AG, London Branch UBS AG, London Branch	5,463		Ordinary
UBS AG, London Branch UBS AG, London Branch	752	(100,1)	Ordingry
UBS AG, Lordon Branch UBS AG, Lordon Branch	4266		Cidinary
UBS AG, London Branch UBS AG, London Branch	357.01		Cignaly
UBS AG, London Branch UBS AG, London Branch	2 155		Codings
USS AG, London Branch USS AG, London Branch			Ordinary
USS AG, London Branch USS AG, London Branch		(13)	Cidinary
UBS AG, London Branch UBS AG, London Branch	7		i i i
USS AG, London Branch USS AG, London Branch		٤	Coding
UBS AG, London Branch UBS AG, London Branch			Ordinary
UBS AG, London Branch UBS AG, London Branch	2,72		Ordinary
UBS AG, London Barnch UBS AG, London Barnch	000 0		Ordinary
UBS AG, London Branch UBS AG, London Branch	060/2		Ordinary
UBS AG, London Banch UBS AG, London Banch	670'		Ordinary
USS AG, London Barnch USS AG, London Barnch	290.01		Ordinary
URS AG, London Banch URS AG, London Banch	12,907		Ordinary
UBS AG, London Barnch UBS AG, London Barnch	2,776	(375)	Ordinary
1985 AG, London Barnch 1985 AG, London Barnch	9,603		Ordinary
UBS AG, London Barnot. UBS AG, London Barnot.	233		Ordinary
UBS AG, London Banch UBS AG, London Banch	516,21		Ordinary
UBS AG, London Banch UBS AG, London Banch	69	(8)	Ordinary
UBS AG, London Banch UBS AG, London Banch	806,01		Ordinary
UBS AG, London Banch UBS AG, London Banch	2,957		Ordinary
UBS AG, London Banch UBS AG, London Banch	755,8	(166)	Ordinary
UBS AG, London Banch UBS AG, London Banch	1/9/11		Ordinary
UBS AG, London Banch UBS AG, London Banch	8,906		Ordinary
UBS AG, London Banch UBS AG, London Banch	14,111	(1,637)	Ordinary
UBS AG, London Banch UBS AG, London Banch	8,723	`	Ordinary
UBS AG, London Branch UBS AG, London Branch	2,888	(332)	Ordinary
UBS AG, London Banch UBS AG, London Banch	5,198		Ordinary
UBS AG, London Banch UBS AG, London Banch	4,336		Ordinary
UBS AG, London Banch UBS AG, London Banch	1,336		Ordinary
UBS AG, London Banch UBS AG, London Banch	388		Ordinary
USS AC, London Banch USS AG, London Banch	10,180	(1,181)	Ordinary
UBS AG, London Banch UBS AG, London Branch	2,388		Ordinary
UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch	2,787	(323)	Ordinary
UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch	6,964		Ordinary
UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch	868		Ordinary
UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch	4,065		Ordinary
UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch	1,269	(147)	Ordinary
UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch	4,056		Ordinary
UBS AC, London Banch UBS AC, London Banch UBS AC, London Banch UBS AC, London Banch UBS AC, London Banch UBS AC, London Banch UBS AC, London Banch UBS AC, London Banch	121		Ordinary
UBS AC, London Banch UBS AC, London Banch UBS AC, London Banch UBS AC, London Banch UBS AC, London Banch UBS AC, London Banch UBS AC, London Banch	528		Ordinary
UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch	4,539	(526)	Ordinary
UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch	1.286		Ordinary
UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch	1,707		Ordinary
UBS AG, London Branch UBS AG, London Branch	23	(9)	Ordinary
UBS AG, London Branch	CEZ 5	(Ka	Ordinary
( a contract of co	030		Ordinary Ordinary
Do AG Condon Drawin	057		o diligit
198 AC London Branch	44.0		Crainary
Section of Section 1997	4,500		Ordinary
Part 1 Libe AG London Brook	55,133	(007)	Ordinary
Cost Act, Colloct Blanch	7,560		Ordinary
lov-14 UBS AG, London Branch Sell	10,396		

UBS Securities Australia Ltd	Buy	2,532	300	Ordinary
JBS Securities Australia Ltd	Buy	312	37	Ordinary
Securities Australia Ltd	Buy	59	7	Ordinary
UBS Securities Australia Ltd	Buy	277 22	72	Ordinary
itles Australia Lid	ong .	ECE,12	055,2	Ordinary
Upg Securities Australia Ltd	, ind	000,01	048	Ordinary
ines Australia Liu	(na	(/	17	Cidinary
DDS Securities Australia Ltd	opa)	/96'96	4,580	Orginary
OBS Securities Australia Ltd	BOy	1,460	1/3	Ordinary
UBS Securities Australia Ltd	80%	/8,492	9,300	Ordinary
JBS Securities Australia Ltd	Buy	844	100	Ordinary
UBS Securities Australia Ltd	Buy	422	20	Ordinary
UBS Securities Australia Ltd	Buy	15,968	1,892	Ordinary
rities Australia Ltd	Buy	18,796	2,227	Ordinary
UBS Securities Australia Ltd	Buy	1691	20	Ordinary
UBS Securities Australia Ltd	Buv	7,014	831	Ordinary
JRS Securities Australia I to	Bix	3 941	191	Ordinary
IDC Convision Australia (4d	Biss	160	S.C.	Ordinan
sition Australia Led	600	6903	2,6	O Granding
DDS Securities Australia Ltd	Buy	798'9	813	Ordinary
rities Australia Ltd	Buy	0	2	Ordinary
JBS Securities Australia Ltd	Buy	18,300	2,167	Ordinary
rrities Australia Ltd	Buy	56,810	6,727	Ordinary
rities Australia Ltd	Buy	33,986	4,022	Ordinary
irities Australia Ltd	Buy	1,048	124	Ordinary
urities Australia I1d	Bin	3 008	356	Ordinary
mittee Australia 14d	Bar	AT3 67	0 734	Ordinan
Control of Control of Control	62	1000	7.00	oral land
The Maria and The	ens.	622,1	199	Cidinary
urdes Australia Ltd	Buy	686,1	188	Ordinary
UBS Securities Australia Ltd	Buy	17,827	2,111	Ordinary
urities Australia Ltd	Buy	38,923	4,609	Ordinary
UBS Securities Australia Ltd	Buy	7,563	968	Ordinary
UBS Securities Australia Ltd	AN B	24.068	2.850	Ordinary
UBS Securities Australia Ltd	Bav	3 338	395	Ordinary
BC Counties Australia 11d	Broz	10,700	2 224	Cadiosi
the Control of the Co		10,00	277'7	(Billion)
Obs securities Australia Liu	and.	500'/5	4,447	Ordinary
urities Australia Ltd	Buy	18,7/3	677'7	Ordinary
JBS Securities Australia Ltd	Buy	3,490	413	Ordinary
JBS Securities Australia Ltd	Buy	1,276	151	Ordinary
JBS Securities Australia Ltd	Buy	3,110	368	Ordinary
Securities Australia Ltd	Buy	862	102	Ordinary
Securities Australia Ltd	Buy	7,199	852	Ordinary
urities Australia Ltd	Š	845	1001	Ordinary
IRS Securities Australia I to	2 8	227	C	Ordinary
mission Acceptation 144		11 222	200	o de la constante de la consta
annes Australia Liu	ng	227	975'	Ordinary
UB> Securities Australia Ltd	Buy	3,431	400	Ordinary
urities Australia Ltd	Buy	189,822	22,332	Ordinary
UBS Securities Australia Ltd	Buy	1,913	225	Ordinary
urities Australia Ltd	Sell	49,385	(5,810)	Ordinary
urities Australia Ltd	Sell	172,000	(20,000)	Ordinary
urities Australia I td	llay	36	(8)	Ordinary
hal Arrest Management	100	05003	(0)	Coding
Dai Asset Malagerien	1120	00,000	(00,130)	Gulliary
unties Australia Ltd	Sell	191,6	(1,064)	Ordinary
Securities Australia Ltd	Sell	26,271	(3,055)	Ordinary
UBS Securities Australia Ltd	Buy	74,038	8,604	Ordinary
urities Australia Ltd	Self	119,722	(13,905)	Ordinary
rities Australia I to	jes	487 963	(FE 190)	Ordinary
Anti-		303 634	10,000	
uriues Australia Ltd	ing	676'/6	10,334	Ordinary
irities Australia Ltd	Sell	361,465	(41,983)	Ordinary
irities Australia Ltd	Sell	90,984	(10,567)	Ordinary
UBS Securities Australia Ltd	Buy	33,316	3,874	Ordinary
UBS Securities Australia Ltd	Sell	2.618.420	(304 241)	Ordinary
IRS AG London Branch	7	0	(1)	Ordinary
LOTIOUS DISTRICT	Harri Control	5	Lines of	Ordinary
UBS AG, London Branch	iiac	200,71	(2,037)	Ordinary
UBS AIG, LONGON Branch	seil	1/6//1	(7,086)	Ordinary
London Branch	Sell	5,108	(594)	Ordinary
JBS AG, London Branch	Sell	4,008	(465)	Ordinary
URS AG Tondon Branch	lla?	15 748	(1,828)	Ordinary
Do AG London Branch	103	099	(25)	Cadinas
LOUIDOIT BIAITCI	Jac.	2000	(ca)	Ordinary
UBS AG, London Branch	Sell	224	(56)	Ordinary
UBS AG, London Branch	Sell	456	(23)	Ordinary
IRS AG London Branch	lay.	980 9		
Color District	100		(706)	Ordinass
The second second		COO'S	(206)	Ordinary
UBS AG, LONGON Branch	Sell	680'9	(706)	Ordinary Ordinary

USS Securities Australia Ltd   Sell   USS Securities Australia Ltd   Sell   USS Securities Australia Ltd   Sell   Sell   Securities Australia Ltd   Sell   Sell   Securities Australia Ltd   Sell   Sell   Securities Australia Ltd   Sell   Sell   Securities Australia Ltd   Sell   Sell   Securities Australia Ltd   Sell   Sell   Securities Australia Ltd   Sell   Sell   Securities Australia Ltd   Sell   Sell   Securities Australia Ltd   Sell   Sell   Sell   Securities Australia Ltd   Sell   Sell   Securities Australia Ltd   Sell   Sell   Securities Australia Ltd   Sell   Securities Australia Ltd   Sell   Sell   Sell   Securities Australia Ltd   Sell   Sell   Securities Australia Ltd   Sell   Sell   Securities Australia Ltd   Sell   Sell   Sell   Securities Australia Ltd   Sell   Sell   Sell   Sell   Sell   Securities Australia Ltd   Sell   Sell   Securities Australia Ltd   Sell   Securities Australia Ltd   Sell   Securities Australia Ltd   Sell   Securities Australia Ltd   Sell   Securities A	06-Nov-14	UBS Securities Australia Ltd	Sell	1,50	(39)	Contract of
10   Securiter Autorials Ltd   568   568   569	14	UBS Securities Australia Ltd	Sell	405	(47)	Ordinary
BS Securiter Authrish LTG         Seid         7.15         CRS           BS Securiter Authrish LTG         Seid         1.76         (1.99)           BS Securiter Authrish LTG         Seid         1.76         (1.99)           BS Securiter Authrish LTG         Seid         4.78         (1.99)           BS Securiter Authrish LTG         Seid         4.78         (1.99)           BS Securiter Authrish LTG         Seid         4.78         (1.90)           BS Securiter Authrish LTG         Seid         4.78         (1.90)           BS Securiter Authrish LTG         Seid         4.78         (1.90)           BS Securiter Authrish LTG         Seid         5.178         (1.90)           BS Securiter Authrish LTG         Seid         5.17	14	UBS Securities Australia Ltd	Sell	69	(8)	Ordinary
(1) St. Scuringer, Australia Ltd         544         1 1494         (199           (10) St. Scuringer, Australia Ltd         544         1 1494         (199           (10) St. Scuringer, Australia Ltd         544         1 77         (199           (10) St. Scuringer, Australia Ltd         544         4 57.4         (199           (10) St. Scuringer, Australia Ltd         544         4 57.4         (190           (10) St. Scuringer, Australia Ltd         544         4 57.4         (190           (10) St. Scuringer, Australia Ltd         544         4 57.4         (190           (10) St. Scuringer, Australia Ltd         544         5 5.14         (190           (10) St. Scuringer, Australia Ltd         544         5 5.14         (190           (10) St. Scuringer, Australia Ltd         544         5 5.14         (190           (10) St. Scuringer, Australia Ltd         544         5 5.14         (190           (10) St. Scuringer, Australia Ltd         544         5 5.14         (190           (10) St. Scuringer, Australia Ltd         544         5 5.14         (170           (10) St. Scuringer, Australia Ltd         544         5 5.14         (170           (10) St. Scuringer, Australia Ltd         544         5 5.14         (170	÷ ;	UBS Securities Australia Ltd	Sell	2,155	(250)	Ordinary
80   Securiter Autorials Ltd   54   11   15   11   11   11   11   11	7	UBS Securities Australia Ltd	Sell	1,198	(139)	Ordinary
100   100	4 5	UBS Securities Australia Ltd	Sell	164	(61)	Ordinary
1955 Securine Australia Ltd   541   542   545	-14	UBS Securities Australia Itd	- Sell	2/1	(07)	Ordinary
USS Securine Australia Ltd   Sel	-14	UBS Securities Australia Ltd	Sell	44.514	(5, 154)	Ordinary
Bits Securine Automital total Serial Serial   Securine Automital total Serial   Securine Automital total Serial   Securine Automital total Secur	-14	UBS Securities Australia Ltd	Sell	3,517	(408)	Ordinary
Bits Securine Australia Led         Sei         483         (5.6)           106 Securine Australia Led         Sei         2,196         (2.20)           106 Securine Australia Led         Sei         2,196         (2.20)           106 Securine Australia Led         Sei         3,19         (6.356)           106 Securine Australia Led         Sei         3,19         (6.356)           106 Securine Australia Led         Sei         2,980         (6.114)           106 Securine Australia Led         Sei         2,980         (6.114)           106 Securine Australia Led         Sei         2,980         (6.114)           107 Securine Australia Led         Sei         2,980         (6.114)           108 Securine Australia Led         Sei         2,980         (6.210)           108 Securine Australia Led         Sei         3,790         (7.20)           108 Securine Australia Led         Sei         3,790         (7.20)           109 Securine Australia Led         Sei         3,790         (7.20)           109 Securine Australia Led         Sei         3,790         (7.20)           109 Securine Australia Led         Sei         3,700         (7.20)           100 Securine Australia Led         Sei	-14	UBS Securities Australia Ltd	Self	95	(11)	Ordinary
(1) S. Securities Autorials Ltd         Sell         7,155         (25)           (1) S. Securities Autorials Ltd         Sell         5,134         (5,99)           (1) S. Securities Autorials Ltd         Sell         5,134         (5,99)           (1) S. Securities Autorials Ltd         Sell         7,693         (6,99)           (1) S. Securities Autorials Ltd         Sell         7,693         (6,29)           (1) S. Securities Autorials Ltd         Sell         8,690         (7,29)	14	UBS Securities Australia Ltd	Sell	483	(29)	Ordinary
IBS Securities Autorials Ltd   Sell   1 4	UBS Securities Australia Ind	lias	2,155	(250)	Ordinary	
BES Securines Automita Let   Self   Self   Self   Self   Securines Automita Let   Self   Se	4-	UBS Securities Australia Ltd	Sell	55.134	(6) 396	Ordinary
(BS) Securine Amerina Ltd         SMI         7,653         (688)         (688)           (BS) Securine Amerina Ltd         SMI         2,664         (2014) </td <td>-14</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>319</td> <td>(37)</td> <td>Ordinary</td>	-14	UBS Securities Australia Ltd	Sell	319	(37)	Ordinary
URS Securities Autorials (14)   Self   76,956   G.450   G.45	-14	UBS Securities Australia Ltd	Sell	7,663	(888)	Ordinary
URS Securities Australia Ltrd         Sell         2.6,874         (3.19)           URS Securities Australia Ltrd         Sell         2.444         (3.89)           URS Securities Australia Ltrd         Sell         65.778         (4.29)           URS Securities Australia Ltrd         Sell         65.778         (7.258)           URS Securities Australia Ltrd         Sell         65.778         (7.258)           URS Securities Australia Ltrd         Sell         65.778         (7.258)           URS Securities Australia Ltrd         Sell         6.728         (7.258)           URS Securities Australia Ltrd         Sell         7.258         (7.258)           URS Securities Australia Ltrd         Sell         7.278         (7.258)           URS Securities Australia Ltrd         Sell         7.278         (7.258)           URS Securities Australia Ltrd         Sell         5.81         (7.258)	14	UBS Securities Australia Ltd	Sell	2,986	(346)	Ordinary
URS Securities Australia Ltd         Sell         7,548         (842)           URS Securities Australia Ltd         Sell         853         (102)           URS Securities Australia Ltd         Sell         873         (102)           URS Securities Australia Ltd         Sell         873         (102)           URS Securities Australia Ltd         Sell         873         (102)           URS Securities Australia Ltd         Sell         875         (102)           URS Securities Australia Ltd         Sell         875         (102)           URS Securities Australia Ltd         Sell         876         (102)           URS Securities Australia Ltd         Sell         876         (102)           URS Securities Australia Ltd         Sell         876         (102)           URS Securities Australia Ltd         Sell         870         (102)           URS Securities Australia Ltd         Sell	-14	UBS Securities Australia Ltd	Sell	26,874	(3,114)	Ordinary
USS Securities Australia Ltd         Sell         65.92         (4.25)         (9.87)           USS Securities Australia Ltd         Sell         8.73         (10.02)           USS Securities Australia Ltd         Sell         3.70         (4.35)           USS Securities Australia Ltd         Sell         1.9         (2.216)           USS Securities Australia Ltd         Sell         1.0         (3.28)           USS Securities Australia Ltd         Sell         1.1         (3.28)         (3.28)           USS Securities Australia Ltd         Sell         1.1         (3.28)         (3.28)           USS Securities Australia Ltd         Sell         1.1         (3.28)         (3.28)           USS Securities Australia Ltd         Sell         1.0         (3.28)         (3.28)           USS Securities Australia Ltd         Sell         3.0         (3.29)         (3.28)           USS Securities Australia Ltd         Sell         3.0         (3.28)         (3.28) <t< td=""><td>-14</td><td>UBS Securities Australia Ltd</td><td>Sell</td><td>2,494</td><td>(289)</td><td>Ordinary</td></t<>	-14	UBS Securities Australia Ltd	Sell	2,494	(289)	Ordinary
USS Securities Australia Ltd   Sell   Sell   Securities Australia Ltd   Sell   Sell   Securities Australia Ltd   Sell   Securities Australia Ltd   Sell   Securities Australia Ltd   Sell   Securities Australia Ltd   Sell   Sell   Securities Australia Ltd   Sell   Sell   Sell   Securities Australia Ltd   Sell   Sell   Securities Australia Ltd   Sell   Sell   Securities Australia Ltd   Sell   Sell   Sell   Securities Australia Ltd   Sell   Sell   Securities Australia Ltd   Sell   Sell   Securities Australia Ltd   Sell   Sell   Sell   Securities Australia Ltd   Sell   Sell   Sell   Securities Australia Ltd   Sell   Sell   Securities Australia Ltd   Sell   Sell   Sell   Securities Australia Ltd   Sell   Sell   Sell   Securities Australia Ltd   Sell   Sell   Sell   Sell   Securities Australia Ltd   Sell   Sell   Sell   Securities Australia Ltd   Sell   Sell   Sell   Sell   Sell   Sell   Securities Australia Ltd   Sell   Sel	-14	UBS Securities Australia Ltd	Sell	867'/	(844)	Ordinary
UBS Securities Australia Ltd         Seil         37-90         (435)           UBS Securities Australia Ltd         Seil         189         (231)           UBS Securities Australia Ltd         Seil         189         (231)           UBS Securities Australia Ltd         Seil         189         (231)           UBS Securities Australia Ltd         Seil         20-34         (231)           UBS Securities Australia Ltd         Seil         20-36         (239)           UBS Securities Australia Ltd         Seil         20-36         (239)           UBS Securities Australia Ltd         Seil         20-36         (239)           UBS Securities Australia Ltd         Seil         88-96         (1,630)           UBS Securities Australia Ltd         Seil         66-95         (1,630)           UBS Securities Australia Ltd         Seil         66-95         (1,630)           UBS Securities Australia Ltd         Seil         66-95         (1,630)           UBS Securities Australia Ltd         Seil         36-73         (1,630)           UBS Securities Australia Ltd         Seil         36-73         (1,630)           UBS Securities Australia Ltd         Seil         36-73         (1,630)           UBS Securities Austr	-14	UBS Securities Australia Ltd	Sell	879	(102)	Ordinary
UBS Securities Australia Ltd         Sell         6.94 Pm         (5) Bd           UBS Securities Australia Ltd         Sell         1.92         (2) Bd           UBS Securities Australia Ltd         Sell         1.93         (2) Bd           UBS Securities Australia Ltd         Sell         1.91         (2) Bd           UBS Securities Australia Ltd         Sell         1.91         (2) Bd           UBS Securities Australia Ltd         Sell         1.91         (2) Bd           UBS Securities Australia Ltd         Sell         1.92         (2.52)           UBS Securities Australia Ltd         Sell         1.92         (2.52)           UBS Securities Australia Ltd         Sell         1.93         (1.032)           UBS Securities Australia Ltd         Sell         2.87         (1.032)           UBS Securities Australia Ltd         Sell         2.87         (1.032)           UBS Securities Australia Ltd         Sell         2.87         (1.032)           UBS Securities Australia Ltd         Sell         2.23         (1.691)           UBS Securities Australia Ltd         Sell         2.23         (1.691)           UBS Securities Australia Ltd         Sell         2.23         (1.691)           UBS Securities Aus	-14	UBS Securities Australia Ltd	Seti	3,750	(435)	Ordinary
UBS Securities Australia Ltd         Sell         198         (2)           UBS Securities Australia Ltd         Sell         43         (6)           UBS Securities Australia Ltd         Sell         19,102         (2,16)           UBS Securities Australia Ltd         Sell         2,034         (6,28)           UBS Securities Australia Ltd         Sell         8,295         (6,28)           UBS Securities Australia Ltd         Sell         8,295         (7,29)           UBS Securities Australia Ltd         Sell         8,278         (7,29)           UBS Securities Australia Ltd         Sell         3,270         (7,29)           UBS Securities Australia Ltd         Sell         3,272         (7,29)           UBS Securities Australia Ltd         Sell         3,272         (7,29)           UBS Securities Australia Ltd         Sell         3,272         (7,29)           UBS Securities Australi	-14	UBS Securities Australia Ltd	Seil	63,478	(7,364)	Ordinary
UBS Securities Autarials Ltd         Sell         784         (5)           UBS Securities Autarials Ltd         Sell         19107         (2,216)           UBS Securities Autarials Ltd         Sell         19,002         (2,216)           UBS Securities Autarials Ltd         Sell         2,034         (7,276)           UBS Securities Autarials Ltd         Sell         28,232         (1,022)           UBS Securities Autarials Ltd         Sell         28,232         (1,022)           UBS Securities Autarials Ltd         Sell         28,232         (1,022)           UBS Securities Autarials Ltd         Sell         1,156         (1,022)           UBS Securities Autarials Ltd         Sell         2,233         (1,022)           UBS Securities Autarials Ltd         Sell         1,255         (1,022)           UBS Securities Autarials Ltd         Sell         1,233         (1,022)           UBS Securities Autarials Ltd         Sell         1,156         (1,022)           UBS Securities Autarials Ltd         Sell         3,702         (1,022)           UBS Securities Autarials Ltd         Sell         3,702         (1,022)           UBS Securities Autarials Ltd         Sell         3,702         (1,022)	-14	UBS Securities Australia Ltd	Sell	198	(23)	Ordinary
USS Securities Autorials Ltd         Sell         19,102         (2.94)           UBS Securities Autorials Ltd         Sell         2,034         (2.95)           UBS Securities Autorials Ltd         Sell         39,003         (2.53)           UBS Securities Autorials Ltd         Sell         39,003         (2.53)           UBS Securities Autorials Ltd         Sell         4,305         (2.53)           UBS Securities Autorials Ltd         Sell         4,305         (2.53)           UBS Securities Autorials Ltd         Sell         4,305         (1.256)           UBS Securities Autorials Ltd         Sell         4,305         (1.652)           UBS Securities Autorials Ltd         Sell         1,155         (1.650)           UBS Securities Autorials Ltd         Sell         1,156         (1.650)           UBS Securities Autorials Ltd         Sell         1,156         (1.650)           UBS Securities Autorials Ltd         Sell         1,158         (1.650)           UBS Securities Autorials Ltd         Sell         1,158         (1.650)           UBS Securities Autorials Ltd         Sell         1,158         (1.650)           UBS Securities Autorials Ltd         Sell         2,234         (7.50)           UB	4 5	UBS Securities Australia Ltd	lies i	43	(2)	Ordinary
USS Securities Australia Ltd         Sell         2,034         4,250           UBS Securities Australia Ltd         Sell         8,995         (1,529)           UBS Securities Australia Ltd         Sell         8,996         (1,032)           UBS Securities Australia Ltd         Sell         8,996         (1,032)           UBS Securities Australia Ltd         Sell         4,875         (1,032)           UBS Securities Australia Ltd         Sell         4,306         (1,032)           UBS Securities Australia Ltd         Sell         4,306         (1,032)           UBS Securities Australia Ltd         Sell         2,233         (1,458)           UBS Securities Australia Ltd         Sell         2,233         (1,458)           UBS Securities Australia Ltd         Sell         3,702         (4,400)           UBS Securities Australia Ltd         Sell         3,702         (4,400)           UBS Securities Australia Ltd         Sell         3,702         (4,400)           UBS Securities Australia Ltd         Sell         3,702         (4,300)           UBS Securities Australia Ltd         Sell         3,702         (4,300)           UBS Securities Australia Ltd         Sell         5,802         (7,500)	-14	UBS Securities Australia 11d	130	10 103	(16)	Ordinary
UBS Securities Australia Ltd         Sell         45.55         (4.597)           UBS Securities Australia Ltd         Sell         42.55         (6.253)           UBS Securities Australia Ltd         Sell         43.55         (7.022)           UBS Securities Australia Ltd         Sell         (1.022)         (1.022)           UBS Securities Australia Ltd         Sell         (1.25)         (1.022)           UBS Securities Australia Ltd         Sell         (1.25)         (1.523)           UBS Securities Australia Ltd         Sell         (1.25)         (1.650)           UBS Securities Australia Ltd         Sell         (2.473)         (2.593)           UBS Securities Australia Ltd         Sell         (2.473)         (4.500)           UBS Securities Australia Ltd         Sell         (2.473)         (4.500)           UBS Securities Australia Ltd         Sell         (3.673)         (4.500)           UBS Securities Australia Ltd         Sell         (3.753)         (4.500)           UBS Securities Australia Ltd         Sell         (3.754)         (4.500)           UBS Securities Australia Ltd         Sell         (3.754)         (4.500)           UBS Securities Australia Ltd         Sell         (3.754)         (4.500) <td>-14</td> <td>UBS Securities Australia Ltd</td> <td>les</td> <td>2.034</td> <td>(9:2/2)</td> <td>Ordinary</td>	-14	UBS Securities Australia Ltd	les	2.034	(9:2/2)	Ordinary
UBS Securities Australia Ird         SFII         45.255         (5.239)           UBS Securities Australia Ird         SFII         8.896         (1.032)           UBS Securities Australia Ird         SFII         6.65         (1.032)           UBS Securities Australia Ird         SFII         6.65         (1.032)           UBS Securities Australia Ird         SFII         6.65         (1.032)           UBS Securities Australia Ird         SFII         4.305         (1.032)           UBS Securities Australia Ird         SFII         24.734         (1.659)           UBS Securities Australia Ird         SFII         24.734         (1.659)           UBS Securities Australia Ird         SFII         1.156         (1.630)           UBS Securities Australia Ird         SFII         1.158         (1.630)           UBS Securities Australia Ird         SFII         SFII         (1.630)           UBS Securities Australia Ird         SFII         SFII         (1.630)	-14	UBS Securities Australia Ltd	Sell Sell	19,603	(4,597)	Ordinary
UBS Securities Australia Ltd         Sell         2292         (3.28a)           UBS Securities Australia Ltd         Sell         8.86         (1.020)           UBS Securities Australia Ltd         Sell         11,165         (1.050)           UBS Securities Australia Ltd         Sell         4,305         (1.050)           UBS Securities Australia Ltd         Sell         4,305         (1.650)           UBS Securities Australia Ltd         Sell         1,2,533         (1.450)           UBS Securities Australia Ltd         Sell         2,233         (1.650)           UBS Securities Australia Ltd         Sell         3,702         (4,500)           UBS Securities Australia Ltd         Sell         3,702         (4,500)           UBS Securities Australia Ltd         Sell         1,180         (1,691)           <	-14	UBS Securities Australia Ltd	Sell	45,255	(5,253)	Ordinary
UBS Securities Australia Ltd         Sell         8.956         (1,022)           UBS Securities Australia Ltd         Sell         6.465         (1,022)           UBS Securities Australia Ltd         Sell         1,1,165         (1,259)           UBS Securities Australia Ltd         Sell         43.55         (5.627)           UBS Securities Australia Ltd         Sell         2,4,734         (7.57)           UBS Securities Australia Ltd         Sell         3,4,500         (4,000)           UBS Securities Australia Ltd         Sell         3,4,500         (4,000)           UBS Securities Australia Ltd         Sell         3,1,80         (1,631)           UBS Securities Australia Ltd         Sell         3,1,80         (1,631)           UBS Securities Australia Ltd         Sell         3,1,80         (1,691)           UBS Securities Australia Ltd         Sell         3,178         (1,691)           UBS Securities Australia Ltd         Sell         3,178         (1,691)           UBS Securities Australia Ltd         Sell         3,172         (4,300)           UBS Securities Australia Ltd         Sell         3,172         (4,300)           UBS Securities Australia Ltd         Sell         3,172         (7,520)	-14	UBS Securities Australia Ltd	Sell	28,292	(3,284)	Ordinary
UBS Securities Australia Ltd         Sell         (1)           UBS Securities Australia Ltd         Sell         (1,265)           UBS Securities Australia Ltd         Sell         (1,265)           UBS Securities Australia Ltd         Sell         (1,253)           UBS Securities Australia Ltd         Sell         (1,253)           UBS Securities Australia Ltd         Sell         (1,253)           UBS Securities Australia Ltd         Sell         (1,257)           UBS Securities Australia Ltd         Sell         (1,273)           UBS Securities Australia Ltd         Sell         (1,273)           UBS Securities Australia Ltd         Sell         (1,273)           UBS Securities Australia Ltd         Sell         (1,578)           UBS Securities Australia Ltd         Sell         (1,578)           UBS Securities Australia Ltd         Sell         (1,578)           UBS Securities Australia Ltd         Sell         (2,520)           UBS Securities Australia Ltd         Sell         (2,520)           UBS Securit	41.	UBS Securities Australia Ltd	Sell	8,896	(1,032)	Ordinary
UBS Securities Australia Ltd         Sell         4,305         (1,25)           UBS Securities Australia Ltd         Sell         4,305         (6,620)           UBS Securities Australia Ltd         Sell         4,305         (6,620)           UBS Securities Australia Ltd         Sell         1,2533         (1,458)           UBS Securities Australia Ltd         Sell         1,2573         (1,691)           UBS Securities Australia Ltd         Sell         1,1576         (4,600)           UBS Securities Australia Ltd         Sell         1,1576         (4,600)           UBS Securities Australia Ltd         Sell         1,158         (1,631)           UBS Securities Australia Ltd         Sell         1,158         (1,631)           UBS Securities Australia Ltd         Sell         1,158         (1,630)           UBS Securities Australia Ltd         Sell         2,158         (2,649)           UBS Securities Australia Ltd         Sell         2,232         (2,649)	4 5	UBS Securities Australia Ltd	Jen Co	6,465	(750)	Ordinary
UBS Securities Australia Ltd         Sell         445,78         (1,620)           UBS Securities Australia Ltd         Sell         1,355         (1,489)           UBS Securities Australia Ltd         Sell         2,233         (2,871)           UBS Securities Australia Ltd         Sell         34,500         (4,000)           UBS Securities Australia Ltd         Sell         31,538         (1,601)           UBS Securities Australia Ltd         Sell         31,538         (1,601)           UBS Securities Australia Ltd         Sell         3,702         (4,000)           UBS Securities Australia Ltd         Sell         3,702         (4,300)           UBS Securities Australia Ltd         Sell         3,702         (1,81)           UBS Securities Australia Ltd         Sell         3,702         (1,80)           UBS Securities Australia Ltd         Sell         3,702         (1,80)           UBS Securities Australia Ltd         Sell         3,702         (3,60)           UBS Securities Australia Ltd         Sell         3,702         (3,60)           UBS Securities Australia Ltd         Sell         3,702         (3,60)           UBS Securities Australia Ltd         Sell         2,533         (7,50)           U	14	UBS Securities Australia Ltd	Sei	11 165	(1)	Ordinary
UBS Securities Australia Ltd         Sell         1,2553         (1,588)           UBS Securities Australia Ltd         Sell         24,734         (2,87)           UBS Securities Australia Ltd         Sell         24,734         (2,87)           UBS Securities Australia Ltd         Sell         34,500         (4,000)           UBS Securities Australia Ltd         Sell         31,838         (3,633)           UBS Securities Australia Ltd         Sell         3,722         (4,300)           UBS Securities Australia Ltd         Sell         3,722         (5,61)           UBS Securities Australia Ltd         Sell         5,58         (5,61)           UBS Securities Australia Ltd         Sell         6,530         (7,34)           UBS Securities Australia Ltd         Sell         6,530         (7,34)	-14	UBS Securities Australia Ltd	Seil	48 578	(05.47)	Ordinary
UBS Securities Australia Ltd         Sell         1,253         (1,489)           UBS Securities Ltd         Sell         1,233         (1,489)           UBS Securities Australia Ltd         Sell         1,233         (2,591)           UBS Securities Australia Ltd         Sell         1,189         (1,691)           UBS Securities Australia Ltd         Sell         1,189         (1,631)           UBS Securities Australia Ltd         Sell         1,189         (1,632)           UBS Securities Australia Ltd         Sell         1,189         (1,181)           UBS Securities Australia Ltd         Sell         1,189         (1,181)           UBS Securities Australia Ltd         Sell         1,153         (2,542)           UBS Securities Australia Ltd         Sell         1,153         (2,542)           UBS Securities Australia Ltd         Sell         1,53         (2,542)           UBS Securitie	-14	UBS Securities Australia Ltd	Sell	4,305	(200)	Ordinary
UBS Securities Australia Ltd         Sell         2.734         (2.59)           UBS Securities Australia Ltd         Sell         1.45.76         (1.691)           UBS Securities Australia Ltd         Sell         3.702         (4.000)           UBS Securities Australia Ltd         Sell         3.702         (4.000)           UBS Securities Australia Ltd         Sell         3.702         (4.000)           UBS Securities Australia Ltd         Sell         3.702         (4.00)           UBS Securities Australia Ltd         Sell         3.702         (4.00)           UBS Securities Australia Ltd         Sell         3.702         (4.00)           UBS Securities Australia Ltd         Sell         3.748         (5.649)           UBS Securities Australia Ltd         Sell         1.53         (5.649)           UBS Securities Australia Ltd         Sell         1.67         (4.00)           UBS Securities Australia Ltd         Sell         1.65         (5.649)           UBS Securities Australia Ltd         Sell         1.67         (5.601)           UBS Securities Australia Ltd         Sell         6.53         (7.26)           UBS Securities Australia Ltd         Sell         6.53         (7.26)           UBS Secur	44	UBS Securities Australia Ltd	Sell	12,553	(1,458)	Ordinary
UBS Securities Australia Ltd         Sell         14,576         (1,691)           UBS Securities Australia Ltd         Sell         34,500         (4,000)           UBS Securities Australia Ltd         Sell         31,538         (1,601)           UBS Securities Australia Ltd         Sell         1,180         (1,37)           UBS Securities Australia Ltd         Sell         1,180         (1,37)           UBS Securities Australia Ltd         Sell         3,229         (3,59)           UBS Securities Australia Ltd         Sell         1,180         (1,131)           UBS Securities Australia Ltd         Sell         1,178         (1,540)           UBS Securities Australia Ltd         Sell         1,678         (1,640)           UBS Securities Australia Ltd         Sell         2,650         (2,530)           UBS Securities Australia Ltd         Sell         1,650         (1,640) <t< td=""><td>4 3</td><td>UBS Securities Australia Ltd</td><td>Sell</td><td>24,734</td><td>(2,871)</td><td>Ordinary</td></t<>	4 3	UBS Securities Australia Ltd	Sell	24,734	(2,871)	Ordinary
UBS Securities Australia Ltd         Sell         34,500         (4,000)           UBS Securities Australia Ltd         Sell         31,538         (4,500)           UBS Securities Australia Ltd         Sell         31,538         (4,500)           UBS Securities Australia Ltd         Sell         31,732         (4,30)           UBS Securities Australia Ltd         Sell         3,722         (4,30)           UBS Securities Australia Ltd         Sell         3,229         (7,31)           UBS Securities Australia Ltd         Sell         3,722         (4,30)           UBS Securities Australia Ltd         Sell         1,67,38         (7,542)           UBS Securities Australia Ltd         Sell         1,67,38         (7,542)           UBS Securities Australia Ltd         Sell         1,536         (7,561)           UBS Securities Australia Ltd         Sell         1,536         (7,561)           UBS Securities Australia Ltd         Sell         1,536         (7,561)           UBS Securities Australia Ltd         Sell         2,530         (7,58)           UBS Securities Australia Ltd         Sell         1,720         (7,541)           UBS Securities Australia Ltd         Sell         1,720         (7,542)	124	UBS Securities Australia Ltd	Sell	2,233	(523)	Ordinary
UBS Securities Australia Ltd         Sell         31,538         (4,500)           UBS Securities Australia Ltd         Sell         1,180         (137)           UBS Securities Australia Ltd         Sell         1,180         (130)           UBS Securities Australia Ltd         Sell         1,538         (1,540)           UBS Securities Australia Ltd         Sell         2,153         (1,540)           UBS Securities Australia Ltd         Sell         2,153         (2,560)           UBS Securities Australia Ltd         Sell         2,276         (7,77)           UBS Securities Australia Ltd         Sell         2,276         (7,24)           UBS Securities Australia Ltd         Sell         2,276         (7,24)           UBS Securi	1 2	Ups counties Australia Ltd	Nell Sell	14,576	(1,691)	Ordinary
USS Securities Australia Ltd         Seril         1,180         (137)           UBS Securities Australia Ltd         Seril         1,180         (137)           UBS Securities Australia Ltd         Seril         3,722         (430)           UBS Securities Australia Ltd         Seril         3,723         (275)           UBS Securities Australia Ltd         Seril         1,738         (7,500)           UBS Securities Australia Ltd         Seril         1,673         (1,544)           UBS Securities Australia Ltd         Seril         1,673         (1,544)           UBS Securities Australia Ltd         Seril         1,650         (2,561)           UBS Securities Australia Ltd         Seril         1,650         (7,580)           UBS Securities Australia Ltd         Seril         6,530         (7,580)           UBS Securities Australia Ltd         Seril         2,530         (7,580)           UBS Securities Australia Ltd         Seril         1,377         (7,590)           UBS Securities Australia Ltd         Seril         1,530         (7,580)           UBS Securities Australia Ltd         Seril         1,500         (1,591)           UBS Securities Australia Ltd         Seril         6,588         (8000)	14	UBS Securities Australia Ltd	Self	34,500	(4,000)	Ordinary
UBS Securities Australia Ltd         Self         3.702         (4.30)           UBS Securities Australia Ltd         Self         3.723         (1.381)           UBS Securities Laturalia Ltd         Self         3.229         (1.549)           UBS Securities Australia Ltd         Self         3.702         (3.50)           UBS Securities Australia Ltd         Self         3.702         (4.30)           UBS Securities Australia Ltd         Self         3.702         (5.661)           UBS Securities Australia Ltd         Self         1.9596         (5.661)           UBS Securities Australia Ltd         Self         1.9596         (5.601)           UBS Securities Australia Ltd         Self         2.320         (7.58)           UBS Securities Australia Ltd         Self         2.325         (7.58)           UBS Securities Australia Ltd         Self         2.325         (7.58)           UBS Securities Australia Ltd         Self         2.325         (7.58)           UBS Securities Australia Ltd         Self         2.255         (7.54)           UBS Securities Australia Ltd         Self         (7.32)         (7.32)           UBS Securities Australia Ltd         Self         (6.31)           UBS Securities Australia Ltd	-14	UBS Securities Australia Ltd	Sell	1180	(137)	Ordinary
UBS Securities Australia Ltd         Sell         10,168         (1,181)           UBS Securities Australia Ltd         Sell         3,122         (375)           UBS Securities Australia Ltd         Sell         1,533         (3,500)           UBS Securities Australia Ltd         Sell         1,533         (1,500)           UBS Securities Australia Ltd         Sell         4300         (1,500)           UBS Securities Australia Ltd         Sell         4357         (430)           UBS Securities Australia Ltd         Sell         4357         (430)           UBS Securities Australia Ltd         Sell         7158         (5,500)           UBS Securities Australia Ltd         Sell         215.38         (7,500)           UBS Securities Australia Ltd         Sell         215.38         (7,500)           UBS Securities Australia Ltd         Sell         22,756         (7,500)           UBS Securities Australia Ltd         Sell         22,255         (7,500)           UBS Securities Australia Ltd         Sell         22,265         (7,24)           UBS Securities Australia Ltd         Sell         22,265         (7,34)           UBS Securities Australia Ltd         Sell         (7,34)           UBS Securities Australia Lt	-14	UBS Securities Australia Ltd	Sell	3 702	(430)	Ordinary
UBS Securities Australa Ltd         Sell         3.259         (375)           UBS Securities Australa Ltd         Sell         3.1418         (3.649)           UBS Securities Australa Ltd         Sell         1.658         (1.344)           UBS Securities Australa Ltd         Sell         1.658         (1.344)           UBS Securities Australa Ltd         Sell         1.658         (1.661)           UBS Securities Australa Ltd         Sell         1.9596         (2.766)           UBS Securities Australa Ltd         Sell         2.530         (7.661)           UBS Securities Australa Ltd         Sell         2.530         (7.651)           UBS Securities Australa Ltd         Sell         1.3318         (1.544)           UBS Securities Australa Ltd         Sell         2.2567         (2.530)           UBS Securities Australa Ltd         Sell         1.3318         (1.545)           UBS Securities Australa Ltd         Sell         1.3218         (1.540)           UBS Securities Australa Ltd         Sell         1.320         (1.340)           UBS Securities Australa Ltd         Sell         1.650         (1.340)           UBS Securities Australa Ltd         Sell         (1.340)           UBS Securities Australa Ltd	14	UBS Securities Australia Ltd	Sell	10,168	(1,181)	Ordinary
UBS Securities Australia Ltd         Sell         31,418         (3,649)           UBS Securities Australia Ltd         Sell         21,553         (2,620)           UBS Securities Australia Ltd         Sell         16,738         (1,942)           UBS Securities Australia Ltd         Sell         568         (6,661)           UBS Securities Australia Ltd         Sell         13,596         (7,061)           UBS Securities Australia Ltd         Sell         13,596         (7,061)           UBS Securities Australia Ltd         Sell         6,530         (7,061)           UBS Securities Australia Ltd         Sell         2,647         (7,061)           UBS Securities Australia Ltd         Sell         2,653         (7,583)           UBS Securities Australia Ltd         Sell         2,653         (7,543)           UBS Securities Australia Ltd         Sell         1,650         (7,743)           UBS Securities Australia Ltd         Sell         6,334         (7,543)           UBS Securities Australia Ltd         Sell         6,334         (7,343)           UBS Securities Australia Ltd         Sell         6,334         (7,343)           UBS Securities Australia Ltd         Sell         (6,001)           UBS Securities Aus	-14	UBS Securities Australia Ltd	Sell	3,229	(375)	Ordinary
UBS Securities Australia Ltd         Sell         1,5329         (520)           UBS Securities Australia Ltd         Sell         16,733         (1,542)           UBS Securities Australia Ltd         Sell         43570         (430)           UBS Securities Australia Ltd         Sell         43575         (600)           UBS Securities Australia Ltd         Sell         71539         (601)           UBS Securities Australia Ltd         Sell         71539         (7,206)           UBS Securities Australia Ltd         Sell         71539         (7,007)           UBS Securities Australia Ltd         Sell         72,265         (7,540)           UBS Securities Australia Ltd         Sell         77,76         (7,540)           UBS Securities Australia Ltd         Sell         7,776         (7,540)           UBS Securities Australia Ltd         Sell         7,776         (7,240)           UBS Securities Australia Ltd         Sell         7,776         (7,240)	-14	UBS Securities Australia Ltd	Sell	31,418	(3.649)	Ordinary
UBS Securities Australia Ltd         Sell         1,528         (1,540)           UBS Securities Australia Ltd         Sell         1,702         (430)           UBS Securities Australia Ltd         Sell         1,578         (650)           UBS Securities Australia Ltd         Sell         1,956         (2,261)           UBS Securities Australia Ltd         Sell         1,556         (7,58)           UBS Securities Australia Ltd         Sell         2,530         (7,58)           UBS Securities Australia Ltd         Sell         2,530         (7,58)           UBS Securities Australia Ltd         Sell         2,533         (7,58)           UBS Securities Australia Ltd         Sell         1,3318         (1,54)           UBS Securities Australia Ltd         Sell         1,530         (1,54)           UBS Securities Australia Ltd         Sell         1,500         (1,54)           UBS Securities Australia Ltd         Sell         6,334         (7,54)           UBS Securities Australia Ltd         Sell         6,334         (7,24)           UBS Securities Australia Ltd         Sell         (8,000)           UBS Securities Australia Ltd         Sell         (6,001)           UBS Securities Australia Ltd         Sell	-14	UBS Securities Australia Ltd	Sell	2,153	(250)	Ordinary
URS Securities Autoritals Ltd         Sell         3702         (430)           URS Securities Autoritals Ltd         Sell         569         (66)           USS Securities Autoritals Ltd         Sell         (15,661)         (7,661)           URS Securities Autoritals Ltd         Sell         (5,500)         (7,269)           URS Securities Autoritals Ltd         Sell         (5,500)         (7,269)           URS Securities Autoritals Ltd         Sell         (7,769)         (7,269)           URS Securities Autoritals Ltd         Sell         (7,776)         (7,549)           URS Securities Autoritals Ltd         Sell         (7,776)         (7,526)           URS Securities Autoritals Ltd         Sell         (7,776)         (5,520)           URS Securities Autoritals Ltd         Sell         (7,776)         (5,520)           URS Securities Autoritals Ltd         Sell         (7,776)         (5,520)           URS Securities Autorials Ltd         Sell         (6,530)         (7,734)           URS Securities Autorials Ltd         Sell         (6,530)         (7,706)           URS Securities Autorials Ltd         Sell         (6,530)         (7,706)           URS Securities Autorials Ltd         Sell         (6,530)	-14	UBS Securities Australia Ltd	Sell	16,738	(1,944)	Ordinary
UBS Securities Australia Ltd         Sell         666           UBS Securities Australia Ltd         Sell         43,575         (6,601)           UBS Securities Australia Ltd         Sell         19,596         (2,276)           UBS Securities Australia Ltd         Sell         2,1538         (2,500)           UBS Securities Australia Ltd         Sell         2,627         (707)           UBS Securities Australia Ltd         Sell         2,527         (2,647)           UBS Securities Australia Ltd         Sell         2,225         (2,587)           UBS Securities Australia Ltd         Sell         2,225         (2,588)           UBS Securities Australia Ltd         Sell         2,225         (2,589)           UBS Securities Australia Ltd         Sell         2,2265         (2,589)           UBS Securities Australia Ltd         Sell         1,5501         (1,912)           UBS Securities Australia Ltd         Sell         1,5501         (1,912)           UBS Securities Australia Ltd         Sell         1,5501         (1,912)           UBS Securities Australia Ltd         Sell         1,500         (1,912)           UBS Securities Australia Ltd         Sell         (1,613)         (1,613)           UBS Securities Aus	-14	UBS Securities Australia Ltd	Sell	3,702	(430)	Ordinary
UBS Securities Australia Ltd         Seli         4.65 Feb         (5.061)           UBS Securities Australia Ltd         Seli         19.566         (7.269)           UBS Securities Australia Ltd         Seli         6.530         (7.58)           UBS Securities Australia Ltd         Seli         6.591         (7.07)           UBS Securities Australia Ltd         Seli         7.2,265         (2.87)           UBS Securities Australia Ltd         Seli         7.2,265         (2.530)           UBS Securities Australia Ltd         Seli         7.776         (5.530)           UBS Securities Australia Ltd         Seli         6.334         (7.24)           UBS Securities Australia Ltd         Seli         6.334         (7.74)           UBS Securities Australia Ltd         Seli         6.334         (7.04)           UBS Securities Australia Ltd         Seli         6.334         (7.04)           UBS Securities Australia Ltd         Seli         (6.33)         (7.24)           UBS Securities Australia Ltd         Seli         (6.33)         (7.34)           UBS Securities Australia Ltd         Seli         (6.33)         (7.34)           UBS Securities Australia Ltd         Seli         (6.33)         (7.34)           <	4	UBS Securities Australia Ltd	Sell	568	(99)	Ordinary
UBS Securities Australia Ltd         Seil         1959G         (2.26)           UBS Securities Australia Ltd         Seil         6.530         (7.26)           UBS Securities Australia Ltd         Seil         2.15.38         (2.500)           UBS Securities Australia Ltd         Seil         1.31.8         (7.54)           UBS Securities Australia Ltd         Seil         1.22.265         (7.58)           UBS Securities Australia Ltd         Seil         1.55.01         (7.54)           UBS Securities Australia Ltd         Seil         1.65.01         (7.76)         (5.53.9)           UBS Securities Australia Ltd         Seil         6.33.4         (7.76)         (5.53.0)           UBS Securities Australia Ltd         Seil         6.33.4         (7.76)         (7.34)           UBS Securities Australia Ltd         Seil         6.33.4         (7.00)           UBS Securities Australia Ltd         Seil         6.33.4         (7.00)           UBS Securities Australia Ltd         Seil         (6.530)         (7.500)           UBS Securities Australia Ltd         Seil         (7.000)         (7.500)         (7.500)           UBS Securities Australia Ltd         Seil         (7.500)         (7.500)         (7.500)         (7.500)	7	UBS Securities Australia Ltd	Sell	43,575	(5,061)	Ordinary
UBS Securities Australia Ltd         Sell         (5.50         (782)           UBS Securities Australia Ltd         Sell         (5.50         (707)           UBS Securities Australia Ltd         Sell         (5.52         (707)           UBS Securities Australia Ltd         Sell         (2.583)         (7.543)           UBS Securities Australia Ltd         Sell         (2.583)         (7.545)           UBS Securities Australia Ltd         Sell         (7.776)         (7.545)           UBS Securities Australia Ltd         Sell         (7.776)         (7.746)           UBS Securities Australia Ltd         Sell         (7.776)         (7.746)           UBS Securities Australia Ltd         Sell         (7.776)         (7.740)           UBS Securities Australia Ltd         Sell         (7.720)         (7.720)           UBS Securities Australia Ltd         Sell         (7.720)         (7.200)           UBS Securities Australia Ltd         Sell         (7.720)         (7.320)	4:	UBS Securities Australia Ltd	Sell	19,596	(2,276)	Ordinary
UBS Securities Australia Ltd         Sell         2.5.00           UBS Securities Australia Ltd         Sell         2.5.00           UBS Securities Australia Ltd         Sell         2.2.265         (2.8.7)           UBS Securities Australia Ltd         Sell         2.2.265         (2.5.83)           UBS Securities Australia Ltd         Sell         2.2.265         (2.5.83)           UBS Securities Australia Ltd         Sell         4.5.501         (1.5.50)           UBS Securities Australia Ltd         Sell         6.334         (7.94)           UBS Securities Australia Ltd         Sell         6.888         (800)           UBS Securities Australia Ltd         Sell         (1.7.20         (2.00)           UBS Securities Australia Ltd         Sell         (6.334)         (7.34)           UBS Securities Australia Ltd         Sell         (6.00)         (6.00)           UBS Securities Australia Ltd         Sell         (6.00)         (6.00)           UBS Securities Australia Ltd         Sell         (6.00)         (6.00)           UBS Securities Australia Ltd         Sell         (6.00)         (6.13)           UBS Securities Australia Ltd         Sell         (6.13)           UBS Securities Australia Ltd         Sell	4 :	UBS Securities Australia Ltd	Seil	6,530	(758)	Ordinary
UBS Securities Australia Ltd         Sell         6,591         (7,547)           UBS Securities Australia Ltd         Sell         13,318         (1,548)           UBS Securities Australia Ltd         Sell         13,318         (1,548)           UBS Securities Australia Ltd         Sell         47,776         (5,538)           UBS Securities Australia Ltd         Sell         (1,912)         (1,912)           UBS Securities Australia Ltd         Sell         (6,501)         (1,912)           UBS Securities Australia Ltd         Sell         (800)         (800)           UBS Securities Australia Ltd         Sell         (1,720)         (2,002)           UBS Securities Australia Ltd         Sell         (1,720)         (2,002)           UBS Securities Australia Ltd         Sell         (6,01)         (6,01)           UBS Securities Australia Ltd         Sell         (6,02)         (6,03)           UBS Securities Australia Ltd         Sell         (6,03)         (6,13)           UBS Securities Australia Ltd         Sell         (6,13)         (6,13)           UBS Securities Australia Ltd         Sell         (6,13)         (6,13)           UBS Securities Australia Ltd         Sell         (1,104)         (6,13)	4	UBS Securities Australia Ltd	Sell	21,538	(2,500)	Ordinary
UBS Securities Australia Ltd         Sell         13,318         (1,548)           UBS Securities Australia Ltd         Sell         22,256         (2,583)           UBS Securities Australia Ltd         Sell         27,256         (5,580)           UBS Securities Australia Ltd         Sell         (6,501)         (1,912)           UBS Securities Australia Ltd         Sell         6,888         (800)           UBS Securities Australia Ltd         Sell         (800)         (1,912)           UBS Securities Australia Ltd         Sell         (800)         (1,912)           UBS Securities Australia Ltd         Sell         (800)         (1,912)           UBS Securities Australia Ltd         Sell         (6,598)         (800)           UBS Securities Australia Ltd         Sell         (6,598)         (6,500)           UBS Securities Australia Ltd         Sell         (6,730)         (6,730)           UBS Securities Australia Ltd         Sell         (7,730)         (7,730)	1	LIBS Securities Australia Ltd	liac los	6,091	(/0/)	Ordinary
USS Securities Australia Ltd         Sell         22,265         (1,349)           USS Securities Australia Ltd         Sell         22,265         (2,583)           USS Securities Australia Ltd         Sell         47,776         (5,580)           USS Securities Australia Ltd         Sell         6,334         (7,24)           USS Securities Australia Ltd         Sell         6,888         (800)           USS Securities Australia Ltd         Sell         (8,980)         (800)           USS Securities Australia Ltd         Sell         (1,720)         (2,000)           USS Securities Australia Ltd         Sell         (1,720)         (2,000)           USS Securities Australia Ltd         Sell         (1,320)         (2,500)           USS Securities Australia Ltd         Sell         (1,320)         (2,500)           USS Securities Australia Ltd         Sell         (1,320)         (2,500)           USS Securities Australia Ltd         Sell         (1,320)         (2,320)           USS Securities Australia Ltd         Sell         (1,530)           USS Securities Australia Ltd         Sell         (1,320)	17	UBS Securities Australia Ind	Jac Jac Jac Jac Jac Jac Jac Jac Jac Jac	12,52/	(2,847)	Ordinary
UBS Securities Australia trd         Sell         47,776         (5,506)           UBS Securities Australia trd         Sell         16,501         (1912)           UBS Securities Australia trd         Sell         6,334         (734)           UBS Securities Australia trd         Sell         6,808         (800)           UBS Securities Australia trd         Sell         17,220         (2,000)           UBS Securities Australia trd         Sell         17,220         (2,000)           UBS Securities Australia trd         Sell         17,220         (2,000)           UBS Securities Australia trd         Sell         20,595         (2,392)           UBS Securities Australia trd         Sell         5,278         (613)           UBS Securities Australia trd         Sell         1,130         (730)           UBS Securities Australia trd         Sell         1,130         (730)           UBS Securities Australia trd         Sell         1,130         (730)	-14	UBS Securities Australia Itd	3	22, 265	(563.5)	Ordinary
UBS Securities Australia Ltd         Sel         16,501         (1,912)           UBS Securities Australia Ltd         Sel         6,888         (800)           UBS Securities Australia Ltd         Sel         6,888         (800)           UBS Securities Australia Ltd         Sel         1,720         2,000)           UBS Securities Australia Ltd         Sel         1,720         2,000)           UBS Securities Australia Ltd         Sel         5,278         (6,13)           UBS Securities Australia Ltd         Sel         5,278         (6,13)           UBS Securities Australia Ltd         Sel         1,130         1,130           UBS Securities Australia Ltd         Sel         1,119         1,130	14	UBS Securities Australia Ltd	Sell	47 776	(5.536)	Ordinary
UBS Securities Australia Ltd         Sell         6,334         (734)           UBS Securities Australia Ltd         Sell         6,888         (800)           UBS Securities Australia Ltd         Sell         1,720         (2,000)           UBS Securities Australia Ltd         Sell         7,209         (2,000)           UBS Securities Australia Ltd         Sell         7,595         (7,332)           UBS Securities Australia Ltd         Sell         5,738         (6,13)           UBS Securities Australia Ltd         Sell         1,130         (7,300)           UBS Securities Australia Ltd         Sell         1,130         (7,300)	-14	UBS Securities Australia Ltd	Sell	16,501	(1.912)	Ordinary
UBS Securities Australia Ltd         Sell         6.888         (800)           UBS Securities Australia Ltd         Sell         17,220         (2,000)           UBS Securities Australia Ltd         Sell         17,220         (2,000)           UBS Securities Australia Ltd         Sell         20,595         (6.13)           UBS Securities Australia Ltd         Sell         5,278         (6.13)           UBS Securities Australia Ltd         Sell         1,130         (7.30)           UBS Securities Australia Ltd         Sell         1,130         (7.30)	-14	UBS Securities Australia Ltd	Self	6,334	(734)	Ordinary
UBS Securities Australia Ltd   Sell   17,270   (2,000)     UBS Securities Australia Ltd   Sell   17,270   (2,000)     UBS Securities Australia Ltd   Sell   20,595   (2,392)     UBS Securities Australia Ltd   Sell   5,278   (613)     UBS Securities Australia Ltd   Sell   2,153   (613)     UBS Securities Australia Ltd   Sell   2,153   (2,30)     UBS Securities Australia Ltd   Sell   1,119   (1,30)     UBS Securities Australia Ltd   Sell   2,178   (1,30)     UBS Securities Australia Ltd   3,178   (1,30)	14	UBS Securities Australia Ltd	Sell	988'9	(800)	Ordinary
UBS Securities Australia Ltd         Sell         1720         (2,000)           UBS Securities Australia Ltd         Sell         7,0595         (2,302)           UBS Securities Australia Ltd         Sell         5,278         (6,13)           UBS Securities Australia Ltd         Sell         1,150         (7,570)           UBS Securities Australia Ltd         Sell         1,110         (7,570)	14	UBS Securities Australia Ltd	Sell	888'9	(008)	Ordinary
UBS Securities Australia Ltd         Sell         20,595         (2,392)           UBS Securities Australia Ltd         Sell         5,278         (613)           UBS Securities Australia Ltd         Sell         5,153         (260)           UBS Securities Australia Ltd         Sell         1,119         /13 nm	14	UBS Securities Australia Ltd	Sell	17,220	(2,000)	Ordinary
UBS Securities Australia Ltd         Sell         5.278         (613)           UBS Securities Australia         541         2,153         (250)           UBS Securities Australia         541         2,153         (250)           UBS Securities Australia         541         2,153         (250)	14	UBS Securities Australia Ltd	Sell	20,595	(2,392)	Ordinary
UBS Securities Australia Itd Sell 2,153 (250)	14	UBS Securities Australia Ltd	Sell	5,278	(613)	Ordinary
UBS Securities Australia Ltd Sell 1119	4	UBS Securities Australia Ltd	lles.	2,153	(050)	Ordinary
(10)	:					

06-Nov-14	UBS AG, London Branch	Sell	8,473	(683)	Ordinary
36-Nov-14	UBS AG, London Branch	Sell	3,909	(454)	Ordinary
6-Nov-14	UBS AG, London Branch	Sell	15,317	(1,779)	Ordinary
U6-Nov-14	UBS AG, London Branch	ii S	1,765	(205)	Ordinary
6-Now-14	PIRS AG LONDON Branch	Soil	166'07	(2,438)	Ordinary
6-Nov-14	URS AG London Branch	Coll	45	(C)	Ordinary
6-Nov-14	URS AG London Branch	Sell	300,53	(200,3)	Ordinary
6-Nov-14	UBS AG. London Branch	Sell	17 694	(2007)	Ordinary
6-Nov-14	UBS AG. London Branch	Sell	851	(16)	Ordinary
6-Nov-14	LIRS AG London Branch	Cell	16.635	(0.00 1/	Ordinary
6.Now.14	HRC AG London Branch	100	0000	(700)	Granais
S-Nov-14	IIRC AG London Branch	la C	9696		Colling
No. 44	Arrest Company Or 2011	100	030'7	(FOE)	Collidia
	USS AG, LOHOOTI BIRTICH	Sell Sell	465		Ordinary
9-Nov-14	UBS AG, London Branch	Sell	2,118		Ordinary
2-NOV-14	UBS AG, London Branch	Sell	3,005		Ordinary
5-Nov-14	UBS AG, London Branch	Sell	146	(17)	Ordinary
5-Nov-14	UBS AG, London Branch	Seli	861		Ordinary
5-Nov-14	UBS AG, London Branch	Sell	215		Ordinary
5-Nov-14	UBS AG, London Branch	Se	465		Ordinary
06-Nov-14	UBS AG. London Branch	Sel	4 055	(471)	Ordinan
06-Nov-14	UBS AG. London Branch	Sell	4 055		Ordinary
6-Nov-14	URS AG London Branch	le y	177		Ordinary
N5-Nov-14	LIBS AG London Branch	lo o	010		Coulingry
DE Mon. 14	Libe A. London Branch	100	610		Gullrary
11.00.14	OBS AG, LONGOI BIRDING	1 AC	505,1		Ordinary
Up-Nov-14	UBS AG, London Branch	Sell	22,115	(2,573)	Ordinary
-Nov-14	UBS AG, London Branch	Sell	2,579		Ordinary
-Nov-14	UBS AG, London Branch	Sell	16,116		Ordinary
06-Nov-14	UBS AG, London Branch	Sell	12,274		Ordinary
-Nov-14	UBS AG, London Branch	Sell	21.180		Ordinary
06-Nov-14	UBS AG. London Branch	las	5 834		Ordinary
DE-Now-14	IIBS AG London Branch	Coll	10 505	(1 C3 t)	(1)
06-Now-14	LIBC AG London Branch	700	200,00	(4,0,1)	Selling S
1 1001	OBS AG, EGIGGI BIRICI	nac .	13,016		Ordinary
DO-NOV-14	UBS AG, London Branch	Sell	14,114		Ordinary
-Nov-14	UBS AG, London Branch	Sell	1,012		Ordinary
-Nov-14	UBS AG, London Branch	Sell	6,246		Ordinary
-Nov-14	UBS AG, London Branch	Sell	3,029	(323)	Ordinary
-Nov-14	U85 AG. London Branch	Sell	17	(0)	Ordinary
-Nov-14	LIRS AG London Branch	(0)	755	(00/	Ordinary
Nov. 14	100 AC London Brook	197	CC/	(00)	Cidinary
100	Loc AC, Edited Digital	190	5,449	(402)	Ordinary
-IAOV-14	UBS AG, London Branch	Sell	/23	(84)	Ordinary
-Nov-14	UBS AG, London Branch	Sell	103	(12)	Ordinary
-Nov-14	UBS AG, London Branch	Sell	955	(111)	Ordinary
-Nov-14	UBS AG, London Branch	Sell	898'8	(673)	Ordinary
-Nov-14	LIRS AG London Branch	(a)	1360	(150)	Ordinan
06-Nov-14	HIRS AG London Branch	130	90	(100)	Codinary
No. 1	1100 AC 11-0		00	0	Collidiy
NOV-14	UBS AG, London Branch	Sell	6,734	(783)	Ordinary
-Nov-14	UBS AG, London Branch	Sell	318	(37)	Ordinary
-Nov-14	UBS AG, London Branch	Sell	10,157	(1,181)	Ordinary
-Nov-14	UBS AG. London Branch	las	53 373	1017.07	Ordinan
OK-Now-14	1	100	0	(4)	O Time
1100	Coo Act, Edition Blanch	liac l	7		Ordinary
NOV-14	Ş	Sell	14,457	(1,681)	Ordinary
-Nov-14	UBS AG, London Branch	Sel	17,533	(2,034)	Ordinary
-Nov-14	UBS AG. London Branch	Seli	45 789	(5.312)	Ordinary
06.Nov.14	LIRS AG London Branch	Coll	74.408	(000 0)	1000
11.00	Žİ:	liac	/4,408	(8,632)	Ordinary
Nov-14	UBS AG, London Branch	Sell	5,724	(664)	Ordinary
-Nov-14	UBS AG, London Branch	Sell	18,059	(2,095)	Ordinary
Nov-14	UBS AG London Branch	le?	8 025	(031)	Ordinary
	2 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7	100	6,000	(166)	Collialy
-NOV-14	UBS AG Australia Branch	Sei	1,329,853	(154,260)	Ordinary
-Nov-14	UBS AG Australia Branch	Sell	224,254	(26,013)	Ordinary
Nov-14	HRS AG Australia Branch	× 8	300 305	25 440	Cedian
	COS AS A CALLED STATE OF THE ST	fno	990,990	22,440	Galhary
1404	ODS AG AUSKIRIN DIRIKII	Sell	2,545,151	(562,759)	Ordinary
-Nov-14	UBS AG Australia Branch	Buy	862,274	100,000	Ordinary
Nov-14	UBS Wealth Management Australia Ltd	Buy	3,985	462	Ordinary
		Enter Into Demotive			,
06-Nov-14	UBS Securities Australia Ltd	Contract	4/N	220.000	Ontions
Nime 14	Securities Australia 140	Coll	C16 903	000'027	Chundre
1	UBS Securities Australia Ltd	Sell	625,812	(72,600)	Ordinary
	UBS Securities Australia Ltd	Sell	388	(42)	Ordinary
	UBS Securities Australia Ltd	jes	PCC	.90	Ordinary
	DID DESCRIPTION CONTROL COO	170	477	(97)	Oromary
-	UBS Securities Australia Ltd	Sell	2,827	(328)	Ordinary
Nov-14	UBS Securities Australia Ltd	Sell	207	(24)	Ordinary
Nov-14	UBS Securities Australia Ltd	IleS	319	(37)	Ordinary
	200 Dec 200 Contract	1130	0.0	(/6)	Ciciliary
Nov-14	UBS Securities Australia Ltd	Sell	7,629	(882)	Ordinary
-Nov-14	IURS Securities Australia Ltd	las	4 310	(SOO)	Ordinary
06-Nov-14	UBS Securities Australia Ltd	Self	4,310	(200)	Ordinary

v-14	UBS Securities Australia Ltd	les	19.286	1355 67	Ordinan
v-14	UBS Securities Australia Ltd	Sell	15.051	(1.745)	Ordinary
w-14	UBS Securities Australia Ltd	Sell	5,894	(683)	Ordinary
v-14	UBS Securities Australia Ltd	Sell	932	(108)	Ordinary
y-14	UBS Securities Australia Ltd	Sell	4,315	(200)	Ordinary
2v-14	UBS Securities Australia Ltd	Sell	2,986	(346)	Ordinary
4.14	UBS Securities Australia Ltd	Sell	2,627	(652)	Ordinary
1 1 2	UBS Securities Australia Ltd	Self	7,638	(882)	Ordinary
1 1	LIBS Securities Australia Ltd	Jet Sell	46,378	(5,374)	Ordinary
5v-14	UBS Securities Australia Ltd	38	170,72	(13,131)	Ordinary
3v-14	UBS Securities Australia Ltd	Sell	5,000	(606)	Ordinary
Dv-14	UBS Securities Australia Ltd	Sell	9.372	(1 086)	Ordinary
DV-14	UBS Securities Australia Ltd	Sell	734	(82)	Ordinary
2v-14	UBS Securities Australia Ltd	Sell	2,183	(223)	Ordinary
2v-14	UBS Securities Australia Ltd	Sell	95	(11)	Ordinary
24-14	UBS Securities Australia Ltd	Sell	328	(38)	Ordinary
3v-14	UBS Securities Australia Ltd	Sell Sell	1,821	(211)	Ordinary
ov-14	UBS Securities Australia Ltd	Sell	3,050	(423)	Ordinary
ov-14	UBS Securities Australia Ltd	Sell	293	(34)	Ordinary
36-Nov-14	UBS Securities Australia Ltd	Seli	10,235	(1,186)	Ordinary
04-14	UBS Securities Australia Ltd	Sell	2,158	(250)	Ordinary
06-Nov-14	UBS Securities Australia Ltd	Sell	104	(12)	Ordinary
Ov-14	UBS Securities Australia Ltd	les Sell	3,046	(323)	Ordinary
DV-14	USS Securities Australia Itd	- S	66	(11)	Ordinary
ov-14	UBS Securities Australia Ltd	Tes.	1 976	(924)	Ordinary
06-Nov-14	UBS Securities Australia Ltd	Sell	311	(36)	Ordinary
ov-14	UBS Securities Australia Ltd	Sell	5,014	(581)	Ordinary
96-Nov-14	UBS Securities Australia Ltd	Sell	3,046	(353)	Ordinary
96-Nov-14	UBS Securities Australia Ltd	Sell	98	(10)	Ordinary
Jo-Nov-14	UBS Securities Australia Ltd	Sell Sell	35	(4)	Ordinary
14	UBS Securities Australia Ital	ii A	1,407	(163)	Ordinary
06-Nov-14	UBS Securities Australia Ind	را الم	32,035	(3,772)	Ordinary
ov-14	UBS Securities Australia Ltd	3	74 805	(565)	Ordinary
0v-14	UBS Securities Australia Ltd	Sell	14.878	(1,725)	Ordinary
ov-14	UBS Securities Australia Ltd	Sell	9,910	(1,149)	Ordinary
Jv-14	UBS Securities Australia Ltd	Sell	11,169	(1,295)	Ordinary
	UBS Securities Australia Ltd	Sell	3,045	(323)	Ordinary
- 1	UBS Securities Australia Ltd	Sell	8,625	(1,000)	Ordinary
	UBS Securities Australia Ltd	Sell	5,072	(288)	Ordinary
1	UBS Securities Australia Ltd	Sell	8,625	(1,000)	Ordinary
	UBS Securities Australia Ltd		224	(26)	Ordinary
ŀ	UBS Securities Australia Ltd	Sell	3,517	(408)	Ordinary
06-Nov-14	URS Securities Australia (10)	lle de	319	(37)	Ordinary
	UBS Securities Australia Ltd	Sell	722	(32)	Ordinary
	UBS Securities Australia Ltd	Seli	2 181	(25)	Ordinary
	UBS Securities Australia Ltd	Sell	3 155	(366)	Ordinary
П	UBS Securities Australia Ltd	Jes.	12,930	(1.500)	Ordinary
	UBS Securities Australia Ltd	Sell	448	(52)	Ordinary
v-14	UBS Securities Australia Ltd	Sell	69	8)	Ordinary
06-Nov-14	UBS Securities Australia Ltd	Sell	3,146	(365)	Ordinary
Dv-14	UBS Securities Australia Ltd	Sell	3,707	(430)	Ordinary
24-14	UBS Securities Australia Ltd	Sell	78	(6)	Ordinary
0v-14	UBS Securities Australia Ltd	Sell	362	(42)	Ordinary
Uo-Nov-14	UBS Securities Australia Ltd	Sell	69	(8)	Ordinary
7.14	UBS Securities Australia Ltd	Sell	43	(2)	Ordinary
DE-Nov-14	UBS Securities Australia Ltd	Sell Sell	1,198	(139)	Ordinary
DE-Nov-14	LIBS Cocumition Australia Ltd	iiac.	2,155	(250)	Ordinary
x-14	UBS Securities Australia Itd	E S	71867	(3,459)	Ordinary
06-Nov-14	UBS Securities Australia Ltd	Sell	10.491	(7171)	Ordinary
w-14	UBS Securities Australia Ltd	Sell	19.180	(2.2.5)	Ordinary
w-14	UBS Securities Australia Ltd	Sell	3.444	(400)	Ordinary
	UBS Securities Australia Ltd	Sell	3.702	(430)	Ordinary
	UBS Securities Australia Ltd	Sell	1,025	(119)	Ordinary
Ш	UBS Securities Australia Ltd	Sell	2,153	(250)	Ordinary
	UBS Securities Australia Ltd	Sell	2,247	(261)	Ordinary
	The state of the s				
	IU65 Securities Australia LTG	les.	1.343	(158)	Ordinary

	Nov-14	LIRS Securities Australia Itcl	Sell	14,370	(699'1)	Ordinary
Mail of Securities Australia Ltd   Self   Securities Australia Ltd   Self   Securities Australia Ltd   Self   Securities Australia Ltd   Self   Securities Australia Ltd   Self   Securities Australia Ltd   Self   Securities Australia Ltd   Self   Securities Australia Ltd   Self   Securities Australia Ltd   Self   Self   Securities Australia Ltd   Self   Self   Securities Australia Ltd   Self   Self   Securities Australia Ltd   Self   Self   Securities Australia Ltd   Self   Self   Securities Australia Ltd   Self   Self   Securities Australia Ltd   Self   Self   Securities Australia Ltd   Self   Self   Securities Australia Ltd   Self   Self   Securities Australia Ltd   Self   Self   Securities Australia Ltd   Self   Self   Securities Australia Ltd   Self   Self   Securities Australia Ltd   Self   Self   Securities Australia Ltd   Self   Self   Securities Australia Ltd   Self   Self   Securities Australia Ltd   Self   Self   Securities Australia Ltd   Self   Self   Securities Australia Ltd   Self   Self   Securities Australia Ltd   Self   Self   Securities Australia Ltd   Self   Securities Australia Ltd   Self						
1875 Securine Australia Ltd   584   17,278   1	Nov-14	UBS Securities Australia Ltd	Sell	16,307	(1,894)	Ordinary
USS SCANINGER AUXILIARIS   VAIL   V	Nov-14	UBS Securities Australia Ltd	Sel.	12,338	(1,433)	Ordinary
10.55 Securine Audition 1.05 Sept.   19.55 Securine Audition 1.05	Nov-14	UBS Securities Australia Ltd	sei	16,445	(016,1)	Ordinary
USS SCANING AUGUSTAL (1998)   1918	Nov-14	UBS Securities Australia Ltd	Sell	4,305	(200)	Ordinary
USS Securines Australia List   USS Securines Australia List	Nov-14	UBS Securities Australia Ltd	Sell	35,025	(4,068)	Ordinary
185 Sourine Australia Lot         Seld         6,248         (17)           185 Sourine Australia Lot         Seld         6,248         (17)           185 Sourine Australia Lot         Seld         5,29         (18)           185 Sourine Australia Lot         Seld         7,173         (18) <td>Nov-14</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>22,330</td> <td>(2,589)</td> <td>Ordinary</td>	Nov-14	UBS Securities Australia Ltd	Sell	22,330	(2,589)	Ordinary
USS Securines Automisa Into         See         33.92         (1.91)           USS Securines Automisa Into         Sea         3.20         (6.93)           USS Securines Automisa Into         Sea         6.93         (1.91)           USS Securines Automisa Into         Sea         6.93         (1.91)           USS Securines Automisa Into         Sea         8.20         (6.93)           USS Securines Automisa Into         Sea         8.12         (1.93)           USS Securines Automisa Into         Sea         1.37         (1.93)           USS Securines Automisa Into         Sea	Nov-14	UBS Securities Australia Ltd	Sell	6,348	(736)	Ordinary
USS Securing Automitals Ltd   Sea	Nov-14	UBS Securities Australia Ltd	Set	96	(11)	Ordinary
USS Securine Automitation         Pair         658         (68)           USS Securine Automitation         Sail         650         (68)           USS Securine Automitation         Sail         8,129         (643)           USS Securine Automitation         Sail         8,129         (643)           USS Securine Automitation         Sail         8,127         (759)           USS Securine Automitation         Sail         1,172         (758)           USS Securine Automitation         Sail         1,172         (758)           USS Securine Automitation         Sail         1,157         (1,18)           USS Securine Automitation         Sail         1,157<	Nov-14	UBS Securities Australia Ltd	Seil	33,937	(3,937)	Ordinary
USS Securine Autorials (14)   Seal	NOV-14	UBS Securities Australia Ltd	Sell	52	(9)	Ordinary
USS Securities Australia Ltd         Seal         8,00         (5,09)           USS Securities Australia Ltd         Seal         1,772         (1,17)           USS Securities	Nov-14	UBS Securities Australia Ltd	Sell	69	(8)	Ordinary
USS Securities Australia Ltd   Seal   1,129   (14)   (14)   (15)   (14	Nov-14	UBS Securities Australia Ltd	Sell	200	(28)	Ordinary
USS Securities Australia Ltd         Seal         15,472         (1799)           USS Securities Australia Ltd         Seal         1,772         (279)           USS Securities Australia L	Nov-14	UBS Securities Australia Ltd	Sell	8,129	(843)	Ordinary
USS Securities Australia Ltd   Seil   7,473 (179) (179)     USS Securities Australia Ltd   Seil   7,473 (179) (179)     USS Securities Australia Ltd   Seil   Seil   7,737 (179) (179)     USS Securities Australia Ltd   Seil   Seil   1,737 (179) (179)     USS Securities Australia Ltd   Seil   Seil   1,737 (179) (179)     USS Securities Australia Ltd   Seil   Seil   1,737 (179) (179)     USS Securities Australia Ltd   Seil   Seil   1,737 (179) (179)     USS Securities Australia Ltd   Seil   Seil   1,737 (179) (179)     USS Securities Australia Ltd   Seil   Seil   1,737 (179) (179)     USS Securities Australia Ltd   Seil   Seil   1,737 (179) (179)     USS Securities Australia Ltd   Seil   Seil   1,737 (179) (179)     USS Securities Australia Ltd   Seil   Seil   1,737 (179) (179)     USS Securities Australia Ltd   Seil   Seil   1,737 (179) (179)     USS Securities Australia Ltd   Seil   Seil   1,737 (179) (179)     USS Securities Australia Ltd   Seil   Seil   1,737 (179) (179) (179)     USS Securities Australia Ltd   Seil   Seil   1,737 (179)	Nov-14	UBS Securities Australia Ltd	Sell	95	(11)	Ordinary
UNIS Scorrince Australia Ltd         Seli         1,712         (818)           UNIS Scorrince Australia Ltd         Seli         1,722         (219)           UNIS Scorrince Australia Ltd         Seli         1,732         (718)           UNIS Scorrince Australia Ltd         Seli         1,732         (1,781)           UNIS Scorrince Australia Ltd         Seli         1,732         (1,782)           UNIS Scorrince Australia Ltd         Seli         1,732         (1,782)           UNIS Scorrince Australia Ltd         Seli         1,732         (1,782)           UNIS	Nov-14	UBS Securities Australia Ltd	Sell	15,473	(562'1)	Ordinary
UBS Securities Australia Ind 1         Sell         1 R888         (7.19)           UBS Securities Australia Ind 1         Sell         6.320         (7.28)           UBS Securities Australia Ind 1         Sell         310         (1.48)           UBS Securities Australia Ind 1         Sell         1.15         (1.48)           UBS Securities Australia Ind 1         Sell         1.15         (1.58)           UBS Securities Australia Ind 2         Sell         1.12         (1.58)	Nov-14	'UBS Securities Australia Ltd	Seil	7,172	(832)	Ordinary
UBS Securine Australia Ind         Sell         6.362         (7.8)           UBS Securine Australia Ind         Sell         17.0         GO           UBS Securine Australia Ind         Sell         1.0         SBO         (1.82)           UBS Securine Australia Ind         Sell         1.0         (1.82)         (1.82)           UBS Securine Australia Ind         Sell         1.0         (1.82)         (1.82)           UBS Securine Australia Ind         Sell         1.0         (1.92)         (1.92)           UBS Securine Australia Ind         Sell         1.0         (1.92)         (1.92)           UBS Securine Australia Ind         Sell         1.0         (1.92)         (1.92)           UBS Securine Australia Ind         Sell         Sell <td>Nov-14</td> <td>UBS Securities Australia Ltd</td> <td>Seli</td> <td>1,888</td> <td>(512)</td> <td>Ordinary</td>	Nov-14	UBS Securities Australia Ltd	Seli	1,888	(512)	Ordinary
UBS Securities Australia Ltd         Seld         121         (1.4)           UBS Securities Australia Ltd         Seld         2155         C520           UBS Securities Australia Ltd         Seld         15,809         (1.820)           UBS Securities Australia Ltd         Seld         1,979         (1.930)           UBS Securities Australia Ltd         Seld         1,979         (1.930)           UBS Securities Australia Ltd         Seld         1,979         (1.930)           UBS Securities Australia Ltd         Seld         1,970         (1.412)           UBS Securities Australia Ltd         Seld         1,970         (1.412)           U	Nov-14	UBS Securities Australia Ltd	Sell	6,362	(738)	Ordinary
URS Securities Australia Ltd         Sell         2150         CGS0           LMS Securities Australia Ltd         Sell         1,088         (1,187)           LMS Securities Australia Ltd         Sell         1,088         (1,187)           LMS Securities Australia Ltd         Sell         1,088         (1,187)           LMS Securities Australia Ltd         Sell         1,198         (1,187)           LMS Securities Australia Ltd         Sell         1,270         (1,187)           LMS Securities Australia Ltd         Sell         1,270         (1,417)           LMS Securities Australia Ltd         Sell         1,270         (1,417)           LMS Securities Australia Ltd         Sell         1,270         (1,417)           LM	Nov-14	UBS Securities Australia Ltd	Sell	121	(14)	Ordinary
URS Securities Australia Led         Sell         1958         C250           URS Securities Australia Led         Sell         10,380         (1,381)           URS Securities Australia Led         Sell         1,580         (1,381)           URS Securities Australia Led         Sell         2,731         (1,682)           URS Securities Australia Led         Sell         2,731         (1,682)           URS Securities Australia Led         Sell         2,731         (1,682)           URS Securities Australia Led         Sell         1,879         (1,731)           URS Securities Australia Led         Sell         1,879         (1,732)           URS Securities Australia Led         Sell         1,879         (1,731)           URS Securities Australia Led         Sell         1,879         (1,732)           URS Securities Australia Led         Sell         1,879         (1,732)           URS Securities Australia Led         Sell         2,1,780         (1,732) <td< td=""><td>Nov-14</td><td>UBS Securities Australia Ltd</td><td>Sell</td><td>310</td><td>(9E)</td><td>Ordinary</td></td<>	Nov-14	UBS Securities Australia Ltd	Sell	310	(9E)	Ordinary
URS Securities Autorials Ltd         Sell         17,0 RO         (0,182)           URS Securities Autorials Ltd         Sell         15,800         (1,834)           USS Securities Autorials Ltd         Sell         5,326         (1,934)           USS Securities Autorials Ltd         Sell         5,326         (1,934)           USS Securities Autorials Ltd         Sell         15,327         (1,935)           USS Securities Autorials Ltd         Sell         22,731         (1,935)           USS Securities Autorials Ltd         Sell         1,935         (1,935)           USS Securities Autorials Ltd         Sell         1,935         (1,935)           USS Securities Autorials Ltd         Sell         8,022         (1,935)           USS Securities Autorials Ltd         Sell         8,022         (1,935)           USS Securities Autorials Ltd         Sell         8,022         (1,935)           USS Securities Autorials Ltd         Sell         1,935         (1,035)           USS Securities Autorials Ltd         Sell         3,155         (1,455)           USS Securities Autorials Ltd         Sell         3,155         (1,455)           USS Securities Autorials Ltd         Sell         3,155         (1,455)	Nov-14	UBS Securities Australia Ltd	Sell	2,155	(052)	Ordinary
UBS Securities Autoriale Ltd         Sell         15.80         (1.82)           UBS Securities Autoriale Ltd         Sell         5,326         (619)           UBS Securities Autoriale Ltd         Sell         1,532         (619)           UBS Securities Autoriale Ltd         Sell         1,63         (619)           UBS Securities Autoriale Ltd         Sell         1,73         (7.20)           UBS Securities Autoriale Ltd         Sell         1,73         (7.20)           UBS Securities Autoriale Ltd         Sell         1,13         (7.21)           UBS Securities Autoriale Ltd         Sell         1,12         (7.21)           UBS Securities Autoriale Ltd         Sell         1,12         (7.21)           UBS Securities Autoriale Ltd         Sell         1,12         (7.22)           UBS Securities Autoriale Ltd         Sell         1,12         (7.22)           UBS Securities Autoriale Ltd         Sell         1,2,19         (1.656)           UBS Securities A	Nov-14	UBS Securities Australia Ltd	Sell	10,180	(1,181)	Ordinary
USS Securities Australia Ltd         Sell         9 154         (1,092)           USS Securities Australia Ltd         Sell         5.356         (1,619)           USS Securities Australia Ltd         Sell         1.798         (1,519)           USS Securities Australia Ltd         Sell         1.798         (1,319)           USS Securities Australia Ltd         Sell         1.799         (1,319)           USS Securities Australia Ltd         Sell         1.750         (1,329)           USS Securities Australia Ltd         Sell         1.749         (1,329)           USS Securities Australia Ltd         Sell         1.750         (1,430) <t< td=""><td>Nov-14</td><td>UBS Securities Australia Ltd</td><td>Seli</td><td>15,809</td><td>(1.834)</td><td>Ordinary</td></t<>	Nov-14	UBS Securities Australia Ltd	Seli	15,809	(1.834)	Ordinary
USS Securities Australia Ltd         Sell         5.356         (619)           USS Securities Australia Ltd         Sell         2.3.731         (2.9.37)           USS Securities Australia Ltd         Sell         2.2.731         (2.5.23)           USS Securities Australia Ltd         Sell         1.7.99         (1.3.9)           USS Securities Australia Ltd         Sell         1.7.99         (1.3.9)           USS Securities Australia Ltd         Sell         4.6.7.22         (1.9.9)           USS Securities Australia Ltd         Sell         4.6.7.22         (1.9.9)           USS Securities Australia Ltd         Sell         1.0.2         (1.9.9)           USS Securities Australia Ltd         Sell         1.0.2         (1.9.9)           USS Securities Australia Ltd         Sell         1.0.2         (1.0.2)           USS Securities Australia Ltd         Sell         2.7.3         (1.0.2) <tr< td=""><td>Nov-14</td><td>UBS Securities Australia Ltd</td><td>Sell</td><td>9,154</td><td>(1,062)</td><td>Ordinary</td></tr<>	Nov-14	UBS Securities Australia Ltd	Sell	9,154	(1,062)	Ordinary
LORS Securities Autarials Ltd         Sell         154         (198           UBS Securities Autarials Ltd         Sell         22,731         (2,637)           UBS Securities Autarials Ltd         Sell         1,879         (7,18)           UBS Securities Autarials Ltd         Sell         1,879         (7,18)           UBS Securities Autarials Ltd         Sell         1,879         (7,18)           UBS Securities Autarials Ltd         Sell         1,979         (7,19)           UBS Securities Autarials Ltd         Sell         1,970         (7,19)           UBS Securities Autarials Ltd         Sell         1,970         (7,19)           UBS Securities Autarials Ltd         Sell         1,970         (1,93)           UBS Securities Autarials Ltd         Sell         1,970         (1,93)           UBS Securities Autarials Ltd         Sell         1,279         (1,415)           UBS Securities Autarials Ltd         Sell         1,279         (1,415)           UBS Securities Autarials Ltd         Sell         1,279         (1,416)           UBS Securities Autarials Ltd         Sell         1,279         (1,416)           UBS Securities Autarials Ltd         Sell         1,279         (1,416)           UBS Secur	Nov-14	UBS Securities Australia Ltd	Sell	5,336	(619)	Ordinary
UBS Securities Australia Ltd         Sel         3.5         (A)           UBS Securities Australia Ltd         Sel         1,158         (1.35)           UBS Securities Australia Ltd         Sel         1,158         (1.35)           UBS Securities Australia Ltd         Sel         1,158         (1.35)           UBS Securities Australia Ltd         Sel         46,772         (1.35)           UBS Securities Australia Ltd         Sel         1,158         (1.35)           UBS Securities Australia Ltd         Sel         1,654         (1.05)           UBS Securities Australia Ltd         Sel         1,654         (1.05)           UBS Securities Australia Ltd         Sel         1,654         (1.05)           UBS Securities Australia Ltd         Sel         1,752         (1.65)           UBS Securities Australia Ltd         Sel         1,743         (1.65)           UBS Securities Australia Ltd <td>Nov-14</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>164</td> <td>(61)</td> <td>Ordinary</td>	Nov-14	UBS Securities Australia Ltd	Sell	164	(61)	Ordinary
UBS Securities Australia Ltd         Sell         1,198         (1.59)           UBS Securities Australia Ltd         Sell         4,6772         (5.428)           UBS Securities Australia Ltd         Sell         4,6772         (5.428)           UBS Securities Australia Ltd         Sell         8,6772         (5.428)           UBS Securities Australia Ltd         Sell         1,12         (1.93)           UBS Securities Australia Ltd         Sell         1,664         (1.93)           UBS Securities Australia Ltd         Sell         1,666         (1.03)           UBS Securities Australia Ltd         Sell         1,666         (1.03)           UBS Securities Australia Ltd         Sell         1,666         (1.040)           UBS Securities Australia Ltd         Sell         3,155         (1.400)           UBS Securities Australia Ltd         Sell         3,155         (1.400)           UBS S	Nov-14	UBS Securities Australia Ltd	Seff	35	(5)	Ordinary
UBS Securities Australia Ltd         Sell         11,98         (139)           UBS Securities Australia Ltd         Sell         66,772         (6,25)           UBS Securities Australia Ltd         Sell         66,772         (6,242)           UBS Securities Australia Ltd         Sell         112         (10)           UBS Securities Australia Ltd         Sell         112         (10)           UBS Securities Australia Ltd         Sell         1664         (10)           UBS Securities Australia Ltd         Sell         1664         (10)           UBS Securities Australia Ltd         Sell         17,87         (10)           UBS Securities Australia Ltd         Sell         17,87         (10)           UBS Securities Australia Ltd         Sell         17,87         (10)           UBS Securities Australia Ltd         Sell         1,198         (1,10)           UBS Securities Australia Ltd         Sell         1,157         (1,159)           UBS Securities Australia Ltd         Sell         1,158         (1,400)           UBS Securities Australia Ltd         Sell         1,158         (1,400)           UBS Securities Australia Ltd         Sell         1,12         (1,600)           UBS Securities Australia Ltd	Nov-14	UBS Securities Australia Ltd	Seli	22.731	(2.637)	Ordinary
UBS Securities Australia Ltd         Sell         (1879)         (2.18)           UBS Securities Australia Ltd         Sell         4.6772         (5.426)           UBS Securities Australia Ltd         Sell         8.647         (1.93)           UBS Securities Australia Ltd         Sell         1.92         (1.93)           UBS Securities Australia Ltd         Sell         35.22         (1.93)           UBS Securities Australia Ltd         Sell         35.22         (1.93)           UBS Securities Australia Ltd         Sell         33.928         (1.650)           UBS Securities Australia Ltd         Sell         33.928         (1.650)           UBS Securities Australia Ltd         Sell         17.529         (853)           UBS Securities Australia Ltd         Sell         17.520         (853)           UBS Sec	Nov-14	UBS Securities Australia Ltd	Sell	1 198	(139)	Ordinary
UBS Securities Australia Ltd         Sell         46,772         (5,426)           UBS Securities Australia Ltd         Sell         139         (13)           UBS Securities Australia Ltd         Sell         190         (10)           UBS Securities Australia Ltd         Sell         190         (22)           UBS Securities Australia Ltd         Sell         33,928         (10)           UBS Securities Australia Ltd         Sell         7,27         (1,650)           UBS Securities Australia Ltd         Sell         7,27         (1,650)           UBS Securities Australia Ltd         Sell         7,27         (1,650)           UBS Securities Australia Ltd         Sell         7,430         (1,60)           UBS Securities Australia	Nov-14	UBS Securities Australia Ltd	Sell	1879	(218)	Ordinary
UBS securities Australia Ltd         Sell         1.92         (1.93)           UBS securities Australia Ltd         Sell         1.72         (1.93)           UBS securities Australia Ltd         Sell         1.96         (1.01)           UBS securities Australia Ltd         Sell         3.92         (1.02)           UBS securities Australia Ltd         Sell         3.92         (1.03)           UBS securities Australia Ltd         Sell         3.92         (1.03)           UBS securities Australia Ltd         Sell         3.92         (1.03)           UBS Securities Australia Ltd         Sell         1.7,197         (1.656)           UBS Securities Australia Ltd         Sell         1.7,57         (1.415)           UBS Securities Australia Ltd         Sell         2.7,53         (1.656)           UBS Securities Australia Ltd         Sell         2.7,53         (1.656)           UBS Securities Australia Ltd         Sell         2.7,53         (1.656)           UBS Securities Australia Ltd         Sell         3.155         (1.66)           UBS Securities Australia Ltd         Sell         2.7,43         (1.656)           UBS Securities Australia Ltd         Sell         3.755         (3.65)           UBS Sec	Nov-14	UBS Securities Australia Ltd	Sell	46,777	(5,426)	Ordinary
UBS Securities Australia Ltd         Sell         (15)           UBS Securities Australia Ltd         Sell         190         (10)           UBS Securities Australia Ltd         Sell         190         (10)           UBS Securities Australia Ltd         Sell         33,928         (10)           UBS Securities Australia Ltd         Sell         33,928         (10)           UBS Securities Australia Ltd         Sell         12,75         (1,656)           UBS Securities Australia Ltd         Sell         12,75         (1,656)           UBS Securities Australia Ltd         Sell         7,620         (883)           UBS Securities Australia Ltd         Sell         1,620         (1,66)           UBS Securities Australia Ltd         Sell         1,2,73         (1,40)           UBS Securities Australia Ltd         Sell         1,2,20         (1,40)           UBS Securities Australia Ltd         Sell         1,2,20         (1,40)           UBS Securities Australia Ltd         S	Nov-14	UBS Securities Australia Ltd	Sell	8 042	(633)	Ordinary
UBS Securities Australia Ltd         Sell         112         (13)           UBS Securities Australia Ltd         Sell         86         (13)           UBS Securities Australia Ltd         Sell         86         (10)           UBS Securities Australia Ltd         Sell         39,28         (14)           UBS Securities Australia Ltd         Sell         39,28         (16)           UBS Securities Australia Ltd         Sell         12,197         (16)           UBS Securities Australia Ltd         Sell         21,783         (16)           UBS Securities Australia Ltd         Sell         21,783         (16)           UBS Securities Australia Ltd         Sell         21,783         (16)           UBS Securities Australia Ltd         Sell         31,55         (16)           UBS Securities Australia Ltd         Sell         31,55         (14,40)           UBS Securities Australia Ltd         Sell         31,55         (14,40)           UBS Securities Australia Ltd         Sell         30,685         (10)           UBS Securities Australia Ltd         Sell         30,685         (14,20)           UBS Securities Australia Ltd         Sell         30,685         (14,20)           UBS Securities Australia Ltd <td>Nov-14</td> <td>IIBS Securities Australia 11d</td> <td>S.</td> <td>20,0</td> <td>(10)</td> <td>Codings</td>	Nov-14	IIBS Securities Australia 11d	S.	20,0	(10)	Codings
UBS Securities Australia Ltd         Sell         150         (42)           UBS Securities Australia Ltd         Sell         150         (42)           UBS Securities Australia Ltd         Sell         1664         (10)           UBS Securities Australia Ltd         Sell         33,528         (42)           UBS Securities Australia Ltd         Sell         12,197         (16,650)           UBS Securities Australia Ltd         Sell         12,197         (16,650)           UBS Securities Australia Ltd         Sell         7,698         (8,837)           UBS Securities Australia Ltd         Sell         7,698         (8,837)           UBS Securities Australia Ltd         Sell         7,698         (8,837)           UBS Securities Australia Ltd         Sell         1,757         (1,656)           UBS Securities Australia Ltd         Sell         3,793         (4,400)           UBS Securities Australia Ltd         Sell         1,2120         (1,400)           UBS Securities Australia Ltd         Sell         1,2120         (1,400)           UBS Securities Australia Ltd         Sell         3,793         (4,401)           UBS Securities Australia Ltd         Sell         3,793         (4,401)           UBS Secu	Now 14	IBS Societies Australia 14d	lay.	112	(61)	Contract of
UBS Securities Australia Ltd         Sell         156         (10)           UBS Securities Australia Ltd         Sell         1664         (10)           UBS Securities Australia Ltd         Sell         35,228         (10)           UBS Securities Australia Ltd         Sell         33,928         (10)           UBS Securities Australia Ltd         Sell         17,97         (1,656)           UBS Securities Australia Ltd         Sell         7,527         (1,656)           UBS Securities Australia Ltd         Sell         7,527         (1,656)           UBS Securities Australia Ltd         Sell         7,620         (885)           UBS Securities Australia Ltd         Sell         7,620         (885)           UBS Securities Australia Ltd         Sell         7,620         (883)           UBS Securities Australia Ltd         Sell         7,620         (883)           UBS Securities Australia Ltd         Sell         7,439         (1,60)           UBS Securities Australia Ltd         Sell         7,439         (1,60)           UBS Securities Australia Ltd         Sell         3,759         (4,40)           UBS Securities Australia Ltd         Sell         3,759         (4,40)           UBS Securities Australi	Nov-14	LIRS Securities Australia Ind	Spil	7001	(CL)	Ordinary
UBS Securities Australia Ltd         Sell         1,664         (10)           UBS Securities Australia Ltd         Sell         33,928         (42)           UBS Securities Australia Ltd         Sell         33,928         (40)           UBS Securities Australia Ltd         Sell         13,197         (1,656)           UBS Securities Australia Ltd         Sell         12,197         (1,656)           UBS Securities Australia Ltd         Sell         7,629         (853)           UBS Securities Australia Ltd         Sell         7,698         (853)           UBS Securities Australia Ltd         Sell         1,570         (440)           UBS Securities Australia Ltd         Sell         1,570         (440)           UBS Securities Australia Ltd         Sell         1,570         (440)           UBS Securities Australia Ltd         Sell         1,770         (440)           UBS Securities Australia Ltd         Sell         1,770         (440)           UBS Securities Australi	1 1	DD Securities Augustin Lea	100	061	(77)	Ordinary
UBS Securities Australia Ltd         Sell         1,654         (13)           UBS Securities Australia Ltd         Sell         33,928         (10)           UBS Securities Australia Ltd         Sell         33,928         (10,550)           UBS Securities Australia Ltd         Sell         1,277         (1,650)           UBS Securities Australia Ltd         Sell         7,629         (685)           UBS Securities Australia Ltd         Sell         7,629         (685)           UBS Securities Australia Ltd         Sell         7,629         (685)           UBS Securities Australia Ltd         Sell         7,439         (1,650)           UBS Securities Australia Ltd         Sell         3,155         (366)           UBS Securities Australia Ltd         Sell         3,155         (440)           UBS Securities Australia Ltd         Sell         7,439         (4,40)           UBS Securities Australia Ltd         Sell         7,439         (4,40)           UBS Securities Australia Ltd         Sell         1,583         (4,40)           UBS Securities Australia Ltd         Sell         3,755         (4,40)           UBS Securities Australia Ltd         Sell         3,756         (3,60)           UBS Securities A		OBS Securities Australia Liu	120	98	(01)	Ordinary
UBS Securities Australia Ltd         Sell         39.28         (42)           UBS Securities Australia Ltd         Sell         39.28         (100)           UBS Securities Australia Ltd         Sell         14,275         (1,650)           UBS Securities Australia Ltd         Sell         12,197         (1,415)           UBS Securities Australia Ltd         Sell         21,783         (833)           UBS Securities Australia Ltd         Sell         7,620         (833)           UBS Securities Australia Ltd         Sell         7,620         (833)           UBS Securities Australia Ltd         Sell         7,620         (833)           UBS Securities Australia Ltd         Sell         3,155         (366)           UBS Securities Australia Ltd         Sell         3,155         (366)           UBS Securities Australia Ltd         Sell         3,155         (366)           UBS Securities Australia Ltd         Sell         3,735         (440)           UBS Securities Australia Ltd         Sell         3,735         (440)           UBS Securities Australia Ltd         Sell         3,735         (430)           UBS Securities Australia Ltd         Sell         3,755         (430)           UBS Securities Austral	10v-14	UBS Securities Australia Ltd	liac liac	1,654	(193)	Ordinary
UBS Securities Australia Ltd         Sell         13,528         (10)           UBS Securities Australia Ltd         Sell         14,275         (16,56)           UBS Securities Australia Ltd         Sell         12,197         (16,56)           UBS Securities Australia Ltd         Sell         7,629         (16,57)           UBS Securities Australia Ltd         Sell         7,698         (832)           UBS Securities Australia Ltd         Sell         7,698         (832)           UBS Securities Australia Ltd         Sell         7,698         (832)           UBS Securities Australia Ltd         Sell         3,155         (1660)           UBS Securities Australia Ltd         Sell         1,2,120         (1,408)           UBS Securities Australia Ltd         Sell         1,393         (4,400)           UBS Securities Australia Ltd         Sell         3,756         (1,400)		Cop Securities Australia Ciu	1120	796	(47)	Ordinary
UBS Securities Australia Ltd         Sell         33.528         (3.556)           UBS Securities Australia Ltd         Sell         12.797         (1.415)           UBS Securities Australia Ltd         Sell         7.629         (883)           UBS Securities Australia Ltd         Sell         7.620         (884)           UBS Securities Australia Ltd         Sell         7.720         (1.666)           UBS Securities Australia Ltd         Sell         7.720         (1.466)           UBS Securities Australia Ltd         Sell         2.75         (4.48)           UBS Securities Australia Ltd         Sell         2.75         (4.68)           UBS Securities Australia Ltd         Sell         2.74         (3.85)           UBS Securities Aus	40.14	UBS Securities Australia Ltd	Sell	98	(10)	Ordinary
UBS Securities Australia Ltd         Sell         14.275         (1.650           UBS Securities Australia Ltd         Sell         7.629         (855)           UBS Securities Australia Ltd         Sell         7.629         (853)           UBS Securities Australia Ltd         Sell         7.629         (833)           UBS Securities Australia Ltd         Sell         7.629         (834)           UBS Securities Australia Ltd         Sell         3.155         (1.660)           UBS Securities Australia Ltd         Sell         3.155         (1.660)           UBS Securities Australia Ltd         Sell         3.753         (1.660)           UBS Securities Australia Ltd         Sell         3.753         (1.600)           UBS Securities Australia Ltd         Sell         3.753         (1.600)           UBS Securities Australia Ltd         Sell         3.753         (1.600)           UBS Securities Australia Ltd         Sell         3.756         (3.83)           UBS Sec	Nov-14	UBS Securities Australia Ltd	Self	33,928	(3,936)	Ordinary
UBS Securities Australia Ltd         Sell         17,197         (14,15)           UBS Securities Australia Ltd         Sell         7,529         (855)           UBS Securities Australia Ltd         Sell         7,529         (855)           UBS Securities Australia Ltd         Sell         7,620         (859)           UBS Securities Australia Ltd         Sell         3,155         (1,560)           UBS Securities Australia Ltd         Sell         3,155         (1,560)           UBS Securities Australia Ltd         Sell         7,439         (1,660)           UBS Securities Australia Ltd         Sell         7,439         (1,660)           UBS Securities Australia Ltd         Sell         3,759         (1,400)           UBS Securities Australia Ltd         Sell         3,799         (4,400)           UBS Securities Australia Ltd         Sell         3,799         (4,00)           UBS Securities Australia Ltd         Sell         3,799         (4,00)           UBS Securities Australia Ltd         Sell         3,799         (4,00)           UBS Securities Australia Ltd         Sell         3,750         (4,00)           UBS Securities Australia Ltd         Sell         3,750         (4,00)           UBS Secu	40v-14	UBS Securities Australia Ltd	Sell	14,275	(1,656)	Ordinary
UBS Securities Australia Ltd         Sell         7.622         (855)           UBS Securities Australia Ltd         Sell         21,783         (852)           UBS Securities Australia Ltd         Sell         7,620         (832)           UBS Securities Australia Ltd         Sell         1,550         (838)           UBS Securities Australia Ltd         Sell         3155         (366)           UBS Securities Australia Ltd         Sell         30,0865         (4,563)           UBS Securities Australia Ltd         Sell         37,39         (4,405)           UBS Securities Australia Ltd         Sell         37,39         (4,405)           UBS Securities Australia Ltd         Sell         37,39         (4,407)           UBS Securities Australia Ltd         Sell         37,79         (4,407)           UBS Securities Australia Ltd         Sell         37,79         (4,407)           UBS Securities Australia Ltd         Sell         3,776         (4,438)           UBS Se	40v-14	UBS Securities Australia Ltd	Sell	12,197	(1,415)	Ordinary
UBS Securities Australia Ltd         Sell         2,783         (2,527)           UBS Securities Australia Ltd         Sell         7,629         (882)           UBS Securities Australia Ltd         Sell         1,557 (1,589)         (882)           UBS Securities Australia Ltd         Sell         3,155 (1,589)         (862)           UBS Securities Australia Ltd         Sell         3,155 (1,589)         (863)           UBS Securities Australia Ltd         Sell         7,439 (1,589)         (863)           UBS Securities Australia Ltd         Sell         1,280 (1,598)         (440)           UBS Securities Australia Ltd         Sell         3,733 (440)         (1,10)           UBS Securities Australia Ltd         Sell         3,733 (440)         (1,10)           UBS Securities Australia Ltd         Sell         3,733 (440)         (1,10)           UBS Securities Australia Ltd         Sell         3,776 (4,40)         (1,60)           UBS Securities Australia Ltd         Sell <t< td=""><td>Nov-14</td><td>UBS Securities Australia Ltd</td><td>Sell</td><td>7,629</td><td>(882)</td><td>Ordinary</td></t<>	Nov-14	UBS Securities Australia Ltd	Sell	7,629	(882)	Ordinary
UBS Securities Australia Ltd         Sell         7,509         (893)           UBS Securities Australia Ltd         Sell         7,500         (894)           UBS Securities Australia Ltd         Sell         13,571         (1,589)           UBS Securities Australia Ltd         Sell         3,155         (366)           UBS Securities Australia Ltd         Sell         7,270         (368)           UBS Securities Australia Ltd         Sell         7,270         (440)           UBS Securities Australia Ltd         Sell         3,755         (440)           UBS Securities Australia Ltd         Sell         3,756         (438)           UBS Securities Australia Ltd         Sell         5,744         (5,53)           UBS Securities Australia	Nov-14	UBS Securities Australia Ltd	Sell	21,783	(2,527)	Ordinary
UBS Securities Australia (Ld.)         Sell         7,670         (884)           UBS Securities Australia (Ld.)         Sell         3,155         (1,689)           UBS Securities Australia (Ld.)         Sell         3,155         (1,680)           UBS Securities Australia (Ld.)         Sell         3,155         (1,683)           UBS Securities Australia (Ld.)         Sell         1,2,120         (1,405)           UBS Securities Australia (Ld.)         Sell         1,2,120         (1,405)           UBS Securities Australia (Ld.)         Sell         3,793         (4,401)           UBS Securities Australia (Ld.)         Sell         1,793         (1,10)           UBS Securities Australia (Ld.)         Sell         1,793         (1,10)           UBS Securities Australia (Ld.)         Sell         1,793         (1,400)           UBS Securities Australia (Ld.)         Sell         3,776         (4,401)           UBS Securities Australia (Ld.)         Sell         3,776         (4,402)           UBS Securities Australia (Ld.)         Sell         2,974         (3,59)           UBS Securities Australia (Ld.)         Sell         3,755         (3,60)           UBS Securities Australia (Ld.)         Sell         3,755         (3,60)	Nov-14	UBS Securities Australia Ltd	Sell	869'2	(688)	Ordinary
UBS Securities Australia Ltd         Sell         13,671         (1,586)           UBS Securities Australia Ltd         Sell         3,155         (366)           UBS Securities Australia Ltd         Sell         3,155         (368)           UBS Securities Australia Ltd         Sell         7,439         (862)           UBS Securities Australia Ltd         Sell         12,720         (1,405)           UBS Securities Australia Ltd         Sell         12,720         (1,405)           UBS Securities Australia Ltd         Sell         12,720         (1,405)           UBS Securities Australia Ltd         Sell         1,993         (1,13)           UBS Securities Australia Ltd         Sell         1,983         (1,13)           UBS Securities Australia Ltd         Sell         2,750         (3,24)           UBS Securities Australia Ltd         Sell         2,750         (3,24)           UBS Securities Australia Ltd         Sell         2,750         (3,26)           UBS Securities Australia Ltd         Sell         2,74         (3,25)           UBS Securities Australia Ltd         Sell         3,155         (3,66)           UBS Securities Australia Ltd         Sell         3,155         (3,66)           UBS Secur	Vov-14	UBS Securities Australia Ltd	Sell	7.620	(884)	Ordinary
UBS Securities Australia Ltd         Sell         3,155         (Beb)           UBS Securities Australia Ltd         Sell         3,155         (Beb)           UBS Securities Australia Ltd         Sell         12,730         (Beb)           UBS Securities Australia Ltd         Sell         12,730         (440)           UBS Securities Australia Ltd         Sell         17,730         (440)           UBS Securities Australia Ltd         Sell         17,730         (440)           UBS Securities Australia Ltd         Sell         991         (115)           UBS Securities Australia Ltd         Sell         1,983         (2,20)           UBS Securities Australia Ltd         Sell         3,776         (448)           UBS Securities Australia Ltd         Sell         3,756         (448)           UBS Securities Australia Ltd         Sell         3,155         (204)           UBS Securities Australia Ltd         Sell         3,155         (266)           UBS Securities Australia	Nov-14	UBS Securities Australia Ltd	Sell	13,671	(1.586)	Ordinary
UBS Securities Australia Ltd         Sell         3.155         (3-69)           UBS Securities Australia Ltd         Sell         3.935         (3-533)           UBS Securities Australia Ltd         Sell         12,130         (1,406)           UBS Securities Australia Ltd         Sell         3,793         (4,400)           UBS Securities Australia Ltd         Sell         1,793         (1,10)           UBS Securities Australia Ltd         Sell         1,883         (1,13)           UBS Securities Australia Ltd         Sell         2,70         (2,43)           UBS Securities Australia Ltd         Sell         2,70         (3,43)           UBS Securities Australia Ltd         Sell         3,145         (3,64)           UBS Securities Australia Ltd         Sell         3,145         (3,64)           UBS Securities Australia Ltd         Sell         3,155         (3,64)           UBS Securities	Vov-14	UBS Securities Australia Ltd	Sell	3,155	(356)	Ordinary
UBS Securities Australia Ltd         Sell         30,0865         (7,563)           UBS Securities Australia Ltd         Sell         7,33         (863)           UBS Securities Australia Ltd         Sell         7,33         (4440)           UBS Securities Australia Ltd         Sell         991         (10)           UBS Securities Australia Ltd         Sell         991         (11)           UBS Securities Australia Ltd         Sell         991         (11)           UBS Securities Australia Ltd         Sell         2,75         (438)           UBS Securities Australia Ltd         Sell         3,76         (438)           UBS Securities Australia Ltd         Sell         3,76         (438)           UBS Securities Australia Ltd         Sell         3,76         (394)           UBS Securities Australia Ltd         Sell         2,974         (383)           UBS Securities Australia Ltd         Sell         3,155         (60)           UBS Securities Australia Ltd         Sell         3,155         (60)           UBS Securities Australia Ltd         Sell         3,155         (50)           UBS Securities Australia Ltd         Sell         3,155         (50)           UBS Securities Australia Ltd	Vov-14	UBS Securities Australia Ltd	Sell	3 155	(386)	Ordinary
UBS Securities Australia Ltd         Sell         7,439         (9,63)           UBS Securities Australia Ltd         Sell         1,270         (1,406)           UBS Securities Australia Ltd         Sell         3,793         (4,406)           UBS Securities Australia Ltd         Sell         95         (1,00           UBS Securities Australia Ltd         Sell         95         (1,00           UBS Securities Australia Ltd         Sell         3,776         (2,30)           UBS Securities Australia Ltd         Sell         3,776         (3,80)           UBS Securities Australia Ltd         Sell         2,750         (3,80)           UBS Securities Australia Ltd         Sell         2,750         (3,80)           UBS Securities Australia Ltd         Sell         2,750         (3,80)           UBS Securities Australia Ltd         Sell         3,745         (3,65)           UBS Securities Australia Ltd         Sell         3,755         (6,60)           UBS Securities Australia Ltd         Sell         3,155         (3,60)           UBS Securities Australia Ltd         Sell         3,155         (3,60)           UBS Securities Australia Ltd         Sell         3,155         (3,60)           UBS Securities Aust	Jov-14	IIRS Securities Australia 14d	9	30.00	(000)	Ordinary
UBS Securities Australia Ltd         Sell         1,732         (1,405)           UBS Securities Australia Ltd         Sell         3,739         (440)           UBS Securities Australia Ltd         Sell         3,739         (440)           UBS Securities Australia Ltd         Sell         1,981         (1,15)           UBS Securities Australia Ltd         Sell         207         (2,4)           UBS Securities Australia Ltd         Sell         3,756         (438)           UBS Securities Australia Ltd         Sell         2,750         (3,59)           UBS Securities Australia Ltd         Sell         3,766         (3,59)           UBS Securities Australia Ltd         Sell         2,750         (3,59)           UBS Securities Australia Ltd         Sell         5,750         (3,59)           UBS Securities Australia Ltd         Sell         5,750         (3,59)           UBS Securities Australia Ltd         Sell         5,750         (3,504)           UBS Securities Australia Ltd         Sell         5,750         (3,60)           UBS Securities Australia Ltd         Sell         3,155         (3,60)           UBS Securities Australia Ltd         Sell         3,155         (3,60)           UBS Securities Au	dov.14	I IRC Securities Australia 11d		20,000	(600,6)	Ordinary
USS Securities Australia Ltd         Sell         1,42,0         (1,40,0)           USS Securities Australia Ltd         Sell         86         (10)           USS Securities Australia Ltd         Sell         931         (10)           USS Securities Australia Ltd         Sell         931         (11)           USS Securities Australia Ltd         Sell         3,776         (438)           USS Securities Australia Ltd         Sell         3,776         (438)           USS Securities Australia Ltd         Sell         2,574         (348)           USS Securities Australia Ltd         Sell         2,574         (348)           USS Securities Australia Ltd         Sell         2,574         (348)           USS Securities Australia Ltd         Sell         3,756         (406)           USS Securities Australia Ltd         Sell         3,155         (360)           USS Securities Australia Ltd<	1 10	Constitution Australia Library	200	004.7	(508)	Orginary
URS Securities Australia Ltd         Sell         3,793         (440)           URS Securities Australia Ltd         Sell         3,793         (440)           URS Securities Australia Ltd         Sell         1,983         (115)           URS Securities Australia Ltd         Sell         3,776         (24)           URS Securities Australia Ltd         Sell         2,750         (24)           URS Securities Australia Ltd         Sell         2,750         (2,94)           URS Securities Australia Ltd         Sell         3,043         (35)           URS Securities Australia Ltd         Sell         3,155         (36)           URS Securities Australia Ltd		Obs Securities Australia Ltd	nec 3	071,21	(1,406)	Ordinary
URS Securities Australia Ltd         Seal         86         (10)           URS Securities Australia Ltd         Seal         991         (115)           URS Securities Australia Ltd         Seal         1,983         (230)           USS Securities Australia Ltd         Seal         3,776         (438)           UBS Securities Australia Ltd         Seal         3,776         (438)           UBS Securities Australia Ltd         Seal         1,805         (2,04)           UBS Securities Australia Ltd         Seal         1,805         (2,04)           UBS Securities Australia Ltd         Seal         2,974         (343)           UBS Securities Australia Ltd         Seal         5,04         (343)           UBS Securities Australia Ltd         Seal         3,155         (360)           UBS Securities Australia Ltd         Seal         2,230         (360)           UBS Securities Australia Ltd         Seal         2,225         (160)           UBS Securities Australia Ltd	4OV-14	UBS Securities Australia Ltd	Sell	3,793	(440)	Ordinary
UBS Securities Australia Ltd         Seri         991         (115)           UBS Securities Australia Ltd         Sell         1,983         (230)           UBS Securities Australia Ltd         Sell         207         (243)           UBS Securities Australia Ltd         Sell         2,756         (438)           UBS Securities Australia Ltd         Sell         2,756         (359)           UBS Securities Australia Ltd         Sell         2,044         (348)           UBS Securities Australia Ltd         Sell         5,074         (35)           UBS Securities Australia Ltd         Sell         5,074         (35)           UBS Securities Australia Ltd         Sell         3,043         (55)           UBS Securities Australia Ltd         Sell         3,155         (360)           UBS Securities Australia Ltd         Sell         24,903         (2,889)           UBS Securities Australia Ltd <td>40v-14</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>98</td> <td>(10)</td> <td>Ordinary</td>	40v-14	UBS Securities Australia Ltd	Sell	98	(10)	Ordinary
UBS Securities Australia Irtd         Seld         1,988         (230)           UBS Securities Australia Irtd         Sell         207         (24)           UBS Securities Australia Irtd         Sell         3,776         (438)           UBS Securities Australia Irtd         Sell         3,756         (319)           UBS Securities Australia Irtd         Sell         2,974         (348)           UBS Securities Australia Irtd         Sell         2,974         (345)           UBS Securities Australia Irtd         Sell         5,043         (35)           UBS Securities Australia Irtd         Sell         3,155         (36)           UBS Securities Australia Irtd         Sell         24,903         (2,889)           UBS Securities Australia Irtd         Sell         24,903         (2,889)           UBS Securities Australia Irtd         Sell         24,903         (2,889)           UBS Securitie	lov-14	UBS Securities Australia Ltd	Sell	166	(115)	Ordinary
UBS Securities Australia Ltd         Sell         3,776         (24)           UBS Securities Australia Ltd         Sell         3,776         (438)           UBS Securities Australia Ltd         Sell         2,750         (439)           UBS Securities Australia Ltd         Sell         2,94         (348)           UBS Securities Australia Ltd         Sell         2,94         (348)           UBS Securities Australia Ltd         Sell         5,94         (35)           UBS Securities Australia Ltd         Sell         5,04         (50)           UBS Securities Australia Ltd         Sell         5,04         (70)           UBS Securities Australia Ltd         Sell         3,155         (60)           UBS Securities Australia Ltd         Sell         3,155         (70)           UBS Securities Australia Ltd         Sell         3,155         (107)           UBS Securities Australia Ltd         Sell         3,155         (107)           UBS Securities Australia Ltd         Sell         3,155         (107)           UBS Securities Australia Ltd         Sell         24,903         (2,89)           UBS Securities Australia Ltd         Sell         24,903         (2,89)           UBS Securities Australia Ltd	Nov-14	UBS Securities Australia Ltd	Sell	1,983	(230)	Ordinary
USS Securities Australia Ltd         Sell         3,776         (438)           UBS Securities Australia Ltd         Sell         2,75G         (438)           UBS Securities Australia Ltd         Sell         2,934         (345)           UBS Securities Australia Ltd         Sell         3,043         (345)           UBS Securities Australia Ltd         Sell         5,043         (35)           UBS Securities Australia Ltd         Sell         5,0         (6)           UBS Securities Australia Ltd         Sell         6,0         (7)           UBS Securities Australia Ltd         Sell         3,155         (60)           UBS Securities Australia Ltd         Sell         3,155         (60)           UBS Securities Australia Ltd         Sell         3,155         (60)           UBS Securities Australia Ltd         Sell         3,155         (369)           UBS Securities Australia Ltd         Sell         3,155         (369)           UBS Securities Australia Ltd         Sell         24,903         (2,889)           UBS Securities Australia Ltd         Sell         24,903         (3,69)           UBS Securities Australia Ltd         Sell         24,903         (3,69)           UBS Securities Australia Ltd	40v-14	UBS Securities Australia Ltd	Sell	202	(7.0)	Ordinary
UBS Securities Australia trd         Sell         2,750         (319)           UBS Securities Australia trd         Sell         18,050         (2,094)           UBS Securities Australia trd         Sell         2,974         (345)           UBS Securities Australia trd         Sell         3,043         (333)           UBS Securities Australia trd         Sell         52         (6)           UBS Securities Australia trd         Sell         60         (7)           UBS Securities Australia trd         Sell         3,155         (366)           UBS Securities Australia trd         Sell         3,155         (366)           UBS Securities Australia trd         Sell         3,155         (369)           UBS Securities Australia trd         Sell         3,155         (369)           UBS Securities Australia trd         Sell         3,155         (369)           UBS Securities Australia trd         Sell         24,903         (2,889)           UBS Securities Australia trd         Sell         24,903         (2,889)           UBS Securities Australia trd         Sell         24,903         (2,889)           UBS Securities Australia trd         Sell         24,74         (55)	Jov-14	IJRS Securities Australia 11d	وا	3 776	(430)	Colinary
UBS Securities Australia Ltd         Sell         18,050         (2,034)           UBS Securities Australia Ltd         Sell         2,974         (345)           UBS Securities Australia Ltd         Sell         2,974         (345)           UBS Securities Australia Ltd         Sell         52         (67)           UBS Securities Australia Ltd         Sell         67         (67)           UBS Securities Australia Ltd         Sell         3,155         (366)           UBS Securities Australia Ltd         Sell         24,903         (2,889)           UBS Securities Australia Ltd         Sell         24,903         (2,889)           UBS Securities Australia Ltd         Sell         24,903         (2,889)           UBS Securities Australia Ltd         Sell         24,903         (5,889)           UBS Securities Australia Ltd         Sell         27,3         (5,50)	Jour. 1.4	LIRS Securities Australia 11d	100	3.350	(010)	Ordinary.
UBS Securities Australia Ltd         Sell         2.974         (4.034)           UBS Securities Australia Ltd         Sell         3.043         (533)           UBS Securities Australia Ltd         Sell         5.0         (6)           UBS Securities Australia Ltd         Sell         3.155         (366)           UBS Securities Australia Ltd         Sell         24,903         (2,882)	17.70	IDC Societies Australia 14d	Coll	10000	(519)	Gellary
UBS Securities Australia Ltd         Sell         2.74         Usb           UBS Securities Australia Ltd         Sell         5.0         (6)           UBS Securities Australia Ltd         Sell         60         (7)           UBS Securities Australia Ltd         Sell         60         (7)           UBS Securities Australia Ltd         Sell         3,155         (366)           UBS Securities Australia Ltd         Sell         3,155         (206)           UBS Securities Australia Ltd         Sell         3,155         (107)           UBS Securities Australia Ltd         Sell         3,155         (107)           UBS Securities Australia Ltd         Sell         24,903         (2,889)           UBS Securities Australia Ltd         Sell         24,74         (55)	14	I IRC Cocurities Australia Ind	100	000'01	(2,034)	Ordinary
URS Securities Australia Ltd         Sell         5.39           UBS Securities Australia Ltd         Sell         60           UBS Securities Australia Ltd         Sell         60           UBS Securities Australia Ltd         Sell         3,155         (50           UBS Securities Australia Ltd         Sell         3,155         (20           UBS Securities Australia Ltd         Sell         3,155         (20           UBS Securities Australia Ltd         Sell         3,155         (30           UBS Securities Australia Ltd         Sell         3,155         (107)           UBS Securities Australia Ltd         Sell         24,903         (2,889)           UBS Securities Australia Ltd         Sell         24,903         (2,889)           UBS Securities Australia Ltd         Sell         24,903         (3,55)           UBS Securities Australia Ltd         Sell         24,903         (3,55)	11 10	Obsiderations Australia Liu	150	2,9/4	(345)	Ordinary
USS Securities Australia Ltd         Sell         52         (b)           UBS Securities Australia Ltd         Sell         60         (70           UBS Securities Australia Ltd         Sell         3,155         (366)           UBS Securities Australia Ltd         Sell         172         (20)           UBS Securities Australia Ltd         Sell         3,155         (107)           UBS Securities Australia Ltd         Sell         24,903         (2,889)           UBS Securities Australia Ltd         Sell         24,903         (2,889)           UBS Securities Australia Ltd         Sell         24,403         (2,889)           UBS Securities Australia Ltd         Sell         24,403         (2,889)           UBS Securities Australia Ltd         Sell         24,74         (55)	POV-14	Obs securities Australia Ltd	100	3,043	(353)	Ordinary
UBS Securities Australia Ltd         Sell         60         (7)           UBS Securities Australia Ltd         Sell         3,155         (366)           UBS Securities Australia Ltd         Sell         3,155         (206)           UBS Securities Australia Ltd         Sell         3,155         (369)           UBS Securities Australia Ltd         Sell         3,155         (369)           UBS Securities Australia Ltd         Sell         24,903         (2,899)           UBS Securities Australia Ltd         Sell         24,903         (2,899)           UBS Securities Australia Ltd         Sell         24,903         (3,69)           UBS Securities Australia Ltd         Sell         24,903         (3,69)	- NO.	UBS Securities Australia Ltd	Sell	52	(9)	Ordinary
UBS Securities Australia Ltd         Sell         3,155         (366)           UBS Securities Australia Ltd         Sell         1,72         (20)           UBS Securities Australia Ltd         Sell         3,155         (366)           UBS Securities Australia Ltd         Sell         222         (107)           UBS Securities Australia Ltd         Sell         2,493         2,889           UBS Securities Australia Ltd         Sell         2,493         2,889           UBS Securities Australia Ltd         Sell         4,74         (55)	OV-14	UBS Securities Australia Ltd	Sell	09	6	Ordinary
UBS Securities Australia Ltd         Sell         722         (20)           UBS Securities Australia Ltd         Sell         3,155         (360)           UBS Securities Australia Ltd         Sell         92.2         (107)           UBS Securities Australia Ltd         Sell         24,903         (2,889)           UBS Securities Australia Ltd         Sell         24,903         (2,889)           UBS Securities Australia Ltd         Sell         24,903         (5,59)	lov-14	UBS Securities Australia Ltd	Sell	3,155	(396)	Ordinary
UBS Securities Australia Ltd         Sell         3,155         (366)           UBS Securities Australia Ltd         Sell         922         (107)           UBS Securities Australia Ltd         Sell         24,903         (2,889)           UBS Securities Australia Ltd         Sell         310         (36)           UBS Securities Australia Ltd         Sell         310         (55)	lov-14	UBS Securities Australia Ltd	Sell	172	(0Z)	Ordinary
UBS Securities Australia Ltd         Sell         242         (107)           UBS Securities Australia Ltd         Sell         24,903         (2,889)           UBS Securities Australia Ltd         Sell         310         (3,889)           UBS Securities Australia Ltd         Sell         474         (55)	lov-14	UBS Securities Australia Ltd	Sell	3,155	(996)	Ordinary
USS Securities Australia Ltd         Self         2.4,903         (2,889)           UBS Securities Australia Ltd         Self         2.4,903         (2,889)           UBS Securities Australia Ltd         Self         310         (39)           UBS Securities Australia Ltd         Self         474         (55)	lov-14	UBS Securities Australia Ltd	Sell	922	(102)	Ordinary
UBS Securities Australia tod         Sell         310         (36)           UBS Securities Australia Ltd         Sell         4.74         (55)	lov-14	UBS Securities Australia Ltd	Self	24.903	(2.889)	Ordinary
UBS Securities Australia Ltd Sell 474 (55)	lov-14	UBS Securities Australia Ltd	Sell	310	1967	Confinant.
Son Securities Australia Ltd (55)	lov-14					Cighiary
	<u>+</u>	Library Australia Australia 14d	Icall	7.67	(05)	Crantary

UBS Securities Australia Ltd	Sell	2,795	(325)	Ordinary
UBS Securities Australia Ltd	Sell	2,950	(343)	Ordinary
UBS Securities Australia Ltd	Sell	40,884	(4,754)	Ordinary
UBS Securities Australia Ltd	Sell	3,853	(448)	Ordinary
UBS Securities Australia Ltd	yell	4,498	(523)	Ordinary
UBS Securities Australia Ltd	Sell	95	31	Ordinary
UBS Securities Australia Ltd	Sell	577	(67)	Ordinary
UBS Securities Australia Ltd	Sell	2,614	(304)	Ordinary
S Securities Australia Ltd	lles.	38,992	(4.534)	Ordinary
S Securities Australia Ltd	lleS	98	(01)	Ordinary
UBS Securities Australia Ltd	Sell	5 083	(591)	Ordinan
IBC Securities Australia 14d	9	4 676	(100)	in in
The second control of the second control of	100	6,0,4	(79C)	Ordinary
oo oeculities Australia Ltd	130	777'1	(142)	Ordinary
55 Securities Australia Ltd	liac li	37,255	(4,332)	Ordinary
BS Securities Australia Ltd	Se!	2,150	(250)	Ordinary
3S Securities Australia Ltd	Sell	1,652	(192)	Ordinary
BS Securities Australia Ltd	Sell	1,385	(161)	Ordinary
BS Securities Australia Ltd	Sell	54.490	(A 23.6)	Ordinary
RS Securities Australia I to	les	15 903	(0707)	- Transport
UBS Securities Australia Itd	lla?	5 530	(040)	Ordinary
RS Securities Australia 11d	=9	010	(60)	O Cities y
On the state of th		010	(/c)	Ordinary
bs securities Australia Ltd	100	2/5	(32)	Ordinary
BS Securities Australia Ltd	lec	15,299	(1,779)	Ordinary
BS Securities Australia Ltd	Sell	1,135	(132)	Ordinary
BS Securities Australia Ltd	Sell	1.058	(123)	Ordinan
BS Securities Australia Ltd	les	10.157	(101)/	Ordinary
BS Securities Australia 11d	9	009 0	(101,11)	Ciciliary
IDC Cocuration Australia 14d	200	5507	(00+)	Ciciliary
200 Seculities Abstralia LtO	1100	212	(57)	Ordinary
B3 Securities Mastralia Ltd	100	070'9	(00/)	Ordinary
JBS Securities Australia Ltd	Sell	103	(12)	Ordinary
JBS Securities Australia Ltd	Sell	8,824	(1,026)	Ordinary
BS Securities Australia Ltd	las	6.158	(716)	Ordinary
BS Securities Australia Ltd	les.	1 231	(571)	Ordinan
RS Securities Australia 11d	وا	, and	(1)	O'chillian
DC Commission Constraint and		06	7117	Clumary
DO DECURISES MUSICIANIA ENG		27/	(38)	Ordinary
DS Securities Australia Ltd	Į,	5,384	(929)	Ordinary
BS Securities Australia Ltd	Sell	95	(11)	Ordinary
BS Securities Australia Ltd	Sell	5,100	(263)	Ordinary
IBS Securities Australia Ltd	Sell	127,418	(14.816)	Ordinary
BS Securities Australia Ltd	Sell	3.070	(357)	Ordinary
BS Securities Australia Ltd	Sell	9595	(458)	Ordinan
IRS Securities Australia Ind	و	000.71	(000 0	Contract of the contract of th
BS Securities Australia 11d	# P	1001	(2,000)	o diam's
Day Securities According to	130	610,1	(8 1)	Ordinary
65 Securities Australia Ltd	Sell	267	(31)	Ordinary
BS Securities Australia Ltd	Sell	20,434	(2,376)	Ordinary
UBS Securities Australia Ltd	Sell	533	(62)	Ordinary
BS Securities Australia Ltd	Sell	6.072	(206)	Ordinary
BS Securities Australia Ltd	Sell	16 693	(1941)	Ordinary
BS Securities Australia Ltd	les	8 170	(050)	Ordinary
DC Committee Americal 144	=30	0/1/0	(006)	Ordinary
les securines Australia Ltd	PC	206	(24)	Ordinary
UBS Securities Australia Ltd	sell	6,054	(704)	Ordinary
UBS Securities Australia Ltd	Seli	220	(64)	Ordinary
UBS Securities Australia Ltd	Sett	2,778	(323)	Ordinary
UBS Securities Australia Ltd	Sell	1.746	(203)	Ordinary
UBS Securities Australia Ltd	Sell	22 300	(5 693)	Ordinary
UBS Securities Australia I1d	Sell	17	(0)	Ordinani
DC Cocusting Australia 14d	3	470.74	(300.0)	O'CHING)
ODS Securities Australia Liu	100	15,451	((58/,1)	Ordinary
UBS Securities Australia Ltd	Sell	8,832	(1,027)	Ordinary
UBS Securities Australia Ltd	Sell	396	(46)	Ordinary
JBS Securities Australia Ltd	Sell	4.756	(496)	Ordinary
UBS Securities Australia (16)	lla'S	16.009	(5,072)	Ordinary
IDC Cocumition Americal 144	Call	550,01	(7/0'1)	Cidilary
DO DECULINES AUSILIARIA LLU		7/1	(nz)	Ordinary
UBS Securities Australia Ltd	Jell Jell	45,456	(5,403)	Ordinary
UBS Securities Australia Ltd	Sell	22,781	(2,649)	Ordinary
UBS Securities Australia Ltd	Selt	23,065	(2,682)	Ordinary
UBS Securities Australia Ltd	Sel	13 5/15	(1 575)	Ordinan
PC Securities Australia Ital	9	10,000	(0.000)	Commany
OBS Securities Australia Lid		006/61	(4,5,4)	Ordinary
85 Securities Australia Ltd	Xei	9,013	(1,048)	Ordinary
BS Securities Australia Ltd	les.	3,982	(463)	Ordinary
UBS Securities Australia Ltd	Sell	2 391	(978)	Ordinary
DC Committee Australia 144	Coll	100/1	(5) (7)	Oldinary
Bs securities Australia Ltd			1	
	-	553	(29)	Ordinary
BS Securities Australia Ltd	Sel	533	(62)	Ordinary

15-Nov-14 16-Nov-14 16-Nov-14 16-Nov-14	UBS Securities Australia Ltd UBS Securities Australia Ltd	Sell	30,247	25.55	
06-Nov-14	Opp permittee Australia LID	9	400	(0.12)	Commany
06-Nov-14	LIRS Securities Australia 11d	3	10 168	(00)	Grainary
	UBS Securities Australia Ltd	Jag.	1214		Ordinary
06-Nov-14	UBS Securities Australia Ltd	Seil	34,561		Ordinary
D6-Nov-14	UBS Securities Australia Ltd	Sell	1,300		Ordinary
36-Nov-14	UBS Securities Australia Ltd	Sell	10,883	(1,264)	Ordinary
06-Nov-14	UBS Securities Australia Ltd	Sell	319		Ordinary
Nov-14	UBS Securities Australia Ltd	Sell	5,011	(285)	Ordinary
lov-14	UBS Securities Australia Ltd	Sell	11		Ordinary
U5-Nov-14	UBS Securities Australia Ltd	Nell Nell	13,053		Ordinary
20-100v-14	Upo Securities Australia Ltd	Į, j	401,20 407,01		Orginary
06-Nov-14	LIBS Securities Australia Itd	Sell	197,201	(007,21)	Ordinary
D6-Nov-14	UBS Securities Australia Ltd	Sell	288	(103)	Ordinary
06-Nov-14	UBS Securities Australia Ltd	Sell	968	(104)	Ordinary
06-Nov-14	UBS Securities Australia Ltd	Sell	4,060		Ordinary
D6-Nov-14	UBS Securities Australia Ltd	Sell	1,017		Ordinary
	UBS Securities Australia Ltd	Sell	3,043	(828)	Ordinary
	UBS Securities Australia Ltd	Sell	4,060		Ordinary
- 1	UBS Securities Australia Ltd	Sell	4,060	(471)	Ordinary
	UBS Securities Australia Ltd	Seil	1,017		Ordinary
1	UBS Securities Australia Ltd	Sel	45,031	(5,224)	Ordinary
1	UBS Securities Australia Ltd	Ser.	4,060	(471)	Ordinary
De-Nov-14	UBS Securities Australia Ltd	Sell	48,556	(5,633)	Ordinary
ı	LIBC Cocuration Augustic 14d	Coll	1,044	(171)	Guinary
6-Nov-14	LIBS Securities Australia 11d	وا	7 DEC	(127)	Ordinary
ŀ	UBS Securities Australia Ltd	les	4 065	(471)	Ordinary
6-Nov-14	UBS Securities Australia Ltd	Sell	4 305	(500)	Ordinary
Vov-14	UBS Securities Australia Ltd	Sell	7.620	(885)	Ordinary
JOV-14	UBS Securities Australia Ltd	Sell	113,600	(13,194)	Ordinary
6-Nov-14	UBS Securities Australia Ltd	Sell	1,756	(204)	Ordinary
36-Nov-14	UBS Securities Australia Ltd	Sell	1,326	(154)	Ordinary
36-Nov-14	UBS Securities Australia Ltd	Sell	108,337	(12,590)	Ordinary
06-Nov-14	UBS Securities Australia Ltd	Sell	119,196	(13,852)	Ordinary
D6-Nov-14	UBS Securities Australia Ltd	Sell	602	(02)	Ordinary
D6-Nov-14	UBS Securities Australia Ltd	Sell	61,156	(7.107)	Ordinary
Jov-14	UBS Securities Australia Ltd	Sell	14,310	(1,663)	Ordinary
06-Nov-14	UBS Securities Australia Ltd	Sell	17,468	(2,030)	Ordinary
lov-14	UBS Securities Australia Ltd	Buy	15.944	1.854	Ordinary
06-Nov-14	UBS Securities Australia Ltd	Sell	29 249	(3 405)	Ordinary
lov-14	UBS Securities Australia Ltd	- Se	£ 00 07	(5,707)	Ordinary
lov-14	UBS Securities Australia Ltd	Sell	797.17	(7.537)	Ordinary
06-Nov-14	UBS Securities Australia I td	Sell	3,029	(353)	Ordinary
	UBS Securities Australia Ltd	Sell	1 012	(118)	Ordinary
l	HBC Societies Australia Itcl	lay	13	(6)	Ordinary
Nov-14	LIBS Securities Australia 11d	Sell	100	(0)	Ordinary
1	LIBC Countries Australia 1+d	اق	4 041	(1/4/)	Ordinary
Nov-14	LIBC Cognition Australia 1+d	ll Soll	140,4	(471)	Crainary
ļ	TIDE Countries Australia Ind	Coll	10,01	(471)	Ordinary
	11RS Securities Australia Ltd	Sell	9,02	(353)	Ordinary
ı	IDS Countries Australia 144	Sell	50,0	(323)	Ordinary
1 10	UBS Securities Australia Ltd	Call	210,1	(118)	Ordinary
11-NON-90	Tibe Countries Australia Ltd	Sell	50,6	(353)	Ordinary
De Mov 14	Upo Countries Australia Ltd	Coll	210,1	(118)	Ordinary
1 1	Upo Securities Australia Ltd	Jet S	210,1	(911)	Orginary
UB-NOV-14	UBS Securities Australia Ltd	361	210,1	(118)	Ordinary
DV-14	UBS Securities Australia Ltd	liac I	210,1	(118)	Ordinary
06-Nov-14	UBS Securities Australia Ltd	Self	1,012	(118)	Ordinary
06-Nov-14	UBS Securities Australia Ltd	Sell	4,041	(471)	Ordinary
06-Nov-14	UBS Securities Australia Ltd	Sell	34	(4)	Ordinary
06-Nov-14	UBS Securities Australia Ltd	Sell	21,450	(2,500)	Ordinary
06-Nov-14	UBS Securities Australia Ltd	les!	24,307	(2,833)	Ordinary
96-Nov-14	UBS Securities Australia Ltd	Sell	71/1	(3,354)	Ordinary
36-Nov-14	UBS Securities Australia Ltd	Sell	6,855	(667)	Ordinary
36-Nov-14	UBS Securities Australia Ltd	Sell	10,245	(1,194)	Ordinary
06-Nov-14	UBS Securities Australia Ltd	Sell	4,295	(200)	Ordinary
06-Nov-14	UBS Securities Australia Ltd	Sell	12,885	(1,500)	Ordinary
36-Nov-14	UBS Securities Australia Ltd	Sell	7,602	(882)	Ordinary
06-Nov-14	UBS Securities Australia Ltd	Sell	71,666	(8,343)	Ordinary
ov-14	UBS Securities Australia Ltd	Sell	197,793	(23,026)	Ordinary
06-Nov-14	UBS Securities Australia Ltd	Sell	1.084	(126)	Ordinary
1:	כמה שניים שמני מוש דיים	= 1		1/021/	A ALID PROPERTY

07-Nov-14	UBS Securities Australia Ltd	, J	6 000 96	(1)	Ordinary
П	UBS Securities Australia Ltd	Sell	50,880	(5,855)	Ordinary
	UBS Securities Australia Ltd	Buy	28,912	3,327	Ordinary
-	UBS Securities Australia Ltd	Sett	1,660	(161)	Ordinary
	UBS Securities Australia Ltd	Buy	99,170	11,412	Ordinary
١	UBS Securities Australia Ltd	Buy	195,073	22,448	Ordinary
	UBS Securities Australia Ltd	Buy	31,545	3,630	Ordinary
-	UBS Securities Australia Ltd	Buy	3,415	393	Ordinary
1	UBS Securities Australia Ltd	Buy	286,831	33,007	Ordinary
1	UBS Securities Australia Ltd	Buy	11,749	1,352	Ordinary
ı	UBS Securities Australia (10	Bury	50,411	5,801	Ordinary
07-Nov-14	UBS Securities Australia Ltd	Buy	13.765	1 584	Ordinary
1	UBS Securities Australia Ltd	Buy	495	57	Ordinary
łl	UBS Securities Australia Ltd	Buy	40,382	4,647	Ordinary
Ш	UBS Securities Australia Ltd	Buy	19	4	Ordinary
	UBS Securities Australia Ltd	Buy	7,256	835	Ordinary
- 1	UBS Securities Australia Ltd	Buy	61	7	Ordinary
07-Nov-14	UBS Securities Australia Ltd	Buy	93,296	10,736	Ordinary
07-Nov-14	UBS Securities Australia Ltd	Buy	317,246	36,507	Ordinary
07-Nov-14	URS Securities Australia Ltd	R <sub>IIV</sub>	10673	57,55	Ordinary
107-Nov-14	UBS Securities Australia Itd	Ruy	127,55	4,053	Ordinary
07-Nov-14	UBS Securities Australia Ltd	Buy	297.902	34 781	Ordinary
07-Nov-14	UBS Securities Australia Ltd	Buy	7,795	268	Ordinary
07-Nov-14	UBS Securities Australia Ltd	Buy	19	7	Ordinary
07-Nov-14	UBS Securities Australia Ltd	Buy	19	7	Ordinary
07-Nov-14	USS Securities Australia 11d	Buy	119 971	10 60	Ordinary
07-Nov-14	UBS Securities Australia Ltd	Buy	069'8	1,000	Ordinary
07-Nov-14	UBS Securities Australia Ltd	Buy	19	7	Ordinary
07-Nov-14	UBS Securities Australia Ltd	lles.	110,120	(12,672)	Ordinary
09-Nov-14	UBS Securities Australia Ltd	Contract	AN.	(11 400)	Ontions
10-Nov-14	UBS Global Asset Management	Buy	6,364	718	Ordinary
10-Nov-14	UBS Global Asset Management	Buy	169,388	19,112	Ordinary
10-Nov-14	UBS Securities Australia Ltd	lles de	2,358,975	(271,343)	Ordinary
10-Nov-14	UBS Securities Australia Ltd	Sell	28,256	3,193	Ordinary
10-Nov-14		Sell	601,674	(508'24)	Ordinary
П	UBS Securities Australia Ltd	Buy	48,739	5,501	Ordinary
10-Nov-14	UBS Securities Australia Ltd	Seii	146,875	(16,813)	Ordinary
1	UBS Securities Australia Ltd	Sell	396,904	(44,854)	Ordinary
	UBS Securities Australia Ltd	Je Je	845,407	(95,524)	Ordinary
10-Nov-14	UBS Securities Australia Ltd	Buy	7.759	(2,000,0)	Ordinary
	UBS Securities Australia Ltd	Buy	108,764	12,284	Ordinary
П	UBS Securities Australia Ltd	Sell	5,275	(594)	Ordinary
-	UBS Securities Australia Ltd	Sell	66,364	(7,500)	Ordinary
1	UBS Securities Australia Ltd	Sell	219,918	(25,067)	Ordinary
l	UBS Securities Australia Ltd	36	391,692	(44,239)	Ordinary
	UBS Securities Australia (1d	Sell	545 758	(484,147)	Ordinary
П	UBS Securities Australia Ltd	Stock returned	WA	(000'2)	Ordinary
	UBS AG, London Branch	Sell	396,904	(44,854)	Ordinary
10-Nov-14	UBS AG, London Branch	Buy	492,988	44,854	Ordinary
ı	UBS AG Australia Branch	Jell Sell	466,377	(52,595)	Ordinary
1	UBS AG Australia Branch	lle S	11 158	(8,869)	Ordinary
10-Nov-14	UBS AG Australia Branch	Sell	669	(62)	Ordinary
10-Nov-14	UBS AG Australia Branch	Buy	660.563	74.455	Ordinary
10-Nov-14	UBS AG Australia Branch	Sell	402,547	(45,480)	Ordinary
10-Nov-14	UBS AG Australia Branch	Sell	1,863,518	(210,466)	Ordinary
10-Nov-14	UBS AG Australia Branch	Sell	518,230	(58,529)	Ordinary
10-Nov-14	UBS AG Australia Branch IIIRS Wealth Management Australia Ltd	liac l	129,130	(14,584)	Ordinary
10-Nov-14	UBS Securities Australia Ltd	Sell	1,373	(155)	Ordinary
10-Nov-14	UBS Securities Australia Ltd	Se⊞	988	(100)	Ordinary
10-Nov-14	UBS Securities Australia Ltd	Sell	7,983	(901)	Ordinary
10-Nov-14	UBS Securities Australia Ltd	100	4,811	(543)	Ordinary
Ι	UBS Securities Australia Ltd	- X	25,433	(2,843)	Ordinary
10-Nov-14	UBS Securities Australia Ltd	ll se	7,761	(876)	Ordinary

06-Nov-14	35 Securities Australia (1d	Seli	348 000	(40,000)	Ordinary
	UBS Securities Australia Ltd	Sell	072.97	(9 196)	Ordinary
П	UBS Securities Australia Ltd	Sell	086'9		Ordinary
36-Nov-14 UE	35 Securities Australia Ltd	Sell	35,497	(4,118)	Ordinary
ā	35 Securities Australia Ltd	Sell	129,300		Ordinary
	3S Securities Australia Ltd	Sell	52		
06-Nov-14	35 Securities Australia Ltd	Sell	86,200		1
	So securities Australia Liu	buy Coll	25,400		
	S Securities Australia 11d	les.	785 013	(A57 52)	
06-Nov-14	SS Securities Australia Ltd	Sell	127,774		l
ı	35 Securities Australia Ltd	Sell	50.884		
13	\$5 Securities Australia Ltd	Buy	2.896		
13	UBS Securities Australia Ltd	Buy	22,179		
13	UBS Securities Australia Ltd	Sell	2,026		
06-Nov-14 ∪E	UBS Securities Australia Ltd	Seli	302		
3	UBS Securities Australia Ltd	Self	1,905		
n	15 Securities Australia Ltd	Sell	1,603		
ñ	UBS Securities Australia Ltd	Sell	166,142	(19,274)	
5	UBS Securities Australia Ltd	Sell	29,756		
Š	UBS Securities Australia Ltd	Sell	309	(36)	Ordinary
ĵ	JBS Securities Australia Ltd	Buy	7,801,100		Ordinary
ă	UBS Securities Australia Ltd	Sell	3,103,200	0	Ordinary
1	is Giobal Asset Management	les.	49,709	9)	Ordinary
-Nov-14	UBS Securities Australia Ltd	Seil	4,251		Ordinary
5	JBS Securities Australia Ltd	Buy	11,383		Ordinary
1	UBS Securities Australia Ltd	Buy	43,550	2,000	Ordinary
5	JBS Securities Australia Ltd	Buy	67,472		
5 5	USS Securities Australia LTG	lac	879,87	(5,594)	
5 2	Obstantials Australia Ltd	D. m.	006,007		
9 9	Sociation Australia Ind	(pa)	45,450	5,000	Ordinary
ğ		N. C.	155,555	500.01	
3 8	S Securities Australia 11d	100	PO 344	(690.6)	Ordinary
NBO NBO	S Securities Australia Ltd	Ang.	7 610 180	300,005	Ordinary
UBS	S Securities Australia Ltd	Sell	25,215,2 2715,352		Ordinary
S	S Securities Australia Ltd	les.	266,617,7		Ordinary
nBS	S Securities Australia Ltd	Sell	3 852 480		Ordinary
-Nov-14	S Securities Australia Itd	Stock returned	DAY.		Ordinary
Ī		: 1	CE9 P36	90.754	Ordinary
15	S AG. London Branch	Sell	788 988		Ordinary
15	S AG Australia Branch	[las	0550		Ordinary
5 5	S AG Australia Branch	Buy	565,476	65,000	Ordinary
15	S AG Australia Branch	Sell	1 802 943		Ordinary
ñ	S AG Australia Branch	Sell	67.874		Ordinary
3	S AG Australia Branch	Sell	3,041,445	(350,093)	Ordinary
15	S Wealth Management Australia Ltd	Buy	5,994	689	Ordinary
15	S Securities Australia Ltd	Buy	367,833	42.672	Ordinary
3	UBS Securities Australia Ltd	Buy	435	950	Ordinary
5	S Securities Australia Ltd	Buy	400		Ordinary
9	S Securities Australia Ltd	Sell	5,475		Ordinary
155	UBS Securities Australia Ltd	Sell	21,160	(2,435)	Ordinary
15	S Securities Australia Ltd	Sell	58.006		Ordinary
15	S Securities Australia Ltd	Sell	243		Ordinary
	S Securities Australia Ltd	Sell	43,450	(5,000)	Ordinary
7-Nov-14 UB	S Securities Australia Ltd	Seli	11,471	(1,320)	Ordinary
	S Securities Australia Ltd	Sell	8,186	(942)	Ordinary
3	S Securities Australia Ltd	Sell	32,153	(3.700)	Ordinary
B)	S Securities Australia Ltd	Sell	10,01	(1,152)	Ordinary
UB	5 Securities Australia Ltd	Sell	17	(2)	Ordinary
OBS	Securitie	Sell	23,350	(2,687)	Ordinary
NBS	Securitie	Sell	148	(21)	Ordinary
UBS	Securitie	Sell	5,440	(929)	Ordinary
UBS		Sell	20,873	(2,402)	Ordinary
<u>s</u>	Securitie	Sell	4,623	(532)	Ordinary
OBS	Securities	Sell	10,237	(1,178)	Ordinary
NBS	Securities	Sell	6		Ordinary
ngs	Securities	Sell	91,784	)	Ordinary
SS.	Securities	Sell	14,199		Ordinary
SB S	Securities	Sei	40,243	(4,631)	Ordinary
OBS	s Securities Australia Ltd	Sell	19,364		Ordinary
8	5 Securities Australia Ltd	Sell	478		Ordinary
NB.	s Securities Australia Ltd	Sell	7,821		Ordinary
an B	UBS Securities Australia Ltd	Sell	5,770	(664)	Ordinary

10-Nov-14 10-Nov-14		9	30 585	1037 67	Ordinan
4	UBS Securities Australia Itel	Į (	50,565	(3,452)	Ordinary
	UBS Securities Australia Ltd	Sell	881.136	(99 451)	Ordinary
7	UBS Securities Australia Ltd	Buy	637	73	Ordinary
0-Nov-14	UBS Securities Australia Ltd	Sell	419	(48)	Ordinary
4	UBS Securities Australia Ltd	Sell	288	(33)	Ordinary
4	UBS Securities Australia Ltd	Buy	26,520	3,000	Ordinary
4	UBS Securities Australia Ltd	Buy	8,840	1,000	Ordinary
4	UBS Securities Australia Ltd	Sell	22,138	(2,500)	Ordinary
4	UBS Securities Australia Ltd	Sell	1,513	(171)	Ordinary
4	UBS Securities Australia Ltd	Sell	44,275	(2,000)	Ordinary
4	UBS Securities Australia Ltd	Sell	3,248	(367)	Ordinary
4	UBS Securities Australia Ltd	Sell	1,071	(121)	Ordinary
4	UBS Securities Australia Ltd	Sell	3,850	(435)	Ordinary
14	UBS Securities Australia Ltd	Sell	20,859	(2,357)	Ordinary
4	UBS Securities Australia Ltd	Sell	1,336	(151)	Ordinary
4	UBS Securities Australia Ltd	Sell	3,894	(440)	Ordinary
2	UBS Securities Australia Ltd	Sell	6,372	(720)	Ordinary
4	UBS Securities Australia Ltd	Sell	5,965	(674)	Ordinary
4	UBS Securities Australia Ltd	Sell	35	(4)	Ordinary
4	UBS Securities Australia Ltd	Sell	4,655	(526)	Ordinary
4	UBS Securities Australia Ltd	Sell	2,797	(316)	Ordinary
4	UBS Securities Australia Ltd	Sell	6,222	(203)	Ordinary
14	UBS Securities Australia Ltd	Sell	2,319	(292)	Ordinary
4	UBS Securities Australia Ltd	Sell	1,009	(114)	Ordinary
14	UBS Securities Australia Ltd	Sell	1,310	(148)	Ordinary
10 Nov 14	UBS Securities Australia Ltd	Sell	1,292	(146)	Ordinary
14	UBS Securities Australia Ltd	Sell	8,939	(1,010)	Ordinary
4	UBS Securities Australia Ltd	Seil	1,292	(146)	Ordinary
0-NOV-14	UBS Securities Australia Ltd	Ser	4,204	(475)	Ordinary
0-Nov-14	UBS Securities Australia Ltd	Sel.	4,823	(545)	Ordinary
1 3	UBS Securities Australia Ltd	Self	2,655	(300)	Ordinary
*	UBS Securities Australia Ltd	Dell.	080'/	(800)	Ordinary
2 :	UBS Securities Australia Ltd		49,843	(5,632)	Ordinary
10-Nov-14	UBS Securities Australia Ltd	J. Sell	42,108	(4,758).	Ordinary
4	UBS Securities Australia Ltd	Į,	1,053	(119)	Ordinary
1 5	OBS Securities Australia Ltd	100	51,649	(5,836)	Ordinary
1 2	UBS Securities Australia Ltd	E 3	115	(13)	Ordinary
	LIRK Countries Australia 144	loc Jon	804,2	(6/2)	Ordinary
<u>.</u>	UDS Countries Accurate Ltd	1120	1,434	(791)	Ordinary
	UBS Societies Australia Ltd	1361	14,886	(789,1)	Ordinary
	OBS Securities Australia Ltd	100	12,538	(1,428)	Ordinary
2 5	Obs securities Australia Ltd	000	7,873	(887)	Ordinary
	LIDE Cocuries Australia Ltd	1140	21,845	(7,457)	Ordinary
	PIEC Countries Australia Ltd	lles.	14,139	(885)1)	Ordinary
1 5	UBS Securities Australia Ltd	ne S	4,315	(487)	Ordinary
<u> </u>	UBS Securities Australia Ltd	ll Sell	792	(30)	Ordinary
<u> </u>	UBS Securities Australia Ltd	Nell Sell	21,243	(2,399)	Ordinary
2 5	Ups Securities Australia Ltd	liac d	8,825	(966)	Ordinary
1 2	UBS Securities Australia Ltd	100	515,1	(1/1)	Ordinary
4 2	UBS Securities Australia LTO	Nell Sell	815	(92)	Ordinary
	UBS Securities Australia Ltd	) Sell	6,601	(745)	Ordinary
	Los securities Australia Ltd	nec nec	1/8/1	(717)	Ordinary
O Nov. 14	Los Courieiro Australia Ltd	1100	177	(57)	Ordinary
1 5	Libe countries Australia Ltd	100	507,6	(644)	Ordinary
<u>.</u>	UBS Securities Australia Ltd	Ē,	8,049	(606)	Ordinary
0-NOV-14	UBS Securities Australia Ltd	Seil	3,511	(366)	Ordinary
4 .	UBS Securities Australia Ltd	) sell	5,567	(628)	Ordinary
4	UBS Securities Australia Ltd	Seil	3,280	(370)	Ordinary
0-Nov-14	UBS Securities Australia Ltd	Sell	816	(95)	Ordinary
10-Nov-14	UBS Securities Australia Ltd	Sell	44,325	(2,000)	Ordinary
0-Nov-14	UBS Securities Australia Ltd	Sell	44,325	(5,000)	Ordinary
0-Nov-14	UBS Securities Australia Ltd	Sell	44,600	(5,031)	Ordinary
4	UBS Securities Australia Ltd	Sell	24,547	(2,769)	Ordinary
10-Nov-14	UBS Securities Australia Ltd	Sell	78,021	(8,801)	Ordinary
4	UBS Securities Australia Ltd	Sell	26,258	(2,962)	Ordinary
10-Nov-14	UBS Securities Australia Ltd	Sell	541	(19)	Ordinary
0-Nov-14	UBS Securities Australia Ltd	Sell	700	(62)	Ordinary
0-Nov-14	UBS Securities Australia Ltd	Sell	3,608	(407)	Ordinary
0-Nov-14	UBS Securities Australia Ltd	Sell	399	(45)	Ordinary
J-Nov-14	UBS Securities Australia Ltd	Seli	13,200	(1,489)	Ordinary
4	IJRS Global Asset Management	روا	1 840	(27.47)	in in
1-1404-14	Obs Ground Asset Invariagement	7511	10001	(417)	Cluridiy

11   12   12   13   13   13   13   13	1-Nov-14	UBS Securities Australia Ltd	Buy	5,196	009	Ordinary
Bits Securine Australia (14)         Bay         (675)         (676)           Bits Securine Australia (14)         Bay         (750)         (750)           Bits Securine Australia (14)         <	-14	UBS Securities Australia Ltd	Buy	4,044	467	Ordinary
Bits Securities Autorials Ltd         Bits Securities Autorials Ltd         Bits Autorials Ltd <td>4</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>6,053</td> <td>669</td> <td>Ordinary</td>	4	UBS Securities Australia Ltd	Buy	6,053	669	Ordinary
Bits Securine Australia Ltd         Bity         668         1 87           Bits Securine Australia Ltd         Bity         1 673         2 60           Bits Securine Australia Ltd         Bity         1 673         2 60           Bits Securine Australia Ltd         Bity         1 673         2 60           Bits Securine Australia Ltd         Bity         1 673         2 60           Bits Securine Australia Ltd         Bity         1 673         2 60           Bits Securine Australia Ltd         Bity         1 673         2 60           Bits Securine Australia Ltd         Bity         2 70         2 60           Bits Securine Australia Ltd         Bity         2 70         2 70           Bits Securine Australia Ltd         Bity         2 70         2 70           Bits Securine Australia Ltd         Bity         2 70         2 70           Bits Securine Australia Ltd         Bity         2 70         2 70           Bits Securine Australia Ltd         Bity         2 70         2 70           Bits Securine Australia Ltd         Bity         2 70         2 70           Bits Securine Australia Ltd         Bity         2 70         2 70           Bits Securine Australia Ltd         Bity         2	14	UBS Securities Australia Ltd	Buy	12,990	1,500	Ordinary
(8) Securite Autrials Led         (8) Securities Autrials Led         (8) Securities Autrials Led         (8) Securities Autrials Led         (8) Securities Autrials Led         (8) Securities Autrials Led         (8) Securities Autrials Led         (8) Securities Autrials Led         (8) Securities Autrials Led         (8) Securities Autrials Led         (8) Page Securities Autrials Led         (8) Securities Autrials Led         (8) Page Securities Autrials Led         (8) Page Securities Autrials Led         (8) Page Securities Autrials Led         (8) Page Securities Autrials Led         (8) Page Securities Autrials Led         (8) Page Securities Autrials Led         (8) Page Securities Autrials Led         (8) Page Securities Autrials Led         (8) Page Securities Autrials Led         (8) Page Securities Autrials Led         (8) Page Securities Autrials Led         (8) Page Securities Autrials Led         (8) Page Securities Autrials Led         (8) Page Secu	14	UBS Securities Australia Ltd	Buy	658	9/	Ordinary
QUESTIONING ALMERIALISTIC         BOTATION 17,370         2,000           DESTIGNMENT ALMERIALISTIC         BOTATION 17,370         2,000           DESTIGNMENT ALMERIALISTIC         BOTATION 17,370         1,400           DESTIGNMENT ALMERIALISTIC         BOTATION 17,370         1,500           DESTIGNMENT ALMER	-14	UBS Securities Australia Ltd	Buy	16,082	1,857	Ordinary
USS SCROTTER ALACTION   USA	17	UBS Securities Australia (14	any.	17,320	2,000	Ordinary
USS SCROTTER Adminish 1544   Say	114	LIBS Securities Australia Ltd	Son A	1,247	144	Ordinary
UBS Securine Australia Ltd   Boy   Carlot   Ca	4-14	UBS Securities Australia Ind	Rick	6/5	8/ 503	Ordinary
Bis Securine Australia Ltd   Buy   1,156   1,555   1	v-14	UBS Securities Australia Ltd	Buy	4.408	5002	Ordinary
UBS Securities Australia Ltd   Bay   18,886   2159     UBS Securities Australia Ltd   Bay   18,986   24,995   24,995     UBS Securities Australia Ltd   Bay   1,995   1,995   2,995     UBS Securities Australia Ltd   Bay   1,995   1,995   2,995     UBS Securities Australia Ltd   Bay   1,995   1,995   2,995   2,995     UBS Securities Australia Ltd   Bay   1,995   1,995   2	4-14	UBS Securities Australia Ltd	Buy	2,165	250	Ordinary
(BS) Securines Autralia (LOI Bay)         Bay         (19,58)         2,158           (BS) Securines Autralia (LOI Bay)         (ASP)         4,259         4,55           (BS) Securines Autralia (LOI Bay)         (ASP)         4,50         5,7           (BS) Securines Autralia (LOI Bay)         (ASP)         2,62         1,50           (BS) Securines Autralia (LOI Bay)         (ASP)         2,62         1,42           (BS) Securines Autralia (LOI Bay)         (ASP)         1,62         2,12           (BS) Securines Autralia (LOI Bay)         (ASP)         1,62         2,12           (BS) Securines Autralia (LOI Bay)         (ASP)         1,63         2,12           (BS) Securines Autralia (LOI Bay)         (ASP)         1,62         2,24           (BS) Securines Autralia (LOI Bay)         (ASP)         1,63         2,24           (BS) Securines Autralia (LOI Bay)         (ASP)         2,12         2,24           (BS) Securines Autralia (LOI Bay)         (ASP)         2,12         2,24 <td>v-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>61</td> <td>7</td> <td>Ordinary</td>	v-14	UBS Securities Australia Ltd	Buy	61	7	Ordinary
OFF SECURITY AND ADMINISTRATION OF ADMINIST	4-14	UBS Securities Australia Ltd	Buy	18,688	2,158	Ordinary
(BS) Securines Autorials (14)         (BW)         (20,23)         (2,53)           (BS) Securines Autorials (14)         (BW)         (2,022)         (2,53)           (BS) Securines Autorials (14)         (BM)         (2,022)         (2,53)           (BS) Securines Autorials (14)         (BM)         (3,24)         (3,53)           (BS) Securines Autorials (14)         (BM)         (3,24)         (3,54)           (BS) Securines Autorials (14)         (BM)         (3,24)         (3,64)           (BS) Securines Autorials (14)         (BM)         (3,14)         (3,14)	1 4	UBS Securities Australia Ind	Buy	4,295	496	Ordinary
(B) Securities Autralia Ind         Bay         17,003         1,393           (B) Securities Autralia Ind         Bay         1,248         1,548           (B) Securities Autralia Ind         Bay         1,548         1,548           (B) Securities Autralia Ind         Bay         1,549         1,54           (B) Securities Autralia Ind         Bay         1,647         1,54           (B) Securities Autralia Ind         Bay         1,627         1,64           (B) Securities Autralia Ind         Bay         1,627         1,64           (B) Securities Autralia Ind         Bay         1,627         2,64           (B) Securities Autralia Ind         Bay         1,627         2,64           (B) Securities Autralia Ind         Bay         1,629         1,67           (B) Securities Autralia Ind         Bay         1,620         2,51           (B) Securities Autralia Ind         Bay         1,620         2,61           (B) Securities Autralia Ind         Bay         1,62         2,61           (B) Securities Autralia Ind         Bay         1,73         1,73           (B) Securities Autralia Ind         Bay         1,73         1,73           (B) Securities Autralia Ind         Bay         1,73 </td <td>v-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>701</td> <td>75</td> <td>Ordinary</td>	v-14	UBS Securities Australia Ltd	Buy	701	75	Ordinary
URS Securities Australia Ltd         Bay         24,923         2,873           URS Securities Australia Ltd         Bay         1,546         2,873           URS Securities Australia Ltd         Bay         1,546         1,546           URS Securities Australia Ltd         Bay         1,524         1,546           URS Securities Australia Ltd         Bay         1,2,548         1,427           URS Securities Australia Ltd         Bay         1,2,548         1,427           URS Securities Australia Ltd         Bay         2,1822         2,281           URS Securities Australia Ltd         Bay         2,1822         2,281           URS Securities Australia Ltd         Bay         2,1822         2,281           URS Securities Australia Ltd         Bay         2,1822         2,282           URS Securities Australia Ltd         Bay         2,1822         2,282           URS Securities Australia Ltd         Bay         2,182         2,082           092           URS Securi	v-14	UBS Securities Australia Ltd	Buy	12.063	- FOF 1	Ordinary
Use Securines Australia Ltd	v-14	UBS Securities Australia Ltd	Buy	24,923	2.878	Ordinary
(18) Securities Australia (Led         Buy         73.98         1,546           (18) Securities Australia (Led         Buy         75.91         34.52         34.61           (18) Securities Australia (Led         Buy         7.25         34.61         1.42           (18) Securities Australia (Led         Buy         2.74         2.75         3.62           (18) Securities Australia (Led         Buy         2.74         2.75         3.62           (18) Securities Australia (Led         Buy         2.74         2.75         3.62           (18) Securities Australia (Led         Buy         2.75         2.75         3.62         4.60           (18) Securities Australia (Led         Buy         2.75         2.82         4.60         1.75         3.75	w-14	UBS Securities Australia Ltd	Buy	25	m	Ordinary
DES Securities Australia Ltd         BMY         3,15,2         847           DES Securities Australia Ltd         BMY         3,15,2         847           DES Securities Australia Ltd         BMY         1,2,56         1,422           DES Securities Australia Ltd         BMY         7,0,38         2,52,1           DES Securities Australia Ltd         BMY         7,1,52,6         1,52,6           DES Securities Australia Ltd         BMY         7,1,52,6         1,52,6           DES Securities Australia Ltd         BMY         7,1,52,6         2,52,1           DES Securities Australia Ltd         BMY         7,1,52,6         2,52,1 <t< td=""><td>w-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>13,388</td><td>1,546</td><td>Ordinary</td></t<>	w-14	UBS Securities Australia Ltd	Buy	13,388	1,546	Ordinary
USS Securities Autorials Ltd         BMY         57,24         374           USS Securities Autorials Ltd         BMY         172         374           USS Securities Autorials Ltd         BMY         17,20         37           USS Securities Autorials Ltd         BMY         61,80         25,51           USS Securities Autorials Ltd         BMY         61,82         2,51           USS Securities Autorials Ltd         BMY         61,82         2,51           USS Securities Autorials Ltd         BMY         18,82         2,51           USS Securities Autorials Ltd         BMY         1,62         2,51           USS Securities Autorials Ltd         BMY         1,63         2,52           USS Securities Autorials Ltd         BMY         1,63         2,52           USS Securities Autorials Ltd         BMY         1,73         2,60           USS Securities Autorials Ltd         BMY         4,25         3,60           USS Securities Autorials Ltd         BMY         4,25	14	UBS Securities Australia Ltd	Buy	753	87	Ordinary
UBS Securities Australia Ltd         Bby         1,23         1,23         1,23           UBS Securities Australia Ltd         Buy         1,23,58         1,427         2,04         1,03         1,427         1,03	w-14	URS Securities Australia 11d	Buy	3,152	36	Ordinary
UBS Securities Australia Ltd         Bay         1,2358         1,472           UBS Securities Australia Ltd         Bay         2,044         2,56           UBS Securities Australia Ltd         Bay         2,044         2,52           UBS Securities Australia Ltd         Bay         2,152         2,244           UBS Securities Australia Ltd         Bay         2,152         2,244           UBS Securities Australia Ltd         Bay         2,152         2,244           UBS Securities Australia Ltd         Bay         1,52         2,24           UBS Securities Australia Ltd         Bay         1,52         2,24           UBS Securities Australia Ltd         Bay         1,75         1,69           UBS Securities Australia Ltd         Bay         7,73         1,69           UBS Securities Australia Ltd         Bay         7,75         2,86           UBS Securities Australia Ltd         Bay         7,75         2,97           UBS Securities Australia Ltd         Bay	w-14	UBS Securities Australia Ltd	Buv	26 701	- 5	Orginary
UBS Securities Australia Ltd         Buy         2,044         2,36           UBS Securities Australia Ltd         Buy         2,1822         2,21           UBS Securities Australia Ltd         Buy         2,1822         2,21           UBS Securities Australia Ltd         Buy         2,1520         2,482           UBS Securities Australia Ltd         Buy         2,1520         2,685           UBS Securities Australia Ltd         Buy         2,500         2,686           UBS Securities Australia Ltd         Buy         2,7737         6,688           UBS Securities Australia Ltd         Buy         2,7737         6,688           UBS Securities Australia Ltd         Buy         2,7737         6,589           UBS Securities Australia Ltd         Buy         2,7737         6,589           UBS Securities Australia Ltd         Buy         2,7737         6,589           UBS Securities Australia Ltd         Buy         2,7737         6,590           UBS Securities Australia Ltd         Buy         2,7737         6,500           UBS Securities Australia Ltd         Buy         2,7737         6,500           UBS Securities Australia Ltd         Buy         2,7737         6,500         1,600           UBS Securi	w-14	UBS Securities Australia Ltd	Buy	12 358	1 427	Ordinary
UBS Securities Autorials Ltd         Bay         2.821           UBS Securities Autorials Ltd         Bay         2.821         2.521           UBS Securities Autorials Ltd         Bay         2.822         2.821           UBS Securities Autorials Ltd         Bay         2.15.20         2.485           UBS Securities Autorials Ltd         Bay         2.75.21         2.685           UBS Securities Autorials Ltd         Bay         2.75.72         6.689           UBS Securities Autorials Ltd         Bay         2.75.77         6.689           UBS Securities Autorials Ltd         Bay         2.75.77         6.689           UBS Securities Autorials Ltd         Bay         2.75.70         6.680           UBS Securities Autorials Ltd         Bay         2.75.70         6.680           UBS Securities Autorials Ltd         Bay         7.748         1.748           UBS Securities Autorials Ltd <t< td=""><td>w-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>2,044</td><td>236</td><td>Ordinary</td></t<>	w-14	UBS Securities Australia Ltd	Buy	2,044	236	Ordinary
UBS Securities Australia Ltd         Buy         226         33           UBS Securities Australia Ltd         Buy         21,832         2,521           UBS Securities Australia Ltd         Buy         11,250         2,485           UBS Securities Australia Ltd         Buy         10,383         1,199           UBS Securities Australia Ltd         Buy         10,288         1,10           UBS Securities Australia Ltd         Buy         2,737         6,693           UBS Securities Australia Ltd         Buy         2,737         6,693           UBS Securities Australia Ltd         Buy         2,737         6,693           UBS Securities Australia Ltd         Buy         5,737         6,693           UBS Securities Australia Ltd         Buy         5,737         6,693           UBS Securities Australia Ltd         Buy         5,737         6,693           UBS Securities Australia Ltd         Buy         2,890         1,10           UBS Securities Australia Ltd         Buy         2,890         1,10           UBS Securities Australia Ltd         Buy         2,890         1,00           UBS Securities Australia Ltd         Buy         2,890         1,00           UBS Securities Australia Ltd         Buy	w-14	UBS Securities Australia Ltd	Buy	61	7	Ordinary
UBS Securities Australia Ltd         BMy         21 832         2,221           UBS Securities Australia Ltd         BMy         21 832         2,221           UBS Securities Australia Ltd         BMy         21 832         2,231           UBS Securities Australia Ltd         BMy         21 824         400           UBS Securities Australia Ltd         BMy         3 464         400           UBS Securities Australia Ltd         BMy         3 759         88           UBS Securities Australia Ltd         BMy         7,737         6,638           UBS Securities Australia Ltd         BMy         7,759         88           UBS Securities Australia Ltd         BMy         7,759         88           UBS Securities Australia Ltd         BMy         7,759         88           UBS Securities Australia Ltd         BMy         7,748         1,748           UBS Securities Australia Ltd         BMy         7,748         1,748           UBS Securities Australia Ltd         BMy	×-14	UBS Securities Australia Ltd	Buy	286	33	Ordinary
UBS Securities Australia Ltd         BMy         715,20         2,485           UBS Securities Australia Ltd         BMy         71,250         2,485           UBS Securities Australia Ltd         BMy         7,737         2,698         17,99           UBS Securities Australia Ltd         BMy         7,737         6,698         17,09           UBS Securities Australia Ltd         BMy         7,737         6,698         1,79           UBS Securities Australia Ltd         BMy         7,737         6,698         1,79           UBS Securities Australia Ltd         BMy         7,737         6,698         1,79           UBS Securities Australia Ltd         BMy         7,737         6,698         1,70           UBS Securities Australia Ltd         BMy         7,737         6,698         1,00           UBS Securities Australia Ltd         BMy         7,12         1,00         1,00           UBS Securities Australia Ltd         BMy         7,12         1,00         1,00           UBS Securities Australia Ltd         BMy         7,14         1,00         1,00           UBS Securities Australia Ltd         BMy         2,80         1,00         1,00           UBS Securities Australia Ltd         BMy         2,	×-14	UBS Securities Australia Ltd	Buy	21,832	2,521	Ordinary
USS Securities Australia Ltd         Buy         1,12,20         2,189           USS Securities Australia Ltd         Buy         3,444         4,00           USS Securities Australia Ltd         Buy         3,444         4,00           USS Securities Australia Ltd         Buy         256         1,29           USS Securities Australia Ltd         Buy         2,50         1,29           USS Securities Australia Ltd         Buy         2,737         6,698           USS Securities Australia Ltd         Buy         2,737         6,698           USS Securities Australia Ltd         Buy         2,500         2,500           USS Securities Australia Ltd         Buy         2,500         2,500           USS Securities Australia Ltd         Buy         2,870         3,33           USS Securities Australia Ltd         Buy         2,870         3,37           USS Securities Australia Ltd         Buy         2,870         3,30           USS Securities Australia Ltd         Buy	177	Libs Securities Australia Ltd	ens)	182	21	Ordinary
UBS Securities Australia Ltd         Buy         1,232         1,198           UBS Securities Australia Ltd         Buy         1,068         124           UBS Securities Australia Ltd         Buy         170         13           UBS Securities Australia Ltd         Buy         57,737         6,698           UBS Securities Australia Ltd         Buy         57,737         6,698           UBS Securities Australia Ltd         Buy         57,737         6,698           UBS Securities Australia Ltd         Buy         4,276         88           UBS Securities Australia Ltd         Buy         2,15,500         9,70           UBS Securities Australia Ltd         Buy         3,60         9,70           UBS Securities Australia Ltd         Buy         3,60         9,70           UBS Securities Australia Ltd         Buy         3,60         9,70           UBS Securities Australia Ltd         Buy         1,15         1,1           UBS Securities Australia Ltd         Buy         3,60         1,0           UBS Securities Australia Ltd         Buy         3,70         4,0           UBS Securities Australia Ltd         Buy         1,1         1,1           UBS Securities Australia Ltd         Buy	14-14	UBS Securities Australia Ltd	Buy	21,520	2,485	Ordinary
UBS Securities Australia Ltd         Bluy         1,088         124           UBS Securities Australia Ltd         Buy         57,737         6,698           UBS Securities Australia Ltd         Buy         47,757         6,698           UBS Securities Australia Ltd         Buy         4,759         88           UBS Securities Australia Ltd         Buy         4,756         88           UBS Securities Australia Ltd         Buy         7,12         13           UBS Securities Australia Ltd         Buy         7,87         9           UBS Securities Australia Ltd         Buy         7,87         9           UBS Securities Australia Ltd         Buy         7,87         9           UBS Securities Australia Ltd         Buy         2,870         1,000           UBS Securities Australia Ltd         Buy         8,620         1,000           UBS Securities Australia Ltd         Buy	N-14	UBS Securities Australia Ltd	Buv	3 464	961,1	Ordinary
UBS Securities Australia Ltd         Buy         150         29           USS Securities Australia Ltd         Buy         57777         6698           USS Securities Australia Ltd         Buy         759         88           USS Securities Australia Ltd         Buy         215,500         25,000           USS Securities Australia Ltd         Buy         215,500         25,000           USS Securities Australia Ltd         Buy         78         40           USS Securities Australia Ltd         Buy         112         21           USS Securities Australia Ltd         Buy         112         35           USS Securities Australia Ltd         Buy         112         36           USS Securities Australia Ltd         Buy         155         100           USS Securities Australia Ltd         Buy         5,189         602           USS Securities Australia Ltd         Buy         5,189         6	N-14	UBS Securities Australia Ltd	Buy	1,068	124	Ordinary
UBS Securities Auxtraila Ltd         Buy         57737         6.588           UBS Securities Auxtraila Ltd         Buy         57737         6.688           UBS Securities Auxtraila Ltd         Buy         4276         488           UBS Securities Auxtraila Ltd         Buy         4276         486           UBS Securities Auxtraila Ltd         Buy         4276         25,000           UBS Securities Auxtraila Ltd         Buy         328         27           UBS Securities Auxtraila Ltd         Buy         346         13           UBS Securities Auxtraila Ltd         Buy         360         37           UBS Securities Auxtraila Ltd         Buy         360         37           UBS Securities Auxtraila Ltd         Buy         362         100           UBS Securities Auxtraila Ltd         Buy         362         100           UBS Securities Auxtraila Ltd         Buy         370         428           UBS Securities Auxtraila Ltd         Buy         370         420           UBS Securities Auxtraila Ltd         Buy         370         420           UBS Securities Auxtraila Ltd         Buy         370         420           UBS Securities Auxtraila Ltd         Buy         370         420 </td <td>w-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>250</td> <td>29</td> <td>Ordinary</td>	w-14	UBS Securities Australia Ltd	Buy	250	29	Ordinary
UBS Securities Auttraile Ltd         Buy         57777         6,598           UBS Securities Auttraile Ltd         Buy         5737         6,98         10           UBS Securities Auttraile Ltd         Buy         215,500         2,600         10           UBS Securities Auttraile Ltd         Buy         215,500         2,500         10           UBS Securities Auttraile Ltd         Buy         2,15,500         2,000         10           UBS Securities Auttraile Ltd         Buy         2,870         3,33         10           UBS Securities Auttraile Ltd         Buy         2,870         3,33         10           UBS Securities Auttraile Ltd         Buy         2,870         1,000         10           UBS Securities Auttraile Ltd         Buy         2,870         1,000         1,000         1,000           UBS Securities Auttraile Ltd         Buy         2,870         1,00	w-14	UBS Securities Australia Ltd	Buy	112	13	Ordinary
UBS Securities Australia Ltd         BNy         759         88           UBS Securities Australia Ltd         BNy         759         88           UBS Securities Australia Ltd         BNy         4.276         6.56           UBS Securities Australia Ltd         BNy         78         40           UBS Securities Australia Ltd         BNy         78         40           UBS Securities Australia Ltd         BNy         78         40           UBS Securities Australia Ltd         BNy         2.870         33.3           UBS Securities Australia Ltd         BNy         2.870         33.3           UBS Securities Australia Ltd         BNy         2.870         3.3           UBS Securities Australia Ltd         BNy         8.620         1.000           UBS Securities Australia Ltd         BNy         8.620         1.020           UBS Securities Australia Ltd         BNy         8.726 <td>w-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>57,737</td> <td>869'9</td> <td>Ordinary</td>	w-14	UBS Securities Australia Ltd	Buy	57,737	869'9	Ordinary
UBS Securities Australia Ltd         Buy         4,276         88           UBS Securities Australia Ltd         Buy         21,550         25,000           UBS Securities Australia Ltd         Buy         21,550         25,000           UBS Securities Australia Ltd         Buy         21,550         27,000           UBS Securities Australia Ltd         Buy         345         40           UBS Securities Australia Ltd         Buy         370         37           UBS Securities Australia Ltd         Buy         370         36           UBS Securities Australia Ltd         Buy         8,20         1,000           UBS Securities Australia Ltd         Buy         7,89         6,02           UBS Securities Australia Ltd         Buy         3,707         4,30           UBS Securities Australia Ltd         Buy         3,707         4,90           UBS Securities Australia Ltd         Buy         3,707         4,90           UBS Securities Australia Ltd         Buy         <	4-14	UBS Securities Australia Ltd	Buy	948	110	Ordinary
UBS Securities Australia Ltd         Buy         215,500         26,500           UBS Securities Australia Ltd         Buy         215,500         2,078           UBS Securities Australia Ltd         Buy         78         40           UBS Securities Australia Ltd         Buy         78         40           UBS Securities Australia Ltd         Buy         2,870         333           UBS Securities Australia Ltd         Buy         2,870         345           UBS Securities Australia Ltd         Buy         2,870         333           UBS Securities Australia Ltd         Buy         2,870         345           UBS Securities Australia Ltd         Buy         2,870         1,000           UBS Securities Australia Ltd         Buy         8,620         1,000           UBS Securities Australia Ltd         Buy         8,620         1,000           UBS Securities Australia Ltd         Buy         5,189         6,022           UBS Securities Australia Ltd         Buy         3,100         4,30           UBS Securities Australia Ltd         Buy         3,189         6,02           UBS Securities Australia Ltd         Buy         5,189         6,02           UBS Securities Australia Ltd         Buy	1 2	UBS Securities Australia Ltd	Buy	759	88	Ordinary
USS Securities Australia Ltd         Buy         \$1.75,000         \$25,000           USS Securities Australia Ltd         Buy         \$1.12         \$1.3           USS Securities Australia Ltd         Buy         \$1.2         \$1.3           USS Securities Australia Ltd         Buy         \$1.2         \$1.2           USS Securities Australia Ltd         Buy         \$1.2         \$1.2           USS Securities Australia Ltd         Buy         \$1.5         \$1.0           USS Securities Australia Ltd         Buy         \$1.5         \$1.0           USS Securities Australia Ltd         Buy         \$1.5         \$1.0           USS Securities Australia Ltd         Buy         \$1.5         \$1.6           USS Securities Australia Ltd         Buy         \$1.2         \$1.0           USS Securities Australia Ltd         Buy         \$1.2         \$1.2           USS Securities Australia Ltd         Buy         \$1.2         \$1.2           USS Securities Australia Ltd         Buy         \$1.2         \$1.2           USS Securities Australia Ltd         Buy         \$1.7         \$1.2           USS Securities Australia Ltd         Buy         \$2.18         \$6.2           USS Securities Australia Ltd         Buy	W-14	UBS Securities Australia Ltd	Buy	4,276	496	Ordinary
USS Securities Australia Ltd         Buy         78         9           USS Securities Australia Ltd         Buy         345         9           USS Securities Australia Ltd         Buy         346         9           USS Securities Australia Ltd         Buy         378         9           USS Securities Australia Ltd         Buy         370         373           USS Securities Australia Ltd         Buy         155         18           USS Securities Australia Ltd         Buy         8,200         1,000           USS Securities Australia Ltd         Buy         8,200         1,000           USS Securities Australia Ltd         Buy         8,200         1,000           USS Securities Australia Ltd         Buy         7,88         6,02           USS Securities Australia Ltd         Buy         7,88         6,02           USS Securities Australia Ltd         Buy         3,107         4,99           USS Securities Australia Ltd         Buy         5,173	21.0	I IRS Societies Australia 144	Bitt	215,500	25,000	Ordinary
UBS Securities Australia Ltd         Buy         78         9           UBS Securities Australia Ltd         Buy         245         40           UBS Securities Australia Ltd         Buy         2870         333           UBS Securities Australia Ltd         Buy         2870         136           UBS Securities Australia Ltd         Buy         155         18           UBS Securities Australia Ltd         Buy         8620         1,000           UBS Securities Australia Ltd         Buy         8,189         6,22           UBS Securities Australia Ltd         Buy         3,189         6,02           UBS Securities Australia Ltd         Buy         3,786         4,26           UBS Securities Australia Ltd         Buy         5,189         6,02           UBS Securities Australia Ltd         Buy         5,189         6,02           UBS Securities Australia Ltd         Buy         5,189         6,02           UBS Securities Australia Ltd         Buy         5,189 <td>v-14</td> <td>UBS Securities Australia 14d</td> <td>66</td> <td>96,010</td> <td>8/6/6</td> <td>Ordinary</td>	v-14	UBS Securities Australia 14d	66	96,010	8/6/6	Ordinary
UBS Securities Australia Ltd         Bby         345         40           UBS Securities Australia Ltd         Bby         333         181         21           UBS Securities Australia Ltd         Bby         155         155         181           UBS Securities Australia Ltd         Bby         155         1,000           UBS Securities Australia Ltd         Bby         1,448         168           UBS Securities Australia Ltd         Bby         293         1,000           UBS Securities Australia Ltd         Bby         341         4           UBS Securities Australia Ltd         Bby         3,100         4           UBS Securities Australia Ltd         Bby         3,100         4           UBS Securities Australia Ltd         Bby         3,100         4           UBS Securities Australia Ltd         Bby         3,700         4,30           UBS Securities Australia Ltd         Bby         3,700         4,30           UBS Securities Australia Ltd         Bby         5,189         6,52           UBS Securities Australia Ltd         Bby         5,189         6,52           UBS Securities Australia Ltd         Bby         5,189         6,92           UBS Securities Australia Ltd <t< td=""><td>v-14</td><td>UBS Securities Australia Ltd</td><td>800</td><td>78</td><td>2 0</td><td>Ordinary</td></t<>	v-14	UBS Securities Australia Ltd	800	78	2 0	Ordinary
UBS Securities Australia Ltd         Buy         2,870         333           UBS Securities Australia Ltd         Buy         161         27           UBS Securities Australia Ltd         Buy         155         18           UBS Securities Australia Ltd         Buy         8,200         1,000           UBS Securities Australia Ltd         Buy         1,448         1,600           UBS Securities Australia Ltd         Buy         7,849         602           UBS Securities Australia Ltd         Buy         7,889         602           UBS Securities Australia Ltd         Buy         3,707         4,99           UBS Securities Australia Ltd         Buy         3,707         4,90           UBS Securities Australia Ltd         Buy         3,707         4,90           UBS Securities Australia Ltd         Buy         5,189         6,02           UBS Securities Australia Ltd         Buy         6,78         1,93           UBS Securities Australia Ltd         Buy         6	v-14	UBS Securities Australia Ltd	Buy	345	40	Ordinary
UBS Securities Australia Ltd         Buy         181         21           UBS Securities Australia Ltd         Buy         155         168           UBS Securities Australia Ltd         Buy         8620         1,000           UBS Securities Australia Ltd         Buy         662         1,000           UBS Securities Australia Ltd         Buy         5,89         602           UBS Securities Australia Ltd         Buy         5,189         602           UBS Securities Australia Ltd         Buy         3,49         6,2           UBS Securities Australia Ltd         Buy         3,70         4,30           UBS Securities Australia Ltd         Buy         3,70         4,50           UBS Securities Australia Ltd         Buy         5,189         602           UBS Securities Australia Ltd         Buy         1,238	v-14	UBS Securities Australia Ltd	Buy	2,870	333	Ordinary
UBS Securities Australia Ltd   Buy   155   158	v-14	UBS Securities Australia Ltd	Buy	181	21	Ordinary
UBS Securities Australia Ltd         BM         155         18           UBS Securities Australia Ltd         BMy         6,220         1,000           UBS Securities Australia Ltd         BMy         6,220         1,000           UBS Securities Australia Ltd         BMy         7,889         6,02           UBS Securities Australia Ltd         BMy         7,889         6,02           UBS Securities Australia Ltd         BMy         3,707         4,39           UBS Securities Australia Ltd         BMy         5,18         6,02           UBS Securities Australia Ltd         BMy         5,18         6,02           UBS Securities Australia Ltd         BMy         6,70         1,32           UBS Securities Australia Ltd         BMy         6,97         1,67           UBS Securities Australia Ltd         BMy         6,97         1,67           UBS Securities Australia Ltd         BMy <th< td=""><td>w-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>310</td><td>36</td><td>Ordinary</td></th<>	w-14	UBS Securities Australia Ltd	Buy	310	36	Ordinary
UBS Securities Australia Ltd         Buy         1,000           UBS Securities Australia Ltd         Buy         662         1,000           UBS Securities Australia Ltd         Buy         662         1,00           UBS Securities Australia Ltd         Buy         5,189         602           UBS Securities Australia Ltd         Buy         3,189         6,02           UBS Securities Australia Ltd         Buy         3,707         4,30           UBS Securities Australia Ltd         Buy         3,707         4,30           UBS Securities Australia Ltd         Buy         5,189         6,02           UBS Securities Australia Ltd         Buy         6,076         1,538           UBS Securities Australia Ltd         Buy         6,074         8,09           UBS Securities Australia Ltd         Buy         6,074	v-14	UBS Securities Australia Ltd	Buy	155	18	Ordinary
UBS Securities Australia Ltd         Bay         1,448         168           UBS Securities Australia Ltd         Bay         662         100           UBS Securities Australia Ltd         Bay         5,789         602           UBS Securities Australia Ltd         Bay         3,189         602           UBS Securities Australia Ltd         Bay         1,12         4           UBS Securities Australia Ltd         Bay         3,707         430           UBS Securities Australia Ltd         Bay         5,189         602           UBS Securities Australia Ltd         Bay         5,189         662           UBS Securities Australia Ltd         Bay         6,974         809           UBS Securities Australia Ltd         Bay         6,974         809           UBS Securities Australia Ltd         Bay         6,974         809           UBS Securities Australia Ltd         Bay         6,974         1,7           UBS Securities Australia Ltd         Bay         6,974	v-14	UBS Securities Australia Ltd	Buy	8,620	1,000	Ordinary
UBS Securities Australia Ltd         Bby         58.2         100           UBS Securities Australia Ltd         Bby         5.189         60.2           UBS Securities Australia Ltd         Bby         5.189         60.2           UBS Securities Australia Ltd         Bby         3.4         4           UBS Securities Australia Ltd         Bby         3.707         4.30           UBS Securities Australia Ltd         Bby         4.726         4.90           UBS Securities Australia Ltd         Bby         5.189         60.2           UBS Securities Australia Ltd         Bby         1,231         1.938           UBS Securities Australia Ltd         Bby         1,6706         1,938           UBS Securities Australia Ltd         Bby         16,706         1,938           UBS Securities Australia Ltd         Bby         1,6706         1,938           UBS Securities Australia Ltd         Bby         1,6706         1,6706           UBS Securities Australia Ltd         Bby	v-14	UBS Securities Australia Ltd	Buy	1,448	168	Ordinary
UBS Securities Autralia Ltd         Buy         23-3         3-4           USS Securities Autralia Ltd         Buy         51-89         60-2           USS Securities Autralia Ltd         Buy         34         4           USS Securities Autralia Ltd         Buy         172         13           USS Securities Autralia Ltd         Buy         3707         4:30           USS Securities Autralia Ltd         Buy         5,189         60-2           USS Securities Autralia Ltd         Buy         1,221         1,338           USS Securities Autralia Ltd         Buy         1,221         1,538           USS Securities Autralia Ltd         Buy         1,670-6         1,538           USS Securities Autralia Ltd         Buy         1,670-6         1,538           USS Securities Autralia Ltd         Buy         6,974         80-9           USS Securities Autralia Ltd         Buy         6,974         80-9           USS Securities Autralia Ltd         Buy         5,97	v-14	UBS Securities Australia Ltd	Buy	862	100	Ordinary
UBS Securities Australia Ltd         BMy         5,18.9         60.2           UBS Securities Australia Ltd         BMy         3.4         4           UBS Securities Australia Ltd         BMy         11.2         1.3           UBS Securities Australia Ltd         BMy         3.70.7         4.90           UBS Securities Australia Ltd         BMy         3.70.7         4.90           UBS Securities Australia Ltd         BMy         5.189         6.02           UBS Securities Australia Ltd         BMy         5.189         6.02           UBS Securities Australia Ltd         BMy         7.0         1.30           UBS Securities Australia Ltd         BMy         6.974         8.69           UBS Securities Australia Ltd         BMy         6.974         8.9           UBS Securities Australia Ltd         BMy <t< td=""><td>v-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>293</td><td>34</td><td>Ordinary</td></t<>	v-14	UBS Securities Australia Ltd	Buy	293	34	Ordinary
UBS Securities Australia Ltd         Bluy         34         4           UBS Securities Australia Ltd         Buy         78         9           UBS Securities Australia Ltd         Buy         3,707         430           UBS Securities Australia Ltd         Buy         4,276         430           UBS Securities Australia Ltd         Buy         5,189         602           UBS Securities Australia Ltd         Buy         1,271         130           UBS Securities Australia Ltd         Buy         6,974         809           UBS Securities Australia Ltd         Buy         6,974         809           UBS Securities Australia Ltd         Buy         6,974         809           UBS Securities Australia Ltd         Buy         6,974         87           UBS Securities Australia Ltd         Buy         6,974         87           UBS Securities Australia Ltd         Buy         5,27         64           UBS Securities Australia Ltd         Buy         5,52         <	v-14	UBS Securities Australia Ltd	Buy	5,189	602	Ordinary
UBS Securities Australia Ltd         BMy         78         9           UBS Securities Australia Ltd         BMy         1172         13           UBS Securities Australia Ltd         BMy         3,707         4,30           UBS Securities Australia Ltd         BMy         5,189         6,52           UBS Securities Australia Ltd         BMy         5,189         6,02           UBS Securities Australia Ltd         BMy         5,78         6,68           UBS Securities Australia Ltd         BMy         1,121         1,53           UBS Securities Australia Ltd         BMy         1,670         1,538           UBS Securities Australia Ltd         BMy         6,974         809           UBS Securities Australia Ltd         BMy         6,974         809           UBS Securities Australia Ltd         BMy         6,974         809           UBS Securities Australia Ltd         BMy         5,96         1,78           UBS Securities Australia Ltd         BMy         5,97         6,41           UBS Securities Australia Ltd         BMy         5,96         6,41           UBS Securities Australia Ltd         BMy         5,96         6,41           UBS Securities Australia Ltd         BMy         5,96 <td>v-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>34</td> <td>4</td> <td>Ordinary</td>	v-14	UBS Securities Australia Ltd	Buy	34	4	Ordinary
UBS Securities Australia Ltd   Buy   3,707   439   13   14   14   15   15   15   15   15   15	4 4	UBS Securities Australia Ltd	Buy	78	6	Ordinary
UBS Securities Australia Ltd         Buy         4,276         430           UBS Securities Australia Ltd         Buy         5,189         602           UBS Securities Australia Ltd         Buy         5,189         602           UBS Securities Australia Ltd         Buy         5,27         65           UBS Securities Australia Ltd         Buy         1,271         1,53           UBS Securities Australia Ltd         Buy         6,970         1,938           UBS Securities Australia Ltd         Buy         6,974         889           UBS Securities Australia Ltd         Buy         6,974         899           UBS Securities Australia Ltd         Buy         6,974         899           UBS Securities Australia Ltd         Buy         5,57         64           UBS Securities Australia Ltd         Buy         5,57         64           UBS Securities Australia Ltd         Buy         5,52         64           UBS Securities Australia Ltd         Buy         5,52         64           UBS Securities Australia Ltd         Buy         7,653         64           UBS Securities Australia Ltd         Buy         7,653         64           UBS Securities Australia Ltd         Buy         7,563	4 2	UBS Securities Australia Ltd	Buy	112	13	Ordinary
USS Securities Australia Ltd   Buy   1,217   47,90     USS Securities Australia Ltd   Buy   1,217   1,20     USS Securities Australia Ltd   Buy   1,217   1,20     USS Securities Australia Ltd   Buy   1,217   1,20     USS Securities Australia Ltd   Buy   1,27   1,238     USS Securities Australia Ltd   Buy   1,27   1,238     USS Securities Australia Ltd   Buy   1,27   1,238     USS Securities Australia Ltd   Buy   1,27   1,27     USS Securities Australia Ltd   1,27   1,27     USS Securities Australia Ltd   1,27   1,27     USS Securities Australia Ltd   1,27   1,27     USS Securities Australia Ltd   1,27   1,27     USS Securities Australia Ltd   1,27   1,27     USS Securities Australia Ltd   1,27   1,27     USS Securities Australia Ltd   1,27   1,27     USS Securities Australia Ltd   1,27   1,27     USS Securities Australia Ltd   1,27   1,27     USS Securities Australia Ltd   1,27   1,27     USS Securities Australia Ltd   1,27   1,27     USS Securities Australia Ltd   1,27   1,27     USS Securities Australia Ltd   1,27     USS Securities Australia Ltd   1,27     USS Securities Australia Ltd   1,27     USS Securities Australia Ltd   1,27     USS Securities Australia Ltd   1,27     USS Securities Australia Ltd   1,27     USS Securities Australia Ltd   1,27     USS Securities Australia Ltd   1,27     USS Securities Australia Ltd   1,27     USS Securities Australia Ltd   1,27     USS Securities Australia Ltd   1,27     USS Securities Australia Ltd   1,27     USS Securities Australia Ltd   1,27     USS Securities Australia Ltd   1,27     USS Securit	77	UBS Securities Australia Ltd	suy.	3,707	430	Ordinary
USE Securities Australia Ltd         BM         1,189         DAZ           1 (18) Securities Australia Ltd         BMy         1,171         1,30           1 (18) Securities Australia Ltd         BMy         1,670         1,98           1 (18) Securities Australia Ltd         BMy         1,670         1,58           1 (18) Securities Australia Ltd         BMy         1,670         1,69           1 (18) Securities Australia Ltd         BMy         6,974         809           1 (18) Securities Australia Ltd         BMy         6,974         809           1 (18) Securities Australia Ltd         BMy         5,2         64           1 (18) Securities Australia Ltd         BMy         5,2         64           1 (18) Securities Australia Ltd         BMy         7,663         86           1 (18) Securities Australia Ltd         BMy         7,663         86           1 (18) Securities Australia Ltd         BMy         7,663         86           1 (18) Securities Australia Ltd         BMy         5,763         86	1/1	1985 Securities Australia Ltd	śng	4,276	436	Ordinary
UBS Securities Australia Ltd         BM         1,721         1,73           UBS Securities Australia Ltd         BM         16,706         1,938           UBS Securities Australia Ltd         BM         4,870         565           UBS Securities Australia Ltd         BM         5,974         809           UBS Securities Australia Ltd         BM         5,974         809           UBS Securities Australia Ltd         BM         26         64           UBS Securities Australia Ltd         BM         552         64           UBS Securities Australia Ltd         BM         7,663         68	14	UBS Securities Australia Ltd	Buy.	5,189	709	Ordinary
UBS Securities Australia Ltd         697         16,705         1,538           UBS Securities Australia Ltd         8My         4870         565           UBS Securities Australia Ltd         80y         586         68           UBS Securities Australia Ltd         8My         6,974         60           UBS Securities Australia Ltd         8My         56         7           UBS Securities Australia Ltd         8My         55         64           UBS Securities Australia Ltd         8My         552         64           UBS Securities Australia Ltd         8My         52         64           UBS Securities Australia Ltd         8My         52         64	-14	UBS Securities Australia Ind	(i) B	121,1	06	Ordinary
UBS Securities Australia Ltd         Bay         4,870         565           UBS Securities Australia Ltd         Bay         586         68           UBS Securities Australia Ltd         Bay         6,974         809           UBS Securities Australia Ltd         Bay         25         3           UBS Securities Australia Ltd         Bay         52         64           UBS Securities Australia Ltd         Bay         552         64	1-14	UBS Securities Australia Ltd	Buy	16 706	1 038	Ordinary
UBS. Securities Australia Ltd         Buy         68         68           UBS. Securities Australia Ltd         Buy         6,974         809           UBS. Securities Australia Ltd         Buy         26         17           UBS. Securities Australia Ltd         Buy         52         64           UBS. Securities Australia Ltd         Buy         7,663         86           UBS. Securities Australia Ltd         Buy         52         64           UBS. Securities Australia Ltd         Buy         52         64	-14	UBS Securities Australia Ltd	Buy	4,870	565	Ordinary
UBS Securite Australia Ltd         Bay         6,974         879           UBS Securite Australia Ltd         Bay         25         3           UBS Securities Australia Ltd         Buy         55         64           UBS Securities Australia Ltd         Buy         552         64           UBS Securities Australia Ltd         Buy         7,663         89           UBS Securities Australia Ltd         Buy         52         64	-14	UBS Securities Australia Ltd	Buy	286	89	Ordinary
UBS Securities Australia Ltd         Buy         26         3           UBS Securities Australia Ltd         Buy         147         17           UBS Securities Australia Ltd         Buy         552         64           UBS Securities Australia Ltd         Buy         7,663         899           UBS Securities Australia Ltd         Buy         52         6	-14	UBS Securities Australia Ltd	Buy	6,974	608	Ordinary
UBS Securities Australia Ltd         Buy         147         17           UBS Securities Australia Ltd         Buy         552         64           UBS Securities Australia Ltd         Buy         7,663         869           UBS Securities Australia Ltd         Buy         52         6	-14	UBS Securities Australia Ltd	Buy	52	m	Ordinary
UBS Securities Australia Ltd         Bby         552         64           LUBS Securities Australia Ltd         Bby         7,663         689           LBS Securities Australia Ltd         Buy         52         6	-14	UBS Securities Australia Ltd	Buy	147	17	Ordinary
UBS Securities Australia Ltd         Buy         7,663         889           UBS Securities Australia Ltd         Buy         52         6	-14	UBS Securities Australia Ltd	Buy	552	64	Ordinary
UBS Securities Australia Ltd Buy 52 6	-14	UBS Securities Australia Ltd	Buy	7 663	000	
				200,	1 488	Ciginary

7	The second secon	(2)	SCS. / -	7,061	Ordinary
11-Nov-14	UBS Securities Australia Ltd	Buy	78	σ	Ordinary
Jov-14	UBS Securities Australia Ltd	Buy	23,249	2,694	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	216	25	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	2,252	261	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	751	87	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	52	9	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	4,315	200	Ordinary
Vov-14	UBS Securities Australia Ltd	Buy	10,960	1,270	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	28,893	3,348	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	4,315	200	Ordinary
VOV-14	UBS Securities Australia Ltd	Buy	4,280	496	Ordinary
Vov-14	UBS Securities Australia Ltd	Buy	155	18	Ordinary
Vov-14	UBS Securities Australia Ltd	Buy	1,579	183	Ordinary
Vov-14	UBS Securities Australia Ltd	Buy	4,315	200	Ordinary
11-Nov-14	UBS Securities Australia Ltd	Buy	112	13	Ordinary
Vov-14	UBS Securities Australia Ltd	Buy	725	88	Ordinary
Vov-14	UBS Securities Australia Ltd	Buy	1,182	137	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	10,960	1.270	Ordinary
1-Nov-14	UBS Securities Australia Ltd	Buy	759	88	Ordinary
1-Nov-14	UBS Securities Australia Ltd	8uy	293	34	Ordinary
1-Nov-14	UBS Securities Australia Ltd	Buy	5.125	265	Ordinary
1-Nov-14	UBS Securities Australia Ltd	Buy	604	02	Ordinary
1-Nov-14	UBS Securities Australia Ltd	Buy	44 160	5117	Craining
Nov-14	UBS Securities Australia Ltd	Buv	282	23	Crdinary
11-Nov-14	UBS Securities Australia Ltd	Buv	367	3 6	in in it
1-Nov-14	UBS Securities Australia Itd	26	1703	74.00	Ciginaly
1-Nov-14	IIRS Securities Australia I to	811	25.1	000	Ordinary
1-Nov-14	LIRC Securities Australia 14d	8.00	042'/1	2,000	Orginary
1-Nov-14	PIDC Countries Australia 14d	600	057	67	Ordinary
1.Nov.14	FIRS Securities Australia 144	65	591,0	660	Ordinary
1 2	The carrier Account to	on,	066,12	7,500	Ordinary
1	Obsisecutives Australia Lib	on)	1,233	143	Ordinary
-NOV-14	Obsideration Australia Ltd	gnà	21,455	2,489	Ordinary
0v-14	UBS securities Australia Ltd	Buy	62,754	7,280	Ordinary
lov-14	UBS Securities Australia Ltd	Buy	741	98	Ordinary
lov-14	UBS Securities Australia Ltd	Buy	23,214	2,693	Ordinary
lov-14	UBS Securities Australia Ltd	Buy	5,146	265	Ordinary
lov-14	UBS Securities Australia Ltd	Buy	276	32	Ordinary
Jov-14	UBS Securities Australia Ltd	Buy	35.825	4 156	Ordinary
tov-14	UBS Securities Australia Ltd	Buy	1 974	229	Ordinary
lov-14	UBS Securities Australia Ltd	Buy	629 2	885	Ordinary
lov-14	UBS Securities Australia Ltd	Buy	33 721	3 912	Colinary
Jov-14	UBS Securities Australia Ltd	Buv	5 310	617	Ordinary.
lov-14	UBS Securities Australia Ltd	Buv	11 200	1 300	Commen
Jov-14	UBS Securities Australia Itd	Bliv	1 4	505,1	Ordinary
17	FIRS Securities Australia 14d	S. S.	550 .		Ordinary
1	I Do Countries Australia Ind	ing	1/8/1	/17	Ordinary
- NON- III	Obs securities Australia Ltd	any	431	20	Ordinary
4	UBS Securities Australia Ltd	Buy	1,914	222	Ordinary
94-14	UBS Securities Australia Ltd	Buy	8,473	686	Ordinary
Ov-14	UBS Securities Australia Ltd	Buy	3,155	366	Ordinary
lov-14	UBS Securities Australia Ltd	Buy	4,284	497	Ordinary
lov-14	UBS Securities Australia Ltd	Buy	207	24	Ordinary
lov-14	UBS Securities Australia Ltd	Buy	23,705	2.750	Ordinary
iov-14	UBS Securities Australia Ltd	Buy	871	101	Ordinary
lov-14	UBS Securities Australia Ltd	Buy	3 931	45k	Ordinan
lov-14	UBS Securities Australia Ltd	Buv	250	ar	Ordinan
lov-14	UBS Securities Australia Ltd	Buv	715	6	Crains
Jov-14	LIBS Securities Australia 14d	200	303 07	300	Cidillary
	COO Occurred Australia Ltd	(no	42,635	4,946	Ordinary
1	ODS Securities Australia Ltd	, ing	2,155	250	Ordinary
JOV-14	UBS Securities Australia Ltd	Buy	612	71	Ordinary
lov-14	UBS Securities Australia Ltd	Buy	78	Ø.	Ordinary
lov-14	UBS Securities Australia Ltd	Buy	3,327	386	Ordinary
11-Nov-14	UBS Securities Australia Ltd	Buy	276	32	Ordinary
1-Nov-14	UBS Securities Australia Ltd	Buy	604	70	Ordinary
1-Nov-14	UBS Securities Australia Ltd	Buy	4.263	494	Ordinary
1-Nov-14	UBS Securities Australia Ltd	Buy	3.338	387	Ordinary
lov-14	LIBS Securities Australia Itd	Bliv	1 284	O S	Ordinary
1.Nov.1.4	LIBS Societies Australia 144	Res	100,	200	Cidinaly
1-Nov-14	LIDE Committee Australia Hd	Buy	457,1 AGD A	UC. 1	Chambry
1-Nov-14	IURS Securities Australia Ltd	Buy	4,234	533	Ordinary
1.Nov.14	Line Consister Australia 14d	and born	cco'i	751	Ofcinary
- A-	THE PROPERTY OF THE PROPERTY O				
	on accuracy regulated and	puy	465	75	Ordinary
11-Nov-14	UBS Securities Australia Ltd	Buy	465	54	Ordinary

	2	200	41,386	4	
UBS Securities Australia Ltd			007.00		
UBS Securities Australia Ltd		Buy	67478		Ordinary
Securities Austra		Rin	2017	760	
ODS Securities Australia Ltd		buy.	(10,2 (10,2		l
securifies Austra		ha	33,590		
UBS Securities Australia Ltd		Buy	20,538		
securfies Austra		BUy	965		
UBS Securities Australia Ltd		Buy	9,430		
UBS Securities Australia Ltd		Buy	12,938		
UBS Securities Australia Ltd		Buy	15.767		
Securities Australia Ltd		Buv	483		
105 Securities Austral		Riv	1 225		Ordina.
I DC Countition Austral		Riny	203		
accounted Addition	N. 1	(Fo	/86	00	
JBS securities Austral	la Ltd	Buy	138		
UBS Securities Australia Ltd	la Ltd	Buy	3,159	366	Ordinary
UBS Securities Australia Ltd	ia Ltd	Buy	35	4	Ordinary
JBS Securities Australia Ltd		Buy	9.424	1.092	Ordinary
JBS Securities Australia Ltd		Buv	2 158		
DC Cocuration Australia		Bir	30.153		
Securities Austral		000	120,120	#6#'n	
Car Security		600	0/1/0		Colliary
OBS Securities Australia Ltd		ána	52,479	n'a	Ordinary
UBS Securities Australia Ltd		Buy	224	26	Ordinary
UBS Securities Australia Ltd		Buy	95	-	Ordinary
UBS Securities Australia Ltd		Buy	1,493	173	Ordinary
Securities Australia Ltd		Buy	173	20	Ordinary
IBS Securities Australia I to		BLW	212 963	3 45	Ordinary
Cocurition Auctor		Biss	030.01		
DDS Securities Australia Ltd		and,	002,11	7	Ordinary
securities Austral		ána.	4,280	490	Ordinary
securities Austral		Buy	015,1		Ordinary
Securities Austral		Buy	48,078		Ordinary
Securities Austral		Buy	9,545	1,106	Ordinary
Securities Austral		Buy	2,149		
Securities Austral		Buy	397	46	
Securities Austral		Buy	3.159		
Securities Austral		Biv	777		
Societion Austral		Bir	101		
Securities Austral		900	0		١
securities Austral		hna	104	71	Ordinary
Securities Austral		on's	69	0	Ordinary
Securities Austral		ons	915,52	/567	Ordinary
Securities Austral		euy	2,045		Ordinary
UBS Securities Australia Ltd		Ruy	173	20	Ordinary
Securities Austral		Buy	397,377		Ordinary
UBS Securities Australia Ltd		Buy	491		Ordinary
JBS Securities Australia Ltd		Buy	268		Ordinary
UBS Securities Australia Ltd		Buy	11,521	1,335	Ordinary
JBS Securities Australia Ltd		Buy	21,877	2,535	Ordinary
JBS Securities Australia Ltd		Buy	3,417	386	Ordinary
IRS Securities Australia Ltd		Buv	2 158		Ordinary
by I offerburk softium 2.3 201		D. 0.	2,130		Countries
secondes Ausural		500	90402		Ordinary
OBS Securities Australia Ltd		ony.	14,6/1	00/1	Ordinary
UBS Securities Australia Ltd		gnà	11,/3/		Ordinary
JBS Securities Australia Ltd		Buy	535		Ordinary
UBS Securities Australia Ltd		Buy	12,712	1,473	Ordinary
JBS Securities Australia Ltd		Buy	7,448	863	Ordinary
UBS Securities Australia Ltd		Buy	785	Б	Ordinary
Securities Australi		Buv	100		Ordinary
Let Cocuration A section 2011		BIN	110		(1)
Securities Australia		(2)	711		Ordinary
UBS securities Australia Ltd		euy	1,994		Ordinary
UBS Securities Australia Ltd		Buy	16,250	1,883	Ordinary
UBS Securities Australia Ltd		Buy	35	4	Ordinary
<b>UBS Securities Australia Ltd</b>		Buy	104	12	Ordinary
JBS Securities Australia Ltd		Buy	518	9	Ordinary
JBS Securities Australia Ltd		Buv	159		Ordinary
IRS Securities Australia Ltd		Buy	328		Ordinary
IBS Securities Australia 11d		Rin	202	69	Ordinan
In Comment American		(2)	000		Counary
Securities Australia Ltd		, och	35	9	Urdinary
Securities Australia Ltd		Buy	78	o.	Ordinary
ecurities Australi		Buy	27,210	3,153	Ordinary
Securities Australi		Buy	190	22	Ordinary
bell citation Assistance 2001		, and	150	77	Coding
Securines Australia		600	12)	1	Cionaly
Securities Australi		suy	138	ç	
Securities Australi			-	1 ~ 1	Ordinary
		Ang	31.612	3.663	Ordinary

Nov-14	ILIRS Securities Australia Ltd	25.0			Crainary
1-Nov-14	UBS Securities Australia Ltd	Buy	3 155	366	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	69		Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	1,241	144	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	4.991		Ordinary
1-Nov-14	UBS Securities Australia Ltd	Buy	319		Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	1,698		Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	810		Ordinary
-Nov-14	UBS Securities Australia Ltd	Buy	15,326		Ordinary
-Nov-14	UBS Securities Australia Ltd	Buy	14,559	1,689	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	15,326		Ordinary
Nov-14	UBS Securities Australia Ltd	Виу	759	88	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	767	88	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	26		Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	8,827	1,024	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	7,275		Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	2,983		Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	3,969		Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	1826		Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	1,326	154	Ordinary
11-Nov-14	UBS Securities Australia Ltd	Виу	1,998		Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	164		Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	5,080		Ordinary
11-Nov-14	UBS Securities Australia Ltd	Buy	8,610		Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	58 841		Ordinan
Nov-14	UBS Securities Australia Ltd	Buy	152 251		Ordinary
-Nov-14	UBS Securities Australia Ltd	Buy	2.419		Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	2 919		Ordinary
11-Nov-14	UBS Securities Australia Ltd	Buy	1 076	125	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	2 153		Codings
1-Nov-14	UBS Securities Australia Ltd	Buy	7 7 491		Ordinary
1-Nov-14	UBS Securities Australia Itd	Buv	113		Ordinary
1-Nov-14	UBS Securities Australia Ltd	Buv	861		Ordinary
Nov-14	UBS Securities Australia Itd	Buv	1 100	72.	Ordinary
1-Nov-14	LIBS Securities Australia 11d	Bur	OCT.		Cidinaly
Nov.14	HRS Securities Australia 11d	Sur.	505.5		Orginary
Nov.14	LIBS Societies Australia 14d	S. P.	5,56,5		Ordinary
Nov.14	IIIDS Societies Australia 144	Sec. of	445		Ordinary
Nov.14	I IRS Societies Australia Ind	Sec.	517	57	Ordinary
Nov. 14	I IDC Cocuration American 1+al	6	201,1	65	Ordinary
1	ODS Securities Australia Ltd	ing.	S		Ordinary
NOV-14	UBS Securities Australia Ltd	Buy	17	2	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	465	72	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	12,732	1,477	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	6	ı	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	17	7	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	362	42	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	2.566	298	Ordinary
11-Nov-14	UBS Securities Australia Ltd	Buy	146	17	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	146	17	Ordinan
Nov-14	UBS Securities Australia Itd	Buv	1		Ordinary
Mon-14	LIBS Securities Australia 144	(no	17	5	Ordinary
No. 14	Upo Securities Australia Ltd	Buy	301	35	Ordinary
1 Nov. 14	UBS Securities Australia Ltd	eug	8,448	086	Ordinary
NOV-14	Upo Securities Australia Ltd	ony o	4,681	543	Ordinary
1000	OBS SECURITIES AUSTRAING LIQ	en)	1,508	1/5	Ordinary
NOV-14	UBS Securities Australia Ltd	gnà	14,525	1,685	Ordinary
1-Nov-14	UBS Securities Australia Ltd	Buy	34	4	Ordinary
1-Nov-14	UBS Securities Australia Ltd	Buy	706	82	Ordinary
11-Nov-14	UBS Securities Australia Ltd	Buy	21,732	2,524	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	2,936	341	Ordinary
11-Nov-14	UBS Securities Australia Ltd	Buy	12,485	1.450	Ordinary
1-Nov-14	UBS Securities Australia Ltd	Buv	8 507	880	Ordinan
1-Nov-14	LIBS Securities Australia 11d	Bry	13.064	000	in in in
1-Nov-14	LIDS Countries Australia Ltd	600	196,61	079'1	Crainary
Now 14	LIDE Countries Australia Ltd	Pag.	USC, I	1,345	Ordinary
11/0/17	Tibe Committee Australia List	600	6	- :	Ordinary
1	Cop Securities Australia Ltu	, and	198	57	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	5,594	649	Ordinary
lov-14	UBS Securities Australia Ltd	Buy	1,302	151	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	267	31	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	8.620	1 000	Ordinary
1-Nov-14	UBS Securities Australia Ltd	Buy	25	9	Ordinary
Nov-14	LIBS Securities Australia Ltd	Bit	43		(ioning)
11 Nov. 17	CO DESCRIPTION CONTRACTOR	(20)	***		
Vov-14				2	Ordinary
	UBS Securities Australia Ltd	Buy	34	v 4	Ordinary

г		_			_	Ι-	т-	_	_	_	1	Т	<b>.</b>	Т		_	1	_	-1		т	_	_	_	_	1	Т	т-			_	-	r-1	-	_	-	_	_	_	1	_			_	_					_	,	_		_		_		_				<b>,</b>	_		_		_		_	_
Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Cidinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary
1,218	1,154	33	1,152	92	-	1 199	1058	1.146	100	008	5	244	R	6 8	1 467	(0 <del>4</del> )	10	- 6	7	1,703	3),	6,5	7/100	7,007	3,101	2002	1771.	22	100	463	366	118	6,382	76	136	350	300	366	366	-	м	4	253	130	948	17	7 0	744	4	250	248	1,707	262	1,804	28	000,5	5//	1,332	1 000	Δ.	594	1,299	92	602	20	2	265	265	250	338
10,505	9,947	284	0,930	260	6	10.347	9.131	628.6	863	6.904	483	2 103	218	700 9	12 660	72,800		9 (	1/1	14,697	14,6/1	1 101	1,484	214.77	3 159	3,650	35	190	43	3,996	3,159	1,018	22,077	959	1,174	3 150	31.09	3.159	3,159	6	26	35	2,183	1,122	181,8	235	787	6.421	35	2,158	2,140	14,731	2,261	15,569	242	43,150	980'9	11,495	77171	150	5.123	11,210	260	5,189	172	17	2,284	2,284	2,155	2.914
Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buv	Buv	Buv	Blav	Bir	Bir	Bissi	600	ony Britis	ána a	any Prince	P. D. D.	Bus,	RIM	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	any bus	Buy	Bitk	Buy	Buy	8uy	Buy	Buy	Buy	Buy	any Bury	Bury	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy.	and,	any Bus	Riny	Buy	Buy	Buy	Buy	Buy	8uy	Buy	Buy	Buy	Buy	Buy
s Ltd	b Ltd	a Ltd	ltd	a Ltd	ı Ltd	Ltd	pt1	Ltd	p rtq	) Itd	11d	Ltd	110	1	120	100	200		100	27	100	110	ורום	1149	P11	ltd	Ital	Ltd																										Ltd	. 14	D27	217	710		141	Ltd	Ltd	Ltd	Ltd						
UBS Securities Australia Ltd	UBS Securities Australia	UBS Securities Australia Lto	UBS Securities Australia	UBS Securities Australia	UBS Securities Australia	UBS Securities Australia	UBS Securities Australia	UBS Securities Australia Ltd	UBS Securities Australia	UBS Securities Australia	UBS Securities Australia	UBS Securities Australia	LIRS Securities Australia	LRS Securities Australia   td	I IBS Securities Australia	1100 Securities Australia Ltd	I IDC Countries Australia	Upo Committee Australia Ltd	UBS Securities Australia	UBS Securities Australia Ltd	UBS Securities Australia	1005 Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	I IBS Securities Australia Ltd	UBS Securities Australia	URS Securities Australia Ltd	UBS Securities Australia	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia	UBS Securities Australia	UBS Securities Australia	UBS Securities Australia	LIRS Securities Australia	IIRS Securities Australia	UBS Securities Australia	UBS Securities Australia	UBS Securities Australia	UBS Securities Australia Ltd	UBS Securities Australia	UBS Securities Australia	UBS Securities Australia	UBS Securities Australia	UBS Securities Australia	UBS Securities Australia	UBS Securities Australia	UBS Securities Australia Ltd	<b>UBS Securities Australia</b>	<b>UBS Securities Australia</b>	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia	UBS Securities Australia Ltd	1185 Securities Australia Ltd	URS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia	UBS Securities Australia Ltd	UBS Securities Australia
11-Nov-14	11-Nov-14	11-Nov-14	11-Nov-14	11-Nov-14	11-Nov-14	11-Nov-14	11-Nov-14	11-Nov-14	11-Nov-14	11-Nov-14	11-Nov-14	11-Nov-14	11-Nov-14	11-Nov-14	11-Nov-14	11-Now-14	11-Nov-14	11 Nov. 14	1-NOV-14	1 - NOV 14	11-NOV-14	11 1001 14	11 No. 14	11-Nov-14	11-Nov-14	11-Nov-14	11-Nov-14	11-Nov-14	11-Nov-14	11-Nov-14				-	1				ı		11-Nov-14		ı	-	١	11-Nov-14	11-Nov-14	11-Nov-14	11-Nov-14	11-Nov-14	11-Nov-14	11-Nov-14	11-Nov-14	11-Nov-14	11-Nov-14	11-NOV-14	1 Nov-14	11 NOV-14	11-Nov-14	11-Nov-14	11-Nov-14	11-Nov-14	11-Nov-14	11-Nov-14	11-Nov-14	11-Nov-14	11-Nov-14		11-Nov-14	

1	1 100 Care minister A. manufactural				
11-Nov-14	1985 Securities Australia Ind	Bus	719	5 1	Ordinary
Nov-14	UBS Securities Australia Itd	Biv	1190	130	Cramary
Jan-14	IIRS Securities Australia 14d	Bis.	063 1	138	Ordinary
14	HIRC Securities Australia Ital	D. C.	859,1	081	Ordinary
1 10	Coo Securities Adstraira Liu	an)	069	28	Ordinary
10.00	Oppositions Australia Ltd	Buy	/17/1	141	Ordinary
	Obs Securities Australia Ltd	bog.	609	56	Ordinary
10. 14 10. 14	Obs securities Australia Ltd	Buy	8/	6	Ordinary
1	Obs securities Australia Ltd	buy	379	44	Ordinary
404-14	UBS Securities Australia Ltd	Buy	483	56	Ordinary
10v-14	Obs securities Australia Ltd	nec	98	(10)	Ordinary
1-NOV-14	UBS Securities Australia Ltd	Zell	372,337	(43,295)	Ordinary
Nov-14	UBS Securities Australia Ltd	Sell	4,059	(472)	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	181	21	Ordinary
Nov-14	UBS Securties Australia Ltd	Buy	10,294	1,197	Ordinary
1-Nov-14	UBS Securities Australia Ltd	Buy	860	100	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	86.000	10,000	Ordinary
1-Nov-14	UBS Securities Australia Ltd	Buy	119 987	13 052	Ordinan
Nov-14	UBS Securities Australia Ltd	3	750 050	100°C	or and
11-Nov-14	LIBS Securities Australia 144	200	035.055	(201,12)	Ordinary
Nov-14	LIBS Securities Australia Itd	(a)	200,002	00,000	Granialy
Nov-14	LIBS Securities Australia ted	D. S.	75t/ / /	(3,004)	Ordinary
1-Now-14	188 Securities Australia 144	Riv	97	6 50	Ordinary
1-Nov-14	1185 Securities Australia 144	Cell	12,300	/05'1	Ordinary
No. 14	Public Countries Australia Ltd	150	15,764	(1,833)	Ordinary
NOV-14	UBS Securities Australia Ltd	éng	6	-	Ordinary
1000-14	UBS Securities Australia Ltd	Buy	30,195	3,511	Ordinary
NOV-14	UBS Securities Australia Ltd	Sell	70,219	(8,165)	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	290,972	33,834	Ordinary
1-Nov-14	UBS Securities Australia Ltd	Sell	98	(10)	Ordinary
Nov-14	UBS Securities Australia Ltd	Sell	98	(10)	Ordinary
-Nov-14	UBS Securities Australia Ltd	Buy	113 460	13 193	Ordinary
Nov-14	UBS Securities Australia Ltd	Buv	25,800	000	Ordinan
11-Nov-14	UBS Securities Australia Ltd	i s	25,000	(30,000)	Cedinary
Nov-14	IIRS Securities Australia IIA	3	25,000	(000,000)	Ordinary
low 14	In the Commerce American Commerce Comme	11-3	//	6	Ordinary
10.14	Obs securities Australia LIG	nac .	1/2,000	(20,000)	Ordinary
404-14	UBS Securities Australia Ltd	Sell	4,248	(494)	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	30,005	3,489	Ordinary
Nov-14	UBS Securities Australia Ltd	Sell	1,720,000	(200,000)	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	5,186	603	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	17,604	2,047	Ordinary
40v-14	UBS Securities Australia Ltd	Buy	360,529	41,922	Ordinary
Vov-14	UBS Securities Australia Ltd	Buy	37.187	4 2 2 1	Ordinary
11-Nov-14	UBS Securities Australia Ltd	Buy	14 757	1675	Ordinary
1-Nov-14	UBS Securities Australia Ltd	Buy	320	CP	Ordinary
1-Nov-14	UBS Securities Australia Ltd	Buv	50 728	5.758	Ordinan
Nov-14	UBS Securities Australia Itd	) Jes	238	(20)	ording.
Jow-14	LIBA Cocuration Australia 144	200	250	(77)	Ordinary
1-Now-14	TIDS Cocuration Augustical and	find	3,542	402	Ordinary
1	ODS Securities Australia Ltd	on)	806	011	Ordinary
1-NOV-14	UBS Securities Australia Ltd	auy	3,106	353	Ordinary
14 14	UBS Securities Australia Ltd	Buy	1,507	171	Ordinary
1-Nov-14	UBS Securities Australia Ltd	Buy	476	54	Ordinary
1-Nov-14		Buy	180,351	20,448	Ordinary
Vov-14	UBS Securities Australia Ltd	Buy	10,064	1.14	Ordinary
1-Nov-14		Buy	40 546	7 597	Ordinan
11-Nov-14	UBS Securities Australia Ltd	Buv	886	113	Ordinan
Jov-14	LIBS Securities Australia 11d	Birv	2 233	350	County of the last
low-14	Let either Australia	Busy	100 101	COC	Cicinary
100.11	1905 Contracts Accepted and	, and	131,205	14,859	Ordinary
1	Obs Securities Australia Liu	ponk	8/0′1	771	Ordinary
40V-14	UBS Securities Australia Ltd	Buy	1,660	188	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	751	82	Ordinary
Vov-14	UBS Securities Australia Ltd	Buy	3,757	425	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	4,761	538	Ordinary
Vov-14	UBS Securities Australia Ltd	Buy	115	13	Ordinary
Nov-14	UBS Securities Australia (1d	Buy	040	106	Ordinary
Jose 14	1985 Controller Australia 14d	) and	0000	3	Ordinary
1 2	Coo Securities Australia Liu	fna	307'1	36	Ordinary
41-70	UBS Securities Australia Ltd	Buy	914	103	Ordinary
Ov-14	UBS Securities Australia Ltd	Buy	1,073	121	Ordinary
04-14	UBS Securities Australia Ltd	Buy	35	4	Ordinary
Ov-14	UBS Securities Australia Ltd	Buy	1,118	126	Ordinary
lov-14	UBS Securities Australia Ltd	Buy	3,314	374	Ordinary
ov-14	UBS Securities Australia Ltd	Buy	1 933	218	
	ODD December Treatment was				200
•			226	017	Orginary
40.70	UBS Securities Australia Ltd	Buy	4.093	462	Ordinary

11-Mov-14   UBS Securities Australia Ltd   Buy   11-Mov-14   UBS Securities Australi	267 462 3.151 3.151 3.151 3.152 2.250 3.153 3.15	31 36 36 36 36 428 36 428 428 431 1513 1513 161 171 172 202 202 203 204 214 217 218 219 210 210 210 210 210 210 210 210	Odinay Odinay
185 Securities Australia Ltd 186 Securities Australia Ltd 187 Securities A	672 465 465 3 (151 3 (151 3 (151 3 (151 3 (151 3 (151 3 (152 3 (1	78 866 366 366 366 366 366 366 367 371 1.513 3.771 1.598 1.702 202 202 202 202 202 202 202 202 202	Odinay Odinay
UBS Securities Australia Ltd UBS Securities UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities UBS Securities Australia Ltd UBS Securities UBS Securities UBS Securities UBS Securities UBS Securities UBS Securities UBS Securities UBS Securities UBS Securities UBS Securities UBS Securities UBS Securities UBS Secur	3 465 3 151 3 151 3 151 3 14 3 14 3 165 3 160 2 73 30 4 20 2 3 60 1 1,598 1 1,598 1 1,598 1 1,598 1 1,007 4 4,007 1 1,007 1 1,007	366 366 366 36 478 478 478 478 478 560 560 560 1,001 1,513	Odinary Odinary
185 Securities Australia Ltd 186 Securities Australia Ltd 187 Securities Australia Ltd 188 Securities Australia Ltd 188 Securities Australia Ltd 188 Securities A	3.151 3.162 3.162 3.163 3.163 3.163 27.302 27.302 40.777 40.777 1.177 1.177 1.179 2.206 2.206 2.206 2.206	366 366 366 367 367 3771 1,513	Odinary Odinary
1.05 Securines Australia Ltd 1.05 Securines Australia Ltd 1.05 Securines Australia Ltd 1.05 Securines Australia Ltd 1.05 Securines Australia Ltd 1.05 Securines Australia Ltd 1.05 Securines Australia Ltd 1.05 Securines Australia Ltd 1.05 Securines Australia Ltd 1.05 Securines Australia Ltd 1.05 Securines Australia Ltd 1.05 Securines Australia Ltd 1.05 Securines Australia Ltd 1.05 Securines Australia Ltd 1.05 Securines Australia Ltd 1.05 Securines Australia Ltd 1.05 Securines Australia Ltd 1.05 Securines Australia Ltd 1.05 Securines Australia Ltd 1.06 Securines Australia Ltd 1.07 Securines Australia Ltd 1.08 Securines Australia Ltd 1.09 Securines Australia Ltd 1.00 Securines A	3.65 3.65 3.68 3.685 3.685 3.685 3.687 4.627 4.627 1.1558 1.1558 1.1598 4.622 2.388 2.388 2.388 2.388 2.288 2.288 2.288 2.288 2.288	366 478 478 478 4736 4736 4736 1513 1513 1513 1519 1519 171 172 202 202 202 203 203 204 216 216 217 217 217 217 217 217 217 217	Odinary Odinary
185 Securities Australia Ltd 186 Securities Australia Ltd 187 Securities A	3.151 3.14 3.14 3.168 3.168 2.7730 40,777 40,777 40,777 40,777 1,177 13,778 4,469 1,179 1,007 4,469 2,269 1,179 1,799 1,799 2,269 2,273 1,179 1,261	36 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	Odinary Odinary
185 Securities Australia Ltd 186 Securities Australia Ltd 187 Securities A	34 365 365 365 13,027 14,027 11,177 11,178 12,028 11,528 11,528 11,027 1,0	3 4 4 7 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	3.65 3.65 3.65 3.65 4.82 6.99 1.100 1.007	428 428 478 4776 4776 4776 4776 4776 4776 4776	Ocidinary Ocidinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Lucial Ltd UBS Securities Luci	7.302 7.302 13.027 40.777 4.322 4.322 4.322 1.1,538 1.1,538 1.007 4.469 2.388 2.388 2.286 2.269 2.273 1.1,539	3,478 1,573 4,736 1,568 1,568 1,568 1,568 1,568 1,568 1,568 1,568 1,568 1,568 1,568 1,667 1,	Octorary Octorary
U.BS. Securities Australia Ltd U.BS. Securities Australia Ltd	13.027 19.027 40.777 9.307 9.307 11.538 11.539 1.739 2.388 1.739 2.388 2.388 2.388 1.739 2.388 1.739 2.268 2.268	1,513 4,736 1,081 1,	Odinay Odinay
1.05 Securities Australia Ltd 1.06 Securities Australia Ltd 1.06 Securities Australia Ltd 1.07 Securities Australia Ltd 1.08 Securities Australia Ltd 1.08 Securities Australia Ltd 1.08 Securities Australia Ltd 1.08 Securities Australia Ltd 1.08 Securities Australia Ltd 1.08 Securities Australia Ltd 1.08 Securities Australia Ltd 1.08 Securities Australia Ltd 1.08 Securities Australia Ltd 1.08 Securities Australia Ltd 1.09 S	1,177 40,777 40,777 1,177 4,827 4,822 4,822 1,1558 1,1007 1,007	4,7513 4,7513 1,081 1,081 1,081 1,098	Odinay Odinay
185 Securities Australia Ltd 186 Securities Australia Ltd 187 Securities A	40,777 1,177 1,177 1,272 9,307 1,588 1,1588 1,1007 1,007 1,739 2,288 2,266 2,266 2,266 2,273 1,2,631 1	4,756 1,081 1,081 1,081 1,547 1,548 1,548 1,540 1,500 1,500 1,600	Odinary Odinary
1085 Securities Australia (1d) 1085 Securities (1d) 1086 Secur	9,177 9,177 9,177 11,528 11,528 1,1007 1,1007 4,145 1,739 2,368 2,368 2,066 2,	1,081 560 1,081 1,347 1,598 1,177 202 202 202 2,03 2,403 1,667 1,6	Odnary Odnary
UBS Securities Australia tot UBS Securities A	4,8307 4,8302 999 11,538 13,775 1,007 4,469 1,739 2,388 2,066 2,066 2,069 2,073 1,2,631 1,2,63	100 100 110 110 110 110 110 110	Ociniary Ociniary
UBS Securities Australia Ltd UBS Securities Australia Ltd	4 822 999 11,599 11,599 1,375 1,375 1,739	1560 1.1347 1.1347 1.158 1.158 1.158 2.15 2.02 3.305 1.063 1.467 2.403 1.467 1	Ordinary Ordinary
UBS Securines Australia Ltd UBS Securines Australia Ltd	11,558 11,558 13,775 1,007 1,007 1,007 1,739 2,8,456 2,8,456 2,006 2,006 2,006 2,006 2,006 2,006 1,007	116 1.588 1.588 1.17 2.05 2.05 2.005 1.053 1.053 1.053 1.467	Ordinary Ordinary
UBS Securities Australia (rd UBS Securities A	11,588 13,775 10,775 1,075 1,738 1,738 1,738 2,845 2,845 2,066 2,066 2,066 2,066 2,066 2,066 2,066 1,2631 1	1,347 1,588 1,77 2,75 2,75 3,305 1,053 1,403 1,403 1,403 1,403 1,403 1,403 1,403 1,403 1,403	Odinary Odinary
UIS Securities Australia (1d) UIS Securities UIS (1d) UIS Securities UIS (1d) UIS Securities UIS	13.775 1.007 4.468 2.368 1.739 2.866 2.066 2.060 2.060 2.273	1,598 117 117 117 202 202 3,305 1,053 2,403 2,403 1,67 1,63 1,63 1,63 1,63 1,63 1,63 1,63 1,63	Ordinary Ordinary
UBS Securities Australia Utd UBS Securities Australia Ltd	1,007 4,469 2,368 2,1739 28,456 20,066 2,066 1,2,631 1,2,531 1,2,131	117 202 202 202 3,305 1,053 1,467 1,467 1,467 1,47 1,487 1,487	Odinay Odinay
UBS Securities Australia Ltd UBS Securities Australia Ltd	4,469 2,388 1,739 1,739 9,066 20,690 1,2,631 1,2,631 1,2,631 1,2,631 1,2,631 1,2,631	202 202 202 3305 1,053 1,467 2,403 1,467 2,64 2,64 1,317	Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	2,388 1,739 1,739 2,066 2,066 2,069 1,2,631 1,2,631 1,2,131	275 202 3,305 1,053 2,403 1,467 264 1,47 1,47 1,487 1,487	Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	1,739 2,845 9,066 20,660 2,531 12,531 12,13	202 3,305 1,053 2,403 1,467 264 1,47 1,327	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	28,456 9,066 20,650 12,651 2,273	3,305 1,053 2,403 1,467 264 264 147 1,331	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	20,050 9,066 20,690 12,631 2,273	1,053 2,403 1,467 1,467 264 1,327 1,327	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	20,690 20,690 12,631 2,273	2,403 1,467 264 264 14 1,327	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
USS Securities Australia Ltd USS Securities Australia Ltd	12,631 12,631 2,273 121	1,467 264 264 1,327 1,327	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia tot UBS Securities A	2,273	1,327	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	121	1,327	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	- 77	1,327	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	20, 44	1,327	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Ltd UBS Securities Australia Ltd	624,11	1.43	Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia trd UBS Securities Australia trd	17,371		Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities UBS Securit	930	801	Ordinary Ordinary Ordinary
UBS Securities Australia (14 UBS UBS UBS UBS	732	82	Ordinary Ordinary Ordinary
UBS Securities Australia (1d) UBS Securities (Australia (1d)	3,995	464	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	129	15	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	2,153	250	7
UBS Securities Australia Ltd UBS Securities Australia Ltd	2,445	284	Ordinary
UBS Securities Australia trd UBS Securities Australia trd	181	21	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	1.403	163	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	1.403	163	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	o	-	Ordinary
UBS Securities Australia tot UBS Securities A	100	- 61	Ordinary
ul 185 securities Australia Ltd UBS Securities Australia Ltd	103	71	Ordinary
UNS Securines Australia Ltd UNS Securines Australia Ltd	5,594	649	Ordinary
UBS Securines Australia to UBS Securines Austral	46.5	649	Ordinary
1095 Securines Australia to 1085 Securines Australia Ltd 1085 Securines Au	155	18	Ordinary
UNS Securine Australia Ltd UNS Securines Australia Ltd	4,931	572	Ordinary
UBS Securities Australia tod UBS Securities Australia tod	8,172	948	Ordinary
UBS Securines Australia (1d) UBS Securines Australia (1d)	7,172	832	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	3,784	439	Ordinary
UBS Securities Australia Utd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Luts UBS Securities UBS	1,233	143	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Lastralia Ltd UBS Securities Australia Ltd	6,465	750	Ordinary
U35 Securities Australia Ltd U85 Securities Australia Ltd U85 Securities Australia Ltd U85 Securities Australia Ltd U85 Securities Australia Ltd U85 Securities Australia Ltd U85 Securities Australia Ltd U85 Securities Australia Ltd U85 Securities Australia Ltd U85 Securities Australia Ltd U85 Securities Australia Ltd U85 Securities Australia Ltd U85 Securities Australia Ltd U85 Securities Australia Ltd U85 Securities Australia Ltd U85 Securities Australia Ltd U85 Securities Australia Ltd U85 Securities Australia Ltd	14.620	1696	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Lateralia Ltd UBS Securities Lateralia Ltd UBS Securities Lateralia Ltd	52	90,	Ordinan
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	2 155	250	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities UBS	50		Ordinary
UBS Securities Australia tot UBS Securities A	OFCE	.00	Ordinary
USS Securities Australia tot USS Securities A	1691	100	Codinary
UBS Securities Australia tro UBS Securities Australia tro UBS Securities Australia tro UBS Securities Australia tro UBS Securities Australia tro UBS Securities Australia tro UBS Securities Australia tro UBS Securities Australia tro UBS Securities Australia tro UBS Securities Australia tro UBS Securities Australia tro UBS Securities Australia tro UBS Securities Australia tro UBS Securities Australia tro UBS Securities Australia tro UBS Securities Australia tro	170,1	8	Ordinary
UBS Securities Australia tro UBS Securities Australia tro UBS Securities Australia tro UBS Securities Australia tro UBS Securities Australia tro UBS Securities Australia tro UBS Securities Australia tro UBS Securities Australia tro UBS Securities Australia tro UBS Securities Australia tro UBS Securities Australia tro UBS Securities Australia tro	13,395	455,1	Ordinary
Uso Securities Australia Ltd. UBS Securities Australia Ltd UBS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd	4,353	coc.	Ordinary
UBS Securities Australia (1d) UBS Securities Australia (1d) UBS Securities Australia (1d) UBS Securities Australia (1d) UBS Securities Australia (1d) UBS Securities Australia (1d) UBS Securities Australia (1d) UBS Securi	121	14	Ordinary
UBS Securities Australia Ird UBS Securities Australia Ird UBS Securities Australia Ird UBS Securities Australia Ird UBS Securities Australia Ird UBS Securities Australia Ird UBS Securities Australia Ird	121	14	Ordinary
USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd	10,988	1,274	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	52	9	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	5 482	989	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	5,474	635	Ordinary
UBS Securities Australia Ltd	1 601	201	Ordinan
	160'1	8	Ordinary
	8,811	1,021	Ordinary
	9,741	1,130	Ordinary
	414	48	Ordinary
	4,413	512	Ordinary
	4,280	496	Ordinary
	501	25,0	Ordinary
	is:	8	Ciulinary
UBS Securities Australia Ltd Buy	302	35	Ordinary
	2.439	200	Codeman

12-Nov-14   UBS AG, London Branch   Sell   12-Nov-14   UBS AG, London Branch   Buy   12-Nov-14   UBS AG, London Branch   Buy   12-Nov-14   UBS AG, London Branch   Buy   12-Nov-14   UBS AG, London Branch   Sell   12-Nov-14   UBS AG, London Branch   Sell   12-Nov-14   UBS AG, London Branch   Sell   12-Nov-14   UBS AG, London Branch   Sell   12-Nov-14   UBS AG, London Branch   Sell   12-Nov-14   UBS AG, London Branch   Sell   12-Nov-14   UBS AG, London Branch   Sell   12-Nov-14   UBS AG, London Branch   Sell   12-Nov-14   UBS AG, London Branch   Sell   12-Nov-14   UBS AG, London Branch   Sell   12-Nov-14   UBS Securities Advantagement Australia Ltd   Sell   12-Nov-14   UBS Securities Australia Ltd   Sell   Securities Australia Ltd   Sell   Securities Australia Ltd   Sell   Securities Australia Ltd   Sell   Securities Australia Ltd   Buy   12-Nov-14   UBS Securities Australia Ltd   Buy   12-Nov-14   UBS Securities Australia Ltd   Buy   12-Nov-14   UBS Securities Australia Ltd   Buy   12-Nov-14   UBS Securities Australia Ltd   Buy   12-Nov-14   UBS Securities Australia Ltd   Buy   12-Nov-14   UBS Securities Australia Ltd   Buy   12-Nov-14   UBS Securities Australia Ltd   Buy   12-Nov-14   UBS Securities Australia Ltd   Buy   12-Nov-14   UBS Securities Australia Ltd   Buy   12-Nov-14   UBS Securities Australia Ltd   Buy   13-Nov-14   UBS Securities Australia Ltd   Sell   13-Nov-14   UBS Securities Australia Ltd   Sell   13-Nov-14   UBS Securities Australia Ltd   Sell   13-Nov-14   UBS Securities Australia Ltd   Sell   13-Nov-14   UBS Securities Australia Ltd   Sell   13-Nov-14   UBS Securities Australia Ltd   Sell   13-Nov-14   UBS Securities Australia Ltd   Sell   13-Nov-14   U	1437,247   141,434   141,434   141,434   141,434   141,434   141,434   143,641   143	(157,147) (13,234) (13,234) (13,234) (13,500) (1	Ordinary Ordinary
UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG, Australia Branch UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Securites Australia Ltd UBS Securites Australia Ltd UBS Securites Australia Ltd UBS Securites Australia Ltd UBS Securites Australia Ltd UBS Securites Australia Ltd UBS Securites Australia Ltd UBS Securites Australia Ltd UBS Securites Australia Ltd UBS Securites Australia Ltd UBS Securites Australia Ltd UBS Securites Australia Ltd UBS Securites Lauralia Ltd UBS Securites Australia Ltd UBS Securites Lauralia Ltd UBS Weath Management Australia L		(1328) (1328) (1389) (1389) (1389) (1389) (1389) (1389) (1389) (1483) (1	Odrinay Odrinay
185 AG, London Banch 186 AG, London Banch 186 AG Australia Banch 186 AG Australia Banch 188 AG Australia Banch 188 AG Australia Banch 188 AG Australia Banch 188 AG Australia Banch 188 AG Australia Banch 188 AG Australia Banch 188 AG Australia Banch 188 AG Australia Banch 188 Securities Australia Ltd 188 Securities Australia Ltd 188 Securities Australia Ltd 188 Securities Australia Ltd 188 Securities Australia Ltd 188 Securities Australia Ltd 188 Securities Australia Ltd 188 Securities Australia Ltd 188 Securities Australia Ltd 188 Securities Australia Ltd 188 Securities Australia Ltd 188 Securities Australia Ltd 188 Securities Australia Ltd 189 Securities Ltd 180 Securities Australia Ltd 180 Securities Ltd 180 Securities Ltd 180 Securities Ltd 180 Securities Ltd 180 Securities Ltd 180 Securities Ltd 180 Securities Ltd 180 Securities Ltd 180 Secu		(3.724) (3.724) (3.729) (1.350) (1.350) (1.350) (1.350) (1.350) (1.350) (1.453	Octinas Octina
1985 AG, Lanckon Barach, 1985 AG, Australia Branch 1985 AG, Australia Branch 1985 AG, Australia Branch 1985 AG, Australia Branch 1985 AG, Australia Branch 1985 AG, Australia Branch 1985 AG, Australia Branch 1985 AG, Australia Branch 1985 Wealth Management Australia Ltd 1985 Wealth Management Australia Ltd 1985 Securities Lond London Branch 1985 Securities Australia Ltd 1985 Securities London Branch 1985 Securities London Branch 1985 Securities London Branch 1985 Securities London Branch 1985 Securities London Branch 1985 Securities London Branch 1985 Securities London Branch 1985 Securities London Branch 1985 Securities London Branch 1985 Securities London Branch 1985 Securities London Branch 1985 Securities London Branch 1985 Securities London Branch 1985 Securities London Branch 1985 Securities London Branch 1985 Securities London B		(13.34) (1.52) (1.52) (1.52) (1.52) (1.182) (1	Oddinay Oddinay
185 AG Australia Branch 185 AG Australia Branch 185 AG Australia Branch 185 AG Australia Branch 185 AG Australia Branch 185 AG Australia Branch 185 AG Australia Branch 185 Securia Australia Ltd 185 Securia Branch 185 Securia Branch 185 Securia Branch 185 Securia Australia Ltd 185 Securia Branch 185 Securia Branch 185 Securia Branch 185 Securia Branch 185 Securia Branch 185 Securia Branch 185 Securia Branch 185 Securia Australia Ltd 185 Securia Branch 185 Securia Branch 185 Securia Australia Ltd 185 Securia Branch 185 Securia Branch 185 Securia Australia Ltd 185 Securia Branch 185 Securia Branch 185 Securia Australia Ltd 185 Securia Branch 185 Securia Branch 185 Securia Australia Ltd 185 Securia Branch 185 Securia Branch 185 Securia Australia Ltd 185 Securia Branch 185 Securia Branch 185 Securia Australia Ltd 185 Securia Branch 185 Securia Branch 185 Securia Australia Ltd 185 Securia Branch 185 Securia Bran		(3.125) (15.27) (1.3980) (2.524) (8.327) (8.327) (1.459) (9.038) (9.038) (9.038) (1.459) (1.459) (9.038) (1.459) (1.45	Odinas Od
UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS AG Australia Branch UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Securities Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Managemen		(1,130) (1,130) (1,132) (2,524) (3,327) (4,132) (4,132) (4,132) (4,132) (4,132) (4,132) (4,132) (4,132) (4,132) (4,132) (4,132) (4,132) (5,133) (5,134) (5,134) (5,134) (7,134	Ocinary Ocinary
185 AG Australia Branch 185 AG Australia Branch 185 AG Australia Branch 185 AG Australia Branch 185 AG Australia Branch 185 AG Australia Branch 185 Securities Australia Ltd 185 Securities Lud 186 Securities Lud 187 Securities Lud 188 Securities Lud 188 Securities Lud 188 Securities Lud 188 Securities Lud 188 Securities Lud 188 Securities Lud 188 Securities Lud 188 Securities Lud 188 Securities Lud 188 Securities Lud 188 Securities Lud 188 Securities Lud 188 Securities Lud 188 Securities Lud 188 Securities Lud 188 Securities Lud 188		(1,390) (1,320) (1,320) (1,327) (1,327) (1,629) (9,039	Ordinary Ordinary
198 Securities Branch 198 Securities Branch 198 Securities Branch 198 Wealth Management Australia Ltd 198 Wealth Management Australia Ltd 198 Wealth Management Australia Ltd 198 Securities Australia Ltd 198 Weelth Management Australia Ltd 198 Weelth Management Australia Ltd 198 Weelth Management Australia Ltd 198 Weelth Management Australia Ltd 198 Weelth Management Australia Ltd 198 Weelth Management Australia Ltd 198 Weelth Management Australia Ltd 198 Weelth Management Australia Ltd 198 Weelth Management Australia Ltd 198 Weelth Management Australia Ltd 198 Weelth Management Australia Ltd 198 Weelth Management Australia Ltd 198 Weelth Management Australia Ltd 198 Weelth Management Australia Ltd 198 Weelth Management Australia Ltd 198 Weelth Management Australia Ltd 198 Weelth Managemen		(13,080) (2,524) (6,327) (1,482) (9,038) (9,038) (1,459) (1,45	Odelano Odelan
1985 AG Australia Branch 1985 AG Australia Branch 1985 Weath Management Australia Ltd 1985 Weath Management Australia Ltd 1985 Weath Management Australia Ltd 1985 Securities Australia Ltd 1985 Weath Management Australia Ltd 1985 Weath Management Australia Ltd 1985 Weath Management Australia Ltd 1985 Weath Management Australia Ltd 1985 Weath Management Australia Ltd 1985 Weath Management Australia Ltd 1985 Weath Management Australia Ltd 1985 Weath Management Australia Ltd 1985 Weath Management Australia Ltd 1985 Weath Management Australia Ltd 1985 Weath Management Australia Ltd 1985 Weath Management Australia Ltd 1985 Weath Management Australia Ltd 1985 Weath Management Australia Ltd 1985 Weath Management Australia Ltd 1985 Weath Management Australia Ltd 1985 Weath Management Australia Ltd 1985 Weath Management Australia Ltd 1985 Weath Management Australia Ltd 1985		(8.327) (1.1827) (1.1827) (1.4528) (9.038) (9.	Ocidings Oci
USS Weath Management Australia Ltd USS Weath Management Australia Ltd USS Weath Management Australia Ltd USS Weath Management Australia Ltd USS Weath Management Australia Ltd USS Securities Lud USS Securities Ustralia Ltd USS Securities Australia Ltd USS Securities Lud USS Securities Ustralia Ltd USS Securities Ustralia Ltd USS Securities Ltd USS Securities Ustralia Ltd USS Securities Ustralia Ltd USS Securities Ustralia Ltd USS Securities Ustralia Ltd USS Securities Ustralia Ltd USS Weath Management Australia Ltd		(8.437) (1.182) (9.638) (9.038	Octina) Octina)
USS Weath Management Australia Ltd USS Weath Management Australia Ltd USS Securities USB USB USB USB USB USB USB USB USB USB		(843) (843) (943) (9,038) (9,0	Odina) Odina)
185 Vestin Management Australia Ltd 188 Vestin Management Australia Ltd 188 Vestin Management Australia Ltd 188 Securities Australia Ltd 188 Weeth Management Australia Ltd 189 Weeth Management Australia Ltd 189 Weeth Management Australia Ltd 189 Weeth Management Australia Ltd 189 Weeth Management Australia Ltd 189 Weeth Management Australia Ltd 189 Weeth Management Australia Ltd 189 Weeth Management Australia Ltd 189 Weeth Management Australia Ltd 189 Weeth Management Australia Ltd 189 Weeth Management Australia Ltd 180 Weeth Management Australia Ltd 180 Weeth Management Australia Ltd 181 Weeth Management Australia Ltd 181 Weeth Management Australia Ltd 181 Weeth Management Australia Ltd 181 Weeth Wanag	2 - 2	(843) (9,038) (9,038) (9,038) (9,038) (13) (13) (13) (13) (13) (14) (14) (15) (15) (15) (15) (16) (16) (16) (16) (16) (16) (16) (16	Octina) Octina)
UBS Securities Australia Ltd UBS Weethi Management Australia Ltd UBS Weethi Management Australia Ltd UBS Weethi Management Australia Ltd UBS Weethi Management Australia Ltd UBS Weethi Management Australia Ltd UBS Weethi Management Australia Ltd UBS Weethi Management Australia Ltd UBS Weethi Management Australia Ltd UBS Weethi Management Australia Ltd UBS Weethi Management Australia Ltd UBS Weethi Management Australia Ltd UBS Weethi Management Australia Ltd UBS Weethi Management Australia Ltd UBS Weethi Management Australia Ltd UBS Weethi Management Australia Ltd UBS Weethi Management Australia Ltd UBS Weethi Management Australia Ltd UBS Weethi Management Australia Ltd		(9.038) (9.038) (9.038) (1.1) (1.1) (1.2)	Odinary Odinary
UBS Securities Australia tota UBS Weethir Management Australia tota UBS Weethir Management Australia tota UBS Weethir Management Australia tota UBS Weethir Management Australia tota UBS Weethir Management Australia tota UBS Weethir Management Australia tota UBS Weethir Management Australia tota UBS Weethir Management Australia tota UBS Weethir Management Australia tota UBS Weethir Management Australia tota UBS Weethir Management Australia tota UBS Weethir Management Australia tota UBS Weethir Management Australia tota UBS Weethir Management Australia tota UBS Weethir Management Australia tota UBS Weethir Management Australia tota UBS Weethir Management Australia tota UBS Weethir Management Australia tota UBS Weethir Management Australia tota UBS Weethir Management Australia t		(9.038) (9.038) (13) (13) (14) (15) (15) (15) (16) (16) (17) (17) (17) (18) (17) (18) (18) (18) (18) (19) (19) (19) (19) (19) (19) (19) (19	Octinary Octinary
UBS Securines Australia trid UBS Weeth Management Australia trid UBS Weeth Management Australia trid	7 7	(33) (43) (5,444 (744	Ociniary Ociniary
UBS Securities Australia tot UBS Weeth Management Australia tot UBS W		5,444 1 2,598 2,788 2,788 2,191 8,603 5,97 2,510 1 1,000 10,000 1	Octinary Octinary
UBS Securities Australia tot UBS Weeth Management Australia tot UBS Weeth Management Australia tot	7	2,444 2,948 2,948 5,783 3,193 2,191 2,191 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	Octinary Octinary
UBS Securines Australia Ltd UBS Securities Australia Ltd UBS Weeth Management Australia Ltd	7	5,444 1 2,598 2,788 2,191 8,603 597 1 551 1 1 10,000 10,00	Ordinary Ordinary
UBS Securines Australia Ltd UBS Securines UB	2 - 2	2,998 5,783 5,783 2,191 8,603 8,603 1,512 1,512 1,513 1,000	Odinaly Odinaly
UBS Securities Australia Ltd UBS Weeth Management Australia Ltd UBS Weeth Management Australia Ltd		2,998 5,783 1 338 2,191 8,603 5,97 5,97 1 1 1 1 1 1,000 1,00	Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary Odinary
UBS Securities Australia trid UBS Weethir Management Australia trid UBS Weethir Management Australia trid	1 2 2	5,783 138 2,191 8,603 597 597 11 11 11 10,000	Ordina) Ordina) Ordina) Ordina) Ordina) Ordina) Ordina) Ordina) Ordina) Ordina) Ordina) Ordina) Ordina) Ordina)
UBS Securities Australia Ltd UBS Securities Lud UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Ltd UBS Weeth Management Australia Ltd UBS Weeth Management Australia Ltd UBS Weeth Management Australia Ltd UBS Weeth Management Australia Ltd UBS Weeth Management Australia Ltd UBS Weeth Management Australia Ltd UBS Weeth Management Australia Ltd UBS Weeth Management Australia Ltd UBS Weeth Management Australia Ltd UBS Weeth Management Australia Ltd UBS Weeth Management Australia Ltd UBS Weeth Management Australia Ltd UBS Weeth Management Australia Ltd UBS Weeth Management Australia Ltd UBS Weeth Management Australia Ltd UBS Weeth Management Australia Ltd UBS Weeth Management Australia Ltd UBS Weeth Management Australia Ltd	7 2	2,131 2,131 8,603 8,603 597 2,531 1,8,633 1,8,633 1,8,633 1,0,000 1,000	ordina) ordina) ordina) ordina) ordina) ordina) ordina) ordina) ordina) ordina) ordina) ordina) ordina) ordina) ordina)
UBS Securines Australia Ltd UBS Securines Ludical Ltd UBS Securines Ludical Ltd UBS Securines Ludical Ltd UBS Securines Ludical Ltd UBS Securines Ludical Ltd UBS Securines Ludical Ltd UBS Securines Ludical Ltd UBS Securines Ludical Ltd UBS Securines Ludical Ltd UBS Weeth Management Australia Ltd	1 2	338 2,191 8,603 597 2,513 1,8,633 2,5,810 2,5,810 10,000 10,000 (857) (857)	Ordinary Ordinary
UBS Securines Australia Ltd UBS Securines UB	2	2,191 8,602 5,57 2,51 1 1 1,8,633 2,5,810 1,000 1,000 1,000 1,000 6,55 6,55 6,55 6,53 6,53 6,53 6,53 6,53	Ordinay Ordinay Ordinay Ordinay Ordinay Ordinay Ordinay Ordinay Ordinay Ordinay Ordinay Ordinay Ordinay Ordinay
UBS Securines Australia Ltd UBS Weeth Management Australia Ltd UBS Weeth Management Australia Ltd	5 7	8,603 8,603 597 2,51 1 1 1,000 1,000 1,000 1,000 1,000 1,000 655 655	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securines Australia Ltd UBS Securines UBS The	2 7	8,603 8,603 257 251 1 18,633 18,633 25,810 10,000 10,000 (857) 655 655 657	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
1985 Securities Australia Ltd  URS Wealth Management Australia Ltd  URS Wealth Management Australia Ltd	7 7 7	597 2 551 1 1 1 18,633 25,810 10,000 (857) 655 655	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securines Australia tod UBS Weeth Management Australia tod UBS Weeth Management Australia tod	2 - 2	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd	2	551 1 1 1 18,633 25,810 10,000 (857) 655 655	Ordinar Ordinar Ordinar Ordinar Ordinar Ordinar Ordinar Ordinar Ordinar Ordinar Ordinar Ordinar
UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines UB	5	18,633 25,810 10,000 (857) 655 7,000 10,000	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Weeth Management Australia Ltd UBS Weeth Management Australia Ltd	2 2	18.633 25,810 10,000 (857) 655 7,000	Ordinan Ordinan Ordinan Ordinan Ordinan Ordinan Ordinan Ordinan
UBS Securities Australia Ltd UBS Keath Management Australia Ltd UBS Weath Management Australia Ltd	Α	18,633 25,810 1 10,000 (857) 655 (230)	Ordinar Ordinar Ordinar Ordinar Ordinar Ordinar Ordinar Ordinar
UBS Securines duration to UBS Securities duration to UBS Securities duration to UBS Securities duration to UBS Securities duration to UBS Securities duration to UBS General Management UBS Securities duration to UBS Securities duration to UBS Securities duration to UBS Securities duration to UBS Securities duration to UBS Securities duration to UBS Securities duration to UBS Securities duration to UBS Securities duration to UBS Securities duration to UBS Actual Branch UBS ACTUAL DURATION TO UBS ACTUAL DURATION TO UBS ACTUAL DURATION TO UBS Weelth Management Australia Ltd UBS Weelth Management Austral		25,810 10,000 (857) 655 (230)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Gocurities Australia Ltd UBS Gocurities Australia Ltd UBS Securities Ltd UBS Securities Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Mealth Management Australia Ltd UBS Wealth Management Australia Ltd		10,000 (857) 655 (230)	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
Less Secontres Australia Ltd  UBS Secontres Australia Ltd  UBS Secontres Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Index  UBS Australia Banch  UBS Act London Branch  UBS Act London Branch  UBS Weelth Management Australia Ltd		10,000 (857) (555 (230)	Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Gotchal Aspert Management UBS Securities Australia tot UBS Securities Australia tot UBS Securities Australia tot UBS Securities Australia tot UBS Securities Australia tot UBS Securities Australia tot UBS Securities Australia tot UBS Securities Australia tot UBS Securities Australia tot UBS Securities Australia tot UBS Securities Australia tot UBS Securities Australia tot UBS Securities Bench UBS Securities Bench UBS Mealth Management Australia tot UBS Wealth Management Australia tot		(857) (857) (530)	Ordinary Ordinary Ordinary Ordinary Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Banch UBS Securities Banch UBS Act London Ranch UBS AG, London Ranch UBS AG, London Ranch UBS AG, London Ranch UBS AG, London Ranch UBS AG, London Ranch UBS AG, London Ranch UBS Securities Ltd UBS Weelth Management Australia Ltd		(230)	Ordinary Ordinary Ordinary
UBS Security Australia 11d UBS Security Australia 11d UBS Security Australia 11d UBS Security Australia 11d UBS Security Australia 11d UBS Securities Australia 11d UBS Securities Australia 11d UBS Securities Australia 11d UBS Securities Australia 11d UBS Securities Australia 11d UBS Securities Australia 11d UBS Securities Australia 11d UBS Securities UBS The Third India 11d UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS Wealth Management Australia 11d		(230)	Ordinary Ordinary Ordinary
UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Act London Branch UBS Act London Branch UBS Act London Branch UBS Act London Branch UBS Act London Branch UBS Act London Branch UBS Act London Branch UBS Act London Branch UBS Wealth Management Australia Ltd		(230)	Ordinary Ordinary Ordinary
UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines UBS The Third Securines UBS AGE London Branch UBS AGE, London Branch UBS AGE, London Branch UBS AGE, London Branch UBS AGE, London Branch UBS AGE, London Branch UBS AGE, London Branch UBS AGE, London Branch UBS Securines Management Australia Ltd UBS Weelth Management Australia Ltd			Ordinary Ordinary
UBS Securines Australia tod UBS Securines Australia tod UBS Securines Australia tod UBS Securines Australia tod UBS Securines Australia tod UBS Securines Australia tod UBS Securines Australia tod UBS Securines Australia tod UBS Securines Australia tod UBS Securines Australia tod UBS Action Branch UBS Action Branch UBS Meath Management Australia tod UBS Weath Management Australia tod		(7,952)	Ordinary
UBS Securines Australia tod UBS Securines Australia tod UBS Securines Australia tod UBS Securines Australia tod UBS Securines Australia tod UBS Securines Australia tod UBS Securines Australia tod UBS Securines Australia tod UBS Securines Banch UBS AG Australia Banch UBS AG Australia Banch UBS AG Australia Banch UBS AG Australia Banch UBS Wealth Management Australia tod	123,776	14,760	Ordinary
UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Banch UBS AG, London Branch UBS AG, Branch UBS AG, Branch UBS Wealth Management Australia Ltd		(167,405)	
UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS AGE London Rearch UBS AGE London Rearch UBS AGE London Rearch UBS AGE London Rearch UBS AGE Australia Banch UBS AGE Australia Banch UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd		(866'64)	Ordinary
UBS Securities abustralia tud UBS Securities Australia tud UBS Securities Australia tud UBS Securities Australia tud UBS Securities Australia tud UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS Weeth Management Australia tud		4,962	Ordinary
UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS Securines Australia Ltd UBS AG, London Banch UBS AG, London Banch UBS AG, London Banch UBS Wealth Management Australia Ltd		20,660	Ordinary
UBS Securines Australia Etd UBS Securines Australia Etd UBS AG. Lordon Branch UBS AG. Lordon Branch UBS AG. Australia Branch UBS AG. Australia Branch UBS AG. Australia Branch UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd		(9,435)	Ordinary
UBS Securities and UBS Active Australia and UBS Act. London Bench 1955 Act. London Bench 1955 Act. London Bench 1955 Act. London Bench 1955 Act. London Bench 1955 Act. London Bench 1955 Act. London Bench 1955 Act. London Bench 1955 Act. London Bench 1955 Act. London Bench 1955 Act. 195		8,600	Ordinary
UBS AG, London Branch UBS AG, London Branch UBS AG, London Branch UBS AG, Australia Branch UBS AG Australia Branch UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd		(1.491)	Ordinan
UBS AG, London Rench. UBS AG, Australia Branch UBS AG, Australia Branch UBS Weath Management Australia Ltd	y 1,731,740	167.405	Ordinary
UBS AG AUSTRAIR BRANCH UBS AGA AUSTRAIR BRANCH UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd		(167.405)	Ordinan
UBS Meah Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd	150	(00,01)	Crainary
UES Weath Management Australia Ltd UES Weath Management Australia Ltd		(080,81)	Orginary
UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd		(2/7)	Ordinary
USE Weath Management Australia Ltd USE Weath Management Australia Ltd USE Weath Management Australia Ltd USE Weath Management Australia Ltd USE Weath Management Australia Ltd USE Weath Management Australia Ltd USE Weath Management Australia Ltd USE Weath Management Australia Ltd USE Weath Management Australia Ltd USE Weath Management Australia Ltd USE Weath Management Australia Ltd USE Weath Management Australia Ltd USE Weath Management Australia Ltd USE Weath Management Australia Ltd USE Weath Management Australia Ltd USE Weath Management Australia Ltd USE Weath Management Australia Ltd USE Weath Management Australia Ltd USE Weath Management Australia Ltd	2,308	(276)	Ordinary
USS Westin Wanagement Australia Ltd USS Westin Management Australia Ltd		(898)	Ordinary
USS Weath Management Australia Ltd USS Weath Management Australia Ltd USS Weath Management Australia Ltd USS Weath Management Australia Ltd USS Weath Management Australia Ltd USS Weath Management Australia Ltd USS Weath Management Australia Ltd USS Weath Management Australia Ltd USS Weath Management Australia Ltd USS Weath Management Australia Ltd USS Weath Management Australia Ltd USS Weath Management Australia Ltd USS Weath Management Australia Ltd	3,270	(391)	Ordinary
UBS Weath Management Australia tod UBS Weath Management Australia tod UBS Weath Management Australia tod UBS Weath Management Australia tod UBS Weath Management Australia tod UBS Weath Management Australia tod UBS Weath Management Australia tod UBS Weath Management Australia tod UBS Weath Management Australia tod UBS Weath Management Australia tod UBS Weath Management Australia tod UBS Weath Management Australia tod UBS Weath Management Australia tod UBS Weath Management Australia tod UBS Weath Management Australia tod		(12,658)	Ordinary
UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd		(202)	Ordinary
UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd UBS Weath Management Australia Ltd	18,657	(2,231)	Ordinary
UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd		(273)	Ordinary
UBS Wealth Management Australia 110 UBS Wealth Management Australia 110 UBS Wealth Management Australia 110 UBS Wealth Management Australia 110 UBS Wealth Management Australia 110 UBS Wealth Management Australia 110		(323)	Ordinary
UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd	11 54,215	(6,483)	Ordinary
UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd		(564)	Ordinary
UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd		(564)	Ordinary
UBS Wealth Management Australia Ltd UBS Wealth Management Australia Ltd		660	ording.
UBS Wealth Management Australia Ltd		7250	Colliary
COS **Carrillora ragement Australia Ltd		(07)	Ordinary
I IDC Month & Annonger Australia 14d	71	(15,084)	Ordinary
DO MERCHING HIGH		(841)	Ordinary
Obs wediti Management Australia Ltd		(357)	Ordinar
Obsignation of the state of the		(2,386)	Ordinary
UBS Wealth Management Australia Ltd	1,254	(150)	Ordinan
UBS Wealth Management Australia Ltd		(202)	Ordinan
UBS Wealth Management Australia Ltd		(929)	Ordinan
I IDC Month h Annoncom Annothing		(070)	Colories,
- [	3,922	(469)	Ordinary

11-Nov-14			2006		1000
	UBS Securities Australia Ltd	gc <sup>x</sup>	2,534	286	Ordinary
-Nov-14	UBS Securities Australia Ltd	Buy	1,214	137	Ordinary
-Nov-14	UBS Securities Australia Ltd	Buy	390	44	Ordinary
-Nov-14	UBS Securities Australia Ltd	Buy	488	55	Ordinary
1-Nov-14	UBS Securities Australia 11d	Buv	843	S	Ordinary
-Nov-14	IUBS Securities Australia 1td	Buy	1 419	160	Ordinary
11-Nov-14	UBS Securities Australia Itd	Buy	7.776	178	Coning
Nov-14	LIBS Securities Australia I to	BIN	5535	300	Ordinary
11-Nov-14	I IBS Securities Australia I to	Bis	2 044	702	Coming
Nov-14	LIBS Securities Australia 11d	Bin	0		California
Now 14	IID Counting Augustia 14d	B <sub>137</sub>	220.01	12.5	ig in
11 Nov. 14	Did all places A solitores 2 2013	in a	805,81	716'1	Ordinary
NOV-14	UBS Securities Australia Ltd	buy	Ď.		Ordinary
NOV-14	UBS Securities Australia Ltd	Buy	2,130	241	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	1,450	164	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	029	9/	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	3,034	344	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	2,002	227	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	889	78	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	009	89	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	388	44	Ordinary
1-Nov-14	UBS Securities Australia Ltd	Buy	3 405	386	Ordinary
11-Nov-14	UBS Securities Australia Itd	Buv	2 346	290	Ordinary
No. 14	1100 Committee Australia (+d	D. 0	505.	3	Oldinary
NOV-14	Obs securities Australia Ltd	and a	1,103	571	Ordinary
11-Nov-14	UBS Securities Australia Ltd	Buy	503	25	Ordinary
11-Nov-14	UBS Securities Australia Ltd	Buy	2,238	254	Ordinary
11-Nov-14	UBS Securities Australia Ltd	Buy	2,731	310	Ordinary
11-Nov-14	UBS Securities Australia Ltd	Buy	352	40	Ordinary
1-Nov-14	UBS Securities Australia Ltd	Buy	829	77	Ordinary
Nov-14	UBS Securities Australia 11d	Buv	UUU 88	10,000	or all
11 May 14	I I DC Countries Australia 144	Burg	000 0	000'0	500
1	CDD Securities Australia Ltd	ing	EU8,2	918	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	1,248	142	Ordinary
1-Nov-14	UBS Securities Australia Ltd	Buy	3,409	388	Ordinary
1-Nov-14	UBS Securities Australia Ltd	Buy	27,892	3,175	Ordinary
1-Nov-14	UBS Securities Australia I to	Buy	1 854	211	Ordinary
1-Nov-14	I IBC Contrition Australia I to	Bin	410 00	000	
	Did alle mone con in the control of	600	t   t, 2t	070'4	Cidinary
4Ov-14	UBS SECURIDES AUSTRAINA LTG	acy	0//01	977'	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	22,788	2,594	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	1,555	177	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	694	79	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	4,325	492	Ordinary
Nov-14	LIRS Securities Australia Ltd	Buv	19 962	177.0	Ordinary
Now 14	LIPS Countries Australia 144	Bin	000	15	0.1
	COO Decarines Australia Etc	500	FOC	ò	Ordinary
NOV-14	USS Securities Australia LIO	and bear	768,51	185,	Ordinary
VOV- (4	UBS Securities Australia Ltd	eny	1,2/5	145	Ordinary
Nov-14	UBS Securities Australia Etd	Виу	34,512	3,924	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	24,090	2,739	Ordinary
1-Nov-14	UBS Securities Australia Ltd	Buy	941	101	Ordinary
Nov-14	Securities	Buy	3,536	402	Ordinary
Nov-14	UBS Securities Australia Ltd	Sell	1,772,000	(200,000)	Ordinary
		Enter Into Derivative			
11-Nov-14	UBS Securities Australia Ltd	Contract	A.N	(80,300)	Options
		Enter Into Derivative			
11-Nov-14	UBS Securities Australia Ltd	Contract	NA.	36,400,000	Options
		Enter Into Derivative			
11-Nov-14	UBS Securities Australia Ltd	Contract	N/A	3,200,000	Options
		Enter Into Derivative			
	UBS Securities Australia Ltd	Contract	NA	36,400,000	Options
2-Nov-14	UBS Global Asset Management	Buy	3,712	435	Ordinary
l	UBS Global Asset Management	Transfer-Out	4/N	(127)	Ordinary
	URS Securities Australia Ltd	Sell	2 146 813	(578 753)	Colinary
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	COO COOLINGS	100	2,140,813	(240,733)	Ordinary
4Ov- 14	UBS Securities Australia Ltd	buy	787,462	33,639	Ordinary
Nov-14	UBS Securities Australia Ltd	Sell	52,505	(6,124)	Ordinary
Nov-14	UBS Securities Australia Ltd	Sell	2,551	(295)	Ordinary
Jov-14	UBS Securities Australia Ltd	Seil	113.808	(13.234)	Ordinary
Jov-14	UBS Securities Australia Ltd	Sell	887 427	(103 206)	Ordinary
2-Nov-14	UBS Securities Australia Itd	100	727,100	(CN1 731)	Colinary C
14	Tibo Countries Australia Ind	20	000 00	(/51,101)	Coming
40A-14	Obs securities Australia Ltd	341	34,120	(4,000)	Ordinary
40v-14	UBS Securities Australia Ltd	Ruy	4,334	50C	Ordinary
40A   14	UBS Securities Australia Ltd	Buy	20,400	10,000	Ordinary
Ov-14	(UBS Securities Australia Ltd	Buy	40717	26 506 1	O-dimen
12-Nov-14			OCC, CIL	onc'nc	Orannary
	UBS Securities Australia Ltd	Buy	11,219	1 300	Ordinary

13-Nov-14	UBS Securities Australia Ltd	Buy	752	06	Ordinary
13-Nov-14	UBS Securities Australia Ltd	Buy	75	6	Ordinary
	UBS Securities Australia Ltd	Buy	777	93	Ordinary
	UBS Securities Australia Ltd	Ruy Bus	6,270	750	Ordinary
	UBS Securities Australia Ltd	ouy Post	2,090	250	Ordinary
	IRS Securities Australia Ind	Buy	2,508	300	Ordinary
	URS Securities Australia Itd	S S	PC7'	200	Orginary
ı	UBS Securities Australia Itd	Birv	500	420	Cidinary
l	LIBS Sequities Australia Itol	Bire	10001	//	Countrient
l	UBS Securities Australia Itd	SIN.	755.31	077	Ordinary
l	UBS Securities Australia Itd	Buy	0		o diling
	LIBS Securities Australia 11d	Birv	300	-	Orginary
	I IBS Securities Australia 11d	Buy	200	<del>4</del> (	Orginary
ĺ	II IBS Societies Australia Itd	Buy	0000	23	Crainary
ı	URS Securities Australia It d	SI K	0,520	/9/1	Ordinary
	LIRS Securities Australia 14d	fro a	134	2	Ordinary
	100 Countries Australia 144	100	2000	clc	Ordinary
	I IDS Sociation Australia 140	Paris Paris	/8C'1	191	Ordinary
l	LIBS Securities Australia Ltd	eco)	2,516	301	Ordinary
1	Upo Securities Australia Ltd	Buy	1,689	202	Ordinary
ı	ions securities Australia Ltd	buy	803	96	Ordinary
1	UBS Securities Australia Ltd	Buy	1,053	126	Ordinary
	UBS Securities Australia Ltd	Buy	48,103	5,754	Ordinary
3-Nov-14	UBS Securities Australia Ltd	Buy	20,440	2,445	Ordinary
3-Nov-14	UBS Securities Australia Ltd	Buy	1,605	192	Ordinary
	UBS Securities Australia Ltd	Buy	2,667	319	Ordinary
	UBS Securities Australia Ltd	Buy	2,374	787	Ordinary
	UBS Securities Australia Ltd	Buy	293	35	Ordinary
	UBS Securities Australia Ltd	Buy	41.800	2002	Ordinary
3-Nov-14	UBS Securities Australia Ltd	Buy	418	G	Ordinan
	UBS Securities Australia Ltd	Buy	468	9	Creinan
	UBS Securities Australia Ltd	Buy	15 357	1 837	Ordioan
	UBS Securities Australia Ltd	Buy	276	22	Ordinary
13-Nov-14	UBS Securities Australia Ltd	Buy	17	_	Ordinac
	UBS Securities Australia Ltd	Buy	1 547	185	Ordinary
	UBS Securities Australia Ltd	Buy	21.769	2,604	Codings
	UBS Securities Australia Ltd	Buy	051	0,	Colores
l	UBS Securities Australia 11d	Riv	410.0	0	S C C
	UBS Securities Australia 11d	Buy	36	010	Ordinary
l	LIBS Securities Australia 14d	Biry	080 1	007	Ordinary
l	LIBS Countries Australia 1+d	600	080,4	\$20	Orginary
l	The contines Australia Ltu	ing	76	11	Ordinary
ł	UBS Securities Australia Ltd	Buy	50,879	980'9	Ordinary
1	UBS Securities Australia Ltd	Any	5,610	671	Ordinary
-	UBS Securities Australia Ltd	Buy	5,183	920	Ordinary
1	UBS Securities Australia Ltd	Buy	40,061	4,792	Ordinary
1	UBS Securities Australia Ltd	Buy	234	28	Ordinary
1	UBS Securities Australia Ltd	Buy	268	32	Ordinary
ı	UBS Securities Australia Ltd	Buy	30,196	3,612	Ordinary
	UBS Securities Australia Ltd	Buy	23,943	2,864	Ordinary
3-Nov-14	UBS Securities Australia Ltd	Buy	836	100	Ordinary
	UBS Securities Australia Ltd	Buy	7,323	876	Ordinary
ı		Buy	25,306	3,027	Ordinary
1	UBS Securities Australia Ltd	Buy	3,403	407	Ordinary
1	UBS Securities Australia Ltd	Buy	1,455	174	Ordinary
	UBS Securities Australia Ltd	Buy	2,466	295	Ordinary
3-Nov-14	UBS Securities Australia Ltd	Buy	44,522	5,332	Ordinary
	UBS Securities Australia Ltd	Buy	593	71	Ordinary
ı	UBS Securities Australia Ltd	Buy	418	50	Ordinary
3-NOV-14	UBS Securities Australia Ltd	guy	73,204	8,767	Ordinary
	UBS Securities Australia Ltd	Buy	19,247	2,305	Ordinary
1	UBS Securities Australia Ltd	Buy	84,327	10,099	Ordinary
1	UBS Securities Australia Ltd	Buy	8,676	1,039	Ordinary
	UBS Securities Australia Ltd	Buy	106,696	12,778	Ordinary
	UBS Securities Australia Ltd	Buy	230,427	27,596	Ordinary
	UBS Securities Australia Ltd	Buy	16,700	2,000	Ordinary
	UBS Securities Australia Ltd	Buy	41,750	2,000	Ordinary
	UBS Securities Australia Ltd	Buy	169,797	20,335	Ordinary
	UBS Securities Australia Ltd	Buy	16.725	2.003	Ordinary
	UBS Securities Australia Ltd	Buy	34.527	4135	Ordinary
	UBS Global Asset Management	Sell	1 760	(215)	Ordinary
l	UBS Securities Australia Itd	Biv	10,434	1 270	Country
	Cap Securines Australia Lid	nai	0,424		The second
				U/4,1	Ordinary
1	IUBS Securities Australia Ltd	Sell	57.340	(000.7)	Ordinary

A 0 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1					Chalipary
13-Nov-14	UBS Wealth Management Australia Ltd	138	0461	(030)	Ordinary
13-Nov-14	UBS Wealth Management Australia Ltd	Sell	1,430	(171)	Ordinary
13-Nov-14	UBS Wealth Management Australia Ltd	Sell	1,907	(228)	Ordinary
13-Nov-14	UBS Wealth Management Australia Ltd	Sell	2,509	(300)	Ordinary
3-Nov-14	UBS Wealth Management Australia Ltd	Sell	2.450	(293)	Ordinary
3-Nov-14	UBS Wealth Management Australia Ltd	Sell	2,693	(322)	Ordinary
13-Nov-14	UBS Wealth Management Australia Ltd	Sell	3,964	(474)	Ordinary
13-Nov-14	UBS Wealth Management Australia Ltd	Sell	3,738	(447)	Ordinary
13-Nov-14	UBS Wealth Management Australia Ltd	Sell	3,638	(435)	Ordinary
3-Nov-14	UBS Wealth Management Australia Ltd	Sell	4,006	(479)	Ordinary
13-Nov-14	UBS Wealth Management Australia Ltd	Sell	953	(114)	Ordinary
13-Nov-14	UBS Wealth Management Australia Ltd	Sell	104,098	(12,448)	Ordinary
3-Nov-14	UBS Wealth Management Australia Ltd	Sell	1,865	(223)	Ordinary
3-Nov-14	UBS Securities Australia Ltd	Buy	1,177,400	140,000	Ordinary
3 Nov 14	UBS Securities Australia Ltd	buy.	584,500	000'0/	Ordinary
3-Nov-14	UBS Securities Australia Ltd	Bin	899'5	436	Ordinary
3-Nov-14	UBS Securities Australia Itol	Buy	0 122	1004	Ordinary
3-Nov-14	UBS Securities Australia Ind	Buy	167 173	10 866	Ordinary
3-Nov-14	UBS Securities Australia Itd	Buv	1153	137	Ordinary
3-Nov-14	UBS Securities Australia Ltd	Buy	5,049	009	Ordinary
3-Nov-14	UBS Securities Australia Ltd	Buy	14,794	1,758	Ordinary
	UBS Securities Australia Ltd	Buy	25,245	3,000	Ordinary
	UBS Securities Australia Ltd	Buy	41,553	4,938	Ordinary
	UBS Securities Australia Ltd	Buy	28,527	3,390	Ordinary
1	UBS Securities Australia Ltd	Buy	52,956	6,293	Ordinary
3-Nov-14	UBS Securities Australia Ltd	suy	2,870	341	Ordinary
ı	UBS Securities Australia Ltd	Busy	816	109	Ordinary
	UBS Securities Australia 14d	Buv	3,688	00	Ordinary
	UBS Securities Australia Ltd	Buy	266'9	831	Ordinary
3-Nov-14	UBS Securities Australia Ltd	Buy	r.	9	Ordinary
	UBS Securities Australia Ltd	Buy	4,303	511	Ordinary
	UBS Securities Australia Ltd	Buy	17,581	2,088	Ordinary
3-Nov-14		Buy	2,105	250	Ordinary
3-Nov-14	UBS Securities Australia Ltd	Buy	42	2	Ordinary
l	UBS Securities Australia Ltd	Buy	261	31	Ordinary
3-Nov-14	I IDS Securities Australia 140	Buy	108	07	Ordinary
l	UBS Securities Australia I td	Buv	17 387	2 065	Ordinary
l	UBS Securities Australia Ltd	Buv	3.814	453	Ordinary
3-Nov-14	UBS Securities Australia Ltd	Buv	749	08	Ordinary
3-Nov-14	UBS Securities Australia Ltd	Buy	1.314	156	Ordinary
3-Nov-14	UBS Securities Australia Ltd	Buy	16,882	2,005	Ordinary
13-Nov-14	UBS Securities Australia Ltd	Buy	3,621	430	Ordinary
3-Nov-14	UBS Securities Australia Ltd	Buy	2,543	302	Ordinary
3-Nov-14	UBS Securities Australia Ltd	Buy	2,526	300	Ordinary
13-Nov-14	UBS Securities Australia Ltd	Buy	51	9	Ordinary
3-Nov-14	UBS Securities Australia Ltd	Buy	2,105	250	Ordinary
3-Nov-14	UBS Securities Australia Ltd	Buy	2,599	309	Ordinary
3-Nov-14	UBS Securities Australia Ltd	Buy	126	15	Ordinary
3-Nov-14	UBS Securities Australia Ltd	Buy	16,460	1,956	Ordinary
3-Nov-14	UBS Securities Australia Ltd	Buy	421	20	Ordinary
ı	UBS Securities Australia Ltd	Buy	88	10	Ordinary
	IUBS Securities Australia Ltd	Buy	29	80	Ordinary
	UBS Securities Australia Etd	Buy	1,742	207	Ordinary
13-Nov-14	UBS Securities Australia Ltd	Buy	353	45	Ordinary
	UBS Securities Australia Ltd	Buy	732	87	Ordinary
3-Nov-14	UBS Securities Australia Ltd	suy	896	115	Ordinary
5-NOV-14	USS Securities Australia Ltd	Buy	679	74	Ordinary
13-Now-14	UBS Securities Australia Ltd	But	1,498	8/1	Ordinary
3-Nov-14	URS Securities Australia Ind	Buy	2,0,2	667	Crainary
13-Nov-14	UBS Securities Australia Ltd	Buy	2711	270	Ordinary
3-Nov-14	UBS Securities Australia Ltd	Buy	3,829	455	Ordinary
3-Nov-14	UBS Securities Australia Ltd	Buy	1,582	188	Ordinary
3-Nov-14	UBS Securities Australia Ltd	Buy	109	13	Ordinary
3-Nov-14	UBS Securities Australia Ltd	Buy	3,112	370	Ordinary
3-Nov-14	UBS Securities Australia Ltd	Buy	8,625	1,025	Ordinary
	UBS Securities Australia Ltd	Buy	4,199	499	Ordinary
	UBS Securities Australia Ltd	Buy	2,432	585	Ordinary
13-Nov-14	UBS Securities Australia Ltd	Buy	1,986	236	Ordinary
l				-	

7-Nov-14	Obsideration Australia Ltd	Stock returned	- V	(000 000 0	Ordinary
1	IIDC AC London Branch	i d		(000'000'0)	(
-Nov-14	UBS AG London Branch	(c)	1 700 370	208,411	Ordinary
Nov-14	UBS AG Australia Branch	200	202,007,1	(40.411)	Ordinary
-Nov-14	UBS AG Australia Branch	Biv	1 136 623	(49,700)	Ordinary
-Nov-14	I IBS AG Australia Branch	Bree	22,001,7	000,000	Ordinary
-Nov-14	UBS AG Australia Branch	Birk	099 331	20,400	Ordinary
-Nov-14	UBS Securities Australia Ltd	Buy	811,000	100,000	Ordinary
-Nov-14	UBS Securities Australia Ltd	Buy	25 908	2 165	Ordinary
-Nov-14	UBS Securities Australia ITd	Buy	50,52	CD1 'C	Oldinaly
-Nov-14	UBS Securities Australia Ltd	Buy	12 278	1 500	Ordings
-Nov-14	UBS Securities Australia Ltd	Buy	3 765	096	Ordinan
-Nov-14	UBS Securities Australia Ltd	Buy	1 523	381	Ordinary
-Nov-14	UBS Securities Australia Ltd	Buy	12,728	1 500	Codings
-Nov-14	UBS Securities Australia Ltd	Buy	16.526	2 019	Ordinary
-Nov-14	UBS Securities Australia Ltd	Buy	3,989	487	Ordinary
Nov. 14	UBS Securities Australia Ltd	Buy	38.273	4.676	Ordinary
-Nov-14	UBS Securities Australia Ltd	Buy	3,143	384	Ordinary
-Nov-14	UBS Securities Australia Ltd	Buy	213	36	Ordinary
-Nov-14	UBS Securities Australia Ltd	Buy	00	-	Ordinary
-Nov-14	UBS Securities Australia Ltd	Buy	843	103	Ordinary
7-Nov-14	UBS Securities Australia Ltd	Buy	28,738	3.511	Ordinary
-Nov-14	UBS Securities Australia Ltd	Buy	1.014	124	Ordinary
-Nov-14	UBS Securities Australia Ltd	Buy	986	010	
-Nov-14	UBS Securities Australia Itd	Buy	3 031	01.0	Cidillary
7-Nov-14	UBS Securities Australia (1d	N B	21,014	6/4	Ordinary
7-Nov-14	URS Securities Australia (14)	200	0000	606,2	Ordinary
Nov-14	I IRC Securities Australia 144	0	2,039	015	Ordinary
Nov-14	I IRS Securities Australia 14d	Riv	104,11	1,403	Ordinary
Nov-14	IIRS Securities Australia 14d	8.5	4 404	305	Orginary
Now 14	II IDC Cognetion Assets in 14d	p	1,491	781	Ordinary
1 - NOV-1-1	Opp Securities Australia Ltd	, and	410	20	Ordinary
NO4-14	Obsiderations Australia Lto	Buy	7,645	934	Ordinary
14Ov-14	UBS Securities Australia Ltd	Buy	8	11	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	221	27	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	889	88	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	3,168	387	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	7,923	896	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	1,096	134	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	2,348	287	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	4,213	515	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	4,502	920	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	10,641	1.308	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	121 500	15,000	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	2000	000,00	Ordinary.
7-Nov-14	URS Securities Australia 11d	8.18	C20,2	000	Ordinary
Nov. 14	I IDC Countries Australia 143	ion o	40,/2/	870'5	Ordinary
7 May 14	Opposition Services Australia Liu	Buy	551	89	Ordinary
NOV 14	UBS securities Australia Ltd	Buy	219,632	27,115	Ordinary
7-NOV-14	UBS Securities Australia Ltd	Buy	591	73	Ordinary
NOV-14	UBS Securities Australia Ltd	Buy	510	63	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	348	43	Ordinary
Nov-14	UBS Securities Australia Ltd	Sell	19,805	(2.445)	Ordinary
18-Nov-14	UBS Global Asset Management	Sell	2.114	(264)	Ordinary
Nov-14	UBS Securities Australia Ltd	Sell	9.882	(1.219)	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	3.200	400	Ordinary
Nov-14	UBS Securities Australia Ltd	Sell	108 194	(13 320)	Ordinan
Nov-14	UBS Securities Australia Ltd	Buv	481 097	50.955	Ordinan
Nov-14	UBS Securities Australia Itd	Şeili	15 808	(4.040)	Coloring
Nov-14	I IBS Securities Australia Itcl	200	1140	(848)	Ordinary
Nov-14	LIBS Socientias Australia Ind	130	1,148	(141)	Ordinary
1	DISPLANTAGE COO.	100	/06,031	(67, 18)	Ordinary
NOV-14	Obs securities Australia Ltd	Buy	194,578	24,000	Ordinary
8-NOV-14	UBS Securities Australia Ltd	Sel	55,202	(6,770)	Ordinary
NOV-14	UBS Securities Australia Ltd	Buy	277,288	34,480	Ordinary
NOV-14	UBS Securities Australia Ltd	Ser.	1,555	(190)	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	59,680	7,460	Ordinary
Nov-14	UBS AG Australia Branch	Buy	926,123	115,000	Ordinary
Nov-14	UBS AG Australia Branch	Buy	46,526	2,698	Ordinary
18-Nov-14	UBS AG Australia Branch	Sell	3,227,891	(400,000)	Ordinary
Nov-14	UBS AG Australia Branch	Buy	119,751	14,800	Ordinary
		Enter Into Derivative			
8-Nov-14	UBS Securities Australia Ltd	Contract	NVA	40,000,000	Options
		Enter into Derivative			
10 10 14					

1985 Securities Australia Ltd 1985 Securities Australia Ltd 1985 Securities Australia Ltd 1985 Securities Australia Ltd 1985 Securities Australia Ltd 1985 Securities Australia Ltd 1985 Geberal Australia Ltd 1985 Geberal Australia Ltd 1985 Securities Australia Ltd 1985 Securit	Sel	900 48 48 77 216 37,214 33,254 593 3,165 502,548 25,843	(101) (101) 6 9 9 9 (27) (4,646) 74	Ordinary Ordinary Ordinary Ordinary					
is ltd is ltd	80 y   80 y	48 72 216 37,214 37,214 593 593 502,649 502,649	(27) (27) (4,646)	O O di					
is tid is tid	Block   Bloc	216 216 37,214 37,214 593 3165 502,649 502,649 502,648	(27) (4,646) 74						
is itd la tid	Self	216 37,214 593 3,165 502,649 25,483	(27) (4,646) 74	ip o					
la tid la tid	Sel	37,214 593 3,165 502,649 25,483	(4,646)	je je					
in it it in it it in in it in in it in it in in it in in it in in in it in in in in in in in in in in in in in	See	37,214 593 3,165 502,649 25,483	(4,546)	o o					
in a lud  le Lid	Self	3,165 502,649 25,483	74						
in agentern. In agentern. In a lid In a	Self	502,649 502,649 25,483		Crain					
la Lid la Lid	See	25,483	(387)	Orgi					
la lid la lid la lid la lid la lid la lid mch mch mch mch mch mch mch mch	Sept   Sept	25,483	(61,882)	Ordin					
ia lid la lid la lid la lid la lid la lid la lid la lid la lid la lid ment Australia lid ment Australia lid ment Australia lid la lid	18   18   18   18   18   18   18   18		3,152	Ordin					
is litid le litid	Sel   Sel	760'0	(269)	Ordin					
is littld list lit	Se	135,826	16,601	Ordin					
le lid le lid le lid nch nch nch nch nch nch nch nch nch nch	See	135,932	(16,601)	Ordinary					
is it it it is it it is it it it is it it is it it is it it is it it is it it it is it is it	Self	28,282	(3,562)	Ordinary					
nch inch inch inch inch inch inch inch i	Seli	148,092	(18,400)	Ordinary					
north  The control of the control of	Blay Self Self Self Self Self Self Self Self	236,616	(29.563)	Ordinary					
rech merr Australia Ltd nerr Australia Ltd nerr Australia Ltd nerr Australia Ltd nerr Australia Ltd is Ltd	Sel	398.219	20 000	Ordinary					
nent Australia Ltd nent Australia Ltd lis Ltd la Ltd	Sel	402 557	(000,03)						
ment Australia Lid nent Australia Lid le Lid	Buy Buy Buy Buy Sell Sell Sell Sell Sell Sell Sell Sel	100,200	(37.00)	5 6					
rent Australia Ltd la Ltd	Buy Sell Sell Sell Sell Sell Sell Sell Sel	23 444	(0 1)	Ordinary					
is tid is tid	Sell Sell Sell Sell Buy	4 2 4 5	#C0'7	Organary					
is to is	Sell Sell Sell Sell Sell Sell Sell Sell	1,454	081	Ordinary					
is it is it	Sell Sell Sell Sell Sell Sell Sell Sell	8,120	(1,000)	Ordin					
is it it is it is	Sell Sell Sell Sell Sell Sell Sell Sell	1,470,600	(180,000)	Ordin					
is it d in tit d is it d is it d is it d is it d is it d is it d is it d is it d is it d is it d is it d is it d is it d is it d is it d is it d	Sell Sell Sell Sell Sell Call	311,220	(38,000)	Ordin					
is it of  is it of  is it of  is it of  is it of  is it of  is it of  is it of  is it of  is it of  is it of  is it of  is it of  is it of	Sell Sell Buy Sell Sell	96,557	(11,804)	Ordin					
in tid	Sell Sell Buy Sell Sell	450,464	(690'55)	Ordin					
is it id is id is it id is if id is it id is it id is it id is it id is it id is it id is it id is it id is it id is it id is it id is it id is it id is it if it is it id is	Sell Buy Sell	126.6	(1719)	Ordir					
a iid b iid b iid b iid b iid b iid b iid	Buy Sett	8 434	(1 001)	Carlo C					
ia tid la tid la tid la tid la tid la tid la tid la tid	18.0	40.088	(100.4)	5 6					
ia trd ia trd ia trd ia trd ia trd ia trd ia trd ia trd ia trd ia trd ia trd ia trd	Cell	101 057	100,001	5					
a rro la lrd la lrd la lrd la lrd la lrd		/87101	(17,452)	- Ja					
la ird la ird la ird la ird la ird	1100	5,522	(675)	Ö					
la Itd la Itd la Itd la Itd	Sell	40,197	(4,914)	Ordir					
is ttd is ttd is ttd is ttd	Sell	3,943	(482)	Ordir					
ia Ltd ia Ltd ia Ltd ia Ltd	Sell	626'5	(726)	Ordir					
ia Ltd ia Ltd ia Ltd	Sell	9.775	(1, 195)	ÖrÖ					
ia Ltd ia Ltd	Sell	6473	(1 152)	2					
ia Ltd	Sell	10 665	(20,0)	Ordinary					
200	9	0010	(1,000)	5 6					
	llac llac	8,180	(1,000)	Ordin Figure					
la Ltd	liac	1,592	(200)	Ordin					
ia Ltd	liec	40,325	(2,066)	Ordin					
ia Ltd	Sell	28,497	(3,580)	Ordin					
ia Ltd	Sell	6,933	(871)	Ordinary					
ia Ltd	Sell.	2,603	(327)	Ordinary					
ia Ltd	llas	21,253	(2,670)	Ordin					
ia Ltd	Sell	12,227	(1,536)	Ordinary					
ia Ltd	Sell	469,266	(58,953)	Ordinary					
ia Ltd	Sell	144,585	(18,164)	Ordinary					
ia Ltd	Sell	4.259	(535)	Ordinary					
ia Ltd	Sell	14.033	(1 763)	Ordinary					
a Itd	lle?	4 030	(50.3)	5					
ia Itd	la's	73.916	(600.0)	Ordinary					
ia Ito	9	100	(70°C)	ordinary.					
10	le)	200	(47)	Cicinaly					
pagement	9	0000	(E)	ordinary.					
7+1	NI B	000 001	(6/3)	Sign of the second					
	ing	02,840	1000,2	Ordinary					
מרום	ana .	33,5/4	3,957	Ordina					
a Ltd	Sell	333,438	(39,379)	Ordina					
ia Ltd	Sell	30,510	(3,617)	Ordinary					
ia Ltd	Sell	6,348	(05/)	Ordina					
a Ltd	Sell	122,811	(14,367)	Ordina					
a Ltd	Sell	156 188	(18 45m)	Ordin					
altd	la S	209.08	(10 EGE)	California					
72	lla)	200,200	(10,000)	5					
	Bro	121,001	(12,000)	Cidinary					
7	,,	201,621	14,507	5					
2010	Coll	493,507	28,028	Cogn					
a Ltd	ne l	2/9/6	(11,397)	Ordin					
a Ltd	ne.	315,945	(37,405)	Ordin					
a Ltd	Sell	254,047	(30,116)	Ordinary					
a Ltd	Sell	1,184,332	(139,825)	Ordin					
a Ltd	Sell	293,953	(34,957)	Ordin					
auto	Sell	560.456	(SEC 002)	Cipio					
201		904,000	(20,002)	Orgina					
a Ltd	Sett	4,383,346	(516,204)	Ordina					
-	a	136 960	00000						
ch	Buy	125,969	12,000	Ordinary					
	UBS Securities Australia tod UBS Securities Australia tod	a titch   Sell    a titch   Sell    b titch   Sell    b titch   Sell    a titch   Se		Sell   40,325     Sell   28,457     Sell   28,457     Sell   21,233     Sell   12,227     Sell   44,586     Sell   4,033     Sell   4,033     Sell   4,033     Sell   1,028     Sell   2,339     Sell   33,438     Sell   33,438     Sell   33,540     Sell   34,330     Sell   33,438     Sell   33,540     Sell   34,330     Sell   33,540      Ш	line countries Australia 11d	Sell	75		Ordinary
-----------	--	--	-----------	----------	----------				
l	ורובה את הוובה שת הוות בית								
١	UBS Securities Australia Ltd	Sell	41	(2)	Ordinary				
		Sell	73		Ordinary				
		Buy	1,040,000		Ordinary				
		Buy	159,800		Ordinary				
- 1		Buy	352,792		Ordinary				
-		Seil	28,067		Ordinary				
18-Nov-14		Buy	760'/		Ordinary				
ı		and a	18,081		Ordinary				
		, pag	19,473		Ordinary				
ı		Den	705,472		Ordinary				
		Call	30,209		Ordinary				
ı		Sell	3460		Ordinary				
1		lac lac	2,400		Ordinary				
		Sell	50,047		Ordinary				
ı		Sell Sell	505 9		Ordinary				
		Sell	213 586		Ordinary				
1 2		Buv	58.085		Ordinary				
19-Nov-14	Securities Australia IIId	Stock returned	AVN		Ordinary				
14	UBS AG. London Branch	Seli	726.643	(100 16)	Ordinary				
9-Nov-14		Buy	885,923		Ordinary				
9-Nov-14	AG Australia Branch	Sell	244,574		Ordinary				
9-Nov-14		Buy	211,012		Ordinary				
9-Nov-14	UBS Wealth Management Australia Ltd	Sell	18,511	(2,324)	Ordinary				
		Enter Into Derivative							
9-Nov-14		Contract	NA	160,000	Options				
1		Sell	99/	(96)	Ordinary				
		Soll	9)	(7)	Ordinary				
9-14Ov-14		Soil	1,50	(10)	Ordinary				
ı		Coll	21/72	(960)	Cidinaly				
ı		a de la companya de l	351,0	(000)	Ordinary				
9-Nov-14		Seli	3 160		Ordinary				
ı		Sell	3 160	(968)	Ordinary				
		Sell	3.695		Ordinary				
		Sell	4,197		Ordinary				
		Sell	3,990		Ordinary				
		Sell	5,410		Ordinary				
		Sell	589		Ordinary				
		Sell	305		Ordinary				
		Sell	24		Ordinary				
İ		Seli	4,309		Ordinary				
-		Sell	2,394		Ordinary				
1		Seil	3,160	1	Ordinary				
1	UBS Securities Australia Ltd	Seil	3,760	(396)	Ordinary				
1		Sell Coll	707		Ordinary				
		Nell Coll	4,309		Ordinary				
		Nell Nell	44/		Ordinary				
		100	3,431		Ordinary				
		Sell	712,85		Ordinary				
19-Nov-14		Sell	0/6,11		Ordinary				
9-NOV-14		Coll	207'C		Ordinary				
2 2		la de	8 3		Ordinary				
19-Nov-14		les	7/5	2	Ordinary				
14		Sell	3 152	(305)	Ordinary				
4	Securities Australia Itd	llas	20,10	(181)	Ordinary				
4	JUBS Securities Australia Ltd	Sell	120	(15)	Ordinary				
9-Nov-14	Securities Australia Ltd	Sell	503	(63)	Ordinary				
4		Sell	16	0	Ordinary				
4		Sell	10 159	(1 273)	Ordinary				
4		Sell	431	(54)	Ordinary				
19-Nov-14		Sell	5,195	(651)	Ordinary				
4		Sell	168	(21)	Ordinary				
9-Nov-14		Sell	3,224	(404)	Ordinary				
19-Nov-14		Sell	351	(44)	Ordinary				
4		Sell	56	(0)	Ordinary				
19-Nov-14		Sell	2,817	(353)	Ordinary				
4		Sell	4,708	(065)	Ordinary				
4	UBS Securities Australia Ltd	les.	869'5	(714)	Ordinary				
19-Nov-14	UBS Securities Australia Itd	Jeil College	1001		,				
ا	COO COCCUSION CONTRACTOR CONTRACT		5	ICO	Ordinary				

21-Nov-14	COS SECURIOR MINISTER		603:04	(765'6)	Ordinary
1-Nov-14	UBS Securities Australia Ltd	Sell	4.748	(554)	Ordinary
Nov-14	UBS Securities Australia Ltd	Sell	40,270	(4,699)	Ordinary
Nov-14	UBS Securities Australia Ltd	les	69	(8)	Ordinary
Nov-14	UBS Securities Australia Ltd	Sell	49,706	(2,800)	Ordinary
Nov-14	UBS Securities Australia Ltd	Sell	13,181	(1,538)	Ordinary
NOV-14	UBS Securities Australia Ltd	Sell	69	(8)	Ordinary
21 Nov-14	UBS Securities Australia Ltd	Nell Control	7,422	(998)	Ordinary
Nov. 14	Upo Securities Australia Ltd	hog	11/,186	13,674	Ordinary
Nov-14	I IRS Securities Australia 14d	1197	18/,67	(3,475)	Ordinary
Nov-14	UBS Securities Australia Ind	3	062,412	(25,000)	Ordinary
21-Nov-14	UBS Securities Australia Ltd	Sell	0/5/9	(000,1)	Crainary
Nov-14	UBS Securities Australia Ltd	Sell	12	(6)	Ordinary
21-Nov-14	UBS Securities Australia Ltd	Sell	13 155	(1 535)	Craining
21-Nov-14	UBS Securities Australia Ltd	Sell	69	(8)	Ordinary
Nov-14	UBS Securities Australia Ltd	Sell	2.014	(232)	Ordinary
Nov-14	UBS Securities Australia Ltd	Sell	112 593	(13 138)	Gerigo
Nov-14	UBS Securities Australia Ltd	Self	223	(00)	Ordinary
21-Nov-14	UBS Securities Australia Ltd	Sell	21 091	(12) (2)	Ordinary
Nov-14	UBS Securities Australia Ltd	Sell	160 945	(18 780)	Ordinary
1-Nov-14	UBS Securities Australia Ltd	Sell	10.832	(1.764)	Creinary
1-Nov-14	UBS Securities Australia Ltd	Sell	3 719	(027)	Ordinary
21-Nov-14	UBS Securities Australia Ltd	Sell	98 744	(11 522)	Contrary.
21-Nov-14	UBS Securities Australia Ltd	Sell	180	(22)	Coloniary
1-Nov-14	UBS Securities Australia Ltd	Se	309 480	(25, 11.2)	Ordinary
1-Nov-14	UBS Securities Australia Ltd	Sell	473.415	(50,12)	ordinary.
1-Nov-14	UBS Securities Australia Ltd	Buy	1 155	138	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	5,533	567	in in in
1-Nov-14	UBS Securities Australia Ltd	Sef	14 544	(1 711)	Cremon
Nov-14	UBS Securities Australia Ltd	Seil	86.496	(3/1/01)	Carling.
Nov-14	UBS Securities Australia Ltd	Sell	86.496	(10.176)	Compa
Nov-14	UBS Securities Australia Ltd	Sell	74.7	(0.77)	Cardinary
Nov-14	UBS Securities Australia Ltd	Sell	2 890	(340)	Ordinan
Nov-14	UBS Securities Australia Ltd	Sell	3.817	(0.0)	Ordinary
1-Nov-14	UBS Securities Australia Ltd	Sell	30,618	(3 600)	Ordinary
Nov-14	UBS Securities Australia Ltd	Sell	1275	(150)	Ordinary
Nov-14	UBS Securities Australia Ltd	Set	5.015	(250)	Ordinary
Nov-14	UBS Securities Australia Ltd	Sell	3.477	(409)	Ordinary
Nov-14	UBS Securities Australia Ltd	Sell	3.944	(464)	Ordinary
Nov-14	UBS Securities Australia Ltd	Sell	94,312	(11,089)	Ordinary
1-Nov-14	UBS Securities Australia Ltd	Sell	17,401	(2,046)	Ordinary
Nov-14	UBS Securities Australia Ltd	Sell	2,118	(249)	Ordinary
Nov-14	UBS Securities Australia Ltd	Sell	5,389	(634)	Ordinary
Nov-14	UBS Securities Australia Ltd	Zet.	6,588	(2775)	Ordinary
Nov-14	UBS Securities Australia Ltd	Sell	3.460	(402)	Ordinary
Nov-14	UBS Securities Australia Ltd	Sell	5.508	(648)	Ordinary
21-Nov-14	UBS Securities Australia Ltd	Sell	42.525	(5,000)	Ordinary
21-Nov-14	UBS Securities Australia Ltd	Sell	278	(101)	Control
Nov-14	UBS Securities Australia Ltd	195	0.350	(101)	Colliary
21-Nov-14	UBS Securities Australia Itd	Sell.	000 5	(101,1)	Ordinary
Nov-14	UBS Securities Australia Itd	les.	5,000	((64)	Cignaly
21-Nov-14	LIPS Securities Australia 14d	Coff	210,0	(199)	Ordinary
1-Nov-14	LIRS Securities Australia Itd	Sell Sell	1110,1	(1/8)	Crainary
Nov-14	LIRS Securities Australia 11d	la d	9,118	(603)	Ordinary
Vov.14	IIBS Securities Australia 14d	le?	2,139	(707)	Ordinary
14-VOV	IIRS Securities Australia Itd	la y	2,545	(00/)	Ordinary
21-Nov-14	URS Securities Australia Itd		102	(71)	Ordinary
1-Now-14	LIPS Securities Australia 144	100	4,884	(9/5)	Ordinary
21 NOW 14	I IDS Countries Australia I+A	100	10	(q)	Ordinary
21 1/201-14	Obs securines Absulate tild	i ac	916	(108)	Ordinary
21 Nov. 14	LIDS Countries Australia Ltd	nac o	416	(49)	Ordinary
21 Nov. 14	Opp Securities Australia Ltd	iac i	12,398	(1,462)	Ordinary
21 Nov 14	UBS Securities Australia Ltd	11-0	9,226	(1,088)	Ordinary
1 Nov. 14	Tipe committee Abstratia Ltd	1100	9,913	(1,169)	Ordinary
100-14	Obs securities Australia Ltd	Sell	17,572	(2,071)	Ordinary
21-NOW-14	UBS Securities Australia Ltd	>ell	260	(99)	Ordinary
101.14	Obs securities Australia Ltu	11-00	687	(34)	Ordinary
10.14	Obs securities Australia Ltd	liac liac	2,190	(258)	Ordinary
21-NOV-14	UBS Securities Australia Ltd	i e	11,948	(1,409)	Ordinary
- No. 14	Obs Securities Australia Ltd	Jen Jen Jen Jen Jen Jen Jen Jen Jen Jen	9,845	(1.161)	Ordinary
-NOV-14	UBS Securities Australia Ltd	9			
			670	(62)	Ordinary
Z1-Nov-14	UBS Securities Australia Ltd	Jes Jes	670	(62)	Ordinary

Securities Aristralia 140

ESS Scentifier Australia Ltd   SMI   CR02	24-Nov-14 24-Nov-14 24-Nov-14	IIRS Securities Australia 11d		225	(000,1)	55
185   Scientific Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Self   Securitie Authorish Ltd   Self   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie Authorish Ltd   Self   Securitie	24-Nov-14 24-Nov-14 24-Nov-14	and occurred control and	Seil	8,875	(610,1)	Ordinary
(b) Securitar Australa LEG         \$41         \$5.50         (7.99)           (b) Securitar Australa LEG         \$41         \$5.50         (7.99)           (b) Securitar Australa LEG         \$41         \$6.50         (7.99)           (b) Securitar Australa LEG         \$41         \$6.50         (7.99)           (b) Securitar Australa LEG         \$41         \$6.50         \$6.50           (c) Securitar Australa LEG         \$41         \$6.50         \$6.50           (c) Securitar Australa LEG         \$41         \$6.50         \$6.50           (c) Securitar Australa LEG         \$42         \$6.50         \$6.50           (c) Securitar Australa LEG         \$42         \$6.50         \$6.47           (c) Securitar Australa LEG         \$42         \$6.50         \$6.50           (c) Securitar Australa LEG         \$42         \$6.50         \$6.50           (c) Securitar Australa LEG         \$42         \$6.50         \$6.50           (c) Securitar Australa LEG         \$42         \$6.5	24-Nov-14 24-Nov-14	UBS Securities Australia Ltd	Sell	232,855	(26,765)	Ordinan
(b) S scorring Apprials (14)         SMI         (5) 16         (70)           (b) S scorring Apprials (14)         SMI         (70)         (70)           (b) S scorring Abstralia (14)         SMI         (70)         (70)           (b) S scorring Abstralia (14)         SMI         (70)         (70)           (b) S scorring Abstralia (14)         SMI         (70)         (70)           (b) S scorring Abstralia (14)         SMI         (70)         (70)           (b) S scorring Abstralia (14)         SMI         (70)         (70)           (c) S scorring Abstralia (14)         SMI         SMI         (71)           (c) S scorring Abstralia (14)         SMI <t< td=""><td>24-Nov-14</td><td>UBS Securities Australia Ltd</td><td>Sell</td><td>5,203</td><td>(865)</td><td>Ordinan</td></t<>	24-Nov-14	UBS Securities Australia Ltd	Sell	5,203	(865)	Ordinan
65 Securiter Autmin   145   541   104   105		UBS Securities Australia Ltd	Sell	6,516	(749)	Ordinan
(b) Streamer Australia Ltd         Sall         47,928         (5,598)           (b) Streamer Australia Ltd         Sall         7,928         (7,09)           (b) Streamer Australia Ltd         Sall         7,928         (7,07)           (b) Streamer Australia Ltd         Sall         6,074         (7,07)           (c) Streamer Australia Ltd         Sall         6,074         (7,07)           (c) Streamer Australia Ltd         Sall         6,074         (7,07)           (c) Streamer Australia Ltd         Sall         6,074         (7,07)           (c) Streamer Australia Ltd         Sall         6,074         (7,07)           (c) Streamer Australia Ltd         Sall         6,074         (7,07)           (c) Streamer Australia Ltd         Sall         6,074         (7,07)           (c) Streamer Australia Ltd         Sall         1,075         (1,07)           (c) Streamer Australia Ltd         <	24-Nov-14	UBS Securities Australia Ltd	Sell	104	(12)	Ordinary
USS Securine Automists 104   SM   SM   SM   SM   SM   SM   SM   S	1	UBS Securities Australia Ltd	Je J	47,928	(2,509)	Ordinary
Color Security Automitate to 15 Per 15   15   15   15   15   15   15   15	1	UBS Securities Australia Ltd		87	(10)	Ordinary
Bits Securine Australia Lists   Securine Austr		UBS Securities Australia Ltd	Sell Coll	215,542	(24,775)	Ordinary
Bits Securine Authorish total   Set   195, 200, 200, 201, 201, 201, 201, 201, 201	ı	LIPS Securities Australia Ind	1 3 S	510,59	(7,473)	Ordinary
BIS Securities Australia Lind   SM   SM   SM   SM   SM   SM   CM   CM	ı	URS Securities Australia Itol	Tel.	/sn/87	(3,225)	Ordinary
163 Securities Australia Ltd   541   542   543		URS Securities Australia 11d	اق	R	0.0	Ordinary
USS Securines Autorials (14)   Seal   Color		UBS Securities Australia Itd	les.	66	(4.1)	Ordinary
USS Securines Australia Ltd   Sell   Color	l	UBS Securities Australia Ltd	Sell	R &	(10)	Ordinary
USS Securities Autoriale Ltd   Self   Color	l	UBS Securities Australia Ltd	Sell	44	(6)	Ordinary
URS Securities Autorials Ltd         Sint         7.359         (67.04)           URS Securities Autorials Ltd         Sell         5.547         (5.78)           URS Securities Autorials Ltd         Sell         2.542         (5.78)           URS Securities Autorials Ltd         Sell         2.542         (5.78)           URS Securities Autorials Ltd         Sell         2.657         (6.78)           URS Securities Autorials Ltd         Sell         2.67         (7.06)           URS Securities Autorials Ltd         Sell         2.61         (7.00)           URS Securities Autorials Ltd         Sell         2.61         (7.00)           URS Securities Autorials Ltd         Sell         2.61         (7.00)           URS Securities Autoria		UBS Securities Australia Ltd	Sell	40.255	(4 677)	Ordinary
UBS Securities Australia Ltd         Sell         6,524         (5,246)           UBS Securities Australia Ltd         Sell         4,524         (5,28)           UBS Securities Australia Ltd         Sell         2,242         (5,58)           UBS Securities Australia Ltd         Sell         2,247         (1,087)           UBS Securities Australia Ltd         Sell         2,471         (1,087)           UBS Securities Australia Ltd         Sell         3,741         (4,081)           UBS Securities Australia Ltd         Sell         3,741         (4,082)           UBS Securities Australia Ltd         Sell         3,741         (4,083)           UBS Securities Australia Ltd         Sell         3,741         (4,083)           UBS Securities Australia Ltd         Sell         1,375         (1,082)           UBS Securities Australia Ltd         Sell         1,326         (1,083)           UBS Securities Australia Ltd         Sell         1,326         (1,019)           UBS Securities Australia Ltd         Sell         1,326         (1,019)           UBS Securities Australia Ltd         Sell         1,326         (1,010)           UBS Securities Australia Ltd         Sell         2,520         (1,020)		UBS Securities Australia Ltd	Sell	7,369	(847)	Ordinary
UBS Securine Australia Lind         Sell         4 554         C528           UBS Securine Australia Lind         Sell         22 420         C528           UBS Securine Australia Lind         Sell         2,422         C108           UBS Securine Australia Lind         Sell         2,422         C1089           UBS Securine Australia Lind         Sell         2,422         C1089           UBS Securine Australia Lind         Sell         2,422         C1089           UBS Securine Australia Lind         Sell         2,421         C1089           UBS Securine Australia Lind         Sell         2,421         C1089           UBS Securine Australia Lind         Sell         2,551         C1089           UBS Securine Australia Lind         Sell         2,561         C1089           UBS Securine Australia Lind         Sell         2,562         C1089           UBS Securine Australia Lind         Sell         2,661         C1089           UBS Securine Australia Lind	24-Nov-14	UBS Securities Australia Ltd	Sell	15,173	(1,744)	Ordinary
URS Securities Australia Ltd         Sell         4,62,429         0,52,429         0,52,629           URS Securities Australia Ltd         Sell         4,627         (1,687)           URS Securities Australia Ltd         Sell         2,421         (1,082)           URS Securities Australia Ltd         Sell         2,421         (1,082)           URS Securities Australia Ltd         Sell         2,421         (1,082)           URS Securities Australia Ltd         Sell         2,510         (1,082)           URS Securities Australia Ltd         Sell         1,375         (1,082)           URS Securities Australia Ltd         Sell         1,375         (1,078)           URS Securities Australia Ltd         Sell         1,375         (1,078)           URS Securities Australia Ltd         Sell         2,510         (1,078)	24-Nov-14	UBS Securities Australia Ltd	Setl	4,594	(228)	Ordinary
UBS Securities Australia Ltd         Sell         9,457         (1,087)           UBS Securities Australia Ltd         Sell         9,427         (1,087)           UBS Securities Australia Ltd         Sell         9,427         (1,087)           UBS Securities Australia Ltd         Sell         9,427         (1,088)           UBS Securities Australia Ltd         Sell         9,427         (1,088)           UBS Securities Australia Ltd         Sell         1,340         (1,088)           UBS Securities Australia Ltd         Sell         1,340         (1,088)           UBS Securities Australia Ltd         Sell         1,340         (1,088)           UBS Securities Australia Ltd         Sell         1,540         (1,011)           UBS Securities Australia Ltd         Sell         2,510         (1,011) <t< td=""><td>ı</td><td>UBS Securities Australia Ltd</td><td>Sell</td><td>22,429</td><td>(2,578)</td><td>Ordinary</td></t<>	ı	UBS Securities Australia Ltd	Sell	22,429	(2,578)	Ordinary
URS Securities Australia Ltd         Sell         9,457         (1,080)           URS Securities Australia Ltd         Sell         7,021         (1,080)           URS Securities Australia Ltd         Sell         2,402         (1,080)           URS Securities Australia Ltd         Sell         2,401         (7,290)           URS Securities Australia Ltd         Sell         2,401         (7,290)           URS Securities Australia Ltd         Sell         2,601         (7,090)           URS Securities Australia Ltd         Sell         1,340         (1,010)           URS Securities Australia Ltd         Sell         1,344         (1,020)           URS Securities Australia Ltd         Sell         1,344         (1,010)           URS Securities Australia Ltd         Sell         1,552         (1,010)           URS Securities Australia Ltd         Sell         1,552         (1,010)           URS Securities Australia Ltd         Sell         1,552         (1,010)           URS Securities Australia Ltd         Sell         1,562         (1,010)           URS Securities Australia Ltd         Sell         2,601         (1,010)           URS Securities Australia Ltd         Sell         2,601         (1,010) <t< td=""><td>1</td><td>UBS Securities Australia (td</td><td>Sell</td><td>4,863</td><td>(529)</td><td>Ordinary</td></t<>	1	UBS Securities Australia (td	Sell	4,863	(529)	Ordinary
USS Securities Australia Ltd         Sell         9,422         (1,083)           UBS Securities Australia Ltd         Sell         2,401         (7,082)           UBS Securities Australia Ltd         Sell         9,205         (1,083)           UBS Securities Australia Ltd         Sell         9,205         (1,083)           UBS Securities Australia Ltd         Sell         1,375         (1,081)           UBS Securities Australia Ltd         Sell         1,375         (1,082)           UBS Securities Australia Ltd         Sell         1,375         (1,011)           UBS Securities Australia Ltd         Sell         1,520         (1,011)           UBS Securities Australia Ltd         Sell         1,629         (1,011)           UBS Securities Australia Ltd         Sell         2,610         (1,011)           UBS Securities Australia Ltd         Sell         2,601         (1,011) <t< td=""><td></td><td>USS Securities Australia Ltd</td><td>Sell Sell</td><td>9,457</td><td>(1,087)</td><td>Ordinary</td></t<>		USS Securities Australia Ltd	Sell Sell	9,457	(1,087)	Ordinary
USS Securities Australia Ltd         Sell         2,400         (1,058)           USS Securities Australia Ltd         Sell         2,400         (1,058)           USS Securities Australia Ltd         Sell         2,400         (1,058)           USS Securities Australia Ltd         Sell         1,374         (1,058)           USS Securities Australia Ltd         Sell         1,390         (1,058)           USS Securities Australia Ltd         Sell         1,390         (1,058)           USS Securities Australia Ltd         Sell         1,390         (1,011)           USS Securities Australia Ltd         Sell         1,509         (1,011)           USS Securities Australia Ltd         Sell         2,610         (1,011)           USS Securities Australia Ltd         Sell         2,610         (1,011)           USS Securities Australia Ltd         Sell         2,610         (1,020)           USS Securities Australia Ltd         Sell         2,709         (1,020) <t< td=""><td></td><td>UBS Securities Australia Itd</td><td>Sell</td><td>1707/</td><td>(807)</td><td>Ordinary</td></t<>		UBS Securities Australia Itd	Sell	1707/	(807)	Ordinary
USS Securities Australia Ltd   Sell   2,250 (1,058)   USS Securities Australia Ltd   Sell   3,741 (430)   (158)   (1	ı	UBS Securities Australia Ltd	Sell	2,442	(1,083)	Orginary
UBS Securities Autarials Ltd         Sell         2,510         (500)           USS Securities Autarials Ltd         Sell         1,340         (154)           USS Securities Autarials Ltd         Sell         1,340         (154)           USS Securities Autarials Ltd         Sell         1,340         (154)           USS Securities Autarials Ltd         Sell         1,509         (1,610)           USS Securities Autarials Ltd         Sell         1,609         (1,611)           USS Securities Autarials Ltd         Sell         2,610         (1,630)           USS Se		UBS Securities Australia Ltd	Sell	9 205	1058	O O
UBS Securities Autarials Ltd         Sell         1,375         (158)           UBS Securities Autarials Ltd         Sell         1,375         (1,526)           UBS Securities Autarials Ltd         Sell         7,552         (1686)           UBS Securities Autarials Ltd         Sell         7,552         (1686)           UBS Securities Autarials Ltd         Sell         7,564         (1,011)           UBS Securities Autarials Ltd         Sell         2,510         (1,026)           UBS Securities Autarials Ltd         Sell         2,510         (1,011)           UBS Securities Autarials Ltd         Sell         2,510         (1,026)           UBS Securities Autarials Ltd         Sell         2,510         (1,026)           UBS Securities Autarials Ltd         Sell         2,510         (1,026)           UB	П	UBS Securities Australia Ltd	Sell	2,610	(008)	Ordinary
UBS Securities Australia Ltd         Sell         1375         (156)           UBS Securities Australia Ltd         Sell         1,240         (156)           UBS Securities Australia Ltd         Sell         7,552         (168)           UBS Securities Australia Ltd         Sell         9,013         (1,60)           UBS Securities Australia Ltd         Sell         1,609         (1,61)           UBS Securities Australia Ltd         Sell         2,610         (1,60)           UBS Securities Australia Ltd         Sell         2,610         (1,07)           UBS Securities Australia Ltd         Sell         2,610         (1,07)           UBS Securities Australia Ltd         Sell         1,792         (1,07)           UBS Securities Australia Ltd         Sell         1,772         (1,07)           UBS Securities Australia Ltd         Sell         1,772         (1,07)           UBS Securities Au	24-Nov-14	UBS Securities Australia Ltd	Sell	3,741	(430)	Ordinan
UBS Securities Australia Ltd         Sell         1,340         (1,66           UBS Securities Australia Ltd         Sell         1,340         (1,66           UBS Securities Australia Ltd         Sell         1,572         (1,66           UBS Securities Australia Ltd         Sell         2,610         (1,010           UBS Securities Australia Ltd         Sell         2,610         (1,000           UBS Securities Australia Ltd         Sell         2,709         (1,000           UBS Securities Australia Ltd         Sell         2,709         (1,000           UBS Securities A	24-Nov-14	UBS Securities Australia Ltd	Sell	1,375	(158)	Ordinary
UBS Securities Australia Ltd         Sell         7,552         (1,056)           UBS Securities Australia Ltd         Sell         7,552         (8686)           UBS Securities Australia Ltd         Sell         1,444         (1,056)           UBS Securities Australia Ltd         Sell         1,609         (1,011)           UBS Securities Australia Ltd         Sell         2,610         (1,011)           UBS Securities Australia Ltd         Sell         2,610         (300)           UBS Securities Australia Ltd         Sell         2,600         (1,075)           UBS Securities Australia Ltd         Sell         1,732         (1,075)           UBS Securities Australia Ltd         Sell         1,732         (1,075)           UBS Securities Australia Ltd         Sell         1,732         (1,075)           UBS Securit	24-Nov-14	UBS Securities Australia Ltd	Sell	1,340	(154)	Ordinary
UBS Securities Australia Ltd         Sell         1,592         (166)           UBS Securities Australia Ltd         Sell         1,603         (160)           UBS Securities Australia Ltd         Sell         1,603         (1,601)           UBS Securities Australia Ltd         Sell         2,610         (1,601)           UBS Securities Australia Ltd         Sell         2,610         (1,601)           UBS Securities Australia Ltd         Sell         2,610         (300)           UBS Securities Australia Ltd         Sell         2,610         (430)           UBS Securities Australia Ltd         Sell         1,792         (1,073)           UBS Securities Australia Ltd         Sell         1,792         (1,073)           UBS Securities Australia Ltd         Sell         1,792         (2,000)           UBS Securities Australia Ltd         Sell         1,792         (2,000)           UBS Securities Australia Ltd         Sell         1,792         (2,001)           UBS Securiti	24 NOV- 14	UBS Securities Australia Ltd	yell Sell	139	(16)	Ordinary
UBS Securities Australia Ltd         Sell         1,444         (1,010)           UBS Securities Australia Ltd         Sell         8,796         (1,011)           UBS Securities Australia Ltd         Sell         1,690         (1,011)           UBS Securities Australia Ltd         Sell         2,610         (1,010)           UBS Securities Australia Ltd         Sell         2,610         (1,000)           UBS Securities Australia Ltd         Sell         2,097         (2,31)           UBS Securities Australia Ltd         Sell         2,007         (2,30)           UBS Securities Australia Ltd         Sell         2,007         (1,000)           UBS Securities Australia Ltd         Sell         2,807         (1,000)           UBS Securities Australia Ltd         Sell         2,807         (1,000)           UBS Securities Australia Ltd         Sell         6,53         (1,000)           UBS Securities Australia Ltd         Sell         6,53         (1,000)           UBS Securities Australia Ltd         Sell         1,543         (1,800)           UBS Securities Australia Ltd         Sell         2,264         (1,800)           UBS Securities Australia Ltd         Sell         2,264         (1,800)           UB	24-Nov-14	LIBS Securities Australia 14d	Sell	266,1	(898)	Ordinary
URS Securities Australia Ltd         Sell         1,100           URS Securities Australia Ltd         Sell         1,609         (1,010           URS Securities Australia Ltd         Sell         2,610         (300           URS Securities Australia Ltd         Sell         2,600         (1,000           URS Securities Australia Ltd         Sell         3,55         (1,075)           URS Securities Australia Ltd         Sell         3,55         (1,075)           URS Securities Australia Ltd         Sell         1,732         (1,075)           URS Securities Australia Ltd         Sell         1,732         (1,075)           URS Securities Australia Ltd         Sell         1,742         (2,02)           URS Securities Australia Ltd         Sel	24-Nov-14	UBS Securities Australia 11d	13	1,444	(450,1)	Ordinary
UBS Securities Australia Ltd         Sell         1,073         1,101           UBS Securities Australia Ltd         Sell         2,610         (300)           UBS Securities Australia Ltd         Sell         2,610         (300)           UBS Securities Australia Ltd         Sell         2,697         (241)           UBS Securities Australia Ltd         Sell         2,610         (300)           UBS Securities Australia Ltd         Sell         1,729         (1,000)           UBS Securities Australia Ltd         Sell         1,729         (1,000)           UBS Securities Australia Ltd         Sell         1,739         (1,000)           UBS Securities Australia Ltd         Sell         1,739         (1,000)           UBS Securities Australia Ltd         Sell         1,739         (1,000)           UBS Securities Australia Ltd         Sell         2,709         (1,000)           UBS Securities Australia Ltd         Sell         3,709         (1,500)           UBS Securities		UBS Securities Australia Itd	Sell	207.0	(199)	Ordinary
UBS Securities Australia Ltd         Sell         2,610         (300)           UBS Securities Australia Ltd         Sell         2,610         (300)           UBS Securities Australia Ltd         Sell         2,097         (241)           UBS Securities Australia Ltd         Sell         2,097         (241)           UBS Securities Australia Ltd         Sell         2,610         (33)           UBS Securities Australia Ltd         Sell         2,610         (33)           UBS Securities Australia Ltd         Sell         2,807         (1,000)           UBS Securities Australia Ltd         Sell         6,35         (1,000)           UBS Securities Australia Ltd         Sell         6,35         (1,000)           UBS Securities Australia Ltd         Sell         1,782         (2,06)           UBS Securities Australia Ltd         Sell         2,709         (1,800)           UBS Securities Australia Ltd         Sell         2,709         (1,800)           UBS Securities Australia Ltd         Sell         2,264         (1,800)           UBS Securities Australia Ltd         Sell         2,264         (1,800)           UBS Securities Australia Ltd         Sell         3,764         (1,500)           UBS Securities		IUBS Securities Australia Ltd	Sell	1 609	(1011)	Ordinary
UBS Securities Australia (1d)         Sell         2,510         C000           UBS Securities Australia (1d)         Sell         2,610         (2001)           UBS Securities Australia (1d)         Sell         374         (300)           UBS Securities Australia (1d)         Sell         2,610         (300)           UBS Securities Australia (1d)         Sell         8,700         (1,075)           UBS Securities Australia (1d)         Sell         8,700         (1,075)           UBS Securities Australia (1d)         Sell         8,700         (1,075)           UBS Securities Australia (1d)         Sell         1,792         (1,075)           UBS Securities Australia (1d)         Sell         1,742         (2,087)           UBS Securities Australia (1d)         Sell         1,742         (2,097)           UBS Securities Australia (1d)         Sell         1,742         (2,097)           UBS Securities Australia (1d)         Sell         1,742         (2,150)		UBS Securities Australia Ltd	Sell	0.000	(100)	Crainary
UBS Securities Australia (14d)         Sell         2,097         (241)           UBS Securities Australia (14d)         Sell         3,741         (404)           UBS Securities Australia (14d)         Sell         3,741         (409)           UBS Securities Australia (14d)         Sell         2,610         (300)           UBS Securities Australia (14d)         Sell         8,700         (1,000)           UBS Securities Australia (14d)         Sell         8,700         (1,000)           UBS Securities Australia (14d)         Sell         1,792         (1,000)           UBS Securities Australia (14d)         Sell         1,792         (1,000)           UBS Genuties Australia (14d)         Sell         1,792         (1,000)           UBS Securities Australia (14d)         Sell         1,792         (1,000)           UBS Securities Australia (14d)         Sell         1,732         (1,000)           UBS Securities Australia (14d)         Sell         1,733         (1,000)           UBS Securities Australia (14d)         Sell         1,733         (1,000)           UBS Securities Australia (14d)         Sell         1,733         (2,100)           UBS Securities Australia (14d)         Sell         1,743         (1,000)		UBS Securities Australia Ltd	Sell	2,610	(300)	Ordinan
UBS Securities Australia (1d)         Sell         7,604         (874)           UBS Securities Australia (1d)         Sell         3741         (430)           UBS Securities Australia (1d)         Sell         2.610         (33)           UBS Securities Australia (1d)         Sell         2.610         (33)           UBS Securities Australia (1d)         Sell         8.700         (1,000)           UBS Securities Australia (1d)         Sell         63.53         (1,075)           UBS Securities Australia (1d)         Sell         63.53         (1,075)           UBS Securities Australia (1d)         Sell         7.702         (1,075)           UBS Securities Australia (1d)         Sell         2.703         (1,075)           UBS Securities Australia (1d)         Sell         2.743         (1,075)           UBS Securities Australia (1d)         Sell         1.574         (1,075)           UBS Securities Australia (1d)         Sell         1.574         (2,08)           UBS Securities Australia (1d)         Sell         1.574         (2,187)           UBS Securities Australia (1d)         Sell         1.542         (2,187)           UBS Securities Australia (1d)         Sell         1.394         (5,156)		UBS Securities Australia Ltd	Sell	2,032	(200)	Ordinary
UBS Securities Australia (EU)         Sell         3741         (430)           UBS Securities Australia (EU)         Sell         2,510         (23)           UBS Securities Australia (EU)         Sell         2,610         (300)           UBS Securities Australia (EU)         Sell         8,700         (1,075)           UBS Securities Australia (EU)         Sell         8,700         (1,075)           UBS Securities Australia (EU)         Sell         1,792         (2,050)           UBS Securities Australia (EU)         Sell         1,792         (2,050)           UBS Gobbal Asset Management         Buy         1,774         (3,050)           UBS Gobbal Asset Management         Buy         2,709 (1,050)         (3,050)           UBS Gobbal Asset Management         Buy         2,709 (1,050)         (3,050)           UBS Gobbal Asset Management         Buy         2,709 (1,070)         (3,050)           UBS Securities Australia (EU)         Sell         1,742         (3,050)		UBS Securities Australia Ltd	Sell	7,604	(874)	Ordinary
UBS Securities Australia Ltd         Sell         2,510         (32)           UBS Securities Australia Ltd         Sell         2,510         (300)           UBS Securities Australia Ltd         Sell         2,500         (1,000)           UBS Securities Australia Ltd         Sell         9,353         (1,000)           UBS Securities Australia Ltd         Sell         1,792         (2,00)           UBS Securities Australia Ltd         Buy         1,179         1,507           UBS Geouties Australia Ltd         Sell         2,709         (1,000)           UBS Securities Australia Ltd         Sell         2,709         (1,000)           UBS Securities Australia Ltd         Sell         2,709         (1,800)           UBS Securities Australia Ltd         Buy         1,733         (1,800)           UBS Securities Australia Ltd         Buy         2,254         2,709           UBS Securities Australia Ltd         Buy         2,254         2,709           UBS Securities Australia Ltd         Buy         2,245         2,709           UBS Securities Australia Ltd         Buy         2,254         2,709           UBS Securities Australia Ltd         Sell         1,861,589         2,71,809           UBS Securities Au		UBS Securities Australia Ltd	Sell	3.741	(057)	Ordinary
UBS Securities Australia Ltd         Sell         2.610         (300)           UBS Securities Australia Ltd         Sell         2.801         (1,020)           UBS Securities Australia Ltd         Sell         8.700         (1,000)           UBS Securities Australia Ltd         Sell         63.53         (1,075)           UBS Securities Australia Ltd         Buy         1,678         (1,075)           UBS Securities Australia Ltd         Buy         1,678         (1,075)           UBS Securities Australia Ltd         Sell         27,031         (1,820)           UBS Securities Australia Ltd         Sell         27,032         (1,820)           UBS Securities Australia Ltd         Sell         27,633         (1,820)           UBS Securities Australia Ltd         Sell         27,633         (1,820)           UBS Securities Australia Ltd         Buy         23,449         27,032           UBS Securities Australia Ltd         Buy         23,449         27,032           UBS Securities Australia Ltd         Buy         23,449         27,032           UBS Securities Australia Ltd         Buy         23,459         27,432           UBS Securities Australia Ltd         Buy         1,837         27,432           UBS		UBS Securities Australia Ltd	Sell	200	(23)	Ordinary
UBS Securities Australia titot         Seli         8.07         (1,079)           UBS Securities Australia titot         Seli         8.700         (1,079)           UBS Securities Australia titot         Seli         65.3         (1,079)           UBS Securities Australia titot         Seli         1,792         (2,000)           UBS Securities Australia titot         Buy         1,779         (1,079)           UBS Securities Australia titot         Buy         33,075         3,873           UBS Securities Australia titot         Seli         1,742         (1,082)           UBS Securities Australia titot         Seli         1,742         (1,082)           UBS Securities Australia titot         Seli         1,742         (1,082)           UBS Securities Australia titot         Buy         2,314         (1,182)           UBS Securities Australia titot         Seli         1,245         (2,103)           UBS Securities Australia titot         Buy         2,244         2,304           UBS Securities Australia titot         Seli         1,867         (1,150)           UBS Securities Australia titot         Buy         1,867         (1,150)           UBS Securities Australia titot         Seli         2,971         (1,150)		UBS Securities Australia Ltd	Sell	2,610	(300)	Ordinary
UBS Securities Australia tot         Sell         8700         (1,000)           UBS Securities Australia tot         Sell         8700         (1,000)           UBS Securities Australia tot         Sell         1,927         (75)           UBS Securities Australia tot         Buy         1,179         13,621           UBS Securities Australia tot         Buy         1,177         13,621           UBS Securities Australia tot         Sell         27,031         (1,820)           UBS Securities Australia tot         Sell         27,031         (1,820)           UBS Securities Australia tot         Buy         23,075         3,873           UBS Securities Australia tot         Buy         23,764         (1,820)           UBS Securities Australia tot         Buy         23,764         (2,180)           UBS Securities Australia tot         Buy         23,264         2,871           UBS Securities Australia tot         Buy         23,264         2,871           UBS Securities Australia tot         Buy         334,799         35,000           UBS Securities Australia tot         Sell         1,861,889         27,743           UBS Securities Australia tot         Sell         2,271         (1,5,756)           UBS Se	١	UBS Securities Australia Ltd	Sell	2,801	(322)	Ordinary
URS Securities Australia tod         Sell         (1,075)           URS Securities Australia tod         Sell         (1,075)           URS Securities Australia tod         Buy         1,078         (1,075)           URS Securities Australia tod         Buy         1,078         (1,075)           URS Securities Australia tod         Sell         27,037         (1,820)           URS Securities Australia tod         Sell         27,037         (1,820)           URS Securities Australia tod         Sell         175,433         (1,820)           URS Securities Australia tod         Sell         27,037         (3,189)           URS Securities Australia tod         Buy         24,246         28,21           URS Securities Australia tod         Buy         24,246         28,11           URS Securities Australia tod         Buy         2,345         27,422           URS Securities Australia tod         Buy         1,837,764         (15,50)           URS Securities Australia tod         Sell         33,764         (15,50)           URS Securities Australia tod         Sell         3,376         (15,50)           URS Securities Australia tod         Sell         3,376         (15,50)           URS Securities Australia tod         <	1	UBS Securities Australia Ltd	Sell	8,700	(1,000)	Ordinary
URS Securities Australia Ltd         Self         (55)         (75)           URS Securities Australia Ltd         Buy         1,792         (206)           URS Golds Asset Management         Buy         13,779         (3,692)           URS Securities Australia Ltd         Buy         33,075         3,873           URS Securities Australia Ltd         Self         27,091         (3,169)           URS Securities Australia Ltd         Self         27,091         (3,169)           URS Securities Australia Ltd         Self         33,075         35,947           URS Securities Australia Ltd         Self         33,745         (3,102)           URS Securities Australia Ltd         Buy         23,145         (3,102)           URS Securities Australia Ltd         Self         23,454         (2,150)           URS Securities Australia Ltd         Self         23,454         (2,150)           URS Securities Australia Ltd         Self         1,865,888         (2,150)           URS Securities Australia Ltd         Self         7,452         (2,150)           URS Securities Australia Ltd         Self         33,799         (2,150)           URS Securities Australia Ltd         Self         3,876         (4,232) <td< td=""><td></td><td>UBS Securities Australia Ltd</td><td>- X-</td><td>9,353</td><td>(1,075)</td><td>Ordinary</td></td<>		UBS Securities Australia Ltd	- X-	9,353	(1,075)	Ordinary
UBS Securities Australia Ltd         Sell         1,578         1,050           UBS Geolan Asset Management         Buy         1,1779         1,362           UBS Geolan Asset Management         Buy         1,1779         1,362           UBS Securities Australia Ltd         Buy         27,091         (1,820)           UBS Securities Australia Ltd         Buy         30,7843         35,33           UBS Securities Australia Ltd         Buy         27,091         (1,820)           UBS Securities Australia Ltd         Buy         22,264         2,81           UBS Securities Australia Ltd         Buy         24,264         2,81           UBS Securities Australia Ltd         Buy         24,264         2,81           UBS Securities Australia Ltd         Buy         24,264         2,81           UBS Securities Australia Ltd         Buy         1,861,588         217,422           UBS Securities Australia Ltd         Buy         134,794         (15,559)           UBS Securities Australia Ltd         Sell         1,367         (15,559)           UBS Securities Australia Ltd         Sell         1,367         (15,759)           UBS Securities Australia Ltd         Sell         227,731         (34,849)           UBS Secur	ı	UBS Securities Australia Ltd	ik 3	653	(75)	Ordinary
UBS Securities Australia Ltd         Buy         15.73         15.62           UBS Securities Australia Ltd         Bluy         27.09         3.873           UBS Securities Australia Ltd         Sell         27.09         (1.820)           UBS Securities Australia Ltd         Buy         27.09         (1.820)           UBS Securities Australia Ltd         Buy         23.49         (1.820)           UBS Securities Lucia Australia Ltd         Buy         23.49         27.032           UBS Securities Ltd         Buy         23.49         27.022           UBS Securities Ltd         Buy         23.49         27.022           UBS Securities Ltd         Buy         23.49         27.022           UBS Securities Ltd         Buy         1.85.758         (1.15.00)           UBS Securities Lastralia Ltd         Buy         1.85.758         (1.15.00)           UBS Securities Ltd         Sell         1.83.754         (1.15.00)           UBS Securities Ltd         Sell         33.00         8.73           UBS Securities Ltd         Sell         3.754         (1.15.00)           UBS Securities Ltd         Sell         3.754         (1.15.00)           UBS Securities Ltd         Sell         2.773.78		LIBS Global Accet Management	Sen.	26/1	(206)	Ordinary
UBS Securities Australia Ltd         Bby         33,075         3,507           UBS Securities Australia Ltd         Self         2,709         3,507           UBS Securities Australia Ltd         Self         2,739         (3,169)           UBS Securities Australia Ltd         Self         30,923         (1,820)           UBS Securities Australia Ltd         Buy         30,923         2,013           UBS Securities Australia Ltd         Buy         2,214,497         2,013           UBS Securities Australia Ltd         Buy         2,2464         2,821           UBS Securities Australia Ltd         Buy         1,861,588         2,17,422           UBS Securities Australia Ltd         Buy         7,432         (7,1500)           UBS Securities Australia Ltd         Buy         7,432         (7,1500)           UBS Securities Australia Ltd         Sell         7,756         (7,1500)           UBS Securities Australia Ltd         Sell         7,756         (7,1500)           UBS Securities Australia Ltd         Sell         3,756         (7,1500)           UBS Securities Australia Ltd         Sell         3,756         (7,1500)           UBS Securities Australia Ltd         Sell         3,756         (7,1500)		UBS Global Asset Management	Rus	8/0/1	S 5	Ordinary
UBS Securities Australia Ltd         Sell         27,091         (3,169)           UBS Securities Australia Ltd         Boy         307,843         35,007           UBS Securities Australia Ltd         Buy         23,764         20,037           UBS Securities Australia Ltd         Buy         23,264         2,871           UBS Securities Australia Ltd         Buy         24,264         2,871           UBS Securities Australia Ltd         Buy         24,264         2,871           UBS Securities Australia Ltd         Buy         1,861,589         217,422           UBS Securities Australia Ltd         Buy         334,799         33,000           UBS Securities Australia Ltd         Buy         1,861,589         217,422           UBS Securities Australia Ltd         Buy         334,799         33,000           UBS Securities Australia Ltd         Buy         32,27         (15,150)           UBS Securities Australia Ltd         Sell         32,27         (15,150)           UBS Securities Australia Ltd         Sell         323,76         (15,156)           UBS Securities Australia Ltd         Sell         227,731         (34,849)           UBS Securities Australia Ltd         Sell         227,731         (24,849)	l	UBS Securities Australia Ltd	Buy	570 EE	5,02	Ordinary
UBS Securities Australia Ltd         Sell         15,433         (1,820)           UBS Securities Australia Ltd         Buy         173,425         (20,187)           UBS Securities Australia Ltd         Buy         24,349         27,032           UBS Securities Australia Ltd         Buy         24,246         28,11           UBS Securities Australia Ltd         Buy         24,246         2,811           UBS Securities Australia Ltd         Buy         1,861,588         217,422           UBS Securities Australia Ltd         Buy         1,861,789         33,000           UBS Securities Australia Ltd         Buy         1,867,784         (15,500)           UBS Securities Australia Ltd         Sell         1,827,784         (15,500)           UBS Securities Australia Ltd         Sell         34,784         (15,500)           UBS Securities Australia Ltd         Sell         320,131         (34,849)           UBS Securities Australia Ltd         Sell         220,131         (34,849)	ı	UBS Securities Australia Ltd	Sell	150.72	(3 169)	Ordinary
UBS Securities Australia Ltd         BMy         30,942         35,947           UBS Securities Australia Ltd         6My         20,1457         20,187           UBS Securities Australia Ltd         6My         23,1457         2,1032           UBS Securities Australia Ltd         8My         24,264         2,821           UBS Securities Australia Ltd         8My         1,861,589         217,422           UBS Securities Australia Ltd         8My         1,861,589         217,422           UBS Securities Australia Ltd         8My         1,861,799         33,000           UBS Securities Australia Ltd         8My         7,442         15,000           UBS Securities Australia Ltd         8My         7,442         15,000           UBS Securities Australia Ltd         8M         7,442         15,000           UBS Securities Australia Ltd         8M         7,432         16,000           UBS Securities Australia Ltd         8M         2,617         17,422           UBS Securities Australia Ltd         8M         2,717         2,617           UBS Securities Australia Ltd         8M         2,717         2,618           UBS Securities Australia Ltd         8M         2,717,422         2,617,422           UBS Securi	25-Nov-14	UBS Securities Australia Ltd	Sell	15,433	(1,820)	Ordinary
UBS Securines Australia trd         Sell         173,425         (20.187)           UBS Securines Australia trd         Buy         231,459         27,022           UBS Securities Australia trd         Sell         24,564         2,821 trd           UBS Securities Australia trd         Buy         24,564         2,821 trd           UBS Securities Australia trd         Buy         1,861,589         217,422           UBS Securities Australia trd         Sell         1,861,589         217,402           UBS Securities Australia trd         Sell         1,861,589         217,402           UBS Securities Australia trd         Sell         1,347,64         (15,560)           UBS Securities Australia trd         Sell         3,320         (40,353)           UBS Securities Australia trd         Sell         3,320         (40,353)           UBS Securities Australia trd         Sell         2,277,56         (40,353)           UBS Securities Australia trd         Sell         2,277,56         (40,353)           UBS Securities Australia trd         Sell         2,277,58         (40,432)           UBS Securities Australia trd         Sell         2,277,58         (27,432)           UBS Securities Australia trd         Sell         2,277,31	25-Nov-14	UBS Securities Australia Ltd	Buy	307,843	35,947	Ordinary
UBS Securities Australia Ltd         Buy         23,1497         27,022           UBS Securities Australia Ltd         Buy         23,264         2,7032           UBS Securities Australia Ltd         Buy         1,661,586         2,71,432           UBS Securities Australia Ltd         Buy         1,661,586         2,71,432           UBS Securities Australia Ltd         Buy         1,861,786         (71,500)           UBS Securities Australia Ltd         Sell         1,32,784         (71,500)           UBS Securities Australia Ltd         Sell         3,784         (71,506)           UBS Securities Australia Ltd         Sell         3,784         (71,506)           UBS Securities Australia Ltd         Sell         3,784         (71,506)           UBS Securities Australia Ltd         Sell         2,277,131         (34,644)           UBS Securities Australia Ltd         Sell         2,277,131         (34,644)           UBS Securities Australia Ltd         Sell         2,277,131         (34,644)           UBS Securities Australia Ltd         Sell         2,277,131         (37,432)           UBS Securities Branch         Buy         1,535         1,437           UBS Advantalia Branch         Buy         1,537         1,537  <	25-Nov-14	UBS Securities Australia Ltd	Sell	173,425	(20,187)	Ordinary
UBS Securities Australia Ltd         BM         2.454         2.821           UBS Securities Australia Ltd         BM         1.661.589         2(17.422)           UBS Securities Australia Ltd         BM         1.661.589         2(17.422)           UBS Securities Australia Ltd         BM         7.472         33.0759         39.000           UBS Securities Australia Ltd         BM         7.472         (7.15.000)           UBS Securities Australia Ltd         Sell         7.472         (7.15.000)           UBS Securities Australia Ltd         Sell         7.476         (7.15.000)           UBS Securities Australia Ltd         Sell         3.46.359         (40.335)           UBS Securities Australia Ltd         Sell         2.57.131         (3.4848)           UBS Securities Australia Ltd         Sell         2.73.0         9.57           UBS Securities Australia Ltd         Sell         2.73.0         9.57           UBS Securities Australia Ltd         Sell         2.73.0         1.73.823           UBS Securities Branch         Sell         2.73.7         1.43.7           UBS Actualia Branch         Buy         1.437         1.437	25-Nov-14	UBS Securities Australia Ltd	Buy	231,497	27,032	Ordinary
URS Securities Australia trd         Buy         1,861,588         271,432           UBS Securities Australia trd         Buy         334,799         27,402           UBS Securities Australia trd         Buy         334,799         39,000           UBS Securities Australia trd         Buy         334,799         37,050           UBS Securities Australia trd         Sell         134,764         (71,500)           UBS Securities Australia trd         Sell         343,564         (75,756)           UBS Securities Australia trd         Sell         227,335,170         (27,353)           UBS Securities Australia trd         Sell         2277,818         (27,342)           UBS Securities Australia trd         Sell         2277,818         (27,432)           UBS Securities Australia trd         Sell         2277,818         (27,432)           UBS Securities Australia trd         Sell         2277,818         (27,432)           UBS Securities Australia trd         Buy         1,535         1,437           UBS Act Australia Benrch         Buy         1,5358         1,437	25-Nov-14	UBS Securities Australia Ltd	Buy	24,264	2,821	Ordinary
URS Securities Australia Ltd         Budy         1,861,588         217,432           URS Securities Australia Ltd         Sell         1,867,799         39,000           URS Securities Australia Ltd         Sell         182,784         (71,500)           URS Securities Australia Ltd         Sell         34,744         (15,00)           URS Securities Australia Ltd         Sell         34,764         (15,766)           URS Securities Australia Ltd         Sell         320,131         (34,848)           URS Securities Australia Ltd         Sell         220,131         (34,848)           URS Securities Australia Ltd         Sell         220,131         (34,848)           URS Securities Australia Ltd         Sell         220,131         (34,848)           URS Securities Australia Ltd         Sell         227,131         (34,848)           URS Securities Branch         Buy         1,535         1,437	25-Nov-14	UBS Securities Australia Ltd	le.	4,390	(515)	Ordinary
UBS Securities Australia Ltd         Self         345,090           UBS Securities Australia Ltd         Sull         183,784         (21,500)           UBS Securities Australia Ltd         Sull         74,923         (75,756)           UBS Securities Australia Ltd         Sull         348,359         (40,358)           UBS Securities Australia Ltd         Buy         8,730         (40,383)           UBS Securities Australia Ltd         Sull         2,971         (40,484)           UBS Securities Australia Ltd         Sull         2,375,170         (27,848)           UBS Securities Australia Ltd         Sull         2,277,818         (271,432)           UBS Securities Australia Ltd         Sull         2,277,818         (271,432)           UBS Securities Australia Ltd         Sull         2,277,818         (271,432)           UBS AG, London Branch         Buy         15,742         1,437           UBS AG, London Branch         Buy         15,754         1,437           UBS AG, London Branch         Buy         15,754         1,437	25-Nov-14	UBS Securities Australia Ltd	Buy	1,861,588	217,432	Ordinary
URS Securities Australia Ltd         Sell         74,923         (71,500)           URS Securities Australia Ltd         Sell         134,764         (75,756)           URS Securities Australia Ltd         Sell         343,564         (75,756)           URS Securities Australia Ltd         Buy         8,230         60,533           URS Securities Australia Ltd         Sell         229,131         (64,848)           URS Securities Australia Ltd         Sell         2,277,818         (27),432           URS Securities Rench         Buy         15,724         1,826           URS AG, London Ranch         Buy         15,724         1,878           URS AG, Australia Bench         Buy         1,3356         1,437	25-NOV-14	USS Securities Australia Ltd	Buy	334,799	39,000	Ordinary
USS Securities Autorials Ltd         Self         14,724.2         18,749.1           UNES Securities Autorials Ltd         Self         34,754.1         (16,359.6)           USE Securities Autorials Ltd         Boy         8,230.         (40,359.6)           USE Securities Autorials Ltd         Self         237,131.         (34,849)           USE Securities Autorials Ltd         Self         237,131.         (34,849)           USE Securities Autorials Ltd         Self         237,131.         (37,849)           USE AGA Autorials Banch         Self         2,277,818.         (27),432.           USE AGA Autorials Banch         Boy         1,857,24         (27,432.           USE AGA Autorials Banch         Boy         1,337.         (37)	25.Nov.14	UBS Securities Australia Ind	laci Bin	183,/84	(21,500)	Ordinary
UBS Securities Australia Ltd         Sell         348,359         (40,555)           UBS Securities Australia Ltd         Sell         8,230         957           UBS Securities Australia Ltd         Sell         29,7131         (34,848)           UBS Securities Australia Ltd         Sell         2,335,170         (27,848)           UBS AG, London Branch         Sell         2,277,518         (271,432)           UBS AG, London Branch         Buy         15,734         1,836           UBS AG, Australia Branch         Buy         15,734         1,836           UBS AG, Australia Branch         Buy         15,734         1,836           UBS AG, Australia Branch         Buy         12,358         1,437	25-Nov-14	UBS Securities Australia Ltd	Sell	134 764	8,745	Ordinary
UBS Securities Australia Ltd         Buy         8,230         957           USS Securities Australia Ltd         \$541         2,335,170         (34,849)           USS Securities Australia Ltd         \$61         2,335,170         (273,832)           UBS AG, London Branch         \$61         2,277,818         (271,432)           UBS AG, London Branch         Buy         1,861,588         217,432           UBS AG, Australia Branch         Buy         15,734         1,836           UBS AG, Australia Branch         Buy         15,734         1,836           UBS AG, Australia Branch         Buy         1,2,358         1,437	25-Nov-14	UBS Securities Australia Ltd	Sell	348 359	(40.353)	Ordinary
UBS Securities Australie Ltd         Sell         297,131         (34,848)           U6S Securities Australia Ltd         Sell         2,377,818         (273,827)           U6S AG, Lordon Banch         Sell         2,277,818         (273,432)           UBS AG, Lordon Banch         Buy         1,861,588         217,432           USS AG Australia Branch         Buy         1,877,24         1,836           USS AG Australia Branch         Buy         1,2358         1,337	25-Nov-14	UBS Securities Australia Ltd	Buy	8.230	957	Ordinary
UBS Securities Australia Ltd         Sell         2,335,170         (273,872)           UBS AG, London Banch         Sell         2,275,818         (217,422)           UBS AG, London Banch         Buy         1,661,588         217,432           UBS AG Australia Banch         Buy         15,734         17,734           UBS AG Australia Branch         Buy         12,734         1,437           UBS AG Australia Branch         Buy         12,356         1,437	25-Nov-14	UBS Securities Australia Ltd	Seli	297,131	(34.848)	Ordinary
UBS AG, London Branch         Sell         2,277,518         (2,17/32)           UBS AG, London Branch         Buy         1,681,588         217,432           UBS AG Australia Branch         Buy         15,734         1,836           UBS AG Australia Branch         Buy         1,235         1,337	25-Nov-14	UBS Securities Australia Ltd	Setl	2,335,170	(273,872)	Ordinary
UBS AGE Australia Branch         Bay         1,561,588         217,432           UBS AGE Australia Branch         Bay         15,734         1,836           UBS AGE Australia Branch         Bay         12,358         1,437	25-Nov-14	UBS AG, London Branch	Sell	2,277,818	(217,432)	Ordinary
UBS AG Australia Branch         Buy         15,734         1,836           UBS AG Australia Branch         Buy         12,358         1,437		UBS AG, London Branch	Buy	1,861,588	217,432	Ordinary
UBS AG Australia Branch Buy 12,358 1,437	İ	UBS AG Australia Branch	Buy	15,734	1,836	Ordinary
		UBS AG Australia Branch	Buy	12,358	1,437	Ordinary

21-Mov-14	I IBC Securities Australia Ind	Yoli	30E CF		
21-Nov-14	UBS Securities Australia Ltd	Sell	8 192	(996)	Ordinary
21-Nov-14	UBS Securities Australia Ltd	Sell	1,153		
21-Nov-14	UBS Securities Australia Ltd	Seli	3 663		Ordinary
21-Nov-14	UBS Securities Australia Itd	Sell	10,625	(1 202)	
21-Nov-14	SIRC Countries Australia 144	Call	5,55		
Z 1-NON-14	ODS SECURITIES AUSTRAIIA LIIG	liac	1,713		
21-Nov-14	UBS Securities Australia Ltd	Sell	6,190		
21-Nov-14	UBS Securities Australia Ltd	Sell	39,924		
21-Nov-14	UBS Securities Australia Ltd	Sell	39,135		
21-Nov-14	UBS Securities Australia Ltd	Sell	386'8		
21-Nov-14	UBS Securities Australia Ltd	Sell	1,755		
21-Nov-14	UBS Securities Australia Ltd	Sell	1 247		
21-Nov-14	IUBS Securities Australia (10	Sell	1 247		
21-Nov-14	I IBC Cocurition Australia (14)	Coll	010 10		
21-Nov-14	LIBS Securities Augustin 144	Coll	800/10	2	
21 Nov. 14	Table Countries Australia Ltd	100	ACE'C		
1 - NOA-13	ODS SECULIES AUSTRAIR LIG	2411	887		Ordinary
Z I-NOV- 14	UBS Securities Australia Ltd	Sel	5,207	(614)	
21-NOV-12	UBS Securities Australia Ltd	Sell	5,207		Ordinary
1-Nov-14	UBS Securities Australia Ltd	Sell	3,273		
	UBS Securities Australia Ltd	Self	9,964		
	UBS Securities Australia Ltd	Sel	0/9		
ĺ	UBS Securities Australia Ltd	Sell	1.068		ļ
ı	UBS Securities Australia Ltd	Sell	082		l
١	UBS Securities Australia (1d	lla?	2002		ı
21-Nov-14	LIRS Securities Australia 11d	lay.	200,6		
ı	Lat all materials A section of O.C.	***	60	(/)	
ı	Opp Secondes Australia LIO	261	580,1		
ı	ODS SECURIORS AUSTRAINA LIC	Sei.	11/	(2)	
1	UBS Securities Australia Ltd	sell	8,886	Ğ	Ordinary
-	UBS Securities Australia Ltd	Sell	203		
- 1	UBS Securities Australia Ltd	Sell	20,804	(2,462)	
1	UBS Securities Australia Ltd	Sell	1,411		
	UBS Securities Australia Ltd	Sell	760		
	UBS Securities Australia Ltd	Sell	8 886	11	
ı	UBS Securities Australia Ltd	Sell	1 166		
ı	UBS Securities Australia Ltd	وا	144 977		
L	LIDS Contraction Australia Ltd	100	116'44	(1)	
1 Mov 14	1700 Commission Authorities and	100	006.1	(525)	
	1100 Countries Australia Ltd	100	560		
ı	COO COUNTY OF THE PART OF THE		4,183	(495)	
1	Obs securities Australia Ltd	liac 3	3,642		Ordinary
1	The second contract of the second contract of	100	2,530	(6/7)	
1	OD DECURIES AUSTRIA LIG	100	eed'e		
100	Coo Seculities Australia Litu	Jac.	5,535		
21-Nov-14	UBS Securities Australia Ltd	Sell	1,208	(143)	Ordinary
-	UBS Securities Australia Ltd	Seil	5,315		
	UBS Securities Australia Ltd	Sell	16,131		
11-Nov-14	UBS Securities Australia Ltd	Sell	4,149		
	UBS Securities Australia Ltd	Sell	38 786		
	ILIBS Securities Australia Itd	la y	10.047	(1,190)	
ı	1100 Cocuration Australia 144	Jell Sell	10,04/		١
	COD Securines Australia Etc	llac llac	4,191		Ordinary
١	UBS Securities Australia Ltd	Sell	209'		
١	UBS Securities Australia Ltd	Sell	10,478		
21-Nov-14	UBS Securities Australia Ltd	Sell	50,700		
1-Nov-14	UBS Securities Australia Ltd	Sell	147,554	(1)	
1-Nov-14	UBS Securities Australia Ltd	Sell	4.724		
21-Nov-14	UBS Securities Australia Ltd	Sell	870	(103)	Ordinary
		Enter Into Derivative			
21-Nov-14	UBS Securities Australia Ltd	Contract	N/A	(750,000)	Options
:		Enter Into Derivative			
21-Nov-14	UBS Securities Australia Ltd	Contract	NA	375,000	Options
		Enter Into Derivative			
21-Nov-14	UBS Securities Australia Ltd	Contract	AVA	(33,400)	Options
	UBS Securities Australia Ltd	Sell	478.147	(59,233)	l
	UBS Securities Australia Ltd	Seli	AP 096		
	UBS Securities Australia Ltd	Sell	864 778		l
ı	UBS Securities Australia I td	Blw	50,508	5 040	
ı	UBS Securities Australia Ltd	Buy	305 565		
ı	IIBS Securities Australia Ital	Coli	200,000		Cidinary
	I ID Contribut Australia I ad		116.0	(07/)	Ordinary
1	IDC Counties Australia to	6.50	166,332		Oromary
1	DDS Securities Australia Ltd	ocy	556		Ordinary
1	UBS Securities Australia Ltd	Sell	438,376		Ordinary
-	UBS Securities Australia Ltd	Buy	197,694		Ordinary
-Nov-14	UBS Wealth Management Australia Ltd	Buy	15,796		Ordinary
1-Nov-14	UBS Wealth Management Australia Ltd	Sell	14 688		Ordinary
24-Nov-14	IRS Wealth Management Australia 144	N. S.	15 640	1 906	Ordinary
24-MOV-14	UBS Wealth Management Australia Lto	euy	15,540		Ordinary

25-Nov-14	UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy	3,338	390	Ordinary
	Securities Australia Ltd Securities Australia Ltd		15 520	1,820	Curiary
	Securities Australia Ltd	r and	10.7/21		Ordinary
		Buy	30,123	3 519	Ordinary
	UBS Securities Australia Ltd	Buy	10.067	1176	Ordinary
	Securities Australia Ltd	Buy	4114	480	Ordinary
	UBS Securities Australia Ltd	Buy	3 985	465	Ordinan
	JBS Securities Australia Ltd	Buy	1 680	196	Ordinary
	Securities Australia Ltd	Buy	13 969	1 630	Ordinary
	UBS Securities Australia Ltd	Buy	3 634	474	Ordinary
	Securities Australia Ltd	Buy	2.571	300	Ordinary
	Securities Australia Ltd	Buy	463	25	Ordinary
	Securities Australia Ltd	Buy	5.913	069	Ordinary
	Securities Australia Ltd	Buy	14,929	1,740	Ordinary
	securities Australia Ltd	Buy	4,350	203	Ordinary
	securities Australia Ltd	Buy	9,084	1,060	Ordinary
	Securities Australia Ltd	8uy	26,558	3,099	Ordinary
	Securities Australia Ltd	Buy	7.023	018	Ordinary
	Securities Australia Ltd	Buy	27.018	3 149	Ordinary
	Securities Australia Ltd	Buy	21.500	2.500	Ordinary
	Securities Australia Ltd	Buy	4 825	561	Ordinary
l	Securities Australia Ltd	Buy	10.062	1170	Ording
	Securities Australia Ltd	Buy	212	904	Ordinary
	securities Australia Ltd	Buv	2 300	OCC	ordinary ordinary
	securities Australia Ltd	Buy	25,000	0.000	Ordinary
l	securities Australia I1d	Buv	11 042	1 204	Ordinary
5-Nov-14 UBS S	Securities Australia Ltd	Buy	1 305	1,204	Ordinary
ı	Securities Australia Itd	Rin	101.30	500,	Cidinaly
	UBS Securities Australia Ltd	Buv	23,553	702 C	Ordinary
ĺ	JBS Securities Australia Itd	Buv	10.146	7,444	Ordinary
ŀ	JBS Securities Australia 11d	Buv	5 730	1222	Ordinary
	ecurities Australia I to	Buy	2,72	ò	Ordinary
1	ecurities Australia I to	NI N	C12	67	Orginary
ı	Aprilia Anstralia (14	Braz	100.0	0,347	Ordinary
1	Apprication Australia 144	D. 10	107.6	707	Ordinary
1	According Australia Ital	on's	202	2.4	Ordinary
ı	econoces Australia Liu	on's	549	62	Ordinary
Т	ecurius Australia Ltd	ána	36,379	4,235	Ordinary
П	eculius Australia Lid	buy	2,594	302	Ordinary
ı	Continue Australia Ltd	620	768'/	919	Ordinary
Т	econines Australia cid	ing	52,253	6,083	Ordinary
Т	ecurines Australia Ltd	gnà	32,405	3,768	Ordinary
-1	ecurities Australia Ltd	Buy	14,955	1,739	Ordinary
ļ	ecurities Australia Ltd	Buy	7,379	828	Ordinary
- 1	ecurities Australia Ltd	Buy	3,784	440	Ordinary
- 1	ecunties Australia Ltd	Buy	866	116	Ordinary
1	ecurities Australia Ltd	Buy	1,797	505	Ordinary
	ecurities Australia Ltd	Buy	593	69	Ordinary
25-Nov-14 UBS Se	UBS Securities Australia Ltd	Buy	5,925	689	Ordinary
	JBS Securities Australia Ltd	Buy	26.058	3.030	Ordinary
	JBS Securities Australia Ltd	Buy	34	4	Ordinary
-	UBS Securities Australia Ltd	Buy	76.841	8 035	Colingo
25-Nov-14 UBS Se	JBS Securities Australia Ltd	Buv	68 680	2007	Ordinary
r	JBS Securities Australia Ltd	Buv	36	200	in in it
	ecurities Australia Ltd	Sell	45 202	(5.75G)	Ordinary
Γ	UBS Securities Australia Ltd	Buv	25,318	2 044	Ordinary
	UBS Securities Australia I1d	Buv	11 782	1 370	Colling
Ī	IRC Securities Australia 14d	Bird	20 7,11	0/0,1	Cidinaly
Ī	UBS Securities Australia Ltd	Buy	055	3,/15	Ordinary
T	BS Securities Australia Ind	Riv	51 408	4.0	Ordinary
Ī	DO Coorrespondent August 1-1	9.00	064,430	755,0	Ordinary
Ť	DOS Securities Acedemic Liu	ond.	768'1	077	Ordinary
T	econices Australia Lite	eu,	513,049	36,401	Ordinary
25-NOV-14	Jos Securities Australia Ltd	Buy	55,823	6,491	Ordinary
1	JBS Securities Australia Ltd	gný	4,971	578	Ordinary
		Enter Into Derivative			
1	JBS Securries Australia Ltd	Contract	N/A	40,200	Options
-1	lobal Asset Management	Je.	164,599	(18,996)	Ordinary
-1	lobal Asset Management	JE SE	650,442	(75,066)	Ordinary
-	lobal Asset Management	Sell	2,348	(270)	Ordinary
- 1	lobal Asset Management	Buy	65,820	7,563	Ordinary
ì	ecurities Australia Ltd	les	4,624	(534)	Ordinary
26-Nov-14 UBS Se	UBS Securities Australia Ltd	les.	17,320	(2,000)	Ordinary
ı	acurities Australia Ltd	Buv	3.46.4	000	Ordinan
1	בכתוווא אמציים הים	suy	3,404	400	Ordinary

URS AGA Matrials Brainch         Sell         443,400           URS AGA Matrials Brainch         Sell         48,000           URS AGA Matrials Brainch         Sell         48,000           URS Securities Australia Ltd         BMY         9,240           URS Securities Australia Ltd         BMY         11,067           URS Securities Australia Ltd         BMY         1,058           URS Securities Australia Ltd         BMY <t< th=""><th></th><th>(9,930) Ordinary</th><th></th><th></th><th></th><th>İ</th><th></th><th></th><th>١</th><th>۱</th><th></th><th>368 Ordinary</th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th>931 Ordinary</th><th></th><th>20 Ordinary</th><th>51 Ordinary</th><th>8 Ordinary</th><th></th><th>667 Ordinary</th><th></th><th>1,112 Ordinary</th><th>803 Ordinary</th><th></th><th></th><th>523 Ordinary</th><th>H</th><th>602 Ordinary</th><th></th><th></th><th></th><th></th><th>640 Ordinary</th><th></th><th></th><th>İ</th><th></th><th></th><th>339 Ordinary</th><th></th><th></th><th><math>\perp</math></th><th></th><th>L</th><th></th><th></th><th>370 Ordinary</th><th>22 Ordinary</th><th></th><th></th><th>1,066 Ordinary</th><th></th><th>153 Ordinary 486 Ordinary</th></t<>		(9,930) Ordinary				İ			١	۱		368 Ordinary								931 Ordinary		20 Ordinary	51 Ordinary	8 Ordinary		667 Ordinary		1,112 Ordinary	803 Ordinary			523 Ordinary	H	602 Ordinary					640 Ordinary			İ			339 Ordinary			$\perp$		L			370 Ordinary	22 Ordinary			1,066 Ordinary		153 Ordinary 486 Ordinary
	443 400	84,659	430,000	9,240	9,240	11,067	10,948	4,539	2,550	10,659	0/5,5	3 128	4,165	3,086	10,200	2,550	4,505	3,043	3,655	7,914	1,190	1726	434	89	2,312	5,670	1,274	9,452	6,817	2,479	5,054	4,435	933	5,105	153	1,753	295	8,462	5,414	5,446	238	2,547	6,147	10,236,000	2 893	16,703	19,770	1,981	1,535	20,723	4,074	13,246	3,160	188	7,780	3,160	9107	1.307	1,307
Austa alsa Branch Australia Branch Australia Branch Australia Branch Interes Australia Ltd Interes Interes Interes Interes Interes Interes Interes Interes Interes Interes Interes Interes Interes Interes Interes Interes Interes Interes Interes	Isell	Sell	Sell	Buy	Buy	Buy	Buy	Buy	Buy	Buy Buy	Buy.	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	8uy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	8uy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Sell	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	lank lank	Buy	Buy Buy
그러리라라라라라라라라라라라라라라라라라라라라라라라라라라라라라라라라라라라라	Australia Branch	Australia Branch	UBS Securities Australia Ltd	UBS Securities Australia Ltd	curities Australia Ltd	curities Australia Ltd	curities Australia Ltd	curities Australia Ltd	curities Australia Ltd	Curries Australia Ltd	Utilies Australia Ltd	Curities Australia Ltd	urities Australia Ltd	urities Australia Ltd	Securities Australia Ltd	curities Australia Ltd	curities Australia Ltd	curties Australia Ltd	unities Australia Ltd	JBS Securities Australia Ltd	Securities Australia Ltd	Curities Australia Ltd	Securities Australia Ltd	curties Australia Ltd	urities Australia Ltd	Securities Australia Ltd	urities Australia Ltd	unities Australia Ltd	urities Australia Ltd	urities Australia Ltd	urities Australia Ltd	urities Australia Ltd	urities Australia Ltd	unities Australia Ltd	urities Australia Ltd	urities Australia Ltd	urities Australia Ltd	urities Australia Ltd	UBS Securities Australia Ltd	urities Australia Ltd	urities Australia Ltd	urities Australia Ltd	urities Australia Ltd	urities Australia Ltd	urities Australia Ltd	UBS Securities Australia Ltd	urities Australia Ltd	urities Australia Ltd	urities Australia Ltd	Securities Australia Ltd	Securities Australia Ltd	urities Australia (td	urities Australia Ltd	Securities Australia Ltd	JBS Securities Australia Ltd	urities Australia Ltd	urrities Australia Ltd	urities Australia Ltd	JBS Securities Australia Ltd JBS Securities Australia Ltd

15   15   15   15   15   15   15   15	27-Nov-14	UBS Securities Australia Ltd	Buy	3,354	[E]	Ordinary
4         LOSS Scortines Auctions Land         Bioly         1,954         1,9	7-Nov-14	UBS Securities Australia Itd	Riv	34		Ordinary
4 UBS Scorringes Australia LEG         Bigs         A LAST         A SATE <td>27-Nov-14</td> <td>UBS Securities Australia 11d</td> <td>, a</td> <td>70</td> <td></td> <td>Ordinary</td>	27-Nov-14	UBS Securities Australia 11d	, a	70		Ordinary
4         Life Sciencies Australia Led         Bigar         15.00           4         Life Sciencies Australia Led         Bigar         15.00           4         Life Sciencies Australia Led         Bigar         15.00           4         Life Sciencies Australia Led         Bigar         15.00           4         Life Sciencies Australia Led         Bigar         15.00           4         Life Sciencies Australia Led         Bigar         15.00           4         Life Sciencies Australia Led         Bigar         15.00           105 Sciencies Australia Led         Bigar         15.00         25.00           105 Sciencies Australia Led         Bigar         2.00         2.00           106 Sciencies Australia Led         Bigar	27-Nov-14	UBS Securities Australia Ital	Buy	88'/		Ordinary
4 USS Scentifies Autralia Ltd         BAY         2,513         2,129           4 USS Scentifies Autralia Ltd         BAY         2,514         0,0284           4 USS Scentifies Autralia Ltd         BAY         2,517         1,028           4 USS Scentifies Autralia Ltd         BAY         2,517         1,028           4 USS Scentifies Autralia Ltd         BAY         2,210         2,523           4 USS Scentifies Autralia Ltd         BAY         2,210         2,521           4 USS Scentifies Autralia Ltd         BAY         2,520         2,627           4 USS Scentifies Autralia Ltd         BAY         3,210         2,521           5 USS Scentifies Autralia Ltd         BAY         3,210         2,521           6 USS Scentifies Autralia Ltd         BAY         3,100         2,670           6 USS Scentifies Autralia Ltd         BAY         3,100         3,100           7 USS Scentifies Autralia Ltd         BAY         3,100         3,100           8 USS Scentifies Autralia L	27-Nov-14	UBS Securities Australia (1d	Riiv		4	Ordinary
4 URS Scorring Authorise Lived         Biny         255.51.7         2,51.7           4 URS Scorring Authorise Lived         Biny         2,52.7         2,52.7           4 URS Scorring Authorise Lived         Biny         2,52.7         2,52.7           4 URS Scorring Authorise Lived         Biny         1,188         1,188           4 URS Scorring Authorise Lived         Biny         2,179         2,179           5 URS Scorring Authorise Lived         Biny         2,179         2,179           6 URS Scorring Authorise Lived         Biny         2,179         2,179           6 URS Scorring Authorise Lived         Biny         2,179         2,179           7 URS Scorring Authorise Lived         Biny         2,179         2,179           8 URS Scorring Authorise Lived         Biny         2,179         2,179           9 URS Scorring Authorise Lived         Biny         2,179         1,179           1 URS Scorring Authorise Lived         Biny         2,179         1,179           1 URS Scorring Authorise Lived         Biny         2,170         1,179           1 URS Scorring Authorise Lived         Biny         2,170         1,179           1 URS Scorring Authorise Lived         Biny         2,170         1,179	27-Nov-14	UBS Securities Australia Ltd	Birv	1000		
4 UBS Securines Autoritals LLD         Bay         \$2.74         \$1.28           4 UBS Securines Autoritals LLD         Bay         \$1.49         \$1.51.23         \$1.52.93           4 UBS Securines Autoritals LLD         Bay         \$1.52.93         \$1.52.93         \$1.52.93           4 UBS Securines Autoritals LLD         Bay         \$1.20         \$1.52.93         \$1.52.93           4 UBS Securines Autoritals LLD         Bay         \$1.20         \$2.00         \$2.00           4 UBS Securines Autoritals LLD         Bay         \$1.20         \$2.00         \$2.00           4 UBS Securines Autoritals LLD         Bay         \$2.00         \$2.00         \$2.00           5 UBS Securines Autoritals LLD         Bay         \$2.00         \$2.00         \$2.00           6 UBS Securines Autoritals LLD         Bay         \$2.00         \$2.00         \$2.00           6 UBS Securines Autoritals LLD         Bay         \$2.00         \$2.00         \$2.00           6 UBS Securines Autoritals LLD         Bay         \$2.00         \$2.00         \$2.00           6 UBS Securines Autoritals LLD         Bay         \$2.00         \$2.00         \$2.00           7 UBS Securities Autoritals LLD         Bay         \$2.00         \$2.00         \$2.00         \$2	7-Nov-14	UBS Securities Australia Ltd	Buv	20,01		
4 (185 Securine Autorials Ltd         Bay         (2,50)         (2,50)           4 (185 Securine Autorials Ltd         Bay         (2,13)         (2,50)           4 (185 Securine Autorials Ltd         Bay         (2,13)         (2,50)           4 (185 Securine Autorials Ltd         Bay         (2,13)         (2,50)           4 (185 Securine Autorials Ltd         Bay         (2,50)         (2,50) <t< td=""><td>7-Nov-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>15/ 28</td><td></td><td></td></t<>	7-Nov-14	UBS Securities Australia Ltd	Buy	15/ 28		
a USE Securine Automita LOT         Bay         (4.15)         (1.55)           4 USE Securine Automita LOT         Bay         2.00         2.00           5 Securine Automita LOT         Bay         2.00         2.00           6 Securine Automita LOT         Bay         2.00         2.00           7 Securine Automita LOT         Bay         2.00         2.00           8 Securine Automita LOT         Bay         2.00         2.00           9 Securine Automita LOT         Bay         2.00         2.00           10 Securine Automita LOT         Bay         2.00         2.00           10 Securine Automita LOT         Bay         2.00         2.00           10 Securine Automita LOT         Bay         2.00         2.00           10 Securine Automita LOT         Bay         2.00         2.00           10 Securine Automita LOT         Bay         2.00         2.00           10 Securine Automita LOT         Bay         1.00         2.00           10 Securine Automita LOT         Bay         1.00         2.00           10 Securine Automita LOT         Bay         1.00         2.00           10 Securine Automita LOT         Bay         2.00         2.00           10	7-Nov-14	UBS Securities Australia Ltd	Buy	207.406		j
4. OSS Securines Automista Let         Bay         2,62,9         2,62,9           4. OSS Securines Automista Let         Bay         1,124         2,51           5. OSS Securines Automista Let         Bay         2,71         2,71           6. OSS Securines Automista Let         Bay         2,71         2,71           6. OSS Securines Automista Let         Bay         2,71         2,71           6. OSS Securines Automista Let         Bay         2,72         2,71           6. OSS Securines Automista Let         Bay         2,72         2,71           6. OSS Securines Automista Let         Bay         2,62         4,72           6. OSS Securines Automista Let         Bay         2,62         4,72           6. OSS Securines Automista Let         Bay         2,72         1,72           6. OSS Securines Automista Let	7-Nov-14	UBS Securities Australia Ltd	Buv	1445		
4 USE Securities Autorials Ltd         Bay         17.15         17.5           4 USE Securities Autorials Ltd         Bay         1.11.6         1.75           4 USE Securities Autorials Ltd         Bay         2.11.6         2.75.33           4 USE Securities Autorials Ltd         Bay         2.11.6         2.75.33           5 Securities Autorials Ltd         Bay         2.11.6         2.75.33           6 Securities Autorials Ltd         Bay         2.15.9         2.75           7 Securities Autorials Ltd         Bay         2.15.9         2.75           8 Securities Autorials Ltd         Bay         2.10.9         2.15.9           9 Securities Autorials Ltd         Bay         2.10.9         2.10.9           10 Securities Autorials Ltd         Bay         2.10.9         2.10.9           10 Securities Autorials Ltd         Bay         2.10.9         1.10.9           10 Securities Autorials Ltd	7-Nov-14	UBS Securities Australia Ltd	Buv	2696		
A LOSS Securines Automita Let         Blay         1,128         1,252           A LOSS Securines Automita Let         Blay         2,150         2,50         2,50           A LOSS Securines Automita Let         Blay         2,150         2,50         2,50         2,50           A LOSS Securines Automita Let         Blay         2,50	7-Nov-14	UBS Securities Australia Ltd	Buv	191 53		
4         UBS Securities Australia Ltd         Bay         2,170         2,19           4         UBS Securities Australia Ltd         Boy         2,176         2,19           4         UBS Securities Australia Ltd         Boy         2,176         2,17           4         UBS Securities Australia Ltd         Boy         2,17         2,17           4         UBS Securities Australia Ltd         Boy         2,17         2,17           5         Light Securities Australia Ltd         Boy         2,17         2,17           6         Light Securities Australia Ltd         Boy         1,17         2,17         1,17           1         Light Securities Australia Ltd         Boy         1,17         2,17         1,17           1         Light Securities Australia Ltd         Boy         2,17         1,17         1,17           1         Light Securities Australia Ltd         Boy         2,17         1,17         1,17           1         Light Securities Australia Ltd         Boy         2,17         1,17         1,17           1         Light Securities Australia Ltd         Boy         2,17         1,17         1,17           1         Light Securities Australia Ltd         Boy <t< td=""><td>7-Nov-14</td><td>UBS Securities Australia Ltd</td><td>Buv</td><td>1,181</td><td></td><td>İ</td></t<>	7-Nov-14	UBS Securities Australia Ltd	Buv	1,181		İ
4 UBS Securities Australia Ltd         Bay         3.250         6.57           4 UBS Securities Australia Ltd         Bay         3.250         6.57           4 UBS Securities Australia Ltd         Bay         3.250         6.57           4 UBS Securities Australia Ltd         Bay         3.25         6.7           4 UBS Securities Australia Ltd         Bay         1.28         1.71           5 UBS Securities Australia Ltd         Bay         1.28         1.75           1 UBS Securities Australia Ltd         Bay         1.28         1.75           1 UBS Securities Australia Ltd         Bay         1.75         1.75           1 UBS Securities Australia Ltd         Bay         1.75         1.75           1 UBS Securities Australia Ltd         Bay         1.75         1.75           1 UBS Securities Australia Ltd         Bay         1.75         1.75           1 UBS Securities Australia Ltd         Bay         1.75         1.75           1 UBS Securities Australia Ltd         Bay         1.75         1.75           1 UBS Securities Australia Ltd         Bay         1.75         1.75           1 UBS Securities Australia Ltd         Bay         1.75         1.75           1 UBS Securities Australia Ltd	7-Nov-14	UBS Securities Australia Ltd	Buv	2010		
4 UBS Securities Australia LEG         BMJ         5,277         5,27           4 UBS Securities Australia LEG         BMJ         7,27         91           4 UBS Securities Australia LEG         BMJ         7,27         91           4 UBS Securities Australia LEG         BMJ         7,27         91           4 UBS Securities Australia LEG         BMJ         1,275         4,27           5 UBS Securities Australia LEG         BMJ         1,275         1,75           6 UBS Securities Australia LEG         BMJ         1,73         1,75           6 UBS Securities Australia LEG         BMJ         1,73         1,75           6 UBS Securities Australia LEG         BMJ         1,73         1,75           6 UBS Securities Australia LEG         BMJ         1,73         1,75           6 UBS Securities Australia LEG         BMJ         1,73         1,75           6 UBS Securities Australia LEG         BMJ         1,73         1,75           6 UBS Securities Australia LEG         BMJ         1,73         1,75           6 UBS Securities Australia LEG         BMJ         1,73         1,75           6 UBS Securities Australia LEG         BMJ         1,73         1,75           6 UBS Securities Australia LEG <t< td=""><td>7-Nov-14</td><td>UBS Securities Australia Ltd</td><td>Buv</td><td>3 100</td><td></td><td>Ordinary</td></t<>	7-Nov-14	UBS Securities Australia Ltd	Buv	3 100		Ordinary
USE Securities Australia Ltd   Buy   2007	7-Nov-14	UBS Securities Australia Ltd	Ruy	2,136		Ordinary
4 UBS Securities Australia Ltd         Boy         724         973           4 LBS Securities Australia Ltd         Boy         1,376         4,78         1,97           4 LBS Securities Australia Ltd         Boy         1,376         1,78         1,78         1,78           4 LBS Securities Australia Ltd         Boy         1,78         1,78         1,78         1,78           5 LBS Securities Australia Ltd         Boy         1,78         1,78         1,78         1,78           6 LBS Securities Australia Ltd         Boy         1,78         1,78         1,78         1,78           6 LBS Securities Australia Ltd         Boy         2,70         3,79         1,79         1,78           6 LBS Securities Australia Ltd         Boy         2,70         3,70         1,70         1,78           6 LBS Securities Australia Ltd         Boy         2,70         3,70         1,70         1,78           6 LBS Securities Australia Ltd         Boy         2,70         3,70         1,70         1,78           7 LBS Securities Australia Ltd         Boy         1,70         1,78         1,70         1,78           8 LBS Securities Australia Ltd         Boy         1,70         1,78         1,70         1,78 <td>7-Nov-14</td> <td>UBS Securities Australia Ltd</td> <td>Buv</td> <td>0/0/0</td> <td></td> <td>Ordinary</td>	7-Nov-14	UBS Securities Australia Ltd	Buv	0/0/0		Ordinary
Miles Securities Australia Ltd   Buy   1,275   1,428   1,171	7-Nov-14	UBS Securities Australia Ltd	Buy	266		Ordinary
4 UBS Securine Australia Led         Buy         1,525         4,24           4 UBS Securine Australia Led         Buy         1,235         1,156           4 UBS Securine Australia Led         Buy         1,235         1,156           4 UBS Securine Australia Led         Buy         1,471         1,68           4 UBS Securine Australia Led         Buy         5,79         7,90           5 UBS Securine Australia Led         Buy         5,70         7,70           6 UBS Securine Australia Led         Buy         2,700         3,842           6 UBS Securine Australia Led         Buy         2,700         3,842           6 UBS Securine Australia Led         Buy         2,700         3,840           6 UBS Securine Australia Led         Buy         2,700         3,70           6 UBS Securine Australia Led         Buy         2,700         3,70           6 UBS Securine Australia Led         Buy         2,700         3,71           1 UBS Securine Australia Led         Buy         2,700         3,71           1 UBS Securine Australia Led         Buy         2,700         3,71           1 UBS Securine Australia Led         Buy         2,700         3,71           1 UBS Securine Australia Led         Buy	7-Nov-14	UBS Securities Australia Itd	Riv	*//		Ordinary
USS Securities Australia Ltd   Buy   1,500	7-Nov-14	UBS Securities Australia 11d	Buy	966		Ordinary
USS Securities Australia Ltd   Buy   1,270   1,520	7-Nov-14	U8S Securities Australia Itd	Surv.	3,002		Ordinary
USS Securities Australia Ltd   Buy   1,542   1,543	7-Nov-14	UBS Securities Australia Itd	Brin	1,326		Ordinary
UBS Securities Australia Ltd   Buy   Ca178   175   1	7-Nov-14	UBS Securities Australia I I d	Brick	1,428		Ordinary
Ligib Securities Australia Ltd         Buy         6,379         756           108 Securities Australia Ltd         Buy         6,375         756           108 Securities Australia Ltd         Buy         1,411         168           108 Securities Australia Ltd         Buy         2,033         3.18           108 Securities Australia Ltd         Buy         2,043         3.18           108 Securities Australia Ltd         Buy         2,056         2.36           108 Securities Australia Ltd         Buy         1,564         1.84           108 Securities Australia Ltd         Buy         1,564         1.84           108 Securities Australia Ltd         Buy         1,564         1.84           108 Securities Australia Ltd         Buy         1,564         1.84           108 Securities Australia Ltd         Buy         1,564         1.87           108 Securities Australia Ltd         Buy         1,564         1.87           108 Securities Australia Ltd         Buy         1,564         1.75           108 Securities Australia Ltd         Buy         1,564         1.75           108 Securities Australia Ltd         Buy         1,564         1.75           108 Securities Australia Ltd         Buy	7-Nov-14	URS Securities Australia Ind	B. D.	1/8		Ordinary
Libbs Securities Australia Ltd         Buy         0,179         170           1 (BS Securities Australia Ltd         Buy         1,411         108           1 (BS Securities Australia Ltd         Buy         2,437         3,342           1 (BS Securities Australia Ltd         Buy         2,437         3,342           1 (BS Securities Australia Ltd         Buy         2,703         3,342           1 (BS Securities Australia Ltd         Buy         1,518         1,55           1 (BS Securities Australia Ltd         Buy         1,518         1,55           1 (BS Securities Australia Ltd         Buy         1,519         2,14           1 (BS Securities Australia Ltd         Buy         1,519         2,14           1 (BS Securities Australia Ltd         Buy         1,519         2,14           1 (BS Securities Australia Ltd         Buy         1,519         2,14           1 (BS Securities Australia Ltd         Buy         1,519         2,14           1 (BS Securities Australia Ltd         Buy         1,595         3,582           1 (BS Securities Australia Ltd         Buy         1,595         3,582           1 (BS Securities Australia Ltd         Buy         1,595         3,782           1 (BS Securities Austral	7-Nov-14	UBS Securities Australia Ind	Bin	866		Ordinary
UBS Securities Australia Ltd         Buy         198         108           UBS Securities Australia Ltd         Buy         1,411         108           UBS Securities Australia Ltd         Buy         2,703         318           UBS Securities Australia Ltd         Buy         1,564         184           UBS Securities Australia Ltd         Buy         1,704         1,704           UBS Securities Australia Ltd         Buy         2,704         3,71           UBS Securities Australia Ltd         Buy         2,704         3,71           UBS Securities Australia Ltd         Buy	7-Nov-14	UBS Securities Australia 11d	D. II	6,375		Ordinary
(18)         Securities Australia (14)         (14)         (16)           (18)         Securities Australia (14)         (14)         (14)         (15)           (18)         (18)         (17)         (17)         (17)           (18)         (18)         (18)         (18)         (18)           (18)         (18)         (18)         (18)         (18)           (18)         (18)         (18)         (18)         (18)           (18)         (18)         (18)         (18)         (18)           (18)         (18)         (18)         (18)         (18)           (18)         (18)         (18)         (18)         (18)           (18)         (18)         (18)         (18)         (18)           (18)         (18)         (18)         (18)         (18)           (18)         (18)         (18)         (18)         (18)           (18)         (18)         (18)         (18)         (18)           (18)         (18)         (18)         (18)         (18)           (18)         (18)         (18)         (18)         (18)           (18)         (18)         (18)         (1	7-Nov-14	UBS Securities Australia Ited	ing.	816		Ordinary
UBS Securities Australia Ltd         Buy         287         18           UBS Securities Australia Ltd         Buy         287         3.42           UBS Securities Australia Ltd         Buy         2.703         3.18           UBS Securities Australia Ltd         Buy         1,554         1.84           UBS Securities Australia Ltd         Buy         1,79         2.14           UBS Securities Australia Ltd         Buy         1,87         2.14           UBS Securities Australia Ltd         Buy         2,661         3.7           UBS Securities Australia Ltd         Buy         2,661         3.7           UBS Securities Australia Ltd         Buy         2,661         3.7           UBS Securities Australia Ltd         Buy         1,097         2.13           UBS Securities Australia Ltd         Buy         1,097         2.34           UBS Securities Australia Ltd         Buy         1,097         1,098           UBS Securities Australia Ltd         Buy         1,09	7-Nov-14	I IRS Securities Australia 14d	bou.	1,411		Ordinary
UBS Securities Australia Ltd         Buy         2,743         3,342           UBS Securities Australia Ltd         Buy         2,703         3,142           UBS Securities Australia Ltd         Buy         1,554         1,554         1,552           UBS Securities Australia Ltd         Buy         1,554         1,552         1,544         1,552           UBS Securities Australia Ltd         Buy         1,519         2,71         1,552         1,544         1,552           UBS Securities Australia Ltd         Buy         1,519         2,71         1,552         1,417         1,552           UBS Securities Australia Ltd         Buy         1,619         2,74         1,552         1,417         1,552           UBS Securities Australia Ltd         Buy         1,629         1,627	7-Nov-14	UBS Securities Australia 144	D	136		Ordinary
UBS Securities Australia Ltd         Buy         6,715         791           UBS Securities Australia Ltd         Buy         6,715         791           UBS Securities Australia Ltd         Buy         1,524         1,584           UBS Securities Australia Ltd         Buy         1,582         1,877           UBS Securities Australia Ltd         Buy         1,819         2.7           UBS Securities Australia Ltd         Buy         1,819         2.7           UBS Securities Australia Ltd         Buy         2,661         3.13           UBS Securities Australia Ltd         Buy         2,667         3.7           UBS Securities Australia Ltd         Buy         2,667         3.7           UBS Securities Australia Ltd         Buy         2,667         3.7           UBS Securities Australia Ltd         Buy         2,677         3.59           UBS Securities Australia Ltd         Buy         2,677         3.59           UBS Securities Australia Ltd         Buy         1,037         1,132           UBS Securities Australia Ltd         Buy         2,671         1,132           UBS Securities Australia Ltd         Buy         1,037         1,132           UBS Securities Australia Ltd         Buy	7-Nov-14	UBS Securities Australia 17d	Busy	28,407		Ordinary
UBS Securities Australia Ltd         Buy         1,518         759           UBS Securities Australia Ltd         Buy         1,546         759           UBS Securities Australia Ltd         Buy         1,546         759           UBS Securities Australia Ltd         Buy         1,547         7,152           UBS Securities Australia Ltd         Buy         1,619         2,14           UBS Securities Australia Ltd         Buy         1,619         2,14           UBS Securities Australia Ltd         Buy         2,665         3,13           UBS Securities Australia Ltd         Buy         2,665         3,13           UBS Securities Australia Ltd         Buy         2,665         3,13           UBS Securities Australia Ltd         Buy         1,679         1,570           UBS Securities Australia Ltd         Buy         1,694         1,754         1,754           UBS Securities Australia Ltd         Buy         1,694         1,754         1,754           UBS Securities Australia Ltd         Buy         1,694         1,754         1,754           UBS Securities Australia Ltd         Buy         1,694         1,754         1,754           UBS Securities Australia Ltd         Buy         1,694         1,75	7-Nov-14	I IRC Securities Australia 1+4	ona)	2,703		Ordinary
USE Securities Australia Ltd         Buy         1,318         156           USE Securities Australia Ltd         Buy         1,564         184           USE Securities Australia Ltd         Buy         2,006         236           USE Securities Australia Ltd         Buy         1,564         1,627           USE Securities Australia Ltd         Buy         1,819         2,14           USE Securities Australia Ltd         Buy         1,667         313           USE Securities Australia Ltd         Buy         1,675         1,675           USE Securities Australia Ltd         Buy         2,661         313           USE Securities Australia Ltd         Buy         2,661         314           USE Securities Australia Ltd         Buy         2,661         317           USE Securities Australia Ltd         Buy         2,661         32           USE Securities Australia Ltd         Buy         1,624         4,18           USE Securities Australia Ltd         Buy         1,624         1,564         1,564           USE Securities Australia Ltd         Buy         1,627         3,64         1,628           USE Securities Australia Ltd         Buy         1,627         3,64         1,627	ı	one securines Australia LLO	suy	6,715		Ordinary
USES Securities Australia Ltd         Buy         1,664         184           USES Securities Australia Ltd         Buy         2,006         2,26           USES Securities Australia Ltd         Buy         1,617         2,14           USES Securities Australia Ltd         Buy         1,819         2,14           USES Securities Australia Ltd         Buy         1,819         2,14           USES Securities Australia Ltd         Buy         2,661         313           USES Securities Australia Ltd         Buy         2,661         313           USES Securities Australia Ltd         Buy         2,661         313           USES Securities Australia Ltd         Buy         2,661         313           USES Securities Australia Ltd         Buy         2,661         1,87           USES Securities Australia Ltd         Buy         2,661         1,87           USES Securities Australia Ltd         Buy         1,037         1,122           USES Securities Australia Ltd         Buy         1,037         1,122           USES Securities Australia Ltd         Buy         1,037         1,126           USES Securities Australia Ltd         Buy         1,037         1,126           USES Securities Australia Ltd         <	1	Obs securities Australia Ltd	Buy	1,318		Ordinary
UBS Securities Australia Ltd         Buy         1,005         2.56           UBS Securities Australia Ltd         Buy         1,172         1,522           UBS Securities Australia Ltd         Buy         1,173         2,1           UBS Securities Australia Ltd         Buy         1,619         2,1           UBS Securities Australia Ltd         Buy         2,661         3,13           UBS Securities Australia Ltd         Buy         2,661         3,1           UBS Securities Australia Ltd         Buy         2,661         3,1           UBS Securities Australia Ltd         Buy         7,7         9           UBS Securities Australia Ltd         Buy         7,683         1,182           UBS Securities Australia Ltd         Buy         7,047         1,182           UBS Securities Australia Ltd         Buy         1,047         1,182           UBS Securities Australia Ltd         Buy         1,037         1,182           UBS Securities Australia Ltd         Buy         1,037         1,182           UBS Securities Australia Ltd         Buy         2,047         6,134           UBS Securities Australia Ltd         Buy         2,047         6,134           UBS Securities Australia Ltd         Buy	ı	Obsideration Australia Ltd	Buy	1,564		Ordinary
UBS Securities Australia Ltd         Buy         14,127         1,652           UBS Securities Australia Ltd         Buy         1,779         2,1           UBS Securities Australia Ltd         Buy         1,877         2,1           UBS Securities Australia Ltd         Buy         1,5955         1,877         2,1           UBS Securities Australia Ltd         Buy         2,661         3,1         2,1           UBS Securities Australia Ltd         Buy         2,661         3,1         2,2           UBS Securities Australia Ltd         Buy         2,661         3,1         2,2           UBS Securities Australia Ltd         Buy         2,667         3,2         3,2         3,2           UBS Securities Australia Ltd         Buy         2,667         1,182         3,2	1	UBS Securities Australia Ltd	Buy	2,006		Ordinary
UBS Securities Australia Ltd         Bluy         179         27           UBS Securities Australia Ltd         Bluy         179         21           UBS Securities Australia Ltd         Bluy         1,817         214           UBS Securities Australia Ltd         Bluy         2,655         1,877           UBS Securities Australia Ltd         Bluy         2,657         1,387           UBS Securities Australia Ltd         Bluy         2,67         3           UBS Securities Australia Ltd         Bluy         2,67         3           UBS Securities Australia Ltd         Bluy         2,67         3           UBS Securities Australia Ltd         Bluy         1,684         1,582           UBS Securities Australia Ltd         Bluy         1,694         1,520           UBS Securities Australia Ltd         Bluy         1,694         1,520           UBS Securities Australia Ltd         Bluy         1,694         1,520           UBS Securities Australia Ltd         Bluy         1,694         1,524           UBS Securities Australia Ltd         Bluy         1,694         1,524           UBS Securities Australia Ltd         Bluy         1,754         1,754           UBS Securities Australia Ltd         Bluy	1	UBS Securities Australia Ltd	Buy	14,127		Ordinary
UBS Securities Australia Ltd         Buy         179         21           UBS Securities Australia Ltd         Buy         1,819         214           UBS Securities Australia Ltd         Buy         1,877         37           UBS Securities Australia Ltd         Buy         2,661         371           UBS Securities Australia Ltd         Buy         2,263         1,877           UBS Securities Australia Ltd         Buy         2,263         3,29           UBS Securities Australia Ltd         Buy         2,263         3,29           UBS Securities Australia Ltd         Buy         2,263         3,108           UBS Securities Australia Ltd         Buy         2,263         1,108           UBS Securities Australia Ltd         Buy         2,264         1,108           UBS Securities Australia Ltd         Buy         2,263         1,108           UBS Securities Australia Ltd         Buy         2,241         6,402           UBS Securities Australia Ltd         Buy         2,343         1,764           UBS Securities Australia Ltd         Buy         2,343         1,764           UBS Securities Australia Ltd         Buy         2,343         1,764           UBS Securities Australia Ltd         Buy	1	UBS Securities Australia Ltd	Buy	179		Ordinary
UBS Securities Australia tod         Buy         1,819         214           UBS Securities Australia tod         Buy         1,619         214           UBS Securities Australia tod         Buy         2,661         313           UBS Securities Australia tod         Buy         2,661         313           UBS Securities Australia tod         Buy         2,77         9           UBS Securities Australia tod         Buy         2,77         32           UBS Securities Australia tod         Buy         1,727         32           UBS Securities Australia tod         Buy         1,0047         1,152           UBS Securities Australia tod         Buy         1,0047         1,152           UBS Securities Australia tod         Buy         1,0047         1,152           UBS Securities Australia tod         Buy         1,0047         1,176         1,176           UBS Securities Australia tod         Buy         1,0047         1,004	-	UBS Securities Australia Ltd	Buy	179		ording.
UBS Securities Australia Ltd         Buy         15,955         1,877           UBS Securities Australia Ltd         Buy         2,661         313           UBS Securities Australia Ltd         Buy         2,661         313           UBS Securities Australia Ltd         Buy         77         2           UBS Securities Australia Ltd         Buy         77         3           UBS Securities Australia Ltd         Buy         2657         3,589           UBS Securities Australia Ltd         Buy         10,047         1,182           UBS Securities Australia Ltd         Buy         2,661         1,2570           UBS Securities Australia Ltd         Buy         1,037         1,2570           UBS Securities Australia Ltd         Buy         1,546         1,2570           UBS Securities Australia Ltd         Buy         1,546         1,2570           UBS Securities Australia Ltd         Buy         1,546         1,284           UBS Securities Australia Ltd         Buy         1,546         1,248           UBS Securities Australia Ltd         Buy         2,343         1,41           UBS Securities Australia Ltd         Buy         2,343         1,05           UBS Securities Australia Ltd         Buy	١	UBS Securities Australia Ltd	Buy	1819		Ordinary
UBS Securities Australia Ltd         Buy         15,955         1,877           UBS Securities Australia Ltd         Buy         2,661         313           UBS Securities Australia Ltd         Buy         2,661         313           UBS Securities Australia Ltd         Buy         222         32           UBS Securities Australia Ltd         Buy         26,677         32           UBS Securities Australia Ltd         Buy         26,677         3,596           UBS Securities Australia Ltd         Buy         26,677         1,827           UBS Securities Australia Ltd         Buy         1,077         1,828           UBS Securities Australia Ltd         Buy         1,037         1,828           UBS Securities Australia Ltd         Buy         1,037         1,848           UBS Securities Australia Ltd         Buy         7,769         1,848           UBS Securities Australia Ltd         Buy         7,769         1,848           UBS Securities Australia Ltd         Buy         7,769         3,14           UBS Securities Australia Ltd         Buy         7,769         3,14           UBS Securities Australia Ltd         Buy         2,343         1,00           UBS Securities Australia Ltd         Buy		UBS Securities Australia Ltd	Buy	o		Codings
UBS Securities Australia Ltd         Buy         2,661         313           UBS Securities Australia Ltd         Buy         7,7         9           UBS Securities Australia Ltd         Buy         77         9           UBS Securities Australia Ltd         Buy         7,7         9           UBS Securities Australia Ltd         Buy         2,663         1,182           UBS Securities Australia Ltd         Buy         10,047         1,182           UBS Securities Australia Ltd         Buy         1,684         1,156           UBS Securities Australia Ltd         Buy         1,037         1,2570           UBS Securities Australia Ltd         Buy         1,037         1,284           UBS Securities Australia Ltd         Buy         1,037         1,284           UBS Securities Australia Ltd         Buy         1,544         1,484           UBS Securities Australia Ltd         Buy         2,779         1,848           UBS Securities Australia Ltd         Buy         2,779         1,548           UBS Securities Australia Ltd         Buy         2,779         1,548           UBS Securities Australia Ltd         Buy         2,776         2,849           UBS Securities Australia Ltd         Buy		UBS Securities Australia Ltd	Buy	15 955		i i
UBS Securities Australia Ltd         Buy         Cod         71           UBS Securities Australia Ltd         Buy         272         32           UBS Securities Australia Ltd         Buy         272         32           UBS Securities Australia Ltd         Buy         286.637         1,682           UBS Securities Australia Ltd         Buy         286.637         1,589           UBS Securities Australia Ltd         Buy         286.637         1,750           UBS Securities Australia Ltd         Buy         1,687         1,750           UBS Securities Australia Ltd         Buy         1,750         1,750           UBS Securities Australia Ltd         Buy         1,7764         1,886           UBS Securities Australia Ltd         Buy         2,8447         6,402           UBS Securities Australia Ltd         Buy         2,343         1,764           UBS Securities Australia Ltd         Buy         2,343         2,745           UBS Securities Australia Ltd         Buy         2,333         2,745           UBS Securities Australia Ltd         Buy         2,333         2,745           UBS Securities Australia Ltd         Buy         2,333         2,745           UBS Securities Australia Ltd         Buy<		UBS Securities Australia Ltd	Buy	2 661		Oldillary
UBS Securities Australia Ltd         Bby         77         91           UBS Securities Australia Ltd         Buy         10047         32           UBS Securities Australia Ltd         Buy         10047         3,599           UBS Securities Australia Ltd         Buy         10047         1,182           UBS Securities Australia Ltd         Buy         10047         1,270           UBS Securities Australia Ltd         Buy         1,034         1,270           UBS Securities Australia Ltd         Buy         1,034         1,034           UBS Securities Australia Ltd         Buy         1,034         1,034           UBS Securities Australia Ltd         Buy         7,769         5,441           UBS Securities Australia Ltd         Buy         7,769         5,441           UBS Securities Australia Ltd         Buy         7,769         5,441           UBS Securities Australia Ltd         Buy         2,333         2,745           UBS Securities Australia Ltd         Buy         2,376         5,447           UBS Securities Australia Ltd         Buy         2,376         3,43           UBS Securities Australia Ltd         Buy         2,276         1,09           UBS Securities Australia Ltd         Buy		UBS Securities Australia Ltd	Buy	604	212	Orginary
UBS Securities Australia Ltd         Buy         277         37           UBS Securities Australia Ltd         Buy         1003         3           UBS Securities Australia Ltd         Buy         28,637         1,582           UBS Securities Australia Ltd         Buy         28,637         1,582           UBS Securities Australia Ltd         Buy         15,946         1,270           UBS Securities Australia Ltd         Buy         1,5946         1,276           UBS Securities Australia Ltd         Buy         1,5946         1,276           UBS Securities Australia Ltd         Buy         1,037         1,284           UBS Securities Australia Ltd         Buy         1,037         1,284           UBS Securities Australia Ltd         Buy         2,343         1,442           UBS Securities Australia Ltd         Buy         2,333         2,745           UBS Securities Australia Ltd         Buy         2,333         2,745           UBS Securities Australia Ltd         Buy         2,506         3,00           UBS Securities Australia Ltd         Buy         2,276         3,00           UBS Securities Australia Ltd         Buy         2,276         3,00           UBS Securities Australia Ltd         Buy </td <td></td> <td>UBS Securities Australia I td</td> <td>Roy</td> <td>100</td> <td></td> <td>Crdinary</td>		UBS Securities Australia I td	Roy	100		Crdinary
UBS Securities Australia Ltd         Buy         10047         1,182           UBS Securities Australia Ltd         Buy         10,047         1,182           UBS Securities Australia Ltd         Buy         28,657         3,599           UBS Securities Australia Ltd         Buy         1,068         1,757           UBS Securities Australia Ltd         Buy         1,037         1,757           UBS Securities Australia Ltd         Buy         1,037         1,764           UBS Securities Australia Ltd         Buy         1,037         1,764           UBS Securities Australia Ltd         Buy         1,037         1,764           UBS Securities Australia Ltd         Buy         1,764         1,848           UBS Securities Australia Ltd         Buy         1,764         1,848           UBS Securities Australia Ltd         Buy         2,843         3,741           UBS Securities Australia Ltd         Buy         2,879         3,745           UBS Securities Australia Ltd         Buy         2,879         3,745           UBS Securities Australia Ltd         Buy         2,879         3,741           UBS Securities Australia Ltd         Buy         2,276         1,09           UBS Securities Australia Ltd	ı	UBS Securities Australia Ind	But	/	6	Ordinary
UBS Securities Australia Ltd         Buy         10.047         1.82           UBS Securities Australia Ltd         Buy         26.57         3.59           UBS Securities Australia Ltd         Buy         3.6918         4.108           UBS Securities Australia Ltd         Buy         1.087         1.2570           UBS Securities Australia Ltd         Buy         1.087         1.384           UBS Securities Australia Ltd         Buy         1.764         1.384           UBS Securities Australia Ltd         Buy         1.576         1.384           UBS Securities Australia Ltd         Buy         1.576         1.384           UBS Securities Australia Ltd         Buy         1.576         1.48           UBS Securities Australia Ltd         Buy         1.578         1.48           UBS Securities Australia Ltd         Buy         2.343         1.76           UBS Securities Australia Ltd         Buy         2.349         1.00           UBS Securities Australia Ltd         Buy         2.360         1.00           UBS Securities Australia Ltd         Buy         2.360         1.00           UBS Securities Australia Ltd         Buy         2.350         1.00           UBS Securities Australia Ltd         Buy	1	UBS Securities Australia Ind	Ring	2//2	32	Ordinary
UBS Securities Australia Ltd         Buy         26.53.7         1.88.7           UBS Securities Australia Ltd         Buy         26.63.7         3.108           UBS Securities Australia Ltd         Buy         1.03.7         1.25.70           UBS Securities Australia Ltd         Buy         1.03.7         1.22.70           UBS Securities Australia Ltd         Buy         5.44.7         6.40.2           UBS Securities Australia Ltd         Buy         5.44.7         6.40.2           UBS Securities Australia Ltd         Buy         7.769         9.4           UBS Securities Australia Ltd         Buy         7.769         9.4           UBS Securities Australia Ltd         Buy         7.769         9.3           UBS Securities Australia Ltd         Buy         7.769         9.3           UBS Securities Australia Ltd         Buy         7.769         9.3           UBS Securities Australia Ltd         Buy         7.769         9.3           UBS Securities Australia Ltd         Buy         7.769         9.3           UBS Securities Australia Ltd         Buy         2.376         9.0           UBS Securities Australia Ltd         Buy         2.260         10.0           UBS Securities Australia Ltd <td< td=""><td>ı</td><td>UBS Securities Australia Ind</td><td>Bur</td><td>26</td><td>3</td><td>Ordinary</td></td<>	ı	UBS Securities Australia Ind	Bur	26	3	Ordinary
URS Securities Australia Ltd         Buy         28.637         3.959           URS Securities Australia Ltd         Buy         10.68.45         4.106           URS Securities Australia Ltd         Buy         10.68.45         1.25.70           URS Securities Australia Ltd         Buy         1.03.7         1.72           URS Securities Australia Ltd         Buy         5.41.7         6.41.7           URS Securities Australia Ltd         Buy         5.47.69         9.14           URS Securities Australia Ltd         Buy         7.769         9.14           URS Securities Australia Ltd         Buy         7.769         9.14           URS Securities Australia Ltd         Buy         7.267.98         3.38           URS Securities Australia Ltd         Buy         7.267.98         3.48           URS Securities Australia Ltd         Buy         7.269         3.24           URS Securities Australia Ltd         Buy         2.28         3.54           URS Securities Australia Ltd         Buy         2.260         3.00           URS Securities Australia Ltd         Buy         2.260         3.00           URS Securities Australia Ltd         Buy         2.260         3.00           URS Securities Australia Ltd	1	I IRC Committee Australia Ital	ing.	10,047	1,182	Ordinary
USS Securities Australia Ltd         Buy         3,4918         4,108           USS Securities Australia Ltd         Buy         15,946         1,270           USS Securities Australia Ltd         Buy         1,594         1,876           USS Securities Australia Ltd         Buy         5,447         6,402           USS Securities Australia Ltd         Buy         5,447         6,402           USS Securities Australia Ltd         Buy         2,447         6,402           USS Securities Australia Ltd         Buy         2,759         9,14           USS Securities Australia Ltd         Buy         2,333         2,745           USS Securities Australia Ltd         Buy         2,333         2,745           USS Securities Australia Ltd         Buy         2,343         2,745           USS Securities Australia Ltd         Buy         2,38         3,54           USS Securities Australia Ltd         Buy         2,28         3,54           USS Securities Australia Ltd         Buy         2,28         3,54           USS Securities Australia Ltd         Buy         2,26         5,03           USS Securities Australia Ltd         Buy         2,26         5,03           USS Securities Australia Ltd         Buy <td>1</td> <td>Dispersional Control of the Control</td> <td>Buy</td> <td>28,637</td> <td>3,369</td> <td>Ordinary</td>	1	Dispersional Control of the Control	Buy	28,637	3,369	Ordinary
UBS Securities Australia Ltd         Buy         10.55 MG           UBS Securities Australia Ltd         Buy         1.570           UBS Securities Australia Ltd         Buy         1.037           UBS Securities Australia Ltd         Buy         1.770           UBS Securities Australia Ltd         Buy         1.770           UBS Securities Australia Ltd         Buy         7.769           UBS Securities Australia Ltd         Buy         7.769           UBS Securities Australia Ltd         Buy         7.769           UBS Securities Australia Ltd         Buy         7.769           UBS Securities Australia Ltd         Buy         7.769           UBS Securities Australia Ltd         Buy         2.343           UBS Securities Australia Ltd         Buy         2.260           UBS Securities Australia Ltd         Buy         4.276           UBS Securities Australia Ltd         Buy         4.276           UBS Securities Australia Ltd         Buy         4.276           UBS Securities Australia Ltd         Buy         2.550           UBS Securities Australia Ltd         Buy         2.250           UBS Securities Australia Ltd         Buy         2.550           UBS Securities Australia Ltd         Buy		190 Commission Australia Liu	Buy	34,918		Ordinary
UBS Securities Australia Ltd         Buy         15,946         1,876           UBS Securities Australia Ltd         Buy         1,037         1,122           UBS Securities Australia Ltd         Buy         5,447         6,402           UBS Securities Australia Ltd         Buy         5,447         6,402           UBS Securities Australia Ltd         Buy         7,709         1,184           UBS Securities Australia Ltd         Buy         7,709         1,484           UBS Securities Australia Ltd         Buy         7,309         3,48           UBS Securities Australia Ltd         Buy         2,343         1,14           UBS Securities Australia Ltd         Buy         3,96         1,00           UBS Securities Australia Ltd         Buy         3,96         1,00           UBS Securities Australia Ltd         Buy         2,26         3,00           UBS Securities Australia Ltd         Buy         2,250         3,00           UBS Securities Australia Ltd         Buy         2,550         3,00           UBS Securities Australia Ltd         Buy         2,550         3,00           UBS Securities Australia Ltd         Buy         2,550         3,00           UBS Securities Australia Ltd         Buy	L	Line Securities Australia Ltd	Buy	106,845		Ordinary
USS Securities Australia Ltd         Buy         1,724         1,224           USS Securities Australia Ltd         Buy         1,726         1,348           USS Securities Australia Ltd         Buy         56,417         6,402           USS Securities Australia Ltd         Buy         7,769         1,848           USS Securities Australia Ltd         Buy         7,769         1,848           USS Securities Australia Ltd         Buy         7,769         1,848           USS Securities Australia Ltd         Buy         7,769         1,745           USS Securities Australia Ltd         Buy         1,990         2,343           USS Securities Australia Ltd         Buy         315         317           USS Securities Australia Ltd         Buy         315         315           USS Securities Australia Ltd         Buy         4,276         100           USS Securities Australia Ltd         Buy         4,276         100           USS Securities Australia Ltd         Buy         2,550         300           USS Securities Australia Ltd         Buy         2,550         300           USS Securities Australia Ltd         Buy         2,550         300           USS Securities Australia Ltd         Buy	1001	Upo Securities Australia Ltd	Buy	15,946		Ordinary
URS Securities Australia Ltd         Blwy         1,384           UBS Securities Australia Ltd         Buy         54,477         6,402           UBS Securities Australia Ltd         Buy         15,708         15,48           UBS Securities Australia Ltd         Buy         7,679         914           UBS Securities Australia Ltd         Buy         7,879         914           UBS Securities Australia Ltd         Buy         7,879         3,78           UBS Securities Australia Ltd         Buy         7,879         3,43           UBS Securities Australia Ltd         Buy         2,34         1,10           UBS Securities Australia Ltd         Buy         2,86         1,10           UBS Securities Australia Ltd         Buy         4,276         5,03           UBS Securities Australia Ltd         Buy         2,550         3,00           UBS Securities Australia Ltd         Buy         2,550         3,00           UBS Securities Australia Ltd         Buy         2,550         3,00           UBS Securities Australia Ltd         Buy         2,550         3,00           UBS Securities Australia Ltd         Buy         2,550         3,00           UBS Securities Australia Ltd         Buy         2,550	-NOV-14	UBS Securities Australia Ltd	Buy	1,037		Ordinary
UBS Securities Australia Ltd         Bluy         54,47         6,402           UBS Securities Australia Ltd         Buy         57,695         1,846           UBS Securities Australia Ltd         Buy         7,769         91,4           UBS Securities Australia Ltd         Buy         23,333         2,745           UBS Securities Australia Ltd         Buy         23,333         2,745           UBS Securities Australia Ltd         Buy         53         1,1           UBS Securities Australia Ltd         Buy         2,96         3,34           UBS Securities Australia Ltd         Buy         85,0         1,00           UBS Securities Australia Ltd         Buy         4,276         503           UBS Securities Australia Ltd         Buy         2,56         1,00           UBS Securities Australia Ltd         Buy         2,50         1,00           UBS Securities Australia Ltd         Buy         2,50         1,00           UBS Securities Australia Ltd         Buy         2,50         1,00           UBS Securities Australia Ltd         Buy         2,50         1,00           UBS Securities Australia Ltd         Buy         2,50         3,44           UBS Securities Australia Ltd         Buy	-1404-14	UBS Securities Australia Ltd	Buy	11,764		Ordinary
URS Securities Australia Ltd         BM         115708         1588           URS Securities Australia Ltd         BMy         7759         914           URS Securities Australia Ltd         BMy         7369         914           URS Securities Australia Ltd         BMy         7379         755           URS Securities Australia Ltd         BMy         19.904         2.343           URS Securities Australia Ltd         BMy         288         3.73           URS Securities Australia Ltd         BMy         2.88         3.75           URS Securities Australia Ltd         BMy         4.276         5.03           URS Securities Australia Ltd         BMy         4.276         5.03           URS Securities Australia Ltd         BMy         4.276         5.03           URS Securities Australia Ltd         BMy         4.276         5.03           URS Securities Australia Ltd         BWy         2.550         3.00           URS Securities Australia Ltd         BWy         2.550         3.00           URS Securities Australia Ltd         BWy         2.652         3.10           URS Securities Australia Ltd         BWy         2.652         3.00           URS Securities Australia Ltd         BWy <t< td=""><td>NOv-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>54,417</td><td></td><td>Ordinary</td></t<>	NOv-14	UBS Securities Australia Ltd	Buy	54,417		Ordinary
URS Securities Australia Ltd         Bby         7769         914           URS Securities Australia Ltd         Buy         28,798         3,388           URS Securities Australia Ltd         Buy         23,333         2,745           URS Securities Australia Ltd         Buy         19,904         2,343           URS Securities Australia Ltd         Buy         315         37           URS Securities Australia Ltd         Buy         80         10           URS Securities Australia Ltd         Buy         80         10           URS Securities Australia Ltd         Buy         4,276         503           URS Securities Australia Ltd         Buy         2,550         30           URS Securities Australia Ltd         Buy         2,550         30           URS Securities Australia Ltd         Buy         2,550         30           URS Securities Australia Ltd         Buy         2,652         30           URS Securities Australia Ltd         Buy         2,652         31           URS Securities Australia Ltd         Buy         2,652         34           URS Securities Australia Ltd         Buy         2,652         34           URS Securities Australia Ltd         Buy         2,652	-NOV-14	UBS Securities Australia Ltd	Buy	15,708	1,848	Ordinary
URS Securities Australia Ltd         Bby         28,789         3.388           URS Securities Australia Ltd         Buy         23,333         2,745           URS Securities Australia Ltd         Buy         19,904         2,343           URS Securities Australia Ltd         Buy         19,904         2,343           URS Securities Australia Ltd         Buy         315         33           URS Securities Australia Ltd         Buy         2,80         100           URS Securities Australia Ltd         Buy         4,276         50           URS Securities Australia Ltd         Buy         4,276         100           URS Securities Australia Ltd         Buy         2,550         30           URS Securities Australia Ltd         Buy         2,550         30           URS Securities Australia Ltd         Buy         2,652         31           URS Securities Australia Ltd         Buy         2,652         34           URS Securities Australia Ltd         Buy         2,652         34           URS Securities Australia Ltd         Buy         2,652         34           URS Securities Australia Ltd         Buy         2,652         34           URS Securities Australia Ltd         Buy         383 <td>-NOV-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>694,7</td> <td>914</td> <td>Ordinary</td>	-NOV-14	UBS Securities Australia Ltd	Buy	694,7	914	Ordinary
UBS Securities Australia Ltd         Buy         23,333         2,745           UBS Securities Australia Ltd         Buy         93         11           UBS Securities Australia Ltd         Buy         19,904         2,343           UBS Securities Australia Ltd         Buy         315         37           UBS Securities Australia Ltd         Buy         850         100           UBS Securities Australia Ltd         Buy         4,276         503           UBS Securities Australia Ltd         Buy         4,276         30           UBS Securities Australia Ltd         Buy         2,550         300           UBS Securities Australia Ltd         Buy         2,652         312           UBS Securities Australia Ltd         Buy         2,652         549           UBS Securities Australia Ltd         Buy         2,652         549           UBS Securities Australia Ltd         Buy         2,652         549           UBS Securities Australia Ltd         Buy         383	-Nov-14	UBS Securities Australia Ltd	Buy	28,798	3388	Ordinary
UBS Securities Australia Ltd         Buy         93         11           UBS Securities Australia Ltd         Buy         19,904         2,343           UBS Securities Australia Ltd         Buy         315         37           UBS Securities Australia Ltd         Buy         60         100           UBS Securities Australia Ltd         Buy         60         100           UBS Securities Australia Ltd         Buy         2,50         30           UBS Securities Australia Ltd         Buy         2,50         30           UBS Securities Australia Ltd         Buy         2,50         30           UBS Securities Australia Ltd         Buy         2,602         30           UBS Securities Australia Ltd         Buy         2,602         30           UBS Securities Australia Ltd         Buy         2,602         30           UBS Securities Australia Ltd         Buy         2,602         30           UBS Securities Australia Ltd         Buy         2,602         30           UBS Securities Australia Ltd         Buy         2,602         30           UBS Securities Australia Ltd         Buy         2,602         30           UBS Securities Australia Ltd         Buy         383         4,50 <td>-Nov-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>23,333</td> <td>2 745</td> <td>Ordinary</td>	-Nov-14	UBS Securities Australia Ltd	Buy	23,333	2 745	Ordinary
UBS Securities Australia (14d)         Buy         19 904         2.3.43           UBS Securities Australia (14d)         Buy         19 904         2.3.43           UBS Securities Australia (14d)         Buy         286         150           UBS Securities Australia (14d)         Buy         4.276         100           UBS Securities Australia (14d)         Buy         4.276         503           UBS Securities Australia (14d)         Buy         2.580         300           UBS Securities Australia (14d)         Buy         2.580         1,058           UBS Securities Australia (14d)         Buy         2.652         333         1,058           UBS Securities Australia (14d)         Buy         2.061         2,449         540         540           UBS Securities Australia (14d)         Buy         2.062         333         1,058         54           UBS Securities Australia (14d)         Buy         2.061         54         54         54           UBS Securities Australia (14d)         Buy         2.061         54         54         54           UBS Securities Australia (14d)         Buy         2.061         54         54         54           UBS Securities Australia (14d)         Buy         2.0	-Nov-14	UBS Securities Australia Ltd	Buy	88	11	Ordinary
UBS Securities Australia Ltd         Buy         315         77           UBS Securities Australia Ltd         Buy         850         100           UBS Securities Australia Ltd         Buy         4,276         503           UBS Securities Australia Ltd         Buy         4,276         503           UBS Securities Australia Ltd         Buy         2,550         30           UBS Securities Australia Ltd         Buy         2,550         30           UBS Securities Australia Ltd         Buy         2,550         30           UBS Securities Australia Ltd         Buy         2,652         31           UBS Securities Australia Ltd         Buy         2,652         34           UBS Securities Australia Ltd         Buy         2,652         34           UBS Securities Australia Ltd         Buy         2,652         34           UBS Securities Australia Ltd         Buy         2,652         54           UBS Securities Australia Ltd         Buy         383         45           UBS Securities Australia Ltd         Buy         69         1           UBS Securities Australia Ltd         Buy         69         1           UBS Securities Australia Ltd         Buy         69         1     <	-Nov-14	UBS Securities Australia Ltd	Buy	19 904		Contract
UBS Securities Australia Ltd         Buy         298         35           UBS Securities Australia Ltd         Buy         298         35           UBS Securities Australia Ltd         Buy         4,276         100           UBS Securities Australia Ltd         Buy         2,550         30           UBS Securities Australia Ltd         Buy         2,550         300           UBS Securities Australia Ltd         Buy         2,550         300           UBS Securities Australia Ltd         Buy         2,652         312           UBS Securities Australia Ltd         Buy         2,652         312           UBS Securities Australia Ltd         Buy         2,652         34           UBS Securities Australia Ltd         Buy         2,652         34           UBS Securities Australia Ltd         Buy         2,652         54           UBS Securities Australia Ltd         Buy         383         45           UBS Securities Australia Ltd         Buy         383         45           UBS Securities Australia Ltd         Buy         40         10           UBS Securities Australia Ltd         Buy         40         10           UBS Securities Australia Ltd         Buy         40         10	Nov-14	UBS Securities Australia Ltd	Buy	315	-	, and a
UBS Securities Australia Ltd         Buy         600         100           UBS Securities Australia Ltd         Buy         4,276         503           UBS Securities Australia Ltd         Buy         2,550         300           UBS Securities Australia Ltd         Buy         1,326         1,556           UBS Securities Australia Ltd         Buy         9,333         1,098           UBS Securities Australia Ltd         Buy         2,662         349           UBS Securities Australia Ltd         Buy         2,662         312           UBS Securities Australia Ltd         Buy         2,682         312           UBS Securities Australia Ltd         Buy         2,682         349           UBS Securities Australia Ltd         Buy         383         445           UBS Securities Australia Ltd         Buy         383         45           UBS Securities Australia Ltd         Buy         69         1           UBS Securities Australia Ltd         Buy         69         1           UBS Securities Australia Ltd         Buy         69         1           UBS Securities Australia Ltd         Buy         600         1	-Nov-14	UBS Securities Australia Ltd	Buv	300		Cidinary
UBS Securities Australia Ltd         Bby         4,276         103           USS Securities Australia Ltd         Buy         2,276         503           UBS Securities Australia Ltd         Buy         2,550         30           UBS Securities Australia Ltd         Buy         1,526         1,566           UBS Securities Australia Ltd         Buy         2,652         313           UBS Securities Australia Ltd         Buy         2,652         312           UBS Securities Australia Ltd         Buy         2,652         342           UBS Securities Australia Ltd         Buy         2,652         342           UBS Securities Australia Ltd         Buy         383         45           UBS Securities Australia Ltd         Buy         383         45           UBS Securities Australia Ltd         Buy         383         45           UBS Securities Australia Ltd         Buy         409         71	-Nov-14	UBS Securities Australia Ltd	Buv	CHO		Orginary
UBS Securities Australia Ltd         Buy         2.90         30.3           UBS Securities Australia Ltd         Buy         2.550         30.0           UBS Securities Australia Ltd         Buy         1,326         1,556           UBS Securities Australia Ltd         Buy         9,333         1,058           UBS Securities Australia Ltd         Buy         2,652         31.2           UBS Securities Australia Ltd         Buy         2,652         31.2           UBS Securities Australia Ltd         Buy         2,652         34.9           UBS Securities Australia Ltd         Buy         45         45           UBS Securities Australia Ltd         Buy         383         45           UBS Securities Australia Ltd         Buy         60         1           UBS Securities Australia Ltd         Buy         60         1           UBS Securities Australia Ltd         Buy         60         1           UBS Securities Australia Ltd         Buy         60         1	-Nov-14	UBS Securities Australia Ltd	Buv	375 /	3 5	Ordinary
UBS Securities Australia Ltd         Bay         2.550         300           UBS Securities Australia Ltd         Buy         1.556         300           UBS Securities Australia Ltd         Buy         9,333         1,058           UBS Securities Australia Ltd         Buy         2,652         312           UBS Securities Australia Ltd         Buy         2,0817         2,449           UBS Securities Australia Ltd         Buy         56         54           UBS Securities Australia Ltd         Buy         383         45           UBS Securities Australia Ltd         Buy         60         1           UBS Securities Australia Ltd         Buy         60         1           UBS Securities Australia Ltd         Buy         60         1           UBS Securities Australia Ltd         Buy         60         1	-Nov-14	UBS Securities Australia IId	Biry	4,270	503	Ordinary
LBS Securities Australia Ltd         Boy         13,226         300           USS Securities Australia Ltd         Buy         13,226         1,556           USS Securities Australia Ltd         Buy         2,652         317           US Securities Australia Ltd         Buy         2,652         312           USS Securities Australia Ltd         Buy         2,081         2,449           USS Securities Australia Ltd         Buy         383         4,5           USS Securities Australia Ltd         Buy         61         1           USS Securities Australia Ltd         Buy         62         54           USS Securities Australia Ltd         Buy         61         1	-Nov-14	UBS Securities Australia Ltd	Buv	0 2 5 5	- 000	Ordinary
USE Securities Australia Ltd         Buy         1,556           USE Securities Australia Ltd         Buy         9,333         1,058           USE Securities Australia Ltd         Buy         2,652         312           USE Securities Australia Ltd         Buy         2,0817         2,449           USE Securities Australia Ltd         Buy         459         54           USE Securities Australia Ltd         Buy         45         45           USE Securities Australia Ltd         Buy         60         1           USE Securities Australia Ltd         Buy         60         1	-Nov-14	UBS Securities Australia Itd	Riv.	2000,5	300	Orginary
UBS Securities Australia Ltd         Buy         2,552         1,088           UBS Securities Australia Ltd         Buy         2,652         1,088           UBS Securities Australia Ltd         Buy         2,0817         2,449           UBS Securities Australia Ltd         Buy         383         45           UBS Securities Australia Ltd         Buy         45         45           UBS Securities Australia Ltd         Buy         67         1           UBS Securities Australia Ltd         Buy         67         1           UBS Securities Australia Ltd         Ruy         67         1	Nov-14	UBS Securities Australia Ltd	Brie	277'61	1,556	Ordinary
URS Securities Australia Ltd         Buy         2,652         31.2           URS Securities Australia Ltd         Buy         20,817         2,449           USS Securities Australia Ltd         Buy         459         54           UBS Securities Australia Ltd         Buy         383         45           UBS Securities Australia Ltd         Buy         383         45           UBS Securities Australia Ltd         Buy         9         1           UBS Securities Australia Ltd         Buy         65         1	Nov-14	HDC Countities Australia 14d	ouy o	2,333	1,098	Ordinary
USS Securities Australia Ltd         Buy         2.0817         2.449           UBS Securities Australia Ltd         Buy         4.59         5.4           UBS Securities Australia Ltd         Buy         383         4.5           UBS Securities Australia Ltd         Buy         383         4.5           UBS Securities Australia Ltd         Buy         9         1           UBS Securities Australia Ltd         Ruy         6.0         1	Now.14	Line Countries Australia Ind	óna	7597	312	Ordinary
UBS Securities Australia Ltd         Buy         459         54           UBS Securities Australia Ltd         Buy         383         45           UBS Securities Australia Ltd         Buy         9         1           UBS Securities Australia Ltd         Buy         occ         1	Nov-14	11DC Consisting Australia 14d	Buy	718'02	2,449	Ordinary
100	Nov. 14	LIBC Committee Australia 14d	, and	459	54	Ordinary
UBS Securities Australia Ltd Buy 9 1	1	Obsider Australia Ltd	Buy	383	45	Ordinary
UBS Securities Australia Ltd	Nov-14	UBS Securities Australia Ltd	Buy	0	-	Ordinary
ממן הרייתונים אתמושות דנת	Nov-14	UBS Securities Australia Ltd	Buv	850	001	Ordinary

26-New-14 UBS Securities Australia (ut 26-New-14 UBS Securities Australia (ut 26-New-14 UBS Securities Australia (ut 26-New-14 UBS Securities Australia (ut 26-New-14 UBS Securities Australia (ut 26-New-14 UBS Securities Australia (ut 26-New-14 UBS Securities Australia (ut 26-New-14 UBS Securities Australia (ut 26-New-14 UBS Securities Australia (ut 26-New-14 UBS Securities Australia (ut 26-New-14 UBS Securities Australia (ut 26-New-14 UBS Securities Australia (ut 26-New-14 UBS Securities Australia (ut 26-New-14 UBS Securities Australia (ut 26-New-14 UBS Securities Australia (ut 26-New-14 UBS Securities Australia (ut 26-New-14 UBS Securities Australia (ut 26-New-14 UBS Securities Australia (ut 26-New-14 UBS Securities Australia (ut 27-New-14 UBS UBS UBS UBS UBS UBS UBS UBS UBS UBS	UBS Securities Australia (td	Sell	132,969		
1111111111	ACTIVITIES QUICTED 140		375.0		
1	Control Control	Seil	616,2		Ordinary
	ecurties Australia Ltd	Sell	32,763		Ordinary
	UBS Securities Australia Ltd	Buy	13,065		Ordinary
1	ecurties Australia Ltd	Sel	358,545		Ordinary
	UBS Securities Australia Ltd	Buy	90,405		Ordinary
	UBS AG Australia Branch	Sell	999,362		Ordinary
	G Australia Branch	Sell	161,334		Ordinary
11111	veann Management Australia Ltd	Sell	13,899		Ordinary
	ecurities Australia Ltd	Sell	870,000		Ordinary
11111	ecuriles Australia Ltd	sei	5,925		
	ecurines Australia Ltd	liac o	5,246		
111	Scurings Australia Ltd	on.	38,463		
П	looal Asset Management	Buy	1,813		
1	Obs securities Australia Ltd	Sell	14,556		
	ecurities Australia Ltd	guy	186,844	22,060	
Ť	UBS Securities Australia Ltd	Sell	125,789		
Ť	JBS Securities Australia Ltd	Sell	150,867		
ı	Job Securities Australia Ltd	Sell	293,204		
1	ecurities Australia Ltd	Sell	301,200		
ı	scurities Australia Ltd	Buy	889'58		
ı	Upo Securites Australia Lto	) sell	112,050	(13,232)	Ordinary
1	curiles Australia Ltd	Buly	111,968		
1	commes Australia Ltd	Seil	402,170	(47,000)	
ı	curries Australia Ltd	auy	47,322		
	curries Australia Ltd	Buy	148,254		
1	curries Australia Ltd	Buy	2,339		
7-140v-14 UBS Se	curries Austraria Ltd	Sell	51,103	(5,901)	Ordinary
	curities Australia Ltd	Buy	686,909	80,790	Ordinary
1	JBS AG, London Branch	Buy	364,403	34,370	Ordinary
	5, London Branch	Sell	293,204	(34,370)	Ordinary
	5 Australia Branch	Sell	1,101,230	(130,000)	Ordinary
1	UBS AG Australia Branch	Buy	847,186	100,000	Ordinary
-	5 Australia Branch	Buy	629,751	18,497	Ordinary
	5 Australia Branch	Buy	588.472	69 210	Ordinary
İ	5 Australia Branch	Buy	58.249	6814	Ordinan
	5 Australia Branch	Seli	49 886	(5,869)	Ordinany
	5 Australia Branch	Buy	7 124 987	365,010	ordinary.
	UBS AG Australia Branch	Buy	374.835	43 782	Ordinary
	a Australia Branch	Buy	780.867	917 708	Ordinary
	UBS Wealth Management Australia Ltd	Buy	3.638	477	Ordinary
	UBS Wealth Management Australia Ltd	Transfer-Out	4/Z	(875)	Ordinary
_	UBS Securities Australia Ltd	Buy	847 000	100 000	Ordinary
	curities Australia Ltd	Buy	75 571	3 1 2 6	Coding
7-Nov-14 UBS Sec	UBS Securities Australia Ltd	Buy	1010	2,120	Cidillaly
7-Nov-14 UBS Sec	curities Australia Ltd	Buy	500 M	1 766	Guinary
San	Securities Australia Ltd	Buy	345	00/1	Ordinary
UBS	UBS Securities Australia Ltd	Buy	047 047	67	Ordinary
7-Nov-14 UBS Sec	UBS Securities Australia Ltd	Buv	1,630	2,516	Orginary
Ī	UBS Securities Australia Ltd	Riv	000,1	761	Ordinary
Ī	Counties Australia 11d	D. 1	3,150	3/1	Ordinary
1	or rices Australia Liu	and.	2,547	300	Ordinary
ı	urines Australia Ltd	ana	1,206	142	Ordinary
İ	Onlines Australia LIO	ana	2,572	303	Ordinary
7 Nov-14	Obsideratives Australia Lto	Buy	26,990	3,179	Ordinary
l	Urines Australia Ltd	Ang	298	35	Ordinary
	JBS Securities Australia Ltd	Buy	4,990	282	Ordinary
i	urities Australia Ltd	Buy	28,475	3,350	Ordinary
	urties Australia Ltd	Buy	35,471	4,173	Ordinary
7-Nov-14 UBS Sect	UBS Securities Australia Ltd	Buy	35,471	4,173	Ordinary
	urities Australia Ltd	Buy	31.603	3 718	Ordinary
	urities Australia Ltd	Buy	32 334	708.5	Ordinan
	Securities Australia Ltd	Buy	33.856	2 083	Ordinary
	UBS Securities Australia Ltd	Buy	32,001	500.0	Ordinary
27-Nov-14 UBS Sect	UBS Securities Australia Ltd	Buv	00000	0000	Countries
l	urities Australia Ltd	Buv	966	2000	Colliday
ı	urities Australia I tol	Rem	001	0	Grainary
	mides Australia Lid	ĥ.	34	4	Ordinary
	military Australia   140	, ind	4,124	486	Ordinary
١	unities Australia Ltd	gn)	93	11	Ordinary
1	unties Australia Ltd	Buy	3,880	457	Ordinary
-	urities Australia Ltd	Buy	5,493	647	Ordinary
	urities Australia Ltd	Buy	17,659	2,080	Ordinary
	urities Australia Ltd	Buy	6.384	757	Ordinary
	rritiae Australia 14d	Bin			Common

Securities Australia Ltd	Buy	2,550	300	i
UBS Securities Australia Ltd	Buy	7,319	861	Ordinary
Australia Ltd	Buy	1,573	185	Ordinary
Australia Ltd	Buy	28,934	3,404	Ordinary
Australia Ltd	Buy	644	76	Ordinary
s Australia Ltd	Buy	73,506	8,663	Ordinary
s Australia Ltd	Buy	69,017	8,134	Ordinary
Australia Ltd	Buy	42,985	2,066	Ordinary
Australia Ltd	Buy	3,241	382	Ordinary
Australia Ltd	Buy	24,708	2,912	Ordinary
Australia Ltd	Buy	2,758	325	Ordinary
Australia Ltd	Buy	969	75	Ordinary
Australia Ltd	Buy	42	5	Ordinary
Australia Ltd	Buy	42,985	2,066	Ordinary
UBS Securities Australia Ltd	Buy	839	66	Ordinary
Australia Ltd	Buy	4,586	543	Ordinary
UBS Securities Australia Ltd	Buy	2,665	315	Ordinary
Australia Ltd	Buy	79,297	3,463	Ordinary
Australia Ltd	Buy	30,735	3,633	Ordinary
Australia Ltd	Buy	3,765	445	Ordinary
UBS Securities Australia Ltd	Buy	2,580	305	Ordinary
Australia Ltd	Buy	2.538	300	Ordinan
Australia Ltd	Buv	1 227	145	Ordinary
Australia I tol	Bit	4 040	C#1	Cidinary
LIBS Securities Australia Ltd	Birc	nth'i	080	Ordinary
ustralia Ltd	end o	6,49/	768	Ordinary
tusti alia Lito	pon	697'1	150	Ordinary
ustralia Ltd	Buy	1,982	234	Ordinary
ustralia Ltd	Buy	4,201	496	Ordinary
ustralia Ltd	Buy	13,967	1,649	Ordinary
ustralia Ltd	Buy	2,541	300	Ordinary
ustralia Ltd	Buy	14,898	1,761	Ordinary
ustralia Ltd	Buy	7,563	895	Ordinary
ustralia Ltd	Buy	1,741	206	Ordinary
ustralia Ltd	Buy	811	96	Ordinary
ustralia Ltd	Buy	1,039	123	Ordinary
ustralia Ltd	Виу	2,308	273	Ordinary
ustralia Ltd	Buy	769	91	Ordinary
vustralia Ltd	Buy	515	19	Ordinary
sustralia Ltd	Sell	10,537	(1.244)	Ordinary
ustralia Ltd	Sell	78,856	(9,310)	Ordinary
Australia Ltd	Buy	11,409	1.347	Ordinary
sustralia Ltd	Buy	9,241	1001	Ordinary
ustralia Ltd	Buy	414 259	48 909	Ordinan
ustralia Ltd	Buy	277.067	32 711	Ordinary
ustralia Ltd	lleS	896.400	(009 6D)	Ordinary
UBS Securities Australia Ltd	Sell	42,500	(2.000)	Ordinary
	Enter Into Derivative			
UBS Securities Australia Ltd	Contract	NA	(50,000,000)	Options
JBS Securities Australia I1d	Enter Into Derivative Contract	× N	1000	
	Enter loto Derivativo	W.Y.	(nng/ss)	Options
UBS Securities Australia Ltd	Contract	N/A	009'66	Options
UBS Securities Australia Ltd	Enter Into Derivative Contract	N/A	(500,000)	Options
V 2000	Enter Into Derivative			
Australia Ltd	Contract	N/A	20,000,000	Options
UBS Securities Australia Ltd	Enter into Derivative Contract	ΑN	2,000	Ontions
IRS Securities Australia Ind	Enter Into Derivative			
Australia Ltu	Contract	NA AA	80,300	Options
JBS Securities Australia Ltd	Contract	AVA	(00 Z 8)	Control
JBS Global Asset Management	Buy	1626	200	Copies
JBS Securities Australia Ltd	Sell	207 702	25 240)	Ordinary
UBS Securities Australia Ltd	Buy	114,491	14.500	Ordinary
UBS Securities Australia Ltd	Buy	210,095	26,463	Ordinary
UBS Securities Australia Ltd	Sell	334,221	(42,113)	Ordinary
JBS Securities Australia Ltd	Buy	11,313	1,444	Ordinary
Securities Australia Ltd	Sell	392,918	(49,994)	Ordinary
Securities Australia Ltd	Buy	140,040	18,000	Ordinary
Securities Australia Ltd	), vell	175,932	(02 500)	
UBS Securities Australia Ltd	Buy		(000)	Ordinary
		40,042	5,065	Ordinary

					Clumary
27-Nov-14	UBS Securities Australia Ltd	Buy	13.413	1.578	Ordinary
27-Nov-14	UBS Securities Australia Ltd	Buy	850		Ordinary
27-Nov-14	UBS Securities Australia Ltd	Buy	468		Ordinary
27-Nov-14	UBS Securities Australia Ltd	Влу	17		Ordinary
27-Nov-14	UBS Securities Australia Ltd	Buy	238	28	Ordinary
27-Nov-14	UBS Securities Australia Ltd	Buy	34		Ordinary
27-Nov-14	UBS Securities Australia Ltd	Buy	1,896	223	Ordinary
27-Nov-14	UBS Securities Australia Ltd	Buy	111		Ordinary
27 Nov. 14	Obsider Australia Ltd	an's	43		Ordinary
27-NOV-14	UBS Securities Australia Ltd	Buy	13,634		Ordinary
7 No. 14	Obs securities Australia Ltd	and a	9,274		Ordinary
7-Nov-14	UBS Counties Australia Ltd	and D	2,023		Ordinary
7 Mov. 14	upo comines Australia Ltd	euy.	820	100	Ordinary
7-Now-14	LIBS Securities Australia Ind	Buny	515,1		Ordinary
7-Nov-14	URS Securities Australia Ind	Riv	3,842		Ordinary
7-Nov-14	UBS Securities Australia 11d	Buy	503	001	Ordinary
7-Nov-14	UBS Securities Australia Itd	Biry	505		Ordinary
7-Nov-14	UBS Securities Australia Itd	Ruy	010		Ordinary
7-Nov-14	TUBS Securities Australia Ind	Buy	200		Ordinary
7-Nov-14	UBS Securities Australia Ltd	Buv	906		Ordinary
7-Nov-14	UBS Securities Australia Ltd	Buy	247		Ordinary
7-Nov-14	UBS Securities Australia Ltd	Buv	2 845	570	Ordinan
7-Nov-14	UBS Securities Australia Ltd	Buy	850	1001	Ordinary
7-Nov-14	UBS Securities Australia Ltd	Buy	on the same		Ordinary
7-Nov-14	UBS Securities Australia Ltd	Buy	119	14	Ordinary
7-Nov-14	UBS Securities Australia Ltd	Buy	77	6	Ordinary
7-Nov-14	UBS Securities Australia Ltd	Buy	3,528	415	Ordinary
ı	UBS Securities Australia Ltd	Buy	1,445	170	Ordinary
İ	UBS Securities Australia Ltd	Buy	17	2	Ordinary
-[	UBS Securities Australia Ltd	8uy	264	1E	Ordinary
-	UBS Securities Australia Ltd	Buy	2,941	346	Ordinary
-	UBS Securities Australia Ltd	Buy	16,167	1,902	Ordinary
1	UBS Securities Australia Ltd	Buy	1,794	211	Ordinary
ı	UBS Securities Australia Ltd	Buy	1,998	235	Ordinary
	UBS Securities Australia Ltd	Buy	459	54	Ordinary
7-NOV-14	UBS Securities Australia Ltd	euy	43	5	Ordinary
7 Nov 14	UBS Securities Australia Ltd	an)	145	17	Ordinary
7-Nov-14	Securities Australia	land.	/1	2	Ordinary
Most 14	The Commission August and Line	ouy.	008	100	Ordinary
Now 14	Tipo Commissor Australia Indi	ina a	\$	10	Ordinary
No. 14	Upo Securities Australia Ltd	and.	5,891	693	Ordinary
Now 14	Day Selection Australia tea	ind.	75977	312	Ordinary
7-Nov-14	Tibe countries Australia Ltd	pro)	3,859	454	Ordinary
	LIBS Securities Australia 11d	Bin	1,233	145	Ordinary
7-Nov-14	LIBC Countries Australia Ind	P. I	055,5	300	Ordinary
ı	URC Securities Australia 14d	p. n.	05.6	/5777	Ordinary
27-Nov-14	LIBS Securities Australia Ltd	Blac	2,763	325	Ordinary
7-Now-14	Tibo Countries Australia Ltd	Buy B.m.	3,740	440	Ordinary
27-Nov-14	Tibe Securities Australia Ltd	áng a	/60′1	129	Ordinary
7 Mov. 14	Obs Securities Australia Ltd	ank	893	105	Ordinary
-Nov-14	UBS Securities Australia Ltd	Buy	298	35	Ordinary
7-NOV-14	UBS Securities Australia Ltd	gray	4,097	482	Ordinary
7-NOV-14	UBS Securities Australia Ltd	Buy	2,610	307	Ordinary
27 Nov-14	USS Securities Australia Ltd	Buy	10,914	1,284	Ordinary
1404	UBS Securities Australia LTG	grý.	4,463	525	Ordinary
1	UBS Securifies Australia Ltd	Buy	10,685	1,257	Ordinary
ı	UBS Securities Australia Ltd	Buy	27,481	3,233	Ordinary
1	UBS Securities Australia Ltd	Buy	26,860	3,160	Ordinary
/-Nov-14	UBS Securities Australia Ltd	Buy	2,066	243	Ordinary
-Nov-14	UBS Securities Australia Ltd	Buy	10,047	1,182	Ordinary
-Nov-14	UBS Securities Australia Ltd	Buy	3,553	418	Ordinary
-Nov-14	UBS Securities Australia Ltd	8uy	4,225	497	Ordinary
-Nov-14	UBS Securities Australia Ltd	Buy	13,209	1,554	Ordinary
7-Nov-14	UBS Securities Australia Ltd	Buy	22,559	2,654	Ordinary
-Nov-14	UBS Securities Australia Ltd	Buy	806'8	1,048	Ordinary
-Nov-14	UBS Securities Australia Ltd	Buy	4,531	533	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	1,479	174	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	5,993	202	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	4,225	497	Ordinary
Nov-14	UBS Securities Australia Ltd	Buy	57,384	15/29	Ordinary
7-Nov-14	UBS Securities Australia Ltd	Buy	12,529	1.474	
					Orginary

List Securities Automists   Boy   CSSSS   CSSS			3	7,354	305	Ordinary
May         2573         339           May         2573         329           May         2573         329           May         2574         100           May         789         100           May         789         100           May         789         100           May         780         72           May         780         72           May         780         72           May         780         780           May         780         780           May         780         780           May         780         780           May         780         780           May         780         780           May         780         780           May         780         780           May         780         780           May         780         780           May         780         780           May         780         780           May         780         780           May         780         780           May         780         780	3-Nov-14	UBS Securities Australia Ltd	Buy	355	45	Ordinan
Miss Securines Automita Lett         Bay         6.50T         RS           Miss Securines Automita Lett         Bay         7.88         1.00           Miss Securines Automita Lett         Bay         7.89         1.00           Miss Securines Automita Lett         Bay         7.89         7.00           Miss Securines Automita Lett         Bay         7.00         7.00           Miss Securines Automita Lett         Bay         7.00         7.00           Miss Securines Automita Lett         Bay         7.00         7.00           Miss Securines Automita Lett         Bay         7.00         7.00           Miss Securines Automita Lett         Bay         7.00         7.00           Miss Securines Automita Lett         Bay         7.00         7.00           Miss Securines Automita Lett         Bay         7.00         7.00           Miss Securines Automita Lett         Bay         7.00         7.00           Miss Securines Automita Lett         Bay         7.00         7.00           Miss Securines Automita Lett         Bay         7.00         7.00           Miss Securines Automita Lett         Bay         7.00         7.00           Miss Securines Automita Lett         Bay         7.00 <td>3-Nov-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>2,593</td> <td>329</td> <td>Ordinary</td>	3-Nov-14	UBS Securities Australia Ltd	Buy	2,593	329	Ordinary
May         7,880         1,000           May         7,880         1,000           May         7,880         1,000           May         7,880         1,000           May         7,890         1,000           May         7,890         2,000           May         7,800         2,000           May         7,800         3,000           May         8,000         3,000           May         8,000         3,000           May         8,000         3,000           May         8,000         3,000           May         8,000         3	3-Nov-14	UBS Securities Australia Ltd	Buy	6,501	825	Ordinary
185 Securine Autorials Let   899   778   778   710	3-Nov-14	UBS Securities Australia Ltd	Buy	088'2	1,000	Ordinary
Mail Securine Augustia 147   894   410   512   512   513   514	3-Nov-14	UBS Securities Australia Ltd	Buy	882	100	Ordinary
185 Securine Australia Ltd   804   20,755   37,001   181 Securine Australia Ltd   804   20,755   32,001   33,001   31,	3-Nov-14	UBS Securities Australia Ltd	Buy	410	52	Ordinary
URS Securine Auctivation   Elegy   20/64   300   300   408   300   408   300   408   300   408   300   408   300   408   300   408   300   408   300   409   400   408   400	3-Nov-14	UBS Securities Australia Ltd	Buy	55	7	Ordinary
URS Securites Auctivale Let   Bay   236   310	3-Nov-14	UBS Securities Australia Ltd	Buy	30,764	3.904	Ordinary
URS scruttles Australia Ltd   Buy   3,155   241   1401   1855   241   241	8-Nov-14	UBS Securities Australia Ltd	Buy	236	30	Ordinary
URS Securine Autoriale Ltd   Buy   13,128   234   241     URS Securine Autoriale Ltd   Buy   1,100   1,401     URS Securine Autoriale Ltd   Buy   1,100   1,401     URS Securine Autoriale Ltd   Buy   1,100   1,401     URS Securine Autoriale Ltd   Buy   1,202   1,202     URS Securine Autoriale Ltd   Buy   1,202   1,203     URS Securine Autoriale Ltd   Buy   1,202   1,203     URS Securine Autoriale Ltd   Buy   1,202   1,203     URS Securine Autoriale Ltd   Buy   1,203   1,203     URS Securine Autoriale Lt	3-Nov-14	UBS Securities Australia Ltd	Buy	69	oc	Ordinary
URS Securite Australia Ltd   Buy   1,199   1,401   1,995   1	-Nov-14	UBS Securities Australia Ltd	Buy	3,128	397	Ordinary
UBS Securities Australia LIC         Buy         1103         140           UBS Securities Australia LIC         Buy         110         140           UBS Securities Australia LIC         Buy         150         140           UBS Securities Australia LIC         Buy         2.22-2         150           UBS Securities Australia LIC         Buy         2.34-3         370           UBS Securities Australia LIC         Buy         2.34-3         370           UBS Securities Australia LIC         Buy         2.34-3         370           UBS Securities Australia LIC         Buy         2.34-3         370           UBS Securities Australia LIC         Buy         2.34-3         370           UBS Securities Australia LIC         Buy         2.34-3         370           UBS Securities Australia LIC         Buy         2.34-3         370           UBS Securities Australia LIC         Buy         2.34-3         370           UBS Securities Australia LIC         Buy         2.34-3         370           UBS Securities Australia LIC         Buy         2.34-3         370           UBS Securities Australia LIC         Buy         2.34-3         370           UBS Securities Australia LIC         Buy <th< td=""><td>1-Nov-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>1 899</td><td>241</td><td>Ordinary</td></th<>	1-Nov-14	UBS Securities Australia Ltd	Buy	1 899	241	Ordinary
Big         Byp         118         22           Big         Big         118         20           Big         Big         118         25           Big         Big         128         284         128           Big         Big         128         26           Big         Big         128         26           Big         Big         128         23         42           Big         Big         128         23         42           Big         Big         128         23         42           Big         Big         128         23         42           Big         Big         138         23         42           Big         Big         128         23         42           Big         Big         128         23         42           Big         Big         128         128         128           Big         Big         128         128         128           Big         Big         128         128         128           Big         Big         128         128         128           Big         Big         <	1-Nov-14	UBS Securities Australia Ltd	Buy	1.103	140	Ordinan
UBS Securities Australia Ltd   Bay   110   120   150	1-Nov-14	UBS Securities Australia Ltd	Buy	158	000	Ordinan
UBS Securities Australia Ltd         Bay         2.864         300           UBS Securities Australia Ltd         Bay         1.28         500           UBS Securities Australia Ltd         Bay         3.128         500           UBS Securities Australia Ltd         Bay         3.128         3.07           UBS Securities Australia Ltd         Bay         3.23         4.09           UBS Securities Australia Ltd         Bay         3.23         4.09           UBS Securities Australia Ltd         Bay         3.23         4.09           UBS Securities Australia Ltd         Bay         3.24         4.02           UBS Securities Australia Ltd         Bay         3.24         4.09           UBS Securities Australia Ltd         Bay         3.24         4.09           UBS Securities Australia Ltd         Bay         3.24         4.05           UBS Securities Australia Ltd         Bay         3.24         4.05           UBS Securities Australia Ltd         Bay         2.03         3.09           UBS Securities Australia Ltd         Bay         2.03         3.09           UBS Securities Australia Ltd         Bay         2.03         3.09           UBS Securities Australia Ltd         Bay         3.00 </td <td>-Nov-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>110</td> <td>14</td> <td>Ordinary</td>	-Nov-14	UBS Securities Australia Ltd	Buy	110	14	Ordinary
UBS Securities Australia Ltd         Bay         1,229         1,58           LOSS Securities Australia Ltd         Bay         6,625         1,50           LOSS Securities Australia Ltd         Bay         8,64         4,049           LOSS Securities Australia Ltd         Bay         3,123         40           LOSS Securities Australia Ltd         Bay         3,123         40           LOSS Securities Australia Ltd         Bay         3,123         40           LOSS Securities Australia Ltd         Bay         3,123         40           LOSS Securities Australia Ltd         Bay         3,123         40           LOSS Securities Australia Ltd         Bay         3,123         40           LOSS Securities Australia Ltd         Bay         6,23         3,17           LOSS Securities Australia Ltd         Bay         1,20         3,17           LOSS Securities Australia Ltd         Bay         1,20         3,10           LOSS Securities Australia Ltd         Bay         1,20         3,10           LOSS Securities Australia Ltd         Bay         2,435         2,435           LOSS Securities Australia Ltd         Bay         2,435         2,435           LOSS Securities Australia Ltd         Bay	-Nov-14	UBS Securities Australia Ltd	Buy	2 364	1 00	Crainary
USS Securities Autorials Ltd   Bay   1,28   1,28   1,09   1,00	-Nov-14	UBS Securities Australia 11d	Buy	1 220	81	0.00
USS Securities Avaitable 14d   Blyy   81,826   4,049     USS Securities Avaitable 14d   Blyy   31,228   4,049     USS Securities Avaitable 14d   Blyy   3,123   4,049     USS Securities Avaitable 14d   Blyy   3,123   4,049     USS Securities Avaitable 14d   Blyy   3,123   4,049     USS Securities Avaitable 14d   Blyy   3,123   4,049     USS Securities Avaitable 14d   Blyy   1,520   1,040     USS Securities Avaitable 14d   Blyy   1,520   1,040     USS Securities Avaitable 14d   Blyy   1,520   1,040     USS Securities Avaitable 14d   Blyy   1,520   1,040     USS Securities Avaitable 14d   Blyy   1,050   1,050     USS Securities Avaitable 1	-Nov-14	UBS Securities Australia 11d	Biv	6371	007	Orginary
LBS Securities Australia Ltd         Byy         8,820         1,118           LBS Securities Australia Ltd         Bay         3,128         3,97           LBS Securities Australia Ltd         Bay         3,128         3,97           LBS Securities Australia Ltd         Bay         2,564         3,00           LBS Securities Australia Ltd         Bay         2,564         300           LBS Securities Australia Ltd         Bay         2,564         300           LBS Securities Australia Ltd         Bay         2,563         3,40           LBS Securities Australia Ltd         Bay         2,563         3,60           LBS Securities Australia Ltd         Bay         2,563         3,60           LBS Securities Australia Ltd         Bay         2,563         3,60           LBS Securities Australia Ltd         Bay         2,563         3,60           LBS Securities Australia Ltd         Bay         2,563         3,60           LBS Securities Australia Ltd         Bay         2,563         1,50           LBS Securities Australia Ltd         Bay         2,563         1,50           LBS Securities Australia Ltd         Bay         2,563         1,50           LBS Securities Australia Ltd         Bay	-Nov-14	UBS Securities Australia 11d	Rus	20010	0/0/0	Ordinary
UBS Securities Australia Ltd         Bay         3,128         409           UBS Securities Australia Ltd         Bay         3,123         409           UBS Securities Australia Ltd         Bay         3,323         409           UBS Securities Australia Ltd         Bay         2,364         300           UBS Securities Australia Ltd         Bay         6,329         791           UBS Securities Australia Ltd         Bay         6,329         794           UBS Securities Australia Ltd         Bay         6,329         3,77           UBS Securities Australia Ltd         Bay         2,263         376           UBS Securities Australia Ltd         Bay         2,263         376           UBS Securities Australia Ltd         Bay         2,263         376           UBS Securities Australia Ltd         Bay         2,263         376           UBS Securities Australia Ltd         Bay         2,635         376           UBS Securities Australia Ltd         Bay         2,635         376           UBS Securities Australia Ltd         Bay         2,635         376           UBS Securities Australia Ltd         Bay         2,635         3,734           UBS Securities Australia Ltd         Bay         2,6	-Nov-14	LIBS Securities Australia 11d	Bin	200,15	4,049	Ordinary
UBS Securities Autorials Ltd         Buy         3,128         459           UBS Securities Autorials Ltd         Buy         3,128         459           UBS Securities Autorials Ltd         Buy         2,584         459           UBS Securities Autorials Ltd         Buy         2,584         459           UBS Securities Autorials Ltd         Buy         2,584         177           UBS Securities Autorials Ltd         Buy         40,285         177           UBS Securities Autorials Ltd         Buy         40,285         177           UBS Securities Autorials Ltd         Buy         40,285         2,687         177           UBS Securities Autorials Ltd         Buy         40,285         2,687         170           UBS Securities Autorials Ltd         Buy         2,286         8,69         17,70           UBS Securities Autorials Ltd         Buy         2,286         1,50         1,50           UBS Securities Autorials Ltd         Buy         2,283         1,50         1,50           UBS Securities Autorials Ltd         Buy         2,283         1,50         1,50           UBS Securities Autorials Ltd         Buy         2,283         1,50         1,50           UBS Securities Autorials Ltd	-Nov-14	LIBS Securities Australia 14d	Big	100,0	0 10	Orginary
USS Securities Australia Ltd         Bay         3,1223         409           USS Securities Australia Ltd         Bay         3,344         300           USS Securities Australia Ltd         Bay         6,238         707           USS Securities Australia Ltd         Bay         6,233         707           USS Securities Australia Ltd         Bay         6,233         707           USS Securities Australia Ltd         Bay         6,075         5,177           USS Securities Australia Ltd         Bay         7,045         894           USS Securities Australia Ltd         Bay         2,045         894           USS Securities Australia Ltd         Bay         2,045         896           USS Securities Australia Ltd         Bay         2,045         896           USS Securities Australia Ltd         Bay         2,045         896           USS Securities Australia Ltd         Bay         2,045         896           USS Securities Australia Ltd         Bay         2,045         896           USS Securities Australia Ltd         Bay         2,045         896           USS Securities Australia Ltd         Bay         2,045         896           USS Securities Australia Ltd         Bay         2,0	Nov. 14	IIIR Socurities Australia 144	, and	3,128	765	Ordinary
USS Securities Australia Ltd         BM         CAS Securities Australia Ltd         BM         CAS Securities Australia Ltd         BM         CAS SECURITIES Australia Ltd         <	Now-14	URC Securities Australia 344	(no	3,423	409	Ordinary
UBS Securities Australia Ltd         Buy         6,334         300           UBS Securities Australia Ltd         Buy         6,233         791           UBS Securities Australia Ltd         Buy         6,233         791           UBS Securities Australia Ltd         Buy         1,395         1,77           UBS Securities Australia Ltd         Buy         1,395         1,77           UBS Securities Australia Ltd         Buy         2,485         1,77           UBS Securities Australia Ltd         Buy         2,983         3,94           UBS Securities Australia Ltd         Buy         2,983         3,98           UBS Securities Australia Ltd         Buy         2,485         3,09           UBS Securities Australia Ltd         Buy         2,483         3,09           UBS Securities Australia Ltd         Buy         2,483         3,08           UBS Securities Australia Ltd         Buy         2,483         3,08           UBS Securities Australia Ltd         Buy         2,733         3,08           UBS Securities Australia Ltd         Buy         2,735         1,730           UBS Securities Australia Ltd         Buy         2,735         1,730           UBS Securities Australia Ltd         Buy	Alou 14	And Securities Additional Little	, and	331	42	Ordinary
USS Securities Australia Ltd         Buy         6.233         791           USS Securities Australia Ltd         Buy         1.395         1.77           USS Securities Australia Ltd         Buy         1.899         2.41           USS Securities Australia Ltd         Buy         1.899         2.41           USS Securities Australia Ltd         Buy         1.899         2.41           USS Securities Australia Ltd         Buy         7.045         3.90           USS Securities Australia Ltd         Buy         7.045         3.90           USS Securities Australia Ltd         Buy         7.045         3.90           USS Securities Australia Ltd         Buy         7.045         3.90           USS Securities Australia Ltd         Buy         7.045         3.90           USS Securities Australia Ltd         Buy         7.045         3.00           USS Securities Australia Ltd         Buy         7.045         3.00           USS Securities Australia Ltd         Buy         7.070         4.08           USS Securities Australia Ltd         Buy         7.070         4.08           USS Securities Australia Ltd         Buy         7.070         4.08           USS Securities Australia Ltd         Buy	No. 14	Upp Countries Australia Ltd	ing a	2,364	300	Ordinary
UBS Securities Australia Ltd         BMY         1,395         170           UBS Securities Australia Ltd         BMY         1,395         177           UBS Securities Australia Ltd         BMY         1,395         1,177           UBS Securities Australia Ltd         BMY         2,368         2,177           UBS Securities Australia Ltd         BMY         2,368         3,94           UBS Securities Australia Ltd         BMY         2,368         3,96           UBS Securities Australia Ltd         BMY         2,368         3,96           UBS Securities Australia Ltd         BMY         2,368         8,96           UBS Securities Australia Ltd         BMY         2,368         8,96           UBS Securities Australia Ltd         BMY         3,544         4,45           UBS Securities Australia Ltd         BMY         3,544         4,45           UBS Securities Australia Ltd         BMY         3,544         4,65           UBS Securities Australia Ltd         BMY         3,544         4,65           UBS Securities Australia Ltd         BMY         3,544         4,65           UBS Securities Australia Ltd         BMY         3,544         4,65           UBS Securities Australia Ltd         BMY	Nov. 14	Libe Consists Australia Ltd	ing.	6,233	791	Ordinary
UBS Securities Australia Ltd         BMY         40,795         177           UBS Securities Australia Ltd         BMY         40,795         177           UBS Securities Australia Ltd         BMY         1,895         241           UBS Securities Australia Ltd         BMY         7,045         894           UBS Securities Australia Ltd         BMY         7,045         895           UBS Securities Australia Ltd         BMY         7,045         896           UBS Securities Australia Ltd         BMY         7,045         193           UBS Securities Australia Ltd         BMY         7,045         193           UBS Securities Australia Ltd         BMY         7,045         1,050           UBS Securities Australia Ltd         BMY         4,058         5,05           UBS Securities Australia Ltd         BMY         4,058         5,05           UBS Securities Australia Ltd         BMY         4,058         1,000           UBS Securities Australia Ltd         BMY         4,056         1,000           UBS Securities Australia Ltd         BMY         4,056         1,000           UBS Securities Australia Ltd         BMY         7,046         1,000           UBS Securities Australia Ltd         BMY		CB3 Securities Australia Ltd	ony	835	106	Ordinary
UBS Securities Australia Ltd         Buy         40,795         5,177           UBS Securities Australia Ltd         Buy         1,899         2,177           UBS Securities Australia Ltd         Buy         2,963         394           UBS Securities Australia Ltd         Buy         2,963         394           UBS Securities Australia Ltd         Buy         2,963         396           UBS Securities Australia Ltd         Buy         1,506         1,208           UBS Securities Australia Ltd         Buy         1,506         1,208           UBS Securities Australia Ltd         Buy         1,506         1,506           UBS Securities Australia Ltd         Buy         1,508         1,508           UBS Securities Australia Ltd         Buy         1,508         4,65           UBS Securities Australia Ltd         Buy         2,713         1,000           UBS Securities Australia Ltd         Buy         2,713         1,000           UBS Securities Australia Ltd         Buy         2,713         1,000           UBS Securities Australia Ltd         Buy         2,713         1,000           UBS Securities Australia Ltd         Buy         7,713         1,143           UBS Securities Australia Ltd         Buy	100	UBS Securities Australia Ltd	Buy	1,395	177	Ordinary
UBS Securities Australia Ltd         BMy         1,899         241           UBS Securities Australia Ltd         BMy         3,940         500           UBS Securities Australia Ltd         BMy         7,045         894           UBS Securities Australia Ltd         BMy         7,045         895           UBS Securities Australia Ltd         BMy         7,243         3,06           UBS Securities Australia Ltd         BMy         2,435         3,06           UBS Securities Australia Ltd         BMy         2,435         3,06           UBS Securities Australia Ltd         BMy         2,435         3,06           UBS Securities Australia Ltd         BMy         3,544         4,05           UBS Securities Australia Ltd         Sell         1,55         1,000           UBS Securities Australia Ltd         Sell         7,73         1,000           UBS Securities Australia Ltd         Sell         7,756         1,000           UBS Securities Australia Ltd         Sell         7,130         1,000           UBS Securities Australia Ltd         Sell         1,152         1,000           UBS Securities Australia Ltd         Sell         1,152         1,000           UBS Securities Australia Ltd         BMy <td>-1404-14</td> <td>Upo securities Australia Ltd</td> <td>Buy</td> <td>40,795</td> <td>5,177</td> <td>Ordinary</td>	-1404-14	Upo securities Australia Ltd	Buy	40,795	5,177	Ordinary
UBS Securities Australia Ltd         BMy         39.0         500           UBS Securities Australia Ltd         BMy         7.045         89.4           UBS Securities Australia Ltd         BMy         7.045         89.4           UBS Securities Australia Ltd         BMy         1.505         1.905           UBS Securities Australia Ltd         BMy         1.505         1.905           UBS Securities Australia Ltd         BMy         6.982         8.86           UBS Securities Australia Ltd         BMy         6.982         8.86           UBS Securities Australia Ltd         BMy         6.982         1.505           UBS Securities Australia Ltd         BMy         6.982         1.856           UBS Securities Australia Ltd         Sell         7.737         (1.080)           UBS Securities Australia Ltd         Sell         7.737         (1.080)           UBS Securities Australia Ltd         Sell         7.736         (1.080)           UBS Securities Australia Ltd         Sell         7.736         (1.080)           UBS Securities Australia Ltd         BMy         7.154         (1.080)           UBS Securities Australia Ltd         BMy         7.154         (1.080)           UBS Securities Australia Ltd	-Nov-14	UBS Securities Australia Ltd	Buy	1,899	241	Ordinary
UBS Securities Australia Ltd         BMY         7,045         894           UBS Securities Australia Ltd         BMY         2,065         376           UBS Securities Australia Ltd         BMY         2,2435         376           UBS Securities Australia Ltd         BMY         2,435         376           UBS Securities Australia Ltd         BMY         2,435         376           UBS Securities Australia Ltd         BMY         2,733         376           UBS Securities Australia Ltd         BMY         3,544         4,056           UBS Securities Australia Ltd         BMY         3,544         4,056           UBS Securities Australia Ltd         Sell         7,735         1,056           UBS Securities Australia Ltd         Sell         7,735         1,056           UBS Securities Australia Ltd         Sell         7,736         1,056           UBS Securities Australia Ltd         Sell         7,736         1,056           UBS Securities Australia Ltd         Sell         1,153         1,056           UBS Securities Australia Ltd         BMY         1,153         1,056           UBS Securities Australia Ltd         BMY         1,153         1,057           UBS Securities Australia Ltd         BMY	-NOV-14	UBS Securities Australia Ltd	Buy	3,940	200	Ordinary
UBS Securine Australia Ltd         Buy         2.95         3.76           UBS Securine Australia Ltd         Buy         1,505         3.76           UBS Securine Australia Ltd         Buy         1,505         4.96           UBS Securine Australia Ltd         Buy         3,908         4.96           UBS Securine Australia Ltd         Buy         4,098         4.09           UBS Securine Australia Ltd         Buy         4,098         5.00           UBS Securine Australia Ltd         Buy         4,098         5.00           UBS Securine Australia Ltd         Sell         7,713         (65)           UBS Securine Australia Ltd         Sell         7,720         (7,000           UBS Securine Australia Ltd         Sell         7,730         (7,000           UBS Securine Australia Ltd         Sell         7,730         (7,596           UBS Securine Australia Ltd         Sell         7,730         (7,596           UBS Securine Australia Ltd         Buy         2,730         (7,596           UBS Securine Australia Ltd         Buy         2,730         (7,596           UBS Securine Australia Ltd         Buy         2,730         (7,596           UBS Securine Australia Ltd         Buy         2,720	-Nov-14	UBS Securities Australia Ltd	Buy	7,045	894	Ordinary
UBS Securities Australia Ltd         Buy         2.2.1         2.8           UBS Securities Australia Ltd         Buy         2,435         1.505         1.51           UBS Securities Australia Ltd         Buy         2,435         3.95         4.96           UBS Securities Australia Ltd         Buy         6,982         8.66         8.50         8.66         8.65           UBS Securities Australia Ltd         Buy         6,982         8.66         8.50         8.60         8.50         8.60         8.50         8.60         8.50         8.60         8.50         8.60         8.50         8.60         8.50         8.60         8.50         8.60         8.50         8.60         8.50         8.60         8.50         8.60         8.50         8.60         8.50         8.60         8.50         8.60         8.50         8.50         8.60         8.50         8.50         8.60         8.50         8.60         8.50         8.60         8.50         8.60         8.50         8.50         8.50         8.50         8.50         8.50         8.50         8.50         8.50         8.50         8.50         8.50         8.50         8.50         8.50         8.50         8.50         8.50         8	-Nov-14	UBS Securities Australia Ltd	Buy	2,963	376	Ordinary
UBS Securine Australia Ltd         Buy         1,509         191           UBS Securine Australia Ltd         Buy         2,435         309           UBS Securine Australia Ltd         Buy         6,982         8,98           UBS Securine Australia Ltd         Buy         4,096         5,98           UBS Securine Australia Ltd         Buy         4,096         5,00           UBS Securine Australia Ltd         Sell         7,37         3,00           UBS Securine Australia Ltd         Sell         7,37         3,00           UBS Securine Australia Ltd         Sell         7,30         1,000           UBS Securine Australia Ltd         Sell         7,30         1,000           UBS Securine Australia Ltd         Sell         7,30         1,000           UBS Securine Australia Ltd         Buy         1,156         2,65           UBS Securine Australia Ltd         Buy         1,156         2,68           UBS Securine Australia Ltd         Buy         1,156         2,68           UBS Securine Australia Ltd         Buy         1,150         1,450           UBS Securine Australia Ltd         Buy         1,150         2,59           UBS Securine Australia Ltd         Buy         1,160 <t< td=""><td>Nov-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>221</td><td>28</td><td>Ordinary</td></t<>	Nov-14	UBS Securities Australia Ltd	Buy	221	28	Ordinary
UBS Securities Australia Ltd         Buy         1,2435         309           UBS Securities Australia Ltd         Buy         6,982         866           UBS Securities Australia Ltd         Buy         6,982         866           UBS Securities Australia Ltd         Buy         6,582         866           UBS Securities Australia Ltd         Sell         6,783         1,288           UBS Securities Australia Ltd         Sell         6,783         1,000           UBS Securities Australia Ltd         Sell         7,713         1,000           UBS Securities Australia Ltd         Sell         7,596         1,000           UBS Securities Australia Ltd         Sell         7,596         1,000           UBS Securities Australia Ltd         Sell         7,596         1,000           UBS Securities Australia Ltd         Sell         1,004         1,000           UBS Securities Australia Ltd         Sell         1,004         1,000           UBS Securities Australia Ltd         Buy         1,153         1,00           UBS Securities Australia Ltd         Buy         1,153         1,00           UBS Securities Australia Ltd         Buy         1,153         1,00           UBS Securities Australia Ltd <td< td=""><td>-Nov-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>1 505</td><td>191</td><td>Ordinary</td></td<>	-Nov-14	UBS Securities Australia Ltd	Buy	1 505	191	Ordinary
UBS Securities Australia Ltd         Buy         3,508         4,568           UBS Securities Australia Ltd         Buy         4,058         5,20           UBS Securities Australia Ltd         Buy         4,058         5,20           UBS Securities Australia Ltd         Sell         1,289         4,058           UBS Securities Australia Ltd         Sell         7,713         1,209           UBS Securities Australia Ltd         Sell         7,713         1,000           UBS Securities Australia Ltd         Sell         7,700         1,000           UBS Securities Australia Ltd         Sell         7,700         1,000           UBS Securities Australia Ltd         Sell         7,700         1,000           UBS Securities Australia Ltd         Sell         7,130         2,457           UBS Securities Australia Ltd         Sell         1,15,248         1,130           UBS Securities Australia Ltd         Sell         1,15,248         1,145,000           UBS Securities Australia Ltd         Buy         2,1,359         2,457           UBS Securities Australia Ltd         Buy         1,1,529         2,457           UBS Securities Australia Ltd         Buy         1,1,62,500         1,1,000           UBS Securities	-Nov-14	UBS Securities Australia Ltd	Buy	2.435	308	Ordinary
UBS Securities Australia Ltd         Buy         6,725         826           UBS Securities Australia Ltd         Buy         6,725         826           UBS Securities Australia Ltd         Buy         3,544         445           UBS Securities Australia Ltd         Sell         6,732         1,209           UBS Securities Australia Ltd         Sell         79,700         10,000           UBS Securities Australia Ltd         Sell         79,700         10,000           UBS Securities Australia Ltd         Sell         79,700         10,000           UBS Securities Australia Ltd         Sell         7,550         10,000           UBS Securities Australia Ltd         Sell         1,94,28         1,1,680           UBS Securities Australia Ltd         Sell         1,19,438         1,1,680           UBS Securities Australia Ltd         Buy         1,19,438         1,1,680           UBS Securities Australia Ltd         Buy         1,19,438         1,16           UBS Securities Australia Ltd         Buy         1,299         2,685           UBS Securities Australia Ltd         Buy         1,299         2,586           UBS Securities Australia Ltd         Buy         40,293         5,516           UBS Securities Aus	-Nov-14	UBS Securities Australia Ltd	Buv	aco n	307	diameter of
UBS Securities Australia Ltd         BMY         4,784         680           UBS Securities Australia Ltd         BMY         1,352         1,298           UBS Securities Australia Ltd         Sell         1,69         1,298           UBS Securities Australia Ltd         Sell         678         (673)           UBS Securities Australia Ltd         Sell         7,73         (700)           UBS Securities Australia Ltd         Sell         7,73         (700)           UBS Securities Australia Ltd         Sell         7,70         (700)           UBS Securities Australia Ltd         Sell         7,70         (700)           UBS Securities Australia Ltd         Sell         1,19,339         (7,467)           UBS Securities Australia Ltd         Sell         1,19,328         (1,486)           UBS Securities Australia Ltd         Sell         1,19,328         (1,486)           UBS Securities Australia Ltd         Buy         2,134         2,547           UBS Securities Australia Ltd         Buy         4,226         1,050           UBS Securities Australia Ltd         Buy         4,224         3,54           UBS Securities Australia Ltd         Buy         4,224         3,54           UBS Securities Australia Ltd </td <td>-Nov-14</td> <td>IIRS Securities Australia 14d</td> <td>N B</td> <td>200,0</td> <td>064</td> <td>Grainary</td>	-Nov-14	IIRS Securities Australia 14d	N B	200,0	064	Grainary
UBS Securities Australia Ltd         BM         3.54         4.45           UBS Securities Australia Ltd         Sell         3.544         4.45           UBS Securities Australia Ltd         Sell         5.713         (.60)           UBS Securities Australia Ltd         Sell         7.73         (.60)           UBS Securities Australia Ltd         Sell         7.97,00         (.00)           UBS Securities Australia Ltd         Sell         7.97,00         (.00)           UBS Securities Australia Ltd         Sell         7.96         (.63)           UBS Securities Australia Ltd         Sell         7.96         (.63)           UBS Securities Australia Ltd         Sell         1.965         7.633           UBS Securities Australia Ltd         Sell         1.966         7.565           UBS Securities Australia Ltd         Buy         1.966         7.566           UBS Securities Australia Ltd         Buy         1.796         2.566           UBS Securities Australia Ltd         Buy         1.796         2.566           UBS Securities Australia Ltd         Buy         1.796         2.566           UBS Securities Australia Ltd         Buy         3.050         1.40,090           UBS Securities Australia Ltd	-Nov-14	URS Securities Australia Itd	Bir	4,000	000	Ordinary
USS Securities Australia Ltd         Buy         1,329         1,329           USS Securities Australia Ltd         Sell         678         689           USS Securities Australia Ltd         Sell         73         689           USS Securities Australia Ltd         Sell         7,900         10,000           USS Securities Australia Ltd         Sell         7,900         10,000           USS Securities Australia Ltd         Sell         7,900         10,000           USS Securities Australia Ltd         Sell         7,590         10,000           USS Securities Australia Ltd         Sell         1,19,428         1,1080           USS Securities Australia Ltd         Buy         1,169,428         1,45,500           USS Securities Australia Ltd         Buy         1,169,428         1,45,500           USS Securities Australia Ltd         Buy         1,168,290         1,45,500           USS Securities Australia Ltd         Buy         1,168,290         1,45,500           USS Securities Australia Ltd         Buy         1,231         1,66           USS Securities Australia Ltd         Buy         4,2,242         3,118           USS Securities Australia Ltd         Buy         4,2,242         3,118           USS Se	-Nov-14	ILIBS Securities Australia 11th	Blay	40.20	026	Ordinary
UBS Securities Australia Ltd         Sell         678         478           UBS Securities Australia Ltd         Sell         678         (82)           UBS Securities Australia Ltd         Sell         7370         (1000)           UBS Securities Australia Ltd         Sell         7,559         (637)           UBS Securities Australia Ltd         Sell         7,559         (637)           UBS Securities Australia Ltd         Sell         8,559         (637)           UBS Securities Australia Ltd         Buy         1,19,625         (7,639)           UBS Securities Australia Ltd         Buy         2,1369         (7,639)           UBS Securities Australia Ltd         Buy         2,1369         (7,639)           UBS Securities Australia Ltd         Buy         2,1369         (7,639)           UBS Securities Australia Ltd         Buy         1,163,500         (4,950)           UBS Securities Australia Ltd         Buy         4,2784         5,569           UBS Securities Australia Ltd         Buy         4,2784         5,569           UBS Securities Australia Ltd         Buy         3,724         5,59           UBS Securities Australia Ltd         Buy         2,2,24         5,59           UBS Securities Australi	Nov-14	LIBC Counting Australia (44)	P. S.	200,01	867'	Ordinary
USS Securities Australia Ltd         Sell         CF3         (85)           USS Securities Australia Ltd         Sell         7.73         (80)           USS Securities Australia Ltd         Sell         7.970         (1000           USS Securities Australia Ltd         Sell         7.859         (572)           USS Securities Australia Ltd         Sell         7.859         (1080)           USS Securities Australia Ltd         Sell         1.9438         (1,1080)           USS Securities Australia Ltd         Sell         1.19,438         (1,1080)           USS Securities Australia Ltd         Buy         1,189,622         2,685           USS Securities Australia Ltd         Buy         1,163,250         1,45,068           USS Securities Australia Ltd         Buy         1,163,250         1,45,068           USS Securities Australia Ltd         Buy         1,163,250         1,45,068           USS Securities Australia Ltd         Buy         40,2933         5,269           USS Securities Australia Ltd         Buy         40,2933         5,269           USS Securities Australia Ltd         Buy         2,242         3,118           USS Securities Australia Ltd         Buy         2,242         3,789           U	Now.14	LIPS Securition Australia Ltd	, and	3,544	445	Ordinary
UBS Securities Australia Ltd         Self         7.73         (340)           UBS Securities Australia Ltd         Self         7.73         (340)           UBS Securities Australia Ltd         Self         7.59         (527)           UBS Securities Australia Ltd         Self         7.86         (572)           UBS Securities Australia Ltd         Self         7.86         (573)           UBS Securities Australia Ltd         Self         1.182         (1.086)           UBS Securities Australia Ltd         Self         1.182         2.487           UBS Securities Australia Ltd         Buy         2.1359         2.487           UBS Securities Australia Ltd         Buy         2.1359         2.487           UBS Securities Australia Ltd         Buy         4.2284         5.56           UBS Securities Australia Ltd         Buy         4.2284         5.435           UBS Securities Australia Ltd         Buy         4.2284         5.435           UBS Securities Australia Ltd         Buy         4.2284         5.435           UBS Securities Australia Ltd         Buy         2.4242         5.757           UBS Securities Australia Ltd         Buy         2.4242         7.77           UBS Securities Australia Ltd	Now-14	USC Counties Australia 144	100	261	(02)	Ordinary
UBS Securities Australia Ltd         BM         79,703         1,000           UBS Securities Australia Ltd         Sell         7,970         1,000           UBS Securities Australia Ltd         Sell         7,559         (572)           UBS Securities Australia Ltd         Sell         2,0,985         (7,633)           UBS Securities Australia Ltd         Sell         1,9,625         (7,639)           UBS Securities Australia Ltd         Buy         1,9,625         2,687           UBS Securities Australia Ltd         Buy         5,8,13         7,394           UBS Securities Australia Ltd         Buy         1,19,628         1,4,696           UBS Securities Australia Ltd         Buy         1,19,628         1,4,696           UBS Securities Australia Ltd         Buy         1,19,628         2,456           UBS Securities Australia Ltd         Buy         1,291         1,66           UBS Securities Australia Ltd         Buy         3,050         1,40,993           UBS Securities Australia Ltd         Buy         2,284         2,569           UBS Securities Australia Ltd         Buy         2,24,24         3,118           UBS Securities Australia Ltd         Buy         2,42,42         3,118           UBS Secur	Nov. 1.4	Total State A section 2 2001		0/0	(68)	Ordinary
UBS Securities Australia Ltd         Sell         7,00           UBS Securities Australia Ltd         Sell         7,866         (987)           UBS Securities Australia Ltd         Sell         2,559         (573)           UBS Securities Australia Ltd         Sell         1,656         (7,686)           UBS Securities Australia Ltd         Sell         1,168,08         (1,686)           UBS Securities Australia Ltd         Buy         1,163,08         1,46,08           UBS Securities Australia Ltd         Buy         1,163,08         1,46,08           UBS Securities Australia Ltd         Buy         1,683,73         1,46,08           UBS Securities Australia Ltd         Buy         1,63,09         5,256           UBS Securities Australia Ltd         Buy         4,2,264         4,77           UBS Securities Australia Ltd         Buy         4,2,264         4,77           UBS Securities Australia Ltd         Buy         4,2,264         4,77           UBS Securities Australia Ltd         Buy         2,4,42         3,118           UBS Securities Australia Ltd         Buy         2,4,42         3,118           UBS Securities Australia Ltd         Buy         2,2,42         3,118           UBS Securities Australia Ltd <td>NOV-14</td> <td></td> <td>Sell</td> <td>2,713</td> <td>(340)</td> <td>Ordinary</td>	NOV-14		Sell	2,713	(340)	Ordinary
UBS Securities Australia Ltd         Sell         4,556         (987)           UBS Securities Australia Ltd         Sell         4,556         (533)           UBS Securities Australia Ltd         Sell         20,985         (7,633)           UBS Securities Australia Ltd         Sell         19,662         2,467           UBS Securities Australia Ltd         Buy         19,662         2,467           UBS Securities Australia Ltd         Buy         1,63,950         1,4566           UBS Securities Australia Ltd         Buy         3,550         1,43           UBS Securities Australia Ltd         Buy         3,550         1,56           UBS Securities Australia Ltd         Buy         3,550         1,50           UBS Securities Australia Ltd         Buy         3,550         1,50           UBS Securities Australia Ltd         Buy         3,550         1,75           UBS Securities Australia Ltd         Buy         2,724         3,118           UBS Securities Australia Ltd         Buy         2,026         1,700           UBS Securities Australia Ltd         Buy         2,022         3,118           UBS Securities Australia Ltd         Buy         2,022         3,78           UBS Securities Australia Ltd	- KOA- 14	Obsideration Australia Ltd	Buy	79,700	10,000	Ordinary
UBS Securities Australia Ltd         Sell         (559         (572)           UBS Securities Australia Ltd         Sell         1,559         (7,533)           UBS Securities Australia Ltd         Buy         1,9428         (1,686)           UBS Securities Australia Ltd         Buy         2,1359         2,685           UBS Securities Australia Ltd         Buy         2,1359         2,685           UBS Securities Australia Ltd         Buy         1,163,289         1,150           UBS Securities Australia Ltd         Buy         1,163,280         1,45           UBS Securities Australia Ltd         Buy         1,231         1,50           UBS Securities Australia Ltd         Buy         4,2,264         3,50           UBS Securities Australia Ltd         Buy         4,2,264         3,50           UBS Securities Australia Ltd         Buy         4,2,94         3,50           UBS Securities Australia Ltd         Buy         2,24,22         3,118           UBS Securities Australia Ltd         Buy         2,24,22         3,118           UBS Securities Australia Ltd         Buy         2,24,22         3,118           UBS Securities Australia Ltd         Buy         1,724         2,56           UBS Securities Austral	-Nov-14	UBS Securities Australia Ltd	Sell	7,866	(282)	Ordinary
UBS Securities Australia Ltd         Sell         C.6.33           UBS Securities Australia Ltd         Buy         19 662         7.467           UBS Securities Australia Ltd         Buy         19 662         7.467           UBS Securities Australia Ltd         Buy         21,359         2.686           UBS Securities Australia Ltd         Buy         58,213         7.304           UBS Securities Australia Ltd         Buy         1,63,550         14,60,00           UBS Securities Australia Ltd         Buy         42,284         5,26           UBS Securities Australia Ltd         Buy         42,284         5,29           UBS Securities Australia Ltd         Buy         42,284         5,49           UBS Securities Australia Ltd         Buy         8,50         1,50           UBS Securities Australia Ltd         Buy         8,27,28         1,50           UBS Securities Australia Ltd         Buy         8,27,20         1,10,100           UBS Securities Australia Ltd         Buy         2,424         2,57           UBS Securities Australia Ltd         Buy         2,084         2,56           UBS Securities Australia Ltd         Buy         2,084         2,56           UBS Securities Australia Ltd         Buy <td>-Nov-14</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>4,559</td> <td>(572)</td> <td>Ordinary</td>	-Nov-14	UBS Securities Australia Ltd	Sell	4,559	(572)	Ordinary
UBS Securities Australia Ltd         Sell         19,652         (1,089)           UBS Securities Australia Ltd         Sell         19,428         (1,4366)           UBS Securities Australia Ltd         Buy         58,213         (1,4366)           UBS Securities Australia Ltd         Buy         58,213         7,304           UBS Securities Australia Ltd         Buy         1,163,350         1,45,608           UBS Securities Australia Ltd         Buy         1,231         6,505           UBS Securities Australia Ltd         Buy         40,993         5,256           UBS Securities Australia Ltd         Buy         40,993         5,266           UBS Securities Australia Ltd         Buy         8,247,000         1,00,000           UBS Securities Australia Ltd         Buy         8,247,000         1,707           UBS Securities Australia Ltd         Buy         24,242         3,118           UBS Securities Australia Ltd         Buy         2,242         3,561           UBS Securities Australia Ltd         Buy         2,024         1,770           UBS Securities Australia Ltd         Buy         2,024         1,737           UBS Securities Australia Ltd         Buy         2,024         1,737           UBS Secu	Nov-14	UBS Securities Australia Ltd	Sell	20,985	(2,633)	Ordinary
UBS Securities Australia Ltd         Bby         19,662         2,467           UBS Securities Australia Ltd         Buy         21,359         1(4,469           UBS Securities Australia Ltd         Buy         58,213         7,304           UBS Securities Australia Ltd         Buy         58,213         7,304           UBS Securities Australia Ltd         Buy         1,62,356         149,606           UBS Securities Australia Ltd         Buy         42,284         5,435           UBS Securities Australia Ltd         Buy         42,284         5,435           UBS Securities Australia Ltd         Buy         40,593         5,569           UBS Securities Australia Ltd         Buy         8,547         1,100,000           UBS Securities Australia Ltd         Buy         2,42,22         1,100,000           UBS Securities Australia Ltd         Buy         2,42,22         1,730           UBS Securities Australia Ltd         Buy         2,42,22         1,730           UBS Securities Australia Ltd         Buy         2,687         2,565           UBS Securities Australia Ltd         Buy         2,788         1,735           UBS Securities Australia Ltd         Buy         2,729         1,735           UBS Securities	-Nov-14	UBS Securities Australia Ltd	Sell	8,655	(1.086)	Ordinary
UBS securities Australia Ltd         Self         (1,9,428)         (1,4,696)           UBS Securities Australia Ltd         Buy         58,213         7,304           UBS Securities Australia Ltd         Buy         1,163,950         1,48,608           UBS Securities Australia Ltd         Buy         1,163,950         1,48,608           UBS Securities Australia Ltd         Buy         4,284         5,455           UBS Securities Australia Ltd         Buy         4,284         5,456           UBS Securities Australia Ltd         Buy         40,993         5,269           UBS Securities Australia Ltd         Buy         3,749         4,797           UBS Securities Australia Ltd         Buy         24,242         3,118           UBS Securities Australia Ltd         Buy         24,242         3,118           UBS Securities Australia Ltd         Buy         24,242         3,118           UBS Securities Australia Ltd         Buy         2,424         2,565           UBS Securities Australia Ltd         Buy         2,022         3,176           UBS Securities Australia Ltd         Buy         2,022         3,78           UBS Securities Australia Ltd         Buy         2,022         3,78           UBS Securities Au	Nov-14	UBS Securities Australia Ltd	Buv	19,662	737.7	Ordinon
UBS Securities Australia Ltd         Buy         21,329         1,4320           UBS Securities Australia Ltd         Buy         21,339         2,685           UBS Securities Australia Ltd         Buy         1,63,550         149,608           UBS Securities Australia Ltd         Buy         4,2,284         5,435           UBS Securities Australia Ltd         Buy         4,2,284         5,435           UBS Securities Australia Ltd         Buy         4,2,284         5,435           UBS Securities Australia Ltd         Buy         40,993         5,269           UBS Securities Australia Ltd         Buy         8,547,700         1,100,000           UBS Securities Australia Ltd         Buy         8,547,700         1,100,000           UBS Securities Australia Ltd         Buy         2,42,22         1,737           UBS Securities Australia Ltd         Buy         2,42,22         1,737           UBS Securities Australia Ltd         Buy         2,687         2,565           UBS Securities Australia Ltd         Buy         2,084         2,565           UBS Securities Australia Ltd         Buy         2,729         1,735           UBS Securities Australia Ltd         Buy         2,729         1,735           UBS Secu	-Nov-14	UBS Securities Australia Itd	و	110 430	7,407	Cidinary
UBS Securities Australia Ltd         Buy         58,213         7,304           UBS Securities Australia Ltd         Buy         58,213         7,304           UBS Securities Australia Ltd         Buy         1,62,950         149,506           UBS Securities Australia Ltd         Buy         4,224         5,435           UBS Securities Australia Ltd         Buy         4,224         5,435           UBS Securities Australia Ltd         Buy         4,729         4,797           UBS Securities Australia Ltd         Buy         8,547,000         1,100,000           UBS Securities Australia Ltd         Buy         2,422         3,118           UBS Securities Australia Ltd         Buy         2,422         3,118           UBS Securities Australia Ltd         Buy         2,022         3,118           UBS Securities Australia Ltd         Buy         1,7248         2,227           UBS Securities Australia Ltd         Buy         1,7248         2,527           UBS Securities Australia Ltd         Buy         2,026         1,735           UBS Securities Australia Ltd         Buy         2,026         1,735           UBS Securities Australia Ltd         Buy         2,026         1,735           UBS Securities Australia L	Nov-14	FID Consister Australia 14d		119,438	(14,986)	Ordinary
UBS Securities Australia Ltd         Buy         1,63,950         1,93,04           UBS Securities Australia Ltd         Buy         1,63,950         1,49,608           UBS Securities Australia Ltd         Buy         1,63,950         1,49,608           UBS Securities Australia Ltd         Buy         42,284         5,435           UBS Securities Australia Ltd         Buy         42,284         5,69           UBS Securities Australia Ltd         Buy         40,993         5,59           UBS Securities Australia Ltd         Buy         24,242         3,118           UBS Securities Australia Ltd         Buy         24,242         3,118           UBS Securities Australia Ltd         Buy         2,262         2,275           UBS Securities Australia Ltd         Buy         1,776         2,265           UBS Securities Australia Ltd         Buy         20,084         2,565           UBS Securities Australia Ltd         Buy         20,084         2,565           UBS Securities Australia Ltd         Buy         2,084         3,787           UBS Securities Australia Ltd         Buy         2,229         1,776           UBS Securities Australia Ltd         Buy         2,529         2,290           UBS Securities Austra	1100	Socialities Australia Liu	buy	665,12	2,685	Ordinary
UBS Securities Australia Ltd         Bluy         1,163.950         149.050           UBS Securities Australia Ltd         Buy         1,63.950         149.050           UBS Securities Australia Ltd         Buy         4,284         5,435           UBS Securities Australia Ltd         Buy         4,284         5,435           UBS Securities Australia Ltd         Buy         4,093         5,769           UBS Securities Australia Ltd         Buy         8,547,000         1,100,000           UBS Securities Australia Ltd         Buy         8,547,000         1,100,000           UBS Securities Australia Ltd         Buy         2,422         3,118           UBS Securities Australia Ltd         Buy         1,728         2,227           UBS Securities Australia Ltd         Buy         1,728         2,227           UBS Securities Australia Ltd         Buy         1,728         2,227           UBS Securities Australia Ltd         Buy         2,089         5,685           UBS Securities Australia Ltd         Buy         2,080         1,735           UBS Securities Australia Ltd         Buy         2,080         1,735           UBS Securities Australia Ltd         Buy         2,080         1,735           UBS Securities	-NOV-14	Obsideration Australia Ltd	Buy	58,213	7,304	Ordinary
UBS Securities Australia Ltd         Bby         1,163,950         14,950.8           UBS Securities Australia Ltd         Buy         3,050         332           UBS Securities Australia Ltd         Buy         42,284         5,435           UBS Securities Australia Ltd         Buy         40,993         5,269           UBS Securities Australia Ltd         Buy         40,993         5,269           UBS Securities Australia Ltd         Buy         24,242         3,118           UBS Securities Australia Ltd         Buy         24,242         3,118           UBS Securities Australia Ltd         Buy         1,7248         2,256           UBS Securities Australia Ltd         Buy         1,776         2,266           UBS Securities Australia Ltd         Buy         1,776         2,266           UBS Securities Australia Ltd         Buy         1,776         2,266           UBS Securities Australia Ltd         Buy         2,094         1,735           UBS Securities Australia Ltd         Buy         2,229         1,735           UBS Securities Australia Ltd         Buy         2,229         3,737           UBS Securities Australia Ltd         Buy         2,229         3,775           UBS Securities Australia Ltd <td>NOV-14</td> <td>UBS Securities Australia Etd</td> <td>Buy</td> <td>888</td> <td>11</td> <td>Ordinary</td>	NOV-14	UBS Securities Australia Etd	Buy	888	11	Ordinary
UBS Securities Australia Ltd         Buy         3.050         392           UBS Securities Australia Ltd         Buy         40.284         5,435           UBS Securities Australia Ltd         Buy         40.284         5,269           UBS Securities Australia Ltd         Buy         37,249         4,797           UBS Securities Australia Ltd         Buy         37,249         4,797           UBS Securities Australia Ltd         Buy         2,422         3,118           UBS Securities Australia Ltd         Buy         77,687         2,227           UBS Securities Australia Ltd         Buy         1,7248         2,227           UBS Securities Australia Ltd         Buy         1,726         2,227           UBS Securities Australia Ltd         Buy         1,726         2,527           UBS Securities Australia Ltd         Buy         1,736         2,527           UBS Securities Australia Ltd         Buy         2,687         877           UBS Securities Australia Ltd         Buy         2,529         377           UBS Securities Australia Ltd         Buy         2,529         377           UBS Securities Australia Ltd         Buy         39,77         4,681           UBS Securities Australia Ltd <td< td=""><td>Nov-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>1,163,950</td><td>149,608</td><td>Ordinary</td></td<>	Nov-14	UBS Securities Australia Ltd	Buy	1,163,950	149,608	Ordinary
UBS Securities Australia Ltd         Buy         42,884         5,435           UBS Securities Australia Ltd         Buy         1,291         1,66           UBS Securities Australia Ltd         Buy         3,249         1,797           UBS Securities Australia Ltd         Buy         8,547,000         1,707           UBS Securities Australia Ltd         Buy         24,242         3,118           UBS Securities Australia Ltd         Buy         1,776         2,257           UBS Securities Australia Ltd         Buy         1,776         2,267           UBS Securities Australia Ltd         Buy         1,776         2,267           UBS Securities Australia Ltd         Buy         2,028         1,778           UBS Securities Australia Ltd         Buy         2,226         1,735           UBS Securities Australia Ltd         Buy         2,226         1,735           UBS Securities Australia Ltd         Buy         2,729         3,73           UBS Securities Australia Ltd         Buy         2,729         3,77           UBS Securities Australia Ltd         Buy         39,776         5,080           UBS Securities Australia Ltd         Buy         3,977         4,88           UBS Securities Australia Ltd	Nov-14	UBS Securities Australia Ltd	Buy	3,050	392	Ordinary
UBS Securities Australia Ltd         Buy         1,291         1,66           UBS Securities Australia Ltd         Buy         37,249         5,769           UBS Securities Australia Ltd         Buy         37,249         4,799           UBS Securities Australia Ltd         Buy         2,422         3,118           UBS Securities Australia Ltd         Buy         2,422         3,118           UBS Securities Australia Ltd         Buy         1,7248         2,227           UBS Securities Australia Ltd         Buy         1,7248         2,527           UBS Securities Australia Ltd         Buy         1,734         2,527           UBS Securities Australia Ltd         Buy         1,736         2,527           UBS Securities Australia Ltd         Buy         1,736         2,527           UBS Securities Australia Ltd         Buy         1,736         2,529           UBS Securities Australia Ltd         Buy         2,529         337           UBS Securities Australia Ltd         Buy         39,776         5,080           UBS Securities Australia Ltd         Buy         39,767         5,080           UBS Securities Australia Ltd         Buy         39,507         7,586	Nov-14	UBS Securities Australia Ltd	Buy	42.284	5.435	Ordinary
UBS Securities Australia Ltd         Buy         40,993         5,269           UBS Securities Australia Ltd         Buy         37,249         4,797           UBS Securities Australia Ltd         Buy         8,547,000         1,700           UBS Securities Australia Ltd         Buy         22,424         3,118           UBS Securities Australia Ltd         Buy         7,7687         3,511           UBS Securities Australia Ltd         Buy         1,770         2,26           UBS Securities Australia Ltd         Buy         1,770         2,26           UBS Securities Australia Ltd         Buy         1,770         2,26           UBS Securities Australia Ltd         Buy         2,0984         2,559           UBS Securities Australia Ltd         Buy         2,59         3,73           UBS Securities Australia Ltd         Buy         2,59         3,77           UBS Securities Australia Ltd         Buy         39,776         5,080           UBS Securities Australia Ltd         Buy         39,776         5,080           UBS Securities Australia Ltd         Buy         39,776         5,687           UBS Securities Australia Ltd         Buy         39,776         5,687           UBS Securities Australia Ltd	Nov-14	UBS Securities Australia Ltd	Buy	1 201	331	Continue
USS Securities Australia Ltd         Buy         4,795 9         5,205 9           UBS Securities Australia Ltd         Buy         8,547,000         1,100,000           UBS Securities Australia Ltd         Buy         7,248         3,118           UBS Securities Australia Ltd         Buy         7,248         3,561           UBS Securities Australia Ltd         Buy         1,7248         2,227           UBS Securities Australia Ltd         Buy         1,7248         2,556           UBS Securities Australia Ltd         Buy         1,736         2,556           UBS Securities Australia Ltd         Buy         1,736         2,556           UBS Securities Australia Ltd         Buy         2,559         877           UBS Securities Australia Ltd         Buy         2,559         377           UBS Securities Australia Ltd         Buy         2,559         377           UBS Securities Australia Ltd         Buy         2,559         377           UBS Securities Australia Ltd         Buy         2,559         377           UBS Securities Australia Ltd         Buy         39,76         5,080           UBS Securities Australia Ltd         Buy         39,76         5,080           UBS Securities Australia Ltd	Nov.12	IIRS Securities Australia Ind	6-10	163'1	200	Cidinary
URS Securities Australia Ltd         Buy         8,347,000         1,700           URS Securities Australia Ltd         Buy         8,347,000         1,100           URS Securities Australia Ltd         Buy         2,242         3,118           URS Securities Australia Ltd         Buy         1,7248         2,527           URS Securities Australia Ltd         Buy         1,774         2,56           URS Securities Australia Ltd         Buy         1,585         1,735           URS Securities Australia Ltd         Buy         2,00         877           URS Securities Australia Ltd         Buy         5,697         877           URS Securities Australia Ltd         Buy         2,59         377           URS Securities Australia Ltd         Buy         2,59         377           URS Securities Australia Ltd         Buy         2,59         377           URS Securities Australia Ltd         Buy         2,59         377           URS Securities Australia Ltd         Buy         39,77         4,88           URS Securities Australia Ltd         Buy         39,77         4,88           URS Securities Australia Ltd         Buy         39,77         4,88           URS Securities Australia Ltd         Buy	1 1 1 1 1 1	one permiss Australia Liu	out.	40,993	697'5	Ordinary
UBS Securities Australia Ltd         Buy         8,547,000         1,100,000           UBS Securities Australia Ltd         Buy         24,247,000         1,100,000           UBS Securities Australia Ltd         Buy         77,687         3,551           UBS Securities Australia Ltd         Buy         1,770         22,27           UBS Securities Australia Ltd         Buy         1,770         2,565           UBS Securities Australia Ltd         Buy         1,776         2,565           UBS Securities Australia Ltd         Buy         6,667         877           UBS Securities Australia Ltd         Buy         2,573         323           UBS Securities Australia Ltd         Buy         2,573         323           UBS Securities Australia Ltd         Buy         2,573         323           UBS Securities Australia Ltd         Buy         2,573         323           UBS Securities Australia Ltd         Buy         2,573         323           UBS Securities Australia Ltd         Buy         3,977         4,88           UBS Securities Australia Ltd         Buy         3,677         4,88           UBS Securities Australia Ltd         Buy         5,680         7,88           UBS Securities Australia Ltd	NOV-14	UBS Securities Australia Ltd	Buy	37,249	4,797	Ordinary
UBS Securities Australia Ltd         Buy         2-4.2         3.118           UBS Securities Australia Ltd         Buy         27,687         3.51           USS Securities Australia Ltd         Buy         1,7248         2,227           UBS Securities Australia Ltd         Buy         1,7248         2,527           UBS Securities Australia Ltd         Buy         1,358         2,565           UBS Securities Australia Ltd         Buy         6,867         877           UBS Securities Australia Ltd         Buy         6,867         877           UBS Securities Australia Ltd         Buy         5,687         877           UBS Securities Australia Ltd         Buy         5,080         39,77         4,88           UBS Securities Australia Ltd         Buy         3,977         4,88         3,677         4,88           UBS Securities Australia Ltd         Buy         3,677         4,88         3,677         4,88           UBS Securities Australia Ltd         Buy         5,980         7,586         3,67         4,88	NOV-14	UBS Securities Australia Ltd	Buy	8,547,000	1,100,000	Ordinary
UBS Securities Australia Ltd         Bky         27,687         3,561           UBS Securities Australia Ltd         Bky         17,248         2,227           UBS Securities Australia Ltd         Bky         1,770         2,266           UBS Securities Australia Ltd         Bky         2,084         2,565           UBS Securities Australia Ltd         Bky         6,887         1,778           UBS Securities Australia Ltd         Bky         2,529         1,778           UBS Securities Australia Ltd         Bky         2,529         3,23           UBS Securities Australia Ltd         Bky         39,776         5,080           UBS Securities Australia Ltd         Bky         39,776         5,080           UBS Securities Australia Ltd         Bky         39,776         5,080           UBS Securities Australia Ltd         Bky         39,776         5,080	Nov-14	UBS Securities Australia Ltd	Buy	24,242	3,118	Ordinary
UBS Securities Australia Uch         Buy         17,248         2,227           U65 Securities Australia Uch         Buy         1,724         2,527           U65 Securities Australia Uch         Buy         1,770         256           U65 Securities Australia Uch         Buy         13,585         1,735           U65 Securities Australia Uch         Buy         6,867         877           U65 Securities Australia Uch         Buy         2,529         373           U65 Securities Australia Uch         Buy         2,529         377           U65 Securities Australia Uch         Buy         39,75         5,080           U65 Securities Australia Uch         Buy         39,75         4,88           U65 Securities Australia Uch         Buy         3,677         4,88           U65 Securities Australia Uch         Buy         5,080         7,586	Nov-14	UBS Securities Australia Ltd	Buy	27,687	3,561	Ordinary
URS Securities Australia Ltd         Bby         1,770         226           USS Securities Australia Ltd         Bby         2,0084         2,565           URS Securities Australia Ltd         Buy         13,588         1,735           URS Securities Australia Ltd         Buy         6,867         877           URS Securities Australia Ltd         Buy         2,529         323           URS Securities Australia Ltd         Buy         39,776         5,080           URS Securities Australia Ltd         Buy         39,776         4,88           URS Securities Australia Ltd         Buy         5,980         7,580	Nov-14	UBS Securities Australia Ltd	Buy	17.248	7227	Ordinary
UBS Securities Australia Ltd         Bby         20,084         2,565           UBS Securities Australia Ltd         Bby         13,585         1,735           UBS Securities Australia Ltd         Buy         6,867         877           UBS Securities Australia Ltd         Buy         2,529         323           UBS Securities Australia Ltd         Buy         3,677         483           UBS Securities Australia Ltd         Buy         5,080         483           UBS Securities Australia Ltd         Buy         5,360         7,586	Nov-14	UBS Securities Australia Ltd	Buy	1770	226	Ordinan
UBS Securities Australia Ltd         BM         CONDESS         1,735           UBS Securities Australia Ltd         BM         6,867         877           UBS Securities Australia Ltd         BM         5,59         873           UBS Securities Australia Ltd         BM         39,75         5,080           UBS Securities Australia Ltd         BM         39,75         5,080           UBS Securities Australia Ltd         BM         36,77         483           UBS Securities Australia Ltd         BM         5,080         7,586	Nov-14	IIRS Securities Australia I to	Blac	400 OC	200	Ordinary
UBS Securities Australia Ltd         Buy         1,735           UBS Securities Australia Ltd         Buy         6,867         877           UBS Securities Australia Ltd         Buy         2,529         323           UBS Securities Australia Ltd         Buy         3,776         483           UBS Securities Australia Ltd         Buy         3,767         483           UBS Securities Australia Ltd         Buy         5,980         7,586           UBS Securities Australia Ltd         Buy         59,360         7,586	No.17	High Constitute Australia Ind	pro.	20,084	7,505	Ordinary
UBS Securities Australia Ltd         Buy         6,867         877           UBS Securities Australia Ltd         Buy         2,529         323           UBS Securities Australia Ltd         Buy         39,776         5,080           UBS Securities Australia Ltd         Buy         33,777         483           UBS Securities Australia Ltd         Buy         59,800         7,586	1	Obsiserations Abstrated Ltd	pny	13,585	1,735	Ordinary
UBS Securities Australia Ltd         Buy         2.539         323           UBS Securities Australia Ltd         Buy         39,756         5,080           UBS Securities Australia Ltd         Buy         3,767         483           UBS Securities Australia Ltd         Buy         59,360         7,586	NOV-14	UBS Securities Australia Ltd	Buy	6,867	877	Ordinary
UBS Securities Australia Ltd         Buy         39,776         5,080           UBS Securities Australia Ltd         Buy         3,767         483           UBS Securities Australia Ltd         Buy         59,360         7,586	Nov-14	UBS Securities Australia Ltd	Buy	2,529	323	Ordinary
UBS Securities Australia Ltd Buy 3,767 483 UBS Securities Australia Ltd Buy 59,360 7,586	Nov-14	UBS Securities Australia Ltd	Buy	39,776	5,080	Ordinary
UBS Securities Australia Ltd Buy 59.360 7 5.86	Nov-14	UBS Securities Australia Ltd	Buy	797.5	1000	1
OSC CONTRACT CANADA CONTRACT C	Nov-14				2	Ciginary
	11-57	U De Constitute Assets 140	·a	0000	100	Ordinary

Buy	
Sing.	$  \cdot  $
liac d	
on)	-
(ag	ı
Sell	ı
Sell	
Sell	1
Sell	ı
Buy	- 1
Buy	
Je K	
and S	
100	
Sel	
Riv	1
Buv	1
Buv	1
BUV	ı
Bry	ı
Buy	
Buy	ı
8uv	1
Buy	1
2 2	ı
, a	1
Berry	
6 2	1
B. 0.	П
Big	1
í a	1
D	н
B. W.	1
Ruy	1
Buy	
BG.	1
Buy	1
Buy	ı
Buy	ı
Buy	
Buy	
Buy	
Buy	
Buy	
Buy	
Buy	
Buy	
Buy	
Buy	
Buv	
Buv	1
Buy	
Ri IV	1
, in	1
in a	
1	
ord.	1
guy.	i
Buy	Ì
Buy	
Buy	
Buy	
Buy	ı
Buv	
Rix	
o co	
eny	
Buy	
Buy	
Buy	- 1
Buy	
) id	
	Sell

1	Con Sell Control of the Control of t	ang.	17,054	2,167	Ordinary
28-Nov-14	UBS Securities Australia I to	Buy	103	7.5	1
28-Nov-14	UBS Securities Australia Ltd	8uv	707	72	Orginary
28-Nov-14	UBS Securities Australia Ltd	Buv	1 556	100	Crainary
28-Nov-14	UBS Securities Australia Ltd	Buv	270	96.	Ordinary
28-Nov-14	UBS Securities Australia Ltd	Buy	7 361	300	Ordinary
28-Nov-14	UBS Securities Australia Ltd	Buy	6.021	765	Ordinary
28-Nov-14	UBS Securities Australia Ltd	Buy	2.455	312	Ordinary
28-Nov-14	UBS Securities Australia Ltd	Buy	19.730	2 507	Ordinary
28-Nov-14	UBS Securities Australia Ltd	Buy	4 350	554	Ordinan
28-Nov-14	UBS Securities Australia Ltd	Buy	433	55	Ordinary
28-Nov-14	UBS Securities Australia Ltd	Buy	755 5	424	Ordinary
8-Nov-14	UBS Securities Australia Ltd	Buy	4.810	612	Ordinary
8-Nov-14	UBS Securities Australia Ltd	Buy	7.766	988	Ordinary
8-Nov-14	UBS Securities Australia Ltd	Buy	1.038	132	Ordinary
8-Nov-14	UBS Securities Australia Ltd	Buy	3 592	457	Company
28-Nov-14	UBS Securities Australia Ltd	Buy	78.550	10,000	Ordinary
8-Nov-14	UBS Securities Australia Ltd	Buy	4 150	825	Ordinary
28-Nov-14	UBS Securities Australia Ltd	Buy	28.702	3 647	Ordinary
28-Nov-14	UBS Securities Australia Ltd	Buy	31,397	3 997	Ordinary
28-Nov-14	UBS Securities Australia Ltd	Buy	2.629	PEE	Ordinary
8-Nov-14	UBS Securities Australia Ltd	Buy	1.487	189	Ordinary
28-Nov-14	UBS Securities Australia Ltd	Buy	α		Ordinan
8-Nov-14	UBS Securities Australia Ltd	Buy	3 423	435	Ordinary
8-Nov-14	UBS Securities Australia Ltd	Buy	3 935	000	Ordinan
8-Nov-14	UBS Securities Australia Ltd	Buy	21.195	2020	Commission
8-Nov-14	UBS Securities Australia Ltd	Buv	7 361	000	Ordinary
8-Nov-14	UBS Securities Australia Ltd	Buy	27 315	3 473	Sedinary
8-Nov-14	UBS Securities Australia Ltd	Buy	15 706	1 007	Ordinary
8-Nov-14	UBS Securities Australia Ltd	Buy	24 437	3 107	Ordinary
8-Nov-14	UBS Securities Australia Ltd	Buy	34.213	4 350	Ordinary
	UBS Securities Australia Ltd	Buy	1 243	821	Ordinary
ı	UBS Securities Australia Ltd	Buy	14 221	1 807	Ordinary
	UBS Securities Australia Ltd	Buy	285	300	Countries
ı	UBS Securities Australia Ltd	Buv	1 157	147	Ordinary.
	UBS Securities Australia Ltd	AN B	136.0	74	Crainary
l	UBS Securities Australia Itd	Bir	7 434	300	Coloring
ı	I IRS Securities Australia Ind	and and	174/	545	Ordinary
8-Nov-14	I IBS Securities Australia (194	(C)	304	571	Ordinary
ı	IIRS Securities Australia Ital	Buy	110,2	067	Ordinary
ı	IIBS Socientias Australia Ital	6.4	670	OS .	Ordinary
8-Nov-14	I IRC Societies Australia Ind	for d	055,01	2,161	Ordinary
	TIDE Committee Australia Ind	(ng	716	116	Ordinary
	Obsider Australia Ltd	, any	7,192	915	Ordinary
	UBS Securities Australia Ltd	Buy	2,523	321	Ordinary
1	UBS Securities Australia Ltd	Buy	2,523	321	Ordinary
1	UBS Securities Australia Ltd	Buy	1,680	214	Ordinary
ı	UBS Securities Australia Ltd	Buy	1,272	162	Ordinary
١	UBS Securities Australia Ltd	Buy	10,111	1,288	Ordinary
-	UBS Securities Australia Ltd	Buy	13,031	1,660	Ordinary
	UBS Securities Australia Ltd	Buy	8,447	1.076	Ordinary
i	UBS Securities Australia Ltd	Buy	149	- 10	Ordinary
	UBS Securities Australia Ltd	Buy	1,130	144	Ordinary
1-Dec-14	UBS Global Asset Management	Sell	5.043	(692)	Ordinary
01-Dec-14	UBS Securities Australia Ltd	Self	224.838	(78 374)	Ordinary
1-Dec-14	UBS Securities Australia Ltd	Sell	583 59	(8 5.02)	Ordinary.
1-Dec-14	UBS Securities Australia Ltd	Buv	85007	(C. 2/2)	Codings
1-Dec-14	UBS Securities Australia Itd	3	10,253	1003.0	Ordinary
1-Dec-14	UBS Securities Australia Itol	818	0140	(2,500)	Ordinary
1.Dec.14	IIRS Securities Australia 144	(Sell	0,143	1,083	Ordinary
01.Dec.14	I DC Counties Augustic Lea	1120	353,045	(47,529)	Ordinary
1 200	OBS Securities Abstralia Ltd	a constant	057,68	(11,702)	Ordinary
1-DEC-14	UBS Securities Australia Ltd	Seil	111,660	(14,682)	Ordinary
01-Dec-14	UBS Securities Australia Ltd	Buy	106,898	14,423	Ordinary
01-Dec-14	IUBS Securities Australia Ltd	Sell	105,121	(14,423)	Ordinary
1-Dec-14	UBS Securities Australia Ltd	Sell	446,622	(60,000)	Ordinary
01-Dec-14	UBS Securities Australia Ltd	Buy	112,698	15,000	Ordinary
1-Dec-14	UBS Securities Australia Ltd	Sell	2,534	(342)	Ordinary
1-Dec-14	UBS Securities Australia Ltd	Buy	331,740	44.665	Ordinary
1-Dec-14	UBS Securities Australia Ltd	Buy	291.366	39.056	Ordinary
01-Dec-14	UBS Securities Australia Ltd	Sell	87.390	(11.459)	Ordinary
01-Dec-14	IUBS Securities Australia Ltd	Buv	11,665	003 1	Ordinary
	UBS Securities Australia Itd	Biv	107 053	100.01	i de la
	I Do Committee Assessment And	Lange bossessie	20,000	104(0)	Ordinary
I	UBS Securities Australia Ltd	Stock borrowed	A/N	250,000	Ordinary

UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	áng u	010,0	449	Ordinary
UBS Securities Australia Ltd	IBUy	2.349	3001	Ordenary
Securities Australia 11d	Buy	5,724	731	Ordinary
Security Control and	Buy	1,879	240	Ordinary
Securities Australia Ltd	Buy	1,488	190	Ordinary
JBS Securities Australia Ltd	Buy	1,584	202	Ordinary
Securities Australia Ltd	Buy	2,352	300	Ordinary
UBS Securities Australia Ltd	Buy	65,516	8,362	Ordinary
Securities Australia Ltd	Buy	3,915	200	Ordinary
Securities Australia Ltd	Buy	1,490	190	Ordinary
Securities Australia Ltd	Buy	1,882	240	Ordinary
Securities Australia Ltd	Buy	201,449	25,695	Ordinary
Securities Australia Ltd	Buy	2,031	259	Ordinary
Securities Australia Ltd	Buv	8 553	1001	l
Securities Australia Ltd	Buy	2 897	370	l
S Securities Australia 11d	Riv	000 9	120	
IBS Securities Australia Itd	Bis	3 900	100	
IBC Correction Australia 14d	6.8	2007	Q P	Ordinary
Contract A contract Con	fina	555.6	404	Ordinary
S Securities Australia Ltd	áng.	501,61	2,440	Ordinary
UBS securities Australia Ltd	BUY	78,350	10,000	Ordinary
UBS Securities Australia Ltd	Buy	87,964	11,227	Ordinary
JBS Securities Australia Ltd	Buy	78,350	10,000	Ordinary
S Securities Australia Ltd	Buv	25,449	3746	Ordinary
IRS Securities Australia Ind	Biv	000	047'5	O'dinay
Securities Additional Co	624	000	4/	Orginary
UBS Securities Australia Ltd	hay	2,352	300	Ordinary
JBS Securities Australia Ltd	Buy	2,031	528	Ordinary
UBS Securities Australia Ltd	Buy	1,074	137	Ordinary
UBS Securities Australia Ltd	Buy	1215	145	Ordinary
S Securities Australia Ltd	Bit	200	200	, and a
2 C C - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1	600		04.7	Cidinaly
o securities Australia Lio	pna	086	125	Ordinary
S Securities Australia Ltd	Buy	13,552	1,722	Ordinary
S Securities Australia Ltd	Buy	3,935	200	Ordinary
UBS Securities Australia Ltd	Buy	1.055	134	Ordinary
S Securities Australia Ltd	Buy	23.610	3,000	Ordinary
S Securities Australia 14d	A I d	1,061	, Line	
Secondary Adams of Co.	fina	700'1	135	Orginary
s securities Australia Ltd	hag	105,462	13,409	Ordinary
S Securities Australia Ltd	Buy	53,380	6,787	Ordinary
5 Securities Australia Ltd	Buy	54.229	6.895	Ordinary
UBS Securities Australia Ltd	Buv	25 380	700 E	Ordinary
S Securities Australia I to	All B	000000	0000	(min)
Con announce Con constant	in a	100,02	2,300	Ordinary
S Secul lifes Australia LID	, and	63,282	8,046	Ordinary
UBS Securities Australia Ltd	Buy	2,361	300	Ordinary
UBS Securities Australia Ltd	Buy	645	82	Ordinary
URS Securities Australia 11d	Buv	1 456	101	Ordinaci
IBS Socurities Australia Ital	200	200	6	Country
Securities Australia Ltd	ín.	47/	76	Orginary
Obs securities Australia Ltd	kng.	/8/	100	Ordinary
Securities Australia Ltd	Buy	1,889	240	Ordinary
Securities Australia Ltd	Buy	3.974	505	Ordinary
UBS Securities Australia Ltd	BLIV	2 353	990	Configura
Day of the World of the Party o		5000	667	Clumary
Obs securities Australia Ltd	óno	7,888	36/	Ordinary
UBS Securities Australia Ltd	Buy	818	104	Ordinary
S Securities Australia Ltd	Buy	105,603	13.427	Ordinary
UBS Securities Australia Ltd	AG B	22 431	2,852	Ordinary
S Societies Australia Ltd	200	100 000	2007	0
Commence Com	in d	129,431	3,/42	Ordinary
Obs securities Australia Ltd	any	37,957	4,823	Ordinary
5 Securities Australia Ltd	Buy	787	100	Ordinary
5 Securities Australia Ltd	Buy	74566	3,811	Ordinary
5 Securities Australia Ltd	Buv	A1 960	5 335	Ordinary
Contration Assessed and	77.0	000	0000	200
o securities Australia Lito	for	845(1)	· /61	Ordinary
S Securities Australia Ltd	Buy	4,525	575	Ordinary
S Securities Australia Ltd	Buy	1,889	240	Ordinary
UBS Securities Australia Ltd	Buy	3.904	496	Ordinary
S Securities Australia I to	Riv	000	133	
C Securities Australia 14d	a a	010	707	o de la la la la la la la la la la la la la
Contradict Assets to the	600	0 0	3	Ordinary
o securines Appli alla Liu	eng.	3,730	4/4	Ordinary
UBS Securities Australia Ltd	Buy	2,361	300	Ordinary
Securities Australia Ltd	Buy	3,113	396	Ordinary
Securities Australia I to	N.B.	956	20	Ordinani
Countries Australia Inch	ec.)	606	/#	Ordinary
s securities Australia Ltd	euy	29,577	3,763	Ordinary
UBS Securities Australia Ltd	Виу	857	109	Ordinary
Securities Australia Ltd	Buy	2 795	797	Ordinary
A to the state of	(1)		767	C CHILLIAN
Securities Australia Ltd	euy.	1,234	157	Ordinary
Securities Australia Ltd	Buv	13 182	1 675	Ordinan

10   10   10   10   10   10   10   10	An An An An An An An An An An An An An A	3,482 9,270 4,836	1,246	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	50 A A A A A A A A A A A A A A A A A A A	4.836	1,245	
UBS Securities Australia tot UBS Securities A	uy Vuy Vuy	4,630	010	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	uy uy	1115		Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Lustralia Ltd	án án	כבר ל		Cranary
186 Securities Australia Ltd 185 Securities Australia Ltd 186 Securities Australia Ltd 187 Securities A	A1	757,7		Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd		199	001	Cranary
UBS Securites Australia Ltd UBS Securites Australia Ltd UBS Securities Australia Ltd		200 02		Ordinary
UBS Securities Australia Ltd UBS Securities UBS Securities Australia Ltd	, M	1116		Cidillary
UBS Securines Australia Ltd UBS Securines Australia Ltd		בייכי כ	000	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd		750.05	200	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd		50,03	5,092	Ordinary
1985 Securities Australia Ltd 1985 Securities Australia Ltd 1985 Securities Australia Ltd 1985 Securities Australia Ltd 1985 Securities Australia Ltd 1985 Securities Australia Ltd 1985 Securities Australia Ltd 1985 Securities Australia Ltd 1985 Securities Australia Ltd 1985 Securities Australia Ltd 1985 Securities Australia Ltd 1985 Securities Lude Ltd 1985 Securities		76	51.3	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd		75/7	0.00	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd		6 150	828	Ordinary
UNS Securities Australia Ltd UNS Securities Australia Ltd		סבר שנ	678	Ordinary
UNS Securities Australia Ltd UNS Securities Australia Ltd	4	55,05		Ordinary
USS Securities Australia Ltd USS Securities Australia Ltd	1	30,012		Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	6	15,529	7	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	,	2,734		Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	Ár.	5,673	764	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	A-	289		Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	, h	787		Ordinary
UNS Securities Australia Ltd UNS Securities Australia Ltd	Á	2,315		Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	á á	6,233	840	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	Á	47,970	6,465	Ordinary
1985 Securities Australia Ltd 1985 Securities Australia Ltd	ly.	5,164		Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	Ár	8,466		Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	Ár	237		Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	ŕ	326		Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	, fr	3,209	433	Ordinary
UBS Securities Autorials Ltd  UBS Securities Autorials Ltd	Ár	22	m	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	=	6.380	(198)	Ordinary
UBS Securities Autorala Ltd  UBS Securities Autorala Ltd	-	177 474	(1)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd		2 2 2 3		Conditions
UBS Securities Australia Ltd UBS Securities Australia Ltd		27,2	(2001)	Occuracy
UBS Securities Australia Ltd UBS Securities Australia Ltd		000.31		O COLUMN
UBS Securities Australia Ltd UBS Securities Australia Ltd		520,51		O COLLEGE
UBS Securities Australia Ltd UBS Securities Australia Ltd		27,2		Ordinary
USS Securities Australia Ltd USS Securities Australia Ltd		716,701	(14,563)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd		19,094		Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd		455,75		Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd		59,124		Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd		81,517		Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	=	54,352	(7,335)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	-	9,233		Ordinary
UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd  UBS Securities Australia Ltd	-	43,074	(5,813)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd		52	(2)	Ordinari
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd		25 00	(C 4 4 2)	ordina)
USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd USS Securities Australia Ltd		00,00	(144(0)	Cloring
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd		42,176	(769°C)	Ordinary
UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod	= =	84,911		Ordinary
UBS Securities Australia (1d UBS Securities Australia (1d UBS Securities Australia (1d UBS Securities Australia (1d UBS Securities Australia (1d UBS Securities Australia (1d UBS Securities Australia (1d UBS Securities Australia (1d UBS Securities Australia (1d UBS Securities Australia (1d UBS Securities Australia (1d UBS Securities Australia (1d UBS Securities Australia (1d UBS Securities Australia (1d UBS Securities Australia (1d	=	7,314		Ordinary
UBS Securities Australia (Ed. UBS) Securities Australia (Ed. U		4,446		Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	=	24.097		Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd		10.767		Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd		000 1000	(1 190 000)	Ordinan
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	>	1 023 400	140,000	S S
USS Securities Australia tod USS Securities Australia tod USS Securities Australia tod USS Securities Australia tod USS Securities Australia tod USS Securities Australia tod USS Securities Australia tod USS Securities Australia tod		00000	000,040	Cidinary
UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod UBS Securities Australia tod		16,388	7,248	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd		128,355	(17,607)	Ordinary
UBS Securities Australia Itd UBS Securities Australia Itd UBS Securities Australia Itd UBS Securities Australia Itd UBS Securities Australia Itd	y	116	15	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	λ.	23		Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS Securities Australia Ltd	λ	31	4	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd	Enter Into Derivative			
UBS Securities Australia Ltd UBS Securities Australia Ltd	אווופרו	AN	(50,000,000)	Options
UBS Securities Australia Ltd	Enter Into Derivative Contract	Q'N	(30,000,000)	Continue
UBS Securities Australia Ltd	Enter Into Derivative		(application)	S CONTRACTOR
	Contract	AVA	(15,000,000)	Options
100 C	Enter Into Derivative			
Obs securities Australia Ltd	Contract	NA	(15,000,000)	Options
	Enter Into Derivative			
U1-Dec-14 UBS Securities Australia Ltd Con	Contract	N/A	(200,000)	Options
Ente	Enter Into Derivative		_	

68 AG, Lendon Baneth         Boy         18,733         14,602           68 AG, Lendon Baneth         Sel         17,743,322         (48,000)           68 AG, Lendon Baneth         Sel         17,743,322         (48,000)           68 AG, Lendon Baneth         Sel         17,743,322         (48,000)           68 AG, Lendon Baneth         Boy         18,243         18,350           68 AG, Lendon Baneth         Boy         18,243         18,350           68 AG, Lendon Baneth         Boy         18,243         18,350           68 AG, Lendon Baneth         Boy         18,243         18,350           68 AG, Lendon Baneth         Boy         18,243         18,350           68 AG, Lendon Baneth         Boy         18,243         18,350           68 AG, Lendon Baneth         Boy         18,243         18,350           68 AG, Lendon Baneth         Boy         18,243         18,350           68 AG, Lendon Baneth         Boy         18,243         18,350           68 AG, Lendon Baneth         Boy         18,243         18,350           68 AG, Lendon Baneth         Boy         18,243         18,350           68 AG, Lendon Baneth         Boy         18,243         18,350	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinari.	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Options		Options		Options	Ordinary                     Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary																	
Buy   Buy	14,682	(240,000)	(000'09)	(1,320)	18,536	44,387	44,387	18,000	21,000	12,000	000 6	55 500	27 000	020,75	06 / 72	06/10	6,324	(005'1)	1,100,000		(110,000,000)	000 000	200,000,000	2,485	450.000	242	375	114	290	152	126	300	152	1367	157	274	1,284	8	184	760	300		351	2,278	147	736	151	288	300	148	298	287	158	298	216	201	1,031	15	220	1,555	300	2,134	278	2,071	3,511	111	871
	136,723	1,764,382	441,095	10,034	137,418	334,547	332,060	134,502	156,899	89,427	068'99	414 713	275 730	206.245	CF2,002	577,504	40,094 AVN	2	N/A		N/A	****	10101	10,001	3 339 000	1,803	2,794	849	2,161	1,132	7,010	2,235	1,132	21 744	1.170	2,041	995'6	09	1,371	1,430	1.074	7	2,615	16,971	1,096	5,483	1,125	2,146	2,235	1,096	566.6	2,141	1,179	2,223	1,609	1,497	7,681	112	1,039 000 C1	152 235	2,235	15,898	2,071	15,429	26,139	826	6.480
85 AG, London Branch 85 AG, London Branch 85 AG, Australia Branch 85 AG, Australia Branch 86 AG, Australia Branch 86 AG, Australia Branch 86 AG, Australia Branch 86 AG, Australia Branch 86 AG, Australia Branch 86 AG, Australia Branch 86 AG, Australia Branch 88 AG, Australia Branch 88 AG, Australia Branch 88 AG, Australia Branch 88 AG, Australia Branch 88 AG, Australia Branch 88 AG, Australia Branch 88 AG, Australia Branch 88 AG, Australia Branch 88 AG, Australia Branch 89 AG, Australia Branch 89 AG, Australia Branch 89 AG, Australia Branch 89 AG, Australia Branch 89 AG, Australia Branch 80 AG, Australia Branch 80 AG, Australia Branch 80 AG, Australia Branch 80 AG, Australia Branch 80 AG, Australia Branch 80 AG, Australia Branch 80 AG, Australia Branch 80 Securities Australia Ltd 80 Securities Australia Ltd 81 Securities Australia Ltd 82 Securities Australia Ltd 83 Securities Australia Ltd 85 Securities Australia	Buy	Sell	Sell	Sell	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	N. B.	800	Rusy	Birk	Transfer-Out	Enter Into Derivative	Contract	Enter Into Derivative	Contract	Enter Into Derivative	The state of the s	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	suy	Duy Buv	Bits	Buy	Buy	Buy	Buy	Buy	kna a	Buy	Burn	Buy	Buy	Buv	Buy                          - 86																					
	181	ង្គារ	IBS AG Australia Branch	IBS AG Australia Branch	IBS AG Australia Branch	IBS AG Australia Branch	IBS AG Australia Branch	PS AG Australia Branch	IBS AG Australia Branch	IBS AG Australia Branch	BS AG Australia Branch	BS AG Australia Branch	BS AG Australia Branch	BS AG Australia Branch	RS AG Australia Branch	RS AG Australia Branch	BS Wealth Management Australia I td		UBS Securities Australia Ltd		BS Securities Australia Ltd	Let is instance American Section 200	DS Securities Australia 14d	BS Securities Australia Ltd	BS Securities Australia Ltd	BS Securities Australia Ltd	BS Securities Australia Ltd	BS Securities Australia Ltd	BS Securities Australia Ltd	BS Securities Australia Ltd	BS Securities Australia Ltd	BS Securities Australia Ltd	BS Securities Australia Ltd	BS Securities Australia Itd	BS Securities Australia Ltd	BS Securities Australia Ltd	BS Securities Australia Ltd	BS Securities Australia Ltd	BS Securities Australia Ltd	BS Securities Australia Hd	85 Securities Australia Ltd	85 Securities Australia Ltd	BS Securities Australia Ltd	BS Securities Australia Ltd	35 Securities Australia Ltd	35 Securities Australia Ltd	35 Securities Australia Ltd	So Securities Australia Ltd	SS Securities Australia Ind	35 Securities Australia Ltd	35 Securities Australia Ltd	35 Securities Australia Ltd	3S Securities Australia Ltd	35 Securities Australia Ltd	35 Securities Australia Ltd	35 Securities Australia Ltd	35 Securities Australia Ltd	SS Securities Australia Ltd	55 Securities Australia Ltd	35 Securities Australia Ltd	35 Securities Australia Ltd	35 Securities Australia Ltd	35 Securities Australia Ltd	35 Securities Australia Ltd	55 Securities Australia Ltd	s Securities Australia Ltd	15 Securities Australia Ltd

Sell         641           Sell         641           Sell         642           Sell         684           Sell         684           Sell         684           Sell         123           Sell         128           Sell         128           Sell         138           Sell         143           Sell         143           Sell         143           Sell         143           Sell         143           Sell         143           Sell         474           Sell         474           Sell         474           Sell         474           Sell         474           Sell         474           Sell         474           Sell         474           Sell         474           Sell         474           Sell         474           Sell         474           Sell         474           Sell         474           Sell         474           Sell         474           Sell         473 <th>61</th> <th>(8)</th> <th>Ordinary</th>	61	(8)	Ordinary
Sell         661         (86)           Sell         53         (11)           Sell         23         (11)           Sell         289         (123)           Sell         289         (123)           Sell         289         (123)           Sell         289         (123)           Sell         289         (123)           Sell         289         (123)           Sell         289         (123)           Sell         289         (123)           Sell         289         (123)           Sell         172         (123)           Sell         172         (123)           Sell         172         (123)           Sell         173         (123)           Sell         173         (123)           Sell         173         (123)           Sell         173         (123)           Sell         173         (123)           Sell         173         (123)           Sell         (124)         (125)           Sell         (125)         (126)           Sell         (126)         (127)	641		
Sept		(84)	Ordinary
Self   253   (17)     Self   234   (17)	929	(98)	Ordinary
Sell	53	(2)	Ordinary
Self         644         (11)           Self         644         (12)           Self         1,235         (152)           Self         1,235         (152)           Self         1,235         (153)           Self         1,235         (153)           Self         1,235         (153)           Self         1,235         (153)           Self         1,235         (153)           Self         1,235         (153)           Self         1,235         (153)           Self         1,235         (153)           Self         1,235         (153)           Self         1,235         (154)           Self         4,187         (154)           Self         4,187         (154)           Self         4,187         (154)           Self         4,187         (154)           Self         4,187         (154)           Self         4,187         (154)           Self         4,187         (154)           Self         4,187         (154)           Self         4,187         (154)           Self         4,187<	237	(31)	Ordinary
Sept	847	0110	Ordinary
Sell         1,255         (162)           Sell         2,829         (173)           Sell         1,259         (173)           Sell         1,275         (168)           Sell         1,275         (168)           Sell         1,275         (169)           Sell         1,275         (164)           Sell         1,275         (164)           Sell         1,275         (164)           Sell         1,275         (164)           Sell         1,275         (164)           Sell         1,275         (164)           Sell         1,275         (164)           Sell         1,275         (164)           Sell         1,275         (164)           Sell         1,275         (164)           Sell         1,275         (164)           Sell         1,275         (164)           Sell         1,275         (164)           Sell         1,275         (164)           Sell         1,275         (164)           Sell         1,275         (164)           Sell         1,275         (164)           Sell	769	(10)	Ordinary
Sept	300 1	(10)	i i
See	(62,1	(791)	Ordinary
Self   2,000   (1/2)	2,829	(371)	Ordinary
Sell         1,256         (376)           Sell         1,276         (141)           Sell         1,976         (142)           Sell         1,976         (143)           Sell         2,886         (1728)           Sell         680         (193)           Sell         680         (193)           Sell         1,475         (143)           Sell         1,425         (143)           Sell         1,425         (143)           Sell         1,432         (144)           Sell         1,432         (146)           Sell         1,432         (146)           Sell         1,432         (146)           Sell         1,432         (147)           Sell         1,432         (148)           Sell         1,432         (147)           Sell         1,432         (147)           Sell         1,432         (147)           Sell         1,432         (147)           Sell         1,432         (147)           Sell         1,432         (147)           Sell         1,432         (147)           Sell         1,4	938	(123)	Ordinary
Self         1,1776         (16)           Self         1,076         (16)           Self         2,886         (17)           Self         647         (193)           Self         647         (194)           Self         (17)         (194)           Self         (17)         (194)           Self         (17)         (194)           Self         (184)         (194)           Self         (184)         (194)           Self         (194)         (194)           Self         (194)         (194)           Self         (194)         (194)           Self         (194)         (194)           Self         (194)         (194)           Self         (194)         (194)           Self         (194)         (194)           Self         (194)         (194)           Self         (194)         (194)           Self         (194)         (194)           Self         (194)         (194)           Self         (194)         (194)           Self         (194)         (194)           Self         (194) <td>2,869</td> <td>(376)</td> <td>Ordinaty</td>	2,869	(376)	Ordinaty
Self         1,076         (141)           Self         2,895         (142)           Self         2,895         (143)           Self         1,475         (143)           Self         1,475         (143)           Self         1,253         (143)           Self         1,253         (143)           Self         1,253         (143)           Self         1,253         (143)           Self         1,253         (144)           Self         1,253         (144)           Self         1,253         (144)           Self         1,253         (144)           Self         1,253         (144)           Self         1,253         (144)           Self         1,253         (144)           Self         1,123         (144)           Self         1,123         (144)           Self         1,123         (144)           Self         1,123         (144)           Self         1,123         (145)           Self         1,123         (144)           Self         1,124         (144)           Self	122	(16)	Ordinary
Sell         7-26         (152)           Sell         7-26         (104)           Sell         1,275         (104)           Sell         1,275         (104)           Sell         1,275         (104)           Sell         1,275         (104)           Sell         1,275         (104)           Sell         1,275         (104)           Sell         1,275         (104)           Sell         1,275         (104)           Sell         1,275         (104)           Sell         1,275         (104)           Sell         1,275         (104)           Sell         1,173         (104)           Sell         1,173         (104)           Sell         1,173         (107)           Sell         1,173         (107)           Sell         1,173         (107)           Sell         1,173         (107)           Sell         1,173         (107)           Sell         1,173         (107)           Sell         1,173         (107)           Sell         1,173         (107)           Sell         1,	3501	(0.10)	Control of
Self	0/0,1	(141)	Orginary
Self   2586	/48	(86)	Ordinary
Sell         2.886         (3-9)           Sell         642         (64)           Sell         642         (64)           Sell         (64)         (63)           Sell         (64)         (63)           Sell         (73)         (73)           Sell         (73)         (73)           Sell         (73)         (74)           Sell         (73)         (74)           Sell         (73)         (74)           Sell         (74)         (74)           Sell         (74)         (74)           Sell         (74)         (74)           Sell         (74)         (75)           Sell         (74)         (75)           Sell         (74)         (75)           Sell         (74)         (75)           Sell         (74)         (75)           Sell         (75)         (75)           Sell         (75)         (75)           Sell         (75)         (75)           Sell         (75)         (75)           Sell         (75)         (75)           Sell         (75)         (75)	952	(125)	Ordinary
Sell         755         (104)           Sell         1,253         (13)           Sell         1,253         (13)           Sell         1,253         (14)           Sell         1,253         (14)           Sell         1,253         (14)           Sell         1,253         (14)           Sell         1,253         (14)           Sell         4,187         (13)           Sell         4,187         (14)           Sell         4,187         (14)           Sell         4,187         (14)           Sell         1,172         (23)           Sell         1,172         (24)           Sell         1,172         (24)           Sell         1,172         (24)           Sell         1,172         (25)           Sell         1,172         (25)           Sell         1,172         (25)           Sell         1,172         (25)           Sell         1,172         (25)           Sell         1,172         (25)           Sell         1,172         (25)           Sell         1,172         (	2 896	(975)	Ordinary
Sell         642         (84)           Sell         642         (84)           Sell         73         (13)           Sell         73         (14)           Sell         73         (14)           Sell         73         (14)           Sell         73         (14)           Sell         73         (14)           Sell         73         (14)           Sell         73         (14)           Sell         73         (14)           Sell         43         (16)           Sell         4,187         (16)           Sell         4,187         (16)           Sell         4,187         (14)           Sell         4,187         (14)           Sell         1,172         (20)           Sell         1,172         (20)           Sell         1,172         (20)           Sell         1,172         (20)           Sell         1,183         (14)           Sell         1,183         (14)           Sell         1,123         (12)           Sell         1,123         (20)	305	000	
Sell   680   683   684   685	CE!	(101)	Ordinary
Sell         680         (89)           Sell         79         (13)           Sell         79         (13)           Sell         499         (14)           Sell         439         (14)           Sell         439         (14)           Sell         439         (14)           Sell         439         (14)           Sell         439         (14)           Sell         439         (14)           Sell         439         (14)           Sell         439         (14)           Sell         439         (14)           Sell         439         (14)           Sell         439         (14)           Sell         439         (14)           Sell         431         (15)           Sell         1,123         (14)           Sell         1,133         (14)           Sell         1,132         (120)           Sell         1,132         (120)           Sell         1,132         (120)           Sell         1,123         (120)           Sell         1,123         (120)	750	(88)	Ordinary
Sell         1,253         (193)           Sell         1,253         (134)           Sell         1,253         (164)           Sell         1,253         (164)           Sell         1,253         (164)           Sell         4,283         (164)           Sell         4,187         (165)           Sell         4,187         (165)           Sell         4,187         (163)           Sell         4,187         (163)           Sell         4,187         (163)           Sell         1,123         (173)           Sell         1,123         (173)           Sell         1,123         (173)           Sell         1,123         (173)           Sell         1,123         (173)           Sell         1,123         (173)           Sell         1,123         (173)           Sell         1,123         (173)           Sell         1,123         (173)           Sell         1,123         (173)           Sell         1,123         (173)           Sell         1,123         (173)           Sell	089	(68)	Ordinary
Sell         99         (1)           Sell         176         (23)           Sell         1,75         (24)           Sell         439         (44)           Sell         439         (44)           Sell         236         (44)           Sell         236         (44)           Sell         236         (44)           Sell         473         (56)           Sell         473         (56)           Sell         473         (56)           Sell         473         (56)           Sell         481         (4)           Sell         481         (4)           Sell         481         (4)           Sell         1,53         (23)           Sell         1,53         (23)           Sell         1,73         (23)           Sell         1,73         (23)           Sell         1,73         (23)           Sell         1,73         (23)           Sell         1,73         (23)           Sell         1,73         (23)           Sell         1,73         (23)           Se	1,475	(193)	Ordinary
Sell         99         (13)           Sell         1,253         (164)           Sell         1,253         (164)           Sell         206         (27)           Sell         428         (164)           Sell         428         (164)           Sell         438         (165)           Sell         418         (165)           Sell         418         (165)           Sell         418         (162)           Sell         418         (162)           Sell         418         (162)           Sell         1,123         (173)           Sell         1,123         (173)           Sell         1,123         (173)           Sell         1,123         (173)           Sell         1,123         (173)           Sell         1,123         (173)           Sell         1,123         (173)           Sell         1,123         (173)           Sell         1,123         (173)           Sell         1,123         (173)           Sell         1,123         (173)           Sell         1,123	8	ε	Ordinary
Sell         1,756         (22)           Sell         4,953         (164)           Sell         4,99         (64)           Sell         2,96         (47)           Sell         2,96         (47)           Sell         4,137         (56)           Sell         4,137         (56)           Sell         4,137         (54)           Sell         4,137         (54)           Sell         4,137         (54)           Sell         4,137         (54)           Sell         4,137         (54)           Sell         4,137         (54)           Sell         4,137         (54)           Sell         4,137         (54)           Sell         1,133         (1,1)           Sell         1,132         (1,2)           Sell         1,232         (22)           Sell         1,232         (22)           Sell         1,232         (22)           Sell         1,232         (22)           Sell         1,232         (22)           Sell         1,232         (22)           Sell         1,232	8	(13)	Ordinari
Seil	367	(CI)	, G. G.
Self	9/1	(53)	Ordinary
Seri         368         (44)           Seri         366         (47)           Seri         428         (44)           Seri         428         (48)           Seri         438         (48)           Seri         418         (18)           Seri         418         (18)           Seri         418         (18)           Seri         418         (19)           Seri         418         (19)           Seri         418         (19)           Seri         418         (19)           Seri         1,123         (19)           Seri         1,123         (19)           Seri         1,123         (19)           Seri         1,123         (147)           Seri         1,123         (147)           Seri         1,123         (147)           Seri         1,123         (15)           Seri         1,123         (147)           Seri         1,123         (147)           Seri         1,123         (147)           Seri         1,123         (147)           Seri         1,123         (140) </td <td>1,253</td> <td>(164)</td> <td>Ordinary</td>	1,253	(164)	Ordinary
Sell         336         (44)           Sell         206         (27)           Sell         428         (58)           Sell         4187         (58)           Sell         4187         (58)           Sell         4187         (58)           Sell         4187         (62)           Sell         4187         (62)           Sell         4187         (62)           Sell         4187         (62)           Sell         4187         (62)           Sell         4187         (62)           Sell         4187         (62)           Sell         4187         (62)           Sell         4187         (62)           Sell         1,722         (22)           Sell         1,722         (22)           Sell         1,722         (22)           Sell         1,722         (22)           Sell         1,722         (22)           Sell         1,722         (23)           Sell         1,722         (23)           Sell         1,722         (23)           Sell         1,722         (23) <td>489</td> <td>(64)</td> <td>Ordinary</td>	489	(64)	Ordinary
Sell         4.08         C27           Sell         4.28         C66           Sell         4.18 7         C62           Sell         4.18 7         C63           Sell         31 40 7         C63           Sell         31 40 7         C63           Sell         4.18 7         C63           Sell         4.18 7         C63           Sell         1.551         C(19)           Sell         1.521         C(29)           Sell         1.521         C(20)           Sell         1.722         C(27)           Sell         1.722         C(27)           Sell         1.722         C(27)           Sell         1.722         C(27)           Sell         1.722         C(27)           Sell         1.722         C(27)           Sell         1.722         C(27)           Sell         1.722         C(27)           Sell         1.722         C(27)           Sell         1.722         C(27)           Sell         1.722         C(27)           Sell         1.722         C(27)           Sell         2.283	336	(44)	Ordinary
Sell   478   C56     Sell   474   (42)     Sell   478   (42)     Sell   588   (478)     Sell   588   (478)     Sell   588   (478)     Sell   588   (489)     Sell   689   (489)     Sell   788   (489)	506	(2)	Ordinary
Sell         138         (18)           Sell         474         (62)           Sell         474         (62)           Sell         474         (62)           Sell         11         (19)           Sell         2,246         (29)           Sell         4,87         (63)           Sell         1,551         (19)           Sell         1,521         (147)           Sell         1,521         (139)           Sell         1,521         (147)           Sell         1,722         (227)           Sell         1,722         (227)           Sell         1,722         (227)           Sell         1,722         (227)           Sell         1,722         (227)           Sell         1,722         (227)           Sell         1,722         (227)           Sell         1,722         (227)           Sell         2,22         (427)           Sell         3,52         (400)           Sell         3,52         (400)           Sell         1,409         (400)           Sell         1,409	428	(3.5)	, deligination
Sell	000	(00)	Couling
Sell	or !	(0)	Ordinary
Sell         4,187         (44)           Sell         1,13         (43)           Sell         1,2346         (234)           Sell         1,551         (43)           Sell         1,551         (43)           Sell         1,531         (43)           Sell         1,532         (23)           Sell         1,732         (23)           Sell         1,732         (23)           Sell         1,732         (23)           Sell         1,732         (23)           Sell         1,732         (23)           Sell         1,732         (23)           Sell         1,732         (23)           Sell         1,732         (23)           Sell         1,732         (23)           Sell         1,732         (23)           Sell         1,732         (23)           Sell         1,732         (23)           Sell         1,732         (23)           Sell         1,732         (23)           Sell         1,732         (23)           Sell         2,840         (17)           Sell         1,732 <td< td=""><td>474</td><td>(62)</td><td>Ordinary</td></td<>	474	(62)	Ordinary
Sell	4,187	(548)	Ordinary
Sell	8	(1)	Ordinary
Sell         145         (19)           Sell         2,246         (29)           Sell         481         (29)           Sell         1,123         (23)           Sell         1,123         (23)           Sell         1,123         (23)           Sell         1,732         (25)           Sell         1,732         (25)           Sell         1,732         (25)           Sell         1,732         (25)           Sell         1,732         (25)           Sell         1,732         (25)           Sell         1,732         (25)           Sell         1,732         (25)           Sell         1,732         (25)           Sell         1,732         (25)           Sell         1,338         (72)           Sell         1,338         (40)           Sell         2,289         (400)           Sell         3,522         (400)           Sell         1,489         (40)           Sell         1,489         (40)           Sell         1,489         (40)           Sell         1,489         (4	31	(5)	Ordinary
Sell         2,246         (234)           Sell         628         (83)           Sell         1,551         (63)           Sell         1,526         (83)           Sell         1,531         (63)           Sell         1,732         (227)           Sell         1,732         (227)           Sell         1,732         (227)           Sell         1,521         (73)           Sell         1,522         (227)           Sell         1,527         (227)           Sell         1,527         (53)           Sell         2,933         (31)           Sell         2,527         (530)           Sell         2,527         (530)           Sell         3,533         (407)           Sell         3,533         (407)           Sell         3,533         (407)           Sell         3,533         (407)           Sell         1,489         (407)           Sell         1,489         (407)           Sell         1,499         (408)           Sell         1,499         (409)           Sell         3,485 <td>777</td> <td>805</td> <td>in the second</td>	777	805	in the second
Sell	0.00	(E)	Ciciliary
Sell   C25     Sell   1,551   C(23)     Sell   1,732   C(23)     Sell   1,732   C(23)     Sell   1,732   C(23)     Sell   1,732   C(23)     Sell   1,732   C(23)     Sell   1,732   C(23)     Sell   2,832   C(3)     Sell   2,832   C(3)     Sell   2,832   C(3)     Sell   2,832   C(3)     Sell   2,832   C(3)     Sell   2,243   C(3)     Sell   2,263   C(3)     Sell   3,022   C(3)     Sell   3,022   C(3)     Sell   3,022   C(3)     Sell   3,022   C(3)     Sell   3,022   C(40)     Sell   3,022	7,246	(294)	Ordinary
Sell         GGG         (82)           Sell         1,551         (747)           Sell         1,123         (747)           Sell         1,732         (257)           Sell         1,732         (277)           Sell         1,732         (277)           Sell         1,732         (277)           Sell         1,732         (277)           Sell         1,732         (273)           Sell         1,732         (273)           Sell         1,732         (273)           Sell         1,732         (273)           Sell         1,388         (72)           Sell         1,388         (72)           Sell         1,388         (407)           Sell         2,283         (407)           Sell         3,522         (400)           Sell         1,489         (400)           Sell         1,489         (400)           Sell         1,489         (400)           Sell         1,489         (400)           Sell         1,489         (400)           Sell         1,499         (400)           Sell         1,499	481	(63)	Ordinary
Self         1,551         (203)           Self         1,123         (1,47)           Self         4,851         (25)           Self         1,732         (227)           Self         1,732         (227)           Self         7,827         (227)           Self         7,827         (27)           Self         7,827         (85)           Self         7,827         (86)           Self         7,827         (87)           Self         2,827         (72)           Self         2,827         (72)           Self         2,823         (73)           Self         3,633         (40)           Self         3,633         (40)           Self         3,632         (40)           Self         3,632         (40)           Self         3,632         (40)           Self         1,499         (40)           Self         1,499         (40)           Self         3,622         (40)           Self         3,485         (40)           Self         3,485         (40)           Self         3,485	929	(82)	Ordinary
Sell         4,123         (47)           Sell         1,122         (25)           Sell         1,722         (227)           Sell         1,722         (227)           Sell         1,722         (277)           Sell         1,722         (277)           Sell         1,722         (27)           Sell         7,52         (59)           Sell         7,52         (59)           Sell         1,382         (7,21)           Sell         1,382         (7,21)           Sell         1,382         (7,21)           Sell         7,28         (607)           Sell         7,55         (607)           Sell         3,52         (400)           Sell         3,52         (400)           Sell         1,489         (400)           Sell         1,489         (400)           Sell         1,489         (400)           Sell         1,489         (400)           Sell         1,489         (400)           Sell         1,489         (400)           Sell         1,499         (400)           Sell         1,499 <td>1.551</td> <td>(502)</td> <td>Ordinary</td>	1.551	(502)	Ordinary
Sell         4,851         (635)           Sell         1,732         (25)           Sell         1,732         (227)           Sell         1,732         (227)           Sell         443         (85)           Sell         7,827         (227)           Sell         7,827         (89)           Sell         7,827         (89)           Sell         7,827         (89)           Sell         7,827         (89)           Sell         7,827         (89)           Sell         7,827         (89)           Sell         2,823         (47)           Sell         3,63         (467)           Sell         3,63         (467)           Sell         3,63         (407)           Sell         3,63         (407)           Sell         3,63         (407)           Sell         3,63         (407)           Sell         3,62         (400)           Sell         3,62         (400)           Sell         1,499         (406)           Sell         1,499         (406)           Sell         3,485	1173	(147)	Company
Self	100	(12.3)	500
Sell	100'#	(000)	Ordinary
Sell	161	(25)	Ordinary
Sell         1,72         6273           Sell         443         (58)           Sell         160         (77)           Sell         7,587         (995)           Sell         7,587         (995)           Sell         7,587         (77)           Sell         7,587         (78)           Sell         7,587         (78)           Sell         7,587         (78)           Sell         7,587         (78)           Sell         7,587         (78)           Sell         7,587         (78)           Sell         3,563         (40)           Sell         3,563         (40)           Sell         3,563         (40)           Sell         3,563         (40)           Sell         3,563         (40)           Sell         7,59         (40)           Sell         7,59         (40)           Sell         7,59         (40)           Sell         7,59         (40)           Sell         7,59         (40)           Sell         7,59         (40)           Sell         7,50         (40)	1,732	(227)	Ordinary
Sell         443         (58)           Sell         38         (51)           Sell         7 597         (93)           Sell         7 592         (94)           Sell         4 532         (94)           Sell         2 983         (171)           Sell         1 386         (178)           Sell         2 923         (178)           Sell         2 283         (991)           Sell         2 22         (52)           Sell         3 563         (473)           Sell         3 563         (467)           Sell         3 563         (467)           Sell         3 563         (467)           Sell         3 563         (467)           Sell         3 562         (467)           Sell         3 562         (467)           Sell         3 562         (467)           Sell         1 4499         (496)           Sell         1 4499         (496)           Sell         1 4499         (496)           Sell         1 4499         (496)           Sell         1 4499         (496)           Sell         2 296 </td <td>1,732</td> <td>(722)</td> <td>Ordinary</td>	1,732	(722)	Ordinary
Sell         38         (5)           Sell         160         (7)           Sell         7.587         (955)           Sell         923         (171)           Sell         2.383         (391)           Sell         1,358         (178)           Sell         7,587         (391)           Sell         7,23         (302)           Sell         4,631         (62)           Sell         2,282         (407)           Sell         3,563         (407)           Sell         3,563         (400)           Sell         3,562         (400)           Sell         3,562         (400)           Sell         3,562         (400)           Sell         3,562         (400)           Sell         1,53         (40)           Sell         1,53         (40)           Sell         1,499         (196)           Sell         3,485         (405)           Sell         3,485         (405)           Sell         3,485         (405)           Sell         3,485         (405)           Sell         3,485	443	(58)	Ordinary
Self   160 (27)     Self   4.532 (935)     Self   2.983 (935)     Self   2.983 (931)     Self   1,386 (178)     Self   2,383 (931)     Self   2,383 (931)     Self   2,383 (931)     Self   2,383 (931)     Self   3,583 (467)     Self   3,583 (467)     Self   3,583 (467)     Self   3,583 (467)     Self   3,583 (467)     Self   3,583 (467)     Self   3,583 (467)     Self   3,583 (467)     Self   3,583 (467)     Self   3,583 (467)     Self   1,499 (198)     Self   1,499 (198)     Self   1,499 (198)     Self   1,499 (198)     Self   1,499 (198)     Self   1,499 (198)     Self   1,499 (198)     Self   3,485 (495)     Self   4,485 (495)	86	(5)	Ordinan
Sell	95	(c)	Ordinary
Sell         7,597         (959)           Sell         7,529         (171)           Sell         1,382         (171)           Sell         1,383         (178)           Sell         2,283         (32)           Sell         2,283         (778)           Sell         2,28         (677)           Sell         7,55         (677)           Sell         3,52         (400)           Sell         3,52         (400)           Sell         3,52         (400)           Sell         3,52         (400)           Sell         1,499         (400)           Sell         1,499         (400)           Sell         1,499         (400)           Sell         1,499         (400)           Sell         1,499         (400)           Sell         1,499         (400)           Sell         3,485         (400)           Sell         3,485         (400)           Sell         3,485         (400)           Sell         3,485         (400)           Sell         3,485         (400)           Sell         3,485 </td <td>160</td> <td>(21)</td> <td>Ordinary</td>	160	(21)	Ordinary
Sell         4552         (594)           Sell         923         (731)           Sell         1,383         (712)           Sell         7,32         (723)           Sell         7,32         (723)           Sell         7,53         (720)           Sell         7,55         (69)           Sell         3,563         (467)           Sell         3,563         (467)           Sell         3,52         (400)           Sell         3,62         (400)           Sell         3,62         (400)           Sell         1,53         (20)           Sell         1,439         (196)           Sell         1,439         (196)           Sell         1,439         (40)           Sell         3,485         (40)           Sell         3,485         (40)           Sell         3,485         (40)           Sell         3,485         (40)           Sell         3,485         (40)           Sell         3,485         (40)           Sell         3,485         (40)           Sell         3,485 <t< td=""><td>7,597</td><td>(962)</td><td>Ordinary</td></t<>	7,597	(962)	Ordinary
Sell         923         (71)           Sell         2.983         (91)           Sell         1,386         (778)           Sell         2.283         (92)           Sell         2.21         (22)           Sell         2.22         (32)           Sell         7.55         (33)           Sell         3.63         (40)           Sell         3.62         (40)           Sell         3.62         (40)           Sell         3.26         (40)           Sell         3.26         (40)           Sell         3.26         (40)           Sell         1.499         (40)           Sell         1.499         (40)           Sell         1.499         (40)           Sell         1.499         (40)           Sell         1.499         (40)           Sell         3.485         (40)           Sell         3.485         (40)           Sell         3.485         (40)           Sell         3.485         (40)           Sell         3.485         (40)           Sell         3.485         (40)	4,532	(594)	Ordinary
Sell         2.983         (391)           Sell         7.38         (778)           Sell         7.21         (29)           Sell         4.631         (20)           Sell         7.52         (33)           Sell         7.55         (99)           Sell         3.523         (467)           Sell         2.289         (30)           Sell         3.52         (400)           Sell         3.62         (400)           Sell         3.62         (400)           Sell         1.53         (20)           Sell         1.53         (20)           Sell         1.439         (196)           Sell         1.439         (196)           Sell         3.485         (405)           Sell         3.485         (405)           Sell         3.485         (405)           Sell         3.485         (405)           Sell         3.485         (405)           Sell         3.485         (405)           Sell         3.485         (405)           Sell         3.485         (405)           Sell         4.499	826	(121)	Ordinary
Sell         (178)           Sell         (178)           Sell         (22)         (52)           Sell         221         (29)           Sell         322         (52)           Sell         322         (53)           Sell         322         (53)           Sell         2.289         (30)           Sell         3.63         (40)           Sell         3.52         (40)           Sell         3.52         (40)           Sell         3.52         (40)           Sell         1.499         (10)           Sell         1.499         (10)           Sell         1.499         (40)           Sell         1.499         (40)           Sell         1.499         (40)           Sell         2.295         (300)           Sell         1.499         (40)           Sell         3.485         (40)           Sell         3.485         (40)           Sell         3.485         (40)           Sell         3.485         (40)           Sell         4.499         4.499           Sell	680 0	(301)	Ordinaso
Sell	030	(166)	(igiligi)
Sell	000	(8/1)	Ordinary
Sell         4.531         (2.9)           Sell         4.531         (6.07)           Sell         2.22         (3.9)           Sell         2.289         (4.09)           Sell         2.289         (4.00)           Sell         3.623         (4.00)           Sell         3.623         (4.00)           Sell         3.52         (4.00)           Sell         3.52         (4.00)           Sell         1.53         (2.0)           Sell         1.499         (1.96)           Sell         1.499         (4.15)           Sell         2.295         (3.00)           Sell         3.485         (4.5)           Sell         1.499         (4.95)           Sell         3.485         (4.95)           Sell         3.485         (4.95)           Sell         3.485         (4.95)           Sell         3.485         (4.95)           Sell         4.15         (4.95)           Sell         4.15         (4.95)           Sell         4.15         (4.95)           Sell         4.16         (4.19)           Sell	707	(35)	Ordinary
Sell         4531         (607)           Sell         252         (53)           Sell         352         (53)           Sell         3,63         (467)           Sell         2,289         (300)           Sell         3,032         (400)           Sell         3,022         (400)           Sell         3,62         (400)           Sell         153         (50)           Sell         1,53         (20)           Sell         1,439         (198)           Sell         2,295         (300)           Sell         3,485         (405)           Sell         3,485         (405)           Sell         3,485         (405)           Sell         3,485         (405)           Sell         3,485         (405)           Sell         3,485         (405)           Sell         3,485         (405)           Sell         3,485         (405)           Sell         3,485         (405)           Sell         3,485         (405)           Sell         3,485         (405)           Sell         3,485	221	(53)	Ordinary
Self   252 (33)     Self   3,563 (467)     Self   3,563 (467)     Self   3,663 (467)     Self   3,662 (407)     Self   3,662 (407)     Self   3,662 (407)     Self   3,662 (407)     Self   3,662 (407)     Self   3,662 (407)     Self   1,439 (105)     Self   1,439 (105)     Self   1,439 (105)     Self   3,485 (405)     Self   3,485 (405)     Self   3,485 (405)     Self   3,485 (405)     Self   3,485 (405)     Self   4,499 (405)     Self   4,490 (405)	4,631	(607)	Ordinary
Sel   755 (99)     Sel   2.289 (400)     Sel   2.289 (400)     Sel   3.052 (400)     Sel   3.052 (400)     Sel   3.052 (400)     Sel   3.052 (400)     Sel   1.59 (196)     Sel   1.59 (196)     Sel   1.499 (196)     Sel   2.295 (300)     Sel   3.485 (405)     Sel	252	(33)	Ordinary
Self	336	(60)	6
Sell	1000	(66)	Cidinary
Sell	5,553	(46/)	Ordinary
Sell	2,289	(300)	Ordinary
Sell   3.052 (400)     Sell   637 (400)     Sell   637 (427)     Sell   750 (427)     Sell	168	(22)	Ordinary
Sell 38.2 (50)     Sell 3.262 (427)     Sell 1.52 (20) (488)     Sell 7.50 (488)     Sell 7.50 (488)     Sell 7.50 (488)     Sell 7.50 (488)     Sell 7.50 (488)     Sell 7.50 (488)     Sell 7.50 (498)     Sell 7.50 (40)	3 052	(000)	Ordina
Sell   3,262 (427)     Sell   3,262 (427)     Sell   1,439 (196)     Sell   1,439 (196)     Sell   2,295 (300)     Sell   3,485 (465)     Sell   3,485 (465)     Sell   3,485 (465)     Sell   3,485 (465)     Sell   3,485 (465)     Sell   3,485 (495)     Sell   4,485 (495)	1000	(0.10)	6
Sell	700	(ac)	Ordinary
Sell   3.262   (427)     Sell   153   (20)     Sell   750   (196)     Sell   1,499   (196)     Sell   3,485   (465)     Sell   3,485   (465)     Sell   447   (4)     Sell   44	/\$9	(98)	Ordinary
Sell   750 (20)     Sell   750 (408)     Sell   2,285 (405)     Sell   3,185 (40)     Sell   3,185 (40)     Sell   3,185 (40)     Sell   4,185 (40)     Sell   4,185 (40)     Sell   4,185 (40)     Sell   4,185 (40)     Sell   4,185 (40)     Sell   4,185 (40)     Sell   4,185 (40)     Sell   4,185 (40)     Sell   4,185 (40)     Sell   4,185 (40)     Sell   5,185 (40)	3,262	(427)	Ordinary
Sell   750   (36)     Sell   1,499   (196)     Sell   2,295   (30)     Sell   3,485   (455)     Sell   3,485   (455)     Sell   146   (19)     Sell   757   (4)	153	(20)	Ordinary
Sell	COF	000	
Sel	nc/	(98)	Ordinary
Sei   2.955   (300)     Sei   2.465   (455)     Sei   31   (4)     Sei   146   (19)     Sei   146   (19)	1,499	(1961)	Ordinary
Sell   3,485   (455)	2.295	(300)	Ordinan
Sell	2 400	(400)	Control of
Sel	3,485	(455)	Crainary
Sell (19)	31	(4)	Ordinary
1617 C21C [162]	146	(01)	Ordinan
		(61)	A III
(66) /5/	757	(66)	Ordinary
	146 757 765		(99) (001)
			١
	╾┟╾╎╾│╾╿╼╿╼╿╼┞╼┞╼┞╼╀╼┦╼╿╼╿╼╟╼╟╼╟╼╟╼╟╼╟╼╟╼╟╼╟╼╟╼╟╌╟╌╟╌╟╌╟╌╟╌╟╌╟		955  2,856  475  640  640  1,475  8  8  99  99  99  99  99  99  99  99

		TENTER 1010 DELIVERINE			
01-Dec-14	UBS Securities Australia Ltd	Contract	N/A	(150,000)	Options
01-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	NA	15,000,000	Options
01-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	ΑVA	(300,000)	Options
01-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	AW	30,000,000	Options
01-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	NA	15,000,000	Options
01-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	Α/N	110,000,000	Options
01-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	N/A	(3,200,000)	Options
01-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	Ϋ́	(200,000,000)	Options
01-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	Ž	500 000 000	orito C
100	Late of the state	Enter Into Derivative		000	Schiolis
02-Dec-14	UBS Global Asset Management	Sell	N/A	(212)	Options
02-Dec-14	UBS Securities Australia Ltd	Sell	225,839	(29,318)	Ordinary
02-Dec-14	UBS Securities Australia Ltd	Buy	322,452	41,816	Ordinary
	UBS Securities Australia Ltd	Sell	307,572	(40,000)	Ordinary
	UBS Securities Australia Ltd	Buy	32,376	4,200	Ordinary
1	UBS Securities Australia Ltd	Sell	939,780	(122,399)	Ordinary
	UBS Securities Australia Ltd	Buy	180,177	23.500	Ordinary
П	UBS Securities Australia Ltd	Sell	95,560	(12,500)	Ordinary
-	UBS Securities Australia Ltd	Buy	516,096	67,221	Ordinary
	I IRS Securities Australia Ltd	Sell Bro	4,873	(637)	Ordinary
П	UBS Securities Australia Ltd	Self	578,754	(75,240)	Ordinary
02-Dec-14	UBS Securities Australia Ltd	Stock returned	A/N	(30,000)	Ordinary
	UBS AG London Branch	Sell	3,653	(480)	Ordinary
02-Dec-14	UBS AG, London Branch	Sell	723	(36)	Ordinary
02-Dec-14	UBS AG, London Branch	Sell	449	(65)	Ordinary
02-Dec-14	UBS AG London Branch	Sell	495	(59)	Ordinary
02-Dec-14	UBS AG, London Branch	Sell	754	(66)	Ordinary
$\ $	UBS AG, London Branch	Sell	8/9	(68)	Ordinary
	UBS AG, London Branch	Sell	831	(109)	Ordinary
02-Dec-14	UBS AG London Branch	Self	754	(66)	Ordinary
	UBS AG, London Branch	Sell	175	(191)	Ordinary
02-Dec-14	UBS AG, London Branch	Sell	2,863	(375)	Ordinary
02-Dec-14	UBS AG, London Branch	Sell	2,710	(355)	Ordinary
02-Dec-14	UBS AG, London Branch	F	683	(110)	Ordinary
02-Dec-14	UBS AG, London Branch	Sell	1.841	(1)	Ordinary
	UBS AG, London Branch	Sell	1,267	(166)	Ordinary
	UBS AG, London Branch	Sell	1,634	(214)	Ordinary
	UBS AG, London Branch	Sell	80 4	9	Ordinary
02-Dec-14	UBS AG, London Branch	Sell	2.352	(308)	Ordinary
02-Dec-14	UBS AG, London Branch	Sell	2,466	(323)	Ordinary
02-Dec-14	UBS AG, London Branch	Seli	3,634	(476)	Ordinary
02-Dec-14	UBS AG, London Branch	Sel.	199	(26)	Ordinary
	USS AG, London Branch	Sell	1,208	(158)	Ordinary
	UBS AG, London Branch	Sell	3,060	(401)	Ordinary
	UBS AG, London Branch	Sell	359	(47)	Ordinary
1	UBS AG, London Branch	Sell	56	(13)	Ordinary
02-Dec-14	UBS AG, London Branch	Sell	73	(97)	Ordinary
02-Dec-14	UBS AG, London Branch	Sell	122	(16)	Ordinary
02-Dec-14	UBS AG, London Branch	Sell	199	(26)	Ordinary
02-Dec-14	UBS AG, London Branch	Sell	802	(105)	Ordinary
	UBS AG, London Branch	Sell	2,170	(284)	Ordinary
	URS AG London Branch	las	33	(34)	Ordinary
			67	í)	

Ordinary	Ordinary	Cretions	Ordinary	Cidinary	Orginary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Cromary
(1,291)	(0.670)	(100)	(1 078)	(0/0,1)	(05/)	(2,714)	(100,1)	(801)	(510,11)	(93)	(67)	(()	(1,327)	(1,313)	(9)	(193)	(27)	(4,352)	(30)	(467)	(2,571)	(2,329)	(110)	(72)	(48)	(57)	(1,956)	(457)	(19)	(492)	(581)	(007)	(8,17,18)	(3,307)	(1,401)	(3,697)	(124)	(47)	(129)	(4,147)	(5,382)	(7,00,6)	(1,182)	(122,399)	95,721	72 569	(7,065)	(092)	8,000	842	(154)	(466)	(151)	(401)	(286)	(366)	(176)	(432)	(140)	(1961)	(108)	(10)	(2.128)	(41)	(230)	(674)	(652)	(338)	(000)	500
10,005	20 626	1 392	8 328	075'5	950.00	906,02	060,0	632	010,7	/1/	577	875,5	10,158	10,058	46	1,478	202	33,315	230	3,577	19,707	17,863	1,304	552	371	440	15,100	3,528	14/	3,567	4,485	5,404	15,203	50001	9 575	28.559	954	362	866	31,932	41,441	69,893	9,101	939,780	1,162,/38	2,0,0,5	54.245	5.845	61,976	6,500	1,178	3,565	1,155	3,068	4,483	2,804	1,348	3,305	2/0,1	797.0	8 487	77	16.279	314	4,065	5,163	4,994	2,589	291	
Sell	Sell	les	Sell	i i	1120	190	199	Coll	lis.	9	1100	lac	Xell	Sell	Sell	Sell	Selt	Sell	Sell	Sell	Sell	Sell	Sell	Sell	Sell	Sell	) sell	Sell Sell	near and a	liac liac	nec.	ilac ilac	Jen Z	100	Sell Sell	Sell	Sell	Sell	Sell	Sell	Sell	Seil	3.	Serie	and B. Inc.	(F)	- Rel	Sell	Buy	Buy	Sell	Sell	Sell	Sell	Nell Sell	Sell Sell	Jac Jac Jac Jac Jac Jac Jac Jac Jac Jac	llac llac	Jell Sell	Sell	Seil	Seli	Self	Sell	Sell	Sell	Sell	Sell	Seil	-14.1
UBS AG, London Branch	UBS AG, London Branch	UBS AG. London Branch	UBS AG, London Branch	IIBS AG London Branch	I IBC AG London Branch	UBS AG London Branch	URS AG Tondon Branch	URS AG London Branch	IRS AG Toodon Branch	IRS AG London Branch	DEC AC. London Broads	Des Act, London Blanch	UBS Au, London Branch	UBS AG, London Branch	UBS AG, London Branch	UBS AG, London Branch	UBS AG, London Branch	ĕ	ৰ।	UBS AG, London Branch	UBS AG, London Branch	ä١	¥١	ă١	₹Į:	UBS AG, LONGON Branch	UBS Act, London Branch	Ups Act, Landon Branch	USS ACT LONGON BRANCH	US AC LONGS BIRITI	UBS AG, LONGON BRANCH	UBS AG, LONDON Blanch	URS AG London Branch	URS AG London Branch	UBS AG, London Branch	UBS AG, London Branch	UBS AG, London Branch	UBS AG, London Branch	UBS AG, London Branch	UBS AG, London Branch	UBS AG, London Branch	UBS AG, London Branch	UBS AG, London Branch	UBS AG, London Branch	USC AG London Branch	UBS AG. London Branch	UBS AG Australia Branch	UBS AG Australia Branch	UBS AG Australia Branch	UBS Wealth Management Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	obs securities Australia Ltd	Obs Securities Australia Ltd	UBS Securities Australia Ind	UBS Securities Australia Itd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	
02-Dec-14	02-Dec-14						02-Dec-14	02-Dec-14	02-Dec-14	02-Dec-14	02 Oct 14	2000	02-Dec-14	02-Dec-14	02-Dec-14	02-Dec-14	02-Dec-14	02-Dec-14	02-Dec-14	02-Dec-14	02-Dec-14	02-Dec-14	02-Dec-14	1	T		1	l	02-Dec-14	Ì		02-Dec-14	02-Dec-14	02-Dec-14	02-Dec-14			02-Dec-14	02-Dec-14	02-Dec-14	02-Dec-14	02-Dec-14	02-Dec-14	02-Dec-14	١	1	02-Dec-14					1	1	1	T	02-Dec-14	T	Ī	T	Ī	Ī		Γ	Ī	Ī	Ī	02-Dec-14		Γ	1

	Control Control	201	(11,1)	00.0	Ordinary
ec-14	AG, L	Sell	61	Ď	Sign of Sign o
-Dec-14	UBS AG, London Branch	Seil	1,169	(153	) Ordinary
02-Dec-14	UBS AG, London Branch	Sell	703	(92	Ordinary
ec-14	UBS AG, London Branch	le S	1,147		
ec-14	UBS AG, London Branch	lac sell	115		
ec-14	UBS AG, London Branch	Jac.	3,789		
100	UBS AG, LONDON BRANCH	(Sell	767.7	(30)	
bec.14	LIRS AG London Branch	والع	5,705		
PC-14	URS AG Tondon Branch	Sell	082		
02-Dec-14	UBS AG, London Branch	Sell	268		
02-Dec-14	UBS AG, London Branch	Sell	842		
ec-14	UBS AG, London Branch	Sell	145		L
ec-14	UBS AG, London Branch	Sell	245		
02-Dec-14	UBS AG, London Branch	Sell	8		
ec-14	UBS AG, London Branch	Sell	2,203		
02-Dec-14	UBS AG, London Branch	Sell	146		
02-Dec-14	UBS AG, London Branch	Self	3,733		
-Dec-14	UBS AG, London Branch	Sell	115		
02-Dec-14	UBS AG, London Branch	Sell	406		
-Dec-14	UBS AG, London Branch	Sell	909	(62)	
02-Dec-14	UBS AG, London Branch	Sell	35		
02-Dec-14	UBS AG, London Branch	Sell	69		L
-Dec-14	UBS AG, London Branch	Sell	214		L
32-Dec-14	UBS AG, London Branch	Sell	4,106		
-Dec-14	UBS AG, London Branch	Sell	1,102		
32-Dec-14	UBS AG, London Branch	Sell	3,182		
02-Dec-14	UBS AG, London Branch	Sell	9,619		Ordinary
02-Dec-14	UBS AG, London Branch	Sell	817		
02-Dec-14	UBS AG, tondon Branch	Sell	4,278		
ec-14	UBS AG, London Branch	Sell	2,043	(267)	Ordinary
ec-14	UBS AG, London Branch	Sell	764		Ordinary
ec-14	UBS AG, London Branch	Sell	542	(71)	
ec-14	UBS AG, London Branch	Sell	222	(53)	
ec-14	UBS AG, London Branch	Sell	15		
02-Dec-14	UBS AG, London Branch	Sell	959'9		Ordinary
PC-14	UBS AG, London Branch	Sell	10,742		
02-Dec-14	UBS AG, London Branch	Sell	689		
02-Dec-14	UBS AG, London Branch	Sell	265		
sc-14	UBS AG, London Branch	Sell	872		
sc-14	UBS AG, London Branch	Sell	2,540		l
ic-14	UBS AG. London Branch	Sell	2 823		
14	UBS AG. London Branch	Sell	122		
x-14	UBS AG. London Branch	Sell	521		Ordinary
sc-14	UBS AG, London Branch	Seil	5 508		l
6-14	UBS AG. London Branch	Sell	777		l
21.7	RS AG London Branch	Cell	102		
02-Dec-14	I IRS AG London Branch	100	261	(07)	Ordinary
* :	OBS AG, LONGON BIRTH	100	1,143		ı
02-Dec-14	UBS AG, London Branch	Dell C-1	810,7		1
02-Dec-14	UBS AG, London Branch	net.	5,016		1
3C-14	UBS AG, London Branch	Sell	346		
02-Dec-14	UBS AG, London Branch	Sell	974		
c-14	UBS AG, London Branch	Sell	2,516		
32-Dec-14	UBS AG, London Branch	Sell	6,297		Ordinary
32-Dec-14	UBS AG, London Branch	Sell	453		İ
02-Dec-14	UBS AG, London Branch	Sell	828		
02-Dec-14	UBS AG, London Branch	Sell	1 220		
02-Dec-14	UBS AG. London Branch	Sell	25		Ordinary
02-Dec-14	LIBS AG London Branch	ay.	637		
02-Dec-14	list of tonden Branch	Coll	1 1 1 1 1	(10)	
Day 14	100 AC LONGOI BIRDING	100	517,1	(523)	
Doc 14	LIBS ACT LONGON DISHON	100	560'1	(/47)	
41.5	UBS AG, London Branch	seil	1,423	(185)	
02-Dec-14	UBS AG, London Branch	Sell	23	(3)	
2-Dec-14	UBS AG, London Branch	Sell	2,686	(347)	Ordinary
32-Dec-14	UBS AG, London Branch	Sell	8	(1)	
-Dec-14	UBS AG, London Branch	Sell	11,578	(264,1)	
-Dec-14	UBS AG, London Branch	Sell	565'9	(851)	
2-Dec-14	UBS AG, London Branch	Sell	098		
-Dec-14	UBS AG. London Branch	Sell	869		
-Dec-14	UBS AG, London Branch	Sell	3.069	(968)	L
Dec-14	LIRS AG London Branch	lley	7151		
1	TO SOLD THE PARTY OF SOLD THE		1,2,1	(/כו)	
4	TOTAL TODGE STREET		000 +	000	
	UBS AG, LONGON BIANCH	Self	1,829	(236)	Ordinary

02-Dec-14	UBS Securities Australia Itd		3		5
-		Sell	16,592	(2,166)	Ordinary
	UBS Securities Australia Ltd	Sell	790	(34)	Ordinary
₽ :	UBS Securities Australia Ltd	Sell	4,491	(587)	Ordinar
4	UBS Securities Australia Ltd	Sell	1,308	(171)	Ordinar
4	UBS Securities Australia Ltd	Sell	1,071	(140)	Ordinar
4 5	UBS Securities Australia Ltd	Sell	11,773	(1,539)	Ordinan
<u>,</u>	UBS Securities Australia Ltd	Self	4,644	(607)	Ordinar
<u>.                                     </u>	1105 Securities Australia Ltd	New Comment	6,311	(825)	Ordinar
4	UBS Securities Australia Itd	100	120,127	(2,631)	Ordinan
4	UBS Securities Australia I1d	3	11,040	(1,623)	Orginary
4	UBS Securities Australia Ltd	Sell	825	(50)	Colora
4	UBS Securities Australia Ltd	Sell	15.185	(1 985)	Ordinary
14	UBS Securities Australia Ltd	Sell	1,767	(231)	Ordinan
02-Dec-14	UBS Securities Australia Ltd	}e⊪	25,911	(3,387)	Ordinary
4	UBS Securities Australia Ltd	Sett	28	(11)	Ordinary
4	UBS Securities Australia Ltd	Sell	5,316	(694)	Ordinary
4	UBS Securities Australia Ltd	Sell	1,157	(151)	Ordinary
<u>.</u>	UBS Securities Australia Ltd	Sell	8,380	(1,094)	Ordinary
22-Dec-14	UBS Securities Australia Ltd	Sell	2,436	(318)	Ordinan
<u> </u>	UBS Securities Australia Ltd	Sell	2,788	(364)	Ordinary
4	LIRS Securities Australia Ind	Sell	8/3	(114)	Ordinan
4	UBS Securities Australia 11d	Sell	130	(24)	Ordinan
14	UBS Securities Australia I td		000	(18)	Ordinary
2	UBS Securities Australia Ltd	Ser	1 475	(201)	O C
02-Dec-14	UBS Securities Australia Ltd	Sell	444	(58)	Ordinar
2-Dec-14	UBS Securities Australia Ltd	Sell	828	(112)	Ordinary
- 1	UBS Securities Australia Ltd	lles.	2,298	(300)	Ordinary
	UBS Securities Australia Ltd	Sell	3,784	(494)	Ordinary
	UBS Securities Australia Ltd	Sell	8,426	(1,100)	Ordinary
-	UBS Securities Australia Ltd	Sell	5,554	(725)	Ordinary
	UBS Securities Australia Ltd	Sell	214	(28)	Ordinary
1	UBS Securities Australia Ltd	Jet I	23	<u>e</u>	Ordinary
	LIRS Securities Australia 14d	Jell Sell	400,51	(1,700)	Ordinary
ı	UBS Securities Australia Itd	Spill	7 550	(1,470)	Ordinary
02-Dec-14	UBS Securities Australia Ltd	Sell	000//	(1,000)	Ordinary
ı	UBS Securities Australia Ltd	Sell	7 650	(1 000)	Coling
	UBS Securities Australia Ltd	Sell	3 198	(817)	Ordinary
П	UBS Securities Australia Ltd	Sell	10.733	(1 403)	Ordinary
	UBS Securities Australia Ltd	Sell	168	(22)	Ordinary
	UBS Securities Australia Ltd	Self	222	(52)	Ordinary
14	UBS Securities Australia Ltd	Sell	145	(61)	Ordinary
4	UBS Securities Australia Ltd	Sell	191	(22)	Ordinary
14	UBS Securities Australia Ltd	Sell	834	(601)	Ordinary
02-Dec-14	UBS Securities Australia Ltd	Sell	8	(1)	Ordinary
14	UBS Securities Australia Ltd	Sell	155	(20)	Ordinary
14	UBS Securities Australia Ltd	Sell	5,740	(743)	Ordinary
14	UBS Securities Australia Ltd	Sett	4,238	(549)	Ordinary
02-Dec-14	UBS Securities Australia Ltd	Sell	4,748	(615)	Ordinary
2-Dec-14	UBS Securities Australia Ltd	Sell	83,492	(10,808)	Ordinary
2-Dec-14		Sell	4,956	(642)	Ordinary
02-Dec-14	UBS Securities Australia Ltd	Sell	6,172	(662)	Ordinary
02-Dec-14	UBS Securities Australia Ltd	Sell	108	(14)	Ordinary
4	UBS Securities Australia Ltd	Sell	162	(21)	Ordinary
4	UBS Securities Australia Ltd	Sell	929	(82)	Ordinary
02-Dec-14	UBS Securities Australia Ltd	Seil	088	(114)	Ordinary
4	UBS Securities Australia Ltd	Sell	1,197	(155)	Ordinary
02-Dec-14	UBS Securities Australia Ltd	Sell	11,812	(1,530)	Ordinary
4	UBS Securities Australia Ltd	Sell	53	(3)	Ordinary
02-Dec-14	UBS Securities Australia Ltd	Sell	394	(51)	Ordinary
4	UBS Securities Australia Ltd	Self	5,902	(765)	Ordinary
4	UBS Securities Australia Ltd	Sell	8	(1)	Ordinary
4	UBS Securities Australia Ltd	Sell	33,823	(4,384)	Ordinary
4	UBS Securities Australia Ltd	Sell	2,621	(340)	Ordinary
4	UBS Securities Australia Ltd	Sell	54,105	(5,013)	Ordinary
4	UBS Securities Australia Ltd	Sell	47,355	(6,138)	Ordinary
02-Dec-14	UBS Securities Australia Ltd	Sell	34,270	(4,442)	Ordinary
4	UBS Securities Australia Ltd	Sell	2.375	(308)	Ordinary
4					
<b>1</b>	INDO COCINTIAC ANTERIAL TO	Leg-	0 440	(0.00)	

l
l
İ
١

20. Dec. 14         UBS Securities Australia Ltd           20. Dec. 14         UBS Securities Austr	Section   Sect	15.82 15.30 67.537 2.310 2.125 3.819 18.134 3.823 4.543 4.543 15.56 1.566 1.566 2.256 2.256 7.310	(592) (2000) (2000) (300) (200) (200) (235) (235) (235) (236) (236) (237) (237) (237) (237) (237) (238	Ordinary Ordinary Ordinary Ordinary Ordinary
		15,400 67,337 2,135 2,125 3,819 18,134 18,134 30,633 4,543 1,686 1,586 1,586 1,586 2,356 2,356 2,356 2,356 7,931 7,931	(2,000) (8,771) (900) (276) (276) (2,355) (82) (82) (82) (82) (82) (82) (82) (82	Ordinary Ordinary Ordinary Ordinary
		67.337 2,125 2,125 3,819 18,334 300 300 30 4,543 4,543 16,543 1,586 1,586 2,356 2,356 7,310	(857) (860) (862) (862) (823) (823) (823) (823) (823) (823) (823) (823) (823) (823)	Ordinary Ordinary Ordinary
		2,25 2,25 3,819 18,134 300 216 2,533 4,543 1,546 1,586 1,586 1,586 2,256 2,256 2,256 7,93	(300) (276) (486) (2.355) (2.355) (4.236) (4.238) (4.238) (500)	Ordinary Ordinary
		2,125 3,191 18,134 300 30,03 32,633 4,543 16,543 1,586 1,586 1,586 2,356 2,356 2,356 2,356 7,93	(276) (495) (2,235) (4,238) (4,238) (4,238) (2,25) (2,25)	Ordinary
		3.819 18.181 300 2.633 4.543 16.543 1.586 2.356 2.356 7.93	(496) (2,355) (39) (4,238) (4,238) (290) (206)	Ordinary
		300 216 32.653 4,543 15.86 1,586 1,586 2,376 2,376 2,370 793	(4,238) (4,238) (590) (2,129) (206)	
		216 2163 32,633 4,543 16,363 1,586 2,310 2,310 793	(4,238) (4,238) (590) (2,125) (206)	Ordinary
		32,513 4,543 16,363 1,586 2,356 2,356 2,350 793	(4,238) (4,238) (590) (2,125) (206)	Ordinary
		4,543 16,363 1,586 2,356 2,310 2,310	(590)	Ordinary
		16,363 1,586 2,356 2,310 793	(2,125)	Ordinary
		1,586 2,356 2,310 793	(206)	Ordinary
		2,356 2,310 793		Ordinary
		2,310	(908)	Ordinary
		793	(300)	Ordinary
			(103)	Ordinary
		3,819	(496)	Ordinary
		631	(82)	Ordinary
		462	(09)	Ordinary
		23,123	(3,003)	Ordinary
		2,426	(315)	Ordinary
		102,1	(156)	Ordinary
		8//8	(1,140)	Ordinary
		17,833	(2,316)	Ordinary
		48 495	(8,700)	Ordinary
		1 563	(203)	Ordinary
		2,310	(300)	Ordinary
		39,216	(5.093)	Ordinary
		13,529	(1,757)	Ordinary
		932	(121)	Ordinary
		12,513	(1,625)	Ordinary
		3,819	(496)	Ordinary
		118,519	(15,405)	Ordinary
		14,094	(1,828)	Ordinary
		2,282	(362)	Ordinary
		22,576	(2,932)	Ordinary
		21,167	(2,749)	Ordinary
		4,012	(521)	Ordinary
		1,371	(178)	Ordinary
		762	(66)	Ordinary
		12,305	(1,598)	Ordinary
		3,819	(496)	Ordinary
		50,019	(6,496)	Ordinary
		6,984	(206)	Ordinary
1		1,301	(169)	Ordinary
	er l	47,725	(6,198)	Ordinary
		97,459	(12,657)	Ordinary
1111		3,611	(469)	Ordinary
111		30,977	(4,023)	Ordinary
11	S Se	8,647	(1,123)	Ordinary
1		51,236	(6,654)	Ordinary
		4	(9)	Ordinary
1		982	(11)	Ordinary
UZ-Dec-14 UBS Securities Australia Ltd		11,211	(1,456)	Ordinary
1		13,090	(1,700)	Ordinary
	Sell	109,579	(14,231)	Ordinary
١		70,547	(9,162)	Ordinary
-		154,000	(20,000)	Ordinary
1		44,421	(2,769)	Ordinary
		826'22	(3,627)	Ordinary
		282,913	(36,742)	Ordinary
	Sell	 187,903	(24,403)	Ordinary
14 UBS Securities Australia Ltd		956'6	(1,293)	Ordinary
	₩.	94,518	(12,275)	Ordinary
02-Dec-14 UBS Securities Australia Ltd		60.915	(7.911)	Ordinary
02-Dec-14 UBS Securities Australia Ltd		73 593	(3 OSA)	Ordinan
		43.474	(5.646)	Ordinary
		385 000	(000005)	Ordinary
I BY	<u>.</u>	107 202	(45 054)	Ordinary.
I		762'/01	(13,334)	Crainary
02-Dec-14 UBS Securities Australia Ltd		134,673	(17,490)	Ordinary

25 S S S S S S S S S S S S S S S S S S S	3 3 3	20,339 (2,6 3,824 (6	(2,638) Ordinary (496) Ordinary (702) Ordinary
2	202		
2	m u		
2	un.		
2 S S S S S S S S S S S S S S S S S S S			
19 S S S S S S S S S S S S S S S S S S S			
2 S S S S S S S S S S S S S S S S S S S			
10 10 10 10 10 10 10 10 10 10 10 10 10 1			
2			ı
Sell Sell Sell Sell Sell	17		
Sell Sell Sell Sell Sell Sell Sell Sell			l
Sell Sell Sell	2		
Sell Sell Sell	2		
Sell			-
261	40		
1170	5		
170	/1		
100			
Coll			١
SAII	2		ŀ
liac lines	87		
III S			
Sell Sell	6	3	
Sell	5	5,975	(776) Ordinary
llac			
Sell Sell	22	22,954 (2,9	
Sell	56		
Sell	4		
Sell	•		(131) Ordinary
Sell			(15) Ordinary
Sell	11,	11,188 (1,4	(1,453) Ordinary
Sell	12.		(1,607) Ordinary
Self	1		(219) Ordinary
Sell			
Seli	2		(286) Ordinary
Sell			(3) Ordinary
Sell			
Sell			
Sell	17.		
Sell	14	14,177 (1,8	(1,840) Ordinary
Sell	14		
Sell	20		
Sell	25.		
Seil	40,		
Sell			(88) Ordinary
Sell	6		
Sell			
Sell	m		
Sell	2		
Sell			
Cp.			
Co			trong ordinary
(Ja)			
Sell	) o		OEA) Ordinary
130	'C		
nec.			(246) Ordinary
Sell	1,		
Seil	2,		(300) Ordinary
Sell	24,		166) Ordinary
Sell	-		(180) Ordinary
Nell Sell			(206) Ordinary
S. C.	02		L
	(1)		
	ie.		
) aci	4		
Sell	36,	9)	
Sel	1,		(223) Ordinary
Sell	88	ε)	
Sell	, T		(251) Ordinary
Sell			(36) Ordinary
Sell			
lles	ox.		(1110) Ordinary
100	7	1,17	1

03-Dec-14	COS SECULIES MOSIBILITY		000	(88)	Ordinary
	UBS Securities Australia Ltd	les.	2.750	(655)	Ordinan
1	UBS Securities Australia Ltd	Sell	2.750	(656)	Ordinary
ı	UBS Securities Australia Ltd	Sell	726	(119)	Ordinary
1	I IBS Securities Australia 11d		1 473	(6   1)	Ordinary
ı	198 Securities Australia 144	9	7/4/	(691)	Ordinary
ı	LIBS Securities Australia 11d	8	1 321	(5)	Oldinaly
	IUBS Securities Australia Itd	105	16,007	(0/1)	Clumary
	LIBS Securities Australia Ltd	195	000	(600,4)	Clumary
	I IRC Committee Australia 14d	Coll	200 FFE C	(60)	Cidinary
03-Dec-14	I IBS Securities Australia Ind	Sea	121,6	(479)	Ordinary
l	LIDE Cocumition Australia 144	100	200	(60)	Ordinary
ı	I Do Countries Australia ted	1173	802	(103)	Ordinary
1	Upo Securities Australia Liu	) Sel	42	(83)	Ordinary
	Open Seconder Australia Ltd	251	/67	(33)	Ordinary
١	Opposed in the second s	nac nac	9	(2)	Ordinary
ı	Obsiderations Australia Ltd	) ell	15,9/9	(2,046)	Ordinary
1	Obs securities Australia Ltd	) dell	15,938	(2,046)	Ordinary
Į	UBS Securities Australia Ltd	Sell	15,796	(2,033)	Ordinary
-	UBS Securities Australia Ltd	Sell	544	(20)	Ordinary
03-Dec-14	UBS Securities Australia Ltd	Sell	132	(71)	Ordinary
-1	UBS Securities Australia Ltd	Sell	31	(4)	Ordinary
1	UBS Securities Australia Ltd	Sell	210	(27)	Ordinary
-	UBS Securities Australia Ltd	Sell	1,040	(134)	Ordinary
١	UBS Securities Australia Ltd	Sell	124	(16)	Ordinary
	UBS Securities Australia Ltd	Sell	1,110	(143)	Ordinary
	UBS Securities Australia Ltd	Sell	784	(101)	Ordinary
1	UBS Securities Australia Ltd	Seil	861		Ordinary
	UBS Securities Australia Ltd	Sell	402		Ordinary
	UBS Securities Australia Ltd	Sell	896		Ordinary
03-Dec-14	UBS Securities Australia Ltd	Sell	23		Ordinary
3-Dec-14	UBS Securities Australia Ltd	Sell	1.277		Ordinary
03-Dec-14	UBS Securities Australia Ltd	Test Se	333		Ordinan
03-Dec-14	LIBS Securities Australia Ltd	100	3 970		O'dellay.
33-Dec-14	LIBS Securities Australia Itd	9	900.1		Cidicialy
3-Dec-14	I IBS Socienties Australia Ind	===	CEE 11		Clouding
2000	Dis Decorres Australia Etc.	1100	5//11		Ordinary
	Obsiderations Australia Ltd		69	(6)	Ordinary
İ	USS Securities Australia Ltd	Sell	15	(2)	Ordinary
03-Dec-14	USS Securities Australia Ltd	) Aei	9,264	(1,200)	Ordinary
1	UBS Securities Australia Ltd	Sea	1,112	(144)	Ordinary
1	UBS Securities Australia Ltd	Seil	10,468	(1,356)	Ordinary
3-Dec-14	UBS Securities Australia Ltd	Sell	8	9	Ordinary
3-Dec-14	UBS Securities Australia Ltd	Sell	5.211	(675)	Ordinary
	UBS Securities Australia Ltd	Sell	16 106	(2.089)	Ordinary
l	UBS Securities Australia Ltd	Sell	1812	(350)	Ordinan
l	UBS Securities Australia Itol	9	3001	(200)	Collinary
03-Dec-14	LIBS Constitute Australia 144	Coll	006,1	(007)	Ordinary
	Consider Australia Library	llac llac	955	(44)	Ordinary
03-Dec-14	UBS Securities Australia Ltd	) Sell	85	(11)	Ordinary
1	UBS Securities Australia Ltd	Seli	672'6	(1,197)	Ordinary
1	UBS Securities Australia Ltd	Sell	1,511	(961)	Ordinary
	UBS Securities Australia Ltd	Sell	378	(49)	Ordinary
	UBS Securities Australia Ltd	Self	67,223	(8,753)	Ordinary
	UBS Securities Australia Ltd	Sell	1,152	(150)	Ordinary
	UBS Securities Australia Ltd	Sell	1,152	(150)	Ordinary
ı	UBS Securities Australia Ltd	Sell	4,992	(650)	Ordinary
	UBS Securities Australia Ltd	Sell	63,022	(8.206)	Ordinary
	UBS Securities Australia Ltd	Sell	54,459	(160.7)	Ordinary
	UBS Securities Australia Ltd	Sell	180.698	(73 667)	Ordinary
	UBS Securities Australia Ltd	Seli	99 769	(3 800)	Ordinary
1	UBS Securities Australia Itd	و	73 502	woo c)	Cardina
	TIDE Countries Australia Lau	Hos	760'67	(060'c)	Ordinary
1	Obs securities Australia Ltd	nac.	181	(25)	Ordinary
1	UBS Securities Australia Ltd	Sell	5,700	(747)	Ordinary
03-Dec-14	UBS Securities Australia Ltd	Sell	23	(E)	Ordinary
-	UBS Securities Australia Ltd	Sell	1,236	(162)	Ordinary
3-Dec-14	UBS Securities Australia Ltd	Sell	954	(125)	Ordinary
3-Dec-14	UBS Securities Australia Ltd	Sell	961	(126)	Ordinary
3-Dec-14	UBS Securities Australia Ltd	Sell	11.872	(1.556)	Ordinary
3-Dec-14	UBS Securities Australia Ltd	Sell	2 030	090	Ordinary
03-Der-14	I IRS Securities Australia 11d	Sell	oeo c	(004)	The state of
Dec.14	IRC Counties Australia 14d	Sell	200,2	(003)	Cidinary
03-Dec-14	UBS Securities Australia Ltd	190	262 1	(00%)	Orginary
Pres-14	UBS Securities Australia Ind	1103	1 U.A.C.	(700)	Orginary
P-Dec- 14	UBS Securities Australia Liu	ישנו	1103		
03-Dec-14			6.1.5	(1,234)	Ordinary
	UBS Securities Australia Ltd	lles	7,249	(1,234)	Ordinary

			00000	Ordinary
UBS Securities Australia Ltd	Buy	156,188	20,000	
intes Australia Ind	Sall	206 570	20,058	Ordinary
irties Australia Itch	Seell	12 072	(40,030)	Ordinary
urities Australia Ltd	Buv	10.100	1 313	Colorary
urities Australia I tol	(cell	200 000	WOT OC	Ordinary
inities Australia I to	رقاا	2 536	(OCC)	Ordinary)
urities Australia Ltd	lias	141 747	(18 435)	Ordinary
mities Australia I td	Alig	24.452	2 170	Ordinary
Application Australia 140	110	1000	2,7,5	Ordina.
wither Australia 14d	Coll	200 300	000	Cidillaly
article Australia 14d	la o	200 030	(171,121)	Commany
unities Australia 14d	2.0	757,007	(32,340)	Countries
Urities Australia Lto	, and	4/5,1/2	61,770	Ordinary
, London Branch	lauy ma	7597/1	18,435	Ordinary
, London Branch	Sell	141,747	(18,435)	Ordinary
5, London Branch	Sell	29,771	(3,178)	Ordinary
i, London Branch	Buy	24,452	3,178	Ordinary
5 Australia Branch	Sell	18,202	(1,371)	Ordinary
3 Australia Branch	Sell	14.417	(1.873)	Ordinary
Australia Branch	Sell	19312	0.5140	Orchpary
ealth Management Australia Itch	Sel	1 751	(000)	Ordinary
oalth Management Augralia ind	Coll	1,634	(0.40)	Countries
cattli Maliagellielit Adstralia Cto		1,0,1	(0) 7)	Ordinary
curries Australia Ltd	Sell	194	(25)	Ordinary
Securities Australia Ltd	Sell	622	(80)	Ordinary
UBS Securities Australia Ltd	Sell	575	(74)	Ordinary
UBS Securities Australia Ltd	Sell	5.518	(712)	Ordinary
Securities Australia Itd	lay	2,020	(300)	Ordinary.
Countries Australia 14d	la S	2,020	(020)	Crostary
Contribe Australia Ltd	1127	2,930	(3/8)	Ordinary
Securities Australia Ltu	100	2,505	(331)	Ordinary
UBS Securities Australia Ltd	liac	1,265	(163)	Ordinary
UBS Securities Australia Ltd	Sell	5,425	(002)	Ordinary
UBS Securities Australia Ltd	Sell	1,318	(071)	Ordinary
UBS Securities Australia Ltd	Sell	23,405	(3,020)	Ordinary
curities Australia Ltd	Sell	3.515	(453)	Ordinary
curities Australia Ltd	Sell	1.086	(140)	Ordinary
curities Australia Ltd	Seli	70	(6)	Ordinary
curities Australia Ltd	Self	20	(6)	Ordinary
curities Australia Ltd	Seli	1.593	(202)	Ordinary
UBS Securities Australia Ltd	Sell	7.718	(626)	Ordinary
curities Australia Ltd	Sell	13 621	(1, 753)	Ordinary
Curties Australia 14d	le y	1 501	(300)	Ordinani
Contract Acceptance and	130	100,1	(207)	Ordinary
Day securities Australia Liu	120	6/1//	(924)	Oromary
curines Australia Ltd	oeil	1,756	(977)	Ordinary
curities Australia Ltd	Seil	1,166	(150)	Ordinary
curities Australia Ltd	Sell	5,851	(753)	Ordinary
UBS Securities Australia Ltd	Sell	10,287	(1,324)	Ordinary
curities Australia Ltd	Sell	1,166	(150)	Ordinary
curities Australia Ltd	Sell	1 027	(041)	Ordinary
trities Δustralia I tol	(Sp.)	1 052	(361)	Ordinary
Contract Australia 144	100	5.00	(102)	Oldinaly
courses Australia Ltu	100	4,400	(4/0)	Ordinary
Cutties Australia Ltd	100	29,716	(2,822)	Ordinary
UBS Securities Australia Ltd	Sell	1,642	(211)	Ordinary
curties Australia Ltd	Sell	8,317	(1,069)	Ordinary
curities Australia Ltd	Sell	11,888	(1,528)	Ordinary
curities Australia Ltd	Sell	5,236	(673)	Ordinary
curities Australia Ltd	Sell	179	(23)	Ordinary
curities Australia I tel	Sell	2 5 7 5	(122/)	Ordinary
IDS Consisting Augustin Lea	100	2002	(101)	Cromings
Culties Australia Liu		09/	(101)	Ordinary
UBS Securities Australia Ltd	Sell	1,120	(144)	Ordinary
JBS Securities Australia Ltd	Sell	490	(63)	Ordinary
JBS Securities Australia Ltd	Sell	514	(99)	Ordinary
UBS Securities Australia 11d	Sell	888	(117)	Ordioan
IDC Committee Australia Leal	03	8		Cielling.
IDC Countries Australia 144	livo.	60	0 6	Ordinary
nities Australia tid	100	80	(UZ)	Ordinary
UBS Securities Australia Ltd	Sell	16	(Z)	Ordinary
UBS Securities Australia Ltd	Sell	148	(61)	Ordinary
UBS Securities Australia Ltd	Sell	1,379	(177)	Ordinary
UBS Securities Australia Ltd	Sell	343	(77)	Ordinary
DO Committee Australia 144	100	9	(4.1)	Cicinary
Unities Australia Ltd	Jen C	x	(3)	Ordinary
UBS Securities Australia Ltd	Sell	31	(4)	Ordinary
UBS Securities Australia Ltd	Seli	47	(9)	Ordinary
urities Australia Itd	Sell			
מונופה שפסיםוום ליני		140	101/	Configuracy,
		148	(19)	Ordinary

UBS Securities Australia Ltd					Ordinary
UBS Securities Australia Ltd	s.		9,874	(1,289)	Ordinary
UBS Securities Australia Ltd	Sel		1,150	(150)	Ordinary
UBS Securities Australia Ltd	Se.		26	(12)	Ordinary
UBS Securities Australia Ltd	Sel	_	20,085	(2,622)	Ordinary
UBS Securities Australia Ltd	Sel	_	4,490	(282)	Ordinary
UBS Securities Australia Ltd	Sel		10,906	(1.421)	Ordinary
UBS Securities Australia Ltd	35		2.934	(382)	Ordinary
UBS Securities Australia Ltd	3.		1.183	(154)	Ordinary
UBS Securities Australia Ltd	Sell		3 3 3 5 5	(224)	Ordinary
UBS Securities Australia Ind			11 420	(00)	y Clumary
I IBS Securities Australia 110			101.00	(004,1)	Ordinary
I IPC Cocumition Australia 140			20,181	(/90'5)	Crdinary
Libs Securities Australia Ltd			24,983	(3,253)	Ordinary
on a second sale and the	200		1,736	(1,006)	Ordinary
UBS Securities Australia Ltd	×.		77	(10)	Ordinary
UBS Securities Australia Ltd	Se		191	(21)	Ordinary
UBS Securities Australia Ltd	8		9.520	(1.738)	Ordinary
UBS Securities Australia Ltd	35		7 330	(200)	
URS Securities Australia 11d	9		100.0	(000)	Cicinaly
I IBS Securities Australia 1+d	199		185,5	(1410)	Ordinary
TIPE Cocumition Australia Lea			0,20,5	(4/1)	Ordinary
OBS Securities Australia LIU			15,495	(2,015)	Ordinary
UBS Securities Australia Ltd	EX.		1,976	(257)	Ordinary
UBS Securities Australia Ltd	Sell		9,359	(1,217)	Ordinary
UBS Securities Australia Ltd	Sell	_	15,265	(1 985)	Ordinary
UBS Securities Australia 11d	leS.		45 000	(020.2)	
IRS Securities Australia 14d	0		506,64	(0/6/6)	Ordinary
THE COMMENT OF THE PARTY OF THE			78,547	(2,306)	Ordinary
Obsisecurities Australia LIG	le c		45,294	(2,890)	Ordinary
UBS Securities Australia Ltd	Sell		28,845	(3,751)	Ordinary
UBS Securities Australia Ltd	Sell		2,807	(392)	Ordinary
UBS Securities Australia Ltd			2,245	(262)	Ordinary
UBS Securities Australia Ltd	Sell		1776	(124)	Ordinan
UBS Securities Australia Ltd	Se		1569	(800)	i i
UBS Securities Australia 110			545 55	(404)	Countries
LIDS Committee Australia Libe			57,243	(4,843)	Ordinary
OBS Securities Australia LLO			87/'5	(742)	Ordinary
Obs securities Australia Ltd		,	34,871	4,517	Ordinary
UBS Securities Australia Ltd	Buy	,	31	4	Ordinary
UBS Securities Australia Ltd		_	4,605	(865)	Ordinary
UBS Securities Australia Ltd			1 348	(175)	Ordinan
UBS Securities Australia Ltd			2 605	10000	in in
LIBC Cocurities Australia 14d			550,3	(OCC)	Crarriery
Did Bill Bill Services Coo	200		cl	(2)	Ordinary
Upo securities Australia Ltd			42,743	(5,551)	Ordinary
UBS Securities Australia (td			77	(10)	Ordinary
UBS Securities Australia Ltd			69	(6)	Ordinary
UBS Securities Australia Ltd			393	(51)	Ordinary
UBS Securities Australia Ltd	lles.		25	6	Ordinary
UBS Securities Australia Ltd	Sell		6.627	(098)	Ordinati
IBS Securities Australia 11d	Sell		2,000	(000)	Ciuliary
The Commission Access to the			4,620	(nng)	Ordinary
COS Secondes Australia Liu	100		8,832	(1,147)	Ordinary
UBS Securities Australia Ltd	э <u>х</u>		770	(100)	Ordinary
UBS Securities Australia Ltd	3		54,601	(160'2)	Ordinary
UBS Securities Australia Ltd	Sell		924	(120)	Ordinary
UBS Securities Australia Ltd	llas		3.465	(450)	Ordinary
UBS Securities Australia Ltd	Sell		3 511	(950)	Ordinan
1985 Secritides Australia	3		0.440	(000)	O Girlary
185 Securities Australia 11d	3		2,75	(010)	Ciginary
Did officers & continuous 2011	4		0	(CC)	Ordinary
Opp Securities Australia Ltd	B.		49,942	(6,486)	Ordinary
Obsiderations Australia Ltd	- N		13,298	(1,727)	Ordinary
UBS Securities Australia Ltd	Sell		5,198	(675)	Ordinary
UBS Securities Australia Ltd	Sell		12,559	(1,631)	Ordinary
UBS Securities Australia Ltd	Sell		27.012	(3.508)	Ordinary
UBS Securities Australia Ltd	Sell		23.916	(3.106)	Ordinary
LIBS Securities Australia Ltd	ay.		1 032	(00.10)	, in 19
I IBS Sociutities Australia 11d	ley		2007	600	1
IBS Societies Australia Itd	Col		7,007	(349)	Orginary
IDC Consistent American Land	100		0,000	(50/)	Ordinary
Obs securities Australia Ltd	IIAC		693	(06)	Ordinary
UBS Securities Australia Ltd	Sell		1,240	(191)	Ordinary
UBS Securities Australia Ltd	Sell		2,225	(583)	Ordinary
UBS Securities Australia Ltd	Sell		19,989	(2.596)	Ordinary
JBS Securities Australia Ltd	3		185	(VC)	Ordinan
IUBS Securities Australia 11d	Sp		2000	(57)	Countries
TIDE Contraction Assessment and			250,6	(1, 73)	Ordinary
UBS Securities Australia Ltd	Seri				
IDC Camping Acceptant			#ZO'I	(133)	Ordinary
COS SECULIES ACSUBIS LIC	Sell		1,024	(133)	Ordinary

JBS Securities Australia Ltd	Sell	6,746	(875)	Ordinary
UBS Securities Australia Ltd	Sell	5.474	(7,0/5)	Ordinary
es Australia Ltd	Sell	46	(9)	Ordinary
JBS Securities Australia Ltd	Sell	1,681	(218)	Ordinary
UBS Securities Australia Ltd	lles.	39	(5)	Ordinary
ies Australia Ltd	Sell	663	(98)	Ordinary
ies Australia Ltd	Sell	362	(47)	Ordinary
les Australia Ltd	Sell Sell	817	(106)	Ordinary
ies Australia Itd	130	1,696	(220)	Ordinary
UBS Securities Australia Ltd	Setl	12,482	(1,619)	Ordinary
es Australia Ltd	Sell	10,301	(1,336)	Ordinary
es Australia Ltd	Sell	3,431	(445)	Ordinary
es Australia Ltd	Sell Sell	1 737	(2,0/4)	Ordinary
es Australia Ltd	Sell	6,944	(006)	Ordinary
es Australia Ltd	Sell	2,455	(318)	Ordinary
es Australia Ltd	Sell	4,871	(631)	Ordinary
es Australia Ltd	Net!	2,100	(271)	Ordinary
es Australia 14d	Sen	558	(116)	Ordinary
es Australia Ital	Cell	14,061	(524)	Ordinary
UBS Securities Australia Ltd	Sell	200	(004)	Ordinary
es Australia Ltd	Sell	47	(9)	Ordinary
es Australia Ltd	Sell	29,202	(3.768)	Ordinary
UBS Securities Australia Ltd	Sell	1,132	(146)	Ordinary
es Australia Ltd	Sell	5,076	(655)	Ordinary
UBS Securities Australia Ltd	Sell	1,976	(255)	Ordinary
UBS Securities Australia Ltd	Sell	7,355	(646)	Ordinary
UBS Securities Australia Ltd	Sell	225	(53)	Ordinary
UBS Securities Australia Ltd	Self	999'9	(829)	Ordinary
UBS Securities Australia Ltd	Nett College	7,755	(1,000)	Ordinary
UBS Securities Australia 11d	Sell	5,980	(900)	Ordinary
es Australia Ltd	Sell	427	(55)	Ordinary
es Australia Ltd	Sell	9.306	(1 200)	Ordinary
es Australia Ltd	Sell	427	(22)	Ordinary
es Australia Ltd	Sell	19,388	(2,500)	Ordinary
es Australia Ltd	Sell	2,418	(312)	Ordinary
es Australia Ltd	Jen C	18,976	(2,447)	Ordinary
es Australia Ltd	Sell	57.2	(757)	Ordinary
es Australia Ltd	Sell	34.161	(4 405)	Ordinary
es Australia Ltd	Sell	6.980	(006)	Ordinary
es Australia Ltd	Sell	7,755	(1,000)	Ordinary
UBS Securities Australia Ltd	Sell	5,979	(771)	Ordinary
UBS Securities Australia Ltd	Sell	816	(105)	Ordinary
UBS Securities Australia Ltd	Sell	202	(26)	Ordinary
es Australia Ltd	Sell	210	(27)	Ordinary
UBS Securities Australia Ltd	Sell	47	(9)	Ordinary
UBS Securities Australia Ltd	Sell	2,347	(302)	Ordinary
JBS Securities Australia Ltd	Sell	1,406	(181)	Ordinary
Securities Australia Ltd	Sell Sell	5,905	(760)	Ordinary
Obs securities Australia Ltd	- Ne	4,819	(621)	Ordinary
Securities Australia Ltd	libo	4,035	(520)	Ordinary
Securities Australia Liu	(Se)	1,901	(245)	Ordinary
Securities Australia Ltd	iac i	1,546	(661)	Ordinary
or Australia Ind	Series Contraction of the Contra	4,775	(519)	Ordinary
IRS Securities Australia (14)	lley.	13,401	(2,008)	Ordinary
Anticipal in	133	30,700	(4,000)	Orginary
se Australia Ind	A10	54,400	(8,435)	Ordinary
UBS Securities Australia Ltd	Buv	17.643	1 652	Ordinary
es Australia Ltd	Buy	90.858	12 000	Ordinary
UBS Securities Australia Ltd	Self	115,541	(15,000)	Ordinary
UBS Securities Australia Ltd	Sell	448,652	(58,237)	Ordinary
es Australia Ltd	Sell	273,492	(35,460)	Ordinary
es Australia Ltd	Вцу	164,253	21,566	Ordinary
tralia Branch	Buy	1,173,839	150,000	Ordinary
tralia Branch	Sell	262 900	/24 E10)	Ordinary
UBS Securities Australia Ltd UBS Securities Australia Ltd UBS AG Australia Branch UBS AG Australia Branch	Seil Buy Buy Sell	273,4 164,2 1,173,8 262,90	28 23 25	32 (35,460) 53 21,566 39 150,000 00 (34,519)
			10.40	1/2

04-Dec-14	UBS Securities Australia Ltd	Buy	3,523	466	Ordinary
04-Dec-14	UBS Securities Australia Ltd	Buy	26,634	3,523	Ordinary
04-Dec-14	UBS securities Australia Ltd	Buy	3,735	494	Ordinary
04-Dec-14	UBS Securities Australia Ltd	Buy	2,948	390	Ordinary
	not plant continue on the	Enter Into Derivative	OS.	/71	Ordinary
04-Dec-14	UBS Securities Australia Ltd	Contract	N/A	825,199	Options
04-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	NA.	825 199	Ontions
04-Dec-14	I.BS Securities Australia Ind	Enter Into Derivative	V.	100	
		Enter Into Derivative	72	661,626	Options
04-Dec-14	UBS Securities Australia Ltd	Contract	N/A	825,199	Options
04-Dec-14	UBS Securities Australia Ltd	Contract	ΝΑ	825,199	Options
04-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	ΑN	825,199	Options
04-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	ΑN	825.200	Options
04-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	d Z	825.200	Ontions
04-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	δ <sup>2</sup>	825,200	Continue
04-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	¥2	825.200	Ontions
04-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	WA	825.200	Options
04-Dec-14	UBS Securities Australia Ltd	Enter into Derivative Contract	ΝΑ	825,200	Options
04-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	δN	825 200	Cotions
04-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	NA NA	825,200	Options
04-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	₹ Z	825.200	Oations
04-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	N.A.	825,200	Options
04-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	₹ Ž	825.200	Options
04-Dec-14	UBS Securities Australia Ltd	Enter into Derivative Contract	NA NA	825,200	Options
04-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	ΑN	825.200	Options
04-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	NA	825,200	Options
04-Dec-14	UBS Securities Australia Ltd	Enter into Derivative Contract	N/A	825,200	Options
04-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	ΝΑ	825,199	Options
04-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	ΝΑ	825,199	Options
04-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	ΝΑ	825,199	Options
04-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	ΝΑ	825,199	Options
04-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	NA	825,199	Options
04-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	Ŋ	825,199	Options
04-Dec-14	UBS Securities Australia [td	Enter Into Derivative Contract	***	825 200	Socitor
04-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	ΑN	825,200	Options
04-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	Š	825 200	Ontions
05-Dec-14	UBS Global Asset Management	Buy	4,909	639	Ordinary
05-Dec-14	UBS Securities Australia Ltd	Sell	19,658	(2,549)	Ordinary
05-Dec-14	UBS Securities Australia Ltd	Seli	23,056	3,000	Ordinary
05-Dec-14	UBS Securities Australia Ltd	Buy	40,633	5,324	Ordinary
05-Dec-14	UBS Securities Australia Ltd	Buy Sell	9,604	1,239	Ordinary
3	מול שנותר שמינותו ליווות מולים	JCII	277,001	(000,44)	Ofdinary

Deceid   185 Securities Australia Ltd     Out-Deceid   185 Securit		i celi		(:2)	
	ecurities Australia Ltd	JA:	147	(61)	Ordinary
	UBS Securities Australia Ltd	Sell	1,085	(140)	Ordinary
	securities Australia Ltd	Sell	543	(70)	Ordinary
	securities Australia Ltd	llec 3	1,085	(140)	Ordinary
	Obstantiation Australia Ltd	la C	930	(071)	Ordinary
	UBS Securities Australia Ltd	Jes III	086	(120)	Ordinary
	ecurities Australia Ltd	Sell	395	(51)	Ordinary
	ecurities Australia Ltd	Sell	395	(51)	Ordinary
	ecurities Australia Ltd	Sell	597	(7.7)	Ordinary
	ecurities Australia Ltd	Sell	564	(34)	Ordinary
	ecurities Australia Ltd	Sell	256	(33)	Ordinary
	ecurities Australia Ltd	Sell	310	(40)	Ordinary
	ecurities Australia Ltd	Sell	233	(30)	Ordinary
	ecurities Australia Ltd	Sell	310	(40)	Ordinary
	ecurities Australia Ltd	Sell	411	(53)	Ordinary
	ecurities Australia Ltd	Sell	62	(8)	Ordinary
	ecurities Australia Ltd	Sell	140	(18)	Ordinary
	ecurities Australia Ltd	Sel!	372	(48)	Ordinary
١	ecurities Australia Ltd	Sell	02	(6)	Ordinary
	ecurities Australia Ltd	Seli	163	(12)	Ordinary
1	UBS Securities Australia Ltd	Sell	233	(30)	Ordinary
1	UBS Securities Australia Ltd	Sell	47	(9)	Ordinary
34-Dec-14 UBS S	UBS Securities Australia Ltd	Sell	601	(14)	Ordinary
OBS	Securities Australia Ltd	Sell	02	(6)	Ordinary
Ī	JBS Securities Australia Ltd	Sell	233	(30)	Ordinary
OBS	Securities Australia Ltd	Sell	02	6)	Ordinary
4-Dec-14 UBS S	JBS Securities Australia Ltd	Sell	069	(68)	Ordinary
34-Dec-14 UBS S	Securities Australia Ltd	Sell	2,527	(326)	Ordinary
-	JBS Securities Australia Ltd	Sell	23	(3)	Ordinary
UBS S	JBS Securities Australia Ltd	195	532	(89)	Ordinary
OBS S	UBS Securities Australia Ltd	Sell	23,490	(3,000)	Ordinary
	ecurities Australia Ltd	Sell	799	(102)	Ordinary
	ecurities Australia Ltd	Sell	18,408	(2,354)	Ordinary
	UBS Securities Australia Ltd	Sell	771,122	(29,818)	Ordinary
04-Dec-14 UBS S	ecurities Australia Ltd	Sell	6,209	(794)	Ordinary
	ecurities Australia Ltd	Pel	3,879	(496)	Ordinary
	Securities Australia Ltd	Seii	2,064	(264)	Ordinary
en E	ecurities Australia Ltd	Sell	2,002	(256)	Ordinary
UBS S	ecurities Australia Ltd	Sell	6,209	(794)	Ordinary
UBS S	S Securities Australia Ltd	Sell	2,330	(298)	Ordinary
UBS S	ecurities Australia Ltd	Sell	4,692	(009)	Ordinary
OB	ecurities Australia Ltd	Sell	1,697	(717)	Ordinary
8	ecurities Australia Ltd	Self	905'9	(832)	Ordinary
04-Dec-14 UBS S	UBS Securities Australia Ltd	Sell	149	(19)	Ordinary
Γ	Securities Australia Ltd	Sell	00	(8)	Ordinary
UBS S	UBS Securities Australia Ltd	Sell	41.157	(5.263)	Ordinary
UBS S	UBS Securities Australia Ltd	Sell	56.484	(7.223)	Ordinary
UBS S	UBS Securities Australia Ltd	Sell	3 910	(500)	Ordinary
S Sen	UBS Securities Australia Ltd	Sell	2,119	(027)	Ordinary
UBS Se	Securities Australia Ltd	Sell	23	6	Ordinary
UBS S	Securities Australia Ltd	Sell	1 300	(3,66)	Ordinary
II RS S	JRS Securities Australia J to	Bin	23 502	2 013	Ordinary
IBSS	JBS Securities Australia Ltd	( J	95 95	(130,00)	Ordinan
OBS Se	UBS Securities Australia Ltd	Self	5,818	(144)	Ordinary
UBS S	UBS Securities Australia Itd	Self	734 900	(000 05)	Ordinary
UBS Se	UBS Securities Australia Ltd	Sell	12.724	(1 675)	Ordinary
UBS Se	Securities Australia Ltd	Seil	9.239	(1.180)	Ordinary
Ī	UBS Securities Australia Ltd	Sell	175.0371	(22.354)	Ordinary
04-Dec-14 UBS Se	ecurities Australia Ltd	Sell	46 980	(6,000)	Ordinary
UBS Se	UBS Securities Australia Ltd	Sell	602.6	(1.240)	Ordinary
IJBS Se	JBS Securities Australia Itd	Sell	21 078	(2, 2, 7)	Ordinary
UBS Se	Securities Australia Ltd	Seli	156.600	(200,00)	Ordinary
	Securities Australia Ltd	Buy	0058	1,000	Ordinary
UBS Se	ecurities Australia Itd	Buy	0000	1	Ordinary
IR. C.	IRS Securities Australia 11d	Riv	10.049	1 205	Section 5
22 2011	IDS Securities Australia 14d	2 2	57.00	70E C	Ordinally Commany
3 20	UDS Sociation Australia 14d	Bing	507	\$7,0	Croimary
2000	Countries Australia Ltd	ino d	50/	16 3	Ordinary
200	JBS Securities Australia Ltd	any	9,3/3	1,211	Ordinary
	UBS Securities Australia Ltd	Buy	897	116	Ordinary
UBS Se	ecurities Australia Ltd	Buv	4 143	536	Ordinary
2 200	Culties Account and	(na	4,143	956	Oldinary

Ordinary	Grainary	Ordinary	Ordinary	(in in in	Ordinary	O Grandy Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	- 1		1	1	Ordinary	1	H	١.	- 1	-1	1	1	Ordinary	Ordinary Contrary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary																		
(640)	(1,020)	243	4 014	auc	(00 AC)	1 168	(10.574)	(21,045)	(1,654)	(2,000)	230	(25,000)	(31,928)	62,309	(332)	(3,032)	12,000	(22,290)	(38,918)	(41,433)	31,090	95,400	300,000	(218,876)	200,000	(550,672)	1,743,035	(55,182)	1500	(1531)	186	34	186	80	34	80	101	44	155	166	147	113	111	100 000	425	38	426	668	7,178	12	1,123	297	11,539	425	40,000	(14,602)	1 707 1	100,17	(11,540)	(6,832)	(9,578)	(10,000)	(2,000)	(308)	70,000	200,000
5,024	36.126	1 908	31,510	1 633	196 250	9 169	83,006	165,203	12,984	15,700	1,677	182,250	241,331	465,722	2,520	22,895	90,280	169,268	295,715	314,626	231,530	696,614	2,224,694	1,619,885	1,489,637	4,063,169	12,920,551	408,218	11.430	11635	1,414	258	1,414	809	258	608	768	334	1,178	1,262	1,117	829	844	741 000	3,132	280	3,140	6,626	52,902	3.987	8,277	2,189	85,042	3,132	292,400	106,449	12 400	14 514	84,127	49,805	69,824	72,900	14,580	2,245	510,300	1.458.000
li de	Sell	Buy	Buy	Suv	Sell	Buy	Seli	les	Sell	Self	Buy	Sell	Sell	Buy	Sell	Sell	Buy	Sell	Sell	Sell	Buy	Buy	Buy	Sell	Buy	les.	Buy	- J	Buv	lies.	Buy	Buy	Buy	Buy	eny	Buy	an.	Buy	Buy	Buy	Buy	Buy	Riv	Buy	Seil	Buv	Sell	Sell	Sett	Sefi	Sell	Sell	Sell	Buy	Buy											
UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Global Asset Management	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS AG Australia Branch	UBS AG Australia Branch	UBS AG Australia Branch	UBS AG Australia Branch	UBS AG Australia Branch	UBS AG Australia Branch	USS Wealth Management Australia 14d	UBS Wealth Management Australia Ltd	UBS Wealth Management Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia ctd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Itd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Itd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd
08-Dec-14	Γ	08-Dec-14	08-Dec-14	08-Dec-14		08-Dec-14			08-Dec-14	38-Dec-14	39-Dec-14	39-Dec-14	39-Dec-14	9-Dec-14			li		09-Dec-14	-	ı	-	-	ı	ı	ı		9-Dec-14	T	39-Dec-14	П	9-Dec-14				9-Dec-14	T		Ш					39-Dec-14			09-Dec-14	T	T							79-Dec-14	T	l	П	19-Dec-14			Ī	Ī	-Dec-14	09-Dec-14

				(000,00)	l
05-Dec-14	UBS Securities Australia Ltd	Buy	168,467		
05-Dec-14	UBS Securities Australia Ltd	Seli	429,953	(665,53)	Ordinary
05-Dec-14	UBS Securities Australia Ltd	Sell	127,21		
05-Dec-14	UBS Securities Australia Ltd	Sell	511,835		
05-Dec-14	UBS Securities Australia Ltd	Stack borrowed	ΑVA		
05-Dec-14	UBS AG, London Branch	8uy	232,664		
05-Dec-14	UBS AG, London Branch	Sell	188,776		
05-Dec-14	UBS AG Australia Branch	Buy	353,440		
05-Dec-14	UBS AG Australia Branch	Seli	241,401		
)5-Dec-14	UBS Wealth Management Australia Ltd	Sell	671		
)5-Dec-14	UBS Wealth Management Australia Ltd	Sell	488		
05-Dec-14	UBS Wealth Management Australia Ltd	Buy	2.976		
5-Dec-14	UBS Wealth Management Australia Ltd	Sell	701		
5-Dec-14	LiBS Wealth Management Australia Ltd	Buv	19,659	2570	
5-Dec-14	LIRS Wealth Management Australia 1th	Rin	1 784		Ordinary
5-Dec-14	IDS Wealth Management Australia Ho	Col	1,404		1
DE Dec 14	DOS Woolth Management Australia and	lo lo	6///		1
#1 - Day 10	Upo Wealth Management Australia Liu	Den.	950		
13-Dec-14	UBS Wearth Management Australia Ltd	Ang .	2,264		
35-Dec-14	UBS Wealth Management Australia Ltd	sell	1,516		
13-Dec-14	UBS wearth Management Australia Ltd	Jac.	625		
25-Dec-14	UBS Wealth Management Australia Ltd	pn)	2,922		
5-Dec-14	UBS Wealth Management Australia Ltd	lles.	922	(121)	Ordinary
5-Dec-14	UBS Wealth Management Australia Ltd	Ruy	5,561		
5-Dec-14	UBS Wealth Management Australia Ltd	Sell	12,733		
5-Dec-14	UBS Securities Australia Ltd	Buy	5,319		
0-Dec-14	Obs securines Australia Ltd	and a	1,524,000		
3-Dec-14	UBS Securities Australia Ltd	gna	19,615		
5-Dec-14	UBS Securities Australia Ltd	Buy	376		Ordinary
D-Dec. 14	Opp Securities Australia LIU	ána á	9/8/9/		
D-08C-14	USS Securities Australia Ltd	De la	45,074		
3-Dec-14	ODS SECULINES AUSTRAINA LIG	ána .	868,67		
75-Dec-14	UBS Securities Australia Ltd	) sel	5,230	(681)	1
03-Det-14	ODS Securities Australia Ltd	1120	16/1		1
5-Dec-14	Obs securines Australia Ltd	liac	61,440		Ordinary
05-Dec-14	UBS Securities Australia Ltd	suy.	38,400		
- Dec-14	Obside Asset Intend	300	/89'1		
00-Dec-14	ODS Securities Australia LIO	S-III	160/61		Ordinary
00-Dec-14	on partnines Australia tio	261	560,8		Ordinary
1000	Dis Granding Australia Little	D	ooc'c)		Ordinary
Dec 14	DIS Commission Australia Little	lou,	140,000	007'07	Ordinary
8-Dec-14	Obs securities Australia Ltd	Sell	192,953		Ordinary
Dec 14	Obs Securities Adstraina Ltd	, inc	577'/		Ordinary
D. Dec. 14	ODS Securities Assigned Liu	(na)	250,6		Ordinary
9-Dec-14	ODS SECURITIES AUSTRAING LIO	360	8,507		Ordinary
5-Dec-14	UBS Securities Australia Ltd	) ell	050,081	(55,655)	Ordinary
9-Dec-14	UBS Securities Australia Ltd	any.	19,460		Ordinary
8-Uec-14	UBS Securities Australia Ltd	Sell	6,152		Ordinary
08-Dec-14	OBS Securities Australia Ltd	la C	36,894	(4,772)	Ordinary
8-Dec-14	UBS Securities Australia Ltd	S	112,741		Ordinary
8-Dec-14	UBS Securities Australia Ltd	Stock borrowed	N/A	0.0	Ordinary
08-Dec-14	UBS Securities Australia Ltd	Stock returned	NA		Ordinary
8-Dec-14	UBS AG Australia Branch	Sell	115,435		Ordinary
8-Dec-14	UBS AG Australia Branch	Seil	172,386		Ordinary
08-Dec-14	UBS Securities Australia Ltd	Seil	1,457,800		Ordinary
08-Dec-14	UBS Securities Australia Ltd	Buy	5,321		Ordinary
08-Dec-14	UBS Securities Australia Ltd	Buy	5.258		Ordinary
08-Dec-14	UBS Securities Australia Itd	Buv	27572		Ordinan
Poc 14	Let Clearly August 200		011.00		Cidinary
0.000	Con Security Actions Co.	100	18,330		Ordinary
P-1-14	ODS SECULINES Australia LIO	buy	23,494		Ordinary
09-Dec-14	OBS Securities Australia Ltd	an)	6,139		Ordinary
0-Dec-14	UBS Securities Australia Ltd	ana.	164/	348	Ordinary
08-Dec-14	UBS Securities Australia Ltd	Buy	5,895	750	Ordinary
00-Dec-14	UBS Securities Australia Ltd	buy D:m	1,65,1	210	Ordinary
D-10-14	OBS Securities Australia Ltd	in a	216	911	Ordinary
00 D-14	Obs securities Australia Ltd	éna.	5,854	7/8	Ordinary
8-Uec-14	UBS Securities Australia Ltd	Buy	25,969	3,304	Ordinary
08-Dec-14	UBS Securities Australia Ltd	Buy	1,556	861	Ordinary
08-Dec-14	UBS Securities Australia Ltd	Buy	78,600	10,000	Ordinary
38-Dec-14	UBS Securities Australia Ltd	Buy	7,962	1,013	Ordinary
38-Dec-14	UBS Securities Australia Ltd	Buy	37,689	4,795	Ordinary
38-Dec-14	UBS Securities Australia Ltd	Buy	17,048	2,169	Ordinary
08-Dec-14	UBS Securities Australia Ltd	Sell	2.834	(361)	Ordinary
		1)**	400		,
	The second secon				

0.000				292	Ordinary
10-Dec-14	UBS Securities Australia Ltd	Buy	2,086		
ı	Obsiderations Australia Ltd	buy	9,531	1,334	Ordinary
	OBS Securities Australia Ltd	hay	6,924	696	Ordinary
1	UBS Securities Australia Ltd	Buy	1,072	150	Ordinary
1	UBS Securities Australia Ltd	Buy	3,744	524	Ordinary
	UBS Securities Australia Ltd	Buy	4.487	628	Ordinary
	UBS Securities Australia Ltd	Buy	6,476	206	Ordinary
	UBS Securities Australia Ltd	Buy	98	12	Ordinary
	UBS Securities Australia Ltd	Buy	143	20	Ordinan
ı	UBS Securities Australia Ltd	Buv	1 085	152	Orolinani
	UBS Securities Australia Itd	Buy	73	300	Con Control
	I IBS Securities Australia Ind	Pare .	occ	0 0	Cidinaly
ì	LIBS Securities Australia 14d	Berry	8/7	AC S	Orginary
	Tibs Countries Australia Ind	, in 8	981	97	Ordinary
1	ODS SECULINES AUSTRAINA LIO	ína	14	2	Ordinary
ı	UBS Securities Australia Ltd	pri	1,292	181	Ordinary
	UBS Securities Australia Ltd	BUY	114	16	Ordinary
-	UBS Securities Australia Ltd	Buy	1,042	146	Ordinary
١	UBS Securities Australia Ltd	Buy	2,770	388	Ordinary
10-Dec-14	UBS Securities Australia Ltd	Buy	2,109	295	Ordinary
10-Dec-14	UBS Securities Australia Ltd	Buy	844	118	Ordinary
10-Dec-14	UBS Securities Australia Ltd	Buy	6,135	828	Ordinary
10-Dec-14	UBS Securities Australia Ltd	Buy	2 679	375	Ordinary
10-Dec-14	UBS Securities Australia Ltd	Buv	3 582	503	Ordinan
10-Dec-14	UBS Securities Australia Ltd	) N	1000	1000	Ordinary
10-Dec-14	LIBS Securities Australia Itd	BIV	\$ 000	200	Ordinaly
10-Dec-14	LISS Securities Australia 11d	× (6)	2,0,1	000	Ordinary
10-Dec-14	LIBS Securities Australia 14d	6.8	OHC'S	8	Ordinary
Doc 14	Cos securities Australia Little	buy.	2021	691	Ordinary
0.000	ODS Securities Australia Ltd	(ng	315	4	Ordinary
1000	opp Securities Australia LLC	hog	3,618	206	Ordinary
O-Dec-14	UBS Securities Australia Ltd	Buy	1,873	292	Ordinary
	UBS Securities Australia Ltd	Buy	1,931	270	Ordinary
	UBS Securities Australia Ltd	Buy	98	12	Ordinary
ı	UBS Securities Australia Ltd	Buy	2,141	562	Ordinary
	UBS Securities Australia Ltd	Buy	14,427	2,015	Ordinary
	UBS Securities Australia Ltd	Buy	2.779	389	Ordinary
	UBS Securities Australia Ltd	Buy	1.072	150	Ordinary
l	UBS Securities Australia Ltd	Buy	385	10	Ordinan
l	FIRS Securities Australia 14d	218	9000	î	Cidiliary
0-Dec-14	ILIRS Sequenties Australia Ind	3	3000	1000	Ordinary
l	TIDS Committee Ameteralia Lead	100	200,2	(014)	Cidinary
l	OCS Securities Australia Liu	200	17/	(101)	Ordinary
1	OBS Securities Australia Ltd	Sell	4,227	(265)	Ordinary
	UBS Securities Australia Ltd	Self	29	(4)	Ordinary
١	UBS Securities Australia Ltd	Sell	2,049	(287)	Ordinary
-	UBS Securities Australia Ltd	Sell	100	(14)	Ordinary
	UBS Securities Australia Ltd	Sell	322	(45)	Ordinary
	UBS Securities Australia Ltd	Sell	5.726	(802)	Ordinary
	UBS Securities Australia Ltd	ll es	050	(32)	Ordinan
10-Dec-14	UBS Securities Australia Itd	3	008 9	(220)	Ordinaly
0-Dec-14	IIRS Securities Australia Itol	3	9010	(000)	Cidillary
l	TIDE Commission Authority and		2,120	(967)	Ordinary
l	one securines Australia tito	100	/	Ξ	Ordinary
	UBS Securities Australia Ltd	Buy	15,823	2,213	Ordinary
10-Dec-14	UBS Securities Australia Ltd	Buy	9,173	1,283	Ordinary
0-Dec-14	UBS Securities Australia Ltd	Buy	236	33	Ordinary
10-Dec-14	UBS Securities Australia Ltd	Buy	1,058	148	Ordinary
0-Dec-14	UBS Securities Australia Ltd	Buy	23	00	Ordinary
0-Dec-14	UBS Securities Australia Ltd	Buy	415	28	Ordinary
	UBS Securities Australia Ltd	Buy	4,276	298	Ordinary
	UBS Securities Australia Ltd	Buy	179	25	Ordinary
	UBS Securities Australia Ltd	Buy	114	16	Ordinary
	UBS Securities Australia Itd	Buy	107	î	Ordinan
	UBS Securities Australia 11d	R <sub>IN</sub>	302	0 00	Contract of
	I IBS Societies Australia 144	(20)	000 325 21	67	Ordinary
10 Dec 14	100 Counting Australia 144	, a	000,875,71	≩ '	Ordinary
	TIDE Commission Australia Lan	, no.	24,555	5,509	Ordinary
1	Cop permittee Australia Ltd	po)	4,720	160	Ordinary
1 - 1 - 1	OBS SECURITIES AUSTRAINA LIG	houy	72,274	3,486	Ordinary
0-Dec-14	UBS Securities Australia Ltd	Buy	10,331	1,425	Ordinary
10-Dec-14	UBS Securities Australia Ltd	Buy	732,852	101,083	Ordinary
	UBS Securities Australia Ltd	Buy	49,075	6,769	Ordinary
	UBS Securities Australia Ltd	Buy	98.274	13.555	Ordinary
10-Dec-14	UBS Securities Australia Ltd	Buy	4.125	569	Ordinary
l	LIBS Securities Australia Ltd	Buy	153 323	07.4.75	,
ı	מבים שבים שפני פוום דים	000	1.00.00		
77 - 101	The second of th	-	000 000	21,140	Ordinary
	UBS Securities Australia Ltd	Buy	322,690	44,509	Ordinary

	UBS Securities Australia Ltd UBS Securities Australia Banch UBS AG Australia Banch UBS	Self	1,069 2,023 2,28,848 41,239 2,3,601 19,410 118,765 2,33,429 2,33,539 1,43,639 1,43,639 1,43,639 1,43,639 1,43,639 1,43,639 1,43,639 1,43,639 1,43,639 1,50,6	(141) (693) 22,000 (559) (7,697) (7,697) (7,697) (1,69	Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	Assiste Management states and assisted Lide Se Australia Lide Se Australia Lide Se Australia Lide Se Australia Lide Se Australia Lide Se Australia Lide Se Australia Lide Se Australia Lide Se Australia Lide Se Australia Lide Se Australia Lide Se Australia Lide Se Australia Lide Se Australia Lide Se Australia Lide Se Australia Lide Se Australia Lide Sentch Coden Rearch Coden Rearch Coden Rearch Coden Rearch Carlia Benach Carlia	Self	\$1023 \$1028 AB \$1236 AB \$1360 \$13		Odinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	se Australia Ltd  Be Australia Bench  Be Australia Bench  Be Bench	Self   Se	23 641 10410		Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	Se Australia Ltd Se Aus	Self	23.661 19.410 118.765 223.429 233.539 233.539 233.539 23.539 23.634 13.499 13.499 13.499 13.499 14.499 15.61.64 14.595 15.04.650 1.504.6		Ordinary Ordinary Ordinary Ordinary Ordinary Ordinary
	Se Australia Ltd Se Australia Ltd Se Australia Ltd Se Australia Ltd Se Australia Ltd Se Australia Ltd Se Australia Ltd Se Australia Ltd Se Australia Ltd Se Australia Ltd Se Australia Ltd Se Australia Ltd Se Australia Ltd Se Australia Ltd Se Australia Ltd Se Australia Ltd Se Australia Ltd Se Australia Ltd Se Se Australia Ltd Se Se Australia Ltd Se Se Se Se Se Se Se Se Se Se Se Se Se S	Self	19,410 118,765 118,765 1235,429 2335,539 21,583 21,583 21,633 21,633 21,623 118,765 118,765 118,765 1,561,64 1,6		Ordinary Ordinary Ordinary Ordinary Ordinary
	as Australia Ltd as Australia Ltd as Australia Ltd as Australia Ltd as Australia Ltd as Australia Ltd as Australia Ltd as Australia Ltd as Australia Ltd as Australia Ltd don Bearch don Bearch don Bearch arial Bearch	Buy   Buy	33,401 118,765 233,539 21,883 21,883 21,884 31,894 118,765		Ordinary Ordinary Ordinary
	Se Australia Ltd Se Australia Ltd Se Australia Ltd Se Australia Ltd Se Australia Ltd Se Australia Ltd Se Australia Ltd Se Australia Ltd Se Australia Ltd Se Australia Ltd Se Australia Ltd Se Australia Ltd Se Australia Ltd Se Australia Ltd Se Australia Ltd Se Australia Ltd Se Australia Ltd Se Se Australia Ltd Se Se Se Se Se Se Se Se Se Se Se Se Se S	Seif	118,765 233,429 233,539 51,982 51,6346 51,73,633 118,765 118,7		Ordinary Ordinary Ordinary
	se Australia Ltd  se Australia Ltd  se Australia Ltd  se Australia Ltd  se Australia Ltd  se Australia Ltd  se Australia Ltd  se Australia Ltd  se Australia Ltd  se Australia Ltd  se Australia Ltd  se Australia Ltd  se Australia Ltd  conden Banch  realia Banch	Buy   Buy	233,429 23,539 23,539 21,5346 23,463 25,463 25,162 25,162 25,163		Ordinary
1     1   1   1   1   1   1   1   1   1	se Australia Ltd se Australia Ltd se Australia Ltd se Australia Ltd se Australia Ltd se Australia Ltd se Australia Ltd se Australia Ltd se Australia Ltd don Bearch don Bearch don Bearch se alia Bearch	Self     Buy	233,539 51,883 83,493 173,673 173,673 173,673 118,65 118,65 118,65 118,65 118,65 118,65 118,65 118,65 118,65 128,64 138,23 13		Ordinary
	se Australia Ltd  Australia Ltd  Se Australia Ltd  Se Australia Ltd  Se Australia Ltd  Se Australia Ltd  Se Australia Ltd  Se Australia Ltd  Se Australia Ltd  Se Australia Ltd  Se Australia Ltd  Se Australia Ltd  Se Australia Ltd  Se Australia Ltd  Se Australia Ltd  Se Australia Sench  tralia Bench	Buy   Buy	51,983 316,346 43,453 5,175,303 113,498 118,765 118,765 118,765 1,504,650 1,504		
	se Australia Ltd  Se Australia Ltd  Se Australia Ltd  Se Australia Ltd  Se Australia Ltd  Se Australia Ltd  Se Australia Ltd  Se Australia Ltd  Se Australia Ltd  Se Australia Ltd  Se Se Australia Ltd  Se Se Se Se Se Se Se Se Se Se Se Se Se S	Buy   Buy	316,346 5,76,303 173,673 173,673 14,98 1,67,616 1,	43,915 (5,999) 714,959 23,988 1,876	Ordinary
<u> </u>	se Australia Ltd  Ex-Australia Ltd  Se Australia Ltd  Se Australia Ltd  Good Bench  don Bench  don Bench  Trail Bench	Self     Buy     Buy     Buy     Buy     Buy     Self     Self     Self     Self     Self     Self     Buy	43.493 5,176.503 173,673 173,673 118,765 118,765 1,143,389 1,504,650 1,504,6	(5,999) 714,959 23,988 1,876	Ordinary
	se Australia Ltd  Se Australia Ltd  Godon Banch  Godon Banch  Traila Banch  Traila Banch  Traila Banch  Traila Banch  Traila Banch  Traila Banch  Traila Banch  Traila Banch  Traila Banch  Traila Banch  Traila Banch  Traila Banch  Traila Banch  Traila Banch  Traila Banch  Traila Banch  Traila Banch  Traila Banch  Traila Banch	Buy   Buy   Buy   Buy   Buy   Buy   Sell   Sell   Sell   Sell   Sell   Buy   Buy   Buy   Sell   Buy   Sell   Buy   Sell   Buy   Sell   Buy   Sell   Buy   Sell   Buy   Sell   Buy   Sell   Se	176.303 13.698 14.6594 118.765 118.765 156.164 1.504.650	714,959 23,988 1,876	Ordinary
	se Australia Lid Australia Lid don Barnch Tralia Branch	8 buy 8 buy	173,673 114,598 114,594 118,765 1,601,650 1,009,538 1,00	23,988	Ordinary
	Se Australia Ltd don Berrich don Berrich ratalia Berrich	Buy   Buy	13.499 1.46.554 118.055 1.56.164 1.434.399 1.504.650 1.504.650 1.808.237 2.521.029 1.584.370 1.407.318	1,876	Ordinary
	don Beach  don Beach  raile Beach  raile Beach  raile Beach  raile Beach  raile Beach  raile Beach  raile Beach  raile Beach	Bloy Sell Sell Sell Sell Bloy Bloy Bloy Bloy Bloy Bloy Bloy Bl	146,954 118,765 1186,164 1,434,389 1,504,650 3,582,00 2,521,029 1,984,370 1,984,370 1,984,370		Ordinary
	dodor Banch clais Banch trais Banch trais Banch crais Banch crais Banch trais Banch trais Banch trais Banch trais Banch	Sedi Sedi Sedi Sedi Sedi Sedi Sedi Sedi	118.765 1,634.389 1,504.650 1,0795.38 1,0795.38 1,0795.38 1,086.237 2,551,00 1,564.370	16,438	Ordinary
	ratella Branch ratella Branch ratella Branch ratella Branch ratella Branch ratella Branch ratella Branch ratella Branch ratella Branch ratella Branch ratella Branch ratella Branch ratella Branch ratella Branch ratella Branch	Self Self Self Self Buy Buy Buy Buy Buy Buy Buy Buy Buy Buy	156 164 1,434,389 1,504,650 1,079,538 1,009,237 1,504,370 1,504,370 1,504,370	(16,438)	Ordinary
	ratela Barch egala Barch egala Barch eraila Barch eraila Barch eraila Barch eraila Barch eraila Barch eraila Barch eraila Barch eraila Barch eraila Barch eraila Barch eraila Barch eraila Barch	Self Self Self Buy Buy Self Self Self Self Self Self Self Self	1,434,389 1,504,650 388,260 1,079,538 1,808,237 2,521,029 1,984,370 1,584,370 1,567,318	(21,723)	Ordinary
	erala Branch Branch Textila Branch Textila Branch Textila Branch Textila Branch Textila Branch Textila Branch	Self Self Buy Buy Self Self Self Buy Buy Buy Buy Buy Buy Buy Buy Buy Buy	1,504,650 358,260 1,079,538 1,808,237 2,521,029 1,984,370 5,780	)	Ordinary
	araia Barch razia Barch razia Barch raia Barch raia Barch raia Barch	Self Self Buy Buy Self Self Suy Self Suy Self Suy Suy Suy Suy Suy Suy Suy Suy Suy Suy	358,260 1,079,538 1,808,237 2,521,029 1,984,370 1,780 1,780		Ordinary
	East Branch Trails Branch Trails Branch Trails Branch Trails Branch Trails Branch Trails Branch Trails Branch Trails Branch	Boy Buy Boy Sell Buy Buy Buy Buy Buy Buy	1,079,538 1,808,237 2,521,029 1,984,370 1,5780 1,5780		Ordinary
	raila Branch raila Branch raila Branch raila Branch raila Branch raila Branch raila Branch raila Branch	Buy Buy Sell Buy Buy Buy Sell Buy Sell Buy	1,808,237 2,521,029 1,984,370 5,780		Ordinary
	ralia Branch Iralia Branch Iralia Branch Iralia Branch Iralia Branch	Buy Sell Sell Buy Buy Sell Sell Buy	2,521,029 1,984,370 5,780	250,000	Ordinary
	tralia Branch Italia Branch Italia Branch Italia Branch	Sell Buy Buy Buy Sell Buy Buy Sell	1,984,370 5,780		Ordinary
	tralia Branch tralia Branch tralia Branch	Buy Sell Buy Buy Sell Buy	5,780		Ordinary
	tralia Branch tralia Branch tralia Branch	Sell Buy Buy Buy Sell Buy	1 550 318		Ordinary
	tralia Branch	Buy Buy Buy Sell Buy			Ordinary
	ralia Branch	Buy Buy Sell Buy	3,591,446		Ordinary
ı		Buy Sell Buy	1806,031		Ordinary
-	ralia Branch	Sell Buy	12,800		Ordinary
	Management Australia Ltd	Buy	25,379		Ordinary
10-Dec-14 UBS Wealth N	Management Australia Ltd		18,733	2,620	Ordinary
	s Australia Etd	Buy	29	4	Ordinary
١	UBS Securities Australia Ltd	Buy	29	4	Ordinary
10-Dec-14 UBS Securities	Securities Australia Ltd	Buy	59	4	Ordinary
	UBS Securities Australia Ltd	Buy	1,550	217	Ordinary
٦	s Australia Ltd	Buy	1,550	217	Ordinary
MBS	Securities Australia Ltd	Buy	1,550	217	Ordinary
٦	Securities Australia Ltd	Buy	4,783	699	Ordinary
	UBS Securities Australia Ltd	Buy	5,506	077	Ordinary
Ī	UBS Securities Australia Ltd	Buy	4,783	699	Ordinary
0-Dec-14 UBS Securities	UBS Securities Australia Ltd	Buy	479	29	Ordinary
OBS	Securities Australia Ltd	Buy	5,173	572	Ordinary
0-Dec-14 UBS Securities	JBS Securities Australia Ltd	Buy	572	80	Ordinary
Ī	Securities Australia Ltd	Buy	1,625	722	Ordinary
	JBS Securities Australia Ltd	Buy	758	106	Ordinary
	UBS Securities Australia Ltd	Buy	086	137	Ordinary
l	s Australia Ltd	Buy	16.259	2 2 7 4	Ordinary
	s Australia Ltd	Buy	844	118	Ordinary
	s Australia I td	Buy	922	32	Ordinary
	s Australia Itd	Bliv	177	16.4	Ordinary
0-Dec-14 I/BS Securities	Australia Itd	Rin	1130	101	Ordinary
l	August 14d	Rew	cac e	357	Countries
l	A Australia Liu	and p.m.	603 •	5/1	Ordinary
O Dec 14 LIBS Societies	UDS Consisting Australia and	in a	200,1	<del>+77</del>	Ordinary
l	Actual of the second of the se	623	636	9 (	Ordinally
Т	August and Lid	1110	5,009	432	Ordinary
1	S Australia Ltd	Buy	3,465	485	Ordinary
-	s Australia Ltd	Buy	2,915	408	Ordinary
1	s Australia Ltd	Buy	5,523	773	Ordinary
-	s Australia Ltd	Buy	1,072	150	Ordinary
	s Australia Ltd	Buy	1,072	150	Ordinary
	s Australia Ltd	Buy	1,386	194	Ordinary
	s Australia Ltd	Buy	400	95	Ordinary
l	s Australia Ltd	Buy	203	17	Ordinary
	s Australia Ltd	Buv	1 393	195	Ordinary
10-Dec-14 UBS Securities	UBS Securities Australia 11d	Buv	1 480	202	Ordinary
ı	Australia Itd	BIN	177	6	Ordinary
	Australia I th	Risk	808.6	2 700	Ordinary
	Anetralia I to	Birv	200,1	150	Ordinary
	Australia I td	Riv	AA1 C	002	Ordinary
10-Dec-14 UBS Securities Australia 11d	Acceptance and	Rin	2,144	300	Ordinary
	Australia Ltd	grij	5,115	436	Ordinary

iBS Securities Australia Ltd	Sell	006	(125)	Ordinary
BS Securities Australia Ltd	Sell	216	(30)	Ordinary
UBS Securities Australia Ltd	Sell	828	(115)	Ordinary
35 Securities Australia 14d	Riv	1 100 000	(103)	Ordinary
IBS Securities Australia 14d	Rin	1 747	200,000	Crainery
IRS Securities Australia Itd	Rin	0 541	547	Ordinary
S Securities Australia Itd	Riv	723	R I	Colling
25 Securities Australia 14d	Cell	1200	(400,17	O CITIES O
Of Countries Australia and	170	107.6	(205,1)	Ordinary
ODS Securities Australia Ltd	les.	/59/1	(507)	Ordinary
2) Securities Australia Liu		3	(A)	Ordinary
Obsisted Australia Ltd	) All	63	(6)	Ordinary
UBS Securities Australia Ltd	) Set	7,741	(1,098)	Ordinary
ss securities Australia Ltd	Sell	3,652	(518)	Ordinary
35 Securities Australia Ltd	Sell	12,126	(1,720)	Ordinary
35 Securities Australia Ltd	Sell	3,391	(481)	Ordinary
35 Securities Australia Ltd	lleS.	2.834	(402)	Ordinary
SS Securities Australia Ltd	lloS	2 940	(247)	
OC Convictor Australia Leal	100	25,240	(/17)	Cranary
of Countries Actually and	173	107	(37)	Ordinary
o securities Australia Liu	"ac"	7,164	(307)	Ordinary
S securities Australia Ltd	sell	169	(24)	Ordinary
BS Securities Australia Ltd	Sell	113	(16)	Ordinary
BS Securities Australia Ltd	Sell	2,834	(402)	Ordinary
SS Securities Australia 11d	leS	12.058	(1 020)	Commen
IDC Committee Australia 1+4	1170	12,000	(000'1)	Countary
So Securities Abstratia Ltd	ii o	12,82/	(1,822)	Ordinary
JBS Securities Australia Ltd	Sell	3,492	(496)	Ordinary
JBS Securities Australia Ltd	Sell	141	(20)	Ordinary
BS Securities Australia Ltd	Sell	85	(12)	Ordinary
JBS Securities Australia Ltd	Sell	4.040	(573)	Ordinary
IBS Securities Australia 144	Cell	2144	500	100
DC Countries Australia 144	100	3,144	(0000)	Ordinary
es securites Australia Liu	1361	015,5	(326)	Ordinary
JBS Securities Australia Ltd	, ser	494	(02)	Ordinary
UBS Securities Australia Ltd	Sell	5,421	(694)	Ordinary
BS Securities Australia Ltd	Sell	874	(124)	Ordinary
BS Securities Australia Ltd	Sell	42	(9)	Ordinary
BS Securities Australia I to	les	7	9	Confinent
BS Securities Australia 1+d	Coll	030 6	(1.7)	or and
20 Securities Australia Lid	130	Opor's	(434)	Ordinary
Obs securities Australia Ltd	100	717	(30)	Ordinary
bs securities Australia Ltd	III-X	3,264	(463)	Ordinary
Bs securties Australia Ltd	sell	8,502	(1,206)	Ordinary
BS Securities Australia Ltd	Buy	988,400	140,000	Ordinary
JBS Securities Australia Ltd	Sell	205	(62)	Ordinary
Securities Australia	Sell	10.258	(1.453)	Ordinary
BS Securities Australia Itd	Sell	346	(00)	in in
130 Commission American 14d	100	240.44	(64)	Ciginary
by securines Australia Ltd	100	4,84/	(2,106)	Ordinary
bs securities Australia Ltd	Sell	12/	(18)	Ordinary
UBS Securities Australia Ltd	Sell	268	(38)	Ordinary
35 Securities Australia Ltd	Sell	17,694	(2,508)	Ordinary
UBS Securities Australia Ltd	Sell	3,137	(445)	Ordinary
UBS Securities Australia Ltd	Seil	9214	(7 307)	Ordinan
IRS Securities Australia Ind	19	555	(105,17)	Occinary
So Securices Australia Ltd	100	233	(33)	Orginary
JBS Securities Australia Ltd	Z.	4,117	(284)	Ordinary
UBS Securities Australia Ltd	Sell	4,082	(679)	Ordinary
BS Securities Australia Ltd	Sell	1,466	(208)	Ordinary
JBS Securities Australia Ltd	Sell	3 626	(515)	Ordinary
IRS Securities Australia Itol	3	14 005	(0.150)	Contract of
Lat Commission A section of CO.	= 3	000	(4,130)	Countries of
as securines Appropriate	361	861,1	(164)	Ordinary
UBS Securities Australia Ltd	sell	1,723	(244)	Ordinary
JBS Securities Australia Ltd	Sell	21,022	(2,984)	Ordinary
JBS Securities Australia Ltd	lles	9,173	(1,303)	Ordinary
UBS Securities Australia Ltd	Sell	5.308	(754)	Ordinary
UBS Securities Australia itd	Sell	3 497	(30%)	Ordina
BS Securities Australia 144	197	47.0	(6.3)	Company
1100 Committee Assets Inch	200	7.01	(10)	or and
no allegation can library can	100	7,814	(01,17)	Ordinary
UBS Securities Australia Ltd	Seil	374	(23)	Ordinary
UBS Securities Australia Ltd	Sell	2,841	(403)	Ordinary
UBS Securities Australia Ltd	Sell	2017	(116)	Ordinary
200 000 000 000 000 000 000 000 000 000	====	È .	(0.10)	Cicinary
UBS Securities Australia Ltd	Seil	1,429	(203)	Ordinary
UBS Securities Australia Ltd	Sell	275	(68)	Ordinary
UBS Securities Australia Ltd	Sell	1 408	(002)	Ordinary
DC Countries Australia 140	100	200	(0)	100
DS SECULIES AUSTRAIR LLG	- N			
			(45)	Ordinary
BS Securities Australia Ltd	Sell	6.442	(45)	Ordinary

10-Dec-14	PIRC Securities Australia 14d	<u>a</u>			
10-Dec-14	UBS Securities Australia Ltd	Sell	8,916	(1,213)	Ordinary
10-Dec-14	UBS Securities Australia Ltd	Sell	904	(123)	Ordinary
	UBS Global Asset Management	===	1,520	(216)	Ordinary
	UBS Global Asset Management	Sell	555,420	(78,921)	Ordinary
- 1	UBS Securities Australia Ltd	Sell	547,299	(77,945)	Ordinary
١	UBS Securities Australia Ltd	Se∯	99,430	(14,087)	Ordinary
- 1	UBS Securities Australia Ltd	Buy	31,009	4,442	Ordinary
1	UBS Securities Australia Ltd	Seit	21,210	(3,030)	Ordinary
1	UBS Securities Australia Ltd	Sei.	54,062	(/,pp4)	Ordinary
	USS Securities Australia Ltd	on)	11/101	1,44	Ordinary
Т	UDS Securities Australia Ltd	Bin	14,233	120 504	Ordinary
	USC Countries Australia Ltd	Bind	350 /	+0C,021	Countary
Т	UBS Securities Australia Itd	Sell	1 106 052	(156 909)	Ordinary
1	UBS Securities Australia Ltd	Buy	78 292	11.150	Ordinary
ı	UBS Securities Australia Ltd	Sell	1,565	(222)	Ordinary
	UBS Securities Australia Ltd	Sell	66,762	(9,461)	Ordinary
	UBS Securities Australia Ltd	Sell	55,528	(7,883)	Ordinary
	UBS Securities Australia Ltd	Buy	49,722	7,100	Ordinary
H	UBS Securities Australia Ltd	Self	175,907	(24,904)	Ordinary
11-Dec-14	UBS Securities Australia Ltd	Buy	353,866	50,156	Ordinary
1	UBS Securities Australia Ltd	Sell	210,507	(30,131)	Ordinary
11-Dec-14	UBS Securities Australia Ltd	Buy Buz	500,15	4,489	Ordinary
11-Dec-14	UBS AG, LONGOI Branch	Sell	1 106 052	606,001	Ordinary
1-Dec-14	HIRS AG Australia Branch	lias lias	60,000,000	(8,647)	Ordinary
1-Dec-14	UBS AG Australia Branch	Sell	2,124	(304)	Ordinary
11-Dec-14	UBS AG Australia Branch	Buy	18,003	2,585	Ordinary
11-Dec-14	UBS AG Australia Branch	Buy	57,010	8,186	Ordinary
11-Dec-14	UBS AG Australia Branch	Sell	1,691,129	(240,000)	Ordinary
11-Dec-14	UBS AG Australia Branch	Sell	424,183	(60,734)	Ordinary
11-Dec-14	UBS AG Australia Branch	Sell	245,169	(35,103)	Ordinary
11-Dec-14	UBS AG Australia Branch	Sell	517,975	(74,163)	Ordinary
11-Dec-14	LIBS Wealth Management Australia Ltd	Jac.	4,715	(100)	Ordinary
11-Dec-14	UBS Wealth Management Australia Ltd	Buv	11 427	1 614	Ordinary
	, and the same of	Enter Into Decivative			(min)
11-Dec-14	UBS Securities Australia Ltd	Contract	N/A	(30,100)	Options
1-Dec-14	UBS Securities Australia Ltd	Sell	1,296,655	(178,849)	Ordinary
1-Dec-14	UBS Securities Australia Ltd	Sell	14,220	(1,975)	Ordinary
1-Dec-14	UBS Securities Australia Ltd	Sell	9,151	(1,271)	Ordinary
1-Dec-14	UBS Securities Australia Ltd	Sell	5,918	(822)	Ordinary
1-Dec-14	UBS Securities Australia Ltd	Jiac J	58,404	(9,494)	Cronary
1-Dec-14	UBS Securries Australia Ltd	) Sell	7,430	(350,1)	Ordinary
1-Dec-14	UDS Securities Australia Ltd	lip lip	190,1	(1363)	Ordinary
	LIRS Securities Australia 11d	les	4 910	(689)	Ordinary
1	UBS Securities Australia Ltd	Sell	1,037	(441)	Ordinary
	UBS Securities Australia Ltd	Sell	080'1	(150)	Ordinary
	UBS Securities Australia Ltd	Seli	3,571	(496)	Ordinary
	UBS Securities Australia Ltd	Sell	418	(85)	Ordinary
П	UBS Securities Australia Ltd	Sell	7,574	(1,052)	Ordinary
	UBS Securities Australia Ltd	Sell	360	(20)	Ordinary
	UBS Securities Australia Ltd	Sell	3,571	(496)	Ordinary
11-Dec-14	UBS Securities Australia Ltd	Sell	1,015	(141)	Ordinary
1-Dec-14	UBS Securities Australia Ltd	Sell	5,134	(713)	Ordinary
1-Dec-14	UBS Securities Australia Ltd	Sell	16,027	(2,226)	Ordinary
	UBS Securities Australia Ltd	Sell.	5,044	(00/)	Ordinary
ł	UBS Securities Australia Ltd	lac lac	707')	(10/)	Organary
	TIDS Countries Australia Ltd	Je Je	019,01	(2,100)	Ordinary
11-Dec-14	UBS Securities Australia Ltd	ias S	65	(0)	Ordinary
1-Dec-14	UBS Securities Australia Ltd	Sell	19.994	(177.0)	Ordinary
	UBS Securities Australia Ltd	Sell	4,190	(582)	Ordinary
	UBS Securities Australia Ltd	Sell	247	(92)	Ordinary
	UBS Securities Australia Ltd	Sell	14	(2)	Ordinary
	UBS Securities Australia Ltd	Se#	1,138	(158)	Ordinary
	UBS Securities Australia Ltd	Self	2,254	(313)	Ordinary
	UBS Securities Australia Ltd	Sell	21,600	(000'E)	Ordinary
	UBS Securities Australia Ltd	Sell	15,106	(2,098)	Ordinary
	UBS Securities Australia Ltd	Sell	28,930	(4,018)	Ordinary
11-Dec-14	UBS Securities Australia Ltd	Sell	4,406	(612)	Ordinary

11-Dec-14   108 Securities Australia Ltd     11-Dec-14   108 Sec	la lid  la lid		5,348 1,357 35,008 35,008 43,790 43,790 1,501	(193) (193) (193) (193) (193) (193) (193) (193) (193) (193) (193)	Ordinary Ordinary Ordinary Ordinary
	la lid  la lid	Se	36,008 36 37 43,790 2,095 15,507 15,507 13,29 4,991 4,991 2,991 4,991 2,991 4,991 13,294 4,991 13,294 4,991 13,294 4,991 13,294 4,991 13,294 4,991 13,294 14,694 13,294 13,294 14,694 13,294 14,694 13,294 13,294 14,694 13,294 14,694 13,294 13,294 13,294 14,694 13,294 13,294 13,294 14,694 13,294 13,294 14,694 14,694 14,694 15,694 16,694 17,694 18,	(193) (193) (193) (193) (193) (193) (193) (193) (193)	Ordinary Ordinary
	is it id  is it id	Section   Sect	36,006 55 56 43,790 15,501 15,501 15,501 16,504 16,91 1,329 4,991 4,991 4,991 4,991 4,991 1,329 285 285 3,006 4,991 4,991 1,329 4,991 1,329 4,991 1,329 4,991 1,329 2,325 2,32	(5,122) (5,122) (8) (6,229) (2,205) (7,6)	Ordinary
	la lid  la lid	S   S   S   S   S   S   S   S   S   S	35 43,790 2,095 115,501 15,501 16,804 1,329 1,32	(5) (8) (6,229) (2,298) (2,205)	Ordinary
	la lid  la lid	Section   Sect	43,790 2,095 15,500 15,500 10,500 11,400 13,200 4,901 4,901 4,901 4,901 13,200 4,901 4,901 4,901 13,200 14,200 14,	(8) (6,229) (2,298) (2,205)	(igiligi)
	is it it it it it it it it it it it it it	Selicition   Sel	43.790 2.095 2.095 15.501 15.501 91.4 91.7 1.329 4.991 4.991 4.991 4.991 4.991 4.991 4.991 7.85 7.	(6,229) (298) (2,205)	Ordinary
	is in tird  is in	Section   Sect	2,095 11,507 5,34 3,088 3,088 3,088 1,329 4,991 4,991 4,991 4,991 1,329 4,991 1,329 1,329 1,329 1,329 1,329 1,329 1,329 1,329 1,489	(298)	Ordinary
	is it id  is it id		15.501 3.058 3.058 914 914 91329 4.991 4.991 4.991 4.991 4.991 7.88 4.991 7.88 7.88 7.88 7.88 7.88 7.88 7.88 7.8	(2,205)	Ordinary
	is it id  is it	See	534 3,058 914 914 964 4,991 4,991 255 765 133 133 133 133	(24)	Ordinary
	la lid  la lid	See	3.058 914 9,694 1,329 4,901 2,91 2,945 4,64 1,33 1,33 1,33 1,33 1,46 1,33 1,33 1,33 1,33 1,33 1,33 1,33 1,3		Ordinan
	is it it it it it it it it it it it it it	S   S   S   S   S   S   S   S   S   S	914 9694 1329 4,991 91 285 766 4169 133 91 285 785 785 133	(132)	Ordinary
		Selin   Seli	9,694 1,329 4,991 91 295 7,897 1,33 1,33 1,33 1,33 1,33 1,33 1,33 1,3	(130)	Ordinary
	is it id  is it	S   S   S   S   S   S   S   S   S   S	1,329 4,991 91 7295 464 464 4,169 133 5,827	(1,379)	Ordinary
	is it de la la la la la la la la la la la la la	See	4,991 91 795 785 464 4,169 133 5,827	(189)	Ordinary
	be ltd  be ltd	S   S   S   S   S   S   S   S   S   S	91 295 4464 4,169 133 5,827	(710)	Ordinary
	a tid  a tid  a tid  b tid  b tid  b tid  b tid  b tid  b tid  b tid  c tid	Sell	295 464 4.169 133 5.827	(13)	Ordinary
	be ltd  ltd  ltd  ltd  ltd  ltd  ltd  ltd	Self   Se	464 4,169 133 5,827	(42)	Ordinary
	a litd a litd b litd		4,169 133 5,827	(99)	Ordinary
	a littd  b littd  b littd  b littd  b littd  b littd  b littd  b littd  b littd  b littd  b littd  b littd  b littd  b littd  b littd  b littd  b littd  b littd  b littd	Sell   Sell	133	(£65)	Ordinary
	a littld  b littld  b littld  b littld  b littld  b littld  b littld  b littld  b littld  b littld  b littld  b littld  c littld  c littld  c littld  c littld  c littld  c littld  c littld  c littld  c littld	S   S   S   S   S   S   S   S   S   S	5,827	(19)	Ordinary
	bitd bitd bitd bitd bitd bitd bitd bitd	See		(830)	Ordinary
	is it to is it	Self	5,054	(720)	Ordinary
	is lated in the control of the contr	Self	7,427	(1,058)	Ordinary
	in it id it id in it id in in it id in in it id in in in in in it id in in in in in in in in in in in in in		25,181	(3,582)	Ordinary
	is it d  is it d  is it d  is it d  is it d  is it d  is it d  is it d  is it d  is it d  is it d  is it d  is it d  is it d  is it d	Sel	4,921	(200)	Ordinary
	a trd  a trd  b trd  b trd  c trd  c trd  b trd  c trd  c trd  c trd  c trd  c trd  c trd  c trd  c trd  c trd  c trd  c trd  c trd  c trd  c trd	Sell Sell Sell Sell Sell Sell Sell Sell	1,765	(251)	Ordinary
	a litté a litt	Sell Sell Sell Sell Sell Sell Sell	6,693	(952)	Ordinary
		Sell Sell Sell Sell Sell	70	(10)	Ordinary
	is little little little little little little little little little little little little	See	1,533	(218)	Ordinary
		Sell Sell Sell Sell	1,055	(150)	Ordinary
	a trd	Sell Sell Sell	28	(12)	Ordinary
	is itd as itd as itd as itd as itd as itd as itd as itd	Sell	211	(30)	Ordinary
	ia itd ia itd a itd a itd a itd a itd a itd a itd a itd	Sell	5,976	(820)	Ordinary
	a trd a trd a trd a trd a trd a trd a trd	leal lead	1,083	(154)	Ordinary
	a trd a trd a trd a trd	1	206	(72)	Ordinary
	is itd a Ltd a Ltd b Ltd	Sell	640	(16)	Ordinary
	a Ltd a Ltd a Ltd	Sell	23,155	(3,289)	Ordinary
	a Ltd a Ltd	Sell	504,845	(117.17)	Ordinary
	a Ltd	Buy	43,761	6,216	Ordinary
	7	Buy	1,841,284	261,546	Ordinary
	a Ltd	Buy	24,640	3,500	Ordinary
	a Ltd	Buy	2,478	352	Ordinary
	a Ltd	Sell	10,150,000	(1,450,000)	Ordinary
		Enter Into Derivative			
	a rto	Contract	ΑN	100	Options
	7	Enter Into Derivative			
	anomout	Compact	N/A	15,000	Options
	agement	a d	30,871	(4,319)	Ordinary
П	0 Ltd	ileo	152,832	(21,714)	Ordinary
	a Liu	100	806,6	(1,411)	Ordinary
	a Liu	lie C	14,120	(2,000)	Ordinary
Т	a Lto	- Ne	246,195	(34,545)	Ordinary
Н	a Ltd	Buy	31,692	4,495	Ordinary
2-Dec-14 Obs securities Australia	arto	euy	13,940	2,000	Ordinary
	artd	liac li	18,507	(2,640)	Ordinary
1	a Ltd	Seil	1,586	(225)	Ordinary
-	a Ltd	Buy	335,138	47,431	Ordinary
1	a Ltd	is.	399,353	(56,312)	Ordinary
Z-Dec-14 UBS Securities Australia	a Ltd	la la	269,993	(38,046)	Ordinary
1	a Utd	lles.	59,047	(8,500)	Ordinary
-1	e Ltd	Buy	313,199	45,245	Ordinary
-	a Ltd	Sell	1,750	(248)	Ordinary
1	e Ltd	Buy	168,285	24,091	Ordinary
-	ch	Buy	82,045	11,871	Ordinary
- 1	÷	Buy	333,018	47,700	Ordinary
- 1	ch	lleS	84,619	(11,871)	Ordinary
	÷	Buy	185,275	26.635	Ordinary
	£	Buy	27,783	3.994	Ordinary
12-Dec-14 UBS AG Australia Branch	6	Buy	84,412	12,135	Ordinary
Н	ch	Sell	2,117,211	(300,000)	Ordinary
	ent Australia Ltd	Seli	1,913	(271)	Ordinary
ı	opt Australia I to	Sell	010	(1.72)	or and a
1	ent Australia Lto	Sell	819	(116)	Ordinary

Dec-14	UBS Securities Australia Ltd	Sell	1,005	(143)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	18,630	(2,650)	Ordinary
11-Dec-14	UBS Securities Australia Ltd	Sell	8	6)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	1,244	(771)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	232	(33)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	3,473	(464)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	2,946	(419)	Ordinary
Dec-14	UBS Securities Australia Ltd	Buy	2,810,000	400,000	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	26,367	(3,756)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	1,957	(278)	Ordinary
Pec-14	UBS Securities Australia Ltd	Sell	14,129	(2,007)	Ordinary
Dec-14	UBS Securities Australia Ltd	Seil	7,293	(1,036)	Ordinary
Dec-14	UBS Securities Australia Ltd	Self	828	(122)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	148	(21)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	148	(21)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	3,520	(200)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	528	(22)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	1,005	(143)	Ordinary
1-Dec-14	UBS Securities Australia Ltd	Sell	3,142	(447)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	396	(25)	Ordinary
-Dec-14	UBS Securities Australia Ltd	Sell	520	(74)	Ordinary
-Dec-14	UBS Securities Australia Ltd	Sell	4,141	(685)	Ordinary
-Dec-14	UBS Securities Australia Ltd	Sell	5,954	(847)	Ordinary
1-Dec-14	UBS Securities Australia Ltd	Sell	1,666	(237)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	3,142	(447)	Ordinary
Dec-14	UBS Securities Australia Ltd	Buy	1,476,300	210,000	Ordinary
1-Dec-14	UBS Securities Australia Ltd	Sell	1,772	(252)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	10,615	(015'1)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	3,016	(429)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	2,109	(300)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	408'6	(1,395)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	1,104	(157)	Ordinary
-Dec-14	UBS Securities Australia Ltd	Seil	11,157	(1,587)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	3,459	(492)	Ordinary
Dec-14	UBS Securities Australia Ltd	les!	710	(101)	Ordinary
1-Dec-14	UBS Securities Australia Ltd	Sell	3.142	(447)	Ordinary
1-Dec-14	UBS Securities Australia Ltd	Sell	11 989	(1 703)	Ordinary
Jec-14	UBS Securities Australia Ltd	Sell	121 21	(2 439)	Ordinary
1-Dec-14	UBS Securities Australia Ltd	Sell	7.821	(1111)	Ordinary
1-Dec-14	UBS Securities Australia Ltd	Sell	827 E	(494)	Ordinary
1-Dec-14	UBS Securities Australia Ltd	Sell	155	(20)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	1 679	(734)	Ordinary
1-Dec-14	UBS Securities Australia Ltd	Sell	75,00	(800 //	Ordinary
1-Dec-14	IIBS Securities Australia 14d	/al	120,02	(4.00)	Gullary
Dec-14	LIRS Securities Australia 11d	la's	900	(0)	Ordinary
1-Dec-14		3	203 C	(74)	Guillary
Dec. 14	A IBC Cocuration Australia 1td	100	180,6	(508)	Ordinary
1000	ODS SECURIES AUSTRALIS	i ac	16,25/	(2,306)	Ordinary
1 Dec 14	Obs Securities Australia Ltd	iii.	1,008	(143)	Ordinary
Jec- 14	UBS Securities Australia Ltd	) Set	43,916	(6,238)	Ordinary
1-Dec-14	UBS Securities Australia Ltd	Sell	70	(10)	Ordinary
Jec-14	UBS Securities Australia Ltd	Sell	2,316	(328)	Ordinary
1-Dec-14	UBS Securities Australia Ltd	Sell	176	(25)	Ordinary
Jec-14	UBS Securities Australia Ltd	Sell	2,443	(347)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	4,097	(285)	Ordinary
1-Dec-14	UBS Securities Australia Ltd	Sell	3,016	(429)	Ordinary
1-Dec-14	UBS Securities Australia Ltd	Sell	3,487	(496)	Ordinary
1-Dec-14	UBS Securities Australia Ltd	Sell	7,719	(1,098)	Ordinary
bec-14	UBS Securities Australia Ltd	Sell	3,930	(653)	Ordinary
Jec-14	UBS Securities Australia Ltd	Sell	18 770	(0.670)	Ordinary
lec-14	UBS Securities Australia Ltd	Sei	18 524	(2 635)	Ordinary
ec-14	UBS Securities Australia Ltd	Sell	13,724	(707.1)	Ordinary
Pc-14	UBS Securities Australia Itd	- N	25,25	(15,13)	Ordinary
1-Dec-14	LIRS Securities Australia Ind	1	500 0	(3,621)	Ordinary
100.14	198 Securities Australia 11d	liac J	1,803	(01,10)	Ordinary
1 1 1	Cos Securities Australia Ltd	iiac ii	610,1	(145)	Ordinary
FC-14	securities	Nell Nell	3,107	(442)	Ordinary
I-Dec-14	UBS Securities Australia Ltd	Sell	16,788	(2,388)	Ordinary
1	UBS Securities Australia Ltd	Sell	2,868	(408)	Ordinary
-Dec-14	UBS Securities Australia Ltd	Sell	11,501	(1,636)	Ordinary
	UBS Securities Australia Ltd	Sell	1,589	(526)	Ordinary
	UBS Securities Australia Ltd	Sell	1,237	(176)	Ordinary
11-Dec-14	UBS Securities Australia Ltd	llas	or.o		
				(3.12)	Ordinary
	UBS Securities Australia Ltd	Sell	6.067	(118)	Ordinary

UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy Buy	16,444	2,316	Ordinary
	Buy	2,666	798	Ordinary
	Buy	7	-	Ordinary
	Buy	5.148	775	Ordinary
	Buy	285	82	Ordinary
	Suy	7,100	1,000	Ordinary
	Buy	5,34U 6,395	863	Ordinary
	Buy	7,105	1,000	Ordinary
	Buy	7,105	1,000	Ordinary
ĺ	Buy	6,395	006	Ordinary
UBS Securities Australia Ltd	Buly	1,201	169	Ordinary
	Buy	290'6	712	Ordinary
	Buv	800.0	200	Ordinary
	Buy	5,008	819	Ordinary
	Buy	2,211	311	Ordinary
	Buy	5,418	762	Ordinary
	Buy	9,200	1,294	Ordinary
1	Buy	1,578	222	Ordinary
	Buy	1,459	202	Ordinary
	Buy	2,199	309	Ordinary
	Buy	4,635	651	Ordinary
	(na	21,360	3,000	Ordinary
	Buy	2,286	321	Ordinary
	Buy	3,062	430	Ordinary
	Buy	3,980	828	Ordinary
	Buy	3,532	496	Ordinary
	Buy	1,068	150	Ordinary
	Buy	3,069	431	Ordinary
	Buy	10,958	1,539	Ordinary
ĺ	Buy	3,069	431	Ordinary
	gna	1,488	209	Ordinary
	ind.	08301	1,52,1	Ordinary
	ood)	4,614	38	Ordinary
	pro)	3,432	482	Ordinary
	ing.	871	18	Ordinary
	6.0	029,11	1,632	Ordinary
	and b	12,083	1,697	Ordinary
	śng	128	80	Ordinary
	ana	/,903	1,110	Ordinary
UBS Securities Australia Ltd	euy	2,983	419	Ordinary
	had	2,/34	387	Ordinary
	pan	377	53	Ordinary
	gnà	128	18	Ordinary
	ans	235	33	Ordinary
	euy	10,688	1,500	Ordinary
	Buy	10,688	1,500	Ordinary
	Buy	3,536	496	Ordinary
	Buy	8,328	1,168	Ordinary
	Buy	242	34	Ordinary
	Buy	242	34	Ordinary
	Buy	1,882	264	Ordinary
	Buy	30.488	4276	Ordinary
	Buy	7	- 	Ordinary
UBS Securities Australia Ltd	Buy	649	16	č
	Buy	5 355	751	Ordinary
	Buv	4 570	641	Crolinger
	Series Control	1001	+	Countries
	(Fig.	1,034	54	Ordinary
	, and	18,310	895'7	Ordinary
	Buy	3,914	549	Ordinary
	Buy	1,430	201	Ordinary
	Buy	3,560	200	Ordinary
	Buy	169	46	Ordinary
	Buy	463	9	Ordinary
	Buy	1 431	100	Ordinan
	Buy	3.745	526	Ordinary
	Buy	19 964	2 804	Ordinan
	Buv	4578	643	Ordinary
HIDC Commission Americal and	D, 00	O.C.	3	A DI JICAN
				,
	3	2,021	284	Ordinary

12-Dec-14	UBS Wealth Management Australia Ltd	Sell	31,692	(4.489)	Ordinary
12-Dec-14	UBS Wealth Management Australia Ltd	Sell	29,158	(4,130)	Ordinary
12-Dec-14	UBS Wealth Management Australia Ltd	Sell	6/9/6	(1,371)	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	¥.	600,000,000	Options
12-Dec-14	UBS Securities Australia Ltd	Seil	3,042	(434)	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Sell	1,743	(249)	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Sell	1,372	(196)	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Sell	77	(11)	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Sell	946	(135)	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Sell	491	(20)	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Sei	189	(27)	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Sell	189	(27)	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Sell	1,437	(205)	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Sell	617	(88)	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Sell	259	(37)	Ordinary
12-Dec-14	IUBS Securities Australia Ltd	B 3	259	(37)	Ordinary
12-Dec-14	UBS Securities Australia Ltd	lie C	175	(25)	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Seil	442	(63)	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Sell	617	(88)	Ordinary
	UBS Securities Australia Ltd	Sell	351	(20)	Ordinary
1	UBS Securities Australia Ltd	la C	7 22 266	(38)	Ordinary
12-Dec-14	LIBS Countries Australia Ltd	Bin	3,525,000	(500,000)	Ordinary
1	LIBS Securities Australia Hd	Bir	007.30	2010	Ordinary
	LIBS Securities Australia Itd	Bir	3 533	300	Ordinary
	UBS Securities Australia 11d	Buy	3,532	490	Ordinary
l	UBS Securities Australia Ltd	Buv	268	108	Ordinary
l	UBS Securities Australia Ltd	Buy	2.510	353	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Buy	12,762	1,795	Ordinary
	UBS Securities Australia Ltd	Buy	213	30	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Buy	2,097	295	Ordinary
12-Dec-14		Buy	5,432	764	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Buy	412	28	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Buy	405	57	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Buy	20,491	2,882	Ordinary
12-Dec-14	LIBS Securities Australia 174	Riv	907	33	Ordinary
12-Dec-14	UBS Securities Australia (1d	Buv	5,10	252	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Buy	9,648	1,357	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Buy	9,215	1,296	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Buy	19,254	2,708	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Buy	2,501	352	Ordinary
12-Dec-14	Obsideration Australia Ltd	puy.	7,268	319	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Buy	1,116	157	Ordinary
12-Dec-14	UBS Securities Australia Ind	Bin	10.345	1,745	Ordinary
12-Dec-14	UBS Securities Australia 11d	Birk	900 6	009	Ordinary
12-Dec-14	UBS Securities Australia Ind	Buv	3 150	090	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Buv	776	£ 13	Ordinary
	UBS Securities Australia Ltd	Buv	10.658	1 500	Ordinary
ł	UBS Securities Australia Ltd	Buy	996	136	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Buy	1.457	205	Ordinary
	UBS Securities Australia Ltd	Buy	2.508	353	Ordinary
1	UBS Securities Australia Ltd	Buy	16.275	2.289	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Buy	3,235	455	Ordinary
	UBS Securities Australia Ltd	Buy	4,107	578	Ordinary
	UBS Securities Australia Ltd	Buy	26,755	3,763	Ordinary
	UBS Securities Australia Ltd	Buy	7,259	1,021	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Buy	8,527	1,201	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Buy	2,599	396	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Buy	4,892	689	Ordinary
12-Dec-14	UBS Securities Australia Ltd	suy.	3,451	486	Ordinary
12-Dec-14	USS Securities Australia Ltd	Bun	6,305 L	888	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Buy	5.247	730	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Buy	1.477	208	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Buy	4,466	629	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Buy	3,507	494	Ordinary
	UBS Securities Australia Ltd	Buy	128	18	Ordinary
	UBS Securities Australia Ltd	Buy	10,082	1,420	Ordinary
1	UBS Securities Australia Ltd	Buy	620'.	066	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Buy	305	43	Ordinary
1	UBS Securities Australia Ltd	Buy	11,559	1,628	Ordinary

2-Dec-14	U65 Securities Australia Ltd	Buy	129	18	Ordinary
1	ILIBC Securities Australia 14d	P. III	, , , , ,		
2-Dec-14	UBS Securities Australia 11d	Siz S	21,034	3,030	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Ang	1 7/0		Ordinary
2-Dec-14	UBS Securities Australia Ltd	8uv	057.5 057.5		Ordinary
2-Dec-14	UBS Securities Australia Ltd	× 8	1240		Clumary
2-Dec-14	UBS Securities Australia Etd	Buy	1749	245	Ordinary
2-Dec-14	UBS Securities Australia Ltd	Buy	7 956		Colingo
2-Dec-14	UBS Securities Australia Ltd	Buy	614		Ordinary
2-Dec-14	UBS Securities Australia Ltd	Buy	129	100	Ordinary
2-Dec-14	UBS Securities Australia Ltd	Buy	727.51		Ordinan
2-Dec-14	UBS Securities Australia Ltd	BUV	4 283		Ordinary
2-Dec-14	UBS Securities Australia Itd	Bitv	020		Oldinary
2-Dec-14	LIRS Securities Australia Itd	Ban	975	/5/	Ordinary
2-Dec-14	URS Securities Australia Itd	200	621		Ordinary
-Dec-14	LIDS Commission Augustin 14d	P	9,8,4		Ordinary
Dec. 17	IIIDS Convision Australia 14d	p.:.	4,038		Ordinary
	Upo occolines Australia Liu	and c	161,11	2,397	Ordinary
- Cec-14	UBS Securities Australia Ltd	any	3,639		Ordinary
2-Dec-14	UBS Securities Australia Ltd	Buy	13,926	+	Ordinary
2-Dec-14	UBS Securities Australia Ltd	Buy	608'9		Ordinary
-Dec-14	UBS Securities Australia Ltd	Sell	11,091		Ordinary
2-Dec-14	UBS Securities Australia Ltd	Sell	18,580	(2,595)	Ordinary
2-Dec-14	UBS Securities Australia Ltd	Sell	21		
-Dec-14	UBS Securities Australia Ltd	Sell	4,260		
2-Dec-14	UBS Securities Australia Ltd	Sell	6,535		Ordinary
-Dec-14	UBS Securities Australia Ltd	Sell	6.535		
-Dec-14	UBS Securities Australia Ltd	Sell	4 962		Ordinan
2-Dec-14	UBS Securities Australia Ltd	Sell	3 146		
2-Dec-14	UBS Securities Australia Ltd	Sell	4 626	(647)	Ordinary
-Dec-14	UBS Securities Australia Ltd	Sell	4 667		
-Dec-14	UBS Securities Australia 13d	وا	4,663		١
-Dec-14	I IRS Securities Australia 14d	199	4,002		
Pec.14	LIDS Countries America 14d	100	4,000		Ordinary
Pac 14	TIDE Committee Augustic Land	11-3	36,308	(19,064)	
-Cec-14	Obside Australia Ltd	Seit	1,773		
-Dec-14	UBS Securities Australia Ltd	Buy	5,834		
-Dec-14	UBS Securities Australia Ltd	Sell	136,265		Ordinary
-Dec-14	UBS Securities Australia Ltd	Buy	11,261		Ordinary
-Dec-14	UBS Securities Australia Ltd	Sell	19,951,500		
2-Dec-14	UBS Securities Australia Ltd	Buy	1,035	150	ļ
-Dec-14	UBS Securities Australia Ltd	Buy	5.410		
-Dec-14	UBS Securities Australia Ltd	Buy	3 961	177	
-Dec-14	UBS Securities Australia Ltd	Buv	7.818	1 123	l
-Dec-14	UBS Securities Australia Itd	NT B	721	501	Ordinary
-Dec-14	I IRC Securities Australia I td	100	100	00	Oraniary
Dec-14	Libe counting Australia Lea	sns.	421	19	Ordinary
000.14	OBS SECURITIES AUSTRAINA LTD	Buy	4,464	647	Ordinary
-Dec-14	UBS Securities Australia Ltd	Buy	731	106	Ordinary
-Dec-14	UBS Securities Australia Ltd	Buy	14	2	Ordinary
-Dec-14	UBS Securities Australia Ltd	Buy	6,361	901	Ordinary
2-Dec-14	UBS Securities Australia Ltd	Buy	614	28	Ordinary
-Dec-14	UBS Securities Australia Ltd	Buy	1,956	77.7	Ordinary
-Dec-14	UBS Securities Australia Ltd	Buy	1,659	225	Ordinary
-Dec-14	UBS Securities Australia Ltd	Sell	3 497	(967)	Ordinary
-Dec-14	UBS Securities Australia Ltd	Sell	5 294	1030	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Sell	18114	(2 573)	Ordinan
-Dec-14	UBS Securities Australia Ltd	Buv	14 120	000 c	Ordinary
-Dec-14	UBS Securities Australia 11d	Se	0901	WOLL/	tin in
-Dec-14	ILIBS Securities Australia 14d	S. C.	003/1	(6/1)	o dinialy
Dec. 14	I BC Consisting Australia 144	foo	3,530	004	Ordinary
200.00	DO SCHOOL STORY OF THE PARTY OF	1130	0,1	(145)	Ordinary
2-Dec-14	Upo seculities Australia Ltd	Ne II	1,030	(148)	Ordinary
-Dec-14	UBS Securities Australia Ltd	Sell	2,895	(416)	Ordinary
-Dec-14	UBS Securities Australia Ltd	Seli	1,322	(061)	Ordinary
-Dec-14	UBS Securities Australia Ltd	Sell	5,283	(657)	Ordinary
-Dec-14	UBS Securities Australia Ltd	Sell	9,814	(1,410)	Ordinary
-Dec-14	UBS Securities Australia Ltd	les	4,795	(689)	Ordinary
-Dec-14	UBS Securities Australia Ltd	lleS.	4,026	(8/5)	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Sell	7	0	Ordinary
-Dec-14	LIBS Securities Australia I tel	و	100	(1)	Ordinary
Dec-14	II IBS Securities Australia Ital	9	25	(07)	Cluinary
200	I IDS Consisted According to a		CC	(c)	Ordinary
1	ODS SECURIORS AUSTRALIA	i ac	5,700	(819)	Ordinary
ı	ODS DECUMES MUSICALITY	i ac	/24	(104)	Ordinary
-	UBS Securities Australia Ltd	Şeil	180	1/36/1	
12-Dec-14	2 TO 10 TO 1			(40)	Ordinary
	UBS Securities Australia Ltd	Buy	3,085	(26)	Ordinary

8 8 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9
_
Buy
Ang.
ána.
eny
1
-

		2	3,365	(472)	Ordinary
15-Dec-14	UBS Securities Australia Ltd	200	410		
-040-14	Obs Securities Australia LIG	Jen Jen	649	(16)	Ordinary
- Fee	UBS Securities Australia Ltd	nec.	200	(28)	Ordinary
-Dec-14	UBS Securities Australia Ltd	Sell	12,896	(1,810)	Ordinary
-Dec-14	UBS Securities Australia Ltd	Sell	171		Ordinary
-Dec-14	UBS Securities Australia Ltd	Buy	4,949,000		Ordinary
-Dec-14	UBS Securities Australia Ltd	Sell	1,068		Ordinary
-Dec-14	UBS Securities Australia Ltd	Sell	699	(54)	Ordinary
-Dec-14	UBS Securities Australia Ltd	Sell	406	(57)	Ordinary
-Dec-14	UBS Securities Australia Ltd	Sell	526	(98)	Ordinary
-Dec-14	UBS Securities Australia Ltd	Sell	066	(139)	Ordinary
-Dec-14	UBS Securities Australia Ltd	Sell	13,905	(1,953)	Ordinary
-Dec-14	UBS Securities Australia Ltd	Setl	1,346	(189)	Ordinary
-Dec-14	UBS Securities Australia Ltd	Self	4,578	(643)	Ordinary
-Dec-14	UBS Securities Australia Ltd	Sell	1,517	(213)	Ordinary
-Dec-14	UBS Securities Australia Ltd	Sell	36	(5)	Ordinary
-Dec-14	UBS Securities Australia Ltd	lles	478	(29)	Ordinary
-Dec-14	UBS Securities Australia Ltd	Sell	2.274	(319)	Ordinary
-Dec-14	UBS Securities Australia Ltd	Sell	1,533	(215)	Ordinary
15-Dec-14	UBS Securities Australia Ltd	Sell	1,070	(150)	Ordinary
5-Dec-14	UBS Securities Australia Ltd	Sell	200	(86)	Ordinan
5-Dec-14	UBS Securities Australia Ltd	Sell	98	(5)	Ordinan
5-Dec-14	UBS Securities Australia Ltd	Sell	5 184	(5,01)	Ordinary
5-Dec-14	UBS Securities Australia itd	lles	103	400	Couling
5-Dec-14	UBS Securities Australia Ltd	Sell	550 1	(62.5)	Cidillary
-Dec-14	LIBS Securities Australia Ltd	9	2037	(6/1)	Ordinary
5-Dec-14	UBS Securities Australia I to	<u></u>	1001	(070)	Ordinary
-Dec-14	I IRS Securities Australia I+d	Coll	100,1	(057)	Ordinary
5-Dec-14	LAS Counties Australia Ltd	100	1807	(ncl.)	Ordinary
S. Doc 14	LAD Counties Australia Lad		670,1	(977)	Ordinary
1 2 2 2	opp secondes Australia Ltd	1700	609'5	(6/2)	Ordinary
3-Dec-14	UBS Securities Australia Ltd	>eii	6,106	(848)	Ordinary
5-Dec-14	UBS Securities Australia Ltd	Sell	4,932	(982)	Ordinary
P-Dec-14	UBS Securities Australia Ltd	Sell	4,932	(982)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	6,376	(882)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	5,141	(714)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	2,255	(313)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	4 932	(685)	Ordinan
5-Dec-14	UBS Securities Australia Ltd	Sel	7,552	(003)	Criminaly
Dec-14	IJBS Securities Australia Itd	3	4,3 3	(666)	Crainary
Day 1.4	TIDE Consider American Lead	3	770'%	(970)	Ordinary
1	Opp Securities Australia LLG	llac l	5,854	(813)	Ordinary
Dec- 14	UBS Securities Australia Ltd	sell	15,844	(2,199)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	609'5	(677)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	4.723	(959)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	159	(20)	Ordinary
15-Dec-14	UBS Securities Australia Ltd	Sell	3 490	(77)	Ordinary
Dec-14	UBS Securities Australia 14d	Soil	363	(+O+)	Ordinary
5-Dec-14	IIIDS Committee Australia 14d	1120	555	(74)	Ordinary
200.14	Opp Securities Australia Little	241	2,711	(375)	Ordinary
2-Dec-14	Obsiseduriles Australia Ltd	) Sell	319	(44)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	551	(20)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	131	(18)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	355	(67)	Ordinary
5-Dec-14	UBS Securities Australia Ltd	Sell	1,581	(218)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	65	(5)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	3178	(057)	Ordinany
5-Dec-14	UBS Securities Australia Ltd	35	145	(00)	Ordinary
5-Dec-14	UBS Securities Australia Itd	3	1,674	1000	0.00
5.Dec-14			1/0'	(057)	Ordinary
1 2	Obsiderations Abstralla Ltd	- C	3,443	(473)	Ordinary
2-0-6-14	UBS Securities Australia Ltd	Sell	204	(28)	Ordinary
5-Dec-14	UBS Securities Australia Ltd	Sell	160	(22)	Ordinary
5-Dec-14	UBS Securities Australia Ltd	Sell	7/7	(9)	Ordinary
5-Dec-14	UBS Securities Australia Ltd	Sell	204	(28)	Ordinary
5-Dec-14	UBS Securities Australia Ltd	Sell	87	(12)	Ordinary
5-Dec-14	UBS Securities Australia Ltd	Sell	292	(36)	Ordinary
Dec-14	UBS Securities Australia Ltd	9,	204	(80)	Ordinary
Dec-14	LIBS Securities Australia 11d	P	116	(20)	S S S S S S S S S S S S S S S S S S S
Dac-14	Line Countries Australia 14d	100	2 6	W17	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	138	(19)	Ordinary
15-Dec-14	UBS Securities Australia Ltd	Sell	792	(36)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	792	(36)	Ordinary
5-Dec-14	UBS Securities Australia Ltd	Se#	160	(22)	Ordinary
Dec-14	UBS Securities Australia Ltd	Sell	160	(20)	Ordinary
15-Dec-14	UBS Securities Australia Itd	Sell	NOC	(27)	5 0
	טבים שוני איירי איירי איירי איירי	Je.11	- 45	1/2/1	
Doc.12			100	14-67	Ordinary
1	UBS Securities Australia Ltd	Sell	182	(25)	Ordinary

ا					
2-Dec-14	UBS Securities Australia Ltd	Sell	2,667	(1,100)	Ordinary
2-Dec-14	UBS Securities Australia Ltd	Buy	13,611	1,950	Ordinary
12-Dec-14	UBS Securities Australia Ltd	BCV	1 710	287	Ordinary
	UBS Securities Australia Ltd	Sell	2.858	(410)	Ordinary
	UBS Securities Australia Ltd	Sell	17,355	(2,490)	Ordinary
12-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	Ψ.V.	300,000,000	Ontions
12-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	d Z	3 000 000	Ontions
12-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	Ž	(000 000 002)	pations
2-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	A N	(600,000,000)	option o
4-Dec-14	UBS Securities Australia Ind	Enter Into Derivative	VIII VIII	(15,000)	e control
5-Dec-14	UBS Global Asset Management	Sell	27.772	(3.912)	Ordinary
	UBS Global Asset Management	Buy	115,436	15,895	Ordinary
5-Dec-14	UBS Securities Australia Ltd	Buy	48,504	6,949	Ordinary
	TUBS Securities Australia Ltd	Seil	20,892	(2,947)	Ordinary
П	UBS Securities Australia Ltd	Buy	174,924	24,316	Ordinary
	UBS Securities Australia Ltd	Sell	171,995	(23,792)	Ordinary
	UBS Securities Australia Ltd	Buy	1,082,090	151,479	Ordinary
1	UBS Securities Australia Ltd	Selles	1,314	(185)	Ordinary
l	UBS Securities Australia Ltd	Sell	11,330	(1,600)	Ordinary
	UBS Securities Australia Ltd	Buy	188,037	26,000	Ordinary
	UBS Securities Australia Ltd	Seil	180,735	(25,422)	Ordinary
	UBS Securities Australia Ltd	Suy	431,817	62,251	Ordinary
	UBS AG. London Branch	Sel	732 552	(31,842)	Ordinary
Ш	UBS AG, London Branch	Buy	284,630	32,611	Ordinary
	UBS AG Australia Branch	Sell	16,212	(2,328)	Ordinary
15-Dec-14	UBS AG Australia Branch	Buy	29,075	4,000	Ordinary
	UBS Securities Australia Ltd	les.	713	(100)	Ordinary
5-Dec-14	UBS Securities Australia Ltd	Sell	1,390	(195)	Ordinary
	UBS Securities Australia Ltd	Sell	2,335	(328)	Ordinary
	UBS Securities Australia Ltd	Sell	1,832	(257)	Ordinary
5-Dec-14	UBS Securities Australia 11d	Sell	1 196	(390)	Ordinary
	UBS Securities Australia Ltd	Seli	1,680	(236)	Ordinary
	UBS Securities Australia Ltd	Sell	1,104	(155)	Ordinary
	UBS Securities Australia Ltd	Sell	648	(91)	Ordinary
5-Dec-14	UBS Securities Australia Itd	Sell Sell	1 047	(104)	Ordinary
5-Dec-14	UBS Securities Australia Ltd	Sell	1,032	(145)	Ordinary
5-Dec-14	UBS Securities Australia Ltd	Sell	1,737	(244)	Ordinary
5-Dec-14	UBS Securities Australia Ltd	Sell	1,069	(150)	Ordinary
	UBS Securities Australia Ltd	Seil	370	(52)	Ordinary
	UBS Securities Australia Ltd	Sell	4,600	(646)	Ordinary
	UBS Securities Australia Ltd	las las	2895	(791)	Ordinary
5-Dec-14	UBS Securities Australia Ltd	Sell	3,532	(496)	Ordinary
	UBS Securities Australia Ltd	Sell	20,271	(2,845)	Ordinary
5-Dec-14	UBS Securities Australia Ltd	Sell	10,688	(1,500)	Ordinary
	UBS Securities Australia Ltd	Sell	3,541	(497)	Ordinary
5-Dec-14	UBS Securities Australia Ltd	Jan Jan	905,0	(886)	Ordinary
l	UBS Securities Australia 11d	lies.	G 83	(217)	Ordinary
	UBS Securities Australia Ltd	Sell	2 293	(322)	Ordinary
	UBS Securities Australia Ltd	Sell	1,069	(150)	Ordinary
	UBS Securities Australia Ltd	Sell	786	(138)	Ordinary
15-Dec-14	UBS Securities Australia Ltd	Sell	2,930	(411)	Ordinary
l	UBS Securities Australia Ltd	Sell	2,467	(346)	Ordinary
	UBS Securities Australia Ltd	Sell	7,465	(1,047)	Ordinary
1	UBS Securities Australia Ltd	Seil	984	(138)	Ordinary
	UBS Securities Australia Ltd	<u></u>	10.688	(1 500)	Ordinary
15-Dec-14	UBS Securities Australia Ltd	les	7 175	(967)	Ordinary
l			67,17	(0.7)	
	II IBC Convertion Australia I to	<u></u>	1241	(474)	Ordinari

Dec. 14   Bit Securine Authority 10   Byta   1949		The second secon	iec	22,210	1/207/21	
Control Statistical Location (Control Series)         9.99         9.90         9.	15-Dec-14	UBS Securities Australia Ltd	Buy	3,376	490	Ordinary
Securine Australia Ltd	13-Dec-14	UBS Securities Australia Ltd	Buy	086	142	Ordinary
Dis Scorring Australia LEG   849   257   258	13-Dec-14	UBS Securities Australia Ltd	Buy	3,422	496	Ordinary
December Auchie   1949   2,020   2,020   2,030   2,0	15-Dec-14	UBS Securities Australia Ltd	Buy	55		Ordinary
(b) S SCOTIME AURTING INC.         SM         5.20         7.88           (b) S SCOTIME AURTING INC.         SMITH AURTING INC.         SMITH AURTING INC.         (106)         (106)         (106)         (107)         (108)           (b) S SCOTIME AURTING INC.         SMITH AURTING INC.         SMITH AURTING INC.         SMITH AURTING INC.         (107)         (108)         (107)         (108)           (b) S SCOTIME AUGUST INC.         SMITH AURTING INC.         SMITH AURTING INC.         SMITH AURTING INC.         SMITH AURTING INC.         SMITH AURTING INC.         (107)         (107)         (108)           (b) S SCOTIME AUGUST INC.         SMITH AURTING INC. <td>15-Dec-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>2,022</td> <td></td> <td>Ordinary</td>	15-Dec-14	UBS Securities Australia Ltd	Buy	2,022		Ordinary
Bit Securines Australia ILM         Self         1,17         (106)           Bit Securines Australia ILM         Self         1,17         (108)           Bit Securines Australia ILM         Self         1,2035         (1288)           Bit Securines Australia ILM         Self         1,17,705         (1572)           Bit Securines Australia ILM         Self         1,17,705         (1573)           Bit Securines Australia ILM         Self         1,17,705         (1573)           Bit Securines Australia ILM         Self         1,17,705         (1573)           Bit Securines Australia ILM         Self         1,17,705         (1573)           Bit Securines Australia ILM         Self         1,17,705         (1573)           Bit Securines Australia ILM         Self         1,17,705         (15,202)           Bit Securines Australia ILM         Self         1,17,705         (15,202)           Bit Securines Australia ILM         Self         1,17,705         (15,202)           Bit Securines Australia ILM         Self         1,17,705         (15,202)           Bit Securines Australia ILM         Self         1,17,705         (15,202)           Bit Securines Australia ILM         Self         1,17,705         (15,202)	15-Dec-14	UBS Securities Australia Ltd	Buy	5,230		Ordinary
USS Securine Australia tet de   Se1   1,10	15-Dec-14	UBS Securities Australia Ltd	Sell	7		Ordinary
USS securine Autorials (14   241   1,275   1,088   1,088   1,085   1,000   1	15-Dec-14	UBS Securities Australia Ltd	Sell	1,167	(166)	Ordinary
LOS Securine Australia Led         SP61         7,05,000         (10,000)           LUSS Securines Australia Led         SP61         7,05,000         (10,000)           LUSS Securines Australia Led         SP61         7,05,000         (10,000)           LUSS Securines Australia Led         SP61         7,55,000         (10,000)           LUSS Securines Australia Led         SP61         7,55,000         (10,000)           LUSS Securines Australia Led         SP61         7,55,000         (10,000)           LUSS Securines Australia Led         SP61         7,500         (10,000)           LUSS Securines Australia Led         SP61         7,500         (10,000)           LUSS Securines Australia Led         SP61         7,500         (10,000)           LUSS Securines Australia Led         SP61         7,600         (10,000)           LUSS Securines Australia Led         SP61         8,900         7,600         (10,000)           LUSS Securines Australia Led         SP61         8,900         7,600         (10,000)           LUSS Securines Australia Led         SP61         8,900         7,100         (10,000)           LUSS Securines Australia Led         SP61         SP61         7,100         (10,000)           LUSS Securin	15-Dec-14	UBS Securities Australia Ltd	Sell	2,025	(288)	Ordinary
USS Securine Auditalia to 14   24   716,000 (100,000   1015 Securine Auditalia to 14   24   716,000 (100,000   1015 Securine Auditalia to 14   24   717,775 (1015) (1015	15-Dec-14	UBS Securities Australia Ltd	Sell	1,322		Ordinary
Loss countries Australia Led         Skal         1,108         (1,57)         (2,513)           Loss countries Australia Led         Skal         1,74,83         (1,538)           Loss countries Australia Led         Skal         1,53,84         (2,513)           Loss countries Australia Led         Skal         1,53,84         (2,513)           Loss countries Australia Led         Skal         1,43,564         (2,00,004)           Loss countries Australia Led         Skal         1,43,564         (2,00,004)           Loss countries Australia Led         Skal         1,43,564         (2,00,004)           Loss countries Australia Led         Skal         1,43,564         (2,00,004)           Loss countries Australia Led         Skal         1,53,720         (2,53,30)           Loss countries Australia Led         Skal         1,52,200         (5,50,004)           Loss countries Australia Led         Skal         2,14,63,801         (7,500)           Loss countries Australia Led         Skal         2,14,63,801         (7,500)           Loss countries Australia Led         Skal         2,14,63,801         (7,500)           Loss countries Australia Led         Skal         2,14,63,801         (7,500)           Los Countries Australia Led <td< td=""><td>13-Dec-14</td><td>UBS Securities Australia Ltd</td><td>Seff</td><td>706,000</td><td></td><td>Ordinary</td></td<>	13-Dec-14	UBS Securities Australia Ltd	Seff	706,000		Ordinary
USS Securine Australia ted   Sail   17,775   (2,516)   (1,52)	16-Dec-14	UBS Global Asset Management	Sell	1,108		Ordinary
URS Scorrings Automate Ind         Seld         1,9,9,9         1,9,9,9         1,9,9,9         1,9,9,9         1,9,9,9         1,0,9,9	16-Dec-14	UBS Securities Australia Ltd	Sell	17,775		Ordinary
UBS Scourines Autorinia Ltd         Bay         1,554         8,00           UBS Scourines Autorinia Ltd         Bay         140,554         7,00           UBS Scourines Autorinia Ltd         Bay         1,40,547         1,00           UBS Scourines Autorinia Ltd         Bay         1,40,547         1,00           UBS Scourines Autorinia Ltd         Bay         1,40,547         1,00           UBS Scourines Autorinia Ltd         Bay         1,40,547         1,00           UBS Scourines Autorinia Ltd         Bay         2,140,537         1,00           UBS Scourines Autorinia Ltd         Bay         2,140,338         0,0           UBS Scourines Autorinia Ltd         Bay         2,140,338         0,0           UBS Scourines Autorinia Ltd         Bay         2,140,338         0,0           UBS Scourines Autorinia Ltd         Bay         2,140,338         0,0           UBS Scourines Autorinia Ltd         Bay         2,140,338         0,0           UBS Scourines Autorinia Ltd         Bay         2,140,338         0,0           UBS Scourines Autorinia Ltd         Bay         2,140,338         0,0           UBS Scourines Autorinia Ltd         Bay         2,140,338         0,0           UBS Scourines Autorinia Ltd <td>16-Dec-14</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>34,989</td> <td></td> <td>Ordinary</td>	16-Dec-14	UBS Securities Australia Ltd	Sell	34,989		Ordinary
(b)         (b) <td>16-Dec-14</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>5,964</td> <td></td> <td>Ordinary</td>	16-Dec-14	UBS Securities Australia Ltd	Buy	5,964		Ordinary
URS Scorrings Autorials Ltd         Sivil         14.3.54.         (20.00.4)           URS Scorrings Autorials Ltd         Sivil         17.68.720         (23.28.4)           USS Scorrings Autorials Ltd         Sivil         2.46.527         (23.28.4)           USS Scorrings Autorials Ltd         Sivil         2.46.527         (23.28.4)           USS Scorrings Autorials Ltd         Sivil         2.46.527         (23.28.4)           USS Scorrings Autorials Ltd         Buy         2.46.527         (23.28.4)           USS Scorrings Autorials Ltd         Buy         2.16.20.20         (5.00.0)           USS Scorrings Autorials Ltd         Buy         2.16.20.20         (5.00.0)           USS Scorrings Autorials Ltd         Buy         2.16.20.20         (5.00.0)           USS Scorrings Autorials Ltd         Buy         2.16.20.20         (5.00.0)           USS Scorrings Autorials Ltd         Buy         2.16.20.20         (5.00.0)           USS Scorrings Autorials Ltd         Buy         7.00.00         (1.00.00           USS Scorrings Autorials Ltd         Buy         7.00.00         (1.00.00           USS Scorrings Autorials Ltd         Buy         7.00.00         (1.00.00           USS Scorrings Autorials Ltd         Buy         7.00.00	16-Dec-14	UBS Securities Australia Ltd	Buy	143,584		Ordinary
UBS Scourines Auteriale Ltd         Bay         158,202         C33,844           UBS Scourines Auteriale Ltd         Self         1,657,709         C33,844           UBS Scourines Auteriale Ltd         Self         5,502         C6,000           UBS Scourines Auteriale Ltd         Self         5,502         C6,000           UBS Scourines Auteriale Ltd         Self         5,502         C6,000           UBS Scourines Auteriale Ltd         Boby         2,517,392         C1,000           UBS Scourines Auteriale Ltd         Boby         2,517,392         C1,000           UBS Scourines Auteriale Bunch         Boby         2,517,392         C1,000           UBS Scourines Auteriale Bunch         Self         2,517,392         C1,000           UBS Scourines Auteriale Bunch         Self         1,507,719         C1,000           UBS Scourines Auteriale Bunch         Self         1,507,719         C1,000           UBS Scourines Auteriale Bunch         Self         1,507,719         C1,000           UBS Scourines Auteriale Bunch         Self         1,507,70         C1,000           UBS Scourines Auteriale Ltd         Buy         7,709         C1,000           UBS Scourines Auteriale Ltd         Buy         7,700         C1,000	16-Dec-14	UBS Securities Australia Ltd	Sell	141,364		Ordinary
URS Securities Australia Loid         Sell         1,657,720         (23,540)           URS Securities Australia Loid         Bay         4,956         (23,620)           URS Securities Australia Loid         Sell         35,520         (5,000)           URS Securities Australia Loid         Bay         475,223         (5,200)           URS Securities Australia Loid         Bay         2,023,235         (5,200)           URS Securities Australia Loid         Bay         2,023,235         (5,200)           URS Securities Australia Loid         Bay         2,023,235         (5,200)           URS ACAL LORGIN Branch         Sell         2,023,235         (5,200)           URS ACAL LORGIN Branch         Sell         2,023,235         (5,200)           URS ACAL LORGIN Branch         Sell         1,023,000         (5,200)           URS ACAL LORGIN Branch         Sell         1,023,000         (5,200)           URS ACAL LORGIN Branch         Sell         1,020,000         (7,000)         (7,000)           URS ACAL LORGIN Branch         Sell         1,000         (7,000)         (7,000)         (7,000)           URS ACAL LORGIN Branch         Sell         1,000         (7,000)         (7,000)         (7,000)           URS ACA	16-Dec-14	UBS Securities Australia Ltd	Buy	124,837		Ordinary
URS Securities Australia Ltd         Sell         2,146,381         G04,204           URS Securities Australia Ltd         Sell         35,500         C5,000           URS Securities Australia Ltd         Sell         231,703         C1,2,560           URS Securities Australia Ltd         Bay         2,145,324         G0,500           URS Securities Australia Ltd         Bay         2,243         G0,500           URS Securities Australia Ltd         Bay         1,527,719         C1,2560           URS Securities Australia Ltd         Bay         1,627,719         C1,5260           URS Actionation Banch         Sell         1,627,719         C1,5260           URS Actionate Australia Ltd         Bay         1,627,719         C1,5260           URS Actionate Australia Ltd         Bay         7,000         C1,5260           URS Securities Australia Ltd         Bay         7,000         C1,5260           URS Securities Australia Ltd         Bay         7,000         C1,527           URS Securities Australia Ltd         Bay         7,000         C1,526           URS Securities Australia Ltd         Bay         7,000         C1,527           URS Securities Australia Ltd         Bay         7,000         C1,527           <	16-Dec-14	UBS Securities Australia Ltd	Seli	1,657,720		Ordinary
UBS Securities Australia (Led         Bial         35,509         6,100           UBS Securities Australia (Led         Sial         35,509         6,000           UBS Securities Australia (Led         Sial         22,179         (5,509)           UBS Securities Australia (Led         Buy         2,522,352         5,309           UBS Securities Australia (Led         Buy         2,522,352         3,045,243           UBS Securities Australia (Led         Buy         2,522,352         3,045,243           UBS Securities Australia (Led         Buy         2,522,352         3,045,243           UBS AGE, London Barrch         Sell         0,05,260         (6,500)           UBS AGE, London Barrch         Sell         0,05,260         (6,500)           UBS AGE, London Barrch         Sell         0,05,260         (6,500)           UBS AGE, London Barrch         Sell         0,05,260         (6,500)           UBS AGE, London Barrch         Sell         0,05,260         (6,500)           UBS AGE, London Barrch         Sell         0,05,260         (6,500)           UBS AGE, London Barrch         Sell         0,05,260         (6,500)           UBS AGE, London Barrch         Sell         0,05,260         (6,500)	16-Dec-14	UBS Securities Australia Ltd	Sell	2,146,381		Ordinary
UBS Securities Australia (Log         Sell         255,00           UBS Securities Australia (Log         Sell         251,70         C52,60           UBS Securities Australia (Log         Sell         221,00         C12,590           UBS Securities Australia (Log         Sell         2,02,23         C12,590           UBS AGE Location Branch         Boy         2,02,23         C12,390           UBS AGE Location Branch         Sell         1,02,23         C12,390           UBS AGE Location Branch         Sell         1,02,23         C12,390           UBS AGE Location Branch         Sell         1,02,23         C12,390           UBS AGE Location Branch         Sell         1,02,23         C12,390           UBS AGE Location Branch         Sell         1,02,23         C12,390           UBS AGE Location Branch         Sell         1,02,20         C12,390           UBS AGE Location Branch         Sell         1,02,20         C12,20           UBS AGE Location Branch         Sell         1,02,20         C12,20           UBS AGE Location Branch         Sell         1,02         C1,02           UBS AGE Location Branch         Sell         1,02         C1,03           UBS AGE Location Branch         Sell <t< td=""><td>16-Dec-14</td><td>UBS Securities Australia Ltd</td><td>Buy</td><td>43,549</td><td></td><td>Ordinary</td></t<>	16-Dec-14	UBS Securities Australia Ltd	Buy	43,549		Ordinary
URS Securities Autorials Ltd         Sell         25.17.93         (35.609)           URS Securities Autorials Ltd         Buy         2.523.32         (3.509)           URS ACC LONDON Branch         Sell         2.102.22         (3.539)           URS ACC LONDON Branch         Sell         2.106.299         (3.523.32         (3.523.32)           URS ACC LONDON Branch         Sell         2.233.92         (3.533.82)         (3.523.32)           URS ACC LONDON Branch         Sell         2.106.292         (3.523.32)	16-Dec-14	UBS Securities Australia Ltd	Sell	35,502		Ordinary
UBS Securities Autorials Ltd.         Sell         92,702         (12,959)           UBS Securities Autorials Ltd.         Buy         475,047         67,042           UBS AGE, Lordoor Barrch         Buy         2,033,022         200,624           UBS AGE, Lordoor Barrch         Buy         2,033,022         200,624           UBS AGE, Lordoor Barrch         Buy         2,033,022         200,624           UBS AGE, Lordoor Barrch         Sell         7,032         2,033,022         200,624           UBS AGE, Lordoor Barrch         Sell         7,032         2,033,022         2,034,022         2,034,022         2,034,022         2,034,022         2,034,022         2,034,022         2,034,022         2,034,022         2,034,022         2,034,022         2,034,022         2,034,022         2,034,022         2,034,022         2,034,022	16-Dec-14	UBS Securities Australia Ltd	Sell	251,791		Ordinary
UBS Securities Australia Ltd         Bluy         2,823,282         20,329           UBS AG, London Barrich         Stell         2,146,399         C0A,624           UBS AG, London Barrich         Stell         2,146,399         C0A,624           UBS AG, London Barrich         Stell         2,146,399         C0A,624           UBS AG, London Barrich         Stell         2,146,399         C0A,624           UBS AG, London Barrich         Stell         2,146,399         C0A,280         (BLS)           UBS AG, London Barrich         Stell         C0A,280         (BLS)         C0A,280         (BLS)           UBS AG, London Barrich         Stell         RS         C0A,280         (BLS)         C0A,280         (BLS)           UBS AG, London Barrich         Stell         Stell         Stell         C0A,280         (BLS)           UBS AG, London Barrich         Stell         Stell         Stell         C0A,000         (BLS)         C0A,000         UBS,240         C0A,000         UBS,240         C0A,000         UBS,240         C0A,000         UBS,240         C0A,000         UBS,240         C0A,000         UBS,240         C0A,000         UBS,240         C0A,000         UBS,240         C0A,000         UBS,240         C0A,000         UBS,240<	16-Dec-14	UBS Securities Australia Ltd	Sell	92,102		Ordinary
UBS AGE, United patrich         Bild         2,1463.99         2,043.922         390,424           UBS AGE, United patrich         Bay         2,1463.99         (33.244)         (33.2	16-Dec-14	UBS Securities Australia Ltd	Buy	475,047		Ordinary
URS AGE (London Branch)         Self         2.033.062         233.0624           URS AGE (London Branch)         Boyl         1.085.7719         2.033.062         233.062           URS AGE (London Branch)         Self         1.085.7719         (203.844)           URS AGE Australia Branch         Self         1.085.7719         (203.844)           URS AGE Australia Branch         Self         1.087.7719         (203.844)           URS AGE Australia Branch         Self         1.087.7719         (205.00)           URS Weelth Management Australia Ltd         Buy         7.00         (117.429)           URS Weelth Management Australia Ltd         Buy         7.00         1.00           URS Weelth Management Australia Ltd         Buy         7.00         1.00           URS Securities Australia Ltd         Buy         7.00         1.00           URS Securities Australia Ltd         Buy         2.243         3.15           URS Securities Australia Ltd         Buy         2.743         4.56           URS Securities Australia Ltd         Buy         2.743         4.56           URS Securities Australia Ltd         Buy         2.743         4.56           URS Securities Australia Ltd         Buy         2.743         4.59 </td <td>16-Dec-14</td> <td>UBS AG, London Branch</td> <td>Buy</td> <td>2,632,352</td> <td></td> <td>Ordinary</td>	16-Dec-14	UBS AG, London Branch	Buy	2,632,352		Ordinary
URS AGE, incoden Rearch         Bluy         2,033,062         233,844           UBS AGE, incoden Rearch         Sell         2,033,062         (117,459)           UBS AGE, incoden Rearch         Sell         (117,459)         (117,459)           UBS AGE, incoden Rearch         Sell         (117,459)         (117,459)           UBS AGE, incoden Rearch         Sell         (117,459)         (117,459)           UBS AGE, incoden Rearch Assistals Ltd         Sell         (117,459)         (117,459)           UBS Weelth Management Australia Ltd         Buy         70,000         (107,000)           UBS Weelth Management Australia Ltd         Buy         70,000         (10,000)           UBS Securities Australia Ltd         Buy         70,000         (10,000)           UBS Securities Australia Ltd         Buy         70,000         (10,000)           UBS Securities Australia Ltd         Buy         70,000         (10,000)           UBS Securities Australia Ltd         Buy         70,000         (10,000)           UBS Securities Australia Ltd         Buy         70,000         (10,000)           UBS Securities Australia Ltd         Buy         70,000         (10,000)           UBS Securities Australia Ltd         Buy         70,000         (	16-Dec-14	UBS AG, London Branch	Self	2,146,369		Ordinary
USS AGO Australia Branch         Soll         1,656,719         C233,844           USS AGO Australia Branch         Sell         1,656,719         (6,510)           USS AGO Australia Branch         Sell         1,652         (6,510)           USS AGO Australia Branch         Sell         1,652         (6,510)           USS AGO Australia Branch         Sell         1,652         (6,510)           USS AGO Australia Branch         Sell         1,830         (1,11,142)           USS Weelth Management Australia Ltd         Buy         2,242         3,542           USS Weelth Management Australia Ltd         Buy         2,243         3,55           USS Securities Australia Ltd         Buy         2,243         3,15           USS Securities Australia Ltd         Buy         2,243         3,15           USS Securities Australia Ltd         Buy         2,243         3,15           USS Securities Australia Ltd         Buy         2,150         7,35           USS Securities Australia Ltd         Buy         2,150         7,57           USS Securities Australia Ltd         Buy         2,150         7,57           USS Securities Australia Ltd         Buy         2,150         7,57           USS Securities Australia Ltd <td>16-Dec-14</td> <td>UBS AG, London Branch</td> <td>Buy</td> <td>2,033,062</td> <td></td> <td>Ordinary</td>	16-Dec-14	UBS AG, London Branch	Buy	2,033,062		Ordinary
UBS Activities Martial is Branch         Sell         66,02,200           UBS Activities Branch         Sell         66,02,200           UBS Activities Branch         Sell         66,02           UBS Activities Branch         Sell         13,206           UBS Activities Branch         Sell         13,206           UBS Activities Australia Ited         Buy         1,059           UBS Securities Australia Ited         Buy         2,243         315           UBS Securities Australia Ited         Buy         2,243         315           UBS Securities Australia Ited         Buy         2,243         315           UBS Securities Australia Ited         Buy         2,243         315           UBS Securities Australia Ited         Buy         2,243         315           UBS Securities Australia Ited         Buy         2,243         315           UBS Securities Australia Ited         Buy         2,243         315           UBS Securities Australia Ited         Buy         2,133         456           UBS Securities Australia Ited         Buy         2,133         456           UBS Securities Australia Ited         Buy         2,133         456           UBS Securities Australia Ited         Buy <td< td=""><td>16-Dec-14</td><td>UBS AG, London Branch</td><td>sell</td><td>1,657,719</td><td></td><td>Ordinary</td></td<>	16-Dec-14	UBS AG, London Branch	sell	1,657,719		Ordinary
UBS Securities Marchine Branch         Sell         786,488         (111,745)           UBS AG Australia Branch         Sell         786,621         (6,616)           UBS Weethin Management Australia Ltd         Sell         13,304         (1,800)           UBS Weethin Management Australia Ltd         Buy         70,000         (7,604)           UBS Securities Australia Ltd         Buy         70,000         (7,604)           UBS Securities Australia Ltd         Buy         70,000         (7,604)           UBS Securities Australia Ltd         Buy         2,243         315           UBS Securities Australia Ltd         Buy         2,243         315           UBS Securities Australia Ltd         Buy         2,243         315           UBS Securities Australia Ltd         Buy         2,243         315           UBS Securities Australia Ltd         Buy         2,133         (7,356)           UBS Securities Australia Ltd         Buy         2,193         (7,356)           UBS Securities Australia Ltd         Buy         1,10,091         (1,571)           UBS Securities Australia Ltd         Buy         1,20,404         2,523           UBS Securities Australia Ltd         Buy         1,20         2,574           UB	16-Dec-14	UBS AG Australia Branch	Sell	609,280		Ordinary
UBS Securities Branch         Sell         66.61           UBS Meath Management Australia Ltd         Buy         19.306         (1.800)           UBS Weath Management Australia Ltd         Buy         19.304         (1.800)           UBS Securities Australia Ltd         Buy         700,000         (0.604)           UBS Securities Australia Ltd         Buy         700,000         (0.605)           UBS Securities Australia Ltd         Buy         700,000         (0.605)           UBS Securities Australia Ltd         Buy         700,000         (0.605)           UBS Securities Australia Ltd         Buy         700,000         (0.600)           UBS Securities Australia Ltd         Buy         3,552         4,600           UBS Securities Australia Ltd         Buy         3,552         4,600           UBS Securities Australia Ltd         Buy         3,552         4,600           UBS Securities Australia Ltd         Buy         1,1001         (1,511)           UBS Securities Australia Ltd         Buy         1,76,41         2,540         4,550           UBS Securities Australia Ltd         Buy         1,76,41         2,514         4,650           UBS Securities Australia Ltd         Buy         1,76,42         2,524	- 1	UBS AG Australia Branch	Sell	788,488		Ordinary
UBS Securities Australia Ltd         Sell         113.30G         (18.90)           UBS Weath Management Australia Ltd         Buy         19.30d         2.74d           UBS Securities Australia Ltd         Buy         70,000         2.74d           UBS Securities Australia Ltd         Buy         70,000         10,000           UBS Securities Australia Ltd         Buy         7,000         1,000           UBS Securities Australia Ltd         Buy         2,243         315           UBS Securities Australia Ltd         Buy         2,243         315           UBS Securities Australia Ltd         Buy         2,243         315           UBS Securities Australia Ltd         Buy         2,243         315           UBS Securities Australia Ltd         Buy         3,275         465           UBS Securities Australia Ltd         Buy         3,275         465           UBS Securities Australia Ltd         Buy         11,091         (1,571)           UBS Securities Australia Ltd         Buy         12,000         1,561           UBS Securities Australia Ltd         Buy         1,562         2,733           UBS Securities Australia Ltd         Buy         1,562         2,733           UBS Securities Australia Ltd		UBS AG Australia Branch	Sell	46,621		Ordinary
UBS Securities Australia Ltd         BUW         19.3 904         2.74.2           UBS Weath Management Australia Ltd         Buy         700,000         10,000           UBS Securities Australia Ltd         Buy         702,103         10,000           UBS Securities Australia Ltd         Buy         5.243         315           UBS Securities Australia Ltd         Buy         5.243         315           UBS Securities Australia Ltd         Buy         2.243         315           UBS Securities Australia Ltd         Buy         2.243         315           UBS Securities Australia Ltd         Buy         2.243         315           UBS Securities Australia Ltd         Buy         2.1933         0.7350           UBS Securities Australia Ltd         Buy         2.1933         0.7350           UBS Securities Australia Ltd         Buy         3.240         2.513           UBS Securities Australia Ltd         Buy         3.244         4.566           UBS Securities Australia Ltd         Buy         1.75,861         2.249           UBS Securities Australia Ltd         Buy         1.75,43         2.249           UBS Securities Australia Ltd         Buy         1.75,40         2.75           UBS Securities Australia Lt		UBS Wealth Management Australia Ltd	Sell	13,306		Ordinary
UBS Securities Australia Ltd         Sell         18.966         (2.694)           UBS Securities Australia Ltd         Buy         77,000         10,000           UBS Securities Australia Ltd         Buy         2,243         38.9           UBS Securities Australia Ltd         Buy         2,243         315           UBS Securities Australia Ltd         Buy         2,243         315           UBS Securities Australia Ltd         Buy         2,243         315           UBS Securities Australia Ltd         Buy         2,243         315           UBS Securities Australia Ltd         Buy         2,133         4,650           UBS Securities Australia Ltd         Buy         32,440         2,733           UBS Securities Australia Ltd         Buy         32,440         4,550           UBS Securities Australia Ltd         Buy         32,440         4,550           UBS Securities Australia Ltd         Buy         32,440         4,550           UBS Securities Australia Ltd         Buy         32,440         4,550           UBS Securities Australia Ltd         Buy         32,440         1,50           UBS Securities Australia Ltd         Buy         3,418         4,89           UBS Securities Australia Ltd	- 1	UBS Wealth Management Australia Ltd	Buy	19,304		Ordinary
UBS Securities Australia Let         Buy         700,000         1,000 00           UBS Securities Australia Let         Buy         7,709         1,085           UBS Securities Australia Let         Buy         6,323         890           UBS Securities Australia Let         Buy         2,243         315           UBS Securities Australia Let         Buy         2,243         315           UBS Securities Australia Let         Self         3,775         4,66           UBS Securities Australia Let         Buy         3,775         4,66           UBS Securities Australia Let         Buy         2,11,800         30,000           UBS Securities Australia Let         Buy         32,440         2,515           UBS Securities Australia Let         Buy         32,440         4,556           UBS Securities Australia Let         Buy         17,6,861         2,61           UBS Securities Australia Let         Buy         176,861         2,61           UBS Securities Australia Let         Buy         176,633         2,63           UBS Securities Australia Let         Buy         1,76,743         2,68           UBS Securities Australia Let         Buy         2,517         3,69           UBS Securities Australia Let		UBS Wealth Management Australia Ltd	Seli	18,966		Ordinary
UBS Securities Australia Ltd         Buy         7,799         1,085           UBS Securities Australia Ltd         Buy         7,793         1,085           UBS Securities Australia Ltd         Buy         2,243         315           UBS Securities Australia Ltd         Buy         3,522         4,65           UBS Securities Australia Ltd         Buy         3,75         4,65           UBS Securities Australia Ltd         Buy         2,1,933         7,735           UBS Securities Australia Ltd         Buy         2,1,933         7,735           UBS Securities Australia Ltd         Buy         3,2,40         4,559           UBS Securities Australia Ltd         Buy         17,686         2,493         7,700           UBS Securities Australia Ltd         Buy         7,768         2,493         7,700           UBS Securities Australia Ltd         Buy         7,768         2,493         7,701           UBS Securities Australia Ltd         Buy         7,767         1,763         2,243           UBS Securities Australia Ltd         Buy         7,767         1,767         2,249           UBS Securities Australia Ltd         Buy         7,767         1,767         2,249           UBS Securities Australia Ltd		UBS Securities Australia Ltd	Buy	700,000		Ordinary
UBS Securities Australia trd         Buy         6.323         890           UBS Securities Australia trd         Buy         2,443         315           UBS Securities Australia trd         Buy         3,275         4,66           UBS Securities Australia trd         Buy         3,752         4,60           UBS Securities Australia trd         Buy         3,752         4,60           UBS Securities Australia trd         Buy         2,118,00         30,00           UBS Securities Australia trd         Buy         8,22         1,13           UBS Securities Australia trd         Buy         1,12,601         1,13           UBS Securities Australia trd         Buy         1,12,602         1,13           UBS Securities Australia trd         Buy         473,767         7,613           UBS Securities Australia trd         Buy         473,767         7,613           UBS Securities Australia trd         Buy         4,73,767         7,613           UBS Securities Australia trd         Buy         2,572         3,69           UBS Securities Australia trd         Buy         4,73,767         7,79           UBS Securities Australia trd         Buy         2,572         3,69           UBS Securities Australia trd		UBS Securities Australia Ltd	Buy	2,709		Ordinary
UBS Securities Australia tetd         Buy         2,243         315           UBS Securities Australia tetd         Buy         2,243         315           UBS Securities Australia tetd         Buy         3,532         4,660           UBS Securities Australia tetd         Sell         11,091         (1,571)           UBS Securities Australia tetd         Buy         21,130         (7,356)           UBS Securities Australia tetd         Buy         21,130         (7,356)           UBS Securities Australia tetd         Buy         403         20,000           UBS Securities Australia tetd         Buy         403         20,000           UBS Securities Australia tetd         Buy         403         20,000           UBS Securities Australia tetd         Buy         40,300         7,070           UBS Securities Australia tetd         Buy         1,76,861         24,910           UBS Securities Australia tetd         Buy         1,76,743         26,34           UBS Securities Australia tetd         Buy         1,54,74         26,34           UBS Securities Australia tetd         Buy         1,54,7         26,34           UBS Securities Australia tetd         Buy         2,572         3,41         4,68		UBS Securities Australia Ltd	Buy	6,323		Ordinary
UBS Securities Australia Ltd         Buy         2,243         315           UBS Securities Australia Ltd         Buy         3,275         4,66           UBS Securities Australia Ltd         SMI         3,275         4,60           UBS Securities Australia Ltd         Buy         3,175         4,60           UBS Securities Australia Ltd         Buy         211,800         30,000           UBS Securities Australia Ltd         Buy         19,404         2,733           UBS Securities Australia Ltd         Buy         19,404         2,733           UBS Securities Australia Ltd         Buy         1,76,861         2,4910           UBS Securities Australia Ltd         Buy         1,76,861         2,4910           UBS Securities Australia Ltd         Buy         1,76,743         2,593           UBS Securities Australia Ltd         Buy         47,25         2,693           UBS Securities Australia Ltd         Buy         1,76,743         2,693           UBS Securities Australia Ltd         Buy         2,573         3,61           UBS Securities Australia Ltd         Buy         2,573         4,89           UBS Securities Australia Ltd         Buy         2,51         2,53           UBS Securities Australia Ltd		UBS Securities Australia Ltd	Buy	2,243		Ordinary
UBS Securities Australia Ltd         Bby         35.22         4.65           UBS Securities Australia Ltd         By         35.22         4.60           UBS Securities Australia Ltd         Buy         211,800         30.00           UBS Securities Australia Ltd         Buy         8.21         1.13           UBS Securities Australia Ltd         Buy         8.22         1.13           UBS Securities Australia Ltd         Buy         1.50,60         1.73           UBS Securities Australia Ltd         Buy         1.76,661         2.45,00           UBS Securities Australia Ltd         Buy         1.76,661         2.45,00           UBS Securities Australia Ltd         Buy         4.72,767         67,681           UBS Securities Australia Ltd         Buy         4.72,767         67,681           UBS Securities Australia Ltd         Buy         2.572         3.68           UBS Securities Australia Ltd         Buy         2.572         3.68           UBS Securities Australia Ltd         Buy         2.572         3.68           UBS Securities Australia Ltd         Buy         2.572         3.69           UBS Securities Australia Ltd         Buy         2.572         3.69           UBS Securities Australia Ltd	-	UBS Securities Australia Ltd	Buy	2,243		Ordinary
UBS Securities Australia Ltd         Buy         3.75         4.60           UBS Securities Australia Ltd         Buy         51,933         (1,531)           UBS Securities Australia Ltd         Buy         211,830         (7,386)           UBS Securities Australia Ltd         Buy         21,832         (7,386)           UBS Securities Australia Ltd         Buy         19,404         2,733           UBS Securities Australia Ltd         Buy         176,861         24,503           UBS Securities Australia Ltd         Buy         175,861         24,503           UBS Securities Australia Ltd         Buy         175,861         24,503           UBS Securities Australia Ltd         Buy         175,662         24,910           UBS Securities Australia Ltd         Buy         175,432         25,249           UBS Securities Australia Ltd         Buy         1,75,432         25,249           UBS Securities Australia Ltd         Buy         1,75,43         25,349           UBS Securities Australia Ltd         Buy         1,75,43         25,349           UBS Securities Australia Ltd         Buy         2,572         1,700           UBS Securities Australia Ltd         Buy         2,572         24,18           UBS Secur	1	UBS Securities Australia Ltd	Buy	3,532		Ordinary
UBS Securities Australia Ltd         Self         11.091         (1.571)           UBS Securities Australia Ltd         Buy         51,933         (7,350)           UBS Securities Australia Ltd         Buy         271,800         30,000           UBS Securities Australia Ltd         Buy         802         113           UBS Securities Australia Ltd         Buy         72,440         4,569           UBS Securities Australia Ltd         Buy         72,440         4,569           UBS Securities Australia Ltd         Buy         72,742         4,569           UBS Securities Australia Ltd         Buy         7,578         5,581           UBS Securities Australia Ltd         Buy         7,578         5,581           UBS Securities Australia Ltd         Buy         1,767,48         5,581           UBS Securities Australia Ltd         Buy         1,767,48         2,572           UBS Securities Australia Ltd         Buy         1,677         2,61           UBS Securities Australia Ltd         Buy         1,677         2,61           UBS Securities Australia Ltd         Buy         2,219         3,61           UBS Securities Australia Ltd         Buy         2,219         1,67           UBS Securities Australia Ltd <td>- {</td> <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>3,275</td> <td></td> <td>Ordinary</td>	- {	UBS Securities Australia Ltd	Buy	3,275		Ordinary
UBS Securities Australia Ltd         Sell         \$51,933         \$7,356           UBS Securities Australia Ltd         Buy         211,800         30,000           UBS Securities Australia Ltd         Buy         19,404         2,133           UBS Securities Australia Ltd         Buy         19,404         2,133           UBS Securities Australia Ltd         Buy         19,404         2,513           UBS Securities Australia Ltd         Buy         176,861         24,910           UBS Securities Australia Ltd         Buy         472,575         57,543           UBS Securities Australia Ltd         Buy         472,575         57,543           UBS Securities Australia Ltd         Buy         1,674         2,549           UBS Securities Australia Ltd         Buy         1,674         2,549           UBS Securities Australia Ltd         Buy         1,674         2,549           UBS Securities Australia Ltd         Buy         1,670         2,671           UBS Securities Australia Ltd         Buy         2,572         3,541           UBS Securities Australia Ltd         Buy         2,572         2,649           UBS Securities Australia Ltd         Buy         2,572         2,649           UBS Securities Australia		UBS Securities Australia Ltd	Sell	11,091		Ordinary
UBS Securities Australia Ltd         Buy         211,800         30,000           UBS Securities Australia Ltd         Buy         19,404         2,733           UBS Securities Australia Ltd         Buy         19,404         2,733           UBS Securities Australia Ltd         Buy         1,32,400         4,599           UBS Securities Australia Ltd         Buy         1,75,661         24,910           UBS Securities Australia Ltd         Buy         1,75,661         24,910           UBS Securities Australia Ltd         Buy         1,75,661         24,910           UBS Securities Australia Ltd         Buy         47,320         7,070           UBS Securities Australia Ltd         Buy         1,584         2,522           UBS Securities Australia Ltd         Buy         1,588         2,623           UBS Securities Australia Ltd         Buy         3,418         4,88           UBS Securities Australia Ltd         Buy         3,418         4,88           UBS Securities Australia Ltd         Buy         3,418         4,88           UBS Securities Australia Ltd         Buy         3,418         4,88           UBS Securities Australia Ltd         Buy         2,49         1,07           UBS Securities Australia Ltd		UBS Securities Australia Ltd	Sell	51,933		Ordinary
UBS Securities Australia Ltd         Buy         80.2         113           UBS Securities Australia Ltd         Buy         19.404         2.733           UBS Securities Australia Ltd         Buy         32.400         4.569           UBS Securities Australia Ltd         Buy         17.6661         24.910           UBS Securities Australia Ltd         Buy         17.6661         24.910           UBS Securities Australia Ltd         Buy         472.767         67.681           UBS Securities Australia Ltd         Buy         472.767         67.643           UBS Securities Australia Ltd         Buy         2.572         3.68           UBS Securities Australia Ltd         Buy         3.418         4.68           UBS Securities Australia Ltd         Buy         3.418         4.68           UBS Securities Australia Ltd         Buy         3.418         4.68           UBS Securities Australia Ltd         Buy         3.418         4.69           UBS Securities Australia Ltd         Buy         7.49         1.07           UBS Securities Australia Ltd         Buy         2.719         3.41           UBS Securities Australia Ltd         Buy         2.719         3.71           UBS Securities Australia Ltd		UBS Securities Australia Ltd	Buy	211.800		Ordinary
UBS Securities Australia Ltd         Buy         19.404         2,733           UBS Securities Australia Ltd         Buy         3.24 do         2,733           UBS Securities Australia Ltd         Buy         3.24 do         4.569           UBS Securities Australia Ltd         Buy         7.76 861         2.450           UBS Securities Australia Ltd         Buy         175 682         2.450           UBS Securities Australia Ltd         Buy         4.75 861         2.450           UBS Securities Australia Ltd         Buy         4.75 42         2.5.249           UBS Securities Australia Ltd         Buy         7.57 43         2.5.249           UBS Securities Australia Ltd         Buy         7.57 43         2.5.249           UBS Securities Australia Ltd         Buy         7.57 43         2.5.249           UBS Securities Australia Ltd         Buy         7.67 43         2.5.249           UBS Securities Australia Ltd         Buy         7.407         7.00           UBS Securities Australia Ltd         Buy         7.407         7.00           UBS Securities Australia Ltd         Buy         7.407         7.53           UBS Securities Australia Ltd         Buy         7.407         7.53           UBS Securities		UBS Securities Australia Ltd	Buy	7		Ordinary
URS Securities Australia Ltd         Buy         19,40a         2,733           URS Securities Australia Ltd         Buy         32,440         4,569           URS Securities Australia Ltd         Buy         7,76,861         24,910           URS Securities Australia Ltd         Buy         473,677         17,613           URS Securities Australia Ltd         Buy         473,677         67,681           URS Securities Australia Ltd         Buy         473,677         67,681           URS Securities Australia Ltd         Buy         176,748         5,524           URS Securities Australia Ltd         Buy         2,572         3,68           URS Securities Australia Ltd         Buy         3,418         4,88           URS Securities Australia Ltd         Buy         3,418         4,88           URS Securities Australia Ltd         Buy         1,407         2,01           URS Securities Australia Ltd         Buy         1,407         2,01           URS Securities Australia Ltd         Buy         2,19         3,14           URS Securities Australia Ltd         Buy         2,19         3,14           URS Securities Australia Ltd         Buy         2,219         3,14           URS Securities Australia Ltd		UBS Securities Australia Ltd	Buy	802	113	Ordinary
URS Securities Australia Ltd         Bby         433         7.61           URS Securities Australia Ltd         Bby         32,44d         4,569           URS Securities Australia Ltd         Buy         176,861         24,910           URS Securities Australia Ltd         Buy         473,777         67,681           URS Securities Australia Ltd         Buy         176,483         26,249           URS Securities Australia Ltd         Buy         1,674         25,249           URS Securities Australia Ltd         Buy         1,674         25,249           URS Securities Australia Ltd         Buy         1,678         26,34           URS Securities Australia Ltd         Buy         3,411         4,88           URS Securities Australia Ltd         Buy         3,411         4,88           URS Securities Australia Ltd         Buy         7,497         1,07           URS Securities Australia Ltd         Buy         2,419         3,41           URS Securities Australia Ltd         Buy         2,419         3,41           URS Securities Australia Ltd         Buy         2,419         3,41           URS Securities Australia Ltd         Buy         2,419         3,41           URS Securities Australia Ltd		UBS Securities Australia Ltd	Buy	19.404	2 733	Ordinary
UBS Securities Australia Ltd         BBy         32,440         4,569           UBS Securities Australia Ltd         Buy         175,685         17,681           UBS Securities Australia Ltd         Buy         473,767         17,681           UBS Securities Australia Ltd         Buy         473,767         67,681           UBS Securities Australia Ltd         Buy         176,742         52,440           UBS Securities Australia Ltd         Buy         1,584         26,349           UBS Securities Australia Ltd         Buy         3,418         488           UBS Securities Australia Ltd         Buy         3,418         488           UBS Securities Australia Ltd         Buy         3,418         488           UBS Securities Australia Ltd         Buy         3,418         488           UBS Securities Australia Ltd         Buy         1,407         201           UBS Securities Australia Ltd         Buy         2,419         3,41           UBS Securities Australia Ltd         Buy         2,219         3,41           UBS Securities Australia Ltd         Buy         2,219         3,41           UBS Securities Australia Ltd         Buy         2,219         3,41           UBS Securities Australia Ltd         <		UBS Securities Australia Ltd	Buy	433	19	Colingo
UBS Securities Australia Ltd         Bby         178,861         24,910           UBS Securities Australia Ltd         Buy         175,652         17,613           UBS Securities Australia Ltd         Buy         473,767         67,681           UBS Securities Australia Ltd         Buy         160,443         25,643           UBS Securities Australia Ltd         Buy         1,673         269           UBS Securities Australia Ltd         Buy         3,418         269           UBS Securities Australia Ltd         Buy         3,418         489           UBS Securities Australia Ltd         Buy         3,418         489           UBS Securities Australia Ltd         Buy         3,418         489           UBS Securities Australia Ltd         Buy         7,497         201           UBS Securities Australia Ltd         Buy         7,497         201           UBS Securities Australia Ltd         Buy         2,419         3,41           UBS Securities Australia Ltd         Buy         2,419         3,14           UBS Securities Australia Ltd         Buy         2,419         3,14           UBS Securities Australia Ltd         Buy         2,219         3,15           UBS Securities Australia Ltd         Buy	16-Dec-14	UBS Securities Australia Ltd	Buy	32 440	7 550	Cicinaly
UBS Securities Australia Ltd         Bay         172,052         17,613           UBS Securities Australia Ltd         Buy         47,376         17,613           UBS Securities Australia Ltd         Buy         47,376         17,613           UBS Securities Australia Ltd         Buy         176,423         7,500           UBS Securities Australia Ltd         Buy         1,577         368           UBS Securities Australia Ltd         Buy         3,418         488           UBS Securities Australia Ltd         Buy         4,388         627           UBS Securities Australia Ltd         Buy         4,388         627           UBS Securities Australia Ltd         Buy         4,388         627           UBS Securities Australia Ltd         Buy         1,407         201           UBS Securities Australia Ltd         Buy         1,407         201           UBS Securities Australia Ltd         Buy         2,19         346           UBS Securities Australia Ltd         Buy         2,19         346           UBS Securities Australia Ltd         Buy         2,19         347           UBS Securities Australia Ltd         Buy         2,219         347           UBS Securities Australia Ltd         Buy	16-Dec-14	UBS Securities Australia Itd	Ruv	176 971	010 10	Ordinary
UBS Securities Australia Ltd         Buy         473,677         67,681           UBS Securities Australia Ltd         Buy         473,677         67,681           UBS Securities Australia Ltd         Buy         176,482         7,070           UBS Securities Australia Ltd         Buy         2,572         368           UBS Securities Australia Ltd         Buy         3,418         489           UBS Securities Australia Ltd         Buy         3,411         489           UBS Securities Australia Ltd         Buy         1,407         201           UBS Securities Australia Ltd         Buy         1,407         201           UBS Securities Australia Ltd         Buy         2,419         3,41           UBS Securities Australia Ltd         Buy         2,419         3,41           UBS Securities Australia Ltd         Buy         2,419         3,41           UBS Securities Australia Ltd         Buy         2,419         3,41           UBS Securities Australia Ltd         Buy         2,319         3,14           UBS Securities Australia Ltd         Buy         2,319         3,14           UBS Securities Australia Ltd         Buy         2,319         3,17           UBS Securities Australia Ltd         Buy <td>16-Dec-14</td> <td>UBS Securities Australia Itd</td> <td>Bliv</td> <td>135 063</td> <td>17.610</td> <td>Ordinary</td>	16-Dec-14	UBS Securities Australia Itd	Bliv	135 063	17.610	Ordinary
USS Securities Australia tot 1085 Securities Securities Australia tot 1085 Securities Securities Securities Securities Securities Securities Securities Securities Securities Securities Securiti	16.Dec.14	A IRS Societies Australia 144	600	260,621	510,71	Ordinary
UBS Securities Australia tot         Buy         176,749         7,070           UBS Securities Australia tot         Buy         2,572         368           UBS Securities Australia tot         Buy         2,572         368           UBS Securities Australia tot         Buy         3,415         488           UBS Securities Australia tot         Buy         3,411         488           UBS Securities Australia tot         Buy         4,383         6,07           UBS Securities Australia tot         Buy         1,407         201           UBS Securities Australia tot         Buy         2,419         3,46           UBS Securities Australia tot         Buy         2,419         3,46           UBS Securities Australia tot         Buy         2,219         3,47           UBS Securities Australia tot         Buy         2,219         3,47           UBS Securities Australia tot         Buy         2,219         3,47           UBS Securities Australia tot         Buy         2,219         3,47           UBS Securities Australia tot         Buy         2,219         3,47           UBS Securities Australia tot         Buy         2,219         3,70           UBS Securities Australia tot         Contract	16-Dec-14	I IRC Securities Australia 144	on d	4/3,/6/	189'/9	Ordinary
UBS Securities Australia Ltd         Buy         17,043         23,848           UBS Securities Australia Ltd         Buy         1,572         2,588           UBS Securities Australia Ltd         Buy         3,418         269           UBS Securities Australia Ltd         Buy         3,418         489           UBS Securities Australia Ltd         Buy         3,411         489           UBS Securities Australia Ltd         Buy         7,439         107           UBS Securities Australia Ltd         Buy         1,407         201           UBS Securities Australia Ltd         Buy         2,419         3,45           UBS Securities Australia Ltd         Buy         2,419         3,45           UBS Securities Australia Ltd         Buy         2,419         3,45           UBS Securities Australia Ltd         Buy         2,419         3,45           UBS Securities Australia Ltd         Buy         2,331         3,13           UBS Securities Australia Ltd         Buy         2,319         3,14           UBS Securities Australia Ltd         Sell         1,76         2,59           UBS Securities Australia Ltd         Contract         1,76         2,79           UBS Securities Australia Ltd         Sell	16-Dec-14	Libe Convictor Australia Lea	hoo a	084,84	0,070	Ordinary
URS Securities Australia Ltd         Buy         2.572         368           UBS Securities Australia Ltd         Buy         1.838         2.69           UBS Securities Australia Ltd         Buy         3.411         4.89           UBS Securities Australia Ltd         Buy         3.411         4.88           UBS Securities Australia Ltd         Buy         4.383         6.77           UBS Securities Australia Ltd         Buy         7.49         1.07           UBS Securities Australia Ltd         Buy         2.219         3.41           UBS Securities Australia Ltd         Buy         2.219         3.45           UBS Securities Australia Ltd         Buy         2.219         3.45           UBS Securities Australia Ltd         Buy         2.319         3.37           UBS Securities Australia Ltd         Buy         2.319         3.37           UBS Securities Australia Ltd         Buy         2.319         3.37           UBS Securities Australia Ltd         Buy         2.719         3.71           UBS Securities Australia Ltd         Buy         2.7419         2.75           UBS Securities Australia Ltd         Contract         Contract         2.7458         58.770           UBS Securities Australia	15 Dec 14	Opp Securities Australia Liu	on's	1/6,/43	25,249	Ordinary
UBS Securities Australia Ltd         Buy         1838         263           UBS Securities Australia Ltd         Buy         3,418         4,88           UBS Securities Australia Ltd         Buy         4,383         627           UBS Securities Australia Ltd         Buy         7,49         107           UBS Securities Australia Ltd         Buy         7,49         107           UBS Securities Australia Ltd         Buy         2,419         201           UBS Securities Australia Ltd         Buy         2,419         3,46           UBS Securities Australia Ltd         Buy         2,419         3,17           UBS Securities Australia Ltd         Buy         2,331         3,17           UBS Securities Australia Ltd         Buy         2,331         3,17           UBS Securities Australia Ltd         Buy         2,331         3,17           UBS Securities Australia Ltd         Buy         2,331         3,13           UBS Securities Australia Ltd         Sell         1,76         2,5           UBS Securities Australia Ltd         Contract         1,76         2,7           UBS Securities Australia Ltd         Sell         5,01         6,00           UBS Securities Australia Ltd         Sell	10-Dec-14	UBS Securities Australia Ltd	BUJ	2,572	368	Ordinary
UBS Securities Australia 100         Buy         3.418         489           UBS Securities Australia 100         Buy         3.411         488           UBS Securities Australia 100         Buy         4,383         627           UBS Securities Australia 100         Buy         1,407         201           UBS Securities Australia 100         Buy         1,407         201           UBS Securities Australia 100         Buy         2,719         3.46           UBS Securities Australia 100         Buy         2,219         3.46           UBS Securities Australia 100         Buy         2,219         3.31           UBS Securities Australia 100         Buy         2,219         3.31           UBS Securities Australia 100         Buy         2,331         3.33           UBS Securities Australia 100         Buy         2,331         3.35           UBS Securities Australia 100         Buy         2,331         3.50           UBS Securities Australia 100         Buy         Contract         NA         2,000           UBS Securities Australia 100         Contract         Contract         800         601         600           UBS Securities Australia 100         Buy         601         601         601	16-Dec-14	UBS Securities Australia Ltd	Buy	1,838	263	Ordinary
URS Securities Australia Ltd         Buy         3,411         488           URS Securities Australia Ltd         Buy         3,411         488           URS Securities Australia Ltd         Buy         7,49         107           URS Securities Australia Ltd         Buy         1,407         201           URS Securities Australia Ltd         Buy         2,419         3,46           URS Securities Australia Ltd         Buy         2,419         3,46           URS Securities Australia Ltd         Buy         2,219         3,47           URS Securities Australia Ltd         Buy         2,219         3,47           URS Securities Australia Ltd         Buy         2,219         3,47           URS Securities Australia Ltd         Buy         2,331         1,081,710         (153,000)           URS Securities Australia Ltd         Contract         NA         2,000)           URS Securities Australia Ltd         Sell         5,012         (660)           URS Securities Australia Ltd         Sell         5,012         (660)	16-Uec-14	UBS Securities Australia Ltd	Buy	3,418	489	Ordinary
UBS Securities Australia Ltd         Buy         4383         G27           UBS Securities Australia Ltd         Buy         749         107           UBS Securities Australia Ltd         Buy         1,407         201           UBS Securities Australia Ltd         Buy         2,419         32           UBS Securities Australia Ltd         Buy         2,419         346           UBS Securities Australia Ltd         Buy         2,219         317           UBS Securities Australia Ltd         Buy         2,319         317           UBS Securities Australia Ltd         Sell         1,76         7,53,000           UBS Securities Australia Ltd         Enter Into Demative         1,76         2,79           UBS Securities Australia Ltd         Contract         5,012         (601)           UBS Securities Australia Ltd         Sell         5,012         (601)           UBS Securities Australia Ltd         Sell         5,012         (661)	16-Dec-14	UBS Securities Australia Ltd	Buy	3,411	488	Ordinary
UBS Securities Australia Ltd         Buy         1407         107           UBS Securities Australia Ltd         Buy         1,407         201           UBS Securities Australia Ltd         Buy         1,547         22           UBS Securities Australia Ltd         Buy         2,119         346           UBS Securities Australia Ltd         Buy         2,131         333           UBS Securities Australia Ltd         Buy         2,331         333           UBS Securities Australia Ltd         Buy         2,331         333           UBS Securities Australia Ltd         Buy         2,331         176         25           UBS Securities Australia Ltd         Contract         NA         2,000           UBS Securities Australia Ltd         Soft Contract         NA         2,000           UBS Securities Australia Ltd         Soft Contract         8,000         8,000           UBS Securities Australia Ltd         Soft Contract         8,000         8,000	16-Dec-14	UBS Securities Australia Ltd	Buy	4,383	627	Ordinary
UBS Securities Australia Ltd         Buy         1,407         201           UBS Securities Australia Ltd         Buy         154         22           UBS Securities Australia Ltd         Buy         2,419         345           UBS Securities Australia Ltd         Buy         2,319         317           UBS Securities Australia Ltd         Sell         1,681,710         (153,000)           UBS Securities Australia Ltd         Sell         1,76         25           UBS Securities Australia Ltd         Contract         NA         (2,000)           UBS Securities Australia Ltd         Sell         50.2         (660)           UBS Securities Australia Ltd         Sell         50.2         (660)           UBS Securities Australia Ltd         Sell         50.2         (660)	16-Dec-14	UBS Securities Australia Ltd	Buy	749	107	Ordinary
UBS Securities Australia Ltd         Buy         154         22           UBS Securities Australia Ltd         Buy         2,419         346           UBS Securities Australia Ltd         Buy         2,319         337           UBS Securities Australia Ltd         Buy         2,331         333           UBS Securities Australia Ltd         Buy         2,331         176         25           UBS Securities Australia Ltd         Contract         NA         (2,000)           UBS Securities Australia Ltd         Sell         5,012         (691)           UBS Securities Australia Ltd         Buy         427,458         58,770	16-Dec-14	UBS Securities Australia Ltd	Buy	1,407	201	Ordinary
UBS Securities Australia Ltd         Buy         2.419         3.46           UBS Securities Australia Ltd         Buy         2.219         3.17           UBS Securities Australia Ltd         Buy         2.331         3.33           UBS Securities Australia Ltd         Buy         7.331         5.335           UBS Securities Australia Ltd         Buy         7.081.710         (153,000)           UBS Securities Australia Ltd         Frinter Into Dervative         NA         (2,000)           UBS Securities Australia Ltd         Sell         5,012         (660)           UBS Securities Australia Ltd         Sull         4,27,456         58,770	16-Dec-14	UBS Securities Australia Ltd	Buy	154	22	Ordinary
UBS Securities Australia Ltd         Buy         2,219         317           UBS Securities Australia Ltd         Self         1,081,710         (153,000)           UBS Securities Australia Ltd         Buy         1,081,710         (153,000)           UBS Securities Australia Ltd         Enter Into Dernative         35         25           UBS Securities Australia Ltd         Contact         NA         (2,000)           UBS Securities Australia Ltd         Sell         5,012         (691)           UBS Securities Australia Ltd         Buy         427,458         58,770	16-Dec-14		Buy	2,419	346	Ordinary
UBS Securities Australia Ltd         Bby         2,331         333           UBS Securities Australia Ltd         Buy         7,681,710         (155,000)           UBS Securities Australia Ltd         Enter Into Dernative         176         25           UBS Securities Australia Ltd         Contract         NA         (2,000)           UBS Securities Australia Ltd         Sell         5,012         (691)           UBS Securities Australia Ltd         Buy         427,458         58,770	16-Dec-14	UBS Securities Australia Ltd	Buy	2,219	317	Ordinary
UBS Securities Australia trid   Self   1,081,710   (153,000)   UBS Securities Australia trid   Enter Into Derivative   LBS Securities Australia trid   Self   Securities Australia trid   Self   Securities Australia trid   Supplementary	16-Dec-14	UBS Securities Australia Ltd	Buy	2.331	333	Ordinary
UBS Securities Australia Ltd         Buy Enter Into Dervative         176         25           UBS Securities Australia Ltd         Enter Into Dervative Contact         NA         (2,000)           UBS Securities Australia Ltd         Sell         (691)           UBS Securities Australia Ltd         Buy         427,458         58,770	16-Dec-14	UBS Securities Australia Ltd	Self	1 081 710	(153 000)	Ordinary
UBS Securities Australia Ltd         Enter Into Dervative         NA         (2,000)           UBS Securities Australia Ltd         Sell         5,012         (691)           UBS Securities Australia Ltd         Buy         427,458         56,770	16-Dec-14	UBS Securities Australia Ltd	Buy	9/1		Ordinary
UBS Securities Australia Ltd         Contact         NA         (2,000)           UBS Securities Australia Ltd         Sell         5,012         (691)           UBS Securities Australia Ltd         Buy         427,458         58,770			Enter Into Derivative			(1000)
UBS Securities Australia Ltd         Sell         5,012         (691)           UBS Securities Australia Ltd         Buy         427,458         58,770	16-Dec-14	UBS Securities Australia Ltd	Contract	φ <sub>N</sub>	(2 000)	Options
UBS Securities Australia Ltd Buy 427,458 58,770	17-Dec-14	UBS Securities Australia Ltd	Sell	5012	(504)	a Para
DOS SECURIOS AUSTRALIA DAY 427,458 58,770	17.Dec. 14	Libe Countries Australia Ind		210'S	(160)	Ordinary
	1/-Dec-14	UBS Securines Australia Ltd	BUy	1827.4581	58 770 1	

De Cacimina Autralia del	976		
	366	(100)	
JBS Securities Australia Ltd   Sell	51		Ordinary
	124	(2)	
	284		
JBS Securities Australia Ltd	328		
	100		
UBS Securities Australia Ltd	100		
	138		
IRS Securities Australia Itd	700		
	757		
	171		
UBS Securities Australia Ltd	102		
	182		
	305	(42)	
	1,141		
UBS Securities Australia Ltd	632		ĺ
UBS Securities Australia Ltd	1,323		Ordinary
UBS Securities Australia Ltd	15		
	ď		l
	364		l
	100		
Upo percurius Australia Ltd	1,432		
	109	(15)	
	465		
	233		
	1,207		
	66	(6)	
UBS Securities Australia Ltd	29		
Ltd	4,304		
UBS Securities Australia Ltd	298	(17)	Ordinary
Ltd	29	(4)	Ordinary
Ltd	5,168	14)	Ordinary
Ltd Sell	496		Ordinary
UBS Securities Australia Ltd Sell	774	(106)	Ordinary
	15	Ø	Ordinary
	7	(L)	Ordinary
	87	(21)	Ordinary
	751 9	(EV8)	Ordinary
Lt.	0 262		Ordinary Ordinary
	7 265		
	3 654	(E03)	l
	1 850		
	520	(0.5)	Codingo
Sell	000	(40.0)	Codinary
	225	(171)	Contrally
UBS Securities Australia Ltd	1004	(89)	Ordinary
lua eny	4,/92	099	Ordinary
ETG BUJ	690'09	8,274	Ordinary
Ltd	98,925	13,626	Ordinary
Ltd Buy	3,630	200	Ordinary
	2,026	279	Ordinary
Ltd Buy	51,060	7,033	Ordinary
	2,018	278	Ordinary
Securities Australia Ltd Buy	33,062	4,554	Ordinary
td Buy	7 260	1 000	Ordinary
Securities Australia Ltd	300 6	97.6	Ordinary
	93 105	247 44	Control of the contro
	501,58		Ordinary
ing of	5,48		Ordinary
	62,093		Ordinary
	19,522		Ordinary
	58,733		Ordinary
curities Australia Ltd	292.800		Ordinary
	69		Ordinan
2.8	000	0000	Ordinary
	onc's		Cidinary
IIRS Securities Australia 144	017.6		Ordinary
	1/5'5		Orginary
	868'1	5/2	Ordinary
	3,422		Ordinary
	138		Ordinary
	6,201		Ordinary
	2,719		Ordinary
	3,402		Ordinary
	1.960		Ordinary
	2 133		Ordinary
	7 554		Ordinary
	4,554		Ordinary
S Securities Australia Ltd Sell Sourcivies Australia Ltd Buy		1,960 2,133 4,554	1,960 284 2,133 (310) 4,554 661

	Toro Securines Australia Ltd	100	67	(4)	Ordinar
	UBS Securities Australia Ltd	les	1,685	(1,614)	Colore
	UBS Securities Australia Ltd	Sell	72		Ordinary
	UBS Securities Australia Ltd	Sell	17,108		Ordinary
	UBS Securities Australia Ltd	Sell	17.108	(2.363)	Ordinary
	UBS Securities Australia Ltd	Sell	17.065		Ordinan
	UBS Securities Australia Ltd	Sell	43	(9)	Ordinary
	UBS Securities Australia Ltd	Sell	22	6	Ordinan
	UBS Securities Australia Ltd	Seif	11128	31/	Ordinary
	UBS Securities Australia Ltd	Ties.	33		Codings
l	UBS Securities Australia Ltd	Self	43	2 9	Commen
ı	UBS Securities Australia 11d	3	17 108		Clainary
1	IRS Securities Australia 144		501,71	(5,363)	Ordinary
1	URS Securities Australia 144	ll o	17,043	(2,354)	Ordinary
1	DE Courier Australia Ltd	- F	60	6	Ordinary
ľ	DOS Countries Australia Led	list of	36	(2)	Ordinary
	Dia Seutifica Australia Liu	100	22	(9)	Ordinary
ı	Upo securities Australia Ltd	Jer.	7	(3)	Ordinary
1	UBS Securities Australia ttd	Sell	14,161	(1,956)	Ordinary
	UBS Securities Australia Ltd	Seal	65	(6)	Ordinary
ı	UBS Securities Australia Ltd	Sell	7	(3)	Ordinary
1	UBS Securities Australia Ltd	Sell	22	(3)	Ordinary
-	UBS Securities Australia Ltd	Sell	21,459	(5,964)	Ordinary
	UBS Securities Australia Ltd	Sell	51	(2)	Ordinary
	UBS Securities Australia Ltd	Sell	18,643	(2,575)	Ordinary
7-Dec-14	UBS Securities Australia Ltd	Sell	22	®	Ordinary
	UBS Securities Australia Ltd	Sell	29		Ordinary
	UBS Securities Australia Ltd	Sell	4.445	9)	Ordinary
	UBS Securities Australia Ltd	-š.	3 743		Codings
	UBS Securities Australia Ltd	Seil	5,178	(10)	0.00
Dec-14	UBS Securities Australia Ind	الع	22,00		Cidinary
Jec-17	IRC Socurities Australia 14d	Coll	061,22		Ordinary
7-Dec-14	IRC Securities Australia Ltd	los	60/10		Ordinary
Dec. 14	UBC Countition Americality 144	Jell Coll	900%	(818,1)	Ordinary
11.00	Obs securines Australia LIU	nac l	16,514		Ordinary
/-Dec-14	UBS Securities Australia Ltd	Sell	16,514		Ordinary
/-Dec-14	UBS Securities Australia Ltd	Sell	16,746		Ordinary
7-Dec-14	UBS Securities Australia Ltd	Selt	18,607		Ordinary
7-Dec-14	UBS Securities Australia Ltd	Sell	23,204		Ordinary
lec-14	UBS Securities Australia Ltd	Sell	22		Ordinary
	UBS Securities Australia Ltd	Sell	23,154	(3.198)	Ordinary
	UBS Securities Australia Ltd	Sell	72	(10)	Ordinan
Г	UBS Securities Australia Ltd	Sell	23 175	(100.5)	Ordinan
17-Dec-14	UBS Securities Australia itd	وا	53,423	(107'0)	Cidinary
Ī	IRS Securities Australia Ital	[9]	17.15	(1)	Country
ı	IRS Securities Australia 14d	Coll	511,71	(7,354)	Ordinary
1	TIPS Conviction Australia 1-4	les.	8,439	(1,164)	Ordinary
ı	UDC Countries Australia Lad	liac line	7,530	(349)	Ordinary
1	UBS Securities Australia Ltd	Nell Sell	4,053	(523)	Ordinary
/-Dec-14	UBS Securities Australia Ltd	Sell	1,204	(199)	Ordinary
1	UBS Securties Australia Ltd	Sell	334	(46)	Ordinary
	UBS Securities Australia Ltd	Sell	1,218	(168)	Ordinary
	UBS Securities Australia Ltd	Sell	28,007	(3.863)	Ordinary
	UBS Securities Australia Ltd	Self	15	(6)	Ordinary
	UBS Securities Australia Ltd	Sell	25.4	(3.6)	Ordinary
l	UBS Securities Australia Itd	lay.	5.74	(503)	S G
1	URS Securities Australia 144	Sell	550,5	(/60)	Ordinary
ı	IDC Countries Australia 140	Jen Sell	916	(44)	Ordinary
	DES SECULIES AUSTRAIR L'U	lie C	29,515	(4,071)	Ordinary
ı	Obs securities Australia Ltd	iik.	8,214	(1,133)	Ordinary
	UBS Securities Australia Ltd	Jell	47,836	(865'9)	Ordinary
7-Dec-14	UBS Securities Australia Ltd	Sell	5,967	(823)	Ordinary
	UBS Securities Australia Ltd	Seil	43,116	(5,947)	Ordinary
	UBS Securities Australia Ltd	Seil	31,654	(4.366)	Ordinary
Ī	UBS Securities Australia Ltd	Seil	15 348	(7117)	Contract
	IIBS Securities Australia Itd	3	059 10	(11.7)	Collinary
	IRS Securities Australia 144	200	0/0/13	(696,7)	Orginary
7 Per 17	On Securities Australia to a	llac l	95	(5)	Ordinary
ı	DBS Securities Australia Ltd	Nell Nell	6,525	(006)	Ordinary
ŀ	UBS Securities Australia Ltd	Sell	29,653	(4,090)	Ordinary
7-Dec-14	UBS Securities Australia Ltd	Sell	68,273	(9.417)	Ordinary
7-Dec-14	UBS Securities Australia Ltd	Sell	37,504	(5.173)	Ordinary
7-Dec-14	UBS Securities Australia Ltd	Sell	22.758	(3.139)	Ordinary
	UBS Securities Australia Ltd	Sell	18 473	(875.0)	Ordinan
l	JBS Securities Australia I1d	Sell	11 200	(9 5 45)	Creditary
l	IRS Cocurities Australia 144	Soll	602,1	(350)	Orginary
	ODS Securities Australia Ltd	Sell	2,719	(375)	Ordinary
17-Dec-14	UBS Securities Australia Ltd	Sell	10,795	(1,489)	Ordinary
	JBS Securities Australia Ltd	Sell	1341	(185)	Ordinary

UBS Securities Australia 11d Sell	170 913	(23.415)	Ordinary
	1/0,913	14.075	Ordinary
Securities Australia Ltd Sell	1 325	(183)	Ordinary
	77.430	(10.638)	Ordinary
Securities Australia Ltd Buy	78,360	11,000	Ordinary
UBS Securities Australia Ltd	060'898	(20,000)	Ordinary
	101,741	(14,001)	Ordinary
	20,432	(2,811)	Ordinary
	131,953	(18,206)	Ordinary
p).	7,173	984	Ordinary
UBS AL, London Branch	94,222	10,638	Ordinary
	77,430	(10,638)	Ordinary
	14,589	(2,004)	Ordinary
	10,066	(1,375)	Ordinary
UBS Wealth Management Australia Ltd	9,363	(1,279)	Ordinary
	181	(25)	Ordinary
	2,266	(313)	Ordinary
Securities Australia Ltd Sell	304	(42)	Ordinary
UBS Securities Australia Ltd Sell	1,086	(150)	Ordinary
	3,743	(517)	Ordinary
Securities Australia Ltd Sell	1,086	(120)	Ordinary
Securities Australia Ltd Sell	1,238	(171)	Ordinary
Securities Australia Ltd Sell	4,243	(586)	Ordinary
	2.331	(322)	Ordinary
	6 292	(986)	Ordinary
S Securities Australia Ltd	268	(32)	Ordinan
	23.55	(505)	Crainery
	2,000	(202)	Countary
	500.0	(2/1)	Guillary
	102,6	(0++1)	Collinary
	340	(47)	Ordinary
	261	(36)	Ordinary
Ltd	3,569	(493)	Ordinary
	4,112	(268)	Ordinary
	1,151	(159)	Ordinary
	11,142	(1,539)	Ordinary
	340	(47)	Ordinary
	15,211	(2,101)	Ordinary
UBS Securities Australia Ltd	174	(24)	Ordinary
Ltd Sell	4,112	(268)	Ordinary
	3,591	(496)	Ordinary
UBS Securities Australia Ltd	43	(9)	Ordinary
	22,878	(3,160)	Ordinary
	10,520	(1,453)	Ordinary
	20,591	(2,844)	Ordinary
Ltd	22,915	(3,165)	Ordinary
UBS Securities Australia Ltd	22,871	(3,159)	Ordinary
JBS Securities Australia Ltd Sell	6,516	(006)	Ordinary
	3.207	(443)	Ordinary
	51	6	Ordinary
	928 91	(2) 334)	Ordinary
	270,01	(400/	Ordinary
	13 63	(000)	Ordinary
TIDE Countries Australia (+A	occ.	(000,1)	Cidinary
	67	t)	Cicinaly
	2000	1000	Orderary
	2,043	(393)	Ordinary
	50/'C	(QK/)	Ordinary
	568,01	(1,499)	Ordinary
	68/'/	(7,45/)	Ordinary
	7	(E)	Ordinary
	13,676	(1,889)	Ordinary
JBS Securities Australia Ltd	59	(4)	Ordinary
	59	(4)	Ordinary
UBS Securities Australia Ltd	19,309	(2,667)	Ordinary
Ltd	65	(6)	Ordinary
UBS Securities Australia Ltd   Sell	13,691	(1,891)	Ordinary
UBS Securities Australia Ltd	22	(3)	Ordinary
	13,662	(1,887)	Ordinary
Securities Australia Ltd Sell	59	(4)	Ordinary
	43	(9)	Ordinary
	3,178	(439)	Ordinary
	905'6	(1,313)	Ordinary
	3,555	(491)	Ordinary

77-Dec-14 Uf 17-Dec-14 Uf 17-Dec-14 Uf 17-Dec-14 Uf 17-Dec-14 Uf 17-Dec-14 Uf 17-Dec-14 Uf 17-Dec-14 Uf 17-Dec-14 Uf	UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy	153		Ordinary
	DO DECOLUES AUSUBIIB LICE	ann.	7,544		Ordinary
	KY Vol. 1716 O. 1617 Sept. 1	Bully	***		
	DO Securities Australia Ltd	puy B <sub>111</sub> ,	344		Ordinary
	bs securities Australia Ltd	puy	7,368	-	Ordinary
	Bs Securities Australia Ltd	Buy	3,626	496	Ordinary
	BS Securities Australia Ltd	Buy	1,075		Ordinary
1	BS Securities Australia Ltd	Buy	4,547	622	Ordinary
	UBS Securities Australia Ltd	Buy	8,575	1	Ordinary
l	BS Securities Australia Ltd	Buy	2,156		Ordinary
Į	BS Securities Australia Ltd	Buy	3,626	496	Ordinary
	8S Securities Australia Ltd	Buy	2,924		Ordinary
1	UBS Securities Australia Ltd	Buy	263		Ordinary
	UBS Securities Australia Ltd	Виу	541		Ordinary
	BS Securities Australia Ltd	Buy	1.000		Ordinary
	BS Securities Australia Ltd	Buv	92E E	321/	Creining
l	BS Securities Australia Itd	Riv	230,5	00.0	Ordinary
l	BS Securities Australia Itd	Ruy	0.55	27C'+	Couriery
	85 Securities Australia Itcl	No. of	25,240	340	Ordinary
	IRS Securities Australia Itol	Bira	07.1	2,609	Orginary
7-Dec-14	IDS Securities Australia Ltd	no.	2,540	348	Ordinary
۱	of Countries Australia (10	bus,	335	46	Ordinary
ı	bo securities Australia LIC	ond.	5,052	269	Ordinary
ı	BS Securities Australia Ltd	Buy	350	48	Ordinary
	85 Securities Australia Ltd	Buy	146	20	Ordinary
7-Dec-14 UE	85 Securities Australia Ltd	Buy	6,161	844	Ordinary
	UBS Securities Australia Ltd	Buy	14,074	1,928	Ordinary
7-Dec-14 UB	3S Securities Australia Ltd	Buy	23	10	Ordinary
	UBS Securities Australia Ltd	Buy	7,441	1,020	Ordinary
	35 Securities Australia Ltd	Buy	3.217	441	Ordinary
17-Dec-14 UB	JBS Securities Australia Ltd	Buy	1.453	199	Ordinary
Ī	UBS Securities Australia Ltd	Buv	130	2	Ordinary
	IBS Securities Australia 14d	N. I.	2 4 2 4	500	Clumary
l	26 Sperindian Australia 14d	100	#2)'Z	67	Ordinary
1	OC Compiler Australia Inch	100	2,438	334	Ordinary
ı	Section of the sectio	foo	5,553	54/	Ordinary
	so securities Australia Lto	Buy	1,672	229	Ordinary
-	35 Securities Australia Ltd	Buy	372	51	Ordinary
-	3S Securities Australia Ltd	Buy	365	20	Ordinary
	3S Securities Australia Ltd	Buy	3.015	413	Ordinary
	35 Securities Australia Ltd	Buv	002.8	1137	Confine
	35 Securities Australia Itd	Buy	4 175	151	O Gillary
	35 Securities Australia Itol	Ruy	7 150	100	
ı	S Cocumition Americal and	6.00	601.7	706	Ordinary
	Socialities Australia Ltd	an)	3,621	496	Ordinary
ı	so securities Australia Ltd	BUY	8,088	1,108	Ordinary
	35 Securities Australia Ltd	Buy	2,088	586	Ordinary
	35 Securities Australia Ltd	Buy	3,621	496	Ordinary
	3S Securities Australia Ltd	Buy	80	~	Ordinary
l	35 Securities Australia 11d	Blav	37.5	O.F.	(in in in
	SC Counties Australia 144	600	0/5	5	Ordinary
١	so securines Australia Ltd	any	117	16	Ordinary
1/-Dec-14 UB	UBS Securities Australia Ltd	Buy	15	2	Ordinary
	55 Securities Australia Ltd	Buy	751	103	Ordinary
	55 Securities Australia Ltd	Buy	408	a,	Ordinari
١	I IRC Countries Australia 11d	Ruy	1000	OC .	Orginary
	o permittee Australia Liu	on's	786'6	1,370	Ordinary
1	OBS Securities Australia Ltd	Buy	109	15	Ordinary
	is Securities Australia Ltd	Buy	117	16	Ordinary
	iS Securities Australia Ltd	Buy	098	118	Ordinary
	S Securities Australia Ltd	Buy	882	121	Ordinan
Γ	UBS Securities Australia Ltd	Bix	001	3	
Ī	Commission Australia Isal		coc	60	Orginary
7-Dec-14	UBS Securities Australia Ltd	Buy	474	65	Ordinary
1	UBS Securities Australia Ltd	Buy	853	117	Ordinary
_	JBS Securities Australia Ltd	Buy	21,586	2,961	Ordinary
7-Dec-14 UB	JBS Securities Australia Ltd	Buy	22	-	Ordinary
7-Dec-14 UB		Ruy	OFF	000	in in
Ī		( ) d	011.3	2005	Ordinary
Ť	S securities Australia Ltd	ana.	2,916	400	Ordinary
/-Dec-14 UB	JBS Securities Australia Ltd	Buy	722	66	Ordinary
	JBS Securities Australia Ltd	Buy	2,916	400	Ordinary
7-Dec-14 UB:	JBS Securities Australia Ltd	Buy	372	70	Ordinary
ľ	JBS Securities Australia Ltd	Buv	901	12	Ordinary)
Ť	IRS Sociention Australia Ind	1	60.	2 1	Clurary
1	כ סברתווובי שמאני שוש דות	anà	343	47	Ordinary
1	UBS Securities Australia Ltd	Buy	6,192	850	Ordinary
	S Securities Australia Ltd	Buy	50.046	6 865	Ordinan
7-Dec-14 UB	UBS Securities Australia Itd	Rin	303 0	0,000	Ordinary
	Committee Australia 144	Duy	2,508	***	Ordinary
1	UBS Securities Australia Ltd	Buy	80/	104	Ordinary
17-Dec-14 UBS		Buy	2,785	382	Ordinary
١			1 300	100	7

Se
Sell
Sell
3
Sell
Sell
Sell
Sell
Sell
Sell
Sell
Sell
Sel
Se
Sell
35
Sell
Too
2
Sel
Sell
Sell
Sel
lay.
3
2
Sei
Sell
Sell
3
100
200
Sel
Sell
Sell
3
199
100
lac ,
Sei
Sell
Sell
Sell
Sel
Sell
وق
Joseph
100
Ne l
Şe
Seil
Sell
le o
100
Jes
Se
Sell
Biz
200
1
ĵο .
Buy
BÇ
ا
l Buy
Buy
Buv
2.0
έnα
Buy Buy
Buv
Pi ii
1
BUJ
BG
Sis
î,
8rà
Buy
E.
+
ğ
ä
ė
Buy

17-Dec-14	UBS Securities Australia Ltd	Buv	11 708	1,605	Ordinary
		,	00/1		
١	UBS Securities Australia Ltd	Buy	48,891		Ordinary
	UBS Securities Australia Ltd	Buy	3,939		
	UBS Securities Australia Ltd	Buy	2,882	395	Ordinary
1	UBS Securities Australia Ltd	Buy	819	112	Ordinary
ı	UBS Securities Australia Ltd	Buy	380	52	Ordinary
ļ	UBS Securities Australia Ltd	Buy	51	7	Ordinary
1	UBS Securities Australia Ltd	Buy	4,199	574	
1	UBS Securities Australia Ltd	Buy	373	51	Ordinary
1	UBS Securities Australia Ltd	Buy	10,021	1,369	Ordinary
Ì	UBS Securities Australia Ltd	Buy	6,654	606	Ordinary
1	UBS Securities Australia Ltd	suy	1,435	196	Ordinary
1	UBS Securities Australia Ltd	Buy	1,362	186	Ordinary
١	UBS Securities Australia Ltd	Buy	1,200	164	
	UBS Securities Australia Ltd	Buy	373	51	
Т	UBS Securities Australia Ltd	Buy	168	23	
1	UBS Securities Australia Ltd	Buy	2,738	374	
1	UBS Securities Australia Ltd	Buy	403	55	
	IOBS Securities Australia Ltd	Buy	366	50	Ordinary
ı	Upo Securities Australia Ltd	any	10,533	1,439	
7.Dec 14	Upo comition Australia Ita	any P.:.	110	15	
7-Dec-14	UBS Securities Australia Ltd	ony or	1,340	183	
7-Dec 14	Dia Biblio A serings Coo	buy	3,631	496	
7-Dec 14	USE Countries Australia Ltd	buy	16777	313	
1	UBS Securities Australia Ltd	ing.	985,9	006	Ordinary
1	HBC Countries Australia Ltd	buy B	4,363	596	ı
ı	I IDS Consisting Australia Lad	, ind	717	29	
П	I IRC Securities Australia 14d	ind.	0/1/5	433	1
1	LIBS Securities Australia 14d	Ser.	100	75	
ı	LIBC Countries Australia 14d	Din.	/=	91	
1	110C Commission Australia Ltd	no.	826	/11/	Ordinary
1	ups securities Australia Ltu	body	56	13	
Т	Up3 Securities Australia Ltd	Buy	4,958	9/9	Ordinary
1	Dos securities Australia LTO	buy	176	24	
-	UBS Securities Australia Ltd	Buy	580	79	
1	UBS Securities Australia Ltd	Buy	909'9	006	
-	UBS Securities Australia Ltd	Buy	1,652	225	
-	UBS Securities Australia Ltd	Buy	2,929	399	Ordinary
-	UBS Securities Australia Ltd	Buy	404	55	Ordinary
-1	UBS Securities Australia Ltd	Buy	2,092	285	Ordinary
1	UBS Securities Australia Ltd	Buy	12,155	1,656	Ordinary
-1	UBS Securities Australia Ltd	Sell	198	(27)	Ordinary
- 1	UBS Securities Australia Ltd	Buy	9,055	1,237	Ordinary
١	UBS Securities Australia Ltd	Buy	692'5	787	Ordinary
- 1	UBS Securities Australia Ltd	Buy	2,833	387	Ordinary
ı	UBS Securities Australia Ltd	Buy	483	99	Ordinary
	UBS Securities Australia Ltd	Buy	2		Ordinary
17-Dec-14	UBS Securities Australia Ltd	Buy	483	99	Ordinary
	UBS Securities Australia Ltd	Buy	366		Ordinary
ı	UBS Securities Australia Ltd	Buy	461		Ordinan
	UBS Securities Australia Itd	Ruv	12		Ordinary
l	UBS Securities Australia Itd	Buv	321		
	LIRS Securities Australia 14d	S. W.	0/1		
7-Dec-14	UBS Securities Australia Itd	Rus	5000		
1	URS Securities Australia (44)	Risc	(oc's		
7-Dec-14	LIRS Securities Australia 14d	Risc	202.0		Ì
7-Dec-14	HRS Securities Australia 144	Bush	coc'c		
7-Dec-14	IRS Societies Australia 14d	Bury	1/5/4		
7-Dec-1/	IBC Securities Australia Ind	900	180,1		
7 Doc 14	The Countries Australia Ltd	ouy.	300	200	
7 Day 14	obsider Abstraile LID	ana.	6,339	998	
T. Dec. 12	Securities Australia Liu	any.	816,1	792	
7-Dec-14	UBS Securities Australia Ltd	Buy	461	63	
7-Dec-14	UBS Securities Australia Ltd	Buy	886	135	
/-Dec-14	UBS Securities Australia Ltd	Buy	59	80	Ordinary
/-Dec-14	UBS Securities Australia Ltd	Buy	37	5	
7-Dec-14	UBS Securities Australia Ltd	Buy	73	10	
7-Dec-14	UBS Securities Australia Ltd	Buy	24,637	3,368	
7-Dec-14	UBS Securities Australia Ltd	Buy	4,389	009	
-Dec-14	UBS Securities Australia Ltd	Buy	56	13	
7-Dec-14	UBS Securities Australia Ltd	Buy	5,040	689	Ordinary
	UBS Securities Australia Ltd	Bry	086	134	Ordinary
Γ	IIDC Cocuration Australia 144	(50)	000	2	Oldstary
17-Dec-14	UBS Securities Australia Ltd	Buy	U8Z'L	172	
					Orginary

UBS Securities Australia Ltd Buy	5.263 467 1,225	722 64 64	Ordinary Ordinary Ordinary
Buy Buy	416	3,104	Ordinary
Buy	2,909	399	Ordinary
Buy	160	22	Ordinary
Buy	1,771	243	Ordinary
Buy	103,183	14,154	Ordinary
Buy	525	72	Ordinary
Buy	1 094	00/	Ordinary
Buy	211	29	Ordinary
Buy	1,094	150	Ordinary
Buy	2,916	400	Ordinary
and B	85	80 0	Ordinary
Buv	474	59	Ordinary
Buy	7		Ordinary
Buy	40.153	5.508	Ordinary
Buy	6,138	842	Ordinary
acy.	28,701	3,937	Ordinary
Buy	99	6	Ordinary
Buy	758	104	Ordinary
Buy	15,965	2,187	Ordinary
les	733,000	(100,000)	Ordinary
Buy	160	22	Ordinary
Buy	2,916	400	Ordinary
Buy	2,916	400	Ordinary
Ruy	2,916	400	Ordinary
euy Puy	204	28	Ordinary
(ng	49,662	6,803	Ordinary
, suy	2,000	274	Ordinary
gray.	554	2/2	Ordinary
Buy	503	69	Ordinary
Buy	1,750	240	Ordinary
Buy	459	63	Ordinary
Buy	3,055	419	Ordinary
Buy	4,461	612	Ordinary
any	255	35	Ordinary
Buy	3,594	493	Ordinary
Buy	3,339	458	Ordinary
Buy	2,916	400	Ordinary
Buy	6,452	882	Ordinary
Buy	7,370	1,011	Ordinary
Buy	416	57	Ordinary
Buv	5 132	707	Ordinary
Buv	50	12	Ordinan
Bita	30.249	0107	Ordinary
Blay	100		O-disease
Bine	200	17	orania)
6	000	Co Silver	Collary
600	7,300	755	Ordinary
600	756,05	757'4	Ordinary
gnà	18,210	2,498	Ordinary
Buy	24,385	3,345	Ordinary
Buy	3,404	797	Ordinary
Buv	755	×	Ordinan
and a	001	5	Commany
ing	67/	200	Ordinary
Buy	1,094	150	Ordinary
Buy	1,094	150	Ordinary
ATTR	7 285	1000	Ordinary
,	200 t	000,1	O'dillay.
fng	760'i	Uč.	Ordinary
Buy	32,609	4,470	Ordinary
Buy	11,315	1.551	Ordinary
Ris	OVC C	555	Ordinary
D D	646/2	244	Guinary
Paris,	DC.	0 8	Gulnary
and "	140	07	Ordinary
Buy	2,320	318	Ordinary
Buy	117	16	Ordinary

17-Dec-14 17-Dec-14	UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy	4,541	617	Ordinary
Dec-14	II IRS Securities Australia Ltd	ă			
700	On Security manual and	00)	6,204	843	Ordinary
	UBS Securities Australia Ltd	Buy	5,262	715	Ordinary
Jec-14	UBS Securities Australia Ltd	Buy	4,387	296	Ordinary
Dec-14	UBS Securities Australia Ltd	Buy	11,069	1,504	Ordinary
ec-14	UBS Securities Australia Ltd	Buy	3,165	430	Ordinary
Dec-14	UBS Securities Australia Ltd	Buy	21,086	2,865	Ordinary
ec-14	UBS Securities Australia Ltd	Buy	1,017	139	Ordinary
ec-14	UBS Securities Australia Ltd	Buy	4,733	947	Ordinary
ec-14	UBS Securities Australia Ltd	Buy	651	08	Ordinary
7-Dec-14	UBS Securities Australia Ltd	Buy	205	38	Ordina
7-Dec-14	UBS Securities Australia Ltd	Buv	10.973	100	Ordinary
7-Dec-14	UBS Securities Australia Ltd	Buv	4 055	25.7	Ordinary
7-Dec-14	UBS Securities Australia Ltd	Buv	6 171	273	Ordinary
7-Dec-14	UBS Securities Australia Ltd	Buy	3 631	907	Ordinary
7-Dec-14	LIBS Securities Australia 11d	Biss	1,000	064	Orginary
7-Dec-14	185 Securities Australia 14d	in a	4,839	199	Ordinary
.Dec. 14	198 Securities Australia 144	Birm	202.	1,b3,1	Ordinary
114	I I DC Cocuries Acceptable Ltd	bod.	4,143	266	Ordinary
1	Obsideration Australia Ltd	any	10,973	1,500	Ordinary
7-Dec 17	Libs commiss Australia Lib	Buy	730,097	31,434	Ordinary
	ODS SECURITIES AUSTRAING LICE	śna	8,638	1,180	Ordinary
Dec 14	Opp permittee Australia Liu	and a	520	71	Ordinary
2.Dec 14	Ups Counties Australia Ltd	buy b	132	82	Ordinary
1	Page Securities Australia Ltd	eug.	12,8/6	1,759	Ordinary
17-Dec-14	Libs committee Australia Ltd	buy.	3,243	443	Ordinary
Т	Opp Securities Australia Ltd	ons	11,353	1,551	Ordinary
Т	Obsideration Augustian Ltd	Buy	2,291	313	Ordinary
П	UBS Securities Australia Ltd	suy	527	72	Ordinary
Т	UBS Securities Australia Ltd	Buy	6,888	941	Ordinary
Т	UBS Securities Australia Ltd	Buy	1,025	140	Ordinary
	UBS Securities Australia Ltd	Buy	12,986	1,774	Ordinary
1	UBS Securities Australia Ltd	Buy	6,734	920	Ordinary
-1	UBS Securities Australia Ltd	Buy	1,025	140	Ordinary
- 1	UBS Securities Australia Ltd	Buy	58,904	8,047	Ordinary
	UBS Securities Australia Ltd	Buy	4,399	109	Ordinary
	UBS Securities Australia Ltd	Buy	1,427	195	Ordinary
1	UBS Securities Australia Ltd	Buy	65	8	Ordinary
ı	UBS Securities Australia Ltd	Buy	184.867	25.255	Ordinary
	UBS Securities Australia Ltd	Buy	15,972	2 182	Ordinary
7-Dec-14	UBS Securities Australia Ltd	Buv	3 499	478	Ordinan
7-Dec-14	UBS Securities Australia Ltd	Buv	2 284	212	Ordinary
7-Dec-14	UBS Securities Australia Ltd	Buy	3.631	716	Ordinary
7-Dec-14	UBS Securities Australia Ltd	Buy	2717	000	Ordinary
c-14	UBS Securities Australia Ltd	Buv	10,372	747	Ordinary
7-Dec-14	UBS Securities Australia Ltd	Buy	1025	140	Ordinary
c-14	UBS Securities Australia 11d	Buy	8598	1 100	Ordinary
7-Dec-14	UBS Securities Australia 14d	Buy	9,030	1.40	Ordinary
7-Dec-14	UBS Securities Australia Itd	Buy	620,1	11	Ordinary
14	LBS Securities Australia 14d	Brite	10		Ordinary
7-Dec-14	LIPS Securities Australia 144	an)	17,T	234	Ordinary
	Upo Securities Australia Liu	(ng	45,435	6,207	Ordinary
Dec 14	Obs Securities Australia Ltd	ouy o	702	96	Ordinary
Dec 14	Obs Securities Australia Ltd	ána d	/3,15/	10,001	Ordinary
2 Dec 14	Ubs Seturities Australia Ltd	euy	1,523	208	Ordinary
1 2	Obsideration Australia Ltd	Buy	886	135	Ordinary
7.05.14	UBS Securities Australia Ltd	pny	2,320	317	Ordinary
6-14	UBS Securities Australia Ltd	Buy	2,467	337	Ordinary
/-Dec-14	UBS Securities Australia Ltd	Buy	29,895	4,084	Ordinary
4	UBS Securities Australia Ltd	Виу	1,025	140	Ordinary
c-14	UBS Securities Australia Ltd	Buy	8,374	1,144	Ordinary
17-Dec-14	UBS Securities Australia Ltd	Buy	7,218	986	Ordinary
c-14	UBS Securities Australia Ltd	Buy	889	26	Ordinary
	UBS Securities Australia Ltd	Buy	952	130	Ordinary
	UBS Securities Australia Ltd	Buy	4,385	599	Ordinary
7-Dec-14	UBS Securities Australia Ltd	Buy	1.749	239	Ordinary
	UBS Securities Australia Ltd	Buy	8.667	1 184	Ordinary
ĺ	UBS Securities Australia 11d	Buy	24 827	NOC 5	Ordinary
	UBS Securities Australia Itd	Bir	350	160,0	Cidinaly
	LIBS Countries Australia 1+4	Ben	000	OC :	Ordinary
	Day Billian A using a coo	pod)	366	20	Ordinary
ı	195 Securities Australia Lid	buy	215,569	29,530	Ordinary
1	UBS Securities Australia Ltd	Buy	32,281	4,422	Ordinary
-	UBS Securities Australia Ltd	Buy	2,081	285	Ordinary
	UBS Securities Australia Ltd	Buy	207.196	78 383	O.d.
17-Dec-14			10011100		- Indipar

Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinan	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary
2,888	3,598	6,000	2,133	6	2,040	807	3,285	100	m	450	2	1.000	1 409	1 409	349	003.	2000	-	133	169	8	7.000	1,900	3,511	100	76	446	16/'/	2	88	1,127	737	2 2 2	7,20,0	88	1,133	10	7,940	38	446	1,470	61	9	т.	7	1 250	864	643	1,011	431	778	82	843	197	843	261,1	21	12	342	9	2,696	4,677	200	419	611	000	220 C
21,126	26,265	43,770	15,560	99	14,882	5,887	23,964	730	22	3,283	15	7.295	10.279	10.279	2546	10.043	1.050	7	026	1 233	693	51,065	13,861	25,613	730	708	45,25	20,0/9	15	642	8,221	5,376	20 00 00	387	584	8,265	73	57,962	2,407	3.254	10,724	139	44	22	15	908 0	9,090	4,713	7,411	3,159	5,703	623	6,179	1,444	6/1/9	8,759	154	88	2,509	44	19,816	34,376	307	3075	4.485	982	10 150
Buy	Buy B.::	auy a	suy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Buv	Biv	Buv	Buy	SUV	BUS	Buy	Buy	Buy	Buy	Buy	Buy	Buy Buy	Bliv	Buy	Buy	Buy	Buy	buy Buy	Buy	Buy	Buy	Buy	Suy.	Buy	Buy	Buy	Buy	Buy	Buy	buy Buy	Buv	Buy	Buy	Buy	Buy	3uy	Buy	Suy	Buy	Sur.	ouy Black	Ank	Suy	Suy	Buy	Juy	Buy	Riw	3	Buy	ASI	
alia Lto	alia Ltd	alia Ltd	alia Lto	slia Ltd	alia Ltd	ilia Ltd	ilia Ltd	tiia Ltd	liattd	ilia Ltd	lia Ltd	lia Ltd	lia Ltd	lia Ltd	lia Ltd	la Itd	lia Ltd	lia Ltd	lia Ltd	la Ltd	lia Ltd	lia Ltd	lia Ltd	lia Ltd	la Ltd	lia Ltd																																									
UBS Securities Austra	UBS Securities Austra	UBS Securities Austra	UBS Securities Austra	UBS Securities Austra	UBS Securities Austra	UBS Securities Austra	UBS Securities Austra	UBS Securities Australia Ltd	UBS Securities Austra	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Austra	UBS Securities Austra	UBS Securities Austra	UBS Securities Austra	LIBS Securities Australia Ltd	U8S Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Austra	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Austra	UBS Securities Austral	UBS Securities Austral	UBS Securities Austral	UBS Securities Austral	UBS Securities Austral	UBS Securities Austral	UBS Securities Austral	UBS Securities Australia Ltd	UBS Securities Austral	UBS Securities Austral	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Austral	UBS Securities Australia Ltd	UBS Securities Austral	UBS Securities Australia Ltd	UBS Securities Australia 11d	UBS Securities Australia Ltd	UBS Securities Australi	UBS Securities Australia Ltd	UBS Securities Australi	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australi	UBS Securities Australi	I IRS Securities Australi	UBS Securities Australi	UBS Securities Australi	UBS Securities Australi	UBS Securities Australia Ltd	UBS Securities Australia	UBS Securities Australia	UBS Securities Australia	UBS Securities Australia	UBS Securities Australia Ltd	UBS Securities Australia	I IDC Committee Australia
1/-Dec-14	17 Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-060-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14	17-Dec-14						

				4,504	Ordinary
UBS Se	UBS Securities Australia Ltd	Buy	4,617	602	Ordinary
ac can	curries Australia Ltd	Sell	1,548,000	(200,000)	Ordinary
e con	curties Australia (1d	Buy	33,531	4,412	Ordinary
US) Se	curties Australia Ltd	Buy	24,540	3,229	Ordinary
UBS Se	curities Australia Ltd	Buy	00	-	Ordinary
UBS Se	curities Australia Ltd	Buy	585	38	Ordinary
UBS Se	curities Australia Ltd	Buy	1.307	172	Ordinary
UBS Se	curities Australia Ltd	Buy	10.655	1.407	Ordinan
UBS Se	curities Australia Ltd	Buv	308 776	14 306	
IRS Se	Curities Australia 11d	Bus	3000	2000	Commercia
S val	Control America Inch	600	000.0	400,	Ordinary
	San Australia Ltd	200	73	8	Ordinary
ŝ	curries Australia Ltd	Buly	76	10	Ordinary
8 8	cunties Australia Ltd	Buy	11,119	1,463	Ordinary
UBS Se	curities Australia Ltd	Buy	10,563	1,403	Ordinary
UBS Se	curities Australia Ltd	Buy	11.962	1 574	Ordinan
UBS Se	curties Australia Ltd	Buy	10.640	1 400	Ordina
UBS Se	Curities Australia Itd	Buy	9000	324 4	in in it
IRS Se	Parities Australia 14d	Ruiv	Out.	2,1	oldina.
2000	Courses Australia Lid	, and	Učš.	46	Ordinary
200	UBS Securities Australia Ltd	Buy	91	12	Ordinary
UBS Se	Securities Australia Ltd	Buy	54,408	7,159	Ordinary
UBS Se	UBS Securities Australia Ltd	Buy	17.081	2 2 4 9	Ordinary
UBS Se	UBS Securities Australia Ltd	Buy	25.006	3 286	Ordinary
UBS Se	UBS Securities Australia Ltd	Silv	13515	205.0	
300	Countries Australia et al	500	402,12	20,72	Ordinary
60	Dos securites Australia cro	any	24,283	3,193	Ordinary
SS SS	UBS Securities Australia Ltd	Buy	15,187	1,997	Ordinary
UBS Se	UBS Securities Australia Ltd	Buy	15	2	Ordinary
UBS Se	cunties Australia Ltd	Buv	46	4	Ordinan
UBS Se	cunties Australia I to	200	26.333	70.0	Contract of the contract of th
I IBS Co.	I IBC Countries Australia 1+d	0	267,02	3,4447	Ordinary
200	Collines Australia Lid	buy	22,39b	2,943	Ordinary
8	cunties Australia Ltd	Buy	25,904	3,404	Ordinary
UBS Sec	curties Australia Ltd	Buy	167	22	Ordinary
UBS Se	curities Australia Ltd	Buy	027.720	12 841	Ordinary
UBS Sec	curities Australia I td	Buy	100	30	1000
3000	continue Australia ( au		06:	07	Clarky
	Collines Australia Lito	Suy	80/	53	Ordinary
000	curries Australia Ltd	Buy	6,415	843	Ordinary
JUBS Sec	curities Australia Ltd	8uy	31,003	4.074	Ordinary
UBS Sec	curities Australia Ltd	Buy	131 554	17 287	Ordinan
UBS Sec	curities Australia I td	Buy	000	901	1
IRS Se	Curities Australia I tol	Biss	072 514	200	Colliany
IRC Ce	IRS Securities Australia 14d	100	0.000	C76'41	General
3	Commission Assessed in the Land	600	4,240	228	Ordinary
	Course Agricultural	(no	3,135	715	Ordinary
082 X6	JBS Securities Australia Ltd	Buy	3,904	513	Ordinary
UBS Sec	JBS Securities Australia Ltd	Buy	1,157	152	Ordinary
UBS Sec	UBS Securities Australia Ltd	Buy	578	76	Ordinary
UBS Sec	Securities Australia Itd	200	07.23	0.00	0
loc con	The state of the s	(20)	17//21	12,838	Ordinary
200	Des securities Australia Ltd	eny	91,238	12,005	Ordinary
UBS Sec	JBS Securities Australia Ltd	Buy	17,066	2,247	Ordinary
UBS Sec	JBS Securities Australia Ltd	Buy	49,854	6.564	Ordinary
UBS Sec	JBS Securities Australia Ltd	Buy	2 964	Š	Ordinan
100 Ce	(DC Counties Australia 24d	Box	1000	2	Cicinaly
200	Control Control	000	12,183	1,603	Ordinary
NBS Se	UBS Securities Australia LTd	suy	8,185	1,077	Ordinary
UBS Sec	curties Australia Ltd	Buy	3,466	456	Ordinary
UBS Sec	JBS Securities Australia Ltd	8uy	8,026	1,056	Ordinary
UBS Sec	JBS Securities Australia Ltd	Buy	129	17	Ordinary
UBS Sec	JBS Securities Australia Ltd	Buy	570	75	Ordinary
UBS Sec	Turities Australia It d	Riv	4 788	60	
	Day of the control of	ío d	4,700	930	Ordinary
Dac con	Culties Australia LLG	end.	7,204	790	Ordinary
oes sec	Curines Australia Ltd	Buy	43,198	5,684	Ordinary
OBS Se	Curties Australia Ltd	Buy	62,416	8,218	Ordinary
UBS Sec	curities Australia Ltd	Buy	46,337	6,101	Ordinary
UBS Sec	curities Australia Ltd	Buy	3.770	496	Ordinary
UBS Sec	curities Australia Ltd	Buy	7 205	976	Ordinary
UBS Sec	Turities Australia Itcl	NI S	0.014	1 100	
JRS Se	LIBS Securities Australia 11d	21.00	000	20,5	Coursely
l RS Se	unities Australia (14	Burk	217 53	2,00	Countries
I IDC Co.	unition Australia Lea	- C	202.01	0,2,0	Orginary
600	Unities Abstralla Ltd	ing	49,757	6,547	Ordinary
UBS Sec	unties Australia Ltd	Buy	10,237	1,347	Ordinary
UBS Sec	curities Australia Ltd	Buy	8	-	Ordinary
UBS Sec	UBS Securities Australia Itd	Buv	14.410	300 1	Ordinary
200	Control Address in Lead	600	014,41	068,1	Ordinary
or can	dithes Australia Liu	Áng	73,4/p	3,091	Ordinary
9 9 9	curities Australia Ltd	Buy	113 000	14 888	Ordina
100				f pop't)	CELLETY
200	curities Australia Ltd	Buy	471	629	Ordinary

-		;;	002 13	8.400	Ordinary
	UBS Securities Australia Ltd	kng	0.30, 10		
	Securities	8uy	2,073	284	Ordinary
	UBS Securities Australia Ltd	ony Bun	2,073	1887	Ordinary
	Securities	B <sub>10</sub>	1300	7000	Ordinary
	Seconda	, in a	ODE'/	000'1	Ordinary
	35 Securities Australia Ltd	éna P	700,02	7,740	Ordinary
	SS Securities Australia Itd	Riv	36.471	900 /	Ordinary
	35 Securities Australia Ltd	Buy	306.184	41 943	Ordinary
	UBS Securities Australia Ltd	Buy	241,316	33,057	Ordinary
	35 Securities Australia Ltd	Buy	65,547	8,979	Ordinary
	BS Securities Australia Ltd	Buy	1000'21	2,000	Ordinary
	35 Securities Australia Ltd	Buy	179	25	Ordinary
	BS Securities Australia Ltd	Buy	13,064	1,822	Ordinary
	BS Securities Australia Ltd	Buy	14,534	2,027	Ordinary
	UBS Securities Australia Ltd	Buy	14,591	2,035	Ordinary
	UBS Securities Australia Ltd	Buy	205	70	Ordinary
	35 Securities Australia Ltd	Buy	11,199	1,563	Ordinary
	BS Securities Australia Ltd	Sell	416	(28)	Ordinary
	SS Securities Australia Ltd	Sell	20	6	Ordinary
	UBS Securities Australia Ltd	Sell	416	(28)	Ordinary
	BS Securities Australia Ltd	Sell	20	8	Ordinary
	RS Securities Australia I td	3	OCE.	(43)	Ordinary
	S Securities Australia 14d	100	127	(014)	Crolings
	as securities Australia Liu	1175	100	(001)	Cumary
	UBS Securities Australia Ltd	) ell	956	(133)	Ordinary
	bs securities Australia Ltd	ne c	SOS	(43)	Ordinary
	JBS Securties Australia Ltd	Sell	955	(133)	Ordinary
	JBS Securities Australia Ltd	Sell	416	(28)	Ordinary
	UBS Securities Australia Ltd	Sell	416	(28)	Ordinary
	UBS Securities Australia Ltd	Sell	1,134	(158)	Ordinary
	UBS Securities Australia Ltd	Sell	1,429	(199)	Ordinary
	35 Securities Australia Ltd	Sell	698	(121)	Ordinary
	35 Securities Australia Ltd	Sell	1,946	(172)	Ordinary
	S Securities Australia Ltd	Sell	237	(33)	Ordinary
	35 Securities Australia Ltd	Sel	2.053	(786)	Ordinary
	35 Securities Australia Ltd	Seli	502 1	(238)	Ordinary
	S Securities Australia Itd	Sell	909	(063)	Ordings
	20 Countries Australia 14d	la de	903 C	(06)	Countries
	LIBS Securities Australia Itd	le S	283	(305)	Ordinary
	Of Cocumition Australia 14d	Coll	5574	(601)	Coding
	OC Cocurition Authorities 184	Coll	1221	(ECC)	o delialy
	or Committee Australia to a	les.	1771	(/47)	Ordinary
	35 Securities Australia Ltd	liac i	495	(FQ)	Ordinary
	35 Securities Australia Ltd	Sell	22	(3)	Ordinary
	35 Securities Australia Ltd	Sell	860	(120)	Ordinary
	35 Securities Australia Ltd	Seli	158	(22)	Ordinary
	iS Securities Australia Ltd	Sell	495	(69)	Ordinary
	is Securities Australia Ltd	Sell	143	100	Ordinary
	is Securities Australia Ltd	Sell	43	(9)	Ordinary
	IRS Securities Australia 1th	Sell	150	(0)	Ordinary
	Counties Australia Ind	100	20110	(02.1.5)	Ordinary.
	so securities Australia Lid	100	601,42	(3,1/3)	Ordinary
	UBS Securities Australia Ltd	Buy	15,040	2,000	Ordinary
	35 Securities Australia Ltd	Seli	100,624	(13,361)	Ordinary
	UBS Securities Australia Ltd	sell	31,770	(4,160)	Ordinary
	35 Securities Australia Ltd	Buy	53,901	7,044	Ordinary
	3S Securities Australia Ltd	Buy	17,428	2,283	Ordinary
	UBS Securities Australia Ltd	Buy	440,245	57,759	Ordinary
	35 Securities Australia Ltd	Seil	5,848	(292)	Ordinary
	UBS Securities Australia Itd	Seil	448	(54)	Ordinany
	IBS Securities Australia I to	Sell	F449 U23	(50 906)	Ordinary
	C Cocurities Australia 144	2	162 64	00000	Ordinary.
	Cop Securities Adadaila Lid	fng	150,21	100'6	Ordinary
	so perdittes Australia Lito	200	245,942	(17,636)	Ordinary
	UBS Securities Australia Ltd	===	010,62	(3,852)	Ordinary
	5 Securities Australia Ltd	Sell	789,788	(52,373)	Ordinary
	UBS Securities Australia Ltd	Seli	681,532	(89,765)	Ordinary
	S Securities Australia Ltd	Sell	84,648	(11,149)	Ordinary
	JBS AG Australia Branch	Buy	408,940	53,492	Ordinary
	JBS AG Australia Branch	Sell	2,110,559	(278,011)	Ordinary
	S Wealth Management Australia Itd	Se	17 328	(2.280)	Ordinary
	INS Wealth Management Australia 14d	Pun	2767	(207)	Ordinary
	S Wealth Management Australia Liu	pon)	4,7/3	070	Organary
	UBS Securities Australia Lto	fond	57	3	Ordinary
	S Securities Australia Ltd	Buy	14,331	1,866	Ordinary
	S Securities Australia Ltd	Biv.	400		
		-	691	22	Ordinary

Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinan	Ordinan	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinan	Ordinan	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary	Ordinary								
4,385	80	4224	405	1 107	4	2 120	150	367	843	535.5	448	278	120	150	843	150	199	180	2,101	2	797	414	1 497	262	405	707	33	1,514	2,980	5,479	7987	1 963	700.2	2,113	266	2995	3,528	1,822	956	388	150	283	983	2 + 72	3,404	51.5	2.155	3,481	2,155	2,960	2,146	5,100	2,562	3,153	533	2,121	135	119	104	-	104	185	426	330	353	513
33,282	19	32.060	3.074	8 402	114	1 139	1.139	3.770	6.407	25.604	3.405	6.407	412	1.139	6.407	1.139	5,024	1,368	15,957	15	6,057	3, 140	11 377	6019	3,078	5,350	251	11,506	22,648	41,640	277,72	26,062	15.405	16,059	4,304	4,304	26,830	13,865	7.270	2,951	1,141	7,476	1,4/6	365.31	75.887	16 366	16.389	26,473	16,389	22,511	16,320	38,786	19,484	23,979	4,056	18,130	1.027	906	792	8	792	1,408	3,242	2,511	2,686	3,909
Buy	Buy	Buy	Buy	Buy	Buv	Buy	ank	S S S S S S S S S S S S S S S S S S S	Buv	Buy	Buy	Buy	Buy	Buy	Buy	Buy	Birk	Buv	Buy	Buy	Buy	Buy	Buy	Sur.	Buy                           Buy	Buy	Buy	Buy	Buy	any	Buy	Buy	Buy																																	
UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	USS Securities Australia 11d		UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia 11d	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	IBS Securities Australia Itd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	JBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	UBS Securities Australia Ltd	IBS Securities Australia Ltd	IBS Securities Australia Ltd	IBS Securities Australia Ltd	IBS Securities Australia Ltd	155 Securities Australia Ltd	DBS Securities Australia Ltd	BS Securities Australia Ltd	UBS Securities Australia Ltd	BS Securities Australia Ltd	UBS Securities Australia Ltd	BS Securities Australia Ltd	85 Securities Australia Ltd	55 Securities Australia Ltd	bs securities Australia Ltd	85 Securities Australia Ltd	85 Securities Australia Lto
	18-Dec-14	l		1	l					18-Dec-14		ĺ					18-Dec-14		Ī	18-Dec-14	T	Ī	Ī	Ī	18-Dec-14 ∪		ĺ			1	8-Dec-14	Ĺ		8-Dec-14	i	-	8-Dec-14	-		П	-	8-Dec-14			l								18-Dec-14 UI			18-Dec-14			-	8-Dec-14 UE	1		ı	ı		18-Dec-14 UB

	מבי היינונים ביינונים	600	/50,52	760,8	Ordinary
ec-14	UBS Securities Australia Ltd	Buv	18 977	7 407	Ordinan
ec-14	UBS Securities Australia Ltd	Buy	25 607	7.316	Ordinary
ec-14	UBS Securities Australia Ltd	Buy	26,666	3,511	Ordinary
ec-14	UBS Securities Australia Ltd	Buy	26,393	3,475	Ordinary
ec-14	UBS Securities Australia Ltd	Buy	31,403	4,132	Ordinary
ec-14	UBS Securities Australia Ltd	Buy	38	5	Ordinary
PEC 14	UBS Securities Australia Ltd	Buy	82,787	10,893	Ordinary
PC-14	UBS Securities Australia Ltd	Buy	53	7	Ordinary
Pec-14	UBS Securities Australia Itd	Buy	188,77	3,6/1	Ordinary
bec-14	UBS Securities Australia Ltd	Buy	920	121	Ordinary
18-Dec-14	UBS Securities Australia Ltd	Buy	28	1.	Ordinary
Jec-14	UBS Securities Australia Ltd	Buy	524	69	Ordinary
ec-14	UBS Securities Australia Ltd	Buy	89,619	11,792	Ordinary
ec-14	UBS Securities Australia Ltd	Buy	2,675	352	Ordinary
Jec-14	UBS Securities Australia (td	Buy	213	28	Ordinary
Jec. 14	UBS Securities Australia Ltd	Buy	3,504	461	Ordinary
Dec-14	UBS Securities Australia Ltd	Buv	534	35/	Ordinary
18-Dec-14	UBS Securities Australia Ltd	Buy	243	33	Ordinary
Jec-14	UBS Securities Australia Ltd	Buy	18.643	2 453	Ordinary
Jec-14	UBS Securities Australia Ltd	Buy	89,201	11,737	Ordinary
Dec-14	UBS Securities Australia Ltd	Buy	22,101	2,908	Ordinary
18-Dec-14	UBS Securities Australia Ltd	Buy	524	69	Ordinary
8-Dec-14	Upo Securities Australia Ita	Buy	160	21	Ordinary
8-Dec-14	URS Securities Australia 11d	Buy	140.00	468	Ordinary
Dec-14	UBS Securities Australia Ltd	Suv	10,045	17 950	Ordinary
Dec-14	UBS Securities Australia Ltd	Buy	1 292	170	Ordinary
Jec-14	UBS Securities Australia Ltd	Buy	2,835	373	Ordinary
Jec-14	UBS Securities Australia Ltd	Buy	24,320	3,200	Ordinary
Jec-14	UBS Securities Australia Ltd	Buy	471	62	Ordinary
Jec-14	UBS Securities Australia Ltd	Buy	631	83	Ordinary
41.0	Upo comition Australia Ltd	euy n	631	83	Ordinary
Dec. 14	IIIRC Countries Australia Ind	in a	808/8	1,159	Ordinary
Dec-14	UBS Securities Australia (1d	(c) (c)	1 004	٥	Ordinary
18-Dec-14	UBS Securities Australia Ltd	Buv	3118	144	Ordinary
Dec-14	UBS Securities Australia Ltd	Buy	54.577	2 407	Ordinary
ec-14	UBS Securities Australia Ltd	Buy	187	22	Ordinary
Pec-14	UBS Securities Australia Ltd	Buy	631	77	Ordinary
ec-14	UBS Securities Australia Ltd	Buy	17.412	2 291	Ordinary
ec-14	UBS Securities Australia Ltd	Buy	2,538	334	Ordinary
ec-14	UBS Securities Australia Ltd	Buy	3,542	466	Ordinary
ec-14	UBS Securities Australia Ltd	Buy	882	116	Ordinary
ec-14	UBS Securities Australia Ltd	Buy	6,848	106	Ordinary
8-Dec-14	UBS Securities Australia Ltd	Buy	23	3	Ordinary
ec-14	UBS Securities Australia Ltd	suy	1,953	257	Ordinary
8-Dec-14	ILIBS Securities Australia Ltd	áng Marin	8,026	1,056	Ordinary
8-Dec-14	LIRS Securities Australia Ind	buy Bur	84,314	11,094	Ordinary
ec-14	UBS Securities Australia 11d	Buv	242	83	Ordinary
ec-14	UBS Securities Australia Ltd	Buy	25,618	3 372	Ordinary
18-Dec-14	UBS Securities Australia Ltd	Buy	18,494	2,435	Ordinary
ec-14	UBS Securities Australia Ltd	Buy	26,036	3.428	Ordinary
ec-14	UBS Securities Australia Ltd	Buy	204,037	26,847	Ordinary
ec-14	UBS Securities Australia Ltd	Buy	144	19	Ordinary
ec-14	UBS Securities Australia Ltd	Buy	19,334	2,544	Ordinary
ec-14	UBS Securities Australia Ltd	Buy	2,683	353	Ordinary
ec-14	UBS Securities Australia Ltd	Buy	137	18	Ordinary
ec-14	UBS Securities Australia Ltd	Buy	1,140	150	Ordinary
ec-14	UBS Securities Australia Ltd	Buy	8,031	1,056	Ordinary
ec-14	UBS Securities Australia Ltd	Buy	686,108	90,218	Ordinary
18-Dec-14	UBS Securities Australia Ltd	Buy	45,136	5,935	Ordinary
4 :	UBS Securities Australia Ltd	Buy	38	5	Ordinary
PC-14	UBS Securities Australia Ltd	Buy	38	5	Ordinary
4 2	UBS Securities Australia Ltd	Buy	132,616	17,438	Ordinary
4 5	UBS Securities Australia Ltd	auy	37,903	4,984	Ordinary
14	LIBS Securities Australia Ltd	Buy	31,431	4,133	Ordinary
	Con reconners was also tro	àna	14707		
	In the Commission Assessed in Land		70*/65	5,189	Ordinary

UBS Securities Australia Ltd   Buy	960 151 1544 1516 22,335 22,335 21,159 21,160 30,884 30,884 30,884 31,985	126 2, 933 2, 933 2, 933 2, 933 2, 933 2, 933 2, 933 2, 933 2, 933 3, 900 3, 900 1, 1, 623 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1
	22,315 22,315 22,316 22,316 2,116 2,116 335 335 335 335 335 335 335 335 335 33	2.993 2.78 2.78 2.78 2.78 2.78 2.666
	1,180 1,180 1,180 1,180 1,180 1,180 1,180 1,180 1,180 1,180 1,180 1,140 1,140 1,141 1,	252 278 278 278 278 278 278 278 278 278 27
	2,175  1,644  1,169  80,894  80,894  31,895  80,894  3,895  1,1590  1,10,89  1,10,99  1,114,150  1,	2.82 2.84 1.650 1.
	1,180 487 487 487 483 1,180 1,180 1,180 1,180 1,180 1,180 1,180 1,180 1,180 1,198 8,833 1,296 1,141 20,163 1,141 20,163 1,141 20,163 1,141 20,163 1,141 20,163 1,141 20,163 1,141 20,163 1,141 20,163 1,141 20,163 1,141 20,163 1,141 20,163 2,683 2,683 2,683 2,683 2,683 2,683 2,683 3,704 3,704 2,801	10,630 10,630 11,630 11,630 11,630 11,630 12,000 13,000 13,000 13,000 13,000 13,000 13,000 13,000 13,000 13,000 13,000 13,000 13,000 13,000 13,000 13,000 13,000 14,335 10,000
	1,180 487 80,884 111,409 111,409 110,611 110,611 111,419 1	155 (167 ) 1
	4.00  4.00	10,630 14,739 14,739 14,739 1,1623 1,
	80, 844 80, 844 11, 1098 8, 893 11, 1098 11, 1098 11, 10611 12, 1061 11, 10611 12, 1061 11, 10611 12, 1061 11, 10611	10,630 1,630 1,167 1,187
	90.894 90.894 90.894 10.894 111,098 8.893 11,590 11,590 11,590 11,590 11,590 11,590 11,590 11,590 11,190 11	1,653 1,653
	25.41 11.038 8.833 8.833 8.833 11.530 11.530 11.651 20.66 20.635 20.683 20.883	14,299 1,162
	111,090 111,090 111,090 111,090 11,090 1,000 1,0	14,599 1,167
	11,088  8,88  8,80204  76  1,408  2,208  1,0611  20,004  114,150  11,013  2,033  2,033  3,034  30,038  30,038  2,038  30,038  30,038  2,038  30,038  2	14,599 1,1627 1,1629 1,
	11,409 11,409 1,409 1,409 1,00	1,167 1,167 1,83 1,969 1,453 1,453 1,150 1,500 1
	1,590 1,1590 1,1690 1,1690 2,286 2,286 3,0,163 2,0,163 2,0,163 3,1,500	1653 1653 166 170 170 170 170 170 170 170 170 170 170
	1408 2.266 2.266 30,204 76 10,611 10,611 10,611 11,611 20,163 11,4150	300 300 300 100 100 100 115,000
	2.286 2.296 7.60 7.60 7.60 7.60 7.60 7.60 7.60 7.6	300 309 100 100 100 100 100 100 100 1
	30,204  76  76  76  76  77  78  78  78  78  78	3,969 10 10 14,535 1,455 1,455 1,500
	76 609 110,613 110,613 110,613 114,150	14,550 15,000 15
	609 110,611 20,103 114,150 114,150 114,150 114,150 114,150 114,150 114,150 114,150 11,141 26,701 26,703 20,404 30,508 20,808 30,508 30,508 30,508 30,508 30,508 30,508 30,508 30,508 30,508 20,406 20,400 20,	14,335 2,646 15,000 15,
	10,617 20,163 20,163 20,163 30,14150 114,150 38,050 31,952 20,440 11,140 11,140 20,035 31,952 20,440 31,952 20,440 31,952 20,440 31,952 20,440 31,952 20,440 31,952 20,440 31,952 20,440 31,952 20,440 20,036 31,952 20,038	14.535 2.646 15.000 15.000 15.000 15.000 15.000 1000
	20,103 129 4 4 14,150 114,150 114,150 114,150 114,150 114,150 114,150 114,150 114,150 114,150 114,150 114,150 114,150 11,263 12,633 2,633 2,633 2,633 2,633 30,400 1,563	15,000 15
	14,150 114,150 114,150 114,150 114,150 114,150 114,150 114,150 114,150 114,150 11,411 11,41 11,411 11,411 11,411 11,411 11,411 11,411 11,411 11,411 11,411 1	15,000 15,000 15,000 15,000 15,000 15,000 15,000 1000 1
	1.29 4.40 1.14.150 1.	17 15,000 15,000 15,000 15,000 15,000 15,000 105 110
	114,150 114,150 114,150 114,150 114,150 114,150 114,150 11,41 11,41 11,41 11,41 11,41 11,41 11,41 11,41 11,41 11,41 11,41 11,42 12,643 13,64 13,64 14,66 14,60 14,60 14,60 14,60 14,60 14,60 14,60 17,	15,000 15,000 15,000 5,000 10,
	114,150 114	15,000 15,000 15,000 5,000 192 192 193 193 194 195 195 195 195 195 195 195 195 195 195
	114,150 14,150 18,030 18,030 1,141 1,141 20,430 20,430 20,430 20,430 30,538 30,538 30,538 30,538 30,538 30,538 30,538 30,538 20,736	15,000 5,000 792 319 319 105 2,886 2,886 3,512 3,512 3,512 3,512 3,512 3,512 3,512 3,512 4,612 6,612 3,512 4,612 6
	144150 144150 146050 14050 14050 14050 14050 14050 14050 150	15,000   5,000   5,000   319   319   105   2,686   2,686   3,511   3,511   3,512   3,522   3
	38,000 6,035 2,431 2,431 20,745 2,683 2,683 2,683 2,683 2,683 2,683 2,683 2,683 2,683 3,004 304 304 304 304 307 41,283 30,588 30,588 2,20 2,00	5,000 719 719 719 710 710 710 710 710 710 710 710
	6.035 6.035 8.00 31,902 20,440 1,141 1,141 26,703 26,703 26,703 26,703 26,703 30,4 30,4 30,4 30,4 30,4 30,4 30,4 30,4 30,4 30,4 30,4 30,4 30,4 30,6	792 105 106 106 106 106 106 106 106 106 106 106
	2,472 2,472 800 31,902 20,440 1,144 1,144 26,705 2,633 2,633 2,633 2,633 2,633 410 395 304 304 304 305 12,563 41,263 31,263 41,263 31,263 41,263 31,263 41,263 32,046 22,046 22,046 22,048 22,048 22,048 22,048 22,048 22,048 22,048 22,048 22,048 22,048 22,048 22,048 22,048 22,048 22,048 22,048 22,048 22,048 22,048 23,048 23,048 23,048 23,048 23,048 23,048 24,068 24,068 26,078 26,078 26,078 26,078 26,078 27,078 28,0	319 4 200 2 686 2 686 3 511 3 511 3 51 3 52 3 53 3 53 3 53 4 40 6
	2,421 2,431 2,431	105 4200 4200 105 150 150 150 150 150 150 1
	31,902 20,440 1,440 1,440 1,440 26,741 26,742 26,743 26,743 26,743 26,743 26,743 26,743 26,743 26,743 26,744 26,114 2	4,200 2,686 1567 157 3,522 3,822 3,822 3,822 3,822 3,822 3,822 3,822 3,822 3,822 3,822 3,822 3,822 3,822 4,000 6,0
	20,490 20,400 20,400 26,78 28,888 2,683 2,683 2,683 394 395 394 395 394 395 394 395 396 41,283 39,298 45,600 45,600 45,600 45,600 45,600 220 45,600 220 45,600 220 45,600 220 45,600 220 45,600 220 45,600 220 45,600 220 45,600 220 45,600 220 220 220 220 220 238,714 28,018	2,086 150 150 151 152 3,51 3,52 3,53 3,53 3,53 3,53 3,53 3,53 3,53
	26.785 26.785 26.785 26.785 26.785 26.888 26.883 2.683 2.683 2.683 30.588 30.588 30.588 30.588 30.588 30.588 30.588 22.683 22.683 22.801 28.018 28.01	2,086 150 3,511 3,512 3,801 77 77 3,801 54 54 54 50 6,046 6,
	26.705 26.705 26.705 26.83 26.838 26.83 2.863 2.863 304 304 304 304 304 304 304 304 304 30	3,517 3,522 3,522 3,801 3,53 3,53 3,53 3,53 3,53 3,53 3,53 3,5
	26.701 26.701 26.703 26.888 28.888 28.888 30.633 410 410 410 41.83	3,511 3,522 3,801 70 3,801 3,53 3,53 5,2 40 5,04 6,046 1,653 1,653 4,056 1,653 4,056 1,653 4,056
	28,883 28,883 2,683 2,683 3,563 304 395 304 30,508 30,508 30,508 30,508 30,508 22,018 22,018 22,018 22,018 22,018 22,018 22,018 22,018 23,018	3,801 70 70 353 353 353 54 60 60 60 60 60 60 60 60 60 60 60 60 60
	28.888 28.888 59.82 2.683 30.4 30.5 30.4 30.5 30.4 30.5	3.801 70 70 353 353 54 60 60 1,653 1,653 4,036 1,653
	2.683 2.683 410 395 305 305 30.58 61,150 61,150 61,150 61,150 720 45,630 45,630 12,633 7,04 28,016 2	70 353 353 353 54 60 40 80,46
	2.683 2.683 410 395 384 384 385 38,1263 31,283 31,283 32,704 28,018 22,037 28,018 28,0	353 353 54 40 40 8,046 1,653 5,432 4,036
	2,683 410 396 396 396 30,6150 61,150 12,563 41,263 41,263 30,598 22,038 22,038 24,660 7,90 7,	353 54 52 50 60 60 8,046 1,653 7,053 7,053
	410 395 304 304 30,156 41,263 41,263 30,598 30,598 45,600 45,600 20,476 28,018 28,018 28,018 28,018 28,018 28,018 28,018 28,018 28,018 28,018 28,018 28,018 28,018 29,281 1,018 28,	54 40 50 8,046 1,653 5,432 6,036
	395 304 304 305 12,633 14,183 4,183 4,183 20,536 20,0476 20,0476 20,0476 38,704 2,638 1,038 2,675 2,675 1,036 2,675 2,675 1,036 2,675 2,75	52 40 50 8,046 1,653 5,432 5,432
	304 307 61,150 12,563 41,283 30,598 30,598 220 45,500 45,500 20,018	8,046 1,653 5,432 5,432
	380 61,150 12,563 12,563 12,038 20,047 20,048 2	50 8,046 1,653 5,432
	17,563 17,563 41,263 41,263 42,563 45,563 45,660 46,660 28,704 28,704 28,704 28,704 28,704 28,704 28,704 28,704 1,018 2,21,811 1,018 2,21,811 2,21,811 1,018 2,21,811 1,018 2,21,811 1,018 2,21,811 1,018 2,21,811 1,018 1,018 2,21,811 1,018 1,018 2,21,811 1,018 1	8,046 1,653 5,432
	12,563 10,583 30,588 30,088 20,476 20,018 20,018 20,018 20,018 20,018 22,018 22,018 22,018 24,660 790 790 790 790 790 790 790 79	1,653
	41,233 20,588 20,598 20,598 21,538 21,538 28,704 28,704 28,704 28,704 28,704 28,704 28,704 23,716 1,517 1,517 2,517 1,51	5,432
	220 46,203 46,203 46,203 46,203 46,00	5,432
	20,588 220 45,630 45,630 20,476 28,014 28,014 28,014 28,014 1,018 23,211 1,018 23,211 1,018 23,615 1,036	4026
	220 45,520 1038 20,478 20,478 28,704 28,704 28,704 28,704 28,704 28,704 28,704 28,704 28,704 28,704 28,704 31,335 31,332 31,332	4,020
	2 (338 2 (338 2 (338 2 (346) 3 (376) 3 (376) 2 (378) 2 (378) 2 (378) 1 (366) 2 (378) 1 (366) 2 (378) 1 (366) 2 (378) 3 (378) 3 (378) 1	53
	2 ( ) ( ) ( ) ( ) ( ) ( ) ( ) ( ) ( ) (	6,004
	2,038 20,476 38,704 28,018 28,018 2,808 1,078 2,075 1,008 2,075 1,406 2,38,716 2,075 1,406 2,38,716 1,036 31,338	2
	20,476 38,704 28,018 28,018 24,860 790 1,018 232,811 13,612 1,055 1,036 238,716 1,036 31,332	2.770
	28,704 28,018 28,018 28,018 29,0 790 790 28,23,811 2,675 2,675 2,675 2,675 31,335 31,335	7 696
	28,018 22,860 24,860 1,018 23,211 1,562 1,562 1,562 1,036 1,036 31,332 17,328	2005
	20,010 20	260,0
	24,860 790 790 1,018 232,811 1,406 238,716 238,716 11,328 11,332	2000
	74860 790 1,018 232,811 1,3612 1,2675 2,675 2,675 1,036 31,332 17,328	S.
	1,790 1,018 222,811 1,862 2,675 1,406 238,716 1,0396 31,335 1,7328	3,271
	232,811 13,612 13,675 1,406 238,716 31,335 17,328	104
	232,811 13,612 2,615 1,406 238,716 10,936 31,332 17,328	134
	13,612 2,675 1,406 238,716 10,336 31,335 17,328	30.633
	2,675 1,406 238,716 10,336 31,335 17,328	1 791
	1,406 238,716 10,936 31,335 17,328	35.3
	238,716 10,936 31,335 17,328	101
	236,710 10,936 31,335 17,328	001
	10,935 31,335 17,328	01410
	31,335	1,439
Securities Australia Ltd	17,328	4,123
		2,280
JBS Securities Australia Ltd Buv	73 302 1	3 066
	25,025 25,045	2,446
	CHC,C2	0.410
	37,610	4,952
	45,570	6,000
	000.00	000
ממי מרכתו ורכי שתפונים דות	20,000	4,000

8-Dec-14	LIBS Securities Australia 11d	Biv	707 517		
8-Dec-14	UBS Securities Australia Ltd	Buy	228,150		Ordinary
18-Dec-14	ĕ	Buy	228,150	30,000	
Dec-14	UBS Securities Australia Ltd	Buy	162,732		
Dec-14	UBS Securities Australia Ltd	Buy	26,701		
18-Dec-14	Securities	Buy	26,701		
Dec-14	Securities	Buy	26,473		
Dec-14	UBS Securities Australia Ltd	Buy	64,217		Ordinary
8-Dec-14	UBS Securities Australia Ltd	Buy	37,865		
0-Dec-14		euy euy	38,185		
Dec 14	Those securities Abstrated Ltd	and Bir	52,825		
10 Dec 14	Tipe Committee Australia Ltd	oc.	161,602		
1000	LIDE Countries Australia Ltd	ind.	1/5,854	5/,//4	
19 Dec 14	Libe countries Australia Ltd	ána d	5,354		Ordinary
Dec 14	Tipo Countries Australia Ltd	body B	005,08/	100,000	
11.00	Social Securities Australia Ltd	on's	44,231		İ
0-Dec-14	Obs Securities Australia Ltd	ank	51,448		
1	Obsideration Australia Ltd	ána a	52,277		Ordinary
8-Dec-14	UBS Securities Australia Ltd	Buy	199,380		
8-Dec-14	UBS Securities Australia Ltd	Buy	53,235		
8-Dec-14	UBS Securities Australia Ltd	Buy	26,701		Ordinary
9-Dec-14	Obs securities Australia Ltd	ank	94,066		
P.Dec.14	Upo Counties Australia Ltd	buy Bur	82,78	10,886	
8-Dec-14	Hos Countries Australia Ind	buy Day	32,169		Ordinary
8-Dec-14	IDS Countries Australia Ind	pon)	989/09		
8.Dec-14	IIBS Securities Australia Ind	Bin	8,329		
8-Dec-14	LiRS Securities Australia 14d	Buy	326 660 6		Ordinary
8-Dec-14	LIBS Securities Australia 14d	R IN	905,220,2		
8-Dec-14	URS Securities Australia 144	Bin	9,2,6		
18-Dec-14	LIBS Securities Australia (14	Bis	67,12		
8-Dec-14	IDC Countries Australia (+d	Bib	776'14		Ordinary
	The Countries Australia Ltd	Duy.	/18,414	34,280	Ordinary
18-Dec-14	TIRC Countries Australia 14d	B by	102,05		
	TIPE Countries Australia I tal	Dust	304,800		
ı	Obsider Australia Ltd	lou)	435,534		
	USS SECURITIES AUSTRAINS LIQ	acy	/57/1	2,270	
1000	Upo Commission A mariful and	ouy D	000,182		
ı	DI Securities Australia Dia	boy.	816.7		
8-Dec-14	UBS Securities Australia Ltd	buy.	158,111	14,6/6	Ordinary
ı	Opp Securities Australia Ltd	puy.	01777	067	Ordinary
ı	Coo Securities Australia Ltd	pon	2,195		Ordinary
10-Dec-14	Ubb Securities Australia Ltd	eny	2,202		Ordinary
١	UBS Securities Australia Ltd	Buy	76,200		Ordinary
-	UBS Securities Australia Ltd	Sell	1,905,000		Ordinary
18-Dec-14	UBS Securities Australia Ltd	Sell	104,130		Ordinary
ı	UBS Securities Australia Ltd	Buy	240,000		
- 1	UBS Securities Australia Ltd	Sell	25,500	(3,000)	
	UBS Securities Australia Ltd	Sell	25,500		
	UBS Securities Australia Ltd	Sell	155.000		Ordinary
l	UBS Securities Australia Ltd	Sell	100 880	(13 000)	Ordinary
	LIBS Securities Australia I to	Riv	47 500		Countaly
	IDC Committee Australia 144	Coll	42,500		Ordinary
l	ODS SECULIARS AUStralia LIG	llac	7,182,424		Ordinary
18-Dec-14	UBS Securities Australia Ltd	Buy	66,672		Ordinary
8-Dec-14	UBS Securities Australia Ltd	Buy	1,690	223	Ordinary
18-Dec-14	UBS Securities Australia Ltd	Buy	6,072	801	Ordinary
	UBS Securities Australia Ltd	Buy	31,628	4,167	Ordinary
	UBS Securities Australia Ltd	Buy	1.459	761	Ordinary
	UBS Securities Australia 11d	BUV	1715	300	Ordinan
18-Dec-14	IIBS Securities Australia Ltd	200	200 5	077	Cidinary
12.5	IDC Countries Australia (40)	62	5,855	300	Cromary
T	Obs securities Australia Ltd	ani	524	69	Ordinary
8-Dec-14	UBS Securities Australia Ltd	Buly	5,533	729	Ordinary
	UBS Securities Australia Ltd	Buy	6,452	850	Ordinary
	UBS Securities Australia Ltd	Buy	6,945	915	Ordinary
	UBS Securities Australia Ltd	Buy	6,945	915	Ordinary
8-Dec-14	UBS Securities Australia Ltd	Buy	388	15	Ordinary
8-Dec-14	UBS Securities Australia Ltd	Buy	6,604	698	Ordinary
	UBS Securities Australia Ltd	Buy	6999	020	Ordinary
Γ	UBS Securities Australia Ltd	Buy	6 954	015	Ordinary
Ī	IRS Securities Australia 14d	Riv	3 808	102	diam's
Ī	IDC Countries Australia 144	(52)	ono's	IOC :	Ordinary
18-Dec-14	IRC Societies Australia I tol	Bin.	2, 140	414	Ordinary
T	DC Socurities Australia I+4	100	705,307	3,018	Ordinary
T	obs securities Australia Ltd	buy	6,323	832	Ordinary
T	JBS Securities Australia Ltd	Buy	6,566	864	Ordinary
18-Dec-14	UBS Securities Australia Ltd	Buy	2 561	755	Ordinary
7	מבים והכי נוסיו מוסי מוסי מוסי	ina)	1100'7	100	Ordinary

	Obsideration Australia LTD				
19-Dec-14	UBS Securities Australia Ltd	Buy	9,426	1,237	Ordinary
	UBS Securities Australia Ltd	Buy	2,309	303	Ordinary
-	UBS Securities Australia Ltd	Buy	3,147	413	Ordinary
-	UBS Securities Australia Ltd	Buy	23	m	Ordinary
	UBS Securities Australia Ltd	Buy	8,039	1,055	Ordinary
١	UBS Securities Australia Ltd	Buy	46,878	6,152	Ordinary
ŀ	UBS Securities Australia Ltd	Buy	152	20	Ordinary
1	UBS Securities Australia Ltd	Buy	762	100	Ordinary
1	UBS Securities Australia Ltd	Buy	8,908	1,169	Ordinary
	UBS Securities Australia Ltd	BUy	11,186	1,468	Ordinary
	UBS Securities Australia Ltd	Buy	66	13	Ordinary
	UBS Securities Australia Ltd	Buy	335	44	Ordinary
-	UBS Securities Australia Ltd	Buy	4,549	265	Ordinary
1	UBS Securities Australia Ltd	Buy	3,574	469	Ordinary
Į	UBS Securities Australia Ltd	Buy	3,734	490	Ordinary
	UBS Securities Australia Ltd	Buy	792	104	Ordinary
	UBS Securities Australia Ltd	Buy	2.903	381	Ordinary
	UBS Securities Australia Ltd	Buv	g	1	Caloro
	UBS Securities Australia Ltd	Buy	2000	2 00	Ordinary
l	UBS Securities Australia Ind	Briv	02.3	25	Colonial
l	UBS Securities Australia Ital	Bin	0	٥١٠	Ordinary
ĺ	I IBS Securities Australia 144	Bree	0 110	-   5	Ordinary
19.Dec.14	LIDE Controller Australia 144	600	7007	37	Ordinary
1	Obs securities Australia Ltd	hay	343	45	Ordinary
1	UBS Securities Australia Ltd	Buy	343	45	Ordinary
	UBS Securities Australia Ltd	Buy	4,221	554	Ordinary
9-Dec-14	UBS Securities Australia Ltd	Buy	312	41	Ordinary
9-Dec-14	UBS Securities Australia Ltd	Buy	1,081	142	Ordinary
	UBS Securities Australia Ltd	Buy	2/29	68	Ordinary
	UBS Securities Australia Ltd	Buy	221	200	Ordinan
19-Dec-14	UBS Securities Australia Ltd	Buy	342	44	Ordinal y
l	UBS Securities Australia Itd	Bliv	240	130.	Cidiliary
ļ	UBS Securities Australia 11d	Bliv	3,047	1,294	Ordinary
Ł	HDC Committee Australia 144	600	2,400	324	Ordinary
١	OBS Securities Australia Ltd	pan	21,979	2,892	Ordinary
1	UBS Securities Australia Ltd	Buy	3,856	202	Ordinary
1	UBS Securities Australia Ltd	Buy	768,995	100,918	Ordinary
1	UBS Securities Australia Ltd	Buy	75,878	088'6	Ordinary
-	UBS Securities Australia Ltd	Sell	123	(91)	Ordinary
i	UBS Securities Australia Ltd	Buy	2,341	306	Ordinary
	UBS Securities Australia Ltd	Buy	57,168	7,473	Ordinary
i	UBS Securities Australia Ltd	Buy	6.571	859	Ordinary
ı	UBS Securities Australia Ltd	Buy	7 222	700	Ordinan
ı	UBS Securities Australia Ltd	Buy	1 308	171	Ordinan
	UBS Securities Australia Ltd	Buv	7001	143	O'dinay.
	UBS Securities Australia Ltd	Buy	1 033	,	Ordinary
l	UBS Securities Australia Itd	Billy	10011	150	Ordinary
l	UBS Securities Australia I1d	Biv	1,904	200	Ordinary
1	UBS Securities Australia Itd	Buy	11157	190	Ordinary
	LIRS Securities Australia 14d	Pin	4 2 2 4	140	Ordinary
9-Dec-14	URS Securities Australia 144	fna	4,544	295	Ordinary
ı	Upo securities Australia Ltd	euy	168	22	Ordinary
l	UBS Securities Australia Ltd	Buy	1,293	169	Ordinary
١	UBS Securities Australia Ltd	Buy	2,272	297	Ordinary
9-Dec-14	UBS Securities Australia Ltd	Buy	237	31	Ordinary
	UBS Securities Australia Ltd	Buy	612	80	Ordinary
9-Dec-14	UBS Securities Australia Ltd	Buy	9,700	1,268	Ordinary
	UBS Securities Australia Ltd	Buy	4,628	909	Ordinary
	UBS Securities Australia Ltd	Buy	742	97	Ordinary
	UBS Securities Australia Ltd	Buy	59,999	7.843	Ordinary
19-Dec-14	UBS Securities Australia Ltd	Buy	1,278	167	Ordinary
	UBS Securities Australia Ltd	Buy	1086	142	Ordinan
	UBS Securities Australia Ltd	Biv	2007	24.0	Collina,
	UBS Securities Australia Itcl	Riv	1150	7 .	Orginary
	INC Securition Australia (44)	600	011	2	Ordinary
	Obsideration Augustical Ltd	one o	1/,388	2,273	Ordinary
ĺ	One Securities Australia Ltd	eu,	10	xo	Ordinary
1	Obsiseruntes Australia LIO	Buy	8,599	1,124	Ordinary
ı	UBS Securities Australia Ltd	Buy	3,794	496	Ordinary
	UBS Securities Australia Ltd	Buy	3,068	401	Ordinary
	UBS Securities Australia Ltd	Buy	2,525	330	Ordinary
	UBS Securities Australia Ltd	Buy	18,475	2,415	Ordinary
	UBS Securities Australia Ltd	Buy	5,531	723	Ordinary
	UBS Securities Australia Ltd	Buy	18 873	237.0	
ı	מבים המבים המבים המבים המבים	2			
			200	7,49/	Ordinary
1	UBS Securities Australia Ltd	Buv	11.020	1,407	Ordinary

	I 190 A.C. Landon Brownith	les	193,140	(20,491)	Ordinary
9-Dec-14	USS ACT A LEGICAL BLANCH				
9-Dec-14	1185 AG Australia Branch	Sol	20,491	7/9/7	Ordinary
19-Dec-14	UBS AG Australia Branch	Sell	910 439	(000,00)	Ordinary
9-Dec-14	UBS AG Australia Branch	Sell	149 245	(10,020)	Ordinary
14	URS AG Australia Branch	Transfer-In	C#7,841	1,000 510	Ordinary
19-Dec-14	UBS AG Australia Branch	dS.	020 000	(127 104)	Ordinary
6.14	UBS AG Australia Branch	Sell	SK 738	(10,17)	Ordinary
c-14	UBS AG Australia Branch	Sell	922	(118)	Ordinary
c-14	UBS Wealth Management Australia Ltd	Sell	1 981	(350)	Coding
c-14	UBS Wealth Management Australia Ltd	Sell	3,549	(464)	Ordinary
c-14	UBS Wealth Management Australia 1td	Sell	3 207	(424)	Ordinary
6.14	UBS Wealth Management Australia Ltd	Sell	900 9	(107)	Ordinary
6-14	UBS Wealth Management Australia Ltd	Biv	000,0	200	Signal S
ı	UBS Wealth Management Australia Ltd	Sell	43.057	(5,670)	Ordinary
L	URS Wealth Management Australia (14)	Bix	3 300	(50,00)	, and a
1	URS Securities Australia Ltd	Bird	502,6	074	Oroinary
ı	HRS Sequities Australia 144	Rin	100	9 5	Ordinary
П	1190 Cocurities Australia 14d	n d	/04	502	Ordinary
	Tipo Cocurities Australia Lita	bod.	5,332	758	Ordinary
9.Dec.14	URS Securities Australia 14d	Black	312	4 (	Ordinary
	IIIR Securities Australia 14d	8.5	20,000	1747	Croinary
П	LIDE Countries Australia 14d	100	30,023	2000	Ordinary
1	IIRS Securities Australia 14d	(a)	200	ý lý	Ordinary
ı	IIIRS Securities Australia Ital	66	2,473	575	Ordinary
	LIRS Securities Australia Itd	Biv	2 724	200	Crainary
L	UBS Securities Australia Itd	Blay	24 499	0000	Cidinary
9-Dec-14	UBS Securities Australia Inc	Bliv	17 507	505.5	Ordinary
ı	188 Securities Australia Ind	2 2	100'0	20207	Ordinary
ı	LIBS Societies Australia (14	Bush	2,073	2/3	Ordinary
9-Dec-14	All Committee Augustic Ltd	600	141.	061	Ordinary
П	Tipo Complete Australia Ltd	900	1,841	955,	Ordinary
Ì	USS Securities Australia Ltd	and a	141,1	150	Ordinary
ı	ODS Securities Adstrain LLC	, and	/,564	\$20	Ordinary
3-Dec-14	Obsiderations Australia Ltd	any.	3,526	464	Ordinary
9-Dec-14	OBS Securities Australia Ltd	auy.	205	27	Ordinary
1	Obs securities Australia Ltd	and,	198	56	Ordinary
1	Obs securities Australia Lto	an)	198	56	Ordinary
9-Dec-14	UBS Securities Australia Ltd	Buy	53	7	Ordinary
9-Dec-14	UBS Securities Australia Ltd	kink	312	41	Ordinary
	Obs securities Australia Ltd	buy.	434	۶/	Ordinary
9-Dec-14	UBS Securities Australia Ltd	gn)	91	12	Ordinary
4	UBS Securities Australia Ltd	suy	30	4	Ordinary
3-Dec-14	UBS Securities Australia Ltd	Buy	167	22	Ordinary
9-Dec-14	UBS Securities Australia Ltd	Buy	4,733	622	Ordinary
9-Dec-14	UBS Securities Australia Ltd	Buy	5,319	669	Ordinary
>14	UBS Securities Australia Ltd	Buy	753	66	Ordinary
19-Dec-14	UBS Securities Australia Ltd	Buy	36,704	4,820	Ordinary
19-Dec-14	UBS Securities Australia Ltd	Buy	3,914	514	Ordinary
14	UBS Securities Australia Ltd	Buy	1.783	234	Ordinary
-14	UBS Securities Australia Ltd	Buy	1 722	300	Cadinary
514	I IBS Securities Australia I tol	Renv	1001	227	Countries
	LIDS Securities Augustin 14d	fna	183	477	Ordinary
1	Upo Securities Australia Ltd	any	4,071	535	Ordinary
<u>.</u>	UBS Securities Australia Ltd	Buy	18,543	2,435	Ordinary
-14	IUBS Securities Australia Ltd	Buy	1,020	134	Ordinary
-14	UBS Securities Australia Ltd	Buy	1,615	212	Ordinary
-14	UBS Securities Australia Ltd	Buy	831	109	Ordinary
14	UBS Securities Australia Ltd	Buy	10,289	1.352	Ordinary
9-Dec-14	UBS Securities Australia Ltd	Buy	1 142	150	Ordinary
44	UBS Securities Australia I td	Buy	7 503	102	Ordinary
-Dec-14	UBS Securities Australia Itd	B. P.	30.754	2003	Commany
9-Dec-14	IIIDS Cocurring Australia 14d	500	59,/54	717'C	Ordinary
9-Dec-14	TIDE Countries Australia 14d	Bin	4,473	/90	Gramary
<u>.</u>  ;	Obs Securities Australia Ltd	hna	4,168	547	Ordinary
9-Dec-14	UBS Securities Australia Ltd	Buy	4,991	655	Ordinary
4	UBS Securities Australia Ltd	Buy	17,663	2,318	Ordinary
١	UBS Securities Australia Ltd	Buy	11,567	1,518	Ordinary
	UBS Securities Australia Ltd	Buy	1,991	655	Ordinary
19-Dec-14	UBS Securities Australia Ltd	Buy	no		Ordinary
	UBS Securities Australia I td	Buv	3 520	137	Ordinary
	I IBC Securities Australia I to	B. E.	020'6	707	Ordinary
l	FIRE Compiliar Australia 144	6.0		7	Crainary
ı	ora Security Australia Billion	no.	3,147	413	Ordenary
1	Obsideration Australia Ltd	ana	2.74		
			CF1,21	\$ 0.	Ordinary
	UBS Securities Australia Ltd	Buy	16	1,534	Ordinary

Observation Authority Letter         Bayer         3155         150         150           Observation Authority Letter         Bayer         3143         188         188         0.00           Observation Authority Letter         Bayer         3143         3143         388         0.00           Observation Authority Letter         Bayer         3143         3143         389         0.00           Observation Authority Letter         Bayer         3143         3143         389         0.00           Observation Authority Letter         Bayer         3143         3143         389         0.00           Observation Authority Letter         Bayer         3143         3143         389         0.00           Observation Authority Letter         Bayer         3143         3143         389         0.00           Observation Authority Letter         Bayer         3143         3143         389         0.00           Observation Authority Letter         Bayer         3143         3143         389         0.00           Observation Authority Letter         Bayer         3143         3143         389         0.00           Observation Authority Letter         Bayer         3143         3143         3143	UBS Securities Australia Ltd UBS Securities Australia Ltd	Buy	760 6	487	Ordinary
By         64         64         681         68           By         681         3143         389           By         681         343         389           By         681         343         389           By         400         4008         408           By         400         4008         408           By         400         3143         389           By         400         3143         389           By         41757         389           By         41757         389           By         41757         389           By         41757         389           By         41757         389           By         41757         389           By         41757         389           By         41757         389           By         41757         389           By         41757         389           By         41757         389           By         41757         389           By         41758         389           By         41758         389           By         41	UBS Securities Australia Ltd		5,935		Ordinary
8y         \$143         \$189           9y         \$144         \$184         \$184           9y         \$142         \$189           9y         \$143         \$189           9y         \$143         \$189           9y         \$143         \$189           9y         \$143         \$189           9y         \$143         \$189           9y         \$143         \$189           9y         \$143         \$189           9y         \$143         \$189           9y         \$143         \$189           9y         \$143         \$189           9y         \$143         \$189           9y         \$143         \$189           9y         \$143         \$189           9y         \$143         \$189           9y         \$143         \$189           9y         \$143         \$189           9y         \$143         \$189           9y         \$143         \$189           9y         \$143         \$189           9y         \$144         \$143           9y         \$144         \$144           9y		Виу	99	80	Ordinary
By         6 81         681         681           By         By         3143         389           By         400         3143         389           By         400         3143         389           By         400         3143         389           By         3143         389           By         3143         389           By         3143         389           By         3143         389           By         3143         389           By         3143         389           By         3143         389           By         3143         389           By         3143         389           By         3143         389           By         41757         340           By         41757         340           By         41757         340           By         41757         340           By         41757         340           By         3143         3143           By         3143         3143           By         3143         3144           By         3143<	UBS Securities Australia Ltd	Buy	3,143	389	Ordinary
By         3143         3184         388           By         409         37         128           By         408         408         628           By         3143         389           By         3143         389           By         3143         389           By         3143         389           By         3143         389           By         3143         389           By         3143         389           By         3143         389           By         3143         389           By         3143         389           By         3143         389           By         3143         389           By         4142         389           By         4142         389           By         4142         389           By         4142         389           By         4142         389           By         4142         389           By         4142         389           By         4142         389           By         4142         389	UBS Securities Australia Ltd	Buy	6,811	843	Ordinary
By         99         12           By         94         408         12           By         143         289           By         3142         289           By         3142         289           By         3142         389           By         3142         389           By         3142         389           By         3222         400           By         3222         400           By         3222         400           By         3222         400           By         3222         400           By         3222         400           By         3222         400           By         3222         400           By         3222         400           By         3222         322           By         3222         324           By         3222         324           By         3222         322           By         3222         322           By         3222         322           By         3222         322           By         3222 <th< td=""><td>UBS Securities Australia Ltd</td><td>Buy</td><td>3,143</td><td>389</td><td>Ordinary</td></th<>	UBS Securities Australia Ltd	Buy	3,143	389	Ordinary
By         A 9008         4 96           By         A 9008         4 96           By         3 143         3 89           By         3 143         3 89           By         3 143         3 89           By         3 143         3 89           By         3 143         3 89           By         3 143         3 89           By         4 1727         5 168           By         5 415         3 89           By         5 415         100           By         5 415         100           By         5 415         100           By         5 415         5 168           By         5 415         5 168           By         5 210         5 169           By         2 22         3 5           By         2 22         3 5           By         2 22         3 5           By         3 139         3 6           By         3 139         3 6           By         3 139         3 6           By         3 139         3 6           By         3 139         3 139           By	UBS Securities Australia Ltd	Buy	46	15	Ordinary
By         3143         389           By         3142         389           By         3142         389           By         3142         389           By         3123         400           By         3123         400           By         3123         400           By         3123         400           By         3123         400           By         3123         400           By         3123         400           By         3123         400           By         3123         400           By         3123         400           By         3123         400           By         3123         400           By         3123         322           By         3123         322           By         3123         324           By         3123         324           By         3123         324           By         3123         324           By         3123         324           By         3123         324           By         3123         324	UBS Securities Australia Ltd	Buy	4,008	496	Ordinary
By         3177         387           By         3142         389           By         3143         389           By         3143         389           By         3143         389           By         604         322         400           By         604         5415         755           By         604         5415         755           By         604         5415         604           By         5415         604         755           By         604         5415         604           By         5224         556           By         5220         350           By         5224         526           By         5222         35           By         5222         35           By         5222         35           By         5222         35           By         5220         52           By         5221         52           By         5221         52           By         52         52           By         52         52           By         52 <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>3.143</td> <td>386</td> <td>Ordinary</td>	UBS Securities Australia Ltd	Buy	3.143	386	Ordinary
By         3143         389           By         3142         389           By         6084         6084         673           By         6084         673         600           By         6084         712         600           By         617         671         600           By         27,470         715           By         27,470         671           By         27,470         671           By         27,470         671           By         27,470         671           By         28,270         671           By         28,270         674           By         28,270         671           By         28,270         1,000           By         28,270         1,000           By         28,270         1,000           By         28,270         1,000           By         28,270         1,000           By         28,270         1,000           By         2,023         3,244           By         2,023         3,244           By         2,023         3,244           By </td <td>UBS Securities Australia Ltd</td> <td>Buy</td> <td>3,127</td> <td>387</td> <td>Ordinary</td>	UBS Securities Australia Ltd	Buy	3,127	387	Ordinary
By         3143         389           By         3143         389           By         6084         753           By         6084         753           By         6084         753           By         6084         753           By         5415         756           By         5415         750           By         5415         750           By         527,0         540           By         27,470         100           By         27,242         55           By         27,242         55           By         27,242         55           By         27,242         55           By         27,243         52           By         27,243         52           By         27,243         52           By         27,244         52           By         27,244         52           By         27,244         52           By         27,244         52           By         27,242         52           By         27,242         52           By         27,242	UBS Securities Australia Ltd	Buy	3,143	389	Ordinary
Bay         3,122         400           Bay         3,122         400           Bay         41,757         5,183         400           Bay         41,757         5,183         400           Bay         27,470         1,00         1,00           Bay         27,470         3,404         1,00           Bay         27,470         3,404         1,00           Bay         2,23,75         1,00         1,00           Bay         2,27,00         3,404         1,00           Bay         2,27,00         3,404         1,00           Bay         2,270         3,60         1,00           Bay         2,270         3,60         3,70           Bay         2,270         3,60         3,60           Bay         2,270         3,60         3,60           Bay         2,270         3,60         3,60           Bay         2,23         3,70         3,60           Bay         3,73         4,185         3,60           Bay         2,24         3,50         3,60           Bay         3,72         3,60         3,60           Bay         3,7	UBS Securities Australia Ltd	Buy	3 143	088	ording.
Bay         6,004         753           Bay         6,004         753           Bay         5,415         5,68           Bay         1,291         1,60           Bay         1,291         1,60           Bay         27,420         3,54           Bay         27,420         3,64           Bay         27,420         3,64           Bay         27,242         3,5           Bay         27,242         3,5           Bay         27,242         3,5           Bay         27,242         3,5           Bay         27,242         3,5           Bay         27,243         8,2           Bay         2,2         3,5           Bay         2,2         3,5           Bay         2,2         3,5           Bay         2,2         3,5           Bay         2,2,2         3,5           Bay         2,2,2         3,5           Bay         2,2,2         3,5           Bay         2,2,4         3,7           Bay         2,2         3,5           Bay         3,1,3         3,2           Ba	UBS Securities Australia Ltd	8uv	3 232	CO.	Cidinary
Buy         6.7         6.0           Buy         5.415         6.7           Buy         5.415         6.71           Buy         5.415         6.71           Buy         5.415         6.71           Buy         5.240         3.400           Buy         5.82         3.5           Buy         5.82         3.5           Buy         2.82         3.5           Buy         2.82         3.5           Buy         2.82         3.5           Buy         2.82         3.5           Buy         2.82         3.5           Buy         2.82         3.5           Buy         2.82         3.5           Buy         2.82         3.5           Buy         2.82         3.5           Buy         2.82         3.5           Buy         3.1         3.8           Buy         3.1         3.8           Buy         3.1         3.8           Buy         3.1         3.8           Buy         3.1         3.8           Buy         3.1         3.8           Buy         3.1 <td>UBS Securities Australia Ltd</td> <td>818</td> <td>2027</td> <td>3 5</td> <td>Cidilary</td>	UBS Securities Australia Ltd	818	2027	3 5	Cidilary
607         617         617         617         618           849         617         610         611         611         611         611         611         611         611         611         611         611         611         611         611         611         611         611         611         612	UBS Securities Australia Itd	Bits	1000	66/	Ordinary
84y         5/415         6/10           84y         5/415         6/11           84y         5/415         100           84y         5/415         100           84y         5/415         100           84y         5/415         100           84y         5/415         5/614           84y         5/415         5/614           84y         5/415         5/614           84y         5/415         5/614           84y         5/415         5/415           84y         5/415         5/42           84y         5/415         5/42           84y         5/42         5/42           84y         5/42         5/42           84y         5/42         5/42           84y         5/42         5/42           84y         5/44         5/50           84y         5/44         5/50           84y         5/45         5/50           84y         5/45         5/50           84y         5/45         5/50           84y         5/45         5/50           84y         5/50         5/50	ISS Societities Australia 14a	600	41,737	2,158	Ordinary
84y         27,470         160           84y         27,470         3,404           84y         1,291         1,60           84y         23,73         6,614           84y         23,73         6,614           84y         23,73         6,614           84y         28,27         35           84y         28,27         35           84y         28,27         6,33           84y         28,27         6,33           84y         28,27         6,33           84y         28,27         6,33           84y         2,345         3,20           84y         3,135         389           84y         3,135         3,89           84y         3,123         3,89           84y         3,123         3,89           84y         3,128         3,89           84y         3,128         3,89           84y         3,128         3,89           84y         3,128         3,89           84y         3,128         3,24           84y         3,128         3,89           84y         3,128         3,128	The contract August 100	and a	5,415	671	Ordinary
Buy         53.74.0         3.404           Buy         27.470         3.404           Buy         23.77         6.614           Buy         23.27         6.614           Buy         23.27         6.614           Buy         28.2         3.5           Buy         28.2         3.5           Buy         28.2         3.70           Buy         28.2         3.70           Buy         28.2         3.70           Buy         28.2         3.70           Buy         28.2         3.70           Buy         2.20         6.53           Buy         3.139         3.89           Buy         3.139         3.89           Buy         3.139         3.89           Buy         3.139         3.89           Buy         3.139         3.89           Buy         3.139         3.84           Buy         3.139         3.89           Buy         3.139         3.89           Buy         3.139         3.89           Buy         3.139         3.84           Buy         3.139         3.84 <t< td=""><td>Open Securities Australia Ltd</td><td>buy</td><td>807</td><td>100</td><td>Ordinary</td></t<>	Open Securities Australia Ltd	buy	807	100	Ordinary
Bay   1,291   160	UBS Securities Australia Ltd	Buy	27,470	3,404	Ordinary
Buy         53,375         6644           Buy         53,375         6644           Buy         72,194         8,346           Buy         72,194         8,346           Buy         72,194         8,346           Buy         72,194         8,346           Buy         72,194         8,346           Buy         72,194         8,346           Buy         7,787         963           Buy         8,070         1,000           Buy         8,070         1,000           Buy         8,070         1,000           Buy         7,787         9639           Buy         7,787         9639           Buy         7,787         9639           Buy         7,787         9639           Buy         7,787         9639           Buy         7,787         9639           Buy         7,784         9639           Buy         7,784         9639           Buy         7,784         9639           Buy         7,784         9639           Buy         7,784         9639      Entre Into Derivative         84,791         1,	UBS Securities Australia Ltd	Buy	1,291	160	Ordinary
Buy         782         35           Buy         782         35           Buy         782         35           Buy         72,194         8,946           Buy         72,265         37,00           Buy         22,855         6,03           Buy         3,139         389           Buy         3,139         389           Buy         3,139         389           Buy         3,139         389           Buy         3,139         389           Buy         3,139         389           Buy         3,139         389           Buy         3,139         389           Buy         7,782         9,639           Buy         7,782         9,639           Buy         7,782         9,639           Buy         7,782         9,639           Buy         7,782         9,639           Buy         7,782         9,639           Buy         7,782         9,639           Buy         7,782         9,639           Buy         7,782         9,639           Buy         8,4791         1,625	UBS Securities Australia Ltd	Buy	53.375	6.614	Ordinary
Buy         782         35           Buy         72,194         8,346           Buy         72,194         8,346           Buy         29,359         3,73           Buy         29,359         3,73           Buy         29,359         3,73           Buy         807         1,000           Buy         807         1,000           Buy         807         1,000           Buy         807         1,000           Buy         807         1,000           Buy         807         1,000           Buy         807         1,000           Buy         807         1,000           Buy         807         1,000           Buy         807         1,000           Buy         807         1,000           Buy         807         1,000           Buy         2,054         3,34           Buy         2,054         3,50           Buy         2,054         3,50           Buy         2,054         3,00           Buy         2,044         3,00           Sell         1,15,419         1,02	UBS Securities Australia Ltd	Buy	282	×.	Ordinary
Buy         72,754         8.46           Buy         5,270         6.33           Buy         5,270         6.34           Buy         5,270         6.35           Buy         5,270         6.37           Buy         2,985         3,703           Buy         8,070         1,000           Buy         8,070         1,000           Buy         8,070         1,000           Buy         8,070         1,000           Buy         8,070         1,000           Buy         8,070         1,000           Buy         8,070         1,000           Buy         8,070         1,000           Buy         8,070         1,000           Buy         8,070         1,000           Buy         8,070         1,000           Buy         2,584         380           Buy         2,023         3,500           Buy         2,023         3,500           Sell         1,15,419         1,4320           Buy         2,024         3,500           Correction         1,13,400         3,500           Entre Into Derivative         <	UBS Securities Australia Ltd	Buy	787	7%	Ordinar.
Buy   72,194   8,346   8,946   8,946   8,946   8,946   8,946   8,946   8,946   8,946   8,946   8,946   8,946   9,946	IRS Sequirities Australia 14d	Blay	202	6	Cidillary
Buy   7,2194   8,346     Buy   2,985   3,700     Buy   2,985   3,700     Buy   2,985   3,700     Buy   2,985   3,700     Buy   2,985   3,700     Buy   8,070   1,000     Buy   8,070   1,000     Buy   8,070   1,000     Buy   8,070   1,000     Buy   8,070   1,000     Buy   8,070   1,000     Buy   8,070   1,000     Buy   8,070   1,000     Buy   8,070   1,000     Buy   2,023   3,85     Buy   2,023   3,85     Buy   2,023   3,85     Buy   2,023   3,85     Buy   2,023   3,85     Buy   2,023   3,85     Buy   3,103   1,15,40   1,42,20     Buy   1,15,419   1,42,20     Buy   1,15,419   1,42,20     Buy   1,15,419   1,42,20     Buy   1,15,419   1,43,20     Buy   1,15,419   1,43,20     Buy   1,15,419   1,43,20     Buy   1,15,419   1,43,20     Buy   1,15,419   1,43,20     Buy   1,15,419   1,43,20     Buy   1,15,419   1,43,20     Buy   1,13,40   1,43,20     Buy   1,13,40   1,43,20     Buy   1,13,40   1,43,20     Buy   1,13,20   1,13,20     Buy   1,13,20	I IDC Countries A research 1 and	per)	797	S	Ordinary
Buy         29,859         3,700           Buy         29,859         3,700           Buy         289         3,700           Buy         289         3,700           Buy         28,700         1,000           Buy         807         1,000           Buy         28,459         7,244           Buy         28,459         7,244           Buy         28,459         7,244           Buy         28,459         7,244           Buy         28,459         7,244           Buy         807         1,000           Buy         28,459         7,244           Buy         2,265         334           Buy         31,233         334           Buy         31,253         359           Buy         2,055         334           Buy         2,054         35,000           Buy         2,053         3,40           Sell         115,413         4,40           Sell         115,413         4,40           Sell         11,541         4,40           Sell         11,541         4,40           Sell         11,541 <t< td=""><td>Opp Securities Australia Ltd</td><td>had</td><td>72,194</td><td>8,946</td><td>Ordinary</td></t<>	Opp Securities Australia Ltd	had	72,194	8,946	Ordinary
Buy   29859   3/202     Buy   29859   3/202     Buy   2,139   3/39     Buy   2,139   3/39     Buy   2,139   3/39     Buy   2,139   3/39     Buy   2,245   3/39     Buy   2,254   3/39     Buy   2,253     Buy   3,264   3/39     Bu	UBS Securities Australia Ltd	Buy	5,270	653	Ordinary
Buy   Sep   73	UBS Securities Australia Ltd	Buy	29,859	3,700	Ordinary
Buy   3,139   389     Buy   8,070   1,000     Buy   8,071   1,000     Buy   8,071   1,000     Buy   8,071   1,000     Buy   7,787   9,639     Buy   7,787   9,639     Buy   7,787   9,639     Buy   7,787   9,639     Buy   7,787   9,639     Buy   7,787   3,800     Buy	UBS Securities Australia Ltd	Buy	685	73	Ordinary
Bayy         16         2           Buy         8,070         1,000           Buy         8,070         1,000           Buy         8,070         1,000           Buy         8,070         1,000           Buy         8,070         1,000           Buy         8,070         7,244           Buy         7,728         9,639           Buy         7,728         9,639           Buy         7,728         9,00           Buy         31,23         387           Buy         3,123         389           Buy         3,128         389           Buy         2,254         366           Buy         2,254         368           Buy         2,254         389           Buy         2,254         368           Buy         2,023         1,500           Sell         1,15,419         1,430           Sell         1,15,419         1,430           Sell         1,15,419         1,430           Sell         1,15,419         1,430           Sell         1,15,419         1,430           Sell         1,15,419 <t< td=""><td>UBS Securities Australia Ltd</td><td>Buy</td><td>3 139</td><td>585</td><td>Cropso</td></t<>	UBS Securities Australia Ltd	Buy	3 139	585	Cropso
Buy   8,070   1,000     Buy   8,070   1,000     Buy   8,33773   1,000     Buy   8,33773   4,153     Buy   77,283   9,003     Buy   77,283   9,003     Buy   77,283   9,003     Buy   77,283   9,003     Buy   77,283   9,000     Buy   7,283   9,000     Buy   7,283   9,000     Buy   7,283   9,000     Buy   7,283   9,000     Buy   7,283   9,000     Buy   7,283   13,23     Buy   7,283   13,200     Buy   7,283   14,320     Buy   7,283   18,300     Cortract Into Derivative   115,419   14,320     Buy   7,283   18,300     Cortract   10,000   14,000     Buy   7,200   14,200     Buy   7,200   19,000     Buy   7,200   19,000     Buy   7,200   19,000     Buy   7,200   19,000     Buy   7,200   19,000     Buy   7,200   25,000	UBS Securities Australia Ltd	Buv	18	5	in in
Buy   2,574   1,000     Buy   2,545   1,244     Buy   33,773   4,145     Buy   3,173   4,145     Buy   3,173   4,145     Buy   3,173   4,145     Buy   3,173   4,145     Buy   3,173   4,143     Buy   3,174   4,143     Buy   3,174   4,143     Buy   3,174   4,143     Buy   3,174   4,143     Buy   3,174   4,143     Buy   3,174   4,144     Buy   3,174   4,144     Buy   3,174   4,144     Buy   3,174   4,144     Buy   3,174   4,144     Buy   3,174   4,144     Buy   3,174   4,144     Buy   3,174   4,144     Buy   3,174   4,144     Buy   3,174   4,144     Buy   3,174   4,144     Buy   3,174   4,144     Buy   3,174   4,144     Buy   3,174   4,144     Buy   3,174   4,144     Buy   3,174   4,144     Buy   3,174   4,144     Buy   3,174   4,144     Buy   3,174   4,144     Buy   3,174   4,174     Buy   3,174   4,174     Buy   3,174   4,174     Buy   3,174   4,174     Buy   3,174   4,174     Buy   3,174   4,174     Buy   3,174   4,174     Buy   3,174   4,174     Buy   3,174     Buy   3,174   3,174     Buy   3,174   3,174     Buy   3,174   3,174     Buy   3,174   3,174     Buy   3,174   3,174     Buy   3,174   3,174     Buy   3,174   3,174     Buy   3,174   3,174     Buy   3,174   3,174     Buy   3,174   3,174     Buy   3,174   3,174     Buy   3,174   3,174     Buy   3,174   3,174     Buy   3,174   3,174     Buy   3,174   3,174     Buy   3,174   3,174     Buy   3,174   3,174     Buy   3,174   3,174     Buy   3,174	UBS Securities Australia 11d	2.2	0100	7 000 .	Cignary
Buy   58,459   7,000     Buy   77,787   7,244     Buy   77,787   7,182     Buy   77,787   7,182     Buy   77,787   7,182     Buy   77,787   7,182     Buy   77,787   7,182     Buy   77,787   7,182     Buy   7,182   7,183     Buy   7,183   7,183     Buy   7,184   7,183     Buy   7,184   7,183     Buy   7,184   7,183     Buy   7,184   7,183     Buy   7,184   7,183     Buy   7,184   7,183     Buy   7,184   7,183     Buy   7,184   7,183     Buy   7,184   7,184     Buy   7,184   7,184     Buy   7,184   7,184     Buy   7,184   7,185     Buy   7,184   7,185     Buy   7,185   84     Buy   7,185	I IRC Contrition Americal and	(2)	0/0'0	000,1	Ordinary
Buy   33,773   4,185     Buy   7,787   9,639     Buy   7,787   9,639     Buy   7,787   9,639     Buy   7,787   9,639     Buy   7,787   9,639     Buy   7,787   9,639     Buy   7,787   9,639     Buy   7,787   9,639     Buy   7,787   9,639     Buy   7,787   9,639     Buy   7,787   9,639     Buy   7,787   9,639     Buy   7,787   14,320     Buy   7,787   14,320     Buy   7,787   14,320     Buy   7,787   14,320     Buy   7,787   14,320     Buy   7,787   14,320     Buy   7,787   14,320     Buy   7,789   14,924   14,320     Buy   7,789   14,924   14,320     Buy   7,789   14,320   16,398     Buy   7,789   14,320   16,398     Buy   7,789   14,320   16,398     Buy   7,789   14,364   14,365     Buy   7,789   16,298     Bu	Dis described control of	book	/08	100	Ordinary
Buy   33,73   4,185     Buy   77,787   9,639     Buy   77,787   9,639     Buy   7,787   9,639     Buy   7,787   9,639     Buy   7,787   9,639     Buy   7,265   334     Buy   2,023   3,34     Buy   2,023   3,34     Buy   2,023   2,507     Buy   2,023   2,507     Buy   115,519   10,520     Sell   44,51   10,520     Sell   44,51   10,520     Sell   44,51   10,520     Sell   44,51   10,520     Sell   7,003     Se	Des securités Australia Ltd	Buy	58,459	7,244	Ordinary
Buy   77,87   9,539     Buy   77,87   9,639     Buy   7,787   9,639     Buy   7,787   9,639     Buy   7,263   9,00     Buy   2,023   344     Buy   2,023   348     Buy   2,023   348     Buy   2,023   348     Buy   2,023   344     Seli   115,419   143,20     Seli   2,500     Cortist   10,000     Cortist   10,000     Cortist   10,000     Buy   2,020   1,500     Buy   2,020   1,500     Buy   2,020   1,500     Buy   2,020   1,500     Buy   2,020   1,500     Buy   2,020   1,500     Buy   2,020   2,500     Buy	Obsiderations Australia Ltd	Buy	33,773	4,185	Ordinary
Buy   7.263   900     Buy   3.125   357     Buy   2.023   3.44     Buy   2.023   3.44     Buy   2.023   3.44     Buy   2.023   2.507     Buy   2.023   2.507     Buy   2.023   2.507     Buy   2.023   2.507     Buy   2.023   2.507     Buy   115,419   10,520     Seli   46,015   2.039     Seli   46,015   2.039     Seli   46,015   2.039     Seli   5,000     Cortiact   700   200     Cortiact   700	UBS Securities Australia Ltd	Buy	787,77	629'6	Ordinary
Buy   31.23   387     Buy   621   77     Buy   2.0595   334     Buy   2.0234   2.507     Buy   2.0234   2.507     Buy   2.0234   2.507     Buy   2.0234   2.507     Buy   2.0234   2.507     Buy   2.0234   2.507     Sell   113.569   2.507     Sell   44.791   1.05.20     Sell   43.20   2.507     Sell   43.20   2.507     Cortiset   1.050   2.507     Cortiset   1.050   2.507     Cortiset   1.050   2.507     Buy   2.020   2.508     Buy	UBS Securities Australia Ltd	Buy	7,263	006	Ordinary
Buy   621   77   899   847   848   849	UBS Securities Australia Ltd	Buy	3.123	387	Ordinary
Buy   3,139   389     Buy   2,254   366     Buy   2,254   366     Buy   2,254   366     Buy   2,254   366     Buy   2,254   366     Buy   2,254   3,139     Seli   115,419   10,520     Seli   19,56   2,403     Seli   19,56   2,403     Seli   19,56   2,500     Seli   19,56   2,500     Cortract   19,56   2,500     Cortract   19,56   2,500     Cortract   19,56   2,500     Seli   2,500   2,500     Cortract   19,501   2,500     Seli   31,092   3,500     Seli   31,092   3,500     Seli   31,092   3,500     Seli   31,092   3,500     Seli   34,366   3,501     Seli   2,500   3,500     Seli   34,366   3,501     Seli   2,602,31   3,603     Seli   2,603   3,500     Seli   2,603   3,500     Seli   3,403   3,500     Seli   3,403   3,500     Seli   3,403   3,500     Seli   3,403   3,500     Seli   3,403   3,500     Seli   3,500   3,500     Seli   3,	UBS Securities Australia Ltd	Buy	621	77	Ordinan
Buy   2,055   334     Buy   2,055   334     Buy   2,055   334     Buy   2,0231   2,570     Buy   2,0231   2,570     Sell   4,573   10,520     Sell   5,473   10,520     Sell   5,473   1,520     Sell   5,473   1,520     Enter into Derivative   58,177   7,718     Enter into Derivative   58,177   7,718     Enter into Derivative   58,177   7,718     Enter into Derivative   58,177   7,718     Enter into Derivative   1,18,419   1,500     Enter into Derivative   1,150   1,500     Enter into Derivative   1,150   1,500     Enter into Derivative   1,150   1,500     Enter into Derivative   1,150   1,032     Enter into Derivative   1,032   1,032     Enter into Derivative   1,032   1,032     Enter into Derivative   1,032   1,032     Enter into Derivative   1,032   1,032     Enter into De	UBS Securities Australia Ltd	Blw	02130	, ,	S C C
Buy   2,954   366     Buy   2,954   366     Buy   2,954   366     Buy   2,954   366     Buy   2,954   366     Buy   2,954   3,509     Sell   119,368   2,403     Sell   115,419   14,320     Sell   115,419   14,320     Sell   115,419   14,320     Sell   115,419   14,320     Contract   Contract   NA   35,000     Contract   Contract   NA   35,000     Contract   Sell   31,092   34,000     Sell   31,092   34,000     Sell   31,092   34,000     Sell   34,966   35,000     Sell   34,966   34,000     Sell   34,966   35,000     Sell   34,966   34,000     Sell   34,966   35,000     Sell   34,966   35,000     Sell   34,966   35,000     Sell   34,966   35,000     Sell   34,966   35,000     Sell   34,966   35,000     Sell   34,966   35,000     Sell   34,966   35,000     Sell   34,966   34,756     Sell   34,364   34,756     Sell   34,364   34,756     Sell   34,501   10,346     Sell   34,501   10,346     Sell   34,501   10,346     Sell   34,501   10,346     Sell   34,501   10,346     Sell   34,501   10,346     Sell   34,501   10,346     Sell   34,501   10,346     Sell   34,502   34,756     Sell   34,506   34,756     Sell   34,756   34,756     Sell   34,756   34,756     Sell   34,756   34,756     Sell   34,756   34,756     Sell   34,756   34,756     Sell   34,756   34,756     Sell   34,	UBS Securities Australia 114	80.00	2010	200	Oldinary
Section	IDS Counties Assets in 1-d	(co	2,055	334	Ordinary
Buy   2,023   2,507     Buy   24,015   2,507     Buy   24,015   10,520     Seli   19,368   2,403     Seli   19,368   2,403     Seli   19,368   2,403     Seli   19,368   2,403     Seli   19,368   2,403     Seli   19,368   2,403     Seli   19,368   2,500     Seli   19,368   3,700     Seli   2,500   3,800     Seli   31,032   3,500     Seli   3,403   3,600     Seli   3,403   3,600     Seli   3,403   3,600     Seli   3,400     Seli   3,400   3,600     Seli   3,400     Seli   3,400     Seli   3,400     Seli   3,400     Seli   3,400     Seli   3,400     Seli   3,400     Seli   3,400     Seli   3,400     Seli   3,400     Seli   3,400     Seli   3,400     Seli   3,400     Seli   3,400     Seli   3,400     Seli   3,400     Seli   3,400     Seli	ODD Securities Australia Liu	, and	2,954	366	Ordinary
Buy	Ubs securities Australia Ltd	any	20,231	2,507	Ordinary
Buy   84/91   10,520	UBS Securities Australia Ltd	Buy	444	52	Ordinary
Self	UBS Securities Australia Ltd	Buy	84,791	10,520	Ordinary
Sell	UBS Securities Australia Ltd	Sell	46,015	(5.709)	Ordinary
Buy   115,419   14,320     Sell   58177   (7,318)     Sell   183   (2.3)     Enter into Derivative   NA   35,000     Contract   183   (2.3)     Contract   183   (2.3)     Enter into Derivative   NA   18,800     Contract   193   (2.0)     Enter into Derivative   NA   18,800     Enter into Derivative   NA   18,800     Enter into Derivative   1,301   (3,000     Sell   31,092   (4,000     Sell   31,540   (4,3,22)     Buy   149,953   (4,3,22)     Buy   197,205   25,000     Buy   197,205   25,588     Buy   272,361   34,41     Sell   18,203   (146,822)     Sell   15,010   (146,822)     Sell   1,501   (146,822)     Sell   1,508   (146,822)     Sell   1	UBS Securities Australia Ltd	Sell	19 368	(2.403)	Cretinan
Sell	UBS Securities Australia Ltd	218	115 440	(4,405)	Ordinary
Self   Self	IDC Cocumition Australia 14d	(C)	214,011	(4,320	Ordinary
First into Definative   183 (23)	UDS Securities Australia Ltd	3611	28,1//	(7,218)	Ordinary
Contract C	ממן מברחוותב שחמו פוום דום	III N	183	(23)	Ordinary
Contract	LBS Securities Australia 14d	Enter into Derivative		4	
Contract   Contract			A91	ODD/SS	Options
Context   Context   NA   18,800	UBS Securities Australia Ltd	Contract	ΥN	50.000	Options
Contract   NuA   18.800		Enter Into Derivative			
Sell   31,092   3,980   3,98	UBS Securities Australia Ltd	Contract	NA	18,800	Options
Self   31,092 (3,947)     Self   31,540 (4,000)     Buy   149,953 (15,998 (3,000)     Self   31,540 (4,000)     Self   31,540 (4,328)     Buy   7,666 (3,328)     Buy   7,006 (3,000)     Buy   7,006 (3,000)     Self   160,231 (2,0246)     Self   28,388 (3,000)     Self   28,388 (3,000)     Self   28,388 (3,000)     Self   27,644 (3,000)     Self   27,644 (4,126)     Self   1,44,118 (2,5,10)     Self   1,44,118 (3,000)     Self   1,501 (10,342 (14,682)     Self   1,501 (14,682)     Self   1,501 (14,682)     Self   1,501 (14,682)     Self   1,501 (14,682)     Self   1,502 (14,682)     Self   3,501 (14,682)	UBS Global Asset Management	Buy	7,901	086	Ordinary
Sell   149.952   16,000     Buy   149.952   16,203     Sell   341.964   16,203     Buy   197.205   25,000     Buy   27,205   25,509     Buy   27,205   25,509     Buy   27,205   25,500     Buy   27,205   25,500     Sell   160,231   20,246     Sell   284,735   36,011     Sell   1,50,331   22,101     Sell   1,50,331   10,246     Sell   1,41,118   22,101     Sell   1,50,413   10,342     Sell   1,50,413   10,342     Sell   1,50,413   10,342     Sell   1,50,413   10,342     Sell   1,50,413   1,46,413     Sell   1,50,413   1,4	UBS Global Asset Management	Self	31,092	(3,947)	Ordinary
Buy   149,953   18,998   19,998   18,998   19,998   18,998   19,	UBS Securities Australia Ltd	Sell	31,540	(4,000)	Ordinary
Sell	UBS Securities Australia Ltd	Buy	149,953	18,998	Ordinary
Buy   7,505   964     Buy   197,205   25,000     Buy   197,205   25,000     Buy   202,068   25,588     Buy   202,068   34,641     Sell   169,231   (202,46)     Sell   24,735   (35,631)     Sell   24,735   (47,66)     Suy   376,845   (47,66)     Suy   81,501   (10,342     Sell   15,202   (146,842)     Sell   15,502   (146,842)     Se	UBS Securities Australia Ltd	Sell	341.964	(43 325)	Ordinary
Buy   197,205   25,500     Buy   202,068   25,588     Buy   273,581   3,641     Sell   160,231   (20,246     Sell   1,60,231   (20,246     Sell   1,74,18   22,101     Sell   1,51,862   (47,565     Sell   1,51,862   (48,822     Sell   1,51,862	UBS Securities Australia I1d	200	9031	(270,01)	Cidinary
Buy   197,055   25,000     Buy   202,068   25,588     Buy   273,581   34,641     Buy   273,581   34,641     Buy   283,308   10,246     Sell   284,735   (20,246     Buy   284,735   (3,669     Sell   1,508     Sell   1,508     Sell   1,508     Sell   1,508     Sell   1,508     Sell   34,641     Sell	100 Committee Australia Ital	Ren	2000,1	to a	Ordinary
Buy   27.0568   25.588	COO Securities Assessing and	boy.	(97,18)	25,000	Ordinary
Buy   273.581   34.641   Buy   273.581   20.7465   Buy   284.735   20.246   Buy   284.735   28	UBS Securities Australia Ltd	buy	202,068	25,588	Ordinary
Sell   169,231 (20,246)     Buy   80,398 (10,246)     Sell   284,735 (36,631)     Sell   276,845 (47,669)     Buy   376,845 (47,669)     Sell   1,52,027 (48,822)     Sell   1,52,027 (19,269)     Sell   1,52,027 (19,269)     Sell   1,52,027 (19,269)     Sell   3,641     Sell   3,641	UBS Securities Australia Ltd	Buy	273,581	34,641	Ordinary
Sell   284,735   10,246   10,247   10	UBS Securities Australia Ltd	Sell	160,231	(20,246)	Ordinary
Sell         284/35         (96.33)           Buy         774,18         22,101           Buy         376,845         (47,66)           Buy         81,501         10,342           Sell         1,508,622         (146,842)           Sell         152,037         (19,682)           Sell         152,037         (19,682)           Buy         273,581         34,641	UBS Securities Australia Ltd	Buy	80,308	10,246	Ordinary
Boy   174,118   22,101     Sell   37,645   174,118   22,101     Buy   31,501   10,342     Sell   1,50,87   (146,842)     Sell   152,037   (146,842)     Sell   152,037   (146,842)     Buy   273,581   34,641	UBS Securities Australia Ltd	Sell	284 735	(150.95)	ordina.
Sell 376.645 (47.766)	UBS Securities Australia (14)	a B	474 140	(100,00)	Cicinaly
See   27,582   (41,789   18,789   18,789   18,789   19,	IBC Countries Australia Lea	Coll	/4,110	22,10	Ordinary
Buy   R1,501   10,342	Obsideration Australia Ltd	200	376,845	(47,766)	Ordinary
Sell   1,588.62 (146,842)   Sell   152,037 (19,555)   Sell   34,641   Sell	UBS Securities Australia Ltd	Buy	81,501	10,342	Ordinary
Sell         152,037         (19,265)           Buy         273,581         34,641	UBS Securities Australia Ltd	Sell	1,158,862	(146,842)	Ordinary
Buy 273,581 34,641	UBS Securities Australia Ltd	lles	152 037	(10.055)	Ordinary
27,3,581 34,641	IRS AG London Reanch	Rin	273 501	11.0,000	Circuia,
	מושים אלי ביותים ושימוים	lacy	13,281	34,641	Ordinary

119-Dec-14	UBS Securities Australia 11d	Buv	0 500		Ordinary
19-Dec-14	UBS Securities Australia Ltd	Buy	5,699	745	Ordinary
19-Dec-14	UBS Securities Australia Ltd	Buy	15,660	2,	Ordinary
19-Dec-14	UBS Securities Australia Ltd	8ny	2,127		Ordinary
19-Dec-14	UBS Securities Australia Ltd	Buy	16.203		Ordinary
19-Dec-14	UBS Securities Australia Ltd	Buy	18,475	2.415	Ordinary
19-Dec-14	UBS Securities Australia Ltd	Buy	15 705		Ordinary
19-Dec-14	UBS Securities Australia Ltd	Buy	3,794		Ordinary
19-Dec-14	UBS Securities Australia Ltd	Buy	10,106		Ordinary
19-Dec-14	UBS Securities Australia Ltd	Buy	8,361		Ordinary
19-Dec-14	UBS Securities Australia Ltd	Buy	5,791		Ordinary
19-Dec-14	UBS Securities Australia Ltd	Buy	2,081		Ordinary
19-Dec-14	UBS Securities Australia Ltd	Buy	1,828		Ordinary
19-Dec-14	UBS Securities Australia Ltd	Buy	214		Ordinary
		Enter Into Derivative			7.00
19-Dec-14	UBS Securities Australia Ltd	Contract	A/A	40,000	Options
		Enter Into Derivative			
19-Dec-14	UBS Securities Australia Ltd	Contract	N/A	25,000	Options
22-Dec-14	UBS Securities Australia Ltd	Seti	14,350		Ordinary
22-Dec-14	UBS Securities Australia Ltd	Buy	164,038		Ordinary
22-Dec-14	UBS Securities Australia Ltd	Sell	866'6		Ordinary
22-Dec-14	UBS Securities Australia Ltd	Sell	139,309		Ordinary
22-Dec-14	UBS Securities Australia Ltd	Buy	2,204,671		Ordinary
22-Dec-14	UBS Securities Australia Ltd	Buy	267,097		Ordinary
22-Dec-14	UBS Securities Australia Ltd	Buy	10,252		Ordinary
22-Dec-14	UBS Securities Australia Ltd	Buy	179,227		Ordinary
22-Dec-14	UBS Securities Australia Ltd	Sel)	8/2/9		Ordinary
22-Dec-14	198 Sociation Australia 144	coll	46,467		Ordinary
22-Dec-14	LIRS Securities Australia Ind	B <sub>IN</sub>	304,342		Ordinary
22-Dec-14		(F)	3/8/363		Ordinary
22-Dec-14	LIBS Securities Australia 144	N. S.	404,003		Ordinary
22-Dec-14	I IBS Sequities Australia I to	(a)	22,22		Cicerary
22-Dec-14	UBS Securities Australia Ltd	Sell	62.010		Ordinary
22-Dec-14	UBS Securities Australia Ltd	Sell	504 092		Cardina
22-Dec-14	UBS Securities Australia Ltd	Sell	522 909	(54 94R)	Ordinary
22-Dec-14	UBS AG Australia Branch	Buy	8.615		Ordinary
	UBS AG Australia Branch	Sell	484,138		Ordinary
	UBS AG Australia Branch	Sell	1,492,758		Ordinary
- 1	UBS AG Australia Branch	Sell	1237,671	(29,537)	Ordinary
22-Dec-14	IUBS AG Australia Branch	Sell	216,677		Ordinary
- 1	UBS Wealth Management Australia Ltd	Buy	686'9	863	Ordinary
22-Dec-14	UBS Wealth Management Australia Ltd	Sell	21,123		Ordinary
- 1	U8S Securities Australia Ltd	Sell	1,286,400	(160,000)	Ordinary
- 1	UBS Securities Australia Ltd	Sell	11,574	(1,408)	Ordinary
- [	UBS Securities Australia Ltd	Sell	3,954	(481)	Ordinary
22-Dec-14	UBS Securities Australia Ltd	Sell	6,708	(822)	Ordinary
- 1	UBS Securities Australia Ltd	Self	3,378	(414)	Ordinary
- 1	UBS Securities Australia Ltd	Seli	1,363	(167)	Ordinary
- 1	UBS Securities Australia Ltd	Sell	22.5	(83)	Ordinary
- 1	UBS Securities Australia Ltd	Sell	465	(57)	Ordinary
- 1	UBS Securities Australia Ltd	Sell	24,480	(000'E)	Ordinary
- 1	UBS Securities Australia Ltd	Seil	5,010	(614)	Ordinary
	U8S Securities Australia Ltd	Sell	141	(18)	Ordinary
22-Dec-14	UBS Securities Australia Ltd	Sell	163		Ordinary
-	UBS Securities Australia Ltd	Sell	1,624	(199)	Ordinary
	UBS Securities Australia Ltd	Sell	45,582		Ordinary
	UBS Securities Australia Ltd	Sell	1,191		Ordinary
	UBS Securities Australia Ltd	Sell	31,612	0	Ordinary
	UBS Securities Australia Ltd	Sell	2,180,950		Ordinary
	UBS Securities Australia Ltd	Buy	1,349		Ordinary
	UBS Securities Australia Ltd	Buy	8,551	1.059	Ordinary
	UBS Securities Australia Ltd	Buy	6.904	852	Ordinary
	UBS Securities Australia Ltd	Buy	1,179	146	Ordinary
	UBS Securities Australia Ltd	Buy	6,016	745	Ordinary
	UBS Securities Australia Ltd	Buy	2,739	339	Ordinary
	UBS Securities Australia Ltd	Buy	4,816	596	Ordinary
22-Dec-14	UBS Securities Australia Ltd	Buy	15,708	1.944	Ordinary
	UBS Securities Australia Ltd	Buy	3.143	389	Ordinary
	UBS Securities Australia Ltd	Buy	2,723	755	Ordinary
	UBS Securities Australia Ltd	Buy	3.143	386	Ordinary
	UBS Securities Australia Ltd	Buy	582	22	Ordinary
	UBS Securities Australia Ltd	Buv	14 083	1 7/43	Ordinan
L	I IDC Cocuration Acceptable 144	1	200,4	0.00	Ordinary
ı	Opposed the Sanata Lib	pu)	3,204	404	Ordinary

22-Dec-14 (BS Securine Australia Ltd 22-Dec-14 (BS Securine Australia) Ltd 22-Dec-14 (BS Securine Australia Ltd 22-Dec-14 (BS Securine Australia) Ltd 22-Dec-14 (BS Securine Australia) Ltd 22-Dec-14 (BS Securine Australia) Ltd 22-Dec-14 (BS Securine Australia) Ltd 22-Dec-14 (BS Securine Australia) Ltd 22-Dec-14 (BS Securine Australia) Ltd 22-Dec-14 (BS Securine Australia) Ltd 22-Dec-14 (BS Securine Australia) Ltd 22-Dec-14 (BS Securine Australia) Ltd 22-Dec-14 (BS Securine Australia) Ltd 22-Dec-14 (BS Securine Australia) Ltd 22-Dec-14 (BS Securine Australia) Ltd 22-Dec-14 (BS Securine Australia) Ltd 22-Dec-14 (BS Securine Australia) Ltd 22-Dec-14 (BS Securine Australia) Ltd 22-Dec-14 (BS Securine Australia) Ltd 22-Dec-14 (BS Securine Australia) Ltd 22-Dec-14 (BS Securine Australia) Ltd 22-Dec-14 (BS Securine Australia) Ltd 22-Dec-14 (BS Securine Austr	Australia (rd Australia (rd Australia (rd Australia (rd Australia (rd Australia (rd Australia (rd Australia (rd Australia (rd Australia (rd Australia (rd Australia (rd Australia (rd Australia (rd Australia (rd Australia (rd	Sell Sell	1,186 570 6,960	(150) (72)	Ordinary
	Australia Led Australia Led Australia Led Australia Led Australia Led Australia Led Australia Led Australia Led Australia Led Australia Led Australia Led	Sell	570	(72)	Ordinary
UBS Securities A UBS SECURITIES A UBS SECURITIES A UBS SE	Votralia Ito Votralia Ito Votralia Ito Votralia Ito Votralia Ito Votralia Ito Votralia Ito Votralia Ito Votralia Ito Votralia Ito Votralia Ito	N	096'9	(881)	C
UBS Securities A UBS SECURITIES A UBS SECURITIES A UBS SE	usstralia Ltd usstralia Ltd usstralia Ltd usstralia Ltd usstralia Ltd usstralia Ltd usstralia Ltd usstralia Ltd	-			Ordinary
185 Securities A 186 Securities A 186 Securities A 187 Securities A 187 Securities A 187 Securities A 188 Securities A	Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd	Je.	4,388	(554)	Ordinary
1085 Securities A 1085 Securities A	vistralia Ltd Australia Ltd Australia Ltd Australia Ltd Australia Ltd		35,913	(4,546)	Ordinary
UBS Securities A UBS Securities A	Australia Ltd Australia Ltd Australia Ltd Australia Ltd	Sell	1,853	(234)	Ordinary
UBS Securities A   UBS SECURITIES A   UBS SECURITIES A   UBS SECURIT	Australia Ltd Australia Ltd Australia Ltd	Sell	1,193	(151)	Ordinary
UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A	vostralia Ltd tustralia Ltd	Sell	16	ଥ	Ordinary
UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A	Australia Ltd	Sell	648	(82)	Ordinary
UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A	Lotenia 1+d	Seil	5,372	(089)	Ordinary
UBS Securities A UBS SECURITIES A UBS SE	מסיו שוום דנם	Sell	909	(64)	Ordinary
UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A	Australia Ltd	Sell	17.175	(2.174)	Ordinary
UBS Securities A UBS SECURITIES A UBS SE	vustralia Ltd	Sell	6,604	(988)	Ording
UBS Securities A UBS SECURITIES SECURITIES SECURITIES SECURITIES SECURITIES SECURITIES A UB	kustralia Ltd	Sell	2 521	(500)	ordinary.
UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A	vustralia Ltd	ية پو	475	(14)	Octoriary
UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A	ustralia Itd	9	4/10	(ng)	Ordinary
UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A	Controlled the	100 J	865'7	(328)	Ordinary
UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A	Data and and and and and and and and and an	1120	2,408	(304)	Ordinary
USS Securities A USS Securities A USS Securities A USS Securities A USS Securities A USS Securities A	Australia Ltd	i Sei	2,354	(862)	Ordinary
UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A	VOSTI GHIG TITO	Sell	38,955	(4,931)	Ordinary
UBS Securities A UBS Securities A UBS Securities A UBS Securities A UBS Securities A	rustralia Ltd	Sell	2,622	(331)	Ordinary
UBS Securities A UBS Securities A UBS Securities A UBS Securities A	kustralia Ltd	Sell	7,119	(006)	Ordinary
UBS Securities A UBS Securities A UBS Securities A	rustralia Ltd	Sell	58,405	(2383)	Ordinary
UBS Securities A UBS Securities A	ustralia Ltd	Sell	4,811	(609)	Ordinary
UBS Securities A	ustralia Ltd	Sell	1.022	(129)	Ordinary
	ustralia Ltd	Setl	15 966	(150.5)	Ordinary
UBS Securities A	Securities Australia Ltd	Sell	6773	(854)	Coding
UBS Securities Australia Ltd	ustralia Ltd	Sell	19,7761	(100)	Ording
UBS Securities Australia Ltd	ustralia Ltd	les.	18 668	(0.000)	College
UBS Securities Australia Ltd	ustralia Ltd	Sell	23 550	(2003)	Cidinary
UBS Securities Australia Itd	ustralia Itd	Spil	2,550	(106,2)	Ordinary
UBS Securities A	ustralia Itd	la)	201,0	(660)	Ordinary
I IBS Sacrimities A	introlia 14d	100	275.0	(08G)	Ordinary
I IDS Securities A.	infinite tra	100	100,1	(602)	Ordinary
( IDC Cocumition )	associate Eric	100	816,81	(2,344)	Ordinary
Obs securities A	ustralia Ltd	Sell	20,050	(2,538)	Ordinary
OBS Securines A	Ustralia Ltd	Sell	45,172	(5,718)	Ordinary
UBS Securities A	ustralia Ltd	Sell	48,933	(6,194)	Ordinary
UBS Securities A	ustralia ttd	Seil	43,174	(5,465)	Ordinary
UBS Securities Ar	ustralia Ltd	Sell	1,186	(150)	Ordinary
UBS Securities Ar	ustralia Ltd	Sell	1,186	(120)	Ordinary
UBS Securities A	ustralia Ltd	Sell	2,781	(352)	Ordinary
UBS Securities A	ustralia Ltd	Sell	648	(82)	Ordinary
UBS Securities Australia Ltd	ustralia Ltd	Sell	3.918	(496)	Ordinary
UBS Securities A	ustralia Ltd	les:	68 548	(8,677)	Ordinan
UBS Securities Australia Ltd	ustralia Ltd	Sell	7 900	(1,000)	Ordinary
UBS Securities Au	ustralia Ltd		262	(25)	Ordinary
UBS Securities At	ustralia Ltd	Sell	OF S	(1001)	Cidilary
UBS Securities Australia Ltd	ustralia Ltd	las.	3 080	(1001)	Ordinary
IIRS Securities Australia 11d	straja Itd	Sell	2,962	(504)	Ordinary
Upo Securities At	ustraile Ltu	100	3,192	(404)	Ordinary
UBS Securities Australia Ltd	ustraila Ltd	Sell	450	(57)	Ordinary
UBS Securities Australia Ltd	ustralia Ltd	Sell	3,523	(446)	Ordinary
UBS Securities As	ustralia Ltd	Sell	1,114	(141)	Ordinary
UBS Securities At	ustralia Ltd	Sell	3,918	(496)	Ordinary
UBS Securities Au	ustralia Ltd	Sell	21.385	(707.6)	Ordinary
UBS Securities Au	ustralia Ltd	Sell	150	(10)	Ordina.
UBS Securities Au	ustralia Ltd	Seli	112 986	(14 302)	Ordinan
UBS Securities Au	ustralia I td	9	764	(17)	O Gillary
I IRS Securities A.	istralia I to	lo lo	150	(11)	Ordinary
I IBS Securities As	otrolin Ital	les de la constant de	CFO.**	(710)	Ordinary
I IDC Countries A.	tradic 140	100	12,354	(595,1)	Ordinary
Opp Securities AL	Stratile LTD	Sell	5,585	(707)	Ordinary
UBS Securities At	ıstralia Ltd	Set	1,525	(193)	Ordinary
UBS Securities Au	ustralia Ltd	Sell	60,664	(629'2)	Ordinary
UBS Securities Au	ıstralia Ltd	Sell	31,241	(3,952)	Ordinary
UBS Securities Au	ustralia Ltd	lleS	16.306	(2.064)	Ordinary
UBS Securities Australia Ltd	stralia Ltd	Sell	158	(20)	Ordinary
UBS Securities AL	stralia Ltd	IteS	4 977	(630)	Ordinary
110C Cocumition A.	0.00	*****	4,977	(059)	Ordinary
OBS Securities Au	Isualia Ltd	)eii	237	(30)	Ordinary
UBS Securities Au	Australia Ltd	Sell	10,886	(1,378)	Ordinary
UBS Securities Australia Ltc	stralia Ltd	Sell	5,001	(633)	Ordinary
UBS Securities Au	stralia Ltd	Sell	4716	(597)	Croinan
UBS Securities Australia 11d	stralia Itd	Coll	10.75	(160)	Cidillary
I IDC Commission A	the state of the s	-	00/61	(705'7)	Ordinary
UBS Securities Au	istralia Ltd	<u></u>	11,329	(1.434)	Ordinary
I IRS Securities Australia 11d	merchin 140	11.3	0000	(2007)	y and a

23-Dec-14 23-Dec-14	UBS AG Australia Branch	Seli	25,951	(3.447)	Ordinary
-					
ı	UBS AG Australia Branch	Sell	54,259	(7,207)	Ordinary
١	UBS AG Australia Branch	Sell	110,177	(14,513)	Ordinary
-	UBS AG Australia Branch	Sell	22,804	(620'E)	Ordinary
	UBS AG Australia Branch	Sell	134,046	(17,657)	Ordinary
	UBS AG Australia Branch	Sell	5,127	(189)	Ordinary
	UBS AG Australia Branch	Sell	523,059	(66,283)	Ordinary
3-Dec-14	UBS AG Australia Branch	Buy	31,897	4,038	Ordinary
ı	UBS AG Australia Branch		32,303	4,090	Ordinary
3-Dec-14	UBS AG Australia Branch	Transfer-Out	Ϋ́N	(000'09)	Ordinary
ı	UBS AG Australia Branch	Sell Sell	57,459	(7,281)	Ordinary
-	UBS AG Australia Branch	Sell	510,372	(64,672)	Ordinary
3-Dec-14	UBS Wealth Management Australia Ltd	Sell	14,023	(1,778)	Ordinary
	UBS Wealth Management Australia Ltd	Sell	12,146	(1,540)	Ordinary
	UBS Wealth Management Australia Ltd	Sell	11,925	(1,512)	Ordinary
	UBS Wealth Management Australia Ltd	Sell	11,057	(1,402)	Ordinary
	UBS Wealth Management Australia Ltd	Buy	3,464	439	Ordinary
	UBS Wealth Management Australia Ltd	Sell	10,481	(1,329)	Ordinary
	UBS Wealth Management Australia Ltd	Seli	23,400	(2,967)	Ordinary
	UBS Wealth Management Australia Ltd	Sell	1,083,893	(137,433)	Ordinary
1	UBS Securities Australia Ltd	Sell	87,429	(11,067)	Ordinary
	UBS Securities Australia Ltd	Sell	73,778	(6886)	Ordinary
3-Dec-14	UBS Securities Australia Ltd	Setl	1,99,1	(252)	Ordinary
3-Dec-14	UBS Securities Australia Ltd	Seil	70,073	(8,870)	Ordinary
3-Dec-14	UBS Securities Australia Ltd	Sell	27,611	(3,495)	Ordinary
3-Dec-14	UBS Securities Australia Ltd	Sell	442	(95)	Ordinary
3-Dec-14	UBS Securities Australia Ltd	Sell	569	(72)	Ordinary
	UBS Securities Australia Ltd	Sell	16	(2)	Ordinary
	UBS Securities Australia Ltd	Sell	5,372	(089)	Ordinary
-	UBS Securities Australia Ltd	Sell	2,789	(323)	Ordinary
	UBS Securities Australia Ltd	Sell	668'08	(3,848)	Ordinary
	UBS Securities Australia Ltd	Sell	580'E	(391)	Ordinary
	UBS Securities Australia Ltd	Sell	26,607	(3.368)	Ordinary
	UBS Securities Australia Ltd	Self	142	(18)	Ordinary
	UBS Securities Australia Ltd	Seil	2.054	(260)	Ordinary
	UBS Securities Australia Ltd	Sell	845	(107)	Ordinary
l	UBS Securities Australia 1td	Sell	E100	(108)	Ordinary
	UBS Securities Australia Ltd	Sell	6 446	(816)	Ordinary
3-Dec-14	UBS Securities Australia Ltd	Sell	2.149	(272)	Ordinary
	UBS Securities Australia Ltd	Seil	3.026	(383)	Ordinary
İ	UBS Securities Australia Ltd	Seil	885	(112)	Ordinary
	UBS Securities Australia Ltd	Sell	4.313	(546)	Ordinary
	UBS Securities Australia Ltd	Sell	1.533	(194)	Ordinary
-Dec-14	UBS Securities Australia Ltd	Sell	9 196	(1 164)	Ordinary
3-Dec-14	UBS Securities Australia Ltd	Self	2 702	(342)	Ordinary
3-Dec-14	UBS Securities Australia Ltd	Sell	509	(769)	Ordinary
3-Dec-14	IIRS Securities Australia Ind	las	5000	(CC)	Ordinary.
-Dec-14	LIRS Securities Australia (14	٥	1 146	(175)	Ordinary
3-Dec-14	LIRS Securities Australia Ital	las	0.75	(240)	Countary
l	Upo Securities Australia Ind	Col	01/2	(344)	Ordinary
1	Opp Securities Australia Ltd	100	5,254	(699)	Ordinary
	UBS Securities Australia Ltd	Sell Sell	1,880	(238)	Ordinary
	UBS Securities Australia Ltd	Sell	2,828	(358)	Ordinary
-Dec-14	UBS Securities Australia Ltd	Sell	1,296	(164)	Ordinary
3-Dec-14	UBS Securities Australia Ltd	Sell	1,462	(182)	Ordinary
3-Dec-14	UBS Securities Australia Ltd	Sell	648	(82)	Ordinary
3-Dec-14	UBS Securities Australia Ltd	Sell	1,153	(146)	Ordinary
3-Dec-14	UBS Securities Australia Ltd	Sell	4,795	(209)	Ordinary
3-Dec-14	UBS Securities Australia Ltd	Sell	1 975	(050)	Ordinary
3-Dec-14	UBS Securities Australia Ltd	Sell	203	(68)	Ordinary
3-Dec-14	LIRC Securities Australia 14d	le l	10,460	(ACE 1)	Countries
23-Dec-14	IDS Cocuration Australia 144	Cell	001.0	(475,1)	Cruinaly
23-Dec-14	IBS Securities Australia Ind	la S	0770	(DC)	Ordinary
23-Dec-14	TIPO Commission Australia 14d	log	94/7	(7.400)	Crairiary
Dec-14	URS Securities Australia 11d	3	45,434	(2,498)	Ordinary
23-Dec-14	UBS Securities Australia Itd	les	20185	(2) 5553	Ordinary
	LIBS Securities Australia 144	19	2110	(000)	Ordinary
۱	UDC Countries Australia 144	lo 3	0117	(006)	Ordinary
	Obsiderations Australia Ltd	lac l	498	(63)	Ordinary
Dec-14	UBS Securities Australia Ltd	- X-	5,562	(704)	Ordinary
23-Dec-14	UBS Securities Australia Ltd	Sell	640	(81)	Ordinary
	UBS Securities Australia Ltd	Self	727	(65)	Ordinary
	UBS Securities Australia Ltd	Sell	5,412	(982)	Ordinary
	UBS Securities Australia Ltd	Sell	3,476	(440)	Ordinary
ı					

UBS Securities Australia Ltd	Sell	3,923	(496)	Ordinary
Securities Australia Ltd	Sell	4,335	(548)	Ordinary
UBS Securities Australia Ltd	Sell	3,267	(413)	Ordinary
Securities Australia tra	nec nec	1,677	(212)	Ordinary
Securities Australia Ltd	Nei:	127	(16)	Ordinary
OBS Securities Australia Ltd	Sell	6,731	(821)	Ordinary
Securities Australia Ltd	Sell	10,030	(1,268)	Ordinary
DS Securities Australia Ltd	) Sell	5,624	(711)	Ordinary
Securities Australia Ltd	lec le	12,759	(1,613)	Ordinary
Obs securities Australia LIO	nec nec	3,599	(455)	Ordinary
Securities Australia LTO	)ell	1,187	(150)	Ordinary
securities Australia Ltd	nac .	14,643	(1,850)	Ordinary
Securities Australia Ltd	Sell	41,092	(5,195)	Ordinary
securities Australia Ltd	, sell	41,464	(5,242)	Ordinary
Securities Australia Ltd	Sell	38,499	(4,864)	Ordinary
Securities Australia Ltd	Sell	28,589	(3.612)	Ordinary
Securities Australia Ltd	Sell	21,711	(2.743)	Ordinary
Securities Australia Ltd	Sel	21.711	(5 7.43)	Ordinan
Securities Australia Ltd	Sell	427	(FA)	Ordinan
UBS Securities Australia Ltd	Sell	3 923	(90/)	ordinary.
Securities Australia Ltd	les	107.0	(430)	o delinary
IBS Securities Australia 144	100	16/6	(100)	Ordinary
IRC Consisting Australia 1+d	100	867,1	(159)	Ordinary
Securities Australia LIO	Sell	3,923	(496)	Ordinary
UBS Securities Australia Ltd	Sell	6,731	(821)	Ordinary
UBS Securities Australia Ltd	Sell	3,599	(455)	Ordinary
UBS Securities Australia Ltd	Sell	13,827	(1 748)	Ordinary
JBS Securities Australia Ltd	Sell	7 942	(1,004)	Ordinary
Securities Australia (10	Sell	9036	(1,004)	Cidilary
URS Securities Australia Ltd	Soll	9/0/9	(1,021)	Ordinary
IRS Securities Australia 1+4	(a)	029(5)	(Z,UUU)	Ordinary
Securities Australia Lite	nan o	25,937	(3,279)	Ordinary
Securities Australia Ltd	Deil	4,453	(263)	Ordinary
UBS Securities Australia Ltd	Sell	3,567	(451)	Ordinary
Securities Australia Ltd	Sell	7,356	(086)	Ordinary
Securities Australia Ltd	Sell	5,352	(089)	Ordinary
Securities Australia Ltd	Sell	6,697	(821)	Ordinary
Securties Australia Ltd	Seil	456	(28)	Ordinary
Securities Australia Ltd	Sell	81,195	(10.317)	Ordinary
Securities Australia Ltd	Sell	3.904	(496)	Ordinary
Securities Australia Ltd	lleS	5.548	(205)	Ordinary
Securities Australia Ltd	Sell	7 484	(051)	Croins
UBS Securities Australia Ltd	Sell	6 697	(851)	Calibra
JBS Securities Australia Ltd	Sell	68 537	(002 0)	Ordinary Ordinary
JBS Securities Australia Ltd	Self	1.4 40.2	(1.020)	Ciginal y
JBS Securities Australia Itcl	9	205/1	(000/1)	Cranacy
IRS Securities Australia Itd	199	0 14	(8/)	Ordinary
IDC Countries Australia I tol		5,745	(/30)	Ordinary
Securines Australia Ltd	Sell.	68,186	(8,664)	Ordinary
Obsisecurities Australia Ltd	Sett	16,330	(2,075)	Ordinary
JBS Securities Australia Ltd	Sell	5,470	(569)	Ordinary
UBS Securities Australia Ltd	Sell	1.118	(142)	Ordinary
JBS Securities Australia Ltd	Sell	41 730	(35.1)	o duling
Securities Australia Ital	la S	06/14	(867°C)	Ordinary
DDC Countries Australia Lea	1100	41,136	(5,217)	Ordinary
Control of the second of the s	100	47,682	(6,051)	Ordinary
Securities Australia Ltd	Sell	37,217	(4,723)	Ordinary
Securities Australia Ltd	Sell	465	(65)	Ordinary
Securities Australia Ltd	Sell	826	(119)	Ordinary
Securities Australia Ltd	Sell	24	Ĉ	Ordinary
Securities Australia Ltd	Sell	1 379	(371)	a de la constante de la consta
Securities Australia Ltd	Sell			0
Securities Australia 14d	Coll	33		Ordinary
Comment of the comment		3,136	(366)	Ordinary
securities Australia Ltd	Sell	4,539	(276)	Ordinary
Securities Australia Ltd	Sell	4,815	(611)	Ordinary
Securities Australia Ltd	Sell	11,528	(1.463)	Ordinary
Securities Australia Ltd	Seil	2.364	(300)	Ordinary
5 Securities Australia Ltd	Sell	8 455	(1 073)	Ordinani
ecurities Australia Ltd	Sell	7.1	(5)	1
Acutities Australia Itel	le de la company	1,	(6)	Ordinary
economic Australia Liu	Jac.	717	(91)	Ordinary
ecurities Australia Ltd	Sell	1,229	(156)	Ordinary
Securities Australia Ltd	Sell	693	(88)	Ordinary
UBS Securities Australia Ltd	Sell	600	(413)	ordinal y
Securities Australia 11d	llay.	496.6	(711)	Clariary
Did all the Control of the Control o		2,304	(300)	Ordinary
UBS Securities Australia Ltd	Sell	3 144	2000	
Agreement According to a			(399)	Cranary
	Seii	3.157	(665)	Ordinary

23-Dec-14	UBS Securities Australia Ltd	Sell	9/8	(111)	Ordinary
23-Dec-14	UBS Securities Australia Ltd	Sed	828	(105)	Ordinary
23-Dec-14	UBS Securities Australia Ltd	Sell	828	(105)	Ordinary
23-Dec-14	UBS Securities Australia Ltd	Sell	2.067	(292)	Ordinary
23-Dec-14	UBS Securities Australia Ltd	Sell	395	(20)	Ordinary
23-Dec-14	UBS Securities Australia Ltd	Sell	18,723	(2,373)	Ordinary
23-Dec-14	UBS Securities Australia Ltd	Sell	2,446	(310)	Ordinary
23-Dec-14	UBS Securities Australia Ltd	Sell	2,446	(310)	Ordinary
23-Dec-14	UBS Securities Australia Ltd	Sell	87	(11)	Ordinary
23-Dec-14	UBS Securities Australia Ltd	Sell	5,113	(648)	Ordinary
23-Dec-14	UBS Securities Australia Ltd	Sell	584	(74)	Ordinary
23-Dec-14	UBS Securities Australia Ltd	Seli	2,399	(304)	Ordinary
23-Dec-14	UBS Securities Australia Ltd	Sell	52,050	(6.597)	Ordinary
23-Dec-14	UBS Securities Australia Ltd	Buy	72,063	9,145	Ordinary
23-Dec-14	UBS Securities Australia Ltd	Buy	109,958	13.954	Ordinary
23-Dec-14	UBS Securities Australia Ltd	Buy	5,161	655	Ordinary
23-Dec-14	UBS Securities Australia Ltd	Buy	75,459	9,576	Ordinary
23-Dec-14	UBS Securities Australia Ltd	Sell	25,360	(3.200)	Ordinary
23-Dec-14	UBS Securities Australia Ltd	Sell	689	(87)	Ordinary
23-Dec-14	UBS Securities Australia Ltd	Sell	4,034	(210)	Ordinary
23-Dec-14	UBS Securities Australia Ltd	Sell	2,468	(312)	Ordinary
23-Dec-14	UBS Securities Australia Ltd	Sell	2,895	(366)	Ordinary
23-Dec-14	UBS Securities Australia Ltd	Sell	3,156	(388)	Ordinary
23-Dec-14	UBS Securities Australia Ltd	Sell	662	(101)	Ordinary
23-Dec-14	UBS Securities Australia Ltd	Sell	111	(14)	Ordinary
23-Dec-14	UBS Securities Australia Ltd	Sell	8	ε	Ordinary
23-Dec-14	UBS Securities Australia Ltd	Sell	14,388	(1,819)	Ordinary
23-Dec-14	UBS Securities Australia Ltd	Sell	10,006	(1,265)	Ordinary
23-Dec-14	UBS Securities Australia Ltd	Sell	13,534	(11/211)	Ordinary
23-Dec-14	UBS Securities Australia Ltd	Enter Into Derivative Contract	d N	20000	Ontions
				200,000	2

1015 Securities Automists Ltd   564   655   65	23-Dec-14 23-Dec-14	UBS Securities Australia Ltd	Seil	41,742	(5,304)	Ordinary
10.55 Securities Australia LEG   5441   5649   6690   66	23-Dec-14	The Contract Acceptance of the				- diameter
10.00   10.0	22 000 10	UBS Securities Australia Ltd	Sell	269'9	(821)	Ordinary
15   Securitar Australia LEG   Serial Contro	23-Dec-14	UBS Securities Australia (10	Sell	3,904	(496)	Ordinary
1975 Securing Augustia Ltd   1978	22 Dec 14	Day Securities Australia Ltd	Jac J	850,01	(1,278)	Ordinary
10.00   1.574   1.050   1.05	23 Dec 14	Ups Securities Australia Ltd	nes.	7,083	(006)	Ordinary
(85) Securine Automita Litt         Sea         4,359         (165)           (85) Securine Automita Litt         Sea         1,359         (165)           (85) Securine Automita Litt         Sea         1,359         (165)           (85) Securine Automita Litt         Sea         1,359         (165)           (85) Securine Automita Litt         Sea         1,359         (165)           (85) Securine Automita Litt         Sea         1,350         (175)           (85) Securine Automita Litt         Sea         1,350         (175)           (85) Securine Automita Litt         Sea         1,350         (175)           (85) Securine Automita Litt         Sea         1,350         (175)           (85) Securine Automita Litt         Sea         1,350         (175)           (85) Securine Automita Litt         Sea         1,350         (175)           (85) Securine Automita Litt         Sea         3,350         (175)           (85) Securine Automita Litt         Sea         3,350         (175)           (85) Securine Automita Litt         Sea         3,350         (175)           (85) Securine Automita Litt         Sea         3,350         (175)           (85) Securine Automita Litt         Sea	23.060-14	Opp Securities Australia Ltd	261	1,5/4	(200)	Ordinary
USS securine Automis as 10         SM         5,184         (1,20)           USS securine Automis as 10         SM         5,184         (1,20)           USS securine Automis as 10         SM         5,184         (1,20)           USS securine Automis as 10         SM         5,184         (1,20)           USS securine Automis as 10         SM         5,184         (1,20)           USS securine Automis as 10         SM         1,186         (1,20)           USS securine Automis as 10         SM         1,186         (1,20)           USS securine Automis as 10         SM         1,186         (1,20)           USS securine Automis as 10         SM         1,186         (1,20)           USS securine Automis as 10         SM         1,186         (1,19)           USS securine Automis as 10         SM         1,186         (1,17)           USS securine Automis as 10         SM         1,186         (1,17)           USS securine Automis as 10         SM         1,186         (1,17)           USS securine Automis as 10         SM         1,186         (1,17)           USS security automis as 10         SM         1,186         (1,17)           USS securities Automis as 10         SM         1,186 </td <td>22.050.14</td> <td>מסכן שביתו וונוס אתאוו שום דות</td> <td>Jac.</td> <td>4,329</td> <td>(550)</td> <td>Ordinary</td>	22.050.14	מסכן שביתו וונוס אתאוו שום דות	Jac.	4,329	(550)	Ordinary
USS Securine Autorials and 10         Sel         \$5,184         G/200           USS Securine Autorials and 10         Sel         \$5,184         G/200           USS Securine Autorials and 10         Sel         \$15,28         G/200           USS Securine Autorials and 10         Sel         \$1,574         G/200           USS Securine Autorials and 10         Sel         \$1,574         G/200           USS Securine Autorials and 10         Sel         \$1,574         G/200           USS Securine Autorials and 10         Sel         \$1,574         G/200           USS Securine Autorials and 10         Sel         \$1,574         G/200           USS Securine Autorials and 10         Sel         \$1,574         G/200           USS Securine Autorials and 10         Sel         \$1,574         G/200           USS Securine Autorials and 10         Sel         \$1,574         G/200           USS Securine Autorials and 10         Sel         \$1,570         G/450           USS Securine Autorials and 10         Sel         \$1,570         G/450           USS Securine Autorials and 10         Sel         \$1,570         G/450           USS Securine Autorials and 10         Sel         \$1,570         G/450           USS Securine	23-Dec-14	UBS Securities Australia Ltd	Sell	1,259	(160)	Ordinary
USS Securine Autorials Ltd   Sel	23-Dec-14	UBS Securities Australia Ltd	Sell	25,184	(3,200)	Ordinary
USS Securine Autoritals (14   584)         1519         (415)           USS Securine Autoritals (14   584)         1503         (415)         (415)           USS Securine Autoritals (14   584)         1503         (415)         (415)           USS Securine Autoritals (14   584)         1   584         (1503)         (42)           USS Securine Autoritals (14   584)         1   584         (1504)         (1504)           USS Securine Autoritals (14   584)         1   584         (1504)         (1504)           USS Securine Autoritals (14   584)         1   584         (1504)         (1504)           USS Securine Autoritals (14   584)         2   524         (1504)           USS Securine Autoritals (14   584)         2   524         (1504)           USS Securine Autoritals (14   584)         2   524         (1504)           USS Securine Autoritals (14   584)         2   524         (1504)           USS Securine Autoritals (14   584)         2   544         2   5204           USS Securine Autoritals (14   584)         2   544         2   5204         (1608)           USS Securine Autoritals (14   584)         2   544         2   5204         (1608)           USS Securine Autoritals (14   584)         2   544         2   5204         (1608)           USS Securine Au	23-Dec-14	UBS Securities Australia Ltd	Sell	25,184	(3,200)	Ordinary
USS Securine Automitate total         Seal         13.266         (1.50)           USS Securine Automitate total         Seal         1,568         (1.50)           USS Securine Automitate total         Seal         1,568         (1.50)           USS Securine Automitate total         Seal         1,578         (1.50)           USS Securine Automitate total         Seal         1,578         (1.50)           USS Securine Automitate total         Seal         1,578         (1.50)           USS Securine Automitate total         Seal         1,578         (1.50)           USS Securine Automitate total         Seal         1,578         (1.50)           USS Securine Automitate total         Seal         1,578         (1.50)           USS Securine Automitate total         Seal         1,578         (1.50)           USS Securine Automitate total         Seal         1,578         (1.50)           USS Securine Automitate total         Seal         1,578         (1.50)           USS Securine Automitate total         Seal         3,500         (1.50)           USS Securities Automitate total         Seal         3,500         (1.50)           USS Securine Automitate total         Seal         3,500         (1.50)	23-Dec-14	UBS Securities Australia Ltd	Sell	519	(99)	Ordinary
UNE Securities Australia Ltd         Seld         1,5,83         C,2,50           UNE Securities Australia Ltd         Seld         1,588         C,2,50           UNE Securities Australia Ltd         Seld         1,574         C,500           UNE Securities Australia Ltd         Seld         6,878         C,600           UNE Securities Australia Ltd         Seld         6,878         C,600           UNE Securities Australia Ltd         Seld         4,777         C,600           UNE Securities Australia Ltd         Seld         4,777         C,600           UNE Securities Australia Ltd         Seld         4,277         C,600           UNE Securities Australia Ltd         Seld         4,277         C,600           UNE Securities Australia Ltd         Seld         2,38         C,620           UNE Securities Australia Ltd         Seld         2,38         C,620           UNE Securities Australia Ltd         Seld         2,38         C,620           UNE Securities Australia Ltd         Seld         2,38         C,620           UNE Securities Australia Ltd         Seld         2,38         C,620           UNE Securities Australia Ltd         Seld         2,30         C,620           UNE Securities Australia Ltd <td>23-Dec-14</td> <td>UBS Securities Australia Ltd</td> <td>Seti</td> <td>3,266</td> <td>(415)</td> <td>Ordinary</td>	23-Dec-14	UBS Securities Australia Ltd	Seti	3,266	(415)	Ordinary
USS Securities Australia Ltd         Seal         1,988         (7,892)           USS Securities Australia Ltd         Seal         1,988         (7,892)           USS Securities Australia Ltd         Seal         2,784         (8,792)           USS Securities Australia Ltd         Seal         2,784         (8,792)           USS Securities Australia Ltd         Seal         3,784         (7,202)           USS Securities Australia Ltd         Seal         3,784         (7,202)           USS Securities Australia Ltd         Seal         3,786         (7,602)           USS Securities Australia Ltd         Seal         3,786         (7,602)           USS Securities Australia Ltd         Seal         3,793         (7,602)           USS Securities Australia Ltd         Seal         3,793         (7,602)           USS Securities Australia Ltd         Seal         3,793         (7,602)           USS Securities Australia Ltd         Seal         3,793         (7,602)           USS Securities Australia Ltd         Seal         3,793         (7,602)           USS Securities Australia Ltd         Seal         2,725         (7,602)           USS Securities Australia Ltd         Seal         2,724         (7,602) <t< td=""><td>23-Dec-14</td><td>UBS Securities Australia Ltd</td><td>Sell</td><td>18,038</td><td>(2,292)</td><td>Ordinary</td></t<>	23-Dec-14	UBS Securities Australia Ltd	Sell	18,038	(2,292)	Ordinary
UNIS Securities Australia Ltd         Seal         6.81         6.82         (6.70)           UNIS Securities Australia Ltd         Seal         2.28         (7.20)           UNIS Securities Australia Ltd         Seal         2.24         (7.00)           UNIS Securities Australia Ltd         Seal         4.777         (6.50)           UNIS Securities Australia Ltd         Seal         4.777         (6.50)           UNIS Securities Australia Ltd         Seal         4.777         (6.50)           UNIS Securities Australia Ltd         Seal         4.777         (6.50)           UNIS Securities Australia Ltd         Seal         6.734         (7.27)           UNIS Securities Australia Ltd         Seal         7.734         (7.27)           UNIS Securities Australia Ltd         Seal         7.734         (7.737)           UNIS Securities Australia Ltd         Seal         7.738         (4.00)           UNIS Securities Australia Ltd         Seal         7.738         (4.00)           UNIS Securities Australia Ltd         Seal         7.738         (4.00)           UNIS Securities Australia Ltd         Seal         7.738         (4.00)           UNIS Securities Australia Ltd         Seal         7.739         (7.19)	23-Dec-14	UBS Securities Australia Ltd	Sell	1,968	(250)	Ordinary
UNIS Securities Automitals Ltd         Seal         CSP8         CSP9           UNIS Securities Automitals Ltd         Seal         22.9         CD0           UNIS Securities Automitals Ltd         Seal         1.57.184         CD0           UNIS Securities Automitals Ltd         Seal         4.27.7         CD0           UNIS Securities Automitals Ltd         Seal         4.27.7         CD0           UNIS Securities Automitals Ltd         Seal         4.27.7         CD0           UNIS Securities Automitals Ltd         Seal         4.27.7         CD0           UNIS Securities Automitals Ltd         Seal         4.27.7         CD0           UNIS Securities Automitals Ltd         Seal         2.2.3         CD0           UNIS Securities Automitals Ltd         Seal         2.2.3         CD0           UNIS Securities Automitals Ltd         Seal         2.2.3         CD0           UNIS Securities Automitals Ltd         Seal         2.2.3         CD0           UNIS Securities Automitals Ltd         Seal         2.2.3         CD0           UNIS Securities Automitals Ltd         Seal         2.2.3         CD0           UNIS Securities Automitals Ltd         Seal         2.2.3         CD0           UNIS Securities Autom	23-Dec-14	UBS Securities Australia Ltd	Sell	14,182	(1,802)	Ordinary
URS Securities Australia Ltd         Seal         2.73         (1.73)           USS Securities Australia Ltd         Seal         2.73         (1.73)           USS Securities Australia Ltd         Seal         4.286         (2.52)           USS Securities Australia Ltd         Seal         4.286         (2.52)           USS Securities Australia Ltd         Seal         4.286         (3.52)           USS Securities Australia Ltd         Seal         3.746         (4.580)           USS Securities Australia Ltd         Seal         3.746         (4.580)           USS Securities Australia Ltd         Seal         3.746         (4.580)           USS Securities Australia Ltd         Seal         3.746         (4.580)           USS Securities Australia Ltd         Seal         3.746         (4.797)           USS Securities Australia Ltd         Seal         3.746         (4.797)           USS Securities Australia Ltd         Seal         3.748         (4.797)           USS Securities Australia Ltd         Seal         3.748         (4.797)           USS Securities Australia Ltd         Seal         3.748         (4.797)           USS Securities Australia Ltd         Seal         3.753         (4.797)           USS S	23-Dec-14	UBS Securities Australia Ltd	Sel	6.878	(874)	Ordinary
USS Securities Australia Ltd   Sea   15.54   (1,720)     USS Securities Australia Ltd   Sea   15.54   (1,721)     USS Securities Australia Ltd   Sea   2.54   (1,721)     USS Securities Australia Ltd   Sea   2.54   (1,721)     USS Securities Australia Ltd   Sea   2.54   (1,721)     USS Securities Australia Ltd   Sea   2.54   (1,721)     USS Securities Australia Ltd   Sea   2.54   (1,722)     USS Securities Australia L	23-Dec-14	UBS Securities Australia Ltd	Se	228	(00)	Ordinary
LESS Securities Australia Lind         Sell         1,5724         (1,000)           LBS Securities Australia Lind         Sell         4,7229         (1,021)           LBS Securities Australia Lind         Sell         4,7229         (1,021)           LBS Securities Australia Lind         Sell         4,7229         (1,021)           LBS Securities Australia Lind         Sell         3,546         (1,021)           LBS Securities Australia Lind         Sell         3,546         (1,021)           LBS Securities Australia Lind         Sell         2,314         (1,021)           LBS Securities Australia Lind         Sell         3,533         (4,056)           LBS Securities Australia Lind         Sell         3,533         (4,056)           LBS Securities Australia Lind         Sell         3,533         (4,056)           LBS Securities Australia Lind         Sell         3,533         (4,056)           LBS Securities Australia Lind         Sell         3,533         (4,056)           LBS Securities Australia Lind         Sell         3,533         (4,056)           LBS Securities Australia Lind         Sell         3,533         (4,056)           LBS Securities Australia Lind         Sell         3,534         (4,056)	23-Dec-14	UBS Securities Australia I to	Sell	191 35	(67)	Ordinary
UBS Securities Australia Ltd         Sell         4,777         (1,720)           UBS Securities Australia Ltd         Sell         4,727         (645)           UBS Securities Australia Ltd         Sell         4,727         (645)           UBS Securities Australia Ltd         Sell         4,727         (645)           UBS Securities Australia Ltd         Sell         3,904         (645)           UBS Securities Australia Ltd         Sell         3,904         (645)           UBS Securities Australia Ltd         Sell         3,904         (645)           UBS Securities Australia Ltd         Sell         3,748         (675)           UBS Securities Australia Ltd         Sell         3,748         (675)           UBS Securities Australia Ltd         Sell         3,748         (675)           UBS Securities Australia Ltd         Sell         3,748         (670)           UBS Securities Australia Ltd         Sell         3,748         (670)           UBS Securities Australia Ltd         Sell         3,748         (670)           UBS Securities Australia Ltd         Sell         3,748         (670)           UBS Securities Australia Ltd         Sell         3,748         (702)           UBS Securities Australia L	23-Dec.14	I IBS Securities Australia 14d	Cell	101,02	(3,200)	Ordinary
URS Securities Australia Ltd         Sell         4,729         (1,73)           URS Securities Australia Ltd         Sell         4,729         (1,620)           URS Securities Australia Ltd         Sell         8,036         (1,620)           URS Securities Australia Ltd         Sell         3,046         (1,620)           URS Securities Australia Ltd         Sell         3,046         (1,620)           URS Securities Australia Ltd         Sell         3,046         (1,620)           URS Securities Australia Ltd         Sell         3,046         (1,620)           URS Securities Australia Ltd         Sell         3,046         (1,620)           URS Securities Australia Ltd         Sell         3,046         (1,620)           URS Securities Australia Ltd         Sell         3,046         (1,620)           URS Securities Australia Ltd         Sell         3,046         (1,620)           URS Securities Australia Ltd         Sell         3,046         (1,620)           URS Securities Australia Ltd         Sell         3,046         (1,620)           URS Securities Australia Ltd         Sell         3,046         (1,620)           URS Securities Australia Ltd         Sell         8,040         (1,620) <td< td=""><td>22 Day 14</td><td>100 Counties Australia and</td><td>1175</td><td>1,0/4</td><td>(200)</td><td>Ordinary</td></td<>	22 Day 14	100 Counties Australia and	1175	1,0/4	(200)	Ordinary
URS Securities Australia Ltd         Sell         4,777         (645)           URS Securities Australia Ltd         Sell         4,279         (645)           URS Securities Australia Ltd         Sell         3,904         (645)           URS Securities Australia Ltd         Sell         3,004         (645)           URS Securities Australia Ltd         Sell         7,235         (1,797)           URS Securities Australia Ltd         Sell         2,314         (7,921)           URS Securities Australia Ltd         Sell         3,793         (4,620)           URS Securities Australia Ltd         Sell         3,793         (4,100)           URS Securities Australia Ltd         Sell         3,793         (4,100)           URS Securities Australia Ltd         Sell         3,793         (4,100)           URS Securities Australia Ltd         Sell         3,793         (4,100)           URS Securities Australia Ltd         Sell         3,793         (4,100)           URS Securities Australia Ltd         Sell         3,793         (4,100)           URS Securities Australia Ltd         Sell         3,793         (4,100)           URS Securities Australia Ltd         Sell         3,793         (4,100)           URS Se	22 Dec 14	UBS Securities Australia Ltd	liac and	9,232	(1,173)	Ordinary
URS Securities Australia Ltd         Sell         4,229         (1,020)           URS Securities Australia Ltd         Sell         8,035         (1,020)           URS Securities Australia Ltd         Sell         3,046         (6,950)           URS Securities Australia Ltd         Sell         2,346         (6,950)           URS Securities Australia Ltd         Sell         2,346         (6,950)           URS Securities Australia Ltd         Sell         2,346         (6,950)           URS Securities Australia Ltd         Sell         2,346         (6,950)           URS Securities Australia Ltd         Sell         2,346         (6,950)           URS Securities Australia Ltd         Sell         3,753         (6,4820)           URS Securities Australia Ltd         Sell         3,753         (6,4820)           URS Securities Australia Ltd         Sell         3,754         (7,090)           URS Securities Australia Ltd         Sell         3,754         (7,090)           URS Securities Australia Ltd         Sell         3,904         (7,090)           URS Securities Australia Ltd         Sell         5,904         (7,090)           URS Securities Australia Ltd         Sell         5,904         (7,090)	23 Dec 14	Obs Securities Adsulating	1120	4,///	(207)	Ordinary
URS Securities Australia Ltd         Sell         9,504         (1,02)           URS Securities Australia Ltd         Sell         3,904         (1,02)           URS Securities Australia Ltd         Sell         3,904         (1,02)           URS Securities Australia Ltd         Sell         3,904         (1,02)           URS Securities Australia Ltd         Sell         7,21         (1,02)           URS Securities Australia Ltd         Sell         7,23         (1,02)           URS Securities Australia Ltd         Sell         3,904         (4,05)           URS Securities Australia Ltd         Sell         3,148         (1,02)           URS Securities Australia Ltd         Sell         3,148         (1,02)           URS Securities Australia Ltd         Sell         3,148         (1,02)           URS Securities Australia Ltd         Sell         3,148         (1,02)           URS Securities Australia Ltd         Sell         3,148         (1,02)           URS Securities Australia Ltd         Sell         3,004         (4,05)           URS Securities Australia Ltd         Sell         3,004         (4,05)           URS Securities Australia Ltd         Sell         3,004         (4,05)           URS Securities	23 Dec 14	OBS SECURIOS Australia Liu	11-10 11-10	4,289	(545)	Ordinary
UBS Securities Australia Ltd         Sell         3,504         (4,505)           UBS Securities Australia Ltd         Sell         3,504         (4,606)           UBS Securities Australia Ltd         Sell         7,225         (1,727)           UBS Securities Australia Ltd         Sell         7,231         (1,629)           UBS Securities Australia Ltd         Sell         7,231         (1,629)           UBS Securities Australia Ltd         Sell         7,231         (1,629)           UBS Securities Australia Ltd         Sell         7,233         (1,629)           UBS Securities Australia Ltd         Sell         7,538         (4,650)           UBS Securities Australia Ltd         Sell         7,538         (4,650)           UBS Securities Australia Ltd         Sell         7,539         (4,690)           UBS Securities Australia Ltd         Sell         8,540         (4,690)           UBS Securities Australia Ltd         Sell         8,540         (4,690)           UBS Securities Australia Ltd         Sell         8,540         (4,690)           UBS Securities Australia Ltd         Sell         8,540         (4,690)           UBS Securities Australia Ltd         Sell         8,540         (4,690) <t< td=""><td>23-Dec-14</td><td>Upo Securities Australia Ltd</td><td>liac .</td><td>8,035</td><td>(1,021)</td><td>Ordinary</td></t<>	23-Dec-14	Upo Securities Australia Ltd	liac .	8,035	(1,021)	Ordinary
USS Securities Australia Ltd         Sell         3,944         (468)           USS Securities Australia Ltd         Sell         3,946         (827)           USS Securities Australia Ltd         Sell         7,25         (1988)           USS Securities Australia Ltd         Sell         7,37         (1988)           USS Securities Australia Ltd         Sell         2,314         (1,297)           USS Securities Australia Ltd         Sell         2,314         (1,297)           USS Securities Australia Ltd         Sell         3,924         (4690)           USS Securities Australia Ltd         Sell         3,733         (4,090)           USS Securities Australia Ltd         Sell         3,733         (1,193)           USS Securities Australia Ltd         Sell         3,904         (4690)           USS Securities Australia Ltd         Sell         3,904         (4690)           USS Securities Australia Ltd         Sell         3,904         (4690)           USS Securities Australia Ltd         Sell         3,904         (4690)           USS Securities Australia Ltd         Sell         3,904         (4690)           USS Securities Australia Ltd         Sell         3,904         (4690)           USS Securitie	23-Dec-14	UBS Securities Australia Ltd	Sell	43,466	(5,523)	Ordinary
UBS Securities Australia Ltd         Sell         23,1466         (8,18)           UBS Securities Australia Ltd         Sell         2,314         (8,200           UBS Securities Australia Ltd         Sell         1,4122         (9,180           UBS Securities Australia Ltd         Sell         37,533         (4,620)           UBS Securities Australia Ltd         Sell         37,533         (4,620)           UBS Securities Australia Ltd         Sell         37,533         (4,620)           UBS Securities Australia Ltd         Sell         37,533         (4,620)           UBS Securities Australia Ltd         Sell         37,633         (4,620)           UBS Securities Australia Ltd         Sell         37,633         (4,630)           UBS Securities Australia Ltd         Sell         37,634         (4,690)           UBS Securities Australia Ltd         Sell         37,634         (4,690)           UBS Securities Australia Ltd         Sell         5,100         (4,690)           UBS Securities Australia Ltd         Sell         5,60         2,37         (4,690)           UBS Securities Australia Ltd         Sell         5,60         2,39         (4,690)           UBS Securities Australia Ltd         Sell         5,60	23-Dec-14	UBS Securities Australia Ltd	Sell	3,904	(496)	Ordinary
UBS Securities Australia Ltd         Self         7.25.14         (3.200           UBS Securities Australia Ltd         Self         7.22.5         (1.757)           UBS Securities Australia Ltd         Self         7.2.3.4         (1.757)           UBS Securities Australia Ltd         Self         7.7.301         (4.656)           UBS Securities Australia Ltd         Self         7.7.301         (4.656)           UBS Securities Australia Ltd         Self         7.7.301         (4.169)           UBS Securities Australia Ltd         Self         5.7.301         (4.169)           UBS Securities Australia Ltd         Self         5.7.00         (4.169)           UBS Securities Australia Ltd         Self         5.7.00         (4.169)           UBS Securities Australia Ltd         Self         5.7.00         (4.169)           UBS Securities Australia Ltd         Self         5.7.00         (4.169)           UBS Securities Australia Ltd         Self         5.7.00         (4.169)           UBS Securities Australia Ltd         Self         5.7.00         (4.169)           UBS Securities Australia Ltd         Self         5.7.00         (4.169)           UBS Securities Australia Ltd         Self         5.8.10         (7.00)	23-Dec-14	UBS Securities Australia Ltd	Sell	30,496	(3,875)	Ordinary
UBS Securities Australia Ltd         Self         1,225         (918)           UBS Securities Australia Ltd         Self         2,314         (1,294)           UBS Securities Australia Ltd         Self         37,933         (4,820)           UBS Securities Australia Ltd         Self         37,934         (4,696)           UBS Securities Australia Ltd         Self         37,934         (4,007)           UBS Securities Australia Ltd         Self         5,100         (4,006)           UBS Securities Australia Ltd         Self         5,100         (4,006)           UBS Securities Australia Ltd         Self         5,100         (4,006)           UBS Securities Australia Ltd         Self         5,100         (4,006)           UBS Securities Australia Ltd         Self         5,100         (4,006)           UBS Securities Australia Ltd         Self         5,200         (4,006)           UBS Securities Australia Ltd         Self         5,200         (4,006)           UBS Securities Australia Ltd         Self         5,200         (4,006)           UBS Securities Australia Ltd         Self         Self         (1,007)           UBS Securities Australia Ltd         Self         Self         (1,007) <td< td=""><td>23-Dec-14</td><td>UBS Securities Australia Ltd</td><td>Sell</td><td>25,184</td><td>(3,200)</td><td>Ordinary</td></td<>	23-Dec-14	UBS Securities Australia Ltd	Sell	25,184	(3,200)	Ordinary
UBS Securities Australia Ltd         Sell         1,142         (1,797)           UBS Securities Australia Ltd         Sell         2,314         (2,699)           UBS Securities Australia Ltd         Sell         3,793         (4,850)           UBS Securities Australia Ltd         Sell         3,793         (4,850)           UBS Securities Australia Ltd         Sell         3,748         (4,109)           UBS Securities Australia Ltd         Sell         3,748         (4,109)           UBS Securities Australia Ltd         Sell         5,100         (6,489)           UBS Securities Australia Ltd         Sell         5,100         (6,489)           UBS Securities Australia Ltd         Sell         5,100         (6,189)           UBS Securities Australia Ltd         Sell         5,100         (7,181)           UBS Securities Australia Ltd         Sell         1,181         (7,181)           UBS Securities Australia Ltd         Sell         1,181         (7,181)           UBS Securities Australia Ltd         Sell         1,181         (7,181)           UBS Securities Australia Ltd         Sell         Sell         (1,181)           UBS Securities Australia Ltd         Sell         Sell         (1,181)	Z3-Dec-14	UBS Securities Australia Ltd	Sell	7,225	(918)	Ordinary
UBS Securities Australia Ltd         Sell         2.3.41         (2.949)           UBS Securities Australia Ltd         Sell         2.7.331         (4.620)           UBS Securities Australia Ltd         Sell         37,633         (4.620)           UBS Securities Australia Ltd         Sell         7,508         (4.60)           UBS Securities Australia Ltd         Sell         6,217         (7.03)           UBS Securities Australia Ltd         Sell         6,217         (7.03)           UBS Securities Australia Ltd         Sell         6,217         (7.03)           UBS Securities Australia Ltd         Sell         6,217         (7.03)           UBS Securities Australia Ltd         Sell         6,217         (7.03)           UBS Securities Australia Ltd         Sell         1,307         (7.181)           UBS Securities Australia Ltd         Sell         1,307         (7.181)           UBS Securities Australia Ltd         Sell         1,307         (7.181)           UBS Securities Australia Ltd         Sell         2,430         (7.181)           UBS Securities Australia Ltd         Sell         2,430         (7.181)           UBS Securities Australia Ltd         Sell         2,430         (7.196)	23-Dec-14	UBS Securities Australia Ltd	Sell	14,142	(1,797)	Ordinary
UBS Securities Australia Ltd         Sell         27.331         (3.469)           UBS Securities Australia Ltd         Sell         37.933         (4.600)           UBS Securities Australia Ltd         Sell         37.933         (4.600)           UBS Securities Australia Ltd         Sell         37.933         (4.600)           UBS Securities Australia Ltd         Sell         37.338         (4.109)           UBS Securities Australia Ltd         Sell         37.90         (4.109)           UBS Securities Australia Ltd         Sell         5.33         (7.03)           UBS Securities Australia Ltd         Sell         5.30         (1.18)           UBS Securities Australia Ltd         Sell         3.90         (1.18)           UBS Securities Australia Ltd         Sell         3.90         (1.18)           UBS Securities Australia Ltd         Sell         3.90         (1.18)           UBS Securities Australia Ltd         Sell         3.90         (1.18)           UBS Securities Australia Ltd         Sell         3.90         (1.70)           UBS Securities Australia Ltd         Sell         3.90         (1.70)           UBS Securities Australia Ltd         Sell         3.90         (1.70)           UBS Securit	23-Dec-14	UBS Securities Australia Ltd	Sell	2,314	(294)	Ordinary
UBS Securities Australia Ltd         Sell         37,933         (4,820)           UBS Securities Australia Ltd         Sell         37,933         (4,820)           UBS Securities Australia Ltd         Sell         7,598         (400)           UBS Securities Australia Ltd         Sell         7,598         (400)           UBS Securities Australia Ltd         Sell         3,304         (400)           UBS Securities Australia Ltd         Sell         3,304         (400)           UBS Securities Australia Ltd         Sell         5,303         (703)           UBS Securities Australia Ltd         Sell         5,304         (4,800)           UBS Securities Australia Ltd         Sell         10,307         (1,181)           UBS Securities Australia Ltd         Sell         3,304         (4,800)           UBS Securities Australia Ltd         Sell         10,307         (1,181)           UBS Securities Australia Ltd         Sell         3,904         (4,900)           UBS Securities Australia Ltd         Sell         3,904         (1,100)           UBS Securities Australia Ltd         Sell         3,904         (1,100)           UBS Securities Australia Ltd         Sell         3,904         (1,100)           UBS	23-Dec-14	UBS Securities Australia Ltd	Sel	27,301	(3.469)	Ordinary
UBS Securities Australia Ltd         Sell         3,924         (469)           UBS Securities Australia Ltd         Sell         3,148         (4,169)           UBS Securities Australia Ltd         Sell         3,148         (4,109)           UBS Securities Australia Ltd         Sell         3,338         (4,109)           UBS Securities Australia Ltd         Sell         5,170         (790)           UBS Securities Australia Ltd         Sell         5,100         (1,181)           UBS Securities Australia Ltd         Sell         5,100         (1,181)           UBS Securities Australia Ltd         Sell         5,100         (1,181)           UBS Securities Australia Ltd         Sell         1,130         (1,181)           UBS Securities Australia Ltd         Sell         1,130         (1,181)           UBS Securities Australia Ltd         Sell         1,130         (1,181)           UBS Securities Australia Ltd         Sell         1,130         (1,181)           UBS Securities Australia Ltd         Sell         2,490         (1,181)           UBS Securities Australia Ltd         Sell         2,722         (4,29)           UBS Securities Australia Ltd         Sell         2,722         (4,21)           UBS	23-Dec-14	UBS Securities Australia Ltd	Sell	37.933	(4 820)	Ordinary
UBS Securities Australia Ltd         Sell         3,148         (400)           UBS Securities Australia Ltd         Sell         37,388         (400)           UBS Securities Australia Ltd         Sell         553         (703)           UBS Securities Australia Ltd         Sell         551         (703)           UBS Securities Australia Ltd         Sell         570         (648)           UBS Securities Australia Ltd         Sell         570         (648)           UBS Securities Australia Ltd         Sell         571         (113)           UBS Securities Australia Ltd         Sell         9,30         (113)           UBS Securities Australia Ltd         Sell         10,30         (1,131)           UBS Securities Australia Ltd         Sell         10,30         (1,131)           UBS Securities Australia Ltd         Sell         10,30         (1,131)           UBS Securities Australia Ltd         Sell         2,72         (3,63)           UBS Securities Australia Ltd         Sell         2,72         (3,63)           UBS Securities Australia Ltd         Sell         2,72         (3,63)           UBS Securities Australia Ltd         Sell         2,72         (4,65)           UBS Securities Australia Ltd <td>23-Dec-14</td> <td>UBS Securities Australia Ltd</td> <td>Sell</td> <td>3.904</td> <td>(496)</td> <td>Ordinary</td>	23-Dec-14	UBS Securities Australia Ltd	Sell	3.904	(496)	Ordinary
UBS Securities Australia Ltd         Sell         3,148         (4.00)           UBS Securities Australia Ltd         Sell         5,338         (4.109)           UBS Securities Australia Ltd         Sell         6,30         (80)           UBS Securities Australia Ltd         Sell         5,31         (1.19)           UBS Securities Australia Ltd         Sell         3,90         (1.18)           UBS Securities Australia Ltd         Sell         3,90         (1.18)           UBS Securities Australia Ltd         Sell         18,100         (2.29)           UBS Securities Australia Ltd         Sell         18,100         (1.18)           UBS Securities Australia Ltd         Sell         19,605         (1.18)           UBS Securities Australia Ltd         Sell         19,605         (1.78)           UBS Securities Australia Ltd         Sell         2,490         (1.18)           UBS Securities Australia Ltd         Sell         2,722         (4.48)           UBS Securities Australia Ltd         Sell         2,722         (4.48)           UBS Securities Australia Ltd         Sell         2,722         (4.24)           UBS Securities Australia Ltd         Sell         2,722         (4.24)           UBS Securities	23-Dec-14	UBS Securities Australia Ltd	Sell	7,508	(DZ-0)	Ordinan
USS Securities Australia Ltd         Sell         5.733         (4,109)           USS Securities Australia Ltd         Sell         5.33         (703)           USS Securities Australia Ltd         Sell         5.100         (680)           USS Securities Australia Ltd         Sell         3.904         (189)           USS Securities Australia Ltd         Sell         93.70         (119)           USS Securities Australia Ltd         Sell         93.00         (189)           USS Securities Australia Ltd         Sell         18.100         (2.297)           USS Securities Australia Ltd         Sell         19.500         (1.13)           USS Securities Australia Ltd         Sell         19.605         (1.480)           USS Securities Australia Ltd         Sell         19.605         (2.480)           USS Securities Australia Ltd         Sell         19.605         (2.480)           USS Securities Australia Ltd         Sell         2.732         (4.95)           USS Securities Australia Ltd         Sell         2.722         (3.64)           USS Securities Australia Ltd         Sell         2.722         (4.65)           USS Securities Australia Ltd         Sell         2.722         (4.65)           USS Secur	23-Dec-14	UBS Securities Australia Hd	lleS	3 140	(000)	Codings
UBS Securities Australia Ltd         Sell         63.23         (7.03)           UBS Securities Australia Ltd         Sell         6.217         (7.03)           UBS Securities Australia Ltd         Sell         6.217         (7.03)           UBS Securities Australia Ltd         Sell         5.904         (1.181)           UBS Securities Australia Ltd         Sell         3.904         (1.181)           UBS Securities Australia Ltd         Sell         1.907         (1.181)           UBS Securities Australia Ltd         Sell         1.907         (1.181)           UBS Securities Australia Ltd         Sell         1.907         (1.181)           UBS Securities Australia Ltd         Sell         1.905         (1.181)           UBS Securities Australia Ltd         Sell         2.490         (1.181)           UBS Securities Australia Ltd         Sell         2.73         (2.788)           UBS Securities Australia Ltd         Sell         2.73         (2.58)           UBS Securities Australia Ltd         Sell         2.73         (2.58)           UBS Securities Australia Ltd         Sell         2.72         (2.58)           UBS Securities Australia Ltd         Sell         2.72         (2.58)           UBS Securi	23-Dec-14	UBS Securities Australia i to	Sell	955 CE	(4 100)	Califord
UBS Securities Australia Ltd         Self         6.377         (780)           UBS Securities Australia Ltd         Self         5.10         (780)           UBS Securities Australia Ltd         Self         5.10         (780)           UBS Securities Australia Ltd         Self         5.904         (1181)           UBS Securities Australia Ltd         Self         18.100         (2.297)           UBS Securities Australia Ltd         Self         18.100         (2.297)           UBS Securities Australia Ltd         Self         18.00         (1.181)           UBS Securities Australia Ltd         Self         1.307         (1.308)           UBS Securities Australia Ltd         Self         2.490         (1.181)           UBS Securities Australia Ltd         Self         2.73         (4.78)           UBS Securities Australia Ltd         Self         2.73         (4.78)           UBS Securities Australia Ltd         Self         2.72         (4.29)           UBS Securities Australia Ltd         Self         2.72         (4.24)           UBS Securities Australia Ltd         Self         2.72         (4.24)           UBS Securities Australia Ltd         Self         2.72         (4.24)           UBS Securities Au	23-Dec-14	LIBS Securities Australia Ltd	Sel	2001	(00)	in in in
UBS Securities Australia Ltd         Sell         5.10         (549)           UBS Securities Australia Ltd         Sell         5.10         (648)           UBS Securities Australia Ltd         Sell         3.90         (1.99)           UBS Securities Australia Ltd         Sell         3.90         (1.99)           UBS Securities Australia Ltd         Sell         8.90         (1.13)           UBS Securities Australia Ltd         Sell         18.90         (1.13)           UBS Securities Australia Ltd         Sell         18.90         (1.13)           UBS Securities Australia Ltd         Sell         3.90         (1.309)           UBS Securities Australia Ltd         Sell         3.90         (1.309)           UBS Securities Australia Ltd         Sell         3.90         (1.609)           UBS Securities Australia Ltd         Sell         5.90         (1.70)           UBS Securities Australia Ltd         Sell         5.00         (1.70)           UBS Securities Australia Ltd         Sell         5.73         (2.78)           UBS Securities Australia Ltd         Sell         5.73         (1.60)           UBS Securities Australia Ltd         Sell         8.70         (1.00)           UBS Securities Australia	23-Dec-14	I IBS Securities Australia I to	00	900	(00)	Commission
USS Securities Australia Ltd         Sell         5,100         (4.48)           USS Securities Australia Ltd         Sell         3,904         (4.99)           USS Securities Australia Ltd         Sell         3,904         (4.99)           USS Securities Australia Ltd         Sell         18,00         (1,181)           USS Securities Australia Ltd         Sell         18,00         (1,181)           USS Securities Australia Ltd         Sell         19,605         (1,181)           USS Securities Australia Ltd         Sell         19,605         (1,181)           USS Securities Australia Ltd         Sell         3,904         (4.99)           USS Securities Australia Ltd         Sell         3,904         (4.99)           USS Securities Australia Ltd         Sell         2,713         (1.50)           USS Securities Australia Ltd         Sell         2,713         (1.50)           USS Securities Australia Ltd         Sell         2,718         (1.60)           USS Securities Australia Ltd         Sell         2,722         (4.48)           USS Securities Australia Ltd         Sell         2,732         (4.55)           USS Securities Australia Ltd         Sell         2,732         (4.55)           USS Sec	23-Dec-14	IIRS Securities Australia 14d	Coll	000	(08)	Ordinary
USS Securities Australia Ltd         Sell         9.10         (1498)           USS Securities Australia Ltd         Sell         9.30         (1991)           USS Securities Australia Ltd         Sell         9.30         (1991)           USS Securities Australia Ltd         Sell         18.00         (2.297)           USS Securities Australia Ltd         Sell         18.00         (1.308)           USS Securities Australia Ltd         Sell         19.605         (1.308)           USS Securities Australia Ltd         Sell         19.605         (1.308)           USS Securities Australia Ltd         Sell         2.450         (1.308)           USS Securities Australia Ltd         Sell         2.73         (6.45)           USS Securities Australia Ltd         Sell         2.73         (6.45)           USS Securities Australia Ltd         Sell         2.72         (6.43)           USS Securities Australia Ltd         Sell         2.72         (6.55)           USS Securities Australia Ltd         Sell         2.72         (6.53)           USS Securities Australia Ltd         Sell         8.20         (1.00)           USS Securities Australia Ltd         Sell         8.20         (1.00)           USS Securities A	23-Dec-14	TIDS Countries August 184	103	0,21	(DE/)	Ordinary
UBS Securities Australia Ltd         Sell         9.97         (189)           UBS Securities Australia Ltd         Sell         9.30C         (1.181)           UBS Securities Australia Ltd         Sell         18,10O         (2.297)           UBS Securities Australia Ltd         Sell         2,49O         (1.181)           UBS Securities Australia Ltd         Sell         19,60S         (1.182)           UBS Securities Australia Ltd         Sell         3,90A         (4.96)           UBS Securities Australia Ltd         Sell         3,90A         (4.96)           UBS Securities Australia Ltd         Sell         3,90A         (4.96)           UBS Securities Australia Ltd         Sell         3,90A         (4.96)           UBS Securities Australia Ltd         Sell         2,73B         (8.49)           UBS Securities Australia Ltd         Sell         5,72B         (6.40)           UBS Securities Australia Ltd         Sell         2,72B         (6.40)           UBS Securities Australia Ltd         Sell         2,72B         (6.51)           UBS Securities Australia Ltd         Sell         2,72B         (1.769)           UBS Securities Australia Ltd         Sell         2,60B         (1.769)           UBS Sec	22 Dec 14	1100 Consider Assessing the	12-3	201,5	(242)	Ordinary
UBS Securities Australia Ltd         Sell         1,59,40         (1,81)           UBS Securities Australia Ltd         Sell         18,100         (1,297)           UBS Securities Australia Ltd         Sell         18,100         (1,297)           UBS Securities Australia Ltd         Sell         10,307         (1,308)           UBS Securities Australia Ltd         Sell         19,605         (2,488)           UBS Securities Australia Ltd         Sell         19,605         (2,488)           UBS Securities Australia Ltd         Sell         2,490         (1,508)           UBS Securities Australia Ltd         Sell         2,732         (643)           UBS Securities Australia Ltd         Sell         2,732         (643)           UBS Securities Australia Ltd         Sell         2,732         (643)           UBS Securities Australia Ltd         Sell         2,732         (643)           UBS Securities Australia Ltd         Sell         2,732         (643)           UBS Securities Australia Ltd         Sell         2,732         (643)           UBS Securities Australia Ltd         Sell         2,732         (643)           UBS Securities Australia Ltd         Sell         8,733         (1,09)           UBS Sec	23 Dec 14	U.D.C. Complete Assessing the	17.0	756	(611)	Crainary
UBS Securities Australia Ltd         Sell         930C         (1,18)           UBS Securities Australia Ltd         Sell         890         (1,13)           UBS Securities Australia Ltd         Sell         2,490         (1,13)           UBS Securities Australia Ltd         Sell         10,307         (1,308)           UBS Securities Australia Ltd         Sell         10,307         (1,308)           UBS Securities Australia Ltd         Sell         3,904         (406)           UBS Securities Australia Ltd         Sell         888         (1,10)           UBS Securities Australia Ltd         Sell         888         (1,10)           UBS Securities Australia Ltd         Sell         888         (1,10)           UBS Securities Australia Ltd         Sell         888         (1,10)           UBS Securities Australia Ltd         Sell         888         (1,10)           UBS Securities Australia Ltd         Sell         1,272         (3,28)           UBS Securities Australia Ltd         Sell         8,20         (1,00)           UBS Securities Australia Ltd         Sell         8,20         (1,00)           UBS Securities Australia Ltd         Sell         8,20         (1,00)           UBS Securities Australia L	23-Dec-14	UBS Securities Australia Ltd	Sell	3,904	(496)	Ordinary
UBS Securities Australia Ltd         Sell         18,100         (2,29)           UBS Securities Australia Ltd         Sell         18,00         (1,30g)           UBS Securities Australia Ltd         Sell         2,490         (1,13)           UBS Securities Australia Ltd         Sell         19,505         (2,480)           UBS Securities Australia Ltd         Sell         3,974         (6,69)           UBS Securities Australia Ltd         Sell         3,974         (6,69)           UBS Securities Australia Ltd         Sell         2,1,391         (6,71)           UBS Securities Australia Ltd         Sell         2,1,391         (6,71)           UBS Securities Australia Ltd         Sell         6,53         (6,43)           UBS Securities Australia Ltd         Sell         6,63         (6,43)           UBS Securities Australia Ltd         Sell         6,63         (6,63)           UBS Securities Australia Ltd         Sell         6,63         (7,7)           UBS Securities Australia Ltd         Sell         8,70         (7,7)           UBS Securities Australia Ltd         Sell         8,70         (7,7)           UBS Securities Australia Ltd         Sell         8,70         (7,7)           UBS Securities A	23-Dec-14	UBS Securities Australia Ltd	Sell	908'6	(1,181)	Ordinary
URS Securities Australia Ltd         Sell         890         (113)           URS Securities Australia Ltd         Sell         2,490         (1130)           URS Securities Australia Ltd         Sell         10,307         (1,309)           URS Securities Australia Ltd         Sell         3,974         (605)           URS Securities Australia Ltd         Sell         3,974         (605)           URS Securities Australia Ltd         Sell         892         (170)           URS Securities Australia Ltd         Sell         882         (170)           URS Securities Australia Ltd         Sell         883         (170)           URS Securities Australia Ltd         Sell         883         (170)           URS Securities Australia Ltd         Sell         883         (170)           URS Securities Australia Ltd         Sell         883         (170)           URS Securities Australia Ltd         Sell         883         (105)           URS Securities Australia Ltd         Sell         883         (105)           URS Securities Australia Ltd         Sell         883         (105)           URS Securities Australia Ltd         Sell         883         (105)           URS Securities Australia Ltd	23-Dec-14	UBS Securities Australia Ltd	Sell	18,100	(2,297)	Ordinary
UBS Securities Australia Ltd         Seil         1,490         (316)           UBS Securities Australia Ltd         Sell         19,605         (7,488)           UBS Securities Australia Ltd         Sell         3,974         (6,505)           UBS Securities Australia Ltd         Sell         3,974         (6,605)           UBS Securities Australia Ltd         Sell         3,974         (6,605)           UBS Securities Australia Ltd         Sell         8,505         (6,71)           UBS Securities Australia Ltd         Sell         6,73         (6,83)           UBS Securities Australia Ltd         Sell         6,51         (6,84)           UBS Securities Australia Ltd         Sell         6,51         (6,84)           UBS Securities Australia Ltd         Sell         2,722         (3,63)           UBS Securities Australia Ltd         Sell         2,722         (3,63)           UBS Securities Australia Ltd         Sell         2,722         (3,63)           UBS Securities Australia Ltd         Sell         8,56         (1,769)           UBS Securities Australia Ltd         Sell         8,56         (1,769)           UBS Securities Australia Ltd         Sell         8,56         (1,769)           UBS Securiti	23-Dec-14	UBS Securities Australia Ltd	Sell	068	(113)	Ordinary
UBS Securities Australia Ltd         Self         11.307         (1.308)           UBS Securities Australia Ltd         Self         19.655         2.4889           UBS Securities Australia Ltd         Self         3.974         (5.65)           UBS Securities Australia Ltd         Self         2.95         (7.89)           UBS Securities Australia Ltd         Self         8.86         (1.10)           UBS Securities Australia Ltd         Self         6.51         (6.43)           UBS Securities Australia Ltd         Self         6.53         (6.43)           UBS Securities Australia Ltd         Self         2.722         (6.43)           UBS Securities Australia Ltd         Self         6.551         (6.43)           UBS Securities Australia Ltd         Self         2.722         (6.43)           UBS Securities Australia Ltd         Self         6.551         (6.43)           UBS Securities Australia Ltd         Self         4.006         (7.749)           UBS Securities Australia Ltd         Self         8.20         (1.05)           UBS Securities Australia Ltd         Self         8.20         (1.05)           UBS Securities Australia Ltd         Self         8.20         (1.05)           UBS Securities A	23-Dec-14	UBS Securities Australia Ltd	Sell	2,490	(316)	Ordinary
UBS Securities Australia tot         Sell         19,605         (2,488)           UBS Securities Australia tot         Sell         3,974         (6,65)           UBS Securities Australia tot         Sell         3,904         (4,65)           UBS Securities Australia tot         Sell         27,139         (2,718)           UBS Securities Australia tot         Sell         8,855         (170)           UBS Securities Australia tot         Sell         6,738         (849)           UBS Securities Australia tot         Sell         2,722         (849)           UBS Securities Australia tot         Sell         2,722         (849)           UBS Securities Australia tot         Sell         4,006         (5,08)           UBS Securities Australia tot         Sell         4,006         (5,08)           UBS Securities Australia tot         Sell         8,51         (1,749)           UBS Securities Australia tot         Sell         8,56         (1,769)           UBS Securities Australia tot         Sell         8,56         (1,60)           UBS Securities Australia tot         Sell         8,56         (1,60)           UBS Securities Australia tot         Sell         8,56         (1,60)           UBS Securities A	23-Dec-14	UBS Securities Australia Ltd	Sell	10,307	(1,308)	Ordinary
UBS Securities Australia Ltd         Sell         3.974         (505)           UBS Securities Australia Ltd         Sell         21,391         (405)           UBS Securities Australia Ltd         Sell         859         (404)           UBS Securities Australia Ltd         Sell         869         (110)           UBS Securities Australia Ltd         Sell         673         (643)           UBS Securities Australia Ltd         Sell         673         (643)           UBS Securities Australia Ltd         Sell         6,531         (643)           UBS Securities Australia Ltd         Sell         2,772         (643)           UBS Securities Australia Ltd         Sell         2,075         (643)           UBS Securities Australia Ltd         Sell         3,23         (10)           UBS Securities Australia Ltd         Sell         3,23         (10)           UBS Securities Australia Ltd         Sell         8,23         (105)           UBS Securities Australia Ltd         Sell         8,23         (105)           UBS Securities Australia Ltd         Sell         8,23         (105)           UBS Securities Australia Ltd         Sell         8,23         (105)           UBS Securities Australia Ltd         <	23-Dec-14	UBS Securities Australia Ltd	Sell	19 605	(2.488)	Ordinary
UBS Securities Australia Ltd         Sell         2,304         (495)           UBS Securities Australia Ltd         Sell         2,394         (495)           UBS Securities Australia Ltd         Sell         2,718         (710)           UBS Securities Australia Ltd         Sell         2,732         (843)           UBS Securities Australia Ltd         Sell         5,722         (843)           UBS Securities Australia Ltd         Sell         5,722         (843)           UBS Securities Australia Ltd         Sell         1,722         (843)           UBS Securities Australia Ltd         Sell         1,009         (710)           UBS Securities Australia Ltd         Sell         1,009         (710)           UBS Securities Australia Ltd         Sell         8,51         (710)           UBS Securities Australia Ltd         Sell         8,52         (105)           UBS Securities Australia Ltd         Sell         8,56         (105)           UBS Securities Australia Ltd         Sell         8,56         (105)           UBS Securities Australia Ltd         Sell         8,56         (105)           UBS Securities Australia Ltd         Sell         8,56         (105)           UBS Securities Australia Ltd	23-Dec-14	UBS Securities Australia Ltd	Sell	3 974	(405)	Ordinary
UBS Securities Australia Ltd         Sell         2,739         (2,738)           UBS Securities Australia Ltd         Sell         888         (170)           UBS Securities Australia Ltd         Sell         888         (170)           UBS Securities Australia Ltd         Sell         6,738         (843)           UBS Securities Australia Ltd         Sell         6,651         (843)           UBS Securities Australia Ltd         Sell         2,722         (345)           UBS Securities Australia Ltd         Sell         2,075         (568)           UBS Securities Australia Ltd         Sell         2,075         (568)           UBS Securities Australia Ltd         Sell         3,23         (41)           UBS Securities Australia Ltd         Sell         8,56         (175)           UBS Securities Australia Ltd         Sell         8,56         (176)           UBS Securities Australia Ltd         Sell         8,56         (176)           UBS Securities Australia Ltd         Sell         8,56         (176)           UBS Securities Australia Ltd         Sell         8,56         (176)           UBS Securities Australia Ltd         Sell         8,28         (176)           UBS Securities Australia Ltd	23-Dec-14	UBS Securities Australia Ltd	Se	200	(900)	Column
UBS Securities Australia Ltd         Sell         (2.7.13)         (2.7.14)           UBS Securities Australia Ltd         Sell         858         (110)           UBS Securities Australia Ltd         Sell         213         (2.7)           UBS Securities Australia Ltd         Sell         2,722         (854)           UBS Securities Australia Ltd         Sell         2,722         (854)           UBS Securities Australia Ltd         Sell         6,571         (843)           UBS Securities Australia Ltd         Sell         3,23         (1,19)           UBS Securities Australia Ltd         Sell         8,50         (7,11)           UBS Securities Australia Ltd         Sell         8,50         (7,10)           UBS Securities Australia Ltd         Sell         8,50         (7,10)           UBS Securities Australia Ltd         Sell         8,50         (7,10)           UBS Securities Australia Ltd         Sell         8,50         (1,00)           UBS Securities Australia Ltd         Sell         8,50         (1,00)           UBS Securities Australia Ltd         Sell         8,50         (1,00)           UBS Securities Australia Ltd         Sell         8,50         (1,00)           UBS Securities Australia L	23-Dec.14	J. IRC Securities Australia 14d	le de la constante de la const	100.50	(0400)	Cignary
URS Securities Australia Ltd         Sell         RSB         (170)           UBS Securities Australia Ltd         Sell         6738         (170)           UBS Securities Australia Ltd         Sell         6738         (843)           UBS Securities Australia Ltd         Sell         2,722         (345)           UBS Securities Australia Ltd         Sell         2,722         (343)           UBS Securities Australia Ltd         Sell         2,722         (343)           UBS Securities Australia Ltd         Sell         2,075         (708)           UBS Securities Australia Ltd         Sell         323         (41)           UBS Securities Australia Ltd         Sell         828         (106)           UBS Securities Australia Ltd         Sell         826         (105)           UBS Securities Australia Ltd         Sell         826         (106)           UBS Securities Australia Ltd         Sell         826         (106)           UBS Securities Australia Ltd         Sell         826         (106)           UBS Securities Australia Ltd         Sell         828         (106)           UBS Securities Australia Ltd         Sell         828         (106)           UBS Securities Australia Ltd         Sel	23-Dec-14	LIDS Committee Australia 14d	lion.	165,13	(2,718)	Ordinary
USS Securities Australia Ltd         Sell         6868         (170)           UBS Securities Australia Ltd         Sell         6738         (874)           UBS Securities Australia Ltd         Sell         6,531         (854)           UBS Securities Australia Ltd         Sell         6,551         (824)           UBS Securities Australia Ltd         Sell         1,050         (7,748)           UBS Securities Australia Ltd         Sell         4,008         (7,09)           UBS Securities Australia Ltd         Sell         8,23         (1,05)           UBS Securities Australia Ltd         Sell         8,26         (1,05)           UBS Securities Australia Ltd         Sell         8,26         (1,05)           UBS Securities Australia Ltd         Sell         8,26         (1,05)           UBS Securities Australia Ltd         Sell         8,26         (1,05)           UBS Securities Australia Ltd         Sell         8,26         (1,05)           UBS Securities Australia Ltd         Sell         8,26         (1,05)           UBS Securities Australia Ltd         Sell         8,26         (1,05)           UBS Securities Australia Ltd         Sell         8,26         (1,05)           UBS Securities Australia Lt	22 Dec 14	Opp Securities Australia Ltg	100	505	(64)	Ordinary
UBS Securities Australia Ltd         Sell         6.738         (878)           UBS Securities Australia Ltd         Sell         6.738         (848)           UBS Securities Australia Ltd         Sell         2.722         (848)           UBS Securities Australia Ltd         Sell         6.551         (828)           UBS Securities Australia Ltd         Sell         4.008         (7.17)           UBS Securities Australia Ltd         Sell         4.008         (7.17)           UBS Securities Australia Ltd         Sell         8.28         (1.05)           UBS Securities Australia Ltd         Sell         8.28         (1.05)           UBS Securities Australia Ltd         Sell         8.28         (1.05)           UBS Securities Australia Ltd         Sell         8.28         (1.05)           UBS Securities Australia Ltd         Sell         8.26         (1.05)           UBS Securities Australia Ltd         Sell         8.28         (1.05)           UBS Securities Australia Ltd         Sell         8.28         (1.05)           UBS Securities Australia Ltd         Sell         8.28         (1.05)           UBS Securities Australia Ltd         Sell         8.28         (1.05)           UBS Securities Australia L	23-0-6-14	Obsiderations Australia Ltd	lisc in	898	(110)	Ordinary
UBS Securities Australia trd         Sell         6,53         (8543)           UBS Securities Australia trd         Sell         6,531         (843)           UBS Securities Australia trd         Sell         6,651         (843)           UBS Securities Australia trd         Sell         6,651         (843)           UBS Securities Australia trd         Sell         1,098         (7,09)           UBS Securities Australia trd         Sell         3,33         (4,1)           UBS Securities Australia trd         Sell         8,28         (1,05)           UBS Securities Australia trd         Sell         8,26         (1,05)           UBS Securities Australia trd         Sell         8,26         (1,05)           UBS Securities Australia trd         Sell         8,26         (1,05)           UBS Securities Australia trd         Sell         8,26         (1,05)           UBS Securities Australia trd         Sell         8,28         (1,05)           UBS Securities Australia trd         Sell         8,28         (1,05)           UBS Securities Australia trd         Sell         8,28         (1,05)           UBS Securities Australia trd         Sell         8,28         (1,05)           UBS Securities Australia trd	23-Dec-14	UBS Securities Australia Ltd	Sell	213	(27)	Ordinary
USS Securities Australia Ltd         Sell         2.772         (3-45)           USS Securities Australia Ltd         Sell         2.772         (3-45)           USS Securities Australia Ltd         Sell         1.3600         (7.789)           USS Securities Australia Ltd         Sell         3.23         (4.1)           USS Securities Australia Ltd         Sell         3.23         (4.1)           USS Securities Australia Ltd         Sell         8.26         (7.1)           USS Securities Australia Ltd         Sell         8.26         (1.05)           USS Securities Australia Ltd         Sell         8.26         (1.05)           USS Securities Australia Ltd         Sell         8.26         (1.05)           USS Securities Australia Ltd         Sell         8.26         (1.05)           USS Securities Australia Ltd         Sell         8.26         (1.05)           USS Securities Australia Ltd         Sell         8.26         (1.05)           USS Securities Australia Ltd         Sell         8.26         (1.05)           USS Securities Australia Ltd         Sell         8.24         (1.05)           USS Securities Australia Ltd         Sell         8.24         (1.05)           USS Securities Australia L	23-Dec-14	UBS Securities Australia Ltd	) Sell	6,738	(854)	Ordinary
UBS Securities Australia It of 1983         Sell         6,651         (843)           UBS Securities Australia It of 1985         Sell         2,075         (2628)           UBS Securities Australia It of 1985         Sell         1,7483         (471)           UBS Securities Australia It of 1985         Sell         3,23         (471)           UBS Securities Australia It of 1985         Sell         8,23         (471)           UBS Securities Australia It of 1985         Sell         8,28         (105)           UBS Securities Australia It of 201         Sell         8,561         (105)           UBS Securities Australia It of 201         Sell         8,561         (105)           UBS Securities Australia It of 201         Sell         8,561         (105)           UBS Securities Australia It of 201         Sell         8,561         (105)           UBS Securities Australia It of 201         Sell         8,561         (105)           UBS Securities Australia It of 201         Sell         8,28         (105)           UBS Securities Australia It of 201         Sell         8,28         (105)           UBS Securities Australia It of 201         Sell         8,28         (105)           UBS Securities Australia It of 201         Sell         8	23-Dec-14	UBS Securities Australia Ltd	Sell	2,722	(345)	Ordinary
UBS Securities Australia (±d)         Sell         1,055         (2675)           UBS Securities Australia (±d)         Sell         1,3800         (1,749)           UBS Securities Australia (±d)         Sell         4,008         (508)           UBS Securities Australia (±d)         Sell         823         (41)           UBS Securities Australia (±d)         Sell         828         (105)           UBS Securities Australia (±d)         Sell         828         (105)           UBS Securities Australia (±d)         Sell         826         (105)           UBS Securities Australia (±d)         Sell         826         (105)           UBS Securities Australia (±d)         Sell         826         (105)           UBS Securities Australia (±d)         Sell         828         (105)           UBS Securities Australia (±d)         Sell         828         (105)           UBS Securities Australia (±d)         Sell         828         (105)           UBS Securities Australia (±d)         Sell         828         (105)           UBS Securities Australia (±d)         Sell         828         (105)           UBS Securities Australia (±d)         Sell         828         (105)           UBS Securities Australia (±d)	23-Dec-14	UBS Securities Australia Ltd	Sell	6,651	(843)	Ordinary
UBS Securities Australia Ltd         Sell         13,800         (1,749)           UBS Securities Australia Ltd         Sell         4,006         (5,008)           UBS Securities Australia Ltd         Sell         323         (41)           UBS Securities Australia Ltd         Sell         828         (105)           UBS Securities Australia Ltd         Sell         835         (106)           UBS Securities Australia Ltd         Sell         836         (106)           UBS Securities Australia Ltd         Sell         828         (105)           UBS Securities Australia Ltd         Sell         828         (105)           UBS Securities Australia Ltd         Sell         828         (105)           UBS Securities Australia Ltd         Sell         828         (105)           UBS Securities Australia Ltd         Sell         828         (105)           UBS Securities Australia Ltd         Sell         828         (105)           UBS Securities Australia Ltd         Sell         824         (107)           UBS Securities Australia Ltd         Sell         824         (107)           UBS Securities Australia Ltd         Sell         824         (107)           UBS Securities Australia Ltd         Sell	23-Dec-14	UBS Securities Australia Ltd	Sell	2,075	(263)	Ordinary
UBS Securities Australia Ltd         Sell         4,008         (5)08           UBS Securities Australia Ltd         Sell         323         (4)1           USS Securities Australia Ltd         Sell         SSO         (7)1           UBS Securities Australia Ltd         Sell         828         (105)           UBS Securities Australia Ltd         Sell         836         (105)           UBS Securities Australia Ltd         Sell         836         (105)           UBS Securities Australia Ltd         Sell         836         (105)           UBS Securities Australia Ltd         Sell         828         (105)           UBS Securities Australia Ltd         Sell         828         (105)           UBS Securities Australia Ltd         Sell         828         (105)           UBS Securities Australia Ltd         Sell         828         (105)           UBS Securities Australia Ltd         Sell         828         (105)           UBS Securities Australia Ltd         Sell         828         (105)           UBS Securities Australia Ltd         Sell         824         (107)           UBS Securities Australia Ltd         Sell         824         (107)           UBS Securities Australia Ltd         Sell	23-Dec-14	UBS Securities Australia Ltd	Self	13,800	(1,749)	Ordinary
UBS Securities Australia Ltd         Sell         323         (41)           UBS Securities Australia Ltd         Sell         550         (71)           UBS Securities Australia Ltd         Sell         828         (105)           UBS Securities Australia Ltd         Sell         836         (105)           UBS Securities Australia Ltd         Sell         836         (105)           UBS Securities Australia Ltd         Sell         836         (105)           UBS Securities Australia Ltd         Sell         828         (105)           UBS Securities Australia Ltd         Sell         828         (105)           UBS Securities Australia Ltd         Sell         828         (107)           UBS Securities Australia Ltd         Sell         828         (107)           UBS Securities Australia Ltd         Sell         828         (107)           UBS Securities Australia Ltd         Sell         844         (107)           UBS Securities Australia Ltd         Sell         9989         (126)	23-Dec-14		Sell	4 008	(508)	Ordinary
USS Securities Australia Ltd         Sell         560         (71)           UBS Securities Australia Ltd         Sell         828         (105)           UBS Securities Australia Ltd         Sell         836         (106)           UBS Securities Australia Ltd         Sell         561         (75)           UBS Securities Australia Ltd         Sell         5641         (75)           UBS Securities Australia Ltd         Sell         828         (105)           UBS Securities Australia Ltd         Sell         828         (105)           UBS Securities Australia Ltd         Sell         844         (107)           UBS Securities Australia Ltd         Sell         844         (107)           UBS Securities Australia Ltd         Sell         844         (107)           UBS Securities Australia Ltd         Sell         894         (107)	23-Dec-14		Seil	323	(41)	Ordinary
UBS Securities Australia Ltd         Sell         828         (105)           UBS Securities Australia Ltd         Sell         836         (105)           UBS Securities Australia Ltd         Sell         836         (105)           UBS Securities Australia Ltd         Sell         836         (105)           UBS Securities Australia Ltd         Sell         828         (159)           UBS Securities Australia Ltd         Sell         828         (105)           UBS Securities Australia Ltd         Sell         828         (107)           UBS Securities Australia Ltd         Sell         844         (107)           UBS Securities Australia Ltd         Sell         844         (107)           UBS Securities Australia Ltd         Sell         844         (107)           UBS Securities Australia Ltd         Sell         844         (107)           UBS Securities Australia Ltd         Sell         9989         (1726)	23-Dec-14		Sell	025		Ordinary.
URS Securities Australia tot         561         626         (105)           URS Securities Australia tot         Self         836         (106)           URS Securities Australia tot         Self         856.1         (105)           URS Securities Australia tot         Self         5,641         (105)           URS Securities Australia tot         Self         828         (105)           URS Securities Australia tot         Self         844         (107)           URS Securities Australia tot         Self         844         (107)           URS Securities Australia tot         Self         844         (107)           URS Securities Australia tot         Self         844         (107)           URS Securities Australia tot         Self         844         (107)           URS Securities Australia tot         Self         844         (107)	23-Dec.14	ILIBS Segurities Australia I to	3	000	(1.7)	Cidillary
UBS Securities Australia Ltd         Sell         836         (108)           UBS Securities Australia Ltd         Sell         5,641         (715)           UBS Securities Australia Ltd         Sell         1,255         (159)           UBS Securities Australia Ltd         Sell         828         (105)           UBS Securities Australia Ltd         Sell         828         (107)           UBS Securities Australia Ltd         Sell         844         (107)           UBS Securities Australia Ltd         Sell         844         (107)           UBS Securities Australia Ltd         Sell         844         (107)           UBS Securities Australia Ltd         Sell         844         (107)           UBS Securities Australia Ltd         Sell         844         (107)           UBS Securities Australia Ltd         Sell         9989         (17,266)	23-Dec-14	LIRK Securities Australia 14d	[0]	970	(501)	Cramary
URS Securities Australia Ltd         Self         \$156         (156)           URS Securities Australia Ltd         Self         \$564         (715)           URS Securities Australia Ltd         Self         \$22         (159)           URS Securities Australia Ltd         Self         \$22         (165)           URS Securities Australia Ltd         Self         \$22         (107)           URS Securities Australia Ltd         Self         \$22         (8)           URS Securities Australia Ltd         Self         \$2         (8)           URS Securities Australia Ltd         Self         \$2         (8)           URS Securities Australia Ltd         Self         \$2         (8)           URS Securities Australia Ltd         Self         \$2         (8)           URS Securities Australia Ltd         Self         \$2         (8)	22 000 14	Data minimum Commission Com	100	000	(on i	Orginary
UBS Securities Australia Ltd         Neil         7.15)           UBS Securities Australia Ltd         Seil         (7.5)           UBS Securities Australia Ltd         Seil         828         (1.05)           UBS Securities Australia Ltd         Seil         844         (107)           UBS Securities Australia Ltd         Seil         844         (107)           UBS Securities Australia Ltd         Seil         844         (107)           UBS Securities Australia Ltd         Seil         844         (107)           UBS Securities Australia Ltd         Seil         844         (107)           UBS Securities Australia Ltd         Seil         9989         (1.266)	PL-Dag-CZ	Obsisecutives Australia Lia	liac .	836	(106)	Ordinary
UBS Securities Australia (10)         Sell         1,255         (159)           UBS Securities Australia (10)         Sell         82.38         (105)           UBS Securities Australia (10)         Sell         84.4         (107)           UBS Securities Australia (10)         Sell         84.4         (107)           UBS Securities Australia (10)         Sell         84.4         (107)           UBS Securities Australia (10)         Sell         84.4         (107)           UBS Securities Australia (10)         Sell         84.4         (107)           UBS Securities Australia (10)         Sell         84.4         (107)	23-Dec-14	UBS Securifies Australia Ltd	sell	5,641	(715)	Ordinary
UBS Securines Australia Ir U         Seli         82.8         (10.5)           UBS Securities Australia Ir U         Seli         84.4         (107)           UBS Securities Australia Ir U         Seli         84.4         (107)           UBS Securities Australia Ir U         Seli         84.4         (107)           UBS Securities Australia Ir U         Seli         84.4         (107)           UBS Securities Australia Ir U         Seli         9989         (17.266)	23-Dec-14	UBS Securities Australia Ltd	Sell	1,255	(159)	Ordinary
LBS Securities Australia Ltd         Sell         844         (107)           UBS Securities Australia Ltd         Sell         63         (8)           UBS Securities Australia Ltd         Sell         844         (107)           UBS Securities Australia Ltd         Sell         9989         (1266)	23-Dec-14	UBS Securities Australia Ltd	Sell	828	(105)	Ordinary
UBS Securities Australia Ltd         Sell         63         (8)           UBS Securities Australia Ltd         Sell         844         (107)           UBS Securities Australia Ltd         Sell         9989         (1.266)	23-Dec-14	UBS Securities Australia Ltd	Sell	844	(107)	Ordinary
UBS Securities Australia Ltd         Sell         844         (107)           UBS Securities Australia Ltd         Sell         9,989         (1,266)	23-Dec-14	UBS Securities Australia Ltd	Sell	69	(8)	Ordinary
UBS Securities Australia Ltd Sell (1.266)	23-Dec-14	UBS Securities Australia Ltd	Sell	844	(107)	Ordinary
	33-Dec-14	UBS Securities Australia Ltd	Sel	0000		
Constitution of the state of th	Dec 10	and the state of t			(3000)	1